		此又行在 <u>收到所有必要的资料及文件後才正式</u> 確	委員會 <u>No. A/YL-SK/410</u>
		This cont is received on The control of the application only upon of all the required information and documents.	25 <u>Form No. S16-I</u> weldge receipt
	APPLIC	CATION FOR PE	RMISSION
	UN	NDER SECTION	16 OF
Т	HE TOV	<b>WN PLANNING (</b>	ORDINANCE
		(CAP. 131)	
根 據	《城市	市規劃條例)	》(第131章)
	第16	條遞交的許	可申請
(1) Construit		Townstowing Unomated	touse(s)":
<ul> <li>(i) Construct 興建「新</li> <li>(ii) Tempora rural are 位於鄉刻 用途/發展</li> <li>(iii) Renewal Regulate 位於鄉刻</li> </ul>	tion of "Ne 界豁免管制 ry use/deve as or Regula 地區或受規 of permissi d Areas 地區或受規	W Territories Exempted F I屋宇」; lopment of land and/or b ated Areas; and 管地區土地上及/或建築 ion for temporary use or l管地區的臨時用途或發展	wilding not exceeding 3 years in 物內進行為期不超過三年的臨時 · development in rural areas or 長的許可續期
<ul> <li>(i) Construct 興建「新</li> <li>(ii) Tempora rural are 位於鄉刻 用途/發展</li> <li>(iii) Renewal Regulate 位於鄉刻</li> <li>Applicant who we Planning Board's i land owner, please https://www.tpb.ge</li> <li>申請人如欲在本述 土地擁有人所對</li> </ul>	tion of "Ne <b>界豁免管制</b> ry use/deve as or Regula 选地區或受規 of permissid d Areas 地區或受規 buld like to pub requirements of te refer to the for ov.hk/en/plan_ar 也報章刊登 <u>申請</u> 指定的其中一	Where the inference is a complete in the provided of the importance of the importan	<b>building not exceeding 3 years in</b> 物內進行為期不超過三年的臨時 <b>r development in rural areas or</b> <b>bo的許可續期</b> local newspapers to meet one of the Town consent of or give notification to the current g the notice in the designated newspapers: t取得現行土地擁有人的同意或通知現行 網址有關在指定的報章刊登通知:

Form No. S16-I 表格第 S16-I 號

For Official Use Only	Application No. 申請編號	A/YL-SK/ FID
請勿填寫此欄	Date Received 收到日期	1 8 FEB 2025

2500351 17, by hund

- 1. The completed form and supporting documents (if any) should be sent to the Secretary, Town Planning Board (the Board), 15/F, North Point Government Offices, 333 Java Road, North Point, Hong Kong. 申請人須把填妥的申請表格及其他支持申請的文件 (倘有),送交香港北角渣華道 333 號北角政府合署 15 樓城市規劃委員會(下稱「委員會」)秘書收。
- 2. Please read the "Guidance Notes" carefully before you fill in this form. The document can be downloaded from the Board's website at <u>http://www.tpb.gov.hk/</u>. It can also be obtained from the Secretariat of the Board at 15/F, North Point Government Offices, 333 Java Road, North Point, Hong Kong (Tel: 2231 4810 or 2231 4835), and the Planning Enquiry Counters of the Planning Department (Hotline: 2231 5000) (17/F, North Point Government Offices, 333 Java Road, North Point, Hong Kong and 14/F, Sha Tin Government Offices, 1 Sheung Wo Che Road, Sha Tin, New Territories). 請先細閱《申請須知》的資料單張,然後填寫此表格。該份文件可從委員會的網頁下載(網址: <u>http://www.tpb.gov.hk/</u>),亦可向委員會秘書處(香港北角渣華道 333 號北角政府合署 15 樓 電話: 2231 4810 或 2231 4835)及規劃署的規劃資料查詢處(熱線: 2231 5000) (香港北角渣華道 333 號北角政府合署 17 樓及新界沙田 上禾輋路 1 號沙田政府合署 14 樓)索取。
- 3. This form can be downloaded from the Board's website, and obtained from the Secretariat of the Board and the Planning Enquiry Counters of the Planning Department. The form should be typed or completed in block letters. The processing of the application may be refused if the required information or the required copies are incomplete. 此表格可從委員會的網頁下載,亦可向委員會秘書處及規劃署的規劃資料查詢處索取。申請人須以打印方式或以正楷填寫表格。如果申請人所提交的資料或文件副本不齊全,委員會可拒絕處理有關申請。

## 1. Name of Applicant 申請人姓名/名稱

(□Mr. 先生 /□Mrs. 夫人 /□Miss 小姐 /□Ms. 女士 / ☑ Company 公司 /□ Organisation 機構 )

Join Bright Warehousing Limited 俊怡嘉貨倉管理有限公司

#### 2. Name of Authorised Agent (if applicable) 獲授權代理人姓名/名稱(如適用)

(□Mr. 先生 /□Mrs. 夫人 /□Miss 小姐 /□Ms. 女士 / ☑ Company 公司 /□ Organisation 機構 )

R-riches Property Consultants Limited 盈卓物業顧問有限公司

3.	Application Site 申請地點	
(a)	Full address / location / demarcation district and lot number (if applicable) 詳細地址/地點/丈量約份及 地段號碼(如適用)	Lots 573 RP and 1710 in D.D. 114, Shek Kong, Yuen Long, New Territories
(b)	Site area and/or gross floor area involved 涉及的地盤面積及/或總樓面面 積	☑Site area 地盤面積
(c)	Area of Government land included (if any) 所包括的政府土地面積(倘有)	N/A sq.m 平方米口About 約

(d)	Name and number of the related statutory plan(s) 有關法定圖則的名稱及編號	Approved Shek Kong Outline Zoning Plan No.: S/YL-SK/9
(e)	Land use zone(s) involved 涉及的土地用途地帶	"Industrial (Group D)" Zone
(f)	Current use(s) 現時用途	Open storage of junk vehicles and vehicle parts (If there are any Government, institution or community facilities, please illustrate on plan and specify the use and gross floor area) (如有任何政府、機構或社區設施,請在圖則上顯示,並註明用途及總樓面面積)

## 4. "Current Land Owner" of Application Site 申請地點的「現行土地擁有人」

The applicant 申請人 -

- □ is the sole "current land owner"<sup>#&</sup> (please proceed to Part 6 and attach documentary proof of ownership). 是唯一的「現行土地擁有人」<sup>#&</sup> (請繼續填寫第 6 部分,並夾附業權證明文件)。
- □ is one of the "current land owners"<sup># &</sup> (please attach documentary proof of ownership). 是其中一名「現行土地擁有人」<sup># &</sup> (請夾附業權證明文件)。
- ✓ is not a "current land owner"<sup>#</sup>. 並不是「現行土地擁有人」<sup>#</sup>。

The application site is entirely on Government land (please proceed to Part 6). 申請地點完全位於政府土地上(請繼續填寫第6部分)。

## 5. Statement on Owner's Consent/Notification 就土地擁有人的同意/通知土地擁有人的陳述

(b) The applicant 申請人 -

has obtained consent(s) of ...... "current land owner(s)"<sup>#</sup>.

已取得 ...... 名「現行土地擁有人」 \* 的同意。

取得同意的日期 (日/月/年)	Io. of 'Current and Owner(s)' 「現行土地擁有 人」數目

(Please use separate sheets if the space of any box above is insufficient. 如上列任何方格的空間不足,請另頁說明)

- 1		_	
- 11		- 1	
		. 1	

has notified	"current land owner(s)"#	
Dimmer December and the		

已通知 ...... 名「現行土地擁有人」#。

Nc La 「 有	o. of 'Current and Owner(s)' 現行土地擁 人」數目	Lot number/address of premises as shown in the record of the Land Registry where notification(s) has/have been given 根據土地註冊處記錄已發出通知的地段號碼/處所地址	Date of notification given (DD/MM/YYYY) 通知日期(日/月/年)
(Plea	ase use separate s	heets if the space of any box above is insufficient. 如上列任何方格的空	E間不足,請另頁說明)
has 已扨	taken reasonabl 采取合理步驟以	e steps to obtain consent of or give notification to owner(s): 取得土地擁有人的同意或向該人發給通知。詳情如下:	力人理止趣
Reas	sonable Steps to	o Obtain Consent of Owner(s) 取得土地擁有人的回意所採取的	的合理步驗
	sent request fo	or consent to the "current land owner(s)" on	(DD/MM/YYYY) <sup>#8</sup>
	古八		
	лх	(口/月/牛)四母一石 坑门工地擁有八」 野诞女不下	可息音
Reas	sonable Steps to	O Give Notification to Owner(s) 向土地擁有人發出通知所採用	和自己的合理步驟
Reas	から sonable Steps to published noti 方令	(口/月/平/问每一名 現门工地擁有人了 郵遞要不同 o Give Notification to Owner(s) <u>向土地擁有人發出通知所採用</u> ces in local newspapers on (DD/MM/YY (日/月/年)在指定報章就申請刊登一次通知 <sup>&amp;</sup>	<u>双的合理步驟</u> YY) <sup>&amp;</sup>
<u>Rea</u> : □ ✓ 06	から <u>sonable Steps to</u> published noti 方合 posted notice i 5/01/2025 - 20	(口/月/平/问号: 名 現门工地擁有人發出通知所採用 o Give Notification to Owner(s) <u>向土地擁有人發出通知所採用</u> ces in local newspapers on(DD/MM/YY (日/月/年)在指定報章就申請刊登一次通知 <sup>&amp;</sup> in a prominent position on or near application site/premises on /01/2025 (DD/MM/YYYY) <sup>&amp;</sup>	n息音 <sup>™</sup> <u>Q的合理步驟</u> YY) <sup>&amp;</sup>
<u>Rea</u> : □ 06	から <u>sonable Steps to</u> published noti 方合 <u></u> posted notice i 5/01/2025 - 20, 方令	(日/月/平/丙基 名 現日工地擁有人了 郵遞要求下 o Give Notification to Owner(s) <u>向土地擁有人發出通知所採取</u> ces in local newspapers on(DD/MM/YY (日/月/年)在指定報章就申請刊登一次通知 <sup>&amp;</sup> in a prominent position on or near application site/premises on /01/2025 (DD/MM/YYYY) <sup>&amp;</sup> (日/月/年)在申請地點/申請處所或附近的顯明位置	<u>2的合理步驟</u> YY) <sup>&amp;</sup> 张出關於該申請的通
<u>Rea:</u> □ □ □ □ □ □ □ □ □ □ □ □ □ □ □ □ □ □ □	sonable Steps to published noti 於 posted notice 5/01/2025 - 20 於 sent notice to p office(s) or run 於 處,或有關的	(日/月/年)内每一名 現日主地擁有人登出通知所採用 o Give Notification to Owner(s) 向土地擁有人發出通知所採用 ces in local newspapers on(DD/MM/YY (日/月/年)在指定報章就申請刊登一次通知 <sup>&amp;</sup> in a prominent position on or near application site/premises on /01/2025 (DD/MM/YYYY) <sup>&amp;</sup> (日/月/年)在申請地點/申請處所或附近的顯明位置 relevant owners' corporation(s)/owners' committee(s)/mutual aid ral committee on23/01/2025(DD/MM/YYYY) <sup>&amp;</sup> (日/月/年)把通知寄往相關的業主立案法團/業主委 的鄉事委員會 <sup>&amp;</sup>	<u>20的合理步驟</u> YY) <sup>&amp;</sup> Rh出關於該申請的通 committee(s)/manager 員會/互助委員會或名
<u>Rea</u> : □ ✓ 06	sonable Steps to published noti 於 posted notice i 5/01/2025 - 20. 於 sent notice to p office(s) or run 於 處,或有關的	(日/月/年/问号: 名 現刊工地擁有人了 郵遞要求下 o Give Notification to Owner(s) <u>自土地擁有人發出通知所採取</u> ces in local newspapers on(DD/MM/YY (日/月/年)在指定報章就申請刊登一次通知 <sup>&amp;</sup> in a prominent position on or near application site/premises on /01/2025 (DD/MM/YYYY) <sup>&amp;</sup> (日/月/年)在申請地點/申請處所或附近的顯明位置 relevant owners' corporation(s)/owners' committee(s)/mutual aid ral committee on23/01/2025 (DD/MM/YYYY) <sup>&amp;</sup> (日/月/年)把通知寄往相關的業主立案法團/業主委 的鄉事委員會 <sup>&amp;</sup>	<u>20的合理步驟</u> YY) <sup>&amp;</sup> Rh出關於該申請的通 committee(s)/manager 員會/互助委員會或行
<u>Rea</u> : □ □ □ 06	sonable Steps to         published noti         於         posted notice is         b)         posted notice is         b)         b)	(日/月/年/问每: 名 現刊工地擁有人引 郵遞要求问 o Give Notification to Owner(s) <u>向土地擁有人發出通知所採用</u> ces in local newspapers on(DD/MM/YY (日/月/年)在指定報章就申請刊登一次通知 <sup>&amp;</sup> in a prominent position on or near application site/premises on /01/2025 (DD/MM/YYYY) <sup>&amp;</sup> (日/月/年)在申請地點/申請處所或附近的顯明位置 relevant owners' corporation(s)/owners' committee(s)/mutual aid ral committee on3/01/2025(DD/MM/YYYY) <sup>&amp;</sup> (日/月/年)把通知寄往相關的業主立案法團/業主委 句鄉事委員會 <sup>&amp;</sup>	<u>2(的合理步驟</u> YY) <sup>&amp;</sup> 张出關於該申請的通 committee(s)/manager 該員會/互助委員會或 <sup>€</sup>
Reas 06 06	sonable Steps to published noti 於 posted notice i 5/01/2025 - 20 於 sent notice to p office(s) or run 於 處,或有關的 ers 其他 others (please 其他(請指明	(日/月/年/问每: 名 現日工地擁有人了 郵遞要求下 o Give Notification to Owner(s) <u>向土地擁有人發出通知所採取</u> ces in local newspapers on(DD/MM/YY (日/月/年)在指定報章就申請刊登一次通知 <sup>&amp;</sup> in a prominent position on or near application site/premises on /01/2025 (DD/MM/YYYY) <sup>&amp;</sup> (日/月/年)在申請地點/申請處所或附近的顯明位置 relevant owners' corporation(s)/owners' committee(s)/mutual aid ral committee on23/01/2025(DD/MM/YYYY) <sup>&amp;</sup> (日/月/年)把通知寄往相關的業主立案法團/業主委 句鄉事委員會 <sup>&amp;</sup>	<u>n息音</u> <u>y的合理步驟</u> YY) <sup>&amp;</sup>
<u>Rea</u> : □ 06 ☑	sonable Steps to         published noti         於         posted notice i         5/01/2025 - 20,         於         sent notice to poffice(s) or run         於         處,或有關的         ers 其他         others (please         其他(請指明)	(日/月/年)问号 名 現刊工地擁有人對此國家不同 of Give Notification to Owner(s) <u>自土地擁有人發出通知所採取</u> ces in local newspapers on(DD/MM/YY (日/月/年)在指定報章就申請刊登一次通知 <sup>&amp;</sup> in a prominent position on or near application site/premises on /01/2025 (DD/MM/YYY) <sup>&amp;</sup> (日/月/年)在申請地點/申請處所或附近的顯明位置 relevant owners' corporation(s)/owners' committee(s)/mutual aid ral committee on <u>23/01/2025</u> (DD/MM/YYYY) <sup>&amp;</sup> (日/月/年)把通知寄往相關的業主立案法團/業主委 句鄉事委員會 <sup>&amp;</sup>	NDE THE THE SECTION OF THE SECTION

Note: May insert more than one「✓」. Information should be provided on the basis of each and every lot (if applicable) and premises (if any) in respect of the application.
 註: 可在多於一個方格內加上「✓」號
申請人須就申請涉及的每一地段(倘適用)及處所(倘有)分別提供資料

6.	Type(s) of Application	n 申請类	頁別		
	Type (i) Change of use 第(i)類 更改現有建築	within existir 物或其部分内	ng building or part thereof 内的用途		
	Type (ii) Diversion of str	version of stream / excavation of land / filling of land / filling of pond as required under Notes of Statutory			
	第(ii)類 根據法定圖則	) 定圖則《註釋》內所要求的河道改道/挖土/填土/填塘工程			
	Type (iii) Public utility in 第(iii)類 公用事業設施:	ic utility installation / Utility installation for private project ]事業設施裝置/私人發展計劃的公用設施裝置			
	Type (iv) Minor relaxatio 第(iv)類 略為放寬於法:	n of stated de 定圖則《註釋	evelopment restriction(s) as provided und <sup>睪</sup> 》內列明的發展限制	er Notes of Statutory Plan(s)	
1	Type (v) Use / developm 第(v)類 上述的(i)至(iii	ent other tha )項以外的用	n (i) to (iii) above 途/發展		
Note 註1 Note 註2	<ol> <li>May insert more than one 「</li> <li>可在多於一個方格內加上「</li> <li>For Development involving colus</li> <li>如發展涉及靈灰安置所用差</li> </ol>	✓」・ ✓」號 mbarium use, plo 食,請填妥於阿	ease complete the table in the Appendix. 付件的表格。		
(i)	<u>For Type (i) applicati</u>	ion 供第(i	) <u>類申請</u>		
(a) [ i	Total floor area involved 涉及的總樓面面積			sq.m 平方米	
(b) I 1 1	Proposed use(s)/development 疑議用途/發展	(If there are a the use and a (如有任何政	any Government, institution or community fac gross floor area) 奴府、機構或社區設施,請在圖則上顯示,	ilities, please illustrate on plan and specify 並註明用途及總樓面面積)	
(c) ] 3	Number of storeys involved 涉及層數		Number of units involv 涉及單位數目	ved	
		Domestic p	oart 住用部分 so	q.m 平方米 □About 約	
(d) I	Proposed floor area 疑議樓面面積	Non-dome:	stic part 非住用部分 s	q.m 平方米   □About 約	
		Total 總計	S	q.m 平方米   □About 約	
(e) I	Proposed uses of different	Floor(s) 樓層	Current use(s) 現時用途	Proposed use(s) 擬議用途	
	floors (if applicable) 不同樓層的擬議用途(如適 用)				
( s ( E	Please use separate sheets if the space provided is insufficient) 如所提供的空間不足,請另頁說 明)				

(ii) For Type (ii) applic	ation 供第(ii)類申請
	<ul> <li>Diversion of stream 河道改道</li> </ul>
л. П	<ul> <li>□ Filling of pond 填塘</li> <li>Area of filling 填塘面積</li></ul>
(a) Operation involved 涉及工程	<ul> <li>□ Filling of land 填土         Area of filling 填土面積sq.m 平方米 □About 約         Depth of filling 填土厚度m 米 □About 約</li> <li>□ Excavation of land 挖土         Area of excavation 挖土面積sq.m 平方米 □About 約         Depth of excavation 挖土派度m 米 □About 約         Cease indicate on site plan the boundary of concerned land/pond(s), and particulars of stream diversion, the extent of filling of land/pond(s) and/or excavation of land)         (請用圖則顯示有關土地/池塘界線,以及河道改道、填塘、填土及/或挖土的細節及/或範圍))     </li> </ul>
(b) Intended use/development 有意進行的用途/發展	
(iii) <u>For Type (iii) applic</u>	ration 供第(iii)類申請
8	□ Public utility installation 公用事業設施裝置
	□ Utility installation for private project 私人發展計劃的公用設施裝置
	Please specify the type and number of utility to be provided as well as the dimensions of each building/structure, where appropriate 請註明有關裝置的性質及數量,包括每座建築物/構築物(倘有)的長度、高度和闊度
	Name/type of installation 裝置名稱/種類Number provision 數量of provision building/structure (m) (LxWxH) 每個裝置/建築物/構築物的尺寸 (米) (長 x 闊 x 高)
(a) Nature and scale 性質及規模	
	(Please illustrate on plan the layout of the installation 請用圖則顯示裝置的布局)

(iv) <u>F</u>	for Type (iv) application 供	<u> </u>
(a) H 日 計	Please specify the proposed r proposed use/development ar 情列明擬議略為放寬的發展[	ninor relaxation of stated development restriction(s) and <u>also fill in the</u> ad development particulars in part (v) below – 艮制 <u>並填妥於第(v)部分的擬議用途/發展及發展細節</u> –
	Plot ratio restriction 地積比率限制	From 由 to 至
	Gross floor area restriction 總樓面面積限制	From 由sq. m 平方米 to 至sq. m 平方米
	Site coverage restriction 上蓋面積限制	From 由% to 至%
	Building height restriction 建築物高度限制	From 由 m 米 to 至 m 米
		From 由 mPD 米 (主水平基準上) to 至
		mPD 米 (主水平基準上)
		From 由 storeys 層 to 至 storeys 層
	Non-building area restriction 非建築用地限制	From 由m to 至 m
	Others (please specify) 其他(請註明)	

(v) <u>For Type (v) application 供第(v)類申請</u>				
(a) Proposed use(s)/development 擬議用途/發展	Proposed Temporary Concrete Period of 5 Years	Batching Plant with Ancillary Facili	ities for a	
	(Please illustrate the details of the propo	sal on a layout plan 請用平面圖說明建議	詳情)	
(b) Development Schedule 發展	影細節表			
Proposed gross floor area (C Proposed plot ratio 擬議地種 Proposed site coverage 擬議	GFA) 擬議總樓面面積 責比率 上蓋面積	4,411 sq.m 平方米 1.0 32 %	☑About 約 ☑About 約 ☑About 約	
Proposed no. of blocks 擬議	座數			
Proposed no. of storeys of ea	ach block 每座建築物的擬議層數	not more than3 storeys 層 □ include 包括storeys of basem □ exclude 不包括storeys of bas	ents 層地庫 ements 層地庫	
Proposed building height of	each block 每座建築物的擬議高度	mPD 米(主水平基準上 … not more than_13… m 米	) □About 約 □About 約	

ł

Ŧ

│	rt 住用部分		
GFA 總	樓面面積	sq. m 平方米	□About 約
number	of Units 單位數目		
average	unit size 單位平均面積	sq. m 平方米	□About 约
estimate	ed number of residents 估計住家數日		
Cotiniau			
↓ Non-domest	ic part 非住用部分	GFA 總樓面面	積
eating p	lace食肆	sq. m 平方米	□About 約
□ hotel 洒	店	sg. m 平方米	□About 約
		(please specify the number of rooms	
		a;	
」 「 office 対	<b>办八</b>		□ A hout 4/1
		sq. m 平方木	
shop an	d services 間佔反服務行業	sq. m 平万禾	山About 約1
Govern	ment, institution or community facilities	(please specify the use(s) and	concerned land
	機構或社區設施	area(s)/GFA(s) 請註明田途及有關	的地面面積/總
		樓面面積)	
			*****
✓ other(s)	其他	(please specify the use(s) and	concerned land
		area(s)/GFA(s) 請註明用途及有關	的地面面積/總
		樓面面積)	
		Concrete batching plant: 4,411 sq	. m (about)
Open space {	木憩用地	(please specify land area(s) 請註明	也面面積)
private	open space 私人休憩用地	sq. m 平方米 口 Not l	ess than 不少於
public o	pen space 公眾休憩用地	sq. m 平方米 口 Not l	ess than 不少於
<b>1</b>			12 41
(c) Use(s) of differ	rent floors (if applicable) 久相區的田滨 (fl	1) (1) (1) (1) (1) (1) (1) (1) (1) (1) (	
(c) Use(s) of differ	rent floors (if applicable) 各棲層的用途 (如	1適用)	
(c) Use(s) of differ [Block number]	rent floors (if applicable) 各樓層的用途 (如 [Floor(s)]	I適用) [Proposed use(s)]	
(c) Use(s) of differ [Block number] [座數]	ent floors (if applicable) 各樓層的用途 (如 [Floor(s)] [層數]	]適用) [Proposed use(s)] [擬議用途]	
(c) Use(s) of differ [Block number] [座數] Please refer to t	rent floors (if applicable) 各樓層的用途 (如 [Floor(s)] [層數] he supplementary statement.	1適用) [Proposed use(s)] [擬議用途]	
(c) Use(s) of differ [Block number] [座數] Please refer to t	rent floors (if applicable) 各樓層的用途 (如 [Floor(s)] [層數] he supplementary statement.	]適用) [Proposed use(s)] [擬議用途]	
(c) Use(s) of differ [Block number] [座數] Please refer to t	rent floors (if applicable) 谷樓層的用途 (如 [Floor(s)] [層數] he supplementary statement.	□適用) [Proposed use(s)] [擬議用途]	
(c) Use(s) of differ [Block number] [座數] Please refer to t	rent floors (if applicable) 谷樓層的用途 (如 [Floor(s)] [層數] he supplementary statement.	]適用) [Proposed use(s)] [擬議用途]	
(c) Use(s) of differ [Block number] [座數] Please refer to t	rent floors (if applicable) 谷樓層的用途 (如 [Floor(s)] [層數] he supplementary statement.	1適用) [Proposed use(s)] [擬議用途]	
(c) Use(s) of differ [Block number] [座數] Please refer to t	rent floors (if applicable) 谷樓層的用途 (如 [Floor(s)] [層數] he supplementary statement.	]適用) [Proposed use(s)] [擬議用途]	
(c) Use(s) of differ [Block number] [座數] Please refer to t	rent floors (if applicable) 各樓層的用途 (如 [Floor(s)] [層數] he supplementary statement.	1適用) [Proposed use(s)] [擬議用途] 写)的擬議用途	
(c) Use(s) of differ [Block number] [座數] Please refer to t (d) Proposed use(s Vehicle parking a	rent floors (if applicable) 各樓層的用途 (如 [Floor(s)] [層數] he supplementary statement.	1適用) [Proposed use(s)] [擬議用途] 有)的擬議用途 ation area	
(c) Use(s) of differ [Block number] [座數] Please refer to t (d) Proposed use(s Vehicle parking a	rent floors (if applicable) 各樓層的用途 (如 [Floor(s)] [層數] he supplementary statement. ) of uncovered area (if any) 露天地方(倘有 and loading/unloading spaces and circula	1適用) [Proposed use(s)] [擬議用途] 写)的擬議用途 ation area	
(c) Use(s) of differ [Block number] [座數] Please refer to t (d) Proposed use(s Vehicle parking a	rent floors (if applicable) 各樓層的用途 (如 [Floor(s)] [層數] he supplementary statement.	1適用) [Proposed use(s)] [擬議用途] 与)的擬議用途 ation area	
(c) Use(s) of differ [Block number] [座數] Please refer to t (d) Proposed use(s Vehicle parking a	rent floors (if applicable) 各樓層的用途 (如 [Floor(s)] [層數] he supplementary statement. ) of uncovered area (if any) 露天地方(倘有 and loading/unloading spaces and circula	1適用) [Proposed use(s)] [擬議用途] 写)的擬議用途 ation area	

8

Part 6 (Cont'd) 第6部分 (續)

<ol> <li>Anticipated Completion Time of the Development Proposal 擬議發展計劃的預計完成時間</li> </ol>
Anticipated completion time (in month and year) of the development proposal (by phase (if any)) (e.g. June 2023) 擬議發展計劃預期完成的年份及月份 (分期 (倘有)) (例:2023 年 6 月)
(Separate anticipated completion times (in month and year) should be provided for the proposed public open space and Government, institution or community facilities (if any)) (由請人須就擬議的公咒休憩用协及政府、機構或社區設施(倘有)提供個別擬議完成的年份及目份)
Mid-2026
~

8. Venicular Access Arr 擬議發展計劃的行	angemen 車通道	nt of the Development Proposal 安排	Δ.
Any vehicular access to the site/subject building? 是否有車路通往地盤/有關 建築物?	Yes 是	<ul> <li>✓ There' is an existing access. (please indicate the street n appropriate) 有一條現有車路。(請註明車路名稱(如適用))</li> <li>Kam Tin Road via a local access</li> <li>□ There is a proposed access. (please illustrate on plan and specin有一條擬議車路。(請在圖則顯示,並註明車路的闊度)</li> </ul>	ame, where
Any provision of parking space for the proposed use(s)? 是否有為擬議用途提供停車 位?	Yes 是	<ul> <li>✓ (Please specify type(s) and number(s) and illustrate on plan)</li> <li>請註明種類及數目並於圖則上顯示)</li> <li>Private Car Parking Spaces 私家車車位</li> <li>Motorcycle Parking Spaces 電單車車位</li> <li>Light Goods Vehicle Parking Spaces 輕型貨車泊車位</li> <li>Medium Goods Vehicle Parking Spaces 中型貨車泊車位</li> <li>Heavy Goods Vehicle Parking Spaces 重型貨車泊車位</li> <li>Others (Please Specify) 其他 (請列明)</li> </ul>	2 N/A N/A N/A 12
Any provision of loading/unloading space for the proposed use(s)? 是否有為擬議用途提供上落客 貨車位?	Yes 是 No 否	<ul> <li>☑ (Please specify type(s) and number(s) and illustrate on plan) 請註明種類及數目並於圖則上顯示) Taxi Spaces 的士車位 Coach Spaces 旅遊巴車位 Light Goods Vehicle Spaces 輕型貨車車位 Medium Goods Vehicle Spaces 中型貨車車位 Heavy Goods Vehicle Spaces 重型貨車車位 Others (Please Specify) 其他 (請列明) Tanker Spaces</li> </ul>	N/A N/A N/A N/A N/A 2

9. Impacts of De	evelopm	ent Proposal 擬議發展計劃的	影響	an a
If necessary, please us justifications/reasons fe 如需要的話,請另頁	se separate or not prov 註明可盡量	sheets to indicate the proposed measur iding such measures. 量減少可能出現不良影響的措施,否則	es to minimise possible a 請提供理據/理由。	dverse impacts or give
Does the development proposal involve alteration of existing building? 擬議發展計劃是否 包括現有建築物的 改動? Does the development proposal involve the operation on the right? 擬議發展是否涉及 右列的工程? (Note: where Type (ii) application is the subject of application, please skip this section. 註:如申請,請跳至下 一條問題。)	Yes 是 No 否 Yes 是	<ul> <li>□ Please provide details 請提供詳</li> <li></li></ul>	情 f concerned land/pond(s), and pa xeavation of land) 象,以及河道改道、填塘、填土 	rticulars of stream diversion, 
	No 否	Depth of excavation 挖土深度	m 米	□About 約
Would the development proposal cause any adverse impacts? 擬議發展計劃會否 造成不良影響?	On envir On traffid On water On draim On slope Affected Landscap Tree Fell Visual In Others (F Please st diameter 請註明盝 直徑及品	L onment 對環境 c 對交通 supply 對供水 age 對排水 s 對斜坡 by slopes 受斜坡影響 be Impact 構成景觀影響 ing 砍伐樹木 npact 構成視覺影響 Please Specify) 其他 (請列明) 	Yes 會 □ Yes 會 □	No 不會 No 不會

Part 9 第 9 部分

10. Justifications 理由
The applicant is invited to provide justifications in support of the application. Use separate sheets if necessary. 現請申請人提供申請理由及支持其申請的資料。如有需要,請另頁說明。
Please refer to the supplementary statement.

11. Declaration 聲明				
I hereby declare that the particulars given in this application are correct and true to the best of my knowledge and belief. 本人謹此聲明,本人就這宗申請提交的資料,據本人所知及所信,均屬真實無誤。				
I hereby grant a permission to the Board to copy all the materials submitted in this application and/or to upload such materials to the Board's website for browsing and downloading by the public free-of-charge at the Board's discretion.本人現准許委員會酌情將本人就此申請所提交的所有資料複製及/或上載至委員會網站,供公眾免費瀏覽或下載。				
Signature       □ Applicant 申請人 / ☑ Authorised Agent 獲授權代理人         簽署       □ Applicant 申請人 / ☑ Authorised Agent 獲授權代理人				
Matthew NG Planning and Development Manager				
Name in Block LettersPosition (if applicable)姓名(請以正楷填寫)職位 (如適用)				
Professional Qualification(s) 專業資格 「 Member 會員 / □ Fellow of 資深會員 「 HKIP 香港規劃師學會 / □ HKIA 香港建築師學會 / □ HKIS 香港測量師學會 / □ HKIE 香港工程師學會 / □ HKILA 香港國境師學會 / □ HKIUD 香港城市設計學會 □ RPP 註冊專業規劃師 Others 其他				
on behalf of 代表    R-riches Property Consultants Limited 盈卓物業顧問有限公司				
☑ Company 公司 / □ Organisation Name and Chop (if applicable) 機構名稱及蓋章(如適用)				
Date 日期 23/01/2025 (DD/MM/YYYY 日/月/年)				

#### <u>Remark 備註</u>

The materials submitted in this application and the Board's decision on the application would be disclosed to the public. Such materials would also be uploaded to the Board's website for browsing and free downloading by the public where the Board considers appropriate.

委員會會向公眾披露申請人所遞交的申請資料和委員會對申請所作的決定。在委員會認為合適的情況下,有關申請 資料亦會上載至委員會網頁供公眾免費瀏覽及下載。

#### Warning 警告

Any person who knowingly or wilfully makes any statement or furnish any information in connection with this application, which is false in any material particular, shall be liable to an offence under the Crimes Ordinance. 任何人在明知或故意的情況下,就這宗申請提出在任何要項上是虛假的陳述或資料,即屬違反《刑事罪行條例》。

#### Statement on Personal Data 個人資料的聲明

- 1. The personal data submitted to the Board in this application will be used by the Secretary of the Board and Government departments for the following purposes:
  - 委員會就這宗申請所收到的個人資料會交給委員會秘書及政府部門,以根據《城市規劃條例》及相關的城市規 劃委員會規劃指引的規定作以下用途:
  - (a) the processing of this application which includes making available the name of the applicant for public inspection when making available this application for public inspection; and
  - 處理這宗申請,包括公布這宗申請供公眾查閱,同時公布申請人的姓名供公眾查閱;以及 (b) facilitating communication between the applicant and the Secretary of the Board/Government departments. 方便申請人與委員會秘書及政府部門之間進行聯絡。
- The personal data provided by the applicant in this application may also be disclosed to other persons for the purposes mentioned in paragraph 1 above.
   申請人就這宗申請提供的個人資料,或亦會向其他人士披露,以作上述第1段提及的用途。
- 3. An applicant has a right of access and correction with respect to his/her personal data as provided under the Personal Data (Privacy) Ordinance (Cap. 486). Request for personal data access and correction should be addressed to the Secretary of the Board at 15/F, North Point Government Offices, 333 Java Road, North Point, Hong Kong. 根據《個人資料(私隱)條例》(第 486 章)的規定,申請人有權查閱及更正其個人資料。如欲查閱及更正個人資料,應向委員會秘書提出有關要求,其地址為香港北角渣華道 333 號北角政府合署 15 樓。

For Developments involving Columbarium Use, please also complete the following: 如發展涉及靈灰安置所用途,請另外填妥以下資料:	
Ash interment capacity 骨灰安放容量 <sup>@</sup>	
Maximum number of sets of ashes that may be interred in the niches 在龕位內最多可安放骨灰的數量 Maximum number of sets of ashes that may be interred other than in niches 在非龕位的範圍內最多可安放骨灰的數量	
Total number of niches 龕位總數	
Total number of single niches 單人龕位總數	
Number of single niches (sold and occupied)       □         單人龕位數目 (已售但未佔用)       □         Number of single niches (sold but unoccupied)       □         單人龕位數目 (已售但未佔用)       □         Number of single niches (residual for sale)       □         單人龕位數目 (待售)       □	
Total number of double niches 雙人龕位總數	
Number of double niches (sold and fully occupied)	
Total no. of niches other than single or double niches (please specify type) 除單人及雙人龕位外的其他龕位總數 (請列明類別)	
Number. of niches (sold and fully occupied)	
Proposed operating hours 擬議營運時間	
<ul> <li>@ Ash interment capacity in relation to a columbarium means – 就靈灰安置所而言,骨灰安放容量指:</li> <li>the maximum number of containers of ashes that may be interred in each niche in the columbarium; 每個龕位內可安放的骨灰容器的最高數目;</li> <li>the maximum number of sets of ashes that may be interred other than in niches in any area in the columbarium; and 在該靈灰安置所並非龕位的範圍內,總共最多可安放多少份骨灰;以及</li> <li>the total number of sets of ashes that may be interred in the columbarium.</li> </ul>	

在該骨灰安置所內,總共最多可安放多少份骨灰。

e = e

## Gist of Application 申請摘要

(Please provide details in both English and Chinese <u>as far as possible</u>. This part will be circulated to relevant consultees, uploaded to the Town Planning Board's Website for browsing and free downloading by the public and available at the Planning Enquiry Counters of the Planning Department for general information.) (請盡量以英文及中文填寫。此部分將會發送予相關諮詢人士、上載至城市規劃委員會網頁供公眾免費瀏覽及

卜載及於規劃署規	劃資料查	至詢處供一般參閱。	· )				
Application No. 申請編號	(For O	fficial Use Only) (請约	勿填寫此欄)				
Location/address 位置/地址	Lots 5	73 RP and 1710 in	D.D. 114, Sheł	Kong,	Yuen Long, I	New Territori	es
Site area 地盤面積					4,411 \$	q.m 平方	米☑ About 約
	(includ	es Government land	of包括政府	土地	N/A	sq.m 平方;	米 □ About 約)
Plan 圖則	Appro	ved Shek Kong Out	tline Zoning Pla	in No.: :	S/YL-SK/9		
Zoning 地帶	"Indus	trial (Group D)" Zor	ie				
Applied use/ development 申請用途/發展	Propos Years	sed Temporary Cor	crete Batching	Plant w	vith Ancillary	Facilities for	a Period of 5
(i) Gross floor are	ea		sq.r	n 平方:	米	Plot R	atio 地積比率
and/or plot rat 總樓面面積及 地積比率	10 之/或	Domestic 住用	N/A	口 A 口 N 不	bout 約 ot more than 下多於	N/A	□About 約 □Not more than 不多於
		Non-domestic 非住用	4,411	☑ A □ N 不	bout 約 ot more than 写多於	1.0	☑About 約 □Not more than 不多於
(ii) No. of blocks 幢數		Domestic 住用			N/A		
		Non-domestic 非住用	N/A				
		Composite 綜合用途			N/A	l.	

(iii)	Building height/No. of storeys 建築物高度/區數	Domestic 住用	N/A	m 米□ (Not more than 不多於)
,	产术切问及7 酒数		N/A	mPD 米(主水平基準上)□ (Not more than 不多於)
			N/A	Storeys(s) 層 □ (Not more than 不多於)
	2	a	v	(□Include 包括/□ Exclude 不包括 □ Carport 停車間 □ Basement 地庫 □ Refuge Floor 防火層 □ Podium 平台)
		Non-domestic 非住用	13	m 米 ☑ (Not more than 不多於)
		×	N/A	mPD 米(主水平基準上) □ (Not more than 不多於)
		8	3	Storeys(s) 層 ☑ (Not more than 不多於)
	, ,		5	(□Include 包括/□ Exclude 不包括 □ Carport 停車間 □ Basement 地庫 □ Refuge Floor 防火層 □ Podium 平台)
1		Composite 綜合用途	N/A	m 米□ (Not more than 不多於)
		hār	N/A	mPD 米(主水平基準上)□ (Not more than 不多於)
		c	. N/A	Storeys(s) 層 □ (Not more than 不多於).
				(□Include 包括/□ Exclude 不包括 □ Carport 停車間 □ Basement 地庫 □ Refuge Floor 防火層 □ Podium 平台)
(iv)	Site coverage 上蓋面積		32	% I About 約
(v)	No. of units 單位數目		N/A	
(vi)	Open space 休憩用地	Private 私人	N/A	sq.m 平方米 🗆 Not less than 不少於
		Public 公眾	N/A	sq.m 平方米 🗆 Not less than 不少於

(vii)	No. of parking spaces and loading /	Total no. of vehicle parking spaces 停車位總數		14	
	unloading spaces	Private Car Parking Spaces 私家直面位		2	
	停車位及上落客貨	Motorcycle Parking Spaces 雪單直由位		N/A	
	車位數目	Light Goods Vehicle Parking Spaces 範型生产工		N/A	
		Light Goods Vehicle Parking Spaces 輕空員单/日車位		N/A	
		Medium Goods Venicle Parking Spaces 中型員車沿車位		12	
		Heavy Goods Vehicle Parking Spaces 重型貨車沿車位		12	
~		Others (Please Specify) 其他 (請列明)	8		
		- 121 			
			2		
		Total no. of vehicle loading/unloading bays/lay-bys 上落客貨車位/停車處總數		2	
		Taxi Spaces 的土車位		N/A	
		Coach Spaces 旅遊巴車位		N/A	
		Light Goods Vehicle Spaces 輕型貨車車位	1.	N/A	
		Medium Goods Vehicle Spaces 中型貨車位		N/A	
		Heavy Goods Vehicle Spaces 重刑貨車車位		N/A	
		Others (Please Specify) 其他 (請列明)			
		Tanker Spaces		2	
				70	_

Submitted Plans, Drawings and Documents 提交的圖則、繪圖及文件		
	<u>Chinese</u>	English
	中文	英文
Plans and Drawings 圖則及繪圖		
Master layout plan(s)/Layout plan(s) 總綱發展藍圖/布局設計圖		$\checkmark$
Block plan(s) 樓宇位置圖		
Floor plan(s) 樓宇平面圖		
Sectional plan(s) 截視圖		
Elevation(s) 立視圖		$\checkmark$
Photomontage(s) showing the proposed development 顯示擬議發展的合成照片		
Master landscape plan(s)/Landscape plan(s) 園境設計總圖/園境設計圖		
Others (please specify) 其他(請註明)		$\checkmark$
Plans showing location/zoning/land status of the Site; Plans showing location/zoning of the	e original pr	emises;
Plans showing dev. phasing/land resumption of HSK/HT NDA; Plan showing alternative si	tes; and Aei	rial photo.
Reports 報告書		
Planning Statement/Justifications 規劃綱領/理據		$\checkmark$
Environmental assessment (noise, air and/or water pollutions)		
環境評估(噪音、空氣及/或水的污染)		
Traffic impact assessment (on vehicles) 就車輛的交通影響評估		$\checkmark$
Traffic impact assessment (on pedestrians) 就行人的交通影響評估		
Visual impact assessment 視覺影響評估		
Landscape impact assessment 景觀影響評估		
Tree Survey 樹木調查		
Geotechnical impact assessment 土力影響評估		
Drainage impact assessment 排水影響評估		1
Sewerage impact assessment 排污影響評估		
Risk Assessment 風險評估		
Others (please specify) 其他(請註明)		
Note: May insert more than one 「ノ」, 註: 可在多於一個方格內加上「ノ」號		

Note: The information in the Gist of Application above is provided by the applicant for easy reference of the general public. Under no circumstances will the Town Planning Board accept any liabilities for the use of the information nor any inaccuracies or discrepancies of the information provided. In case of doubt, reference should always be made to the submission of the applicant. 註: 上述申請摘要的資料是由申請人提供以方便市民大眾參考。對於所載資料在使用上的問題及文義上的歧異,城市規劃委員

 $\hat{\mathcal{T}}_{\rm ec}$ 

.\*

會概不負責。若有任何疑問,應查閱申請人提交的文件。

此為空白頁。

5

This is a blank page.

#### SECTION 16 PLANNING APPLICATION

## PROPOSED TEMPORARY CONCRETE BATCHING PLANT WITH ANCILLARY FACILITIES FOR A PERIOD OF 5 YEARS IN "INDUSTRIAL (GROUP D)" ZONE,

LOTS 573 RP AND 1710 IN D.D. 114 SHEK KONG, YUEN LONG, NEW TERRITORIES

**PLANNING STATEMENT** 

Applicant Join Bright Warehousing Limited

<u>Consultancy Team</u> Planning Consultant: Drainage Consultant: Environmental Consultant: Traffic Consultant:

R-riches Property Consultants Limited EnviroSolutions & Consulting Limited EnviroSolutions & Consulting Limited CKM Asia Limited



January 2025 Version 1.0

#### **CONTENT PAGE**

EXEC	CUTIVE SUMMARY	3
行政	摘要	4
1.	INTRODUCTION	5
	Background	5
2.	JUSTIFICATION	6
	To Facilitate the Relocation of the Applicant's Business Premises Affected by HSK/HT NDA	6
	Applicant's Effort in Identifying Suitable Site for Relocation	6
	Applied Use is the Same as the Affected Business in HSK	6
	To Meet the Growing Demand for Concrete Production	7
	Approval of the Application would not Frustrate the Long-term Planning Intention of the "I(D)" Zone	7
	The Proposed Development is not Incompatible with Surrounding Land Uses	7
3.	SITE CONTEXT	9
	Site Location	9
	Accessibility	9
	Existing Site Condition	9
	Surrounding Area	9
4.	PLANNING CONTEXT	10
	Zoning of the Site	10
	Planning Intention	10
	Statutory Development Restrictions	10
	Previous and Similar Applications	10
	Land Status of the Site	10
5.	DEVELOPMENT PROPOSAL	11
	Development Details	11
	Operation Mode	11
	No Traffic Impact	11
	No Environmental Impact	12
	No Drainage Impact	12
	Fire Safety Aspect	13
6.	CONCLUSION	14



#### APPENDICES

Appendix I	Details of the Affected Business Premises
Appendix II	Details of Alternative Sites for Relocation
Appendix III	Traffic Impact Assessment
Appendix IV	Environmental Assessment
Appendix V	Drainage Proposal

#### LIST OF PLANS

Plan 1	Location Plan
Plan 2	Plan showing the Zoning of the Site
Plan 3	Plan showing the Land Status of the Site
Plan 4	Original Premises – Location
Plan 5	Original Premises – Zoning
Plan 6	Original Premises – HSK/HT NDA Phasing and Land Resumption
Plan 7	Original Premises – HSK/HT NDA Land Resumption
Plan 8	Plan showing Alternative Sites for Relocation
Plan 9	Aerial Photo of the Site
Plan 10	Layout Plan
Plan 11	Elevation Plan

#### LIST OF TABLES

- Table 1Differences between the Original Premises and the Site
- Table 2Development Parameters
- Table 3
   Provision of Parking and Loading/Unloading Spaces



#### EXECUTIVE SUMMARY

- The applicant seeks planning permission from the Town Planning Board (the Board) under Section (S.) 16 of the Town Planning Ordinance (Cap. 131) to use *Lots 573 RP and 1710 in D.D. 114, Shek Kong, Yuen Long, New Territories* (the Site) for 'Proposed Temporary Concrete Batching Plant with Ancillary Facilities for a Period of 5 Years' (the proposed development).
- The Site falls within an area zoned "Industrial (Group D)" ("I(D)") on the Approved Shek Kong Outline Zoning Plan (OZP) No.: S/YL-SK/9. The Site occupies an area of 4,411 m<sup>2</sup> (about). A concrete batching plant with other ancillary facilities with total gross floor area (GFA) of 4,411 m<sup>2</sup> (about) is proposed at the Site. The remaining area is reserved for vehicle parking and loading/unloading spaces and circulation area.
- The Site is accessible from Kam Tin Road via a local access. The operation hours of the proposed development are 07:00 to 23:00 daily.
- Justifications for the proposed development are as follows:
  - the applicant's original premises is affected by land resumption for the development of the Hung Shui Kiu/Ha Tsuen (HSK/HT) New Development Area (NDA);
  - the applicant has spent effort in identifying suitable sites for relocation;
  - the applied use is the same as the applicant's original premises;
  - the proposed development is not incompatible with surrounding land uses; and
  - the proposed development is only on a temporary basis, approval of the application will not frustrate the long-term planning intention of the "I(D)" zone.
- Details of development parameters are as follows:

Application Site Area	4,411 m² (about)	
Covered Area	1,411 m² (about)	
Uncovered Area	3,000 m² (about)	
Plot Ratio	1.0 (about)	
Site Coverage	32% (about)	
Total GFA	4,411 m² (about)	
- Domestic GFA	Not applicable	
- Non-Domestic GFA	4,411 m² (about)	
Building Height	Not more than 13 m	
No. of Storey	Not more than 3 storeys	



Planning Statement 20250122 Ver 1.0 行政摘要 (內文如與英文版本有任何差異,應以英文版本為準)

- 申請人現根據《城市規劃條例》(第131章)第16條,向城市規劃委員會提交有關 新界元朗石崗丈量約份第114約地段第573號餘段及第1710號的規劃申請,於上 述地點作「擬議臨時混凝土配料廠連附屬設施(為期5年)」(擬議發展)。
- 申請地點所在的地區在《石崗分區計劃大綱核准圖編號 S/YL-SK/9》上劃為「工業 (丁類)」地帶。申請地盤面積為4,411 平方米(約)。申請地點將設有混凝土配料 廠及其他附屬設施,總樓面面積合共為4,411 平方米(約),申請地點的其餘地方 將預留作車輛停泊和上落貨位及流轉空間。
- 申請地點可從錦田公路經一條地區道路前往。擬議發展的作業時間為每日上午七時 至下午十一時。
- 擬議發展的申請理據如下:
  - 申請人原來的經營處所受到「洪水橋/廈村新發展區」發展收地影響;
  - 申請人曾經致力尋找合適的搬遷地點;
  - 申請用途與申請人先前受影響的發展場地用途一致;
  - 擬議發展與周邊地方的用途並非不協調;及
  - 擬議發展只屬臨時性質,批出規劃許可則不會影響「工業(丁類)」地帶的長 遠規劃意向。
- 擬議發展的詳情發展參數如下:

<b>由</b> + 1 前 - 4 ·	1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1
甲請地盤面積·	4,411 平方未(約)
上蓋總面積:	1,411 平方米(約)
露天地方面積:	3,000 平方米(約)
地積比率:	1.0(約)
上蓋覆蓋率:	32%(約)
總樓面面積	4,411 平方米(約)
住用總樓面面積:	不適用
非住用總樓面面積:	4,411 平方米(約)
構築物高度:	不多於13米
構築物層數:	不多於3層



## 1. INTRODUCTION

## Background

- 1.1 R-riches Property Consultants Limited has been commissioned by Join Bright Warehousing Limited<sup>1</sup> (the applicant) to make submission on their behalf to the Board under S.16 of the Ordinance in respect to Lots 573 RP and 1710 in D.D. 114, Shek Kong, Yuen Long, New Territories (Plans 1 to 3).
- 1.2 The applicant would like to use the Site for 'Proposed Temporary Concrete Batching Plant with Ancillary Facilities for a Period of 5 Years'. The Site currently falls within an area zoned "I(D)" on the Approved Shek Kong OZP No.: S/YL-SK/9 (Plan 2). According to the Notes of the OZP, the applied use is a Column 2 use within the "I(D)" zone, which requires planning permission from the Board.
- 1.3 In support of the proposal, a set of indicative development plans and drawings are provided with the planning statement. Assessments and proposals (i.e. Traffic Impact Assessment (TIA), Environmental Assessment (EA) and Drainage Proposal) to mitigate potential adverse impacts are submitted for the consideration of relevant government bureaux/departments and members of the Board (Plans 1 to 11 and Appendices I to V).

<sup>&</sup>lt;sup>1</sup> Join Bright Warehousing Limited (the applicant) is authorized by Glorious Concrete (H.K.) Limited (the affected business operator) to facilitate the relocation of the existing affected business premises in Hung Shui Kiu. Details of the affected business operator are provided at **Appendix I**.



## 2. JUSTIFICATIONS

## To facilitate the relocation of the applicant's business premises affected by HSK/HT NDA

- 2.1 The current application is intended to facilitate the relocation of the applicant's affected premises of a concrete batching plant and the ancillary facilities in HSK, i.e. *various lots in D.D. 130*, due to land resumption and to pave way for the development of the HSK/HT NDA (**Appendix I** and **Plan 4**).
- 2.2 According to the Revised Recommended Outline Development Plan (Revised RODP) for the HSK NDA, the affected premises falls within areas zoned "Industry", "Other Specified Uses", "Government" and an area shown as 'Road'. The affected premises currently falls within areas zoned "Industrial" ("I"), "Government, Institution or Community" ("G/IC") and area shown as 'Road' on the Approved Hung Shui Kiu and Ha Tsuen OZP No.: S/HSK/2 (**Plan 5**).
- 2.3 With reference to the implementation programme, the affected premises falls within the land resumption limit for the second phase development of the HSK/HT NDA (Plan 6). As land where the premises will be developed for industrial and government, Institution or community uses, the concerned parcel of land will be resumed and reverted to the Government in 2024 (Plan 7). Therefore, the applicant desperately needs to identify a suitable site for relocation in order to continue its business operation.

#### Applicant's effort in identifying suitable site for relocation

2.4 Whilst the applicant has spent effort to relocate the affected premises to a number of alternative sites in the New Territories, those sites were considered not suitable or impracticable due to various issues such as land use incompatibility, environmental concerns, land ownership or accessibility (Appendix II and Plan 8). After a lengthy site-searching process, the Site was identified for relocation as it is relatively flat, in close proximity to NDAs and easily accessible from Kam Tin Road via a local access.

## Applied use is the same as the affected business in HSK

2.5 The proposed development involves the operation of a concrete batching plant with ancillary facilities to support the daily operation of the Site. The existing concrete batching plant consists of 4 cementitious silos with total capacity of about 480 tonnes. The maximum production capacity is about 1,400 m<sup>3</sup> (about) within a 12-hour production period between 07:00 to 19:00. Details of the difference between the original premises and proposed development are shown at **Table 1** below.



	<b>Original Premises</b>	The Site	Difference
	(a)	(b)	(b) — (a)
Site Area	4,835 m <sup>2</sup>	4,411 m <sup>2</sup>	-424 m², -9%
GFA	3,903 m <sup>2</sup>	4,411 m <sup>2</sup>	+508 m <sup>2</sup> , +13%
Silo Capacity	480 t	2,100 t	+337.5 t <i>, +338%</i>
Max. Production Capacity	117 m³/h	200 m³/h	+83 m³/h <i>, +71%</i>

Table 1: Differences between the Original Premises and the Site

2.6 A large portion of the Site is uncovered (i.e. 3,000 m<sup>2</sup> (about), 68% of the Site) and designated for parking and loading/unloading (L/UL) spaces and circulation area to support the daily operation of the Site. A substantial amount of circulation space is reserved within the Site so as to enhance the Site's overall efficiency, as well as to minimize the potential adverse traffic impact to the surrounding road network.

## To meet the growing demand for concrete production

2.7 Ready-mixed concrete is one of the most commonly-used raw materials for construction projects in Hong Kong. With the continued development across various NDAs in the New Territories, the amount of construction works and the demand for concrete arising from planned development projects will remain high. The continued operation of the concrete batching plant can ensure a steady and sustainable concrete supply and reduce cross-district traffic for concrete delivery.

# Approval of the application would not frustrate the long-term planning intention of the "I(D)" zone

- 2.8 Given that the Site falls within an area zoned "I(D)" on the Approved Shek Kong OZP No.: S/YL-SK/9, and the applied use is a Column 2 use within the "I(D)" zone, approval of the current application on a temporary basis would not frustrate the long-term planning intention of the "I(D)" zone and would better utilize deserted land in the New Territories.
- 2.9 The proposed development is generally in line with the planning intention of the "I(D)" zone. The special background of the application should be considered on its individual merit, of which the approval of the current application would therefore not set an undesirable precedent for the "I(D)" zone.

## The proposed development is not incompatible with surrounding land uses

2.10 The proposed development situates in a relatively remote area, which is far away from sensitive receivers. The surrounding area is considered to be predominately in industrial character comprising various brownfield operations. The proposed development is therefore considered not incompatible with surrounding land uses.



2.11 Furthermore, results of the technical assessments (i.e. TIA, EA and drainage proposal) in Chapter 5 of the Planning Statement have indicated that the significant adverse impact arising from the proposed development are not anticipated. Upon approval of the current application, the applicant will make effort in complying with approval conditions in relation to fire services and drainage aspects, so as to minimize the potential adverse impact arisen from the proposed development.



## 3. SITE CONTEXT

## Site Location

3.1 The Site is located approximately 260 m east of Kam Tin Road; 4.5 km east of Kam Sheung Road MTR Station; 12.5 km south of Kwu Tung North NDA; 12.9 km south of Fanling North NDA; and 13.1 km east of HSK/HT NDA; and 16.7 km east of the original premises in HSK.

## Accessibility

3.2 The Site is accessible from Kam Tin Road via a local access (**Plan 1**).

## Existing Site Condition

3.3 The Site is currently used as an open storage yard for junked vehicles, vehicle parts and construction materials. It is generally flat, unfenced and concrete-formed (Plans 1, 3 and 9).

## Surrounding Area

- 3.4 The Site is predominantly for industrial use and is mainly surrounded by open storage yards and temporary structures for various brownfield operations (**Plans 1**, **3** and **9**).
- 3.5 To its immediate north are some vacant temporary structures, across which is a vegetated slope and some unused/vacant land.
- 3.6 To its immediate east is the site of an approved planning application (No. A/YL-SK/275) for logistics centre. To its further east are some open storage yards, warehouses, workshops and logistics centre (Nos. A/YL-SK/279 and 301).
- 3.7 To its immediate south are some open storage of vehicle, vehicle parts, construction machinery and construction material. To its further south are some temporary structures for storage of construction machinery, vehicle dismantling workshops, selling of vehicle products and the site of an approved logistics centre (No. A/YL-SK/307).
- 3.8 To its immediate west are some open storage of vehicles, vehicle parts, construction machinery and construction material. To its further west is Kam Tin Road, across which are some open storage yards and vehicle repair workshops.



## 4. PLANNING CONTEXT

#### Zoning of the Site

4.1 The Site falls within an area zoned "I(D)" on the Approved Shek Kong OZP No.: S/YL-SK/9 (Plan 2). According to the Notes of the OZP, the applied use is a Column 2 use within the "I(D)" zone, which requires planning permission from the Board.

#### Planning Intention

4.2 The planning intention of the "I(D)" zone is primarily for industrial uses that cannot be accommodated in conventional flatted factories due to extensive land and/or high ceiling requirements. It is also intended for the redevelopment of existing informal industrial uses, which are operated in workshop premises in rural area, to properly designed permanent industrial buildings.

#### Statutory Development Restrictions

4.3 According to the Notes of the OZP, no new development, or addition, alteration and/or modification to or redevelopment of an existing building shall result in a total development and/or redevelopment in excess of a <u>maximum plot ratio of 1.6</u>, a <u>maximum site coverage of 80%</u> and a <u>maximum building height of 13 m</u> within the "I(D)" zone on the OZP.

#### Previous and Similar Applications

- 4.4 The Site is not subject of any previous application.
- 4.5 There is no similar approval for 'Concrete Batching Plant' within the same "I(D)" zone.

#### Land Status of the Site

- 4.6 Detailed survey was conducted by the applicant to delineate the lot boundary of the Site, i.e. Lots 573 RP and 1710 in D.D. 114 (Plan 3). The Site consists of 2 private lots with private land area of 4,332 m<sup>2</sup> (about) of Old Schedule Lot held under Block Government Lease (Lot 573 RP in D.D. 114) and 79 m<sup>2</sup> (about) of New Grant Lot (Lot 1710 in D.D. 114).
- 4.7 Given that there is restriction on the erection of structures without prior approval from the Government, the applicant will submit application for Short Term Waiver (STW) to the Lands Department (LandsD) to make way for the erection of the proposed structure at the Site, after planning approval has been obtained from the Board. No structure is proposed for domestic use.



## 5. DEVELOPMENT PROPOSAL

#### Development Details

5.1 The Site consists of an area of 4,411 m<sup>2</sup> (about). Details of development parameters are shown at **Table 2** below.

Application Site Area4,411 m² (about)Covered Area1,411 m² (about)Uncovered Area3,000 m² (about)Plot Ratio1.0 (about)Site Coverage32% (about)Total GFA4,411 m² (about)- Domestic GFANot applicable- Non-Domestic GFA4,411 m² (about)Building HeightNot more than 13 mNo. of StoreyNot more than 3 storeys	•	
Covered Area1,411 m² (about)Uncovered Area3,000 m² (about)Plot Ratio1.0 (about)Site Coverage32% (about)Total GFA4,411 m² (about)- Domestic GFANot applicable- Non-Domestic GFA4,411 m² (about)Building HeightNot more than 13 mNo. of StoreyNot more than 3 storeys	Application Site Area	4,411 m² (about)
Uncovered Area3,000 m² (about)Plot Ratio1.0 (about)Site Coverage32% (about)Total GFA4,411 m² (about)- Domestic GFANot applicable- Non-Domestic GFA4,411 m² (about)Building HeightNot more than 13 mNo. of StoreyNot more than 3 storeys	Covered Area	1,411 m² (about)
Plot Ratio1.0 (about)Site Coverage32% (about)Site Coverage32% (about)Total GFA4,411 m² (about)- Domestic GFANot applicable- Non-Domestic GFA4,411 m² (about)Building HeightNot more than 13 mNo. of StoreyNot more than 3 storeys	Uncovered Area	3,000 m² (about)
Plot Ratio1.0 (about)Site Coverage32% (about)Total GFA4,411 m² (about)- Domestic GFANot applicable- Non-Domestic GFA4,411 m² (about)Building HeightNot more than 13 mNo. of StoreyNot more than 3 storeys		
Site Coverage32% (about)Site Coverage32% (about)Total GFA4,411 m² (about)- Domestic GFANot applicable- Non-Domestic GFA4,411 m² (about)Building HeightNot more than 13 mNo. of StoreyNot more than 3 storeys	Plot Ratio	1.0 (about)
Total GFA4,411 m² (about)- Domestic GFANot applicable- Non-Domestic GFA4,411 m² (about)Building HeightNot more than 13 mNo. of StoreyNot more than 3 storeys	Site Coverage	32% (about)
Total GFA4,411 m² (about)- Domestic GFANot applicable- Non-Domestic GFA4,411 m² (about)Building HeightNot more than 13 mNo. of StoreyNot more than 3 storeys		
- Domestic GFANot applicable- Non-Domestic GFA4,411 m² (about)Building HeightNot more than 13 mNo. of StoreyNot more than 3 storeys	Total GFA	4,411 m² (about)
• Non-Domestic GFA     4,411 m² (about)       Building Height     Not more than 13 m       No. of Storey     Not more than 3 storeys	- Domestic GFA	Not applicable
Building Height     Not more than 13 m       No. of Storey     Not more than 3 storeys	- Non-Domestic GFA	4,411 m² (about)
Building HeightNot more than 13 mNo. of StoreyNot more than 3 storeys		
No. of Storey Not more than 3 storeys	Building Height	Not more than 13 m
	No. of Storey	Not more than 3 storeys

 Table 2: Development Parameters

5.2 The proposed concrete batching plant comprises 4 silos with total capacity of 2,100 tonnes (about) for storage of materials for concrete production; aggregate storage and handling system; concrete mixing towers; ice handling system; water tank and chiller plant; truck washing facilities; and other ancillary supporting facilities with total GFA of 4,411 m<sup>2</sup> (about), covered area of 1,411 m<sup>2</sup> (about) and a maximum building height of 13 m (about). The remaining open area is reserved for parking and L/UL spaces and circulation area (**Plans 10** and **11**).

## **Operation Mode**

5.3 The proposed concrete batching plant consists of 2 production lines, each with a design production capacity of about 100 m<sup>3</sup> per hour. Operation hours are from 07:00 to 23:00 daily, which consist of a 12-hour production period between 07:00 to 19:00, and a delivery period of raw materials beyond peak hours between 19:00 and 23:00. It is estimated that the Site would be able to accommodate about 10 workers. Visitor is not anticipated at the Site.

#### No Traffic Impact

5.4 The Site is accessible from Kam Tin Road via a local access i.e. Chun Yiu Road, which is proposed to be upgraded to allow 2-lane 2-way traffic with a minimum width of



6 m (**Plan 1**). An 11 m (about) wide ingress/egress is proposed at the northern part of the Site. A total of 16 parking and L/UL spaces will be provided (**Plan 10**). Details of parking and L/UL provision are shown at **Table 3** below:

**Table 3**: Provision of Parking and L/UL Spaces

Type of Parking Spaces	No. of Space	
Parking Spaces for Private Cars		
- 2.5 m (W) x 5 m (L)	Z	
Parking Spaces for Heavy Goods Vehicles	12	
- 11 m (W) x 3.5 m (L)	12	
Type of L/UL Spaces	No. of Space	
L/UL Spaces for Container Vehicles	2	
- 16 m (W) x 3.5 m (L)	Z	

5.5 A TIA has been conducted to assess the potential traffic impact induced by the proposed concrete batching plant on nearby road network during the construction and operational stages (**Appendix III**). The assessment result has indicated that all key junctions and road links perform satisfactorily during the peak hours on a normal weekday in the assessment year of 2030, i.e. 5 years upon obtaining planning permission from the Board. The TIA has concluded that <u>no adverse traffic impact</u> arising from the proposed concrete batching plant is anticipated.

#### No Environmental Impact

5.6 An EA covering assessments on air quality, noise, water quality, waste management and land contamination has been conducted to assess the potential environmental impact induced by the proposed concrete batching plant during the construction and operational stages (**Appendix IV**). With proper implementation of mitigation and control measures, the assessment result has indicated that <u>no</u> adverse <u>environmental impact</u> from the proposed concrete batching plant is anticipated.

## No Drainage Impact

5.7 The applicant has submitted a drainage proposal to review the drainage arrangements for the proposed concrete batching plant (**Appendix V**). The existing hard paving has already been taken into consideration. The potential increase in surface runoff generated by the proposed concrete batching plant is estimated to be minimal. A series of perimeter surface drains with sand trap/catch pit are proposed to collect the cumulative runoff, which will finally connect to the proposed discharge point at the stream to the south of the Site via a stormwater drainage pipe. The drainage proposal has concluded that <u>no adverse drainage impact</u> is envisaged.



## Fire Safety Aspect

5.8 The applicant will submit a fire service installations (FSIs) proposal for the consideration of the Director of Fire Services to enhance fire safety of the Site. Upon receiving the STW approval from LandsD and the General Building Plan approval from the Buildings Department for erection of structures, the applicant will implement the accepted FSIs proposal at the Site.



## 6. CONCLUSION

- 6.1 The current application is intended to facilitate the relocation of an existing concrete batching plant and its ancillary facilities in HSK, which will be affected by the HSK/HT NDA development (Plans 4 to 7). Whilst the applicant attempted to relocate the concrete batching plant to a number of alternative sites in the New Territories, those sites were considered not suitable or impracticable (Appendix II and Plan 8). Having considered a number of factors, including locations of concrete demand, land use compatibility, environmental and traffic impacts and concrete delivery time, the Site is considered suitable for relocation as it is located not far away from the original premises, in close proximity to various NDAs and easily accessible.
- 6.2 Being located within an area zoned "I(D)" on the OZP, the proposed concrete batching plant is generally in line with the planning intention of the "I(D)" zone, which is primarily for industrial uses that cannot be accommodated in conventional flatted factories due to extensive land and/or high ceiling requirements. Hence, approval of the application on a temporary basis would not frustrate the long-term planning intention of "I(D)" zone and better utilize deserted land in the New Territories. Given that the relocation is to facilitate the HSK/HT NDA development, approval of the application can facilitate relocation prior to land resumption, thereby minimize the impact on the implementation programme of the HSK/HT NDA.
- 6.3 The proposed development is situated in an area predominated by industrial-related activities, surrounded by temporary structures for various brownfield operations, and is closely connected to various NDAs via nearby public road network; the proposed development is considered not incompatible with the surrounding areas. Approval of the current application would not set an undesirable precedent within the "I(D)" zone, and should be considered on its own merits.
- 6.4 The proposed development will not create significant nuisance to the surrounding areas. Technical assessments and proposals have been provided to demonstrate that adverse traffic, environmental and drainage impacts arising from the proposed development are <u>not</u> anticipated.
- 6.5 In view of the above, the Board is hereby respectfully recommended to <u>approve</u> the subject application for '**Proposed Temporary Concrete Batching Plant with Ancillary Facilities for a Period of 5 Years**'.

R-riches Property Consultants Limited January 2025



#### LIST OF PLANS

Plan 1	Location Plan
Plan 2	Plan showing the Zoning of the Site
Plan 3	Plan showing the Land Status of the Site
Plan 4	Original Premises – Location
Plan 5	Original Premises – Zoning
Plan 6	Original Premises – HSK/HT NDA Phasing
Plan 7	Original Premises – HSK/HT NDA Land Resumption
Plan 8	Plan showing Alternative Sites for Relocation
Plan 9	Aerial Photos of the Site
Plan 10	Layout Plan
Plan 11	Elevation Plan
























#### LIST OF APPENDICES

Appendix I	Details of the Affected Business Premises
Appendix II	Details of Alternative Sites for Relocation
Appendix III	Traffic Impact Assessment
Appendix IV	Environmental Assessment
Appendix V	Drainage Proposal



## Appendix I

Details of the Affected Business Premises



#### Appendix I – Details of the Affected Business Premises

Company Name:

Glorious Concrete (H.K.) Limited 永輝混凝土 (香港) 有限公司

**Details of Business Premises** 

Location:Various Lots in D.D. 130, Hung Shui Kiu, New TerritoriesUse of Premises:Concrete Batching Plant





**Appendices** 20250115 Ver 1.0







**Appendices** 20250115 Ver 1.0







**Appendices** 20250115 Ver 1.0



## Appendix II

Details of Alternative Sites for Relocation



Alternative Sites/ Application Site	Site 1	Site 2	Site 3	Site 4	Site 5	Application Site
Location	Various Lots in D.D. 99, Chau Tau, San Tin, New Territories	Various Lots in D.D. 130, Lam Tei, Tuen Mun, New Territories	Various Lots in D.D. 104, Ngau Tam Mei, Yuen Long, New Territories	Various Lots in D.D. 103, Kam Tin, Yuen Long, New Territories	Various Lots in D.D. 129, Lau Fau Shan, Yuen Long, New Territories	Various Lots in D.D. 114, Shek Kong, Yuen Long, New Territories
Site Area	4,242m <sup>2</sup> (about)	7,130 m² (about)	530 m² (about)	2,230 m <sup>2</sup> (about)	10,740 m <sup>2</sup> (about)	4,411 m <sup>2</sup> (about)
Accessibility	Accessible from Lok Ma Chau Road via a local access	Accessible from Fuk Hang Tsuen Road via a local access	Accessible from Ngau Tam Mei Road via a local access	Accessible from Kam Tin Road via a local access	Accessible from Deep Bay Road via a local access	Accessible from Kam Tin Road via a local access
Distance from Original Premises	19.8 km (about)	5.3 km (about)	15.2 km (about)	12.5 km (about)	9.1 km (about)	16.7 km (about)
Outline Zoning Plan	Approved San Tin Technopole OZP No. S/STT/2	Approved Lam Tei and Yick Yuen OZP No. S/TM-LTYY/12	Approved Ngau Tam Mei OZP No. S/YL-NTM/14	Approved Kam Tin South OZP No. S/YL-KTS/15	Approved Lau Fau Shan and Tsim Bei Tsui OZP No.: S/YL- LFS/11	Approved Shek Kong OZP No. S/YL-SK/9
Zoning	"Other Specified Uses" annotated "Innovation and Technology"	"Comprehensive Development Area"	"Green Belt"	"Agriculture"	"Green Belt"	"Industrial (Group D)"
Existing Condition	Generally flat, partially covered with vegetation and occupied by vacant temporary structures	Hard-paved and occupied by temporary structures	Hard paved and occupied by temporary structures	Vacant and covered with vegetation	Covered with vegetation and woodland	Vacant, formed and generally flat
Surrounding Area	Surrounded by vehicle park, temporary structures for storage, workshop and agricultural uses; and vacant land covered with vegetation and hard-paving	Surrounded by warehouse, workshop, logistic centre and land covered by residential use	Surrounded by greenhouse and agricultural use	Surrounded by open storage, some GIC uses, woodland and residential use	Surrounded by tree groups, temporary structures for open storage and residential use	Surrounded by open storage yards and temporary structures for brownfield operations
Suitability for Relocation	<ul> <li><u>Not suitable</u> for relocation:</li> <li>12% smaller than the original premises</li> <li>in close proximity to existing fish ponds</li> <li>not compatible with the surrounding area</li> </ul>	<ul> <li><u>Not suitable</u> for relocation:</li> <li>47% larger than the original premises</li> <li>in close proximity to residential developments</li> <li>not compatible with the surrounding area</li> </ul>	<ul> <li><u>Not suitable</u> for relocation:</li> <li>89% smaller than the original premises</li> <li>active agricultural activities in the vicinity</li> <li>not compatible with the surrounding area</li> </ul>	<ul> <li><u>Not suitable</u> for relocation:</li> <li>54% smaller than the original premises</li> <li>in close proximity to residential developments</li> <li>not compatible with the surrounding area</li> </ul>	<ul> <li><u>Not suitable</u> for relocation:</li> <li>122% larger than the original premises</li> <li>tree felling is required</li> <li>not compatible with the surrounding area</li> <li>remote location</li> </ul>	<ul> <li><u>Comparatively Suitable</u> for relocation:         <ul> <li>similar site area</li> <li>already formed, relatively flat and vacant</li> <li>in close proximity to NDAs</li> <li>easily accessible</li> <li>no agricultural activity</li> </ul> </li> </ul>

## Appendix II – Alternative Sites for the Relocation of the Applicant's Original Premises in Hung Shui Kiu



## Appendix III

Traffic Impact Assessment



> Traffic Impact Assessment Final Report 21<sup>th</sup> January 2025

Prepared by: CKM Asia Limited

Prepared for: Join Bright Warehousing Limited

## CONTENTS

CH	IAPTER	PAGE
1.	INTRODUCTION Background Scope of Study Structure of Report	1
2.	EXISTING SITUATION The Subject Site The Road Network Historic Traffic Growth Public Transport Facilities Existing Traffic Flows Performance of the Surveyed Junctions Performance of the Surveyed Road Links	2
3.	THE PROPOSED TEMPORARY CONCRETE BATCHING PLANT Development Schedule Vehicular Access Points Upgrading of Chun Yiu Road Proposed Haulage Route Operation of Proposed Temporary Concrete Batching Plant Internal Transport Facilities Swept Path Analysis	5
4.	TRAFFIC IMPACT Design Year Traffic Forecast 2030 Junction Capacity Analysis 2030 Road Link Capacity Analysis	8
5.	SUMMARY Figures Appendix A – Vehicle Composition of Existing Traffic Flows Appendix B – Capacity Analyses	11
	Appendix C – Swept Path Analysis	

## TABLES

## NUMBER

- 2.1 AADT of ATC Stations Located near the Subject Site
- 2.2 Road-Based Public Transport Services Operating near the Subject Site
- 2.3 Existing Junction Performance
- 2.4 Existing Road Link Performance
- 3.1 Details of Vehicle Movements at Different Time Periods
- 3.2 Results of Traffic Generation Surveys at Other Similar Existing Concrete Batching Plants
- 3.2 Internal Transport Facilities
- 4.1 Details of Other Known Major Planned / Committed Development
- 4.2 Traffic Generation of the Proposed Temporary Concrete Batching Plant
- 4.3 2030 Junction Performance
- 4.4 2030 Road Link Performance

## **FIGURES**

## NUMBER

- 1.1 Location of the Subject Site
- 2.1 Road-Based Public Transport Services operating near the Subject Site
- 2.2 Location of Surveyed Junctions and Area of Influence
- 2.3 Junction of Kam Tin Road / Fan Kam Road
- 2.4 Junction of Kam Tin Road / Chun Yiu Road
- 2.5 Junction of Kam Tin Road / Kam Sheung Road / Sheung Tsuen Bus Terminus
- 2.6 Junction of Kam Tin Road / Route Twisk
- 2.7 Existing Peak Hour Hour Traffic Flows
- 3.1 Proposed Upgrading of Chun Yiu Road
- 3.2 Proposed Haulage Routes
- 3.3 Layout of the Proposed Temporary Concrete Batching Plant
- 4.1 Year 2030 Peak Hour Traffic Flows without the Proposed Temporary Concrete Batching Plan
- 4.2 Year 2030 Peak Hour Traffic Flows with the Proposed Temporary Concrete Batching Plant

## 1.0 INTRODUCTION

## Background

- 1.1 The Applicant, Join Bright Warehousing Limited, is authorised by Glorious Concrete (Hong Kong) Limited, the affected business operator, to facilitate the relocation of their concrete batching plant ("the Existing Concrete Batching Plant"). The Existing Concrete Batching Plant is located in Yick Yuen which is within the planned Hung Shui Kiu / Ha Tsuen New Development Area ("HSK/HT NDA"). The captioned site will be acquired by the Government for construction of the planned HSK/HT NDA.
- 1.2 The Applicant is assisting the affected business operator with a proposal to relocate the Existing Concrete Batching Plant to a site located at Lots 573 RP and 1710 in D.D. 114, Shek Kong, Yuen Long (hereinafter "the Proposed Temporary Concrete Batching Plant"). **Figure 1.1** shows location of the Subject Site.
- 1.3 The Subject Site is currently zoned as "*Industrial (Group D)*" in the Approved Shek Kong Outline Zoning Plan No. S/YL-SK/9, and "*Concrete Batching Plant*" is categorized under Column 2 use. Hence, application for permission under Section 16 of the Town Planning Ordinance (Cap. 131), ("S16 Planning Application") is required for the Proposed Temporary Concrete Batching Plant.
- 1.4 In connection, CKM Asia Limited, a traffic and transportation planning consultancy firm, was commissioned to carry out a Traffic Impact Assessment ("TIA") in support of the S16 planning application for the Proposed Temporary Concrete Batching Plant. This TIA report has been updated in responses to the comments provided by Transport Department during the pre-submission stage.

#### Scope of Study

- 1.5 The main objectives of this study are as follows:
  - To assess the existing traffic condition in the vicinity of the Subject Site;
  - To present the provision of internal transport facilities;
  - To quantify the amount of traffic generated by the Proposed Temporary Concrete Batching Plant;
  - To examine the traffic impact on the local road network;
  - To identify any deficiencies in the road network in accommodating the expected traffic generation; and
  - To recommend traffic management proposal and improvement measures, if necessary.

## Structure of Report

- 1.6 After this introduction, the remaining chapters contain the following:
  - Chapter Two Describes the existing condition and surveys;
  - Chapter Three Outlines the Proposed Temporary Concrete Batching Plant;
  - Chapter Four Presents the traffic impact analyses;
  - Chapter Five Summarises the overall conclusion.

## 2.0 EXISTING SITUATION

## The Subject Site

2.1 The Subject Site is located in Shek Kong to the east of Kam Tin Road. It has a site area of approximately 4,411m<sup>2</sup> and is currently occupied by an open storage. Access to the Subject Site is from an access road known as Chun Yiu Road, and is connected to Kam Tin Road.

## The Road Network

- 2.2 Kam Tin Road is a single carriageway 2-way rural road connecting Lam Kam Road with Tai Po to the east, and it is also connected to Tsuen Wan to the south via Route Twisk. To the west, Kam Tin Road connects Castle Peak Road – Yuen Long, Tsing Long Highway (Route 3) and Fan Kam Road.
- 2.3 Chun Yiu Road is a single track access road connecting Kam Tin Road. At present, Chun Yiu Road is some 3m to 4m wide, and a portion of Chun Yiu Road is often occupied by parked vehicles along the northern side.

## Historic Traffic Growth

2.4 Table 2.1 presents the historic annual average daily traffic ("AADT") from the Annual Traffic Census ("ATC") published by the Transport Department for roads located nearby for the latest 5 years, i.e. from 2019 to 2023.

						,	
Station No.	5254	5463	6110	6207	6208	6212	OVERALL
Road	Kam Tin	Lam Kam	Kam Tin	Kam Tin	Kam	Fan Kam	
	Road	Road	Bypass	Road	Sheung	Road	
			<i>,</i> ,		Road		
From	Fan Kam	Kam	Kam Tin	Fan Kam	Kam Tin	Kam Tin	
	Road	Sheung	Road	Road	Road	Road	
		Road					
То	Kam Sheung	Lam Kam	Kam Tin	Kam Sheung	Kam Tin	Castle	
	Road	Road	Road	Road	Road	Peak	
	E. Junction			W. Junction		Road	
Year		Annu	ial Average	Daily Traffic (v	ehicles / day)		
2019	18,510	19,580	14,990	21,300	8,080	11,660	94,120
2020	18,330	19,660	12,810	21,640	9,400	12,250	94,090
2021	19,040	20,420	12,450	20,490	8,960	12,450	93,810
2022	18,850	20,220	12,980	20,520	9,600	12,400	94,570
2023	15,740	20,900	14,490	21,510	10,460	13,890	96,990
	•	•	Ave	rage Annual Gi	owth (2019 -	- 2023) =	0.75%

TABLE 2.1AADT OF ATC STATIONS LOCATED NEAR THE SUBJECT SITE

2.5 Table 2.1 shows that there is increase in AADT in the past 5 years. The average annual growth between 2019 and 2023, is found to be +0.75% per annum.

#### Public Transport Facilities

2.6 Public transport services are available along Kam Tin Road near the Subject Site, and details of the services are given in Table 2.2, and shown in **Figure 2.1**.

TABLE 2.2ROAD-BASEDPUBLICTRANSPORTSERVICESOPERATINGNEAR THE SUBJECT SITE

		<b>_</b> / • · · ·
Route	Routing	Frequency (minutes)
KMB 51	Tsuen Wan (Nina Tower) ひ Sheung Tsuen (Circular)	15 – 60
KMB 54	Yuen Long (West) ひ Sheung Tsuen (Circular)	20 - 30
KMB 64K	Yuen Long (West) ↔ Tai Po Market Station	6 - 20
KMB 64S	Sheung Tsuen Playground → Kam Sheung Road Station	10 – 15 (4 trips) (A)(1)
KMB 251A	Kam Sheung Road Station ひ Sheung Tsuen (Circular)	15 - 30
KMB 251B	Pat Heung Road ひ Sheung Tsuen (Circular)	20 – 30
LWB E36P	Sheung Tsuen → Skycity	60 (2 trips) (A)(2)
	Skycity → Sheung Tsuen	20 (2 trips) (P)(2)
KMB 251M	Sheung Tsuen → Tsuen Wan Station	60 <sup>(A)(1)</sup>
GMB 72	Yuen Long Tai Hang Street ↔ Lui Kung Tin	10
RS NR917	Fan Kam Road, Yuen Long → Wan Chai	M-F: 5 - 15 (7 trips) (A)(1)
		Sat: 15 - 20 (3 trips) (A)(2)
	Wan Chai → Fan Kam Road, Yuen Long	25 – 30 (4 trips) <sup>(P)(1)</sup>
RS NR918	Lam Kam Road → Wan Chai	M-F: 5 - 25 (8 trips) (A)(1)
		Sat: 15 - 20 (3 trips) (A)(2)
	Wan Chai → Lam Kam Road	25 – 30 (3 trips) (P)(1)

Note: KMB – Kowloon Motor Bus LWB – Long Win Bus

GMB – Green Minibus RS – Resident Services

(A) AM Peak Services only.

(P) Afternoon Peak Services only.

(1) Monday to Friday. No services on Saturdays, Sundays and Public Holidays.

(2) Monday to Saturday. No services on Sundays and Public Holidays

### Existing Traffic Flows

- 2.7 To quantify the existing traffic flows in the vicinity of the Subject Site, manual classified counts were conducted on Tuesday, 11<sup>th</sup> June 2024, and also on Friday, 4<sup>th</sup> October 2024 at the following junctions:
  - J01 Junction of Kam Tin Road / Fan Kam Road;
  - J02 Junction of Kam Tin Road / Chun Yiu Road;
  - J03 Junction of Kam Tin Road / Kam Sheung Road / Sheung Tsuen Bus Terminus; and
  - J04 Junction of Kam Tin Road / Lam Kam Road / Route Twist.
- 2.8 Locations of these surveyed junctions and the area of influence ("AOI") are shown in Figure 2.2, and the existing junction layouts are found in Figures 2.3 2.6.
- 2.9 The traffic counts are classified by vehicle type to enable traffic flows in passenger car units ("pcu") to be calculated. This TIA adopted the higher traffic flows observed in October 2024.
- 2.10 Based on the information of the ATC Core Stations 6207 and 6212 for Kam Tin Road and Fan Kam Road, the AM and PM peak hours in the vicinity of the Subject Site is from 0900 to 1000 hours, and from 1900 to 2000 hours. Whereas, the AM and PM peak hours identified from the surveys at the above junctions are found to be from 0800 to 0900 hours, and from 1700 to 1800 hours respectively.

- 2.11 With reference to the ATC, the traffic surveys conducted, and the expected operation of the Proposed Temporary Concrete Batching Plant, the peak hour considered in this TIA included the following:
  - i) 0800 0900 hours ("AM-8 Peak Hour")
  - ii) 0900 1000 hours ("AM-9 Peak Hour")
  - iii) 1700 1800 hours ("PM Peak Hour"), and
  - iv) 1900 2000 hours ("Evening Peak Hour").
- 2.12 **Figure 2.7** presents the observed peak hour traffic flows in pcu/hour, and the detail of vehicle composition is summarised in **Appendix A.**

## Performance of the Surveyed Junctions

2.13 Existing performance of the surveyed junctions are calculated based on the methods outlined in Volume 2 of the Transport Planning and Design Manual ("TPDM"), which is published by the Transport Department. The results of the performance of junctions are summarised in Table 2.3, and detailed calculations of junction performance are found in **Appendix B**.

Ref.	Junction	Type of	Parameter		Peak	Hour	
		Junction		AM-8	AM-9	PM	Evening
J01	Kam Tin Road / Fan Kam Road	Roundabout	RFC	0.498	0.571	0.506	0.592
J02	Kam Tin Road / Chun Yiu Road	Priority	RFC	0.055	0.054	0.167	0.167
J03	Kam Tin Road / Kam Sheung Road	Priority	RFC	0.81	0.76	0.71	0.73
	/ Sheung Tsuen Bus Terminus						
J04	Kam Tin Road / Lam Kam Road /	Roundabout	RFC	0.551	0.529	0.578	0.587
	Route Twist						

#### TABLE 2.3 EXISTING JUNCTION PERFORMANCE

Note: RFC – Ratio of Flow to Capacity

For priority and roundabout, RFC < 1 indicates the junction operates within capacity, and  $RFC \ge 1$  indicates the junction operates at / over capacity.

2.14 The results in Table 2.3 shows that the junctions analysed operate with capacity.

#### Performance of the Surveyed Road Links

2.15 The existing performance, in terms of Peak Hourly Flows / Design Flow Ratio ("P/Df"), of Kam Tin Road and Chun Yiu Road within the AOI are calculated based on the observed traffic flows and the analysis results are summarized in Table 2.4.

Road Link	Configuration	Design Flow	Peak Hour 2-way Traffic Flows (veh/hr) [Peak Hour Flows / Design Flow Ratio (P/Df)]			
		(veh/hr)	AM-8	AM-9	PM	Evening
Kam Tin Road	Single-2 District	1 <i>,</i> 581 <sup>(1)</sup>	836	798	866	877
	Distributor		[0.529]	[0.505]	[0.548]	[0.555]
Chun Yiu Road	Single Track	100 (2)	49	48	80	80
	Access Road		[0.490]	[0.480]	[0.800]	[0.800]

#### TABLE 2.4EXISTING ROAD LINK PERFORMANCE

(1) Design flow in reference with TPDM for single 2-lane 7.3m carriageway, i.e. 1,700 veh/hr for <u>both</u> direction of flow, and reduced by 7% considering percentage of heavy vehicles is between 15% and 20%.

(2) Design flow in reference with TPDM for single track access road, i.e. 100 veh/hr.

2.16 The results in Table 2.4 show that Kam Tin Road and Chun Yiu Road within the AOI operate with capacity.

## 3.0 THE PROPOSED TEMPORARY CONCRETE BATCHING PLANT

## Development Schedule

3.1 The Proposed Temporary Concrete Batching Plant has 2 production lines with maximum concrete production capacity of 100m<sup>3</sup>/hour/line. Sufficient loading / unloading facilities meeting the operational requirements are provided within the Proposed Temporary Concrete Batching Plant.

## Vehicular Access Point

3.2 The Subject Site has an existing vehicular access point at Chun Yiu Road, which connects to Kam Tin Road. The Proposed Temporary Concrete Batching Plant will continue to access via the existing vehicular access point.

## Upgrading of Chun Yiu Road

3.3 Chun Yiu Road will be upgraded and paved to allow 2-lane 2-way traffic with a minimum carriageway width of 6m between Kam Tin Road and the Subject Site. **Figures 3.1** shows the proposed upgrading of Chun Yiu Road, and location of the vehicular access point for the Subject Site.

## Proposed Haulage Route

3.4 Haulage routes to and from the Subject Site are presented in **Figure 3.2**.

## **Operation of Proposed Temporary Concrete Batching Plant**

3.5 The Proposed Temporary Concrete Batching Plant is expected to operate daily from 0700 to 2300 hours. Details of vehicle movements related to the concrete production and the delivery of raw materials are presented in Table 3.1.

Type of	Type of Vehicle	Typical Vehicle	Traffic	Generation (veh	ı/hour)
Vehicle Movement		Dimension (Approx.)	AM Peak Production (0700 – 0900)	Daytime Operation (0900 – 1900)	Evening Raw Material Delivery (1900 – 2300)
Concrete Prod	luction				
Concrete	Concrete Mixer	Heavy Goods Vehicles: 11m (L) x 2.5m (W)	20 (1)	Max. 20 (2)	4
Raw Material	Delivery		•	•	
Aggregate / Sand	Aggregate/ Sand Truck	Heavy Goods Vehicles: 11m (L) x 2.5m (W)	0	10	11
lce	Ice truck		2	2	0
Admixture	Admixture Truck		0	1	0
Waste	Waste Truck		0	1	0
Cement /	Cement / PFA	Articulated Vehicles:	0	4	1
PFA	Tanker	15.4m (L) x 2.5m (W)			
		TOTAL	22	38	16 <sup>(3)</sup>

#### TABLE 3.1 DETAILS OF VEHICLE MOVEMENTS AT DIFFERENT TIME PERIODS

Note:

(1) Maximum concrete production capacity =  $100m^3$  per production line x 2 production lines ÷ typical capacity of  $10m^3$  per concrete mixer truck = 20 nos.

(2) Maximum 20 nos, may be less if hourly production demand does not reach the maximum production capacity.

(3) Due to environmental concern, traffic generation may be limited to either 4 concrete mixers <u>or</u> 12 deliveries of raw material; but to be conservative, a combined number of 16 vehicles in total are assumed in the TIA.

- 3.6 Table 3.1 shows the following:
  - During the <u>AM peak production</u>, i.e. 0700 to 0900 hours, the Proposed Temporary Concrete Batching Plant is expected to operate at its production capacity, and 20 concrete mixer trucks are generated. Raw materials will not be delivered during this time, except for ice which cannot be produced or stored on-site. Hence, a total of 22 vehicles are generated.
  - During the <u>daytime operation</u>, which is from 0900 to 1900 hours, there are no more than 20 concrete mixers and 18 vehicles delivering raw materials per hour. Hence, a total of no more than 38 vehicles are generated.
  - During the <u>evening raw material delivery period</u> which is from 1900 to 2300 hours, the production of concrete is expected to be reduced to only 4 concrete mixers per hour, **or** the delivery of raw materials is expected to be 12 vehicles per hour due to environmental constraints. To be conservative for traffic analysis, the total of 16 vehicles per hour are assumed.
- 3.7 To ensure traffic generation of the Proposed Temporary Concrete Batching Plant will match the estimation presented in Table 3.1 above, the Applicant will require the operator to keep record of all vehicles entering and leaving Proposed Temporary Concrete Batching Plant; and these records can be provided for review by the Authority upon request.

#### *Traffic Generation Surveys at Other Similar Existing Concrete Batching Plant*

- 3.8 To verify the estimated traffic generation presented in Table 3.1, traffic generation surveys were conducted at the following 2 existing concrete batching plants which are located in Tong Yan San Tsuen, Yuen Long:
  - Site 1: Golik Concrete Limited
  - Site 2: Redland Concrete Ltd
- 3.9 Similar to the Proposed Temporary Concrete Batching Plant, the 2 surveyed concrete batching plants have 2 production lines. Table 3.2 presents the survey results.

Hours	Traffic Generation (veh/hour)								
	Site 1 -	Golik Concrete L	imited	Site 2 -	Redland Concre	ete Ltd			
	Concrete	Raw Material	TOTAL	Concrete	Raw Material	TOTAL			
	Mixer	Delivery		Mixer	Delivery				
0700 – 0800	11	2	13	4	2	6			
0800 - 0900	18	13	31	8	3	11			
0900 – 1000	20	12	32	10	6	16			
1000 – 1100	20	18	38	11	4	15			
1100 – 1200	19	15	34	19	9	28			
1200 – 1300	20	13	33	15	3	18			
1300 – 1400	19	16	35	12	5	17			
1400 – 1500	19	14	33	16	6	22			
1500 – 1600	20	9	29	11	7	18			
1600 – 1700	15	1	16	16	5	21			
1700 - 1800	14	2	16	13	7	20			
1800 – 1900	5	1	6	14	4	18			
1900 – 2000	2	0	2	13	2	15			

TABLE 3.2	RESULTS	OF	TRAFFIC	GENERATION	SURVEYS	AT	OTHER
	SIMILAR O	CON	CRETE BAT	CHING PLANTS	5		

- 3.10 Table 3.2 shows that Site 1 generates more traffic than Site 2, and the following are observed:
  - i) The number of concrete mixers trucks generated peaked at 18 to 20 vehicles per hour between 0800 and 1600 hours;
  - ii) The number of raw material delivery vehicles was no more than 19 vehicles per hour between 0800 and 1600 hours; and
  - iii) The maximum number of vehicles generated was 38, including 20 concrete mixer trucks, and 18 raw material delivery vehicles.
- 3.11 Results of the above traffic generation surveys show that the estimated traffic generation for the Proposed Temporary Concrete Batching Plant found in Table 3.1 is of similar order.

## Internal Transport Facilities

3.12 Internal transport facilities provided for the Proposed Temporary Concrete Batching Plant are presented in **Table 3.3**, and the master layout plan is shown in **Figures 3.3** 

Ref.	Туре	Dimension	Quantity						
	Car Parking Spaces								
CP1 & CP2 Car Parking Spaces 5.0m (L) x 2.5m (W) x 2.4m (H)									
Goods Vehicle Loading / Unloading									
LP1, LP6 & LP12	Raw Material Unloading and Waster Collection	11.0m(L) x 3.5m(W) x 4.7m(H)	3						
LP2 – LP5, & LP7 – LP9	Concrete Mixer Waiting Spaces	11.0m(L) x 3.5m(W) x 4.7m(H)	7						
LP10 & LP11	Mixer Loading Bays	11.0m(L) x 3.5m(W) x 4.7m(H)	2						
LP13 & LP14	Raw Material Unloading	16.0m(L) x 3.5m(W) x 4.7m(H)	2						
		TOTAL	2 + 14						

TABLE 3.3 INTERNAL TRANSPORT FACILITIES

## Swept Path Analysis

- 3.13 Swept path analyses using CAD-based program were carried out to ensure ease of vehicle manoeuvring within the Proposed Temporary Concrete Batching Plant. No manoeuvring issue is found.
- 3.14 In addition, in response to Transport Department comment, swept path of articulated vehicles travelling along the proposed haulage route within the AOI was also carried, and no manoeuvring issue is found also.
- 3.15 The swept path analysis drawings are found in the **Appendix C**

## 4.0 TRAFFIC IMPACT

## Design Year

- 4.1 Should the planning application for the Proposed Temporary Concrete Batching Plant be approved by the Town Planning Board in 2025, the planning permission will expire in 5 years, i.e. 2030. Hence, the traffic assessment year adopted is 2030.
- 4.2 The 2 scenarios for year 2030 assessed are:
  - (i) Year 2030 peak hours without the Proposed Temporary Concrete Batching Plant;
  - (ii) Year 2030 peak hours with the Proposed Temporary Concrete Batching Plant

## Traffic Forecast

- 4.3 The design year traffic flows are estimated as follows:
  - (i) Expected traffic growth from 2024 to 2030 with reference to the historic traffic growth from the ATC;
  - (ii) Traffic generated by other known planned / committed developments located in the vicinity, and
  - (iii) Traffic generation of the Proposed Temporary Concrete Batching Plant.
- 4.4 Details of the above are presented in below paragraphs.

## (i) <u>Traffic Growth Rate</u>

4.5 With reference to Table 2.1, a conservative growth rate of 1.0% per annum is adopted to produce the 2030 traffic flows from 2024.

#### (ii) Other Known Planned / Committed Developments

4.6 A review of public domain, including the Town Planning Board's Statutory Planning Portal 3, etc., was undertaken to identify other known major planned / committed developments located in the vicinity. Only 1 development is found with expected completion on or before the design year, and details are presented in Table 4.1.

## TABLE 4.1DETAILS OF OTHER KNOWN MAJOR PLANNED / COMMITTED<br/>DEVELOPMENT WITHIN THE AOI

Location	Parameters
Proposed House Development at Lots No. 1691 RP (Part) and 1691 S.E in	8 Houses
D.D.114 and Adjoining Government Land, East of Kam Tin Road, Pat Heung,	
Yuen Long	

4.7 In addition, the Kam Tin South Public Housing Development located some 3.5km west of the Subject Site to the south of MTR Kam Sheung Road Station is also taken into consideration.

#### *(iii) Traffic Generation of the Proposed Temporary Concrete Batching Plant*

4.8 With reference to Table 3.1, the estimated traffic generation of the Proposed Temporary Concrete Batching Plant is given in Table 4.2.

## TABLE 4.2TRAFFIC GENERATION OF THE PROPOSED TEMPORARY<br/>CONCRETE BATCHING PLANT

ltem	AM Peak Production (AM-8 Peak Hour)		Daytime ( (AM-) PM Pea	Operation 9 and ak Hour)	Evening Raw Material Delivery (Evening Peak Hour)		
	Generation	Attraction	Generation	Attraction	Generation	Attraction	
Total in veh/hr [From Table 3.1]	22	22	38	38	16	16	
PCU Factor	2.5	2.5	2.5	2.5	2.5	2.5	
Total in	55	55	95	95	40	40	
PCU/Hour	110 (2-way)		190 (2-way)		80 (2-way)		

4.9 **Table 4.2** shows that the Proposed Temporary Concrete Batching Plant is expected to generate 110 (2-way) pcu during the AM peak hour, 190 (2-way) pcu during the PM peak hour, and 80 (2-way) during the evening peak hour.

## 2030 Traffic Flows

4.10 Year 2030 traffic flows for the following cases are derived:

2030 Traffic Flows without	= 2024 Existing Traffic Flows +
the Proposed Temporary	Traffic Growth from 2024 to 2030 +
Concrete Batching Plant [A]	Traffic Generated by Other Development
2030 Traffic Flows with the Proposed Temporary Concrete Batching Plant	= [A] + Traffic Generation of the Proposed Temporary Concrete Batching Plant

4.11 **Figures 4.1 and 4.2** show the year 2030 peak hour traffic flows for the cases without and with the Proposed Temporary Concrete Batching Plant.

#### 2030 Junction Capacity Analysis

4.12 The 2030 junction capacity analyses for the cases without and with the Proposed Temporary Concrete Batching Plant are summarised in Table 4.3 and the detailed calculations are found in **Appendix B**.

Ref.	Junction	Without the Proposed Temporary Concrete Batching Plant		With the Proposed Temporary Concrete Batching Plant					
		AM-8	AM-9	PM	Evening	AM-8	AM-9	PM	Evening
J01	Kam Tin Road / Fan Kam Road	0.545	0.622	0.552	0.646	0.570	0.669	0.597	0.666
J02	Kam Tin Road / Chun Yiu	0.060	0.058	0.189	0.189	0.314	0.549	0.749	0.378
	Road								
J03	Kam Tin Road / Kam Sheung	0.91	0.85	0.78	0.81	0.97	0.95	0.88	0.85
	Road / Sheung Tsuen Bus								
	Terminus								
J04	Kam Tin Road / Lam Kam	0.594	0.571	0.622	0.633	0.637	0.646	0.697	0.665
	Road / Route Twist								

## TABLE 4.32030 JUNCTION PERFORMANCE

Note: RFC – Ratio of Flow to Capacity

For priority and roundabout, RFC < 1 indicates the junction operates within capacity, and  $RFC \ge 1$  indicates the junction operates at / over capacity.

4.13 Table 4.3 shows that the junctions analysed have capacity to accommodate the expected traffic growth to 2030 and the traffic generated by the Proposed Temporary Concrete Batching Plant.

#### 2030 Road Link Capacity Analysis

4.14 Road link capacity analyses for 2030 without and with the Proposed Temporary Concrete Batching Plant are summarised in **Table 4.4**.

Road Link	Configuration	Design Flow	Peak Hour 2-way Traffic Flows [Peak Hour Flows / Design Flow Ratio (P/Df)]					
		(veh/hr)	AM-8	AM-9	PM	Evening		
Without the Proposed Temporary Concrete Batching Plant								
Kam Tin Road	Single-2 District	1,581 <sup>(1)</sup>	903	865	921	926		
	Distributor		[0.571]	[0.547]	[0.583]	[0.586]		
Chun Yiu Road	Single Track	100 (2)	53	52	84	84		
	Access Road		[0.530]	[0.520]	[0.840]	[0.840]		
With the Proposed Temporary Concrete Batching Plant								
Kam Tin Road	Single-2 District	1,581 <sup>(1)</sup>	999	1,031	1,087	995		
	Distributor		[0.632]	[0.652]	[0.688]	[0.629]		
Chun Yiu Road	Single-2 Local	720 <sup>(3)</sup>	171 (4)	250 (4)	287 (4)	177 (4)		
	Road		[0.238]	[0.347]	[0.399]	[0.246]		
1) Design flow in reference with TPDM for single 2-lane 7.3m carriageway, i.e. 1,700 veh/hr for both direction of								

#### TABLE 4.4 2030 ROAD LINK PERFORMANCE

flow, and reduced by 7% considering percentage of heavy vehicles is between 15% and 20%
(2) Design flow in reference with TPDM for single track access road, i.e. 100 veh/hr.

(2) Design now in reference with TPDM for single track access road, i.e. 100 ven/hr.
 (3) Design flow in reference with TPDM for single 2-lane local road, i.e. 800 veh/hr, 2-way, and reduced by 10% considering high percentage of heavy vehicles

(4) In view of very high percentage of heavy vehicles, traffic flows in passenger car unit is adopted.

4.15 Table 4.4 shows that both Kam Tin Road and the upgraded Chun Yiu Road have capacity to accommodate the expected traffic growth in 2030 and the traffic generated by the Proposed Temporary Concrete Batching Plant.

## 5.0 SUMMARY

- 5.1 The Applicant proposes to relocate the concrete batching plant from Yick Yuen, Hung Sui Kiu in Yuen Long to the Subject Site which is in Shek Kong due to land acquisition by the Government for construction of the planned HSK/HT NDA.
- 5.2 The Proposed Temporary Concrete Batching Plant will have 2 production lines, and provides sufficient internal transport facilities including concrete mixer truck waiting spaces and queuing area to ensure that vehicles will not queue back onto the public road. The existing vehicle access to the Subject Site, which is Chun Yiu Road is proposed to be upgraded to 2-lane with a minimum carriageway width of 6m to serve the Proposed Temporary Concrete Batching Plant.
- 5.3 Manual classified counts were conducted at junctions located in the vicinity of the Proposed Temporary Concrete Batching Plant in order to establish the peak hour traffic flows. Currently, the junctions operate with capacities during the AM, PM and evening PM peak hours.
- 5.4 The Proposed Temporary Concrete Batching Plant is expected to be completed by 2026, and the capacity analyses are undertaken for year 2030. All junctions and road link analysed have sufficient capacity to accommodate the expected traffic flow to 2030 and the traffic generated by the Proposed Temporary Concrete Batching Plant.
- 5.5 From traffic engineering viewpoint, the Proposed Temporary Concrete Batching Plant is considered acceptable.

# Figures



\JOB\J7300-J7349\J7342\(2025 01) TIA\_R2\Fig 1 1 RevA.dwg



IOB\J7300-J7349\J7342\(2025 01) TIA\_R2\Fig 2.1 RevA.dwg




4.0 TS TS TS TS TS 43.3 + 43.3 + 43.5 + ET ET ET 43.6	TS TS TS TS TS TS TS TS TS TS TS TS TS T	J. THURSON IN THE STREET S
TS T	4.0 TS TS TS TS TS TS TS TS TS TS TS TS TS	43.3 +
Project Title PROPOSED TEMPORARY CONCRETE BATCHING PLANT WITH ANCILLARY FACILITIES FOR A PERIOD OF 5 YEARS AT LOTS 573 RP AND 1710 IN D.D. 114, SHEK KONG, YUEN LONG, NT Figure Title Figure Title Figure Title Figure Title Figure No.	Project Title     PROPOSED TEMPORARY CONCRETE BATCHING PLANT WITH ANCILLARY FACILITIES FOR A PERIOD OF 5 YEARS AT LOTS 573 RP AND 1710 IN D.D. 114, SHEK KONG, YUEN LONG, NT J7342	Figure No. 2.4 Designed by M.C.Y. S.C.Y. K.C. CKM Asia Limited Traffic and Transportation Planning Consultants 21st Floor: Methodist House: 36 Hennessy Road







\JOB\J7300-J7349\J7342\(2025 01) TIA\_R2\Fig 2.7 4.1 4.2 RevA.dwg







9\J7342\(2025 01) TIA\_R2\Fig 3.3 & SP1XX Rev A dwg \JOB\J7300



4.2 RevA dwg (2025 01) TIA\_R2\Fig 2.7 4.1 7342\( 349/J7 



OB\J7300-J7349\J7342\(2025 01) TIA\_R2\Fig 2.7 4.1 4.2 RevA.dwg

Appendix A – Vehicle Composition of Existing Traffic Flows



\JOB\J7300-J7349\J7342\(2025 01) TIA\_R2\Fig 2.7 4.1 4.2 RevA dwg

						0						ΤΟΤΑΙ
ID	мс	PC	Taxi	Van	LGV	M/HGV	PLB	PrLB	NFB	FBSD	FBDD	(VEH)
J01-01	6	181	14	37	11	30	26	2	3	2	3	315
J01-02	1	44	6	9	7	4	0	2	0	0	0	73
J01-03	0	3	0	0	0	1	0	0	0	0	0	4
J01-04	2	61	3	11	5	4	0	0	0	0	0	86
J01-05	9	170	40	26	14	29	7	4	0	0	4	303
J01-06	0	1	0	0	0	2	0	0	0	0	0	3
J01-07	8	202	20	14	11	21	23	2	1	1	3	306
J01-08	10	201	14	29	18	23	9	1	1	1	7	314
J01-09	0	0	0	0	0	0	0	0	0	0	0	0
J02-01	14	269	15	64	22	18	7	1	4	3	3	420
J02-02	0	8	0	1	3	0	0	0	0	0	0	12
J02-03	0	5	0	1	2	2	0	0	0	0	0	10
J02-04	0	6	0	0	0	0	0	0	0	0	0	6
J02-05	0	13	1	2	5	0	0	0	0	0	0	21
J02-06	22	242	11	51	31	19	8	2	3	2	3	394
J03-01	22	230	10	48	31	21	8	2	3	0	0	375
J03-02	0	33	0	32	5	2	0	1	2	0	0	75
J03-03	5	116	9	21	14	8	8	0	0	0	8	189
J03-04	5	95	16	21	19	16	9	4	4	0	8	197
J03-05	14	241	15	32	20	16	7	0	2	3	3	353
J03-06	0	18	1	3	7	0	0	0	0	3	3	35
J04-01	8	182	15	26	19	15	10	0	0	0	4	279
J04-02	9	185	5	27	11	3	6	0	0	0	0	246
J04-03	0	37	2	6	6	13	0	3	3	2	8	80
J04-04	6	79	2	8	4	2	9	0	0	0	0	110
J04-05	1	11	0	2	0	2	0	0	0	0	0	16
J04-06	0	0	0	0	0	0	0	0	0	0	0	0
J04-07	4	121	0	6	0	5	0	0	0	0	0	136
J04-08	5	265	17	34	6	19	9	0	3	0	6	364
J04-09	0	0	0	0	0	3	0	0	0	0	0	3

TABLE A1COMPOSITION FOR EXISTING TRAFFIC FLOW AT AM-8 PEAK HOUR

	_ 0			10112					7 4 7 1 3			ΤΟΤΑΙ
ID	МС	РС	Taxi	Van	LGV	M/HGV	PLB	PrLB	NFB	FBSD	FBDD	(VEH)
J01-01	6	187	16	35	12	44	25	4	4	2	3	338
J01-02	1	53	9	14	10	5	0	4	0	0	0	96
J01-03	0	4	0	0	0	1	0	0	0	0	0	5
J01-04	3	64	5	16	9	7	0	0	0	0	0	104
J01-05	9	211	39	25	13	28	7	4	0	0	4	340
J01-06	0	1	0	0	0	2	0	0	0	0	0	3
J01-07	8	242	19	13	10	20	22	2	1	1	3	341
J01-08	13	208	20	47	26	30	9	2	1	1	7	364
J01-09	0	0	0	0	0	0	0	0	0	0	0	0
J02-01	13	256	14	60	21	18	7	1	4	3	3	400
J02-02	0	8	0	1	3	0	0	0	0	0	0	12
J02-03	0	5	0	1	2	2	0	0	0	0	0	10
J02-04	0	6	0	0	0	0	0	0	0	0	0	6
J02-05	0	12	1	2	5	0	0	0	0	0	0	20
J02-06	21	231	10	49	29	18	8	2	3	2	3	376
J03-01	21	220	9	47	29	20	8	2	3	0	0	359
J03-02	0	31	0	30	5	2	0	1	2	0	0	71
J03-03	5	111	9	20	13	8	8	0	0	0	8	182
J03-04	5	90	15	20	18	16	9	4	4	0	8	189
J03-05	13	229	14	30	19	16	7	0	2	3	3	336
J03-06	0	17	1	3	7	0	0	0	0	3	3	34
J04-01	9	186	17	22	15	17	9	0	0	0	4	279
J04-02	10	185	5	28	11	3	6	0	0	0	0	248
J04-03	0	1	2	7	7	15	0	3	3	2	8	48
J04-04	8	94	2	9	5	2	9	0	0	0	0	129
J04-05	1	10	0	2	0	2	0	0	0	0	0	15
J04-06	0	0	0	0	0	0	0	0	0	0	0	0
J04-07	4	116	0	6	0	5	0	0	0	0	0	131
J04-08	7	238	22	35	8	19	9	0	3	0	6	347
J04-09	0	0	0	0	0	3	0	0	0	0	0	3

TABLE A2COMPOSITION FOR EXISTING TRAFFIC FLOW AT AM-9 PEAK HOUR

						1						TOTAL
ID	мс	PC	Taxi	Van	LGV	M/HGV	PLB	PrLB	NFB	FBSD	FBDD	(VEH)
J01-01	13	210	7	26	9	10	28	0	3	2	3	311
J01-02	6	57	5	11	9	4	0	0	1	0	0	93
J01-03	0	2	0	0	0	0	0	0	0	0	0	2
J01-04	8	100	4	6	11	8	2	0	1	0	0	140
J01-05	3	290	29	27	24	9	5	0	1	0	3	391
J01-06	0	2	1	0	0	0	0	0	0	0	0	3
J01-07	5	254	33	27	8	15	21	0	0	0	4	367
J01-08	15	194	10	31	8	14	8	1	3	2	3	289
J01-09	0	5	1	0	0	0	0	0	0	0	0	6
J02-01	17	246	15	44	20	27	5	3	1	1	2	381
J02-02	0	12	0	2	4	1	0	0	0	0	0	19
J02-03	0	13	0	4	3	1	0	0	0	0	0	21
J02-04	0	13	0	2	5	2	0	0	0	0	0	22
J02-05	0	13	0	3	2	0	0	0	0	0	0	18
J02-06	26	280	19	56	25	21	7	4	2	2	3	445
J03-01	26	282	17	57	28	22	7	4	2	0	0	445
J03-02	1	40	7	14	4	9	0	2	0	0	0	77
J03-03	5	104	6	27	3	5	7	0	0	0	7	164
J03-04	11	149	13	20	20	8	9	3	0	0	11	244
J03-05	15	206	7	30	19	17	5	1	1	1	2	304
J03-06	0	29	1	5	1	1	0	0	0	1	3	41
J04-01	9	307	14	21	17	8	9	0	1	0	6	392
J04-02	8	120	6	8	8	0	8	0	0	1	0	159
J04-03	2	71	8	11	7	3	2	0	1	2	6	113
J04-04	6	129	2	19	2	2	7	0	0	1	0	168
J04-05	3	15	0	4	0	1	0	0	0	0	0	23
J04-06	0	0	0	0	0	0	0	0	0	0	0	0
J04-07	3	73	0	0	0	1	0	0	0	0	0	77
J04-08	8	212	9	18	7	10	12	2	0	0	7	285
J04-09	0	1	0	0	0	2	0	0	0	0	0	3

TABLE A3COMPOSITION FOR EXISTING TRAFFIC FLOW AT PM PEAK HOUR

					/	0 110 11		<u> </u>				011
ID	мс	PC	Taxi	Van	LGV	M/HGV	PLB	PrLB	NFB	FBSD	FBDD	TOTAL (VEH)
J01-01	25	274	15	47	22	21	29	1	5	2	3	444
J01-02	13	63	10	26	21	7	0	0	2	0	0	142
J01-03	0	4	0	0	0	0	0	0	0	0	0	4
J01-04	13	131	8	8	16	10	2	0	1	0	0	189
J01-05	3	279	30	28	25	9	5	0	1	0	3	383
J01-06	0	2	1	0	0	0	0	0	0	0	0	3
J01-07	5	256	34	28	8	15	22	0	0	0	4	372
J01-08	24	251	20	42	10	18	8	1	3	2	3	382
J01-09	0	5	1	0	0	0	0	0	0	0	0	6
J02-01	16	259	14	44	21	24	5	3	1	1	2	390
J02-02	0	12	0	2	4	1	0	0	0	0	0	19
J02-03	0	13	0	4	3	1	0	0	0	0	0	21
J02-04	0	13	0	2	5	2	0	0	0	0	0	22
J02-05	0	13	0	3	2	0	0	0	0	0	0	18
J02-06	27	282	19	56	26	19	7	4	2	2	3	447
J03-01	27	285	16	57	29	20	7	4	2	0	0	447
J03-02	1	41	7	14	4	9	0	2	0	0	0	78
J03-03	5	107	6	28	3	5	7	0	0	0	7	168
J03-04	10	163	13	21	21	8	9	3	0	0	10	258
J03-05	15	222	7	31	20	16	5	1	1	1	2	321
J03-06	0	30	1	5	1	1	0	0	0	1	3	42
J04-01	9	315	15	23	19	8	9	0	1	0	6	405
J04-02	8	130	6	8	8	0	8	0	0	1	0	169
J04-03	2	51	8	10	7	3	2	0	1	2	6	92
J04-04	12	142	5	28	4	2	7	0	0	1	0	201
J04-05	3	40	0	4	0	1	0	0	0	0	0	48
J04-06	0	0	0	0	0	0	0	0	0	0	0	0
J04-07	3	74	0	0	0	1	0	0	0	0	0	78
J04-08	16	171	19	32	10	20	12	2	0	0	7	289
J04-09	0	1	0	0	0	2	0	0	0	0	0	3

TABLE A4COMPOSITION FOR EXISTING TRAFFIC FLOW AT EVENING PEAK HOUR

Appendix B – Capacity Analyses

lunction:		Kam Tin P	Poad / Ean k	(am Road						lo	h Number	17342
Scopario:			ondition	Valli Noau						- 30		1
Design V	aar.	2024		signed By	· MCV	C	becked By:	WCH		Date:	21 Jonu	1 2025
Design re	-ai.	2024	. De	signed by		C	neckeu by.	WCH		Dale.	21 Janu	ary 2025
AM(08) P	EAK											
Arm	To A	To B	To C	To D	To E	To F	To G	To H	Total	q <sub>c</sub>		
From A	0	358	374						732	94		
From B	359	6	95						460	379		
From C	386	83	5						474	365		
From D												
From E												
From F												
From G												
From H												
Total	745	447	474						1666			
										•	•	
PM (17) P	Peak											
Arm	To A	To B	To C	To D	To E	To F	To G	To H	Total	q <sub>c</sub>		
From A	6	407	325						738	109		
From B	426	3	157						586	333		
From C	354	104	2						460	435		
From D												
From E												
From F												
From G												
From H												
Total	786	514	484						1784			
Lonord					Coordinia	Devened						
Arm	Pood	(in clockwic	o ordor)		Geometric			r (m)	L (m)	D (m)	<i>(</i> 2)	0
	Kom Tin I		e oldel)		From A	e (m)	v (III)	25.0	10.0	D (III) 25	<u>()</u>	0.7
	Ean Kom				FIOIII A From B	7.7	2.0	15.0	10.0	25	40	0.7
	Kom Tin I				From C	7.7	2.9 5.5	100.0	100.0	25	50 60	0.0
					From D	1.1	5.5	100.0	100.0	25	00	0.0
F					From E							
F					From F							
G					From G							
н					From H							
Predictiv	e Equatior	n Q <sub>E</sub> = K(F -	f <sub>c</sub> q <sub>c</sub> )			Limitatio	n					
Q <sub>E</sub>	Entry Cap	pacity				е	Entry Wid	th		4.0 - 15.0	m	
q <sub>c</sub>	Circulatin	g Flow acros	ss the Entry	/		v	Approach	Half Width		2.0 - 7.3 m	ı	
К	= 1-0.003	47(∅-30)-0.	978[(1/r)-0.	05]		r	Entry Rad	ius		6.0 - 100.0	) m	
F	$= 303x_2$	(1				L	Effective L	ength of Fl	are	1.0 - 100.0	) m	
t <sub>c</sub>	$= 0.210t_{D}$	(1+0.2x <sub>2</sub> )				D	Inscribed	Circle Diam	eter	15 - 100 m	ı	
t <sub>D</sub>	= 1+0.5/(	1+M)				Ø	Entry Ang	le		10° - 60°		
М	= exp[(D-	60)/10]				S	Sharpness	s of Flare		0.0 - 3.0		
X <sub>2</sub>	= v+(e-v)/	/(1+2S)										
S	= 1.6(e-v)	)/L										
Ratio-of-I	Flow to Ca	pacity (RFC	C)									
								ζ <sub>E</sub>	Entr	y Flow	RI	=C
Arm	x <sub>2</sub>	Μ	t <sub>D</sub>	K	F	t <sub>c</sub>	AM(08)	PM (17)	AM(08)	PM (17)	AM(08)	PM (17)
From A	5.112	0.030	1.485	0.986	1549	0.631	1469	1460	732	738	0.498	0.506
From B	5.029	0.030	1.485	0.984	1524	0.626	1266	1294	460	586	0.363	0.453
From C	/.591	0.030	1.485	0.935	2300.174	0.786	1883	1831	474	460	0.252	0.251
From D												
From E												
From C												
From H												
	1						1				I	

Junction:		Kam Tin R	Road / Fan I	Kam Road						Jo	b Number:	J7342
Scenario:		Existing C	ondition							-	J01 - P.	2
Design Ye	ear:	2024	De	signed By:	MCY	С	hecked By:	WCH	_	Date:	21 Janu	ary 2025
AM (09) F	Peak											
Arm	To A	To B	To C	To D	To E	To F	To G	To H	Total	q <sub>c</sub>		
From A	0	390	438						828	123		
From B	394	6	119						519	444		
From C	429	111	6						546	400		
From D												
From E												
From F												
From G												
From H												
Total	823	507	563						1893			
DM (19) E	Doak											
Arm	To A	To B	To C	To D	To E	To F	To G	To H	Total	q <sub>c</sub>		
From A	6	413	423						842	170		
From B	419	3	210						632	433		
From C	512	163	4						679	428		
From D												
From E												
From F												
From G												
From H												
Total	937	579	637						2153			
Legend	1				Geometric	Paramet	ers			- / >		
Arm	Road (		e order)		Arm	e (m)	v (m)	r (m)	L (m)	D (m)	Ø (°)	<u>S</u>
					From A	1.1	3.3	35.0	10.0	25	40	0.7
В	Fan Kam				From B	1.1	2.9	15.0	12.5	25	30	0.6
	Kam IIn F	Koad (WB)			From C	1.1	5.5	100.0	100.0	25	60	0.0
					From D							
					FIOILE From E							
					From G							
н					From H							
Predictiv	e Equation	Q <sub>E</sub> = K(F -	f <sub>c</sub> q <sub>c</sub> )			Limitatio	n					
Q <sub>E</sub>	Entry Cap	acity				е	Entry Widt	th		4.0 - 15.0	m	
q <sub>c</sub>	Circulating	g Flow acros	ss the Entry	/		v	Approach	Half Width		2.0 - 7.3 m	1	
K	= 1-0.003	47(Ø-30)-0.	.978[(1/r)-0.	05]		r	Entry Radi	ius		6.0 - 100.0	) m	
⊢ f	$-303x_2$	1+0.22 )				L	Effective L	ength of FI	are	1.0 - 100.0	m	
1 <sub>C</sub>		$(+0.2x_2)$				D	Inscribed (	Circle Diam	leter	15 - 100 m	1	
ч <sub>D</sub>	= 1+0.5/(1	+IVI)				0	Entry Angi	e		10° - 60°		
M	= exp[(D-6	50)/10] (1.28)				S	Sharpness	s of Flare		0.0 - 3.0		
^2 0	= v+(e-v)/	(1+25)										
	= 1.6(e-v)	/L			1							
Ratio-of-I	Flow to Ca	pacity (RFC	C)									
A			+		-	£			Entry	/ Flow	RI	FC
Arm	X <sub>2</sub>	M	ι <sub>D</sub>	<u>к</u>	+		AM(09)	PM(19)	AM(09)	PM(19)	AM(09)	PM(19)
	5.112	0.030	1.405 1.405	0.900	1549	0.031	1451	1422	δ28 510	622	0.071	0.592
From C	5.029	0.030	1.405	0.984	1024	0.020	1057	1232	519	670	0.423	0.313
	1.591	0.030	1.400	0.935	2300.174	0.780	1007	1030	540	0/9	0.294	0.370
From E												
From G												
From H												
	1						1		I			

Junction:		Kam Tin R	oad / Fan ł	Kam Road						Jo	b Number:	J7342
Scenario:		Without Proposed Temporary Concrete Batching Plant     J01 - P. 3       2030     Designed By:     MCV       Checked By:     WCH     Date:       21.January 2025								3		
Design Ye	ear:	2030	De	signed By	: MCY	Checked By: WCH Date: 21 Janua						
AM(08) P	EAK											
Arm	IO A	I o B	To C	ToD	IOE	IOF	To G	IOH	lotal	q <sub>c</sub>		
From A	0	390	408						798	100		
From B	396	6	101						503	413		
From C	425	89	5						519	402		
From D												
From E												
From F												
From G												
From H												
Total	821	485	514						1820			
PM (17) F	Peak											
Arm	To A	To B	To C	To D	To E	To F	To G	To H	Total	q <sub>c</sub>		
From A	6	442	356						804	116		
From B	459	3	168						630	364		
From C	384	111	2						497	468		
From D												
From E												
From F												
From G												
From H												
Total	849	556	526						1931			
Legend	1				Geometric	Paramet	ers					-
Arm	Road (		e order)		Arm	e (m)	v (m)	r (m)	L (m)	D (m)	Ø (°)	S
A	Kam Tin F	Road (EB)			From A	7.7	3.3	35.0	10.0	25	40	0.7
В	Fan Kam	Road (SB)			From B	7.7	2.9	15.0	12.5	25	30	0.6
С	Kam Tin F	Road (WB)			From C	7.7	5.5	100.0	100.0	25	60	0.0
D					From D							
E					From E							
F					From F							
G					From G							
H					From H							
Predictiv	e Equation	$O_{-} = K(F_{-})$	fa)			Limitatio	<b>n</b>					
Q <sub>E</sub>	Entry Can	acity	·c4c/		ז ר		Entry Wid	th		40-150	m	
с <u>ь</u>	Circulating	a Flow acros	e the Entry	,		v	Approach	Half Width		20 - 73 m	,	
90 K		$47(\propto 20)$ 0	070[(1/r) 0	051		v -	Entry Ded			2.0 - 7.0 1	1	
r. E	$= 303x_{0}$	+7( <i>©</i> -30)-0.3	970[(1/1)-0.	05]		r I	Entry Rad	enath of El	are	0.0 - 100.0	) m	
f	$= 0.210t_{-}($	$(1+0.2x_{o})$					Incoribod	Circle Diem	ator	15 100 m	,	
'c t_	- 1+0 5//4	+M)				a a	Entry And			100 600	•	
ч <u>о</u>	- 1+0.5/(1					0				10 - 00		
IVI X-	$= \exp[(D - \theta)]$	(1,28)			L	5	Sharphese	s of Flare		0.0 - 3.0		
^2 0	- v+(e-v)/	(1+23)										
5	= 1.6(e-v)	/L			1							
Ratio-of-	Flow to Ca	pacity (RFC	;)									
٨rm	Xa	М	ta	ĸ	F	f		J <sub>E</sub> DM (17)				-C
From A	5 110	0.030	1 /95	0.096	15/0	-c 0 621	1/65	1/55	700	804	0.545	0.552
From P	5.020	0.030	1 /05	0.900	1594	0.031	12/5	1400	502	620	0.040	0.002
From C	3.029	0.030	1.400	0.904	1024	0.020	1240	12/5	503	407	0.404	0.494
	1.591	0.030	1.405	0.935	2300.174	0.780	1000	1807	919	497	0.280	0.275
From E												
From F												
From G												
гюш Н												

Junction:		Kam Tin R	load / Fan I	Kam Road						Jo	b Number:	J7342
Scenario:		Without Proposed Temporary Concrete Batching Plant     J01 - P. 4       2030     Designed By:     MCY     Checked By:     WCH     Date:     21.January 2025								4		
Design Ye	ear:	2030 Designed By: MCY Checked By: WCH Date: 21 January 20								ary 2025		
AM (09) F	Peak	To D	Ta C	T <sub>2</sub> D	То Г	ТаГ	Ta C	Tall	Tatal		1	
	10 A	10 0	10 C	10 D	TOE	10 F	10 G	10 1	10121	4c		
From A	122	424	476						900	131		
FIOIII B	433	110	6						505	402		
From D	470	119	0						595	439		
From E												
From E												
From G												
From H												
Total	903	549	608						2060			
Total	000	010	000						2000	1	1	
PM (19) F	Peak											
Arm	To A	To B	To C	To D	To E	To F	To G	To H	Total	q <sub>c</sub>		
From A	6	448	460						914	181		
From B	452	3	224						679	470		
From C	551	174	4						729	461		
From D												
From E												
From F												
From G												
From H	4000											
lotal	1009	625	688						2322			
Legend					Geometric	Paramot	are					
Arm	Road (	in clockwise	a order)		Arm		v (m)	r (m)	L (m)	D (m)	Ø (°)	9
	Kam Tin F	ad (FR)			From A	7.7	33	35.0	10.0	25	40	0.7
В	Fan Kam I	Road (SB)			From B	77	2.9	15.0	12.5	25	30	0.6
C	Kam Tin F	Road (WB)			From C	77	5.5	100.0	100.0	25	60	0.0
D					From D							
E					From E							
F					From F							
G					From G							
Н					From H							
Predictiv	e Equation	Q <sub>E</sub> = K(F -	f <sub>c</sub> q <sub>c</sub> )			Limitatio	n					
Q <sub>E</sub>	Entry Cap	acity				е	Entry Wid	th		4.0 - 15.0	m	
q <sub>c</sub>	Circulating	g Flow acros	ss the Entry	/		v	Approach	Half Width		2.0 - 7.3 m	ו	
K	= 1-0.0034	47(Ø-30)-0.	978[(1/r)-0.	.05]		r	Entry Rad	ius		6.0 - 100.0	) m	
⊢ f	$-303X_2$	1+0.2v)				L	Effective L	Length of FI	are	1.0 - 100.0	) m	
і <sub>с</sub> +		$(+0.2x_2)$				D		Circle Diam	leter	15 - 100 m	1	
ч <sub>D</sub>	= 1+0.5/(1	+M)				0	Entry Ang	ie ( Ei		10° - 60°		
M	= exp[(D-6	00)/10] (1,28)			L	8	Sharpness	s of Flare		0.0 - 3.0		
^2 S	= 1.6(0.1)	(1+23) //										
5	= 1.6(e-v)/	/L			1							
Ratio-of-	Flow to Ca	pacity (RFC	C)						1		1	
۸rm	Y.	N/	t_	ĸ	E	f			Entr	y Flow	RI AM(00)	
From A	5 110	0.030	1 / 95	0.096	15/0	10 0 621	11/16	1/15		014		0.646
From R	5 020	0.030	1 485	0.900	159/	0.626	1202	1210	565	679	0.022	0.561
From C	7 501	0.030	1 485	0.004	2300 17/	0.786	1828	1812	595	729	0.325	0.001
From D	1.591	0.030	1.400	0.930	2000.174	0.700	1020	1012	395	129	0.020	0.402
From F												
From F												
From G												
From H												
	1						1		1		1	

Junction:		Kam Tin R	oad / Fan k	Kam Road						Jo	b Number:	J7342
Scenario:		With Propo	osed Temp	orary Con	crete Batchir	ng Plant					J01 - P.	5
Design Ye	ear:	2030	De	signed By	: MCY	С	hecked By:	WCH		Date:	21 Janu	ary 2025
AM(08) P	EAK									1	1	
Arm	To A	To B	To C	To D	To E	To F	To G	To H	Total	Q <sub>c</sub>		
From A	0	390	436						826	127		
From B	396	6	128						530	441		
From C	453	116	5						574	402		
From D												
From E												
From F												
From G												
From H												
Total	849	512	569						1930			
											-	
PM (17) P	Peak											
Arm	To A	То В	To C	To D	To E	To F	To G	To H	Total	q <sub>c</sub>		
From A	6	442	404						852	163		
From B	459	3	215						677	412		
From C	432	158	2						592	468		
From D												
From E												
From F												
From G												
From H												
Total	897	603	621						2121			
					_	_						
Legend	<u> </u>			1	Geometric	Paramet	ers					-
Arm	Road (		e order)		Arm	e (m)	v (m)	r (m)	L (m)	D (m)	Ø (°)	S
A		Road (EB)			From A	1.1	3.3	35.0	10.0	25	40	0.7
В	Fan Kam I	Road (SB)			From B	1.1	2.9	15.0	12.5	25	30	0.6
С	Kam Tin F	Road (WB)			From C	7.7	5.5	100.0	100.0	25	60	0.0
D	0				From D							
E					From E							
F					From F							
G					From G							
н					From H							
Predictive	e Equation	$Q_r = K(F -$	f.g.)			Limitatio	n					
Q <sub>F</sub>	Entry Cap		•6467		ו ר	e	Entry Widt	h		4 0 - 15 0	m	1
a.	Circulating	n Flow acros	ss the Entry	,		v	Approach	Half Width		20-73m	1	
-ic K	- 1 0 003	17(7 30) 0	078[(1/r) 0	051		r	Entry Podi			6.0 100.0	) m	
F	$= 303x_{2}$	τι (∞ <b>-</b> 50)-0.	o i ol( i /i )-0.	551		I	Effective I	enath of FI	are	10-100.0	) m	
f	= 0.210t <sub>p</sub> (	1+0.2x <sub>2</sub> )					Inscribed (	Circle Diam	eter	15 - 100 m	1	
to	= 1+0 5//1	+M)				Ø	Entry Anal	e		10° - 60°	•	
M		SOV/101				6	Sharphoes	of Flore				
X <sub>2</sub>	$= v + (e_v)/i$	(1+2S)			'	- 5	Sharphese			0.0 - 3.0		
5 S	$= 1.6(e_{1}y)$	(1.20)										
	- 1.0(e-v)/				1							
Ratio-of-F	Flow to Ca	pacity (RFC	<b>;</b> )				1					
A			÷	17	-	£			Entry	Flow	R	
Arm	x <sub>2</sub>	IVI	LD	K	+		AIVI(08)	PM (17)	AIVI(U8)	PM (17)	AIVI(08)	PIVI (17)
⊢rom A	5.112	0.030	1.485	0.986	1549	0.631	1448	1426	826	852	0.570	0.597
From B	5.029	0.030	1.485	0.984	1524	0.626	1227	1245	530	677	0.432	0.544
From C	7.591	0.030	1.485	0.935	2300.174	0.786	1855	1807	574	592	0.309	0.328
From D												
From E												
From F												
From G												
From H												

Junction:		Kam Tin R	oad / Fan k	Kam Road						Jo	b Number:	J7342
Scenario:		With Propo	posed Temporary Concrete Batching Plant J01 - P. 6 Designed By: MCY Checked By: WCH Date: 21 January 2							6		
Design Ye	ear:	2030	De	signed By	: MCY	C	hecked By:	WCH		21 Janu	ary 2025	
AM (09) F	Peak											
Arm	To A	To B	To C	To D	To E	To F	To G	To H	Total	q <sub>c</sub>		
From A	0	424	524						948	178		
From B	433	6	173						612	530		
From C	518	166	6						690	439		
From D												
From F												
From F												
From G												
From H												
Total	951	596	703						2250			
										1		
PM (19) P	eak								-			
Arm	To A	To B	To C	To D	To E	To F	To G	To H	Total	q <sub>c</sub>		
From A	6	448	480						934	201		
From B	452	3	244						699	490		
From C	571	194	4						769	461		
From D												
From E												
From F												
From G												
From H												
Total	1029	645	728						2402			
Logond					Coomotrio	Deremet						
Legena	Deed	in electroite	andan)		Geometric			n (ma)	L (ma)	D (m)	<i>(</i> 0)	<u> </u>
	Koau (		e order)			e (m)	v (III) 2 2	25.0	L (III)	D (III)	<u>Ø(')</u>	0.7
					FIOIL A	7.7	2.0	15.0	10.0	25	40	0.7
C	Kom Tin E				From C	7.7	2.9	100.0	100.0	25	50 60	0.0
					From D	1.1	5.5	100.0	100.0	25	00	0.0
					FIOID E							
Г С					From C							
Ч					From H							
	1				TIONTI							
Predictive	e Equation	Q <sub>E</sub> = K(F -	f <sub>c</sub> q <sub>c</sub> )			Limitatio	n					
Q <sub>E</sub>	Entry Cap	acity				е	Entry Widt	th		4.0 - 15.0	m	
q <sub>c</sub>	Circulating	g Flow acros	ss the Entry	/		v	Approach	Half Width		2.0 - 7.3 m	ı	
К	= 1-0.0034	47(∅-30)-0.	978[(1/r)-0.	05]		r	Entry Rad	ius		6.0 - 100.0	) m	
F	= 303x <sub>2</sub>					L	Effective L	ength of Fl	are	1.0 - 100.0	) m	
f <sub>c</sub>	= 0.210t <sub>D</sub> (	1+0.2x <sub>2</sub> )				D	Inscribed	Circle Diam	eter	15 - 100 m	ı	
t <sub>D</sub>	= 1+0.5/(1	+M)				Ø	Entry Angl	le		10° - 60°		
М	= exp[(D-6	60)/10]				S	Sharpness	s of Flare		0.0 - 3.0		
x <sub>2</sub>	= v+(e-v)/	(1+2S)										
S	= 1.6(e-v)/	/L										
Ratio-of-I	Flow to Ca	pacity (RFC	;)									
					_		0	λ <sup>Ε</sup>	Entr	y Flow	RI	=C
Arm	X <sub>2</sub>	M	τ <sub>D</sub>	K	F		AM(09)	PM(19)	AM(09)	PM(19)	AM(09)	PM(19)
From A	5.112	0.030	1.485	0.986	1549	0.631	141/	1402	948	934	0.669	0.666
⊢rom B	5.029	0.030	1.485	0.984	1524	0.626	11/3	1197	612	699	0.522	0.584
From C	/.591	0.030	1.485	0.935	2300.174	0.786	1828	1812	690	769	0.377	0.424
⊢rom D												
⊢rom E												
⊢rom F												
⊢rom G												
From H												















Junctions 9
PICADY 9 - Priority Intersection Module
Version: 9.0.2.5947 © Copyright TRL Limited, 2017
For sales and distribution information, program advice and maintenance, contact TRL: +44 (0)1344 770558 software@trl.co.uk www.trlsoftware.co.uk
The users of this computer program for the solution of an engineering problem are in no way relieved of their responsibility for the correctness of the solution

Filename: J7342\_TIA\_R2.j9

Path: J:\ENG\Job\J73XX\J7342 Kam Tin - S16 for Proposed Concrete Batching Plant\working\(2024 10) TIA\_R2\Junction9 Report generation date: 21/1/2025 17:59:09

```
»Kam Tin Rd / Kam Sheung Rd / Sheung Tsuen BT - 2024 EXISTING, AM (08)
»Kam Tin Rd / Kam Sheung Rd / Sheung Tsuen BT - 2024 EXISTING, AM (09)
»Kam Tin Rd / Kam Sheung Rd / Sheung Tsuen BT - 2024 EXISTING, PM (17)
»Kam Tin Rd / Kam Sheung Rd / Sheung Tsuen BT - 2024 EXISTING, PM (19)
»Kam Tin Rd / Kam Sheung Rd / Sheung Tsuen BT - 2030 WITHOUT, AM (08)
»Kam Tin Rd / Kam Sheung Rd / Sheung Tsuen BT - 2030 WITHOUT, AM (09)
»Kam Tin Rd / Kam Sheung Rd / Sheung Tsuen BT - 2030 WITHOUT, PM (17)
»Kam Tin Rd / Kam Sheung Rd / Sheung Tsuen BT - 2030 WITHOUT, PM (19)
»Kam Tin Rd / Kam Sheung Rd / Sheung Tsuen BT - 2030 WITHOUT, PM (19)
»Kam Tin Rd / Kam Sheung Rd / Sheung Tsuen BT - 2030 WITH, AM (08)
»Kam Tin Rd / Kam Sheung Rd / Sheung Tsuen BT - 2030 WITH, PM (17)
»Kam Tin Rd / Kam Sheung Rd / Sheung Tsuen BT - 2030 WITH, PM (17)
```

#### Summary of junction performance

	AM (08)	AM (09)	PM (17)	PM (19)
	RFC	RFC	RFC	RFC
	Kam Tin Rd / Kam	Sheung Rd / Sheun	ig Tsuen BT [Locked	I] - 2024 EXISTING
Stream B-ACD	0.81	0.76	0.71	0.73
Stream A-D	0.00	0.00	0.00	0.00
Stream D-ABC	0.08	0.08	0.09	0.09
Stream C-B	0.00	0.00	0.00	0.00
	Kam Tin Rd / Kam	] - 2030 WITHOUT		
Stream B-ACD	0.91	0.85	0.78	0.81
Stream A-D	0.00	0.00	0.00	0.00
Stream D-ABC	0.09	0.09	0.09	0.10
Stream C-B	0.00	0.00	0.00	0.00
	Kam Tin Rd / Ka	im Sheung Rd / She	ung Tsuen BT [Lock	ed] - 2030 WITH
Stream B-ACD	0.97	0.95	0.88	0.85
Stream A-D	0.00	0.00	0.00	0.00
Stream D-ABC	0.09	0.09	0.10	0.10
Stream C-B	0.00	0.00	0.00	0.00

There are warnings associated with one or more model runs - see the 'Data Errors and Warnings' tables for each Analysis or Demand Set.

Values shown are the highest values encountered over all time segments. Delay is the maximum value of average delay per arriving vehicle.



## File summary

#### File Description

Title	Proposed Concrete Batching Plant
Location	Kam Tin
Site number	
Date	31/12/2024
Version	
Status	TIA_R2
Identifier	
Client	
Jobnumber	J7342
Enumerator	СКМ
Description	

#### Units

Distance units	Speed units	Traffic units input	Traffic units results	Flow units	Average delay units	Total delay units	Rate of delay units
m	kph	PCU	PCU	perHour	s	-Min	perMin

#### **Analysis Options**

Calculate Queue Percentiles	Calculate residual capacity	RFC Threshold	Average Delay threshold (s)	Queue threshold (PCU)
		1.00	36.00	20.00

## Demand Set Summary

ID	Scenario name	Time Period name	Traffic profile type	Start time (HH:mm)	Finish time (HH:mm)	Time segment length (min)
DEX08	2024 EXISTING	AM (08)	ONE HOUR	08:00	09:30	15
DEX09	2024 EXISTING	AM (09)	ONE HOUR	09:00	10:30	15
DEX17	2024 EXISTING	PM (17)	ONE HOUR	17:00	18:30	15
DEX19	2024 EXISTING	PM (19)	ONE HOUR	19:00	20:30	15
DRF08	2030 WITHOUT	AM (08)	ONE HOUR	08:00	09:30	15
DRF09	2030 WITHOUT	AM (09)	ONE HOUR	09:00	10:30	15
DRF17	2030 WITHOUT	PM (17)	ONE HOUR	17:00	18:30	15
DRF21	2030 WITHOUT	PM (19)	ONE HOUR	19:00	20:30	15
DWT08	2030 WITH	AM (08)	ONE HOUR	08:00	09:30	15
DWT09	2030 WITH	AM (09)	ONE HOUR	09:00	10:30	15
DWT17	2030 WITH	PM (17)	ONE HOUR	17:00	18:30	15
DWT19	2030 WITH	PM (19)	ONE HOUR	19:00	20:30	15

## **Analysis Set Details**

ID	Name	Locked	Network flow scaling factor (%)
AJ03	Kam Tin Rd / Kam Sheung Rd / Sheung Tsuen BT	~	100.000



# Kam Tin Rd / Kam Sheung Rd / Sheung Tsuen BT - 2024 EXISTING, AM (08)

#### **Data Errors and Warnings**

Severity	Area	ltem	Description
Warning	Vehicle Mix		HV% is zero for all movements / time segments. Vehicle Mix matrix should be completed whether working in PCUs or Vehs.

## **Junction Network**

#### Junctions

Junction	Name	Junction Type	Major road direction	Junction Delay (s)	Junction LOS
1	Kam Tin Rd / Kam Sheung Rd / Sheung Tsuen BT	Crossroads	Two-way	9.59	А

#### **Junction Network Options**

Driving side	Lighting
Left	Normal/unknown

## Arms

#### Arms

Arm	Name	Description	Arm type
Α	Kam Tin Road		Major
в	Kam Sheung Road (Arm B) Kam Sheung Road (Arm B) Kam Sheung Road (Arm B) v		Minor
С	Kam Tin Road		Major
D	Sheung Tsuen B/T		Minor

#### **Major Arm Geometry**

Arm	Width of carriageway (m)	Has kerbed central reserve	Has right turn bay	Visibility for right turn (m)	Blocks?	Blocking queue (PCU)
Α	7.80			100.0		-
С	7.80			65.0		-

Geometries for Arm C are measured opposite Arm B. Geometries for Arm A (if relevant) are measured opposite Arm D.

#### **Minor Arm Geometry**

Arm	Minor arm type	Lane width (m)	Visibility to left (m)	Visibility to right (m)
в	One lane	5.00	35	30
D	One lane	5.00	15	20



#### Slope / Intercept / Capacity

#### **Priority Intersection Slopes and Intercepts**

Junction	Stream	Intercept (PCU/hr)	Slope for A-B	Slope for A-C	Slope for A-D	Slope for B-A	Slope for B-C	Slope for B-D	Slope for C-A	Slope for C-B	Slope for C-D	Slope for D-A	Slope for D-B	Slope for D-C
1	A-D	632	-	-	-	-	-	-	0.226	0.322	0.226	-	-	-
1	B-A	605	0.101	0.257	0.257	-	-	-	0.161	0.366	-	0.257	0.257	0.128
1	B-C	772	0.109	0.276	-	-	-	-	-	-	-	-	-	-
1	B-D, nearside lane	605	0.101	0.257	0.257	-	-	-	0.161	0.366	0.161	-	-	-
1	B-D, offside lane	605	0.101	0.257	0.257	-	-	-	0.161	0.366	0.161	-	-	-
1	C-B	612	0.218	0.218	0.312	-	-	-	-	-	-	-	-	-
1	D-A	764	-	-	-	-	-	-	0.273	-	0.108	-	-	-
1	D-B, nearside lane	591	0.158	0.158	0.358	-	-	-	0.251	0.251	0.099	-	-	-
1	D-B, offside lane	591	0.158	0.158	0.358	-	-	-	0.251	0.251	0.099	-	-	-
1	D-C	591	-	0.158	0.358	0.125	0.251	0.251	0.251	0.251	0.099	-	-	-

The slopes and intercepts shown above do NOT include any corrections or adjustments.

Streams may be combined, in which case capacity will be adjusted.

Values are shown for the first time segment only; they may differ for subsequent time segments.

## **Traffic Demand**

#### **Demand Set Details**

ID	Scenario name	Time Period name	Traffic profile type	Start time (HH:mm)	Finish time (HH:mm)	Time segment length (min)
DEX08	2024 EXISTING	AM (08)	ONE HOUR	08:00	09:30	15

 Vehicle mix source
 PCU Factor for a HV (PCU)

 HV Percentages
 2.00

#### **Demand overview (Traffic)**

Arm	Linked arm	Use O-D data	Average Demand (PCU/hr)	Scaling Factor (%)
Α		~	650	100.000
в		✓	311	100.000
С		✓	425	100.000
D		✓	48	100.000

# **Origin-Destination Data**

#### Demand (PCU/hr)

	То					
		Α	в	С	D	
	Α	0	252	398	0	
From	в	224	0	87	0	
	С	425	0	0	0	
	D	48	0	0	0	

## **Vehicle Mix**

#### **Heavy Vehicle Percentages**

	То				
		A	в	С	D
	Α	0	0	0	0
From	в	0	0	0	0
	С	0	0	0	0
	D	0	0	0	0



## **Detailed Demand Data**

## Demand for each time segment

Time Segment	Arm	Demand (PCU/hr)	Demand in PCU (PCU/hr)
	Α	489	489
08:00-08:15	в	234	234
	С	320	320
	D	36	36
	Α	584	584
09.15 09.20	в	280	280
00.15-00.50	С	382	382
	D	43	43
	Α	716	716
09.20 09.45	в	342	342
08:30-08:45	С	468	468
	D	53	53
	Α	716	716
08:45-09:00	в	342	342
08.45-09.00	С	468	468
	D	53	53
	Α	584	584
09:00 09:15	в	280	280
09:00-09:15	С	382	382
	D	43	43
	Α	489	489
09:15-09:30	в	234	234
	С	320	320
	D	36	36

# Results

## Results Summary for whole modelled period

Stream	Max RFC	Max delay (s)	Max Queue (PCU)	Max LOS
B-ACD	0.81	43.28	3.9	E
A-B				
A-C				
A-D	0.00	0.00	0.0	А
D-ABC	0.08	6.17	0.1	А
C-D				
C-A				
С-В	0.00	0.00	0.0	А



# Kam Tin Rd / Kam Sheung Rd / Sheung Tsuen BT - 2024 EXISTING, AM (09)

#### **Data Errors and Warnings**

Severity	Area	ltem	Description
Warning	Vehicle Mix		HV% is zero for all movements / time segments. Vehicle Mix matrix should be completed whether working in PCUs or Vehs.

## **Junction Network**

#### Junctions

Junction	Name	Junction Type	Major road direction	Junction Delay (s)	Junction LOS
1	Kam Tin Rd / Kam Sheung Rd / Sheung Tsuen BT	Crossroads	Two-way	7.74	A

#### **Junction Network Options**

Driving side	Lighting
Left	Normal/unknown

## **Traffic Demand**

#### **Demand Set Details**

ID	Scenario name	Time Period name	Traffic profile type	Start time (HH:mm)	Finish time (HH:mm)	Time segment length (min)
DEX09	2024 EXISTING	AM (09)	ONE HOUR	09:00	10:30	15

Vehicle mix source	PCU Factor for a HV (PCU)			
HV Percentages	2.00			

#### **Demand overview (Traffic)**

Arm	Linked arm	Use O-D data	Average Demand (PCU/hr)	Scaling Factor (%)
Α		~	623	100.000
в		✓	299	100.000
С		✓	407	100.000
D		~	47	100.000

## **Origin-Destination Data**

#### Demand (PCU/hr)

	То				
		Α	в	c	D
	Α	0	243	380	0
From	в	216	0	83	0
	С	407	0	0	0
	D	47	0	0	0

## **Vehicle Mix**



#### Heavy Vehicle Percentages

	То				
		Α	В	С	D
	Α	0	0	0	0
From	в	0	0	0	0
	С	0	0	0	0
	D	0	0	0	0

# **Detailed Demand Data**

#### Demand for each time segment

Time Segment	Arm	Demand (PCU/hr)	Demand in PCU (PCU/hr)
	Α	469	469
09:00-09:15	в	225	225
	С	306	306
	D	35	35
	Α	560	560
09-15 09-20	в	269	269
09:15-09:30	С	366	366
	D	42	42
	Α	686	686
00.20 00.45	в	329	329
09:30-09:45	с	448	448
	D	52	52
	Α	686	686
00.45 40.00	в	329	329
09:45-10:00	С	448	448
	D	52	52
	Α	560	560
40.00 40.45	в	269	269
10.00-10.15	С	366	366
	D	42	42
	Α	469	469
40.45 40.20	в	225	225
10:15-10:30	С	306	306
	D	35	35

# Results

#### **Results Summary for whole modelled period**

Stream	Max RFC	Max delay (s)	Max Queue (PCU)	Max LOS
B-ACD	0.76	34.65	3.0	D
A-B				
A-C				
A-D	0.00	0.00	0.0	A
D-ABC	0.08	6.10	0.1	A
C-D				
C-A				
С-В	0.00	0.00	0.0	A


# Kam Tin Rd / Kam Sheung Rd / Sheung Tsuen BT - 2024 EXISTING, PM (17)

#### **Data Errors and Warnings**

Severity	Area	ltem	Description
Warning	Vehicle Mix		HV% is zero for all movements / time segments. Vehicle Mix matrix should be completed whether working in PCUs or Vehs.

## **Junction Network**

#### Junctions

Junction	Name	Junction Type	Major road direction	Junction Delay (s)	Junction LOS
1	Kam Tin Rd / Kam Sheung Rd / Sheung Tsuen BT	Crossroads	Two-way	5.58	А

#### **Junction Network Options**

Driving side	Lighting	
Left	Normal/unknown	

### **Traffic Demand**

#### **Demand Set Details**

ID	Scenario name	Time Period name	Traffic profile type	Start time (HH:mm)	Finish time (HH:mm)	Time segment length (min)
DEX17	2024 EXISTING	PM (17)	ONE HOUR	17:00	18:30	15

Vehicle mix source	PCU Factor for a HV (PCU)		
HV Percentages	2.00		

#### **Demand overview (Traffic)**

Arm	Linked arm	Use O-D data	Average Demand (PCU/hr)	Scaling Factor (%)	
Α		~	630	100.000	
в		✓	277	100.000	
С		✓	495	100.000	
D		~	49	100.000	

## **Origin-Destination Data**

#### Demand (PCU/hr)

		То					
From		Α	в	c	D		
	Α	0	286	344	0		
	в	186	0	91	0		
	С	495	0	0	0		
	D	49	0	0	0		



	То				
From		Α	В	С	D
	Α	0	0	0	0
	в	0	0	0	0
	С	0	0	0	0
	D	0	0	0	0

## **Detailed Demand Data**

#### Demand for each time segment

Time Segment	Arm	Demand (PCU/hr)	Demand in PCU (PCU/hr)
	Α	474	474
47.00 47.45	в	209	209
17:00-17:15	С	373	373
	D	37	37
	Α	566	566
17:15-17:30	в	249	249
	С	445	445
	D	44	44
	Α	694	694
17:30-17:45	в	305	305
	С	545	545
	D	54	54
	Α	694	694
	в	305	305
17:45-18:00	С	545	545
	D	54	54
	Α	566	566
49.00 49.45	в	249	249
18:00-18:15	С	445	445
	D	44	44
	Α	474	474
40.45 40.20	в	209	209
18:15-18:30	С	373	373
	D	37	37

# Results

Stream	Max RFC	Max delay (s)	Max Queue (PCU)	Max LOS
B-ACD	0.71	28.11	2.3	D
A-B				
A-C				
A-D	0.00	0.00	0.0	A
D-ABC	0.09	6.41	0.1	A
C-D				
C-A				
С-В	0.00	0.00	0.0	A



# Kam Tin Rd / Kam Sheung Rd / Sheung Tsuen BT - 2024 EXISTING, PM (19)

#### **Data Errors and Warnings**

Severity	Area	ltem	Description
Warning	Vehicle Mix		HV% is zero for all movements / time segments. Vehicle Mix matrix should be completed whether working in PCUs or Vehs.

## **Junction Network**

#### Junctions

Junction	Name	Junction Type	Major road direction	Junction Delay (s)	Junction LOS
1	Kam Tin Rd / Kam Sheung Rd / Sheung Tsuen BT	Crossroads	Two-way	6.11	А

#### **Junction Network Options**

Driving side	Lighting	
Left	Normal/unknown	

### **Traffic Demand**

#### **Demand Set Details**

ID	Scenario name	Time Period name	Traffic profile type	Start time (HH:mm)	Finish time (HH:mm)	Time segment length (min)
DEX19	2024 EXISTING	PM (19)	ONE HOUR	19:00	20:30	15

Vehicle mix source	PCU Factor for a HV (PCU)		
HV Percentages	2.00		

#### **Demand overview (Traffic)**

Arm	Linked arm	Use O-D data	Average Demand (PCU/hr)	Scaling Factor (%)
Α		~	660	100.000
в		✓	282	100.000
С		✓	494	100.000
D		~	50	100.000

## **Origin-Destination Data**

#### Demand (PCU/hr)

	То				
		A	в	c	D
	Α	0	300	360	0
From	в	190	0	92	0
	С	494	0	0	0
	D	50	0	0	0



	То				
		Α	в	С	D
	Α	0	0	0	0
From	в	0	0	0	0
	С	0	0	0	0
	D	0	0	0	0

## **Detailed Demand Data**

#### Demand for each time segment

Time Segment	Arm	Demand (PCU/hr)	Demand in PCU (PCU/hr)
	Α	497	497
40.00 40.45	в	212	212
19:00-19:15	С	372	372
	D	38	38
	Α	593	593
10.15 10.20	в	254	254
19:15-19:30	С	444	444
	D	45	45
	Α	727	727
19:30-19:45	в	310	310
	c	544	544
	D	55	55
19:45-20:00	Α	727	727
	в	310	310
	c	544	544
	D	55	55
	Α	593	593
20.00 20.15	в	254	254
20:00-20:15	С	444	444
	D	45	45
	Α	497	497
20.45 20.20	в	212	212
20:15-20:30	С	372	372
	D	38	38

# Results

Stream	Max RFC	Max delay (s)	Max Queue (PCU)	Max LOS
B-ACD	0.73	31.06	2.6	D
A-B				
A-C				
A-D	0.00	0.00	0.0	A
D-ABC	0.09	6.42	0.1	A
C-D				
C-A				
С-В	0.00	0.00	0.0	A



# Kam Tin Rd / Kam Sheung Rd / Sheung Tsuen BT - 2030 WITHOUT, AM (08)

#### **Data Errors and Warnings**

Severity	Area	ltem	Description
Warning	Vehicle Mix		HV% is zero for all movements / time segments. Vehicle Mix matrix should be completed whether working in PCUs or Vehs.

## **Junction Network**

#### Junctions

Junction	Name	Junction Type	Major road direction	Junction Delay (s)	Junction LOS
1	Kam Tin Rd / Kam Sheung Rd / Sheung Tsuen BT	Crossroads	Two-way	16.02	С

#### **Junction Network Options**

Driving side	Lighting	
Left	Normal/unknown	

### **Traffic Demand**

#### **Demand Set Details**

ID	Scenario name	Time Period name	Traffic profile type	Start time (HH:mm)	Finish time (HH:mm)	Time segment length (min)
DRF08	2030 WITHOUT	AM (08)	ONE HOUR	08:00	09:30	15

Vehicle mix source	PCU Factor for a HV (PCU)	
HV Percentages	2.00	

#### **Demand overview (Traffic)**

Arm	Linked arm	Use O-D data	Average Demand (PCU/hr)	Scaling Factor (%)
Α		~	705	100.000
в		✓	331	100.000
С		✓	461	100.000
D		~	51	100.000

## **Origin-Destination Data**

#### Demand (PCU/hr)

	То				
		Α	в	c	D
	Α	0	268	437	0
From	в	238	0	93	0
	С	461	0	0	0
	D	51	0	0	0



	То				
		Α	В	С	D
	Α	0	0	0	0
From	в	0	0	0	0
	С	0	0	0	0
	D	0	0	0	0

## **Detailed Demand Data**

#### Demand for each time segment

Time Segment	Arm	Demand (PCU/hr)	Demand in PCU (PCU/hr)
	Α	531	531
09-00 09-45	в	249	249
08:00-08:15	С	347	347
	D	38	38
	Α	634	634
08-15-08-30	в	298	298
08:15-08:30	С	414	414
	D	46	46
	Α	776	776
08:30-08:45	в	364	364
	С	508	508
	D	56	56
	Α	776	776
08-45-09-00	в	364	364
08.45-09.00	С	508	508
	D	56	56
	Α	634	634
09.00 09.15	в	298	298
09.00-09.15	С	414	414
	D	46	46
	Α	531	531
09-15-09-20	в	249	249
09:15-09:30	С	347	347
	D	38	38

# Results

Stream	Max RFC	Max delay (s)	Max Queue (PCU)	Max LOS
B-ACD	0.91	73.97	6.9	F
A-B				
A-C				
A-D	0.00	0.00	0.0	A
D-ABC	0.09	6.32	0.1	A
C-D				
C-A				
С-В	0.00	0.00	0.0	A



# Kam Tin Rd / Kam Sheung Rd / Sheung Tsuen BT - 2030 WITHOUT, AM (09)

#### **Data Errors and Warnings**

Severity	Area	ltem	Description
Warning	Vehicle Mix		HV% is zero for all movements / time segments. Vehicle Mix matrix should be completed whether working in PCUs or Vehs.

## **Junction Network**

#### Junctions

Junction	Name	Junction Type	Major road direction	Junction Delay (s)	Junction LOS
1	Kam Tin Rd / Kam Sheung Rd / Sheung Tsuen BT	Crossroads	Two-way	11.59	В

#### **Junction Network Options**

Driving side	Lighting
Left	Normal/unknown

### **Traffic Demand**

#### **Demand Set Details**

ID	Scenario name	Time Period name	Traffic profile type	Start time (HH:mm)	Finish time (HH:mm)	Time segment length (min)
DRF09	2030 WITHOUT	AM (09)	ONE HOUR	09:00	10:30	15

Vehicle mix source	PCU Factor for a HV (PCU)	
HV Percentages	2.00	

#### **Demand overview (Traffic)**

Arm	Linked arm	Use O-D data	Average Demand (PCU/hr)	Scaling Factor (%)
Α		~	676	100.000
в		✓	318	100.000
С		✓	442	100.000
D		~	50	100.000

## **Origin-Destination Data**

#### Demand (PCU/hr)

	То				
		Α	в	c	D
	Α	0	258	418	0
From	в	229	0	89	0
	С	442	0	0	0
	D	50	0	0	0



	То				
		Α	В	С	D
	Α	0	0	0	0
From	в	0	0	0	0
	С	0	0	0	0
	D	0	0	0	0

## **Detailed Demand Data**

#### Demand for each time segment

Time Segment	Arm	Demand (PCU/hr)	Demand in PCU (PCU/hr)
	Α	509	509
00.00 00.45	в	239	239
09:00-09:15	С	333	333
	D	38	38
	Α	608	608
00.15 00.20	в	286	286
09:15-09:30	С	397	397
	D	45	45
	Α	744	744
09:30-09:45	в	350	350
	c	487	487
	D	55	55
	Α	744	744
09:45 10:00	в	350	350
09:45-10:00	c	487	487
	D	55	55
	Α	608	608
10.00 10.15	в	286	286
10:00-10:15	С	397	397
	D	45	45
	Α	509	509
10.15 10.20	в	239	239
10:15-10:30	С	333	333
	D	38	38

# Results

Stream	Max RFC	Max delay (s)	Max Queue (PCU)	Max LOS
B-ACD	0.85	53.17	4.8	F
A-B				
A-C				
A-D	0.00	0.00	0.0	A
D-ABC	0.09	6.25	0.1	A
C-D				
C-A				
С-В	0.00	0.00	0.0	A



# Kam Tin Rd / Kam Sheung Rd / Sheung Tsuen BT - 2030 WITHOUT, PM (17)

#### **Data Errors and Warnings**

Severity	Area	ltem	Description
Warning	Vehicle Mix		HV% is zero for all movements / time segments. Vehicle Mix matrix should be completed whether working in PCUs or Vehs.

## **Junction Network**

#### Junctions

Junction	Name	Junction Type	Major road direction	Junction Delay (s)	Junction LOS
1	Kam Tin Rd / Kam Sheung Rd / Sheung Tsuen BT	Crossroads	Two-way	7.56	A

#### **Junction Network Options**

Driving side	Lighting
Left	Normal/unknown

## **Traffic Demand**

#### **Demand Set Details**

ID	Scenario name	Time Period name	Traffic profile type	Start time (HH:mm)	Finish time (HH:mm)	Time segment length (min)
DRF17	2030 WITHOUT	PM (17)	ONE HOUR	17:00	18:30	15

Vehicle mix source	PCU Factor for a HV (PCU)		
HV Percentages	2.00		

#### **Demand overview (Traffic)**

Arm	Linked arm	Use O-D data	Average Demand (PCU/hr)	Scaling Factor (%)
Α		~	676	100.000
в		✓	295	100.000
С		✓	535	100.000
D		~	52	100.000

## **Origin-Destination Data**

#### Demand (PCU/hr)

	То				
		Α	в	c	D
	Α	0	304	372	0
From	в	197	0	98	0
	С	535	0	0	0
	D	52	0	0	0



	То				
		Α	в	С	D
	Α	0	0	0	0
From	в	0	0	0	0
	С	0	0	0	0
	D	0	0	0	0

## **Detailed Demand Data**

#### Demand for each time segment

Time Segment	Arm	Demand (PCU/hr)	Demand in PCU (PCU/hr)
	Α	509	509
47.00 47.45	в	222	222
17:00-17:15	С	403	403
	D	39	39
	Α	608	608
17.15 17.20	в	265	265
17:15-17:30	С	481	481
	D	47	47
	Α	744	744
17:30-17:45	в	325	325
	c	589	589
	D	57	57
	Α	744	744
17:45-18:00	в	325	325
	c	589	589
	D	57	57
	Α	608	608
10.00 10.15	в	265	265
10.00-10:15	С	481	481
	D	47	47
	Α	509	509
10.15 10.20	в	222	222
10.15-10:30	С	403	403
	D	39	39

# Results

Stream	Max RFC	Max delay (s)	Max Queue (PCU)	Max LOS
B-ACD	0.78	38.75	3.3	E
A-B				
A-C				
A-D	0.00	0.00	0.0	A
D-ABC	0.09	6.59	0.1	A
C-D				
C-A				
С-В	0.00	0.00	0.0	A



# Kam Tin Rd / Kam Sheung Rd / Sheung Tsuen BT - 2030 WITHOUT, PM (19)

#### **Data Errors and Warnings**

Severity	Area	ltem	Description
Warning	Vehicle Mix		HV% is zero for all movements / time segments. Vehicle Mix matrix should be completed whether working in PCUs or Vehs.

## **Junction Network**

#### Junctions

Junction	Name	Junction Type	Major road direction	Junction Delay (s)	Junction LOS
1	Kam Tin Rd / Kam Sheung Rd / Sheung Tsuen BT	Crossroads	Two-way	8.72	А

#### **Junction Network Options**

Driving side	Lighting	
Left	Normal/unknown	

## **Traffic Demand**

#### **Demand Set Details**

ID	Scenario name	Time Period name	Traffic profile type	Start time (HH:mm)	Finish time (HH:mm)	Time segment length (min)
DRF21	2030 WITHOUT	PM (19)	ONE HOUR	19:00	20:30	15

Vehicle mix source	PCU Factor for a HV (PCU)	
HV Percentages	2.00	

#### **Demand overview (Traffic)**

Arm	Linked arm	Use O-D data	Average Demand (PCU/hr)	Scaling Factor (%)
Α		~	707	100.000
в		✓	301	100.000
С		✓	534	100.000
D		~	53	100.000

## **Origin-Destination Data**

#### Demand (PCU/hr)

	То				
		Α	в	c	D
	Α	0	318	389	0
From	в	202	0	99	0
	С	534	0	0	0
	D	53	0	0	0



	То				
		Α	в	С	D
	Α	0	0	0	0
From	в	0	0	0	0
	С	0	0	0	0
	D	0	0	0	0

## **Detailed Demand Data**

#### Demand for each time segment

Time Segment	Arm	Demand (PCU/hr)	Demand in PCU (PCU/hr)
	Α	532	532
19:00-19:15	в	227	227
	С	402	402
	D	40	40
	Α	636	636
10.15 10.20	в	271	271
19:15-19:50	С	480	480
	D	48	48
	Α	778	778
19:30-19:45	в	331	331
	c	588	588
	D	58	58
19:45-20:00	Α	778	778
	в	331	331
	С	588	588
	D	58	58
	Α	636	636
20.00 20.45	в	271	271
20:00-20:15	с	480	480
	D	48	48
	Α	532	532
20.45 20.20	в	227	227
20:15-20:30	С	402	402
	D	40	40

# Results

Stream	Max RFC	Max delay (s)	Max Queue (PCU)	Max LOS
B-ACD	0.81	45.05	3.9	E
A-B				
A-C				
A-D	0.00	0.00	0.0	A
D-ABC	0.10	6.60	0.1	A
C-D				
C-A				
С-В	0.00	0.00	0.0	A



# Kam Tin Rd / Kam Sheung Rd / Sheung Tsuen BT - 2030 WITH, AM (08)

#### **Data Errors and Warnings**

Severity	Area	ltem	Description
Warning	Vehicle Mix		HV% is zero for all movements / time segments. Vehicle Mix matrix should be completed whether working in PCUs or Vehs.

## **Junction Network**

#### Junctions

Junction	Name	Junction Type	Major road direction	Junction Delay (s)	Junction LOS
1	Kam Tin Rd / Kam Sheung Rd / Sheung Tsuen BT	Crossroads	Two-way	22.25	С

#### **Junction Network Options**

Driving side	Lighting
Left	Normal/unknown

### **Traffic Demand**

#### **Demand Set Details**

ID	Scenario name	Time Period name	Traffic profile type	Start time (HH:mm)	Finish time (HH:mm)	Time segment length (min)
DWT08	2030 WITH	AM (08)	ONE HOUR	08:00	09:30	15

 Vehicle mix source
 PCU Factor for a HV (PCU)

 HV Percentages
 2.00

#### **Demand overview (Traffic)**

Arm	Linked arm	Use O-D data	Average Demand (PCU/hr)	Scaling Factor (%)
Α		~	760	100.000
в		✓	331	100.000
С		✓	516	100.000
D		~	51	100.000

## **Origin-Destination Data**

#### Demand (PCU/hr)

		То				
		Α	в	c	D	
	Α	0	268	492	0	
From	в	238	0	93	0	
	С	516	0	0	0	
	D	51	0	0	0	



			То		
		Α	В	С	D
	Α	0	0	0	0
From	в	0	0	0	0
	С	0	0	0	0
	D	0	0	0	0

## **Detailed Demand Data**

#### Demand for each time segment

Time Segment	Arm	Demand (PCU/hr)	Demand in PCU (PCU/hr)
	Α	572	572
09-00 09-45	в	249	249
08:00-08:15	С	388	388
	D	38	38
	Α	683	683
09.15 09.20	в	298	298
08.15-08.50	С	464	464
	D	46	46
	Α	837	837
09.20 09.45	в	364	364
08:30-08:45	c	568	568
	D	56	56
	Α	837	837
09.45 00.00	в	364	364
08.45-09.00	c	568	568
	D	56	56
	Α	683	683
09.00 09.15	в	298	298
09:00-09:15	С	464	464
	D	46	46
	Α	572	572
00.45 00.20	в	249	249
09:15-09:30	С	388	388
	D	38	38

# Results

Stream	Max RFC	Max delay (s)	Max Queue (PCU)	Max LOS
B-ACD	0.97	110.43	10.7	F
A-B				
A-C				
A-D	0.00	0.00	0.0	A
D-ABC	0.09	6.51	0.1	A
C-D				
C-A				
С-В	0.00	0.00	0.0	A



# Kam Tin Rd / Kam Sheung Rd / Sheung Tsuen BT - 2030 WITH, AM (09)

#### **Data Errors and Warnings**

Severity	Area	ltem	Description
Warning	Vehicle Mix		HV% is zero for all movements / time segments. Vehicle Mix matrix should be completed whether working in PCUs or Vehs.

## **Junction Network**

#### Junctions

Junction	Name	Junction Type	Major road direction	Junction Delay (s)	Junction LOS
1	Kam Tin Rd / Kam Sheung Rd / Sheung Tsuen BT	Crossroads	Two-way	19.51	С

#### **Junction Network Options**

Driving side	Lighting
Left	Normal/unknown

### **Traffic Demand**

#### **Demand Set Details**

ID	Scenario name	Time Period name	Traffic profile type	Start time (HH:mm)	Finish time (HH:mm)	Time segment length (min)
DWT09	2030 WITH	AM (09)	ONE HOUR	09:00	10:30	15

 Vehicle mix source
 PCU Factor for a HV (PCU)

 HV Percentages
 2.00

#### **Demand overview (Traffic)**

Arm	Linked arm	Use O-D data	Average Demand (PCU/hr)	Scaling Factor (%)
Α		~	771	100.000
в		✓	318	100.000
С		✓	537	100.000
D		~	50	100.000

## **Origin-Destination Data**

#### Demand (PCU/hr)

	То				
		Α	в	c	D
From	Α	0	258	513	0
	в	229	0	89	0
	С	537	0	0	0
	D	50	0	0	0



	То				
		Α	В	С	D
	Α	0	0	0	0
From	в	0	0	0	0
	С	0	0	0	0
	D	0	0	0	0

## **Detailed Demand Data**

#### Demand for each time segment

Time Segment	Arm	Demand (PCU/hr)	Demand in PCU (PCU/hr)
	Α	580	580
00.00 00.45	в	239	239
09:00-09:15	С	404	404
	D	38	38
	Α	693	693
00.15 00.20	в	286	286
09.15-09.50	С	483	483
	D	45	45
	Α	849	849
00.20 00.45	в	350	350
09:30-09:45	c	591	591
	D	55	55
	Α	849	849
09:45 10:00	в	350	350
09.45-10.00	c	591	591
	D	55	55
	Α	693	693
10.00 10.15	в	286	286
10:00-10:15	С	483	483
	D	45	45
	Α	580	580
10.15 10.20	в	239	239
10:15-10:30	С	404	404
	D	38	38

# Results

Stream	Max RFC	Max delay (s)	Max Queue (PCU)	Max LOS
B-ACD	0.95	101.80	9.4	F
A-B				
A-C				
A-D	0.00	0.00	0.0	A
D-ABC	0.09	6.57	0.1	A
C-D				
C-A				
С-В	0.00	0.00	0.0	A



# Kam Tin Rd / Kam Sheung Rd / Sheung Tsuen BT -2030 WITH, PM (17)

#### **Data Errors and Warnings**

Severity	Area	ltem	Description
Warning	Vehicle Mix		HV% is zero for all movements / time segments. Vehicle Mix matrix should be completed whether working in PCUs or Vehs.

## **Junction Network**

#### Junctions

Junction	Name	Junction Type	Major road direction	Junction Delay (s)	Junction LOS
1	Kam Tin Rd / Kam Sheung Rd / Sheung Tsuen BT	Crossroads	Two-way	11.48	В

#### **Junction Network Options**

Driving side	Lighting
Left	Normal/unknown

### **Traffic Demand**

#### **Demand Set Details**

ID	Scenario name	Time Period name	Traffic profile type	Start time (HH:mm)	Finish time (HH:mm)	Time segment length (min)
DWT17	2030 WITH	PM (17)	ONE HOUR	17:00	18:30	15

 Vehicle mix source
 PCU Factor for a HV (PCU)

 HV Percentages
 2.00

#### **Demand overview (Traffic)**

Arm	Linked arm	Use O-D data	Average Demand (PCU/hr)	Scaling Factor (%)
Α		~	771	100.000
в		✓	295	100.000
С		✓	630	100.000
D		~	52	100.000

## **Origin-Destination Data**

#### Demand (PCU/hr)

	То					
		Α	в	c	D	
From	Α	0	304	467	0	
	в	197	0	98	0	
	С	630	0	0	0	
	D	52	0	0	0	



	То				
		Α	в	С	D
	Α	0	0	0	0
From	в	0	0	0	0
	С	0	0	0	0
	D	0	0	0	0

## **Detailed Demand Data**

#### Demand for each time segment

Time Segment	Arm	Demand (PCU/hr)	Demand in PCU (PCU/hr)
	Α	580	580
47.00 47.45	в	222	222
17:00-17:15	С	474	474
	D	39	39
	Α	693	693
17.15 17.20	в	265	265
17:15-17:30	С	566	566
	D	47	47
	Α	849	849
17.20 17.45	в	325	325
17:30-17:45	c	694	694
	D	57	57
	Α	849	849
47.45 49.00	в	325	325
17:45-16:00	С	694	694
	D	57	57
	Α	693	693
49.00 49.45	в	265	265
18:00-18:15	с	566	566
	D	47	47
	Α	580	580
18:15-18:30	в	222	222
10.15-18:30	С	474	474
	D	39	39

# Results

Stream	Max RFC	Max delay (s)	Max Queue (PCU)	Max LOS
B-ACD	0.88	66.79	5.6	F
A-B				
A-C				
A-D	0.00	0.00	0.0	A
D-ABC	0.10	6.96	0.1	A
C-D				
C-A				
С-В	0.00	0.00	0.0	A



# Kam Tin Rd / Kam Sheung Rd / Sheung Tsuen BT - 2030 WITH, PM (19)

#### **Data Errors and Warnings**

Severity	Area	ltem	Description
Warning	Vehicle Mix		HV% is zero for all movements / time segments. Vehicle Mix matrix should be completed whether working in PCUs or Vehs.

## **Junction Network**

#### Junctions

Junction	Name	Junction Type	Major road direction	Junction Delay (s)	Junction LOS
1	Kam Tin Rd / Kam Sheung Rd / Sheung Tsuen BT	Crossroads	Two-way	10.39	В

#### **Junction Network Options**

Driving side	Lighting
Left	Normal/unknown

### **Traffic Demand**

#### **Demand Set Details**

ID	Scenario name	Time Period name	Traffic profile type	Start time (HH:mm)	Finish time (HH:mm)	Time segment length (min)
DWT19	2030 WITH	PM (19)	ONE HOUR	19:00	20:30	15

 Vehicle mix source
 PCU Factor for a HV (PCU)

 HV Percentages
 2.00

#### **Demand overview (Traffic)**

Arm	Linked arm	Use O-D data	Average Demand (PCU/hr)	Scaling Factor (%)
Α		~	747	100.000
в		✓	301	100.000
С		✓	574	100.000
D		~	53	100.000

## **Origin-Destination Data**

#### Demand (PCU/hr)

	То				
		Α	в	c	D
	Α	0	318	429	0
From	в	202	0	99	0
	С	574	0	0	0
	D	53	0	0	0



	То				
		Α	в	С	D
	Α	0	0	0	0
From	в	0	0	0	0
	С	0	0	0	0
	D	0	0	0	0

## **Detailed Demand Data**

#### Demand for each time segment

Time Segment	Arm	Demand (PCU/hr)	Demand in PCU (PCU/hr)
	Α	562	562
40.00 40.45	в	227	227
19:00-19:15	С	432	432
	D	40	40
	Α	672	672
10.15 10.20	в	271	271
19:15-19:50	С	516	516
	D	48	48
	Α	822	822
10.20 10.45	в	331	331
19:30-19:45	c	632	632
	D	58	58
	Α	822	822
40.45 20.00	в	331	331
19:45-20:00	С	632	632
	D	58	58
	Α	672	672
20.00 20.45	в	271	271
20:00-20:15	С	516	516
	D	48	48
	Α	562	562
20:15-20:30	в	227	227
20:15-20:30	С	432	432
	D	40	40

# Results

Stream	Max RFC	Max delay (s)	Max Queue (PCU)	Max LOS
B-ACD	0.85	56.64	4.8	F
A-B				
A-C				
A-D	0.00	0.00	0.0	A
D-ABC	0.10	6.75	0.1	A
C-D				
C-A				
С-В	0.00	0.00	0.0	A



Junction:		Kam Tin R	load / Lam	Kam Road	d / Route Tw	visk				Jo	b Number:	J7342
Scenario:	: Existing Condition									J04 - P.	1	
Design Ye	ear:	2024	. De	signed By	: MCY		Checked By:	WCH	-	Date:	21 Janu	ary 2025
		To B	To C	To D	To F	To F	ToG	To H	Total	a,	1	
From A	121	318	258	10 0	10 2		100		697	26		
From B	121	8	1/1						560	370		
From C	411	10	0						126	5/9		
From D	110	10	0						130	540		
From G												
From H	650	244	200						1202			
Total	000	344	399						1393		l	
PM (17) F	Peak											
Arm	To A	To B	To C	To D	To E	To F	To G	To H	Total	q <sub>c</sub>		
From A	136	427	167						730	30		
From B	318	6	77						401	303		
From C	176	24	0						200	460		
From D												
From E												
From F												
From G												
From H												
Total	630	457	244						1331			
Legend	1			1	Geometri	c Paramet	ers					
Arm	Road (	in clockwis	e order)		Arm	e (m)	v (m)	r (m)	L (m)	D (m)	Ø (°)	S
A	Kam Tin F	Road (EB)			From A	6.0	3.5	100.0	5.5	27	60	0.7
В	Lam Kam	Road WB)			From B	6.0	3.5	25.0	8.0	27	60	0.5
С	Route Twi	sk (NB)			From C	7.0	3.5	65.0	12.0	27	35	0.5
D					From D							
E					From E							
F					From F							
G					From G							
н					From H							
Predictiv	e Equation	Q <sub>E</sub> = K(F -	f <sub>c</sub> q <sub>c</sub> )			Limitatio	n					
Q <sub>E</sub>	Entry Cap	acity			1	е	Entry Wid	th		4.0 - 15.0	m	
q <sub>c</sub>	Circulating	g Flow acros	ss the Entry	/		v	Approach	Half Width		2.0 - 7.3 m	ı	
к	= 1-0.003	47(Ø-30)-0.	978[(1/r)-0.	05]		r	Entry Rad	lius		6.0 - 100.0	) m	
F	= 303x <sub>2</sub>					L	Effective L	Length of F	lare	1.0 - 100.0	) m	
f <sub>c</sub>	= 0.210t <sub>D</sub> (	1+0.2x <sub>2</sub> )				D	Inscribed	Circle Diam	leter	15 - 100 m	ı	
t <sub>D</sub>	= 1+0.5/(1	+M)				Ø	Entry Ang	le		10° - 60°		
М	= exp[(D-6	60)/10]				S	Sharpness	s of Flare		0.0 - 3.0		
x <sub>2</sub>	= v+(e-v)/	(1+2S)										-
S	= 1.6(e-v)	/L										
Ratio-of-	Flow to Ca	nacity (REC										
Itatio-or-			,				Q <sub>E</sub>		Entry Flow	/	RFC	
Arm	x <sub>2</sub>	М	t <sub>D</sub>	к	F	f <sub>c</sub>	AM(08)	PM (17)	AM(08)	PM (17)	AM(08)	PM (17)
From A	4.519	0.037	1.482	0.935	1369	0.593	1266	1264	697	730	0.551	0.578
From B	4.750	0.037	1.482	0.906	1439	0.607	1095	1137	560	401	0.511	0.353
From C	5.310	0.035	1.483	1.017	1609	0.642	1283	1335	136	200	0.106	0.150
From D												
From E												
From F												
From G												
From H												

Junction:		Kam Tin R	oad / Lam	Job Number: J7342								
Scenario:		Existing Co	ondition							J04 - P.	2	
Design Ye	ear:	2024	De	signed By	: MCY	. C	hecked By:	WCH		Date:	21 Janu	ary 2025
AM (09) F	Peak										1	
Arm	IO A	IO B	10 C	To D	IOE	IOF	To G	IOH	l otal	q <sub>c</sub>		
From A	92	318	260						670	25		
From B	394	8	136						538	352		
From C	137	17	0						154	494		
From D												
From E												
From F												
From G												
From H												
Total	623	343	396						1362			
DM (40) F	)											
Arm		To B	To C	To D	To F	To F	ToG	To H	Total	G.	1	
Erom A	115	440	177	10 D	10 L	101	10.0	1011	724	HC FF		
From D	115	442	70						134	202		
From C	335	0	18						419	292		
From D	210	49	0						259	450		
From D												
From E												
From F												
From G												
From H	660	407	255						1412			
TOLAI	000	497	200						1412			
Legend					Geometric	: Paramet	ers					
Arm	Road (	in clockwise	e order)		Arm	e (m)	v (m)	r (m)	L (m)	D (m)	Ø(°)	S
A	Kam Tin F	Road (EB)			From A	6.0	3.5	100.0	5.5	27	60	0.7
В	Lam Kam	Road WB)			From B	6.0	3.5	25.0	8.0	27	60	0.5
c	Route Twi	sk (NB)			From C	7.0	3.5	65.0	12.0	27	35	0.5
D		()			From D							
E					From E							
F					From F							
G					From G							
Н					From H							
	1					1						
Predictiv	e Equation	Q <sub>E</sub> = K(F -	f <sub>c</sub> q <sub>c</sub> )			Limitatio	n					
Q <sub>E</sub>	Entry Cap	acity				е	Entry Widt	h		4.0 - 15.0	m	
q <sub>c</sub>	Circulating	g Flow acros	ss the Entry	/		v	Approach	Half Width		2.0 - 7.3 m	ו	
К	= 1-0.0034	47(Ø <b>-</b> 30)-0.	978[(1/r)-0.	05]		r	Entry Radi	ius		6.0 - 100.0	) m	
F	= 303x <sub>2</sub>					L	Effective L	ength of Fl	are	1.0 - 100.0	) m	
f <sub>c</sub>	= 0.210t <sub>D</sub> (	1+0.2x <sub>2</sub> )				D	Inscribed (	Circle Diam	eter	15 - 100 m	ı	
t <sub>D</sub>	= 1+0.5/(1	+M)				Ø	Entry Angl	е		10° - 60°		
М	= exp[(D-6	50)/10]				S	Sharpness	s of Flare		0.0 - 3.0		
x <sub>2</sub>	= v+(e-v)/	(1+2S)										
S	= 1.6(e-v)/	/L										
Ratio-of-	Flow to Ca	pacity (RFC	;)									
								Q <sub>E</sub>	Entr	y Flow	RI	FC
Arm	x <sub>2</sub>	Μ	t <sub>D</sub>	К	F	f <sub>c</sub>	AM(09)	PM(19)	AM(09)	PM(19)	AM(09)	PM(19)
From A	4.519	0.037	1.482	0.935	1369	0.593	1266	1250	670	734	0.529	0.587
From B	4.750	0.037	1.482	0.906	1439	0.607	1110	1143	538	419	0.485	0.367
From C	5.310	0.035	1.483	1.017	1609	0.642	1313	1338	154	259	0.117	0.194
From D												
From E												
From F												
From G												
From H												

Junction:	n: Kam Tin Road / Lam Kam Road / Route Twisk Job Number: <u>J</u> 7											J7342
Scenario:		Without Pr	oposed Te	mporary C	Concrete Bate	ching Plar	nt		J04 - P.	3		
Design Ye	ear:	2030	De	signed By	: MCY	C	Checked By:	_	Date:	21 Janu	ary 2025	
									-			
AM(08) P	EAK									-	-	
Arm	To A	To B	To C	To D	To E	To F	To G	To H	Total	q <sub>c</sub>	ĺ	
From A	128	349	274						751	27	1	
From B	451	8	150						609	402	1	
From C	125	19	0						144	587	1	
From D											1	
From E											1	
From F											1	
From G											1	
From H											1	
Total	704	376	424						1504		1	
PM (17) F	Peak											
Arm	To A	To B	To C	To D	To E	To F	To G	To H	Total	q <sub>c</sub>		
From A	144	464	177						785	31		
From B	345	6	82						433	321		
From C	187	25	0						212	495		
From D												
From E												
From F												
From G												
From H												
Total	676	495	259						1430			
Legend					Geometric	Paramet	ters					
Arm	Road	(in clockwise	e order)		Arm	e (m)	v (m)	r (m)	L (m)	D (m)	Ø (°)	S
Α	Kam Tin I	Road (EB)			From A	6.0	3.5	100.0	5.5	27	60	0.7
В	Lam Kam	Road WB)			From B	6.0	3.5	25.0	8.0	27	60	0.5
С	Route Tw	isk (NB)			From C	7.0	3.5	65.0	12.0	27	35	0.5
D	(	)			From D							
Е					From E							
F					From F							
G					From G							
Н					From H							
Predictiv	e Equation	n Q <sub>E</sub> = K(F -	f <sub>c</sub> q <sub>c</sub> )			Limitatio	n					
Q <sub>E</sub>	Entry Cap	bacity				е	Entry Wid	th		4.0 - 15.0	m	
q₀	Circulatin	g Flow acros	ss the Entry	/		v	Approach	Half Width		2.0 - 7.3 m	1	
К	= 1-0.003	47(Ø-30)-0.	978[(1/r)-0.	05]		r	Entry Rad	ius		6.0 - 100.0	) m	
F	$= 303x_2$	(4				L	Effective L	ength of F	are	1.0 - 100.0	) m	
t <sub>c</sub>	$= 0.210t_{D}$	(1+0.2x <sub>2</sub> )				D	Inscribed	Circle Diam	leter	15 - 100 m	1	
t <sub>D</sub>	= 1+0.5/(	1+M)				Ø	Entry Angl	le		10° - 60°		
М	= exp[(D-	60)/10]				S	Sharpness	s of Flare		0.0 - 3.0		
x <sub>2</sub>	= v+(e-v),	/(1+2S)										
S	= 1.6(e-v)	)/L										
Ratio-of-	Flow to Ca	pacity (RFC	2)									
		<b></b>	,				Q <sub>E</sub>		Entry Flow	/	RFC	
Arm	x <sub>2</sub>	М	t <sub>D</sub>	К	F	$f_{c}$	AM(08)	PM (17)	AM(08)	PM (17)	AM(08)	PM (17)
From A	4.519	0.037	1.482	0.935	1369	0.593	1265	1263	751	785	0.594	0.622
From B	4.750	0.037	1.482	0.906	1439	0.607	1083	1127	609	433	0.563	0.384
From C	5.310	0.035	1.483	1.017	1609	0.642	1252	1312	144	212	0.115	0.162
From D					-						_	
From E												
From F												
From G												
From H												
	•								_		·	

Junction:	Kam Tin Road / Lam Kam Road / Route TWIST Job Number: J73											J7342
Scenario:		Without Proposed Temporary Concrete Batching Plant									J04 - P.	4
Design Ye	ear:	2030	De	signed By	: MCY	С	hecked By:	WCH		Date:	21 Janu	ary 2025
AM (09) F	Peak											
Arm	To A	To B	To C	To D	To E	To F	To G	To H	Total	Q <sub>c</sub>		
From A	98	349	276						723	26		
From B	433	8	144						585	374		
From C	145	18	0						163	539		
From D												
From E												
From F												
From G												
From H												
Total	676	375	420						1471			
PM (19) F	Peak	T- D	T- 0	T- D	Т. Г	T. F	T- 0	T. 11	Tetel			
Arm	IO A	IOB	100	TOD	IOE	IOF	10 G	IOH	lotal	q <sub>c</sub>		
From A	122	480	188						790	58		
From B	363	6	83						452	310		
From C	223	52	0						275	491		
From D												
From E												
From F												
From G												
From H												
Total	708	538	271						1517			
					0	<b>D</b>						
Legena				1	Geometric	Paramet	ers	( )			~ ~	
Arm	Road (		e order)		Arm	e (m)	v (m)	r (m)	L (m)	D (m)	Ø (°)	<u> </u>
A						0.0	3.5	100.0	5.5	27	60	0.7
В						0.0	3.5	25.0	8.0	27	00	0.5
	Route Twi	SK (NB)			From C	7.0	3.5	65.0	12.0	27	35	0.5
					From D							
E					From E							
F												
G					From G							
					FIOIII H							
Predictiv	e Equation	Q <sub>E</sub> = K(F -	f <sub>c</sub> q <sub>c</sub> )			Limitatio	n					
Q <sub>E</sub>	Entry Cap	acity			ו ר	е	Entry Widt	th		4.0 - 15.0	m	
q <sub>c</sub>	Circulating	g Flow acros	ss the Entry	/		v	Approach	Half Width		2.0 - 7.3 m	ı	
к	= 1-0.003	- 47(Ø-30)-0.	۔ .978[(1/r)-0	051		r	Entry Rad	ius		6 0 - 100 0	) m	
F	$= 303x_2$			]		L	Effective L	Length of Fl	are	1.0 - 100.0	) m	
f <sub>c</sub>	= 0.210t <sub>D</sub> (	(1+0.2x <sub>2</sub> )				D	Inscribed	Circle Diam	eter	15 - 100 m	n	
t <sub>D</sub>	= 1+0.5/(1	(+M)				Ø	Entry Ang	le		10° - 60°		
м	= exp[(D-6	, 50)/101				S	Sharpness	s of Flare		00-30		
X <sub>2</sub>	= v+(e-v)/	(1+2S)			'		enaipheet			010 010		
s	= 1.6(e-v)	(, , /L										
		·			4							
Ratio-of-	Flow to Ca	pacity (RFC	C)						<u> </u>			
Arm	X <sub>2</sub>	М	to	к	F	fa	AM(09)	رE PM(19)	Entry AM(09)	/ Flow PM(19)	AM(09)	-C PM(19)
From A	A 510	0.037	1 /82	0 025	1360	0 503	1266	12/18	702	700	0.571	0.633
From R	4.519	0.037	1 / 82	0.000	1/30	0.090	1/08	1133	585	150	0.522	0.000
From C	5 210	0.037	1.402	1 017	1409	0.007	1090	1315	162	45Z	0.000	0.099
From D	5.510	0.035	1.403	1.017	1009	0.042	1204	1315	103	215	0.127	0.209
From E												
From E												
From C												
	1						1					

Scenario: 2030         With Proposed Temporary Concrete Batching Plant         Jule - P. 5           AM(03) PEAK         2030         Designed By         MCY         Checked By         WCH         Date:         21 January 2025           AM(03) PEAK         Am         To A         To B         To C         To D         To E         To F         To G         To H         Tota         4.4           From A         451         8         150         800         4.57         Form D         144         642           From C         125         19         0         144         642         4.50         4.50         Form F         Form F         4.53         4.64         4.7           From A         233         4.64         177         Form A         8.80         3.1         Form A         4.33         4.64           From B         3.45         6         8.2         1.122         59.0         5.5         2.7         60         0.7           From B         4.24         771         4.85         2.58         1.1225         1.1225         1.1225         1.1225         1.1225         1.1225         1.1225         1.1225         1.1225         1.1225         1.1225	Junction:	Kam Tin Road / Lam Kam Road / Route Twisk Job Number: J											J7342	
Design Verrer:         2030         Designed By         MCY         Checked By         WCH         Date:         21 January 2025           AM(6) PEAK         Am         To A         To B         To C         To C         To C         To C         To C         To B         45         9         9           From A         183         349         274         500         To C         To C         To C         009         457           From B         451         8         150         144         642         609         457           From F         From F         1444         642         11559         1443         416           From F         759         376         424         1559         1212         590           PM (17) Peak         187         25         0         1212         590         31           From B         345         6         82         1325         11525         11525           From C         771         495         259         1525         11525         11525           Lagend         Emm Facel (B)         MCm         0 (m)         2 (r)         S         0.5         5         0         0.5 <td>Scenario:</td> <td></td> <td>With Prop</td> <td>osed Temp</td> <td>orary Conc</td> <td>crete Batch</td> <td>ing Plant</td> <td></td> <td></td> <td></td> <td>J04 - P.</td> <td>5</td>	Scenario:		With Prop	osed Temp	orary Conc	crete Batch	ing Plant				J04 - P.	5		
Add(6)         PEAK           Arm         To A         To B         To C         To F         To G         To H         Total         9.0           From A         135         349         274	Design Ye	ear:	2030	. De	signed By:	MCY	- 0	hecked By:	WCH	-	Date:	21 Janu	ary 2025	
$ \begin{array}{c c c c c c c c c c c c c c c c c c c $	AM(08) P	EAK												
From A         183         349         2/4         B         450         7           From B         451         8         150         609         457         609         457           From C         125         19         0         144         642         6457           From C         750         376         424         1569         144         642           PM (7) Poak         Total         750         376         424         1569         1           From A         230         464         177         803         31         414         642           From A         230         464         177         803         117         803         117           From A         230         464         177         117 <td< td=""><td>Arm</td><td>To A</td><td>To B</td><td>To C</td><td>To D</td><td>To E</td><td>To F</td><td>To G</td><td>To H</td><td>Total</td><td>q<sub>c</sub></td><td></td><td></td></td<>	Arm	To A	To B	To C	To D	To E	To F	To G	To H	Total	q <sub>c</sub>			
From B         451         8         150           From C         125         19         0         144         642           From D         125         19         0         144         642           From F         -         159         -         144         642           From F         -         159         -         159         -           Prom F	From A	183	349	274						806	27			
From C         125         19         0         144         642           From D         Intervention	From B	451	8	150						609	457			
From D         Image: Normal Stress Stre	From C	125	19	0						144	642			
From E       From K	From D	-									-			
From F From F Total       759       376       424       1859         PM (17) Pack	From F													
Priori G From H       Total       759       376       424       1850         PM (17) Peak	From F													
Improve the second sec	From G													
$ \begin{array}{ c c c c c } \hline 100al & 759 & 376 & 424 \\ \hline 1050 & \hline $	From H													
$\begin{array}{c c c c c c c c c c c c c c c c c c c $	Total	759	376	424						1559				
$ \begin{array}{ c c c c c c c c c c c c c c c c c c c$	Total	1.00	010							1000	1			
$ \begin{array}{ c c c c c c c c c c c c c c c c c c c$	PM (17) P	Peak												
From A 239 464 177 From B 345 6 82 From D 187 25 0 From D 187 25 0 From D 7 From B 7 From H 771 495 259 Legend A Kam Road (In clockwise order) A Kam Tin Road (Ed) K Road (In clockwise order) A Kam Tin Road (Ed) B Lam Kam Road WB) C Route Twisk (NB) D 0 E 0 From B 6.0 3.5 100.0 5.5 27 60 0.7 From B 6.0 3.5 100.0 5.5 27 60 0.7 From B 6.0 3.5 26.0 8.0 27 60 0.5 From B 6.0 3.5 26.0 8.0 27 60 0.5 From C 7.0 3.5 65.0 12.0 27 35 0.5 From H From H From H Pedictive Equation $Q_{g} = K(F - f_{q}_{q})$ Limitation $Q = Entry Kadius 6.0 - 100.0 m D Inscribed Circle Diameter 15 - 100 m Q = Entry Angle 10^{-} 60^{-}S Sharpness of Flare 0.0 - 3.0From HArm x_{2} M t_{0} K F t_{0} AM(08) PM (17) AM(08) PM (17) AM(08) PM (17)AM(08) PM (17)AM (17) AM(08) PM (17)AM (17) AM (17) AM (18) 0.170From D AD A$	Arm	To A	To B	To C	To D	To E	To F	To G	To H	Total	q <sub>c</sub>			
From B       345       6       82         From C       187       25       0       212       590         From F       590       1525       1525       1525         Total       771       495       259       1525       1525         Lagend       Arm Road (EB)         A       Kam Tin Road (EB)       Arm       6.0       3.5       100.0       5.5       27       60       0.7         B       Lam Kam Road WB)       From A       6.0       3.5       25.0       8.0       27       60       0.5         D       0       0       From B       6.0       3.5       25.0       8.0       27       60       0.5         From D       From B       6.0       3.5       25.0       8.0       27       60       0.5         From D       From D       From B       From D       From E       From B       From B       6.0       3.5       25.0       8.0       27       60       0.5         G       O       Immation       Immati	From A	239	464	177						880	31			
From C       187       25       0       212       590         From D       From C       187       25       0       1825         From E       From H       1525       1525       1525         Lagend       771       495       259       1525       1525         Lagend       771       495       259       1525       1525         Lagend       771       495       259       1525       50       5       27       60       0.7         A       Kam Tin Road (EB)       From A       6.0       3.5       10.0       5.5       27       60       0.5         Form B       6.0       3.5       25.0       8.0       27       60       0.5         Form B       6.0       3.5       25.0       8.0       27       60       0.5         Form B       Form B       6.0       3.5       65.0       12.0       27       35       0.5         Form B       Form B       Form B       Form B       6.0       0.0       10.0       0.7       10.0       10.0       10.0       10.0       10.0       10.0       10.0       10.0       10.0       10.0       10.0 <td>From B</td> <td>345</td> <td>6</td> <td>82</td> <td></td> <td></td> <td></td> <td></td> <td></td> <td>433</td> <td>416</td> <td></td> <td></td>	From B	345	6	82						433	416			
$ \begin{array}{c c c c c c c c c c c c c c c c c c c $	From C	187	25	0						212	590			
From E From H       Second I Total       771       426       259       1525         Legend       Image: Second II Colockwise order)         Arm       Road (in clockwise order)       Image: Second II Colockwise order)         A       Kam Tin Road (EB)       Arm $e$ (m)       v (m)       r (m)       L (m)       D (m) $\otimes$ (°)       S         B       Lam Kam Road WB)       C       No       6.0       3.5       100.0       5.5       27       60       0.7         Form B       6.0       3.5       25.0       8.0       27       60       0.7         Form C       7.0       3.5       65.0       12.0       27       60       0.7         Form B       6.0       3.5       25.0       8.0       27       60       0.7         Form B       6.0       3.5       25.0       8.0       27       60       0.7         Form B       6.0       3.5       100.1       5.0       12.0       27       60       0.7         Form B       6.0       3.5       12.0       27       60       0.7       7         Form B       6.0       3.5       100.1       <	From D													
From Fe From H         Iso         Iso           Total         771         495         259         1525           Legend         Iso           Am         Road (in clockwise order)         A         A $(m)$ $v$ (m) $r$ (m) $L$ (m) $D$ (m) $O$ (°)         S           A         Kam Tin Road (EB)         A $(m)$ $v$ (m) $r$ (m) $L$ (m) $D$ (m) $O$ (°)         S           D         0         0         From B         6.0         3.5         100.0         5.5         27         60         0.7           B         Lam Kam Road W(B)         D         From B         6.0         3.5         25.0         8.0         2.7         60         0.5           D         0          From B         From D         From E         From E         From E         From F         D         10.50 m         V         Approach Half Width         2.0 - 7.3 m         C         10.100 m         L         Effective Length of Flare         10100.0 m         D	From E													
From G From H       T71       495       259         Isometric Parameters         Lagend         Arm       Road (in clockwise order)         A       Kam Tin Road (EB)         B       Lam Kam Road WB)         C       Route Twisk (NB)         D       0         E       0         From B       6.0       3.5       100.0       5.5       27       60       0.7         From B       6.0       3.5       100.0       5.5       27       60       0.5         From B       6.0       3.5       25.0       8.0       27       60       0.5         From B       From B       6.0       3.5       25.0       8.0       27       60       0.5         From B       From B       6.0       3.5       12.0       27       35       0.5         From B       From F       From B       6.0       12.0       27       35       0.5         Prodictive Equation Q <sub>E</sub> = K(F - f_q_e)       Eintry Width       4.0 - 15.0 m       4.0 - 16.0 m       5       S harpness of Flare       0.0 - 3.0	From F													
From H         construction	From G													
Total         771         495         259           Legend         Geometric Parameters         Its25           Arm         Road (in clockwise order)         A         Kam Tin Road (EB) $\overline{M''}$ $\overline{M'''}$ $\overline{M''''}$ $\overline{M''''''}$ $\overline{M'''''''''}$ $M''''''''''''''''''''''''''''''''''''$	From H													
Geometric Parameters           Arm         Road (in clockwise order)         A         Kam Tin Road (EB)         Trom A         6.0         3.5         100.0         5.5         27         60         0.7           B         Lam Kam Road (WB)         C         Route Twisk (NB)         0         0         0.5         5.5         27         60         0.7           From B         6.0         3.5         25.0         8.0         27         60         0.5           C         Route Twisk (NB)         0         0         7.0         3.5         65.0         12.0         27         35         0.5           From B         From C         7.0         3.5         65.0         12.0         27         35         0.5           From B         From C         7.0         3.5         65.0         12.0         27         35         0.5           From B         From G         From G         From G         From G         From G         From G         6.0         10.0         0.5         10.0         0.5         10.0         0.5         10.0         0.5         10.0         10.5         10.5         10.5         10.5         10.5         10.5	Total	771	495	259						1525				
Geometric Parameters           Arm         Read (in clockwise order)         A         Kam Tin Read (EB)         Form A         6.0         3.5         100.0         5.5         2.7         60         0.7           B         Lam Kam Road (WB)         From A         6.0         3.5         100.0         5.5         2.7         60         0.5           C         Route Twisk (NB)         From D         6.0         3.5         25.0         8.0         2.7         35         0.5           F         0         0         From D         From D         From D         7.0         3.5         65.0         12.0         2.7         35         0.5           F         G         From C         7.0         3.5         65.0         12.0         2.7         35         0.5           F         G         From D         From D         From B         6.0         3.5         65.0         12.0         2.7         35         0.5           F         G         Circulating Flow across the Entry         K         Form H         4.0         -15.0 m         V         Approach Half Width         2.0         7.3 m         F         5         5         5						•	_							
$ \begin{array}{ c c c c c c c c c c c c c c c c c c c$	Legend				I	Geometri	C Paramet	ers			5 ( )	~ ~		
$ \begin{array}{c ccccccccccccccccccccccccccccccccccc$	Arm	Road (		e order)		Arm	e (m)	v (m)	r (m)	L (m)	D (m)	Ø (°)	5	
$ \begin{array}{c c c c c c c c c c c c c c c c c c c $	A		Koad (EB)			From A	6.0	3.5	100.0	5.5	27	60	0.7	
$ \begin{array}{c c c c c c c c c c c c c c c c c c c $	В		Road WB)			From B	6.0	3.5	25.0	8.0	27	60	0.5	
$ \begin{array}{c c c c c c c c c c c c c c c c c c c $	C	Route Twi	SK (NB)			From C	7.0	3.5	65.0	12.0	27	35	0.5	
$\begin{array}{c c c c c c c c c c c c c c c c c c c $	D	0				From D								
$\begin{array}{c c c c c c c c c c c c c c c c c c c $	E					From E								
G H       From G From H         Predictive Equation $Q_{E} = K(F - f_{c}q_{c})$ Limitation $Q_{E}$ Entry Capacity $q_{c}$ 4.0 - 15.0 m $v$ $q_{c}$ Circulating Flow across the Entry $K = 1-0.0347(\mathcal{E}^{-30})-0.978[(1/r)-0.05]$ $F = 303x_{2}$ $t_{0} = 1+0.5/(1+0.2x_{2})$ $t_{0} = 1+0.5/(1+M)$ $M = exp[(D-60)/10]$ $x_{2} = v+(e-v)/(1+2S)$ $S = 1.6(e-v)/L$ Limitation         Circulating Flow across the Entry $M = exp[(D-60)/10]$ $x_{2} = v+(e-v)/(1+2S)$ $S = 1.6(e-v)/L$ Ce       Entry Flow to Capacity (RFC)         Ratio-of-Flow to Capacity (RFC)         Ce       Entry Flow         Ratio-of-Flow to Capacity (RFC)       K       F $f_{c}$ AM(08)       PM (17)       AM(08)       PM (17)       AM(08)       PM (17)         From A       4.519       0.037       1.482       0.935       1369       0.593       1265       1263       806       880       0.637       0.697         From B       4.750       0.037       1.482       0.906       1439       0.607       1052       1075       609       433       0.579       0.403         From B       4.750       0.035       1.483       1.017       1609       0.642       12	F					From F								
H       From H         Predictive Equation $Q_E = K(F \cdot f_c q_c)$ Limitation $Q_E$ Entry Capacity       4.0 - 15.0 m $q_c$ Circulating Flow across the Entry $K$ $4.0 - 15.0 m$ $K = 1-0.00347(\emptyset - 30) - 0.978[(1/r) - 0.05]$ $F$ $= 303x_2$ $6.0 - 100.0 m$ $f_c = 0.210t_0(1+0.2x_2)$ $b$ $E$ $Entry Radius$ $6.0 - 100.0 m$ $t_0 = 1+0.5/(1+M)$ $M$ $e xp([0-60)/10]$ $X_2 = v+(e-v)/(1+2S)$ $S$ $S$ $S = 1.6(e-v)/L$ $S$ Sharpness of Flare $0.0 - 3.0$ $S$ Ratio-of-Flow to Capacity (RFC)         Q <sub>E</sub> Entry Flow       RFC         Arm $X_2$ M $t_0$ $K$ F $f_c$ AM(08)       PM (17)       AM(08)       PM (17)         From A       4.519       0.037       1.482       0.906       1439       0.607       1052       1075       609       433       0.579       0.403         From B       4.750       0.035       1.483       1.017       1609       0.642       1216       1250       144       212       0.118       0.170 <td>G</td> <td></td> <td></td> <td></td> <td></td> <td>From G</td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td>	G					From G								
Predictive Equation $Q_E = K(F \cdot f_c q_c)$ $Q_E$ Entry Capacity $     <<<>>>>$	н					From H								
$ \begin{array}{c c c c c c c c c c c c c c c c c c c $	Predictive	e Equation	Q <sub>E</sub> = K(F -	f <sub>c</sub> q <sub>c</sub> )		_	Limitatio	n					_	
$\begin{array}{c c c c c c c c c c c c c c c c c c c $	Q <sub>E</sub>	Entry Cap	acity			]	е	Entry Widt	th		4.0 - 15.0	m		
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$	q <sub>c</sub>	Circulating	g Flow acros	ss the Entry	/		v	Approach	Half Width		2.0 - 7.3 m	ı		
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$	К	= 1-0.0034	47(Ø-30)-0.	978[(1/r)-0.	05]		r	Entry Radi	ius		6.0 - 100.0	) m		
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	F	= 303x <sub>2</sub>					L	Effective L	ength of F	lare	1.0 - 100.0	) m		
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	f <sub>c</sub>	= 0.210t <sub>D</sub> (	1+0.2x <sub>2</sub> )				D	Inscribed (	Circle Diam	neter	15 - 100 m	ı		
$ \begin{array}{c ccccccccccccccccccccccccccccccccccc$	t <sub>D</sub>	= 1+0.5/(1	+M)				Ø	Entry Angl	е		10° - 60°			
X <sub>2</sub> = v+(e-v)/(1+2S)         S = 1.6(e-v)/L         Ratio-of-Flow to Capacity (RFC)         Q <sub>E</sub> Entry Flow       RFC         Arm X <sub>2</sub> M t <sub>D</sub> K F       f <sub>c</sub> AM(08) PM (17) AM(08) PM (17) AM(08) PM (17)         From A 4.519       0.037       1.482       0.906       1439       0.607       1052       1075       609       433       0.637       0.403         From B 4.750       0.037       1.482       0.906       1439       0.607       1052       1075       609       433       0.579       0.403         From C       5.310       0.035       1.483       1.017       1609       0.403       0.579       0.403       From C       5.310       0.035       1.483       1.017       1609       144       212       0.118       0.170       From F         From G <th colsp<="" td=""><td>М</td><td>= exp[(D-6</td><td>60)/10]</td><td></td><td></td><td></td><td>S</td><td>Sharpness</td><td>s of Flare</td><td></td><td>0.0 - 3.0</td><td></td><td></td></th>	<td>М</td> <td>= exp[(D-6</td> <td>60)/10]</td> <td></td> <td></td> <td></td> <td>S</td> <td>Sharpness</td> <td>s of Flare</td> <td></td> <td>0.0 - 3.0</td> <td></td> <td></td>	М	= exp[(D-6	60)/10]				S	Sharpness	s of Flare		0.0 - 3.0		
S = 1.6(e-v)/L         Ratio-of-Flow to Capacity (RFC)         Arm       X <sub>2</sub> M       t <sub>D</sub> K       F       f <sub>c</sub> AM(08)       PM (17)       AM(08)       PM (17)       AM(08)       PM (17)         From A       4.519       0.037       1.482       0.935       1369       0.593       1265       1263       806       880       0.637       0.697         From B       4.750       0.037       1.482       0.906       1439       0.607       1052       1075       609       433       0.579       0.403         From C       5.310       0.035       1.483       1.017       1609       0.642       1216       1250       144       212       0.118       0.170         From D       From F       From F       From G       From H       I	x <sub>2</sub>	= v+(e-v)/	(1+2S)										-	
Ratio-of-Flow to Capacity (RFC)           Arm         X2         M         to         K         F         fc         AM(08)         PM (17)         AM(08)         D(17)         AM(08)	S	= 1.6(e-v)/	/L											
Arm         X2         M         t_D         K         F         f_c         AM(08)         PM (17)         AM (08)	Ratio-of-	Flow to Ca	nacity (REC											
Arm         x2         M         tD         K         F         fc         AM(08)         PM (17)         A           From B         4.750         0.037         1.483         1.017         1609         0.642         1216         1250         144         212         0.118         0.170           From C         5.310         0.035         1.483         1.017         1609         0.642         12				- 1				Q <sub>E</sub>		Entry Flow	v	RFC		
From A         4.519         0.037         1.482         0.935         1369         0.593         1265         1263         806         880         0.637         0.697           From B         4.750         0.037         1.482         0.906         1439         0.607         1052         1075         609         433         0.579         0.403           From C         5.310         0.035         1.483         1.017         1609         0.642         1216         1250         144         212         0.118         0.170           From D         From F         From F         From G         From H         Image: Second S	Arm	x <sub>2</sub>	Μ	t <sub>D</sub>	к	F	f <sub>c</sub>	AM(08)	PM (17)	AM(08)	PM (17)	AM(08)	PM (17)	
From B       4.750       0.037       1.482       0.906       1439       0.607       1052       1075       609       433       0.579       0.403         From C       5.310       0.035       1.483       1.017       1609       0.642       1216       1250       144       212       0.118       0.170         From D       From F       From F       From G       From H       Image: Constraint of the second	From A	4.519	0.037	1.482	0.935	1369	0.593	1265	1263	806	880	0.637	0.697	
From C       5.310       0.035       1.483       1.017       1609       0.642       1216       1250       144       212       0.118       0.170         From D       From E       From F       From G       Image: Constraint of the constraint of	From B	4.750	0.037	1.482	0.906	1439	0.607	1052	1075	609	433	0.579	0.403	
From D From E From F From G From H	From C	5.310	0.035	1.483	1.017	1609	0.642	1216	1250	144	212	0.118	0.170	
From E From F From G From H	From D													
From F From G From H	From E													
From G From H	From F													
From H	From G													
	From H													

Junction:	: Kam Tin Road / Lam Kam Road / Route TWIST Job Number: J										J7342	
Scenario:		With Prope	osed Temp	orary Con	crete Batchir	ng Plant			_	J04 - P.	6	
Design Ye	ear:	2030	De	signed By	: MCY	С	hecked By:	WCH		Date:	21 Janu	ary 2025
AM (09) F	Peak											
Arm	To A	To B	To C	To D	To E	To F	To G	To H	Total	q <sub>c</sub>		
From A	193	349	276						818	26		
From B	433	8	144						585	469		
From C	145	18	0						163	634		
From D												
From E												
From F												
From G												
From H												
Total	771	375	420						1566			
										<u> </u>		
PM (19) F	Peak											
Arm	To A	To B	To C	To D	To E	To F	To G	To H	Total	q <sub>c</sub>		
From A	162	480	188						830	58		
From B	363	6	83						452	350		
From C	223	52	0						275	531		
From D		02	U						210			
From E												
From E												
From G												
From H												
Total	7/8	538	271						1557			
Total	140	000	211						1007			
l egend					Geometric	Paramet	ers					
Arm	Pood	(in clockwic	o ordor)		Arm		) (m)	r (m)	L (m)	D(m)	Ø (9)	6
	Kam Tin F				From A	6 (III)	3.5	100.0	5.5	27	<u></u> 60	0.7
					From B	6.0	3.5	25.0	0.0	27	60	0.7
		iak (NP)			From C	0.0	3.5	25.0	12.0	27	25	0.5
	Roule IW	ISK (IND)			From D	7.0	5.5	05.0	12.0	21	55	0.5
G L												
Predictiv	e Equation	$0_{-} = K/F_{-}$	fa)			Limitatio	•					
0-	Entry Car		·c4c/		n r		Entry Widt	th		10 150	m	
a.	Circulation	a Flow acros	es the Entr	,		e v		Half Width		2073m	\	
Чc IZ		$47(\propto 20)$ 0		051		v	Approach			2.0 - 7.3 11		
ĸ	= 1-0.003 $= 303x_{-}$	47( <i>∅</i> -30)-0.	978[(1/r)-0.	05]		r	Entry Rad	IUS	oro	6.0 - 100.0	) m	
f	$= 0.00 \Lambda_2$ = 0.010+	(1+0 2v )				L D			aite otor	1.0 - 100.0	, 111	
'c t_		(+ · U.ZA2)						oncie Diam	elei	10° 60°	I	
۳D	= 1+0.5/(	(1+1VI)				Ø	Entry Ang	ie ( Ei		10 - 60		
M	= exp[(D-	60)/10] //4 - 00)			L	S	Sharpness	s of Flare		0.0 - 3.0		_
×2	= v+(e-v)/	(1+25)										
S	= 1.6(e-v)	/L										
Ratio-of-	Flow to Ca	pacity (RFC	5)						<b>-</b> ·		D	-0
				12	-	4			Entry		RI	
Arm	x <sub>2</sub>	М	τ <sub>D</sub>	К	F	Γ <sub>C</sub>	AM(09)	PM(19)	AM(09)	PM(19)	AM(09)	PM(19)
From A	4.519	0.037	1.482	0.935	1369	0.593	1266	1248	818	830	0.646	0.665
From B	4.750	0.037	1.482	0.906	1439	0.607	1046	1111	585	452	0.559	0.407
From C	5.310	0.035	1.483	1.017	1609	0.642	1222	1289	163	275	0.133	0.213
From D												
From E												
From F												
From G												
From H												

Appendix C – Swept Path Analyses











:\JOB\J7300-J7349\J7342\(2025 01) TIA\_R2\Fig 3.3 & SP1XX Rev A.dwg



.UOB\J7300-J7349\J7342\(2025 01) TIA\_R2\Fig 3.3 & SP1XX Rev A.dwg






.UOB\J7300-J7349\J7342\(2025 01) TIA\_R2\Fig 3.3 & SP1XX Rev A.dwg





















### Appendix IV

**Environmental Assessment** 







### J24.00171.HK.01

S.16 Planning Application for Proposed Temporary Concrete Batching Plant with Ancillary Facilities for a Period of 5 Years at Lots 573 RP and 1710 in D.D. 114, Shek Kong, Yuen Long Environmental Assessment

Prepared for: Join Bright Warehousing Limited 24 January 2025



# S.16 Planning Application for Proposed Temporary Concrete Batching Plant with Ancillary Facilities for a Period of 5 Years at Lots 573 RP and 1710 in D.D. 114, Shek Kong, Yuen Long Environmental Assessment

Prenared for

		Join Bright	Warehousi	ing Limi	ted	
For	and on beh	nalf of	The chouse			
Envi	iroSolution	s & Consulting				
Alex Grou	<b>ci BHANJA</b>					
ESC	Project No	. J24.00171.H	(.01			
Deliv	verable No	. D01				
Revi	sion No.	1				
File L	ocation					
https: jobs/j	//envirosc.sha 24.00171.hk.(	arepoint.com/teams/r D1and02 kam tin cbp s	newsharepoint/sha 16/06. deliverable:	red documer s/j24.00171.ł	nts/new share	epoint/05. _ea_v2.docx
Rev.	Description		Prepared	Reviewed	Approved	Date
0	Environmen	ital Assessment	JC	MW	AW	29/07/2024
1	Environmen	tal Assessment	JC	MW	AW	24/01/2025
Distri	bution	Internal	🛛 Confidentia	l	🗆 Public	
This re and d	eport has bee iligence withi	n prepared by EnviroSo in the terms of the Co	olutions & Consulti ontract with Client	ng Limited w t, incorporati	ith all reasona ng our Gene	able skill, care, ral Terms and

This report has been prepared by EnviroSolutions & Consulting Limited with all reasonable skill, care, and diligence within the terms of the Contract with Client, incorporating our General Terms and Conditions of Business and taking account of the resources devoted to it by agreement with Client. We disclaim any responsibility to Client and others in respect or any matters outside the scope of the above. This report is confidential to Client and we accept no responsibility of whatsoever nature to any third parties to whom this report, or any part thereof, is made known.



# CONTENTS

1	I	Introduction	1-1
	1.1	Project Background	1-1
	1.2	Site Description	1-1
	1.3	Objective	1-1
2	1	AIR QUALITY	2-1
	2.1	Environmental Legislation and Standards	2-1
	2.2	Construction Phase Impact	2-3
	2.3	Operation Phase Impact	2-5
	2.4	Cumulative Air Quality Impact Assessment for Operation Phase	2-8
	2.5	Air Dispersion Model	2-8
	2.6	Vehicular Emissions from Open Roads	2-9
	2.7	Industrial Emissions	2-9
	2.8	Particle Size Distribution	2-10
	2.9	Ozone Limiting Method for Short-term Cumulative NO2 Assessment	2-10
	2.10	Jenkin Method for Long-term Cumulative NO2 Assessment	2-11
	2.11	Assessment Results	2-11
	2.12	Conclusion	2-15
3	ſ	NOISE	3-1
	3.1	Environmental Legislation and Standards	3-1
	3.2	Noise Impact of Construction Phase	3-3
	3.3	Noise Impact of Operation Phase	3-5
	3.4	Noise Impact from Industrial Noise	3-8
	3.5	Noise Impact from Off-Site Traffic Noise	3-10
	3.6	Conclusion	3-11
4	١	WATER QUALITY	4-1
	4.1	Environmental Legislation, Standards and Guidelines	4-1
	4.2	Identification of Water Control Zone ("WCZ") and Water Sensitive Receiver ("WSR")	4-2
	4.3	Review of Water Quality Impact	4-2
	4.4	Mitigation Measures	4-3
	4.5	Conclusion	4-6
5	١	WASTE MANAGEMENT	5-1
	5.1	Environmental Legislation and Standards	5-1
	5.2	Review of Waste Management Impacts	5-1
	5.3	Mitigation Measures	5-7
	5.4	Conclusion	5-11
6	I	LAND CONTAMINATION	6-1



6.1	Environmental Legislation and Standards	6-1
6.2	Assessment Methodology	6-1
6.3	Site Appraisal	6-1
6.4	Conclusion	6-2
Meth	odology and Assumptions for Calculation of Vehicular Emission Source	1

# **APPENDICES**

- Appendix A Calculation and Summary of Emission Rates
- Appendix B Landuse Characteristics Parameters for AERMET
- Appendix C EMFAC-HK Model Input
- Appendix D Cumulative Results comparing to Updated Pollutants of Proposed AQOs
- Appendix E Background Noise Monitoring Data
- Appendix F Noise Meter Calibration Certificates
- Appendix G Result of Industrial Noise Impact Assessment
- Appendix H Traffic Forecasts for Year 2031
- Appendix I Result of Traffic Noise Impact Assessment
- Appendix J Calculation of Sewage Generation
- Appendix K Historical Aerial Photographs
- Appendix L Site Walkover Checklist and Site Photos
- Appendix M Correspondence with EPD and FSD

### **FIGURES**

Figure 1-1	Site Location and its Environs	1-2
Figure 1-2	Site Layout Plan	1-3
Figure 2-1	Locations of Emission Points of the Plant	2-16
Figure 2-2	Locations of ASRs	2-17
Figure 3-1	Locations of Representative NSRs and Background Noise Measurement	3-12
Figure 3-2	Locations of Proposed Noise Sources	3-13
Figure 4-1	Representative WSRs within 500m Study Area of the Site	4-7

### **TABLES**

Table 2-1	Hong Kong Air Quality Objectives	2-1
Table 2-2	Concentration Limit for Cement works – Concrete Batching Plant	2-2
Table 2-3	Summary of Identified Emission Points within CBP during Operation	2-6
Table 2-4	Emission Control Measures for EPs	2-6
Table 2-5	Representative ASRs	2-8
Table 2-6	Particle Size Distribution for Different Emission Types	2-10
Table 2-7	Predicted FSP and RSP Concentrations at Representative ASRs (2024 AQOs)	2-12
Table 2-8	Predicted NO <sub>2</sub> Concentrations at Representative ASRs (2024 AQOs)	2-13
Table 3-1	Construction Noise Criteria for Non-Restricted Hours	3-1
Table 3-2	Acceptable Noise Levels for Fixed Noise Source	3-2
Table 3-3	HKPSG Standards for Road Traffic Noise Standards	3-2
Table 3-4	Representative NSRs	3-5
Table 3-5	Noise Criteria of Noise from Fixed Sources of Identified Representative NSRs	3-6



Table 3-6 9	Predicted Fixed Noise Impact Level at Identified Representative NSRs During Operation Phase.3-
Table 3-7	Proposed Operation Phase Noise Mitigation Measures
Table 3-8	Predicted Traffic Noise Impact Level at Identified Representative NSRs During Operation Phase 3-10
Table 4-1	Identified Water Sensitive Receivers
Table 5-1	Estimated Amount of Different Types of Wastes to be Generated During Construction Phase .5-6
Table 6-1	Historical Land Uses of the Project Site6-2
Table D-21	Predicted FSP and RSP Concentrations at Representative ASRs (2025 AQOs)1
Table D-32	Predicted NO <sub>2</sub> Concentrations at Representative ASRs (2025 AQOs)2



## 1 Introduction

#### 1.1 Project Background

- 1.1.1 Land resumption will be carried out to facilitate the development of Hung Shui Kiu/ Ha Tsuen New Development Area. The existing Concrete Batching Plant ("CBP") in Hung Shui Kiu will be affected by the land resumption, thus, Join Bright Warehousing Limited ("the Client") has identified a site location at Lots 573 RP and 1710 in D.D. 114, Shek Kong ("the Site") for the existing CBP relocation ("the Proposed Development"). Upgrading works on Chun Yiu Road is also proposed to facilitate the operation of CBP.
- 1.1.2 The Site is currently zoned as "Industrial (Group D)"("I(D)") on the Approved Shek Kong Outline Zoning Plan ("OZP") No. S/YL-SK/9. 'Concrete Batching Plant' is a Column 2 use within the "I(D)" zone on the OZP. A planning application is required to submit and obtain permission from Town Planning Board under Section 16 of the Town Planning Ordinance.
- 1.1.3 According to Item K.5, Part I, Schedule 2 of the Environmental Impact Assessment Ordinance, "a cement works or concrete batching plant with a total silo capacity of more than 10 000 tonnes in which cement is handled and manufactured" is a Designated Project (DP). The total silo capacity of the proposed CBP will not be more than 5,000 tonnes, hence, it is not considered as DP.
- 1.1.4 EnviroSolutions & Consulting Limited ("ESC") was employed by Join Bright Warehousing Limited to carry out an Environmental Assessment ("EA") to support the abovementioned application.

#### **1.2** Site Description

1.2.1 The site is currently used for storage and maintenance of construction vehicles and machinery. The site surroundings in general are open storage, scrapyard and car repairing workshops. **Figure 1-1** and **Figure 1-2** show the location and tentative layout of the site.

#### 1.3 Objective

- 1.3.1 The objectives of this EA Report are to:
  - Assess the potential environmental impacts arising from the operation of the Proposed Development, in terms of air quality, noise, water quality and waste management.
  - Propose measures for compliance with the List of Guidance Notes on Best Practicable Means (BPM) for Specified Processes for Cement Works (Concrete Batch Plant).
  - Recommend further appropriate measures to mitigate any impacts if necessary.



#### Figure 1-1 Site Location and its Environs





Figure 1-2 Site Layout Plan





# 2 AIR QUALITY

#### 2.1 Environmental Legislation and Standards

#### **Air Quality Objectives**

2.1.1 The Air Quality Objectives ("AQOS") established under the *Air Pollution Control Ordinance* ("APCO") (Cap. 311) are given in **Table 2-1**Table 2-1.

#### Table 2-1Hong Kong Air Quality Objectives

POLLUTANT	AVERAGING TIME	CONCENTRATION LIMIT <sup>[Note 1]</sup> (µg/m <sup>3</sup> )	NUMBER OF EXCEEDANCES ALLOWED
Sulphur Dioxide (SO <sub>2</sub> )	10-minute	500	3
	24-hour	50	3
Respirable Suspended Particulates	24-hour	100	9
("RSP" or "PM <sub>10"</sub> ) <sup>[Note 2]</sup>	Annual	50	Not applicable
Fine Suspended Particulates	24-hour	50	35
("FSP" or "PM <sub>2.5</sub> ") <sup>[Note 3]</sup>	Annual	25	Not applicable
Nitrogen Dioxide (NO <sub>2</sub> )	1-hour	200	18
	Annual	40	Not applicable
Ozone	8-hour	160	9
Carbon Monoxide	1-hour	30,000	0
	8-hour	10,000	0
Lead	Annual	0.5	Not applicable

Notes:

- 1. All measurements of the concentration of gaseous air pollutants, i.e. SO<sub>2</sub>, NO<sub>2</sub>, O<sub>3</sub> and CO, are to be adjusted to a reference temperature of 293 Kelvin and a reference pressure of 101.325 kilopascal.
- 2. RSP means suspended particles in air with a nominal aerodynamic diameter of  $10\mu m$  or less.
- 3. FSP means suspended particles in air with a nominal aerodynamic diameter of  $2.5\mu m$  or less.

#### Air Pollution Control (Construction Dust) Regulation

- 2.1.2 Enacted under Section 43 of the APCO, the *Air Pollution Control (Construction Dust) Regulation* defines notifiable and regulatory works to ensure effective dust abatement measures have been properly implemented to reduce dust emissions for a number of construction activities.
- 2.1.3 The Regulation requires that advance notice is given to EPD for any notifiable work<sup>[Ref.#1]</sup> and the contractor shall ensure that the notifiable and regulatory works are carried out in

<sup>&</sup>lt;sup>1</sup> Notifiable works include site formation, reclamation, demolition of a building, work carried out in any part of a tunnel that is within 100m of any exit to the open air, construction of the foundation of a building, construction of the superstructure of a building and road construction work.



accordance with the Schedule of the Regulation, which also includes dust control and suppression measures.

#### Air Pollution Control (Non-road Mobile Machinery) (Emission) Regulation

2.1.4 This Regulation takes effect on June 2015 and required Non-road Mobile Machinery ("NRMM"), except those exempted, to comply with the prescribed emission standards. From 1 September 2015, all regulated machines sold or leased for use in Hong Kong must be approved or exempted with a proper label in a prescribed format issued by EPD. Starting from 1 December 2015, only approved or exempted NRMMs with a proper label are allowed to be used in specified activities and locations including construction sites, container terminals and back up facilities, restricted areas of the airport, designated waste disposal facilities and specified processes.

#### Air Pollution Control (Fuel Restriction) Regulation

2.1.5 The regulation imposes legal control on the type of fuels allowed for use and their sulphur contents in commercial and industrial processes. Gaseous fuel, conventional solid fuel with a sulphur content not exceeding 1% by weight or liquid fuel with a sulphur content not exceeding 0.005% by weight and a viscosity not more than 6 centistokes at 40°C, such as Ultra Low Sulphur Diesel ("ULSD") are permitted to be used in commercial and industrial processes.

#### Guidance Notes on Best Practicable Means ("BPM") for SPs

2.1.6 A number of GN have been published for SPs which "A Guidance Note on the Technical, Management and Monitoring Requirements for Specified Process – Cement Works (Concrete Batch Plant) BPM 3/2(16)" ("the BPM") dated February 2016 is relevant to the Proposed CBP. With reference to Annex 1 of the BPM, the emissions from non-fugitive fixed emission points of the Proposed CBP shall not exceed the concentration limited in **Table 2-2**Table 2-2.

#### Table 2-2 Concentration Limit for Cement works – Concrete Batching Plant

AIR POLLUTANT	CONCENTRATION LIMIT
Particulate matter	10 mg/m <sup>3</sup> (design standard) <sup>[Note 1]</sup>

Notes:

For the emission points of existing premises with a designed concentration limit of 50 mg/m<sup>3</sup>, the licence holder shall, upon licence renewal, submit an improvement plan to meet the limit of 10 mg/m<sup>3</sup> (design standard). The concentration limit of 10 mg/m<sup>3</sup> (design standard) shall be met by 1 January 2018 for all plants.

#### **Supporting Documents for SP Licence**

2.1.7 The Proposed CBP will have total silo capacity exceeding 50 tonnes. Therefore, it is classified as cement works in respect of a Specified Process ("SP") requiring an application of a SP Licence under Section 14 of APCO.



#### 2.2 Construction Phase Impact

- 2.2.1 Fugitive dust is the major impact that will be generated during construction activities, such as excavation, stockpiling, earth moving, transferring or handling of dusty materials.
- 2.2.2 There is no construction programme at this stage of the project but the duration of construction of the plant is estimated to be about 12 months and assumed to be started at 2025.
- 2.2.3 Based on desktop review, no major concurrent project within in 500m assessment area is anticipated. As advised by the Applicant, the site is already formed. Therefore, construction works including site formation/excavation and use of construction plants such as dump trucks, excavators, etc. will be minimal. With the implementation of dust control measures stipulated in the *Air Pollution Control (Construction Dust) Regulation*, dust generation can be controlled and significant fugitive dust impact is therefore not anticipated. More detailed descriptions for control measures are provided below.
- 2.2.4 Since the Site is fully paved with concrete, no site formation works is required during the construction phase. Excavation works are minimal. About 6 tonnes per day of inert construction and demolition (C&D) materials and less than 1 tonne of non-inert C&D materials will be excavated. Since the works area is limited for construction works, it is anticipated that no more than 20 numbers of construction plant would be in operation during each construction activity.
- 2.2.5 There is about 1 dump truck per day estimated for transportation of excavated material from site formation works. dump truck loaded with excavated material would be covered entirely before leaving the Project Site to ensure that dusty material would not leak from the dump truck according to the APCO requirement. In case temporary stockpiling of small amount of dusty material is required, the stockpile will be covered by tarpaulin sheets or placed in an area sheltered on the top and the 3 sides. The mitigation measures described in as below would also be implemented during the construction phase to minimise impacts on air quality on nearby ASRs.

#### **Emission Control Measures**

- 2.2.6 To avoid adverse air quality impact on the air sensitive uses nearby, good practice and control measures to be implemented during the construction phase are as follows:
- 2.2.7 Provide hard paving on open area, regular watering to reduce dust emissions from exposed site surfaces and unpaved roads, particularly during dry weather.
- 2.2.8 The working area of any excavation or earth moving operation shall be sprayed with water immediately before, during and immediately after the operation so as to maintain the entire surface wet.
- 2.2.9 Frequent watering for particularly dusty areas and areas close to ASRs.
- 2.2.10 Any stockpile of dusty materials shall be either covered entirely by impervious sheeting, placed in an area sheltered on the top and three sides, or sprayed with water so as to maintain the entire surface wet.



- 2.2.11 Where possible, dusty materials shall be sprayed with water immediately prior to any loading, unloading or transfer operation so as to maintain the dusty materials wet.
- 2.2.12 The working area for the uprooting of trees, shrubs, or vegetation or for the removal of boulders, poles, pillars or temporary or permanent structures shall be sprayed with water immediately before, during and immediately after the operation so as to maintain the entire surface wet.
- 2.2.13 All demolished items (including trees, shrubs, vegetation, boulders, poles, pillars, structures, debris, rubbish and other items arising from the site clearance) that may dislodge dust particles shall be covered entirely by impervious sheeting or placed in an area sheltered on the top and three sides within a day of demolition.
- 2.2.14 Tarpaulin covering of all dusty vehicles loads transported to, from and between site locations.
- 2.2.15 Vehicle washing facilities including a high-pressure water jet shall be provided at every discernible or designated vehicle exit point. The area where vehicle washing takes place and the section of the road between the washing facilities and the exit point shall be paved with concrete, bituminous materials or hardcore.
- 2.2.16 Provision of not less than 2.4m high hoarding from ground level along site boundary where adjoins a road, streets or other accessible to the public except for a site entrance or exit.
- 2.2.17 Spray water on the surface of façade grinding work as far as practicable.
- 2.2.18 Equip vacuum cleaner on grinder for façade grinding work as far as practicable.
- 2.2.19 Main haul road shall be sprayed with water so as to maintain the entire surface wet. Imposition of speed controls for vehicles on site haul roads and confine haulage and delivery vehicles to designated roadways inside the site.
- 2.2.20 A portion of any road leading only to a construction site that is within 30m of a discernible or designated vehicle entrance or exit shall be kept clear of dusty materials.
- 2.2.21 Where possible, routing of vehicles and positioning of construction plant should be at the maximum possible distance from the ASRs.
- 2.2.22 Every stock of more than 20 bags of cement or dry Pulverised Fuel Ash ("PFA") should be covered entirely by impervious sheeting or placed in an area sheltered on the top and three sides.
- 2.2.23 Emission from Non-Road Mobile Machinery ("NRMM") is governed under the Air Pollution Control (Non-road Mobile Machinery) (Emission) Regulation, as mentioned in Paragraph
   2.1.4. Each NRMM to be used in construction site shall be obtained with valid label in a prescribed format issued by EPD. The Contractor should avoid the use of exempted NRMM and electric NRMMs shall be adopted as far as practicable. Power supply for the on-site machinery shall be provided as far as practicable.
- 2.2.24 The *Air Pollution Control (Fuel Restriction) Regulations,* in particular not using restricted liquid fuel shall be fully complied with to minimise any aerial emissions.



- 2.2.25 In addition, the EPD's *Recommended Pollution Control Clause ("RPCC") for Construction Contract* in COP should be incorporated in the relevant works contract. The RPCC are generally good engineering practice to minimize inconvenience and environmental nuisance to nearby residents and other sensitive receivers. The general requirements are summarised below:
  - The Contractor shall observe and comply with the APCO and is subsidiary regulations, particularly the Air Pollution Control (Open Burning) Regulation and Air Pollution Control (Construction Dust) Regulation and Air Pollution Control (Smoke) Regulation.
  - The Contractor shall undertake at all times to prevent dust nuisance and smoke as a result of his activities.
  - The Contractor shall ensure that there will be adequate water supply / storage for dust suppression.
  - The Contractor shall devise, arrange methods of working and carrying out the works in such a manner as to minimise dust impacts on the surrounding environment, and shall provide experienced personnel with suitable training to ensure that these methods are implemented.
  - For better smoke control, the Contractor shall not use diesel hammer for percussive piling.
  - Before commencement of any work, the Engineer may require the methods of working, plant, equipment and air pollution control system to be used on the site to be made available for inspection and approval to ensure that they are suitable for the project.
- 2.2.26 No site formation will be carried out for the proposed CBP. Besides, the components/ structures of the proposed CBP will be prefabricated. Considering limited construction scale and dust release, EM&A shall not be necessary.

#### 2.3 Operation Phase Impact

#### **Identification of Air Pollution Sources**

2.3.1 The works and equipment of the proposed CBP for concrete production will be driven by the mains electricity supply. Therefore, particulate matters, i.e., Respirable Suspended Particulates ("RSP" or "PM<sub>10</sub>") and Fine Suspended Particulates ("FSP" or "PM<sub>2.5</sub>") will be the major air pollution concerns during the operation of the CBP. Particulate Matters ("PM") including RSP and FSP are the major pollutants that will be generated during the operating activities of the CBP including transferring and handling of dusty materials, fugitive dust emissions from vehicle movements and dust emissions from Dust Collectors ("DCs"), as well as vehicular emissions from open road networks within the 500m Study Area. On the other hand, Nitrogen Dioxide ("NO<sub>2</sub>") will be emitted from tailpipe emissions during the vehicle travelling within the CBP and also from open road networks within 500m Study Area. The indicative Emission Points ("EPs") within the Site as listed in **Table 2-3** and illustrated on **Figure 2-1**Figure 2- will be the major sources of RSP, FSP and NO<sub>2</sub> during the operation of the CBP.



#### Table 2-3 Summary of Identified Emission Points within CBP during Operation

EMISSION POINTS	DESCRIPTION	POLLUTANT	ТҮРЕ	OPERATION HOURS
EP1-EP14	Dust Collectors (DCs) for silos/enclosed hoppers	RSP, FSP	Ducted	07:00 to 23:00
EP15 and EP16	Aggregate Stockpile	RSP, FSP	Fugitive	07:00 to 23:00
EP17 and EP18	Unloading of aggregate into aggregate receiving hopper	RSP, FSP	Fugitive	07:00 to 23:00
EP19	Pave Roads	RSP, FSP, NO <sub>2</sub>	Fugitive	07:00 to 23:00

Note: The operation hours in this table are indicative and the actual operation hours may be changed subject to the detailed design stage in the future.

#### **Emission Control Measures**

- 2.3.2 Since no diesel generator is needed during the operation of the CBP, there will be no emissions associated with combustion of Ultra-Low Sulphur Diesel ("ULSD"). The major potential emission sources during concrete production rate are fugitive dust emissions generated from loading, unloading, handling, transferring and storing activities. The major potential emission sources during concrete production rate are fugitive dust emissions generated from loading, unloading, handling, transferring and storing activities. The major potential emission sources during concrete production rate are fugitive dust emissions generated from loading, unloading, handling, transferring and storing activities.
- 2.3.3 The following mitigation measures summarised in **Table 2-4** will be implemented at each EP.

EP	DESCRIPTIONS	MITIGATION MEASURES
EP1-EP14	Dust Collectors (DCs)	<ul> <li>Dust collectors shall be installed at the silos and silos shall be equipped with high level alarms to warn of over filling. Seating of pressure relief valves of all silos shall be checked at least once a week. Dust accumulated shall be properly handled.</li> <li>During unloading of cement/PFA/GGBS/silica fume from tankers to the relevant silos via the pneumatic system, dust-laden air will be vented to the dust collector, in order to abate the dust emissions and comply with the BPM3/2(16) limit.</li> </ul>
		<ul> <li>3-side with top shall be enclosed and the front opening</li> </ul>

#### Table 2-4Emission Control Measures for EPs



EP	DESCRIPTIONS	MITIGATION MEASURES
		side of the Mixer Tower (EP13&EP14) shall be covered with flexible curtain. • The weight hopper shall be fully enclosed
EP15 and EP16	Aggregate Stockpile	<ul> <li>The aggregate stockpile areas with aggregate of sizes of</li> <li>5mm, 10mm and 20mm will be enclosed on 3-sides with rigid wall sufficiently higher than the top of the stockpile to prevent whipping. Water sprayings shall be provided regularly to suppress dust emissions.</li> <li>Stockpile areas with</li> </ul>
		aggregates of size ≤5mm should be covered on top, enclosed on 3-sides. Flexible curtain shall be installed in the opening and water sprayings are provided regularly in the stockpile areas and also at the openings to suppress dust emissions.
EP17 and EP18	Unloading of aggregate into aggregate receiving hopper	•The receiving hopper shall be enclosed on top and 3 sides, plastic curtain shall be installed at the front. In no case shall these hoppers be used for material storage. Water spraying device shall be operated during the unloading of aggregate/crushed fines/sand.
EP19	Paved Roads	<ul> <li>All haul roads within the Site shall be paved and adequately wetted.</li> <li>Vehicle cleaning facilities should be provided at the exit of the Site to wash off any dust/mud on the vehicles, including the wheels, prior to leaving the Site</li> </ul>

#### **Identification of Air Sensitive Receivers**

2.3.4 Representatives Air Sensitive Receivers ("ASR") located in the vicinity of the Site have been identified. Some of the structures which were vacant/abandoned houses identified during the site visit are not considered as ASRs.



In total, 14 representative ASRs have been identified within the 500m Study Area from the boundary of the CBP. Table 2-5 summarises the 14 ASRs and Figure 2-2 Figure 2- indicates their location.

ID	DESCRIPTION	STOREY	RECEPTOR HEIGHTS ABOVE GROUND LEVEL (mAG)	DISTANCE FROM THE CBP (m)	PATH GRID
ASR 1	Village House 1	2	1.5, 4.5	80	(33,46)
ASR 2	Auto Repair Shop	2	1.5, 4.5	20	(33,46)
ASR 3	Auto Repair Shop	2	1.5, 4.5	70	(33,46)
ASR 4	Warehouse	3	1.5, 4.5, 7.5	140	(33,46)
ASR 5	Warehouse	3	1.5, 4.5, 7.5	90	(33,46)
ASR 6	Corporate Office	2	1.5, 4.5	125	(33,46)
ASR 7	Village House 2	2	1.5, 4.5	130	(33,46)
ASR 8	Village House 3	1	1.5	170	(33,46)
ASR 9	Auto Repair Shop	2	1.5, 4.5	35	(33,46)
ASR 10	Village House 4	2	1.5, 4.5	215	(33,46)
ASR 11	Construction equipment rental	2	1.5, 4.5	200	(33,46)
ASR 12	Tire Shop	2	1.5, 4.5	205	(33,46)
ASR 13	Village House 5	3	1.5, 4.5, 7.5	320	(33,46)
ASR 14	Food Manufacturer Office	2	1.5, 4.5	180	(33,46)

#### Table 2-5Representative ASRs

2.3.6 According to Paragraph 3.3.10 of Chapter 9 Environment of HKPSG, at least 100 m buffer distance from the dusty uses including CBP should be required for the nearby air-sensitive uses. Among the 14 identified ASR, the distance of ASR1, ASR2, ASR3, ASR5, ASR9 is less than 100m from the CBP. Therefore, an assessment of cumulative air quality impact from the operation of CBP is conducted and provided in **Section 2.4**.

#### 2.4 Cumulative Air Quality Impact Assessment for Operation Phase

#### **Background Air Quality**

- 2.4.1 According to "Guidelines on Assessing the 'TOTAL' Air Quality Impacts", PATH pollutant concentrations as from EPD's Smart Air Modelling Platform ("SAMP") are used as background concentrations of the assessment.
- 2.4.2 The assessment area of 500m from the Plant covered PATH grids (32,46), (33,45) and (33,46). Year 2025 as downloaded from PATH v3.0 is adopted as the assessment year for predicted cumulative impact comparing against the prevailing AQOs.

#### **Air Dispersion Model**

2.4.3 A Gaussian dispersion model AERMOD was used to estimate pollutant concentrations at ASRs. The model was originally developed by the United States Environmental Protection



Agency ("USEPA") and is adopted for evaluating industrial chimney releases (point sources), area and volume sources as well as line sources (i.e. vehicle emission for open roads).

- 2.4.4 AERMET is a meteorological pre-processor developed by USEPA and is used for organising meteorological data into a format suitable for use by AERMOD. Site specific MET data has been downloaded from the Smart Air Modelling Platform ("SAMP"). Details are shown in **Appendix B**Appendix B.
- 2.4.5 The output from MET data consists of two parts; a file with extension ".sfc" is the surface air data; and a file with extension ".pfl" is the upper air data. Data including wind speed, wind direction and temperature in the surface air data from the output file in ".sfc" format were replaced by the original WRF data.

#### Vehicular Emissions from Open Roads

- 2.4.6 The predicted 24-hour traffic flow and vehicle compositions at the identified roads within the assessment area was provided by the traffic consultant for the assessment of the potential air quality impact from the open roads. Two years of traffic data, Year 2026 (the commencing year of the proposed CBP) and Year 2031 (five years after the commencing year, 2026 as five years valid period of SP licence) is provided by the project traffic consultant.
- 2.4.7 NO<sub>2</sub>, RSP and FSP are the key pollutants for vehicular emissions from open roads. Latest EMFAC-HK model as provided in the SAMP with "Zero Emission Vehicle Scenario" has been used to estimate the vehicular emission rates for NO, NO<sub>2</sub>, RSP and FSP to include the consideration of Zero Emission Vehicle ("ZEV") assumptions. The detailed input parameters for the EMFAC-HK model input are summarised in **Appendix C**Appendix C.
- 2.4.8 There is an identified bus terminus within the 500m study area, Sheung Tsuen Bus terminus. According to desktop study and site survey, no bus was observed to stop and switch off engine to stay at the terminus. Broad-bush approach was assumed in the assessment.
- 2.4.9 Traffic flows at Year 2031and the emission Year of Year 2026, the commencing year was adopted for conservative approach. For the estimation of long-term air quality impact of pollutants (annual average), the daily profile of averaged temperature and relative humidity data in each hour for each month (i.e., 24 hours data in each month and for 12 months) as derived from the EMFAC-HK model in the SAMP were adopted for the model input. For short-term air quality impact of pollutants (hourly or daily average), the daily profile of minimum temperature and relative humidity data in each hour for each month were adopted.

#### **Industrial Emissions**

- 2.4.10 The surrounding of the Site is mainly open storage and scrap car services. There is no other industrial emission identified within 500m assessment area. According to the "Guidelines on Assessing the 'TOTAL' Air Quality Impacts", there is no major industrial emission source within 4km is identified
- 2.4.11 The emission rates of the Plant have been summarised in Appendix A.



#### Particle Size Distribution

- 2.4.12 Particle size distributions are different for different types of emissions. Individual particle size distributions for each emission type have therefore been assumed as listed in Table
   2-6Table 2-6, below, in order to better reflect actual emissions conditions.
- 2.4.13 No particle size distribution was adopted for EP1 to EP14 (i.e. those relating to loading of cementitious materials with DCs installed) due to either the emission not being particulate (e.g. fuel combustion, etc.) or the emission sources not being fugitive.

TYPE OF EMISSION ACTIVITY	PARTICLE SZIE DISTRIBUTION			REFERENCE	APPLIED TO EMISSION SOURCES
Aggregate unloading	Mean Particle Size (μm)	RSP %age	FSP %age	Category 3 "Mechanically Generated Aggregate, Unprocessed Ores", Page B.2-13,	EP15, EP16, EP17 and EP18
	0.50	8%	27%	Appendix B.2 Generalized Particle	
	1.50	14%	47%	Size Distributions, AP-42, USEPA (Version 1/95)	
	2.25	8%	27%		
	2.75	6%			
	3.50	14%			
	4.50	10%			
	5.50	8%			
	8.00	33%			
		100%	100%		
Area sources for truck movements such as Paved Road	Mean Particle Size (μm)	RSP %age	FSP %age	Table 13.2.1-1 and page 13.2.1-12 of Section 13.2.1.3 of AP-42, USEPA	EP19
	1.25	24.2%	100%		
	6.25	75.8%			
		100%	100%		

 Table 2-6
 Particle Size Distribution for Different Emission Types

#### Ozone Limiting Method for Short-term Cumulative NO<sub>2</sub> Assessment

- 2.4.14 Ozone Limiting Method ("OLM") has been adopted for conversion of NO from vehiclerelated source and NO<sub>x</sub> from industrial emission sources to NO<sub>2</sub> based on the predicted O<sub>3</sub> level from PATH model as extracted from the SAMP for the short-term cumulative NO<sub>2</sub> assessment.
- 2.4.15 The initial NO<sub>2</sub>/NO<sub>x</sub> ratios for industrial emission has been assumed as 10% according to *"Technical support document (TSD) for NO<sub>2</sub>-related AERMOD modifications"* as published by USEPA. The predicted initial NO concentrations from open roads and 90% of the predicted NOx concentrations from industrial emissions was firstly added together on an hour-to-hour basis and OLM was applied subsequently. The NO<sub>2</sub>/NO<sub>x</sub> conversion has been calculated based on the equation below:

 $[NO2] predicted = [NO2] veh + 0.1 \times [NOx] ind$  $+ Min\{([NO] veh + 0.9 \times [NOx] ind) or (\frac{46}{48} \times [O3] PATH)\}$ 

Where,



[NO2]predicted = predicted NO2 concentration

[NO2]veh = predicted initial NO2 concentration from vehicular emissions

[NOx]ind = initial NOx concentration from industrial sources

[NO]veh = predicted initial NO concentration from vehicular emissions

Min = minimum of the two values in (brackets)

[O3]PATH = representative O3 PATH concentration

#### Jenkin Method for Long-term Cumulative NO<sub>2</sub> Assessment

2.4.16 Jenkin method was adopted for the conversion of cumulative annual average NO<sub>x</sub> to NO<sub>2</sub> by using the empirical relationship in observed annual mean of NO<sub>x</sub> and NO<sub>2</sub> concentrations with reference to the "*Guidance on Choice of Models and Model Parameters*". The empirical relationship is derived from the annual mean observed data by relevant EPD's Air Quality Monitoring Stations ("AQMS") including North (the closest station), the nearest roadside station (Mong Kok) and derived by the SAMP. The resulting curve was adopted for the cumulative annual average NO<sub>x</sub> to NO<sub>2</sub> conversion and the NO<sub>x</sub>to-NO<sub>2</sub> conversion equation using Jenkin method is presented in **Appendix C**Appendix C.

#### 2.5 Assessment Results

2.5.1 The cumulative air quality impact due to vehicular emissions, the Plant operation and background concentrations were evaluated and compared to the prevailing AQOs as described in **Section 2.1**. The results are summarised in **Table 2-7**Table 2-7Table 2-7 and **Table 2-8**Table 2-8. Comparison of new parameters of proposed 2025 AQOs are presented in **Appendix D**Appendix D for reference.



			HEIGHT		RSP, μg/m³		FSP, μg/m³	
ASR ID	DESCRIPTION	PATH GRID	(mPD)	(mAG)	10 <sup>th</sup> HIGHEST DAILY AVERAGE	ANNUAL AVERAGE	36 <sup>th</sup> HIGHEST DAILY AVERAGE	ANNUAL AVERAGE
ASR1	Village House 1	(33,46)	53.1	1.5	56.5	21.0	29.6	13.2
ASR1	Village House 1	(33,46)	56.1	4.5	56.5	21.0	29.6	13.1
ASR2	Auto Repair Shop 1	(33,46)	51.8	1.5	56.6	21.3	29.6	13.3
ASR2	Auto Repair Shop 1	(33,46)	54.8	4.5	56.5	21.3	29.6	13.2
ASR3	Auto Repair Shop 2	(33,46)	54.3	1.5	56.5	21.0	29.5	13.1
ASR3	Auto Repair Shop 2	(33,46)	57.3	4.5	56.4	20.9	29.5	13.1
ASR4	Warehouse 1	(33,46)	53.1	1.5	56.4	20.9	29.5	13.1
ASR4	Warehouse 1	(33,46)	56.1	4.5	56.4	20.9	29.5	13.1
ASR4	Warehouse 1	(33,46)	59.1	7.5	56.4	20.9	29.5	13.1
ASR5	Warehouse 2	(33,46)	56.2	1.5	56.4	20.9	29.5	13.1
ASR5	Warehouse 2	(33,46)	59.2	4.5	56.4	20.9	29.5	13.1
ASR5	Warehouse 2	(33,46)	62.2	7.5	56.4	20.9	29.5	13.1
ASR6	Corporate Office	(33,46)	62.7	1.5	56.4	20.9	29.5	13.1
ASR6	Corporate Office	(33,46)	65.7	4.5	56.4	20.9	29.5	13.1
ASR7	Village House 2	(33,46)	68.5	1.5	56.4	20.9	29.5	13.1
ASR7	Village House 2	(33,46)	71.5	4.5	56.4	20.9	29.5	13.1
ASR8	Village House 3	(33,46)	83.4	1.5	56.4	20.9	29.5	13.1
ASR9	Auto Repair Shop 3	(33,46)	58.6	1.5	57.9	23.2	29.6	13.8
ASR9	Auto Repair Shop 3	(33,46)	61.6	4.5	57.1	22.2	29.6	13.7
ASR10	Village House 4	(33,46)	47	1.5	57.3	21.5	30.2	13.6
ASR10	Village House 4	(33,46)	50	4.5	57.2	21.4	30.0	13.6
ASR11	Construction equipment rental	(33,46)	44.6	1.5	56.6	21.2	29.6	13.3



			HEIGHT		RSP, μg/m <sup>3</sup>		FSP, μg/m³	
ASR ID	DESCRIPTION	PATH GRID	(mPD)	(mAG)	10 <sup>th</sup> HIGHEST DAILY AVERAGE	ANNUAL AVERAGE	36 <sup>th</sup> HIGHEST DAILY AVERAGE	ANNUAL AVERAGE
ASR11	Construction equipment rental	(33,46)	47.6	4.5	56.6	21.1	29.6	13.3
ASR12	Tire Shop	(33,46)	46.2	1.5	56.4	20.9	29.5	13.1
ASR12	Tire Shop	(33,46)	49.2	4.5	56.4	20.9	29.5	13.1
ASR13	Village House 5	(33,46)	43.8	1.5	56.6	21.0	29.5	13.2
ASR13	Village House 5	(33,46)	46.8	4.5	56.6	21.0	29.5	13.2
ASR13	Village House 5	(33,46)	49.8	7.5	56.6	21.0	29.5	13.2
ASR14	Food Manufacturer Office	(33,46)	59.5	1.5	56.6	21.0	29.6	13.2
ASR14	Food Manufacturer Office	(33,46)	62.5	4.5	56.5	21.0	29.6	13.2
	Prevailing AQOs					50	50	25

Table 2-8	Predicted NO <sub>2</sub> Concentrations a	it Representative ASRs	(2024 AQOs)
-----------	--	------------------------	-------------

		DATU			NO₂, μg/m³		
ID	DESCRIPTION	GRID	(mPD)	(mAG)	19 <sup>th</sup> HIGHEST HOURLY AVERAGE	ANNUAL AVERAGE	
ASR1	Village House 1	(33,46)	53.1	1.5	55.3	11.2	
ASR1	Village House 1	(33,46)	56.1	4.5	54.2	10.9	
ASR2	Auto Repair Shop 1	(33,46)	51.8	1.5	58.1	10.8	
ASR2	Auto Repair Shop 1	(33,46)	54.8	4.5	56.0	10.8	
ASR3	Auto Repair Shop 2	(33,46)	54.3	1.5	53.3	10.9	
ASR3	Auto Repair Shop 2	(33,46)	57.3	4.5	52.7	10.6	
ASR4	Warehouse 1	(33,46)	53.1	1.5	53.1	10.4	
ASR4	Warehouse 1	(33,46)	56.1	4.5	53.1	10.4	
ASR4	Warehouse 1	(33,46)	59.1	7.5	52.4	10.4	
ASR5	Warehouse 2	(33,46)	56.2	1.5	51.9	10.3	
ASR5	Warehouse 2	(33,46)	59.2	4.5	51.9	10.3	



		PATH GRID	HEIGHT, (mPD)	HEIGHT (mAG)	NO₂, μg/m³		
ID	DESCRIPTION				19 <sup>th</sup> HIGHEST HOURLY AVERAGE	ANNUAL AVERAGE	
ASR5	Warehouse 2	(33,46)	62.2	7.5	50.4	10.3	
ASR6	Corporate Office	(33,46)	62.7	1.5	49.4	10.2	
ASR6	Corporate Office	(33,46)	65.7	4.5	49.4	10.2	
ASR7	Village House 2	(33,46)	68.5	1.5	49.0	10.2	
ASR7	Village House 2	(33,46)	71.5	4.5	49.0	10.1	
ASR8	Village House 3	(33,46)	83.4	1.5	49.0	10.1	
ASR9	Auto Repair Shop 3	(33,46)	58.6	1.5	64.5	11.9	
ASR9	Auto Repair Shop 3	(33,46)	61.6	4.5	59.9	11.4	
ASR10	Village House 4	(33,46)	47	1.5	168.3	27.1	
ASR10	Village House 4	(33,46)	50	4.5	156.4	24.4	
ASR11	Construction equipment rental	(33,46)	44.6	1.5	98.6	15.0	
ASR 11	Construction equipment rental	(33,46)	47.6	4.5	93.4	14.3	
ASR 12	Tire Shop	(33,46)	46.2	1.5	59.0	10.9	
ASR 12	Tire Shop	(33,46)	49.2	4.5	59.1	10.9	
ASR 13	Village House 5	(33,46)	43.8	1.5	102.5	15.3	
ASR 13	Village House 5	(33,46)	46.8	4.5	100.9	14.9	
ASR 13	Village House 5	(33,46)	49.8	7.5	85.6	13.9	
ASR 14	Food Manufacturer Office	(33,46)	59.5	1.5	65.1	11.5	
ASR 14	Food Manufacturer Office	(33,46)	62.5	4.5	62.6	11.4	
		200	40				



2.5.2 No exceedance of the pollutant concentrations of prevailing AQOs and proposed 2025 AQOs at representative ASRs is predicted.

#### 2.6 Conclusion

- 2.6.1 With the implementation of the recommended mitigation measures and good site practice, adverse air quality impact during the construction phase is not anticipated.
- 2.6.2 For the operation phase, with implementation of mitigation measures and predicted assessment results, the operation of the Proposed CBP will not pose any unacceptable air quality impact.
- 2.6.3 Overall, no adverse air quality impacts are anticipated during the construction and operation phases of the Proposed CBP.



#### *Figure 2-1 Locations of Emission Points of the Plant*



ESC Project J24.00171.HK.01 | D01 | Rev 1


#### Figure 2-2 Locations of ASRs





### 3 NOISE

#### 3.1 Environmental Legislation and Standards

#### Noise Control Ordinance ("NCO", Cap. 400)

- 3.1.1 The principal legislation controlling environmental noise impact is the *Noise Control Ordinance* ("NCO"). The NCO enables regulations and Technical Memoranda ("TMs") to be enacted, which introduces detailed control criteria, measurement procedures and other technical matters. The TMs of NCO include:
  - TM on Noise from Percussive Piling ("PP-TM")
  - TM on Noise from Construction Work other than Percussive Piling ("GW-TM")
  - TM on Noise from Construction Work in Designated Area ("DA-TM")
  - TM for the Assessment of Noise from Places Other Than Domestic Premises, Public Places or Construction Sites ("IND-TM")
- 3.1.2 The Site falls within a Designated Area ("DA") in accordance with EPD's Plan No. EPD/AN/NT-01 for Yuen Long, Tin Shui Wai, Mai Po, Shek Kong and Kwu Tung. Therefore, DA-TM is applicable.
- 3.1.3 In addition, the following requirements are given under the NCO:
  - Hand-held breakers having a mass of above 10kg and any air compressor capable of supplying compressed air at 500kPa or above must be fitted with Noise Emission Label issued under the *Noise Control (Hand Held Percussive Breakers) Regulation and Noise Control (Air Compressors) Regulation* of NCO.
  - Construction Noise Permit ("CNP") must be applied by the Contractor from EPD for any
    percussive piling at any time or any other construction activities conducted within
    restricted hours (for all days 7pm to 7am the next day and at all times on Public
    Holidays or Sundays) as defined in NCO.
- 3.1.4 For noise arising from construction activities (other than percussive piling) during normal working hours (7am to 7pm from Monday to Saturday, not including general holidays), the noise criteria as shown in **Table 3-1** and control measures for construction noise impact during normal working hours can be referred to *Professional Persons Environmental Consultative Committee Practice Note PN 1/24 Minimizing Noise from Construction Activities* ("ProPECC PN1/24").

NOISE SENSITIVE USE	L <sub>eg (30min)</sub> NOISE CRITERIA BETWEEN 0700 AND 1900 ON ANY DAY NOT BEING A SUNDAY OR GENERAL HOLIDAY
All domestic premises	75 dB(A)
Temporary housing accommodation	
Hostels	
Convalescences homes	
Homes for the aged	

#### Table 3-1Construction Noise Criteria for Non-Restricted Hours



NOISE SENSITIVE USE	Leg (30min) NOISE CRITERIA BETWEEN 0700 AND 1900 ON ANY DAY NOT BEING A SUNDAY OR GENERAL HOLIDAY
Places of public worship Courts of law Hospitals and medical clinics	70 dB(A)
Educational institutions (including kindergartens and nurseries)	70 dB(A) (or 65 dB(A) during examination)

3.1.5 For fixed plant noise during operation phase, the requirements of IND-TM shall be complied with. Table 2 of IND-TM stipulates the day, evening and night time Acceptable Noise Levels ("ANLs") for Noise Sensitive Receivers ("NSRs") according to the corresponding Area Sensitive Rating ("ASR"), which is determined by Influencing Factors ("IFs") in accordance with the IND-TM. These are summarised in **Table 3-2**.

#### Table 3-2Acceptable Noise Levels for Fixed Noise Source

	ANL, dB(A)				
TIME PERIOD	ASR "A"	ASR "B"	ASR "C"		
Day (0700 to 1900 hours)	60	65	70		
Evening (1900 to 2300 hours)	00	60			
Night (2300 to 0700 hours)	50	55	60		

#### Hong Kong Planning Standards & Guidelines ("HKPSG")

- 3.1.6 The noise criteria for planned fixed source shall follow the requirements of Table 4.1 of Chapter 9 of HKPSG:
  - 5 dB(A) below the appropriate ANLs shown in Table 2 of IND-TM, and
  - The prevailing background noise levels
- 3.1.7 As recommended in Table 4.1 of Chapter 9 Environment of HKPSG, standards for road traffic noise in terms of  $L_{10(1-hr)}$  for the following uses relying on opened windows for ventilation are shown in **Table 3-3**.

#### Table 3-3 HKPSG Standards for Road Traffic Noise Standards

USES	NOISE STANDARDS L10(1-Hr), dB(A)
All domestic premises including temporary housing accommodation	70
Hotels and hostels	70
Offices	70
Educational institutions including kindergartens, child care centres and all others where unaided voice communication is required	65
Places of public worship and courts of law	55
Hospitals, clinics, convalescences and residential care homes for the elderly, - diagnostic rooms, - wards	55



#### **3.2** Noise Impact of Construction Phase

- 3.2.1 Sources of noise during construction and demolition will be arising from various construction and demolition activities, particularly with the use of Powered Mechanical Equipment ("PME"). The construction and demolition works will be carried out during non-restricted hours, i.e. daytime between 07:00 and 19:00 of any day not being Sunday or general holiday. As such, construction noise will be generated during non-restricted hours only.
- 3.2.2 Given the mitigation measures described in **Paragraph 3.2.3** would be implemented as far as practicable, the noise generated from the construction of the Project is not anticipated to pose any unacceptable noise impacts on the NSRs nearby.

#### Mitigation Measures

- 3.2.3 Construction should be carried out during non-restricted hours as far as practicable. The mitigation measures recommended in ProPECC PN 1/24 should be implemented where applicable. In addition, the following measures and on-site practices are recommended in order to minimise the potential construction noise impacts as far as practicable:
  - The Contractor shall devise, arrange methods of working and carry out the Works in such a manner so as to minimise noise impacts on the surrounding environment, and shall provide experienced personnel with suitable training to ensure that these methods are implemented
  - Quality Powered Mechanical Equipment (QPME) and quieter construction methods should be adopted as far as practicable
  - Use of Non-percussive pile driving methods such as hydraulic press-in method, vibration or jacking method for installing or extracting sheet piles as far as practicable
  - Use of Non-percussive equipment such as hydraulic crusher, sawing, coring machines etc. for demolition and concrete breaking work
  - Close all hoods, cover panels and inspection hatches of powered mechanical plant such as generators, air compressors etc. during operation
  - Provide noise dampening materials inside and outside refuse chutes during building construction
  - Fit mufflers or silencers, and dampening layer with steel collars to hand-held pneumatic breakers
  - Use of non-explosive chemical expansion agents instead of explosive chemicals or expansive compounds
  - Use of prefabricated structure / sections to replace in-situ construction to reduce the amount of mechanical equipment used on site
  - Use of self-compacting concrete (without the aid of a vibrator e.g. poker for compaction) for in-situ concreting
  - Noisy equipment and noisy activities should be located as far away from the NSRs as far as practicable
  - Provide an acoustic screen or enclosure shield the public or NSR from the noisy activities



- 3.2.4 If PME is required for any construction work during restricted hours, a CNP shall be applied for as specified in the NCO. The GW-TM can be referred to for the noise criteria and assessment procedures for obtaining a CNP.
- 3.2.5 In addition, the EPD's Recommended Pollution Control Clauses ("RPCC") for Construction Contracts should be incorporated in the relevant works contract. The RPCC are generally good engineering practices to minimize inconvenience and environmental nuisance to nearby residents and other sensitive receivers. The general requirements as summarised as follows:
  - The Contractor shall observe and comply with the NCO and its subsidiary regulation.
  - The Contractor shall ensure that all plant and equipment to be used on the Site are properly maintained in good operating condition and noisy construction activities shall be effectively sound-reduced by means of silencers, mufflers, acoustic linings and shields, acoustic sheds or screen or other means, to avoid disturbance to nearby noise sensitive receivers.
  - For carrying out any construction work other than percussive piling during the time period from 0700 to 1900 hours on any day not being a general holiday (including Sundays), the Contractor shall comply with the following requirements.
    - The noise level measured at 1m from most affected external façade of the nearby noise sensitive receivers from the construction works alone during any 30-minute period shall not exceed an equivalent sound level ("Leq") of 75dB(A).
    - Should the limits stated in the above be exceeded, the construction shall stop and shall not recommence until appropriate measures acceptable to the Engineer that are necessary for compliance have been implemented.
    - The Contractor shall adopt, where necessary and practicable, the use of quieter construction equipment and/or methods when carrying out the construction works, including demolition works, foundation works, site formation works, road opening works during restricted hours.
  - Before commencement of any work, the Engineer may require the methods of working, plant equipment and sound-reducing measures to be used on the Site to be made available for trial demonstration inspection and approval to ensure that they are suitable for the project.
  - The Contractor shall devise, arrange methods of working and carry out the Works in such a manner so as to minimise noise impacts on the surrounding environment, and shall provide experienced personnel with suitable training to ensure that these methods are implemented.
  - The Contractor shall, when necessary, apply for a construction noise permit in accordance with the Noise Control (General) Regulations prior to the commencement of the relevant part(s) of the works, display the permit as required and provide a copy to the Engineer.
  - Measures that are to be taken to protect adjacent noise sensitive receivers, if necessary, shall include, but not be limited to, adequate noise barriers. The barriers shall be of substantial construction and designed to reduce transmission of noise. The location and details of the barriers shall be submitted to the Engineer for approval before works commence adjacent to NSRs.



3.2.6 With the implementation of the aforementioned mitigation measures, adverse construction noise impact is not anticipated.

#### 3.3 Noise Impact of Operation Phase

#### Identification of Noise Sensitive Receivers

3.3.1 For the on-site operation noise (i.e. industrial noise), the first layer Noise Sensitive Receivers ("NSRs") within 300m from the site boundary were identified. All identified NSRs are existing NSRs and no planned NSR has been identified within the assessment area. NSRs located within industrial zone, so it is considered as ASR "C". The representative NSRs are summarised in **Table 3-4Table 3-4** below and shown in **Figure 3-1**.

NSR NO.	DESCRIPTION	NO. OF STOREYS	DISTANCE FROM THE SITE, m	ASR
IN1	Village House 1, unknown	1	84	С
IN2	Village House 2, unknown	2	143	С
IN3	Village House 3, unknown	1	169	С
IN4	Village House, 25B Kam Tin Road	1	214	С

#### Table 3-4 Representative NSRs

3.3.2 The ASR and ANLs adopted in this EA report are used for assessment purpose only, they should not bind the Noise Control Authority's decision in determining the noise criteria based on the legislation and practices being in force, and contemporary conditions/ situations of adjoining land uses.

#### Prevailing Background Noise

- 3.3.3 Background noise monitoring at the sensitive façade where is representative to the identified NSRs is the most preferable location. The Site is mainly surrounded by private lots (i.e. located between the Site and the identified NSRs). However, the landowner/ occupants refused to allow for background noise measurement. As such, background noise monitoring locations near the representative NSRs were adopted. The location of the background noise monitoring locations is shown in **Figure 3-1**.
- 3.3.4 Locations BN1, BN2 and BN3 have been selected to conduct the background noise monitoring throughout the entire proposed operation hours (i.e. 0700 to 1900) during the weekday under free-field condition. BN1 is located at the roadside of a village access road to the northwest of IN2 and southwest of IN3. The major noise affecting BN1 is the village access road and the open-air parking area which is considered to be comparable to that at IN2 and IN3. BN2 is located at the roadside of an access road to the south of IN1. The major noise source of BN2 is the access road, which has similar environs to that of IN1. BN3 is located at the roadside of Kam Tin Road to the southeast of IN4, which is near the entrance of IN4.
- 3.3.5 A set of minimum L<sub>90(1hour)</sub> has been adopted as prevailing background noise level. The monitoring data is presented in **Appendix E**.
- 3.3.6 With reference to the Technical Memorandum ("TM") issued under the *Noise Control Ordinance* ("NCO"), sound level meters in compliance with the International



Electrotechnical Commission Publications 651: 1979 (Type 1) and 804: 1985 specifications were used for carrying out the noise monitoring. Immediately prior to and following each noise measurement, the accuracy of the sound level meter was checked using an acoustic calibrator generating a known sound pressure level at a known frequency. Measurements were considered acceptable and valid as the calibration level from before and after the noise measurement agrees to within 1.0dB.

3.3.7 Noise measurements were made in accordance with standard acoustical principles and practices in relation to weather conditions. Calibration certificates of the noise monitoring equipment used are provided in **Appendix F**.

#### Noise Criteria

3.3.8 As discussed in **paragraph 3.3.1**, the ASR of all identified NSRs is "C". The noise criteria for the planned fixed noise source were determined with reference to ANL-5 and the measured background noise level, as shown in **Table 3-5**.

Table 3-5	Noise Criteria of Nois	e from Fixed Sources	of Identified Repr	esentative NSRs
		2	, , , ,	

NSR NO.	DESCRIPTION	TIME PERIOD	MEASURED BACKGROUND NOISE LEVEL L90(1HOUR), dB(A)	ANL - 5, dB(A)	NOISE CRITERIA, dB(A)
IN1	Village House 1, unknown		46	65	46
IN2	Village House 2, unknown		48	65	48
IN3	Village House 3, unknown	Day	48	65	48
IN4	Village House, 25B Kam Tin Road		52	65	52

Note: The background noise level was measured at free-field condition. Thus, a façade correction of +3 dB(A) was applied.

#### **Potential Noise Impacts**

3.3.9 The plant will be in operation between 0700 and 2300 per day. Due to site constraints, unloading of aggregates/cement/PFA/GGBS/admixture will not be carried out at the same time with the process of concrete mixing and collection. As such, the major fixed noise sources listed below have been considered under two scenarios in daytime (i.e. 0700 – 1900) and evening (i.e. 1900 – 2300). The proposed noise sources were shown in Figure 3-2.

#### Day time Operation

<u>Scenario 1 – Concrete mixing and collection:</u>

- Concrete Mixer (electric) (2 nos./30 mins; % on time: 100.0%)
- Filter Fan (Mixing unit) Blower (2 nos./30 mins; % on time: 100.0%)



- Concrete Truck (Concrete Collection) (10 nos./30 mins; % on time: 10%)
- Conveyor / Screw Conveyors (3 nos./30 mins; % on time: 100.0%)
- Water / Admixture Pump (1 no./30 mins; % on time: 100.0%)
- Pump (Silo) (4 no./30 mins; % on time: 100.0%)

#### Scenario 2 – Unloading of aggregates/cement/PFA/GGBS/admixture:

- Conveyor / Screw Conveyors (3 nos./30 mins; % on time: 100.0%)
- Water / Admixture Pump (1 no./30 mins; % on time: 100.0%)
- Pump (mounted on unloading tank) (4 nos./30 mins; % on time: 100.0%)
- Filter Fan (Silo) Blower (4 nos./30 mins; % on time: 100.0%)
- Aggregate Truck (7 nos./30 mins)
- Cement/PFA/GGBS/Admixture Tanker (2 nos./30 mins)

#### **Evening Operation**

#### Scenario 1 – Concrete mixing and collection:

- Concrete Mixer (electric) (1 nos./30 mins; % on time: 100.0%)
- Filter Fan (Mixing unit) Blower (1 nos./30 mins; % on time: 100.0%)
- Concrete Truck (Concrete Collection) (2 nos./30 mins; % on time: 10%)
- Conveyor / Screw Conveyors (3 nos./30 mins; % on time: 100.0%)
- Water / Admixture Pump (1 no./30 mins; % on time: 100.0%)
- Pump (Silo) (2 no./30 mins; % on time: 100.0%)

#### Scenario 2 – Unloading of aggregates/cement/PFA/GGBS/admixture:

- Conveyor / Screw Conveyors (3 nos./30 mins; % on time: 100.0%)
- Water / Admixture Pump (1 no./30 mins; % on time: 100.0%)
- Pump (mounted on unloading tank) (2 nos./30 mins; % on time: 100.0%)
- Filter Fan (Silo) Blower (2 nos./30 mins; % on time: 100.0%)
- Aggregate Truck (5 nos./30 mins)
- Cement/PFA/GGBS/Admixture Tanker (1 nos./30 mins)
- 3.3.10 The lists of equipment, utilisation rate and number of items detailed in **Appendix G** have been confirmed by the Applicant and it is considered to be practicable for business-asusual operation of CBP. Loader will be used only if emergency case to transport aggregate from storage area to hopper. Vehicle washing facility will be applied on site during operation, however, the operation time of this facility is rather short each time (i.e. approximately 1 minutes for each vehicle) and its function is not continuous. Aggregate will be stored in aggregate storage areas, which is served as an emergency use. No action will be taken on the aggregates storage except whenever there are insufficient aggregates provided for concrete production. As a result, insignificant noise will be generated due to the occasionally use of the aggregate storage. In conclusion, noise impacts that may be



generated from aggregate storage, vehicle washing and parking are considered insignificant and negligible.

3.3.11 The number of concrete lorry mixers was assumed based on the normal operation practice limitation, that approximate 3 minutes were needed for filling up 1 concrete lorry mixer with newly produced concrete, it takes about 6 minutes for the whole process, including the travel in and out of the site. Thus, the 5 concrete lorry mixers per mixer per 30 minutes per production line as the maximum allowed traffic flow were assumed based on the previous experiences of CBP operator under realistic condition. As mentioned in **Section 0** above, the maximum of 10 concrete lorry mixers per mixer per hour for the maximum concrete production rate of 100m<sup>3</sup> per hour is practical in terms of business operation in the daytime which has been confirmed by the Applicant. Moreover, there will be no queuing of trucks due to the Proposed Development as reported in the Traffic Impact Assessment ("TIA") Report attached to the Planning Statement for this Planning Application.

#### 3.4 Noise Impact from Industrial Noise

#### **Assessment Methodology**

3.4.1 Although the proposed CBP is an industrial use and the noise impact therefore references IND-TM, the characteristics of concrete batching, involving concrete lorry mixer and concrete mixer, are also similar to construction activities. Therefore, the noise impact is proposed to be estimated in accordance with the guidelines given in GW-TM. As such, all items of PME, except truck movement, have been considered at the proposed positions. While due to the long travelling distance of conveyor/ screw conveyor, as a conservative approach, the closest distance between conveyor/ screw conveyor and each NSRs has be adopted and assessed based on the following standard acoustic formula:

SPL = SWL - DC + FC

where:

SPL – Sound Pressure Level at receiver, dB(A)

SWL - Sound Power Level of PME, dB(A)

 $DC - Distance Correction, dB(A) (DC = 20 \times log10(D) + 8)$ 

D – Horizontal distance between NSR and notional source, m

FC – Façade Correction of +3 dB(A)

- 3.4.2 Sound Power Levels (SWLs) of PME are obtained from Table 3 of GW-TM and from Sound Power Levels of Other Commonly Used PME available on EPD's website. Reference is also made to the British Standard 5228 Code of Practice for Noise and Vibration Control on Construction and Open Site Part 1: Noise amended in February 2014 (BS 5228-1:2009+A1:2014). The "Aggregates Truck" to be adopted during the operation stage is lorry, which will be used to transport raw materials to the Site. Its weight is between 5.5 tonne and 38 tonnes, thus, the sound power level of 105dB(A) with reference to EPD's "Sound power levels of other commonly used PME" was adopted in the calculation.
- 3.4.3 There will be no tonal noise/ impulsive noise occurring within the CBP and there will be no intermittent noise because of no piling, concrete breaking or similar process. As such, no corrections for tonality, impulsiveness and intermittency are considered to be required.



- 3.4.4 With regard to the screening effect, a 10 dB(A) reduction was adopted for NSRs without direct line-of-sight to the PME whereas a 5 dB(A) reduction was adopted for NSRs without direct line-of-sight only to the "noisy" part of the PME. For IN1, there is a more than 3m height solid wall surround the existing open storage area which located in between the IN1 (1 storey) and the proposed site. For IN4, there are several existing structures higher than 3m located between IN4 and the proposed site. Therefore, there is no direct line of sight between IN1, IN4 and the entire site.
- 3.4.5 During the operation of the CBP, Cement/ PFA/ GGBS/ Admixture tanker and aggregate truck will not enter the Site at the same time with concrete truck during the day time and evening.

#### Assessment Results

3.4.6 The predicted noise impact results are summarised in Table 3-6 and detailed in AppendixG. The assessment results of industrial noise ranged from 39 dB(A) to 46 dB(A) at both scenarios during daytime and evening operation.

Table 3-6Predicted Fixed Noise Impact Level at Identified Representative NSRs During<br/>Operation Phase

NSR NO.	DESCRIPTION	TIME PERIOD	NOISE CRITERIA (DAY/EVENING)	PREDICTED NOISE LEVEL IN Leq(30min), dB(A) (DAY/EVENING)	EXCEEDANCE
SCENARIO	1				
IN1	Village House 1, unknown		46/40	46/40	No
IN2	Village House 2, unknown		48/47	45/39	No
IN3	Village House 3, unknown	Day/Evening	48/47	45/39	No
IN4	Village House, 25B Kam Tin Road		52/49	39/34	No
SCENARIO	2				
IN1	Village House 1, unknown		46/40	42/40	No
IN2	Village House 2, unknown		48/47	40/40	No
IN3	Village House 3, unknown	Day/Evening	48/47	40/39	No
IN4	Village House, 25B Kam Tin Road		52/49	39/36	No

#### **Mitigation Measures**

3.4.7 The design of the proposed CBP should ensure that the potential industrial noise impacts at the NSRs shall comply with the relevant criteria. Measures as summarised in **Table 3-7** 



will be taken into account in the plant design to avoid noisy activities during the operation of the CBP. As such, unacceptable industrial noise impact is not anticipated.

#### Table 3-7Proposed Operation Phase Noise Mitigation Measures

NOISE SOURCE	PROPOSED MITIGATION MEASURES
Concrete mixer (electric)	Enclosure (minimum surface density of 10kg/m <sup>2</sup> )
Conveyor / Screw Conveyors	Enclosure (minimum surface density of 10kg/m <sup>2</sup> ) Semi-enclosure (minimum surface density of 10kg/m <sup>2</sup> )
Water / Admixture Pump	Semi-enclosure (minimum surface density of 10kg/m <sup>2</sup> )
Pump (silo)	Enclosure (minimum surface density of 10kg/m <sup>2</sup> )
Filter Fan (mixing unit) - blower	Enclosure (minimum surface density of 10kg/m <sup>2</sup> )
Filter Fan (silo) - blower	Semi-enclosure (minimum surface density of 10kg/m <sup>2</sup> )
Pump (mounted on unloading tank)	Enclosure (minimum surface density of 10kg/m <sup>2</sup> )
Truck (Concrete Collection)	Enclosure (minimum surface density of 10kg/m <sup>2</sup> ), with acoustic curtain installed at the entrance/exit

#### 3.5 Noise Impact from Off-Site Traffic Noise

#### **Assessment Methodology**

- 3.5.1 Upgrading works for Chun Yiu Road has been proposed for the operation of CBP. In addition, the operation of CBP would lead to an increase in traffic flow, which would potentially affect the existing representative NSRs in the vicinity of the Site.
- 3.5.2 The traffic noise impact assessment will be carried out in two scenarios, "Before" and "After" in order to compare the traffic noise level at the representative NSRs due to the additional traffic flow. As the proposed CBP is expected to operate in 2026 for a period of 5 years, traffic data in Year 2031 has been adopted in the assessment as the worst case scenario. Traffic forecasts provided by the Project Traffic Consultant were adopted to assess the off-site traffic noise impact. Detailed peak hour traffic forecasts for the assessment year of 2031 is provided in **Appendix H**.
- 3.5.3 Given that PM peak hour traffic flow would represent a more severe traffic condition than AM peak hour traffic flow, "Before" scenario has been compared with the "After" scenario under PM peak hour traffic flow to evaluate the traffic noise impact on the existing representative NSRs.

#### **Assessment Results**

3.5.4 The HKPSG assessment criteria for domestic premises is 70 dB(A). Besides, contribution of less than 1.0 dB(A) is considered insignificant in accordance with the *Road Traffic Noise Impact Assessment under the Environmental Impact Assessment Ordinance* ("GN 12/2023"). The predicted road traffic noise levels at each existing representative NSR under "Before" and "After" scenarios are summarised in **Table 3-89** and shown in **Appendix** I.

Table 3-89Predicted Traffic Noise Impact Level at Identified Representative NSRsDuring Operation Phase



			PREDICTED I IN L10(1	NOISE LEVEL hr), dB(A)		PROJECT
NSR NO.	DESCRIPTION	NOISE CRITERION	BEFORE	AFTER	EXCEEDANCE	CONTRIBUTION, dB(A)
IN1	Village House 1, unknown	70	67.8	67.9	No	0.1
IN2	Village House 2, unknown		54.1 – 54.2	54.4 – 54.5	No	0.3
IN3	Village House 3, unknown		52.6	53.2	No	0.6
IN4	Village House, 25B Kam Tin Road		68.7	69.5	No	0.8

3.5.5 Results show that the additional traffic flow generated from the operation of the proposed CBP would not cause significant contribution (i.e. more than 1.0 dB(A)) to the traffic noise level at the representative NSRs. No noise exceedances are expected at all representative NSRs. As such, adverse traffic noise impact arising from the operation of the proposed CBP is not anticipated.

#### 3.6 Conclusion

- 3.6.1 During the construction phase of the Proposed Development, with the implementation of the noise mitigation measures recommended in **Paragraph 3.2.3**, no adverse noise impact is anticipated.
- 3.6.2 Fixed plant noise is the major potential noise source during the operational phase. Quantitative assessment for the fixed plant noise sources were conducted. The predicted noise level during operation ranges from 39 dB(A) to 46 dB(A) at both Scenarios 1 and 2, which indicated that all the noise levels at NSRs would comply with the relevant noise criteria.
- 3.6.3 Traffic noise impact assessment has been carried out to identify the contribution arising from the additional traffic flow generated during the operation phase. Results show that no noise exceedances are expected at the representative NSRs and the additional traffic flow would not cause significant contribution to the traffic noise level. No adverse traffic noise impact is anticipated at the existing NSRs.
- 3.6.4 Overall, therefore, there will be no adverse noise impact during the construction and operation phases of the Proposed CBP.



#### *Figure 3-1 Locations of Representative NSRs and Background Noise Measurement*



#### J24.00171.HK.01

S.16 Planning Application for Proposed Temporary Concrete Batching Plant with Ancillary Facilities for a Period of 5 Years at Lots 573 RP and 1710 in D.D. 114, Shek Kong, Yuen Long Environmental Assessment



#### Figure 3-2 Locations of Proposed Noise Sources





### 4 WATER QUALITY

#### 4.1 Environmental Legislation, Standards and Guidelines

#### Water Pollution Control Ordinance (Cap. 358)

4.1.1 The Technical Memorandum – Standards for Effluent Discharged into Drainage and Sewerage Systems, Inland and Coastal Waters ("WPCO-TM") is issued under Section 21 of the Water Pollution Control Ordinance ("WPCO"). All discharges into government sewerage systems, marine and inland waters are required to comply with the standards stipulated in the WPCO-TM.

#### Construction Site Drainage, ProPECC PN2/23

4.1.2 With reference to *Professional Persons Environmental Consultative Committee ("ProPECC") Practice Note Construction Site Drainage* ("ProPECC PN2/23"), various guidelines for the handling and disposal of construction site discharges are included. The guidelines include the use of sediment traps, wheel washing facilities for vehicles leaving the Site, adequate maintenance of drainage systems to prevent flooding, overflow, sewage collection and treatment, and comprehensive waste management (collection, handling, transportation, and disposal) procedures.

#### Drainage Plans subject to Comment by the Environmental Protection Department, ProPECC PN1/23

4.1.3 With reference to ProPECC Practice Note Drainage Plan subject to Comment by the Environmental Protection Department – Building (Standards of Sanitary Fitments, Plumbing, Drainage Works and Latrines) Regulations ("ProPECC PN1/23"), various guidelines for the pollution control for discharge to storm drains and foul sewers, such as the use of grease trap for wastewater from the restaurant kitchen, the use of silt removal facilities for open surface channel led to stormwater drains, etc., are included. The guidelines also include the requirements for submission of drainage plans.

#### Protection of Natural Streams/Rivers from Adverse Impacts Arising from Construction Works, ETWB TCW No. 5/2005

4.1.4 Various procedures, guidelines and precautionary mitigation measures were circulated in *Protection of Natural Streams/Rivers from Adverse Impacts Arising from Construction Works* ("ETWB TCW No. 5/2005") to reinforce existing measures for protection of natural streams/rivers from construction works.

#### Hong Kong Planning Standards and Guidelines ("HKPSG")

4.1.5 HKPSG provides various water quality management guidelines for potentially polluting uses such as effluent-producing industries, sewage collection and disposal, as well as some guidelines to minimise water quality impacts on sensitive uses.



# 4.2 Identification of Water Control Zone ("WCZ") and Water Sensitive Receiver ("WSR")

4.2.1 The Site is situated in Deep Bay WCZ. With reference to Annex 14 of the *Technical Memorandum on Environmental Impact Assessment Process* ("EIAO-TM"), several inland water courses within 500m study area were identified as potential WSRs through desktop study and using topographic map of GeoInfo Map. The identified WSRs are listed in **Table 4-1 Table 4-1** and shown in **Figure 4-1Figure 4-1**. Potential water quality impacts during construction and operation phase are discussed below.

WSR ID	DESCRIPTION	ТҮРЕ	STATUS (ACTIVE/INACTIVE)	DISTANCE FROM THE SITE, m
W1	Watercourses to the south	Modified	Active	5
	of the Site	watercourse		
W2	Watercourse to the south	Modified	Active	200
	of the Site	watercourse		
W3	Watercourses to the east	Natural	Active	270
	of the Site	watercourse		
W4	Watercourse to the east of	Modified	Active	140
	the Site	watercourse		
W5	Watercourses to the east	Natural	Active	200
	of the Site	watercourse		
W6	Watercourses to the north	Natural	Active	160
	of the Site	watercourse		
W7	Watercourses to the north	Natural	Active	330
	of the Site	watercourse		
W8	Watercourses to the north	Natural	Active	360
	of the Site	watercourse		
W9	Watercourse to the west	Modified	Active	150
	of the Site	watercourse		

#### Table 4-1 Identified Water Sensitive Receivers

#### 4.3 Review of Water Quality Impact

#### **Construction Phase**

- 4.3.1 Muddy runoff from the Site may be generated during construction phase, especially during the rainy season.
- 4.3.2 Wash water from vehicles and equipment; silt from any on-site stockpiles of soil, cement and grouting materials; and spillage of fuels, oil and lubricants from construction/reinstatement vehicles and plant may generate water quality impacts. If these pollution sources are not properly controlled, it would lead to increased amounts of suspended solids, grease and oil, pH, Biochemical Oxygen Demand ("BOD"), etc. in the water system.
- 4.3.3 There is also the issue of sewage generated by construction workers on-site.
- 4.3.4 Most of the WSRs identified within 500m of the study area are located at an elevation higher than that of the Proposed Development, except the streams to the south of the Site. However, wastewater from construction of the Proposed CBP will be properly handled before discharge. Therefore, with implementation of the recommended mitigation



measures and good practices listed below, adverse water quality impacts from Proposed Development on the WSRs are not anticipated.

#### **Operation Phase**

- 4.3.5 Water sprinklers and wheel washing facilities will be installed for dust suppression. The water sprinklers and wheel washing facilities will be a major source of surface runoff as the entire site will be paved. The plant management should avoid excessive wetting of the ground to minimise generation of surface runoff. All the surface runoff will be collected by the peripheral drainage system and diverted to sedimentation tank or silt removal prior to on-site reuse.
- 4.3.6 Another major source of sewage/wastewater during operation phase would be sewage and grey water from toilets used by on-site staff. There will be around 12 no. of on-site staff and 100 no. of visitors/drivers during the daily operation of the Proposed CBP. The estimated ADWF from the Proposed Development is calculated to be 3.67 m<sup>3</sup>/day.
- 4.3.7 Sewage arising from the Proposed Development would be temporarily stored in sewage holding tanks and then tankered away for off-site disposal in a Sewage Treatment Facility. As advised by the Applicant, two (2) sewage holding tanks with the capacity of  $3m^3$  each will be provided on-site. With the total capacity of  $3.67m^3$ , the proposed sewage holding tanks is sufficient to cater the sewage generation. The details of sewage holding tanks will be further reviewed in the detailed design. Detailed sewage generation calculations have been provided in **Appendix J**.
- 4.3.8 As mentioned in **Paragraph 4.3.4**, most of the identified WSRs are located at a higher elevation than the Site, except the streams to the south of the Site. Nonetheless, with implementation of the recommended mitigation measures and good site practices in **Section 4.4**, adverse water quality impacts on the WSRs from the operation of the Proposed CBP are not anticipated.

#### 4.4 Mitigation Measures

#### **Construction Phase**

- 4.4.1 During construction phase, adequate capacity and number of portable toilets with adequate frequency for offsite disposal to be supplied, maintained and emptied by a licensed collector should be provided for construction workers.
- 4.4.2 The construction contractor shall follow good site practice and be responsible for providing, implementing and maintaining the mitigation measures as specified in ProPECC PN 2/23 for construction site drainage. The key requirements are as follows:
  - Surface run-off from construction sites should be discharged into storm drains via
    adequately designed sand/silt removal facilities such as sand traps, silt traps and
    sediment basins. Temporary construction drainage or earth bunds or sand bag barriers
    should be provided on site to properly direct storm water to such silt removal facilities.
    Perimeter channels at site boundaries should be provided where necessary to
    intercept storm run-off from outside the Site so that it will not wash across the Site.
    Catchpits and perimeter channels should be construction in advance of site formation
    works and earthworks.



- Silt removal facilities, channels and manholes should be maintained and the deposited silt and grit should be removed regularly, at the onset of and after each rainstorm to ensure that these facilities are functioning properly at all times.
- Construction works should be programmed to minimise soil excavation works in rainy reasons (generally from April to September). If soil excavation works could not be avoided in these months or at any time of year when rainstorms are likely, for the purpose of preventing soil erosion, temporarily exposed slope surfaces should be covered (e.g. by tarpaulin), and temporary access roads should be protected by crushed stone or gravel, as excavation proceeds. Intercepting channels should be provided (e.g. along the crest) to prevent storm runoff from washing across exposed soil surfaces. Arrangements should always be in place to ensure that adequate surface protection measures can be safely carried out well before the arrival of a rainstorm.
- Earthworks final surfaces should be well compacted and the subsequent permanent works or surface protection works should be carried out immediately after the final surfaces are formed to prevent erosion caused by rainstorms. Appropriate drainage like intercepting channels should be provided when necessary.
- Measures should be taken to minimise the ingress of rainwater into trenches. If
  excavation of trenches in wet seasons is necessary, they should be dug and backfilled
  in short sections. Rainwater pumped out from trenches should be discharged into
  storm drains via silt removal facilities.
- Open stockpiles of construction materials (e.g. aggregates, sand and fill material) on sites should be covered with tarpaulin or similar fabric during rainstorms. Measures should be taken to prevent the washing away of construction materials, soil, silt or debris into any drainage system.
- Manholes (including newly constructed ones) should always be adequately covered and temporarily sealed so as to prevent silt, construction materials or debris from getting into the drainage system, and to prevent storm run-off from getting into foul sewers. Discharge of surface run-off into foul sewers must always be prevented in order not to unduly overload the foul sewerage system.
- All vehicles and plant should be cleaned before they leave a construction site to ensure no earth, mud, debris and the like is deposited by them on roads. A wheel washing bay should be provided at every site exit if practicable and wash-water should have sand and silt settled out or removed before discharging into storm water drains. The section of construction road between the wheel washing bay and the public road should be paved to reduce vehicle tracking of soil and to prevent site run-off from entering public road drains.
- 4.4.3 In addition, the EPD's RPCC for Construction Contract should be incorporated in the relevant works contract. The RPCC are generally good engineering practice to minimise inconvenience and environmental nuisance to nearby residents and other sensitive receivers. The general requirements are summarised below:
  - The Contractor shall observe and comply with WPCO and its subsidiary regulation.
  - The Contractor shall carry out the Works in such as manner as to minimise adverse impacts on the water quality during execution of the works. In particular the Contractor shall arrange his method of working to minimise the effects on the water quality within and outside the Site, on the transport routes and at the loading, dredging and dumping areas.



- The Contractor shall follow the practices, and be responsible for the design, construction, operation and maintenance of all the mitigation measures as specified in the ProPECC PN 2/23 "Construction Site Drainage" issued by the Director of Environmental Protection. The design of the mitigation measures shall be submitted by the Contractor to the Engineer for approval.
- The Contractor shall not discharge directly or indirectly or cause or permit or suffer to be discharged into any public sewer, stormwater drain, channel, stream-course or sea any trade effluent or foul or contaminated water or cooling or hot water without the prior written consent of the Engineer in consultation with the Director of Environmental Protection and Director of Water Supplies, who may as a condition of granting his consent require to the Contractor to provide, operate and maintain at the Contractor's own expense to the satisfaction of the Engineer suitable works for the treatment and disposal of such trade effluent or foul or contaminated or cooling or hot water. The design of such treatment works shall be submitted to the Engineer for approval not less than one month before commencement of the relevant works.
- If any office, site canteen or site toilet facilities is/are erected, foul water effluent shall be directed to a foul sewer or to a sewage treatment and disposal facilities either directly or indirectly by means of pumping or other means approved by the Engineer.
- 4.4.4 All site discharges should be treated as necessary in accordance with the terms and conditions of the Discharge License.
- 4.4.5 With the implementation of the mitigation measures and good site practices, no adverse water quality impact during construction phase is anticipated.

#### **Operation Phase**

- 4.4.6 Wheel washing facilities will be provided at the site entrance. All vehicles will be cleaned properly before leaving the Site. Wastewater generated from the wheel washing facilities will also be collected and diverted to sedimentation tank for silt removal before being reused on-site.
- 4.4.7 Wastewater generated from the surface runoff due to the use of water sprinkler and wheel washing facilities will be first diverted to a sump pit for collection. After that, the wastewater will be diverted to a sedimentation tank for sedimentation and the slurry generated will be dewatered to produce cement cake. The cement cake arising from the treatment process is inert materials and will be reused in the Public Fill Reception Facilities. The design of the sedimentation tank, including the capacity, treatment methodology, etc., will be provided to EPD for approval during detailed design stage in accordance with *ProPECC PN1/23*.
- 4.4.8 Sewage from toilets will be temporarily stored in wastewater holding tanks and tankered away for off-site disposal at a Sewage Treatment Facility. As discussed in **Paragraph 4.3.7**, sufficient capacity and number of wastewater storage tanks for temporarily storing and tankering away for off-site disposal will be provided.
- 4.4.9 Hence, with provision of adequate number and capacity of sedimentation tank and mobile toilets, no adverse water quality impact during operation phase of the Proposed CBP is anticipated.



#### 4.5 Conclusion

- 4.5.1 During construction phase, portable toilets will be supplied for construction workers. With the implementation of the mitigation measures and good site practices mentioned in **Paragraph 4.4.2** and **4.4.3**, adverse water quality impacts from construction phase are not anticipated.
- 4.5.2 The Contractor shall apply for a Discharge License under the WPCO. All site discharges shall be treated in accordance with the terms and conditions of the Discharge License.
- 4.5.3 During operation phase, no adverse water quality impact is anticipated from wastewater/sewage from employees. Portable toilets will be available for site staff. Sewage generated from toilets will be temporarily stored in sewage holding tanks and tankered away for off-site disposal at a Sewage Treatment Facility. Besides, the wastewater generated from water sprinklers and wheel washing facilities will be collected and diverted to sedimentation tank for silt removal. The treated effluent will be reused onsite.
- 4.5.4 Overall, no adverse water quality impacts are anticipated during the construction or operation phase of the Proposed CBP.



#### *Figure 4-1 Representative WSRs within 500m Study Area of the Site*





#### 5 WASTE MANAGEMENT

#### 5.1 Environmental Legislation and Standards

#### Waste Management

- 5.1.1 The key environmental legislation and standards applicable to waste management in Hong Kong are as follows:
  - Waste Disposal Ordinance (Cap. 354) ("WDO")
  - Waste Disposal (Chemical Waste) (General) Regulation (Cap. 354C)
  - Waste Disposal (Charges for Disposal of Chemical Waste) Regulation (Cap. 354J)
  - Waste Disposal (Charges for Disposal of Construction Waste) Regulation (Cap. 354N)
  - Land (Miscellaneous Provisions) Ordinance (Cap. 28)
  - Public Health and Municipal Services Ordinance (Cap.132BK) Public Cleansing and Prevention of Nuisances Regulation
  - Environmental, Transport and Works Bureau ("ETWB") Technical Circular (Works) No. 19/2005, Environmental Management on Construction Sites
  - ETWB Technical Circular (Works) No. 22/2003A, Additional Measures to improve Site Cleanliness and Control Mosquito Breeding on Construction Sites
  - Development Bureau ("DevB") Technical Circular (Works) No. 6/2010, Trip Ticket System for Disposal of Construction & Demolition Materials
  - Civil Engineering and Development Department ("CEDD") Technical Circulars (CEDD TC No. 11/2019), Management of Construction and Demolition Materials
  - Building Department Practice Note for Authorised Persons, Registered Structural Engineers and Registered Geotechnical Engineers Waste Minimisation – Construction and Demolition Waste ("ADV-19")
  - Building Department Practice Note for Authorised Persons, Registered Structural Engineers and Registered Geotechnical Engineers Waste Minimisation – Provision of Fitments and Fittings in New Buildings ("APP-114")
  - Building Department Practice Note for Registered Contractors ("PNRC 17"), Control of Environmental Nuisance from Construction Sites
  - CEDD Project Administration Handbook for Civil Engineering Works ("PAH")
  - EPD Code of Practice on the Packaging, Labelling and Storage of Chemical Wastes
  - EPD Recommended Pollution Control Clauses ("RPCC") for Construction Contracts

#### 5.2 Review of Waste Management Impacts

#### **Construction Phase**

- 5.2.1 The key potential waste sources during the construction phase are:
  - Inert Construction and Demolition ("C&D") materials (e.g. waste concrete, surplus soil, waste asphalt etc.)



- Non-inert C&D Materials (e.g. wood and plastics)
- Chemical wastes such as waste battery and waste lubricating oil from vehicles/plant maintenance
- General refuse generated by site workers

#### Inert C&D Materials

- 5.2.2 Inert C&D materials are those which do not decompose, such as debris, rubble, earth and concrete, and which are suitable for land reclamation and site formation.
- 5.2.3 As the existing structures within the Site are mainly containers and temporary structures, no building demolition works is required for the Project. Besides, since the Site is fully paved with concrete, no site formation works is required during the construction phase. Therefore, no significant amount of inert C&D materials is anticipated during site preparation stage. The major source of inert C&D materials during construction phase will be the construction of the superstructure for the Proposed Development.
- 5.2.4 In accordance with Section 3.2 of A Guide for Managing and Minimizing Building and Demolition Waste published by the Hong Kong Polytechnic University in May 2001 ("the Guide"), it provides a "waste index" for building waste generation in Hong Kong based on the Gross Floor Area ("GFA") of three different building types as follows:
  - Private Housing Projects 0.250m<sup>3</sup>/m<sup>2</sup> GFA
  - Government Housing Projects 0.174m<sup>3</sup>/m<sup>2</sup> GFA
  - Commercial Office Projects 0.200m<sup>3</sup>/m<sup>2</sup> GFA
- 5.2.5 For estimating of building waste generated from the Proposed CBP, the "waste index" for Commercial Office are adopted. However, as noted above, in addition to inert C&D materials, this "waste index" also include non-inert C&D materials, such as timber formwork, packaging waste and other wastes, and the Guide does not identify what proportion of building waste is inert C&D materials and what proportion is non-inert C&D materials.
- 5.2.6 With reference to Plate 2.12 of EPD's *Monitoring of Solid Waste in Hong Kong Waste Statistics for 2022*, in 2022 92% of construction wastes was either reused on-site or sent to the public fill reception facilities, implying that such construction wastes should be inert C&D materials. The proportion of inert C&D materials in the "waste index" can therefore be estimated by applying the Hong Kong-wide proportion of inert C&D materials in construction waste, i.e. 92%, to the "waste index" as follows:

Waste Index Inert C&D materials (Commercial Office Projects) = 
$$0.92 \times 0.200 \text{m}^3/\text{m}^2 \text{GFA}$$
  
=  $0.184 \text{m}^3/\text{m}^2 \text{GFA}$ 

5.2.7 As the Proposed CBP are mainly comprised of steel structures, assuming the GFA of the Proposed CBP is the same as the site area, i.e. 4,411m<sup>2</sup>, the proportion of inert C&D materials of building waste from the Proposed Development can therefore be estimated as follows:

Building Waste = Waste Index Inert C&D materials (Commercial Office Projects) x GFA



= 0.184 x 4,411

= 812m<sup>3</sup>

- 5.2.8 Assuming the density of inert C&D materials is 1.8 tonnes/m<sup>3</sup>, approx. 1,461 tonnes of building wastes would be generated by the Proposed Development.
- 5.2.9 Assuming the construction period to be one year with six working days a week and four weeks a month, the daily inert C&D material generation rate will be approx. 5.1 tonnes/day (i.e. 1,461 tonnes/ (6 x 4 x 12) days).
- 5.2.10 Inert C&D materials should be reused on-site as far as practicable and efforts should be made to optimise cut and fill requirements during the detailed design. Good site practice and mitigation measures should be implemented, as recommended in **Section 5.3**. The remaining materials should be sent to public fill reception facilities, Fill Bank at Tuen Mun Area 38 and Fill Bank at Tseung Kwan O Area 137.
- 5.2.11 The majority of inert C&D materials generated from construction will be building waste. Since deep excavation and backfilling works is not needed for the Proposed Development, not much of the inert C&D materials will be re-used on site. Hence, the 5.1 tonnes/day inert C&D material will be delivered to public fill reception facilities. The reuse of inert C&D materials in public filling reception facilities would be agreed with relevant authorities before delivery.
- 5.2.12 With the implementation of the recommended good site practice and mitigation measures, no adverse waste impact from the handling, transportation or disposal of inert C&D materials during construction of the Proposed CBP is anticipated.

#### Non-Inert C&D Materials

- 5.2.13 Non-inert C&D materials (or C&D waste) are those which can decompose such as bamboo, timber, vegetation, packaging waste and other organic material, and which are therefore unsuitable for land reclamation.
- 5.2.14 The major source of non-inert C&D material during construction will be the non-inert C&D material component of building waste including timber formwork, packaging waste generated by demolition of existing building and construction of the Proposed CBP.
- 5.2.15 The building waste in the "waste index" provided in the Guide also includes non-inert C&D materials. Since Plate 2.12 of Waste Statistics for 2022 shows that in 2022, 8% of C&D waste was disposed of at landfills. The proportion of non-inert C&D materials (or C&D waste) in the "waste index" can be estimated by applying the Hong Kong-wide proportion of non-inert C&D materials (or C&D waste) in construction waste, i.e. 8%, to the "waste index" as follows:



Waste Index Non-Inert C&D materials (Commercial Office Projects) =  $0.08 \times 0.200 \text{ m}^3/\text{m}^2 \text{ GFA}$ 

 $= 0.016 \text{m}^3/\text{m}^2 \text{GFA}$ 

5.2.16 Given the total GFA of the Proposed Development is approx. 4,411m<sup>2</sup>, the non-inert C&D materials (or C&D waste) components in building waste can therefore be estimated as follows:

Building Waste = Waste Index Non-Inert C&D materials (Commercial Office Projects) x GFA

 $= 71m^{3}$ 

- 5.2.17 Assuming the density of non-inert C&D materials is 1.0 tonnes/m<sup>3</sup>, approx. 71 tonnes of building waste would be generated by the Proposed Development. Assuming the construction period to be one year with six working days a week and four weeks a month, the daily non-inert C&D material generation rate will be approx. 0.2 tonnes/day (i.e. 71 tonnes/ (6 x 4 x 12) days).
- 5.2.18 Non-inert C&D materials generated during construction should be sorted on-site. Recyclable materials, such as metal, paper product, timber and plastic, should be collected by local recyclers for recycling. All non-inert C&D materials should be recycled as far as possible and landfill disposal should be adopted as the last resort. The nearest disposal facility is North East New Territories Landfill ("NENT") Landfill. Disposal of C&D wastes of landfills would be agreed with relevant authorities.
- 5.2.19 It is expected that no more than 10% of the generated non-inert building waste can be recycled or reused. This means that the expected amount of non-inert C&D waste to be reused or recycled on-site is around 7 tonnes at most.
- 5.2.20 If 10% of non-inert C&D materials can be reused/recycled on-site, the surplus non-inert C&D materials mainly comprising building waste will be approx. 64 tonnes in total. Assuming the construction period to be one year with six working days a week and four weeks a month, the total daily non-inert C&D materials for disposal of at NENT Landfill would be approx. 0.2 tonnes/day (i.e. 64 tonnes/ (6 x 4 x 12) days).
- 5.2.21 Considering the above estimation, with the implementation of the recommended good site practice and mitigation measures, no adverse waste impact from the handling, transportation or disposal of non-inert C&D materials (or C&D waste) during construction of the Proposed CBP is anticipated.

#### **General Refuse**

- 5.2.22 General refuse from workers is similar to domestic waste and includes packaging and organic material.
- 5.2.23 The number of workers will depend on the contractor and the construction methods employed. According to the Applicant's experience, the number of construction workers for the Proposed Development should be no more than 20 per day.
- 5.2.24 According to Plate 2.7 of Waste Statistics for 2022, the per capita domestic waste disposal rate in 2022 was 0.93 kg/person/day, although the per worker generation rate of general



refuse will likely be less than this. For a conservative approach, the per capita domestic waste disposal rate in 2022 has been adopted for general refuse generation by construction workers. Since every worker is expected to generate general refuse, the total general refuse generated by construction workers is estimated as follows:

General Refuse/Day	<ul> <li>No. of workers/day x per capita generation rate</li> </ul>
	= 20 workers x 0.93 kg/workers/day
	= 18.6 kg/day
Total General Refuse	= General Refuse/Day x Construction Duration
	= 18.6 kg/day x 6 days/week x 4 weeks/month x 12 months
	= 5,357 kg or 5 tonnes

- 5.2.25 General refuse generated during construction should be sorted on-site. Recyclable materials, such as metal, paper product and plastics should be collected by local recyclers for recycling. All general refuse should be recycled as far as possible and landfill disposal should only be adopted as the last resort.
- 5.2.26 According to Plate 3.2 of Waste Statistics for 2022, in 2022 the recovery rate of domestic waste is approx. 20%. It is therefore assumed that 20% of general refuse, i.e., approx. 1,071kg of general refuse, would be reused and recycled by the recyclers. The surplus general refuse of 4,286 kg or 14.9kg/day (i.e. 18.6kg x 80%) in average would be sent to landfills.
- 5.2.27 Given the above, no adverse waste impact from the handling, transportation or disposal of general refuse from workforce during construction of the Proposed CBP is anticipated.

#### Chemical Waste

- 5.2.28 No hazardous materials or hazardous wastes are expected to be generated during the construction phase. Only limited amount of chemical waste which is expected to be less than 1 tonne of chemical waste including waste batteries, lubricating oil and waste paints may be generated throughout the entire construction phase given the small scale of the works. Other chemical wastes include waste lamp will be generated and the amount will be insignificant.
- 5.2.29 The Contractor shall register as a Chemical Waste Producer under the WDO. All chemical waste shall be stored at a properly designed chemical waste storage area located within the construction site in accordance with EPD's *Code of Practice on the Packaging, Labelling and Storage of Chemical Wastes*. A licensed collector shall be employed to handle and dispose of all chemical wastes, e.g. at the Chemical Waste Treatment Centre ("CWTC") at Tsing Yi, or other facility approved by EPD.
- 5.2.30 Given the above, no adverse waste impact from the handling, transportation or disposal of chemical waste during the construction of the Proposed CBP is anticipated.

#### Summary



## **5.2.31** The type and estimated quantities of different types of wastes generated during the construction phase are summarised in **Table 5-1Table 5-1**.

# Table 5-1Estimated Amount of Different Types of Wastes to be Generated During<br/>Construction Phase

WASTE TYPE	ESTIMATED QUANTITY (TONNES)	KEY SOURCES OF WASTE GENERATION	TREATMENT							
INERT C&D MATERIAL										
Building Waste	1,461	Superstructure Construction	<ol> <li>On-site reuse/recycle</li> <li>Off-site reuse/recycle</li> <li>Delivery to public fill reception facilities</li> </ol>							
NON-INERT C&D MATERIAL										
Building Waste	71	Superstructure Construction	<ol> <li>On-site sorting for reuse/recycle</li> <li>Disposal of at landfill</li> </ol>							
OTHERS										
General Refuse	5	Construction staff	<ol> <li>On-site sorting for reuse/recycle</li> <li>Disposal of at landfill</li> </ol>							
Chemical Waste	<1	Waste batteries, lubricating oil and waste paints, etc.	All to be collected by the licensed chemical waste collector and treated in the CWTC.							

#### **Operation Phase**

- 5.2.32 During the operation phase, the major type of waste generated will be cementitious cake. As advised by the Applicant, about 20m<sup>3</sup> cementitious cake would be generated from daily operation of the Proposed CBP. Besides, municipal Solid Waste ("MSW") from onsite staff and truck drivers as well as chemical waste from the maintenance of the plant equipment will be generated during operation phase.
- 5.2.33 According to Plate 2.7 of Waste Statistic for 2022, the most recent per municipal solid waste disposal rate is 1.51kg/person/day. The estimated total staff of the Proposed Development would be around 12. With six working days per week, the annual MSW generation during the operation phase is expected to be around 5.2 tonnes (i.e. 1.51 kg x 12 x (6 x 4 x 12) days).
- 5.2.34 Plate 3.2 of Waste Statistics for 2022 shows that in 2022, the recovery rate of municipal solid waste is 32%. It is therefore estimated that 32% of commercial waste (i.e. 1.7 tonnes/year) could be reused and recycled by the recyclers. The surplus MSW of 3.5 tonnes/year would be disposed of at the North West New Territories Transfer Station.
- 5.2.35 Since MSW will be collected on a regular basis by registered waste collectors and will be disposed of at a landfill managed by EPD, no adverse waste impacts from handling, transportation or disposal are anticipated. Nevertheless, to minimize MSW generation mitigation measures proposed in **Section 5.3** should be implemented.



- 5.2.36 During the operation phase, it is expected that less than 1 tonne of chemical waste would be generated from the daily operation and maintenance activities of the Proposed CBP. A licensed collector shall be employed to handle and dispose of the chemical wastes, if any.
- 5.2.37 Cementitious cake, the major type of waste, will be also generated from the sedimentation tank for recycling the wastewater from runoff and wheel washing facilities. The cementitious cake is inert materials and will be therefore reused at Public Fill Reception Facilities.
- 5.2.38 With the implementation of the mitigation measures, there should be no adverse waste impact from the handling, transportation or disposal of domestic waste during the operation of the Proposed CBP.

#### 5.3 Mitigation Measures

#### **Construction Phase**

- 5.3.1 Waste management shall be controlled through contractual requirements as well as through statutory requirements.
- 5.3.2 A Waste Management Plan ("WMP") should be developed by the contractor and submitted to the Project Engineer/Architect for approval in accordance with ADV-19 before the commencement of any construction works. The objectives of the WMP will be to identify any potential environmental impacts from the generation of waste at the Site; to recommend appropriate waste handling, collection, sorting, disposal and recycling measures in accordance with requirements of the current regulations; and to categorise and permit segregation of C&D materials where practicable (i.e. inert material/non-inert material) for disposal considerations i.e. public fill/landfill.
- 5.3.3 The contractors should adopt good housekeeping practices with reference to the WMP such as waste segregation prior to disposal. Besides the provision of stockpiling and segregating areas at site, effective collection of site wastes is required to prevent waste materials being blown around by wind, flushed or leached into nearby waters, or creating odour nuisance pest and vermin problems. Waste storage areas should be well maintained and cleaned regularly.
- 5.3.4 A trip-ticket system should be established in accordance with DevB TC(W) No. 6/2010 and the *Waste Disposal (Charges for Disposal of Construction Waste) Regulation* to monitor the disposal of public fill and solid wastes at public filling facilities and landfills, and to control fly-tipping. A trip-ticket system should be included as one of the contractual requirements for the contractor to strictly implement. Dump trucks with mechanical cover shall be used to minimise windblown litter and dust during transportation of waste.
- 5.3.5 Whenever there are excess recyclable construction materials, including bricks, plastics and metals, reuse and recycling should be carried out as far as practicable to minimise the amount of waste disposal. Other inert materials such as concrete, asphalt, etc. should be delivered to public fill. Non-inert and non-recyclable wastes should be disposed at designated landfill site.
- 5.3.6 General refuse should be stored in enclosed bins or compaction units separate from C&D material. A reputable waste collector should be employed by the construction contractor to remove general refuse from the Site, separately from C&D materials. Preferably an



enclosed and covered area should be provided to reduce the occurrence of "wind-blown" materials.

- 5.3.7 Food waste generated during construction and operation would be separated from other waste and recycled as far as practicable, in order to minimise unpleasant odour and potential environmental hygiene issues.
- 5.3.8 For chemical waste, the Contractor should follow the 'trip-ticket' system of which the arrangement of production, collection and disposal in accordance with the *Waste Disposal* (*Chemical Waste*) (*General*) *Regulation*.
- 5.3.9 The mitigation measures for construction phase are recommended based on the waste management hierarchy principles. Recommendations of good site practices, waste reduction measures as well as the waste transportation, storage and collection are described as follows:

#### **Good Site Practice**

- 5.3.10 Adverse impact from waste management implications is not expected, provided that good site practices are strictly implemented. The following good site practices are recommended throughout the construction phase:
  - Nomination of an approved personnel, such as a site manager, to be responsible for the implementation of good site practices, arrangements for collection and effective disposal to an appropriate facility, of all wastes generated at the Site
  - Training of site personnel in site cleanliness, appropriate waste management procedures and concepts of waste reduction, reuse and recycling
  - Provision of sufficient waste disposal points and regular collection for disposal
  - Appropriate measures to minimise windblown litter and dust during transportation of waste by either covering trucks or by transporting wastes in enclosed containers
  - Regular cleaning and maintenance programme for drainage systems, sumps and oil interceptors
  - A Waste Management Plan (WMP) should be prepared by the contractor and submitted to the Engineer for approval

#### Waste Reduction Measures

- 5.3.11 Amount of waste generation can be significant reduced through good management and control. Waste reduction is best achieved at the planning and design phase, as well as by ensuring the implementation of good site practices. The following recommendations are proposed to achieve reduction:
  - Segregate and store different types of waste in different containers, skip or stockpiles to enhance reuse or recycling of materials and their proper disposal
  - Proper storage and site practices to minimise the potential for damage and contamination of construction materials
  - plan and stock construction materials carefully to minimise amount of waste generated and avoid unnecessary generation of waste



- sort out demolition debris and excavated materials from demolition works to recover reusable/recyclable portions (i.e. soil, broken concrete, metal etc.)
- provide training to workers on the importance of appropriate waste management procedures, including waste reduction, reuse and recycling
- 5.3.12 In addition to the above measures, specific mitigation measures are recommended for the specific waste types so as to minimise environmental impacts during handling, transportation and disposal of waste.

#### Storage, Collection and Transportation of Waste

- 5.3.13 Storage of waste on site may induce adverse environmental implications if not properly managed. The following recommendation should be implemented to minimise the impacts:
  - Waste such as soil should be handled and stored well to ensure secure containment
  - Stockpiling area should be provided with covers and water spraying system to prevent materials from wind-blown or being washed away
  - Different locations should be designated to stockpile each material to enhance reuse
- 5.3.14 The collection and transportation of waste from works area to respective disposal sites may also induce adverse environmental impacts if not properly managed. The following recommendation should be implemented to minimise the impacts:
  - Remove waste in timely manner
  - Employ trucks and vessels with cover or enclosed containers for waste transportation
  - Obtain relevant waste disposal permits from the appropriate authorities; including Chemical Waste Permits / licenses under the Waste Disposal Ordinance (Cap 354), Water Pollution Control Ordinance (Cap 358), Public Dumping Licence under the Land (Miscellaneous Provisions) Ordinance (Cap 28), Effluent Discharge Licence under the Water Pollution Control Ordinance
  - Disposal of waste should be done at licensed waste disposal facilities
- 5.3.15 In addition, the EPD's RPCC for Construction Contract should be incorporated in the relevant works contract. The RPCC are generally good engineering practice to minimise inconvenience and environmental nuisance to nearby residents and other sensitive receivers. The general requirements are as follows:
  - The Contractor shall observe and comply with the WDO and its subsidiary.
  - The Contractor shall submit to the Engineer for approval a waste management plan with appropriate mitigation measures including allocation of an area for waste segregation and shall ensure that the day-to-day site operations comply with the approved waste management plan.
  - The Contractor shall minimise the generation of waste from his work. Avoidance and minimisation of waste generation can be achieved through changing or improving design and practices, careful planning and good site management.



- The Contractor shall ensure that different types of wastes are segregated on-site and stored in different containers, skips or stockpiles to facilitate reuse / recycling of waste and, as the last resort, disposal at different outlets as appropriate.
- The reuse and recycling of waste shall be practised as far as possible. The recycled materials shall include paper / cardboard, timber and metal etc.
- The Contractor shall ensure that C&D materials are sorted into public fill (inert portion) and non-inert C&D materials (non-inert portion). The public fill which comprises soil, rock, concrete, brick, cement plaster/mortar, inert building debris, aggregates and asphalt shall be reused such as earth filling, reclamation, site formation works, etc. as far as practicable, and delivered to public fill reception facility as the last resort. The non-inert C&D materials which comprises metal, timber, paper, glass, etc. shall be reused and recycled as far as practicable, and, as the last resort, disposal of at landfills.
- The Contractor shall record the amount of waste generated, recycled and disposed of (including the disposal sites).
- The Contractor shall use a trip ticket system for the delivery of C&D materials to any designated public filling facility and/or landfill.
- Training shall be provided for workers about the concepts of site cleanliness and appropriate waste management procedure, including waste reduction, reuse and recycling.
- The Contractor shall not permit sewage and untreated effluent containing sand, cement, silt or any other suspended or dissolved material to flow from the Site onto any adjoining land, or allow any solid waste including refuse which is not part of the final product from waste processing plants to be deposited anywhere within the Site and the adjoining land. He shall arrange removal of such matter from the Site in a proper manner to the satisfaction of the Engineer in consultation with the EPD.
- The Contractor shall observe and comply with the Waste Disposal (Chemical Waste) (General) Regulation.
- The Contractor shall apply for registration as chemical waste producer under the *Waste Disposal (Chemical Waste) (General) Regulation* when chemical waste is produced. All chemical waste shall be properly stored, labelled, packaged and collected in accordance with the Regulation.
- Fly-tipping or disposal of C&D materials at locations other than the designated locations shall be prohibited. In order to review the trip-ticket record as mentioned in **Paragraph 5.3.4** above, the following measures may be considered when necessary:
  - All dump trucks engaged on-site for delivery of inert and non-inert C&D material from the site to the designated disposal locations, including PFRFs, landfills etc., should be equipped with GPS or equivalent system for tracking and monitoring of their travel routings and parking locations by the Contractor to prohibit illegal dumping and landfilling of materials.
  - The data collected by GPS or equivalent system should be recorded properly for checking and analysis the travel routing and parking locations of dump truck engaged on-site.

#### **Operation Phase**



- 5.3.16 The operator shall encourage reuse and recycling of commercial wastes in line with government policy. The waste management hierarchy shall be adopted by the building management to manage commercial wastes in a suitable manner. The waste management hierarchy is a concept which shows the desirability of various waste management methods and comprises the following in order of preference:
  - Avoidance
  - Minimisation
  - Recycling/reuse
- 5.3.17 MSW such as general refuse, food waste, food packaging, paper, can, plastic bottles, etc., which shall be collected and stored in appropriate waste receptacles with a secure lid to minimize the potential adverse impact due to wind blowing away garbage and to improve hygiene. Recyclable and non-recyclable waste shall be regularly collected by waste collectors and taken off-site for recycling or disposal, respectively.
- 5.3.18 For the cementitious cake, the Contractor should temporarily store the cementitious cake onsite in a waste hopper with impervious sheeting covering before transferred to Public Fill Reception Facilities for reuse.
- 5.3.19 For chemical waste, the Contractor should follow the 'trip-ticket' system of which the arrangement of production, collection and disposal in accordance with the Waste Disposal (Chemical Waste) (General) Regulation.

#### 5.4 Conclusion

- 5.4.1 With the development of WMP and to implement the good site practices recommended therein, the waste generation during construction phase can be greatly reduced. Provided that good site practices as recommended in **Section 5.3** will be followed, there should be no adverse impacts related to the management, handling and transportation of waste during construction phase.
- 5.4.2 During the operation phase, the major type of waste generated will be general waste from onsite staff and truck drivers, and cementitious cake. General waste will be collected on a regular basis by registered waste collectors and will be disposed of at a landfill managed by EPD, while cementitious cake will be reused at Public Fill Reception Facilities. Hence, no adverse waste impacts from handling, transportation, or disposal are anticipated during operation phase.
- 5.4.3 With the implementation of the recommended mitigation measures, adverse waste management impact during the construction and operation phases of the Proposed CBP is not anticipated.



#### 6 LAND CONTAMINATION

#### 6.1 Environmental Legislation and Standards

- 6.1.1 The following legislation, standards and guidelines were taken reference to for conducting the land contamination assessment:
  - EPD Guidance Note for Contaminated Land Assessment and Remediation.
  - EPD Practice Guide for Investigation and Remediation of Contaminated Land.
  - Guidance Manual for Use of Risk-Based Remediation Goals for Contaminated Land Management.

#### 6.2 Assessment Methodology

- 6.2.1 Land contamination assessment would be conducted according to EPD's Practice Guide. Typically, as a part of a complete land contamination assessment, a site appraisal would be conducted to determine whether there is any potential for land contamination in the Site.
- 6.2.2 In the case that potential land contamination issues are identified, a complete land contamination assessment with the following steps should be undertaken:
  - 1. Design a site investigation ("SI") strategy and prepare a Contamination Assessment Plan ("CAP") for EPD's approval
  - 2. Upon EPD's approval of the CAP, conduct SI according to the approved CAP
  - 3. Upon completion of SI, interpret the results and prepare a Contamination Assessment Report ("CAR") for EPD's approval
  - 4. Plan and design remediation strategy and prepare a Remediation Assessment Plan ("RAP") for EPD's approval
  - 5. Carry out remediation works according to the approved RAP
  - 6. Prepare a Remediation Report ("RR") for EPD's endorsement

#### 6.3 Site Appraisal

#### **Review of Historical Use**

**6.3.1** Aerial photographs records provided in **Appendix K** show that the Site was a piece of agricultural land with some vegetation and a temporary structure for storage in 1963. In Year 1973, The Site was occupied by several temporary structures with some vegetation and bare ground. No significant change on the land use was observed until 2013. In Year 2013, the Site was used as an open storage for vehicles and vehicle parts. In 2023, the Site was used as open storage for construction materials and plants. The historical land uses of the Site based on the aerial photographic records are summarised in **Table 6-1Table 6-1**.



es of the Project Site

PHOTO ID	HISTORICAL LAND USES
1963_1963-8394	Majority of the Site was occupied by agricultural land and natural vegetation. A temporary structure with storage use was present. No stressed vegetation was observed.
1973_07328	The Site was occupied by several temporary structures with some vegetation and bare ground. No stressed vegetation was observed.
1982_42976	No major changes were observed
1993_CN05931	No major changes were observed
2003_CW52218	No major changes were observed
2013_CS43741	The site was used as an open storage for abandoned vehicles and vehicle parts.
2023_E188907C	The site was used as open storage for construction materials and plants.

6.3.2 A site visit was conducted on 10 January 2023, based on site observations, the site should be used for open storage for construction materials and plants. The ground was paved with concrete and no crack or oil stain was observed. Therefore, there is no potential for land contamination. The site walkover checklist and site photos are provided in Appendix L.

#### **Dangerous Goods and Incident Records**

- 6.3.3 EPD Regional Office (North) – Yuen Long was contacted on 4 July 20204 to review if any record of registered Chemical Waste Producer ("CWP") or accident spillage / leakage of dangerous or chemical is related to the Site. EPD's reply is still pending and will be provided once available.
- Moreover, Fire Services Department ("FSD") was also contacted to review any current / 6.3.4 past licences for storage of Dangerous Goods ("DG"), registration of DG licence, fire incidents, spillage/leakage of DG etc., relating to the Site. According to FSD's reply dated 30 July 2024, neither records of DG license, fire incidents nor incidents of spillage/ leakage of DG were found in connection with the Site. Therefore, it is considered that no land contamination potential is anticipated replated to the Site. The information request letters and replies from EPD and FSD are attached in Appendix M.

#### 6.4 Conclusion

- 6.4.1 Based on the preliminary review on the historical aerial photographs site visit and relevant information from EPD and FSD, no existing and previous development with potential land contamination activities on the Site is identified. Hence, no land contamination issue is anticipated.
- 6.4.2 As such, steps 1 to 6 of Paragraph 6.2.2 are not required. Hence, no adverse impact from the land contamination issue is anticipated and site investigation is considered not necessary.



### **Appendix A**

## Calculation and Summary of Emission Rates

Summary of emission rates and parameters for emission sources during daytime operation

		Coordinates													Outlet Dim	ension (m)	sion (m)			FSP	NOx	
Emission Points	Descriptions	x	Y	Elevation (mPD)	Type of Source	Ducted / Fugitive?	Stack/Emission Height (mAG)	Stack/Emission Height (mPD)	Exit Temperature (K)	Exit Temperature (K) (For model input)	Flow Rate (m <sup>3</sup> /hour)	Exit Velocity (m/s)	Length of X- axis	Length of Y- axis	Input Angle	Length	Width	Calculated Diameter	Emission Rate (g/s for point source or g/m/s for area source)	Emission Rate (g/s for point source or g/m/s for area source)	Emission Rate (g/s for point source or g/m/s for area source)	Operation Period
EP1	Dust Collector (DC-1) for Cementitious Material Silo	828770.4	832953.2	51.2	Point	D	13.0	64.2	Ambient	0	3,000	25.42	-	-	-	0.25	0.15	0.22	1.08E-03	1.08E-03		0700-2300
EP2	Dust Collector (DC-2) for Cementtitous Material Silo	828762.4	832953.2	51.2	Point	D	13.0	64.2	Ambient	0	3000	25.42	-		-	0.25	0.15	0.22	1.08E-03	1.08E-03		0700-2300
EP3	Dust Collector (DC-3) for Cementitious Material Silo	828753.3	832953.2	51.2	Point	D	13.0	64.2	Ambient	0	3,000	25.42	-	-	-	0.25	0.15	0.22	1.08E-03	1.08E-03		0700-2300
EP4	Dust Collector (DC-4) for Cementitious Material Silo	828748.1	832953.2	51.2	Point	D	13.0	64.2	Ambient	0	3,000	25.42	-	-	-	0.25	0.15	0.22	1.08E-03	1.08E-03		0700-2300
EP5	Dust Collector (DC-5) for Cement/GGBS/PFA Holding Hopper	828740.0	832927.2	51.2	Point	D	13.0	64.2	Ambient	0	1,500	12.71	-	-	-	0.25	0.15	0.22	1.08E-03	1.08E-03		0700-2300
EP6	Dust Collector (DC-6) for Cement/GGBS/PFA Holding Hopper	828735.3	832927.2	51.2	Point	D	13.0	64.2	Ambient	0	1,500	12.71	-	-	-	0.25	0.15	0.22	1.08E-03	1.08E-03		0700-2300
EP7	Dust Collector (DC-7) for Cement/GGBS/PFA Holding Hopper	828737.8	832927.2	51.2	Point	D	13.0	64.2	Ambient	0	1,500	12.71	-	-	-	0.25	0.15	0.22	1.08E-03	1.08E-03		0700-2300
EP8	Dust Collector (DC-8) for Silica Fume Silo	828735.1	832934.8	51.2	Point	D	13.0	64.2	Ambient	0	3,000	25.42	-	-	-	0.25	0.15	0.22	1.08E-03	1.08E-03		0700-2300
EP9	Dust Collector (DC-9) for Silica Fume Silo	828726.2	832938.1	51.2	Point	D	13.0	64.2	Ambient	0	3,000	25.42	-	-	-	0.25	0.15	0.22	1.08E-03	1.08E-03		0700-2300
EP10	Dust Collector (DC-10) for Cement/GGBS/PFA Holding Hopper	828729.0	832945.6	51.2	Point	D	13.0	64.2	Ambient	0	1,500	1.25	-	-	-	0.491	0.562	0.59	1.08E-03	1.08E-03		0700-2300
EP11	Dust Collector (DC-11) for Cement/GGBS/PFA Holding Hopper	828726.3	832945.6	51.2	Point	D	13.0	64.2	Ambient	0	1,500	1.25	-	-	-	0.491	0.562	0.59	1.08E-03	1.08E-03		0700-2300
EP12	Dust Collector (DC-12) for Cement/GGBS/PFA Holding Hopper	828731.5	832945.6	51.2	Point	D	13.0	64.2	Ambient	0	1,500	1.25	-	-	-	0.491	0.562	0.59	1.08E-03	1.08E-03		0700-2300
EP13	Dust Collector (DC-13) for Mixer	828731.7	832942.3	51.2	Point	D	13.0	64.2	Ambient	0	1,500	1.25	-	-	-	0.491	0.562	0.59	2.24E-04	2.24E-04		0700-2300
EP14	Dust Collector (DC-14) for Mixer	828735.1	832930.5	51.2	Point	D	13.0	64.2	Ambient	0	1,500	1.25	-	-	-	0.491	0.562	0.59	2.24E-04	2.24E-04		0700-2300
EP15	Aggregate Stockpile 2	828774.6	832957.0	51.2	Area	F	2	53.2	-	-	-	-	14.5	7.4	90.0	-	-	-	8.19E-07	1.24E-07		0700-2300
EP16	Aggregate Stockpile 1	828797.0	832913.7	51.2	Area	F	2	53.2	-	-	-	-	22.0	15.0	0.0	-	-	-	1.47E-06	2.22E-07		0700-2300
EP17	Unloading of aggregate into aggregate receiving hopper	828774.8	832936.5	51.2	Area	F	3.5	54.7	-	-	-	-	4.5	4.8	0.0	-	-	-	1.11E-04	3.41E-05		0700-2300
EP18	Unloading of aggregate into aggregate receiving hopper	828774.8	832931.5	51.2	Area	F	3.5	54.7	-	-	-	-	4.5	4.8	0.0	-	-	-	1.11E-04	3.41E-05		0700-2300
EP19_1	Paved Road	828741.4	832944.7	51.2	Area	F	0.5	51.7	-	-	-	-	15.7	3.0	-52.0	-	-	-	5.21E-05	1.26E-05		0700-2300
EP19_2	Paved Road	828762.5	832944.1	51.2	Area	F	0.5	51.7	-	-	-	-	20.2	3.0	-89.0	-	-	-	2.39E-05	5.78E-06		0700-2300
EP19_3	Paved Road	828782.5	832944.1	51.2	Area	F	0.5	51.7	-	-	-	-	19.9	3.0	-89.0	-	-	-	2.39E-05	5.78E-06		0700-2300
EP19_4	Paved Road	828788.5	832937.9	51.2	Area	F	0.5	51.7	-	-	-	-	9.7	3.0	-47.0	-	-	-	2.39E-05	5.78E-06		0700-2300
EP19_5	Paved Road	828788.4	832927.6	51.2	Area	F	0.5	51.7	-	-	-	-	11.3	3.0	-2.0	-	-	-	2.39E-05	5.78E-06		0700-2300
EP19_6	Paved Road	828780.9	832918.1	51.2	Area	F	0.5	51.7	-	-	-	-	13.1	3.0	37.0	-	-	-	2.39E-05	5.78E-06		0700-2300
EP19_7	Paved Road	828782.1	832915.7	51.2	Area	F	0.5	51.7	-	-	-	-	34.6	3.0	-90.0	-	-	-	2.39E-05	5.78E-06		0700-2300
EP19_8	Paved Road	828747.3	832915.8	51.2	Area	F	0.5	51.7	-	-	-	-	10.0	3.0	-84.0	-	-	-	6.54E-06	1.58E-06		0700-2300
EP19_9	Paved Road	828736.8	832917.1	51.2	Area	F	0.5	51.7	-	-	-	-	10.3	3.0	-61.0	-	-	-	3.27E-06	7.91E-07		0700-2300
EP19_10	Paved Road	828727.0	832923.5	51.2	Area	F	0.5	51.7	-	-	-	-	16.3	3.0	1.0	-	-	-	3.27E-06	7.91E-07		0700-2300
EP19_11	Paved Road	828736.0	832918.4	51.2	Area	F	0.5	51.7	-	-	-	-	14.4	3.0	1.0	-	-	-	3.27E-06	7.91E-07		0700-2300
EP19_12	Paved Road	828727.4	832939.8	51.2	Area	F	0.5	51.7	-	-	-	-	15.9	3.0	4.0	-	-	-	3.27E-06	7.91E-07		0700-2300
EP19_13	Paved Road	828736.3	832932.4	51.2	Area	F	0.5	51.7	-	-	-	-	24.0	3.0	-19.0	-	-	-	3.27E-06	7.91E-07		0700-2300
EP19_14	Paved Road	828746.0	832917.0	51.2	Area	F	0.5	51.7	-	-	-	-	29.0	3.0	-10.0	-	-	-	5.43E-06	1.31E-06		0700-2300
[																						

Note: Max emission rate for EP19
EP1 to EP4 and EP8 to EP9 Emission from Dust Collectors (DC 1 to DC 4 & DC 8 to DC 9) for Cementiti	ious Ma	aterials Silos		
Maximum Cementitious Material Loading Rate per Silo	=	60	tonnes/hour	(Note 1)
Capacity of Dust Collector (Flow rate)	=	3,000	m <sup>3</sup> /hour	(Note 2)
	=	1	m <sup>3</sup> /s	
Control Efficiency (η%)	=	99.99	%	(Note 2)
Uncontrolled TSP Emission Factor of Cementitious Materials	=	1.57	kg/Mg	(Note 3)
Uncontrolled RSP Emission Factor of Cementitious Materials	=	0.65	kg/Mg	(Note 4)
Uncontrolled FSP Emission Factor of Cementitious Materials	=	0.65	kg/Mg	(Note 5)
Unmitigated TSP Emission Rate of Each Dust Collector of Cementitious Materials per Silo	=	94.2	kg/hour	
Unmitigated RSP Emission Rate of Each Dust Collector of Cementitious Materials per Silo	=	39	kg/hour	
Unmitigated FSP Emission Rate of Each Dust Collector of Cementitious Materials per Silo	=	39	kg/hour	
Mitigated TSP Emission of Dust Collector	=	0.00942	kg/hour	
	=	9.42	g/hour	
	=	0.0026	g/s	
	=	3.14	mg/m <sup>3</sup>	
Mitigated RSP Emission of Dust Collector	=	0.0039	kg/hour	
	=	3.9	g/hour	
	=	0.0011	g/s	
	=	1.30	mg/m <sup>3</sup>	
Mitigated FSP Emission of Dust Collector	=	0.0039	kg/hour	
•	=	3.9	g/hour	
	=	0.0011	g/s	
	=	1.30	mg/m <sup>3</sup>	
EP5 to EP7 and EP10 to EP12 Emissions from Dust Collectors (DC5 to DC7 and DC10 to DC12) for Cer	nent/G	GBS/PFA Holdi	ng Hoppers	
Maximum Cement/GGBS/PFA Consumption Rate	=	60	tonne/hour	(Note 6)

Capacity of Dust Collector (Flow rate)	=	1500	m³/hour	(Note 2)
	=	0.4167	m <sup>3</sup> /s	
Control Efficiency (ŋ%)	=	99.99	%	(Note 2)
Uncontrolled TSP Emission Factor of Cement/PFA/GGBS Loading	=	1.57	kg/Mg	(Notes 7, 3)
Uncontrolled RSP Emission Factor of Cement/PFA/GGBS Loading	=	0.65	kg/Mg	(Notes 7, 4)
Uncontrolled FSP Emission Factor of Cement/PFA/GGBS Loading	=	0.65	kg/Mg	(Notes 7, 5)
Unmitigated TSP Emission Rate of Each Dust Collector of Cementitious Materials	=	94.2	kg/hour	
Unmitigated RSP Emission Rate of Each Dust Collector of Cementitious Materials	=	39	kg/hour	
Unmitigated FSP Emission Rate of Each Dust Collector of Cementitious Materials	=	39	kg/hour	
Mitigated TSP Emission of Dust Collector	=	0.00942	kg/hour	
·	=	9.42	g/hour	
	=	0.0026	g/s	
	=	6.28	mg/m <sup>3</sup>	
Mitigated RSP Emission of Dust Collector	=	0.0039	kg/hour	
	=	3.9	g/hour	
	=	0.0011	g/s	
	=	2.6	mg/m <sup>3</sup>	
Mitigated FSP Emission of Dust Collector	=	0.0039	kg/hour	
	=	3.9	g/hour	
	=	0.0011	g/s	
	=	2.6	mg/m <sup>3</sup>	
		2.0		

## EP13 and EP14 Emissions from Dust Collector (DC 13 and DC 14) for 5m<sup>3</sup> Concrete Mixer

Total Concrete Production Rate	=	100	m <sup>3</sup> /hour/leg	
Density		2.4	tonne/m <sup>3</sup>	
Total Concrete Production Rate	=	240	tonne/hour/leg	
Maximum Cementitious Material Consumption Rate (r1)	=	100	tonne/hour/leg	(Note 8)
		0.028	tonne/s/leg	
Maximum Aggregate/Sand Consumption Rate (r2)	=	200	tonne/hour/leg	(Note 8)
		0.056	tonne/s/leg	
Uncontrolled TSP Emission Factor of Mixer Loading for Cementitious Material (EF <sub>TSP1</sub> )	=	0.286	kg/Mg	(Note 9)
Uncontrolled TSP Emission Factor of Mixer(Weigh Hopper) Loading for Aggregate/Sand (EF <sub>TSP2</sub> )	=	0.0026	kg/Mg	(Note 9)
Unmitigated Total TSP Emission Rate of (ER <sub>TSP</sub> )	=	0.0081	kg/s/mixer	
Uncontrolled RSP Emission Factor of Mixer Loading for Cementitious Material (EF <sub>RSP1</sub> )	=	0.078	kg/Mg	(Note 9)
Uncontrolled RSP Emission Factor of Mixer(Weigh Hopper) Loading for Aggregate/Sand (EF <sub>RSP2</sub> )	=	0.0013	kg/Mg	(Note 9)
Unmitigated Total RSP Emission Rate of (ER <sub>RSP</sub> )	=	0.0022	kg/s/mixer	
Uncontrolled FSP Emission Factor of Mixer Loading for Cementitious Material (EF <sub>FSP1</sub> )	=	0.078	kg/Mg	(Note 10)
Uncontrolled FSP Emission Factor of Mixer(Weigh Hopper) Loading for Aggregate/Sand (EF <sub>FSP2</sub> )	=	0.0013	kg/Mg	(Note 10)
Unmitigated Total FSP Emission Rate of (ER <sub>FSP</sub> )	=	0.0022	kg/s/mixer	
Mitigated Measures: Dust collector DC 13 and DC 14				
Capacity of Dust Collector (Flow rate)	=	1500	m <sup>3</sup> /hour	(Note 2)
	=	0.4167	m³/s	
Control Efficiency (ŋ%)	=	99.99	%	(Note 2)
Mitigated TSP Emission Rate of Dust Collector	=	8.09E-07	kg/s	
		0.001	g/s	
		1.9	mg/m <sup>3</sup>	
Mitigated RSP Emission Rate of Dust Collector	=	2.24E-07	kg/s	
		2.24E-04	g/s	
		0.5	mg/m³	
Mitigated FSP Emission Rate of Dust Collector	=	2.24E-07	kg/s	
		2.24E-04	g/s	
		0.5	malm	

Note: 1. The maximum cementitious material (cement, PFA, GGBS and silica fume) loading rate is provided by the Applicant.

2. The maintain definitions in activity (Lenient, Przy, Golds and sinker in the loading tast is provided by the Applicant:
2. Capacity of Dust Collector is provided by the Applicant as shown Appendix.
3. For TSP (or Total PM), cement unloading rate of 0.36kg/Mg and cement supplement unloaing rate of 1.57kg/Mg are mentioned in
USEPA AP-42 Table 11.12-1 of Section 11.12 version 6/06. The higher value of 1.57kg/Mg is considered for the worst-case scenario.
4. For RSP (or PM<sub>0b</sub>), cement unloading rate of 0.24kg/Mg and cement supplement unloaing rate of 0.55kg/Mg are mentioned in

USEPA AP-42 Table 11.12-1 of Section 11.12 region  $\beta/0.5$ . The higher value of 0.65kg/Mg is considered for the worst-case scenario. 5. For FSP (or PM<sub>2.5</sub>), no data for cementitious material unloading to elevated storage silo is provided in USEPA AP-42.

For Tsy (OF PM<sub>25</sub>), no data for cementuous material unioading to elevate storage silo is provided in USEA AP-42.
 Therefore, the uncontrolled ISP emission factor for cementuous material unioading to silo is assumed to be the same as that of RSP for the worst-case scenario.
 The maximum cement, PFA and GGBS consumption rate is provided by the Applicant.
 Cementitious material is loaded from the relevant silo to presurised tank and then the relevant holding hopper. Because the cementitious material is is conveyed via pneumatic pipe to holding tank, the emission factor of unloading to elevated silo (pneumatic) should be adopted for the worst-case scenario.
 Material consumption rates are different for different concrete products. Therefore, the maximum aggregate and cementitious consumption rates have

been adopted for the worst-case scenario. 9. The emission factors are referred to USEPA AP-42 Table 11.12-1 of Section 11.12 version 6/06.

10. For FSP (or PM25), no data for cementitious material and aggregate/sand unloading to mixer/weigh hopper is provided in USEPA AP-42.

Therefore, the uncontrolled FSP emission factors for cementitious material and aggregate/sand unloading to mixer/weigh hopper are assumed to be the same as that of RSP for the worst-case scenario.

# Environmental Assessment for Concrete Batching Plant Kam Tin Calculation of Emissions from Aggregate Ground Stockpile

Method: USEPA AP-42, Section 13.2.4.3, version 6/06

Emission Factor	=	k(0.0016) *	$\frac{(U/2.2)^{1.3}}{(M/2)^{1.4}}$	– kg/Mg
Wing Speed (U)	=	3.60	m/s	(Note 1)
Moisture Content (M)	=	4.17	%	(Note 2)
k <sub>rsp</sub>	=	0.35	dimensionless	(Note 3)
k <sub>FSP</sub>	=	0.053	dimensionless	(Note 3)
RSP Emission Factor	=	3.80E-04	kg/Mg	
FSP Emission Factor	=	5.75E-05	kg/Mg	
Working hour per day	=	24	hour/day	
Stockpiles (EP15)				
Approximate Amount of Aggregate Fine/Sand Stored	=	400	tonnes/day	(Note 4)
Mitigation Measures : Cover on top with three side enclosure	+ roof + v	vatering		
Receiving Area	=	107.3	m²	
Control Efficiency	=	95	%	(Note 5)
Mitigated RSP Emission Rate	=	0.007595	kg/day	
	=	0.000088	g/s	
	=	8.192E-07	g/m²/s	
Mitigated FSP Emission Rate	=	0.001150	kg/day	
	=	0.000013	g/s	
	=	1.241E-07	g/m²/s	
Stockpiles (EP16)				
Approximate Amount of	=	1100	tonnes/day	(Note 4)
Aggregate Fine/Sand Stored				
Mitigation Measures : Watering				
Receiving Area	=	330	m <sup>2</sup>	
Control Efficiency	=	90	%	(Note 6)
Mitigated RSP Emission Rate	=	0.041771	kg/day	
	=	0.000483	g/s	
	=	1.465E-06	g/m²/s	
Mitigated FSP Emission Rate	=	0.006325	kg/day	
	=	0.000073	g/s	
	=	2.218E-07	g/m²/s	

Noted:

1. The average wind speed is estimated based on the average wind speed for MM5 Grids (33,46) in m/s.

2. The moisture content was assumed to be 4.17% with reference to AP42 Table11.12-1 Note b - moisture content of sand

3. The particle size multipliers are suggested in USEPA AP-42 Section 13.2.4.3.

4. The approximate amounts of stored sand/aggregate fines and aggregates per day are advised by the Applicant.

5. Reference to the Approved EIA report "Road Works at West Kowloon" in 2009, 95% of dust removal efficiency was adopted for unloading aggregate to receiving hopper with enclosure on three sides and a top cover and equipped with water spraying system.

6. Referece to the DRR of Approved SP Licence Renewal "Lam Tei Quarry CBP", 90% of dust removal efficiency was adopted for unloading aggregate receiving hopper with watering.

# Concrete Batching Plant at Kam Tin Calculation of Emissions from Aggregate Transfer

Method: USEPA AP-42 Table 11.12-1 of Section 11.12 version 6/06

Emissions from unloading of aggregates to aggregate receiving hopper (EP17)				
Maximum aggregate loading rate	=	100,000	kg/hr	
		100	tonnes/hour (i.e. Mg/hour)	(Note 1)
Uncontrolled TSP Emission Factor (Aggregate)	=	0.0035	kg/Mg	
Uncontrolled RSP Emission Factor (Aggregate)	=	0.0017	kg/Mg	
Unmitigated Total TSP Emission Rate from EP5	=	0.3500	kg/hr	
Unmitigated Total RSP Emission Rate from EP5	=	0.1700	kg/hr	
Unmitigated Total FSP Emission Rate from EP5	=	0.0525	kg/hr	(Note 2)
Mitigation Measures · With three side enclosure + water spraving				
Pereiving Area	=	21 36	m <sup>2</sup>	(Note 1)
Control Efficience	_	21.50		(Note 1)
Control Enciency	=	95	%	(Note 3)
Mitigated TSP Emission Rate	=	0.0175	kg/hr	
	=	0.004861	g/s	
	=	2.28E-04	g/m <sup>2</sup> /s	
			8//2	
Mitigated RSP Emission Rate	=	0.0085	kg/hr	
	=	0.002361	g/s	
	=	1.11E-04	g/m²/s	
Mitigated FSP Emission Rate	=	0.00263	kg/hr	
	=	7.29E-04	g/s	
	=	3.41E-05	g/m²/s	
Emissions from unloading of aggregates to aggregate receiving hopper (EP18)		400.000	1 - 11 -	
Maximum aggregate loading rate	=	100,000	kg/nr	(Nata 1)
		100	tonnes/hour (i.e. Mg/hour)	(Note 1)
Uncontrolled TSP Emission Factor (Aggregate)	=	0.0035	kg/Mg	
Uncontrolled RSP Emission Factor (Aggregate)	=	0.0017	kg/Mg	
Unmitigated Total TSP Emission Rate from EP5	=	0.3500	kg/hr	
Unmitigated Total RSP Emission Rate from EP5	=	0.1700	kg/hr	
Unmitigated Total FSP Emission Rate from EP5	=	0.0525	kg/hr	(Note 2)
Mitigation Measures : With three side enclosure + water spraying			2	
Receiving Area	=	21.36	m <sup>2</sup>	(Note 1)
Control Efficiency	=	95	%	(Note 3)
Mitigated TSP Emission Rate	=	0.0175	kø/hr	
intigated for Emission fate	=	0 004861	g/s	
	_	2 28F-04	$a/m^2/c$	
	-	2.285-04	g/m /s	
Mitigated RSP Emission Rate	=	0.0085	kg/hr	
-	=	0.002361	g/s	
	=	1.11E-04	$g/m^2/s$	
			B//3	
Mitigated FSP Emission Rate	=	0.00263	kg/hr	
-	=	7.29E-04	g/s	
	_	3 41F-05	$\sigma/m^2/s$	
		J. TIL 05	5/11/3	

Note

1. Information as provided by the Applicant.

 2. FSP emission concentration is assumed to be 15% of TSP in accordance with Page B.2-13, Appendix B2 of AP-42.
 3. Reference to the Approved EIA report "Road Works at West Kowloon" in 2009, 95% of dust removal efficiency was adopted for unloading aggregate to receiving hoj enclosure on three sides and a top cover and equipped with water spraying system.

4. Since the aggregate unloading from barge and truck will not operate at the same time, so as the worst case scenario, the higher pollutant emission rate will be adopted in the assssmemt (i.e. unloading from barge)

## Environmental Assessment for Concrete Batching Plant at Kam Tin

## Traffic Density of the CBP

Sogmont ID	Coord	linates	Road Longth	0700 - 0900				
Segment ID	Х	Y	Koau Lengui	Concrete Truck	Aggregate Truck	Cementitious Tanker	Total	
EP19_1	828741.4	832944.7	15.7	20	4		24	
EP19_2	828762.5	832944.1	20.2	20	2		22	
EP19_3	828782.5	832944.1	19.9	20	2		22	
EP19_4	828788.5	832937.9	9.7	20	2		22	
EP19_5	828788.4	832927.6	11.3	20	2		22	
EP19_6	828780.9	832918.1	13.1	20	2		22	
EP19_7	828782.1	832915.7	34.6	20	2		22	
EP19_8	828747.3	832915.8	10	20			20	
EP19_9	828736.8	832917.1	10.3	10			10	
EP19_10	828727	832923.5	16.3	10			10	
EP19_11	828736	832918.4	14.4	10			10	
EP19_12	828727.4	832939.8	15.9	10			10	
EP19_13	828736.3	832932.4	24	10			10	
EP19 14	828746	832917	29	0	2		2	

## RSP Emission Factors of the Paved Road

Segment ID	Coordinates		Road Length		ctor (g/s/m²)		
Jegmentib	Х	Y	rtoau Lengur	Concrete Truck	Aggregate Truck	Cementitious Tanker	Total
EP19_1	828741.4	832944.7	15.7	1.83E-05	3.66E-06	0.00E+00	2.20E-05
EP19_2	828762.5	832944.1	20.2	1.24E-05	1.24E-06	0.00E+00	1.37E-05
EP19_3	828782.5	832944.1	19.9	1.24E-05	1.24E-06	0.00E+00	1.37E-05
EP19_4	828788.5	832937.9	9.7	1.24E-05	1.24E-06	0.00E+00	1.37E-05
EP19_5	828788.4	832927.6	11.3	1.24E-05	1.24E-06	0.00E+00	1.37E-05
EP19_6	828780.9	832918.1	13.1	1.24E-05	1.24E-06	0.00E+00	1.37E-05
EP19_7	828782.1	832915.7	34.6	1.24E-05	1.24E-06	0.00E+00	1.37E-05
EP19_8	828747.3	832915.8	10	6.54E-06			6.54E-06
EP19_9	828736.8	832917.1	10.3	3.27E-06			3.27E-06
EP19_10	828727	832923.5	16.3	3.27E-06			3.27E-06
EP19_11	828736	832918.4	14.4	3.27E-06			3.27E-06
EP19_12	828727.4	832939.8	15.9	3.27E-06			3.27E-06
EP19_13	828736.3	832932.4	24	3.27E-06			3.27E-06
EP19 14	828746	832917	29		5.88E-07	0.00E+00	5.88E-07

## FSP Emission Factors of the Paved Road

### Coordinates FSP Emission Factor (g/s/m<sup>2</sup>) Road Length Segment ID Aggregate Truck Cementitious Tanker х Y Concrete Truck Total EP19\_1 828741.4 832944.7 15.7 4.43E-06 8.86E-07 0.00E+00 5.31E-06 EP19\_2 828762.5 832944.1 20.2 3.00E-06 3.00E-07 0.00E+00 3.31E-06 EP19\_3 828782.5 832944.1 19.9 3.00E-06 3.00E-07 0.00E+00 3.31E-06 EP19 4 828788.5 832937.9 9.7 3.00E-06 3.00E-07 0.00E+00 3.31E-06 EP19 5 828788.4 832927.6 11.3 3.00E-06 3.00E-07 0.00E+00 3.31E-06 EP19\_6 828780.9 832918.1 13.1 3.00E-06 3.00E-07 0.00E+00 3.31E-06 3.00E-07 0.00E+00 EP19\_7 828782.1 832915.7 34.6 3.00E-06 3.31E-06 EP19 8 828747.3 832915.8 10 1.58E-06 1.58E-06 EP19\_9 828736.8 832917.1 10.3 7.91E-07 7.91E-07 EP19\_10 828727 832923.5 16.3 7.91E-07 7.91E-07 EP19\_11 828736 832918.4 14.4 7.91E-07 7.91E-07 828727.4 832939.8 EP19\_12 15.9 7.91E-07 7.91E-07 EP19 13 828736.3 832932.4 24 7.91E-07 7.91E-07 EP19\_14 828746 832917 29 1.42E-07 0.00E+00 1.42E-07

### Calculation of Dust Suppression Efficiency for Paved Road Method: USEPA, Control of Open Fugitive Dust Sources

Percentage of Dust Suppression by Watering is derived from the equation: C = 100 - (0.8pdt / i)

Where:

<ul> <li>average control efficiency, perc</li> </ul>	ent
--	-----

- C
   =
   average fourtor entents, percent

   P
   =
   potential average hourly daytime evaporation rate, mm/h

   d
   =
   average hourly daytime traffic rate

   i
   =
   application intensity, L/m<sup>2</sup>

   t
   =
   time between applications, h (t =1 for apply water per hour)

 0.0049 x evaporation for annual condition
 1204.1 mm
 47.40542 inch Р Evaporation for annual condition Ρ = 0.2323 mm/h 1.5 L/m<sup>2</sup> 1 h i = = t

С

(Note 1) (Note 2) (Note 3)

(t=1 for apply watering every hour)

Road Segments	average hourly daytime traffic rate (d)	Average control efficiency (%)
EP19_1	56	93.1
EP19_2	38	95.3
EP19_3	38	95.3
EP19_4	38	95.3
EP19_5	38	95.3
EP19_6	38	95.3
EP19_7	38	95.3
EP19_8	20	97.5
EP19_9	10	98.8
EP19_10	10	98.8
EP19_11	10	98.8
EP19_12	10	98.8
EP19_13	10	98.8
EP19 14	18	97.8

# SP Licence Application for Concrete Batching Plant at TKO 137 Area

inounoui	USEPA AP-42, Section 13.2.1.3, Version 1/2011				
	Particulate Emission Factor (g/VKT), E where	=	k (sL) <sup>0.91</sup> (W)	1.02	
	k	=	particle size	multiplier	
	sL W	=	road surface average wei	silt loading ght of the vehicles tr	aveling the road
From Concrete True	-be				
Mitigation measures:	water spraying and wheel washing				
	Average Loaded Weight of Truck Average Unladden Weight of Trucks	=	30 15	tons tons	
	Average Weight of Truck	=	22.5	tons	<i></i>
	k	=	3.23 0.62	g/VKT (for TSP) g/VKT (for RSP)	(Note 1)
		=	0.15	g/VKT (for FSP)	(11 - 0)
	sL = road surface silt loading	=	12	g/m	(Note 2)
EP19_1	Control Efficiency	_	93.1	%	(Note 3)
	Trucks Travelling Frequency	=	20	veh/hr	(Note 3) (Note 4)
	Road Length	=	15.7	m m <sup>2</sup>	
	Area of paved road travelled	=	47.1		
	RSP Emission Factor (uncontrolled) of paved road	=	142.4534 9.8829	g/VKT g/VKT	
		=	9.88E-03	g/veh/m	
		=	1.98E-01 5 49E-05	g/m/hr g/m/s	
		=	8.62E-04	g/s	
		=	<u>1.83E-05</u>	g/s/m²	
	FSP Emission Factor (uncontrolled) of paved road	=	34.4645	g/VKT	
	FSP Emission Factor (controlled) of paved road	=	2.3910 2.39F-03	g/VKT g/veh/m	
		=	4.78E-02	g/m/hr	
		=	1.33E-05 2.09E-04	g/m/s g/s	
		=	<u>4.43E-06</u>	g/s/m <sup>2</sup>	
EP19_2					
	Control Efficiency	=	95.3	%	(Note 3)
	Road Length	=	20.2	m	(Note 4)
	Area of paved road travelled	=	60.6	m²	
	RSP Emission Factor (uncontrolled) of paved road	=	142.4534	g/VKT	
	RSP Emission Factor (controlled) of paved road	=	6.7062 6.71E-03	g/VKT g/veb/m	
		=	1.34E-01	g/m/hr	
		=	3.73E-05 7.53E-04	g/m/s g/s	
		=	<u>1.242E-05</u>	g/s/m <sup>2</sup>	
	FSP Emission Factor (uncontrolled) of paved road	=	34.465	g/VKT	
	FSP Emission Factor (controlled) of paved road	=	1.6225	g/VKT	
		=	1.62E-03 3.24E-02	g/veh/m g/m/hr	
		=	9.01E-06	g/m/s	
		=	<u>3.00E-06</u>	g/s g/s/m <sup>2</sup>	
ED10 2					
LF19_3	Control Efficiency	=	95.3	%	(Note 3)
	Trucks Travelling Frequency Road Length	=	20 19.9	veh/hr m	(Note 4)
	Area of paved road travelled	=	59.7	m <sup>2</sup>	
	RSP Emission Factor (uncontrolled) of paved road	=	142.4534	g/VKT	
	RSP Emission Factor (controlled) of paved road	=	6.7062	g/VKT	
		=	6.71E-03 1.34E-01	g/veh/m g/m/hr	
		=	3.73E-05	g/m/s	
		=	7.41E-04 1.242E-05	g/s g/s/m <sup>2</sup>	
			<u></u>		
	FSP Emission Factor (uncontrolled) of paved road FSP Emission Factor (controlled) of paved road	=	34.465 1.6225	g/VKT g/VKT	
		=	1.62E-03	g/veh/m	
		=	3.24E-02 9.01E-06	g/m/nr g/m/s	
		=	1.79E-04	g/s	
		=	<u>3.00E-06</u>	g/s/m <sup>-</sup>	
EP19_4	Control Efficiency	-	92.5	%	(Note 2)
	Trucks Travelling Frequency	=	20	veh/hr	(Note 4)
	Road Length	=	9.7 20 1	m m <sup>2</sup>	
	Area or paved road travelled	-	27.1		
	RSP Emission Factor (uncontrolled) of paved road RSP Emission Factor (controlled) of paved road	=	142.4534 6.7062	g/VKT g/VKT	
		_	6 71F-03	g/veh/m	

		-	1 2/F-01	a/m/hr	
		_	2 725 05	g/11/11	
		=	3.73E-05	g/m/s	
		=	3.61E-04	g/s	
		=	<u>1.242E-05</u>	g/s/m²	
	FSP Emission Factor (uncontrolled) of paved road	=	34.465	g/VKT	
	FSP Emission Factor (controlled) of paved road	=	1.6225	g/VKT	
		=	1.62E-03	g/veh/m	
		=	3.24E-02	g/m/hr	
		=	9.01E-06	g/m/s	
		_	9.012-00	g/11/3	
		=	8.74E-05	g/s	
		=	<u>3.00E-06</u>	g/s/m²	
EP19_5					
	Control Efficiency	=	95.3	%	(Note 3)
	Trucks Travelling Frequency	=	20	veh/hr	(Note 4)
	Road Length	=	11.3	m	
	Area of payed road travelled	=	33.9	m <sup>2</sup>	
	···· ··· ··· · · · · · · · · · · · · ·				
	BSP Emission Factor (uncontrolled) of payed road	=	142 4534	g/VKT	
	PSP Emission Eactor (controlled) of paved road	_	6 7062	g/VKT	
	Nor Emission ractor (controlled) of paved road	_	0.7002	g/vici	
		=	6.71E-03	g/ven/m	
		=	1.34E-01	g/m/hr	
		=	3.73E-05	g/m/s	
		=	4.21E-04	g/s	
		=	1.242E-05	g/s/m <sup>2</sup>	
	FSP Emission Factor (uncontrolled) of paved road	=	34,465	g/VKT	
	ESP Emission Factor (controlled) of paved road	=	1.6225	g/VKT	
		-	1 62F-03	g/veh/m	
		-	1.02L-03	6/ v C 1/ 111	
		-	3.24E-02	g/11)/Nř	
		=	9.01E-06	g/m/s	
		=	1.02E-04	g/s	
		=	3.00E-06	g/s/m <sup>2</sup>	
EP19_6					
-	Control Efficiency	=	95.3	%	(Note 3)
	Trucks Travelling Frequency	-	20	veb/br	(Note 4)
	Deed Leasth	_	12.1	venijin	(NOLE 4)
	Road Length	-	13.1	2	
	Area of paved road travelled	=	39.3	mf	
	RSP Emission Factor (uncontrolled) of paved road	=	142.4534	g/VKT	
	RSP Emission Factor (controlled) of paved road	=	6.7062	g/VKT	
		=	6.71E-03	g/veh/m	
		=	1.34E-01	g/m/hr	
		-	3 73E-05	g/m/s	
		_	4 88E-04	g/11/3 g/s	
		-	4.00E-04	g/s	
		=	<u>1.242E-05</u>	g/s/m	
	FSP Emission Factor (uncontrolled) of paved road	=	34.465	g/VKT	
	FSP Emission Factor (controlled) of paved road	=	1.6225	g/VKT	
		=	1.62E-03	g/veh/m	
		=	3.24E-02	g/m/hr	
		=	9.01E-06	g/m/s	
		=	1.18E-04	g/s	
		=	3 00E-06	$a/s/m^2$	
			<u>3.002 00</u>	g/ 3/ 111	
ED10 7					
EP19_7	Control Efficiency	_	05.2	0/	(Nata 2)
		=	95.3	%	(Note 3)
	Trucks Travelling Frequency	=	20	ven/hr	(Note 4)
	Road Length	=	34.6	m	
	Area of paved road travelled	=	103.8	m	
	RSP Emission Factor (uncontrolled) of paved road	=	142.4534	g/VKT	
	RSP Emission Factor (controlled) of paved road	=	6.7062	g/VKT	
		=	6.71E-03	g/veh/m	
		=	1.34E-01	g/m/hr	
		=	3.73F-05	g/m/s	
		_	1 205-03	a/s	
		-	1.2.200-000	6/3 ala/2	
		=	<u>1.242E-05</u>	g/s/m <sup>-</sup>	
	FSP Emission Factor (uncontrolled) of paved road	=	34.465	g/VKT	
	FSP Emission Factor (controlled) of paved road	=	1.6225	g/VKT	
		=	1.62E-03	g/veh/m	
		=	3.24E-02	g/m/hr	
		=	9.01E-06	g/m/s	
		=	3,12F-04	g/s	
		-	3 005 00	a/s/m <sup>2</sup>	
		-	3.00E-06	g/ 3/111	
ED10 8					
EN13 <sup>8</sup>	· · · · · · · · · · · · · · · · · · ·		07 F	0/	(11-1-2)
	Control Efficiency	=	97.5	%	(Note 3)
	Trucks Travelling Frequency	=	20	veh/hr	(Note 4)
	Road Length	=	10.0	m	
	Area of paved road travelled	=	30.0	m <sup>2</sup>	
	RSP Emission Factor (uncontrolled) of paved road	=	142.4534	g/VKT	
	RSP Emission Factor (controlled) of paved road	=	3 5296	g/VKT	
		-	3 535-03	g/veh/m	
		-	7 065 02	a/m/hr	
		-	1 065 05	6/ 11/ 11 a/m/c	
		-	1.90E-05	g/11)/5	
		=	1.96E-04	g/s	
		=	<u>6.536E-06</u>	g/s/m²	
	FSP Emission Factor (uncontrolled) of paved road	=	34.465	g/VKT	

		=	8.54E-04	g/veh/m	
		=	1.71E-02	g/m/hr	
		=	4.74E-06	g/m/s	
		=	4.74E-05	g/s	
		=	<u>1.58E-06</u>	g/s/m²	
EP19_9	Control Efficiency	_	00.0	0/	(Nets 2)
	Trucks Travelling Frequency	-	98.8	% veh/hr	(Note 3)
	Road Length	=	10.3	m	(10000 4)
	Area of paved road travelled	=	30.9	m <sup>2</sup>	
			50.5		
	RSP Emission Factor (uncontrolled) of paved road	=	142.4534	g/VKT	
	RSP Emission Factor (controlled) of paved road	=	3.5296	g/VKT	
		=	3.53E-03	g/veh/m	
		=	3.53E-02	g/m/hr	
		=	9.80E-06	g/m/s	
		=	1.01E-04	g/s	
		=	<u>3.268E-06</u>	g/s/m²	
	ESP Emission Eactor (uncontrolled) of payed road	_	34 465	α/\/KT	
	FSP Emission Factor (controlled) of paved road	_	0.8539	g/VKT	
		=	8.54E-04	g/veh/m	
		=	8.54E-03	g/m/hr	
		=	2.37E-06	g/m/s	
		=	2.44E-05	g/s	
		=	7.91E-07	g/s/m <sup>2</sup>	
EP19_10					(11.1.2)
	Control Efficiency	-	98.8	% voh/hr	(Note 3)
	Irucks Iravelling Frequency	-	16.2	wen/nr	(Note 4)
	Area of payed road travelled	_	10.5	m <sup>2</sup>	
	Area or paved road travelled	-	40.9		
	RSP Emission Factor (uncontrolled) of paved road	=	142.4534	g/VKT	
	RSP Emission Factor (controlled) of paved road	=	3.5296	g/VKT	
		=	3.53E-03	g/veh/m	
		=	3.53E-02	g/m/hr	
		=	9.80E-06	g/m/s	
		=	1.60E-04	g/s	
		=	3.268E-06	g/s/m²	
	FCD Emission Factor (uncentralled) of pound road	_	24.405	- h///T	
	FSP Emission Factor (uncontrolled) of paved road	-	34.465	g/VKI g/VKT	
	FSP Emission Factor (controlled) of paved road	_	0.8539 8 54F=04	g/vki g/veh/m	
		=	8.54E-04	g/w/hr	
		=	2.37E-06	g/m/s	
		=	3.87E-05	g/s	
		=	<u>7.91E-07</u>	g/s/m <sup>2</sup>	
EP19_11					
	Control Efficiency	=	98.8	%	(Note 3)
	Trucks Travelling Frequency	-	10	ven/nr	(Note 4)
	Area of payed road travelled	-	14.4	m <sup>2</sup>	
	Area of paved road travelled	-	43.2		
	RSP Emission Eactor (uncontrolled) of payed road	=	142.4534	g/VKT	
	RSP Emission Factor (controlled) of paved road	=	3.5296	g/VKT	
		=	3.53E-03	g/veh/m	
		=	3.53E-02	g/m/hr	
		=	9.80E-06	g/m/s	
		=	1.41E-04	g/s	
		=	1.41E-04 <u>3.268E-06</u>	g/s g/s/m²	
	ECD Emission Easter (secondary) - Consultant	=	1.41E-04 <u>3.268E-06</u>	g/s g/s/m <sup>2</sup>	
	FSP Emission Factor (uncontrolled) of paved road	= = =	1.41E-04 <u>3.268E-06</u> 34.465 0.8529	g/s g/s/m <sup>2</sup> g/VKT g/VKT	
	FSP Emission Factor (uncontrolled) of paved road FSP Emission Factor (controlled) of paved road	= = = =	1.41E-04 3.268E-06 34.465 0.8539 8.54F-04	g/s g/vKT g/vKT g/vkT	
	FSP Emission Factor (uncontrolled) of paved road FSP Emission Factor (controlled) of paved road	= = = =	1.41E-04 <u>3.268E-06</u> 34.465 0.8539 8.54E-04 8.54E-03	g/s g/s/m <sup>2</sup> g/VKT g/VKT g/veh/m g/m/hr	
	FSP Emission Factor (uncontrolled) of paved road FSP Emission Factor (controlled) of paved road	= = = = =	1.41E-04 <u>3.268E-06</u> 34.465 0.8539 8.54E-04 8.54E-03 2.37E-06	g/s g/vKT g/vKT g/veh/m g/m/hr g/m/s	
	FSP Emission Factor (uncontrolled) of paved road FSP Emission Factor (controlled) of paved road	= = = = = =	1.41E-04 3.268E-06 34.465 0.8539 8.54E-04 8.54E-03 2.37E-06 3.42E-05	g/s g/vKT g/vKT g/veh/m g/m/hr g/m/s g/n/s	
	FSP Emission Factor (uncontrolled) of paved road FSP Emission Factor (controlled) of paved road	= = = = = = =	1.41E-04 3.268E-06 34.465 0.8539 8.54E-04 8.54E-03 2.37E-06 3.42E-05 7.91E-07	g/s g/s/m <sup>2</sup> g/VKT g/VKT g/veh/m g/m/hr g/m/s g/s g/s/m <sup>2</sup>	
	FSP Emission Factor (uncontrolled) of paved road FSP Emission Factor (controlled) of paved road		1.41E-04 <u>3.268E-06</u> 34.465 0.8539 8.54E-04 8.54E-03 2.37E-06 3.42E-05 <u>7.91E-07</u>	g/s g/s/m <sup>2</sup> g/VKT g/VKT g/veh/m g/m/hr g/m/hr g/m/s g/s g/s/m <sup>2</sup>	
EP19_12	FSP Emission Factor (uncontrolled) of paved road FSP Emission Factor (controlled) of paved road		1.41E-04 <u>3.268E-06</u> 34.465 0.8539 8.54E-04 8.54E-03 2.37E-06 3.42E-05 <u>7.91E-07</u>	g/s g/s/m <sup>2</sup> g/VKT g/VKT g/veh/m g/m/hr g/m/hr g/m/s g/s g/s/m <sup>2</sup>	
EP19_12	FSP Emission Factor (uncontrolled) of paved road FSP Emission Factor (controlled) of paved road Control Efficiency		1.41E-04 3.268E-06 34.465 0.8539 8.54E-04 8.54E-03 2.37E-06 3.42E-05 7.91E-07 98.8	g/s g/s/m <sup>2</sup> g/VKT g/VKT g/veh/m g/m/hr g/m/s g/s g/s/m <sup>2</sup>	(Note 3)
EP19_12	FSP Emission Factor (uncontrolled) of paved road FSP Emission Factor (controlled) of paved road Control Efficiency Trucks Traveling Bord Location		1.41E-04 <u>3.268E-06</u> 34.465 0.8539 8.54E-04 8.54E-03 2.37E-06 3.42E-05 <u>7.91E-07</u> 98.8 10 15 0	g/s g/s/m <sup>2</sup> g/VKT g/VKT g/vkh/m g/m/hr g/m/s g/s g/s g/s/m <sup>2</sup> % veh/hr m	(Note 3) (Note 4)
EP19_12	FSP Emission Factor (uncontrolled) of paved road FSP Emission Factor (controlled) of paved road Control Efficiency Trucks Travelling Frequency Road Length		1.41E-04 <u>3.268E-06</u> 0.8539 8.54E-04 8.54E-03 2.37E-06 3.42E-05 <u>7.91E-07</u> 98.8 10 15.9 4.7 7.7	g/s g/s/m <sup>2</sup> g/VKT g/VKT g/veh/m g/m/hr g/m/hr g/s/s g/s/m <sup>2</sup> % veh/hr m m <sup>2</sup>	(Note 3) (Note 4)
EP19_12	FSP Emission Factor (uncontrolled) of paved road FSP Emission Factor (controlled) of paved road Control Efficiency Trucks Travelling Frequency Road Length Area of paved road travelled		1.41E-04 <u>3.268E-06</u> 34.465 0.8539 8.54E-04 8.54E-03 2.37E-06 3.42E-05 7.91E-07 98.8 10 15.9 47.7	g/s g/s/m <sup>2</sup> g/VKT g/VKT g/veh/m g/m/hr g/m/s g/s/s/g/s/m <sup>2</sup> % veh/hr m m <sup>2</sup>	(Note 3) (Note 4)
EP19_12	FSP Emission Factor (uncontrolled) of paved road FSP Emission Factor (controlled) of paved road Control Efficiency Trucks Travelling Frequency Road Length Area of paved road travelled RSP Emission Factor (uncontrolled) of paved road		1.41E-04 3.268E-06 34.465 0.8539 8.54E-04 8.54E-03 2.37E-05 7.91E-07 98.8 10 15.9 47.7 142 4534	g/s g/s/m <sup>2</sup> g/VKT g/VKT g/veh/m g/m/hr g/m/s g/s g/s/m <sup>2</sup> % veh/hr m m <sup>2</sup> g/VKT	(Note 3) (Note 4)
EP19_12	FSP Emission Factor (uncontrolled) of paved road FSP Emission Factor (controlled) of paved road Control Efficiency Trucks Travelling Frequency Road Length Area of paved road travelled RSP Emission Factor (uncontrolled) of paved road RSP Emission Factor (controlled) of paved road		1.41E-04 <u>3.268E-06</u> 34.465 0.8539 8.54E-03 8.54E-03 2.37E-06 3.42E-05 7.91E-07 98.8 10 15.9 47.7 142.4534 3.5296	g/s g/s/m <sup>2</sup> g/VKT g/VKT g/veh/m g/m/hr g/m/s g/s/s g/s/m <sup>2</sup> % veh/hr m m <sup>2</sup> g/VKT g/VKT	(Note 3) (Note 4)
EP19_12	FSP Emission Factor (uncontrolled) of paved road FSP Emission Factor (controlled) of paved road Control Efficiency Trucks Travelling Frequency Road Length Area of paved road travelled RSP Emission Factor (uncontrolled) of paved road RSP Emission Factor (uncontrolled) of paved road		1.41E-04 <u>3.268E-06</u> 34.465 0.8539 8.54E-04 8.54E-03 2.37E-06 3.42E-05 <u>7.91E-07</u> 98.8 10 15.9 47.7 142.4534 3.5296 3.53E-03	g/s g/s/m <sup>2</sup> g/VKT g/VKT g/vkh/m g/m/hr g/m/s g/s/s g/s/m <sup>2</sup> % veh/hr m m <sup>2</sup> g/VKT g/VKT g/VKT g/VKT	(Note 3) (Note 4)
EP19_12	FSP Emission Factor (uncontrolled) of paved road FSP Emission Factor (controlled) of paved road Control Efficiency Trucks Travelling Frequency Road Length Area of paved road travelled RSP Emission Factor (uncontrolled) of paved road RSP Emission Factor (controlled) of paved road		1.41E-04 <u>3.268E-06</u> 34.465 0.8539 8.54E-04 8.54E-03 2.37E-06 3.42E-05 <u>7.91E-07</u> 98.8 10 15.9 47.7 142.4534 3.53E-03 3.53E-03	g/s g/s/m <sup>2</sup> g/VKT g/VKT g/veh/m g/m/hr g/m/s g/s g/s g/s/m <sup>2</sup> % veh/hr m m <sup>2</sup> g/VKT g/VKT g/VKT g/veh/m g/m/hr	(Note 3) (Note 4)
EP19_12	FSP Emission Factor (uncontrolled) of paved road FSP Emission Factor (controlled) of paved road Control Efficiency Trucks Travelling Frequency Road Length Area of paved road travelled RSP Emission Factor (uncontrolled) of paved road RSP Emission Factor (controlled) of paved road		1.41E-04 <u>3.268E-06</u> 3.4.465 0.8539 8.54E-04 8.54E-04 8.54E-03 2.37E-06 3.42E-05 7.91E-07 98.8 10 15.9 47.7 142.4534 3.5296 3.53E-02 3.53E-02 9.80E-06	g/s g/s/m <sup>2</sup> g/VKT g/VKT g/veh/m g/m/hr g/s g/s/m <sup>2</sup> % veh/hr m m <sup>2</sup> g/VKT g/VKT g/veh/m g/veh/m g/m/hr g/m/s	(Note 3) (Note 4)
EP19_12	FSP Emission Factor (uncontrolled) of paved road FSP Emission Factor (controlled) of paved road Control Efficiency Trucks Travelling Frequency Road Length Area of paved road travelled RSP Emission Factor (uncontrolled) of paved road RSP Emission Factor (controlled) of paved road		1.41E-04 <u>3.268E-06</u> 34.465 0.8539 8.54E-04 8.54E-03 2.37E-06 3.42E-05 7.91E-07 98.88 10 15.9 47.7 142.4534 3.5296 3.53E-03 3.53E-03 3.53E-03 9.80E-06 1.56E-04	g/s g/s/m <sup>2</sup> g/VKT g/VKT g/veh/m g/m/hr g/m/s g/s/m <sup>2</sup> % veh/hr m m <sup>2</sup> g/VKT g/VKT g/VKT g/VKT g/vh/m g/m/hr g/m/s g/s	(Note 3) (Note 4)
EP19_12	FSP Emission Factor (uncontrolled) of paved road FSP Emission Factor (controlled) of paved road Control Efficiency Trucks Travelling Frequency Road Length Area of paved road travelled RSP Emission Factor (uncontrolled) of paved road RSP Emission Factor (controlled) of paved road		1.41E-04 <u>3.268E-06</u> 34.465 0.8539 8.54E-03 8.54E-03 3.42E-05 <u>7.91E-07</u> 9.8.8 10 15.9 47.7 142.4534 3.5296 3.53E-03 3.53E-02 9.80E-04 <u>3.268E-06</u>	g/s g/s/m <sup>2</sup> g/VKT g/VKT g/veh/m g/m/s g/s/m <sup>2</sup> % veh/hr m m <sup>2</sup> g/VKT g/VKT g/veh/m g/m/hr g/VKT g/veh/m g/m/s g/s/m <sup>2</sup>	(Note 3) (Note 4)
EP19_12	FSP Emission Factor (uncontrolled) of paved road FSP Emission Factor (controlled) of paved road Control Efficiency Trucks Travelling Frequency Road Length Area of paved road travelled RSP Emission Factor (uncontrolled) of paved road RSP Emission Factor (controlled) of paved road		1.41E-04 <u>3.268E-06</u> 34.465 0.8539 8.54E-04 8.54E-05 <u>2.37E-06</u> 3.42E-05 <u>7.91E-07</u> 98.8 10 15.9 47.7 142.4534 3.5296 3.53E-03 3.53E-02 9.80E-06 1.56E-06	g/s g/s/m <sup>2</sup> g/VKT g/VKT g/whr g/m/hr g/m/s g/s/m <sup>2</sup> % veh/hr m m <sup>2</sup> g/VKT g/VKT g/VKT g/VKT g/veh/m g/m/hr g/s g/s/m <sup>2</sup>	(Note 3) (Note 4)
EP19_12	FSP Emission Factor (uncontrolled) of paved road FSP Emission Factor (controlled) of paved road Control Efficiency Trucks Travelling Frequency Road Length Area of paved road travelled RSP Emission Factor (uncontrolled) of paved road RSP Emission Factor (uncontrolled) of paved road		1.41E-04 <u>3.268E-06</u> 34.465 0.8539 8.54E-04 8.54E-03 2.37E-06 3.42E-05 <u>7.91E-07</u> 98.8 10 15.9 47.7 142.4534 3.5266 1.56E-04 <u>3.268E-06</u> 3.4455 0.015	g/s g/s/m <sup>2</sup> g/VKT g/VKT g/veh/m g/m/hr g/m/s g/s/s g/s/m <sup>2</sup> % veh/hr m m <sup>2</sup> g/VKT g/VKT g/VKT g/veh/m g/m/hr g/m/s g/s/m <sup>2</sup>	(Note 3) (Note 4)
EP19_12	FSP Emission Factor (uncontrolled) of paved road FSP Emission Factor (controlled) of paved road Control Efficiency Trucks Travelling Frequency Road Length Area of paved road travelled RSP Emission Factor (uncontrolled) of paved road RSP Emission Factor (controlled) of paved road FSP Emission Factor (uncontrolled) of paved road		1.41E-04 <u>3.268E-06</u> 3.4.465 0.8539 8.54E-04 8.54E-03 2.37E-06 3.42E-05 <u>7.91E-07</u> 98.8 10 15.9 47.7 142.4534 3.53E-03 3.53E-03 3.53E-04 <u>3.268E-06</u> 3.4465 0.8539 8.54E-04	g/s g/s/m <sup>2</sup> g/VKT g/VKT g/veh/m g/m/hr g/m/s g/s g/s/m <sup>2</sup> % veh/hr m <sup>2</sup> g/VKT g/VKT g/veh/m g/v/kT g/s/m <sup>2</sup> g/s/m <sup>2</sup>	(Note 3) (Note 4)
EP19_12	FSP Emission Factor (uncontrolled) of paved road FSP Emission Factor (controlled) of paved road Control Efficiency Trucks Travelling Frequency Road Length Area of paved road travelled RSP Emission Factor (uncontrolled) of paved road RSP Emission Factor (controlled) of paved road FSP Emission Factor (uncontrolled) of paved road FSP Emission Factor (uncontrolled) of paved road		1.41E-04 <u>3.268E-06</u> 3.4.465 0.8539 8.54E-04 8.54E-03 2.37E-06 3.42E-05 <u>7.91E-07</u> 98.8 10 15.9 47.7 142.4534 3.5296 3.53E-02 3.53E-02 3.54E-04 <u>3.268E-06</u> 8.54E-04 8.54E-04	g/s g/s/m <sup>2</sup> g/VKT g/VKT g/veh/m g/m/hr g/m/s g/s/m <sup>2</sup> % % veh/hr m m <sup>2</sup> g/VKT g/VKT g/VKT g/VKT g/VKT g/VKT g/VKT g/VKT	(Note 3) (Note 4)
EP19_12	FSP Emission Factor (uncontrolled) of paved road FSP Emission Factor (controlled) of paved road Control Efficiency Trucks Travelling Frequency Road Length Area of paved road travelled RSP Emission Factor (uncontrolled) of paved road RSP Emission Factor (controlled) of paved road FSP Emission Factor (uncontrolled) of paved road		1.41E-04 3.268E-06 34.465 0.8539 8.54E-03 8.54E-03 3.42E-05 2.37E-06 3.42E-05 7.91E-07 9.88 10 15.9 47.7 142.4534 3.5296 3.53E-03 3.53E-03 3.53E-03 3.53E-04 3.4685 0.8539 8.54E-04 8.54E-03 2.37E-06	g/s g/s/m <sup>2</sup> g/VKT g/VKT g/veh/m g/m/hr g/m/s g/s/m <sup>2</sup> % veh/hr m m <sup>2</sup> g/VKT g/VKT g/VKT g/veh/m g//KT g/VKT g/VKT g/VKT g/VKT g/VKT g/VKT	(Note 3) (Note 4)
EP19_12	FSP Emission Factor (uncontrolled) of paved road FSP Emission Factor (controlled) of paved road Control Efficiency Trucks Travelling Frequency Road Length Area of paved road travelled RSP Emission Factor (uncontrolled) of paved road RSP Emission Factor (controlled) of paved road FSP Emission Factor (uncontrolled) of paved road FSP Emission Factor (uncontrolled) of paved road		1.41E-04 3.268E-06 34.465 0.8539 8.54E-03 2.37E-06 3.42E-05 7.91E-07 98.8 10 98.8 10 98.8 10 9.7 142.4534 3.5266 3.53E-03 3.53E-03 3.53E-03 3.53E-03 3.53E-04 3.56E-04 3.54E-03 2.37E-05 3.77E-05	g/s g/s/m <sup>2</sup> g/VKT g/VKT g/veh/m g/m/hr g/m/hr g/s g/s/m <sup>2</sup> % veh/hr m m <sup>2</sup> g/VKT g/VKT g/veh/m g/m/hr g/s g/s/m <sup>2</sup> g/VKT g/vKT g/vKT g/vKT g/vKT g/vKT g/vKT	(Note 3) (Note 4)
EP19_12	FSP Emission Factor (uncontrolled) of paved road FSP Emission Factor (controlled) of paved road Control Efficiency Trucks Travelling Frequency Road Length Area of paved road travelled RSP Emission Factor (uncontrolled) of paved road RSP Emission Factor (controlled) of paved road FSP Emission Factor (uncontrolled) of paved road FSP Emission Factor (controlled) of paved road		1.41E-04 <u>3.268E-06</u> 34.465 0.8539 8.54E-04 8.54E-05 <u>2.37E-06</u> 3.42E-05 <u>7.91E-07</u> 98.8 10 15.9 47.7 142.4534 3.5296 3.53E-03 3.55E-03 3.55E-03 3.55E-03 3.55E-03 3.55E-03 3.5	g/s g/s/m <sup>2</sup> g/VKT g/VKT g/veh/m g/m/hr g/m/s g/s/m <sup>2</sup> % veh/hr m m <sup>2</sup> g/VKT g/VKT g/VKT g/VKT g/veh/m g/m/hr g/s g/VKT g/VKT g/VKT g/VKT g/VKT g/VKT g/VKT g/VKT g/VKT	(Note 3) (Note 4)

Control Efficiency		00.0	0/	(1)
Control Efficiency	=	98.8	%	(Note
Trucks Travelling Frequency	=	10	veh/hr	(Note -
Road Length	=	24.0	m	
Area of paved road travelled	=	72.0	m <sup>2</sup>	
RSP Emission Factor (uncontrolled) of paved road	=	142.4534	g/VKT	
RSP Emission Factor (controlled) of paved road	=	3.5296	g/VKT	
	=	3.53E-03	g/veh/m	
	=	3.53E-02	g/m/hr	
	=	9.80E-06	g/m/s	
	=	2.35E-04	g/s	
	=	<u>3.268E-06</u>	g/s/m <sup>2</sup>	
FSP Emission Factor (uncontrolled) of payed road	=	34,465	g/VKT	
FSP Emission Factor (controlled) of paved road	=	0.8539	g/VKT	
· · · · · · · · · · · · · · · · · · ·	=	8.54E-04	g/veh/m	
	=	8.54E-03	g/m/hr	
	=	2.37E-06	g/m/s	
	=	5.69E-05	g/s	
	=	7 91F-07	g/s/m <sup>2</sup>	
EP19 14			0, ,	
– Control Efficiency	=	97.8	%	(Note
Trucks Travelling Frequency	=	0	veh/hr	(Note
Road Length	=	29.0	m	
Area of paved road travelled	=	87.0	m <sup>2</sup>	
RSP Emission Factor (uncontrolled) of paved road	=	142.4534	g/VKT	
RSP Emission Factor (controlled) of paved road	=	3.5296	g/VKI	
	=	3.53E-03	g/veh/m	
	=	U.UUE+00	g/m/nr	
	=	U.U0E+00	g/m/s	
	=	0.00E+00	g/s	
	=	0.000E+00	g/s/m²	
ECD Emission Easter (uncontrolled) of payed and	-	34 465	α/\/KT	
FSP Emission Factor (uncontrolled) of paved road	-	0 8500	g/ VNI	
FSP Emission Factor (controlled) of paved road	-	0.0000	g/VNI	
	=	8.54E-04	g/ven/m	
	=	0.00E+00	g/m/nr	
	=	0.00E+00	g/m/s	
	=	0.00E+00	g/s	
	-	0.00E+00	g/s/m	
From Aggregate Truck				
Mitigation measures: water spraying and wheel washing				
Average Loaded Weight of Concrete Lorry Mixer	=	30	tons	
Average Weight of Concrete Lorry Mixer	=	15	tons	
Average Unladden Weight of Concrete Lorry Mixer	=	22.5	tons	
k	=	3.23	g/VKT (for TSP)	(Note
	=	0.62	g/VKT (for RSP)	
	=	0.15	g/VKT (for FSP)	
sL = road surface silt loading	=	12	g/m²	(Note
EP19_1				
Control Efficiency	=	93.1	%	(Note
Trucks Travelling Frequency	=	4	ven/nr	(Note
Koad Length	=	15.7	m 2	
Area of paved road travelled	=	47.1	m	
RSP Emission Factor (uncontrolled) of paved road	=	142.4534	g/VKT	
RSP Emission Factor (controlled) of paved road	=	9.8829	g/VKT	
	=	9.88E-03	g/veh/m	
	=	3.95E-02	g/m/hr	
	=	1.10E-05	g/m/s	
	=	1.72E-04	g/s	
	=	<u>3.66E-06</u>	g/s/m <sup>2</sup>	
		34 4645	g/VKT	
FSP Emission Factor (uncontrolled) of paved road	=	2 2010	CT ( ) / K	
FSP Emission Factor (uncontrolled) of paved road FSP Emission Factor (controlled) of paved road	=	2.3910	g/viti	
FSP Emission Factor (uncontrolled) of paved road FSP Emission Factor (controlled) of paved road	=	2.3910 2.39E-03	g/veh/m	
FSP Emission Factor (uncontrolled) of paved road FSP Emission Factor (controlled) of paved road	= = =	2.3910 2.39E-03 9.56E-03	g/veh/m g/m/hr	
FSP Emission Factor (uncontrolled) of paved road FSP Emission Factor (controlled) of paved road	= = = =	2.3910 2.39E-03 9.56E-03 2.66E-06	g/vkl g/veh/m g/m/hr g/m/s	
FSP Emission Factor (uncontrolled) of paved road FSP Emission Factor (controlled) of paved road	= = = = =	2.3910 2.39E-03 9.56E-03 2.66E-06 4.17E-05	g/vkh g/veh/m g/m/hr g/m/s g/s	
FSP Emission Factor (uncontrolled) of paved road FSP Emission Factor (controlled) of paved road	= = = = =	2.3910 2.39E-03 9.56E-03 2.66E-06 4.17E-05 <u>8.86E-07</u>	g/vk1 g/whn g/m/hr g/m/s g/s g/s/m <sup>2</sup>	
FSP Emission Factor (uncontrolled) of paved road FSP Emission Factor (controlled) of paved road	= = = = =	2.3910 2.39E-03 9.56E-03 2.66E-06 4.17E-05 <u>8.86E-07</u>	g/veh/m g/m/hr g/m/s g/s g/s/m <sup>2</sup>	
FSP Emission Factor (uncontrolled) of paved road FSP Emission Factor (controlled) of paved road EP19_2 Control Efficiency	= = = = =	2.3910 2.39E-03 9.56E-03 2.66E-06 4.17E-05 <u>8.86E-07</u> 95.3	g/veh/m g/m/hr g/m/s g/s g/s/m <sup>2</sup>	(Note :
FSP Emission Factor (uncontrolled) of paved road FSP Emission Factor (controlled) of paved road EP19_2 Control Efficiency Trucks Travelling Frequency		2.3910 2.39E-03 9.56E-03 2.66E-06 4.17E-05 <u>8.86E-07</u> 95.3 2	g/veh/m g/w/hr g/m/hr g/m/s g/s g/s/m <sup>2</sup> % veh/hr	(Note 3 (Note 4
FSP Emission Factor (uncontrolled) of paved road FSP Emission Factor (controlled) of paved road EP19_2 Control Efficiency Trucks Travelling Frequency Road Length	= = = = = = = = = =	2.3910 2.39E-03 9.56E-03 2.66E-06 4.17E-05 <u>8.86E-07</u> 95.3 2 20.2	g/veh/m g/m/hr g/m/s g/s g/s/m <sup>2</sup> % veh/hr m	(Note - (Note -
FSP Emission Factor (uncontrolled) of paved road FSP Emission Factor (controlled) of paved road EP19_2 Control Efficiency Trucks Travelling Frequency Road Length Area of paved road travelled		2.3910 2.392-03 9.56E-03 2.66E-06 4.17E-05 <u>8.86E-07</u> 95.3 2 20.2 60.6	g/veh/m g/m/hr g/m/s g/s g/s/m <sup>2</sup> % veh/hr m m <sup>2</sup>	(Note (Note -
FSP Emission Factor (uncontrolled) of paved road FSP Emission Factor (controlled) of paved road EP19_2 Control Efficiency Trucks Travelling Frequency Road Length Area of paved road travelled		2.3910 2.39E-03 9.56E-03 2.66E-06 4.17E-05 <u>8.86E-07</u> 95.3 2 20.2 60.6	g/veh/m g/m/hr g/m/s g/s g/s/m <sup>2</sup> % veh/hr m m <sup>2</sup>	(Note - (Note -
FSP Emission Factor (uncontrolled) of paved road FSP Emission Factor (controlled) of paved road EP19_2 Control Efficiency Trucks Travelling Frequency Road Length Area of paved road travelled RSP Emission Factor (uncontrolled) of paved road		2.3910 2.39E-03 9.56E-03 2.66E-06 4.17E-05 <u>8.86E-07</u> 95.3 2 95.3 2 20.2 60.6 142.4534	g/veh/m g/m/hr g/m/s g/s g/s/m <sup>2</sup> % veh/hr m m <sup>2</sup> g/VKT	(Note : (Note -
FSP Emission Factor (uncontrolled) of paved road FSP Emission Factor (controlled) of paved road EP19_2 Control Efficiency Trucks Travelling Frequency Road Length Area of paved road travelled RSP Emission Factor (uncontrolled) of paved road RSP Emission Factor (controlled) of paved road		2.3910 2.392-03 9.56E-03 2.66E-06 4.17E-05 <u>8.86E-07</u> 95.3 2 20.2 60.6 142.4534 6.7062	g/veh/m g/m/hr g/m/s g/s g/s/m <sup>2</sup> % veh/hr m m <sup>2</sup> g/VKT g/VKT	(Note (Note
FSP Emission Factor (uncontrolled) of paved road FSP Emission Factor (controlled) of paved road EP19_2 Control Efficiency Trucks Travelling Frequency Road Length Area of paved road travelled RSP Emission Factor (uncontrolled) of paved road RSP Emission Factor (controlled) of paved road		2.3910 2.392-03 9.56E-03 2.66E-06 4.17E-05 8.86E-07 95.3 2 20.2 60.6 142.4534 6.7062 6.71E-03	g/veh/m g/m/hr g/m/s g/s g/s/m <sup>2</sup> % veh/hr m m <sup>2</sup> g/VKT g/VKT g/vE/m	(Note (Note
EP19_2 EP19_2 Control Efficiency Trucks Travelling Frequency Road Length Area of paved road travelled RSP Emission Factor (uncontrolled) of paved road RSP Emission Factor (controlled) of paved road		2.3910 2.392-03 9.56E-03 2.66E-06 4.17E-05 <u>8.86E-07</u> 95.3 2 20.2 60.6 142.4534 6.7062 6.71E-03 1.34E-02	g/veh/m g/m/hr g/m/s g/s g/s/m <sup>2</sup> % veh/hr m m <sup>2</sup> g/VKT g/VKT g/VKT g/veh/m g/m/hr	(Note (Note
FSP Emission Factor (uncontrolled) of paved road FSP Emission Factor (controlled) of paved road EP19_2 Control Efficiency Trucks Travelling Frequency Road Length Area of paved road travelled RSP Emission Factor (uncontrolled) of paved road RSP Emission Factor (controlled) of paved road		2.3910 2.39E-03 9.56E-03 2.66E-06 4.17E-05 <u>8.86E-07</u> 95.3 2 20.2 60.6 142.4534 6.7062 6.71E-03 1.34E-02 3.73E-06	g/veh/m g/m/hr g/m/s g/s g/s/m <sup>2</sup> % veh/hr m <sup>2</sup> g/VKT g/VKT g/VKT g/VKT g/vFh/m g/m/hr g/m/s	(Note (Note
FSP Emission Factor (uncontrolled) of paved road FSP Emission Factor (controlled) of paved road EP19_2 Control Efficiency Trucks Travelling Frequency Road Length Area of paved road travelled RSP Emission Factor (uncontrolled) of paved road RSP Emission Factor (controlled) of paved road		2.3910 2.392-03 9.56E-03 2.66E-03 2.66E-06 4.17E-05 8.86E-07 95.3 2 20.2 60.6 142.4534 6.7062 6.71E-03 1.34E-02 3.73E-06	g/veh/m g/m/hr g/m/s g/s g/s/m <sup>2</sup> % veh/hr m m <sup>2</sup> g/VKT g/vKT g/vKT g/vKT g/w/hr g/m/hr g/m/s g/s	(Note (Note
FSP Emission Factor (uncontrolled) of paved road FSP Emission Factor (controlled) of paved road EP19_2 Control Efficiency Trucks Travelling Frequency Road Length Area of paved road travelled RSP Emission Factor (uncontrolled) of paved road RSP Emission Factor (controlled) of paved road		2.3910 2.392-03 9.56E-03 2.66E-06 4.17E-05 8.86E-07 95.3 2 20.2 60.6 142.4534 6.7062 6.71E-03 1.34E-02 3.73E-05 1.24E-06	g/veh/m g/m/hr g/m/s g/s g/s/m <sup>2</sup> % veh/hr m m <sup>2</sup> g/VKT g/veh/m g/m/kr g/m/s g/s/m <sup>2</sup>	(Note (Note
FSP Emission Factor (uncontrolled) of paved road FSP Emission Factor (controlled) of paved road EP19_2 Control Efficiency Trucks Travelling Frequency Road Length Area of paved road travelled RSP Emission Factor (uncontrolled) of paved road RSP Emission Factor (controlled) of paved road		2.3910 2.392-03 9.562-03 2.662-06 4.17E-05 <u>8.86E-07</u> 95.3 2 20.2 60.6 142.4534 6.7062 6.71E-03 1.34E-02 3.73E-06 7.53E-05 1.24E-06	g/veh/m g/m/hr g/m/s g/s g/s/m <sup>2</sup> % veh/hr m m <sup>2</sup> g/VKT g/VKT g/VKT g/veh/m g/m/hr g/m/s g/s/m <sup>2</sup>	(Note (Note
FSP Emission Factor (uncontrolled) of paved road FSP Emission Factor (controlled) of paved road EP19_2 Control Efficiency Trucks Travelling Frequency Road Length Area of paved road travelled RSP Emission Factor (uncontrolled) of paved road RSP Emission Factor (uncontrolled) of paved road		2.3910 2.392-03 9.562-03 2.662-06 2.662-06 8.8662-07 95.3 2 20.2 60.6 142.4534 6.7062 6.71E-03 1.34E-02 3.73E-06 7.53E-05 1.24E-06 34.46645	g/vkh/m g/m/hr g/m/s g/s g/s/m <sup>2</sup> % veh/hr m m <sup>2</sup> g/VKT g/vkh/m g/w/hr g/s g/s/m <sup>2</sup> g/VKT	(Note (Note
EP19_2 Control Efficiency Trucks Travelling Frequency Road Length Area of paved road travelled RSP Emission Factor (uncontrolled) of paved road RSP Emission Factor (controlled) of paved road FSP Emission Factor (uncontrolled) of paved road		2.3910 2.3920 2.3920 2.662-03 2.662-06 4.172-05 8.8662-07 95.3 2 20.2 60.6 142.4534 6.7062 6.712-03 1.342-02 3.732-06 7.532-05 1.242-06 34.6645 1.6225	g/veh/m g/m/hr g/m/s g/s g/s/m <sup>2</sup> % veh/hr m m <sup>2</sup> g/VKT g/vkT g/vkT g/vkT g/m/hr g/r/s g/s g/s/m <sup>2</sup>	(Note (Note
FSP Emission Factor (uncontrolled) of paved road FSP Emission Factor (controlled) of paved road EP19_2 Control Efficiency Trucks Travelling Frequency Road Length Area of paved road travelled RSP Emission Factor (uncontrolled) of paved road RSP Emission Factor (controlled) of paved road FSP Emission Factor (uncontrolled) of paved road		2.3910 2.3920 2.39203 9.562-03 2.662-06 4.17E-05 8.862-07 95.3 2 20.2 60.6 142.4534 6.7062 6.71E-03 1.34E-02 3.73E-06 7.53E-05 1.24E-06 34.4645 1.6225 1.6225	g/veh/m g/m/hr g/m/s g/s g/s/m <sup>2</sup> % veh/hr m <sup>2</sup> g/VKT g/veh/m g/m/hr g/m/s g/s g/s/m <sup>2</sup> g/VKT g/veh/m g/VKT g/VKT	(Note (Note
FSP Emission Factor (uncontrolled) of paved road FSP Emission Factor (controlled) of paved road EP19_2 Control Efficiency Trucks Travelling Frequency Road Length Area of paved road travelled RSP Emission Factor (uncontrolled) of paved road RSP Emission Factor (controlled) of paved road FSP Emission Factor (uncontrolled) of paved road		2.3910 2.392-03 9.562-03 2.662-06 4.17E-05 <u>8.86E-07</u> 95.3 2 20.2 60.6 142.4534 6.7062 6.71E-03 1.34E-02 3.73E-06 1.24E-06 34.4645 1.6225 1.622E-03 3.24E-03	g/veh/m g/m/hr g/m/s g/s g/s/m <sup>2</sup> % veh/hr m m <sup>2</sup> g/VKT g/veh/m g/m/hr g/m/s g/s g/s/m <sup>2</sup> g/VKT g/vKT g/vKT g/vk/m	(Note (Note
FSP Emission Factor (uncontrolled) of paved road FSP Emission Factor (controlled) of paved road Control Efficiency Trucks Travelling Frequency Road Length Area of paved road travelled RSP Emission Factor (uncontrolled) of paved road RSP Emission Factor (controlled) of paved road FSP Emission Factor (uncontrolled) of paved road FSP Emission Factor (uncontrolled) of paved road		2.3910 2.3910 2.39E-03 9.56E-03 2.66E-06 8.86E-07 95.3 2 20.2 60.6 142.4534 6.7062 6.71E-03 1.34E-02 3.73E-06 1.34E-02 3.73E-06 1.24E-06 34.4645 1.62E-03 3.24E-03 9.01E-07 4.027 - 2 1.62E-03	g/veh/m g/m/hr g/m/s g/s g/s/m <sup>2</sup> % veh/hr m m <sup>2</sup> g/VKT g/vKT g/vKT g/vKT g/s/m <sup>2</sup> g/S/m <sup>2</sup> g/VKT g/VKT g/VKT g/VKT g/VKT g/VKT	(Note - (Note -
FSP Emission Factor (uncontrolled) of paved road FSP Emission Factor (controlled) of paved road Control Efficiency Trucks Travelling Frequency Road Length Area of paved road travelled RSP Emission Factor (uncontrolled) of paved road RSP Emission Factor (controlled) of paved road FSP Emission Factor (uncontrolled) of paved road		2.3910 2.3910 2.39E-03 9.56E-03 2.66E-06 4.17E-05 8.86E-07 95.3 2 20.2 60.6 142.4534 6.7062 6.71E-03 1.34E-02 3.73E-06 7.53E-05 1.24E-06 34.4645 1.6225 1.6225 1.6225-03 3.24E-03 3.24E-03	g/veh/m g/m/hr g/m/s g/s g/s/m <sup>2</sup> % veh/hr m <sup>2</sup> g/VKT g/vKT g/vKT g/vKT g/vKT g/vKT g/vKT g/vKT g/vKT g/vKT	(Note

	Control Efficiency	=	95.3	%	(Note 3)
	Trucks Travelling Frequency	=	2	veh/hr	(Note 4)
	Road Length	=	19.9	m	( )
	Area of payed read travelled	_	E0 7	m <sup>2</sup>	
	Area of paved road travelled	=	59.7		
				h	
	RSP Emission Factor (uncontrolled) of paved road	=	142.4534	g/VKI	
	RSP Emission Factor (controlled) of paved road	=	6.7062	g/VKT	
		=	6.71E-03	g/veh/m	
		=	1.34E-02	g/m/hr	
		=	3.73E-06	g/m/s	
		=	7.41E-05	g/s	
		=	1 24F-06	a/s/m <sup>2</sup>	
			1121200	5/ 5/ 11	
	ESP Emission Eactor (uncontrolled) of payed road	_	34 4645	a ///KT	
	ESP Emission Factor (controlled) of paved road	_	1 6225	g/VKT	
	FSP Emission Factor (controlled) of paved road	_	1.0225	g/vki	
		=	1.62E-03	g/ven/m	
		=	3.24E-03	g/m/hr	
		=	9.01E-07	g/m/s	
		=	1.79E-05	g/s	
		=	3.00E-07	g/s/m²	
EP19_4					
-	Control Efficiency	=	95.3	%	(Note 3)
	Trucks Travelling Frequency	=	2	veh/hr	(Note 4)
	Boad Length	=	97	m	(
	Area of payed read travelled	_	20.1	m <sup>2</sup>	
	Area of paved road travelled	=	29.1	m	
	DCD Facilities Factors (concentrally d) of a conduct d		442 4524	- 6 0/7	
	KSP Emission Factor (uncontrolled) of paved road	=	142.4534	g/VKI	
	RSP Emission Factor (controlled) of paved road	=	6.7062	g/VKT	
		=	6.71E-03	g/veh/m	
		=	1.34E-02	g/m/hr	
		=	3.73E-06	g/m/s	
		=	3.61E-05	g/s	
		=	1.24E-06	g/s/m <sup>2</sup>	
				J	
	ESP Emission Factor (uncontrolled) of paved road	=	34 4645	g/VKT	
	ESP Emission Factor (controlled) of paved road	-	1 6225	g/VKT	
	FSP Emission Factor (controlled) of paved road	_	1.0225	g/vki	
		=	1.62E-03	g/ven/m	
		=	3.24E-03	g/m/hr	
		=	9.01E-07	g/m/s	
		=	8.74E-06	g/s	
		=	3.00E-07	g/s/m <sup>2</sup>	
EP19_5					
	Control Efficiency	=	95.3	%	(Note 3)
	Trucks Travelling Frequency	=	2	veh/hr	(Note 4)
	Road Length	=	11.3	m	
	Area of payed road travelled	_	22.0	m <sup>2</sup>	
	Alea of paved toad travelled	-	55.9		
				h	
	RSP Emission Factor (uncontrolled) of paved road	=	142.4534	g/VKI	
	RSP Emission Factor (controlled) of paved road	=	6.7062	g/VKT	
		=	6.71E-03	g/veh/m	
		=	1.34E-02	g/m/hr	
		=	3.73E-06	g/m/s	
		=	4.21E-05	g/s	
		-	1 24F-06	g/s/m <sup>2</sup>	
		_		8/ •/ ···	
		-			
	ESD Emission Enter (uncentrolled) of payed read	_	24 4645	a h/kt	
	FSP Emission Factor (uncontrolled) of paved road	=	34.4645	g/VKT	
	FSP Emission Factor (uncontrolled) of paved road FSP Emission Factor (controlled) of paved road	-	34.4645 1.6225	g/VKT g/VKT	
	FSP Emission Factor (uncontrolled) of paved road FSP Emission Factor (controlled) of paved road	= = =	34.4645 1.6225 1.62E-03	g/VKT g/VKT g/veh/m	
	FSP Emission Factor (uncontrolled) of paved road FSP Emission Factor (controlled) of paved road	= = =	34.4645 1.6225 1.62E-03 3.24E-03	g/VKT g/VKT g/veh/m g/m/hr	
	FSP Emission Factor (uncontrolled) of paved road FSP Emission Factor (controlled) of paved road		34.4645 1.6225 1.62E-03 3.24E-03 9.01E-07	g/VKT g/VKT g/veh/m g/m/hr g/m/s	
	FSP Emission Factor (uncontrolled) of paved road FSP Emission Factor (controlled) of paved road		34.4645 1.6225 1.62E-03 3.24E-03 9.01E-07 1.02E-05	g/VKT g/VKT g/veh/m g/m/hr g/m/s g/s	
	FSP Emission Factor (uncontrolled) of paved road FSP Emission Factor (controlled) of paved road		34.4645 1.6225 1.62E-03 3.24E-03 9.01E-07 1.02E-05 <u>3.00E-07</u>	g/VKT g/VKT g/veh/m g/m/hr g/m/s g/s g/s	
	FSP Emission Factor (uncontrolled) of paved road FSP Emission Factor (controlled) of paved road		34.4645 1.6225 1.62E-03 3.24E-03 9.01E-07 1.02E-05 <u>3.00E-07</u>	g/VKT g/VKT g/veh/m g/m/hr g/m/s g/s g/s	
EP19_6	FSP Emission Factor (uncontrolled) of paved road FSP Emission Factor (controlled) of paved road		34.4645 1.6225 1.62E-03 3.24E-03 9.01E-07 1.02E-05 <u>3.00E-07</u>	g/VKT g/VKT g/veh/m g/m/hr g/m/s g/s g/s	
EP19_6	FSP Emission Factor (uncontrolled) of paved road FSP Emission Factor (controlled) of paved road Control Efficiency		34.4645 1.6225 1.62E-03 3.24E-03 9.01E-07 1.02E-05 <u>3.00E-07</u> 95.3	g/VKT g/VKT g/veh/m g/m/hr g/m/s g/s g/s/m <sup>2</sup>	(Note 3)
EP19_6	FSP Emission Factor (uncontrolled) of paved road FSP Emission Factor (controlled) of paved road Control Efficiency Trucks Travelling Frequency		34.4645 1.6225 1.62E-03 3.24E-03 9.01E-07 1.02E-05 <u>3.00E-07</u> 95.3 2	g/VKT g/VkT g/veh/m g/m/s g/s g/s/m <sup>2</sup> % veh/hr	(Note 3) (Note 4)
EP19_6	FSP Emission Factor (uncontrolled) of paved road FSP Emission Factor (controlled) of paved road Control Efficiency Trucks Travelling Frequency Road Length	-	34.4645 1.6225 1.62E-03 3.24E-03 9.01E-07 1.02E-05 <u>3.00E-07</u> 95.3 2 13.1	g/VKT g/VkT g/veh/m g/m/hr g/m/s g/s g/s g/s/m <sup>2</sup> % veh/hr m	(Note 3) (Note 4)
EP19_6	FSP Emission Factor (uncontrolled) of paved road FSP Emission Factor (controlled) of paved road Control Efficiency Trucks Travelling Frequency Road Length		34.4645 1.6225 1.62E-03 3.24E-03 9.01E-07 1.02E-05 <u>3.00E-07</u> 95.3 2 13.1 20.2	g/VKT g/VkT/m g/m/hr g/m/s g/s/m <sup>2</sup> % veh/hr m m <sup>2</sup>	(Note 3) (Note 4)
EP19_6	FSP Emission Factor (uncontrolled) of paved road FSP Emission Factor (controlled) of paved road Control Efficiency Trucks Travelling Frequency Road Length Area of paved road travelled		34.4645 1.6225 1.622E-03 3.24E-03 9.01E-07 1.02E-05 <u>3.00E-07</u> 95.3 2 13.1 39.3	g/VKT g/VkT g/veh/m g/m/hr g/m/s g/s g/s/m <sup>2</sup> % veh/hr m m <sup>2</sup>	(Note 3) (Note 4)
EP19_6	FSP Emission Factor (uncontrolled) of paved road FSP Emission Factor (controlled) of paved road Control Efficiency Trucks Travelling Frequency Road Length Area of paved road travelled		34.4645 1.6225 1.622-63 3.24E-03 9.01E-07 1.02E-05 <u>3.00E-07</u> 95.3 2 13.1 39.3	g/VKT g/VkT g/veh/m g/m/hr g/m/s g/s g/s/s g/s/m <sup>2</sup> % veh/hr m m <sup>2</sup>	(Note 3) (Note 4)
EP19_6	FSP Emission Factor (uncontrolled) of paved road FSP Emission Factor (controlled) of paved road Control Efficiency Trucks Travelling Frequency Road Length Area of paved road travelled RSP Emission Factor (uncontrolled) of paved road		34.4645 1.6225 1.62E-03 3.24E-03 9.01E-07 1.02E-05 <u>3.00E-07</u> 95.3 2 13.1 39.3 142.4534	g/VKT g/VkT g/veh/m g/m/hr g/m/hr g/s g/s g/s/s/m <sup>2</sup> % veh/hr m m <sup>2</sup> g/VKT	(Note 3) (Note 4)
EP19_6	FSP Emission Factor (uncontrolled) of paved road FSP Emission Factor (controlled) of paved road Control Efficiency Trucks Travelling Frequency Road Length Area of paved road travelled RSP Emission Factor (uncontrolled) of paved road RSP Emission Factor (controlled) of paved road		34.4645 1.6225 3.24E-03 9.01E-07 1.02E-05 <u>3.00E-07</u> 95.3 2 13.1 39.3 142.4534 6.7062	g/VKT g/VkT g/veh/m g/m/hr g/m/s g/s/g/s/m <sup>2</sup> % veh/hr m m <sup>2</sup> g/VKT g/VKT	(Note 3) (Note 4)
EP19_6	FSP Emission Factor (uncontrolled) of paved road FSP Emission Factor (controlled) of paved road Control Efficiency Trucks Travelling Frequency Road Length Area of paved road travelled RSP Emission Factor (uncontrolled) of paved road		34.4645 1.6225 1.622-03 3.24E-03 9.01E-07 1.02E-05 <u>3.00E-07</u> 95.3 2 13.1 39.3 142.4534 6.7062 6.71E-03	g/VKT g/VkT g/veh/m g/m/s g/s g/s/s/m <sup>2</sup> % veh/hr m m <sup>2</sup> g/VKT g/VKT g/VkT	(Note 3) (Note 4)
EP19_6	FSP Emission Factor (uncontrolled) of paved road FSP Emission Factor (controlled) of paved road Control Efficiency Trucks Travelling Frequency Road Length Area of paved road travelled RSP Emission Factor (uncontrolled) of paved road RSP Emission Factor (controlled) of paved road		34.4645 1.6225 1.62E-03 3.24E-03 9.01E-07 1.02E-05 <u>3.00E-07</u> 95.3 2 13.1 39.3 142.4534 6.7062 6.71E-03 1.34E-02	g/VKT g/Veh/m g/m/hr g/m/hr g/m/hr g/s g/s/s/m <sup>2</sup> % veh/hr m <sup>2</sup> g/VKT g/VKT g/VKT g/veh/m g/m/hr	(Note 3) (Note 4)
EP19_6	FSP Emission Factor (uncontrolled) of paved road FSP Emission Factor (controlled) of paved road Control Efficiency Trucks Travelling Frequency Road Length Area of paved road travelled RSP Emission Factor (uncontrolled) of paved road RSP Emission Factor (controlled) of paved road		34.4645 1.6225 1.622-03 3.24E-03 9.01E-07 1.02E-05 <u>3.00E-07</u> 95.3 2 13.1 39.3 142.4534 6.7062 6.71E-03 1.34E-02 3.73E-06	g/VKT g/VkT g/veh/m g/m/hr g/m/s g/s/m <sup>2</sup> % veh/hr m <sup>2</sup> g/VKT g/VkT g/VkT g/VkTm g/NkT	(Note 3) (Note 4)
EP19_6	FSP Emission Factor (uncontrolled) of paved road FSP Emission Factor (controlled) of paved road Control Efficiency Trucks Travelling Frequency Road Length Area of paved road travelled RSP Emission Factor (uncontrolled) of paved road RSP Emission Factor (controlled) of paved road		34.4645 1.6225 1.622-03 3.24E-03 9.01E-07 1.02E-05 <u>3.00E-07</u> 95.3 2 13.1 39.3 142.4534 6.7062 6.71E-03 1.34E-02 3.73E-06 4.88E-05	g/VKT g/VkT g/veh/m g/m/hr g/m/hr g/s g/s/m <sup>2</sup> % veh/hr m m <sup>2</sup> g/VKT g/VKT g/VKT g/vk/m g/m/hr g/m/s g/s	(Note 3) (Note 4)
EP19_6	FSP Emission Factor (uncontrolled) of paved road FSP Emission Factor (controlled) of paved road Control Efficiency Trucks Travelling Frequency Road Length Area of paved road travelled RSP Emission Factor (uncontrolled) of paved road RSP Emission Factor (controlled) of paved road		34.4645 1.6225 1.622-03 3.24E-03 9.01E-07 1.02E-05 <u>3.00E-07</u> 95.3 2 13.1 39.3 142.4534 6.7062 6.71E-03 1.34E-02 3.73E-06 4.88E-05 1.24E-06	g/VKT g/Veh/m g/m/hr g/m/hr g/m/hr g/s g/s/m <sup>2</sup> % veh/hr m <sup>2</sup> g/VKT g/VKT g/veh/m g/veh/m g/m/hr g/s/s g/s/m <sup>2</sup>	(Note 3) (Note 4)
EP19_6	FSP Emission Factor (uncontrolled) of paved road FSP Emission Factor (controlled) of paved road Control Efficiency Trucks Travelling Frequency Road Length Area of paved road travelled RSP Emission Factor (uncontrolled) of paved road RSP Emission Factor (controlled) of paved road		34.4645 1.6225 1.622-03 3.24E-03 9.01E-07 1.02E-05 3.00E-07 95.3 2 13.1 39.3 142.4534 6.7062 6.71E-03 1.34E-02 3.73E-06 4.88E-05 1.24E-06	g/VKT g/Veh/m g/m/hr g/m/s g/s g/s/s/m <sup>2</sup> % veh/hr m <sup>2</sup> g/VKT g/VKT g/VKT g/veh/m g/m/s g/m/s g/s/m <sup>2</sup>	(Note 3) (Note 4)
EP19_6	FSP Emission Factor (uncontrolled) of paved road FSP Emission Factor (controlled) of paved road Control Efficiency Trucks Travelling Frequency Road Length Area of paved road travelled RSP Emission Factor (uncontrolled) of paved road RSP Emission Factor (controlled) of paved road		34.4645 1.6225 1.622-03 3.24E-03 9.01E-07 1.02E-05 <u>3.00E-07</u> 95.3 2 13.1 39.3 142.4534 6.7062 6.71E-03 1.34E-02 3.73E-06 4.88E-05 <u>1.24E-06</u> 34.4645	g/VKT g/VkT g/veh/m g/m/hr g/m/hr g/s g/s/m <sup>2</sup> % veh/hr m m <sup>2</sup> g/VKT g/VKT g/VKT g/vkh/m g/m/hr g/m/s g/s/m <sup>2</sup>	(Note 3) (Note 4)
EP19_6	FSP Emission Factor (uncontrolled) of paved road FSP Emission Factor (controlled) of paved road Control Efficiency Trucks Travelling Frequency Road Length Area of paved road travelled RSP Emission Factor (uncontrolled) of paved road RSP Emission Factor (controlled) of paved road		34.4645 1.6225 1.62E-03 3.24E-03 9.01E-07 1.02E-05 <u>3.00E-07</u> 95.3 2 13.1 39.3 142.4534 6.7062 6.71E-03 1.34E-02 3.73E-06 4.88E-05 <u>1.24E-06</u> 34.4645 <u>1.6225</u>	g/VKT g/Veh/m g/m/hr g/m/hr g/m/hr g/s g/s/m <sup>2</sup> % veh/hr m <sup>2</sup> g/VKT g/veh/m g/veh/m g/veh/m g/m/hr g/s g/s/m <sup>2</sup>	(Note 3) (Note 4)
EP19_6	FSP Emission Factor (uncontrolled) of paved road FSP Emission Factor (controlled) of paved road Control Efficiency Trucks Travelling Frequency Road Length Area of paved road travelled RSP Emission Factor (uncontrolled) of paved road RSP Emission Factor (controlled) of paved road FSP Emission Factor (uncontrolled) of paved road		34.4645 1.6225 1.622-03 3.24E-03 9.01E-07 1.02E-05 3.00E-07 95.3 2 13.1 39.3 142.4534 6.7062 6.71E-03 1.34E-02 3.73E-06 4.88E-05 1.24E-06 34.4645 1.6225	g/VKT g/veh/m g/m/hr g/m/hr g/s g/s/s/m <sup>2</sup> % veh/hr m m <sup>2</sup> g/VKT g/vKT g/vKT g/vKT g/w/hr g/m/s g/s/m <sup>2</sup> g/VKT	(Note 3) (Note 4)
EP19_6	FSP Emission Factor (uncontrolled) of paved road FSP Emission Factor (controlled) of paved road Control Efficiency Trucks Travelling Frequency Road Length Area of paved road travelled RSP Emission Factor (uncontrolled) of paved road RSP Emission Factor (controlled) of paved road FSP Emission Factor (uncontrolled) of paved road		34.4645 1.6225 1.622-03 3.24E-03 9.01E-07 1.02E-05 <u>3.00E-07</u> 95.3 2 13.1 39.3 142.4534 6.7062 6.71E-03 1.34E-02 3.73E-06 4.88E-05 <u>1.22E-06</u> 34.4645 1.6225 1.6225-03	g/VKT g/VkT g/veh/m g/m/s g/s g/s/m <sup>2</sup> % veh/hr m m <sup>2</sup> g/VKT g/veh/m g/vKT g/vKT g/vKT g/s/kT g/s/kT g/vKT g/vKT	(Note 3) (Note 4)
EP19_6	FSP Emission Factor (uncontrolled) of paved road FSP Emission Factor (controlled) of paved road Control Efficiency Trucks Travelling Frequency Road Length Area of paved road travelled RSP Emission Factor (uncontrolled) of paved road RSP Emission Factor (controlled) of paved road FSP Emission Factor (uncontrolled) of paved road		34.4645 1.6225 1.622-03 3.24E-03 9.01E-07 1.02E-05 <u>3.00E-07</u> 95.3 2 13.1 39.3 142.4534 6.7062 6.71E-03 1.34E-02 3.73E-06 4.88E-05 <u>1.24E-06</u> 34.4645 1.6225 1.62E-03 3.24E-03	g/VKT g/Veh/m g/m/br g/m/s g/s g/s/s/m <sup>2</sup> % veh/hr m <sup>2</sup> g/VKT g/veh/m g/vkT g/veh/m g/m/hr g/s g/s/KT g/VKT g/VKT	(Note 3) (Note 4)
EP19_6	FSP Emission Factor (uncontrolled) of paved road FSP Emission Factor (controlled) of paved road Control Efficiency Trucks Travelling Frequency Road Length Area of paved road travelled RSP Emission Factor (uncontrolled) of paved road RSP Emission Factor (controlled) of paved road FSP Emission Factor (uncontrolled) of paved road		34.4645 1.6225 1.622-03 3.24E-03 9.01E-07 1.02E-05 3.00E-07 95.3 2 13.1 39.3 142.4534 6.7062 6.71E-03 1.34E-02 3.73E-06 4.88E-05 1.6225 1.6225 1.6225 1.6225 1.6225 3.24E-03 3	g/VKT g/Veh/m g/m/hr g/m/hr g/m/hr g/s g/s/s/m <sup>2</sup> % veh/hr m <sup>2</sup> g/VKT g/VKT g/VKT g/veh/m g/NKT g/VKT g/VKT g/VKT g/VKT	(Note 3) (Note 4)
EP19_6	FSP Emission Factor (uncontrolled) of paved road FSP Emission Factor (controlled) of paved road Control Efficiency Trucks Travelling Frequency Road Length Area of paved road travelled RSP Emission Factor (uncontrolled) of paved road RSP Emission Factor (controlled) of paved road FSP Emission Factor (uncontrolled) of paved road		34.4645 1.6225 1.622-03 3.24E-03 9.01E-07 1.02E-05 3.00E-07 95.3 2 13.1 39.3 142.4534 6.7062 6.71E-03 1.34E-02 3.73E-06 4.88E-05 1.6225 1.6225 1.6225 1.6225 1.6225 1.6225 1.6225 1.6225 1.6225 1.6225 1.6225 1.6225 1.6225 1.6225 1.6225 1.622-03 3.24E-03 9.01E-07 1.18E-05	g/VKT g/VkT g/veh/m g/m/s g/s g/s/m <sup>2</sup> % veh/hr m m <sup>2</sup> g/VKT g/vkh/m g/vKT g/vkh/m g/s/s/m <sup>2</sup> g/s/m <sup>2</sup>	(Note 3) (Note 4)
EP19_6	FSP Emission Factor (uncontrolled) of paved road FSP Emission Factor (controlled) of paved road Control Efficiency Trucks Travelling Frequency Road Length Area of paved road travelled RSP Emission Factor (uncontrolled) of paved road RSP Emission Factor (controlled) of paved road FSP Emission Factor (uncontrolled) of paved road		34.4645 1.6225 1.622-03 3.24E-03 9.01E-07 1.02E-05 3.00E-07 95.3 2 13.1 39.3 142.4534 6.7062 6.71E-03 1.34E-02 3.73E-06 1.24E-06 34.4645 1.6225 1.62E-03 3.24E-03 3.24E-03 3.24E-03 3.04E-07 1.18E-05 3.00E-07	g/VKT g/Vktr g/vkh/m g/m/hr g/m/hr g/s g/s/m <sup>2</sup> <sup>%</sup> veh/hr m <sup>2</sup> g/VKT g/vkh/m g/vkh/m g/w/hr g/s/s g/s/m <sup>2</sup>	(Note 3) (Note 4)
EP19_6	FSP Emission Factor (uncontrolled) of paved road FSP Emission Factor (controlled) of paved road Control Efficiency Trucks Travelling Frequency Road Length Area of paved road travelled RSP Emission Factor (uncontrolled) of paved road RSP Emission Factor (controlled) of paved road FSP Emission Factor (uncontrolled) of paved road		34.4645 1.6225 1.622-03 3.24E-03 9.01E-07 1.02E-05 3.00E-07 95.3 2 13.1 39.3 142.4534 6.70E2 6.71E-03 1.34E-02 3.73E-06 4.88E-05 1.24E-06 34.4645 1.6225 1.622E-03 3.24E-03 3.24E-03 9.01E-07 1.18E-05 3.00E-07	g/VKT g/Veh/m g/m/hr g/m/hr g/m/s g/s g/s/m <sup>2</sup> % veh/hr m <sup>2</sup> g/VKT g/VKT g/veh/m g/w/hr g/s/m <sup>2</sup> g/VKT g/VKT g/VKT g/VKT g/VKT g/VKT g/VKT g/VKT g/VKT g/VKT g/VKT	(Note 3) (Note 4)
EP19_6 EP19_7	FSP Emission Factor (uncontrolled) of paved road FSP Emission Factor (controlled) of paved road Control Efficiency Trucks Travelling Frequency Road Length Area of paved road travelled RSP Emission Factor (uncontrolled) of paved road RSP Emission Factor (controlled) of paved road FSP Emission Factor (uncontrolled) of paved road		34.4645 1.6225 1.622-03 3.24E-03 9.01E-07 1.022-05 3.00E-07 95.3 2 13.1 39.3 142.4534 6.7062 6.71E-03 1.34E-02 3.73E-06 4.88E-05 1.6225 1.6225 1.6225 1.6225 1.6225 1.6225 1.6225 1.6225 1.62E-03 3.24E-03 3.24E-03 3.24E-03 3.24E-03 3.24E-03 3.24E-03 3.24E-03 3.24E-03 3.24E-03 3.24E-03 3.24E-03 3.24E-03 3.24E-03 3.24E-03 3.73E-06 3.74E-05 3.24E-03 3.24E-03 3.24E-03 3.24E-03 3.24E-03 3.24E-03 3.24E-03 3.24E-03 3.24E-05 3.24E-	g/VKT g/Veh/m g/m/hr g/m/hr g/s g/s/s g/s/m <sup>2</sup> % veh/hr m <sup>2</sup> g/VKT g/VKT g/VKT g/veh/m g/N/r g/VKT g/VKT g/VKT g/VKT g/VKT g/VKT g/VKT g/VKT g/VKT g/VKT g/VKT	(Note 3) (Note 4)
EP19_6 EP19_7	<form><form></form></form>		34.4645 1.6225 1.622-03 3.24E-03 9.01E-07 1.02E-05 3.00E-07 95.3 2 13.1 39.3 142.4534 6.7062 6.71E-03 1.34E-02 3.73E-06 4.88E-05 1.6225 1.62E-03 3.24E-03 9.01E-07 1.18E-05 3.00E-07 95.3	g/VKT g/Vktr g/vkh/m g/m/hr g/m/hr g/s g/s/m <sup>2</sup> % veh/hr m <sup>2</sup> g/VKT g/VKT g/vkh/m g/s g/s/m <sup>2</sup> g/VKT g/vktm g/vKT g/vKT g/vktm g/vkttm g/vktm} g/vktm	(Note 3) (Note 4)
EP19_6 EP19_7	FSP Emission Factor (uncontrolled) of paved road         FSP Emission Factor (controlled) of paved road         Control Efficiency         Trucks Travelling Frequency         Rad Length         Area of paved road travelled         RSP Emission Factor (uncontrolled) of paved road         RSP Emission Factor (controlled) of paved road		34.4645 1.6225 1.622-03 3.24E-03 9.01E-07 1.022-05 3.00E-07 95.3 2 13.1 39.3 142.4534 6.7062 6.71E-03 1.34E-02 3.73E-06 4.88E-05 1.622E-03 3.24E-03 9.01E-07 1.18E-05 3.00E-07 95.3 2	g/VKT g/Veh/m g/m/hr g/m/s g/s g/s/s/m <sup>2</sup> % veh/hr m <sup>2</sup> g/VKT g/VKT g/VKT g/veh/m g/vkT g/VKT	(Note 3) (Note 4) (Note 3) (Note 4)
EP19_6 EP19_7	FSP Emission Factor (uncontrolled) of paved road FSP Emission Factor (controlled) of paved road Control Efficiency Trucks Travelling Frequency Road Length Area of paved road travelled RSP Emission Factor (uncontrolled) of paved road RSP Emission Factor (controlled) of paved road FSP Emission Factor (controlled) of paved road FSP Emission Factor (controlled) of paved road		34.4645 1.6225 1.622-03 3.24E-03 9.01E-07 1.02E-05 3.00E-07 95.3 2 13.1 39.3 142.4534 6.7062 6.71E-03 1.24E-06 34.4645 1.6225 1.626-03 3.04E-03 3.04E-03 9.01E-07 1.18E-05 3.00E-07 95.3 2 3.4.6	g/VKT g/Veh/m g/m/hr g/m/hr g/s g/s/s/m <sup>2</sup> % veh/hr m <sup>2</sup> g/VKT g/VKT g/VKT g/VKT g/veh/m g/N/kT g/VKT	(Note 3) (Note 4) (Note 3) (Note 4)

	RSP Emission Eactor (uncontrolled) of payed road	=	142 4534	g/VKT	
	RSP Emission Factor (controlled) of paved road	_	6 7062	g/VKT	
	KSP Emission Factor (controlled) of paved road	_	6.71E-02	g/vKi	
		-	1.245.02	g/ven/hr	
		_	1.54E-02	g/11/11	
		-	3./3E-00	g/m/s	
		=	1.29E-04	g/s	
		=	<u>1.24E-06</u>	g/s/m <sup>-</sup>	
	FSP Emission Factor (uncontrolled) of payed road	=	34.4645	g/VKT	
	ESP Emission Factor (controlled) of paved road	=	1 6225	g/VKT	
		=	1.62E-03	g/veh/m	
		-	2 24E-02	g/wch/hr	
		-	9.01E-07	g/m/s	
		=	3.12E-05	g/m,5	
		=	3.00E-07	g/s/m <sup>2</sup>	
			<u></u>	8/ •/ ···	
EP19_14					
	Control Efficiency	=	97.8	%	(Note 3)
	Trucks Travelling Frequency	=	2	veh/hr	(Note 4)
	Road Length	=	29.0	m	
	Area of paved road travelled	=	87.0	m²	
			4 42 4524	- 6 0/7	
	RSP Emission Factor (uncontrolled) of paved road	=	142.4534	g/VKT	
	RSP Emission Factor (controlled) of paved road	=	3.1/66	g/VKI	
		=	3.18E-03	g/ven/m	
		=	0.35E-03	g/m/nr	
		=	1./0E-06	g/111/5	
		=	5.12E-05	g/5	
		=	5.88E-07	g/s/m	
1	ESP Emission Factor (uncontrolled) of naved road	=	34,4645	g/VKT	
	FSP Emission Factor (controlled) of paved road	=	0.7685	g/VKT	
		=	7.69E-04	g/veh/m	
		=	1.54E-03	g/m/hr	
		=	4.27E-07	g/m/s	
		=	1.24E-05	g/s	
		=	1.42E-07	g/s/m <sup>2</sup>	
				0.	
From Cementitious Tanker					
Mitigation measures: water spr	aying and wheel washing				
	Average Loaded Weight of Truck	=	38	tons	
	Average Unladden Weight of Trucks	=	12	tons	
	Average Weight of Trucks	=	25		
	k	=	3.23	g/VKT (for TSP)	(Note 1)
		=	0.62	g/VKT (for RSP)	
		=	0.15	g/VKT (for FSP)	
	sL = road surface silt loading	=	12	g/m2	(Note 2)
FD10_1					
EP19_1	Control Efficiency	_	02.1	0/	(Nata 2)
	Control Efficiency	-	93.1	% 	(Note 3)
	Trucks Travelling Frequency	=	15.7	ven/nr	(Note 4)
	Road Length	-	15.7	2	
	Area of paved road travelled	=	47.1	m	
	RSP Emission Factor (uncontrolled) of payed road	=	158.6154	g/VKT	
	RSP Emission Factor (controlled) of paved road	=	11.0041	g/VKT	
		=	1 10F-02	g/veh/m	
		=	0.00F+00	g/m/hr	
		=	0.00E+00	g/m/s	
		=	0.00E+00	g/11/3	
		=	0.00E+00	g/s/m <sup>2</sup>	
			<u>5.502+00</u>	0/ 5/	
	FSP Emission Factor (uncontrolled) of paved road	=	38.3747	g/VKT	
	FSP Emission Factor (controlled) of paved road	=	2.6623	g/VKT	
		=	2.66E-03	g/veh/m	
		=	0.00E+00	g/m/hr	
		=	0.00E+00	g/m/s	
		=	0.00E+00	g/s	
		=	0.00E+00	g/s/m²	
EP19_2	Control Effect	_	05.2	0/	(Nat - 2)
	Control Efficiency	=	95.3	70 veb/br	(Note 3)
	Trucks Travelling Frequency	-	U 20 2	ven/m	(NOTE 4)
	Kodu Length	_	20.2	m <sup>2</sup>	
	Area of paved road travelled	=	60.6	111	
1	RSP Emission Factor (uncontrolled) of naved road	=	158.6154	g/VKT	
	RSP Emission Factor (controlled) of paved road	=	7,4671	g/VKT	
		=	7.47E-03	g/veh/m	
		=	0.00E+00	g/m/hr	
		=	0.00E+00	g/m/s	
		=	0.00E+00	g/s	
		=	<u>0.00E+00</u>	g/s/m <sup>2</sup>	
	FSP Emission Factor (uncontrolled) of paved road	=	38.3747	g/VKT	
	FSP Emission Factor (controlled) of paved road	=	1.8066	g/VKT	
		=	1.81E-03	g/veh/m	
		=	0.00E+00	g/m/hr	
		=	0.00E+00	g/m/s	
		=	0.00E+00	g/s	
		=	<u>0.00E+00</u>	g/s/m <sup>2</sup>	
EP19 3					
	Control Efficiency	=	95.3	%	(Note 3)

	Road Length	=	19.9	m	
	Area of payed road travelled	=	59.7	m <sup>2</sup>	
	RSP Emission Factor (uncontrolled) of paved road	=	158.6154	g/VKT	
	RSP Emission Factor (controlled) of paved road	=	7.4671	g/VKT	
		=	7.47E-03	g/veh/m	
		=	0.00E+00	g/m/hr	
		=	0.00E+00	g/m/s	
		=	0.00E+00	g/s	
		=	0.00E+00	g/s/m <sup>2</sup>	
	FSP Emission Factor (uncontrolled) of paved road	=	38.3747	g/VKT	
	FSP Emission Factor (controlled) of paved road	=	1.8066	g/VKT	
		=	1.81E-03	g/veh/m	
		-	0.00E+00	g/m/m	
		-	0.00E+00	g/11/3 g/s	
		=	0.005+00	$a/s/m^2$	
			0.002+00	8/ 3/ 111	
EP19 4					
	Control Efficiency	=	95.3	%	(Note 3)
	Trucks Travelling Frequency	=	0	veh/hr	(Note 4)
	Road Length	=	9.7	m	
	Area of paved road travelled	=	29.1	m <sup>2</sup>	
	RSP Emission Factor (uncontrolled) of paved road	=	158.6154	g/VKT	
	RSP Emission Factor (controlled) of paved road	=	7.4671	g/VKT	
		=	7.47E-03	g/veh/m	
		=	0.00E+00	g/m/hr	
		=	0.00E+00	g/m/s	
		=	0.00E+00	g/s	
		=	<u>0.00E+00</u>	g/s/m²	
	FCD Emission Factor (manager P. D. C. 1997)		20 27	-	
	FOR EMISSION FACTOR (uncontrolled) of paved road	=	38.3/47	g/VKI	
	For Emission Factor (controlled) of paved road	-	1,815,02	g/VKI g/veb/~	
		-	1.01E-05	g/ven/n	
		-	0.00E+00	g/m/s	
		=	0.00E+00	g/11/3 g/s	
		=	0.00E+00	$a/s/m^2$	
			0.002100	g/ 3/111	
EP19 5					
-	Control Efficiency	=	95.3	%	(Note 3)
	Trucks Travelling Frequency	=	0	veh/hr	(Note 4)
	Road Length	=	11.3	m	
	Area of paved road travelled	=	33.9	m <sup>2</sup>	
	RSP Emission Factor (uncontrolled) of paved road	=	158.6154	g/VKT	
	RSP Emission Factor (controlled) of paved road	=	7.4671	g/VKT	
		=	7.47E-03	g/veh/m	
		=	0.00E+00	g/m/hr	
		=	0.00E+00	g/m/s	
		=	0.00E+00	g/s	
		=	0.00E+00	g/s/m <sup>-</sup>	
	FCD Emission Factor (uncentralied) of pound road	_	20 2747	~ h///T	
	FSP Emission Factor (uncontrolled) of paved road	-	1 8066	g/VKT	
	Tor Emission factor (controlled) of paved road	-	1.81F=03	g/vici g/veh/m	
		=	0.00E+00	g/weil/ill g/m/hr	
		=	0.00E+00	g/m/s	
		=	0.00E+00	g/s	
		=	0.00E+00	g/s/m <sup>2</sup>	
EP19_6					
	Control Efficiency	=	95.3	%	(Note 3)
	Trucks Travelling Frequency	=	0	veh/hr	(Note 4)
	Road Length	=	13.1	m	
	Area of paved road travelled	=	39.3	m	
			450 000	- 4	
	KSP Emission Factor (uncontrolled) of paved road	=	158.6154	g/VKT	
	KSP Emission Factor (controlled) of paved road	-	7.4671	g/VKI g/vob/	
		-	7.4/E-U3	g/ven/m	
		-	0.00E+00	g/11/11 g/m/s	
		=	0.00E+00	g/s	
		=	0.00E+00	g/s/m <sup>2</sup>	
			<u></u>	0/ -/	
	FSP Emission Factor (uncontrolled) of paved road	=	38.3747	g/VKT	
	FSP Emission Factor (controlled) of paved road	=	1.8066	g/VKT	
		=	1.81E-03	g/veh/m	
		=	0.00E+00	g/m/hr	
		=	0.00E+00	g/m/s	
		=	0.00E+00	g/s	
		=	<u>0.00E+00</u>	g/s/m <sup>2</sup>	
EP19_7			0= c	0/	(1)-( 2)
	Control Efficiency	=	95.3	%	(Note 3)
	Frequency Road Leagth	-	34.6	wen/nf m	(NOLE 4)
	Kodu Length	-	102.0	m <sup>2</sup>	
	Area or paved road travelled	=	103.8		
	RSP Emission Factor (uncontrolled) of paved road	=	158 6154	g/VKT	
	RSP Emission Factor (controlled) of paved road	=	7.4671	g/VKT	
		=	7.47E-03	g/veh/m	
		=	0.00E+00	g/m/hr	

	FSP Emission Factor (uncontrolled) of paved road	= = =	0.00E+00 0.00E+00 <u>0.00E+00</u> 38.3747	g/m/s g/s g/s/m <sup>2</sup> g/VKT	
	FSP Emission Factor (controlled) of paved road	=	1.8066	g/VKT	
		=	1.81E-03	g/veh/m	
		=	0.00E+00	g/m/hr	
		=	0.00E+00	g/m/s	
		=	0.00E+00	g/s	
		=	0.00E+00	g/s/m²	
EP19_14					
	Control Efficiency	=	97.8	%	(Note 3)
	Trucks Travelling Frequency	=	0	veh/hr	(Note 4)
	Koad Length	=	29.0	m 2	
	Area of paved road travelled	=	87.0	m <sup>-</sup>	
	RSP Emission Factor (uncontrolled) of paved road	=	158.6154	g/VKT	
	RSP Emission Factor (controlled) of paved road	=	3.5370	g/VKT	
		=	3.54E-03	g/veh/m	
		=	0.00E+00	g/m/hr	
		=	0.00E+00	g/m/s	
		=	0.00E+00	g/s	
		=	<u>0.00E+00</u>	g/s/m²	
	FSP Emission Factor (uncontrolled) of paved road	=	38.3747	g/VKT	
	FSP Emission Factor (controlled) of paved road	=	0.8557	g/VKT	
		=	8.56E-04	g/veh/m	
		=	0.00E+00	g/m/hr	
		=	0.00E+00	g/m/s	
		=	0.00E+00	g/s	
		=	0.00E+00	g/s/m²	

Note:

Note.
1. Particle size multiplier is referred to Table 13.2-1-1, Section 13.2.1 of USEPA AP42 (ver 1/11) for RSP (PM10) and FSP (PM2.5).
2. Road surfacing silt loading is referred to Table 13.2.1-3 of USEPA AP42 (ver 1/11) for concrete batching.
3. Percentage of dust mitigation efficiency is calculated with reference to Cowherd et al., "Control of Open Fugitive Dust Sources, EPA-450/3-88-008, U.S. Environmental Protection Agency, Research Triangle
4. The vehicle numbers within the site is advised by the applicant.

# SP Licence Application for Concrete Batching Plant at TKO 137 Area Calculation of Emissions from Paved Roads (0900 to 1900) Method: USEPA AP-42, Section 13.2.1.3, version 1/2011

Particulate Emission Factor (g/VKT), E k (sL) $^{0.91}$  (W) $^{1.02}$ = where particle size multiplier k = road surface silt loading sL = w \_ average weight of the vehicles traveling the road From Concrete Trucks Mitigation measures: water spraying and wheel washing Average Loaded Weight of Truck 30 tons = Average Unladden Weight of Trucks Average Weight of Trucks 15 tons 22.5 tons 3.23 g/VKT (for TSP) g/VKT (for RSP) g/VKT (for FSP) 0.62 0.15 g/m<sup>2</sup> sL = road surface silt loading 12 EP19\_1 Control Efficiency 93.1 % Trucks Travelling Frequency 20 veh/hr m m² Road Length = 15.7 Area of paved road travelled \_ 47.1 RSP Emission Factor (uncontrolled) of paved road 142.4534 g/VKT RSP Emission Factor (controlled) of paved road 9.8829 g/VKT 9.88E-03 g/veh/m 1.98E-01 g/m/hr 5.49E-05 g/m/s 8.62E-04 g/s <u>1.83E-05</u> g/s/m<sup>2</sup> FSP Emission Factor (uncontrolled) of paved road 34.4645 g/VKT FSP Emission Factor (controlled) of paved road 2.3910 g/VKT 2.39E-03 g/veh/m 4.78E-02 g/m/hr 1.33E-05 g/m/s = 2.09E-04 g/s = 4.43E-06 g/s/m<sup>2</sup> EP19\_2 Control Efficiency 95.3 % = Trucks Travelling Frequency = 20 veh/hr 20.2 Road Length = m

Area of paved road travelled

RSP Emission Factor (uncontrolled) of paved road RSP Emission Factor (controlled) of paved road

FSP Emission Factor (uncontrolled) of paved road

FSP Emission Factor (controlled) of paved road

(Note 1)

(Note 2)

(Note 3)

(Note 4)

(Note 3)

(Note 4)

m<sup>2</sup>

g/VKT

g/VKT

g/veh/m

g/m/hr

g/m/s

g/VKT

g/VKT

g/veh/m

g/m/hr

g/m/s

g/s 1.242E-05 g/s/m<sup>2</sup>

60.6

142.4534

6.7062

6.71E-03

1.34E-01

3.73E-05

7.53E-04

34.465

1.6225

1.62E-03

3.24E-02

9.01E-06

		=	1.82E-04	g/s	
		=	<u>3.00E-06</u>	g/s/m <sup>2</sup>	
EP19_3					
Con	trol Efficiency	=	95.3	%	(Note 3)
Trucks Travel	ling Frequency	=	20	veh/hr	(Note 4)
	Road Length	=	19.9	m	
Area of paved	road travelled	=	59.7	m <sup>2</sup>	
RSP Emission Factor (uncontrolled)	of paved road	=	142.4534	g/VKT	
RSP Emission Factor (controlled)	of paved road	=	6.7062	g/VKT	
		=	6.71E-03	g/veh/m	
		=	1.34E-01	g/m/hr	
		=	3.73E-05	g/m/s	
		=	7.41E-04	g/s	
		=	<u>1.242E-05</u>	g/s/m <sup>2</sup>	
FSP Emission Factor (uncontrolled)	of paved road	=	34.465	g/VKT	
FSP Emission Factor (controlled)	of paved road	=	1.6225	g/VKT	
		=	1.62E-03	g/veh/m	
		=	3.24E-02	g/m/hr	
		=	9.01E-06	g/m/s	
		=	1.79E-04	g/s	
		=	<u>3.00E-06</u>	g/s/m²	
EP19_4					
Con	trol Efficiency	=	95.3	%	(Note 3)
Trucks Travel	ling Frequency	=	20	veh/hr	(Note 4)
	Road Length	=	9.7	m	
Area of paved	road travelled	=	29.1	m <sup>2</sup>	
RSP Emission Factor (uncontrolled)	of paved road	=	142.4534	g/VKT	
RSP Emission Factor (controlled)	of paved road	=	6.7062	g/VKT	
		=	6.71E-03	g/veh/m	

		-	1 2/F-01	a/m/hr	
		_	2 725 05	g/11/11	
		=	3.73E-05	g/m/s	
		=	3.61E-04	g/s	
		=	<u>1.242E-05</u>	g/s/m²	
	FSP Emission Factor (uncontrolled) of paved road	=	34.465	g/VKT	
	FSP Emission Factor (controlled) of paved road	=	1.6225	g/VKT	
		=	1.62E-03	g/veh/m	
		=	3.24E-02	g/m/hr	
		=	9.01E-06	g/m/s	
		_	9.012-00	g/11/3	
		=	8.74E-05	g/s	
		=	<u>3.00E-06</u>	g/s/m²	
EP19_5					
	Control Efficiency	=	95.3	%	(Note 3)
	Trucks Travelling Frequency	=	20	veh/hr	(Note 4)
	Road Length	=	11.3	m	
	Area of payed road travelled	=	33.9	m <sup>2</sup>	
	···· ··· ··· · · · · · · · · · · · · ·				
	BSP Emission Factor (uncontrolled) of payed road	=	142 4534	g/VKT	
	PSP Emission Eactor (controlled) of paved road	_	6 7062	g/VKT	
	Nor Emission ractor (controlled) of paved road	_	0.7002	g/vici	
		=	6.71E-03	g/ven/m	
		=	1.34E-01	g/m/hr	
		=	3.73E-05	g/m/s	
		=	4.21E-04	g/s	
		=	1.242E-05	g/s/m <sup>2</sup>	
	FSP Emission Factor (uncontrolled) of paved road	=	34,465	g/VKT	
	ESP Emission Factor (controlled) of paved road	=	1.6225	g/VKT	
		-	1 62F-03	g/veh/m	
		-	1.02L-03	6/ v C 1/ 111	
		-	3.24E-02	g/m/nř	
		=	9.01E-06	g/m/s	
		=	1.02E-04	g/s	
		=	3.00E-06	g/s/m <sup>2</sup>	
EP19_6					
-	Control Efficiency	=	95.3	%	(Note 3)
	Trucks Travelling Frequency	-	20	veb/br	(Note 4)
	Deed Leasth	_	12.1	venijin	(NOLE 4)
	Road Length	-	13.1	2	
	Area of paved road travelled	=	39.3	mf	
	RSP Emission Factor (uncontrolled) of paved road	=	142.4534	g/VKT	
	RSP Emission Factor (controlled) of paved road	=	6.7062	g/VKT	
		=	6.71E-03	g/veh/m	
		=	1.34E-01	g/m/hr	
		-	3 73E-05	g/m/s	
		_	4 88E-04	g/11/3 g/s	
		-	4.00E-04	g/s	
		=	<u>1.242E-05</u>	g/s/m	
	FSP Emission Factor (uncontrolled) of paved road	=	34.465	g/VKT	
	FSP Emission Factor (controlled) of paved road	=	1.6225	g/VKT	
		=	1.62E-03	g/veh/m	
		=	3.24E-02	g/m/hr	
		=	9.01E-06	g/m/s	
		=	1.18E-04	g/s	
		=	3 00E-06	$a/s/m^2$	
			<u>3.002 00</u>	g/ 3/ 111	
ED10 7					
EP19_7	Control Efficiency	_	05.2	0/	(Nata 2)
	Control Efficiency	=	95.3	%	(Note 3)
	Trucks Travelling Frequency	=	20	ven/hr	(Note 4)
	Road Length	=	34.6	m	
	Area of paved road travelled	=	103.8	m	
	RSP Emission Factor (uncontrolled) of paved road	=	142.4534	g/VKT	
	RSP Emission Factor (controlled) of paved road	=	6.7062	g/VKT	
		=	6.71E-03	g/veh/m	
		=	1.34E-01	g/m/hr	
		=	3.73F-05	g/m/s	
		_	1 205-03	a/s	
		-	1.2.31=03	6/3 ala/2	
		=	<u>1.242E-05</u>	g/s/m <sup>-</sup>	
	FSP Emission Factor (uncontrolled) of paved road	=	34.465	g/VKT	
	FSP Emission Factor (controlled) of paved road	=	1.6225	g/VKT	
		=	1.62E-03	g/veh/m	
		=	3.24E-02	g/m/hr	
		=	9.01E-06	g/m/s	
		=	3,12F-04	g/s	
		-	3 005 00	a/s/m <sup>2</sup>	
		-	3.00E-06	g/ 3/111	
ED10 8					
EN13 <sup>8</sup>	· · · · · · · · · · · · · · · · · · ·		07 F	0/	(11-1-2)
	Control Efficiency	=	97.5	%	(Note 3)
	Trucks Travelling Frequency	=	20	veh/hr	(Note 4)
	Road Length	=	10.0	m	
	Area of paved road travelled	=	30.0	m <sup>2</sup>	
	RSP Emission Factor (uncontrolled) of paved road	=	142.4534	g/VKT	
	RSP Emission Factor (controlled) of paved road	=	3 5296	g/VKT	
		-	3 535-03	g/veh/m	
		-	7 065 02	a/m/hr	
		-	1 065 05	6/ 11/ 11 a/m/c	
		-	1.90E-05	g/11)/5	
		=	1.96E-04	g/s	
		=	<u>6.536E-06</u>	g/s/m²	
	FSP Emission Factor (uncontrolled) of paved road	=	34.465	g/VKT	

		=	8.54E-04	g/veh/m	
		=	1.71E-02	g/m/hr	
		=	4.74E-06	g/m/s	
		=	4.74E-05	g/s	
		=	<u>1.58E-06</u>	g/s/m²	
EP19_9	Control Efficiency	_	00.0	0/	(Nets 2)
	Trucks Travelling Frequency	-	98.8	% veh/hr	(Note 3)
	Road Length	=	10.3	m	(10000 4)
	Area of paved road travelled	=	30.9	m <sup>2</sup>	
			50.5		
	RSP Emission Factor (uncontrolled) of paved road	=	142.4534	g/VKT	
	RSP Emission Factor (controlled) of paved road	=	3.5296	g/VKT	
		=	3.53E-03	g/veh/m	
		=	3.53E-02	g/m/hr	
		=	9.80E-06	g/m/s	
		=	1.01E-04	g/s	
		=	<u>3.268E-06</u>	g/s/m²	
	ESP Emission Eactor (uncontrolled) of payed road	_	34 465	α/\/KT	
	FSP Emission Factor (controlled) of paved road	_	0.8539	g/VKT	
		=	8.54E-04	g/veh/m	
		=	8.54E-03	g/m/hr	
		=	2.37E-06	g/m/s	
		=	2.44E-05	g/s	
		=	7.91E-07	g/s/m <sup>2</sup>	
EP19_10					(11.1.2)
	Control Efficiency	-	98.8	% voh/hr	(Note 3)
	Irucks Iravelling Frequency	-	16.2	wen/nr	(Note 4)
	Area of payed road travelled	_	10.5	m <sup>2</sup>	
	Area or paved road travelled	-	40.9		
	RSP Emission Factor (uncontrolled) of paved road	=	142.4534	g/VKT	
	RSP Emission Factor (controlled) of paved road	=	3.5296	g/VKT	
		=	3.53E-03	g/veh/m	
		=	3.53E-02	g/m/hr	
		=	9.80E-06	g/m/s	
		=	1.60E-04	g/s	
		=	3.268E-06	g/s/m²	
	FCD Emission Factor (uncentralled) of pound road	_	24.405	- h///T	
	FSP Emission Factor (uncontrolled) of paved road	-	34.465	g/VKI g/VKT	
	FSP Emission Factor (controlled) of paved road	_	0.8539 8 54F=04	g/vki g/veh/m	
		=	8.54E-04	g/w/hr	
		=	2.37E-06	g/m/s	
		=	3.87E-05	g/s	
		=	<u>7.91E-07</u>	g/s/m <sup>2</sup>	
EP19_11					
	Control Efficiency	=	98.8	%	(Note 3)
	Trucks Travelling Frequency	-	10	ven/nr	(Note 4)
	Area of payed road travelled	-	14.4	m <sup>2</sup>	
	Area of paved road travelled	-	43.2		
	RSP Emission Eactor (uncontrolled) of payed road	=	142.4534	g/VKT	
	RSP Emission Factor (controlled) of paved road	=	3.5296	g/VKT	
		=	3.53E-03	g/veh/m	
		=	3.53E-02	g/m/hr	
		=	9.80E-06	g/m/s	
		=	1.41E-04	g/s	
		=	1.41E-04 <u>3.268E-06</u>	g/s g/s/m²	
	ECD Emission Easter (secondary) - Consultant	=	1.41E-04 <u>3.268E-06</u>	g/s g/s/m <sup>2</sup>	
	FSP Emission Factor (uncontrolled) of paved road	= = =	1.41E-04 <u>3.268E-06</u> 34.465 0.8529	g/s g/s/m <sup>2</sup> g/VKT g/VKT	
	FSP Emission Factor (uncontrolled) of paved road FSP Emission Factor (controlled) of paved road	= = = =	1.41E-04 3.268E-06 34.465 0.8539 8.54F-04	g/s g/vKT g/vKT g/vkT	
	FSP Emission Factor (uncontrolled) of paved road FSP Emission Factor (controlled) of paved road	= = = =	1.41E-04 <u>3.268E-06</u> 34.465 0.8539 8.54E-04 8.54E-03	g/s g/s/m <sup>2</sup> g/VKT g/VKT g/veh/m g/m/hr	
	FSP Emission Factor (uncontrolled) of paved road FSP Emission Factor (controlled) of paved road	= = = = =	1.41E-04 <u>3.268E-06</u> 34.465 0.8539 8.54E-04 8.54E-03 2.37E-06	g/s g/vKT g/vKT g/veh/m g/m/hr g/m/s	
	FSP Emission Factor (uncontrolled) of paved road FSP Emission Factor (controlled) of paved road	= = = = = =	1.41E-04 3.268E-06 34.465 0.8539 8.54E-04 8.54E-03 2.37E-06 3.42E-05	g/s g/vKT g/vKT g/veh/m g/m/hr g/m/s g/n/s	
	FSP Emission Factor (uncontrolled) of paved road FSP Emission Factor (controlled) of paved road	= = = = = = =	1.41E-04 3.268E-06 34.465 0.8539 8.54E-04 8.54E-03 2.37E-06 3.42E-05 7.91E-07	g/s g/s/m <sup>2</sup> g/VKT g/VKT g/veh/m g/m/hr g/m/s g/s g/s/m <sup>2</sup>	
	FSP Emission Factor (uncontrolled) of paved road FSP Emission Factor (controlled) of paved road		1.41E-04 <u>3.268E-06</u> 34.465 0.8539 8.54E-04 8.54E-03 2.37E-06 3.42E-05 <u>7.91E-07</u>	g/s g/s/m <sup>2</sup> g/VKT g/VKT g/veh/m g/m/hr g/m/hr g/m/s g/s g/s/m <sup>2</sup>	
EP19_12	FSP Emission Factor (uncontrolled) of paved road FSP Emission Factor (controlled) of paved road		1.41E-04 <u>3.268E-06</u> 34.465 0.8539 8.54E-04 8.54E-03 2.37E-06 3.42E-05 <u>7.91E-07</u>	g/s g/s/m <sup>2</sup> g/VKT g/VKT g/veh/m g/m/hr g/m/hr g/m/s g/s g/s/m <sup>2</sup>	
EP19_12	FSP Emission Factor (uncontrolled) of paved road FSP Emission Factor (controlled) of paved road Control Efficiency		1.41E-04 3.268E-06 34.465 0.8539 8.54E-04 8.54E-03 2.37E-06 3.42E-05 7.91E-07 98.8	g/s g/s/m <sup>2</sup> g/VKT g/VKT g/veh/m g/m/hr g/m/s g/s g/s/m <sup>2</sup>	(Note 3)
EP19_12	FSP Emission Factor (uncontrolled) of paved road FSP Emission Factor (controlled) of paved road Control Efficiency Trucks Traveling Bord Location		1.41E-04 <u>3.268E-06</u> 34.465 0.8539 8.54E-04 8.54E-03 2.37E-06 3.42E-05 <u>7.91E-07</u> 98.8 10 15 0	g/s g/s/m <sup>2</sup> g/VKT g/VKT g/vkh/m g/m/hr g/m/s g/s g/s g/s/m <sup>2</sup> % veh/hr m	(Note 3) (Note 4)
EP19_12	FSP Emission Factor (uncontrolled) of paved road FSP Emission Factor (controlled) of paved road Control Efficiency Trucks Travelling Frequency Road Length		1.41E-04 <u>3.268E-06</u> 0.8539 8.54E-04 8.54E-03 2.37E-06 3.42E-05 <u>7.91E-07</u> 98.8 10 15.9 4.7 7.7	g/s g/s/m <sup>2</sup> g/VKT g/VKT g/veh/m g/m/hr g/m/hr g/s/s g/s/m <sup>2</sup> % veh/hr m m <sup>2</sup>	(Note 3) (Note 4)
EP19_12	FSP Emission Factor (uncontrolled) of paved road FSP Emission Factor (controlled) of paved road Control Efficiency Trucks Travelling Frequency Road Length Area of paved road travelled		1.41E-04 <u>3.268E-06</u> 34.465 0.8539 8.54E-04 8.54E-03 2.37E-06 3.42E-05 7.91E-07 98.8 10 15.9 47.7	g/s g/s/m <sup>2</sup> g/VKT g/VKT g/veh/m g/m/hr g/m/s g/s/s/g/s/m <sup>2</sup> % veh/hr m m <sup>2</sup>	(Note 3) (Note 4)
EP19_12	FSP Emission Factor (uncontrolled) of paved road FSP Emission Factor (controlled) of paved road Control Efficiency Trucks Travelling Frequency Road Length Area of paved road travelled RSP Emission Factor (uncontrolled) of paved road		1.41E-04 3.268E-06 34.465 0.8539 8.54E-04 8.54E-03 2.37E-05 7.91E-07 98.8 10 15.9 47.7 142 4534	g/s g/s/m <sup>2</sup> g/VKT g/VKT g/veh/m g/m/hr g/m/s g/s g/s/m <sup>2</sup> % veh/hr m m <sup>2</sup> g/VKT	(Note 3) (Note 4)
EP19_12	FSP Emission Factor (uncontrolled) of paved road FSP Emission Factor (controlled) of paved road Control Efficiency Trucks Travelling Frequency Road Length Area of paved road travelled RSP Emission Factor (uncontrolled) of paved road RSP Emission Factor (controlled) of paved road		1.41E-04 <u>3.268E-06</u> 34.465 0.8539 8.54E-03 8.54E-03 2.37E-06 3.42E-05 7.91E-07 98.8 10 15.9 47.7 142.4534 3.5296	g/s g/s/m <sup>2</sup> g/VKT g/VKT g/veh/m g/m/hr g/m/s g/s/s g/s/m <sup>2</sup> % veh/hr m m <sup>2</sup> g/VKT g/VKT	(Note 3) (Note 4)
EP19_12	FSP Emission Factor (uncontrolled) of paved road FSP Emission Factor (controlled) of paved road Control Efficiency Trucks Travelling Frequency Road Length Area of paved road travelled RSP Emission Factor (uncontrolled) of paved road RSP Emission Factor (uncontrolled) of paved road		1.41E-04 <u>3.268E-06</u> 34.465 0.8539 8.54E-04 8.54E-03 2.37E-06 3.42E-05 <u>7.91E-07</u> 98.8 10 15.9 47.7 142.4534 3.5296 3.53E-03	g/s g/s/m <sup>2</sup> g/VKT g/VKT g/vkh/m g/m/hr g/m/s g/s/s g/s/m <sup>2</sup> % veh/hr m m <sup>2</sup> g/VKT g/VKT g/VKT g/VKT	(Note 3) (Note 4)
EP19_12	FSP Emission Factor (uncontrolled) of paved road FSP Emission Factor (controlled) of paved road Control Efficiency Trucks Travelling Frequency Road Length Area of paved road travelled RSP Emission Factor (uncontrolled) of paved road RSP Emission Factor (controlled) of paved road		1.41E-04 <u>3.268E-06</u> 34.465 0.8539 8.54E-04 8.54E-03 2.37E-06 3.42E-05 <u>7.91E-07</u> 98.8 10 15.9 47.7 142.4534 3.53E-03 3.53E-03	g/s g/s/m <sup>2</sup> g/VKT g/VKT g/veh/m g/m/hr g/m/s g/s g/s g/s/m <sup>2</sup> % veh/hr m m <sup>2</sup> g/VKT g/VKT g/VKT g/veh/m g/m/hr	(Note 3) (Note 4)
EP19_12	FSP Emission Factor (uncontrolled) of paved road FSP Emission Factor (controlled) of paved road Control Efficiency Trucks Travelling Frequency Road Length Area of paved road travelled RSP Emission Factor (uncontrolled) of paved road RSP Emission Factor (controlled) of paved road		1.41E-04 <u>3.268E-06</u> 3.4.465 0.8539 8.54E-04 8.54E-04 8.54E-03 2.37E-06 3.42E-05 7.91E-07 98.8 10 15.9 47.7 142.4534 3.5296 3.53E-02 3.53E-02 9.80E-06	g/s g/s/m <sup>2</sup> g/VKT g/VKT g/veh/m g/m/hr g/s g/s/m <sup>2</sup> % veh/hr m m <sup>2</sup> g/VKT g/VKT g/veh/m g/v/hr g/m/hr	(Note 3) (Note 4)
EP19_12	FSP Emission Factor (uncontrolled) of paved road FSP Emission Factor (controlled) of paved road Control Efficiency Trucks Travelling Frequency Road Length Area of paved road travelled RSP Emission Factor (uncontrolled) of paved road RSP Emission Factor (controlled) of paved road		1.41E-04 <u>3.268E-06</u> 34.465 0.8539 8.54E-04 8.54E-03 2.37E-06 <b>3.42E-05</b> <b>7.91E-07</b> <b>98.8</b> 10 15.9 47.7 <b>142.4534</b> <b>3.5296</b> <b>3.53E-03</b> <b>3.53E-03</b> <b>3.53E-03</b> <b>3.53E-03</b> <b>3.53E-03</b> <b>3.53E-03</b> <b>3.53E-03</b> <b>3.53E-03</b> <b>3.53E-03</b> <b>3.53E-03</b> <b>3.53E-03</b> <b>3.53E-03</b> <b>3.53E-03</b> <b>3.53E-03</b> <b>3.53E-03</b> <b>3.53E-03</b> <b>3.53E-03</b> <b>3.53E-03</b> <b>3.53E-03</b> <b>3.53E-03</b> <b>3.53E-03</b> <b>3.53E-03</b> <b>3.53E-03</b> <b>3.53E-03</b> <b>3.53E-03</b> <b>3.53E-03</b> <b>3.53E-03</b> <b>3.53E-03</b> <b>3.53E-03</b> <b>3.53E-03</b> <b>3.53E-03</b> <b>3.53E-03</b> <b>3.53E-03</b> <b>3.53E-03</b> <b>3.53E-03</b> <b>3.53E-03</b> <b>3.53E-03</b> <b>3.53E-03</b> <b>3.53E-03</b> <b>3.53E-03</b> <b>3.53E-03</b> <b>3.53E-03</b> <b>3.53E-03</b> <b>3.53E-03</b> <b>3.53E-03</b> <b>3.53E-03</b> <b>3.53E-03</b> <b>3.53E-03</b> <b>3.53E-03</b> <b>3.53E-03</b> <b>3.53E-03</b> <b>3.53E-03</b> <b>3.53E-03</b> <b>3.53E-03</b> <b>3.53E-03</b> <b>3.53E-03</b> <b>3.53E-03</b> <b>3.53E-03</b> <b>3.53E-03</b> <b>3.53E-03</b> <b>3.53E-03</b> <b>3.53E-03</b> <b>3.53E-03</b> <b>3.53E-03</b> <b>3.53E-03</b> <b>3.53E-03</b> <b>3.53E-03</b> <b>3.53E-03</b> <b>3.53E-03</b> <b>3.53E-03</b> <b>3.53E-03</b> <b>3.53E-03</b> <b>3.55E-03</b> <b>3.55E-03</b> <b>3.55E-03</b> <b>3.55E-03</b> <b>3.55E-03</b> <b>3.55E-03</b> <b>3.55E-03</b> <b>3.55E-03</b> <b>3.55E-03</b> <b>3.55E-03</b> <b>3.55E-03</b> <b>3.55E-03</b> <b>3.55E-03</b> <b>3.55E-03</b> <b>3.55E-03</b> <b>3.55E-03</b> <b>3.55E-03</b> <b>3.55E-03</b> <b>3.55E-03</b> <b>3.55E-03</b> <b>3.55E-03</b> <b>3.55E-03</b> <b>3.55E-03</b> <b>3.55E-03</b> <b>3.55E-03</b> <b>3.55E-03</b> <b>3.55E-03</b> <b>3.55E-03</b> <b>3.55E-03</b> <b>3.55E-03</b> <b>3.55E-03</b> <b>3.55E-03</b> <b>3.55E-03</b> <b>3.55E-03</b> <b>3.55E-03</b> <b>3.55E-03</b> <b>3.55E-03</b> <b>3.55E-03</b> <b>3.55E-03</b> <b>3.55E-03</b> <b>3.55E-03</b> <b>3.55E-03</b> <b>3.55E-03</b> <b>3.55E-03</b> <b>3.55E-03</b> <b>3.55E-03</b> <b>3.55E-03</b> <b>3.55E-03</b> <b>3.55E-03</b> <b>3.55E-03</b> <b>3.55E-03</b> <b>3.55E-03</b> <b>3.55E-03</b> <b>3.55E-03</b> <b>3.55E-03</b> <b>3.55E-03</b> <b>3.55E-03</b> <b>3.55E-03</b> <b>3.55E-03</b> <b>3.55E-03</b> <b>3.55E-03</b> <b>3.55E-03</b> <b>3.55E-03</b> <b>3.55E-03</b> <b>3.55E-03</b> <b>3.55E-03</b> <b>3.55E-03</b> <b>3.55E-03</b> <b>3.55E-03</b> <b>3.55E-03</b> <b>3.55E-03</b> <b>3.55E-03</b> <b>3.55E-03</b> <b>3.55E-03</b> <b>3.55E-03</b> <b>3.55E-03</b> <b>3.55E-03</b> <b>3.55E-03</b> <b>3.55E-03</b> <b>3.55E-03</b> <b>3.55E-03</b> <b>3.55E-03</b> <b>3.55E-03</b> <b>3.55E-03</b> <b>3.55E-03</b> <b>3.55E-03</b> <b>3.5E-03</b>	g/s g/s/m <sup>2</sup> g/VKT g/VkT g/veh/m g/m/hr g/m/s g/s/m <sup>2</sup> % veh/hr m m <sup>2</sup> g/VKT g/VKT g/VKT g/VKT g/vh/m g/m/hr g/m/s g/s	(Note 3) (Note 4)
EP19_12	FSP Emission Factor (uncontrolled) of paved road FSP Emission Factor (controlled) of paved road Control Efficiency Trucks Travelling Frequency Road Length Area of paved road travelled RSP Emission Factor (uncontrolled) of paved road RSP Emission Factor (controlled) of paved road		1.41E-04 <u>3.268E-06</u> 34.465 0.8539 8.54E-03 8.54E-03 3.42E-05 <u>7.91E-07</u> 9.8.8 10 15.9 47.7 142.4534 3.5296 3.53E-03 3.53E-02 9.80E-04 <u>3.268E-06</u>	g/s g/s/m <sup>2</sup> g/VKT g/VKT g/veh/m g/m/s g/s/m <sup>2</sup> % veh/hr m m <sup>2</sup> g/VKT g/VKT g/vKT g/vKT g/vKT g/vk/m g/m/hr g/r/s g/s/m <sup>2</sup>	(Note 3) (Note 4)
EP19_12	FSP Emission Factor (uncontrolled) of paved road FSP Emission Factor (controlled) of paved road Control Efficiency Trucks Travelling Frequency Road Length Area of paved road travelled RSP Emission Factor (uncontrolled) of paved road RSP Emission Factor (controlled) of paved road		1.41E-04 <u>3.268E-06</u> 34.465 0.8539 8.54E-04 8.54E-05 <u>2.37E-06</u> 3.42E-05 <u>7.91E-07</u> 98.8 10 15.9 47.7 142.4534 3.5296 3.53E-03 3.53E-02 9.80E-06 1.56E-06	g/s g/s/m <sup>2</sup> g/VKT g/VKT g/veh/m g/m/hr g/m/s g/s/m <sup>2</sup> % veh/hr m m <sup>2</sup> g/VKT g/VKT g/VKT g/VKT g/veh/m g/m/hr g/s g/s/m <sup>2</sup>	(Note 3) (Note 4)
EP19_12	FSP Emission Factor (uncontrolled) of paved road FSP Emission Factor (controlled) of paved road Control Efficiency Trucks Travelling Frequency Road Length Area of paved road travelled RSP Emission Factor (uncontrolled) of paved road RSP Emission Factor (uncontrolled) of paved road		1.41E-04 <u>3.268E-06</u> 34.465 0.8539 8.54E-04 8.54E-03 2.37E-06 3.42E-05 <u>7.91E-07</u> 98.8 10 15.9 47.7 142.4534 3.5266 1.56E-04 <u>3.268E-06</u> 3.4455 0.015	g/s g/s/m <sup>2</sup> g/VKT g/VKT g/veh/m g/m/hr g/m/s g/s/s g/s/m <sup>2</sup> % veh/hr m m <sup>2</sup> g/VKT g/VKT g/VKT g/veh/m g/m/hr g/m/s g/s/m <sup>2</sup>	(Note 3) (Note 4)
EP19_12	FSP Emission Factor (uncontrolled) of paved road FSP Emission Factor (controlled) of paved road Control Efficiency Trucks Travelling Frequency Road Length Area of paved road travelled RSP Emission Factor (uncontrolled) of paved road RSP Emission Factor (controlled) of paved road FSP Emission Factor (uncontrolled) of paved road		1.41E-04 <u>3.268E-06</u> 3.4.465 0.8539 8.54E-04 8.54E-03 2.37E-06 3.42E-05 <u>7.91E-07</u> 98.8 10 15.9 47.7 142.4534 3.53E-03 3.53E-03 3.53E-04 <u>3.268E-06</u> 3.4465 0.8539 8.54E-04	g/s g/s/m <sup>2</sup> g/VKT g/VKT g/veh/m g/m/hr g/m/s g/s g/s/m <sup>2</sup> % veh/hr m <sup>2</sup> g/VKT g/VKT g/veh/m g/n/hr g/m/s g/s/m <sup>2</sup> g/s/m <sup>2</sup>	(Note 3) (Note 4)
EP19_12	FSP Emission Factor (uncontrolled) of paved road FSP Emission Factor (controlled) of paved road Control Efficiency Trucks Travelling Frequency Road Length Area of paved road travelled RSP Emission Factor (uncontrolled) of paved road RSP Emission Factor (controlled) of paved road FSP Emission Factor (uncontrolled) of paved road FSP Emission Factor (uncontrolled) of paved road		1.41E-04 <u>3.268E-06</u> 3.4.465 0.8539 8.54E-04 8.54E-03 2.37E-06 3.42E-05 <u>7.91E-07</u> 98.8 10 15.9 47.7 142.4534 3.5296 3.53E-02 3.53E-02 3.54E-04 <u>3.268E-06</u> 8.54E-04 8.54E-04	g/s g/s/m <sup>2</sup> g/VKT g/VKT g/veh/m g/m/hr g/m/s g/s/m <sup>2</sup> % % veh/hr m m <sup>2</sup> g/VKT g/VKT g/VKT g/VKT g/VKT g/VKT g/VKT g/VKT	(Note 3) (Note 4)
EP19_12	FSP Emission Factor (uncontrolled) of paved road FSP Emission Factor (controlled) of paved road Control Efficiency Trucks Travelling Frequency Road Length Area of paved road travelled RSP Emission Factor (uncontrolled) of paved road RSP Emission Factor (controlled) of paved road FSP Emission Factor (uncontrolled) of paved road		1.41E-04 3.268E-06 34.465 0.8539 8.54E-03 8.54E-03 3.42E-05 2.37E-06 3.42E-05 7.91E-07 9.88 10 15.9 47.7 142.4534 3.5296 3.53E-03 3.53E-03 3.53E-03 3.53E-04 3.4685 0.8539 8.54E-04 8.54E-03 2.37E-06	g/s g/s/m <sup>2</sup> g/VKT g/VKT g/veh/m g/m/hr g/m/s g/s/m <sup>2</sup> % veh/hr m m <sup>2</sup> g/VKT g/VKT g/VKT g/veh/m g//KT g/VKT g/VKT g/VKT g/VKT g/VKT g/VKT	(Note 3) (Note 4)
EP19_12	FSP Emission Factor (uncontrolled) of paved road FSP Emission Factor (controlled) of paved road Control Efficiency Trucks Travelling Frequency Road Length Area of paved road travelled RSP Emission Factor (uncontrolled) of paved road RSP Emission Factor (controlled) of paved road FSP Emission Factor (uncontrolled) of paved road FSP Emission Factor (uncontrolled) of paved road		1.41E-04 3.268E-06 34.465 0.8539 8.54E-03 2.37E-06 3.42E-05 7.91E-07 98.8 10 98.8 10 98.8 10 9.7 142.4534 3.5266 3.53E-03 3.53E-03 3.53E-03 3.53E-03 3.53E-04 3.56E-04 3.54E-03 2.37E-05 3.77E-05	g/s g/s/m <sup>2</sup> g/VKT g/VKT g/veh/m g/m/hr g/m/hr g/s g/s/m <sup>2</sup> % veh/hr m m <sup>2</sup> g/VKT g/VKT g/veh/m g/m/hr g/s g/s/m <sup>2</sup> g/VKT g/vKT g/vKT g/vKT g/vKT g/vKT g/vKT	(Note 3) (Note 4)
EP19_12	FSP Emission Factor (uncontrolled) of paved road FSP Emission Factor (controlled) of paved road Control Efficiency Trucks Travelling Frequency Road Length Area of paved road travelled RSP Emission Factor (uncontrolled) of paved road RSP Emission Factor (controlled) of paved road FSP Emission Factor (uncontrolled) of paved road FSP Emission Factor (controlled) of paved road		1.41E-04 <u>3.268E-06</u> 34.465 0.8539 8.54E-04 8.54E-05 <u>2.37E-06</u> 3.42E-05 <u>7.91E-07</u> 98.8 10 15.9 47.7 142.4534 3.5296 3.53E-03 3.55E-03 3.55E-03 3.55E-03 3.55E-03 3.55E-03 3.5	g/s g/s/m <sup>2</sup> g/VKT g/VKT g/veh/m g/m/hr g/m/s g/s/m <sup>2</sup> % veh/hr m m <sup>2</sup> g/VKT g/VKT g/VKT g/VKT g/veh/m g/s/m <sup>2</sup> g/VKT g/VKT g/VKT g/VKT g/VKT g/VKT g/VKT g/VKT g/VKT g/VKT g/VKT	(Note 3) (Note 4)

Control Efficiency	=	98.8	%	(Note 3
Trucks Travelling Frequency	=	10	ven/hr	(Note -
Koad Length Area of paved road travelled	=	24.0 72.0	m <sup>2</sup>	
RSP Emission Factor (uncontrolled) of paved road	=	142 4534	g/VKT	
RSP Emission Factor (controlled) of paved road	=	3 5296	g/VKT	
····· -·······························	=	3.53E-03	g/veh/m	
	=	3.53E-02	g/m/hr	
	=	9.80E-06	g/m/s	
	=	2.35E-04	g/s	
	=	<u>3.268E-06</u>	g/s/m <sup>2</sup>	
FSP Emission Factor (uncontrolled) of paved road	=	34.465	g/VKT	
FSP Emission Factor (controlled) of paved road	=	0.8539	g/VKT	
	=	8.54E-04	g/veh/m	
	=	8.54E-03	g/m/hr	
	-	2.37E-06	g/m/s a/s	
	=	7.91E-07	g/s/m <sup>2</sup>	
EP19_14			0/	(1)
Control Efficiency Trucks Travelling Frequency	=	97.8 0	% veh/hr	(Note )
Road Length	=	29.0	m	
Area of paved road travelled	=	87.0	m <sup>2</sup>	
RSP Emission Factor (uncontrolled) of paved road	=	142.4534	g/VKT	
RSP Emission Factor (controlled) of paved road	=	3.5296	g/VKT	
	=	3.53E-03	g/veh/m	
	-	U.UUE+00	g/m/nr g/m/c	
	-	0.00E+00	g/111/5 g/s	
	=	0.000E+00	ы» g/s/m²	
		24.465	- 6 11/7	
FSP Emission Factor (uncontrolled) of paved road FSP Emission Factor (controlled) of paved road	=	54.405 0.8539	g/VNI g/VKT	
	=	8.54E-04	g/veh/m	
	=	0.00E+00	g/m/hr	
	=	0.00E+00	g/m/s	
	=	0.00E+00	g/s	
	=	0.00E+00	g/s/m <sup>2</sup>	
From Aggregate Truck				
Mitigation measures: water spraying and wheel washing Average Loaded Weight of Concrete Lorry Miver	=	30	tons	
Average Loaded Weight of Concrete Lorry Mixer	=	15	tons	
Average Unladden Weight of Concrete Lorry Mixer	=	22.5	tons	
k k	=	3.23	g/VKT (for TSP)	(Note
	=	0.62	g/VKT (for RSP)	
	=	0.15	g/VKT (for FSP)	
sL = road surface silt loading	=	12	g/m²	(Note 2
EP19_1				
Control Efficiency	=	93.1	%	(Note 3
Trucks Travelling Frequency	=	28	veh/hr	(Note
Road Length	=	15.7	m <sup>2</sup>	
Area of paved road travelled	-	47.1	m	
RSP Emission Factor (uncontrolled) of paved road	=	142.4534	g/VKT	
RSP Emission Factor (controlled) of paved road	=	9.8829	g/VKI	
	=	3.88E-U3	g/w/br	
	-	2.77E-UI 7.69F-05	в/ 11/ 11 g/m/s	
	=	1.21F-03	g/s	
	=	2.56E-05	g/s/m <sup>2</sup>	
FSP Emission Factor (uncontrolled) of paved road	=	34.4645	g/VKT	
FSP Emission Factor (controlled) of paved road	=	2.3910	g/VKT	
	=	2.39E-03	g/veh/m	
	=	6.69E-02	g/m/hr	
	=	1.86E-05	g/m/s	
	=	2.92E-04	g/s g/s/m <sup>2</sup>	
		J.20E-00	8/ 3/ 111	
EP19_2 Control Efficiency	=	95.3	%	(Note 3
Trucks Travelling Frequency	=	14	veh/hr	(Note 4
Road Length	=	20.2	m	
Area of paved road travelled	=	60.6	m <sup>2</sup>	
	_	142 4524	~ \ /// T	
RSP Emission Factor (uncontrolled) of paved road RSP Emission Factor (controlled) of paved road	=	142.4534 6 7062	g/VKI g/VKT	
KSP ETHISSION PACTOR (CONTROLLED) OF PAVED FORD	-	6.71F-03	g/veh/m	
	_	9.39F-02	g/m/hr	
	=	2.55L-02	g/m/s	
	=	Z		
	=	5.27F-04	g/s	
	= = =	5.27E-04 8.69E-06	g/s g/s/m <sup>2</sup>	
ESP Emission Eactor (uncontrolled) of payed road	-	5.27E-04 8.69E-06	g/s g/s/m <sup>2</sup>	
FSP Emission Factor (uncontrolled) of paved road FSP Emission Factor (controlled) of paved road		2.011-03 5.27E-04 <u>8.69E-06</u> 34.4645 1.6225	g/s g/s/VKT g/VKT	
FSP Emission Factor (uncontrolled) of paved road FSP Emission Factor (controlled) of paved road	= = = = =	2.011-03 5.27E-04 <u>8.69E-06</u> 34.4645 1.6225 1.62E-03	g/s g/s/m <sup>2</sup> g/VKT g/VKT g/veh/m	
FSP Emission Factor (uncontrolled) of paved road FSP Emission Factor (controlled) of paved road	= = = = = = =	2.011-03 5.27E-04 <u>8.69E-06</u> 34.4645 1.6225 1.62E-03 2.27E-02	g/s g/s/m <sup>2</sup> g/VKT g/VKT g/veh/m g/w/hr	
FSP Emission Factor (uncontrolled) of paved road FSP Emission Factor (controlled) of paved road	= = = = = = = = =	2.011-03 5.27E-04 <u>8.69E-06</u> 34.4645 1.6225 1.62E-03 2.27E-02 6.31E-06	g/s g/s/m <sup>2</sup> g/VKT g/VKT g/veh/m g/m/hr g/m/s	
FSP Emission Factor (uncontrolled) of paved road FSP Emission Factor (controlled) of paved road		5.27E-04 8.69E-06 34.4645 1.6225 1.62E-03 2.27E-02 6.31E-06 1.27E-04	g/s g/s/m <sup>2</sup> g/VKT g/VKT g/veh/m g/m/hr g/m/s g/s	
FSP Emission Factor (uncontrolled) of paved road FSP Emission Factor (controlled) of paved road	-	5.27E-04 8.69E-06 34.4645 1.6225 1.62E-03 2.27E-02 6.31E-06 1.27E-04 2.10E-06	g/s g/s/km <sup>2</sup> g/VKT g/VKT g/veh/m g/m/hr g/m/hr g/s g/s g/s/m <sup>2</sup>	

EP19_3					
	Control Efficiency	=	95.3	%	(Note 3)
	Trucks Travelling Frequency	=	14	veh/hr	(Note 4)
	Road Length	=	19.9	m	
	Area of paved road travelled	=	59.7	m²	
			442 4524	- 6 0/7	
	RSP Emission Factor (uncontrolled) of paved road	-	6 7062	g/VKI g/VKT	
	hor Emission factor (controlled) of paved road	-	6.71E-03	g/vRl	
		=	9.39E-02	g/m/hr	
		=	2.61E-05	g/m/s	
		=	5.19E-04	g/s	
		=	8.69E-06	g/s/m <sup>2</sup>	
	FSP Emission Factor (uncontrolled) of paved road	=	34.4645	g/VKT	
	FSP Emission Factor (controlled) of paved road	=	1.6225	g/VKT	
		=	1.62E-03	g/veh/m	
		=	2.27E-02	g/m/hr	
		=	6.31E-06	g/m/s	
		=	1.26E-04	g/s	
		-	2.10E-06	g/s/m	
FP19 4					
	Control Efficiency	=	95.3	%	(Note 3)
	Trucks Travelling Frequency	=	14	veh/hr	(Note 4)
	Road Length	=	9.7	m	
	Area of paved road travelled	=	29.1	m <sup>2</sup>	
	RSP Emission Factor (uncontrolled) of paved road	=	142.4534	g/VKT	
	RSP Emission Factor (controlled) of paved road	=	6.7062	g/VKT	
		=	6.71E-03	g/veh/m	
		=	9.39E-02	g/m/hr	
		=	2.01E-05	g/m/s	
		-	2.53E-04	g/s	
		-	0.09E-06	g/s/m	
	ESP Emission Factor (uncontrolled) of paved road	=	34 4645	g/VKT	
	ESP Emission Factor (controlled) of paved road	=	1.6225	g/VKT	
	· • • • • • • • • • • • • • • • • • • •	=	1.62E-03	g/veh/m	
		=	2.27E-02	g/m/hr	
		=	6.31E-06	g/m/s	
		=	6.12E-05	g/s	
		=	2.10E-06	g/s/m <sup>2</sup>	
EP19_5					(11.1.0)
	Control Efficiency	=	95.3	% 	(Note 3)
	Boad Length	-	14	m	(Note 4)
	Area of payed read travelled	_	22.0	m <sup>2</sup>	
	Alea of paved toad travelled	-	55.9		
	RSP Emission Factor (uncontrolled) of payed road	=	142.4534	g/VKT	
	RSP Emission Factor (controlled) of paved road	=	6.7062	g/VKT	
		=	6.71E-03	g/veh/m	
		=	9.39E-02	g/m/hr	
		=	2.61E-05	g/m/s	
		=	2.95E-04	g/s	
		=	<u>8.69E-06</u>	g/s/m <sup>2</sup>	
	FSP Emission Factor (uncontrolled) of paved road	=	34.4645	g/VKT	
	FSP Emission Factor (controlled) of paved road	=	1.6225	g/VKT	
		=	1.62E-03	g/veh/m	
		_	6.21E-06	g/11/11 g/m/s	
		=	7 13E-05	g/11/3 g/s	
		=	2.10F-06	$g/s/m^2$	
			<u></u>	0/ 5/ 11	
EP19_6					
	Control Efficiency	=	95.3	%	(Note 3)
	Trucks Travelling Frequency	=	14	veh/hr	(Note 4)
	Road Length	=	13.1	m	
	Area of paved road travelled	=	39.3	m²	
	RSP Emission Factor (uncontrolled) of paved road	=	142.4534	g/VKT	
	RSP Emission Factor (controlled) of paved road	=	6.7062	g/VKI	
		-	0./10-03	g/ven/m	
		=	2.61F-02	g/m/s	
		=	3.42E-04	g/s	
		=	8,69F-06	g/s/m <sup>2</sup>	
			2.052.00	0, -1	
	FSP Emission Factor (uncontrolled) of paved road	=	34.4645	g/VKT	
	FSP Emission Factor (controlled) of paved road	=	1.6225	g/VKT	
		=	1.62E-03	g/veh/m	
		=	2.27E-02	g/m/hr	
		=	6.31E-06	g/m/s	
		=	8.27E-05	g/s	
		=	<u>2.10E-06</u>	g/s/m <sup>+</sup>	
ED10 7					
CF 19_1	Control Efficiency	=	05.2	%	(Note 2)
	Trucks Travelling Frequency	=	14	veh/hr	(Note 4)
					(
	Road Length	=	34.6	m	

	RSP Emission Eactor (uncontrolled) of payed road	=	142 4534	ø/VKT	
	RSP Emission Factor (controlled) of paved road	=	6.7062	g/VKT	
		=	6.71E-03	g/veh/m	
		=	9.39E-02	g/m/hr	
		=	2.61E-05	g/m/s	
		=	9.02E-04	g/s	
		=	8.69E-06	g/s/m <sup>2</sup>	
	ESP Emission Eactor (uncontrolled) of payed road	_	34 4645	α/\/KT	
	FSP Emission Factor (controlled) of paved road	-	1 6225	g/VKT	
		-	1.62E-03	g/veh/m	
		=	2.27E-02	g/m/hr	
		=	6.31E-06	g/m/s	
		=	2.18E-04	g/s	
		=	2.10E-06	g/s/m <sup>2</sup>	
ED10 14					
CP19_14	Control Efficiency	=	97.8	%	(Note 3)
	Trucks Travelling Frequency	=	14	veh/hr	(Note 4)
	Road Length	=	29.0	m	
	Area of paved road travelled	=	87.0	m <sup>2</sup>	
	RSP Emission Factor (uncontrolled) of payed road	-	142 4534	α/\/KT	
·	RSP Emission Factor (controlled) of paved road	=	3.1766	g/VKT	
	···· -································	=	3.18E-03	g/veh/m	
		=	4.45E-02	g/m/hr	
		=	1.24E-05	g/m/s	
		=	3.58E-04	g/s	
		=	<u>4.12E-06</u>	g/s/m <sup>2</sup>	
1	FSP Emission Factor (uncontrolled) of paved road	=	34,4645	g/VKT	
	FSP Emission Factor (controlled) of paved road	=	0.7685	g/VKT	
		=	7.69E-04	g/veh/m	
		=	1.08E-02	g/m/hr	
		=	2.99E-06	g/m/s	
		=	8.67E-05	g/s	
		=	<u>9.96E-07</u>	g/s/m²	
From Cementitious Tanker					
Mitigation measures: water spr	aying and wheel washing				
	Average Loaded Weight of Truck	=	38	tons	
	Average Unladden Weight of Trucks	=	12	tons	
	Average Weight of Trucks	-	25		(Nets 1)
	ĸ	-	3.23	g/VKI (for ISP)	(Note 1)
		-	0.02	g/VKT (for FSP)	
	sL = road surface silt loading	=	12	g/m2	(Note 2)
EP19_1					(1)
	Control Efficiency Trucks Travelling Frequency	-	93.1	% veb/br	(Note 3)
	Road Length	-	0 15 7	m	(NOLE 4)
	Area of payed road travelled	=	47.1	m <sup>2</sup>	
I	RSP Emission Factor (uncontrolled) of paved road	=	158.6154	g/VKT	
	RSP Emission Factor (controlled) of paved road	=	11.0041	g/VKT	
		-	1.10E-02	g/veh/m	
		-	8.80E-02	g/m/m	
		-	3.84E-04	g/11/3 g/s	
		=	8.15E-06	g/s/m <sup>2</sup>	
I	FSP Emission Factor (uncontrolled) of paved road	=	38.3747	g/VKT	
	FSP Emission Factor (controlled) of paved road	=	2.6623	g/VKI	
		-	2.00E-U3	g/ven/m g/m/br	
		-	5.92F-06	в/ 11/ 11 g/m/s	
		=	9.29E-05	g/s	
		=	<u>1.97E-06</u>	g/s/m <sup>2</sup>	
			-		
EP19_2	Control Efficiency	=	95 2	%	(Note 3)
	Trucks Travelling Frequency	-	4	veh/hr	(Note 4)
	Road Length	=	20.2	m	(
	Area of paved road travelled	=	60.6	m²	
			450 51-	- 6 0/7	
I	RSP Emission Factor (uncontrolled) of paved road	=	158.6154	g/VKT	
	Nor Emission ractor (controlled) of paved road	-	7.40/1 7.47F-02	g/vni g/veh/m	
		-	2.99F-02	g/m/hr	
		=	8.30E-06	g/m/s	
		=	1.68E-04	g/s	
		=	<u>2.77E-06</u>	g/s/m <sup>2</sup>	
				h 11.000	
I	FSP Emission Factor (uncontrolled) of paved road	=	38.3747	g/VKT	
	For Emission Factor (controlled) of paved road	=	1.8066	g/VKI	
		-	1.81E-03	g/weii/iii g/m/br	
		-	2,01F-06	g/m/s	
		=	4.05E-05	g/s	
		=	6.69E-07	g/s/m <sup>2</sup>	
EP19_3					
	Control Efficiency	=	95.3	%	(Note 3)
	Irucks Iravelling Frequency	=	4	ven/nr	(Note 4)

	Road Length	=	19.9	m	
	Area of paved road travelled	=	59.7	m <sup>2</sup>	
			55.7		
	RSP Emission Factor (uncontrolled) of payed road	=	158.6154	g/VKT	
	RSP Emission Factor (controlled) of paved road	=	7.4671	g/VKT	
	· · · · · · · · · · · · · · · · · · ·	=	7.47E-03	g/veh/m	
		=	2.99E-02	g/m/hr	
		=	8.30E-06	g/m/s	
		=	1.65E-04	g/s	
		=	2.77E-06	g/s/m <sup>2</sup>	
	FSP Emission Factor (uncontrolled) of paved road	=	38.3747	g/VKT	
	FSP Emission Factor (controlled) of paved road	=	1.8066	g/VKT	
		=	1.81E-03	g/veh/m	
		=	7.23E-03	g/m/hr	
		=	2.01E-06	g/m/s	
		=	3.99E-05	g/s	
		=	<u>6.69E-07</u>	g/s/m²	
CD10 4					
EP19_4	Control Efficiency	_	05.2	0/	(Noto 2)
	Trucks Travelling Frequency	_	4	veh/hr	(Note 4)
	Road Length	=	9.7	m	(11010-1)
	Area of payed road travelled	-	29.1	m <sup>2</sup>	
		-	25.1		
	RSP Emission Factor (uncontrolled) of payed road	=	158.6154	g/VKT	
	RSP Emission Factor (controlled) of paved road	=	7.4671	g/VKT	
		=	7.47E-03	g/veh/m	
		=	2.99E-02	g/m/hr	
		=	8.30E-06	g/m/s	
		=	8.05E-05	g/s	
		=	2.77E-06	g/s/m <sup>2</sup>	
	FSP Emission Factor (uncontrolled) of paved road	=	38.3747	g/VKT	
	FSP Emission Factor (controlled) of paved road	=	1.8066	g/VKT	
		=	1.81E-03	g/veh/m	
		=	7.23E-03	g/m/hr	
		=	2.01E-06	g/m/s	
		=	1.95E-05	g/s	
		=	6.69E-07	g/s/m <sup>2</sup>	
EP19_5					4
	Control Efficiency	=	95.3	%	(Note 3)
	Trucks Travelling Frequency	=	4	ven/hr	(Note 4)
	Road Length	=	11.3	m 2	
	Area of paved road travelled	=	33.9	m	
	PSP Emission Eactor (uncontrolled) of payed road	_	158 6154	αΔ/KT	
	RSP Emission Factor (uncontrolled) of paved road	-	7 /671	g/ V K I g///KT	
	tor Emission ractor (controlled) or paved road	-	7 47F-03	g/veh/m	
		=	2 99E-02	g/m/hr	
		=	8.30E-06	g/m/s	
		=	9.38E-05	g/s	
		=	2.77E-06	$g/s/m^2$	
				8/ •/ ···	
	FSP Emission Factor (uncontrolled) of paved road	=	38.3747	g/VKT	
	FSP Emission Factor (controlled) of paved road	=	1.8066	g/VKT	
		=	1.81E-03	g/veh/m	
		=	7.23E-03	g/m/hr	
		=	2.01E-06	g/m/s	
		=	2.27E-05	g/s	
		=	<u>6.69E-07</u>	g/s/m <sup>2</sup>	
EP19_6					
	Control Efficiency	=	95.3	%	(Note 3)
	Trucks Travelling Frequency	=	4	veh/hr	(Note 4)
	Road Length	=	13.1	m 2	
	Area of paved road travelled	=	39.3	m <sup>-</sup>	
	DCD Emission Easter (unservice lie d) of any 1	_	150 645 -	~ \//T	
	nor emission ractor (uncontrolled) of paved road	-	108.0154	g/VKI	
	nor emission ractor (controlled) of paved road	-	7.46/1	g/VKI	
		-	7.47E-03	g/ven/m	
		-	2.39E-02 8.30F-06	g/m/s	
		=	1.09E-04	g/11/3 g/s	
		=	2.77F-06	$g/s/m^2$	
			<u>/L<sup>-</sup>00</u>	6/ 3/ 11	
	FSP Emission Factor (uncontrolled) of paved road	=	38.3747	g/VKT	
	FSP Emission Factor (controlled) of paved road	=	1.8066	g/VKT	
		=	1.81E-03	g/veh/m	
		=	7.23E-03	g/m/hr	
		=	2.01E-06	g/m/s	
		=	2.63E-05	g/s	
		=	<u>6.69E-07</u>	g/s/m <sup>2</sup>	
EP19_7					
	Control Efficiency	=	95.3	%	(Note 3)
	Trucks Travelling Frequency	=	4	veh/hr	(Note 4)
	Road Length	=	34.6	m	
	Area of paved road travelled	=	103.8	m²	
	RSP Emission Factor (uncontrolled) of paved road	=	158.6154	g/VKT	
	KSP Emission Factor (controlled) of paved road	=	/.4671	g/VKT	
		-	7.4/E-03	g/veh/m	
		-	2.JJE=UZ	g/11/11	

	=	8.30E-06	g/m/s	
	=	2.87E-04	g/s	
	=	2.77E-06	g/s/m <sup>2</sup>	
FSP Emission Factor (uncontrolled) of paved road	=	38.3747	g/VKT	
FSP Emission Factor (controlled) of paved road	=	1.8066	g/VKT	
	=	1.81E-03	g/veh/m	
	=	7.23E-03	g/m/hr	
	=	2.01E-06	g/m/s	
	=	6.95E-05	g/s	
	=	6.69E-07	g/s/m <sup>2</sup>	
EP19_14				
Control Efficiency	=	97.8	%	(Note 3)
Trucks Travelling Frequency	=	4	veh/hr	(Note 4)
Road Length	=	29.0	m	
Area of paved road travelled	=	87.0	m <sup>2</sup>	
RSP Emission Factor (uncontrolled) of paved road	=	158.6154	g/VKT	
RSP Emission Factor (controlled) of paved road	=	3.5370	g/VKT	
	=	3.54E-03	g/veh/m	
	=	1.41E-02	g/m/hr	
	=	3.93E-06	g/m/s	
	=	1.14E-04	g/s	
	=	<u>1.31E-06</u>	g/s/m <sup>2</sup>	
FSP Emission Factor (uncontrolled) of paved road	=	38.3747	g/VKT	
FSP Emission Factor (controlled) of paved road	=	0.8557	g/VKT	
	=	8.56E-04	g/veh/m	
	=	3.42E-03	g/m/hr	
	=	9.51E-07	g/m/s	
	=	2.76E-05	g/s	
	=	<u>3.17E-07</u>	g/s/m <sup>2</sup>	

Note:

Particle size multiplier is referred to Table 13.2-1-1, Section 13.2.1 of USEPA AP42 (ver 1/11) for RSP (PM10) and FSP (PM2.5).
 Road surfacing silt loading is referred to Table 13.2.1-3 of USEPA AP42 (ver 1/11) for concrete batching.
 Percentage of dust mitigation efficiency is calculated with reference to Cowherd et al., "Control of Open Fugitive Dust Sources, EPA-450/3-88-008, U.S. Environmental Protection Agency, Research Triangle
 The vehicle numbers within the site is advised by the applicant.

# Calculation of Emissions from Paved Roads (1900 to 2300)

SP Licence Application for Concrete Batching Plant at TKO 137 Area USEPA AP-42, Section 13.2.1.3, version 1/2011 Method Particulate Emission Factor (g/VKT), E k (sL)<sup>0.91</sup> (W)<sup>1.02</sup> = where particle size multiplier k = sL road surface silt loading = w average weight of the vehicles traveling the road \_ From Concrete Trucks Mitigation measures: water spraying and wheel washing Average Loaded Weight of Truck 30 tons Average Unladden Weight of Trucks 15 tons Average Weight of Truck 22.5 tons 3.23 g/VKT (for TSP) 0.62 g/VKT (for RSP) g/VKT (for FSP) 0.15 g/m<sup>2</sup> sL = road surface silt loading 12 EP19\_1 Control Efficiency 93.1 % veh/h Trucks Travelling Frequency 6 Road Length 15.7 m m<sup>2</sup> Area of paved road travelled 47.1 142.4534 RSP Emission Factor (uncontrolled) of paved road g/VKT RSP Emission Factor (controlled) of paved road 9.8829 g/VKT 9.88E-03 g/veh/m 5.93E-02 g/m/hr 1.65E-05 g/m/s 2.59E-04 g/s 5.49E-06 g/s/m<sup>2</sup> FSP Emission Factor (uncontrolled) of paved road 34.4645 g/VKT FSP Emission Factor (controlled) of paved road 2.3910 g/VKT 2.39E-03 g/veh/m 1.43E-02 g/m/hr 3.99E-06 g/m/s 6.26E-05 g/s <u>1.33E-06</u> g/s/m<sup>2</sup> EP19\_2 Control Efficiency 95.3 % Trucks Travelling Frequency 6 veh/h 20.2 Road Length = m m<sup>2</sup> Area of payed road travelled 60.6 RSP Emission Factor (uncontrolled) of paved road 142.4534 g/VKT g/VKT RSP Emission Factor (controlled) of paved road 6.7062 6.71E-03 g/veh/m 4.02E-02 g/m/hr 1.12E-05 g/m/s 2.26E-04 g/s 3.726E-06 g/s/m FSP Emission Factor (uncontrolled) of paved road 34.465 g/VKT FSP Emission Factor (controlled) of paved road 1.6225 g/VKT 1.62E-03 g/veh/m 9.73E-03 g/m/hr 2.70E-06 g/m/s 5.46E-05 g/s = <u>9.01E-07</u> g/s/m<sup>2</sup>

(Note 1)

(Note 2)

(Note 3)

(Note 4)

(Note 3)

(Note 4)

EP19 3 Control Efficiency 95.3 % (Note 3) Trucks Travelling Frequency 6 veh/h (Note 4) Road Length 19.9 m m<sup>2</sup> Area of paved road travelled 59.7 RSP Emission Factor (uncontrolled) of paved road 142.4534 g/VKT RSP Emission Factor (controlled) of paved road 6.7062 g/VKT 6.71E-03 g/veh/m 4.02E-02 g/m/hr 1.12E-05 g/m/s 2.22E-04 g/s 3.726E-06 g/s/m<sup>2</sup> FSP Emission Factor (uncontrolled) of paved road 34.465 g/VKT FSP Emission Factor (controlled) of paved road 1.6225 g/VKT 1.62E-03 g/veh/m 9.73E-03 g/m/hr 2.70E-06 g/m/s = 5.38E-05 g/s <u>9.01E-07</u> g/s/m<sup>2</sup> EP19\_4 Control Efficiency 95.3 % (Note 3) 6 veh/hr Trucks Travelling Frequency (Note 4) 9.7 Road Length m m² Area of paved road travelled 29.1 RSP Emission Factor (uncontrolled) of paved road 142.4534 g/VKT 6.7062 RSP Emission Factor (controlled) of paved road g/VKT 6.71E-03 g/veh/m

		-	4 02E-02	g/m/hr	
		_	4.022-02	g/m/c	
		=	1.12E-05	g/m/s	
		=	1.08E-04	g/s	
		=	3.726E-06	g/s/m <sup>2</sup>	
	ESP Emission Factor (uncontrolled) of payed road	=	34.465	g/VKT	
	ESP Emission Eactor (controlled) of paved road	-	1 6225	g/\/KT	
	Tor Emission racio (controlled) or parea road	_	1 625 02	g/uch/m	
		-	1.022-03	g/ven/m	
		=	9.73E-03	g/m/hr	
		=	2.70E-06	g/m/s	
		=	2.62E-05	g/s	
		=	9.01E-07	g/s/m <sup>2</sup>	
			<u>21012 07</u>	8/ 4/	
FD10 F					
EP19_5					(
	Control Efficiency	=	95.3	%	(Note 3)
	Trucks Travelling Frequency	=	6	veh/hr	(Note 4)
	Road Length	=	11.3	m	
	Area of payed road travelled	=	33.9	m <sup>2</sup>	
	· · · · · · · · · · · · · · · · · · ·				
	PSP Emission Eactor (uncontrolled) of payed road	_	142 4524	a/V/KT	
	RCD Emission Factor (uncontrolled) of paved road	_	6 7062	g/ VKT	
	RSP Emission Factor (controlled) of paved road	=	6.7062	g/vki	
		=	6.71E-03	g/veh/m	
		=	4.02E-02	g/m/hr	
		=	1.12E-05	g/m/s	
		=	1.26E-04	g/s	
		_	2.202.01	6/5 a/a/m <sup>2</sup>	
		-	3.726E-06	g/s/m	
	FSP Emission Factor (uncontrolled) of paved road	=	34.465	g/VKT	
	FSP Emission Factor (controlled) of paved road	=	1.6225	g/VKT	
		=	1.62E-03	g/veh/m	
		=	9 73F-02	g/m/hr	
		-	2 705 00	6/11/11	
		-	2./UE-Ub	g/11/5	
		=	3.06E-05	g/s	
		=	<u>9.01E-07</u>	g/s/m <sup>2</sup>	
EP19 6					
21 25_0	Control Efficiency	_	05.2	0/	(Noto 2)
		-	95.5	70 . /.	(Note 3)
	Trucks Travelling Frequency	=	6	veh/hr	(Note 4)
	Road Length	=	13.1	m	
	Area of payed road travelled	=	39.3	m <sup>2</sup>	
	RSD Emission Easter (uncentrolled) of payed read	_	142 4524	a li /kt	
	RSP Emission Factor (uncontrolled) of paved road	-	142.4554	g/VKT	
	RSP Emission Factor (controlled) of paved road	=	6.7062	g/vki	
		=	6.71E-03	g/veh/m	
		=	4.02E-02	g/m/hr	
		=	1.12E-05	g/m/s	
		=	1.46E-04	g/s	
		_	2,7005,000	6/5 -/-/2	
		-	3.726E-06	g/s/m	
	FSP Emission Factor (uncontrolled) of paved road	=	34.465	g/VKT	
	FSP Emission Factor (controlled) of paved road	=	1.6225	g/VKT	
		=	1.62E-03	g/veh/m	
		_	0 72E-02	g/m/hr	
		-	3.73L-03	g/11/11	
		=	2.70E-06	g/m/s	
		=	3.54E-05	g/s	
		=	<u>9.01E-07</u>	g/s/m <sup>2</sup>	
FP19 7					
LI 15_7	Control Efficiency	_	05.2	0/	(Noto 2)
	Control Enciency	-	95.3	70	(Note 3)
	Trucks Travelling Frequency	=	6	veh/hr	(Note 4)
	Road Length	=	34.6	m	
	Area of paved road travelled	=	103.8	m <sup>2</sup>	
	RSP Emission Factor (uncontrolled) of payed road	=	147 1521	g/VKT	
	RSD Emission Factor (antonicolled) of paved foad	_	142.4334	g/ V N I	
	NOP ETHISSION FACTOR (CONTROLLED) OF DAVED FOOD	-	0.7062	g/vrl	
		=	6./1E-03	g/ven/m	
		=	4.02E-02	g/m/hr	
		=	1.12E-05	g/m/s	
		=	3.87E-04	g/s	
		=	3.726F-06	$g/s/m^2$	
			<u>5.7 202 00</u>	0/ 1/ 11	
	FCD Exclusion From 1 ( )		· · · ·	- 6	
	FSP Emission Factor (uncontrolled) of paved road	=	34.465	g/VKT	
	FSP Emission Factor (controlled) of paved road	=	1.6225	g/VKT	
		=	1.62E-03	g/veh/m	
		=	9.73E-03	g/m/hr	
		=	2,70F-06	g/m/s	
		-	0 365 05	a/s	
		-	3.30E-U5	g/ 5	
		=	<u>9.01E-07</u>	g/s/m²	
EP19_8					
-	Control Efficiency	=	97.5	%	(Note 3)
	Trucks Travelling Frequency	-	د. د	veb/br	(Note 4)
	Trucks Travelling Frequency	=	10.2	ven/nr	(NOTE 4)
	Road Length	=	10.0	m	
	Area of paved road travelled	=	30.0	m <sup>2</sup>	
			-		
	RSP Emission Factor (uncontrolled) of paved road	=	142 4534	g/VKT	
	RSP Emission Eactor (controlled) of paved road	-	2 5200	α/\/⊮T	
	Nor Emission Factor (controlled) of paved foad	_	3.3290	g/vNI	
		=	3.53E-03	g/veh/m	
		=	2.12E-02	g/m/hr	
		=	5.88E-06	g/m/s	
		=	5.88E-05	g/s	
		=	1 0615 00	g/s/m <sup>2</sup>	
		-	T.30TF-00	g/ 3/111	
			···	- 6	
	FSP Emission Factor (uncontrolled) of paved road	=	34.465	g/VKT	
	FSP Emission Factor (controlled) of payed road	=	0.8539	g/VKT	

	=	8.54E-04	g/veh/m	
	=	5.12E-03	g/m/hr	
	=	1.42E-06	g/m/s	
	=	1.42E-05	g/s	
	=	<u>4.74E-07</u>	g/s/m²	
5040.0				
EP19_9	fficiency =	98.8	%	(Note 3)
Trucks Travelling E	requency =	3	veh/hr	(Note 4)
Roa	d Length =	10.3	m	(11010-1)
Area of paved road	travelled =	30.9	m <sup>2</sup>	
	avened	50.5		
RSP Emission Factor (uncontrolled) of pa	ved road =	142.4534	g/VKT	
RSP Emission Factor (controlled) of pa	ved road =	3.5296	g/VKT	
	=	3.53E-03	g/veh/m	
	=	1.06E-02	g/m/hr	
	=	2.94E-06	g/m/s	
	=	3.03E-05	g/s	
	=	<u>9.804E-07</u>	g/s/m²	
			4.000	
FSP Emission Factor (uncontrolled) of pa	ved road =	34.465	g/VKI	
FSP Emission Factor (controlled) of pa	ved road =	0.8539 8.54E-04	g/VKI	
	-	2.54E-04	g/ven/m	
	=	7 12F-07	g/m/s	
	=	7.33E-06	g/s	
	=	2.37E-07	g/s/m <sup>2</sup>	
			8/ -/	
EP19_10				
Control E	fficiency =	98.8	%	(Note 3)
Trucks Travelling F	requency =	3	veh/hr	(Note 4)
Roa	d Length =	16.3	m	
Area of paved road	travelled =	48.9	m <sup>2</sup>	
RSP Emission Factor (uncontrolled) of pa	ved road =	142.4534	g/VKT	
RSP Emission Factor (controlled) of pa	ved road =	3.5296	g/VKT	
	=	3.53E-03	g/veh/m	
	=	1.06E-02	g/m/hr	
	-	2.94E-06	g/m/s	
	-	4.792-03	g/s	
	-	9.804E-07	g/s/m	
ESP Emission Factor (uncontrolled) of pa	ved road =	34,465	g/VKT	
ESP Emission Factor (controlled) of pa	ved road =	0.8539	g/VKT	
	=	8.54E-04	g/veh/m	
	=	2.56E-03	g/m/hr	
	=	7.12E-07	g/m/s	
	=	1.16E-05	g/s	
	=	2.37E-07	g/s/m <sup>2</sup>	
EP19_11				
EP19_11 Control E	fficiency =	98.8	%	(Note 3)
EP19_11 Control E Trucks Travelling F	fficiency = requency =	98.8 3	% veh/hr	(Note 3) (Note 4)
EP19_11 Control E Trucks Travelling F Roa	fficiency = requency = d Length =	98.8 3 14.4	% veh/hr m	(Note 3) (Note 4)
EP19_11 Control E Trucks Travelling Fi Roa Area of paved road	fficiency = requency = d Length = travelled =	98.8 3 14.4 43.2	% veh/hr m <sup>2</sup>	(Note 3) (Note 4)
EP19_11 Control E Trucks Travelling F Roa Area of paved road BSP Emission Eactor (uncontrolled) of pa	fficiency = requency = d Length = travelled =	98.8 3 14.4 43.2	% veh/hr m m <sup>2</sup> g///KT	(Note 3) (Note 4)
EP19_11 Control E Trucks Travelling Fi Roz Area of paved road RSP Emission Factor (uncontrolled) of pa BSP Emission Factor (uncontrolled) of pa	fficiency = requency = d Length = travelled = ved road = ved road =	98.8 3 14.4 43.2 142.4534 3 5296	% veh/hr m m <sup>2</sup> g/VKT g/VKT	(Note 3) (Note 4)
EP19_11 Control E Trucks Travelling Fi Roz Area of paved road RSP Emission Factor (uncontrolled) of pa RSP Emission Factor (controlled) of pa	fficiency = requency = d Length = travelled = ved road = =	98.8 3 14.4 43.2 142.4534 3.5296 3.53E-03	% veh/hr m <sup>2</sup> g/VKT g/VKT g/vKT	(Note 3) (Note 4)
EP19_11 Control E Trucks Travelling F Roa Area of paved road RSP Emission Factor (uncontrolled) of pa RSP Emission Factor (controlled) of pa	fficiency = requency = d Length = travelled = ved road = ved road = =	98.8 3 14.4 43.2 142.4534 3.5296 3.532603 1.06E-02	% veh/hr m <sup>2</sup> g/VKT g/VKT g/VkT g/veh/m g/m/hr	(Note 3) (Note 4)
EP19_11 Control E Trucks Travelling Fi Roa Area of paved road RSP Emission Factor (uncontrolled) of pa RSP Emission Factor (controlled) of pa	fficiency = requency = d Length = travelled = ved road = = = = =	98.8 3 14.4 43.2 142.4534 3.5296 3.53E-03 1.06E-02 2.94E-06	% veh/hr m <sup>2</sup> g/VKT g/VKT g/veh/m g/m/hr g/m/s	(Note 3) (Note 4)
EP19_11 Control E Trucks Travelling F Roa Area of paved road RSP Emission Factor (uncontrolled) of pa RSP Emission Factor (controlled) of pa	fficiency = requency = d Length = travelled = ved road = = = = = =	98.8 3 14.4 43.2 142.4534 3.5296 3.53E-03 1.06E-02 2.94E-06 4.24E-05	% veh/hr m 2 g/VKT g/VKT g/wk/m g/m/hr g/m/s g/s	(Note 3) (Note 4)
EP19_11 Control E Trucks Travelling F Roa Area of paved road RSP Emission Factor (uncontrolled) of pa RSP Emission Factor (controlled) of pa	fficiency = requency = d Length = travelled = ved road = e = = = = = =	98.8 3 14.4 43.2 142.4534 3.5296 3.53E-03 1.06E-02 2.94E-06 4.24E-05 <u>9.804E-07</u>	% veh/hr m g/VKT g/VKT g/veh/m g/m/hr g/m/s g/s g/s/m <sup>2</sup>	(Note 3) (Note 4)
EP19_11 Control E Trucks Travelling F Roa Area of paved road RSP Emission Factor (uncontrolled) of pa RSP Emission Factor (controlled) of pa	fficiency = requency = d Length = travelled = ved road = = = = = =	98.8 3 14.4 43.2 142.4534 3.5296 3.53E-03 1.06E-02 2.94E-06 4.24E-05 <u>9.804E-07</u>	% veh/hr m <sup>2</sup> g/VKT g/VKT g/veh/m g/w/hr g/m/hr g/s g/s	(Note 3) (Note 4)
EP19_11 Control E Trucks Travelling F Roa Area of paved road RSP Emission Factor (uncontrolled) of pa RSP Emission Factor (controlled) of pa	fficiency = requency = d Length = travelled = ved road = = = = = = = = =	98.8 3 14.4 3.2 142.4534 3.5296 3.53E-03 1.06E-02 2.94E-06 4.24E-05 <u>9.804E-07</u> 34.465	% veh/hr m g/VKT g/VKT g/veh/m g/w/hr g/m/s g/s g/s/m <sup>2</sup> g/vet	(Note 3) (Note 4)
EP19_11 Control E Trucks Travelling F Roa Area of paved road RSP Emission Factor (uncontrolled) of pa RSP Emission Factor (controlled) of pa FSP Emission Factor (uncontrolled) of pa	fficiency = requency = d Length = travelled = ved road = = = = = ved road = = = = ved road =	98.8 3 14.4 43.2 142.4534 3.5296 3.53E-03 1.06E-02 2.94E-06 4.24E-05 <u>9.804E-07</u> 34.465 0.8539	% veh/hr m g/VKT g/VKT g/VkT g/w/hr g/m/s g/s g/s/m <sup>2</sup> g/VKT g/VKT g/VKT	(Note 3) (Note 4)
EP19_11 Control E Trucks Travelling Fi Roa Area of paved road RSP Emission Factor (uncontrolled) of pa RSP Emission Factor (controlled) of pa FSP Emission Factor (uncontrolled) of pa	fficiency = equency = d Length = travelled = ved road = = = = = ved road = = = ved road = = =	98.8 3 14.4 43.2 142.4534 3.5296 3.538-03 1.06E-02 2.94E-06 4.24E-05 <u>9.804E-07</u> 34.465 0.8539 8.54E-04 2.56E-02	% veh/hr m g/VKT g/VKT g/veh/m g/m/s g/s/m <sup>2</sup> g/s/m <sup>2</sup> g/VKT g/VKT g/VKT g/VKT	(Note 3) (Note 4)
EP19_11 Control E Trucks Travelling Fi Roa Area of paved road RSP Emission Factor (uncontrolled) of pa RSP Emission Factor (controlled) of pa FSP Emission Factor (uncontrolled) of pa	fficiency = requency = d Length = travelled = ved road = e e = = ved road = = = ved road = = = ved road = =	98.8 3 14.4 43.2 142.4534 3.536-03 1.06E-02 2.94E-06 4.24E-05 <u>9.804E-07</u> 34.465 0.8539 8.54E-04 2.56E-03 7.12F-07	% veh/hr m g/VKT g/VKT g/veh/m g/m/hr g/m/s g/s g/s/s g/s/kT g/VKT g/VKT g/veh/m g/m/hr g/m/hr	(Note 3) (Note 4)
EP19_11 Control E Trucks Travelling Fi Roa Area of paved road RSP Emission Factor (uncontrolled) of pa RSP Emission Factor (controlled) of pa FSP Emission Factor (uncontrolled) of pa	fficiency = requency = d Length = travelled = ved road = e e ved road = e ved road = = = = = = = = = = = = = = = = = = =	98.8 3 14.4 43.2 142.4534 3.5296 3.53E-03 1.06E-02 2.94E-06 4.24E-05 9.804E-07 34.465 0.8539 8.54E-04 2.56E-03 7.12E-07 1.02F-05	% veh/hr m <sup>2</sup> g/VKT g/VKT g/veh/m g/m/hr g/s g/s/kT g/VKT g/VKT g/VKT g/VKT g/veh/m g/m/hr g/m/hr g/m/s g/s	(Note 3) (Note 4)
EP19_11 Control E Trucks Travelling F Roz Area of paved road RSP Emission Factor (uncontrolled) of pa RSP Emission Factor (controlled) of pa FSP Emission Factor (uncontrolled) of pa	fficiency = requency = d Length = travelled = ved road = = = = ved road = = = ved road = = = = = = = = = = = = = = = = = = =	98.8 3 14.4 3.2 142.4534 3.5296 3.53E-03 1.06E-02 2.94E-06 4.24E-05 <u>9.804E-07</u> 34.465 0.8539 8.54E-04 2.56E-03 7.12E-07 1.02E-05 2.37E-07	% veh/hr m g/VKT g/VKT g/VKT g/veh/m g/m/s g/s/m <sup>2</sup> g/VKT g/VKT g/VKT g/VKT g/veh/m g/m/hr g/s g/s	(Note 3) (Note 4)
EP19_11 Control E Trucks Travelling F Roa Area of paved road RSP Emission Factor (uncontrolled) of pa RSP Emission Factor (controlled) of pa FSP Emission Factor (uncontrolled) of pa	fficiency = requency = d Length = travelled = ved road = = = ved road = = ved road = = = = = = = = = = = = = = = = = = =	98.8 3 14.4 3.2 142.4534 3.5296 3.53E-03 1.06E-02 2.94E-06 4.24E-05 <u>9.804E-07</u> 34.465 0.8539 8.54E-04 2.56E-03 7.12E-07 1.02E-05 <u>2.37E-07</u>	% veh/hr m m <sup>2</sup> g/VKT g/VKTg/veh/m g/m/hr g/m/s g/s/m <sup>2</sup> g/VKT g/VKT g/VKT g/veh/m g/m/hr g/m/s g/s	(Note 3) (Note 4)
EP19_11 Control E Trucks Travelling Fi Roa Area of paved road RSP Emission Factor (uncontrolled) of pa RSP Emission Factor (controlled) of pa FSP Emission Factor (uncontrolled) of pa FSP Emission Factor (controlled) of pa	fficiency = requency = d Length = travelled = ved road = = = = ved road = = = ved road = = = = = = = = = = = = = = = = = = =	98.8 3 14.4 43.2 142.4534 3.5296 3.53E-03 1.06E-03 2.94E-06 4.24E-05 <u>9.804E-07</u> 34.465 0.8539 8.54E-04 2.56E-03 7.12E-07 1.02E-05 <u>2.37E-07</u>	% veh/hr m m <sup>2</sup> g/VKT g/VKT g/veh/m g/m/s g/s/m <sup>2</sup> g/VKT g/VKT g/VKT g/veh/m g/m/hr g/m/s g/s g/s/m <sup>2</sup>	(Note 3) (Note 4)
EP19_11 Control E Trucks Travelling Fi Roa Area of paved road RSP Emission Factor (uncontrolled) of pa RSP Emission Factor (uncontrolled) of pa FSP Emission Factor (uncontrolled) of pa FSP Emission Factor (controlled) of pa	fficiency = requency = d Length = travelled = ved road = e ved road = = e ved road = = = = = = = = = = = = = = = = = = =	98.8 3 14.4 43.2 142.4534 3.53E-03 1.06E-02 2.94E-06 4.24E-05 <u>9.804E-07</u> 34.465 0.8539 8.54E-04 2.56E-03 7.12E-07 1.02E-05 <u>2.37E-07</u> 98.8	% veh/hr m 2 g/VKT g/VKT g/veh/m g/m/hr g/s/g/s/m <sup>2</sup> g/VKT g/VKT g/VKT g/veh/m g/m/hr g/m/s g/s/m <sup>2</sup>	(Note 3) (Note 4)
EP19_11 Control E Trucks Travelling Fi Roa Area of paved road RSP Emission Factor (uncontrolled) of pa RSP Emission Factor (controlled) of pa FSP Emission Factor (uncontrolled) of pa FSP Emission Factor (controlled) of pa FSP Emission Factor (controlle	fficiency = requency = d Length = travelled = ved road = ved road = = = ved road = = = = ved road = = = = = = = = = = = = = = = = = = =	98.8 3 14.4 43.2 142.4534 3.53E-03 1.06E-02 2.94E-06 4.24E-05 <u>9.804E-07</u> 34.465 0.8539 8.54E-04 2.56E-03 7.12E-07 1.02E-05 <u>2.37E-07</u> 98.8 3	% veh/hr m <sup>2</sup> g/VKT g/VKT g/VKT g/veh/m g/s g/s/m <sup>2</sup> g/VKT g/VKT g/VKT g/veh/m g/m/hr g/m/hr g/m/s g/s/m <sup>2</sup>	(Note 3) (Note 4) (Note 3) (Note 4)
EP19_11 Control E Trucks Travelling F Roz Area of paved road RSP Emission Factor (uncontrolled) of pa RSP Emission Factor (uncontrolled) of pa FSP Emission Factor (uncontrolled) of pa FSP Emission Factor (controlled) of pa EP19_12 Control E Trucks Travelling F Roz	fficiency = requency = d Length = travelled = ved road = ved road = e ved road = e e travelroad = e e e travelroad = e e e e travelroad = e e e e e e e e e e e e e e e e e e e	98.8 3 14.4 3.2 142.4534 3.5296 3.53E-03 1.06E-02 2.94E-06 4.24E-05 <u>9.804E-07</u> 34.465 0.8539 8.54E-04 2.56E-03 7.12E-07 1.02E-05 <u>2.37E-07</u> 98.8 3 15.9	% veh/hr m g/VKT g/VKT g/vkh/m g/m/hr g/m/s g/s/m <sup>2</sup> g/VKT g/VKT g/VKT g/veh/m g/m/hr g/m/s g/s/m <sup>2</sup>	(Note 3) (Note 4) (Note 3) (Note 4)
EP19_11 Control E Trucks Travelling Fi Roa Area of paved road RSP Emission Factor (uncontrolled) of pa RSP Emission Factor (uncontrolled) of pa FSP Emission Factor (uncontrolled) of pa FSP Emission Factor (controlled) of pa FSP Emission Factor (control	fficiency = requency = d Length = travelled = ved road = e ved road = e ved road = e e travel road = e e e e e e fficiency = requency = d Length = travelled =	98.8 3 14.4 3.2 142.4534 3.5296 3.53E-03 1.06E-02 2.94E-06 4.24E-05 <u>9.804E-07</u> 34.465 0.8539 8.54E-04 2.56E-03 7.12E-07 1.02E-05 <u>2.37E-07</u> 98.8 3 15.9 47.7	% veh/hr m g/VKT g/VKT g/veh/m g/v/hr g/s/m <sup>2</sup> g/VKT g/VKT g/vKT g/vKT g/veh/m g/m/s g/s g/s/m <sup>2</sup> % veh/hr m m <sup>2</sup>	(Note 3) (Note 4) (Note 3) (Note 4)
EP19_11 Control E Trucks Travelling Fi Roa Area of paved road RSP Emission Factor (uncontrolled) of pa RSP Emission Factor (uncontrolled) of pa FSP Emission Factor (uncontrolled) of pa FSP Emission Factor (controlled) of pa FSP Emissic (controlled) of	fficiency = requency = d Length = travelled = ved road = = = ved road = = = ved road = = = = fficiency = requency = equency = travelled =	98.8 3 14.4 3.2 142.4534 3.5296 3.53E-03 1.06E-02 2.94E-06 4.24E-05 <u>9.804E-07</u> 34.465 0.8539 8.54E-04 2.56E-03 7.12E-07 1.02E-05 <u>2.37E-07</u> 98.8 3 15.9 4.7	% veh/hr m m <sup>2</sup> g/VKT g/VKTg/veh/m g/m/s g/s/m <sup>2</sup> g/VKT g/VKT g/VKT g/VKT g/veh/m g/m/s g/s g/s/m <sup>2</sup> % veh/hr m m <sup>2</sup>	(Note 3) (Note 4) (Note 3) (Note 4)
EP19_11 Control E Trucks Travelling Fi Roa Area of paved road RSP Emission Factor (uncontrolled) of pa RSP Emission Factor (uncontrolled) of pa FSP Emission Factor (controlled) of pa FSP Emission Factor (controlled) of pa FSP Emission Factor (controlled) of pa FSP Emission Factor (uncontrolled) of pa	fficiency = requency = d Length = travelled = ved road = ved road = = ved road = = e ved road = = = fficiency = requency = d Length = travelled = ved road =	98.8 3 14.4 43.2 142.4534 3.536-03 1.06E-02 2.94E-06 4.24E-05 <u>9.804E-07</u> 34.465 0.8539 8.54E-04 2.56E-03 7.12E-07 1.02E-05 <u>2.37E-07</u> 98.8 3 15.9 47.7 142.4534	% veh/hr m 2 g/VKT g/VKT g/VKT g/veh/m g/s g/s/m <sup>2</sup> g/VKT g/vKT g/vKT g/vh/m g/m/hr g/m/hr g/s g/s/m <sup>2</sup> % veh/hr m m <sup>2</sup>	(Note 3) (Note 4) (Note 3) (Note 4)
EP19_11 Control E Trucks Travelling Fi Roa Area of paved road RSP Emission Factor (uncontrolled) of pa RSP Emission Factor (uncontrolled) of pa FSP Emission Factor (controlled) of pa FSP Emission Factor (controlled) of pa Control E Trucks Travelling Fi Roa Area of paved road RSP Emission Factor (uncontrolled) of pa RSP Emission Factor (uncontrolled) of pa Control E Trucks Travelling Fi Roa Area of paved road RSP Emission Factor (uncontrolled) of pa RSP Emission Factor (uncontrolled)	fficiency = requency = d Length = travelled = ved road = ved road = = e ved road = = e ved road = = e travelled = fficiency = requency = d Length = travelled = ved road = ved road =	98.8 3 14.4 43.2 142.4534 3.5296 3.53E-03 1.06E-02 2.94E-06 4.24E-05 <u>9.804E-07</u> 34.465 0.8539 8.54E-04 2.56E-03 7.12E-07 1.02E-05 <u>2.37E-07</u> 98.8 3 15.9 47.7	% veh/hr m <sup>2</sup> g/VKT g/VKT g/veh/m g/m/hr g/s g/s/m <sup>2</sup> g/VKT g/VKT g/veh/m g/m/hr g/s/m <sup>2</sup> % veh/hr m m <sup>2</sup> g/VKT g/VKT	(Note 3) (Note 4) (Note 3) (Note 4)
EP19_11 Control E Trucks Travelling F Roz Area of paved road RSP Emission Factor (uncontrolled) of pa RSP Emission Factor (uncontrolled) of pa FSP Emission Factor (uncontrolled) of pa FSP Emission Factor (controlled) of pa Control E Trucks Travelling F Roz Area of paved road RSP Emission Factor (uncontrolled) of pa Control E Roz Area of paved road RSP Emission Factor (uncontrolled) of pa RSP Emission Factor (uncontrolled) of pa	fficiency = requency = d Length = travelled = ved road = e ved road = ved road = e requency = d Length = travelled = ved road = ved road =	98.8 3 14.4 3.5296 3.53E-03 1.06E-02 2.94E-06 4.24E-05 9.804E-07 34.465 0.8539 8.54E-04 2.56E-03 7.12E-07 1.02E-05 2.37E-07 98.8 3 15.9 47.7 142.4534 3.5296 3.53E-03 1.05E-02 3.526-03 1.05E-02 3.53E-03 1.05E-02 3.53E-03 3.5296 3.53E-03 1.05E-02 3.53E-03 1.05E-02 3.53E-03 3.5296 3.53E-03 1.05E-02 3.53E-03 3.5296 3.53E-03 1.05E-02 3.53E-03 3.5296 3.53E-03 3.5296 3.53E-03 3.5296 3.53E-03 3.5296 3.53E-03 3.5296 3.53E-03 3.5296 3.53E-03 3.5296 3.53E-03 3.5296 3.53E-03 3.5296 3.53E-03 3.5296 3.5260 3.526	% veh/hr m 2 g/VKT g/VKT g/veh/m g/m/hr g/s/m <sup>2</sup> g/VKT g/VKT g/veh/m g/m/hr g/s/m <sup>2</sup> g/s/m <sup>2</sup> % veh/hr m m <sup>2</sup> g/VKT g/VKT g/VKT g/VKT	(Note 3) (Note 4) (Note 3) (Note 4)
EP19_11 Control E Trucks Travelling F Roa Area of paved road RSP Emission Factor (uncontrolled) of pa RSP Emission Factor (uncontrolled) of pa FSP Emission Factor (uncontrolled) of pa FSP Emission Factor (controlled) of pa FSP Emission Factor (controlled) of pa RSP Emission Factor (uncontrolled) of pa RSP Emission Factor (	fficiency = requency = d Length = travelled = ved road = = = ved road = = = ved road = = = = fficiency = requency = d Length = travelled = ved road = = =	98.8 3 14.4 3.5296 3.53E-03 1.06E-02 2.94E-06 4.24E-05 <u>9.804E-07</u> <u>34.465</u> 0.8539 8.54E-04 2.56E-03 7.12E-07 1.02E-05 <u>2.37E-07</u> 98.8 3 1.5.9 47.7 142.4534 3.5296 3.53E-03 1.06E-02 2.94E-06	% veh/hr m m <sup>2</sup> g/VKT g/VKT g/VKT g/VKT g/VKT g/VKT g/VKT g/VKT g/VKT g/VKT g/VKT g/VKT g/VKT g/VKT g/VKT	(Note 3) (Note 4) (Note 3) (Note 4)
EP19_11 Control E Trucks Travelling Fi Roa Area of paved road RSP Emission Factor (uncontrolled) of pa RSP Emission Factor (uncontrolled) of pa FSP Emission Factor (uncontrolled) of pa FSP Emission Factor (controlled) of pa Control E Trucks Travelling Fi Roa Area of paved road RSP Emission Factor (uncontrolled) of pa RSP Emission Factor (uncontrolled) of pa	fficiency = equency = d Length = travelled = ved road = = = ved road = = = ved road = = = = travelled = travelled = ved road = ued road = = = = = = = = = = = = = = = = = = =	98.8 3 14.4 43.2 142.4534 3.5296 3.53E-03 1.06E-02 2.94E-06 4.24E-05 <u>9.804E-07</u> 34.465 0.8539 8.54E-04 2.56E-03 7.12E-07 1.02E-05 <u>2.37E-07</u> 98.8 3 15.9 47.7 142.4534 3.5296 3.53E-03 1.06E-02 2.94E-06	% veh/hr m m <sup>2</sup> g/VKT g/VKTg/veh/m g/m/hr g/m/s g/s/m <sup>2</sup> g/VKT g/VKTg/veh/m g/m/s g/s/m <sup>2</sup> % veh/hr m m <sup>2</sup> g/VKT g/VKTg/veh/m g/VKTg/veh/m g/veh/m	(Note 3) (Note 4) (Note 3) (Note 4)
EP19_11 Control E Trucks Travelling Fi Roa Area of paved road RSP Emission Factor (uncontrolled) of pa RSP Emission Factor (uncontrolled) of pa FSP Emission Factor (uncontrolled) of pa FSP Emission Factor (controlled) of pa Control E Trucks Travelling Fi Roa Area of paved road RSP Emission Factor (uncontrolled) of pa RSP Emission Factor (uncontrolled) of pa	fficiency = requency = d Length = travelled = ved road = ved road = = ved road = = ved road = = e travelled = travelled = travelled = ved road = = = = = = = = = = = = = = = = = = =	98.8 3 14.4 3.5296 3.53E-03 1.06E-03 2.94E-06 4.24E-05 9.804E-07 34.465 0.8539 8.54E-04 2.56E-03 7.12E-07 1.02E-05 2.37E-07 98.8 3 15.9 47.7 142.4534 3.5296 3.53E-03 1.06E-02 2.94E-06 4.68E-05	% veh/hr m 2 g/VKT g/VKT g/VKT g/veh/m g/m/hr g/s g/S/KT g/VKT g/VKT g/s/m <sup>2</sup> % veh/hr m m <sup>2</sup> g/VKT g/VKT g/VKT g/VKT g/VKT g/VKT g/VKT g/VKT	(Note 3) (Note 4) (Note 3) (Note 4)
EP19_11 Control E Trucks Travelling Fi Roa Area of paved road RSP Emission Factor (uncontrolled) of pa RSP Emission Factor (uncontrolled) of pa FSP Emission Factor (uncontrolled) of pa FSP Emission Factor (controlled) of pa Control E Trucks Travelling Fi Roa Area of paved road RSP Emission Factor (uncontrolled) of pa RSP Emission Factor (uncontrolled) of pa	fficiency = requency = d Length = travelled = ved road = ved road = e ved road = ved road = e tved road = e travelled = travelled = ved road = ved road = e = = = = = = = = = = = = = = = = = =	98.8 3 14.4 3.2 142.4534 3.5296 3.53E-03 1.06E-02 2.94E-06 4.24E-05 9.804E-07 34.465 0.8539 8.54E-04 2.56E-03 7.12E-07 1.02E-05 2.37E-07 1.02E-05 2.37E-07 1.02E-05 2.37E-07 1.02E-05 2.37E-07 1.02E-05 3.53E-03 3.1.06E-02 2.94E-06 4.68E-05 9.804E-07	% veh/hr m <sup>2</sup> g/VKT g/VKT g/veh/m g/m/hr g/s g/s/m <sup>2</sup> g/VKT g/VKT g/VKT g/s/m <sup>2</sup> % veh/hr m <sup>2</sup> g/VKT g/VKT g/VKT g/VKT g/VKT g/VKT g/VKT g/VKT g/VKT g/VKT	(Note 3) (Note 4) (Note 3) (Note 4)
EP19_11 Control E Trucks Travelling F Roz Area of paved road RSP Emission Factor (uncontrolled) of pa RSP Emission Factor (uncontrolled) of pa FSP Emission Factor (uncontrolled) of pa FSP Emission Factor (controlled) of pa Control E Trucks Travelling F Roz Area of paved road RSP Emission Factor (uncontrolled) of pa RSP Emission Factor (uncontrolled) of pa	fficiency = requency = d Length = travelled = ved road = ved road = ved road = ved road = travelled = travelled = travelled = ved road = ved ro	98.8 3 14.4 43.2 142.4534 3.5296 3.53E-03 1.06E-02 2.94E-06 4.24E-05 <u>9.804E-07</u> 34.465 0.8539 8.54E-04 2.56E-03 7.12E-07 1.02E-05 <u>2.37E-07</u> 1.02E-05 <u>2.37E-07</u> 1.02E-05 <u>2.37E-07</u> 1.42.4534 3.5296 3.53E-03 1.06E-02 2.94E-06 4.68E-05 <u>9.804E-07</u> 34.465	% veh/hr m 2 g/VKT g/VKT g/VKT g/veh/m g/m/hr g/vkf g/VKT g/VKT g/veh/m g/m/hr g/s/m <sup>2</sup> % veh/hr m m <sup>2</sup> g/VKT g/VKT g/VKT g/VKT g/VKT g/VKT	(Note 3) (Note 4) (Note 3) (Note 4)
EP19_11 Control E Trucks Travelling F Roz Area of paved road RSP Emission Factor (uncontrolled) of pa RSP Emission Factor (uncontrolled) of pa FSP Emission Factor (uncontrolled) of pa FSP Emission Factor (controlled) of pa RSP Emission Factor (uncontrolled) of pa	fficiency = requency = d Length = travelled = ved road = e ved road = e ved road = e requency = d Length = travelled = ved road = e ved road = e s ved road = e s ved road = e ved road = e e ved road = e ved road = e e e ved road = e e e e e e e e e e e e e e e e e e e	98.8 3 14.4 3.5296 3.53E-03 1.06E-02 2.94E-06 4.24E-05 9.804E-07 34.465 0.8539 8.54E-04 2.56E-03 7.12E-07 1.02E-05 2.37E-07 1.02E-05 2.34E-05 9.804E-07 2.94E-06 3.53E-03 1.06E-02 2.94E-06 3.53E-03 1.06E-02 2.94E-06 3.53E-03 1.06E-02 2.94E-06 3.53E-03 1.06E-02 2.94E-06 3.53E-03 1.06E-02 2.94E-06 3.53E-03 1.06E-02 2.94E-06 3.53E-03 1.06E-02 2.94E-06 3.53E-03 1.06E-02 2.94E-06 3.53E-03 3.52E-07 3.52E	% veh/hr m m² g/VKT g/VKTg/wh/m g/m/hr g/s/m² g/VKT g/VKT g/veh/m g/m/hr g/m/s g/s/m² % veh/hr m m² g/VKT g/VKT g/VKT g/VKT g/VKT g/m/s g/s/m²	(Note 3) (Note 4) (Note 3) (Note 4)
EP19_11 Control E Trucks Travelling Fi Roa Area of paved road RSP Emission Factor (uncontrolled) of pa RSP Emission Factor (uncontrolled) of pa FSP Emission Factor (controlled) of pa Area of paved road RSP Emission Factor (uncontrolled) of pa RSP Emission Factor (uncontrolled) of pa	fficiency = requency = d Length = travelled = ved road = e ved road = e ved road = e ravelroad = e equency = equency = d Length = travelled = ved road = e e e ued road = e e e ued road = e e e e e e e e e e e e e e e e e e e	98.8 3 14.4 3.5296 3.535-03 1.06E-02 2.94E-06 4.24E-05 <u>9.804E-07</u> 34.465 0.8539 8.54E-04 2.56E-03 7.12E-07 1.02E-05 <u>2.37E-07</u> 1.02E-05 <u>2.37E-07</u> 1.02E-05 <u>2.37E-07</u> 1.02E-05 <u>2.37E-07</u> 1.02E-05 <u>2.37E-07</u> 1.02E-05 <u>2.37E-07</u> 1.02E-05 <u>2.37E-07</u> 1.02E-05 <u>2.37E-07</u> 1.02E-05 <u>2.37E-07</u> 1.02E-05 <u>2.37E-07</u> 1.02E-05 <u>2.37E-07</u> 1.02E-05 <u>2.37E-07</u> 1.02E-05 <u>2.37E-07</u> 1.02E-05 <u>2.37E-07</u> 1.02E-05 <u>2.37E-07</u> 1.02E-05 <u>2.37E-07</u> 1.02E-05 <u>2.37E-07</u> 1.02E-05 <u>2.37E-07</u> 1.02E-05 <u>2.37E-07</u> 1.02E-05 <u>2.37E-07</u> 1.02E-05 <u>2.37E-07</u> 1.02E-05 <u>2.37E-07</u> 1.02E-05 <u>2.37E-07</u> 1.02E-05 <u>2.37E-07</u> 1.02E-05 <u>2.37E-07</u> 1.02E-05 <u>2.37E-07</u> 1.02E-05 <u>2.37E-07</u> 1.02E-05 <u>2.37E-07</u> 1.02E-05 <u>2.37E-07</u> 1.02E-05 <u>2.37E-07</u> 1.02E-05 <u>2.37E-07</u> 1.02E-05 <u>2.37E-07</u> 1.02E-05 <u>2.37E-07</u> 1.02E-05 <u>2.37E-07</u> 1.02E-05 <u>2.37E-07</u> 1.02E-05 <u>2.37E-07</u> 1.02E-05 <u>2.37E-07</u> 1.02E-05 <u>2.37E-07</u> 1.02E-05 <u>2.37E-07</u> 1.02E-05 <u>2.37E-07</u> 1.02E-05 <u>2.37E-07</u> 1.02E-05 <u>2.37E-07</u> 1.02E-05 <u>2.37E-07</u> 1.02E-05 <u>2.37E-07</u> 1.02E-05 <u>3.35E-03</u> 1.06E-02 <u>2.94E-06</u> <u>3.53E-03</u> 1.06E-02 <u>2.94E-06</u> <u>3.53E-03</u> 1.06E-02 <u>2.94E-06</u> <u>3.53E-03</u> 1.06E-02 <u>2.94E-06</u> <u>3.53E-03</u> 1.06E-02 <u>2.94E-06</u> <u>3.53E-03</u> 1.06E-02 <u>2.94E-06</u> <u>3.53E-03</u> 1.06E-02 <u>2.94E-06</u> <u>3.53E-03</u> 1.06E-02 <u>2.94E-06</u> <u>3.53E-03</u> 1.06E-02 <u>2.94E-06</u> <u>3.53E-03</u> 1.06E-02 <u>2.94E-06</u> <u>3.53E-03</u> 1.06E-02 <u>2.94E-06</u> <u>3.53E-03</u> 1.06E-02 <u>2.94E-06</u> <u>3.53E-03</u> <u>3.54E-07</u>	% veh/hr m m <sup>2</sup> g/VKT g/VKTg/Veh/m g/m/hr g/m/s g/s/m <sup>2</sup> g/VKT g/VKTg/Veh/m g/m/hr g/m/s g/s/m <sup>2</sup> % veh/hr m m <sup>2</sup> g/VKT g/VKTg/VKT g/VKT g/vch/m g/m/hr g/r/s g/s/m <sup>2</sup>	(Note 3) (Note 4) (Note 3) (Note 4)
EP19_11 Control E Trucks Travelling Fi Roa Area of paved road RSP Emission Factor (uncontrolled) of pa RSP Emission Factor (uncontrolled) of pa FSP Emission Factor (uncontrolled) of pa FSP Emission Factor (uncontrolled) of pa Area of paved road RSP Emission Factor (uncontrolled) of pa RSP Emission Factor (uncontrolled) of pa RSP Emission Factor (uncontrolled) of pa RSP Emission Factor (uncontrolled) of pa	fficiency = equency = d Length = travelled = ved road = ved road = e ved road = e e travelled = fficiency = d Length = e travelled = ved road = ved road = e e e ued road = e e e e e e e e e e e e e e e e e e e	98.8 3 14.4 43.2 142.4534 3.5296 3.53E-03 1.06E-02 2.94E-06 4.24E-05 <u>9.804E-07</u> 34.465 0.8539 8.54E-04 2.56E-03 7.12E-07 1.02E-05 <u>2.37E-07</u> 1.02E-05 <u>2.37E-07</u> 1.42.4534 3.5296 3.53E-03 1.06E-02 2.94E-06 4.68E-05 <u>9.804E-07</u>	% veh/hr m <sup>2</sup> g/VKT g/VKT g/VKT g/veh/m g/m/hr g/s g/S/MT g/VKT g/VKT g/VKT g/VKT g/VKT g/VKT g/VKT g/VKT g/VKT g/VKT g/VKT g/VKT g/VKT g/VKT g/VKT g/VKT g/VKT g/VKT g/VKT	(Note 3) (Note 4) (Note 3) (Note 4)
EP19_11 Control E Trucks Travelling Fi Roa Area of paved road RSP Emission Factor (uncontrolled) of pa RSP Emission Factor (uncontrolled) of pa FSP Emission Factor (uncontrolled) of pa FSP Emission Factor (uncontrolled) of pa Area of paved road RSP Emission Factor (uncontrolled) of pa RSP Emission Factor (uncontrolled) of pa	fficiency = requency = d Length = travelled = ved road = ved road = ved road = e ved road = e requency = d Length = travelled = ved road = ved ved ved ved ved ved ved ved ved ved	98.8 3 14.4 3.22 142.4534 3.5296 3.53E-03 1.06E-02 2.94E-06 4.24E-05 9.804E-07 34.465 0.8539 8.54E-04 2.56E-03 7.12E-07 1.02E-05 2.37E-07 1.02E-05 2.37E-07 1.02E-05 2.37E-07 1.02E-05 2.37E-07 1.02E-05 3.53E-03 1.06E-02 2.94E-06 4.68E-05 9.804E-07 34.465 0.8539 8.54E-04 2.56E-03 7.12E-07	% veh/hr m <sup>2</sup> g/VKT g/VKT g/VKT g/veh/m g/m/hr g/s g/s/m <sup>2</sup> g/VKT g/VKT g/veh/m g/s g/s/m <sup>2</sup> % veh/hr m <sup>2</sup> g/VKT g/VKT g/VKT g/VKT g/VKT g/veh/m g/m/hr g/m/s g/s g/S/M <sup>2</sup>	(Note 3) (Note 4) (Note 3) (Note 4)
EP19_11 Control E Trucks Travelling F Roz Area of paved road RSP Emission Factor (uncontrolled) of pa RSP Emission Factor (uncontrolled) of pa FSP Emission Factor (uncontrolled) of pa FSP Emission Factor (controlled) of pa Control E Trucks Travelling F Roz Area of paved road RSP Emission Factor (uncontrolled) of pa RSP Emission Factor (uncontrolled) of pa RSP Emission Factor (uncontrolled) of pa	fficiency = requency = d Length = travelled = ved road = ved road = ved road = requency = d Length = travelled = ved road = ved road = ved road = ved road = ved road = ved road = i ved road = i ved road = i i i i i i i i i i i i i i i i i i i	98.8 3 14.4 3.5296 3.53E-03 1.06E-02 2.94E-06 4.24E-05 9.804E-07 34.465 0.8539 8.54E-04 2.56E-03 7.12E-07 1.02E-05 2.37E-07 1.02E-05 2.37E-07 1.02E-05 2.37E-07 1.422.4534 3.53E-03 1.06E-02 2.94E-06 4.68E-05 9.804E-07 34.465 0.8539 8.54E-04 2.56E-03 7.12E-07 1.13E-05	% veh/hr m <sup>2</sup> g/VKT g/VKT g/VKT g/veh/m g/m/hr g/s/s g/s/m <sup>2</sup> g/VKT g/VKT g/VKT g/VKT g/VKT g/VKT g/VKT g/VKT g/VKT g/VKT g/VKT g/VKT g/VKT g/VKT g/VKT	(Note 3) (Note 4) (Note 3) (Note 4)

Control Efficiency	_	00.0	0/	(Nete )
Control Enciency	=	98.8	70 	(Note
Trucks Travelling Frequency	=	3	ven/nr	(Note
Road Length	=	24.0	m 2	
Area of paved road travelled	=	72.0	m	
RSP Emission Factor (uncontrolled) of paved road	=	142.4534	g/VKT	
RSP Emission Factor (controlled) of paved road	=	3.5296	g/VKT	
	=	3.53E-03	g/veh/m	
	=	1.06E-02	g/m/hr	
	=	2.94E-06	g/m/s	
	=	7.06E-05	g/s	
	=	<u>9.804E-07</u>	g/s/m <sup>2</sup>	
FCD Environ Factor (upperturbed) of acted acad	_	24.405	- AUKT	
FSP Emission Factor (controlled) of paved road	-	0 8539	g/VKT	
	=	8 54F-04	g/veh/m	
	=	2 56E-03	g/m/hr	
	=	7 12E-07	g/m/s	
	=	1 71E-05	g/m/s	
	=	2 37F-07	$a/s/m^2$	
EP19 14		2.376-07	g/ 3/ 111	
Control Efficiency	=	97.8	%	(Note
Trucks Travelling Frequency	=	0	veh/hr	(Note
Road Length	=	29.0	m	
Area of paved road travelled	=	87.0	m <sup>2</sup>	
RSP Emission Factor (uncontrolled) of paved road	=	142.4534	g/VKT	
KSP Emission Factor (controlled) of paved road	=	3.5296	g/VNI	
	=	5.53E-U3	g/veii/iii	
	=	0.00E+00	g/11/11F	
	=	U.UUE+00	g/m/s	
	=	U.00E+00	g/s	
	=	0.000E+00	g/s/m <sup>4</sup>	
FSP Emission Factor (uncontrolled) of paved road	-	34 465	g/VKT	
FSP Emission Factor (controlled) of paved road	-	0 8230	g/VKT	
Tor Emission ractor (controlled) of paved road	-	8 54F-04	g/veh/m	
	_	0.045-04	g/m/br	
	_	0.002+00	ъ/ 11/ 11 a/m/s	
	_	0.00E+00	g/11/3	
	_	0.000+00	g/3	
	-	0.002+00	g/s/111	
From Aggregate Truck				
Mitigation measures: water spraying and wheel washing				
Average Loaded Weight of Concrete Lorry Mixer	=	30	tons	
Average Weight of Concrete Lorry Mixer	=	15	tons	
Average Unladden Weight of Concrete Lorry Mixer	=	22.5	tons	
k	=	3.23	g/VKT (for TSP)	(Note
	=	0.62	g/VKT (for RSP)	
	=	0.15	g/VKT (for FSP)	
sL = road surface silt loading	=	12	g/m²	(Note
5740.4				
EP19_1		02.4	0/	(1)
Control Efficiency	=	93.1	% 	(Note
Trucks Travelling Frequency	=	40	ven/nr	(Note
Road Length	=	15.7	m m <sup>2</sup>	
Alea of paved foad travelled	=	47.1	m	
RSP Emission Factor (uncontrolled) of paved road	=	142.4534	g/VKT	
RSP Emission Factor (controlled) of paved road	=	9.8829	g/VKT	
	=	9.88E-03	g/veh/m	
	=	3.95E-01	g/m/hr	
	=	1.10E-04	g/m/s	
	=	1.72E-03	g/s	
	=	3.66E-05	g/s/m <sup>2</sup>	
			h	
FSP Emission Factor (uncontrolled) of paved road	=	34.4645	g/VKT	
FSP Emission Factor (controlled) of paved road	=	2.3910	g/VNI g/veh/m	
	-	2.39E-U3	g/ven/m	
	-	3.30E-UZ	g/11/11	
	-	2.00E-U5	g/11/5 g/c	
	-	4.1/E-U4	g/s	
	_	0.00E-UD	g/ s/ 111	
EP19_2				
EP19_2 Control Efficiency	=	95.3	%	(Note
EP19_2 Control Efficiency Trucks Travelling Frequency	= =	95.3 20	% veh/hr	(Note (Note
EP19_2 Control Efficiency Trucks Travelling Frequency Road Length	= = =	95.3 20 20.2	% veh/hr m	(Note ) (Note )
EP19_2 Control Efficiency Trucks Travelling Frequency Road Length Area of paved road travelled	= = =	95.3 20 20.2 60.6	% veh/hr m m <sup>2</sup>	(Note ) (Note )
EP19_2 Control Efficiency Trucks Travelling Frequency Road Length Area of paved road travelled	= = =	95.3 20 20.2 60.6	% veh/hr m m <sup>2</sup>	(Note (Note
EP19_2 Control Efficiency Trucks Travelling Frequency Road Length Area of paved road travelled RSP Emission Factor (uncontrolled) of paved road	= = =	95.3 20 20.2 60.6 142.4534	% veh/hr m m <sup>2</sup> g/VKT	(Note (Note
EP19_2 Control Efficiency Trucks Travelling Frequency Road Length Area of paved road travelled RSP Emission Factor (uncontrolled) of paved road RSP Emission Factor (controlled) of paved road	= = = =	95.3 20 20.2 60.6 142.4534 6.7062	% veh/hr m m <sup>2</sup> g/VKT g/VKT	(Note (Note
EP19_2 Control Efficiency Trucks Travelling Frequency Road Length Area of paved road travelled RSP Emission Factor (uncontrolled) of paved road RSP Emission Factor (controlled) of paved road	= = = =	95.3 20 20.2 60.6 142.4534 6.7062 6.71E-03	% veh/hr m m <sup>2</sup> g/VKT g/VKT g/veh/m	(Note (Note
EP19_2 Control Efficiency Trucks Travelling Frequency Road Length Area of paved road travelled RSP Emission Factor (uncontrolled) of paved road RSP Emission Factor (controlled) of paved road	= = = =	95.3 20 20.2 60.6 142.4534 6.7062 6.71E-03 1.34E-01	% veh/hr m g/VKT g/VKT g/veh/m g/m/hr	(Note (Note
EP19_2 Control Efficiency Trucks Travelling Frequency Road Length Area of paved road travelled RSP Emission Factor (uncontrolled) of paved road RSP Emission Factor (controlled) of paved road	= = = = = =	95.3 20 20.2 60.6 142.4534 6.7062 6.71E-03 1.34E-01 3.73E-05	% veh/hr m g/VKT g/VKT g/VKT g/veh/m g/m/hr g/m/s	(Note (Note
EP19_2 Control Efficiency Trucks Travelling Frequency Road Length Area of paved road travelled RSP Emission Factor (uncontrolled) of paved road RSP Emission Factor (controlled) of paved road	= = = = = = =	95.3 20 20.2 60.6 142.4534 6.7062 6.71E-03 1.34E-01 3.73E-05 7.53E-04	% veh/hr m m <sup>2</sup> g/VKT g/vkh/m g/m/hr g/m/s g/s	(Note (Note
EP19_2 Control Efficiency Trucks Travelling Frequency Road Length Area of paved road travelled RSP Emission Factor (uncontrolled) of paved road RSP Emission Factor (controlled) of paved road		95.3 20 20.2 60.6 142.4534 6.7062 6.71E-03 1.34E-01 3.73E-05 7.53E-04 <u>1.24E-05</u>	% veh/hr m m <sup>2</sup> g/VKT g/veh/m g/w/kr g/m/hr g/m/s g/s g/s/m <sup>2</sup>	(Note (Note
EP19_2 Control Efficiency Trucks Travelling Frequency Road Length Area of paved road travelled RSP Emission Factor (uncontrolled) of paved road RSP Emission Factor (uncontrolled) of paved road		95.3 20 20.2 60.6 142.4534 6.7062 6.71E-03 1.34E-01 3.73E-05 7.53E-04 <u>1.24E-05</u>	% veh/hr m m <sup>2</sup> g/VKT g/veh/m g/m/hr g/m/s g/s g/s/m <sup>2</sup> g/VKT	(Note (Note
EP19_2 Control Efficiency Trucks Travelling Frequency Road Length Area of paved road travelled RSP Emission Factor (uncontrolled) of paved road RSP Emission Factor (controlled) of paved road		95.3 20 20.2 60.6 142.4534 6.7062 6.71E-03 1.34E-01 3.73E-05 7.53E-04 <u>1.24E-05</u> 34.4645 1.6325	% veh/hr m 2 g/VKT g/VKT g/veh/m g/w/hr g/s g/s/m <sup>2</sup> g/s/KT g/VKT	(Note (Note
EP19_2 Control Efficiency Trucks Travelling Frequency Road Length Area of paved road travelled RSP Emission Factor (uncontrolled) of paved road RSP Emission Factor (controlled) of paved road FSP Emission Factor (uncontrolled) of paved road		95.3 20 20.2 60.6 142.4534 6.7062 6.71E-03 1.34E-01 3.73E-05 7.53E-04 <u>1.24E-05</u> 34.4645 1.6225 1.6225	% veh/hr m m <sup>2</sup> g/VKT g/VKT g/veh/m g/m/s g/m/s g/s/m <sup>2</sup> g/VKT g/VKT g/VKT	(Note (Note
EP19_2 Control Efficiency Trucks Travelling Frequency Road Length Area of paved road travelled RSP Emission Factor (uncontrolled) of paved road RSP Emission Factor (controlled) of paved road FSP Emission Factor (uncontrolled) of paved road		95.3 20 20.2 60.6 142.4534 6.7062 6.71E-03 1.34E-01 3.73E-05 7.53E-04 <u>1.24E-05</u> 34.4645 1.6225 1.62E-03 3.24E-02	% veh/hr m m <sup>2</sup> g/VKT g/vk/m g/v/kT g/v/hr g/v/KT g/vkT g/vkT g/vk/m	(Note
EP19_2 Control Efficiency Trucks Travelling Frequency Road Length Area of paved road travelled RSP Emission Factor (uncontrolled) of paved road RSP Emission Factor (controlled) of paved road FSP Emission Factor (uncontrolled) of paved road		95.3 20 20.2 60.6 142.4534 6.7062 6.71E-03 1.34E-01 3.73E-05 7.53E-04 <u>1.24E-05</u> 34.4645 1.6225 1.6225 1.622-03 3.24E-02 9.01E-06	% veh/hr m m <sup>2</sup> g/VKT g/vKT g/veh/m g/m/s g/s g/s/kT g/vKT g/vKT g/veh/m g/m/hr g/m/hr	(Note
EP19_2 Control Efficiency Trucks Travelling Frequency Road Length Area of paved road travelled RSP Emission Factor (uncontrolled) of paved road RSP Emission Factor (controlled) of paved road FSP Emission Factor (uncontrolled) of paved road		95.3 20 20.2 60.6 142.4534 6.7062 6.71E-03 1.34E-01 3.73E-05 7.53E-04 1.24E-05 34.4645 1.6225 1.6225 1.6225-03 3.24E-02 9.01E-06	% veh/hr m m <sup>2</sup> g/VKT g/VkT g/veh/m g/m/hr g/m/s g/VKT g/VKT g/VKT g/vkh/m g/m/hr g/m/hr g/m/s	(Note
EP19_2 Control Efficiency Trucks Travelling Frequency Road Length Area of paved road travelled RSP Emission Factor (uncontrolled) of paved road RSP Emission Factor (controlled) of paved road FSP Emission Factor (uncontrolled) of paved road		95.3 20 20.2 60.6 142.4534 6.7062 6.71E-03 1.34E-01 3.73E-05 7.53E-04 <u>1.24E-05</u> 34.4645 1.6225 1.6225 1.6225 3.24E-02 9.01E-06 1.82E-04	% veh/hr m m <sup>2</sup> g/VKT g/VKT g/veh/m g/m/s g/s g/s/m <sup>2</sup> g/VKT g/VKT g/VKT g/veh/m g/m/s g/s	(Note (Note

Control Efficiency         =         95.3         %           Trucks Travelling Frequency         =         20         veh/hr           Road Length         =         19.9         m           Area of paved road travelled         =         59.7         m²           RSP Emission Factor (uncontrolled) of paved road         =         6.716-03         g/VKT           =         1.34E-01         g/m/hr         =         1.34E-01         g/m/hr           =         7.41E-04         g/s         =         1.24E-05         g/S/m²           FSP Emission Factor (uncontrolled) of paved road         =         3.73E-05         g/m/hr           =         1.24E-05         g/S/m²          5/m²           FSP Emission Factor (uncontrolled) of paved road         =         3.46645         g/VKT           FSP Emission Factor (controlled) of paved road         =         1.62E-03         g/VkT	(Note 3) (Note 4)
Trucks Travelling Frequency Road Length=20veh/hrRoad Length=19.9mArea of paved road travelled=59.7m²RSP Emission Factor (uncontrolled) of paved road=142.4534g/VKTRSP Emission Factor (controlled) of paved road=6.7062g/VKT=6.71E-03g/veh/r=1.34E-01g/m/hr=3.73E-05g/m/s=7.41E-04g/s=1.24E-05g/s/m²=1.24E-05g/s/m²FSP Emission Factor (uncontrolled) of paved road=34.4645g/VKTFSP Emission Factor (controlled) of paved road=1.6225g/VKTFSP Emission Factor (controlled) of paved road=1.6225g/VKTFSP Emission Factor (controlled) of paved road=1.6225g/VKT	(Note 4) 1
Road Length=19.9mArea of paved road travelled=59.7m²RSP Emission Factor (uncontrolled) of paved road=142.4534g/VKTRSP Emission Factor (controlled) of paved road=6.7062g/VKT=6.71E-03g/VKT=6.71E-03g/VkT=1.34E-01g/m/hr=3.73E-05g/m/hr=1.24E-05g/S/m²=1.24E-05g/VKTFSP Emission Factor (uncontrolled) of paved road=34.4645g/VKTFSP Emission Factor (controlled) of paved road=1.6225g/VKT=1.622-03g/VkT=1.622-03g/VkT	n
Area of paved road travelled       =       59.7       m <sup>2</sup> RSP Emission Factor (uncontrolled) of paved road       =       142.4534       g/VKT         RSP Emission Factor (controlled) of paved road       =       6.7062       g/VKT         =       1.34E-01       g/m/hr         =       7.41E-04       g/s         =       1.24E-05       g/s/m <sup>2</sup> FSP Emission Factor (uncontrolled) of paved road       =       34.4645       g/VKT         FSP Emission Factor (controlled) of paved road       =       34.2625       g/VKT         FSP Emission Factor (controlled) of paved road       =       1.62E-03       g/VkT	n
RSP Emission Factor (uncontrolled) of paved road       =       142.4534       g/VKT         RSP Emission Factor (controlled) of paved road       =       6.7062       g/VKT         =       6.71E-03       g/veh/r         =       1.34E-01       g/m/hr         =       3.73E-05       g/m/s         =       7.41E-04       g/s         =       1.24E-05       g/s/m <sup>2</sup> FSP Emission Factor (uncontrolled) of paved road       =       34.4645       g/VKT         FSP Emission Factor (controlled) of paved road       =       1.6225       g/VKT         =       1.62E-03       g/veh/r       =       1.62E-03       g/veh/r	n
RSP Emission Factor (uncontrolled) of paved road         =         142.4534         g/VKT           RSP Emission Factor (controlled) of paved road         =         6.7062         g/VKT           =         6.71E-03         g/weh/           =         1.34E-01         g/m/hr           =         7.41E-04         g/s           =         7.41E-04         g/s           =         1.24E-05         g/s/m <sup>2</sup> FSP Emission Factor (uncontrolled) of paved road         =         34.4645         g/VKT           FSP Emission Factor (controlled) of paved road         =         1.6225         g/VKT           =         1.622-03         g/veh/r         =         1.622-03         g/veh/r	n
KSP Emission Factor (uncontrolled) of paved road       =       142.43.34       g/VK1         RSP Emission Factor (controlled) of paved road       =       6.706.02       g/VK1         =       1.34E-01       g/m/hr         =       3.73E-05       g/m/s         =       7.41E-04       g/s         =       1.24E-05       g/s/m²         FSP Emission Factor (uncontrolled) of paved road       =       3.46455       g/VKT         FSP Emission Factor (controlled) of paved road       =       3.42625       g/VKT         FSP Emission Factor (uncontrolled) of paved road       =       1.6225       g/VKT	n
RSP Emission Factor (controlled) of paved road       =       6.7062       g/VkT         =       6.716-03       g/veh/r         =       1.34E-01       g/m/rs         =       3.73E-05       g/m/s         =       7.41E-04       g/s         =       1.24E-05       g/s/m <sup>2</sup> FSP Emission Factor (uncontrolled) of paved road       =       34.4645       g/VKT         FSP Emission Factor (controlled) of paved road       =       1.6225       g/VKT         =       1.62E-03       g/veh/r	n
=       6.71E-03       g/veh/i         =       1.34E-01       g/m/hr         =       3.73E-05       g/m/s         =       7.41E-04       g/s         =       1.24E-05       g/s/m²         FSP Emission Factor (uncontrolled) of paved road       =       34.4645       g/VKT         FSP Emission Factor (controlled) of paved road       =       1.622.5       g/VKT         =       1.62E-03       g/veh/r       =       1.62E-03       g/veh/r	n
=       1.34E-01       g/m/hr         =       3.73E-05       g/m/s         =       7.41E-04       g/s         =       1.24E-05       g/s/m²         FSP Emission Factor (uncontrolled) of paved road       =       34.4645       g/VKT         FSP Emission Factor (controlled) of paved road       =       1.6225       g/VKT         FSP Emission Factor (controlled) of paved road       =       1.6225       g/VKT	
= 3.73E-05 g/m/s = 7.41E-04 g/s = 1.24E-05 g/S/m <sup>2</sup> FSP Emission Factor (uncontrolled) of paved road = 34.4645 g/VKT FSP Emission Factor (controlled) of paved road = 1.6225 g/VKT = 1.62E-03 g/veh/r	
=       7.41E-04       g/s         =       1.24E-05       g/s/m <sup>2</sup> FSP Emission Factor (uncontrolled) of paved road       =       34.4645       g/VKT         FSP Emission Factor (controlled) of paved road       =       1.6225       g/VKT         =       1.62E-03       g/veh/r	
= <u>1.24E-05</u> g/s/m <sup>2</sup> FSP Emission Factor (uncontrolled) of paved road = 34.4645 g/VKT FSP Emission Factor (controlled) of paved road = 1.6225 g/VKT = 1.62E-03 g/veh/r	
FSP Emission Factor (uncontrolled) of paved road = 34.4645 g/VKT FSP Emission Factor (controlled) of paved road = 1.6225 g/VKT = 1.62E-03 g/veh/r	
FSP Emission Factor (uncontrolled) of paved road       =       34.4645       g/VKT         FSP Emission Factor (controlled) of paved road       =       1.6225       g/VKT         =       1.62E-03       g/veh/r	
FSP Emission Factor (controlled) of paved road = 1.6225 g/VKT = 1.6225 g/VKT	
= 1.62E-03 g/veh/r	
= 1.62E-03 g/ven/1	
	.1
= 3.24E-02 g/m/hr	
= 9.01E-06 g/m/s	
= 1.79E-04 g/s	
= <u>3.00E-06</u> g/s/m <sup>2</sup>	
EP19_4	
Control Efficiency = 95.3 %	(Note 3)
Trucks Travelling Frequency = 20 yeh/hr	(Note 4)
Road length = 97  m	(
Area of payed read tradition $=$ 20.1 m <sup>2</sup>	
Area of paved road travelled = 29.1 m	
KSP Emission Factor (uncontrolled) of paved road = 142.4534 g/VKT	
RSP Emission Factor (controlled) of paved road = 6.7062 g/VKT	
= 6.71E-03 g/veh/r	a
= 1.34E-01 g/m/hr	
= 3.73E-05 g/m/s	
= 3.61E-04 g/s	
= 1.24F-05 g/s/m <sup>2</sup>	
<u> 12.2.05</u> g/3/11	
FSP Emission Factor (uncontrolled) of payed road = $34.4645$ g/VKT	
ESE Emission action (ancontrolled) of payed road = 1.6225 g/Vit	
FSF EITISSION FACTOR (CONTROLLED) OF PAVENTUAL = 1.022.5 g/VKT	
= 1.62E-03 g/Ven/r	1
= 3.24E-02 g/m/hr	
= 9.01E-06 g/m/s	
= 8.74E-05 g/s	
= <u>3.00E-06</u> g/s/m <sup>2</sup>	
EP19_5	
Control Efficiency = 95.3 %	(Note 3)
Trucks Travelling Frequency = 20 veh/hr	(Note 4)
Road Length = 11.3 m	
Area of payed road travelled $-$ 23.9 $m^2$	
Area of paved road travelleu – 55.5 m	
RSP Emission Factor (uncontrolled) of paved road = 142.4534 g/VK1	
RSP Emission Factor (controlled) of paved road = 6.7062 g/VKT	
= 6.71E-03 g/veh/r	n
= 1.34E-01 g/m/hr	
= 3.73E-05 g/m/s	
= 4.21E-04 g/s	
= 1.24E-05 g/s/m <sup>2</sup>	
ECD Emission Eactor (uncontrolled) of payed road $-$ 24 464E $a$ (V/T	
FSP Emission Factor (uncontrolled) of paved road = 34.4645 g/VKT	
FSP Emission Factor (uncontrolled) of paved road = 34.4645 g/VKT FSP Emission Factor (controlled) of paved road = 1.6225 g/VKT	,
FSP Emission Factor (uncontrolled) of paved road = 34.4645 g/VKT FSP Emission Factor (controlled) of paved road = 1.6225 g/VKT = 1.622-03 g/veh/r	ı
FSP Emission Factor (uncontrolled) of paved road       =       34.4645       g/VKT         FSP Emission Factor (controlled) of paved road       =       1.6225       g/VKT         =       1.62E-03       g/veh/r       =       3.24E-02       g/m/hr	n
FSP Emission Factor (uncontrolled) of paved road         =         34.4645         g/VKT           FSP Emission Factor (controlled) of paved road         =         1.6225         g/VKT           =         1.622-03         g/veh/r           =         3.24E-02         g/m/hr           =         9.01E-06         g/m/s	n
FSP Emission Factor (uncontrolled) of paved road         =         34.4645         g/VKT           FSP Emission Factor (controlled) of paved road         =         1.6225         g/VKT           =         1.62E-03         g/veh/t           =         9.01E-06         g/m/s           =         1.02E-04         g/s	n
FSP Emission Factor (uncontrolled) of paved road         =         34.4645         g/VKT           FSP Emission Factor (controlled) of paved road         =         1.6225         g/VKT           =         1.62E-03         g/wh/r           =         3.24E-02         g/m/hr           =         1.02E-04         g/s           =         1.02E-04         g/s           =         3.00E-06         g/s/m <sup>2</sup>	n
FSP Emission Factor (uncontrolled) of paved road         =         34.4645         g/VKT           FSP Emission Factor (controlled) of paved road         =         1.6225         g/VKT           =         1.62E-03         g/veh/r           =         3.24E-02         g/m/hr           =         9.01E-06         g/m/s           =         1.02E-04         g/s           =         3.00E-06         g/sm <sup>2</sup>	n
FSP Emission Factor (uncontrolled) of paved road       =       34.4645       g/VKT         FSP Emission Factor (controlled) of paved road       =       1.6225       g/VKT         =       1.62E-03       g/veh/r         =       3.24E-02       g/m/r         =       1.02E-06       g/m/s         =       3.00E-06       g/s/m <sup>2</sup>	n
FSP Emission Factor (uncontrolled) of paved road       =       34.4645       g/VKT         FSP Emission Factor (controlled) of paved road       =       1.6225       g/VKT         =       1.62E-03       g/weh/         =       3.24E-02       g/m/s         =       1.02E-04       g/s         =       3.00E-06       g/s/m <sup>2</sup>	n (Note 3)
FSP Emission Factor (uncontrolled) of paved road = 34.4645 g/VKT FSP Emission Factor (controlled) of paved road = 1.6225 g/VKT = 1.62E-03 g/veh/t = 3.24E-02 g/m/hr = 9.01E-06 g/m/s = 1.02E-04 g/s = 3.00E-06 g/s/m <sup>2</sup> EP19_6 Control Efficiency = 95.3 % Trucks Travelling Frequency = 20 veh/hr	n (Note 3) (Note 4)
FSP Emission Factor (uncontrolled) of paved road = 34.4645 g/VKT FSP Emission Factor (controlled) of paved road = 1.6225 g/VKT = 1.62E-03 g/veh/u = 3.24E-02 g/m/hr = 9.01E-06 g/m/s = 1.02E-04 g/s = <u>3.00E-06</u> g/s/m <sup>2</sup> EP19_6 Control Efficiency = 95.3 % Trucks Travelling Frequency = 20 veh/hr Road Leneth = 13.1 m	n (Note 3) (Note 4)
FSP Emission Factor (uncontrolled) of paved road       =       34.4645       g/VKT         FSP Emission Factor (controlled) of paved road       =       1.6225       g/VKT         =       1.62E-03       g/veh/i         =       3.24E-02       g/m/r         =       9.01E-06       g/m/s         =       1.02E-04       g/s         =       3.00E-06       g/s/m²	n (Note 3) (Note 4)
FSP Emission Factor (uncontrolled) of paved road       =       34.4645       g/VKT         FSP Emission Factor (controlled) of paved road       =       1.6225       g/VKT         =       1.622-03       g/veh/u         =       3.24E-02       g/m/hr         =       1.022-04       g/s         =       1.02E-04       g/s         =       3.00E-06       g/s/m²         EP19_6       Control Efficiency       =       95.3       %         Trucks Travelling Frequency       =       20       veh/hr         Road Length       =       13.1       m         Area of paved road travelled       =       39.3       m²	n (Note 3) (Note 4)
FSP Emission Factor (uncontrolled) of paved road       =       34.4645       g/VKT         FSP Emission Factor (controlled) of paved road       =       1.6225       g/VKT         =       1.62E-03       g/veh/u         =       3.24E-02       g/m/hr         =       9.01E-06       g/m/s         =       1.02E-04       g/s         =       3.00E-06       g/s/m <sup>2</sup> EP19_6       Control Efficiency       =       95.3       %         Trucks Travelling Frequency       =       2.0       veh/hr         Road Length       =       13.1       m         Area of paved road travelled       =       39.3       m <sup>2</sup>	n (Note 3) (Note 4)
FSP Emission Factor (uncontrolled) of paved road       =       34.4645       g/VKT         FSP Emission Factor (controlled) of paved road       =       1.6225       g/VKT         =       1.62E-03       g/weh/i         =       9.01E-06       g/m/s         =       1.02E-04       g/s         =       3.00E-06       g/s/m²         EP19_6       Control Efficiency       =       95.3       %         Trucks Travelling Frequency       =       20       veh/hr         Road Length       =       13.1       m         Area of paved road travelled       =       39.3       m²         RSP Emission Factor (uncontrolled) of paved road       =       142.4534       g/VKT	n (Note 3) (Note 4)
FSP Emission Factor (uncontrolled) of paved road       =       34.4645       g/VKT         FSP Emission Factor (controlled) of paved road       =       1.622-03       g/VKT         =       3.24E-02       g/m/hr         =       3.02E-04       g/s         =       3.00E-06       g/s/m²         EP19_6       Control Efficiency       =       95.3       %         Trucks Travelling Frequency       =       20       veh/hr         RSP Emission Factor (uncontrolled) of paved road       =       142.4534       g/VKT         RSP Emission Factor (controlled) of paved road       =       142.4534       g/VKT	n (Note 3) (Note 4)
FSP Emission Factor (uncontrolled) of paved road       =       34.4645       g/VKT         FSP Emission Factor (controlled) of paved road       =       1.6225       g/VKT         =       1.622-03       g/veh/u         =       3.24E-02       g/m/hr         =       1.022-04       g/s         =       1.022-04       g/s         =       3.00E-06       g/s/m²         EP19_6       Control Efficiency       =       95.3       %         Trucks Travelling Frequency       =       20       veh/hr         Road Length       =       13.1       m         Area of paved road travelled       =       39.3       m²         RSP Emission Factor (uncontrolled) of paved road       =       142.4534       g/VKT         RSP Emission Factor (controlled) of paved road       =       6.7062       g/VKT	n (Note 3) (Note 4) 1
FSP Emission Factor (uncontrolled) of paved road       =       34.4645       g/VKT         FSP Emission Factor (controlled) of paved road       =       1.62E-03       g/veh/i         =       1.62E-03       g/weh/i       =       3.24E-02       g/weh/i         =       9.01E-06       g/m/sr       =       3.00E-06       g/s/m²         EP19_6       Control Efficiency       =       95.3       %         Trucks Travelling Frequency       =       20       veh/hr         Road Length       =       13.1       m         Area of paved road travelled       =       39.3       m²         RSP Emission Factor (uncontrolled) of paved road       =       142.4534       g/VKT         RSP Emission Factor (uncontrolled) of paved road       =       142.4534       g/VKT         =       0.71E-03       g/weh/i       =       1.34E-01       g/wh/i	n (Note 3) (Note 4)
FSP Emission Factor (uncontrolled) of paved road       =       34.4645       g/VKT         FSP Emission Factor (controlled) of paved road       =       1.6225       g/VKT         =       1.622-03       g/weh/i         =       3.24E-02       g/m/hr         =       9.01E-06       g/m/s         =       1.022-04       g/s         =       3.00E-06       g/s/m²         EP19_6       Control Efficiency       =       95.3       %         Trucks Travelling Frequency       =       20       veh/hr         RSP Emission Factor (uncontrolled) of paved road       =       13.1       m         Area of paved road travelled       =       39.3       m²         RSP Emission Factor (uncontrolled) of paved road       =       6.716-03       g/v/KT         =       6.716-03       g/v/KT       =       6.716-03       g/v/KT         =       1.34E-01       g/m/kr       =       3.73E-05       g/m/s	n (Note 3) (Note 4)
FSP Emission Factor (uncontrolled) of paved road       =       34.4645       g/VKT         FSP Emission Factor (controlled) of paved road       =       1.622-03       g/VKT         =       1.622-03       g/m/hr         =       3.24E-02       g/m/hr         =       1.022-04       g/s         =       1.022-04       g/s         =       3.00E-06       g/s/m²         EP19_6       Control Efficiency       =       95.3       %         Trucks Travelling Frequency       =       20       veh/hr         Road Length       =       13.1       m         Area of paved road travelled       =       39.3       m²         RSP Emission Factor (uncontrolled) of paved road       =       142.4534       g/VKT         =       6.71E-03       g/wh/h       =       1.34E-01       g/m/hr         =       3.73E-05       g/m/s       =       4.88E-04       g/s	n (Note 3) (Note 4)
FSP Emission Factor (uncontrolled) of paved road       =       34.4645       g/VKT         FSP Emission Factor (controlled) of paved road       =       1.6225       g/VKT         =       1.62E-03       g/weh/i         =       9.01E-06       g/m/sr         =       1.02E-04       g/s         =       1.02E-04       g/s         =       3.00E-06       g/s/m²         EP19_6       Control Efficiency       =       95.3       %         Trucks Travelling Frequency       =       20       veh/hr         Road Length       =       13.1       m         Area of paved road travelled       =       39.3       m²         RSP Emission Factor (uncontrolled) of paved road       =       142.4534       g/VKT         =       1.34E-01       g/m/hr       =       3.73E-05       g/m/hr         =       3.73E-05       g/m/hr       =       3.73E-05       g/m/s	n (Note 3) (Note 4)
FSP Emission Factor (uncontrolled) of paved road       =       34.4645       g/VKT         FSP Emission Factor (controlled) of paved road       =       1.62E-03       g/veh/         =       1.62E-03       g/weh/         =       9.01E-06       g/m/s         =       1.02E-04       g/s         =       3.00E-06       g/s/m²         EP19_6       Control Efficiency       =       95.3       %         Trucks Travelling Frequency       =       20       veh/hr         Road Length       =       13.1       m         Area of paved road travelled       =       39.3       m²         RSP Emission Factor (uncontrolled) of paved road       =       142.4534       g/VKT         =       6.716-03       g/weh/m       =       6.7062       g/Wt         =       3.73E-05       g/m/s       =       3.73E-05       g/mk         =       1.424-01       g/mk       =       3.73E-05       g/mk	n (Note 3) (Note 4)
FSP Emission Factor (uncontrolled) of paved road       =       34.4645       g/VKT         FSP Emission Factor (controlled) of paved road       =       1.622-03       g/VKT         =       3.24E-02       g/m/hr         =       9.01E-06       g/s/s         =       1.022-04       g/s         =       3.00E-06       g/s/m²         EP19_6       Control Efficiency       =       95.3       %         Trucks Travelling Frequency       =       20       veh/hr         Road Length       =       13.1       m         Area of paved road travelled       =       39.3       m²         RSP Emission Factor (uncontrolled) of paved road       =       142.4534       g/VKT         =       6.71E-03       g/veh/r       =       3.73E-05       g/m/hr         =       3.73E-05       g/m/hr       = <td>n (Note 3) (Note 4) n</td>	n (Note 3) (Note 4) n
FSP Emission Factor (uncontrolled) of paved road       =       34.4645       g/VKT         FSP Emission Factor (controlled) of paved road       =       1.62E-03       g/veh/i         =       1.62E-03       g/veh/i       =       3.24E-02       g/weh/i         =       9.01E-06       g/m/s       =       1.02E-04       g/s         =       1.02E-04       g/s       =       3.00E-06       g/s/m²         EP19_6       Control Efficiency       =       95.3       %         Trucks Travelling Frequency       =       20       veh/hr         Road Length       =       13.1       m         Area of paved road travelled       =       39.3       m²         RSP Emission Factor (uncontrolled) of paved road       =       142.4534       g/VKT         =       1.34E-01       g/m/hr       =       3.73E-05       g/m/s         =       1.34E-01       g/m/hr       =       3.73E-05       g/ms/m²         ESP Emission Factor (uncontrolled) of paved road       =       1.24E-05       g/S/m²         ESP Emission Factor (uncontrolled) of paved road       =       1.24E-05       g/VKT         =       52       Emission Factor (uncontrolled) of paved road       =	n (Note 3) (Note 4) n
FSP Emission Factor (uncontrolled) of paved road       =       34.4645       g/VKT         FSP Emission Factor (controlled) of paved road       =       1.62E-03       g/veh/i         =       3.24E-02       g/m/s         =       1.02E-04       g/s         =       3.00E-06       g/s/m²         EP19_6       Control Efficiency       =       95.3       %         Trucks Travelling Frequency       =       20       veh/hr         RSP Emission Factor (uncontrolled) of paved road       =       142.4534       g/VKT         RSP Emission Factor (controlled) of paved road       =       142.4534       g/VKT         =       6.716-03       g/weh/m       =       3.73E-05       g/m/s         =       1.34.401       g/m/hr       =       3.73E-05       g/m/s         =       1.42.4534       g/VKT       =       6.716-03       g/weh/m         =       3.73E-05       g/m/s       =       1.424-534       g//KT         =       3.73E-05       g/m/s       =       1.424-534       g//KT         =       3.73E-05       g/s/m       =       1.424-534       g//KT         =       1.52E       g//KT       =       1.52E	n (Note 3) (Note 4) n
FSP Emission Factor (uncontrolled) of paved road       =       34.4645       g/VKT         FSP Emission Factor (controlled) of paved road       =       1.622-03       g/VKT         =       3.24E-02       g/m/hr         =       9.01E-06       g/m/s         =       1.022-04       g/s         =       3.00E-06       g/s/m²         EP19_6       Control Efficiency       =       95.3       %         Trucks Travelling Frequency       =       20       veh/hr         Road Length       =       13.1       m         Area of paved road travelled       =       39.3       m²         RSP Emission Factor (uncontrolled) of paved road       =       142.4534       g/VKT         =       6.71E-03       g/veh/r       =       1.33E-05       g/m/hr         =       3.73E-05       g/m/hr       =       3.73E-05       g/m/hr         =       3.73E-05       g/s       =       1.424534       g/VKT         =       6.71E-03       g/veh/r       =       1.34E-01       g/m/hr         =       3.73E-05       g/m/hr       =       3.73E-05       g/m/hr         =       1.24E-05       g/s/m²       =	n (Note 3) (Note 4) 1
FSP Emission Factor (uncontrolled) of paved road       =       34.4645       g/VKT         FSP Emission Factor (controlled) of paved road       =       1.6225       g/VKT         =       1.622-03       g/veh/i         =       9.01E-06       g/m/sr         =       1.02E-04       g/s         =       1.02E-04       g/s         =       3.00E-06       g/s/m²         EP19_6       Control Efficiency       =       95.3       %         Trucks Travelling Frequency       =       20       veh/hr         Road Length       =       13.1       m         Area of paved road travelled       =       39.3       m²         RSP Emission Factor (uncontrolled) of paved road       =       6.71E-03       g/vek/i         =       1.34E-01       g/m/hr       =       3.73E-05       g/vek/i         =       1.34E-01       g/m/hr       =       3.73E-05       g/vek/i         =       1.62E-03       g/s/s       =       1.62E-05       g/S/s/m²         =       1.62E-05       g/S/s/m²       =       1.62E-05       g/VKT         =       1.62E-05       g/S/m²       =       1.62E-05       g/VKT	n (Note 3) (Note 4) n
FSP Emission Factor (uncontrolled) of paved road       =       34.4645       g/VKT         FSP Emission Factor (controlled) of paved road       =       1.6225       g/VKT         =       1.62E-03       g/veh/i         =       9.01E-06       g/m/s         =       1.02E-04       g/s         =       3.00E-06       g/s/m²         EP19_6       Control Efficiency       =       95.3       %         Trucks Travelling Frequency       =       20       veh/hr         Road Length       =       13.1       m         Area of paved road travelled       =       39.3       m²         RSP Emission Factor (uncontrolled) of paved road       =       142.4534       g/VKT         =       6.71E-03       g/weh/r       =       3.73E-05       g/ms         =       1.42E-01       g/ms       =       1.42E-05       g/KT         =       3.73E-05       g/s/m²       =       1.42E-05       g/VKT         =       1.42E-05       g/s/s       =       1.42E-05       g/VKT         =       1.62E-03       g/wh/r       =       1.42E-05       g/VKT         =       1.62E-03       g/wh/r       = <t< td=""><td>n (Note 3) (Note 4) n</td></t<>	n (Note 3) (Note 4) n
FSP Emission Factor (uncontrolled) of paved road       =       34.4645       g/VKT         FSP Emission Factor (controlled) of paved road       =       1.6225       g/VKT         =       1.622-03       g/weh/i         =       3.24E-02       g/m/hr         =       9.01E-06       g/m/s         =       1.022-04       g/s         =       3.00E-06       g/s/m²         EP19_6       Control Efficiency       =       95.3       %         Trucks Travelling Frequency       =       20       veh/hr         Road Length       =       13.1       m         Area of paved road travelled       =       39.3       m²         RSP Emission Factor (uncontrolled) of paved road       =       142.4534       g/VKT         =       1.34E-01       g/m/hr       =       3.73E-05       g/m/hr         =       3.73E-05       g/m/hr       =       3.44645       g/VKT         =       1.622E-05       g/s/m²       =       1.6225       g/VKT         =       1.34E-01       g/m/hr       =       3.73E-05       g/m/hr         =       1.62E-05       g/s/m²       =       1.62E-05       g/vKT	n (Note 3) (Note 4) 1
FSP Emission Factor (uncontrolled) of paved road       =       34.4645       g/VKT         FSP Emission Factor (controlled) of paved road       =       1.6225       g/VKT         =       1.6225       g/veh/i         =       9.01E-06       g/m/sr         =       1.02E-04       g/s         =       1.02E-04       g/s         =       3.00E-06       g/s/m²         EP19_6       Control Efficiency       =       95.3       %         Trucks Travelling Frequency       =       20       veh/hr         Road Length       =       13.1       m         Area of paved road travelled       =       39.3       m²         RSP Emission Factor (uncontrolled) of paved road       =       6.71E-03       g/VKT         =       1.34E-01       g/m/hr       =       3.73E-05       g/VKT         =       1.42E-05       g/S       =       1.42E-05       g/VKT         =       1.34E-01       g/m/hr       =       3.73E-05       g/VKT         =       1.42E-05       g/S/s/m²       =       1.42E-05       g/VKT         =       1.42E-05       g/S/s/m²       =       1.42E-05       g/VKT	n (Note 3) (Note 4) n
FSP Emission Factor (uncontrolled) of paved road       =       34.4645       g/VKT         FSP Emission Factor (controlled) of paved road       =       1.62E-03       g/veh/i         =       9.01E-06       g/m/s         =       1.02E-04       g/s         =       3.00E-06       g/s/m²         EP19_6       Control Efficiency       =       95.3       %         Trucks Travelling Frequency       =       20       veh/hr         Road Length       =       13.1       m         Area of paved road travelled       =       39.3       m²         RSP Emission Factor (uncontrolled) of paved road       =       142.4534       g/VKT         =       0.71E-03       g/weh/hr       =       3.73E-05       g/wr/hr         =       1.34E-01       g/wh/hr       =       3.73E-05       g/wr/hr         =       1.34E-01       g/s/m²       =       1.22E-05       g/s/m²         FSP Emission Factor (uncontrolled) of paved road       =       3.46455       g/VKT         =       1.62E-03       g/wh/hr       =       3.24E-02       g/m/hr         =       1.62E-03       g/wh/hr       =       3.24E-02       g/m/hr         <	n (Note 3) (Note 4) 1
FSP Emission Factor (uncontrolled) of paved road       =       34.4645       g/VKT         FSP Emission Factor (controlled) of paved road       =       1.6225       g/VKT         =       1.622-03       g/weh/i         =       9.01E-06       g/m/s         =       1.02E-04       g/s         =       1.02E-04       g/s         =       1.02E-04       g/s         =       3.00E-06       g/s/m²         EP19_6       Control Efficiency       =       95.3       %         Trucks Travelling Frequency       =       20       veh/hr         Road Length       =       13.1       m         Area of paved road travelled       =       39.3       m²         RSP Emission Factor (uncontrolled) of paved road       =       142.4534       g/VKT         =       6.716-03       g/weh/i       =       3.73E-05       g/m/s         =       1.424-01       g/m/hr       =       3.73E-05       g/m/s         =       1.622-5       g/VKT       =       1.622-5       g/VKT         =       1.622-5       g/KT       =       1.622-5       g/VKT         =       1.622-5       g/s/m²       = <td>n (Note 3) (Note 4) n</td>	n (Note 3) (Note 4) n
FSP Emission Factor (uncontrolled) of paved road       =       34.4645       g/VKT         FSP Emission Factor (controlled) of paved road       =       1.6225       g/VKT         =       1.622-03       g/veh/it         =       9.01E-06       g/m/hr         =       9.01E-06       g/m/st         =       1.02E-04       g/s         =       1.02E-04       g/s         =       3.00E-05       g/s/m <sup>2</sup> EP19_6       Control Efficiency       =       95.3       %         Trucks Travelling Frequency       =       20       veh/hr         Road Length       =       13.1       m         Area of paved road travelled       =       39.3       m <sup>2</sup> RSP Emission Factor (uncontrolled) of paved road       =       6.71E-03       g/VKT         =       1.42.4534       g/VKT       =       6.71E-03       g/VKT         =       6.71E-03       g/VKT       =       6.71E-03       g/VKT         =       1.42.4534       g/VKT       =       6.71E-03       g/VKT         =       1.42.4534       g/VKT       =       6.71E-03       g/Wch/r         =       1.462-05       g/VKT <td>n (Note 3) (Note 4) 1</td>	n (Note 3) (Note 4) 1
FSP Emission Factor (uncontrolled) of paved road FSP Emission Factor (controlled) of paved road FSP Emission Factor (controlled) of paved road = $1.62E-03$ g/veh/l = $3.24E-02$ g/m/hr = $9.01E-06$ g/m/s = $1.02E-04$ g/s = $3.00E-05$ g/s/m <sup>2</sup> EP19_6 EP19_7 EP1	n (Note 3) (Note 4) n n (Note 3) (Note 4)
FSP Emission Factor (uncontrolled) of paved road       =       34.4645       g/VKT         FSP Emission Factor (controlled) of paved road       =       1.6225       g/VKT         =       1.62E-03       g/weh/i         =       9.01E-06       g/m/s         =       1.02E-04       g/s         =       1.02E-04       g/s         =       3.00E-06       g/s/m²         EP19_6       Control Efficiency       =       95.3       %         Trucks Travelling Frequency       =       20       veh/hr         Road Length       =       13.1       m         Area of paved road travelled       =       39.3       m²         RSP Emission Factor (uncontrolled) of paved road       =       142.4534       g/VKT         =       6.7162       g/WKT       =       6.7162       g/WKT         =       1.34E-01       g/m/s       =       1.34E-01       g/m/s         =       1.225       g/KT       =       1.32E-05       g/KT         =       3.73E-05       g/m/s       =       1.42E-05       g/KT         =       1.32E-05       g/s/m²       =       1.32E-05       g/KT         =	n (Note 3) (Note 4) n n (Note 3) (Note 3)

R	SP Emission Factor (uncontrolled) of payed road	=	142,4534	g/VKT	
	RSP Emission Factor (controlled) of paved road	=	6.7062	g/VKT	
		=	6.71E-03	g/veh/m	
		=	1.34E-01	g/m/hr	
		=	3.73E-05	g/m/s	
		=	1.29E-03	g/s	
		=	<u>1.24E-05</u>	g/s/m <sup>2</sup>	
F	SP Emission Factor (uncontrolled) of payed road	=	34.4645	g/VKT	
	FSP Emission Factor (controlled) of paved road	=	1.6225	g/VKT	
		=	1.62E-03	g/veh/m	
		=	3.24E-02	g/m/hr	
		=	9.01E-06	g/m/s	
		=	3.12E-04	g/s	
		=	3.00E-06	g/s/m <sup>2</sup>	
EP19_14					
	Control Efficiency	=	97.8	%	(Note 3)
	Trucks Travelling Frequency	=	20	veh/hr	(Note 4)
	Road Length	=	29.0	m 2	
	Area of paved road travelled	=	87.0	m	
R	SP Emission Factor (uncontrolled) of paved road	=	142.4534	g/VKT	
	RSP Emission Factor (controlled) of paved road	=	3.1766	g/VKT	
		=	3.18E-03	g/veh/m	
		=	6.35E-02	g/m/hr	
		=	1.76E-05	g/m/s	
		-	5.12E-04	g/s	
			<u>3.00L-00</u>	6/ 3/	
F	SP Emission Factor (uncontrolled) of paved road	=	34.4645	g/VKT	
	FSP Emission Factor (controlled) of paved road	=	0.7685	g/VKT	
		=	7.69E-04	g/veh/m	
		=	1.54E-02	g/m/nr g/m/s	
		=	4.27E-06 1.24F-04	g/III/S g/S	
		=	1.42E-06	g/s/m <sup>2</sup>	
From Cementitious Tanker	wine and wheel weeking				
willigation measures: water spra	Average Loaded Weight of Truck	-	38	tons	
	Average Unladden Weight of Trucks	=	12	tons	
	Average Weight of Trucks	=	25		
	k	=	3.23	g/VKT (for TSP)	(Note 1)
		=	0.62	g/VKT (for RSP)	
		=	0.15	g/VKT (for FSP)	<i>(</i> , , , , , , , , , , , , , , , , , , ,
	sL = road surface silt loading	=	12	g/m2	(Note 2)
EP19_1					
	Control Efficiency	=	93.1	%	(Note 3)
	Trucks Travelling Frequency	=	8	veh/hr	(Note 4)
	Road Length	=	15.7	m 2	
	Area of paved road travelled	=	47.1	m <sup>2</sup>	
R	SP Emission Factor (uncontrolled) of paved road	=	158.6154	g/VKT	
	RSP Emission Factor (controlled) of paved road	=	11.0041	g/VKT	
		=	1.10E-02	g/veh/m	
		=	8.80E-02	g/m/hr	
		=	2.45E-05	g/m/s	
		=	3.84E-04	g/s	
		-	8.15E-06	g/s/m	
F	SP Emission Factor (uncontrolled) of paved road	=	38.3747	g/VKT	
	FSP Emission Factor (controlled) of paved road	=	2.6623	g/VKT	
		=	2.66E-03	g/veh/m	
		=	2.13E-02	g/m/hr	
		=	5.92E-06	g/m/s	
		-	9.29E-03	g/s g/s/m <sup>2</sup>	
			<u></u>	or 91	
EP19_2					
	Control Efficiency	=	95.3	% veh/hr	(Note 3)
	Trucks Travelling Frequency Road Length	=	4 20.2	m	(NOTE 4)
	Area of paved road travelled	=	60.6	m <sup>2</sup>	
R	SP Emission Factor (uncontrolled) of paved road	=	158.6154	g/VKT	
	RSP Emission Factor (controlled) of paved road	=	7.4671	g/VKT	
		=	7.47E-03	g/veh/m	
		-	2.99E-02 8 30E-06	g/11/11 g/m/s	
		=	1.68E-04	g/s	
		=	2.77E-06	g/s/m <sup>2</sup>	
F	SP Emission Factor (uncontrolled) of paved road	=	38.3747	g/VKT	
	FSP Emission Factor (controlled) of paved road	=	1.8066	g/VKT	
		=	1.81E-03	g/veh/m g/m/br	
		-	7.23E-U3	g/11/11 g/m/s	
		=	4.05E-05	g/s	
		=	6.69E-07	g/s/m <sup>2</sup>	
EP19_3					
	Control Efficiency	=	95.3	% veh/hr	(Note 3)
	mucks mavelling rrequency	-	4	vCI/11	(NOLE 4)

	Road Length	=	19.9	m	
	Area of paved road travelled	=	59.7	m <sup>2</sup>	
			55.7		
	RSP Emission Factor (uncontrolled) of payed road	=	158.6154	g/VKT	
	RSP Emission Factor (controlled) of paved road	=	7.4671	g/VKT	
	· · · · · · · · · · · · · · · · · · ·	=	7.47E-03	g/veh/m	
		=	2.99E-02	g/m/hr	
		=	8.30E-06	g/m/s	
		=	1.65E-04	g/s	
		=	2.77E-06	g/s/m <sup>2</sup>	
	FSP Emission Factor (uncontrolled) of paved road	=	38.3747	g/VKT	
	FSP Emission Factor (controlled) of paved road	=	1.8066	g/VKT	
		=	1.81E-03	g/veh/m	
		=	7.23E-03	g/m/hr	
		=	2.01E-06	g/m/s	
		=	3.99E-05	g/s	
		=	<u>6.69E-07</u>	g/s/m²	
CD10 4					
EP19_4	Control Efficiency	_	05.2	0/	(Noto 2)
	Trucks Travelling Frequency	_	4	veh/hr	(Note 4)
	Road Length	=	9.7	m	(11010-1)
	Area of payed road travelled	-	29.1	m <sup>2</sup>	
		-	25.1		
	RSP Emission Factor (uncontrolled) of payed road	=	158.6154	g/VKT	
	RSP Emission Factor (controlled) of paved road	=	7.4671	g/VKT	
		=	7.47E-03	g/veh/m	
		=	2.99E-02	g/m/hr	
		=	8.30E-06	g/m/s	
		=	8.05E-05	g/s	
		=	2.77E-06	g/s/m <sup>2</sup>	
	FSP Emission Factor (uncontrolled) of paved road	=	38.3747	g/VKT	
	FSP Emission Factor (controlled) of paved road	=	1.8066	g/VKT	
		=	1.81E-03	g/veh/m	
		=	7.23E-03	g/m/hr	
		=	2.01E-06	g/m/s	
		=	1.95E-05	g/s	
		=	6.69E-07	g/s/m <sup>2</sup>	
EP19_5					4
	Control Efficiency	=	95.3	%	(Note 3)
	Trucks Travelling Frequency	=	4	ven/hr	(Note 4)
	Road Length	=	11.3	m 2	
	Area of paved road travelled	=	33.9	m	
	PSP Emission Eactor (uncontrolled) of payed road	_	158 6154	αΔ/KT	
	RSP Emission Factor (uncontrolled) of paved road	-	7 /671	g/ V K I g///KT	
	tor Emission ractor (controlled) or paved road	-	7 47F-03	g/veh/m	
		=	2 99E-02	g/m/hr	
		=	8.30E-06	g/m/s	
		=	9.38E-05	g/s	
		=	2.77E-06	$g/s/m^2$	
				8/ •/ ···	
	FSP Emission Factor (uncontrolled) of paved road	=	38.3747	g/VKT	
	FSP Emission Factor (controlled) of paved road	=	1.8066	g/VKT	
		=	1.81E-03	g/veh/m	
		=	7.23E-03	g/m/hr	
		=	2.01E-06	g/m/s	
		=	2.27E-05	g/s	
		=	<u>6.69E-07</u>	g/s/m <sup>2</sup>	
EP19_6					
	Control Efficiency	=	95.3	%	(Note 3)
	Trucks Travelling Frequency	=	4	veh/hr	(Note 4)
	Road Length	=	13.1	m 2	
	Area of paved road travelled	=	39.3	m <sup>-</sup>	
	DCD Emission Easter (unservice lie d) of any 1	_	150 645 -	~ \//T	
	nor emission ractor (uncontrolled) of paved road	-	108.0154	g/VKI	
	nor emission ractor (controlled) of paved road	-	7.46/1	g/VKI	
		-	7.47E-03	g/ven/m	
		-	2.39E-02 8.30F-06	g/m/s	
		=	1.09E-04	g/11/3 g/s	
		=	2.77F-06	$g/s/m^2$	
			<u>/L<sup>-</sup>00</u>	6/ 3/ 111	
	FSP Emission Factor (uncontrolled) of paved road	=	38.3747	g/VKT	
	FSP Emission Factor (controlled) of paved road	=	1.8066	g/VKT	
		=	1.81E-03	g/veh/m	
		=	7.23E-03	g/m/hr	
		=	2.01E-06	g/m/s	
		=	2.63E-05	g/s	
		=	<u>6.69E-07</u>	g/s/m <sup>2</sup>	
EP19_7					
	Control Efficiency	=	95.3	%	(Note 3)
	Trucks Travelling Frequency	=	4	veh/hr	(Note 4)
	Road Length	=	34.6	m	
	Area of paved road travelled	=	103.8	m²	
	RSP Emission Factor (uncontrolled) of paved road	=	158.6154	g/VKT	
	KSP Emission Factor (controlled) of paved road	=	/.4671	g/VKT	
		-	7.4/E-03	g/veh/m	
		-	2.JJC=UZ	g/11/11	

	=	8.30E-06	g/m/s	
	=	2.87E-04	g/s	
	=	2.77E-06	g/s/m <sup>2</sup>	
FSP Emission Factor (uncontrolled) of paved road	=	38.3747	g/VKT	
FSP Emission Factor (controlled) of paved road	=	1.8066	g/VKT	
	=	1.81E-03	g/veh/m	
	=	7.23E-03	g/m/hr	
	=	2.01E-06	g/m/s	
	=	6.95E-05	g/s	
	=	6.69E-07	g/s/m <sup>2</sup>	
EP19_14				
Control Efficiency	=	97.8	%	(Note 3)
Trucks Travelling Frequency	=	4	veh/hr	(Note 4)
Road Length	=	29.0	m	
Area of paved road travelled	=	87.0	m <sup>2</sup>	
RSP Emission Factor (uncontrolled) of paved road	=	158.6154	g/VKT	
RSP Emission Factor (controlled) of paved road	=	3.5370	g/VKT	
	=	3.54E-03	g/veh/m	
	=	1.41E-02	g/m/hr	
	=	3.93E-06	g/m/s	
	=	1.14E-04	g/s	
	=	<u>1.31E-06</u>	g/s/m <sup>2</sup>	
FSP Emission Factor (uncontrolled) of paved road	=	38.3747	g/VKT	
FSP Emission Factor (controlled) of paved road	=	0.8557	g/VKT	
	=	8.56E-04	g/veh/m	
	=	3.42E-03	g/m/hr	
	=	9.51E-07	g/m/s	
	=	2.76E-05	g/s	
	=	<u>3.17E-07</u>	g/s/m <sup>2</sup>	

Note:

Particle size multiplier is referred to Table 13.2-1-1, Section 13.2.1 of USEPA AP42 (ver 1/11) for RSP (PM10) and FSP (PM2.5).
 Road surfacing silt loading is referred to Table 13.2.1-3 of USEPA AP42 (ver 1/11) for concrete batching.
 Percentage of dust mitigation efficiency is calculated with reference to Cowherd et al., "Control of Open Fugitive Dust Sources, EPA-450/3-88-008, U.S. Environmental Protection Agency, Research Triangle
 The vehicle numbers within the site is advised by the applicant.

EP19_1		EP19_2 to 7	7	EP19_8 to 2	13	EP19_14	
End Hr	Factor	End Hr	Factor	End Hr	Factor	End Hr	Factor
1:00	0	1:00	0	1:00	0	1:00	0
2:00	0	2:00	0	2:00	0	2:00	0
3:00	0	3:00	0	3:00	0	3:00	0
4:00	0	4:00	0	4:00	0	4:00	0
5:00	0	5:00	0	5:00	0	5:00	0
6:00	0	6:00	0	6:00	0	6:00	0
7:00	0	7:00	0	7:00	0	7:00	0
8:00	4.22E-01	8:00	5.72E-01	8:00	1.00E+00	8:00	1.08E-01
9:00	4.22E-01	9:00	5.72E-01	9:00	1.00E+00	9:00	1.08E-01
10:00	1.00E+00	10:00	1.00E+00	10:00	1.00E+00	10:00	1.00E+00
11:00	1.00E+00	11:00	1.00E+00	11:00	1.00E+00	11:00	1.00E+00
12:00	1.00E+00	12:00	1.00E+00	12:00	1.00E+00	12:00	1.00E+00
13:00	1.00E+00	13:00	1.00E+00	13:00	1.00E+00	13:00	1.00E+00
14:00	1.00E+00	14:00	1.00E+00	14:00	1.00E+00	14:00	1.00E+00
15:00	1.00E+00	15:00	1.00E+00	15:00	1.00E+00	15:00	1.00E+00
16:00	1.00E+00	16:00	1.00E+00	16:00	1.00E+00	16:00	1.00E+00
17:00	1.00E+00	17:00	1.00E+00	17:00	1.00E+00	17:00	1.00E+00
18:00	1.00E+00	18:00	1.00E+00	18:00	1.00E+00	18:00	1.00E+00
19:00	1.00E+00	19:00	1.00E+00	19:00	1.00E+00	19:00	1.00E+00
20:00	9.65E-01	20:00	7.92E-01	20:00	3.00E-01	20:00	1.33E+00
21:00	9.65E-01	21:00	7.92E-01	21:00	3.00E-01	21:00	1.33E+00
22:00	9.65E-01	22:00	7.92E-01	22:00	3.00E-01	22:00	1.33E+00
23:00	9.65E-01	23:00	7.92E-01	23:00	3.00E-01	23:00	1.33E+00
0:00	0	0:00	0	0:00	0	0:00	0



# **Appendix B**

# Landuse Characteristics Parameters for AERMET

J24.00171.HK.01	
S.16 Planning Application for Proposed Temporary Concrete Batching Plant with Ancillary	
Facilities for a Period of 5 Years at Lots 573 RP and 1710 in D.D. 114, Shek Kong, Yuen Long	
Environmental Assessment	



# For Roughness (33, 46)

Code 🔽	•	Hong Kong Planning Department Classification	Roughness 🔻	Albedo 🔽	Bowen Ratic 🔻
1	-	Private Residential	1	0.18	1.5
2	2	Public Residential	1	0.18	1.5
3	6	Rural Settlement	0.375	0.165	0.9
11	-	Commercial/Business and Office	1	0.18	1.5
21		Industrial Land	0.7	0.18	1.5
22	2	Industrial Estates/Science and Technology Parks	0.7	0.18	1.5
23	5	Warehouse and Open Storage	0.7	0.18	1.5
31		Government, Institutional and Community Facilities	0.7	0.18	1.5
32	2	Open Space and Recreation	0.04	0.15	1
41		Roads and Transport Facilities	0.7	0.18	1.5
42	2	Railways	0.7	0.18	1.5
43	5	Airport	0.07	0.18	1.5
44	ŀ	Port Facilities	0.7	0.18	1.5
51		Cemeteries/Funeral Facilities	0.7	0.18	1.5
52	2	Utilities	0.7	0.18	1.5
53	5	Vacant Land/Construction in Progress	0.2	0.18	1
54	1	Others	0.2	0.18	1
61	-	Agricultural Land	0.1575	0.18	0.55
62	2	Fish Ponds/Gei Wais	0.001	0.1	0.1
71		Woodland	1.05	0.1625	0.75
72	2	Shrubland	0.3	0.18	1.25
73	5	Grassland	0.065	0.185	0.8
74		Mangrove/Swamp	0.065	0.14	0.225
81		Badland	0.15	0.1625	0.75
83	5	Rocky Shore	0.05	0.2	4.75
91	-	Reservoirs	0.001	0.1	0.1
92	2	Streams and Nullahs	0.001	0.1	0.1
99	)	SZ Residential *	1	0.18	1.5
C	)	Open Sea *	0.001	0.1	0.1
	* N	Ion-PlanD Land Utilization categories			

Angle	-	Group	•	Inverse-dista	Roughness 🔽
	0	0 - 30		5.176452751	0.180888
	30	30 - 60		5.31768931	0.243078
	60	60 - 90		5.176452751	0.512926
	90	90 - 120		5.176452751	0.423659
:	120	120 - 150		5.31768931	0.395184
:	150	150 - 180		5.176452751	0.310706
:	180	180 - 210		5.176452751	0.298283
:	210	210 - 240		5.31768931	0.347882
	240	240 - 270		5.176452751	0.374540
	270	270 - 300		5.176452751	0.539075
:	300	300 - 330		5.31768931	0.285071
	330	330 - 360		5.176452751	0.241476







# For albedo and Brown Ratio (33,46)

Code	<b>•</b> •	Hong Kong Planning Department Classification	Roughness 🔽 A	Albedo 🔽	Bowen Ratic 💌	Grid Count 🔽	Percent 🗾 💌	Albedo x Pei 💌	b^n/Sn 🛛 🔽
	1	Private Residential	1	0.18	1.5	4187	0.42%	0.00075366	1.001699124
	2	Public Residential	1	0.18	1.5	154	0.02%	0.00002772	1.000062444
	3	Rural Settlement	0.375	0.165	0.9	66392	6.64%	0.01095468	0.993029313
	11	Commercial/Business and Office	1	0.18	1.5	772	0.08%	0.00013896	1.000313068
	21	Industrial Land	0.7	0.18	1.5	3297	0.33%	0.00059346	1.001337712
	22	Industrial Estates/Science and Technology Parks	0.7	0.18	1.5	0	0.00%	0	1
	23	Warehouse and Open Storage	0.7	0.18	1.5	30393	3.04%	0.00547074	1.012399546
	31	Government, Institutional and Community Facilities	0.7	0.18	1.5	12931	1.29%	0.00232758	1.005256838
	32	Open Space and Recreation	0.04	0.15	1	2944	0.29%	0.0004416	1
	41	Roads and Transport Facilities	0.7	0.18	1.5	15497	1.55%	0.00278946	1.006303275
	42	Railways	0.7	0.18	1.5	6506	0.65%	0.00117108	1.002641438
	43	Airport	0.07	0.18	1.5	0	0.00%	0	1
	44	Port Facilities	0.7	0.18	1.5	0	0.00%	0	1
	51	Cemeteries/Funeral Facilities	0.7	0.18	1.5	24861	2.49%	0.00447498	1.010131245
	52	Utilities	0.7	0.18	1.5	4589	0.46%	0.00082602	1.001862412
	53	Vacant Land/Construction in Progress	0.2	0.18	1	8747	0.87%	0.00157446	1
	54	Others	0.2	0.18	1	18651	1.87%	0.00335718	1
	61	Agricultural Land	0.1575	0.18	0.55	84997	8.50%	0.01529946	0.950455105
	62	Fish Ponds/Gei Wais	0.001	0.1	0.1	2140	0.21%	0.000214	0.995084588
	71	Woodland	1.05	0.1625	0.75	393599	39.36%	0.063959838	0.89294403
	72	Shrubland	0.3	0.18	1.25	152396	15.24%	0.02743128	1.034591005
	73	Grassland	0.065	0.185	0.8	158439	15.84%	0.029311215	0.965263034
	74	Mangrove/Swamp	0.065	0.14	0.225	20	0.00%	0.000028	0.999970167
	81	Badland	0.15	0.1625	0.75	6	0.00%	0.00000975	0.999998274
	83	Rocky Shore	0.05	0.2	4.75	0	0.00%	0	1
	91	Reservoirs	0.001	0.1	0.1	2275	0.23%	0.0002275	0.994775315
	92	Streams and Nullahs	0.001	0.1	0.1	6207	0.62%	0.0006207	0.985809502
	99	SZ Residential *	1	0.18	1.5	0	0.00%	0	1
	0	Open Sea *	0.001	0.1	0.1	0	0.00%	0	1
				0.171969	0.856381	1000000			
	*	Non-PlanD Land Utilization categories							







# Appendix C EMFAC-HK Model Input


# Methodology and Assumptions for Calculation of Vehicular Emission Source

## Model Year

Two years of traffic data were provided by the traffic consultant. The 24-hourly vehicle flows for the modelling are shown in **Appendix G-1**.

- 2026 (the commencing year of the proposed CBP)
- 2031 (5 years valid period of SP licence)

### Model

EMFAC-HK as built in EPD's Smart Air Modelling Platform ("SAMP") was used to estimate the composite emission factors. "Zero Emission Vehicle" ("ZEV") Scenario was selected in SAMP.

### Vehicle Speed

Vehicle speed on each road link at each hour was provided by the traffic consultant as summarized in **Appendix C-1**. All the vehicle classes on the same road link were assumed to have the same travelling speed.

### Trips

Start emissions of vehicles were assumed for all the assessed road links as conservative approach.

## Composite Emission Factors

The composite emission factors for the assessed roads as estimated through the SAMP are shown in **Appendix C-2**. Based on the PATH meteorological data of temperature and relative humidity for the relevant grids of the assessed roads, three types of composite emission factors were estimated, annual hour minimum, monthly hour minimum and monthly hour average.

For long-term air quality impact, monthly hour average composite emission factors were adopted. For short-term air quality, monthly hour minimum composite emission factors were used.

## Jenkin Method

Jenkin method was adopted for the conversion of cumulative NOx to NO<sub>2</sub> by using the functional form of annual mean of NO<sub>2</sub>-to-NOx with reference to the Review of Methods for NO to NO<sub>2</sub> Conversion in plumes at short range for the long-term cumulative NO<sub>2</sub> assessment. The mentioned functional form is presented as equation below.

$$[NO_2] = \frac{\left([NOx] + [OX] + \frac{J}{k}\right) - \sqrt{([NOx] + [OX] + \frac{J}{k})^2 - 4[NOx][OX]}}{2}$$

Where [NO<sub>2</sub>] is the NO<sub>2</sub> concentration

[NOx] is the NOx concentration

[OX] is the sum of NO<sub>2</sub> concentration and O3 concentration (i.e.  $[OX] = [NO_2] + [O_3]$ )

J is the photolysis rate of NO<sub>2</sub>

k is the rate constant for reaction between NO and  $\mathsf{O}_3$ 

The derivation of cumulative annual average NOx to NO<sub>2</sub> conversion equation using Jenkin Method for this assessment by SAMP are shown in **Appendix C-3**.

Appendix C-1 Traffic Flows



# Figure C-1



#### J24.00171.HK.01

S.16 Planning Application for Proposed Temporary Concrete Batching Plant with Ancillary Facilities for a Period of 5 Years at Lots 573 RP and 1710 in D.D. 114, Shek Kong, Yuen Long Environmental Assessment







Road Name Road ID	D Road Type	Speed Limit (km/hr)	Average Speed (km/hr)	Start Emission Estimated by Broad Brush Approach (Y/N)	Hour	Total Vehicles (Veh/hr)	01 - Private Cars	02 - Taxi	03 - Light Goods Vehicles< =2.5t	04 - Light Goods Vehicles 2.5-3.5t	05 - Light Goods Vehicles> 3.5t	06 - Medium Goods Vehicles< =15t	07 - Medium Goods Vehicles1 5-24t	08 - Public Light Buses	09 - Private Light Bus <=3.5t	10 - Private Light Bus >3.5t	11 - Non- franchise d Bus<=6.4 t	12 - Non- franchise d Bus 6.4- 15t	13 - Non- franchise d Bus 15- 24t	14 - Franchise d Bus Single Deck	15 - Franchise d Bus Double Deck	16 - Motorcycl es	17 - Heavy Goods Vehicles> 24t	18 - Non- franchise d Bus >24t	Total
Chun Yiu Road RD_1_2	2 RR	30	26.0	Y	0000-0100	10	100.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	100.00%
Chun Yiu Road RD_1_2	2 RR	30	26.0	Y	0100-0200	5	100.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	100.00%
Chun Yiu Road RD_1_2 Chun Yiu Road RD_1_2	2 RR 2 RR	30 30	26.0 26.0	Y Y	0200-0300	5 5	100.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	100.00%
Chun Yiu Road RD_1_2	2 RR	30	26.0	Y	0400-0500	5	50.00%	0.00%	0.00%	0.00%	50.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	100.00%
Chun Yiu Road RD_1_2 Chun Yiu Road RD 1 2	2 RR 2 RR	30 30	26.0 26.0	Y	0500-0600	10 10	100.00% 73.35%	0.00%	0.00%	0.00%	0.00% 26.65%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	100.00% 100.00%
Chun Yiu Road RD_1_2	2 RR	30	26.0	Y	0700-0800	25	65.86%	0.00%	0.00%	9.90%	15.36%	0.00%	4.44%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	4.44%	0.00%	100.00%
Chun Yiu Road RD_1_2 Chun Yiu Road RD_1 2	2 RR 2 RR	30 30	26.0 26.0	Y	0800-0900	25 25	60.02% 58.16%	0.00%	0.00%	8.98% 9.46%	23.00% 24.38%	0.00%	4.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	4.00% 4.00%	0.00%	100.00%
Chun Yiu Road RD_1_2	2 RR	30	26.0	Ŷ	1000-1100	25	55.96%	0.00%	0.00%	9.90%	25.26%	0.00%	4.44%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	4.44%	0.00%	100.00%
Chun Yiu Road RD_1_2 Chun Yiu Road RD_1_2	2 RR 2 RR	30 30	26.0 26.0	Y	1100-1200	25 55	48.70% 26.04%	0.00%	0.00%	9.46% 20.35%	33.84% 37.62%	0.00%	4.00% 7.59%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	4.00% 8.40%	0.00%	100.00%
Chun Yiu Road RD_1_2	2 RR	30	26.0	Ŷ	1300-1400	60	25.00%	0.00%	0.00%	19.65%	33.95%	0.00%	10.70%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	10.70%	0.00%	100.00%
Chun Yiu Road RD_1_2 Chun Yiu Road RD_1_2	2 RR	30 30	26.0	Y	1400-1500	60 60	27.70%	0.00%	0.00%	18.25%	35.50%	0.00%	9.60%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	8.95% 7.15%	0.00%	100.00%
Chun Yiu Road RD_1_2	2 RR	30	26.0	Ý	1600-1700	55	30.65%	0.00%	0.00%	18.29%	36.83%	0.00%	4.14%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	10.09%	0.00%	100.00%
Chun Yiu Road RD_1_2	2 RR	30	26.0	Y	1700-1800	55	35.99%	0.00%	0.00%	24.05%	27.93%	0.00%	3.95%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	8.07%	0.00%	100.00%
Chun Yiu Road RD_1_2 Chun Yiu Road RD_1_2	2 RR	30	26.0	Y	1900-2000	45	60.55%	0.00%	0.00%	15.55%	23.44 % 18.60%	0.00%	2.22%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	2.50%	0.00%	100.00%
Chun Yiu Road RD_1_2	2 RR	30	26.0	Y	2000-2100	25	59.38%	0.00%	0.00%	25.26%	9.90%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	5.46%	0.00%	100.00%
Chun Yiu Road RD_1_2 Chun Yiu Road RD_1_2	2 RR 2 RR	30	26.0	Ý	2100-2200	20 15	53.55% 53.33%	0.00%	0.00%	22.15% 28.87%	24.30% 17.80%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	100.00%
Chun Yiu Road RD_1_2	2 RR	30	26.0	Y	2300-0000	15	53.33%	0.00%	0.00%	17.80%	28.87%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	100.00%
Chun Yiu Road RD_3_4 Chun Yiu Road RD 3 4	4 RR 4 RR	30 30	26.0 26.0	Y Y	0000-0100	10 10	80.00% 75.00%	10.00% 25.00%	0.00%	0.00%	10.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	100.00%
Chun Yiu Road RD_3_4	4 RR	30	26.0	Y	0200-0300	5	50.00%	0.00%	0.00%	0.00%	50.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	100.00%
Chun Yiu Road RD_3_4 Chun Yiu Road RD_3_4	4 RR 4 RR	30 30	26.0 26.0	Y	0300-0400	5 5	0.00%	50.00% 0.00%	0.00%	0.00%	50.00% 50.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	100.00%
Chun Yiu Road RD_3_4	4 RR	30	26.0	Ŷ	0500-0600	10	83.35%	0.00%	0.00%	0.00%	16.65%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	100.00%
Chun Yiu Road RD_3_4 Chun Yiu Road RD_3_4	4 RR	30 30	26.0	Y	0600-0700	15 30	83.33% 80.00%	8.33%	0.00%	0.00%	8.33%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	100.00%
Chun Yiu Road RD_3_4	4 RR	30	26.0	Ý	0800-0900	35	72.79%	3.43%	0.00%	6.79%	17.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	100.00%
Chun Yiu Road RD_3_4	4 RR	30	26.0	Y	0900-1000	30	73.33%	3.33%	0.00%	6.67%	16.67%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	100.00%
Chun Yiu Road RD_3_4 RD_3_4	4 RR	30	26.0	Ý	1100-1200	25	60.00%	4.00%	0.00%	8.00%	28.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	100.00%
Chun Yiu Road RD_3_4	4 RR	30	26.0	Y	1200-1300	55	28.92%	0.00%	0.00%	16.67%	38.50%	0.00%	7.95%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	7.95%	0.00%	100.00%
Chun Yiu Road RD_3_4 Chun Yiu Road RD_3_4	4 RR 4 RR	30	26.0	Ý	1400-1400	55	25.57% 31.10%	0.00%	0.00%	16.47%	36.37% 34.48%	0.00%	9.35%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	9.35%	0.00%	100.00%
Chun Yiu Road RD_3_4	4 RR	30	26.0	Y	1500-1600	55	30.26%	0.00%	0.00%	19.00%	35.72%	0.00%	7.51%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	7.51%	0.00%	100.00%
Chun Yiu Road RD_3_4 Chun Yiu Road RD 3 4	4 RR 4 RR	30 30	26.0	Y Y	1600-1700 1700-1800	55 50	32.56% 37.52%	0.00%	0.00%	16.28% 20.92%	36.77% 29.14%	0.00%	4.14% 4.14%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	10.25% 8.28%	0.00%	100.00%
Chun Yiu Road RD_3_4	4 RR	30	26.0	Y	1800-1900	45	52.60%	0.00%	0.00%	16.81%	23.64%	0.00%	2.33%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	4.61%	0.00%	100.00%
Chun Yiu Road RD_3_4 Chun Yiu Road RD 3 4	4 RR 4 RR	30 30	26.0 26.0	Y	1900-2000 2000-2100	45 20	63.10% 63.35%	0.00%	0.00%	13.10% 21.10%	18.47% 10.55%	0.00%	2.67% 0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	2.67% 5.00%	0.00%	100.00% 100.00%
Chun Yiu Road RD_3_4	4 RR	30	26.0	Y	2100-2200	20	59.05%	0.00%	0.00%	18.05%	22.90%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	100.00%
Chun Yiu Road RD_3_4 Chun Yiu Road RD_3_4	4 RR 4 RR	30 30	26.0 26.0	Y	2200-2300	15 15	53.33% 53.33%	0.00%	0.00%	17.80% 17.80%	28.87% 28.87%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	100.00%
Chun Yiu Road RD_5_6	6 RR	30	26.0	Ŷ	0000-0100	10	80.00%	10.00%	0.00%	0.00%	10.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	100.00%
Chun Yiu Road RD_5_6	6 RR	30	26.0	Y	0100-0200	10	75.00%	25.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	100.00%
Chun Yiu Road RD_5_6	6 RR	30	26.0	Ý	0300-0400	5	0.00%	50.00%	0.00%	0.00%	50.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	100.00%
Chun Yiu Road RD_5_6	6 RR	30	26.0	Y	0400-0500	5	50.00%	0.00%	0.00%	0.00%	50.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	100.00%
Chun Yiu Road RD_5_6	6 RR	30	26.0	Y	0600-0700	15	83.33%	8.33%	0.00%	0.00%	8.33%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	100.00%
Chun Yiu Road RD_5_6	6 RR	30	26.0	Y	0700-0800	75	28.54%	1.44%	0.00%	2.88%	4.26%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	62.88%	0.00%	100.00%
Chun Yiu Road RD_5_6 Chun Yiu Road RD_5_6	6 RR	30	26.0	Ý	0900-1000	105	26.68% 17.66%	0.97%	0.00%	2.82% 1.94%	6.96% 4.91%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	62.16% 74.51%	0.00%	100.00%
Chun Yiu Road RD_5_6	6 RR	30	26.0	Y	1000-1100	105	16.00%	1.03%	0.00%	2.00%	5.03%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	75.94%	0.00%	100.00%
Chun Yiu Road RD_5_6 Chun Yiu Road RD 5 6	6 RR 6 RR	30 30	26.0 26.0	Y	1100-1200 1200-1300	105 125	14.80% 11.19%	0.97%	0.00%	1.94% 6.41%	6.91% 15.20%	0.00%	0.00% 3.19%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	75.37% 64.00%	0.00%	100.00% 100.00%
Chun Yiu Road RD_5_6	6 RR	30	26.0	Y	1300-1400	135	10.72%	0.00%	0.00%	6.90%	15.14%	0.00%	4.44%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	62.79%	0.00%	100.00%
Chun Yiu Road RD_5_6 Chun Yiu Road RD 5_6	6 RR 6 RR	30 30	26.0 26.0	Y	1400-1500 1500-1600	135 135	12.52% 12.39%	0.00%	0.00%	6.29% 7.78%	14.01% 14.69%	0.00%	3.83% 3.11%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	63.34% 62.03%	0.00%	100.00%
Chun Yiu Road RD_5_6	6 RR	30	26.0	Ŷ	1600-1700	130	12.85%	0.00%	0.00%	6.42%	14.23%	0.00%	1.56%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	64.94%	0.00%	100.00%
Chun Yiu Road RD_5_6 Chun Yiu Road RD_5_6	6 RR 6 RR	30 30	26.0 26.0	Y Y	1700-1800 1800-1900	130 125	14.51% 18.61%	0.00%	0.00%	8.08% 5.90%	11.25% 8.47%	0.00%	1.62% 0.83%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	64.55% 66.18%	0.00%	100.00% 100.00%
Chun Yiu Road RD_5_6	6 RR	30	26.0	Ŷ	1900-2000	105	24.45%	0.00%	0.00%	5.09%	7.18%	0.00%	1.05%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	62.23%	0.00%	100.00%
Chun Yiu Road RD_5_6	6 RR	30	26.0	Y	2000-2100	80	15.20%	0.00%	0.00%	5.05%	2.55%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	77.20%	0.00%	100.00%
Chun Yiu Road RD_5_6	6 RR	30 30	26.0 26.0	ř Y	2100-2200	75	8.44%	0.00%	0.00%	3.95% 2.85%	5.15% 4.34%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	84.37%	0.00%	100.00%
Chun Yiu Road RD_5_6	6 RR	30	26.0	Y	2300-0000	15	53.33%	0.00%	0.00%	17.80%	28.87%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	100.00%
Chun Yiu Road RD_7_8 Chun Yiu Road RD_7_8	o RR 8 RR	30	26.0 26.0	ř Y	0100-0100	10	80.93% 75.00%	9.53% 12.50%	0.00%	0.00%	9.53% 12.50%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	100.00%

Road Name	Road ID	Road Type	Speed Limit (km/hr)	Average Speed (km/hr)	Start Emission Estimated by Broad Brush Approach (Y/N)	Hour	Total Vehicles (Veh/hr)	01 - Private Cars	02 - Taxi	03 - Light Goods Vehicles< =2.5t	04 - Light Goods Vehicles 2.5-3.5t	05 - Light Goods Vehicles> 3.5t	06 - Medium Goods Vehicles< =15t	07 - Medium Goods Vehicles1 5-24t	08 - Public Light Buses	09 - Private Light Bus <=3.5t	10 - Private Light Bus >3.5t	11 - Non- franchise d Bus<=6.4 t	12 - Non- franchise d Bus 6.4 15t	13 - Non- franchise d Bus 15- 24t	14 - Franchise d Bus Single Deck	15 - Franchise d Bus Double Deck	16 - Motorcycl es	17 - Heavy Goods Vehicles> 24t	18 - Non- franchise d Bus >24t	Total
Chun Yiu Road	RD 7 8	RR	30	26.0	Y	0200-0300	10	83.35%	0.00%	0.00%	0.00%	16.65%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	100.00%
Chun Yiu Road	RD_7_8	RR	30	26.0	Y	0300-0400	5	25.00%	25.00%	0.00%	0.00%	50.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	100.00%
Chun Yiu Road	RD_7_8	RR	30	26.0	Y	0400-0500	10	83.35%	0.00%	0.00%	0.00%	16.65%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	100.00%
Chun Yiu Road	RD 7 8	RR	30	26.0	Ý	0600-0700	25	76.22%	4.26%	0.00%	4.26%	15.26%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	100.00%
Chun Yiu Road	RD_7_8	RR	30	26.0	Y	0700-0800	95	38.43%	1.10%	0.00%	3.29%	6.54%	0.00%	1.09%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	49.54%	0.00%	100.00%
Chun Yiu Road	RD_7_8	RR	30	26.0	Y	0800-0900	100	35.08%	1.08%	0.00%	4.28%	10.70% 9.12%	0.00%	1.04%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	47.82%	0.00%	100.00%
Chun Yiu Road	RD_7_8	RR	30	26.0	Ý	1000-1100	125	23.72%	0.78%	0.00%	3.24%	8.17%	0.00%	0.84%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	63.25%	0.00%	100.00%
Chun Yiu Road	RD_7_8	RR	30	26.0	Y	1100-1200	125	21.87%	0.78%	0.00%	3.24%	10.52%	0.00%	0.84%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	62.75%	0.00%	100.00%
Chun Yiu Road Chun Yiu Road	RD_7_8 RD_7_8	RR	30 30	26.0 26.0	Y	1200-1300	180 190	15.36% 15.83%	0.00%	0.00%	10.80%	21.61%	0.00%	4.56% 6.38%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	47.68% 46.52%	0.00%	100.00%
Chun Yiu Road	RD_7_8	RR	30	26.0	Ŷ	1400-1500	190	16.83%	0.00%	0.00%	10.30%	20.13%	0.00%	5.45%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	47.29%	0.00%	100.00%
Chun Yiu Road	RD_7_8	RR	30	26.0	Y	1500-1600	185	17.06%	0.00%	0.00%	11.00%	20.85%	0.00%	4.41%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	46.68%	0.00%	100.00%
Chun Yiu Road	RD 7 8	RR	30 30	26.0	ř Y	1700-1800	180	21.65%	0.00%	0.00%	9.71%	20.00%	0.00%	2.28%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	49.16%	0.00%	100.00%
Chun Yiu Road	RD_7_8	RR	30	26.0	Ŷ	1800-1900	170	27.59%	0.00%	0.00%	9.21%	12.92%	0.00%	1.25%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	49.04%	0.00%	100.00%
Chun Yiu Road	RD_7_8	RR	30	26.0	Y	1900-2000	145	35.52%	0.00%	0.00%	7.96%	10.13%	0.00%	1.45%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	44.93%	0.00%	100.00%
Chun Yiu Road	RD 7 8	RR	30	26.0	ř Y	2100-2100	105	25.71%	0.00%	0.00%	8.93% 7.35%	3.94%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	64.25%	0.00%	100.00%
Chun Yiu Road	RD_7_8	RR	30	26.0	Ŷ	2200-2300	85	14.47%	0.00%	0.00%	6.06%	6.06%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	73.41%	0.00%	100.00%
Chun Yiu Road	RD_7_8	RR	30	26.0	Y	2300-0000	25	56.70%	0.00%	0.00%	14.92%	28.38%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	100.00%
Kam Tin Road	RD_09_10 RD_09_10	RR	50	40.0	Y	0100-0100	110	64.40%	14.95%	0.00%	4.90%	5.45% 4.70%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	2.20%	2.75%	0.00%	0.00%	100.00%
Kam Tin Road	RD_09_10	RR	50	40.0	Y	0200-0300	95	68.49%	14.56%	0.00%	6.74%	9.05%	1.16%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	100.00%
Kam Tin Road	RD_09_10	RR	50	40.0	Y	0300-0400	75	53.54%	19.61%	0.00%	9.83%	15.59%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	1.44%	0.00%	0.00%	100.00%
Kam Tin Road	RD_09_10 RD_09_10	RR	50 50	40.0	ř Y	0400-0500	95 165	59.64% 59.40%	6.13%	0.00%	7.96%	4.91%	3.71%	1.25%	0.00% 3.65%	0.00%	0.00%	0.63%	0.00% 6.13%	1.25%	0.00% 1.87%	1.25%	1.05%	0.00%	0.00%	100.00%
Kam Tin Road	RD_09_10	RR	50	37.7	Y	0600-0700	410	68.94%	5.71%	0.00%	6.71%	3.73%	1.49%	0.50%	2.69%	0.00%	0.00%	0.24%	0.00%	0.50%	0.76%	1.50%	5.72%	1.50%	0.00%	100.00%
Kam Tin Road	RD_09_10	RR	50	32.1	Y	0700-0800	940	62.93%	3.16%	0.00%	10.17%	3.73%	0.84%	0.75%	1.40%	0.00%	0.00%	0.00%	3.21%	0.40%	0.75%	0.85%	5.37%	6.45%	0.00%	100.00%
Kam Tin Road	RD 09 10	RR	50	32.3	Ý	0900-1000	930	56.62%	3.96%	0.00%	14.22%	6.41%	1.54%	1.50%	1.40%	0.00%	0.00%	0.20%	1.10%	0.20%	0.20%	0.85%	4.03 %	10.14%	0.00%	100.00%
Kam Tin Road	RD_09_10	RR	50	32.8	Y	1000-1100	875	56.97%	4.11%	0.00%	13.17%	7.22%	1.69%	1.26%	1.50%	0.00%	0.00%	0.10%	0.00%	0.20%	0.60%	0.70%	1.49%	11.00%	0.00%	100.00%
Kam Tin Road	RD_09_10	RR	50 50	33.1	Y	1100-1200	850	53.77% 22.51%	3.20%	0.00%	14.46%	9.22%	1.39%	1.20%	1.05%	0.00%	0.00%	0.35%	0.00%	0.00%	0.50%	0.70%	2.09%	12.05%	0.00%	100.00%
Kam Tin Road	RD 09 10	RR	50	30.5	Ý	1300-1400	1095	31.26%	5.26%	0.00%	16.37%	12.75%	8.28%	2.20%	0.80%	0.00%	0.00%	0.00%	0.00%	0.40%	0.45%	0.45%	2.41%	19.56%	0.00%	100.00%
Kam Tin Road	RD_09_10	RR	50	30.8	Y	1400-1500	1060	33.01%	4.71%	0.00%	15.88%	12.60%	8.67%	1.70%	1.05%	0.00%	0.00%	0.00%	0.20%	0.40%	0.30%	0.50%	2.46%	18.50%	0.00%	100.00%
Kam Tin Road Kam Tin Road	RD_09_10	RR	50 50	31.2	Y	1500-1600	1030	33.71% 36.37%	4.21%	0.00%	17.95%	13.35% 12.15%	7.71% 6.31%	1.55%	0.90%	0.00%	0.00%	0.00%	0.10%	0.40%	0.30%	0.50%	2.37%	16.94% 18.62%	0.00%	100.00%
Kam Tin Road	RD_09_10	RR	50	30.3	Ý	1700-1800	1115	37.39%	4.06%	0.00%	17.21%	8.74%	7.87%	0.55%	1.26%	0.00%	0.00%	0.00%	0.00%	0.35%	0.25%	0.45%	6.92%	14.94%	0.00%	100.00%
Kam Tin Road	RD_09_10	RR	50	31.8	Y	1800-1900	970	50.21%	4.67%	0.00%	13.45%	7.63%	3.40%	0.20%	1.57%	0.00%	0.00%	0.00%	0.11%	0.20%	0.31%	0.51%	5.16%	12.60%	0.00%	100.00%
Kam Tin Road Kam Tin Road	RD_09_10 RD_09_10	RR	50 50	32.7 36.0	Y	1900-2000	885 570	60.54% 51.61%	3.61%	0.00%	11.14%	5.69% 2.49%	2.16%	0.45%	1.36%	0.00%	0.00%	0.00%	0.11%	0.20%	0.31%	0.55%	4.56% 5.16%	9.32% 15.37%	0.00%	100.00%
Kam Tin Road	RD_09_10	RR	50	37.1	Ý	2100-2200	470	48.05%	6.48%	0.00%	13.59%	5.39%	1.71%	0.00%	1.51%	0.00%	0.00%	0.00%	0.00%	0.45%	1.27%	0.85%	5.61%	15.10%	0.00%	100.00%
Kam Tin Road	RD_09_10	RR	50	38.3	Y	2200-2300	350	41.43%	10.43%	0.00%	12.70%	4.58%	5.48%	0.00%	0.55%	0.00%	0.00%	0.00%	0.00%	0.00%	0.86%	1.41%	3.19%	19.36%	0.00%	100.00%
Kam Tin Road Kam Tin Road	RD_09_10 RD_11_12	RR	50 50	39.1 40.0	Ý	2300-0000	280	52.74% 73.25%	15.48%	0.00%	10.10%	7.56%	3.97%	0.00%	0.00%	0.00%	0.00%	0.00%	0.39%	0.74%	2.15%	1.08%	5.05% 2.73%	0.74%	0.00%	100.00%
Kam Tin Road	RD_11_12	RR	50	40.0	Ŷ	0100-0200	110	64.10%	15.10%	0.00%	12.25%	4.75%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	1.90%	1.90%	0.00%	0.00%	100.00%
Kam Tin Road	RD_11_12	RR	50	40.0	Y	0200-0300	90	68.15%	13.65%	0.00%	6.80%	10.25%	1.15%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	100.00%
Kam Tin Road	RD 11 12	RR	50	40.0	Ý	0400-0500	95	59.74%	12.97%	0.00%	14.13%	10.95%	1.11%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	1.11%	0.00%	0.00%	100.00%
Kam Tin Road	RD_11_12	RR	50	40.0	Y	0500-0600	170	59.81%	6.20%	0.00%	8.01%	4.35%	3.71%	1.25%	3.69%	0.00%	0.00%	0.61%	6.20%	1.25%	1.83%	1.25%	1.86%	0.00%	0.00%	100.00%
Kam Tin Road Kam Tin Road	RD_11_12 RD_11_12	RR	50 50	37.7	Y	0600-0700	405	68.48% 62.72%	5.53%	0.00%	6.72%	4.03%	1.53%	0.50%	2.74%	0.00%	0.00%	0.24%	0.00%	0.50%	0.76%	1.50%	5.74% 5.38%	1.74%	0.00%	100.00%
Kam Tin Road	RD_11_12 RD_11_12	RR	50	32.2	Ý	0800-0900	920	60.56%	2.86%	0.00%	12.98%	6.56%	0.99%	1.20%	1.65%	0.00%	0.00%	0.35%	0.55%	0.43%	0.55%	0.65%	4.06%	6.84%	0.00%	100.00%
Kam Tin Road	RD_11_12	RR	50	32.7	Y	0900-1000	885	57.04%	2.71%	0.00%	12.63%	6.47%	1.04%	1.25%	1.69%	0.00%	0.00%	0.35%	0.60%	0.20%	0.60%	0.70%	4.02%	10.69%	0.00%	100.00%
Kam Tin Road	RD_11_12 RD_11_12	RR	50 50	32.8	Y	1000-1100	870 845	56.55% 53.35%	3.92%	0.00%	13.18%	7.41% 0.35%	1.73%	1.41%	1.49%	0.00%	0.00%	0.10%	0.00%	0.20%	0.60%	0.70%	1.49%	11.23%	0.00%	100.00%
Kam Tin Road	RD_11_12	RR	50	31.9	Ý	1200-1200	955	32.49%	5.47%	0.00%	17.92%	14.47%	5.70%	1.90%	0.95%	0.00%	0.00%	0.00%	0.10%	0.40%	0.30%	0.50%	2.72%	17.07%	0.00%	100.00%
Kam Tin Road	RD_11_12	RR	50	30.5	Y	1300-1400	1095	31.26%	5.26%	0.00%	16.40%	12.77%	8.25%	2.20%	0.85%	0.00%	0.00%	0.00%	0.00%	0.20%	0.45%	0.45%	2.36%	19.54%	0.00%	100.00%
Kam Tin Road	RD_11_12 RD_11_12	RR	50 50	30.8	Y	1400-1500	1060	32.96%	4.72%	0.00%	16.06%	12.57%	8.69%	1.70%	1.05%	0.00%	0.00%	0.00%	0.21%	0.40%	0.30%	0.45%	2.47%	18.44%	0.00%	100.00%
Kam Tin Road	RD_11_12	RR	50	31.2	Ŷ	1600-1700	1030	36.15%	4.27%	0.00%	15.21%	12.22%	6.33%	0.60%	0.96%	0.00%	0.00%	0.00%	0.00%	0.20%	0.50%	0.50%	4.50%	18.56%	0.00%	100.00%
Kam Tin Road	RD_11_12	RR	50	32.6	Y	1700-1800	895	58.06%	3.61%	0.00%	11.20%	5.65%	2.16%	0.45%	1.36%	0.00%	0.52%	0.23%	0.11%	0.20%	0.31%	0.55%	4.51%	11.07%	0.00%	100.00%
Kam Tin Road	RD 11 12	RR	50 50	31.8 32.7	Y Y	1800-1900	970 880	50.08% 58.37%	4.66% 3.77%	0.00%	13.56%	7.65% 5.98%	3.40% 2.13%	0.20%	1.51% 1.36%	0.00%	0.00%	0.00%	0.11%	0.20%	0.31%	0.51%	5.16% 4.77%	12.65% 9.35%	0.00%	100.00%
Kam Tin Road	RD_11_12	RR	50	36.0	Ý	2000-2100	570	51.53%	7.27%	0.00%	13.40%	2.46%	1.42%	0.35%	1.77%	0.00%	0.00%	0.00%	0.00%	0.00%	0.56%	0.75%	5.11%	15.38%	0.00%	100.00%
Kam Tin Road	RD_11_12	RR	50	37.1	Y	2100-2200	465	47.83%	6.48%	0.00%	13.80%	5.60%	1.71%	0.00%	1.51%	0.00%	0.00%	0.00%	0.00%	0.45%	1.28%	0.85%	5.61%	14.89%	0.00%	100.00%
Kam Tin Road	RD 11 12	RR	50 50	38.3 39.1	r Y	2300-0000	350 280	41.40% 52.74%	10.38%	0.00%	12.70%	4.62% 7.56%	5.48% 3.97%	0.00%	0.55%	0.00%	0.00%	0.00%	0.00%	0.00%	0.86%	1.41%	3.19% 5.05%	0.74%	0.00%	100.00%
Kam Tin Road	RD_13_14	RR	50	40.0	Ý	0000-0100	190	73.30%	11.40%	0.00%	4.90%	5.45%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	2.20%	2.75%	0.00%	0.00%	100.00%
Kam Tin Road	RD_13_14	RR	50	40.0	Y	0100-0200	110	64.40%	14.95%	0.00%	12.15%	4.70%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	1.90%	1.90%	0.00%	0.00%	100.00%
Kam Tin Road	RD_13_14 RD_13_14	RR	50	40.0	Y	0300-0400	75	53.54%	19.61%	0.00%	9.83%	15.59%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	1.44%	0.00%	0.00%	100.00%

Road Name	Road ID	Road Type	Speed Limit (km/hr)	Average Speed (km/hr)	Start Emission Estimated by Broad Brush Approach (Y/N)	Hour	Total Vehicles (Veh/hr)	01 - Private Cars	02 - Taxi	03 - Light Goods Vehicles< =2.5t	04 - Light Goods Vehicles 2.5-3.5t	05 - Light Goods Vehicles> 3.5t	06 - Medium Goods Vehicles< =15t	07 - Medium Goods Vehicles1 5-24t	08 - Public Light Buses	09 - Private Light Bus <=3.5t	10 - Private Light Bus >3.5t	11 - Non- franchise d Bus<=6.4 t	12 - Non- franchise d Bus 6.4 15t	13 - Non- franchise d Bus 15- 24t	14 - Franchise d Bus Single Deck	15 - Franchise d Bus Double Deck	16 - Motorcycl es	17 - Heavy Goods Vehicles> 24t	18 - Non- franchise d Bus >24t	Total
Kam Tin Road	RD 13 14	RR	50	40.0	Y	0400-0500	95	59.64%	13.81%	0.00%	13.81%	10.64%	1.05%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	1.05%	0.00%	0.00%	100.00%
Kam Tin Road	RD_13_14	RR	50	40.0	Y	0500-0600	165	59.40%	6.13%	0.00%	7.96%	4.91%	3.71%	1.25%	3.65%	0.00%	0.00%	0.63%	6.13%	1.25%	1.87%	1.25%	1.88%	0.00%	0.00%	100.00%
Kam Tin Road	RD_13_14 RD_13_14	RR	50 50	37.7	Y	0600-0700	410 940	68.94% 62.93%	5.71%	0.00%	6.71% 10.17%	3.73%	1.49%	0.50%	2.69%	0.00%	0.00%	0.24%	0.00%	0.50%	0.76%	1.50%	5.72%	1.50% 6.45%	0.00%	100.00%
Kam Tin Road	RD_13_14	RR	50	32.3	Ŷ	0800-0900	925	61.13%	2.90%	0.00%	12.92%	6.47%	0.94%	1.10%	1.65%	0.00%	0.00%	0.30%	0.50%	0.20%	0.50%	0.65%	4.03%	6.70%	0.00%	100.00%
Kam Tin Road	RD_13_14	RR	50	32.2	Y	0900-1000	930	56.62%	3.96%	0.00%	14.22%	6.41%	1.54%	1.50%	1.40%	0.00%	0.00%	0.20%	1.10%	0.20%	0.20%	0.85%	1.64%	10.14%	0.00%	100.00%
Kam Tin Road Kam Tin Road	RD_13_14 RD_13_14	RR	50 50	32.8	Y Y	1000-1100	875 850	56.97% 53.77%	4.11%	0.00%	13.17% 14.46%	9.22%	1.69%	1.26%	1.50%	0.00%	0.00%	0.10%	0.00%	0.20%	0.60%	0.70%	1.49%	11.00%	0.00%	100.00%
Kam Tin Road	RD_13_14	RR	50	32.0	Ý	1200-1300	950	32.51%	5.46%	0.00%	17.85%	14.45%	5.72%	1.90%	0.95%	0.00%	0.00%	0.00%	0.10%	0.40%	0.30%	0.50%	2.77%	17.08%	0.00%	100.00%
Kam Tin Road	RD_13_14	RR	50	30.5	Y	1300-1400	1095	31.26%	5.26%	0.00%	16.37%	12.75%	8.28%	2.20%	0.80%	0.00%	0.00%	0.00%	0.00%	0.20%	0.45%	0.45%	2.41%	19.56%	0.00%	100.00%
Kam Tin Road	RD_13_14 RD_13_14	RR	50 50	30.8	Ý	1500-1600	1060	33.01% 33.71%	4.71%	0.00%	15.88%	12.60%	8.67% 7.71%	1.70%	0.90%	0.00%	0.00%	0.00%	0.20%	0.40%	0.30%	0.50%	2.46%	16.94%	0.00%	100.00%
Kam Tin Road	RD_13_14	RR	50	31.2	Y	1600-1700	1030	36.37%	4.26%	0.00%	15.04%	12.15%	6.31%	0.60%	0.95%	0.00%	0.00%	0.00%	0.00%	0.20%	0.50%	0.50%	4.48%	18.62%	0.00%	100.00%
Kam Tin Road	RD_13_14 RD_13_14	RR	50 50	30.3	Y	1700-1800	1115	37.39% 50.21%	4.06%	0.00%	17.21%	8.74% 7.63%	7.87%	0.55%	1.26%	0.00%	0.00%	0.00%	0.00%	0.35%	0.25%	0.45%	6.92% 5.16%	14.94%	0.00%	100.00%
Kam Tin Road	RD_13_14	RR	50	32.7	Ý	1900-2000	885	60.54%	3.61%	0.00%	11.14%	5.69%	2.16%	0.45%	1.36%	0.00%	0.00%	0.00%	0.11%	0.20%	0.31%	0.55%	4.56%	9.32%	0.00%	100.00%
Kam Tin Road	RD_13_14	RR	50	36.0	Y	2000-2100	570	51.61%	7.23%	0.00%	13.30%	2.49%	1.42%	0.35%	1.77%	0.00%	0.00%	0.00%	0.00%	0.00%	0.56%	0.75%	5.16%	15.37%	0.00%	100.00%
Kam Tin Road Kam Tin Road	RD_13_14 RD_13_14	RR	50 50	37.1	Y Y	2100-2200	470	48.05% 41.43%	6.48% 10.43%	0.00%	13.59% 12.70%	5.39% 4.58%	1.71% 5.48%	0.00%	1.51%	0.00%	0.00%	0.00%	0.00%	0.45%	1.27%	0.85%	5.61% 3.19%	15.10% 19.36%	0.00%	100.00%
Kam Tin Road	RD_13_14	RR	50	39.1	Ŷ	2300-0000	280	52.74%	15.48%	0.00%	10.10%	7.56%	3.97%	0.00%	0.00%	0.00%	0.00%	0.00%	0.39%	0.74%	2.15%	1.08%	5.05%	0.74%	0.00%	100.00%
Kam Tin Road	RD_15_16	RR	50	40.0	Y	0000-0100	185	73.25%	11.49%	0.00%	4.91%	5.47%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	2.15%	2.73%	0.00%	0.00%	100.00%
Kam Tin Road	RD_15_16 RD_15_16	RR	50 50	40.0	Ý	0200-0300	90	64.10% 68.15%	13.65%	0.00%	12.25% 6.80%	4.75%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	100.00%
Kam Tin Road	RD_15_16	RR	50	40.0	Y	0300-0400	75	52.84%	19.33%	0.00%	9.64%	16.75%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	1.44%	0.00%	0.00%	100.00%
Kam Tin Road	RD_15_16	RR	50 50	40.0	Y	0400-0500	95 170	59.74%	12.97%	0.00%	14.13%	10.95%	1.11%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	1.11%	0.00%	0.00%	100.00%
Kam Tin Road	RD_15_16	RR	50	37.7	Ý	0600-0700	405	68.48%	5.53%	0.00%	6.72%	4.03%	1.53%	0.50%	2.74%	0.00%	0.00%	0.24%	0.00%	0.50%	0.76%	1.50%	5.74%	1.74%	0.00%	100.00%
Kam Tin Road	RD_15_16	RR	50	32.2	Y	0700-0800	930	62.72%	3.01%	0.00%	10.28%	3.77%	0.84%	0.85%	1.39%	0.00%	0.00%	0.00%	3.22%	0.45%	0.75%	0.85%	5.38%	6.49%	0.00%	100.00%
Kam Tin Road Kam Tin Road	RD_15_16 RD_15_16	RR	50 50	32.3	Y	0800-0900	920 885	60.56% 57.04%	2.86%	0.00%	12.98% 12.63%	6.56% 6.47%	0.99%	1.20%	1.65%	0.00%	0.00%	0.35%	0.55%	0.20%	0.55%	0.65%	4.06%	6.84% 10.69%	0.00%	100.00%
Kam Tin Road	RD_15_16	RR	50	32.8	Ý	1000-1100	870	56.55%	3.92%	0.00%	13.18%	7.41%	1.73%	1.41%	1.49%	0.00%	0.00%	0.10%	0.00%	0.20%	0.60%	0.70%	1.49%	11.23%	0.00%	100.00%
Kam Tin Road	RD_15_16	RR	50	33.1	Y	1100-1200	845	53.35%	3.11%	0.00%	14.49%	9.35%	1.43%	1.30%	1.05%	0.00%	0.00%	0.35%	0.00%	0.00%	0.50%	0.70%	2.13%	12.24%	0.00%	100.00%
Kam Tin Road	RD_15_16 RD_15_16	RR	50	30.5	Ý	1300-1400	1095	32.49%	5.26%	0.00%	16.40%	12.77%	8.25%	2.20%	0.95%	0.00%	0.00%	0.00%	0.10%	0.40%	0.30%	0.50%	2.72%	19.54%	0.00%	100.00%
Kam Tin Road	RD_15_16	RR	50	30.8	Y	1400-1500	1060	32.96%	4.72%	0.00%	16.06%	12.57%	8.69%	1.70%	1.05%	0.00%	0.00%	0.00%	0.21%	0.40%	0.30%	0.45%	2.47%	18.44%	0.00%	100.00%
Kam Tin Road Kam Tin Road	RD_15_16 RD_15_16	RR	50 50	31.2 31.2	Y	1500-1600	1030 1030	33.80% 36.15%	4.21% 4.27%	0.00%	18.11% 15.21%	13.32% 12.22%	7.69% 6.33%	1.55%	0.85%	0.00%	0.00%	0.00%	0.10%	0.40%	0.30%	0.50%	2.32%	16.84% 18.56%	0.00%	100.00%
Kam Tin Road	RD_15_16	RR	50	32.6	Ŷ	1700-1800	895	58.06%	3.61%	0.00%	11.20%	5.65%	2.16%	0.45%	1.36%	0.00%	0.52%	0.23%	0.11%	0.20%	0.31%	0.55%	4.51%	11.07%	0.00%	100.00%
Kam Tin Road	RD_15_16	RR	50	31.8	Y	1800-1900	970	50.08%	4.66%	0.00%	13.56%	7.65%	3.40%	0.20%	1.51%	0.00%	0.00%	0.00%	0.11%	0.20%	0.31%	0.51%	5.16%	12.65%	0.00%	100.00%
Kam Tin Road	RD_15_16 RD_15_16	RR	50 50	32.7	Ý	2000-2000	880 570	58.37% 51.53%	3.77%	0.00%	11.77%	5.98% 2.46%	2.13%	0.44%	1.36%	0.00%	0.58%	0.22%	0.11%	0.24%	0.36%	0.56%	4.77%	9.35% 15.38%	0.00%	100.00%
Kam Tin Road	RD_15_16	RR	50	37.1	Y	2100-2200	465	47.83%	6.48%	0.00%	13.80%	5.60%	1.71%	0.00%	1.51%	0.00%	0.00%	0.00%	0.00%	0.45%	1.28%	0.85%	5.61%	14.89%	0.00%	100.00%
Kam Tin Road Kam Tin Road	RD_15_16 RD_15_16	RR	50 50	38.3 39.1	Y	2200-2300	350 280	41.46% 52 74%	10.38% 15.48%	0.00%	12.70%	4.62% 7.56%	5.48% 3.97%	0.00%	0.55%	0.00%	0.00%	0.00%	0.00%	0.00%	0.86% 2.15%	1.41%	3.19% 5.05%	19.36% 0.74%	0.00%	100.00%
Kam Tin Road	RD_17_18	RR	50	40.0	Ŷ	0000-0100	180	74.39%	11.38%	0.00%	5.11%	5.15%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	1.11%	2.86%	0.00%	0.00%	100.00%
Kam Tin Road	RD_17_18	RR	50	40.0	Y	0100-0200	105	65.61%	14.70%	0.00%	11.76%	4.94%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	1.00%	2.00%	0.00%	0.00%	100.00%
Kam Tin Road	RD 17 18	RR	50	40.0	Y	0200-0300	90 70	53.70%	18.80%	0.00%	10.10%	9.30%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	1.45%	0.00%	0.00%	100.00%
Kam Tin Road	RD_17_18	RR	50	40.0	Y	0400-0500	90	60.55%	13.50%	0.00%	13.50%	10.15%	1.15%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	1.15%	0.00%	0.00%	100.00%
Kam Tin Road Kam Tin Road	RD_17_18 RD_17_18	RR	50 50	40.0	Y Y	0500-0600	160 390	60.28% 68.94%	6.43% 5.70%	0.00%	7.66%	4.46%	3.79%	1.29%	3.85%	0.00%	0.00%	0.61%	6.43% 0.00%	1.29%	1.29%	0.68%	1.94%	0.00%	0.00%	100.00%
Kam Tin Road	RD_17_18	RR	50	32.5	Ŷ	0700-0800	900	62.86%	3.02%	0.00%	10.26%	3.79%	0.88%	0.89%	1.43%	0.00%	0.00%	0.00%	3.33%	0.45%	0.43%	0.43%	5.56%	6.66%	0.00%	100.00%
Kam Tin Road	RD_17_18	RR	50	32.6	Y	0800-0900	890	60.71%	2.83%	0.00%	12.97%	6.53%	1.02%	1.25%	1.69%	0.00%	0.00%	0.34%	0.55%	0.20%	0.32%	0.32%	4.21%	7.06%	0.00%	100.00%
Kam Tin Road	RD_17_18	RR	50	33.2	Ý	1000-1100	840	56.56%	3.89%	0.00%	13.12%	7.36%	1.80%	1.46%	1.53%	0.00%	0.00%	0.14%	0.00%	0.25%	0.38%	0.38%	1.56%	11.58%	0.00%	100.00%
Kam Tin Road	RD_17_18	RR	50	33.4	Y	1100-1200	820	53.40%	3.08%	0.00%	14.48%	9.30%	1.46%	1.35%	1.09%	0.00%	0.00%	0.34%	0.00%	0.00%	0.27%	0.37%	2.21%	12.65%	0.00%	100.00%
Kam Tin Road Kam Tin Road	RD_17_18 RD_17_18	RR	50 50	32.2	Y	1200-1300	935 1075	32.55%	5.05% 4.90%	0.00%	17.79% 16.27%	14.77% 13.01%	5.82% 8.43%	1.95%	1.00%	0.00%	0.00%	0.00%	0.10%	0.40%	0.10%	0.19%	2.82%	17.47%	0.00%	100.00%
Kam Tin Road	RD_17_18	RR	50	31.1	Ŷ	1400-1500	1040	33.11%	4.35%	0.00%	15.88%	12.82%	8.87%	1.75%	1.05%	0.00%	0.00%	0.00%	0.20%	0.40%	0.10%	0.20%	2.51%	18.77%	0.00%	100.00%
Kam Tin Road	RD_17_18	RR	50	31.4	Y	1500-1600	1010	33.85%	3.90%	0.00%	17.98%	13.62%	7.82%	1.60%	0.90%	0.00%	0.00%	0.00%	0.10%	0.40%	0.10%	0.20%	2.36%	17.17%	0.00%	100.00%
Kam Tin Road	RD 17 18	RR	50	31.4	Y	1700-1800	895	58.06%	4.00% 3.61%	0.00%	11.20%	5.65%	2.16%	0.45%	1.36%	0.00%	0.52%	0.23%	0.00%	0.20%	0.19%	0.19%	4.56%	11.07%	0.00%	100.00%
Kam Tin Road	RD_17_18	RR	50	32.0	Y	1800-1900	950	50.29%	4.34%	0.00%	13.44%	7.86%	3.46%	0.20%	1.56%	0.00%	0.00%	0.00%	0.11%	0.20%	0.09%	0.19%	5.31%	12.94%	0.00%	100.00%
Kam Tin Road Kam Tin Road	RD_17_18 RD_17_18	RR	50 50	32.7 36.2	Y Y	1900-2000	880 555	58.37% 52.11%	3.77%	0.00%	11.77% 13.23%	5.98% 2.56%	2.13% 1.43%	0.44%	1.36% 1.81%	0.00%	0.58%	0.22%	0.11%	0.24%	0.36%	0.56%	4.77% 5.25%	9.35% 15.79%	0.00%	100.00%
Kam Tin Road	RD_17_18	RR	50	37.3	Ý	2100-2200	450	48.47%	6.05%	0.00%	13.83%	5.81%	1.78%	0.00%	1.56%	0.00%	0.00%	0.00%	0.00%	0.45%	0.43%	0.43%	5.80%	15.41%	0.00%	100.00%
Kam Tin Road	RD_17_18	RR	50	38.4	Y	2200-2300	340	42.02%	9.85%	0.00%	12.88%	4.76%	5.67%	0.00%	0.60%	0.00%	0.00%	0.00%	0.00%	0.00%	0.29%	0.58%	3.27%	20.07%	0.00%	100.00%
Kam Tin Road	RD 19 20	RR	50 50	39.2 39.2	Y Y	2300-0000 0000-0100	205 270	54.72% 68.59%	14.72% 16.08%	0.00%	3.00%	7.95% 4.88%	4.17%	0.00%	0.00%	0.00%	0.00%	0.00%	0.37%	0.75%	0.75%	0.38%	5.27% 1.50%	0.75%	0.00%	100.00%
Kam Tin Road	RD_19_20	RR	50	40.0	Y	0100-0200	155	59.14%	21.72%	0.00%	8.56%	3.95%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	5.29%	1.35%	0.00%	0.00%	100.00%
Kam Tin Road Kam Tin Road	RD_19_20 RD_19_20	RR	50 50	40.0 ∡0.0	Y	0200-0300	125	63.24% 46.45%	20.79%	0.00%	4.79%	8.80% 15.15%	1.60%	0.78%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	100.00%
Kam Tin Road	RD_19_20	RR	50	40.0	Ý	0400-0500	135	54.46%	19.67%	0.00%	9.87%	9.81%	1.55%	0.00%	1.55%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	1.55%	1.55%	0.00%	0.00%	100.00%
Kam Tin Road	RD_19_20	RR	50	39.6	Y	0500-0600	230	55.95%	9.30%	0.00%	5.35%	4.45%	4.45%	2.25%	5.35%	0.00%	0.00%	1.35%	3.55%	2.20%	1.35%	3.55%	0.90%	0.00%	0.00%	100.00%

Road Name	Road ID	Road Type	Speed Limit (km/hr)	Average Speed (km/hr)	Start Emission Estimated by Broad Brush Approach (Y/N)	Hour	Total Vehicles (Veh/hr)	01 - Private Cars	02 - Taxi	03 - Light Goods Vehicles< =2.5t	04 - Light Goods Vehicles 2.5-3.5t	05 - Light Goods Vehicles> 3.5t	06 - Medium Goods Vehicles< =15t	07 - Medium Goods Vehicles1 5-24t	08 - Public Light Buses	09 - Private Light Bus <=3.5t	10 - Private Light Bus >3.5t	11 - Non- franchise d Bus<=6.4 t	12 - Non- franchise d Bus 6.4 15t	13 - Non- franchise d Bus 15- 24t	14 - Franchise d Bus Single Deck	15 - Franchise d Bus Double Deck	16 - Motorcycl es	17 - Heavy Goods Vehicles> 24t	18 - Non- franchise d Bus >24t	Total
Kam Tin Road	RD 19 20	RR	50	35.8	Y	0600-0700	585	PC 65.98%	8.11%	LGV3 0.00%	LGV4 4.51%	LGV6 3.29%	1.75%	0.70%	<u>PLB</u> 4.15%	0.00%	0.00%	0.49%	0.00%	0.49%	0.49%	4.35%	MC 3.99%	HGV9 1.69%	0.00%	100.00%
Kam Tin Road	RD_19_20	RR	50	28.7	Y	0700-0800	1265	64.52%	4.76%	0.00%	7.40%	3.59%	0.95%	1.30%	2.20%	0.00%	0.00%	0.10%	1.90%	0.78%	0.60%	2.45%	4.13%	5.33%	0.00%	100.00%
Kam Tin Road Kam Tin Road	RD_19_20 RD_19_20	RR	50 50	28.8 28.7	Y Y	0800-0900	1250 1260	62.40% 58.04%	4.36%	0.00%	9.30% 10.26%	6.18% 6.13%	1.15%	1.75% 2.46%	2.60%	0.00%	0.00%	0.60%	0.30%	0.44%	0.40%	1.80%	3.04%	5.68% 8.23%	0.00%	100.00%
Kam Tin Road	RD_19_20	RR	50	29.5	Y	1000-1100	1185	58.05%	6.10%	0.00%	9.46%	6.88%	2.05%	2.00%	2.40%	0.00%	0.00%	0.20%	0.00%	0.34%	0.40%	2.10%	1.15%	8.88%	0.00%	100.00%
Kam Tin Road Kam Tin Road	RD_19_20 RD_19_20	RR	50 50	30.2 29.5	Y Y	1100-1200	1125 1190	55.93% 39.60%	4.92% 6.70%	0.00%	10.51% 12.75%	8.93% 16.20%	1.65% 1.45%	1.85%	1.65%	0.00%	0.00%	0.45%	0.00%	0.19%	0.35%	1.85%	1.55% 2.45%	10.18% 14.95%	0.00%	100.00%
Kam Tin Road	RD_19_20	RR	50	27.9	Y	1300-1400	1335	39.35%	6.43%	0.00%	11.97%	14.62%	2.09%	1.38%	1.94%	0.00%	0.00%	0.00%	0.00%	0.05%	0.55%	1.65%	2.20%	17.78%	0.00%	100.00%
Kam Tin Road Kam Tin Road	RD_19_20 RD_19_20	RR RR	50 50	28.3 28.6	Y	1400-1500	1305 1270	41.24% 42.26%	5.74% 5.09%	0.00%	11.61% 13.01%	14.31% 15.16%	2.23%	1.06% 0.96%	2.40% 1.90%	0.00%	0.00%	0.00%	0.25%	0.15%	0.35%	1.70%	2.20%	16.77% 15.16%	0.00%	100.00% 100.00%
Kam Tin Road	RD_19_20	RR	50	28.3	Ŷ	1600-1700	1300	43.99%	5.01%	0.00%	10.69%	13.54%	1.64%	0.40%	2.25%	0.00%	0.00%	0.00%	0.00%	0.10%	0.55%	1.55%	4.00%	16.29%	0.00%	100.00%
Kam Tin Road Kam Tin Road	RD_19_20 RD 19_20	RR RR	50 50	27.4 28.2	Y Y	1700-1800 1800-1900	1390 1310	46.53% 57.54%	4.91% 5.24%	0.00%	12.38% 8.85%	9.79% 7.88%	1.92% 0.75%	0.34%	2.81% 3.22%	0.00%	0.00%	0.00%	0.00%	0.15%	0.20%	1.60% 1.49%	6.19% 4.30%	13.18% 10.02%	0.00%	100.00% 100.00%
Kam Tin Road	RD_19_20	RR	50	28.9	Ŷ	1900-2000	1245	66.80%	3.88%	0.00%	7.08%	5.67%	0.50%	0.23%	2.77%	0.00%	0.00%	0.00%	0.16%	0.09%	0.31%	1.94%	3.64%	6.92%	0.00%	100.00%
Kam Tin Road Kam Tin Road	RD_19_20 RD 19_20	RR RR	50 50	33.8 35.4	Y Y	2000-2100 2100-2200	780 630	58.05% 54.62%	8.11% 7.19%	0.00%	8.58% 8.92%	2.56% 5.63%	0.37%	0.14%	3.48% 2.72%	0.00%	0.00%	0.00%	0.00%	0.00%	0.68%	2.28% 2.88%	4.08% 4.64%	11.68% 11.50%	0.00%	100.00% 100.00%
Kam Tin Road	RD_19_20	RR	50	37.2	Y	2200-2300	455	47.60%	12.17%	0.00%	8.87%	4.91%	1.32%	0.00%	1.56%	0.00%	0.00%	0.00%	0.00%	0.00%	1.13%	4.44%	2.67%	15.33%	0.00%	100.00%
Kam Tin Road Lam Kam Road	RD_19_20 RD_21_22	RR RR	50 50	37.8 39.7	Y Y	2300-0000 0000-0100	395 195	57.01% 77.05%	16.45% 13.54%	0.00%	6.43% 2.64%	7.42% 2.09%	0.77%	0.00%	0.55%	0.00%	0.00%	0.00%	0.27%	0.27%	1.81% 0.00%	4.12% 3.63%	4.12% 1.05%	0.77%	0.00%	100.00% 100.00%
Lam Kam Road	RD_21_22	RR	50	40.0	Ŷ	0100-0200	110	67.91%	18.37%	0.00%	7.33%	2.73%	0.89%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	2.77%	0.00%	0.00%	0.00%	100.00%
Lam Kam Road Lam Kam Road	RD_21_22 RD_21_22	RR RR	50 50	40.0 40.0	Y Y	0200-0300	95 75	70.64% 57.96%	17.38%	0.00%	4.35% 7.26%	4.34% 7.27%	3.29% 1.44%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	100.00% 100.00%
Lam Kam Road	RD_21_22	RR	50	40.0	Ŷ	0400-0500	95	66.71%	16.63%	0.00%	8.92%	4.40%	2.25%	0.00%	1.09%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	100.00%
Lam Kam Road	RD_21_22 RD_21_22	RR	50 50	39.9 37.6	Y	0500-0600	160 420	67.20% 73.32%	8.36% 6.77%	0.00%	4.51% 3.84%	1.90%	7.09% 2.44%	1.26%	4.53% 3.40%	0.00%	0.00%	0.00%	0.00%	1.94%	0.00%	1.95% 2.66%	1.26%	0.00%	0.00%	100.00%
Lam Kam Road	RD_21_22	RR	50	33.0	Ŷ	0700-0800	855	75.70%	4.24%	0.00%	6.57%	1.80%	1.55%	0.85%	2.01%	0.00%	0.00%	0.00%	0.00%	0.70%	0.00%	1.54%	2.82%	2.22%	0.00%	100.00%
Lam Kam Road	RD_21_22	RR	50 50	33.1	Y	0800-0900	845 795	73.08%	3.92%	0.00%	8.32%	3.08%	1.79%	1.21%	2.26%	0.00%	0.00%	0.00%	0.00%	0.38%	0.00%	1.17%	2.15%	2.65%	0.00%	100.00%
Lam Kam Road	RD_21_22 RD_21_22	RR	50	33.8	Y	1000-1100	780	69.97%	5.62%	0.00%	8.57%	3.43%	3.35%	1.41%	2.17%	0.00%	0.00%	0.00%	0.00%	0.25%	0.00%	1.40%	0.75%	3.09%	0.00%	100.00%
Lam Kam Road	RD_21_22	RR	50 50	34.3	Y	1100-1200	730	68.42% 51.66%	4.54%	0.00%	9.88%	4.68%	2.75%	1.40%	1.52%	0.00%	0.00%	0.00%	0.00%	0.12%	0.00%	1.34%	1.09%	4.27%	0.00%	100.00%
Lam Kam Road	RD_21_22 RD_21_22	RR	50	34.6	Ý	1300-1400	705	49.87%	5.27%	0.00%	10.57%	8.66%	4.70%	1.70%	2.29%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	1.57%	2.14%	13.22%	0.00%	100.00%
Lam Kam Road	RD_21_22	RR	50 50	34.7	Y	1400-1500	695 675	51.77%	4.75%	0.00%	10.13%	8.52%	4.91%	1.29%	2.64%	0.00%	0.00%	0.00%	0.28%	0.00%	0.00%	1.91%	2.06%	11.74%	0.00%	100.00%
Lam Kam Road	RD_21_22 RD_21_22	RR	50	34.5	Ý	1600-1700	695	55.41%	4.22%	0.00%	9.31%	7.97%	3.52%	0.44%	2.63%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	1.61%	3.60%	11.29%	0.00%	100.00%
Lam Kam Road	RD_21_22	RR	50	34.6	Y	1700-1800	700	76.11%	3.05%	0.00%	5.92%	3.13%	1.01%	0.27%	2.88%	0.00%	0.13%	0.13%	0.17%	0.00%	0.00%	1.91%	3.14%	2.16%	0.00%	100.00%
Lam Kam Road	RD_21_22 RD_21_22	RR	50	34.3	Ý	1900-2000	735	69.51%	4.66%	0.00%	8.03%	3.94%	1.62%	0.39%	2.83%	0.00%	0.12%	0.12%	0.12%	0.00%	0.00%	1.73%	4.36%	2.56%	0.00%	100.00%
Lam Kam Road	RD_21_22	RR	50	37.5	Y	2000-2100	430	70.31%	6.30%	0.00%	7.24%	1.43%	0.71%	0.22%	3.74%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	2.37%	3.48%	4.20%	0.00%	100.00%
Lam Kam Road	RD_21_22 RD_21_22	RR	50	39.5	Ý	2200-2300	235	64.01%	10.50%	0.00%	7.90%	3.05%	3.08%	0.00%	1.78%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	4.79%	2.67%	2.22%	0.00%	100.00%
Lam Kam Road	RD_21_22	RR	50	39.6	Y	2300-0000	230	66.85%	12.40%	0.00%	5.33%	4.41%	1.76%	0.00%	0.89%	0.00%	0.00%	0.00%	0.45%	0.00%	0.00%	3.97%	3.06%	0.89%	0.00%	100.00%
Route Twisk	RD_23_24 RD_23_24	RR	50	40.0	Y	0100-0200	55	78.16%	8.08%	0.00%	2.86% 9.97%	1.89%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	1.89%	0.00%	0.00%	100.00%
Route Twisk	RD_23_24	RR	50	40.0	Y	0200-0300	50	76.41%	7.14%	0.00%	7.14%	7.14%	2.17%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	100.00%
Route Twisk	RD_23_24 RD_23_24	RR	50	40.0	Y	0400-0500	45	73.19%	7.28%	0.00%	12.26%	7.28%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	100.00%
Route Twisk	RD_23_24	RR	50	40.0	Y	0500-0600	80	71.93%	4.03%	0.00%	6.65%	1.35%	4.05%	2.68%	6.65%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	2.68%	0.00%	0.00%	100.00%
Route Twisk	RD_23_24 RD_23_24	RR	50	36.5	Y	0700-0800	525	80.70%	1.37%	0.00%	6.73%	1.52%	0.60%	1.17%	2.51%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	5.39%	0.00%	0.00%	100.00%
Route Twisk	RD_23_24	RR	50	36.5	Y	0800-0900	520	78.28%	1.38%	0.00%	8.48%	2.88%	0.75%	1.53%	2.88%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	3.85%	0.00%	0.00%	100.00%
Route Twisk	RD_23_24 RD_23_24	RR	50	36.4	Y	1000-1100	465	77.19%	2.17%	0.00%	9.33%	3.50%	1.51%	1.46%	2.89%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	4.33%	0.00%	0.00%	100.00%
Route Twisk	RD_23_24	RR	50	37.5	Y	1100-1200	430	76.18%	1.67%	0.00%	10.61%	4.45%	1.21%	1.86%	1.88%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	2.14%	0.00%	0.00%	100.00%
Route Twisk	RD_23_24 RD_23_24	RR	50	38.6	Ý	1300-1400	320	53.93%	4.73%	0.00%	17.04%	8.88%	3.15%	1.91%	3.44%	0.00%	0.00%	0.00%	0.00%	0.00%	1.25%	0.00%	3.78%	1.91%	0.00%	100.00%
Route Twisk	RD_23_24	RR	50	38.7	Y	1400-1500	315	56.12%	3.87%	0.00%	16.62%	8.37%	3.20%	1.57%	4.19%	0.00%	0.00%	0.00%	0.00%	0.00%	0.65%	0.00%	3.84%	1.57%	0.00%	100.00%
Route Twisk	RD_23_24 RD_23_24	RR	50	38.6	Y	1600-1700	325	50.37 % 57.89%	3.38%	0.00%	14.90%	7.73%	2.55%	0.61%	4.04%	0.00%	0.00%	0.00%	0.00%	0.00%	1.25%	0.00%	5.89% 6.78%	1.57%	0.00%	100.00%
Route Twisk	RD_23_24	RR	50	38.1	Y	1700-1800	370	76.05%	2.20%	0.00%	8.20%	2.75%	0.50%	0.25%	4.10%	0.00%	0.00%	0.00%	0.00%	0.00%	0.50%	0.00%	5.20%	0.25%	0.00%	100.00%
Route Twisk	RD_23_24 RD_23_24	RR	50	36.7	Y	1900-1900	500	77.95%	2.95%	0.00%	8.05%	4.05% 2.40%	0.40%	0.25%	3.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.40%	0.00%	5.20%	0.55%	0.00%	100.00%
Route Twisk	RD_23_24	RR	50	39.7	Y	2000-2100	220	69.35%	5.15%	0.00%	10.65%	1.40%	0.00%	0.00%	5.60%	0.00%	0.00%	0.00%	0.00%	0.00%	0.90%	0.00%	6.50%	0.45%	0.00%	100.00%
Route Twisk	RD_23_24 RD_23_24	RR	50	40.0	Y	2200-2300	110	63.70%	4.75%	0.00%	13.25%	3.70%	1.85%	0.00%	2.75%	0.00%	0.00%	0.00%	0.00%	0.00%	2.40%	0.00%	4.65%	0.00%	0.00%	100.00%
Route Twisk	RD_23_24	RR	50	40.0	Y	2300-0000	115	65.32%	10.28%	0.00%	7.52%	3.77%	1.90%	0.00%	0.94%	0.00%	0.00%	0.00%	0.00%	0.00%	3.74%	0.00%	6.53%	0.00%	0.00%	100.00%
Access Road	RD_25_26 RD_25_26	RR	30 30	26.0	ř Y	0100-0100	10	40.00%	0.00%	0.00%	40.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	20.00%	0.00%	100.00%
Access Road	RD_25_26	RR	30	26.0	Y	0200-0300	10	40.00%	0.00%	0.00%	40.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	20.00%	0.00%	100.00%
Access Road	RD_25_26 RD_25_26	RR	30 30	26.0 26.0	r Y	0400-0400	10	40.00%	0.00%	0.00%	40.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	20.00% 20.00%	0.00%	100.00%
Access Road	RD_25_26	RR	30	26.0	Y	0500-0600	10	40.00%	0.00%	0.00%	40.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	20.00%	0.00%	100.00%
Access Road	RD_25_26 RD_25_26	RR	30	26.0	ř Y	0700-0800	20	40.00% 55.60%	0.00%	0.00%	40.00% 33.30%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	20.00% 11.10%	0.00%	100.00%

Road Name	Road ID	Road Type	Speed Limit (km/hr)	Average Speed (km/hr)	Start Emission Estimated by Broad Brush Approach (Y/N)	Hour	Total Vehicles (Veh/hr)	01 - Private Cars	02 - Taxi	03 - Light Goods Vehicles< =2.5t	04 - Light Goods Vehicles 2.5-3.5t	05 - Light Goods Vehicles> 3.5t	06 - Medium Goods Vehicles< =15t	07 - Medium Goods Vehicles1 5-24t	08 - Public Light Buses	09 - Private Light Bus <=3.5t	10 - Private Light Bus >3.5t	11 - Non- franchise d Bus<=6.4 t	12 - Non- franchise d Bus 6.4 15t	13 - Non- franchise d Bus 15- 24t	14 - Franchise d Bus Single Deck	15 - Franchise d Bus Double Deck	16 - Motorcycl es	17 - Heavy Goods Vehicles> 24t	18 - Non- franchise d Bus >24t	Total
Access Road	RD 25 26	RR	30	26.0	Y	0800-0900	20	55.60%	0.00%	0.00%	33.30%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	11.10%	0.00%	100.00%
Access Road	RD_25_26	RR	30	26.0	Y	0900-1000	20	55.60%	0.00%	0.00%	33.30%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	11.10%	0.00%	100.00%
Access Road	RD_25_26	RR	30	26.0	Y	1000-1100	20	55.60%	0.00%	0.00%	33.30%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	11.10%	0.00%	100.00%
Access Road	RD_25_26 RD_25_26	RR	30	26.0	Y	1200-1300	20	55.60%	0.00%	0.00%	33.30%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	11.10%	0.00%	100.00%
Access Road	RD_25_26	RR	30	26.0	Y	1300-1400	20	55.60%	0.00%	0.00%	33.30%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	11.10%	0.00%	100.00%
Access Road	RD_25_26	RR	30	26.0	Y	1400-1500	20	55.60%	0.00%	0.00%	33.30%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	11.10%	0.00%	100.00%
Access Road	RD 25 26	RR	30	26.0	Y	1600-1700	20	55.60%	0.00%	0.00%	33.30%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	11.10%	0.00%	100.00%
Access Road	RD_25_26	RR	30	26.0	Y	1700-1800	20	55.60%	0.00%	0.00%	33.30%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	11.10%	0.00%	100.00%
Access Road	RD_25_26	RR	30 30	26.0	Ŷ	1800-1900	20	55.60%	0.00%	0.00%	33.30%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	11.10%	0.00%	100.00%
Access Road	RD_25_26	RR	30	26.0	Ý	2000-2100	10	40.00%	0.00%	0.00%	40.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	20.00%	0.00%	100.00%
Access Road	RD_25_26	RR	30	26.0	Y	2100-2200	10	40.00%	0.00%	0.00%	40.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	20.00%	0.00%	100.00%
Access Road	RD_25_26	RR	30	26.0	Y	2200-2300	10	40.00%	0.00%	0.00%	40.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	20.00%	0.00%	100.00%
Kam Sheung Road	RD_27_28	RR	50	40.0	Ý	0000-0100	100	55.56%	20.16%	0.00%	6.05%	6.10%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	10.11%	2.04%	0.00%	0.00%	100.00%
Kam Sheung Road	RD_27_28	RR	50	40.0	Y	0100-0200	65	41.98%	26.44%	0.00%	14.00%	7.05%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	10.53%	0.00%	0.00%	0.00%	100.00%
Kam Sheung Road Kam Sheung Road	RD_27_28 RD_27_28	RR	50 50	40.0	Y	0200-0300	50 40	46.30% 27.75%	28.15%	0.00%	9.20%	13.95%	2.40%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	100.00%
Kam Sheung Road	RD_27_28	RR	50	40.0	Ý	0400-0500	45	40.00%	24.44%	0.00%	17.78%	13.33%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	4.44%	0.00%	0.00%	0.00%	100.00%
Kam Sheung Road	RD_27_28	RR	50	40.0	Y	0500-0600	85	40.55%	10.82%	0.00%	9.92%	4.89%	4.80%	1.18%	7.29%	0.00%	0.00%	2.35%	7.38%	3.53%	0.00%	7.29%	0.00%	0.00%	0.00%	100.00%
Kam Sheung Road Kam Sheung Road	RD_27_28 RD_27_28	RR	50 50	39.7	Y	0500-0700	220	53.96% 55.10%	10.02% 5.92%	0.00%	13.08%	4.56%	1.80%	0.45%	5.91%	0.00%	0.00%	0.91%	0.00%	0.91%	0.00%	8.24% 4.66%	2.73%	2.73%	0.00%	100.00%
Kam Sheung Road	RD_27_28	RR	50	37.0	Ŷ	0800-0900	475	52.65%	5.33%	0.00%	16.30%	8.52%	1.09%	1.09%	3.65%	0.00%	0.00%	1.09%	0.61%	0.65%	0.00%	3.43%	2.16%	3.43%	0.00%	100.00%
Kam Sheung Road	RD_27_28	RR	50	37.2	Y	0900-1000	455	52.23%	5.34%	0.00%	16.21%	8.43%	1.13%	1.13%	3.79%	0.00%	0.00%	1.13%	0.67%	0.69%	0.00%	3.53%	2.20%	3.53%	0.00%	100.00%
Kam Sheung Road Kam Sheung Road	RD_27_28 RD_27_28	RR	50 50	37.4 37.5	Y Y	1000-1100	440 430	49.66% 47.34%	7.52% 5.89%	0.00%	16.90% 18.68%	9.57% 12.35%	2.05%	1.13%	3.40%	0.00%	0.00%	0.22%	0.00%	0.48%	0.00%	4.08%	0.93%	4.08% 5.25%	0.00%	100.00%
Kam Sheung Road	RD_27_28	RR	50	36.8	Ŷ	1200-1300	495	32.09%	8.35%	0.00%	20.92%	15.07%	5.85%	4.46%	2.25%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	3.44%	2.05%	5.53%	0.00%	100.00%
Kam Sheung Road	RD_27_28	RR	50	36.1	Y	1300-1400	565	31.41%	7.85%	0.00%	19.21%	13.32%	8.37%	5.38%	2.14%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	2.68%	1.98%	7.66%	0.00%	100.00%
Kam Sheung Road	RD_27_28 RD_27_28	RR	50 50	36.2	Ý	1400-1500	530	33.29%	7.20% 6.46%	0.00%	18.75%	13.20%	9.00%	4.23%	2.58%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	2.93%	1.83%	6.98% 5.85%	0.00%	100.00%
Kam Sheung Road	RD_27_28	RR	50	36.6	Ŷ	1600-1700	515	37.59%	6.60%	0.00%	18.31%	13.19%	6.65%	1.55%	2.70%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	2.89%	3.49%	7.02%	0.00%	100.00%
Kam Sheung Road	RD_27_28	RR	50	36.9	Y	1700-1800	480	60.40%	5.25%	0.00%	12.65%	5.70%	2.10%	1.05%	3.40%	0.00%	0.00%	1.05%	0.00%	0.00%	0.00%	3.60%	3.35%	1.45%	0.00%	100.00%
Kam Sheung Road Kam Sheung Road	RD_27_28 RD_27_28	RR	50 50	36.6	Ý	1900-2000	470	51.55% 58.47%	6.95% 5.63%	0.00%	13.59%	8.00% 5.98%	3.40%	1.06%	4.00% 3.41%	0.00%	0.00%	1.08%	0.00%	0.00%	0.00%	3.00%	3.80%	3.00%	0.00%	100.00%
Kam Sheung Road	RD_27_28	RR	50	38.9	Y	2000-2100	295	52.73%	11.35%	0.00%	15.54%	2.78%	1.37%	0.70%	4.49%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	4.48%	3.78%	2.78%	0.00%	100.00%
Kam Sheung Road	RD_27_28	RR	50	39.5	Y	2100-2200	235	49.67%	9.96%	0.00%	17.02%	6.04%	1.74%	0.00%	3.89%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	5.62%	4.72%	1.34%	0.00%	100.00%
Kam Sheung Road	RD 27 28	RR	50	40.0	Ý	2300-0000	160	45.40%	20.65%	0.00%	11.25%	6.90%	3.80%	0.00%	0.65%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	7.55%	3.15%	0.65%	0.00%	100.00%
Sheung Tsuen B/T and Car Park	RD_29	RR	30	26.0	Y	0000-0100	10	57.10%	14.30%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	28.60%	0.00%	0.00%	0.00%	100.00%
Sheung Tsuen B/T and Car Park	RD_29	RR	30 30	26.0	Y	0100-0200	5	50.00%	25.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	25.00%	0.00%	0.00%	0.00%	100.00%
Sheung Tsuen B/T and Car Park	RD_29	RR	30	26.0	Ý	0300-0400	5	50.00%	50.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	100.00%
Sheung Tsuen B/T and Car Park	RD_29	RR	30	26.0	Y	0400-0500	5	100.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	100.00%
Sheung Tsuen B/T and Car Park Sheung Tsuen B/T and Car Park	RD_29 RD_29	RR	30 30	26.0 26.0	Y	0500-0600	5 15	50.00% 53.20%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	25.00% 6.70%	25.00%	0.00%	0.00%	0.00%	100.00%
Sheung Tsuen B/T and Car Park	RD_29	RR	30	26.0	Ŷ	0700-0800	30	63.40%	3.30%	0.00%	6.70%	3.30%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	10.00%	13.30%	0.00%	0.00%	0.00%	100.00%
Sheung Tsuen B/T and Car Park	RD_29	RR	30	26.0	Y	0800-0900	30	62.20%	3.40%	0.00%	10.30%	6.90%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	6.90%	10.30%	0.00%	0.00%	0.00%	100.00%
Sheung Tsuen B/T and Car Park	RD 29	RR	30	26.0	Ý	1000-1100	30	59.30%	3.70%	0.00%	11.10%	7.40%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	7.40%	11.10%	0.00%	0.00%	0.00%	100.00%
Sheung Tsuen B/T and Car Park	RD_29	RR	30	26.0	Y	1100-1200	30	54.00%	3.80%	0.00%	11.50%	11.50%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	7.70%	11.50%	0.00%	0.00%	0.00%	100.00%
Sheung Tsuen B/T and Car Park	RD_29	RR	30 30	26.0	Y	1200-1300	20	27.70%	16.70%	0.00%	27.80%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	11.10%	16.70% 14.30%	0.00%	0.00%	0.00%	100.00%
Sheung Tsuen B/T and Car Park	RD_29	RR	30	26.0	Ý	1400-1500	20	31.60%	15.80%	0.00%	26.30%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	10.50%	15.80%	0.00%	0.00%	0.00%	100.00%
Sheung Tsuen B/T and Car Park	RD_29	RR	30	26.0	Y	1500-1600	20	30.00%	15.00%	0.00%	30.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	10.00%	15.00%	0.00%	0.00%	0.00%	100.00%
Sheung Tsuen B/T and Car Park	RD_29	RR	30	26.0	Y	1600-1700	25	33.30%	14.30%	0.00%	23.80%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	14.30% 9.10%	14.30% 13.60%	0.00%	0.00%	0.00%	100.00%
Sheung Tsuen B/T and Car Park	RD_29	RR	30	26.0	Ŷ	1800-1900	25	42.90%	14.30%	0.00%	19.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	9.50%	14.30%	0.00%	0.00%	0.00%	100.00%
Sheung Tsuen B/T and Car Park	RD_29	RR	30	26.0	Y	1900-2000	20	50.00%	10.00%	0.00%	15.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	10.00%	15.00%	0.00%	0.00%	0.00%	100.00%
Sneung Isuen B/I and Car Park Sheung Tsuen B/T and Car Park	RD 29	RR	30 30	26.0 26.0	Y Y	2000-2100	15 15	35.70% 28.50%	21.40% 14.30%	0.00%	14.30% 14.30%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	14.30% 28.60%	14.30% 14.30%	0.00%	0.00%	0.00%	100.00%
Sheung Tsuen B/T and Car Park	RD_29	RR	30	26.0	Ŷ	2200-2300	10	20.00%	20.00%	0.00%	10.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	20.00%	30.00%	0.00%	0.00%	0.00%	100.00%
Sheung Tsuen B/T and Car Park	RD_29	RR	30	26.0	Y	2300-0000	15 F	23.10%	23.10%	0.00%	7.70%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	30.70%	15.40%	0.00%	0.00%	0.00%	100.00%
Village Access Road	RD_30_31 RD_30_31	RR	30	26.0 26.0	Ϋ́	0100-0100	5 0	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%
Village Access Road	RD_30_31	RR	30	26.0	Y	0200-0300	0	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%
Village Access Road	RD_30_31	RR	30 30	26.0	Y	0300-0400	5	0.00%	0.00%	0.00%	0.00%	100.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	100.00%
Village Access Road	RD_30_31	RR	30	26.0	Ý	0500-0600	5	100.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	100.00%
Village Access Road	RD_30_31	RR	30	26.0	Y	0600-0700	10	58.35%	0.00%	0.00%	0.00%	41.65%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	100.00%
Village Access Road	RD_30_31 RD_30_31	RR	30 30	26.0 26.0	Y	0700-0800	10 15	65.00% 57.80%	0.00%	0.00%	0.00%	35.00% 42.20%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	100.00%
Village Access Road	RD_30_31	RR	30	26.0	Ŷ	0900-1000	15	57.80%	0.00%	0.00%	0.00%	42.20%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	100.00%

PC   TAXL   IGV3   IGV4   IGV6   HGV7   HGV8   PLB   PV4   PV5   NEB6   NEB7   NEB8   EBSD   EBDD   MC   HG	% 0.00% 100.00 % 0.00% 100.00 % 0.00% 100.00
Village Access Road RD 30 31 RR 30 26.0 Y 1000-1100 15 57.80% 0.00% 0.00% 42.20% 0.0	% 0.00% 100.00 % 0.00% 100.00
Village Access Road RD_30_31 RR 30 26.0 Y 1100-1200 20 41.65% 0.00% 0.00% 0.00% 58.35% 0.00%	% 0.00% 100.00 <sup>°</sup>
Village Access Road RD_30_31 RR 30 26.0 Y 1200-1300 10 40.00% 0.00	a 000 100 00
Village Access Road RD_30_51 RR 30 26.0 Y 1300-1400 10 40.00% 0.0	% 0.00% 100.00 % 0.00% 100.00
Village Access Road RD_30_31 RR 30 26.0 Y 1500-1600 10 40.00% 0.00	% 0.00% 100.00
Village Access Road RD_30_31 RR 30 26.0 Y 1600-1700 10 40.00% 0.00	% 0.00% 100.00
Village Access Road RD_30_31 RR 30 26.0 Y 1700-1800 10 50.00% 0.00% 0.00% 50.00% 0.0	% 0.00% 100.00 <sup>4</sup>
Village Access Road RD 30 1 RR 30 26.0 Y 1900-2000 10 75.00% 0.00%	% 0.00% 100.00
Village Access Road RD_30_31 RR 30 26.0 Y 2000-2100 10 100.00% 0.00%	% 0.00% 100.00
Village Access Road RD_30_31 RR 30 26.0 Y 2100-2200 10 50.00% 0.00	% 0.00% 100.00 <sup>4</sup>
Village Access Road RD_30_31 RR 30 26.0 1 2200-2300 10 100.00% 0.00%	% 0.00% 100.00 % 0.00% 100.00
miningen Access Teach 1 200-000 10 1000 10 1000 100 1000 1000	% 0.00% 0.00%
Emission Point 19 Paved Road EP19_1_7 LD 30 15.0 Y 0100-0200 0 0.00%	% 0.00% 0.00%
Emission Point 19 Paved Road EP19_1_7 LD 30 15.0 Y 0200-0300 0 0.00% 0.0	% 0.00% 0.00%
Emission Point 19 Favea Road EP19_1/LD 30 15.0 Y 0.400-4900 0 0.00% 0.00	% 0.00% 0.00% % 0.00% 0.00%
Emission Point 19 Paved Road EP19_17 LD 30 15.0 Y 0500-0600 0 0.00%	% 0.00% 0.00%
Emission Point 19 Paved Road EP19_1_7 LD 30 15.0 Y 0600-0700 0 0.00%	% 0.00% 0.00%
Emission Point 19 Payed Road EP19_1_7 LD 30 15.0 Y 0700-0800 68 0.00% 0.	J% 0.00% 100.00
Emission Fouriti 9 Faveto Road EF19_7/ED 30 15.0 Y 0900-1000 68 0.00% 0.	0% 0.00% 100.00
Emission Point 19 Paved Road E19-1-7 LD 30 15.0 Y 1000-1100 68 0.00% 0.0	0.00% 100.00
Emission Point 19 Paved Road EP19_1_7 LD 30 15.0 Y 1100-1200 68 0.00%	J% 0.00% 100.00 <sup>4</sup>
Emission Point 19 Paved Road EP19_1.7 LD 30 15.0 Y 1200-1300 68 0.00% 0.	J% 0.00% 100.00 <sup>4</sup>
Emission Four 19 Paved Road EP19 17 LD 30 15.0 Y 1400-1500 68 0.00% 0.00	0% 0.00% 100.00
Emission Point 19 Paved Road EP19_1_7 LD 30 15.0 Y 1500-1600 68 0.00%	J% 0.00% 100.00 <sup>4</sup>
Emission Point 19 Paved Road EP19_1_7 LD 30 15.0 Y 1600-1700 68 0.00% 0.	J% 0.00% 100.00
Emission Point 19 Favea Road EP19_1/LD 30 15.0 Y 17/00-1800 68 0.00%	0% 0.00% 100.00 0% 0.00% 100.00
Emission Point 19 Paved Road EP19 17 LD 30 15.0 Y 1900-2000 68 0.00% 0.0	0% 0.00% 100.00
Emission Point 19 Paved Road EP19_1_7 LD 30 15.0 Y 2000-2100 68 0.00% 0.	J% 0.00% 100.00 <sup>6</sup>
Emission Point 19 Paved Road EP19_1_7 LD 30 15.0 Y 2100-2200 68 0.00% 0.	J% 0.00% 100.00
Emission Fouri 19 Faved Road EP19 17 LD 30 15.0 Y 2300-2000 0 0.00% 0.00	% 0.00% 100.00 % 0.00% 0.00%
Emission Point 19 Paved Road EP19_8 LD 30 15.0 Y 0000-0100 0 0.00%	% 0.00% 0.00%
Emission Point 19 Paved Road EP19_8 LD 30 15.0 Y 0100-0200 0 0.00%	% 0.00% 0.00%
Emission Point 19 Favea Road EP19_6 LD 30 15.0 Y 0200-0300 0 0.00%	% 0.00% 0.00% % 0.00% 0.00%
Emission Point 19 Paved Road EP19_8 LD 30 15.0 Y 0400-0500 0 0.00%	% 0.00% 0.00%
Emission Point 19 Paved Road EP19_8 LD 30 15.0 Y 0500-0600 0 0.00%	% 0.00% 0.00%
Emission Point 19 Payed Road EP19_8 LD 30 15.0 Y 0600-0700 0 0.00%	% 0.00% 0.00%
Emission Fouriti 9 Faveto Road EF19_0 ED 30 15.0 Y 080-0900 20 0.00% 0.0	0% 0.00% 100.00
Emission Point 19 Paved Road EP19_8 LD 30 15.0 Y 0900-1000 20 0.00% 0.00	J% 0.00% 100.00 <sup>4</sup>
Emission Point 19 Paved Road EP19_8 LD 30 15.0 Y 1000-1100 20 0.00% 0.00	J% 0.00% 100.00
Emission Point 19 Favea Road EP19_6 LD 30 15.0 Y 1100-1200 20 0.00% 0.00	0% 0.00% 100.00 0% 0.00% 100.00
Emission Point 19 Paved Road EP19_8 LD 30 15.0 Y 1300-1400 20 0.00% 0.00	0% 0.00% 100.00
Emission Point 19 Paved Road EP19_8 LD 30 15.0 Y 1400-1500 20 0.00% 0.00	J% 0.00% 100.00 <sup>4</sup>
Emission Point 19 Paved Road EP19_8 LD 30 15.0 Y 1500-1500 20 0.00% 0.00	J% 0.00% 100.00'
Emission Fourity Paved Road EF19_8 LD 30 15.0 Y 1700-1800 20 0.00%	0% 0.00% 100.00
Emission Point 19 Paved Road EP19_8 LD 30 15.0 Y 1800-1900 20 0.00% 0.00	0.00% 100.00
Emission Point 19 Paved Road EP19_8 LD 30 15.0 Y 1900-2000 20 0.00% 0.00	J% 0.00% 100.00 <sup>4</sup>
Emission Point 19 Paved Road EP19_8 LD 30 15.0 Y 2000-2100 20 0.00% 0.00	J% 0.00% 100.00 0% 0.00% 100.00
Emission Point 19 Paved Road EP19_8 LD 30 15.0 Y 2200-2300 20 0.00% 0.00	0% 0.00% 100.00
Emission Point 19 Paved Road EP19_8 LD 30 15.0 Y 2300-0000 0 0.00%	% 0.00% 0.00%
Emission Point 19 Paved Road EP19_9_13 LD 30 15.0 Y 0000-0100 0 0.00% 0.	% 0.00% 0.00%
Emission Fouriti 9Franza National EP19 913 LD 30 15.0 Y 0200-300 0 0.00%	% 0.00% 0.00%
Emission Point 19 Paved Road EP19_13 LD 30 150 Y 0300-0400 0 0.00%	% 0.00% 0.00%
Emission Point 19 Paved Road EP19_9_13 LD 30 15.0 Y 0400-0500 0 0.00% 0.	% 0.00% 0.00%
Emission Point 19 Havea Road EP119 9 13 LD 30 15.0 Y 0500-0600 0 0.00%	% 0.00% 0.00% % 0.00% 0.00%
Emission Point 19 Paved Road EP19 913 LD 30 15.0 Y 0700-0800 10 0.00% 0.	0% 0.00% 100.00
Emission Point 19 Paved Road EP19_9_13 LD 30 15.0 Y 0800-0900 10 0.00%	J% 0.00% 100.00 <sup>4</sup>
Emission Point 19 Paved Road EP19_9_13 LD 30 15.0 Y 0900-1000 10 0.00% 0	J% 0.00% 100.00
Emission Found is Faved Naeu EF19_9_13 ED 30 13.0 T 1000-1100 10 0.00%	0.00% 100.00 0% 0.00% 100.00

Road Name	Road ID	Road Type	Speed Limit (km/hr)	Average Speed (km/hr)	Start Emission Estimated by Broad Brush Approach (Y/N)	Hour	Total Vehicles (Veh/hr)	01 - Private Cars	02 - Taxi	03 - Light Goods Vehicles< =2.5t	04 - Light Goods Vehicles 2.5-3.5t	05 - Light Goods Vehicles> 3.5t	06 - Medium Goods Vehicles< =15t	07 - Medium Goods Vehicles1 5-24t	08 - Public Light Buses	09 - Private Light Bus <=3.5t	10 - Private Light Bus >3.5t	11 - Non- franchise d Bus<=6.4 t	12 - Non- franchise d Bus 6.4- 15t	13 - Non- franchise d Bus 15- 24t	14 - Franchise d Bus Single Deck	15 - Franchise d Bus Double Deck	16 - Motorcycl es	17 - Heavy Goods Vehicles> 24t	18 - Non- franchise d Bus >24t	Total
Environment 40 Deced	5040.0.40		00	45.0	V	4000 4000	40	PC		LGV3	LGV4	LGV6	HGV7	HGV8	PLB	PV4	PV5	0.00%		0.00%	FBSD	FBDD		HGV9	NFB9	400.00%
Emission Point 19 Paved Road	EP19_9_13		30	15.0	ř	1200-1300	10	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	100.00%	0.00%	100.00%
Emission Point 10 Paved Road	EP19_9_13 EP10_0_13		30	15.0	v	1400 1500	10	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	100.00%	0.00%	100.00%
Emission Point 10 Paved Road	EP10_0_12		20	15.0	v	1500 1600	10	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	100.00%	0.00%	100.00%
Emission Point 19 Paved Road	EP19_9_13		30	15.0	T V	1600-1600	10	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	100.00%	0.00%	100.00%
Emission Point 10 Paved Road	EP19_9_13 EP10_0_13		30	15.0	v	1700 1900	10	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	100.00%	0.00%	100.00%
Emission Point 10 Paved Road	EP10_0_12		20	15.0	v	1900-1000	10	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	100.00%	0.00%	100.00%
Emission Point 10 Paved Road	EP10_0_12		20	15.0	v	1000-1900	10	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	100.00%	0.00%	100.00%
Emission Point 19 Paved Road	EP10 0 13		30	15.0	v	2000-2100	10	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	100.00%	0.00%	100.00%
Emission Point 19 Paved Road	EP10 0 13		30	15.0	v	2100-2200	10	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	100.00%	0.00%	100.00%
Emission Point 19 Paved Road	EP10 0 13		30	15.0	v	2200-2200	10	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	100.00%	0.00%	100.00%
Emission Point 19 Paved Road	EP19 9 13		30	15.0	Ý	2300-0000	0	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%
Emission Point 19 Paved Road	EP19 14	LD	30	15.0	Ŷ	0000-0100	0	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%
Emission Point 19 Paved Road	EP19_14	LD	30	15.0	Ŷ	0100-0200	0	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%
Emission Point 19 Paved Road	EP19_14	I D	30	15.0	Ý	0200-0300	õ	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%
Emission Point 19 Paved Road	FP19_14	ID.	30	15.0	Ŷ	0300-0400	0	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%
Emission Point 19 Paved Road	FP19_14	ID.	30	15.0	Ŷ	0400-0500	0	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%
Emission Point 19 Paved Road	EP19_14	LD	30	15.0	Ý	0500-0600	ō	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%
Emission Point 19 Paved Road	EP19 14	LD	30	15.0	Y	0600-0700	0	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%
Emission Point 19 Paved Road	EP19 14	LD	30	15.0	Y	0700-0800	48	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	100.00%	0.00%	100.00%
Emission Point 19 Paved Road	EP19 14	LD	30	15.0	Y	0800-0900	48	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	100.00%	0.00%	100.00%
Emission Point 19 Paved Road	EP19 14	LD	30	15.0	Y	0900-1000	48	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	100.00%	0.00%	100.00%
Emission Point 19 Paved Road	EP19 14	LD	30	15.0	Y	1000-1100	48	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	100.00%	0.00%	100.00%
Emission Point 19 Paved Road	EP19 14	LD	30	15.0	Y	1100-1200	48	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	100.00%	0.00%	100.00%
Emission Point 19 Paved Road	EP19 14	LD	30	15.0	Y	1200-1300	48	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	100.00%	0.00%	100.00%
Emission Point 19 Paved Road	EP19 14	LD	30	15.0	Y	1300-1400	48	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	100.00%	0.00%	100.00%
Emission Point 19 Paved Road	EP19 14	LD	30	15.0	Y	1400-1500	48	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	100.00%	0.00%	100.00%
Emission Point 19 Paved Road	EP19_14	LD	30	15.0	Y	1500-1600	48	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	100.00%	0.00%	100.00%
Emission Point 19 Paved Road	EP19_14	LD	30	15.0	Y	1600-1700	48	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	100.00%	0.00%	100.00%
Emission Point 19 Paved Road	EP19_14	LD	30	15.0	Y	1700-1800	48	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	100.00%	0.00%	100.00%
Emission Point 19 Paved Road	EP19_14	LD	30	15.0	Y	1800-1900	48	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	100.00%	0.00%	100.00%
Emission Point 19 Paved Road	EP19_14	LD	30	15.0	Y	1900-2000	48	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	100.00%	0.00%	100.00%
Emission Point 19 Paved Road	EP19_14	LD	30	15.0	Y	2000-2100	48	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	100.00%	0.00%	100.00%
Emission Point 19 Paved Road	EP19_14	LD	30	15.0	Y	2100-2200	48	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	100.00%	0.00%	100.00%
Emission Point 19 Paved Road	EP19_14	LD	30	15.0	Y	2200-2300	48	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	100.00%	0.00%	100.00%
Emission Point 19 Paved Road	EP19_14	LD	30	15.0	Y	2300-0000	0	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%

Road Name	Road ID	Road Type	Speed Limit (km/hr)	Average Speed (km/hr)	Start Emission Estimated by Broad Brush Approach (Y/N)	Hour	Total Vehicles (Veh/hr)	01 - Private Cars	02 - Taxi	03 - Light Goods Vehicles< =2.5t	04 - Light Goods Vehicles 2.5-3.5t	05 - Light Goods Vehicles> 3.5t	06 - Medium Goods Vehicles< =15t	07 - Medium Goods Vehicles1 5-24t	08 - Public Light Buses	09 - Private Light Bus <=3.5t	10 - Private Light Bus >3.5t	11 - Non- franchise d Bus<=6.4 t	12 - Non- franchise d Bus 6.4 15t	13 - Non- franchise d Bus 15- 24t	14 - Franchise d Bus Single Deck	15 - Franchise d Bus Double Deck	16 - Motorcycl es	17 - Heavy Goods Vehicles> 24t	18 - Non- franchise d Bus >24t	Total
Chun Yiu Road	RD_1_2	RR	30	26.0	Y	0000-0100	10	100.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	100.00%
Chun Yiu Road	RD_1_2	RR	30	26.0	Y	0100-0200	10	100.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	100.00%
Chun Yiu Road	RD_1_2 RD_1_2	RR	30	26.0	Ŷ	0200-0300	5	100.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	100.00%
Chun Yiu Road	RD_1_2	RR	30	26.0	Ý	0400-0500	5	50.00%	0.00%	0.00%	0.00%	50.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	100.00%
Chun Yiu Road	RD_1_2	RR	30	26.0	Y	0500-0600	10	100.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	100.00%
Chun Yiu Road Chun Yiu Road	RD_1_2 RD_1_2	RR	30	26.0	Y Y	0600-0700	10 25	77.50% 69.00%	0.00%	0.00%	0.00%	22.50% 14.02%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00% 4.00%	0.00%	100.00%
Chun Yiu Road	RD_1_2	RR	30	26.0	Ý	0800-0900	30	61.85%	0.00%	0.00%	8.40%	20.65%	0.00%	4.55%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	4.55%	0.00%	100.00%
Chun Yiu Road	RD_1_2	RR	30	26.0	Y	0900-1000	30	60.60%	0.00%	0.00%	8.70%	21.60%	0.00%	4.55%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	4.55%	0.00%	100.00%
Chun Yiu Road Chun Yiu Road	RD_1_2 RD_1_2	RR	30	26.0	Y Y	1000-1100	25	60.02% 51.90%	0.00%	0.00%	8.98%	23.00%	0.00%	4.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	4.00%	0.00%	100.00%
Chun Yiu Road	RD_1_2	RR	30	26.0	Ŷ	1200-1300	55	27.40%	0.00%	0.00%	20.13%	36.80%	0.00%	7.27%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	8.40%	0.00%	100.00%
Chun Yiu Road	RD_1_2	RR	30	26.0	Y	1300-1400	60	26.30%	0.00%	0.00%	19.20%	33.45%	0.00%	10.70%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	10.35%	0.00%	100.00%
Chun Yiu Road Chun Yiu Road	RD_1_2 RD_1_2	RR	30	26.0	Y Y	1400-1500	60 60	29.05%	0.00%	0.00%	18.05% 21.70%	34.70% 34.70%	0.00%	9.25%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	8.95% 7.15%	0.00%	100.00%
Chun Yiu Road	RD_1_2	RR	30	26.0	Ŷ	1600-1700	55	33.33%	0.00%	0.00%	17.58%	35.37%	0.00%	3.95%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	9.76%	0.00%	100.00%
Chun Yiu Road	RD_1_2	RR	30	26.0	Y	1700-1800	55	37.25%	0.00%	0.00%	23.51%	27.49%	0.00%	3.95%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	7.80%	0.00%	100.00%
Chun Yiu Road Chun Yiu Road	RD_1_2 RD_1_2	RR	30 30	26.0 26.0	Y Y	1800-1900	50 45	52.25% 62.43%	0.00%	0.00%	18.00%	23.00%	0.00%	2.40%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	4.35%	0.00%	100.00%
Chun Yiu Road	RD_1_2	RR	30	26.0	Ŷ	2000-2100	25	61.72%	0.00%	0.00%	23.88%	9.42%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	4.98%	0.00%	100.00%
Chun Yiu Road	RD_1_2	RR	30	26.0	Y	2100-2200	20	56.25%	0.00%	0.00%	21.25%	22.50%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	100.00%
Chun Yiu Road Chun Yiu Road	RD_1_2 RD_1_2	RR	30	26.0	Y Y	2200-2300	15 15	53.33% 53.33%	0.00%	0.00%	28.87%	17.80% 28.87%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	100.00%
Chun Yiu Road	RD_3_4	RR	30	26.0	Ŷ	0000-0100	10	80.00%	10.00%	0.00%	0.00%	10.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	100.00%
Chun Yiu Road	RD_3_4	RR	30	26.0	Y	0100-0200	10	66.70%	16.65%	0.00%	0.00%	16.65%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	100.00%
Chun Yiu Road	RD_3_4	RR	30	26.0	Y	0200-0300	10	75.00%	0.00%	0.00%	0.00%	25.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	100.00%
Chun Yiu Road	RD_3_4	RR	30	26.0	Ý	0400-0500	10	75.00%	0.00%	0.00%	0.00%	25.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	100.00%
Chun Yiu Road	RD_3_4	RR	30	26.0	Y	0500-0600	10	83.35%	0.00%	0.00%	0.00%	16.65%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	100.00%
Chun Yiu Road Chun Yiu Road	RD_3_4 RD_3_4	RR	30	26.0	Y Y	0600-0700	15 35	80.00% 79.57%	6.67% 3.43%	0.00%	0.00%	13.33%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	100.00%
Chun Yiu Road	RD_3_4	RR	30	26.0	Ŷ	0800-0900	35	72.07%	3.07%	0.00%	6.21%	18.64%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	100.00%
Chun Yiu Road	RD_3_4	RR	30	26.0	Y	0900-1000	35	70.79%	3.21%	0.00%	6.50%	19.50%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	100.00%
Chun Yiu Road Chun Yiu Road	RD_3_4 RD_3_4	RR	30 30	26.0 26.0	Y Y	1000-1100	35	69.36% 64.29%	3.43%	0.00%	6.50%	20.43%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	100.00%
Chun Yiu Road	RD_3_4	RR	30	26.0	Ŷ	1200-1300	55	28.09%	0.00%	0.00%	16.29%	40.60%	0.00%	7.51%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	7.51%	0.00%	100.00%
Chun Yiu Road	RD_3_4	RR	30	26.0	Y	1300-1400	65	26.45%	0.00%	0.00%	14.88%	38.45%	0.00%	10.11%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	10.11%	0.00%	100.00%
Chun Yiu Road Chun Yiu Road	RD_3_4 RD_3_4	RR	30 30	26.0	Y	1500-1600	60 60	29.20%	0.00%	0.00%	14.67%	38.13%	0.00%	9.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	9.00%	0.00%	100.00%
Chun Yiu Road	RD_3_4	RR	30	26.0	Ŷ	1600-1700	55	34.10%	0.00%	0.00%	15.16%	37.63%	0.00%	3.75%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	9.35%	0.00%	100.00%
Chun Yiu Road	RD_3_4	RR	30	26.0	Y	1700-1800	55	38.10%	0.00%	0.00%	18.86%	31.07%	0.00%	4.01%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	7.95%	0.00%	100.00%
Chun Yiu Road Chun Yiu Road	RD_3_4 RD_3_4	RR	30 30	26.0	Ý	1900-1900	50 45	51.94% 63.53%	0.00%	0.00%	15.10%	26.30%	0.00%	2.22%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	4.44% 2.39%	0.00%	100.00%
Chun Yiu Road	RD_3_4	RR	30	26.0	Y	2000-2100	25	62.98%	0.00%	0.00%	18.02%	14.02%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	4.98%	0.00%	100.00%
Chun Yiu Road	RD_3_4	RR	30	26.0	Y	2100-2200	20	59.05%	0.00%	0.00%	18.05%	22.90%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	100.00%
Chun Yiu Road	RD_3_4 RD_3_4	RR	30	26.0	Ý	2300-2300	15	48.53%	0.00%	0.00%	16.20%	35.27%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	100.00%
Chun Yiu Road	RD_5_6	RR	30	26.0	Y	0000-0100	10	80.00%	10.00%	0.00%	0.00%	10.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	100.00%
Chun Yiu Road	RD_5_6	RR	30	26.0	Y	0100-0200	10	66.70%	16.65%	0.00%	0.00%	16.65%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	100.00%
Chun Yiu Road	RD_5_6	RR	30	26.0	Ý	0300-0400	5	33.40%	33.30%	0.00%	0.00%	33.30%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	100.00%
Chun Yiu Road	RD_5_6	RR	30	26.0	Y	0400-0500	10	75.00%	0.00%	0.00%	0.00%	25.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	100.00%
Chun Yiu Road Chun Yiu Road	RD_5_6 RD_5_6	RR	30 30	26.0 26.0	Y	0500-0600	10 15	83.35% 80.00%	0.00%	0.00%	0.00%	16.65% 13.33%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	100.00%
Chun Yiu Road	RD_5_6	RR	30	26.0	Ŷ	0700-0800	75	30.58%	1.38%	0.00%	2.82%	4.20%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	61.02%	0.00%	100.00%
Chun Yiu Road	RD_5_6	RR	30	26.0	Y	0800-0900	75	28.30%	1.32%	0.00%	2.64%	7.98%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	59.76%	0.00%	100.00%
Chun Yiu Road Chun Yiu Road	RD_5_6 RD_5_6	RR	30	26.0	Y Y	1000-1000	105	19.09% 17.43%	0.97%	0.00%	1.89%	5.71%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	72.34%	0.00%	100.00%
Chun Yiu Road	RD_5_6	RR	30	26.0	Ŷ	1100-1200	105	16.29%	0.97%	0.00%	1.89%	7.60%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	73.26%	0.00%	100.00%
Chun Yiu Road	RD_5_6	RR	30	26.0	Y	1200-1300	130	10.97%	0.00%	0.00%	6.27%	16.66%	0.00%	3.23%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	62.87%	0.00%	100.00%
Chun Yiu Road Chun Yiu Road	RD_5_6 RD_5_6	RR	30	26.0	Y Y	1300-1400	140	11.77%	0.00%	0.00%	6.69%	16.79% 15.79%	0.00%	4.34%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	60.41% 62.01%	0.00%	100.00%
Chun Yiu Road	RD_5_6	RR	30	26.0	Ŷ	1500-1600	135	12.22%	0.00%	0.00%	7.67%	15.91%	0.00%	3.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	61.20%	0.00%	100.00%
Chun Yiu Road	RD_5_6	RR	30	26.0	Y	1600-1700	135	13.97%	0.00%	0.00%	6.24%	15.47%	0.00%	1.56%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	62.77%	0.00%	100.00%
Chun Yiu Road Chun Yiu Road	RD_5_6 RD_5_6	RR	30 30	26.0 26.0	Y Y	1700-1800	130	15.64% 19.75%	0.00%	0.00%	7.82%	12.39%	0.00%	1.56%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	62.58% 64.00%	0.00%	100.00%
Chun Yiu Road	RD_5_6	RR	30	26.0	Ŷ	1900-2000	105	25.74%	0.00%	0.00%	4.99%	7.92%	0.00%	1.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	60.36%	0.00%	100.00%
Chun Yiu Road	RD_5_6	RR	30	26.0	Y	2000-2100	85	17.08%	0.00%	0.00%	4.89%	3.72%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	74.31%	0.00%	100.00%
Chun Yiu Road Chun Yiu Road	RD 5_6	RR	30 30	26.0 26.0	Y Y	2100-2200	80 75	13.00% 8.44%	0.00%	0.00%	3.95% 2.85%	5.15% 4.34%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	77.90% 84,37%	0.00%	100.00%
Chun Yiu Road	RD_5_6	RR	30	26.0	Ŷ	2300-0000	15	48.53%	0.00%	0.00%	16.20%	35.27%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	100.00%
Chun Yiu Road	RD_7_8	RR	30	26.0	Y	0000-0100	15	80.93%	9.53%	0.00%	0.00%	9.53%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	100.00%
	RD_1_8	ĸĸ	30	20.0	Y	0100-0200	10	15.00%	12.50%	0.00%	0.00%	12.50%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	100.00%

Road Name	Road ID	Road Type	Speed Limit (km/hr)	Average Speed (km/hr)	Start Emission Estimated by Broad Brush Approach (Y/N)	Hour	Total Vehicles (Veh/hr)	01 - Private Cars	02 - Taxi	03 - Light Goods Vehicles< =2.5t	04 - Light Goods Vehicles 2.5-3.5t	05 - Light Goods Vehicles> 3.5t	06 - Medium Goods Vehicles< =15t	07 - Medium Goods Vehicles1 5-24t	08 - Public Light Buses	09 - Private Light Bus <=3.5t	10 - Private Light Bus >3.5t	11 - Non- franchise d Bus<=6.4 t	12 - Non- franchise d Bus 6.4 15t	13 - Non- franchise d Bus 15- 24t	14 - Franchise d Bus Single Deck	15 - Franchise d Bus Double Deck	16 - Motorcycl es	17 - Heavy Goods Vehicles> 24t	18 - Non- franchise d Bus >24t	Total
Chun Yiu Road	RD 7 8	RR	30	26.0	Y	0200-0300	10	83.35%	0.00%	0.00%	0.00%	16.65%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	100.00%
Chun Yiu Road	RD_7_8	RR	30	26.0	Y	0300-0400	5	25.00%	25.00%	0.00%	0.00%	50.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	100.00%
Chun Yiu Road Chun Yiu Road	RD_7_8 RD_7_8	RR	30 30	26.0 26.0	Y Y	0400-0500	10 10	75.00% 87.50%	0.00%	0.00%	0.00%	25.00% 12.50%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	100.00%
Chun Yiu Road	RD_7_8	RR	30	26.0	Ŷ	0600-0700	25	78.26%	4.02%	0.00%	4.02%	13.70%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	100.00%
Chun Yiu Road	RD_7_8	RR	30	26.0	Y	0700-0800	95	39.73%	1.10%	0.00%	3.24%	6.48%	0.00%	1.09%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	48.36%	0.00%	100.00%
Chun Yiu Road	RD_7_8	RR	30	26.0	Y	0900-1000	130	25.48%	0.81%	0.00%	3.23%	8.78%	0.00%	0.80%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	40.30% 60.91%	0.00%	100.00%
Chun Yiu Road	RD_7_8	RR	30	26.0	Y	1000-1100	130	24.67%	0.81%	0.00%	3.23%	9.58%	0.00%	0.80%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	60.91%	0.00%	100.00%
Chun Yiu Road Chun Yiu Road	RD_7_8 RD_7_8	RR	30 30	26.0 26.0	Y	1100-1200	130 190	22.25% 15.51%	0.81%	0.00%	3.23%	12.00% 23.56%	0.00%	0.80%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	60.91% 44.88%	0.00%	100.00%
Chun Yiu Road	RD_7_8	RR	30	26.0	Ŷ	1300-1400	205	16.40%	0.00%	0.00%	11.92%	21.89%	0.00%	5.97%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	43.83%	0.00%	100.00%
Chun Yiu Road	RD_7_8	RR	30	26.0	Y	1400-1500	200	16.94%	0.00%	0.00%	11.29%	22.04%	0.00%	5.15%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	44.59%	0.00%	100.00%
Chun Yiu Road Chun Yiu Road	RD_7_8 RD 7 8	RR	30	26.0	Ý	1600-1600	200	19.21%	0.00%	0.00%	12.39%	22.70%	0.00%	4.12% 2.18%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	43.81% 45.98%	0.00%	100.00%
Chun Yiu Road	RD_7_8	RR	30	26.0	Y	1700-1800	190	22.09%	0.00%	0.00%	14.00%	16.67%	0.00%	2.14%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	45.10%	0.00%	100.00%
Chun Yiu Road Chun Yiu Road	RD_7_8 RD_7_8	RR	30 30	26.0 26.0	Y	1800-1900	175 150	28.26% 36.25%	0.00%	0.00%	10.00% 8.92%	13.48% 10.96%	0.00%	1.15%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	47.10% 42.53%	0.00%	100.00%
Chun Yiu Road	RD_7_8	RR	30	26.0	Ý	2000-2100	110	26.70%	0.00%	0.00%	9.55%	4.75%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	59.00%	0.00%	100.00%
Chun Yiu Road	RD_7_8	RR	30	26.0	Y	2100-2200	105	21.22%	0.00%	0.00%	8.04%	9.09%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	61.66%	0.00%	100.00%
Chun Yiu Road Chun Yiu Road	RD_7_8 RD_7_8	RR	30 30	26.0 26.0	Y	2200-2300	90 30	15.25% 56.05%	0.00%	0.00%	7.05%	5.85% 27.90%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	71.85%	0.00%	100.00%
Kam Tin Road	RD_09_10	RR	50	39.9	Ŷ	0000-0100	200	73.65%	11.65%	0.00%	4.60%	5.55%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	2.00%	2.55%	0.00%	0.00%	100.00%
Kam Tin Road	RD_09_10	RR	50	40.0	Y	0100-0200	120	64.10%	15.75%	0.00%	12.25%	4.40%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	1.75%	1.75%	0.00%	0.00%	100.00%
Kam Tin Road	RD 09 10	RR	50	40.0	Ý	0200-0300	85	50.64%	20.82%	0.00%	10.41%	16.86%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	1.27%	0.00%	0.00%	100.00%
Kam Tin Road	RD_09_10	RR	50	40.0	Y	0400-0500	105	58.94%	13.94%	0.00%	13.94%	11.08%	1.05%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	1.05%	0.00%	0.00%	100.00%
Kam Tin Road Kam Tin Road	RD_09_10 RD_09_10	RR	50 50	40.0 37.4	Y	0500-0600	175 440	59.77% 68.40%	6.41% 5.92%	0.00%	7.61%	4.71%	4.09%	1.71%	3.55%	0.00%	0.00%	0.58%	5.83%	1.15%	1.71%	1.15%	1.73%	0.00%	0.00%	100.00%
Kam Tin Road	RD_09_10	RR	50	31.4	Ý	0700-0800	1005	63.17%	3.31%	0.00%	10.22%	3.68%	0.89%	0.80%	1.50%	0.00%	0.00%	0.00%	3.01%	0.40%	0.70%	0.80%	5.52%	6.00%	0.00%	100.00%
Kam Tin Road	RD_09_10	RR	50	31.5	Y	0800-0900	995	61.07%	3.06%	0.00%	12.92%	6.47%	0.99%	1.10%	1.75%	0.00%	0.00%	0.30%	0.50%	0.20%	0.50%	0.60%	4.04%	6.50%	0.00%	100.00%
Kam Tin Road	RD_09_10 RD_09_10	RR	50 50	31.5	Ý	1000-1000	995 935	56.77%	4.11%	0.00%	14.28%	6.41% 7.22%	1.84%	1.61%	1.40%	0.00%	0.00%	0.20%	0.00%	0.20%	0.20%	0.80%	1.74%	9.64% 10.50%	0.00%	100.00%
Kam Tin Road	RD_09_10	RR	50	32.5	Y	1100-1200	905	54.08%	3.31%	0.00%	14.57%	9.21%	1.44%	1.20%	1.10%	0.00%	0.00%	0.30%	0.00%	0.00%	0.40%	0.70%	2.14%	11.55%	0.00%	100.00%
Kam Tin Road	RD_09_10	RR	50	31.3	Y	1200-1300	1020	32.51%	5.61%	0.00%	17.94%	14.60%	5.92%	1.75%	0.95%	0.00%	0.00%	0.00%	0.10%	0.40%	0.30%	0.50%	2.77%	16.64%	0.00%	100.00%
Kam Tin Road	RD_09_10	RR	50	30.1	Ý	1400-1500	1135	33.12%	4.86%	0.00%	15.93%	12.60%	9.02%	1.60%	1.05%	0.00%	0.00%	0.00%	0.15%	0.35%	0.25%	0.45%	2.46%	18.15%	0.00%	100.00%
Kam Tin Road	RD_09_10	RR	50	30.3	Y	1500-1600	1110	33.82%	4.36%	0.00%	18.04%	13.41%	7.86%	1.45%	0.91%	0.00%	0.00%	0.00%	0.10%	0.40%	0.30%	0.45%	2.36%	16.54%	0.00%	100.00%
Kam Tin Road Kam Tin Road	RD_09_10 RD_09_10	RR	50 50	30.3 29.4	Y	1600-1700	1110 1200	36.32% 37.53%	4.36% 4.16%	0.00%	15.14% 17.30%	12.20% 8.65%	6.52% 8.04%	0.55%	1.05%	0.00%	0.00%	0.00%	0.00%	0.20%	0.45%	0.45%	4.53%	18.24% 14.61%	0.00%	100.00%
Kam Tin Road	RD_09_10	RR	50	31.1	Ŷ	1800-1900	1040	50.22%	4.81%	0.00%	13.49%	7.64%	3.45%	0.20%	1.61%	0.00%	0.00%	0.00%	0.11%	0.20%	0.31%	0.45%	5.30%	12.21%	0.00%	100.00%
Kam Tin Road	RD_09_10	RR	50	32.0	Y	1900-2000	950	60.49%	3.71%	0.00%	11.18%	5.74%	2.22%	0.45%	1.51%	0.00%	0.00%	0.00%	0.11%	0.20%	0.31%	0.55%	4.65%	8.88%	0.00%	100.00%
Kam Tin Road	RD_09_10 RD_09_10	RR	50 50	35.6	Ý	2000-2100 2100-2200	495	51.82% 48.35%	6.71%	0.00%	13.39%	2.50% 5.49%	1.48%	0.35%	1.85%	0.00%	0.00%	0.00%	0.00%	0.00%	1.22%	0.65%	5.11%	14.74%	0.00%	100.00%
Kam Tin Road	RD_09_10	RR	50	38.1	Y	2200-2300	370	41.54%	10.56%	0.00%	13.00%	4.60%	5.69%	0.00%	0.81%	0.00%	0.00%	0.00%	0.00%	0.00%	0.81%	1.31%	3.27%	18.41%	0.00%	100.00%
Kam Tin Road	RD_09_10 RD_11_12	RR	50 50	38.8	Y	2300-0000	305	52.61%	15.25%	0.00%	10.26%	7.68%	4.02%	0.00%	0.32%	0.00%	0.00%	0.00%	0.32%	0.65%	2.00%	0.97%	4.95%	0.97%	0.00%	100.00%
Kam Tin Road	RD_11_12	RR	50	40.0	Ý	0100-0200	115	64.26%	15.19%	0.00%	12.51%	4.45%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	1.80%	1.80%	0.00%	0.00%	100.00%
Kam Tin Road	RD_11_12	RR	50	40.0	Y	0200-0300	100	67.40%	14.75%	0.00%	7.35%	9.45%	1.05%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	100.00%
Kam Tin Road	RD_11_12 RD_11_12	RR	50 50	40.0	Ý	0300-0400	100	52.05% 59.00%	20.05%	0.00%	10.65%	15.95%	1.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	1.30%	0.00%	0.00%	100.00%
Kam Tin Road	RD_11_12	RR	50	40.0	Y	0500-0600	175	59.49%	5.87%	0.00%	8.21%	4.67%	4.07%	1.79%	3.53%	0.00%	0.00%	0.59%	5.87%	1.19%	1.79%	1.19%	1.74%	0.00%	0.00%	100.00%
Kam Tin Road Kam Tin Road	RD_11_12 RD_11_12	RR	50 50	37.4	Y	0600-0700	435	68.37% 62.74%	5.78%	0.00%	6.71% 10.28%	3.68%	1.62%	0.45%	3.04%	0.00%	0.00%	0.24%	0.00%	0.45%	0.71%	1.35%	5.75%	1.85% 6.14%	0.00%	100.00%
Kam Tin Road	RD_11_12	RR	50	31.6	Ý	0800-0900	985	60.61%	2.91%	0.00%	12.98%	6.61%	1.03%	1.35%	1.74%	0.00%	0.00%	0.30%	0.50%	0.20%	0.50%	0.60%	4.06%	6.59%	0.00%	100.00%
Kam Tin Road	RD_11_12	RR	50	32.0	Y	0900-1000	950	57.44%	2.86%	0.00%	12.73%	6.52%	1.04%	1.35%	1.79%	0.00%	0.00%	0.30%	0.50%	0.20%	0.50%	0.65%	3.92%	10.19%	0.00%	100.00%
Kam Tin Road Kam Tin Road	RD_11_12 RD_11_12	RR	50 50	32.2	Y Y	1000-1100	935 905	56.51% 53.61%	4.12%	0.00%	13.19% 14.55%	7.40% 9.39%	1.82%	1.65%	1.64%	0.00%	0.00%	0.10%	0.00%	0.20%	0.50%	0.65%	1.59%	10.63%	0.00%	100.00%
Kam Tin Road	RD_11_12	RR	50	31.3	Ŷ	1200-1300	1020	32.29%	5.61%	0.00%	18.16%	14.33%	5.95%	1.80%	0.96%	0.00%	0.00%	0.00%	0.10%	0.40%	0.30%	0.50%	2.77%	16.82%	0.00%	100.00%
Kam Tin Road	RD_11_12	RR	50	29.6	Y	1300-1400	1175	31.16%	5.36%	0.00%	16.56%	12.68%	8.44%	2.05%	0.95%	0.00%	0.00%	0.00%	0.00%	0.20%	0.45%	0.45%	2.47%	19.23%	0.00%	100.00%
Kam Tin Road	RD 11 12	RR	50	30.1	Y	1500-1600	1105	33.91%	4.86%	0.00%	18.26%	12.43%	8.99% 7.89%	1.45%	0.90%	0.00%	0.00%	0.00%	0.10%	0.35%	0.25%	0.45%	2.47%	16.23%	0.00%	100.00%
Kam Tin Road	RD_11_12	RR	50	30.4	Y	1600-1700	1105	36.20%	4.36%	0.00%	15.25%	12.08%	6.54%	0.55%	1.10%	0.00%	0.00%	0.00%	0.00%	0.20%	0.45%	0.45%	4.54%	18.26%	0.00%	100.00%
Kam Tin Road Kam Tin Road	RD_11_12 RD_11_12	RR	50 50	31.9	Y	1700-1800	955 1040	58.37% 50.13%	3.65%	0.00%	11.24%	5.56% 7.66%	2.22%	0.45%	1.51%	0.00%	0.52%	0.23%	0.11%	0.20%	0.31%	0.55%	4.60%	10.49%	0.00%	100.00%
Kam Tin Road	RD_11_12	RR	50	32.1	Ý	1900-2000	945	58.47%	3.80%	0.00%	11.81%	5.98%	2.23%	0.44%	1.46%	0.00%	0.53%	0.22%	0.11%	0.20%	0.31%	0.56%	4.91%	8.97%	0.00%	100.00%
Kam Tin Road	RD_11_12	RR	50	35.6	Y	2000-2100	610	51.71%	7.56%	0.00%	13.55%	2.50%	1.48%	0.35%	1.85%	0.00%	0.00%	0.00%	0.00%	0.00%	0.51%	0.65%	5.15%	14.70%	0.00%	100.00%
Kam Tin Road Kam Tin Road	RD 11_12 RD 11_12	RR	50 50	36.7 38.1	Y Y	2100-2200 2200-2300	500 375	48.33% 41.25%	6.65% 10.61%	0.00%	13.74% 13.05%	5.46% 4.65%	1.82% 5.72%	0.00%	1.41% 0.81%	0.00%	0.00%	0.00%	0.00%	0.40% 0.00%	1.22% 0.81%	0.79% 1.36%	5.83% 3.27%	14.34% 18.47%	0.00%	100.00%
Kam Tin Road	RD_11_12	RR	50	38.8	Ŷ	2300-0000	305	52.93%	15.35%	0.00%	10.35%	7.40%	4.02%	0.00%	0.32%	0.00%	0.00%	0.00%	0.32%	0.65%	2.04%	0.97%	5.00%	0.65%	0.00%	100.00%
Kam Tin Road	RD_13_14	RR	50	39.9	Y	0000-0100	200	73.65%	11.65%	0.00%	4.60%	5.55%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	2.00%	2.55%	0.00%	0.00%	100.00%
Kam Tin Road	RD 13_14	RR	50 50	40.0	ř Y	0200-0200	120	67.45%	14.80%	0.00%	7.40%	4.40% 9.35%	1.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	100.00%
Kam Tin Road	RD_13_14	RR	50	40.0	Y	0300-0400	85	50.64%	20.82%	0.00%	10.41%	16.86%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	1.27%	0.00%	0.00%	100.00%

Road Name	Road ID	Road Type	Speed Limit (km/hr)	Average Speed (km/hr)	Start Emission Estimated by Broad Brush Approach (Y/N)	Hour	Total Vehicles (Veh/hr)	01 - Private Cars	02 - Taxi	03 - Light Goods Vehicles< =2.5t	04 - Light Goods Vehicles 2.5-3.5t	05 - Light Goods Vehicles> 3.5t	06 - Medium Goods Vehicles< =15t	07 - Medium Goods Vehicles1 5-24t	08 - Public Light Buses	09 - Private Light Bus <=3.5t	10 - Private Light Bus >3.5t	11 - Non- franchise d Bus<=6.4 t	12 - Non- franchise d Bus 6.4 15t	13 - Non- franchise d Bus 15- 24t	14 - Franchise d Bus Single Deck	15 - Franchise d Bus Double Deck	16 - Motorcycl es	17 - Heavy Goods Vehicles> 24t	18 - Non- franchise d Bus >24t	Total
Kam Tin Road	RD 13 14	RR	50	40.0	Y	0400-0500	105	58.94%	13.94%	0.00%	13.94%	11.08%	1.05%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	1.05%	0.00%	0.00%	100.00%
Kam Tin Road	RD_13_14	RR	50	40.0	Ŷ	0500-0600	175	59.77%	6.41%	0.00%	7.61%	4.71%	4.09%	1.71%	3.55%	0.00%	0.00%	0.58%	5.83%	1.15%	1.71%	1.15%	1.73%	0.00%	0.00%	100.00%
Kam Tin Road	RD_13_14	RR	50 50	37.4	Y	0600-0700	440	68.40%	5.92%	0.00%	6.66%	3.68%	1.58%	0.45%	2.99%	0.00%	0.00%	0.24%	0.00%	0.45%	0.70%	1.35%	5.72%	1.85%	0.00%	100.00%
Kam Tin Road	RD_13_14	RR	50	31.5	Ý	0800-0900	995	61.07%	3.06%	0.00%	12.92%	6.47%	0.99%	1.10%	1.75%	0.00%	0.00%	0.30%	0.50%	0.20%	0.50%	0.60%	4.04%	6.50%	0.00%	100.00%
Kam Tin Road	RD_13_14	RR	50	31.5	Y	0900-1000	995	56.77%	4.11%	0.00%	14.28%	6.41%	1.63%	1.61%	1.40%	0.00%	0.00%	0.20%	1.01%	0.20%	0.20%	0.80%	1.74%	9.64%	0.00%	100.00%
Kam Tin Road	RD_13_14 RD_13_14	RR	50 50	32.2	Ý	1100-1100	935 905	57.13% 54.08%	4.16%	0.00%	13.17%	7.22% 9.21%	1.84%	1.31%	1.60%	0.00%	0.00%	0.10%	0.00%	0.20%	0.50%	0.65%	2.14%	10.50%	0.00%	100.00%
Kam Tin Road	RD_13_14	RR	50	31.3	Y	1200-1300	1020	32.51%	5.61%	0.00%	17.94%	14.60%	5.92%	1.75%	0.95%	0.00%	0.00%	0.00%	0.10%	0.40%	0.30%	0.50%	2.77%	16.64%	0.00%	100.00%
Kam Tin Road Kam Tin Road	RD_13_14 RD_13_14	RR	50 50	29.6 30.1	Y	1300-1400	1175	31.26%	5.35% 4.86%	0.00%	16.42%	12.80%	8.48% 9.02%	2.05%	0.95%	0.00%	0.00%	0.00%	0.00%	0.20%	0.40%	0.40%	2.46%	19.22% 18.15%	0.00%	100.00%
Kam Tin Road	RD_13_14	RR	50	30.3	Ý	1500-1600	1110	33.82%	4.36%	0.00%	18.04%	13.41%	7.86%	1.45%	0.91%	0.00%	0.00%	0.00%	0.10%	0.40%	0.30%	0.45%	2.36%	16.54%	0.00%	100.00%
Kam Tin Road	RD_13_14	RR	50	30.3	Y	1600-1700	1110	36.32%	4.36%	0.00%	15.14%	12.20%	6.52%	0.55%	1.05%	0.00%	0.00%	0.00%	0.00%	0.20%	0.45%	0.45%	4.53%	18.24%	0.00%	100.00%
Kam Tin Road	RD_13_14 RD_13_14	RR	50 50	29.4 31.1	ř Y	1800-1900	1200	37.53% 50.22%	4.16%	0.00%	17.30%	8.65% 7.64%	8.04% 3.45%	0.50%	1.61%	0.00%	0.00%	0.00%	0.00%	0.30%	0.25%	0.40%	5.30%	14.61%	0.00%	100.00%
Kam Tin Road	RD_13_14	RR	50	32.0	Y	1900-2000	950	60.49%	3.71%	0.00%	11.18%	5.74%	2.22%	0.45%	1.51%	0.00%	0.00%	0.00%	0.11%	0.20%	0.31%	0.55%	4.65%	8.88%	0.00%	100.00%
Kam Tin Road Kam Tin Road	RD_13_14 RD_13_14	RR	50 50	35.6 36.8	Y	2000-2100	610 495	51.82% 48.35%	7.61% 6.71%	0.00%	13.39% 13.74%	2.50% 5.49%	1.48%	0.35%	1.85%	0.00%	0.00%	0.00%	0.00%	0.00%	0.51%	0.65%	5.11% 5.87%	14.74% 14.15%	0.00%	100.00%
Kam Tin Road	RD_13_14	RR	50	38.1	Ŷ	2200-2300	370	41.54%	10.56%	0.00%	13.00%	4.60%	5.69%	0.00%	0.81%	0.00%	0.00%	0.00%	0.00%	0.00%	0.81%	1.31%	3.27%	18.41%	0.00%	100.00%
Kam Tin Road	RD_13_14	RR	50	38.8	Y	2300-0000	305	52.61%	15.25%	0.00%	10.26%	7.68%	4.02%	0.00%	0.32%	0.00%	0.00%	0.00%	0.32%	0.65%	2.00%	0.97%	4.95%	0.97%	0.00%	100.00%
Kam Tin Road Kam Tin Road	RD_15_16 RD_15_16	RR	50 50	39.9 40.0	Ý	0000-0100	200	73.72% 64.26%	11.33%	0.00%	4.67%	5.67% 4.45%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	2.05%	2.57%	0.00%	0.00%	100.00%
Kam Tin Road	RD_15_16	RR	50	40.0	Y	0200-0300	100	67.40%	14.75%	0.00%	7.35%	9.45%	1.05%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	100.00%
Kam Tin Road Kam Tin Road	RD_15_16 RD_15_16	RR	50 50	40.0 40.0	Y	0300-0400	80 100	52.05% 59.00%	20.05%	0.00%	10.65% 14.00%	15.95% 11.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	1.30%	0.00%	0.00%	100.00%
Kam Tin Road	RD_15_16	RR	50	40.0	Ý	0500-0600	175	59.49%	5.87%	0.00%	8.21%	4.67%	4.07%	1.79%	3.53%	0.00%	0.00%	0.59%	5.87%	1.19%	1.79%	1.19%	1.74%	0.00%	0.00%	100.00%
Kam Tin Road	RD_15_16	RR	50	37.4	Y	0600-0700	435	68.37%	5.78%	0.00%	6.71%	3.68%	1.62%	0.45%	3.04%	0.00%	0.00%	0.24%	0.00%	0.45%	0.71%	1.35%	5.75%	1.85%	0.00%	100.00%
Kam Tin Road	RD_15_16 RD_15_16	RR	50 50	31.5	Ý	0700-0800	985	62.74% 60.61%	3.22%	0.00%	10.28%	3.82% 6.61%	0.94%	1.35%	1.54%	0.00%	0.00%	0.00%	3.02% 0.50%	0.40%	0.70%	0.80%	5.50% 4.06%	6.59%	0.00%	100.00%
Kam Tin Road	RD_15_16	RR	50	32.0	Y	0900-1000	950	57.44%	2.86%	0.00%	12.73%	6.52%	1.04%	1.35%	1.79%	0.00%	0.00%	0.30%	0.50%	0.20%	0.50%	0.65%	3.92%	10.19%	0.00%	100.00%
Kam Tin Road	RD_15_16	RR	50	32.2	Y	1000-1100	935	56.51%	4.12%	0.00%	13.19%	7.40%	1.82%	1.65%	1.64%	0.00%	0.00%	0.10%	0.00%	0.20%	0.50%	0.65%	1.59%	10.63%	0.00%	100.00%
Kam Tin Road	RD_15_16	RR	50	31.3	Ý	1200-1200	1020	32.29%	5.61%	0.00%	18.16%	14.33%	5.95%	1.80%	0.96%	0.00%	0.00%	0.00%	0.10%	0.40%	0.43%	0.50%	2.77%	16.82%	0.00%	100.00%
Kam Tin Road	RD_15_16	RR	50	29.6	Y	1300-1400	1175	31.16%	5.36%	0.00%	16.56%	12.68%	8.44%	2.05%	0.95%	0.00%	0.00%	0.00%	0.00%	0.20%	0.45%	0.45%	2.47%	19.23%	0.00%	100.00%
Kam Tin Road Kam Tin Road	RD_15_16 RD_15_16	RR	50 50	30.1 30.4	Y	1400-1500	1135 1105	33.01% 33.91%	4.86%	0.00%	16.16%	12.43% 13.12%	8.99% 7.89%	1.60%	1.06%	0.00%	0.00%	0.00%	0.16%	0.35%	0.25%	0.45%	2.47%	18.23% 16.59%	0.00%	100.00%
Kam Tin Road	RD_15_16	RR	50	30.4	Ŷ	1600-1700	1105	36.20%	4.36%	0.00%	15.25%	12.08%	6.54%	0.55%	1.10%	0.00%	0.00%	0.00%	0.00%	0.20%	0.45%	0.45%	4.54%	18.26%	0.00%	100.00%
Kam Tin Road	RD_15_16	RR	50	31.9	Y	1700-1800	955	58.37%	3.65%	0.00%	11.24%	5.56%	2.22%	0.45%	1.51%	0.00%	0.52%	0.23%	0.11%	0.20%	0.31%	0.55%	4.60%	10.49%	0.00%	100.00%
Kam Tin Road	RD_15_16	RR	50	32.1	Ý	1900-2000	945	58.47%	3.80%	0.00%	11.81%	5.98%	2.23%	0.20%	1.46%	0.00%	0.53%	0.22%	0.11%	0.20%	0.31%	0.56%	4.91%	8.97%	0.00%	100.00%
Kam Tin Road	RD_15_16	RR	50	35.6	Y	2000-2100	610	51.71%	7.56%	0.00%	13.55%	2.50%	1.48%	0.35%	1.85%	0.00%	0.00%	0.00%	0.00%	0.00%	0.51%	0.65%	5.15%	14.70%	0.00%	100.00%
Kam Tin Road Kam Tin Road	RD_15_16 RD_15_16	RR	50 50	36.7	Y	2100-2200	500 375	48.33% 41.25%	6.65% 10.61%	0.00%	13.74% 13.05%	5.46% 4.65%	1.82% 5.72%	0.00%	1.41%	0.00%	0.00%	0.00%	0.00%	0.40%	1.22% 0.81%	0.79%	5.83%	14.34% 18.47%	0.00%	100.00%
Kam Tin Road	RD_15_16	RR	50	38.8	Ŷ	2300-0000	305	52.93%	15.35%	0.00%	10.35%	7.40%	4.02%	0.00%	0.32%	0.00%	0.00%	0.00%	0.32%	0.65%	2.04%	0.97%	5.00%	0.65%	0.00%	100.00%
Kam Tin Road	RD_17_18	RR	50	39.9	Y	0000-0100	195	73.88%	11.70%	0.00%	4.82%	5.88%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	1.08%	2.65%	0.00%	0.00%	100.00%
Kam Tin Road	RD_17_18	RR	50	40.0	Ý	0200-0300	95	67.36%	14.19%	0.00%	7.59%	9.77%	1.09%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	100.00%
Kam Tin Road	RD_17_18	RR	50	40.0	Y	0300-0400	80	52.75%	20.25%	0.00%	9.45%	16.20%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	1.35%	0.00%	0.00%	100.00%
Kam Tin Road Kam Tin Road	RD_17_18 RD_17_18	RR	50 50	40.0 40.0	Ý	0400-0500	100	58.80% 60.31%	13.30% 6.03%	0.00%	14.40% 8.45%	11.40% 4.18%	1.05% 4.19%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	1.05%	0.00%	0.00%	100.00%
Kam Tin Road	RD_17_18	RR	50	37.6	Y	0600-0700	420	68.63%	5.77%	0.00%	6.73%	3.84%	1.69%	0.45%	3.12%	0.00%	0.00%	0.23%	0.00%	0.45%	0.48%	0.70%	5.98%	1.94%	0.00%	100.00%
Kam Tin Road Kam Tin Road	RD_17_18 RD_17_18	RR	50 50	31.8	Y	0700-0800	965 955	62.73% 60.56%	3.24%	0.00%	10.31%	3.75% 6.53%	0.97%	0.95%	1.58%	0.00%	0.00%	0.00%	3.13%	0.45%	0.43%	0.43%	5.68% 4 17%	6.35% 6.81%	0.00%	100.00%
Kam Tin Road	RD_17_18	RR	50	32.0	Ý	0900-1000	950	57.44%	2.86%	0.00%	12.73%	6.52%	1.04%	1.35%	1.79%	0.00%	0.00%	0.30%	0.50%	0.20%	0.50%	0.65%	3.92%	10.19%	0.00%	100.00%
Kam Tin Road	RD_17_18	RR	50	32.5	Y	1000-1100	905	56.66%	4.00%	0.00%	13.22%	7.31%	1.88%	1.70%	1.69%	0.00%	0.00%	0.09%	0.00%	0.20%	0.32%	0.32%	1.62%	10.99%	0.00%	100.00%
Kam Tin Road Kam Tin Road	RD_17_18 RD_17_18	RR	50 50	32.7	Ý	1100-1200	880 1000	53.62% 32.35%	3.18% 5.45%	0.00%	14.58% 17.98%	9.39% 14.58%	1.51% 6.07%	1.50%	1.15%	0.00%	0.00%	0.34%	0.00%	0.00%	0.21%	0.32%	2.17%	12.04%	0.00%	100.00%
Kam Tin Road	RD_17_18	RR	50	29.9	Y	1300-1400	1150	31.21%	5.10%	0.00%	16.37%	12.92%	8.62%	2.10%	0.95%	0.00%	0.00%	0.00%	0.00%	0.20%	0.20%	0.20%	2.51%	19.62%	0.00%	100.00%
Kam Tin Road	RD_17_18 RD_17_18	RR	50 50	30.3 30.6	Y	1400-1500	1115	33.06% 33.75%	4.65%	0.00%	15.88%	12.62%	9.12%	1.65%	1.05%	0.00%	0.00%	0.00%	0.20%	0.40%	0.10%	0.20%	2.51%	18.57%	0.00%	100.00%
Kam Tin Road	RD_17_18	RR	50	30.6	Ý	1600-1700	1080	36.15%	4.20%	0.00%	15.13%	12.33%	6.67%	0.55%	1.15%	0.00%	0.00%	0.00%	0.00%	0.20%	0.19%	0.19%	4.63%	18.61%	0.00%	100.00%
Kam Tin Road	RD_17_18	RR	50	31.9	Y	1700-1800	955	58.37%	3.65%	0.00%	11.24%	5.56%	2.22%	0.45%	1.51%	0.00%	0.52%	0.23%	0.11%	0.20%	0.31%	0.55%	4.60%	10.49%	0.00%	100.00%
Kam Tin Road	RD 17 18	RR	50 50	31.3	r Y	1900-1900	945	50.24% 58.47%	4.04% 3.80%	0.00%	13.48% 11.81%	7.81% 5.98%	3.51% 2.23%	0.20%	1.00%	0.00%	0.00%	0.00%	0.10%	0.20%	0.10%	0.19%	5.40% 4.91%	12.45% 8.97%	0.00%	100.00%
Kam Tin Road	RD_17_18	RR	50	35.7	Y	2000-2100	595	52.12%	7.24%	0.00%	13.43%	2.55%	1.54%	0.35%	1.85%	0.00%	0.00%	0.00%	0.00%	0.00%	0.19%	0.34%	5.29%	15.10%	0.00%	100.00%
Kam Tin Road Kam Tin Road	RD_17_18 RD_17_18	RR RR	50 50	36.9 38.2	Y	2100-2200	485 360	48.71% 41.53%	6.44% 10.40%	0.00%	13.78% 13.18%	5.67% 4.80%	1.88% 5.92%	0.00%	1.46% 0.86%	0.00%	0.00%	0.00%	0.00%	0.40%	0.43%	0.43%	6.01% 3.36%	14.80% 19.08%	0.00%	100.00%
Kam Tin Road	RD_17_18	RR	50	38.9	Ŷ	2300-0000	295	53.80%	15.24%	0.00%	10.77%	7.66%	4.15%	0.00%	0.37%	0.00%	0.00%	0.00%	0.37%	0.70%	0.71%	0.33%	5.19%	0.70%	0.00%	100.00%
Kam Tin Road	RD_19_20	RR	50	38.9	Y	0000-0100	295	68.18%	15.92%	0.00%	3.11%	4.84%	0.00%	0.70%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	5.91%	1.35%	0.00%	0.00%	100.00%
Kam Tin Road Kam Tin Road	RD 19_20	RR	50 50	40.0	Y Y	0200-0300	165 135	59.25% 61.90%	20.97%	0.00%	8.64% 4.49%	4.33% 8.99%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	5.56% 0.73%	1.25%	0.00%	0.00%	100.00%
Kam Tin Road	RD_19_20	RR	50	40.0	Ŷ	0300-0400	110	45.15%	29.35%	0.00%	7.55%	15.10%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.95%	1.90%	0.00%	0.00%	100.00%
Kam Tin Road	RD_19_20	RR	50	40.0	Y	0400-0500	145	54.66%	19.36%	0.00%	10.09%	10.09%	1.45%	0.00%	1.45%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	1.45%	1.45%	0.00%	0.00%	100.00%
Nam minodu	10_19_20	ININ	50	35.4	'	0000-0000	240	33.34 %	5.00 /0	0.00 /0	J.J4 /0	4.1370	+.50 /0	2.4J/0	J.1J/0	0.00 /0	0.0070	1.21/0	3.30 /0	2.00 /0	1.21/0	3.14/0	1.21/0	0.00 /0	0.0070	100.00%

Image         Image <th< th=""><th>Road Name</th><th>Road ID</th><th>Road Type</th><th>Speed Limit (km/hr)</th><th>Average Speed (km/hr)</th><th>Start Emission Estimated by Broad Brush Approach (Y/N)</th><th>Hour</th><th>Total Vehicles (Veh/hr)</th><th>01 - Private Cars</th><th>02 - Taxi</th><th>03 - Light Goods Vehicles&lt; =2.5t</th><th>04 - Light Goods Vehicles 2.5-3.5t</th><th>05 - Light Goods Vehicles&gt; 3.5t</th><th>06 - Medium Goods Vehicles&lt; =15t</th><th>07 - Medium Goods Vehicles1 5-24t</th><th>08 - Public Light Buses</th><th>09 - Private Light Bus &lt;=3.5t</th><th>10 - Private Light Bus &gt;3.5t</th><th>11 - Non- franchise d Bus&lt;=6.4 t</th><th>12 - Non- franchise d Bus 6.4 15t</th><th>13 - Non- franchise d Bus 15- 24t</th><th>14 - Franchise d Bus Single Deck</th><th>15 - Franchise d Bus Double Deck</th><th>16 - Motorcycl es</th><th>17 - Heavy Goods Vehicles&gt; 24t</th><th>18 - Non- franchise d Bus &gt;24t</th><th>Total</th></th<>	Road Name	Road ID	Road Type	Speed Limit (km/hr)	Average Speed (km/hr)	Start Emission Estimated by Broad Brush Approach (Y/N)	Hour	Total Vehicles (Veh/hr)	01 - Private Cars	02 - Taxi	03 - Light Goods Vehicles< =2.5t	04 - Light Goods Vehicles 2.5-3.5t	05 - Light Goods Vehicles> 3.5t	06 - Medium Goods Vehicles< =15t	07 - Medium Goods Vehicles1 5-24t	08 - Public Light Buses	09 - Private Light Bus <=3.5t	10 - Private Light Bus >3.5t	11 - Non- franchise d Bus<=6.4 t	12 - Non- franchise d Bus 6.4 15t	13 - Non- franchise d Bus 15- 24t	14 - Franchise d Bus Single Deck	15 - Franchise d Bus Double Deck	16 - Motorcycl es	17 - Heavy Goods Vehicles> 24t	18 - Non- franchise d Bus >24t	Total
Kam Tin Raad       RB       50       27.7       Y       0700-0800       1355       64.26%       4.31%       0.00%       5.25%       3.26%       1.05%       1.26%       2.07%       0.00% <th< th=""><th>Kam Tin Road</th><th>RD 19 20</th><th>RR</th><th>50</th><th>35.4</th><th>Y</th><th>0600-0700</th><th>630</th><th>PC 65.89%</th><th>8.00%</th><th>0.00%</th><th>LGV4 4.51%</th><th><u>LGV6</u> 3.39%</th><th>1.65%</th><th>0.65%</th><th>4.15%</th><th>0.00%</th><th>PV5 0.00%</th><th>0.49%</th><th>0.00%</th><th>0.49%</th><th>FBSD 0.49%</th><th>FBDD 4.51%</th><th>MC 3.99%</th><th>HGV9 1.79%</th><th>0.00%</th><th>100.00%</th></th<>	Kam Tin Road	RD 19 20	RR	50	35.4	Y	0600-0700	630	PC 65.89%	8.00%	0.00%	LGV4 4.51%	<u>LGV6</u> 3.39%	1.65%	0.65%	4.15%	0.00%	PV5 0.00%	0.49%	0.00%	0.49%	FBSD 0.49%	FBDD 4.51%	MC 3.99%	HGV9 1.79%	0.00%	100.00%
Name         The Data         P         Sold         Sol	Kam Tin Road	RD_19_20	RR	50	27.7	Y	0700-0800	1355	64.92%	4.66%	0.00%	7.35%	3.59%	1.05%	1.25%	2.40%	0.00%	0.00%	0.10%	1.80%	0.73%	0.50%	2.45%	4.08%	5.13%	0.00%	100.00%
Imam Tan Rade         France         France         Part Solution	Kam Tin Road Kam Tin Road	RD_19_20 RD 19_20	RR	50 50	27.8	Y Y	0800-0900	1345 1350	62.66% 58.40%	4.31% 5.95%	0.00%	9.25% 10.25%	6.18% 6.13%	1.15% 1.80%	1.80%	2.70%	0.00%	0.00%	0.50%	0.30%	0.34%	0.40%	1.90%	2.99%	5.53% 7.88%	0.00%	100.00%
Kam       In Read       Rd.       19, 20       Ref       50       2, 37       Y       100-100       1200       1205	Kam Tin Road	RD_19_20	RR	50	28.6	Y	1000-1100	1275	58.20%	5.91%	0.00%	9.46%	6.93%	2.20%	2.05%	2.40%	0.00%	0.00%	0.20%	0.00%	0.34%	0.40%	2.20%	1.20%	8.53%	0.00%	100.00%
Kam Tin Read         RD         50         28.8         Y         1300-1400         14.07         14.27         24.98         1.94%         0.00%         0.00%         0.05%         0.55%         1.55%         2.28%         17.94%         0.00%         0.00%         0.00%         0.05%         0.55%         1.55%         2.28%         17.94%         0.00%         0.00%         0.00%         0.05%         0.15%         0.25%         1.57%         0.00%         0.00%         0.00%         0.05%         0.15%         0.25%         1.57%         0.00%         0.00%         0.00%         0.05%         0.15%         0.25%         0.05%         0.00%         0.00%         0.00%         0.05%         <	Kam Tin Road Kam Tin Road	RD_19_20 RD 19_20	RR	50 50	29.3 28.6	Y Y	1100-1200	1210 1270	56.39% 39.71%	4.87%	0.00%	10.56%	8.98% 16.21%	1.65% 1.56%	1.80%	1.65%	0.00%	0.00%	0.40%	0.00%	0.15%	0.30%	1.90%	1.55% 2.40%	9.82% 14.91%	0.00%	100.00%
Kam Tin Road       RD_19_20       RR       50       27.3       Y       1440-1500       140       127%       127%       2.43%       101%       2.39%       0.00%       0.00%       0.00%       0.20%       115%       0.25%       17.7%       2.15%       18.7%%       0.00%       100.00%         Kam Tin Road       RD_19_20       RR       50       27.3       Y       1600-1700       1400       5.04%       0.00%       13.22%       10.05%       11.7%       0.35%       0.05%       <	Kam Tin Road	RD_19_20	RR	50	26.8	Y	1300-1400	1440	39.18%	6.32%	0.00%	11.97%	14.62%	2.30%	1.28%	1.94%	0.00%	0.00%	0.00%	0.00%	0.05%	0.50%	1.65%	2.26%	17.94%	0.00%	100.00%
Kam Tin Road         RD, 19 20         RR         50         27.3         Y         1900-1700         1400         44.00%         5.05%         0.00%         1.07%         1.35%         2.30%         0.00%	Kam Tin Road Kam Tin Road	RD_19_20 RD 19_20	RR	50 50	27.3	Y Y	1400-1500	1400 1365	41.26% 42.34%	5.74% 5.04%	0.00%	11.67%	14.27% 15.06%	2.43%	1.01%	2.39%	0.00%	0.00%	0.00%	0.20%	0.15%	0.25%	1.70% 1.75%	2.15%	16.78% 15.07%	0.00%	100.00%
Kam In Nead       FAD	Kam Tin Road	RD_19_20	RR	50	27.3	Y	1600-1700	1400	44.00%	5.05%	0.00%	10.70%	13.50%	1.70%	0.35%	2.30%	0.00%	0.00%	0.00%	0.00%	0.05%	0.50%	1.60%	3.90%	16.35%	0.00%	100.00%
Kam Tin Read       RD 19 20       RR 50       27.9       Y       1900000       1335       67.15%       3.83%       0.00%       7.14%       5.62%       0.71%       0.00%       0.00%       0.01%       0.01%       0.01%       0.01%       0.01%       0.01%       0.01%       0.01%       0.01%       0.00%	Kam Tin Road Kam Tin Road	RD_19_20 RD 19_20	RR	50 50	26.3	Y Y	1700-1800 1800-1900	1495 1410	46.56% 57.70%	4.86% 5.13%	0.00%	12.38% 8.91%	9.79% 7.88%	2.13% 0.85%	0.35%	2.80%	0.00%	0.00%	0.00%	0.00%	0.15%	0.20%	1.60% 1.59%	6.09% 4.25%	13.08% 9.77%	0.00%	100.00%
Kam Tin Road       ND_19_20       RR       50       33.2       Y       2000*       6.20%       7.12%       0.00%       6.27%       2.28%       0.00%	Kam Tin Road	RD_19_20	RR	50	27.9	Y	1900-2000	1335	67.15%	3.83%	0.00%	7.14%	5.62%	0.51%	0.23%	2.76%	0.00%	0.00%	0.00%	0.16%	0.09%	0.31%	1.94%	3.59%	6.68%	0.00%	100.00%
Kam         Tin Road         PD         36.9         Y         2200-2300         480         48.04%         12.08%         0.00%         1.45%         0.00%         0.	Kam Tin Road Kam Tin Road	RD_19_20 RD 19_20	RR	50 50	33.2 34.9	Y Y	2000-2100 2100-2200	835 675	58.20% 54.78%	7.92%	0.00%	8.62% 9.09%	2.52% 5.63%	0.37%	0.14%	3.62%	0.00%	0.00%	0.00%	0.00%	0.00%	0.62%	2.38%	4.08% 4.79%	11.52% 10.92%	0.00%	100.00%
Kam       In Road       RD_19_20       RR       50       37.5       Y       2000-000       21.29       RL       50       37.6       Y       2000-000       21.29       RL       50       37.6       Y       2000-000       21.09       2.44%       1.24%       0.00% <th< td=""><td>Kam Tin Road</td><td>RD_19_20</td><td>RR</td><td>50</td><td>36.9</td><td>Y</td><td>2200-2300</td><td>480</td><td>48.04%</td><td>12.08%</td><td>0.00%</td><td>9.00%</td><td>5.00%</td><td>1.45%</td><td>0.00%</td><td>1.46%</td><td>0.00%</td><td>0.00%</td><td>0.00%</td><td>0.00%</td><td>0.00%</td><td>1.04%</td><td>4.58%</td><td>2.73%</td><td>14.61%</td><td>0.00%</td><td>100.00%</td></th<>	Kam Tin Road	RD_19_20	RR	50	36.9	Y	2200-2300	480	48.04%	12.08%	0.00%	9.00%	5.00%	1.45%	0.00%	1.46%	0.00%	0.00%	0.00%	0.00%	0.00%	1.04%	4.58%	2.73%	14.61%	0.00%	100.00%
Lam Kam Road         RD_21_22         RR         50         40.0         Y         0100-020         120         68.18%         18.15%         0.00%         2.61%         0.82%         0.00%         <	Kam Tin Road Lam Kam Road	RD_19_20 RD_21_22	RR	50 50	37.6 39.6	Y Y	2300-0000	420 210	57.41% 77.05%	16.29%	0.00%	6.48% 2.44%	7.42% 1.94%	0.95%	0.00%	0.45%	0.00%	0.00%	0.00%	0.22%	0.23%	1.70%	4.32% 3.92%	3.82% 0.99%	0.72%	0.00%	100.00%
Lam Kam Road       RD_21_22       RR       50       40.0       Y       0200-0300       105       7       7.14%       7.18%       0.00%       0.	Lam Kam Road	RD_21_22	RR	50	40.0	Y	0100-0200	120	68.18%	18.15%	0.00%	6.87%	2.61%	0.82%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	3.38%	0.00%	0.00%	0.00%	100.00%
Lam Kam Road         RD_21_22         RR         50         40.0         Y         0400-0500         105         64.29%         17.34%         0.00%         2.06%         0.00%	Lam Kam Road Lam Kam Road	RD_21_22 RD_21_22	RR RR	50 50	40.0 40.0	Y	0200-0300 0300-0400	105 75	71.04% 57.75%	17.56% 26.77%	0.00%	4.16% 7.06%	4.16% 7.03%	3.09% 1.39%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00% 0.00%	0.00%	0.00% 0.00%	0.00%	0.00%	100.00%
Lam Kam Road         RD_21_22         RR         50         39.9         Y         0500-0600         170         65.62%         7.48%         0.00%         1.44%         4.83%         0.00%         0.00%         0.00%         2.40%         1.23%         0.00%         <	Lam Kam Road	RD_21_22	RR	50	40.0	Y	0400-0500	105	64.29%	17.34%	0.00%	8.13%	5.10%	2.06%	0.00%	2.06%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	1.03%	0.00%	0.00%	0.00%	100.00%
Lam Kam Road       RD_21_22       RR       50       32.3       Y       0700-0800       92.0       75.52%       4.38%       0.00%       6.55%       1.76%       1.50%       0.86%       2.03%       0.00%       0.00%       0.04%       0.00%       1.50%       3.06%       2.21%       0.00%	Lam Kam Road Lam Kam Road	RD_21_22 RD_21_22	RR RR	50 50	39.9 37.3	Y	0500-0600 0600-0700	170 445	65.62% 73.43%	7.85% 6.96%	0.00%	4.80% 3.86%	2.45% 1.55%	7.18% 2.46%	1.84% 0.47%	4.83% 3.39%	0.00%	0.00%	0.00%	0.00%	1.79% 0.44%	0.00% 0.00%	2.40% 2.71%	1.23% 2.69%	0.00% 2.02%	0.00%	100.00%
Lam Kam Road       RD_21_22       RR       50       32.4       Y       0800-0900       910       72.99%       3.94%       0.00%       1.28%       2.23%       0.00%       0.00%       0.00%       0.01%       0.00%       0.01%       0.00%       0.01%       0.00%       0.01%       0.00%       <	Lam Kam Road	RD_21_22	RR	50	32.3	Ŷ	0700-0800	920	75.52%	4.36%	0.00%	6.55%	1.76%	1.50%	0.86%	2.05%	0.00%	0.00%	0.00%	0.00%	0.64%	0.00%	1.50%	3.06%	2.21%	0.00%	100.00%
Lam Kam Road         RD_21_22         RR         50         33.1         Y         1000-1100         845         69.54%         5.63%         0.00%         3.45%         1.44%         2.12%         0.00%         0.00%         0.00%         0.25%         0.00%         1.42%         0.99%         3.13%         0.00%         100.01%           Lam Kam Road         RD_21_22         RR         50         33.7         Y         1100-1200         790         66.821%         4.66%         0.00%         2.82%         1.38%         1.41%         0.00%	Lam Kam Road	RD_21_22 RD_21_22	RR RR	50 50	32.4 33.0	Y	0800-0900	910 855	72.99% 70.49%	3.94% 5.03%	0.00%	8.22% 8.30%	3.07% 3.03%	1.86%	1.22% 1.33%	2.32%	0.00%	0.00%	0.00%	0.00%	0.31%	0.00%	1.20% 1.26%	2.21%	2.65% 2.83%	0.00%	100.00% 100.00%
Lam Kam Road         RD_21_22         RR         50         33.7         Y         1100-1200         766         68.21%         4.56%         0.00%         2.82%         1.83%         0.00%         0.00%         0.00%         0.12%         0.00%         0.12%         0.00%         0.12%         0.00%         0.12%         0.00%         <	Lam Kam Road	RD_21_22	RR	50	33.1	Ŷ	1000-1100	845	69.54%	5.63%	0.00%	8.57%	3.46%	3.45%	1.44%	2.12%	0.00%	0.00%	0.00%	0.00%	0.25%	0.00%	1.42%	0.99%	3.13%	0.00%	100.00%
Lam Kam Read         RD_21_22         R         50         33.9         Y         1300-1400         770         48.82%         5.38%         0.00%         1.57%         2.21%         0.00% <t< td=""><td>Lam Kam Road</td><td>RD_21_22 RD_21_22</td><td>RR</td><td>50 50</td><td>33.7 35.0</td><td>Y</td><td>1100-1200</td><td>790 665</td><td>68.21% 50.65%</td><td>4.56% 5.71%</td><td>0.00%</td><td>9.80% 11.56%</td><td>4.76% 10.18%</td><td>2.82% 3.46%</td><td>1.38%</td><td>1.51% 2.43%</td><td>0.00%</td><td>0.00%</td><td>0.00%</td><td>0.00%</td><td>0.12%</td><td>0.00%</td><td>1.24% 2.25%</td><td>1.27% 2.29%</td><td>4.33% 9.91%</td><td>0.00%</td><td>100.00%</td></t<>	Lam Kam Road	RD_21_22 RD_21_22	RR	50 50	33.7 35.0	Y	1100-1200	790 665	68.21% 50.65%	4.56% 5.71%	0.00%	9.80% 11.56%	4.76% 10.18%	2.82% 3.46%	1.38%	1.51% 2.43%	0.00%	0.00%	0.00%	0.00%	0.12%	0.00%	1.24% 2.25%	1.27% 2.29%	4.33% 9.91%	0.00%	100.00%
Lam Kam Road         RD_21_22         RR         50         34.0         Y         1400-1500         760         50.78%         4.76%         0.00%         1.61%         2.65%         0.00%         <	Lam Kam Road	RD_21_22	RR	50	33.9	Ŷ	1300-1400	770	48.82%	5.36%	0.00%	10.70%	8.98%	4.96%	1.57%	2.21%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	1.67%	2.06%	13.67%	0.00%	100.00%
Lam Kam Road RD_21_22 RR 50 34.1 Y 1600-1700 755 54.64% 4.10% 0.00% 9.29% 8.35% 0.06% 0.06% 0.00	Lam Kam Road	RD_21_22 RD_21_22	RR	50 50	34.0	Y	1400-1500	760 735	50.78%	4.76%	0.00%	10.21%	8.86% 9.33%	5.14%	1.16%	2.65%	0.00%	0.00%	0.00%	0.28%	0.00%	0.00%	1.88%	1.99%	12.30%	0.00%	100.00%
Lam Kam Road RD_21_22 RR 50 34.0 Y 1700-1800 760 75.57% 3.03% 0.00% 6.01% 3.32% 1.05% 0.27% 2.94% 0.00% 0.13% 0.13% 0.11% 0.00% 0.00% 2.01% 3.19% 2.24% 0.00% 100.00% 100.00% RD_21_22 RR 50 33.7 Y 1800-1900 790 67.89% 4.25% 0.00% 7.48% 0.13% 3.14% 4.54% 0.00% 10.00% 0.00% 0.01% 0.00% 0.00% 0.01% 0.00	Lam Kam Road	RD_21_22 RD_21_22	RR	50	34.1	Ý	1600-1700	755	54.64%	4.10%	0.00%	9.29%	8.35%	3.58%	0.40%	2.50%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	1.71%	3.73%	11.69%	0.00%	100.00%
	Lam Kam Road	RD_21_22	RR	50 50	34.0	Y	1700-1800	760	75.57%	3.03%	0.00%	6.01% 7.62%	3.32%	1.05%	0.27%	2.94%	0.00%	0.13%	0.13%	0.11%	0.00%	0.00%	2.01%	3.19%	2.24%	0.00%	100.00%
Lam Kam Road RD_21_22 RR 50 33.6 Y 1900-2000 800 69.35% 4.64% 0.00% 7.93% 3.92% 1.62% 0.39% 2.93% 0.00% 0.12% 0.12% 0.12% 0.00% 1.93% 4.28% 2.65% 0.00% 100.00%	Lam Kam Road	RD_21_22 RD_21_22	RR	50	33.6	Ý	1900-2000	800	69.35%	4.64%	0.00%	7.93%	3.92%	1.62%	0.39%	2.93%	0.00%	0.12%	0.12%	0.12%	0.00%	0.00%	1.93%	4.28%	2.65%	0.00%	100.00%
Lam Kam Road RD_21_22 RR 50 37.1 Y 2000-2100 470 6941% 6.46% 0.00% 7.52% 1.50% 0.67% 0.22% 3.90% 0.00% 0.00% 0.00% 0.00% 0.00% 0.35% 3.68% 4.29% 0.00% 100.00%	Lam Kam Road	RD_21_22	RR	50	37.1	Y	2000-2100	470	69.41%	6.46%	0.00%	7.52%	1.50%	0.67%	0.22%	3.90%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	2.35%	3.68%	4.29%	0.00%	100.00%
Lami Kami Koad RD_2122 RR 50 36.2 T 2100-2200 360 60.65% 5.65% 0.00% 5.10% 0.61% 5.21% 0.00% 0.0	Lam Kam Road	RD_21_22 RD_21_22	RR	50	39.4	Ý	2200-2200	250	63.09%	10.56%	0.00%	8.14%	3.26%	3.21%	0.00%	1.63%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	5.28%	2.42%	2.23%	0.00%	100.00%
Lam Kam Road RD_21_22 RR 50 39.4 Y 2300-0000 250 66.09% 12.70% 0.00% 5.32% 4.48% 2.06% 0.00% 0.79% 0.00% 0.0	Lam Kam Road	RD_21_22	RR	50	39.4	Y	2300-0000	250	66.09%	12.70%	0.00%	5.32%	4.48%	2.06%	0.00%	0.79%	0.00%	0.00%	0.00%	0.39%	0.00%	0.00%	4.08%	3.30%	0.79%	0.00%	100.00%
RouteTwisk RD_23_24 RR 50 40.0 Y 0100-0200 60 78.45% 9.05% 0.00% 9.00% 0	Route Twisk	RD_23_24 RD_23_24	RR	50	40.0	Ý	0100-0200	60	78.45%	9.05%	0.00%	9.05%	1.73%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	1.73%	0.00%	0.00%	100.00%
Route Twisk RD_23_24 RR 50 40.0 Y 0200-0300 50 75.78% 8.75% 0.00% 6.72% 6.72% 2.03% 0.00%	Route Twisk	RD_23_24	RR	50	40.0	Y	0200-0300	50	75.78%	8.75%	0.00%	6.72%	6.72%	2.03%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	100.00%
RouteTwisk RD_23_24 RR 50 40.0 Y 0400-6500 50 69.97% 91.0% 0.00% 11.43% 01.43% 0.00%	Route Twisk	RD_23_24 RD_23_24	RR	50	40.0	Ý	0300-0400	30 50	62.37% 69.97%	9.10%	0.00%	11.43%	7.00%	0.00%	0.00%	2.73%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	100.00%
Route Twisk RD_23_24 RR 50 40.0 Y 0500-0600 85 71,05% 3,74% 0.00% 6,14% 1.20% 3,53% 2,61% 9,12% 0.00%	Route Twisk	RD_23_24	RR	50	40.0	Y	0500-0600	85	71.05%	3.74%	0.00%	6.14%	1.20%	3.53%	2.61%	9.12%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	2.61%	0.00%	0.00%	100.00%
Route Twisk RD_23_24 RR 50 36.0 Y 0700-0800 570 80.59% 1.55% 0.00% 3.85% 1.55% 0.53% 1.04% 2.65% 0.00% 0.0	Route Twisk	RD_23_24 RD_23_24	RR	50 50	38.9	Ý	0700-0800	260 570	79.80% 80.59%	2.73%	0.00%	3.85% 6.85%	1.58%	0.53%	1.04%	4.65% 2.63%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	5.45% 5.26%	0.00%	0.00%	100.00%
Route Twisk RD_23_24 RR 50 36.1 Y 0800-0900 560 77.88% 1.40% 0.00% 8.60% 2.90% 0.75% 1.48% 3.08% 0.00%	Route Twisk	RD_23_24	RR	50	36.1	Y	0800-0900	560	77.88%	1.40%	0.00%	8.60%	2.90%	0.75%	1.48%	3.08%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	3.93%	0.00%	0.00%	100.00%
RouteTwisk RD_23_24 RR 50 36.7 Y 1000-1100 580 /7.22% 1.37% し.00% 8.69% 3.15% し.72% 1.41% 2.97% し.00% し.00% し.00% し.00% 0.00%	Route Twisk Route Twisk	RD_23_24 RD_23_24	RR	50 50	35.9 36.7	Ý	1000-1000	580 500	77.20% 77.15%	1.37%	0.00%	8.69% 9.48%	3.15% 3.45%	0.72%	1.41% 1.65%	2.97% 3.05%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	4.49% 1.60%	0.00%	0.00%	100.00%
Route Twisk RD_23_24 RR 50 37.1 Y 1100-1200 465 75.72% 1.72% 0.00% 0.65% 4.58% 1.05% 1.72% 2.16% 0.00%	Route Twisk	RD_23_24	RR	50	37.1	Y	1100-1200	465	75.72%	1.72%	0.00%	10.65%	4.58%	1.05%	1.72%	2.16%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	2.40%	0.00%	0.00%	100.00%
RouteTwisk RD 23 24 RR 50 38.4 Y 1200-1400 310 54.00% 4.55% U.00% 17.95% 9.80% 1.95% 1.30% 3.90% U.00% U.00% U.00% U.00% U.00% U.00% 4.05% U.00% 4.05% 17.30% U.00% 100.00% RD 23 24 RR 50 38.4 Y 1300-1400 34 5 53.93% 4.67% 0.00% 16.9% 9.07% 2.95% 1.72% 3.79% 0.00% 0.00% 0.00% 0.00% 0.00% 1.05% U.00% 4.09% 1.7.3% U.00% 100.00%	Route I wisk Route Twisk	RD_23_24 RD_23_24	RR	50 50	38.7	Y Y	1200-1300	310 345	54.00% 53.93%	4.55% 4.67%	0.00%	17.95%	9.80% 9.07%	1.95% 2.95%	1.30%	3.90%	0.00%	0.00%	0.00%	0.00%	0.00%	0.65%	0.00%	4.60% 4.09%	1.30%	0.00%	100.00%
Route Twisk RD_23_24 RR 50 38.4 Y 1400-1500 340 56.15% 4.20% 0.00% 15.95% 8.65% 2.95% 1.45% 4.45% 0.00% 0.00% 0.00% 0.00% 0.00% 0.00% 0.415% 1.45% 1.45% 0.00% 100.00%	Route Twisk	RD_23_24	RR	50	38.4	Y	1400-1500	340	56.15%	4.20%	0.00%	15.95%	8.65%	2.95%	1.45%	4.45%	0.00%	0.00%	0.00%	0.00%	0.00%	0.60%	0.00%	4.15%	1.45%	0.00%	100.00%
Route Twisk RD 23 24 RR 50 38.5 Y 1500-1700 335 55 57.85% U.00% 18.11% 9.01% 2.40% 1.22% 3.65% U.00% U.00% U.00% U.00% U.00% U.00% 0.80% U.00% 3.64% 1.22% U.00% 100.00% Route Twisk RD 23 24 RR 50 38.3 Y 1600-1700 355 57.85% 3.73% 0.00% 1.00% 14.7% 8.07% 1.70% 0.59% 4.25% 0.00% 0.00% 0.00% 0.00% 1.00% 1.00% 1.42% 0.00% 10.00%	Route I wisk Route Twisk	RD_23_24 RD_23_24	RR	50 50	38.5	Y Y	1500-1600	335 355	56.53% 57.85%	3.62%	0.00%	18.11% 14.31%	9.01% 8.07%	2.40%	1.22%	3.65% 4.25%	0.00%	0.00%	0.00%	0.00%	0.00%	0.60%	0.00%	3.64% 6.92%	1.22%	0.00%	100.00%
Route Twisk RD_23_24 RR 50 37.8 Y 1700-1800 400 75.80% 2.25% 0.00% 8.10% 2.75% 0.50% 0.25% 4.30% 0.00% 0.00% 0.00% 0.00% 0.50% 0.00% 5.30% 0.25% 0.00% 100.00%	Route Twisk	RD_23_24	RR	50	37.8	Y	1700-1800	400	75.80%	2.25%	0.00%	8.10%	2.75%	0.50%	0.25%	4.30%	0.00%	0.00%	0.00%	0.00%	0.00%	0.50%	0.00%	5.30%	0.25%	0.00%	100.00%
Route Twisk RD 23 24 RR 50 37./ Y 1900-1900 405 68.05% 3.27% U.00% 10.5% 4.03% 1.00% U.25% 5.30% U.00% U.00% U.00% U.00% U.00% 0.50% U.00% 0.49% U.00% 100.00% Rute Twisk RD 23 24 RR 50 36.3 Y 1900-200 540 77.45% 2.40% 0.00% 6.00% 6.40% 0.40% 0.20% 3.15% 0.00% 0.	Route I wisk Route Twisk	RD_23_24 RD_23_24	RR	50 50	37.7	Y Y	1800-1900	405 540	68.05% 77.45%	3.27%	0.00%	10.54%	4.03% 2.40%	1.00%	0.25%	5.30% 3.15%	0.00%	0.00%	0.00%	0.00%	0.00%	0.50%	0.00%	6.56% 5.40%	0.49%	0.00%	100.00%
Route Twisk RD_23_24 RR 50 39.5 Y 2000-2100 235 69.65% 5.29% 0.00% 10.31% 1.77% 0.00	Route Twisk	RD_23_24	RR	50	39.5	Y	2000-2100	235	69.65%	5.26%	0.00%	10.31%	1.77%	0.00%	0.00%	5.61%	0.00%	0.00%	0.00%	0.00%	0.00%	0.90%	0.00%	6.07%	0.44%	0.00%	100.00%
RouteTwinsk RD 23 24 RR 50 40.0 Y 2100-2200 156 65.3/% 4.52% U.UU% 11.55% 3.31% U.UU% U.UU% U.UU% U.UU% U.UU% 0.00% 0	Route Twisk	RD 23_24 RD 23_24	RR	50 50	40.0 40.0	Y Y	2100-2200 2200-2300	185 115	64.03%	4.92% 8.82%	0.00%	12.25%	3.31% 3.52%	0.00%	0.00%	4.96% 3.50%	0.00%	0.00%	0.00%	0.00%	0.00%	2.20%	0.00%	7.68% 4.38%	0.00%	0.00%	100.00%
Route Twisk RD_23_24 RR 50 40.0 Y 2300-0000 125 64.37% 10.13% 0.00% 8.50% 4.24% 1.70% 0.00% 1.70% 0.00% 0.00% 0.00% 0.00% 0.00% 0.00% 5.97% 0.00% 0.00% 100.00%	Route Twisk	RD_23_24	RR	50	40.0	Y	2300-0000	125	64.37%	10.13%	0.00%	8.50%	4.24%	1.70%	0.00%	1.70%	0.00%	0.00%	0.00%	0.00%	0.00%	3.39%	0.00%	5.97%	0.00%	0.00%	100.00%
Access Road RD 25 26 RR 30 26.0 Y 0000-0100 10 40.00% 0.00% 40.00% 0.00%	Access Road Access Road	RD_25_26 RD_25_26	RR RR	30 30	26.0 26.0	Y Y	0100-0100	10 10	40.00% 40.00%	0.00%	0.00%	40.00% 40.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	20.00% 20.00%	0.00%	100.00%
Access Road RD_25_26 RR 30 26.0 Y 0200-0300 10 40.00% 0.00%	Access Road	RD_25_26	RR	30	26.0	Ŷ	0200-0300	10	40.00%	0.00%	0.00%	40.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	20.00%	0.00%	100.00%
Access Road RD_25_26 RR 30 26.0 Y 0300-0400 10 40.00% 0.00% 40.00% 0.00%	Access Road	RD_25_26 RD_25_26	RR RR	30 30	26.0 26.0	Y	0300-0400	10 10	40.00%	0.00%	0.00%	40.00% 40.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	20.00%	0.00%	100.00%
Access Read RD_25_26 RR 30 26.0 Y 0500-0600 10 40.00% 0.00%	Access Road	RD_25_26	RR	30	26.0	Ý	0500-0600	10	40.00%	0.00%	0.00%	40.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	20.00%	0.00%	100.00%
Access Road         RD_25_26         RR         30         26.0         Y         0600-0700         10         40.00%         0.00% <th< td=""><td>Access Road Access Road</td><td>RD_25_26 RD_25_26</td><td>RR RR</td><td>30 30</td><td>26.0 26.0</td><td>Y Y</td><td>0600-0700 0700-0800</td><td>10 20</td><td>40.00% 60.00%</td><td>0.00% 0.00%</td><td>0.00% 0.00%</td><td>40.00% 30.00%</td><td>0.00% 0.00%</td><td>0.00% 0.00%</td><td>0.00% 0.00%</td><td>0.00% 0.00%</td><td>0.00% 0.00%</td><td>0.00% 0.00%</td><td>0.00% 0.00%</td><td>0.00% 0.00%</td><td>0.00% 0.00%</td><td>0.00% 0.00%</td><td>0.00% 0.00%</td><td>0.00% 0.00%</td><td>20.00% 10.00%</td><td>0.00% 0.00%</td><td>100.00% 100.00%</td></th<>	Access Road Access Road	RD_25_26 RD_25_26	RR RR	30 30	26.0 26.0	Y Y	0600-0700 0700-0800	10 20	40.00% 60.00%	0.00% 0.00%	0.00% 0.00%	40.00% 30.00%	0.00% 0.00%	0.00% 0.00%	0.00% 0.00%	0.00% 0.00%	0.00% 0.00%	0.00% 0.00%	0.00% 0.00%	0.00% 0.00%	0.00% 0.00%	0.00% 0.00%	0.00% 0.00%	0.00% 0.00%	20.00% 10.00%	0.00% 0.00%	100.00% 100.00%

Road Name	Road ID	Road Type	Speed Limit (km/hr)	Average Speed (km/hr)	Start Emission Estimated by Broad Brush Approach (Y/N)	Hour	Total Vehicles (Veh/hr)	01 - Private Cars	02 - Taxi	03 - Light Goods Vehicles< =2.5t	04 - Light Goods Vehicles 2.5-3.5t	05 - Light Goods Vehicles> 3.5t	06 - Medium Goods Vehicles< =15t	07 - Medium Goods Vehicles1 5-24t	08 - Public Light Buses	09 - Private Light Bus <=3.5t	10 - Private Light Bus >3.5t	11 - Non- franchise d Bus<=6.4 t	12 - Non- franchise d Bus 6.4 15t	13 - Non- franchise d Bus 15- 24t	14 - Franchise d Bus Single Deck	15 - Franchise d Bus Double Deck	16 - Motorcycl es	17 - Heavy Goods Vehicles> 24t	18 - Non- franchise d Bus >24t	Total
Access Road	RD 25 26	RR	30	26.0	Y	0800-0900	20	60.00%	0.00%	0.00%	30.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	10.00%	0.00%	100.00%
Access Road	RD_25_26	RR	30	26.0	Y	0900-1000	20	60.00%	0.00%	0.00%	30.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	10.00%	0.00%	100.00%
Access Road	RD_25_26	RR	30	26.0	Y	1000-1100	20	60.00%	0.00%	0.00%	30.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	10.00%	0.00%	100.00%
Access Road	RD 25 26	RR	30	26.0	Ý	1200-1200	20	60.00%	0.00%	0.00%	30.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	10.00%	0.00%	100.00%
Access Road	RD_25_26	RR	30	26.0	Y	1300-1400	20	60.00%	0.00%	0.00%	30.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	10.00%	0.00%	100.00%
Access Road	RD_25_26	RR	30 30	26.0	Y	1400-1500	20	60.00%	0.00%	0.00%	30.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	10.00%	0.00%	100.00%
Access Road	RD_25_26	RR	30	26.0	Ý	1600-1700	20	60.00%	0.00%	0.00%	30.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	10.00%	0.00%	100.00%
Access Road	RD_25_26	RR	30	26.0	Y	1700-1800	20	60.00%	0.00%	0.00%	30.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	10.00%	0.00%	100.00%
Access Road	RD_25_26 RD_25_26	RR	30 30	26.0 26.0	Y	1800-1900	20	60.00% 40.00%	0.00%	0.00%	30.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	10.00%	0.00%	100.00%
Access Road	RD_25_26	RR	30	26.0	Ŷ	2000-2100	10	40.00%	0.00%	0.00%	40.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	20.00%	0.00%	100.00%
Access Road	RD_25_26	RR	30	26.0	Y	2100-2200	10	40.00%	0.00%	0.00%	40.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	20.00%	0.00%	100.00%
Access Road	RD_25_26 RD_25_26	RR	30 30	26.0	ř Y	2200-2300	10	40.00%	0.00%	0.00%	40.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	20.00%	0.00%	100.00%
Kam Sheung Road	RD_27_28	RR	50	40.0	Ŷ	0000-0100	110	55.24%	20.53%	0.00%	5.63%	5.61%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	11.16%	1.84%	0.00%	0.00%	100.00%
Kam Sheung Road	RD_27_28	RR	50	40.0	Y	0100-0200	65	41.28%	27.02%	0.00%	15.85%	6.36%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	9.49%	0.00%	0.00%	0.00%	100.00%
Kam Sheung Road Kam Sheung Road	RD_27_28 RD_27_28	RR	50 50	40.0	ř Y	0200-0300	50 40	45.30% 26.30%	30.90%	0.00%	8.50%	21.10%	2.25%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	100.00%
Kam Sheung Road	RD_27_28	RR	50	40.0	Ŷ	0400-0500	55	37.18%	23.93%	0.00%	18.60%	14.80%	0.00%	0.00%	1.82%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	3.67%	0.00%	0.00%	0.00%	100.00%
Kam Sheung Road	RD_27_28	RR	50	40.0	Y	0500-0600	95	40.37%	12.11%	0.00%	8.97%	6.64%	4.38%	1.09%	7.73%	0.00%	0.00%	2.13%	6.71%	3.22%	0.00%	6.64%	0.00%	0.00%	0.00%	100.00%
Kam Sheung Road	RD 27 28	RR	50	36.6	Y	0700-0800	240 510	55.52%	6.12%	0.00%	12.85%	4.27%	0.76%	0.41%	3.33%	0.00%	0.00%	0.87%	3.55%	1.18%	0.00%	4.73%	3.42%	2.55%	0.00%	100.00%
Kam Sheung Road	RD_27_28	RR	50	36.6	Y	0800-0900	515	52.70%	5.49%	0.00%	16.10%	8.27%	0.98%	0.98%	3.74%	0.00%	0.00%	0.98%	0.61%	0.60%	0.00%	3.57%	2.40%	3.57%	0.00%	100.00%
Kam Sheung Road	RD_27_28	RR	50	36.8	Y	0900-1000	490	52.27%	5.57%	0.00%	15.86%	8.23%	1.04%	1.04%	3.89%	0.00%	0.00%	1.04%	0.61%	0.60%	0.00%	3.67%	2.50%	3.67%	0.00%	100.00%
Kam Sheung Road	RD 27 28	RR	50	37.0	Y	1100-1200	475	49.67%	6.12%	0.00%	18.35%	9.32%	1.35%	1.09%	2.42%	0.00%	0.00%	0.22%	0.00%	0.43%	0.00%	3.53%	1.27%	4.22% 5.69%	0.00%	100.00%
Kam Sheung Road	RD_27_28	RR	50	36.4	Y	1200-1300	535	31.94%	8.25%	0.00%	20.66%	15.08%	5.94%	4.11%	2.45%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	3.59%	2.09%	5.87%	0.00%	100.00%
Kam Sheung Road	RD_27_28	RR	50 50	35.6	Y	1300-1400	610 590	31.25%	7.90%	0.00%	19.10%	13.14%	8.57%	4.98%	2.29%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	2.78%	1.99%	8.01% 7.46%	0.00%	100.00%
Kam Sheung Road	RD_27_28	RR	50	35.9	Ý	1500-1600	580	33.88%	6.45%	0.00%	20.98%	13.92%	7.87%	3.52%	2.24%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	2.98%	1.93%	6.23%	0.00%	100.00%
Kam Sheung Road	RD_27_28	RR	50	36.1	Y	1600-1700	560	37.20%	6.65%	0.00%	18.10%	13.10%	6.60%	1.45%	2.85%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	2.85%	3.80%	7.40%	0.00%	100.00%
Kam Sheung Road	RD_27_28	RR	50 50	36.5	Y	1700-1800	520	60.25% 51.49%	5.25% 6.90%	0.00%	12.45%	5.65%	2.15%	1.00%	3.50%	0.00%	0.00%	1.00%	0.00%	0.00%	0.00%	3.70%	3.50%	1.55%	0.00%	100.00%
Kam Sheung Road	RD_27_28	RR	50	36.7	Ý	1900-2000	505	58.17%	5.57%	0.00%	13.57%	5.98%	2.16%	1.01%	3.60%	0.00%	0.00%	0.98%	0.00%	0.00%	0.00%	3.79%	3.58%	1.58%	0.00%	100.00%
Kam Sheung Road	RD_27_28	RR	50	38.7	Y	2000-2100	315	52.63%	11.20%	0.00%	15.68%	2.58%	1.27%	0.60%	4.49%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	4.49%	4.18%	2.87%	0.00%	100.00%
Kam Sheung Road Kam Sheung Road	RD_27_28 RD_27_28	RR	50 50	39.3 39.9	Y	2100-2200	255 190	49.92% 40.99%	10.05%	0.00%	16.56%	6.04% 5.47%	1.63% 5.99%	0.00%	3.64%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	5.68% 8.76%	4.83% 2.76%	1.63%	0.00%	100.00%
Kam Sheung Road	RD_27_28	RR	50	40.0	Ŷ	2300-0000	180	45.20%	20.95%	0.00%	10.50%	7.00%	4.10%	0.00%	1.20%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	6.95%	3.50%	0.60%	0.00%	100.00%
Sheung Tsuen B/T and Car Park	RD_29	RR	30	26.0	Y	0000-0100	10	57.10%	14.30%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	28.60%	0.00%	0.00%	0.00%	100.00%
Sheung Tsuen B/T and Car Park Sheung Tsuen B/T and Car Park	RD_29 RD_29	RR	30	26.0	ř Y	0200-0300	5	50.00%	25.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	25.00%	0.00%	0.00%	0.00%	100.00%
Sheung Tsuen B/T and Car Park	RD_29	RR	30	26.0	Y	0300-0400	5	50.00%	50.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	100.00%
Sheung Tsuen B/T and Car Park	RD_29	RR	30	26.0	Y	0400-0500	5	100.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	100.00%
Sheung Tsuen B/T and Car Park	RD_29	RR	30	26.0	Ý	0600-0700	20	56.00%	6.30%	0.00%	6.30%	6.30%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	6.30%	18.80%	0.00%	0.00%	0.00%	100.00%
Sheung Tsuen B/T and Car Park	RD_29	RR	30	26.0	Y	0700-0800	35	64.50%	3.20%	0.00%	6.50%	3.20%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	9.70%	12.90%	0.00%	0.00%	0.00%	100.00%
Sheung Tsuen B/T and Car Park Sheung Tsuen B/T and Car Park	RD_29 RD_29	RR	30 30	26.0 26.0	Y	0800-0900	30 30	63.30% 62.20%	3.30%	0.00%	10.00%	6.70% 6.90%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	6.70% 3.40%	10.00% 13.80%	0.00%	0.00%	0.00%	100.00%
Sheung Tsuen B/T and Car Park	RD_29	RR	30	26.0	Ŷ	1000-1100	30	60.80%	3.60%	0.00%	10.70%	7.10%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	7.10%	10.70%	0.00%	0.00%	0.00%	100.00%
Sheung Tsuen B/T and Car Park	RD_29	RR	30	26.0	Y	1100-1200	30	55.60%	3.70%	0.00%	11.10%	11.10%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	7.40%	11.10%	0.00%	0.00%	0.00%	100.00%
Sheung Tsuen B/T and Car Park Sheung Tsuen B/T and Car Park	RD_29 RD_29	RR	30	26.0	Ý	1200-1300	20	31.60%	15.80%	0.00%	26.30%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	13.60%	13.60%	0.00%	0.00%	0.00%	100.00%
Sheung Tsuen B/T and Car Park	RD_29	RR	30	26.0	Ŷ	1400-1500	20	35.00%	15.00%	0.00%	25.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	10.00%	15.00%	0.00%	0.00%	0.00%	100.00%
Sheung Tsuen B/T and Car Park	RD_29	RR	30	26.0	Y	1500-1600	25	33.30%	14.30%	0.00%	28.60%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	9.50%	14.30%	0.00%	0.00%	0.00%	100.00%
Sheung Tsuen B/T and Car Park	RD_29 RD_29	RR	30	26.0	Y	1700-1800	25	39.20%	13.00%	0.00%	25.80%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	8.70%	13.00%	0.00%	0.00%	0.00%	100.00%
Sheung Tsuen B/T and Car Park	RD_29	RR	30	26.0	Y	1800-1900	25	45.50%	13.60%	0.00%	18.20%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	9.10%	13.60%	0.00%	0.00%	0.00%	100.00%
Sheung Tsuen B/T and Car Park	RD_29	RR	30	26.0	Y	1900-2000	25	52.40%	9.50%	0.00%	14.30%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	9.50%	14.30%	0.00%	0.00%	0.00%	100.00%
Sheung Tsuen B/T and Car Park	RD_29	RR	30	26.0	Ý	2100-2200	15	28.50%	14.30%	0.00%	14.30%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	28.60%	14.30%	0.00%	0.00%	0.00%	100.00%
Sheung Tsuen B/T and Car Park	RD_29	RR	30	26.0	Y	2200-2300	15	27.20%	18.20%	0.00%	9.10%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	18.20%	27.30%	0.00%	0.00%	0.00%	100.00%
Sneung Tsuen B/T and Car Park	RD_29 RD_30_21	RR	30 30	26.0	Y	2300-0000	15	23.10%	23.10%	0.00%	7.70%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	30.70%	15.40%	0.00%	0.00%	0.00%	100.00%
Village Access Road	RD_30_31	RR	30	26.0	Ý	0100-0200	0	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%
Village Access Road	RD_30_31	RR	30	26.0	Y	0200-0300	0	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%
Village Access Road Village Access Road	RD_30_31 RD_30_31	RR	30 30	26.0 26.0	Y Y	0300-0400	5 5	0.00%	0.00%	0.00%	0.00%	100.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	100.00%
Village Access Road	RD_30_31	RR	30	26.0	Ŷ	0500-0600	5	100.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	100.00%
Village Access Road	RD_30_31	RR	30	26.0	Y	0600-0700	10	58.35%	0.00%	0.00%	0.00%	41.65%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	100.00%
Village Access Road	RD 30_31	RR	30	26.0 26.0	ř Y	0800-0900	10	57.80%	0.00%	0.00%	0.00%	42.20%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	100.00%
Village Access Road	RD_30_31	RR	30	26.0	Y	0900-1000	15	57.80%	0.00%	0.00%	0.00%	42.20%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	100.00%

Road Name	Road ID	Road Type	Speed Limit (km/hr)	Average Speed (km/hr)	Start Emission Estimated by Broad Brush Approach (Y/N)	Hour	Total Vehicles (Veh/hr)	01 - Private Cars	02 - Taxi	03 - Light Goods Vehicles< =2.5t	04 - Light Goods Vehicles 2.5-3.5t	05 - Light Goods Vehicles> 3.5t	06 - Medium Goods Vehicles< =15t	07 - Medium Goods Vehicles1 5-24t	08 - Public Light Buses	09 - Private Light Bus <=3.5t	10 - Private Light Bus >3.5t	11 - Non- franchise d Bus<=6.4 t	12 - Non- franchise d Bus 6.4 15t	13 - Non- franchise d Bus 15- 24t	14 - Franchise d Bus Single Deck	15 - Franchise d Bus Double Deck	16 - Motorcycl es	17 - Heavy Goods Vehicles> 24t	18 - Non- franchise d Bus >24t	Total
Village Access Road	RD_30_31	RR	30	26.0	Y	1000-1100	15	57.80%	0.00%	0.00%	0.00%	42.20%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	100.00%
Village Access Road	RD_30_31	RR	30	26.0	Y	1100-1200	20	41.65%	0.00%	0.00%	0.00%	58.35%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	100.00%
Village Access Road	RD_30_31	RR	30	26.0	Y	1200-1300	10	40.00%	0.00%	0.00%	0.00%	60.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	100.00%
Village Access Road	RD_30_31 RD_30_31	RR	30	26.0	Ý	1300-1400	10	40.00%	0.00%	0.00%	0.00%	60.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	100.00%
Village Access Road	RD_30_31	RR	30	26.0	Ý	1500-1600	10	40.00%	0.00%	0.00%	0.00%	60.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	100.00%
Village Access Road	RD_30_31	RR	30	26.0	Y	1600-1700	10	40.00%	0.00%	0.00%	0.00%	60.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	100.00%
Village Access Road	RD_30_31	RR	30	26.0	Y	1700-1800	10	50.00%	0.00%	0.00%	0.00%	50.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	100.00%
Village Access Road	RD_30_31 RD_30_31	RR	30	26.0	Ý	1900-1900	10	75.00%	0.00%	0.00%	0.00%	25.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	100.00%
Village Access Road	RD_30_31	RR	30	26.0	Ý	2000-2100	10	100.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	100.00%
Village Access Road	RD_30_31	RR	30	26.0	Y	2100-2200	10	50.00%	0.00%	0.00%	0.00%	50.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	100.00%
Village Access Road	RD_30_31	RR	30	26.0	Y	2200-2300	10	100.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	100.00%
Village Access Road Emission Point 19 Paved Road	RD_30_31 EP19_1_7		30	26.0	Y	2300-0000	10	100.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	100.00%
Emission Point 19 Paved Road	EP19 1 7	LD	30	15.0	Ŷ	0100-0200	õ	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%
Emission Point 19 Paved Road	EP19_1_7	LD	30	15.0	Y	0200-0300	0	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%
Emission Point 19 Paved Road	EP19_1_7	LD	30	15.0	Y	0300-0400	0	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%
Emission Point 19 Paved Road	EP19_1_7 EP10_1_7		30	15.0	Y	0400-0500	0	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%
Emission Point 19 Paved Road	EP19_1_7	LD	30	15.0	Ý	0600-0700	0	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%
Emission Point 19 Paved Road	EP19_1_7	LD	30	15.0	Y	0700-0800	68	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	100.00%	0.00%	100.00%
Emission Point 19 Paved Road	EP19_1_7	LD	30	15.0	Y	0800-0900	68	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	100.00%	0.00%	100.00%
Emission Point 19 Paved Road	EP19_1_7	LD	30	15.0	Y	0900-1000	68	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	100.00%	0.00%	100.00%
Emission Point 19 Paved Road	EP19_1_7		30	15.0	Y	1100-1200	68	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	100.00%	0.00%	100.00%
Emission Point 19 Paved Road	EP19_1_7	LD	30	15.0	Ŷ	1200-1300	68	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	100.00%	0.00%	100.00%
Emission Point 19 Paved Road	EP19_1_7	LD	30	15.0	Y	1300-1400	68	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	100.00%	0.00%	100.00%
Emission Point 19 Paved Road	EP19_1_7	LD	30	15.0	Y	1400-1500	68	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	100.00%	0.00%	100.00%
Emission Point 19 Paved Road	EP19_1_7 EP19_1_7		30	15.0	Ý	1600-1600	68	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	100.00%	0.00%	100.00%
Emission Point 19 Paved Road	EP19 1 7	LD	30	15.0	Ŷ	1700-1800	68	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	100.00%	0.00%	100.00%
Emission Point 19 Paved Road	EP19_1_7	LD	30	15.0	Y	1800-1900	68	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	100.00%	0.00%	100.00%
Emission Point 19 Paved Road	EP19_1_7	LD	30	15.0	Y	1900-2000	68	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	100.00%	0.00%	100.00%
Emission Point 19 Paved Road	EP19_1_7 EP10_1_7		30	15.0	Y	2000-2100	68	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	100.00%	0.00%	100.00%
Emission Point 19 Paved Road	EP19_1_7	LD	30	15.0	Ý	2200-2300	68	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	100.00%	0.00%	100.00%
Emission Point 19 Paved Road	EP19_1_7	LD	30	15.0	Y	2300-0000	0	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%
Emission Point 19 Paved Road	EP19_8	LD	30	15.0	Y	0000-0100	0	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%
Emission Point 19 Paved Road	EP19_8 EP10_8		30	15.0	Y	0100-0200	0	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%
Emission Point 19 Paved Road	EP19 8	LD	30	15.0	Ý	0300-0400	0	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%
Emission Point 19 Paved Road	EP19_8	LD	30	15.0	Y	0400-0500	0	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%
Emission Point 19 Paved Road	EP19_8	LD	30	15.0	Y	0500-0600	0	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%
Emission Point 19 Paved Road	EP19_8	LD	30	15.0	Y	0500-0700	0	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%
Emission Point 19 Paved Road	EP19 8	LD	30	15.0	Ý	0800-0900	20	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	100.00%	0.00%	100.00%
Emission Point 19 Paved Road	EP19_8	LD	30	15.0	Y	0900-1000	20	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	100.00%	0.00%	100.00%
Emission Point 19 Paved Road	EP19_8	LD	30	15.0	Y	1000-1100	20	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	100.00%	0.00%	100.00%
Emission Point 19 Paved Road	EP19_8 EP10_8		30	15.0	Y	1200-1200	20	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	100.00%	0.00%	100.00%
Emission Point 19 Paved Road	EP19 8	LD	30	15.0	Ý	1300-1400	20	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	100.00%	0.00%	100.00%
Emission Point 19 Paved Road	EP19_8	LD	30	15.0	Y	1400-1500	20	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	100.00%	0.00%	100.00%
Emission Point 19 Paved Road	EP19_8	LD	30	15.0	Y	1500-1600	20	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	100.00%	0.00%	100.00%
Emission Point 19 Paved Road	EP19_8	LD	30	15.0	Y	1600-1700	20	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	100.00%	0.00%	100.00%
Emission Point 19 Paved Road	EP19_0 EP19_8	LD	30	15.0	Y	1800-1900	20	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	100.00%	0.00%	100.00%
Emission Point 19 Paved Road	EP19_8	LD	30	15.0	Y	1900-2000	20	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	100.00%	0.00%	100.00%
Emission Point 19 Paved Road	EP19_8	LD	30	15.0	Y	2000-2100	20	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	100.00%	0.00%	100.00%
Emission Point 19 Paved Road	EP19_8	LD	30	15.0	Y	2100-2200	20	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	100.00%	0.00%	100.00%
Emission Point 19 Paved Road	EP19_0		30	15.0	Y	2200-2300	20	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%
Emission Point 19 Paved Road	EP19_9_13	LD	30	15.0	Ŷ	0000-0100	õ	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%
Emission Point 19 Paved Road	EP19_9_13	LD	30	15.0	Y	0100-0200	0	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%
Emission Point 19 Paved Road	EP19_9_13	LD	30	15.0	Y	0200-0300	0	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%
Emission Point 19 Paved Road	EP19 9 13	LD	30	15.0	Y	0400-0400	0	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%
Emission Point 19 Paved Road	EP19_9_13	LD	30	15.0	Ŷ	0500-0600	õ	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%
Emission Point 19 Paved Road	EP19_9_13	LD	30	15.0	Y	0600-0700	0	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%
Emission Point 19 Paved Road	EP19_9_13	LD	30	15.0	Y	0700-0800	10	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	100.00%	0.00%	100.00%
Emission Point 19 Paved Road	EP19_9_13 EP19_9_13		30	15.0	r Y	0900-1000	10	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	100.00%	0.00%	100.00%
Emission Point 19 Paved Road	EP19_9_13	LD	30	15.0	Ŷ	1000-1100	10	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	100.00%	0.00%	100.00%
Emission Point 19 Paved Road	EP19_9_13	LD	30	15.0	Y	1100-1200	10	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	100.00%	0.00%	100.00%

Road Name	Road ID	Road Type	Speed Limit (km/hr)	Average Speed (km/hr)	Start Emission Estimated by Broad Brush Approach (Y/N)	Hour	Total Vehicles (Veh/hr)	01 - Private Cars	02 - Taxi	03 - Light Goods Vehicles< =2.5t	04 - Light Goods Vehicles 2.5-3.5t	05 - Light Goods Vehicles> 3.5t	06 - Medium Goods Vehicles< =15t	07 - Medium Goods Vehicles1 5-24t	08 - Public Light Buses	09 - Private Light Bus <=3.5t	10 - Private Light Bus >3.5t	11 - Non- franchise d Bus<=6.4 t	12 - Non- franchise d Bus 6.4- 15t	13 - Non- franchise d Bus 15- 24t	14 - Franchise d Bus Single Deck	15 - Franchise d Bus Double Deck	16 - Motorcycl es	17 - Heavy Goods Vehicles> 24t	18 - Non- franchise d Bus >24t	Total
Emission Deint 10 Reved Based	ED10 0 12		20	15.0	V	1200 1200	10	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	100.00%	0.00%	100.00%
Emission Point 19 Paved Road	EP19_9_13	LD	30	15.0	ř	1200-1300	10	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	100.00%	0.00%	100.00%
Emission Point 19 Paved Road	EP19_9_13 EP10_0_12		30	15.0	ř	1300-1400	10	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	100.00%	0.00%	100.00%
Emission Point 10 Payed Road	EP10_0_12		20	15.0	v	1500 1600	10	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	100.00%	0.00%	100.00%
Emission Point 10 Payed Road	EP10_0_12		20	15.0	v	1600 1700	10	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	100.00%	0.00%	100.00%
Emission Point 10 Paved Road	EP19_9_13 EP10_0_13		30	15.0	v	1700 1900	10	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	100.00%	0.00%	100.00%
Emission Point 10 Payed Road	EP10_0_12		20	15.0	v	1900-1000	10	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	100.00%	0.00%	100.00%
Emission Point 10 Payed Road	EP10_0_12		20	15.0	v	1000-1900	10	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	100.00%	0.00%	100.00%
Emission Point 19 Paved Road	EP10 0 13		30	15.0	v	2000-2100	10	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	100.00%	0.00%	100.00%
Emission Point 19 Paved Road	EP10 0 13		30	15.0	v	2100-2200	10	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	100.00%	0.00%	100.00%
Emission Point 19 Paved Road	EP10 0 13		30	15.0	v	2200-2200	10	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	100.00%	0.00%	100.00%
Emission Point 19 Paved Road	EP19 9 13	I D	30	15.0	Ý	2300-0000	0	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%
Emission Point 19 Paved Road	FP19 14	I D	30	15.0	Ŷ	0000-0100	0	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%
Emission Point 19 Paved Road	FP19_14	I D	30	15.0	Ŷ	0100-0200	0	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%
Emission Point 19 Paved Road	EP19_14	LD	30	15.0	Ý	0200-0300	0	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%
Emission Point 19 Paved Road	EP19 14	LD	30	15.0	Y	0300-0400	0	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%
Emission Point 19 Paved Road	EP19 14	LD	30	15.0	Y	0400-0500	0	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%
Emission Point 19 Paved Road	EP19 14	LD	30	15.0	Y	0500-0600	0	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%
Emission Point 19 Paved Road	EP19 14	LD	30	15.0	Y	0600-0700	0	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%
Emission Point 19 Paved Road	EP19 14	LD	30	15.0	Y	0700-0800	48	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	100.00%	0.00%	100.00%
Emission Point 19 Paved Road	EP19 14	LD	30	15.0	Y	0800-0900	48	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	100.00%	0.00%	100.00%
Emission Point 19 Paved Road	EP19_14	LD	30	15.0	Y	0900-1000	48	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	100.00%	0.00%	100.00%
Emission Point 19 Paved Road	EP19_14	LD	30	15.0	Y	1000-1100	48	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	100.00%	0.00%	100.00%
Emission Point 19 Paved Road	EP19_14	LD	30	15.0	Y	1100-1200	48	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	100.00%	0.00%	100.00%
Emission Point 19 Paved Road	EP19_14	LD	30	15.0	Y	1200-1300	48	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	100.00%	0.00%	100.00%
Emission Point 19 Paved Road	EP19_14	LD	30	15.0	Y	1300-1400	48	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	100.00%	0.00%	100.00%
Emission Point 19 Paved Road	EP19_14	LD	30	15.0	Y	1400-1500	48	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	100.00%	0.00%	100.00%
Emission Point 19 Paved Road	EP19_14	LD	30	15.0	Y	1500-1600	48	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	100.00%	0.00%	100.00%
Emission Point 19 Paved Road	EP19_14	LD	30	15.0	Y	1600-1700	48	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	100.00%	0.00%	100.00%
Emission Point 19 Paved Road	EP19_14	LD	30	15.0	Y	1700-1800	48	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	100.00%	0.00%	100.00%
Emission Point 19 Paved Road	EP19_14	LD	30	15.0	Y	1800-1900	48	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	100.00%	0.00%	100.00%
Emission Point 19 Paved Road	EP19_14	LD	30	15.0	Y	1900-2000	48	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	100.00%	0.00%	100.00%
Emission Point 19 Paved Road	EP19_14	LD	30	15.0	Y	2000-2100	48	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	100.00%	0.00%	100.00%
Emission Point 19 Paved Road	EP19_14	LD	30	15.0	Y	2100-2200	48	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	100.00%	0.00%	100.00%
Emission Point 19 Paved Road	EP19_14	LD	30	15.0	Y	2200-2300	48	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	100.00%	0.00%	100.00%
Emission Point 19 Paved Road	EP19 14	LD	30	15.0	Y	2300-0000	0	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%	0.00%



# Appendix C-2 Composite Emission Factors

						N	O2 Emissio	n								NO Emission							NOX Em	ission			PN	M25 Emission		PN	110 Emission	
Road Name	Road ID	Hour	A	nnualHourM	in	Mo	onthlyHourN	Лin	Mont	thlyHourAver	age	Ar	nnualHourMir	n	Mo	onthlyHourMi	in	Mon	thlyHourAve	rage	Anr	nualHourMir	ı	Mont	hlyHourAve	rage	An	nualHourMin		An	nualHourMin	
			g/km/veh	g/s/m2	tonne/year	g/km/veh	g/s/m2	tonne/year	g/km/veh	g/s/m2	tonne/year	g/km/veh	g/s/m2 1	tonne/year	g/km/veh	g/s/m2	tonne/year	g/km/veh	g/s/m2	tonne/year	g/km/veh	g/s/m2	tonne/year	g/km/veh	g/s/m2	tonne/year	g/km/veh	g/s/m2 t	onne/year	g/km/veh	g/s/m2 t	tonne/year
Chun Yiu Road	RD_1_2 RD_1_2	0000-0100	0.0030941	7.053E-10	3.886E-06 3.916E-06	0.0025024	5.661E-10	3.143E-06 3.161E-06	0.0022395	5.066E-10	2.813E-06	0.058477	1.298E-08	7.205E-05 7.344E-05	0.0465329	1.04/E-08	5.815E-05 0	0.0410785	9.293E-09	5.159E-05	0.0604669 1	1.368E-08	7.594E-05	0.043318	9.8E-09 9.804F-09	5.44E-05	0.0053993	1.221E-09	5.781E-06	0.0057794	1.307E-09	7.258E-06
Chun Yiu Road	RD 1 2	0200-0300	0.0031179	3 527E-10	1 958E-06	0.0025153	2 845E-10	1 579E-06	0.0022358	2 529E-10	1 404F-06	0.058477	6.614E-09	3.672E-05	0.0465103	5 261E-09	2 921E-05	0.0413651	4.679E-09	2 598E-05	0.0615948 6	5 967E-09	3.868E-05	0.0436009	4 932F-09	2 738F-05	0.0053993	6 107E-10	3 39F-06	0.0057794	6 537E-10	3.629E-06
Chun Yiu Road	RD_1_2	0300-0400	0.5292911	5.987E-08	0.0003324	0.480112	5.431E-08	0.0003015	0.4432902	5.014E-08	0.0002784	1.3610254	1.539E-07	0.0008547	1.2345153	1.396E-07	0.0007752	1.1398552	1.289E-07	0.0007158	1.8903165	2.138E-07	0.001187	1.5831454	1.791E-07	0.0009941	0.04252	4.809E-09	2.67E-05	0.0462	5.226E-09	2.901E-05
Chun Yiu Road	RD_1_2	0400-0500	0.2662633	3.012E-08	0.0001672	0.2405158	2.721E-08	0.000151	0.2230663	2.523E-08	0.0001401	0.7102462	8.034E-08	0.000446	0.6386244	7.224E-08	0.000401 0	0.5914935	6.69E-08	0.0003714	0.9765094 1	1.105E-07	0.0006132	0.8145598	9.214E-08	0.0005115	0.0239596	2.71E-09	1.505E-05	0.0259897	2.94E-09	1.632E-05
Chun Yiu Road	RD_1_2	0500-0600	0.0032592	7.373E-10	4.093E-06	0.0025482	5.765E-10	3.2E-06	0.0022575	5.107E-10	2.835E-06	0.0605969	1.371E-08	7.61E-05	0.04704	1.064E-08	5.908E-05	0.041725	9.439E-09	5.24E-05	0.0638561 1	1.445E-08	8.02E-05	0.0439825	9.95E-09	5.524E-05	0.0053993	1.221E-09	6.781E-06	0.0057794	1.307E-09	7.258E-06
Chun Yiu Road	RD_1_2	0600-0700	0.1219966	2.76E-08	0.0001532	0.1099804	2.488E-08	0.0001381	0.1014413	2.295E-08	0.0001274	0.3541316	8.011E-08	0.0004448	0.3141325	7.106E-08	0.0003945	0.2885297	6.527E-08	0.0003624	0.4761282 1	1.077E-07	0.000598	0.389971	8.822E-08	0.0004898	0.0137514	3.111E-09	1.727E-05	0.014874	3.365E-09	1.868E-05
Chun Yiu Road	RD_1_2 RD_1_2	0700-0800	0.1514035	8.300E-08	0.0004756	0.1395377	1.092E-08	0.0004381	0.1289471	1.1293E-08	0.0004049	0.7395932	4.163E-07	0.0023222	0.8126497	5.515E-07	0.0021406	0.0302405	5.116E-07	0.0019977	1.0680982 3	7 249E-07	0.002/9//	0.7651936	4.328E-07	0.0024025	0.0192973	1.0912-08	8.588F-05	0.0209075	1.182E-08	0.304E-U3
Chun Yiu Road	RD 1 2	0900-1000	0.1986289	1.348E-07	0.0007484	0.1845929	1.253E-07	0.0006955	0.1696401	1.151E-07	0.0006392	0.8845559	6.003E-07	0.0033328	0.826197	5.607E-07	0.0031129	0.7658892	5.198E-07	0.0028857	1.0831848	7.351E-07	0.0040243	0.9355293	6.349E-07	0.0035248	0.0231737	1.573E-08	8.731E-05	0.0251269	1.705E-08	9.467E-05
Chun Yiu Road	RD_1_2	1000-1100	0.1974568	1.117E-07	0.00062	0.1835397	1.038E-07	0.0005763	0.1678817	9.495E-08	0.0005271	0.8506823	4.811E-07	0.0026709	0.7937807	4.489E-07	0.0024923	0.7322645	4.141E-07	0.0022991	1.0481391 5	5.928E-07	0.0032909	0.9001462	5.091E-07	0.0028262	0.0226307	1.28E-08	7.106E-05	0.024537	1.388E-08	7.704E-05
Chun Yiu Road	RD_1_2	1100-1200	0.2447272	1.661E-07	0.0009221	0.2279453	1.547E-07	0.0008588	0.2082965	1.414E-07	0.0007848	0.9996463	6.784E-07	0.0037664	0.9357287	6.35E-07	0.0035256	0.8628407	5.856E-07	0.0032509	1.2443735 8	8.445E-07	0.0046885	1.0711373	7.269E-07	0.0040357	0.0264032	1.792E-08	9.948E-05	0.0286435	1.944E-08 (	0.0001079
Chun Yiu Road	RD_1_2	1200-1300	0.3439628	4.28E-07	0.0023759	0.3213982	3.999E-07	0.0022201	0.2933702	3.65E-07	0.0020265	1.5357629	1.911E-06	0.0106083	1.4434533	1.796E-06	0.0099706	1.3299962	1.655E-06	0.0091869	1.8797257 2	2.339E-06	0.0129842	1.6233664	2.02E-06	0.0112134	0.0363175	4.519E-08 0	0.0002509	0.0394364	4.907E-08	0.0002724
Chun Yiu Road	RD_1_2 RD_1_2	1300-1400	0.3704982	5.029E-07	0.002/919	0.34/5829	4./18E-0/	0.0026192	0.3160364	4.29E-07	0.0023815	1.7594255	2.388E-06	0.013258	1.661952	2.256E-06	0.0125235	1.5290056	2.075E-06	0.0115217	2.1299237	2.891E-06	0.0160499	1.845042	2.504E-06	0.0139032	0.0403022	5.4/E-08 U	0003037	0.0437683	5.941E-08 (	0.0003298
Chun Yiu Road	RD 1 2	1500-1600	0.3237104	4.394E-07	0.0024393	0.3048983	4.433E-07	0.0024940	0.2767268	3.756E-07	0.0022013	1.4594208	1.981E-06	0.0109974	1.3810859	1.875E-06	0.0104071	1.265165	1.717E-06	0.0095336	1.7831312	2.42E-06	0.0148003	1.5418918	2.093E-06	0.0116188	0.0378488	4.691E-08 (	0.0002832	0.0410337	5.094E-08 (	0.0002828
Chun Yiu Road	RD_1_2	1600-1700	0.3185416	3.963E-07	0.0022003	0.2977016	3.704E-07	0.0020564	0.2701646	3.361E-07	0.0018662	1.3972778	1.739E-06	0.0096517	1.3131198	1.634E-06	0.0090704	1.2032413	1.497E-06	0.0083114	1.7158194	2.135E-06	0.011852	1.4734058	1.833E-06	0.0101775	0.0336495	4.187E-08 0	0.0002324	0.0365333	4.546E-08	0.0002524
Chun Yiu Road	RD_1_2	1700-1800	0.2672156	3.325E-07	0.0018458	0.2474582	3.079E-07	0.0017093	0.2258572	2.81E-07	0.0015601	1.2545867	1.561E-06	0.008666	1.1670223	1.452E-06	0.0080612	1.072714	1.335E-06	0.0074098	1.5218023 1	1.893E-06	0.0105118	1.2985712	1.616E-06	0.0089699	0.0292953	3.645E-08 0	0.0002024	0.0317964	3.956E-08 (	0.0002196
Chun Yiu Road	RD_1_2	1800-1900	0.1965379	2.223E-07	0.0012342	0.180797	2.045E-07	0.0011353	0.16606	1.878E-07	0.0010428	0.8812258	9.968E-07	0.0055337	0.8125447	9.191E-07	0.0051024	0.7496151	8.479E-07	0.0047072	1.0777638 1	1.219E-06	0.0067679	0.9156752	1.036E-06	0.00575	0.0221925	2.51E-08 0	0.0001394	0.0240654	2.722E-08	0.0001511
Chun Yiu Road	RD_1_2	1900-2000	0.149329	1.52E-07	0.0008439	0.1372377	1.397E-07	0.0007756	0.1264286	1.287E-07	0.0007145	0.688287	7.007E-07	0.0038899	0.6324636	6.438E-07	0.0035744 0	0.5845079	5.95E-07	0.0033034	0.837616 8	3.527E-07	0.0047339	0.7109364	7.237E-07	0.0040179	0.018128	1.845E-08 (	001505	0.0196402	1.999E-08	0.000111
Chun Yiu Road	RD_1_2 RD_1_2	2100-2200	0.11047.94	6.356E-08	0.0003529	0.1070485	5.772E-08	0.0003301	0.1174215	5 313E-08	0.000311	0.5577405	2 523E-07	0.0021209	0.5013061	2 268F-07	0.0012592	0.3724718	2 074F-07	0.0011515	0.6982211	3 159E-07	0.0024800	0.575841	2 605F-07	0.0021084	0.0156134	7.064F-09	3 922F-05	0.01/2491	7.651E-09	4 248F-05
Chun Yiu Road	RD_1_2	2200-2300	0.1229779	4.173E-08	0.0002317	0.1116426	3.788E-08	0.0002103	0.102611	3.482E-08	0.0001933	0.5730518	1.945E-07	0.0010795	0.5138431	1.744E-07	0.000968	0.4682292	1.589E-07	0.0008821	0.6960297	2.362E-07	0.0013112	0.5708402	1.937E-07	0.0010754	0.0145361	4.933E-09	2.738E-05	0.0157403	5.341E-09	2.965E-05
Chun Yiu Road	RD_1_2	2300-0000	0.1707114	5.793E-08	0.0003216	0.1545121	5.243E-08	0.0002911	0.1431124	4.856E-08	0.0002696	0.6070426	2.06E-07	0.0011436	0.5445227	1.848E-07	0.0010258	0.5018066	1.703E-07	0.0009453	0.777754 2	2.639E-07	0.0014652	0.644919	2.188E-07	0.0012149	0.0176744	5.998E-09	3.33E-05	0.0191531	6.499E-09	3.608E-05
Chun Yiu Road	RD_3_4	0000-0100	0.0565378	1.5E-08	4.286E-05	0.0509436	1.351E-08	3.862E-05	0.0468754	1.243E-08	3.553E-05	0.235512	6.247E-08	0.0001785	0.2013326	5.341E-08	0.0001526	0.1808834	4.798E-08	0.0001371	0.2920497	7.747E-08	0.0002214	0.2277589	6.042E-08	0.0001727	0.0085714	2.274E-09	6.498E-06	0.0092435	2.452E-09	7.007E-06
Chun Yiu Road	RD_3_4	0100-0200	0.0922966	2.448E-08	6.997E-05	0.0830054	2.202E-08	6.292E-05	0.0764914	2.029E-08	5.798E-05	0.3568939	9.46/E-08	0.0002705	0.3042679	8.071E-08	0.0002307 0	0.2/36/89	7.26E-08	0.0002075	0.4491906	1.192E-07	0.0003405	0.3501704	9.289E-08	0.0002654	0.0106809	2.833E-09	8.097E-06	0.0115472	3.063E-09	8.753E-06
Chun Yiu Road	RD_3_4 RD_3_4	0200-0300	0.1343137	2 417F-08	6 908E-05	0.1213548	2 18F-08	6.23E-05	0.1124448	2.963E-08	5.738F-05	0.6611654	8 769E-08	0.0002905	0.5420788	7 503E-08	0.0002593	0.3138341	6.379E-08	0.0002394	0.8434166 1	1.373E-07	0.0003923	0.4282989	8 758F-08	0.0003247	0.0146795	2 117F-09	6.05E-06	0.0158845	2 297F-09	1.204E-05 6 563E-06
Chun Yiu Road	RD 3 4	0400-0500	0.1347493	3.574E-08	0.0001021	0.1215311	3.224E-08	9.213E-05	0.112657	2.988E-08	8.54E-05	0.3848566	1.021E-07	0.0002917	0.3427462	9.092E-08	0.0002598	0.3165492	8.397E-08	0.00024	0.5196059 1	1.378E-07	0.0003939	0.4292062	1.139E-07	0.0003254	0.0146795	3.894E-09	1.113E-05	0.0158845	4.214E-09	1.204E-05
Chun Yiu Road	RD_3_4	0500-0600	0.0911249	2.417E-08	6.908E-05	0.0818764	2.172E-08	6.207E-05	0.0759474	2.015E-08	5.757E-05	0.2778126	7.37E-08	0.0002106	0.2442826	6.48E-08	0.0001852	0.2252246	5.975E-08	0.0001707	0.3689375 9	9.787E-08	0.0002797	0.3011721	7.989E-08	0.0002283	0.0115799	3.072E-09	8.778E-06	0.0125094	3.318E-09	9.483E-06
Chun Yiu Road	RD_3_4	0600-0700	0.0744099	2.961E-08	8.461E-05	0.0667738	2.657E-08	7.593E-05	0.06149	2.447E-08	6.992E-05	0.2689051	1.07E-07	0.0003058	0.2301093	9.156E-08	0.0002617	0.2083644	8.291E-08	0.0002369	0.343315 1	1.366E-07	0.0003904	0.2698544	1.074E-07	0.0003068	0.0099888	3.975E-09	1.136E-05	0.0107835	4.291E-09	1.226E-05
Chun Yiu Road	RD_3_4	0700-0800	0.0631723	5.865E-08	0.0001676	0.057491	5.338E-08	0.0001525	0.0525949	4.883E-08	0.0001395	0.273131	2.536E-07	0.0007247	0.2415574	2.243E-07	0.0006409	0.2180388	2.024E-07	0.0005785	0.3363033 3	3.122E-07	0.0008923	0.2706336	2.513E-07	0.000718	0.0096003	8.913E-09	2.547E-05	0.0103601	9.619E-09	2.749E-05
Chun Yiu Road	RD_3_4 RD_3_4	0900-0900	0.1004278	9.881E-08	0.0002824	0.1019691	9.000E-08 9.467E-08	0.0002591	0.0889919	8.202E-08	0.0002361	0.3729050	3.463E-07	0.0009896	0.3360894	3.12E-07 3.239E-07	0.0008917 0	0.30383331	2.821E-07 2.929E-07	0.0008062	0.4793934 4	4.451E-07 4.59E-07	0.0012719	0.392847	3.04/E-0/	0.0010423	0.0120983	1.1/9E-08	3.309E-05	0.0137329	1.275E-08	3.044E-05 3.741E-05
Chun Yiu Road	RD 3 4	1000-1100	0.1159494	1.077E-07	0.0003076	0.1070359	9.938E-08	0.000284	0.0971319	9.018E-08	0.0002577	0.4005817	3.719E-07	0.0010628	0.3643243	3.383E-07	0.0009666	0.3282967	3.048E-07	0.000871	0.5165311 4	1.796E-07	0.0013705	0.4254286	3.95E-07	0.0011287	0.013392	1.243E-08	3.553E-05	0.0144888	1.345E-08	3.844E-05
Chun Yiu Road	RD_3_4	1100-1200	0.1448996	1.345E-07	0.0003844	0.1341734	1.246E-07	0.000356	0.1217245	1.13E-07	0.000323	0.4686695	4.351E-07	0.0012435	0.4294606	3.987E-07	0.0011394	0.3875295	3.598E-07	0.0010282	0.6135691 5	5.697E-07	0.0016279	0.509254	4.728E-07	0.0013512	0.0154466	1.434E-08	4.098E-05	0.0167258	1.553E-08	4.438E-05
Chun Yiu Road	RD_3_4	1200-1300	0.3546404	5.174E-07	0.0014786	0.3312254	4.833E-07	0.001381	0.3021407	4.408E-07	0.0012597	1.5120765	2.206E-06	0.0063043	1.4209529	2.073E-06	0.0059244	1.309004	1.91E-06	0.0054577	1.8667169	2.724E-06	0.0077829	1.6111447	2.351E-06	0.0067174	0.036741	5.36E-08 (	0.0001532	0.0398965	5.821E-08 (	0.0001663
Chun Yiu Road	RD_3_4	1300-1400	0.3858444	6.653E-07	0.0019012	0.3617642	6.238E-07	0.0017825	0.3286109	5.666E-07	0.0016192	1.7395292	2.999E-06	0.0085713	1.6427529	2.833E-06	0.0080945	1.5107672	2.605E-06	0.0074441	2.1253736	3.665E-06	0.0104725	1.839378	3.172E-06	0.0090633	0.0409572	7.062E-08 0	0.0002018	0.0444808	7.67E-08	0.0002192
Chun Yiu Road	RD_3_4	1400-1500	0.3659678	5.825E-07	0.0015548	0.3434041	5.466E-07	0.0015619	0.3111975	4.953E-07	0.0014154	1.6209381	2.58E-06	0.00/3/26	1.5311187	2.43/E-06	0.0069641	1.4043903	2.235E-06	0.0063877	1.9869059 :	3.162E-06	0.0090371	1./1558/8	2.731E-06	0.0078031	0.035865	5.708E-08 (	0001/5/	0.0419495	6.198F-08 (	0.0001908
Chun Yiu Road	RD 3 4	1600-1700	0.3229988	4.712E-07	0.0013467	0.3017613	4.403E-07	0.0012581	0.2737008	3.993E-07	0.0011411	1.3711909	2.001E-06	0.0057169	1.2882869	1.88E-06	0.0053713	1.1801096	1.722E-06	0.0049203	1.6941897	2.472E-06	0.0070636	1.4538104	2.121E-06	0.0060614	0.0336718	4.913E-08 (	0.0001404	0.0365572	5.334E-08 (	0.0001524
Chun Yiu Road	RD_3_4	1700-1800	0.2835079	4.136E-07	0.001182	0.2624572	3.829E-07	0.0010943	0.2394388	3.493E-07	0.0009983	1.2664331	1.848E-06	0.0052802	1.1786487	1.72E-06	0.0049142	1.08413	1.582E-06	0.0045201	1.5499409	2.261E-06	0.0064622	1.3235687	1.931E-06	0.0055184	0.0304249	4.439E-08 0	0.0001269	0.0330242	4.818E-08	0.0001377
Chun Yiu Road	RD_3_4	1800-1900	0.2104854	2.792E-07	0.0007978	0.1935368	2.567E-07	0.0007336	0.1776613	2.356E-07	0.0006734	0.8913996	1.182E-06	0.0033787	0.8220365	1.09E-06	0.0031158	0.758448	1.006E-06	0.0028747	1.1018851 1	1.461E-06	0.0041765	0.9361093	1.242E-06	0.0035481	0.0230754	3.061E-08	8.746E-05	0.0250258	3.319E-08	9.485E-05
Chun Yiu Road	RD_3_4	1900-2000	0.15557	1.857E-07	0.0005307	0.1428973	1.706E-07	0.0004875	0.1315662	1.571E-07	0.0004488	0.6735439	8.04E-07	0.0022976	0.6188742	7.388E-07	0.0021111	0.5719968	6.828E-07	0.0019512	0.8291138 9	9.897E-07	0.0028283	0.703563	8.399E-07	0.0024	0.0184091	2.198E-08	6.28E-05	0.0199452	2.381E-08	6.804E-05
Chun Yiu Road	RD_3_4	2000-2100	0.1352036	8.966E-08	0.0002562	0.1241208	8.231E-08	0.0002352	0.1147106	7.607E-08	0.0002174	0.677747	4.495E-07	0.0012844	0.6214916	4.122E-07	0.00011778	0.5754602	3.816E-07	0.0010906	0.8129507 5	5.391E-07	0.0015407	0.6901708	4.577E-07	0.001308	0.0171222	1.136E-08	3.245E-05	0.018547	1.23E-08	3.515E-05
Chun Yiu Road	RD 3 4	2200-2300	0.171015	6.805E-08	0.0001945	0.1552201	6.176E-08	0.0001765	0.1425949	5.674E-08	0.0001621	0.6082541	2.42E-07	0.0006916	0.5470873	2.177E-07	0.0006221	0.4998817	1.989E-07	0.0005684	0.7792691 3	3.101E-07	0.0008861	0.6424766	2.556E-07	0.0007306	0.0176744	7.033E-09	2.01E-05	0.0191531	7.621E-09	2.178E-05
Chun Yiu Road	RD_3_4	2300-0000	0.2028138	8.07E-08	0.0002306	0.1835951	7.305E-08	0.0002088	0.1700428	6.766E-08	0.0001934	0.6744663	2.684E-07	0.0007669	0.6060663	2.412E-07	0.0006892	0.5590226	2.224E-07	0.0006357	0.8772801	3.491E-07	0.0009975	0.7290654	2.901E-07	0.000829	0.01991	7.922E-09	2.264E-05	0.0215867	8.589E-09	2.455E-05
Chun Yiu Road	RD_5_6	0000-0100	0.0565378	1.689E-08	1.554E-05	0.0509436	1.522E-08	1.401E-05	0.0468754	1.4E-08	1.289E-05	0.235512	7.034E-08	6.475E-05	0.2013326	6.014E-08	5.535E-05 (	0.1808834	5.403E-08	4.973E-05	0.2920497 8	3.723E-08	8.029E-05	0.2277589	6.803E-08	6.262E-05	0.0085714	2.56E-09	2.357E-06	0.0092435	2.761E-09	2.541E-06
Chun Yiu Road	RD_5_6	0100-0200	0.0922966	2.757E-08	2.538E-05	0.0830054	2.479E-08	2.282E-05	0.0764914	2.285E-08	2.103E-05	0.3568939	1.066E-07	9.812E-05	0.3042679	9.088E-08	8.365E-05 0	0.2736789	8.174E-08	7.524E-05	0.4491906 1	1.342E-07	0.0001235	0.3501704	1.046E-07	9.627E-05	0.0106809	3.19E-09	2.937E-06	0.0115472	3.449E-09	3.175E-06
Chun Yiu Road	RD_5_6	0200-0300	0.1343137	4.012E-08	3.6932-05	0.1213548	3.625E-08	3.336E-05	0.1124448	3.359E-08	3.091E-05	0.5832441	1.145E-07	0.0001054	0.5420786	1.022E-07	9.405E-05 0	0.3158541	9.434E-08	8.684E-05	0.51/55/8 1	1 365 07	0.0001423	0.4282989	1.2/9E-0/	0.0001178	0.0146/95	4.385E-09	4.036E-06	0.0172140	4./44E-09	4.36/E-U6
Chun Yiu Road	RD 5 6	0400-0500	0.1347493	4.025E-08	3.705E-05	0.1215311	3.63E-08	3.341E-05	0.112657	3.365E-08	3.097E-05	0.3848566	1.15E-07	0.0001058	0.3427462	1.024E-07	9.423E-05 (	0.3165492	9.455E-08	8.703E-05	0.5196059 1	1.552E-07	0.0001139	0.4292062	1.282E-07	0.000118	0.0139025	4.385E-09	4.036E-06	0.0173145	4.744E-09	4.367E-06
Chun Yiu Road	RD_5_6	0500-0600	0.0911249	2.722E-08	2.505E-05	0.0818764	2.446E-08	2.251E-05	0.0759474	2.268E-08	2.088E-05	0.2778126	8.298E-08	7.638E-05	0.2442826	7.296E-08	6.716E-05	0.2252246	6.727E-08	6.192E-05	0.3689375	1.102E-07	0.0001014	0.3011721	8.996E-08	8.28E-05	0.0115799	3.459E-09	3.184E-06	0.0125094	3.736E-09	3.439E-06
Chun Yiu Road	RD_5_6	0600-0700	0.0744099	3.334E-08	3.069E-05	0.0667738	2.992E-08	2.754E-05	0.06149	2.755E-08	2.536E-05	0.2689051	1.205E-07	0.0001109	0.2301093	1.031E-07	9.49E-05 0	0.2083644	9.335E-08	8.593E-05	0.343315	1.538E-07	0.0001416	0.2698544	1.209E-07	0.0001113	0.0099888	4.475E-09	4.119E-06	0.0107835	4.831E-09	4.447E-06
Chun Yiu Road	RD_5_6	0700-0800	0.5492961	1.231E-06	0.0011326	0.511046	1.145E-06	0.0010538	0.4766303	1.068E-06	0.0009828	3.3829868	7.578E-06	0.0069756	3.1831802	7.131E-06	0.0065636	3.004867	6.731E-06	0.006196	3.9322829 8	3.809E-06	0.0081083	3.4814973	7.799E-06	0.0071788	0.0676571	1.516E-07 (	0.0001395	0.0735146	1.647E-07 (	0.0001516
Chun Yiu Road	RD_5_6	0800-0900	0.556/605	1.24/E-06	0.001148	0.5201416	1.165E-06	0.0010725	0.4831243	1.082E-06	0.0009962	3.355/808	1.245E-05	0.0069195	3.1690244	7.099E-06	0.0065345	2.9818315	6.68E-06	0.0061485	4 619846 1	5.765E-06	0.0080676	3.4649558	1.289F-05	0.00/144/	0.0677979	1.519E-07 (	0001398	0.073669	1.65E-07 0 2.706E-07 0	0.0001519
Chun Yiu Road	RD 5 6	1000-1100	0.6635809	2.081E-06	0.0019156	0.6229587	1.954E-06	0.0017983	0.5768313	1.809E-06	0.0016652	4.0579897	1.273E-05	0.0117145	3.8493217	1.207E-05	0.0111121	3.6136548	1.133E-05	0.0104318	4.7215706 1	1.481E-05	0.0136301	4.190486	1.314E-05	0.012097	0.0809118	2.538E-07 0	0.0002336	0.0879342	2.758E-07 (	0.0002538
Chun Yiu Road	RD_5_6	1100-1200	0.6687611	2.097E-06	0.0019306	0.628877	1.972E-06	0.0018154	0.5819515	1.825E-06	0.00168	4.0540391	1.271E-05	0.0117031	3.8513455	1.208E-05	0.0111179	3.6138019	1.133E-05	0.0104322	4.7228002 1	1.481E-05	0.0136336	4.1957534	1.316E-05	0.0121122	0.0809975	2.54E-07 0	0.0002338	0.0880281	2.761E-07 (	0.0002541
Chun Yiu Road	RD_5_6	1200-1300	0.657628	2.554E-06	0.0023504	0.6186746	2.402E-06	0.0022112	0.5702204	2.214E-06	0.002038	3.8213585	1.484E-05	0.0136579	3.6314618	1.41E-05	0.0129792	3.3955994	1.318E-05	0.0121362	4.4789866 1	1.739E-05	0.0160083	3.9658198	1.54E-05	0.0141742	0.0776745	3.016E-07 (	0.0002776	0.0844164	3.278E-07 (	0.0003017
Chun Yiu Road	RD_5_6	1300-1400	0.6474774	2.707E-06	0.0024922	0.6105207	2.553E-06	0.0023499	0.5595717	2.34E-06	0.0021538	3.7558853	1.571E-05	0.0144565	3.575918	1.495E-05	0.0137638	3.3281953	1.392E-05	0.0128103	4.4033628 1	1.841E-05	0.0169486	3.8877671	1.626E-05	0.0149641	0.0764183	3.196E-07 (	0.0002941	0.0830496	3.473E-07 (	0.0003197
Chun Yiu Road	RD_5_6	1400-1500	0.651034	2.625E-06	0.0024164	0.6035824	2.479E-06	0.0022814	0.5535312	2.269E-06	0.0020881	3.795015	1.53E-05	0.0140854	3.5168836	1.458E-05	0.0134243 :	3.3626392	1.356E-05	0.01224806	4.446049 1	1.793E-05 1.755E-05	0.0165018	3.9252461	1.583E-05	0.0145688	0.0756365	3.104E-07 (	0002857	0.0836564	3.3/3E-0/ 0 3.315E-07 0	0.0003105
Chun Yiu Road	RD 5 6	1600-1700	0.636238	2.565E-06	0.0023614	0.5989699	2.415E-06	0.0022231	0.5495611	2.216E-06	0.0020397	3.7132081	1.497E-05	0.0137818	3.5304832	1.424E-05	0.0131036	3.2887922	1.326E-05	0.0122065	4.3494462 1	1.754E-05	0.0161432	3.8383533	1.548E-05	0.0142463	0.0754784	3.043E-07 (	0.0002801	0.0820274	3.308E-07 (	0.0003044
Chun Yiu Road	RD_5_6	1700-1800	0.6224525	2.417E-06	0.0022247	0.5812959	2.257E-06	0.0020776	0.5361243	2.082E-06	0.0019162	3.6913777	1.433E-05	0.0131933	3.4853752	1.353E-05	0.0124571	3.260104	1.266E-05	0.0116519	4.3138303	1.675E-05	0.015418	3.7962283	1.474E-05	0.0135681	0.0743008	2.885E-07 0	0.0002656	0.0807459	3.135E-07 (	0.0002886
Chun Yiu Road	RD_5_6	1800-1900	0.6122991	2.286E-06	0.0021042	0.5695551	2.126E-06	0.0019573	0.5292877	1.976E-06	0.001819	3.6722839	1.371E-05	0.0126203	3.4564123	1.29E-05	0.0118784	3.2535629	1.215E-05	0.0111813	4.284583	1.6E-05	0.0147245	3.7828506	1.412E-05	0.0130002	0.0737616	2.754E-07 0	0.0002535	0.0801576	2.993E-07 (	0.0002755
Chun Yiu Road	RD_5_6	1900-2000	0.5721203	1.794E-06	0.0016516	0.5320106	1.668E-06	0.0015358	0.4958717	1.555E-06	0.0014315	3.4555504	1.084E-05	0.0099754	3.251259	1.02E-05	0.0093856	3.0679699	9.622E-06	0.0088565	4.0276707 1	1.263E-05	0.0116269	3.5638417	1.118E-05	0.010288	0.069625	2.184E-07	0.000201	0.0756562	2.373E-07 (	0.0002184
Chun Yiu Road	KD_5_6 RD 5_6	2000-2100	0.0010016	1.673E-06	0.0015307	0.61442/7	1.56E-06	0.0014359	0.5744861	1.459E-06	0.0013425	4.0899373	1.038E-05	0.0095578	3.84/331 4.0302766	9.768E-06	0.0089909	3 8176004	9.241E-U6 9.122E-06	0.0083066	4.7509988 1	1.206E-05	0.0111026	4.21449	1.0/E-05	0.0098489	0.0809185	2.054E-07 (	0001891	0.08/9413	2.233E-07 ( 2.206E-07	0.0002055
Chun Yiu Road	RD 5 6	2200-2300	0.751235	1.683E-06	0.001549	0.6951558	1.557E-06	0.0014334	0.6499645	1.456E-06	0.0013402	4.6263158	1.036E-05	0.0095394	4.3372054	9.716E-06	0.0089432	4.1045569	9.195E-06	0.0084635	5.3775508 1	1.205E-05	0.0110884	4.7545214	1.065E-05	0.0098037	0.0909503	2.037E-07 0	0.0001875	0.0988542	2.214E-07 (	0.0002038
Chun Yiu Road	RD_5_6	2300-0000	0.2028138	9.087E-08	8.364E-05	0.1835951	8.226E-08	7.571E-05	0.1700428	7.618E-08	7.012E-05	0.6744663	3.022E-07	0.0002781	0.6060663	2.715E-07	0.0002499	0.5590226	2.505E-07	0.0002305	0.8772801	3.93E-07	0.0003618	0.7290654	3.266E-07	0.0003007	0.01991	8.92E-09	8.211E-06	0.0215867	9.671E-09	8.902E-06
Chun Yiu Road	RD_7_8	0000-0100	0.0540437	2.352E-08	4.666E-05	0.048683	2.118E-08	4.203E-05	0.0447924	1.949E-08	3.867E-05	0.2271988	9.886E-08	0.0001961	0.1940979	8.445E-08	0.0001676	0.1743592	7.587E-08	0.0001505	0.2812425 1	1.224E-07	0.0002428	0.2191516	9.536E-08	0.0001892	0.0084234	3.665E-09	7.272E-06	0.0090819	3.952E-09	7.841E-06
Chun Yiu Road	RD_7_8	0100-0200	0.0700689	2.033E-08	4.033E-05	0.0629438	1.826E-08	3.623E-05	0.0579835	1.682E-08	3.337E-05	0.2825137	8.195E-08	0.0001626	0.2400276	6.963E-08	0.0001381	0.2157096	6.257E-08	0.0001242	0.3525826 1	1.023E-07	0.0002029	0.273693	7.939E-08	0.0001575	0.0093645	2.716E-09	5.39E-06	0.0101095	2.933E-09	5.819E-06
Chun Yiu Road	RD 7 8	0200-0300	0.0904943	2.020E-U8	3.208E-05 7 745E-05	0.2434235	2.309E-08 3 531F-09	4.7E-05 7.005F-05	0.075035	2.194E-08 3.256F-09	+.303E-05	0.2747719	1.97E-08	0.0001581	0.2433368	1.059E-08	0.0001401 0	0.2241748	9.6705-08	0.000129	0.3052001	1.00E-07	0.0002102 0	0.2998098	0.097E-08	0.0001726	0.0115/99	3.359E-09	6 507F-06	0.0125094	3.029E-09	7.0635-06
Chun Yiu Road	RD 7 8	0400-0500	0.1347493	3.909E-08	7.756E-05	0.1215311	3.525E-08	6.995E-05	0.112657	3.268E-08	6.484E-05	0.3848566	1.116E-07	0.0002215	0.3427462	9.942E-08	0.0001973 (	0.3165492	9.182E-08	0.0001313	0.5196059 1	1.507E-07	0.0002991	0.4292062	1.245E-07	0.0002303	0.0146795	4.258E-09	8.449E-06	0.0158845	4.608E-09	9.142E-06
Chun Yiu Road	RD_7_8	0500-0600	0.0692244	2.008E-08	3.984E-05	0.0621039	1.801E-08	3.574E-05	0.0575803	1.67E-08	3.314E-05	0.2236717	6.488E-08	0.0001287	0.1951201	5.66E-08	0.0001123	0.1794875	5.207E-08	0.0001033	0.2928962 8	8.496E-08	0.0001686	0.2370678	6.877E-08	0.0001364	0.0100394	2.912E-09	5.778E-06	0.010832	3.142E-09	6.234E-06
Chun Yiu Road	RD_7_8	0600-0700	0.0797129	5.781E-08	0.0001147	0.0716648	5.197E-08	0.0001031	0.0660462	4.79E-08	9.503E-05	0.2996749	2.173E-07	0.0004312	0.2602651	1.887E-07	0.0003745	0.2368653	1.718E-07	0.0003408	0.3793878	2.751E-07	0.0005459	0.3029115	2.197E-07	0.0004359	0.01062	7.702E-09	1.528E-05	0.0114699	8.318E-09	1.65E-05
Chun Yiu Road	RD_7_8	0700-0800	0.462328	1.274E-06	0.0025279	0.4298323	1.184E-06	0.0023502	0.4006261	1.104E-06	0.0021905	2.7986728	7.712E-06	0.0153024	2.6309598	7.25E-06	0.0143854	2.4818783	6.839E-06	0.0135703	3.2610008 8	8.986E-06	0.0178303	2.8825044	7.943E-06	0.0157608	0.0570784	1.573E-07 (	0.0003121	0.0620062	1.709E-07	0.000339
Chun Yiu Road	RD 7 8	0900-1000	0.5755403	2.17F-06	0.0027014	0.539559	2.035F-06	0.0025212	0.4064452	1.889F-06	0.0023393	2./519090	1.304F-05	0.015839	3.2768468	1.236F-05	0.024518	2.4394903	1.162F-05	0.0230627	4.0343736	1.521F-05	0.0185404 .	2.0459410	0.255E-Ub 1.351F-05	0.0268107	0.0508077	2.644F-07 (	.0005245	0.0761764	2.873E-07	0.0003556
Chun Yiu Road	RD_7_8	1000-1100	0.5808489	2.19E-06	0.004346	0.5449137	2.055E-06	0.0040771	0.5041333	1.901E-06	0.003772	3.4750803	1.31E-05	0.0260012	3.2938111	1.242E-05	0.0246449	3.0895666	1.165E-05	0.0231167	4.0559292 1	1.529E-05	0.0303472	3.5937	1.355E-05	0.0268887	0.0704027	2.655E-07 0	0.0005268	0.0765029	2.885E-07 (	0.0005724
Chun Yiu Road	RD_7_8	1100-1200	0.5942289	2.241E-06	0.0044461	0.5583741	2.106E-06	0.0041779	0.516205	1.947E-06	0.0038623	3.5097315	1.324E-05	0.0262604	3.3316225	1.256E-05	0.0249278	3.1232363	1.178E-05	0.0233686	4.1039603	1.548E-05	0.0307066	3.6394413	1.372E-05	0.027231	0.0713007	2.689E-07 0	0.0005335	0.0774807	2.922E-07 (	0.0005797
Chun Yiu Road	RD_7_8	1200-1300	0.5536791	3.052E-06	0.0060548	0.5201306	2.867E-06	0.0056879	0.4784093	2.637E-06	0.0052316	3.0635137	1.688E-05	0.033501	2.9056093	1.601E-05	0.0317743	2.7098326	1.494E-05	0.0296334	3.6171929 1	1.994E-05	0.0395558	3.188242	1.757E-05	0.034865	0.0638854	3.521E-07 (	0.0006986	0.0694199	3.826E-07 (	0.0007591
Chun Yiu Road	RD_7_8	1300-1400	0.5506067	3.274E-06	0.0064965	0.5185983	3.084E-06	0.0061188	0.4744855	2.822E-06	0.0055984	3.075699	1.829E-05	0.0362896	2.9236634	1.739E-05	0.0344958	2.7147876	1.614E-05	0.0320313	3.6263057	2.156E-05	0.0427861	3.1892731	1.897E-05	0.0376297	0.0639198	3.801E-07 0	0.0007542	0.0694562	4.13E-07 0	0.0008195
Chun Yiu Road	RD 7 8	1400-1500	0.5377604	3.196E-06	0.0061902	0.519//34	3.015E-06	0.00598584	0.4/4/837	2.754E-U6	0.005361	2 9863383	1.783E-05	0.0303/99	2.9243893	1.097E-05	0.0330628	2.7119009	1.573E-05	0.0312168	3.024/81/ 3	2.103E-05 2.045E-05	0.041/25	3.1118991	1.805F-05	0.0358212	0.0036659	3.705E-07 (	0007352	0.0093975	4.020E-07 (	0.0007807
Chun Yiu Road	RD 7 8	1600-1700	0.5360904	2.955E-06	0.0058624	0.5039805	2.778E-06	0.0055113	0.4614364	2.543E-06	0.005046	2.9834768	1.644E-05	0.0326258	2.831492	1.561E-05	0.0309638	2.6308684	1.45E-05	0.0287698	3.5195672	1.94E-05	0.0384882	3.0923049	1.704E-05	0.0338159	0.0622293	3.43E-07 0	0.0006805	0.0676179	3.727E-07 (	0.0007394
Chun Yiu Road	RD_7_8	1700-1800	0.5058123	2.788E-06	0.0055313	0.4717261	2.6E-06	0.0051586	0.4343401	2.394E-06	0.0047497	2.9089559	1.603E-05	0.0318109	2.7407657	1.511E-05	0.0299716	2.5572869	1.409E-05	0.0279652	3.4147682 1	1.882E-05	0.0373422	2.9916271	1.649E-05	0.0327149	0.059658	3.288E-07 0	0.0006524	0.0648202	3.573E-07	0.0007088
Chun Yiu Road	RD_7_8	1800-1900	0.4945086	2.51E-06	0.0049808	0.4594064	2.332E-06	0.0046272	0.426363	2.164E-06	0.0042944	2.8848282	1.464E-05	0.0290565	2.7101904	1.376E-05	0.0272975	2.5465183	1.293E-05	0.025649	3.3793368 1	1.715E-05	0.0340372	2.9728812	1.509E-05	0.0299433	0.0591335	3.002E-07 0	0.0005956	0.0642468	3.261E-07 (	0.0006471
Chun Yiu Road	RD_7_8	1900-2000	0.4422899	1.924E-06	0.0038184	0.4107625	1.787E-06	0.0035462	0.3823812	1.664E-06	0.0033012	2.6060846	1.134E-05	0.0224991	2.4470369	1.065E-05	0.021126	2.3049641	1.003E-05	0.0198994	3.0483745	1.326E-05	0.0263175	2.6873453	1.169E-05	0.0232006	0.053758	2.339E-07 0	0.0004641	0.0583973	2.541E-07 (	0.0005042
Chun Yiu Road	RD_7_8	2000-2100	0.540026	1.723E-06	0.0035382	0.5016772	1.601E-06	0.0031762	0.4688491 0.5064061	1.496E-06	0.0029683	3.3374586	1.065E-05	0.0211297	3.1357464	1.001E-05	0.0198526	2.9638017	9.457E-06	0.0188824	3.8774846 1 4.1127389 1	1.237E-05	0.0245486	3.4326508	1.095E-05	0.0217324	0.0565218	2.123E-07 (	0004212	0.0722817	2.306E-07 (	0.0004576
Chun Yiu Road	RD_7_8	2200-2300	0.6557329	1.712E-06	0.0033967	0.6065027	1.583E-06	0.0031417	0.5668463	1.48E-06	0.0029362	4.0217063	1.05E-05	0.0208323	3.7669462	9.834E-06	0.0195127	3.5622477	9.3E-06	0.0184523	4.6774392 1	1.221E-05	0.024229	4.1290939	1.078E-05	0.0213886	0.0794571	2.074E-07 (	0.0004116	0.0863533	2.254E-07 (	0.0004473

						NO2 Emission								NO Emission							NOX Em	nission			PI	M25 Emission	n	PI	M10 Emissio	n
Road Name	Road ID	Hour	Α	nnualHourMin		MonthlyHourMin	Mo	nthlyHourAver	age	An	inualHourMin	1	M	onthlyHourM	in	Mont	thlyHourAve	erage	Ar	nnualHourMi	n	Mon	thlyHourAve	erage	An	nnualHourMi	n	An	nnualHourMi	n
			g/km/veh	g/s/m2 tonne/ye	ar g/km/veh	g/s/m2 tonne/y	ar g/km/veh	g/s/m2	tonne/year g/	km/veh	g/s/m2 t	tonne/year	g/km/veh	g/s/m2	tonne/year	g/km/veh	g/s/m2	tonne/year	g/km/veh	g/s/m2	tonne/year	g/km/veh	g/s/m2	tonne/year	g/km/veh	g/s/m2	tonne/year	g/km/veh	g/s/m2	tonne/year
Chun Yiu Koad Kam Tin Road	RD_7_8 RD_09_10	2300-0000	0.1640456	1.428E-07 0.00028	33 0.1484595 32 0.0302185	3 1.292E-07 0.00025 5 1.816E-07 0.00061	63 0.13/5003 48 0.0367868	1.19/E-0/ 1.703E-07	0.0002374 0.5	1175514	1.933E-06 (	0.0009968	0.51/6455	4.505E-07	0.0008938	0.3559452	4.151E-07	0.0008237	0.7413548	6.452E-07	0.0012801	0.5145345	5.348E-07	0.0010611	0.01/1623	1.494E-08	2.963E-05	0.0185946	1.518E-08	3.211E-05
Kam Tin Road	RD_09_10	0100-0100	0.0420303	1.576E-07 0.00038	96 0.035218.	1 1 05F-07 0 00035	48 0.0307808 54 0.0353896	9.83E-08	0.0003329 0.4	1407876	1.933L-00 1 1.224E-06	0.0003437	0.3793809	1.095E-06	0.0033473	0.35559452	1.048E-00	0.0034509	0.4822059	1 339E-06	0.0045355	0.332732	1.018L-00	0.0001300	0.0090923	2 495E-08	8.45E-05	0.0098033	2 723E-08	9.221E-05
Kam Tin Road	RD 09 10	0200-0300	0.0567319	1.313E-07 0.00044	47 0.0512696	5 1.187E-07 0.00040	19 0.0476729	1.104E-07	0.0003737 0.2	2767085	6.405E-07 (	0.0021689	0.2365662	5.476E-07	0.0018542	0.2145045	4.965E-07	0.0016813	0.3334403	7.719E-07	0.0026136	0.2621775	6.069E-07	0.002055	0.0073179	1.694E-08	5.736E-05	0.0079267	1.835E-08	6.213E-05
Kam Tin Road	RD_09_10	0300-0400	0.0836494	1.646E-07 0.00055	73 0.0756631	1.489E-07 0.00050	41 0.0699733	1.377E-07	0.0004662 0.3	3914952	7.703E-07 (	0.0026083	0.3356324	6.604E-07	0.0022361	0.3029318	5.96E-07	0.0020183	0.4751446	9.349E-07	0.0031656	0.3729051	7.337E-07	0.0024845	0.009705	1.91E-08	6.466E-05	0.0105835	2.082E-08	7.051E-05
Kam Tin Road	RD_09_10	0400-0500	0.0690722	1.679E-07 0.00056	85 0.0624415	5 1.518E-07 0.00051	39 0.0581182	1.413E-07	0.0004783 0.	343123	8.34E-07 (	0.0028239	0.2956718	7.186E-07	0.0024334	0.2692587	6.544E-07	0.002216	0.4121953	1.002E-06	0.0033924	0.3273769	7.957E-07	0.0026943	0.0086245	2.096E-08	7.098E-05	0.0093926	2.283E-08	7.73E-05
Kam Tin Road	RD_09_10	0500-0600	0.155439	6.297E-07 0.00213	21 0.1423793	3 5.768E-07 0.0019	3 0.1339652	5.427E-07	0.0018376 0.8	3046537	3.26E-06 (	0.0110373	0.7326022	2.968E-06	0.010049	0.6906529	2.798E-06	0.0094736	0.9600928	3.889E-06	0.0131694	0.8246181	3.34E-06	0.0113111	0.0166506	6.745E-08	0.0002284	0.0181329	7.346E-08	0.0002487
Kam Tin Road	RD_09_10 RD_09_10	0700-0700	0.1352251	3 146E-06 0.01065	27 0.00000000	2 011F-06 0.00225	54 0.062522 67 0.1162893	2 7055-06	0.0021563 0.	348869	1.042E-05 (	0.020708	0.5427643	1 704F-05	0.018/189	0.5090437	1.676E-05	0.01/5559	0.0730103	2 257E-05	0.0232108	0.3713037	1.046F-05	0.0197122	0.0136620	1.414E-07	0.0004788	0.0152606	1.554E-07	0.0005265
Kam Tin Road	RD 09 10	0800-0900	0.1254145	2.889E-06 0.00978	11 0.1167571	1 2.689E-06 0.00910	59 0.1082753	2.494E-06	0.0084444 0.8	3048373	1.854E-05 (	0.0627692	0.7474431	1.722E-05	0.058293	0.6961401	1.603E-05	0.0542919	0.9302518	2.143E-05	0.0725503	0.8044154	1.853E-05	0.0627363	0.0184724	4.255E-07	0.0014407	0.0202237	4.658E-07	0.0015772
Kam Tin Road	RD_09_10	0900-1000	0.1628138	3.75E-06 0.01269	78 0.1524959	3.512E-06 0.01189	31 0.1415796	3.261E-06	0.0110418 0.9	948943	2.291E-05 (	0.0775917	0.933479	2.15E-05	0.072802	0.8722709	2.009E-05	0.0680283	1.1577081	2.666E-05	0.0902896	1.0138505	2.335E-05	0.0790701	0.021244	4.893E-07	0.0016568	0.02315	5.332E-07	0.0018055
Kam Tin Road	RD_09_10	1000-1100	0.1606499	3.477E-06 0.01177	36 0.1506408	3 3.26E-06 0.0110	4 0.1394935	3.019E-06	0.0102231 1.	010904	2.188E-05 (	0.0740861	0.9489833	2.054E-05	0.0695481	0.8847846	1.915E-05	0.0648432	1.171554	2.536E-05	0.0858597	1.0242781	2.217E-05	0.0750663	0.0217261	4.702E-07	0.0015922	0.0236705	5.123E-07	0.0017347
Kam Tin Road	RD_09_10	1100-1200	0.1739708	3.645E-06 0.01234	07 0.1634064	1 3.423E-06 0.01159	13 0.1511466	3.166E-06	0.0107217 1.0	657668	2.233E-05 0	0.0756008	1.0039199	2.103E-05	0.0712136	0.9360671	1.961E-05	0.0664004	1.2397376	2.597E-05	0.0879415	1.0872137	2.278E-05	0.0771221	0.023076	4.834E-07	0.0016369	0.0251587	5.271E-07	0.0017846
Kam Tin Road	KD_09_10 RD_00_10	1200-1300	0.2897662	6.842E-06 0.02316	6/ U.2/2441	0.433E-06 0.02178	15 0.2510/85	5.928E-06	0.0200736 1.	0437719	3.645E-05 0	0.1234237	1.4564249	3.439E-05	0.1164404	1.3534333	3.196E-05	0.1082063	1.8335381	4.329E-05	0.1465904	1.6045118	3./88E-05	0.1282799	0.0322873	7.623E-07	0.0025813	0.0352079	8.313E-07	0.0028149
Kam Tin Road	RD_09_10	1400-1500	0.3248149	8.534E-06 0.02889	6 0.3063987	7 8.05E-06 0.02725	73 0.2847307 83 0.2803226	7.365E-06	0.0202288 1.6	5556459	4.35E-05 (	0.1380310	1.5672345	4.118E-05	0.1394265	1.4472195	3.802E-05	0.1383333	1.9804608	5.203E-05	0.1761885	1.7275421	4.539E-05	0.153688	0.0332338	8.97E-07	0.0032433	0.0371987	9.773E-07	0.0033302
Kam Tin Road	RD_09_10	1500-1600	0.3041434	7.815E-06 0.02646	16 0.287937	7.398E-06 0.02505	16 0.2638347	6.779E-06	0.0229546 1.5	654956	4.022E-05	0.1362041	1.4868556	3.82E-05	0.1293621	1.3740884	3.531E-05	0.119551	1.8696389	4.804E-05	0.1626657	1.6379231	4.209E-05	0.1425056	0.032683	8.398E-07	0.0028436	0.0356128	9.151E-07	0.0030984
Kam Tin Road	RD_09_10	1600-1700	0.2893104	7.434E-06 0.02517	11 0.2721214	4 6.992E-06 0.02367	56 0.2495682	6.413E-06	0.0217134 1.5	541386	3.993E-05	0.135216	1.4667757	3.769E-05	0.1276151	1.3573324	3.488E-05	0.1180931	1.843449	4.737E-05	0.1603871	1.6069005	4.129E-05	0.1398065	0.032838	8.438E-07	0.002857	0.0358624	9.215E-07	0.0031202
Kam Tin Road	RD_09_10	1700-1800	0.2659327	7.387E-06 0.02501	31 0.2477845	5 6.883E-06 0.02330	62 0.2281845	6.338E-06	0.0214626 1.4	1111322	3.92E-05 (	0.1327285	1.3167857	3.658E-05	0.1238545	1.2199037	3.389E-05	0.1147419	1.6770649	4.659E-05	0.1577417	1.4480882	4.022E-05	0.1362046	0.0301252	8.368E-07	0.0028335	0.0329963	9.166E-07	0.0031036
Kam Tin Road	RD_09_10	1800-1900	0.1862401	4.484E-06 0.01518	17 0.1729282	2 4.163E-06 0.01409	66 0.1605897	3.866E-06	0.0130908 1.0	928747	2.631E-05	0.089088	1.0156812	2.445E-05	0.0827954	0.9481601	2.283E-05	0.0772913	1.2791148	3.079E-05	0.1042697	1.1087497	2.669E-05	0.0903821	0.0238993	5.754E-07	0.0019482	0.0261761	6.302E-07	0.0021338
Kam Tin Road	RD_09_10 RD_09_10	2000-2000	0.13633	2 132E-06 0.01030	0.1283/93 04 0.1407491	3 2.823E-06 0.0095	95 0.11964/6	2.031E-06	0.0069093 0.8	1850338	1.9022-05 0	0.0643905	1 0114465	1./04E-05	0.0597154	0.7506885	1.051E-05	0.055843	1 2369273	2.200E-05	0.0746909	1 0855822	1.9146-05	0.0648076	0.0195559	4.3E-07	0.0014562	0.0214251	4./12E-0/ 3.305E-07	0.0015954
Kam Tin Road	RD 09 10	2100-2200	0.1641335	1.881E-06 0.00636	82 0.1523261	L 1.745E-06 0.00591	01 0.1431837	1.641E-06	0.0055554 1.1	1654377	1.335E-05 (	0.0452178	1.0825475	1.24E-05	0.0420018	1.022228	1.171E-05	0.0396614	1.3295712	1.523E-05	0.051586	1.1654117	1.335E-05	0.0452168	0.023964	2.746E-07	0.0009298	0.0262455	3.007E-07	0.0010183
Kam Tin Road	RD_09_10	2200-2300	0.2171837	1.86E-06 0.00629	86 0.2018686	5 1.729E-06 0.00585	44 0.1897623	1.625E-06	0.0055033 1.4	1004296	1.199E-05 (	0.0406142	1.3041462	1.117E-05	0.0378219	1.2328255	1.056E-05	0.0357535	1.6176133	1.385E-05	0.0469128	1.4225878	1.218E-05	0.0412569	0.0265161	2.271E-07	0.000769	0.0289218	2.477E-07	0.0008388
Kam Tin Road	RD_09_10	2300-0000	0.1024243	7.231E-07 0.00244	86 0.0940116	5 6.637E-07 0.00224	75 0.0882859	6.233E-07	0.0021106 0.	693072	4.893E-06	0.0165689	0.6276436	4.431E-06	0.0150047	0.5878389	4.15E-06	0.0140531	0.7954963	5.616E-06	0.0190175	0.6761248	4.774E-06	0.0161637	0.0151401	1.069E-07	0.0003619	0.0166098	1.173E-07	0.0003971
Kam Tin Road	RD_11_12	0000-0100	0.0434735	1.858E-07 0.00068	99 0.0399548	3 1.707E-07 0.0006	4 0.0374853	1.602E-07	0.0005948 0.4	1231742	1.808E-06 (	0.0067151	0.3849766	1.645E-06	0.0061089	0.3614425	1.545E-06	0.0057355	0.4666477	1.994E-06	0.0074049	0.3989277	1.705E-06	0.0063303	0.0098268	4.199E-08	0.0001559	0.010747	4.593E-08	0.0001705
Kam Tin Road	RD_11_12 RD_11_12	0100-0200	0.0420317	1.033E-07 0.00038	35 0.0383668	3 9.428E-08 0.00035	01 0.0359486	8.834E-08	0.000328 0.4	1459499	1.096E-06	0.004069	0.3995181	9.817E-07	0.0036453	0.3721815	9.145E-07	0.0033959	0.4879815	1.199E-06	0.0044525	0.4081301	1.003E-06	0.0037239	0.0091245	2.242E-08	8.325E-05	0.0099585	2.447E-08	9.086E-05
Kam Tin Road	RD_11_12 RD_11_12	0200-0300	0.0373208	1.365E-07 0.00040	58 0.0320002	1 234F-07 0 0004	83 0.0667711	1.033E-07	0.0003837 0.2	8804783	6 504E-07	0.0022005	0.237342	5.574E-07	0.0018879	0.2138103	5.03E-07	0.0017123	0.4603238	7.869E-07	0.0020034	0.2041748	6 171F-07	0.002090	0.0073034	1.575E-08	6.012E-05	0.0073702	1.766E-08	6.557E-05
Kam Tin Road	RD_11_12	0400-0500	0.068365	1.461E-07 0.00054	24 0.0617928	3 1.32E-07 0.00049	03 0.0575092	1.229E-07	0.0004563 0.3	3416865	7.301E-07	0.002711	0.2943214	6.289E-07	0.0023352	0.2679725	5.726E-07	0.0021261	0.4100516	8.762E-07	0.0032534	0.3254816	6.955E-07	0.0025824	0.0085716	1.832E-08	6.801E-05	0.0093332	1.994E-08	7.405E-05
Kam Tin Road	RD_11_12	0500-0600	0.1570447	5.872E-07 0.00218	0.143892	5.381E-07 0.00199	79 0.1354125	5.064E-07	0.0018802 0.8	3212529	3.071E-06 (	0.0114029	0.748703	2.8E-06	0.0103956	0.7062575	2.641E-06	0.0098062	0.9782976	3.658E-06	0.0135834	0.84167	3.147E-06	0.0116864	0.0169754	6.348E-08	0.0002357	0.0184864	6.913E-08	0.0002567
Kam Tin Road	RD_11_12	0600-0700	0.0729721	6.783E-07 0.00251	85 0.0669294	4 6.221E-07 0.0023	1 0.0628732	5.844E-07	0.00217 0.	601688	5.593E-06 (	0.0207664	0.5439584	5.056E-06	0.0187739	0.5101772	4.742E-06	0.017608	0.6746602	6.271E-06	0.0232849	0.5730505	5.326E-06	0.019778	0.013918	1.294E-07	0.0004804	0.0153003	1.422E-07	0.0005281
Kam Tin Road	RD_11_12	0700-0800	0.1382897	2.955E-06 0.01097	21 0.1279795	5 2.735E-06 0.01015	41 0.1189497	2.542E-06	0.0094377 0.8	3497981	1.816E-05 (	0.0674243	0.7853064	1.678E-05	0.0623074	0.7335943	1.568E-05	0.0582045	0.9880878	2.111E-05	0.0783964	0.852544	1.822E-05	0.0676422	0.0193659	4.138E-07	0.0015365	0.021243	4.539E-07	0.0016855
Kam Tin Road	RD_11_12 RD_11_12	0800-0900	0.1289103	3 147E-06 0.01168	5 0.1200413	2.52/E-06 0.00938	14 0.1113372 46 0.1348518	2.343E-00 2.737E-06	0.008/012 0.8	036768	2.017E-05	0.0043298	0.764926	1.010-05	0.0597801	0.7127131	1.5E-U5 1.769E-05	0.0556990	1 1487042	2.004E-05	0.0744048	1 0061163	2.0425-05	0.0644007	0.0188474	3.96/E-07	0.001473	0.0200325	4.343E-07	0.0018115
Kam Tin Road	RD 11 12	1000-1100	0.1648703	3.294E-06 0.01223	08 0.1546167	7 3.089E-06 0.01147	01 0.1431886	2.861E-06	0.0106224 1.0	352051	2.068E-05	0.076796	0.9721802	1.942E-05	0.0721206	0.9066784	1.811E-05	0.0672613	1.2000754	2.398E-05	0.0890268	1.049867	2.097E-05	0.0778837	0.02219	4.433E-07	0.0016462	0.0240337	4.829E-07	0.0017932
Kam Tin Road	RD_11_12	1100-1200	0.1778802	3.44E-06 0.01277	25 0.1670832	2 3.231E-06 0.01199	73 0.1545462	2.989E-06	0.0110971 1.0	0833576	2.095E-05	0.0777895	1.0206919	1.974E-05	0.0732899	0.9517979	1.841E-05	0.068343	1.2612378	2.439E-05	0.0905621	1.1063441	2.139E-05	0.07944	0.0234538	4.535E-07	0.0016841	0.0255671	4.944E-07	0.0018358
Kam Tin Road	RD_11_12	1200-1300	0.2906185	6.334E-06 0.02351	93 0.2732654	1 5.956E-06 0.02211	49 0.2518679	5.489E-06	0.0203833 1.	554315	3.388E-05 (	0.1257882	1.466521	3.196E-05	0.1186832	1.3629664	2.971E-05	0.1103027	1.8449336	4.021E-05	0.1493075	1.6148343	3.52E-05	0.1306859	0.0324282	7.068E-07	0.0026244	0.0353614	7.707E-07	0.0028617
Kam Tin Road	RD_11_12	1300-1400	0.3293354	8.269E-06 0.03070	27 0.3103901	1 7.793E-06 0.02893	65 0.2845077	7.143E-06	0.0265236 1.	255121	4.332E-05 (	0.1608631	1.6324903	4.099E-05	0.1521911	1.5104509	3.792E-05	0.1408138	2.0548475	5.159E-05	0.1915659	1.7949586	4.507E-05	0.1673374	0.0353693	8.88E-07	0.0032974	0.0385369	9.675E-07	0.0035927
Kam Tin Road	RD_11_12 RD_11_12	1400-1500	0.32452	7.87E-06 0.02922	39 0.3061304	1 7.424E-06 0.02756	79 0.2800925	6.793E-06	0.0252231 1.6	590398	4.024E-05 (	0.1494009	1.5704761	3.809E-05	0.1414256	1.4502427	3.517E-05	0.1305982	1.9835597	4.811E-05	0.1786248	1.7303352	4.196E-05	0.1558213	0.0341614	8.285E-07	0.0030763	0.037223	9.027E-07	0.003352
Kam Tin Road	RD_11_12 RD_11_12	1600-1000	0.3030312	6.829E-06 0.02535	R4 0 2720611	1 6 4 2 4 F-06 0 0 2 3 8	23 0 2495213	5.891E-06	0.0230481 1.	5561211	3.674F-05	0.1364289	1.4686124	3.458E-05	0.12387568	1.3590054	3.233E-05	0.1200233	1.8028830	4.358E-05	0.1617873	1.6085267	3.798F-05	0.143078	0.0323035	7.755E-07	0.0028498	0.0358726	8 47F-07	0.003145
Kam Tin Road	RD 11 12	1700-1800	0.1520524	3.103E-06 0.01152	12 0.1419259	2.896E-06 0.01075	39 0.1310721	2.675E-06	0.0099315 0.9	475951	1.934E-05 (	0.0718004	0.8847504	1.805E-05	0.0670386	0.8223599	1.678E-05	0.0623112	1.0996476	2.244E-05	0.0833216	0.953432	1.946E-05	0.0722427	0.0210733	4.3E-07	0.0015967	0.0230732	4.708E-07	0.0017483
Kam Tin Road	RD_11_12	1800-1900	0.1860293	4.134E-06 0.01535	0.1727289	3.838E-06 0.01425	28 0.1604008	3.564E-06	0.0132355 1.0	918369	2.426E-05 (	0.0900932	1.0146471	2.255E-05	0.0837238	0.9471303	2.105E-05	0.0781527	1.2778661	2.84E-05	0.1054434	1.107531	2.461E-05	0.0913882	0.0238611	5.302E-07	0.0019689	0.0261328	5.807E-07	0.0021564
Kam Tin Road	RD_11_12	1900-2000	0.1431162	2.89E-06 0.01073	05 0.132791	2.681E-06 0.00995	64 0.1237228	2.498E-06	0.0092765 0.8	3863997	1.79E-05 (	0.0664603	0.8218012	1.659E-05	0.0616168	0.769	1.553E-05	0.0576579	1.0295158	2.079E-05	0.0771908	0.8927229	1.803E-05	0.0669344	0.0200319	4.045E-07	0.0015019	0.0219531	4.433E-07	0.001646
Kam Tin Road	RD_11_12	2000-2100	0.1508717	1.966E-06 0.00730	19 0.1406332	2 1.833E-06 0.00680	64 0.1320833	1.722E-06	0.0063926 1.0	0855328	1.415E-05	0.052538	1.0110359	1.318E-05	0.0489325	0.9529625	1.242E-05	0.0461218	1.2364044	1.612E-05	0.0598399	1.0850458	1.414E-05	0.0525144	0.0219684	2.863E-07	0.0010632	0.0240442	3.134E-07	0.0011637
Kam Tin Road	RD_11_12 RD_11_12	2200-2200	0.1650179	1.755-06 0.00654	54 0.15314/1 58 0.202068/	1 1.636E-06 0.0060	55 U.1439533 89 0 1908025	1.538E-06	0.005/10/ 1.1	101032	1.248E-05 0	0.0463237	1.0846/34	1.159E-05	0.0430298	1.0242158	1.094E-05 9.949E-06	0.0406314	1.332/201	1.424E-05	0.05287	1.1681691	1.248E-05	0.0463422	0.0240131	2.565E-07 2.137E-07	0.0009526	0.0262971	2.81E-07	0.0010432
Kam Tin Road	RD 11 12	2300-0000	0.0992496	6.468E-07 0.00240	18 0.09107	5.935E-07 0.00220	38 0.0855116	5.573E-07	0.0020693 0.6	5799749	4.431E-06 0	0.0164548	0.6151835	4.009E-06	0.0148869	0.5758744	3.753E-06	0.0139357	0.7792246	5.078E-06	0.0188566	0.661386	4.31E-06	0.016005	0.0148782	9.696E-08	0.00036	0.0163271	1.064E-07	0.0003951
Kam Tin Road	RD_13_14	0000-0100	0.0426905	1.824E-07 0.00063	27 0.0392185	5 1.676E-07 0.00058	12 0.0367868	1.572E-07	0.0005452 0.4	175514	1.784E-06 (	0.0061881	0.3793809	1.621E-06	0.0056224	0.3559452	1.521E-06	0.0052751	0.4602419	1.967E-06	0.0068208	0.392732	1.678E-06	0.0058203	0.0096923	4.142E-08	0.0001436	0.0106	4.53E-08	0.0001571
Kam Tin Road	RD_13_14	0100-0200	0.0414184	1.062E-07 0.00036	83 0.0377831	L 9.688E-08 0.0003	6 0.0353896	9.074E-08	0.0003147 0.4	407876	1.13E-06 (	0.0039195	0.3942052	1.011E-06	0.0035053	0.3668931	9.408E-07	0.0032624	0.4822059	1.236E-06	0.0042878	0.4022827	1.031E-06	0.0035771	0.0089833	2.303E-08	7.988E-05	0.0098033	2.514E-08	8.717E-05
Kam Tin Road	RD_13_14	0200-0300	0.0567319	1.212E-07 0.00042	04 0.0512696	5 1.096E-07 0.00037	99 0.0476729	1.019E-07	0.0003533 0.2	2767085	5.913E-07 (	0.0020504	0.2365662	5.055E-07	0.001753	0.2145045	4.583E-07	0.0015895	0.3334403	7.125E-07	0.0024708	0.2621775	5.602E-07	0.0019427	0.0073179	1.564E-08	5.423E-05	0.0079267	1.694E-08	5.874E-05
Kam Tin Road	RD_13_14 RD_13_14	0300-0400	0.0836494	1.519E-07 0.00052 1.55E-07 0.00053	59 0.0756631 74 0.0624415	1.374E-07 0.0004	58 0.0581182	1.2/1E-0/ 1.304E-07	0.0004407 0.:	3914952	7.508F-07 (	0.0024658	0.3356324	6.096E-07	0.002114	0.3029318	5.502E-07	0.001908	0.4/51446	8.63E-07	0.0029927	0.3729051	5.//3E-0/ 7.345E-07	0.0023487	0.009705	1.763E-08	6.113E-05	0.0105835	1.922E-08	0.000E-05
Kam Tin Road	RD 13 14	0500-0600	0.155439	5.812E-07 0.00201	57 0.1423793	3 5.324E-07 0.00184	63 0.1339652	5.009E-07	0.0017372 0.8	3046537	3.009E-06 (	0.0104343	0.7326022	2.739E-06	0.0095	0.6906529	2.583E-06	0.008956	0.9600928	3.59E-06	0.01245	0.8246181	3.084E-06	0.0106932	0.0166506	6.226E-08	0.0002159	0.0181329	6.78E-08	0.0002351
Kam Tin Road	RD_13_14	0600-0700	0.0725683	6.823E-07 0.00236	6 0.0665557	7 6.257E-07 0.0021	7 0.062522	5.878E-07	0.0020385 0.	600442	5.645E-06 (	0.0195768	0.5427643	5.103E-06	0.0176963	0.5090437	4.786E-06	0.0165968	0.6730103	6.327E-06	0.0219428	0.5715657	5.374E-06	0.0186353	0.0138826	1.305E-07	0.0004526	0.0152606	1.435E-07	0.0004976
Kam Tin Road	RD_13_14	0700-0800	0.1352251	2.904E-06 0.01007	03 0.1251263	3 2.687E-06 0.00931	82 0.1162893	2.497E-06	0.0086601 0.8	3348869	1.793E-05 (	0.0621744	0.7712055	1.656E-05	0.057432	0.7202671	1.547E-05	0.0536386	0.970112	2.083E-05	0.0722447	0.8365564	1.796E-05	0.0622987	0.0190758	4.096E-07	0.0014206	0.0209283	4.494E-07	0.0015585
Kam Tin Road	RD_13_14	0800-0900	0.1254145	2.666E-06 0.00924	67 0.1167571	1 2.482E-06 0.00860	84 0.1082753	2.302E-06	0.0079831 0.8	3048373	1.711E-05 (	0.0593402	0.7474431	1.589E-05	0.0551086	0.6961401	1.48E-05	0.051326	0.9302518	1.978E-05	0.0685869	0.8044154	1.71E-05	0.0593091	0.0184724	3.927E-07	0.001362	0.0202237	4.3E-07	0.0014911
Kam Tin Road	RD_13_14	1000 1100	0.1626136	3.4622-06 0.01200	42 0.1524955	3.2422-06 0.01124	54 0.1415/90 60 0.1204026	3.012-06	0.0104586 0.5	010004	2.115E-05	0.0700280	0.933479	1.985E-05	0.0657499	0.8/22/09	1.8555-05	0.064312	1.157/081	2.4010-05	0.0811602	1.0136505	2.1502-05	0.0700655	0.021244	4.51/E-07	0.0015063	0.02315	4.9222-07	0.001/068
Kam Tin Road	RD_13_14 RD_13_14	1100-1200	0.1739708	3.364E-06 0.01166	54 0.1500408 56 0.1634064	3.16E-06 0.01045	81 0.1511466	2.923E-06	0.010136 1.0	010304	2.021-05 (	0.0714708	1.0039199	1.941E-05	0.0673233	0.9360671	1.81E-05	0.0627731	1.2397376	2.341L-05	0.0831373	1.0242781	2.102E-05	0.072909	0.023076	4.341L-07 4.462E-07	0.0015055	0.0251587	4.865E-07	0.0016872
Kam Tin Road	RD_13_14	1200-1300	0.2897662	6.315E-06 0.02190	0.272441	5.938E-06 0.02059	16 0.2510785	5.472E-06	0.018977 1.5	5437719	3.365E-05 (	0.1166812	1.4564249	3.174E-05	0.1100794	1.3534333	2.95E-05	0.1022951	1.8335381	3.996E-05	0.1385823	1.6045118	3.497E-05	0.1212721	0.0322873	7.037E-07	0.0024403	0.0352079	7.674E-07	0.0026611
Kam Tin Road	RD_13_14	1300-1400	0.3297112	8.278E-06 0.02870	71 0.3107269	9 7.801E-06 0.02705	42 0.2847907	7.15E-06	0.024796 1.3	165461	4.31E-05 0	0.1494552	1.6238468	4.077E-05	0.1413841	1.5022482	3.772E-05	0.1307969	2.0462573	5.138E-05	0.1781623	1.7870389	4.487E-05	0.1555928	0.0352398	8.848E-07	0.0030682	0.0383953	9.64E-07	0.003343
Kam Tin Road	RD_13_14	1400-1500	0.3248149	7.877E-06 0.02731	8 0.3063987	7 7.431E-06 0.02576	92 0.2803226	6.798E-06	0.0235761 1.6	5556459	4.015E-05 0	0.1392455	1.5672345	3.801E-05	0.1318098	1.4472195	3.51E-05	0.1217161	1.9804608	4.803E-05	0.1665635	1.7275421	4.19E-05	0.1452922	0.0341395	8.28E-07	0.0028713	0.0371987	9.021E-07	0.0031285
Kam Tin Road	RD 13_14 RD 13_14	1500-1600	0.2893104	6.862E-06 0.02301	6 0.277171/	0.829E-06 0.02368 1 6.454E-06 0.02734	22 0.2495687 22 0.2495687	0.258E-Ub 5.919E-06	0.0217006 1.5	541386	3.686F-05	0.128/634	1.4667757	3.327E-05 3,479F-05	0.1222952	1.3573320	3.239E-05 3.219E-05	0.11302	1.843449	4.434E-U5	0.153//95	1.6069005	3.805E-U5	0.134/207	0.032838	7.789F-07	0.0026882	0.0358624	6.44/E-0/ 8.506F-07	0.0029292
Kam Tin Road	RD_13_14	1700-1800	0.2659327	6.819E-06 0.02364	67 0.2477845	5 6.353E-06 0.0220	3 0.2281845	5.851E-06	0.0202901 1.4	1111322	3.618E-05	0.1254777	1.3167857	3.376E-05	0.1170885	1.2199037	3.128E-05	0.1084737	1.6770649	4.3E-05	0.1491245	1.4480882	3.713E-05	0.1287639	0.0301252	7.724E-07	0.0026787	0.0329963	8.461E-07	0.002934
Kam Tin Road	RD_13_14	1800-1900	0.1862401	4.139E-06 0.01435	24 0.1729282	2 3.843E-06 0.01332	65 0.1605897	3.569E-06	0.0123757 1.0	928747	2.429E-05	0.0842212	1.0156812	2.257E-05	0.0782724	0.9481601	2.107E-05	0.0730689	1.2791148	2.842E-05	0.0985736	1.1087497	2.464E-05	0.0854446	0.0238993	5.311E-07	0.0018418	0.0261761	5.817E-07	0.0020172
Kam Tin Road	RD_13_14	1900-2000	0.13833	2.808E-06 0.00973	77 0.1283793	3 2.606E-06 0.00903	72 0.1196476	2.429E-06	0.0084226 0.8	3647339	1.755E-05 (	0.0608729	0.8019496	1.628E-05	0.0564532	0.7506883	1.524E-05	0.0528446	1.003064	2.036E-05	0.0706106	0.8703359	1.767E-05	0.0612672	0.0195559	3.97E-07	0.0013766	0.0214251	4.349E-07	0.0015082
Kam Tin Road	RD_13_14	2000-2100	0.1509934	1.968E-06 0.00682	5 0.1407491	L 1.835E-06 0.0063	0.1321942	1.723E-06	0.0059753 1.0	0859338	1.415E-05 (	0.0490852	1.0114465	1.318E-05	0.0457183	0.9533881	1.243E-05	0.0430941	1.2369273	1.612E-05	0.0559103	1.0855822	1.415E-05	0.0490694	0.0219688	2.863E-07	0.000993	0.0240427	3.134E-07	0.0010868
Kam Tin Road	RD_13_14 RD_13_14	2200-2200	0.1641335	1.736E-06 0.00602 1.717E-06 0.00595	J3 0.1523261	1 1.611E-06 0.00558 5 1.596E-06 0.00553	/2 0.143183/ 46 0.1897623	1.514E-06	0.0052519 1.1	100/296	1.233E-05 0	0.0427476	1.0825475	1.145E-05	0.0397073	1.022228	1.081E-05 9.747E-06	0.0374948	1.3295/12	1.406E-05	0.048768	1.1654117	1.233E-05	0.042/46/	0.023964	2.535E-07 2.096E-07	0.000879	0.0262455	2.776E-07	0.0009627
Kam Tin Road	RD 13 14	2300-0000	0.1024243	6.675E-07 0.00231	48 0.0940116	5 6.127E-07 0.00212	47 0.0882859	5.754E-07	0.0019953 0.	693072	4.517E-06 (	0.0156638	0.6276436	4.09E-06	0.014185	0.5878389	3.831E-06	0.0132854	0.7954963	5.184E-06	0.0179786	0.6761248	4.406E-06	0.0152807	0.0151401	9.867E-08	0.0003422	0.0166098	1.082E-07	0.0003754
Kam Tin Road	RD_15_16	0000-0100	0.0434735	2.013E-07 0.00029	66 0.0399548	3 1.85E-07 0.00027	26 0.0374853	1.735E-07	0.0002557 0.4	231742	1.959E-06	0.002887	0.3849766	1.782E-06	0.0026264	0.3614425	1.673E-06	0.0024658	0.4666477	2.16E-06	0.0031836	0.3989277	1.847E-06	0.0027216	0.0098268	4.549E-08	6.704E-05	0.010747	4.975E-08	7.332E-05
Kam Tin Road	RD_15_16	0100-0200	0.0420317	1.119E-07 0.00016	49 0.0383668	3 1.021E-07 0.00015	05 0.0359486	9.57E-08	0.000141 0.4	1459499	1.187E-06 (	0.0017494	0.3995181	1.064E-06	0.0015672	0.3721815	9.908E-07	0.00146	0.4879815	1.299E-06	0.0019142	0.4081301	1.086E-06	0.001601	0.0091245	2.429E-08	3.579E-05	0.0099585	2.651E-08	3.907E-05
Kam Tin Road	RD_15_16	0200-0300	0.0575268	1.332E-07 0.00019	62 0.0520002	2 1.204E-07 0.00017	74 0.0483583	1.119E-07	0.000165 0.2	2781553	6.439E-07 (	0.0009488	0.237942	5.508E-07	0.0008116	0.2158165	4.996E-07	0.0007362	0.3356821	7.77E-07	0.0011451	0.2641748	6.115E-07	0.0009011	0.0073634	1.704E-08	2.512E-05	0.0079762	1.846E-08	2.721E-05
Kam Tin Road	RD_15_16	0300-0400	0.0798454	1.479E-07 0.00021	79 0.072205	1.337E-07 0.0001	0.0667711	1.237E-07	0.0001822 0.3	3804783	7.046E-07 (	0.0010383	0.3260591	6.038E-07	0.0008898	0.2942491	5.449E-07	0.000803	0.4603238	8.525E-07	0.0012562	0.3610201	6.686E-07	0.0009852	0.0094717	1.754E-08	2.585E-05	0.0103297	1.913E-08	2.819E-05
Kam Tin Road	RD_15_16 RD_15_16	0500-0500	0.006305	6 362E-07 0.00023	75 0.001/928	5 829F-07 0.0002	0 0.0575092	1.331E-07 5.485E-07	0.0001982 0.3	2212520	3.327E-06 (	0.0011055	0.2943214	3.033E-07	0.001004	0.20/9/25	0.203E-07	0.0009141	0.4100516	3.963E-07	0.0013987	0.3254610	7.534E-07 3.41E-06	0.0011103	0.0085716	1.984E-08	2.9246-05	0.0093332	2.10E-08	0.0001104
Kam Tin Road	RD 15 16	0600-0700	0.0729721	7.348E-07 0.00108	28 0.0669294	1 6.739E-07 0.00099	31 0.0628732	6.331E-07	0.0009329 0.	601688	6.059E-06	0.0089281	0.5439584	5.477E-06	0.0080714	0.5101772	5.137E-06	0.0075702	0.6746602	6.793E-06	0.0100109	0.5730505	5.77E-06	0.0085031	0.013918	1.401E-07	0.0002065	0.0153003	1.541E-07	0.000227
Kam Tin Road	RD_15_16	0700-0800	0.1382897	3.201E-06 0.00471	72 0.1279795	5 2.962E-06 0.00436	55 0.1189497	2.753E-06	0.0040575 0.8	8497981	1.967E-05	0.0289876	0.7853064	1.818E-05	0.0267877	0.7335943	1.698E-05	0.0250238	0.9880878	2.287E-05	0.0337048	0.852544	1.973E-05	0.0290813	0.0193659	4.483E-07	0.0006606	0.021243	4.917E-07	0.0007246
Kam Tin Road	RD_15_16	0800-0900	0.1289163	2.939E-06 0.00433	15 0.1200415	5 2.737E-06 0.00403	33 0.1113372	2.539E-06	0.0037409 0.8	3231429	1.877E-05	0.0276572	0.764926	1.744E-05	0.0257011	0.7127131	1.625E-05	0.0239468	0.9520592	2.171E-05	0.0319887	0.8240503	1.879E-05	0.0276877	0.0188474	4.297E-07	0.0006333	0.0206325	4.704E-07	0.0006932
Kam Tin Road	RD_15_16	0900-1000	0.1550275	3.409E-06 0.00502	38 0.1452033	3 3.193E-06 0.00470	54 0.1348518	2.965E-06	0.00437 0.9	9936768	2.185E-05	0.0322007	0.932116	2.05E-05	0.0302058	0.8712645	1.916E-05	0.0282339	1.1487042	2.526E-05	0.0372245	1.0061163	2.213E-05	0.0326038	0.0219806	4.834E-07	0.0007123	0.0240337	5.285E-07	0.0007788
Kam Tin Road	RD_15_16	1000-1100	0.1648703	3.568E-06 0.00525	84 0.1546167	/ 3.346E-06 0.00493	13 0.1431886	3.099E-06	0.0045668 1.0	1833576	2.241E-05 0	0.0330168	0.9721802	2.104E-05	0.0310067	0.9056784	1.962E-05	0.0289176	1.2000754	2.597E-05	0.0382751	1.049867	2.272E-05	0.0334844	0.02219	4.803E-07	0.0007077	0.025567*	5.232E-07	0.000771
Kam Tin Road	RD_15_16 RD_15_16	1200-1200	0.2906185	6.862E-06 0.00549	16 0.273265/	1 6.452E-06 0.0051		5.947F-06	0.0087633 1	554315	3.67E-05	0.0540799	1.466521	3.463F-05	0.0510252	1.3629664	3.218F-05	0.0474777	1.8449336	4.356F-05	0.0641915	1.6148343	3.813F-05	0.0561856	0.0234536	7.657F-07	0.0011283	0.0353614	8.349F-07	0.0007893
Kam Tin Road	RD_15 16	1300-1400	0.3293354	8.958E-06 0.0132	0.3103901	1 8.442E-06 0.01244	06 0.2845077	7.738E-06	0.0114032 1.3	255121	4.693E-05	0.0691596	1.6324903	4.44E-05	0.0654312	1.5104509	4.108E-05	0.0605398	2.0548475	5.589E-05	0.0823596	1.7949586	4.882E-05	0.0719431	0.0353693	9.62E-07	0.0014176	0.0385369	1.048E-06	0.0015446
Kam Tin Road	RD_15_16	1400-1500	0.32452	8.526E-06 0.01256	42 0.3061304	8.043E-06 0.01185	22 0.2800925	7.359E-06	0.0108441 1.6	5590398	4.359E-05	0.0642317	1.5704761	4.126E-05	0.0608028	1.4502427	3.81E-05	0.0561479	1.9835597	5.211E-05	0.0767959	1.7303352	4.546E-05	0.066992	0.0341614	8.975E-07	0.0013226	0.037223	9.78E-07	0.0014411
Kam Tin Road	RD_15_16	1500-1600	0.3030312	7.751E-06 0.01142	21 0.2868911	1 7.338E-06 0.01081	37 0.2628887	6.724E-06	0.009909 1.5	598524	3.99E-05	0.0587953	1.4814608	3.789E-05	0.0558405	1.3690725	3.502E-05	0.0516042	1.8628836	4.765E-05	0.0702174	1.6319612	4.174E-05	0.0615133	0.0325055	8.314E-07	0.0012252	0.0354202	9.06E-07	0.0013351
Kam Tin Road	RD_15_16	1600-1700	0.2892402	7.398E-06 0.01090	23 0.2720611	L 6.959E-06 0.01025	48 0.2495213	6.382E-06	0.0094052 1.5	5561211	3.98E-05 (	0.0586546	1.4686124	3.757E-05	0.0553562	1.3590054	3.476E-05	0.0512248	1.8453613	4.72E-05	0.0695569	1.6085267	4.114E-05	0.0606299	0.0328466	8.402E-07	0.0012381	0.0358726	9.176E-07	0.0013521
Kam Tin Road	RU_15_16 RD 15 14	1800-1900	0.1520524	4 478F-06 0.00495	95 0.1419255 95 0.1777704	3.13/E-06 0.00462	3* 0.1310/21 77 0.1604009	2.898E-UB 3.861E-06	0.0042098 0.9	1018360	2.095E-05	0.030809	1 0146471	1.950E-05	0.0288218	0.8223599	2.978E-02	0.0207893	1.09904/6	2.431E-05 3.076F-0F	0.0358223	0.353432	2.1085-05	0.0302005	0.0210733	+.059E-07	0.0008465	0.0230732	6 291F-07	0.0007516
Kam Tin Road	RD 15 16	1900-2000	0.1431162	3.131E-06 0.00461	34 0.132791	2.905E-06 0.00428	05 0.1237228	2.706E-06	0.0039882 0.8	3863997	1.939E-05 (	0.0285731	0.8218012	1.798E-05	0.0264908	0.769	1.682E-05	0.0247888	1.0295158	2.252E-05	0.0331865	0.8927229	1.953E-05	0.028777	0.0200319	4.382E-07	0.0006457	0.0219531	4.802E-07	0.0007077
Kam Tin Road	RD_15_16	2000-2100	0.1508717	2.13E-06 0.00313	93 0.1406332	2 1.986E-06 0.00292	63 0.1320833	1.865E-06	0.0027484 1.0	0855328	1.533E-05	0.0225876	1.0110359	1.428E-05	0.0210374	0.9529625	1.346E-05	0.0198291	1.2364044	1.746E-05	0.0257269	1.0850458	1.532E-05	0.0225774	0.0219684	3.102E-07	0.0004571	0.0240442	3.395E-07	0.0005003
Kam Tin Road	RD_15_16	2100-2200	0.1650179	1.91E-06 0.00281	45 0.1531471	1.773E-06 0.0026	2 0.1439533	1.666E-06	0.0024552 1.1	1677022	1.352E-05 (	0.0199158	1.0846734	1.255E-05	0.0184997	1.0242158	1.185E-05	0.0174686	1.3327201	1.543E-05	0.0227303	1.1681691	1.352E-05	0.0199238	0.0240131	2.779E-07	0.0004096	0.0262971	3.044E-07	0.0004485

						N	IO2 Emission									NO Emission							NOX En	nission			PI	M25 Emission	1	PN	110 Emissior	n
Road Name	Road ID	Hour	A	nnualHourMi	n	Mo	onthlyHourM	in	Mont	thlyHourAve	age	A	nnualHourMi	in	M	onthlyHourM	in	Mont	thlyHourAve	rage	Ar	nualHourMi	n	Mon	thlyHourAve	rage	An	nualHourMin	a	Anr	nualHourMir	n
			g/km/veh	g/s/m2	tonne/year	g/km/veh	g/s/m2	tonne/year	g/km/veh	g/s/m2	tonne/year	g/km/veh	g/s/m2	tonne/year	g/km/veh	g/s/m2	tonne/year	g/km/veh	g/s/m2	tonne/year	g/km/veh	g/s/m2	tonne/year	g/km/veh	g/s/m2	tonne/year	g/km/veh	g/s/m2 1	tonne/year	g/km/veh	g/s/m2	tonne/year
Kam Tin Road	RD_15_16 RD_15_16	2200-2300	0.2183573	1.895E-06 7.007E-07	0.002/932	0.2029684	1./62E-06 6.43E-07	0.0025963	0.1908025	1.656E-06 6.037E-07	0.0024407	1.4101032	1.224E-05 4.801E-06	0.0180376	1.3133226	1.14E-05 4 343E-06	0.016/996	1.2416071	1.078E-05	0.0158823	1.6284605	1.414E-05	0.0208308	1.4324096	1.243E-05	0.0183229	0.0266682	2.315E-07 I	0.0003411	0.0290874 .	2.525E-07 1 153E-07	0.0003721
Kam Tin Road	RD_13_10 RD_17_18	2300-0000	0.0332450	1.664F-07	0.0010320	0.0355291	1.521F-07	0.0003473	0.0330653	1.416E-07	0.0001947	0.3234189	4.801E-00	0.0019042	0.2892311	4.343E-00	0.0004003	0.267466	4.000E-00	0.0015748	0.3622942	1.551E-06	0.0021331	0.3005314	1 287E-06	0.000381	0.0148782	3 59E-08	4 937F-05	0.0091834	3.931E-08	5.407E-05
Kam Tin Road	RD 17 18	0100-0200	0.0375646	9.484E-08	0.0001304	0.0340508	8.597E-08	0.0001182	0.0317627	8.019E-08	0.0001103	0.3530507	8.914E-07	0.0012259	0.309746	7.82E-07	0.0010755	0.2853497	7.204E-07	0.0009908	0.3906153	9.862E-07	0.0013563	0.3171124	8.006E-07	0.0011011	0.0077631	1.96E-08	2.696E-05	0.0084805	2.141E-08	2.945E-05
Kam Tin Road	RD_17_18	0200-0300	0.0594947	1.241E-07	0.0001707	0.0537546	1.121E-07	0.0001542	0.0500098	1.043E-07	0.0001434	0.2838357	5.92E-07	0.0008141	0.2423017	5.054E-07	0.000695	0.2199899	4.588E-07	0.000631	0.3433305	7.161E-07	0.0009848	0.2699997	5.631E-07	0.0007745	0.0075139	1.567E-08	2.155E-05	0.0081401	1.698E-08	2.335E-05
Kam Tin Road	RD_17_18	0300-0400	0.0802532	1.41E-07	0.0001938	0.0726235	1.276E-07	0.0001754	0.0671479	1.179E-07	0.0001622	0.3769664	6.621E-07	0.0009105	0.3233	5.678E-07	0.0007809	0.2916258	5.122E-07	0.0007044	0.4572197	8.03E-07	0.0011044	0.3587737	6.301E-07	0.0008666	0.0094848	1.666E-08	2.291E-05	0.0103456	1.817E-08	2.499E-05
Kam Tin Road	RD_17_18	0400-0500	0.070872	1.556E-07	0.000214	0.0641556	1.408E-07	0.0001937	0.0596896	1.31E-07	0.0001802	0.3492134	7.667E-07	0.0010544	0.3011338	6.611E-07	0.0009092	0.2739737	6.015E-07	0.0008272	0.4200854	9.223E-07	0.0012684	0.3336634	7.325E-07	0.0010074	0.0087817	1.928E-08	2.651E-05	0.0095639	2.1E-08	2.888E-05
Kam Tin Road	RD_17_18	0500-0600	0.15555325	5./3E-0/ 6.449E-07	0.0007881	0.1404037	5.242E-07	0.000721	0.1519001	4.923E-07	0.000677	0.7294355	2.722E-06	0.0057441	0.0598899	2.403E-00 4.285E-06	0.0053871	0.0191187	2.311E-06	0.0055025	0.882908	5.451E-06	0.0045321	0.7510188	2.803E-06	0.0058549	0.0155471	5./28E-08	0.00016	0.0138972	0.24E-08	0.0001762
Kam Tin Road	RD 17 18	0700-0800	0.1368188	2.899E-06	0.0039864	0.1263192	2.676E-06	0.0036805	0.1175439	2.49E-06	0.0034248	0.7968707	1.688E-05	0.023218	0.7334866	1.554E-05	0.0213712	0.6853464	1.452E-05	0.0199686	0.9336895	1.978E-05	0.0272044	0.8028903	1.701E-05	0.0233934	0.0183733	3.893E-07	0.0005353	0.0201758	4.274E-07	0.0005879
Kam Tin Road	RD_17_18	0800-0900	0.1295927	2.717E-06	0.0037367	0.1203224	2.523E-06	0.0034694	0.1115543	2.339E-06	0.0032166	0.7942344	1.665E-05	0.0229014	0.735248	1.542E-05	0.0212005	0.6841505	1.434E-05	0.0197272	0.9238271	1.937E-05	0.0266381	0.7957048	1.668E-05	0.0229438	0.018393	3.856E-07	0.0005304	0.0201429	4.223E-07	0.0005808
Kam Tin Road	RD_17_18	0900-1000	0.155211	3.237E-06	0.004452	0.145198	3.028E-06	0.0041648	0.134609	2.807E-06	0.0038611	0.9958353	2.077E-05	0.0285641	0.9322385	1.944E-05	0.0267399	0.8699015	1.814E-05	0.0249519	1.1510463	2.401E-05	0.0330161	1.0045105	2.095E-05	0.028813	0.0219806	4.584E-07	0.0006305	0.0240337	5.013E-07	0.0006894
Kam Tin Road	RD_17_18	1000-1100	0.1655303	3.289E-06	0.0045231	0.1550584	3.081E-06	0.004237	0.1437865	2.857E-06	0.003929	1.0043616	1.996E-05	0.0274441	0.9416852	1.871E-05	0.0257314	0.8786955	1.746E-05	0.0240103	1.1698919	2.324E-05	0.0319672	1.022482	2.032E-05	0.0279392	0.0217284	4.317E-07	0.0005937	0.0236721	4.703E-07	0.0006468
Kam Tin Road	RD_17_18 RD_17_18	1200-1200	0.1/5629	5.393E-06	0.0046665	0.1645296	3.1/9E-06	0.0043716 0	0.1524427	2.945E-06	0.0040504	1.0353111	2E-05 3 347E-05	0.0275083	1 4336	1.8/9E-05 3 147E-05	0.0258432	1 33/2500	1./54E-05 2.020E-05	0.0241211	1.2109401	2.339E-05 3.00E-05	0.054877	1.060273	2.048E-05	0.0281715	0.0225722	4.361E-07 I	0.0005997	0.024588	4./5E-0/ 7.667E-07	0.0006533
Kam Tin Road	RD 17 18	1300-1400	0.3329928	8.407E-06	0.0115623	0.3129676	7.902E-06	0.0108669	0.2881973	7.276E-06	0.0100069	1.702212	4.298E-05	0.0591046	1.6062252	4.055E-05	0.0557717	1.4918369	3.766E-05	0.0517999	2.0352048	5.138E-05	0.0706669	1.7800342	4.494E-05	0.0618068	0.0351062	8.863E-07	0.001219	0.038252	9.658E-07	0.0013282
Kam Tin Road	RD_17_18	1400-1500	0.3280819	8.031E-06	0.011045	0.3087676	7.558E-06	0.0103948	0.2830501	6.929E-06	0.009529	1.6421142	4.02E-05	0.0552826	1.5510443	3.797E-05	0.0522166	1.4343613	3.511E-05	0.0482885	1.9701961	4.823E-05	0.0663276	1.7174114	4.204E-05	0.0578175	0.0340369	8.332E-07	0.0011459	0.0370888	9.079E-07	0.0012486
Kam Tin Road	RD_17_18	1500-1600	0.3008228	7.166E-06	0.0098549	0.283111	6.744E-06	0.0092746	0.2602419	6.199E-06	0.0085254	1.5160239	3.611E-05	0.0496645	1.4313125	3.409E-05	0.0468893	1.3263032	3.159E-05	0.0434493	1.8168467	4.328E-05	0.0595193	1.5865451	3.779E-05	0.0519747	0.0317488	7.563E-07	0.0010401	0.0346081	8.244E-07	0.0011338
Kam Tin Road	RD_17_18	1600-1700	0.2851486	6.761E-06	0.0092983	0.2674597	6.342E-06	0.0087215	0.2464285	5.843E-06	0.0080357	1.4973928	3.55E-05	0.048828	1.4094171	3.342E-05	0.0459593	1.3093524	3.105E-05	0.0426963	1.7825414	4.226E-05	0.0581264	1.5557808	3.689E-05	0.050732	0.0318705	7.557E-07	0.0010393	0.0348268	8.258E-07	0.0011357
Kam Tin Road	RD_17_18 RD_17_18	1/00-1800	0.1522237	3.192E-06	0.0043893	0.141/056	2.9/1E-06 3.9F-06	0.0053634	0.1310729	2.748E-06	0.0037794	1.0653130	1.991E-05 2.386E-05	0.0273769	0.8837623	1.853E-05	0.0254829	0.8223786	1.724E-05 2.067E-05	0.023/129	1.1016/19	2.31E-05 2.806E-05	0.031/662	0.9534515	1.999E-05 2.43E-05	0.0274923	0.0210733	4.418E-07 1	0.0006076	0.0250/32 4	4.838E-07	0.0006653
Kam Tin Road	RD 17 18	1900-2000	0.1431584	2.97E-06	0.0040847	0.1328652	2.757E-06	0.003791	0.1241443	2.576E-06	0.0035422	0.8876924	1.842E-05	0.0253282	0.8224062	1.706E-05	0.0234654	0.7714675	1.601E-05	0.022012	1.0308507	2.139E-05	0.0294129	0.8956118	1.858E-05	0.0255542	0.0200319	4.156E-07	0.0005716	0.0219531	4.555E-07	0.0006264
Kam Tin Road	RD_17_18	2000-2100	0.1518862	1.984E-06	0.0027286	0.1414739	1.848E-06	0.0025416	0.1329299	1.736E-06	0.0023881	1.0506304	1.372E-05	0.0188746	0.9764663	1.276E-05	0.0175422	0.9201327	1.202E-05	0.0165302	1.2025166	1.571E-05	0.0216032	1.0530626	1.376E-05	0.0189183	0.0214504	2.802E-07	0.0003854	0.0234858	3.068E-07	0.0004219
Kam Tin Road	RD_17_18	2100-2200	0.1660785	1.768E-06	0.002432	0.1542889	1.643E-06	0.0022594	0.1446709	1.54E-06	0.0021185	1.102812	1.174E-05	0.0161493	1.0233671	1.09E-05	0.0149859	0.9634384	1.026E-05	0.0141083	1.2688905	1.351E-05	0.0185813	1.1081093	1.18E-05	0.0162268	0.0229172	2.44E-07	0.0003356	0.0251134	2.674E-07	0.0003678
Kam Tin Road	RD_17_18	2200-2300	0.219576	1.735E-06	0.0023867	0.2037868	1.611E-06	0.0022151	0.1921212	1.518E-06	0.0020883	1.3276522	1.049E-05	0.014431	1.2320738	9.738E-06	0.0133921	1.1667541	9.221E-06	0.0126821	1.5472282	1.223E-05	0.0168177	1.3588753	1.074E-05	0.0147704	0.0253952	2.007E-07	0.000276	0.0277071	2.19E-07	0.0003012
Kam Tin Road	RD_17_18 RD_19_20	2300-0000	0.0967116	0.204E-07	0.0008614	0.0683749	3.881F-07	0.0007872	0.0592436	3.504E-07	0.0007377	0.5451259	5.4E-06	0.0048376	0.482627	5.059F-06	0.0042988	0.4478161	2.9E-06	0.0039887	0.0398370	4.144E-06	0.005699	0.5300434	5.195F-06	0.0047264	0.0125439	0.124E-08 0	0.0002138	0.0170749	0.935E-08 1.065E-07	0.0001229
Kam Tin Road	RD 19 20	0100-0200	0.0611522	2.133E-07	0.0004658	0.0570535	1.99E-07	0.0004346	0.0543746	1.896E-07	0.0004142	0.8454531	2.948E-06	0.0064397	0.7821809	2.728E-06	0.0059578	0.7447136	2.597E-06	0.0056724	0.9066053	3.162E-06	0.0069055	0.7990882	2.787E-06	0.0060866	0.0143596	5.008E-08	0.0001094	0.0156306	5.451E-08	0.0001191
Kam Tin Road	RD_19_20	0200-0300	0.0718815	2.051E-07	0.000448	0.0654603	1.868E-07	0.0004079	0.0612865	1.749E-07	0.0003819	0.4324079	1.234E-06	0.0026948	0.3797202	1.083E-06	0.0023664	0.3514955	1.003E-06	0.0021905	0.5042894	1.439E-06	0.0031427	0.4127821	1.178E-06	0.0025725	0.0091759	2.618E-08	5.718E-05	0.0099487	2.839E-08	6.2E-05
Kam Tin Road	RD_19_20	0300-0400	0.0808975	1.881E-07	0.0004108	0.0733577	1.705E-07	0.0003725	0.0680198	1.581E-07	0.0003454	0.4967838	1.155E-06	0.0025226	0.431342	1.003E-06	0.0021903	0.3935894	9.151E-07	0.0019986	0.5776813	1.343E-06	0.0029334	0.4616092	1.073E-06	0.002344	0.0102928	2.393E-08	5.227E-05	0.011248	2.615E-08	5.712E-05
Kam Tin Road	RD_19_20	0400-0500	0.0765081	2.345E-07	0.0005121	0.0696998	2.136E-07	0.0004665	0.0652176	1.999E-07	0.0004365	0.5163533	1.582E-06	0.0034563	0.4558643	1.397E-06	0.0030514	0.4223495	1.294E-06	0.0028271	0.5928614	1.817E-06	0.0039684	0.4875671	1.494E-06	0.0032636	0.0104988	3.217E-08	7.028E-05	0.0114464	3.508E-08	7.662E-05
Kam Tin Road	RD_19_20 RD_19_20	0500-0600	0.1655278	8.5/1E-0/ 1.254E-06	0.0018721	0.087113	7.909E-07	0.001/2/6 0	0.1443/44	1.098F-06	0.0016329	1.1338074	5.8/1E-Ub 1 28E-05	0.0128233	1.0440165	5.406E-06	0.0118078	0.9911246	5.132E-06	0.0112096	1.2993351	6./28E-06	0.0146954	1.135499	5.88E-06	0.0128425	0.0212799	1.102E-07 0	0.0002407	0.0231421	1.198E-07 2 738E-07	0.0002617
Kam Tin Road	RD 19 20	0700-0800	0.1380785	3.954E-06	0.008637	0.127834	3.661E-06	0.0079961	0.1192747	3.416E-06	0.0074608	1.0160746	2.91E-05	0.0635565	0.9415646	2.696E-05	0.0588959	0.8844903	2.533E-05	0.0553258	1.154153	3.305E-05	0.0721935	1.0037649	2.875E-05	0.0627866	0.0219847	6.296E-07	0.0013752	0.0240602	6.89E-07	0.001505
Kam Tin Road	RD_19_20	0800-0900	0.1386971	3.943E-06	0.0086116	0.1288847	3.664E-06	0.0080024	0.1195897	3.4E-06	0.0074253	0.9774152	2.779E-05	0.0606871	0.9083796	2.582E-05	0.0564008	0.8483648	2.412E-05	0.0526745	1.1161123	3.173E-05	0.0692988	0.9679545	2.752E-05	0.0600997	0.0215136	6.116E-07	0.0013358	0.023508	6.683E-07	0.0014596
Kam Tin Road	RD_19_20	0900-1000	0.1721863	4.913E-06	0.0107307	0.1610327	4.595E-06	0.0100356	0.1492168	4.258E-06	0.0092992	1.1693217	3.336E-05	0.0728724	1.0960594	3.127E-05	0.0683067	1.0239491	2.922E-05	0.0638128	1.3415081	3.828E-05	0.0836031	1.1731659	3.347E-05	0.073112	0.0243483	6.947E-07	0.0015174	0.0265205	7.567E-07	0.0016528
Kam Tin Road	RD_19_20	1000-1100	0.1711931	4.613E-06	0.0100761	0.1604112	4.323E-06	0.0094415	0.1488381	4.011E-06	0.0087603	1.170872	3.155E-05	0.0689152	1.0999766	2.964E-05	0.0647424	1.0291122	2.773E-05	0.0605715	1.3420651	3.617E-05	0.0789913	1.1779503	3.174E-05	0.0693318	0.0244729	6.595E-07	0.0014404	0.0266176	7.173E-07	0.0015667
Kam Tin Road	RD_19_20 RD_19_20	1200-1200	0.1821932	4.659E-06	0.0101768	0.1705949	4.363E-06	0.009529	0.15/9685	4.04E-06	0.0088237	1.1831/02	3.026E-05	0.0660888	1.1126261	2.845E-05 3.92E-05	0.0621484	1.0399285	2.659E-05	0.0580877	1.3653634	3.492E-05	0.0762656	1.197897	3.063E-05	0.0669114	0.0251996	6.444E-07 I 8 714E-07	0.0014076	0.0274236	7.013E-07	0.0015318
Kam Tin Road	RD 19 20	1300-1400	0.2860191	8.705E-06	0.0190131	0.2684939	8.172E-06	0.0178481	0.2468409	7.513E-06	0.0164087	1.7051789	5.19E-05	0.1133516	1.6083677	4.895E-05	0.1069161	1.4945082	4.549E-05	0.0993473	1.991198	6.06E-05	0.1323647	1.7413491	5.3E-05	0.115756	0.0357235	1.087E-06	0.0023747	0.0389002	1.184E-06	0.0025859
Kam Tin Road	RD_19_20	1400-1500	0.2771974	8.202E-06	0.0179148	0.2605715	7.71E-06	0.0168403	0.2384669	7.056E-06	0.0154117	1.6305912	4.825E-05	0.1053825	1.539461	4.555E-05	0.0994929	1.4244652	4.215E-05	0.0920609	1.9077886	5.645E-05	0.1232973	1.662932	4.921E-05	0.1074726	0.0342635	1.014E-06	0.0022144	0.0373072	1.104E-06	0.0024111
Kam Tin Road	RD_19_20	1500-1600	0.2568605	7.41E-06	0.0161855	0.2414101	6.965E-06	0.0152119	0.2215308	6.391E-06	0.0139593	1.5254937	4.401E-05	0.0961254	1.4402522	4.155E-05	0.0907542	1.3358473	3.854E-05	0.0841753	1.7823542	5.142E-05	0.1123109	1.5573781	4.493E-05	0.0981346	0.0322178	9.295E-07	0.0020301	0.035122	1.013E-06	0.0022131
Kam Tin Road	RD_19_20	1600-1700	0.252857	7.482E-06	0.0163417	0.2367448	7.005E-06	0.0153004	0.2176029	6.439E-06	0.0140633	1.5382253	4.552E-05	0.099413	1.4467294	4.281E-05	0.0934998	1.3436247	3.976E-05	0.0868363	1.7910822	5.3E-05	0.1157548	1.5612276	4.62E-05	0.1008996	0.0330763	9.787E-07	0.0021377	0.0360849	1.068E-06	0.0023321
Kam Tin Road	RD_19_20	1700-1800	0.2195578	6.937E-06	0.0151525	0.2037773	6.439E-06	0.0140635	0.1876884	5.93E-06	0.0129531	1.3830558	4.37E-05	0.0954501	1.2862228	4.064E-05	0.0887672	1.1939112	3.772E-05	0.0823965	1.6026136	5.064E-05	0.1106026	1.3815996	4.365E-05	0.0953496	0.0299534	9.464E-07 (	0.0020672	0.0327586	1.035E-06	0.0022608
Kam Tin Road	RD 19 20	1900-2000	0.1192811	3.366E-06	0.007351	0.1106391	3.122E-06	0.0068184	0.1033536	2.916E-06	0.0063695	0.8951305	2.526E-05	0.0551649	0.830175	2.342E-05	0.0511618	0.7801021	2.201E-05	0.048076	1.0144115	2.862E-05	0.0625159	0.8834556	2.493E-05	0.0544454	0.0240004	5.656E-07	0.0012355	0.0219371	6.19E-07	0.0013519
Kam Tin Road	RD_19_20	2000-2100	0.1324725	2.338E-06	0.0051063	0.1234993	2.18E-06	0.0047604	0.1161958	2.051E-06	0.0044789	1.1281928	1.991E-05	0.0434876	1.0509113	1.855E-05	0.0405087	0.9927541	1.752E-05	0.0382669	1.2606653	2.225E-05	0.0485939	1.1089499	1.957E-05	0.0427458	0.0224852	3.968E-07	0.0008667	0.0245781	4.338E-07	0.0009474
Kam Tin Road	RD_19_20	2100-2200	0.1439412	2.054E-06	0.0044852	0.1340386	1.912E-06	0.0041767	0.1259778	1.797E-06	0.0039255	1.2189947	1.739E-05	0.037984	1.136403	1.621E-05	0.0354104	1.0745938	1.533E-05	0.0334845	1.3629359	1.944E-05	0.0424692	1.2005716	1.713E-05	0.03741	0.0243623	3.476E-07	0.0007591	0.0266188	3.798E-07	0.0008294
Kam Tin Road	RD_19_20	2200-2300	0.1778059	1.804E-06	0.0039399	0.1657571	1.682E-06	0.0036729	0.1569277	1.592E-06	0.0034773	1.5258535	1.548E-05	0.0338103	1.4264406	1.447E-05	0.0316075	1.359145	1.379E-05	0.0301164	1.7036594	1.728E-05	0.0377502	1.5160727	1.538E-05	0.0335936	0.0279365	2.834E-07	0.000619	0.0304416	3.088E-07	0.0006745
Kam Lin Koad	RD_19_20 RD_21_22	2300-0000	0.0880222	7.814E-07	0.001/066	0.0813967	1.649E-07	0.0015782	0.0770363	6.838E-07	0.0014935	0.9245233	2 726E-06	0.01/9251	0.8524987	7.568E-06	0.0165287	0.8097947	7.188E-06	0.015/00/	1.0125455	2 902E-06	0.0196318	0.886831	7.872E-06	0.01/1943	0.01/9696	1.595E-07 I	0.0003484	0.0118385	1./44E-0/ 5.651E-08	0.0003808
Lam Kam Road	RD 21 22	0100-0200	0.0460789	1.257E-07	0.0003412	0.042727	1.165E-07	0.0003164	0.0405572	1.106E-07	0.0003003	0.5770229	1.574E-06	0.0042727	0.5276055	1.439E-06	0.0039068	0.4992281	1.362E-06	0.0036966	0.6231018	1.7E-06	0.0046139	0.5397853	1.472E-06	0.0039969	0.0102716	2.802E-08	7.606E-05	0.0111333	3.037E-08	8.244E-05
Lam Kam Road	RD_21_22	0200-0300	0.0500036	1.193E-07	0.000324	0.0452775	1.081E-07	0.0002934	0.0422706	1.009E-07	0.0002739	0.2537758	6.057E-07	0.0016442	0.2130146	5.084E-07	0.0013801	0.1923529	4.591E-07	0.0012463	0.3037794	7.251E-07	0.0019682	0.2346235	5.6E-07	0.0015201	0.0059799	1.427E-08	3.874E-05	0.0064665	1.543E-08	4.19E-05
Lam Kam Road	RD_21_22	0300-0400	0.0511229	8.716E-08	0.0002366	0.0460972	7.859E-08	0.0002133	0.0426317	7.268E-08	0.0001973	0.3091677	5.271E-07	0.0014308	0.2574073	4.388E-07	0.0011913	0.2290384	3.905E-07	0.00106	0.3602907	6.142E-07	0.0016674	0.2716701	4.632E-07	0.0012573	0.0061708	1.052E-08	2.856E-05	0.0066831	1.139E-08	3.093E-05
Lam Kam Road	RD_21_22	0400-0500	0.0564714	1.348E-07	0.0003659	0.0514365	1.228E-07	0.0003333	0.0481749	1.15E-07	0.0003121	0.4063842	9.7E-07	0.002633	0.3549256	8.471E-07	0.0022996	0.3272125	7.81E-07	0.00212	0.4628556	1.105E-06	0.0029989	0.3753874	8.96E-07	0.0024322	0.0080311	1.917E-08	5.203E-05	0.0087005	2.077E-08	5.637E-05
Lam Kam Road	RD_21_22 RD_21_22	0500-0600	0.0750527	4.598E-07	0.001248	0.1099579	4.249E-07	0.0011535	0.1041393	4.024E-07	0.0010924	0.7834581	5.028E-06	0.0082185	0.7160885	2.767E-06	0.00/511/	0.5838083	2.619E-06	0.00/1103	0.9024328	3.48/E-06	0.0094665	0.7819545	3.022E-06	0.0082027	0.0152365	5.888E-08 1 1 393E-07	0.0001598	0.0150186	6.405E-08	0.0001739
Lam Kam Road	RD 21 22	0700-0800	0.0699961	1.464E-06	0.0039736	0.0649107	1.357E-06	0.0036849	0.0607881	1.271E-06	0.0034509	0.5786933	1.21E-05	0.032852	0.5322556	1.113E-05	0.0302158	0.4989648	1.043E-05	0.0283259	0.6486894	1.357E-05	0.0368256	0.5597529	1.171E-05	0.0317768	0.0135049	2.824E-07	0.0007667	0.0147795	3.091E-07	0.000839
Lam Kam Road	RD_21_22	0800-0900	0.0821318	1.699E-06	0.0046119	0.076378	1.58E-06	0.0042888	0.0710513	1.47E-06	0.0039897	0.5907911	1.222E-05	0.0331742	0.545642	1.129E-05	0.030639	0.5085412	1.052E-05	0.0285557	0.6729229	1.392E-05	0.0377861	0.5795925	1.199E-05	0.0325454	0.0137141	2.837E-07	0.0007701	0.014976	3.098E-07	0.0008409
Lam Kam Road	RD_21_22	0900-1000	0.0872732	1.696E-06	0.0046044	0.0816462	1.587E-06	0.0043075	0.0758436	1.474E-06	0.0040014	0.6212198	1.207E-05	0.0327746	0.5771982	1.122E-05	0.0304521	0.5372597	1.044E-05	0.028345	0.708493	1.377E-05	0.037379	0.6131033	1.192E-05	0.0323463	0.0141746	2.755E-07	0.0007478	0.015487	3.01E-07	0.0008171
Lam Kam Road	RD_21_22	1000-1100	0.1019261	1.958E-06	0.0053146	0.0955395	1.835E-06	0.0049816	0.0887955	1.706E-06	0.0046299	0.6687262	1.284E-05	0.0348683	0.6243357	1.199E-05	0.0325537	0.5825064	1.119E-05	0.0303727	0.7706523	1.48E-05	0.0401828	0.6713019	1.289E-05	0.0350026	0.0146159	2.807E-07	0.0007621	0.0158977	3.054E-07	0.0008289
Lam Kam Road	RD_21_22 RD_21_22	1200-1200	0.1074943	1.93E-06	0.0052401	0.1008029	1.81E-06	0.0049139	0.0936015	1.681E-06	0.0045628	0.68/3316	1.234E-05	0.0335057	0.643842	1.156E-05	0.0313857	1.0214606	1.079E-05	0.0293033	0.794826	1.427E-05	0.038/458	1 1832172	1.248E-05	0.0338662	0.015227	2./34E-0/ I 3.575E-07	0.0007423	0.0257485	2.9/8E-07 3.802E-07	0.0008084
Lam Kam Road	RD 21 22	1300-1400	0.2179415	3.815E-06	0.0103551	0.2054535	3.596E-06	0.0097618	0.1901076	3.327E-06	0.0090327	1.2905353	2.259E-05	0.0613177	1.2209214	2.137E-05	0.0580101	1.1407405	1.997E-05	0.0542005	1.5084768	2.64E-05	0.0716729	1.3308481	2.329E-05	0.0632331	0.0262196	4.589E-07	0.0012458	0.0285687	5E-07	0.0013574
Lam Kam Road	RD_21_22	1400-1500	0.2106102	3.638E-06	0.0098769	0.1987828	3.434E-06	0.0093222	0.1831373	3.164E-06	0.0085885	1.2389176	2.14E-05	0.0581007	1.1731503	2.027E-05	0.0550165	1.0919934	1.887E-05	0.0512105	1.4495278	2.504E-05	0.0679776	1.2751307	2.203E-05	0.059799	0.0252287	4.358E-07	0.0011831	0.0274889	4.749E-07	0.0012891
Lam Kam Road	RD_21_22	1500-1600	0.1924435	3.215E-06	0.008728	0.1815078	3.033E-06	0.0082321	0.1675117	2.799E-06	0.0075973	1.1427893	1.909E-05	0.0518297	1.0811339	1.806E-05	0.0490334	1.0074859	1.683E-05	0.0456932	1.3352328	2.231E-05	0.0605578	1.1749975	1.963E-05	0.0532905	0.0236472	3.951E-07	0.0010725	0.0257704	4.306E-07	0.0011688
Lam Kam Road	RD_21_22	1600-1700	0.180346	3.095E-06	0.0022157	0.1695638	2.91E-06	0.0078996	0.1568443	2.692E-06	0.007307	1.111671	1.908E-05	0.0368387	1.0483056	1.799E-05	0.0240067	0.9788	1.68E-05	0.032344	1.292017	2.217E-05	0.0200544	1.1356444	1.949E-05	0.0529072	0.01234982	4.033E-07 (	0.0006324	0.0256741	4.406E-07	0.00011961
Lam Kam Road	RD_21_22 RD_21_22	1800-1900	0.0665707	1.185E-06	0.0032157	0.0039017	1.104E-06	0.0029968	0.0593259	1.025E-06	0.0027822	0.572298	9.887E-06	0.0208587	0.6413378	9.175E-06 1.152E-05	0.0249067	0.6006194	8.503E-00 1.079E-05	0.023244	0.0408087	1.10/E-05	0.0300544	0.5549732	9.588E-00	0.0200202	0.0152939	2.297E-07	0.0006234	0.0169697	2.515E-07 3.047E-07	0.0006826
Lam Kam Road	RD_21_22	1900-2000	0.0814393	1.481E-06	0.0040202	0.0756582	1.376E-06	0.0037348	0.0708681	1.289E-06	0.0034984	0.6344107	1.154E-05	0.0313174	0.5856272	1.065E-05	0.0289092	0.5500648	1E-05	0.0271537	0.71585	1.302E-05	0.0353376	0.6209329	1.129E-05	0.0306521	0.0143807	2.615E-07	0.0007099	0.0157806	2.87E-07	0.000779
Lam Kam Road	RD_21_22	2000-2100	0.071224	7.609E-07	0.0020656	0.0664247	7.097E-07	0.0019264	0.0626347	6.692E-07	0.0018165	0.6898028	7.37E-06	0.0200054	0.6389769	6.827E-06	0.0185314	0.6028199	6.44E-06	0.0174828	0.7610268	8.131E-06	0.0220711	0.6654547	7.11E-06	0.0192993	0.0142414	1.522E-07	0.000413	0.015572	1.664E-07	0.0004516
Lam Kam Road	RD_21_22	2100-2200	0.0696799	5.702E-07	0.0015479	0.0647974	5.303E-07	0.0014394	0.0610005	4.992E-07	0.0013551	0.6873764	5.625E-06	0.0152694	0.636014	5.205E-06	0.0141285	0.6003221	4.913E-06	0.0133356	0.7570563	6.195E-06	0.0168173	0.6613226	5.412E-06	0.0146907	0.0144079	1.179E-07	0.0003201	0.0157874	1.292E-07	0.0003507
Lam Kam Road	RD_21_22	2200-2300	0.0992009	5.637E-07	0.0015303	0.0925651	5.26E-07	0.0014279	0.087838	4.992E-07	0.001355	0.9607851	5.46E-06	0.0148215	0.8953227	5.088E-06	0.0138116	0.8535092	4.85E-06	0.0131666	1.059986	6.024E-06	0.0163518	0.9413472	5.35E-06	0.0145216	0.0175022	9.946E-08	0.00027	0.0190777	1.084E-07	0.0002943
Route Twisk	RD_21_22 RD_23_24	2300-0000	0.0786592	4.47E-07 4.373E-08	2 594E-05	0.0728461	4.14E-07 3.914F-08	2 321E-05	0.0090358	3.923E-07 3.595E-08	2 132E-05	0.1429084	4.222E-06 3.56E-07	0.0002111	0.1223218	3.02F-07	0.00105695	0.1093663	2 7F-07	0.0001601	0.1619071	4.009E-00 3.998E-07	0.00126748	0.1239262	4.095E-06 3.06E-07	0.0001815	0.0144447	1 44F-08	8 541F-06	0.006401	1 581F-08	9 373F-06
Route Twisk	RD 23 24	0100-0200	0.0172559	2.13E-08	1.263E-05	0.0153153	1.891E-08	1.121E-05 (	0.0141213	1.743E-08	1.034E-05	0.1767267	2.182E-07	0.0001294	0.1482584	1.83E-07	0.0001085	0.1331076	1.643E-07	9.746E-05	0.1939826	2.395E-07	0.000142	0.1472289	1.818E-07	0.0001078	0.0054735	6.757E-09	4.007E-06	0.0059782	7.38E-09	4.377E-06
Route Twisk	RD_23_24	0200-0300	0.0531286	5.466E-08	3.242E-05	0.0481577	4.954E-08	2.938E-05	0.0449196	4.621E-08	2.741E-05	0.2414069	2.484E-07	0.0001473	0.2077899	2.138E-07	0.0001268	0.1896226	1.951E-07	0.0001157	0.2945355	3.03E-07	0.0001797	0.2345421	2.413E-07	0.0001431	0.0070069	7.209E-09	4.275E-06	0.0075826	7.801E-09	4.626E-06
Route Twisk	RD_23_24	0300-0400	0.0602307	3.718E-08	2.205E-05	0.0544728	3.363E-08	1.994E-05	0.0503619	3.109E-08	1.844E-05	0.3083819	1.904E-07	0.0001129	0.2648279	1.635E-07	9.695E-05	0.2389705	1.475E-07	8.748E-05	0.3686127	2.275E-07	0.0001349	0.2893324	1.786E-07	0.0001059	0.0079256	4.892E-09	2.901E-06	0.0085918	5.304E-09	3.145E-06
Route Twisk Route Twick	RD_23_24	0400-0500	0.0437355	4.5E-08	2.668E-05	0.0393966	4.053E-08	2.404E-05	0.0365674	3.762E-08	2.231E-05	0.270172	2.78E-07	0.0001648	0.2298383	2.365E-07	0.0001402	0.2077879	2.138E-07	0.0001268	0.3139075	3.23E-07	0.0001915	0.2443553	2.514E-07	0.0001491	0.007007	7.209E-09	4.275E-06	0.0075866	7.805E-09	4.629E-06
Route Twisk	RD_23_24 RD_23_24	0600-0600	0.0344206	1.204E-07 1.841F-07	0.0001092	0.0309388	1.655F-07	9.816F-05	0.02887	1.542F-07	9.141E-05	0.2557967	1.368F-06	0.0004394	0.2161291	1.156F-06	0.0005835	0.34145/	1.053F-06	0.0006246	0.49243/3	1.553F-06	0.0009208	0.225696	1.207F-06	0.0004156	0.0079516	4.254F-08	2.523F-05	0.008804	4.71E-08	2.793F-05
Route Twisk	RD_23_24	0700-0800	0.0308545	3.619E-07	0.0002146	0.0281245	3.299E-07	0.0001956	0.0260439	3.055E-07	0.0001811	0.2453374	2.877E-06	0.0017064	0.2155946	2.529E-06	0.0014996	0.1966165	2.306E-06	0.0013676	0.2761919	3.239E-06	0.001921	0.2226604	2.611E-06	0.0015487	0.0082582	9.686E-08	5.744E-05	0.0091252	1.07E-07	6.347E-05
Route Twisk	RD_23_24	0800-0900	0.0427188	4.922E-07	0.0002919	0.0393112	4.53E-07	0.0002686	0.0362805	4.18E-07	0.0002479	0.2892997	3.333E-06	0.0019769	0.2586881	2.981E-06	0.0017677	0.2358421	2.718E-06	0.0016116	0.3320186	3.826E-06	0.0022688	0.2721227	3.136E-06	0.0018595	0.008882	1.023E-07	6.069E-05	0.0097541	1.124E-07	6.665E-05
Route Twisk	RD_23_24	0900-1000	0.0430999	5.144E-07	0.000305	0.0398897	4.761E-07	0.0002823	0.0367139	4.381E-07	0.0002598	0.2894683	3.455E-06	0.0020487	0.26075	3.112E-06	0.0018455	0.2371251	2.83E-06	0.0016783	0.3325682	3.969E-06	0.0023537	0.273839	3.268E-06	0.0019381	0.0090725	1.083E-07	6.421E-05	0.0099837	1.191E-07	7.066E-05
Route Twisk	RD_23_24	1000-1100	0.050959	5.243E-07	0.0003109	0.0473851	4.875E-07	0.0002891	0.0437241	4.498E-07	0.0002668	0.3029578	3.117E-06	0.0018484	0.2753171	2.832E-06	0.0016798	0.2514203	2.587E-06	0.001534	0.3539169	3.641E-06	0.0021593	0.2951444	3.036E-06	0.0018008	0.0086033	8.851E-08	5.249E-05	0.0093601	9.63E-08	5.711E-05
Route Twisk	RD 23_24	1200-1200	0.096516	5.09/E-0/ 6.156F-07	0.0003023	0.090077	+./39E-U/ 5.742F-07	0.000281 0	0.0456245	+.305E-U7 5.284F-07	0.0002589	0.548727	2.979E-06 3.5E-06	0.001/669	0.2045218	2.722E-06 3.221F-06	0.0010144	0.4618716	2.946F-06	0.0017479	0.5046/27	3.469E-06	0.0020692	0.544711	2.920E-U6 3.474F-06	0.001/35	0.0091553	8.608F-08	5.105F-05	0.0148033	9.442F-08	3.0/c-05 5.6E-05
Route Twisk	RD_23_24	1300-1400	0.1085834	7.708E-07	0.0004571	0.1017206	7.221E-07	0.0004282	0.0934724	6.635E-07	0.0003935	0.6253324	4.439E-06	0.0026326	0.5798764	4.116E-06	0.0024412	0.5312613	3.771E-06	0.0022366	0.7339158	5.21E-06	0.0030897	0.6247337	4.435E-06	0.0026301	0.0147908	1.05E-07	6.227E-05	0.0161929	1.149E-07	6.817E-05
Route Twisk	RD_23_24	1400-1500	0.1021634	7.147E-07	0.0004239	0.0958034	6.702E-07	0.0003975	0.0875776	6.127E-07	0.0003633	0.5611118	3.925E-06	0.002328	0.5191768	3.632E-06	0.002154	0.4717938	3.301E-06	0.0019574	0.6632752	4.64E-06	0.0027518	0.5593714	3.913E-06	0.0023208	0.0135778	9.499E-08	5.633E-05	0.0148753	1.041E-07	6.172E-05
Route Twisk	RD_23_24	1500-1600	0.0962959	6.638E-07	0.0003936	0.0901951	6.217E-07	0.0003687	0.082559	5.691E-07	0.0003375	0.5339152	3.68E-06	0.0021826	0.4936864	3.403E-06	0.0020181	0.4491444	3.096E-06	0.001836	0.6302111	4.344E-06	0.0025762	0.5317034	3.665E-06	0.0021735	0.0130952	9.027E-08	5.353E-05	0.0143309	9.878E-08	5.858E-05
Koute Twisk	RD_23_24	1600-1700	U.0839849	6.135E-07	U.0003638	u.0783076	5.72E-07	U.0003392 (	U.0717922	5.244E-07	U.000311	U.5296522	3.869E-06	U.0022944	U.4869126	3.557E-06	0.0021093	U.4441049	3.244E-06	U.0019238	U.6136371	4.482E-06	U.0026582	U.5158971	3.768E-06	U.0022348	U.0137538	1.005E-07	5.958E-05	U.015172	1.108E-07	6.572E-05
Route Twisk	RD 23_24	1800-1900	0.0352327	2.9E-07 4.15F-07	0.0002461	0.0323261	2.001E-U/ 3.811F-07	0.0001578 0	0.0296022	2.430E-U7 3.52F-07	0.0001445	0.3627779	2.361E-U6 3.023E-06	0.0014122	0.3235361	2.119E-06 2.696F-06	0.0012564	0.2948629	1.914E-U6 2.457F-06	0.001135	0.3245599	2.0/1E-U6 3.438F-06	0.0015842	0.2021299	2.15/E-U6 2.809F-06	0.0012795	0.0102006	7.294E-08 8.501F-08	4.325E-05 5.041F-05	0.0112906	0.053E-08 9,409F-08	4.770E-US
Route Twisk	RD_23_24	1900-2000	0.0315321	3.504E-07	0.0002078	0.0287316	3.192E-07	0.0001893	0.0265813	2.953E-07	0.0001752	0.2674158	2.971E-06	0.0017621	0.2358686	2.621E-06	0.0015542	0.2152013	2.391E-06	0.001418	0.2989479	3.322E-06	0.0019699	0.2417826	2.686E-06	0.0015932	0.0086341	9.593E-08	5.689E-05	0.0095419	1.06E-07	6.288E-05
Route Twisk	RD_23_24	2000-2100	0.0300218	1.452E-07	8.609E-05	0.0273071	1.32E-07	7.831E-05 (	0.0252935	1.223E-07	7.253E-05	0.3332748	1.612E-06	0.0009557	0.2938321	1.421E-06	0.0008426	0.2680065	1.296E-06	0.0007685	0.3632965	1.757E-06	0.0010418	0.2932999	1.418E-06	0.0008411	0.008995	4.349E-08	2.579E-05	0.0099753	4.823E-08	2.861E-05

| David Name   | Deed ID   |   |  |  
   |   | N  | O2 Emission  | _   
   |  |   |  
  | 4   |  | -   | N  | IO Emission  
   | _  |   |   |  
  | 4.   |  | NOX Em   
   | nission  |  |   
  | PM2  
   | 5 Emission   |  
   | PN   | 110 Emission   |  |
|--|---|---|--
--	---	--
---|---|---
--|---|--|--
--|---|---
---|--|--
--|--|--
--
--
--	--	--
Road Name	Road ID	Hour
   | .onne/year j  | g/km/veh   | g/s/m2   | tonne/year  
   | g/km/veh   | g/s/m2  | age<br>tonne/year  
  | g/km/veh  | g/s/m2   | tonne/year  | g/km/veh   | g/s/m2   
   | n<br>tonne/year  | g/km/veh  | g/s/m2  | tonne/year   
  | g/km/veh   | g/s/m2   | i<br>tonne/year  
   | g/km/veh   | g/s/m2 tonne,  | /year g/kr  
  | m/veh g  
   | g/s/m2 to  | onne/year  
   | g/km/veh   | g/s/m2 t   | onne/year  |
| Route Twisk  | RD_23_24  | 2100-2200   | 0.0370224  | 1.409E-07 8  
   | 3.358E-05   | 0.03365  | 1.281E-07  | 7.596E-05   
   | 0.031133   | 1.185E-07   | 7.028E-05  
  | 0.4258852   | 1.621E-06  | 0.0009614   | 0.3784806  | 1.441E-06  
   | 0.0008544  | 0.3473423   | 1.322E-06   | 0.0007841  
  | 0.4629076  | 1.762E-06  | 0.001045   
   | 0.3784753  | 1.441E-06 0.000  | 8544 0.01   
  | 12624 4.2  
   | 287E-08 2  | 2.542E-05  
   | J.0125039  | 4.76E-08   | 2.823E-05  |
| Route Twisk  | RD 23 24  | 2300-2300   | 0.0517419  | 1.331E-07  
   | 7.892E-05 (   | 0.0470358  | 1.21E-07   | 7.174E-05   
   | 0.0419343  | 1.132E-07   | 6.715E-05  
  | 0.4103332   | 1.283E-06  | 0.0007607   | 0.4446942  | 1.144E-06  
   | 0.0006783  | 0.4138332   | 1.064E-06   | 0.0006312 (  
  | ).5504687  | 1.416E-06  | 0.0008396  
   | 0.4578539  | 1.178E-06 0.000  | 5324 0.01<br>5984 0.01  
  | 123043 3.  
   | .165E-08 1   | 1.877E-05  
   | 0.0135713  | 3.491E-08  | 2.07E-05   |
| Access Road  | RD_25_26  | 0000-0100   | 0.2101043  | 4.748E-08 6  
   | 6.623E-05 0   | 0.1945707  | 4.397E-08  | 6.134E-05   
   | 0.182159   | 4.116E-08   | 5.742E-05  
  | 1.5079585   | 3.408E-07  | 0.0004754   | 1.3994006  | 3.162E-07  
   | 0.0004412  | 1.31356   | 2.968E-07   | 0.0004141  
  | 1.7180627  | 3.882E-07  | 0.0005416  
   | 1.495719   | 3.38E-07 0.000   | 4715 0.02   
  | 187423 6.4   
   | 495E-09 9  | 3.061E-06  
   | J.0311943  | 7.049E-09  | 9.834E-06  |
| Access Road<br>Access Road   | RD_25_26<br>RD_25_26  | 0100-0200   | 0.2104738  | 4.756E-08 6<br>4.774E-08   
   | 3.635E-05 0<br>6.661E-05 (  | 0.1939551<br>0.1943722   | 4.383E-08<br>4.392E-08   | 6.114E-05<br>6.127E-05  
   | 0.1822722  | 4.119E-08<br>4.13E-08   | 5.746E-05<br>5.762E-05   
  | 1.5107487<br>1.5164147  | 3.414E-07<br>3.427E-07   | 0.0004763   | 1.3952808  | 3.153E-07<br>3.16E-07  
   | 0.0004399  | 1.3144501 1.3179172   | 2.97E-07<br>2.978E-07   | 0.0004144 :  
  | L.7212224<br>L.7276955   | 3.889E-07<br>3.904E-07   | 0.0005426  
   | 1.4967223  | 3.382E-07 0.000<br>3.391E-07 0.000   | 4718 0.02<br>4731 0.02  
  | .87423 6.4<br>287423 6./   
   | 495E-09 9.<br>495E-09 9  | 9.061E-06 0  
   | 0.0311943<br>0.0311943   | 7.049E-09 9<br>7.049E-09 9   | 9.834E-06<br>9.834E-06   |
| Access Road  | RD_25_26  | 0300-0400   | 0.2114008  | 4.777E-08 f  
   | 5.664E-05 C   | 0.1949966  | 4.406E-08  | 6.147E-05   
   | 0.1826882  | 4.128E-08   | 5.759E-05  
  | 1.5174587   | 3.429E-07  | 0.0004784   | 1.40251  | 3.169E-07  
   | 0.0004421  | 1.3173761   | 2.977E-07   | 0.0004153  
  | 1.7288595  | 3.907E-07  | 0.000545   
   | 1.5000643  | 3.39E-07 0.000   | 4729 0.02   
  | 287423 6./   
   | .495E-09 9   | 9.061E-06  
   | 0.0311943  | 7.049E-09  | 9.834E-06  |
| Access Road  | RD_25_26  | 0400-0500   | 0.2117103  | 4.784E-08 6  
   | 3.674E-05 0   | 0.1948698  | 4.404E-08  | 6.143E-05   
   | 0.1831787  | 4.139E-08   | 5.775E-05  
  | 1.5195947   | 3.434E-07  | 0.000479  | 1.4016272  | 3.167E-07  
   | 0.0004419  | 1.3207642   | 2.985E-07   | 0.0004164  
  | 1.731305   | 3.912E-07  | 0.0005458  
   | 1.5039429  | 3.398E-07 0.000  | 4741 0.02   
  | 187423 6.4   
   | 495E-09 9  | 3.061E-06  
   | J.0311943  | 7.049E-09 9  | 9.834E-06  |
| Access Road  | RD_25_26  | 0600-0700   | 0.2121238  | 4.793E-08 (  
   | 6.687E-05 C   | 0.1946593  | 4.399E-08  | 6.137E-05   
   | 0.1825249  | 4.125E-08   | 5.754E-05  
  | 1.5225476   | 3.443E-07  | 0.0004803   | 1.4001525  | 3.164E-07  
   | 0.0004414  | 1.3162026   | 2.974E-07   | 0.0004104  
  | 1.7346714  | 3.92E-07   | 0.0005468  
   | 1.4987276  | 3.387E-07 0.000  | 4725 0.02   
  | 287423 6.4   
   | .495E-09 9   | 9.061E-06  
   | 0.0311943  | 7.049E-09 9  | 9.834E-06  |
| Access Road  | RD_25_26  | 0700-0800   | 0.1158919  | 5.238E-08 7  
   | 7.307E-05 0   | 0.1071294  | 4.842E-08  | 6.754E-05   
   | 0.0997456  | 4.508E-08   | 6.289E-05  
  | 0.8818796   | 3.986E-07  | 0.000556  | 0.8131847  | 3.675E-07  
   | 0.0005127  | 0.7567101   | 3.42E-07  | 0.0004771 0  
  | 0.9977715  | 4.509E-07  | 0.0006291  
   | 0.8564557  | 3.871E-07 0.00   | 054 0.01  
  | 17947 8.1  
   | 111E-09 1  | 1.132E-05  
   | 0.019445   | 8.788E-09  | L.226E-05  |
| Access Road<br>Access Road   | RD_25_26<br>RD_25_26  | 0800-0900   | 0.1155777 0.1151931  | 5.223E-08 7<br>5.206E-08   
   | 7.287E-05 0<br>7.263E-05 (  | 0.1073345<br>0.1075436   | 4.851E-08<br>4.86E-08  | 6.767E-05<br>6.781E-05  
   | 0.0994427  | 4.494E-08<br>4.492E-08  | 6.27E-05   
  | 0.8790138   | 3.973E-07<br>3.959E-07   | 0.0005542   | 0.814504   | 3.681E-07<br>3.688E-07   
   | 0.0005135  | 0.7544149   | 3.41E-07<br>3.408E-07   | 0.0004757 (  
  | ).9945915<br>).9912544   | 4.495E-07<br>4.48E-07  | 0.0006271 0.000625   
   | 0.8538576  | 3.859E-07 0.000<br>3.857E-07 0.000   | 5383 0.01<br>1538 0.01  
  | 17947 8.1<br>17947 8.1   
   | 111E-09 1<br>.111E-09 1  | 1.132E-05<br>1.132E-05   
   | 0.019445   | 8.788E-09 1<br>8.788E-09 1   | L.226E-05  |
| Access Road  | RD_25_26  | 1000-1100   | 0.1151789  | 5.205E-08  
   | 7.262E-05 0   | 0.1076438  | 4.865E-08  | 6.787E-05   
   | 0.0995253  | 4.498E-08   | 6.275E-05  
  | 0.8754145   | 3.956E-07  | 0.0005519   | 0.8165086  | 3.69E-07   
   | 0.0005148  | 0.7548696   | 3.412E-07   | 0.0004759  
  | 0.9905934  | 4.477E-07  | 0.0006246  
   | 0.8543949  | 3.861E-07 0.000  | 5387 0.01   
  | 17947 8.1  
   | .111E-09 1   | 1.132E-05  
   | 0.019445   | 8.788E-09  | L.226E-05  |
| Access Road  | RD_25_26<br>RD_25_26  | 1100-1200   | 0.1152946  | 5.211E-08 7  
   | /.269E-05 0   | 0.1077616  | 4.87E-08   | 6.794E-05   
   | 0.0995712  | 4.5E-08   | 6.278E-05  
  | 0.8758501   | 3.958E-07  | 0.0005522   | 0.8173962  | 3.694E-07  
   | 0.0005154  | 0.755241  | 3.413E-07   | 0.0004762 0  
  | 0.9911447  | 4.479E-07  | 0.0006249  
   | 0.8548122  | 3.863E-07 0.000<br>3.866E-07 0.000   | 539 0.01  
  | 17947 8.1  
   | 111E-09 1<br>111E-09 1   | 1.132E-05  
   | 0.019445   | 8.788E-09 1  | L.226E-05  |
| Access Road  | RD_25_26  | 1300-1400   | 0.1154146  | 5.216E-08  
   | 7.277E-05 C   | 0.1083058  | 4.895E-08  | 6.829E-05   
   | 0.0996108  | 4.502E-08   | 6.28E-05   
  | 0.8767991   | 3.963E-07  | 0.0005528   | 0.8214199  | 3.712E-07  
   | 0.0005179  | 0.7554935   | 3.410E-07   | 0.0004763  
  | ).9922137  | 4.484E-07  | 0.0006256  
   | 0.8551044  | 3.865E-07 0.000  | 5391 0.01   
  | 17947 8.2  
   | .111E-09 1   | 1.132E-05  
   | 0.019445   | 3.788E-09  | L.226E-05  |
| Access Road  | RD_25_26  | 1400-1500   | 0.1155346  | 5.222E-08 7  
   | 7.284E-05 0   | 0.1085964  | 4.908E-08  | 6.847E-05   
   | 0.0994503  | 4.495E-08   | 6.27E-05   
  | 0.8777321   | 3.967E-07  | 0.0005534   | 0.8236072  | 3.722E-07  
   | 0.0005193  | 0.754235  | 3.409E-07   | 0.0004755  
  | 0.9932667  | 4.489E-07  | 0.0006262  
   | 0.8536853  | 3.858E-07 0.000  | 5382 0.01   
  | 17947 8.2  
   | 111E-09 1  | 1.132E-05  
   | 0.019445   | 8.788E-09  | L.226E-05  |
| Access Road<br>Access Road   | RD_25_26<br>RD_25_26  | 1500-1600<br>1600-1700  | 0.1154146  | 5.216E-08 7<br>5.205E-08   
   | 7.262E-05 (   | 0.1084418<br>0.1078348   | 4.901E-08<br>4.874E-08   | 6.837E-05<br>6.799E-05  
   | 0.0995411  | 4.499E-08<br>4.482E-08  | 6.276E-05 (  
  | 0.8767991<br>0.8749071  | 3.963E-07<br>3.954E-07   | 0.0005528   | 0.8224396  | 3.717E-07<br>3.696E-07   
   | 0.0005185  | 0.7549338   | 3.412E-07<br>3.399E-07  | 0.000476 0   
  | ).9922137<br>).9900817   | 4.484E-07<br>4.475E-07   | 0.0006256  
   | 0.8544748  | 3.862E-07 0.000<br>3.847E-07 0.000   | 5387 0.01<br>5367 0.01  
  | 17947 8.1<br>17947 8.1   
   | 111E-09 1<br>.111E-09 1  | 1.132E-05<br>1.132E-05   
   | 0.019445   | 8.788E-09 1<br>8.788E-09 1   | L.226E-05  |
| Access Road  | RD_25_26  | 1700-1800   | 0.1158431  | 5.235E-08  
   | 7.304E-05 0   | 0.1076242  | 4.864E-08  | 6.786E-05   
   | 0.0992124  | 4.484E-08   | 6.255E-05  
  | 0.8809883   | 3.982E-07  | 0.0005555   | 0.8165888  | 3.691E-07  
   | 0.0005149  | 0.7526761   | 3.402E-07   | 0.0004746  
  | 0.9968314  | 4.505E-07  | 0.0006285  
   | 0.8518885  | 3.85E-07 0.000   | 5371 0.01   
  | 17947 8.1  
   | .111E-09 1   | 1.132E-05  
   | 0.019445   | 8.788E-09  | L.226E-05  |
| Access Road  | RD_25_26  | 1800-1900   | 0.1158431  | 5.235E-08 7  
   | /.304E-05 0   | 0.1074408  | 4.856E-08  | 6.774E-05   
   | 0.0997708  | 4.509E-08   | 6.29E-05   
  | 0.8809883   | 3.982E-07  | 0.0005555   | 0.8153457  | 3.685E-07  
   | 0.0005141  | 0.7568788   | 3.421E-07   | 0.0004772 0  
  | 0.9968314  | 4.505E-07  | 0.0006285  
   | 0.8566497  | 3.872E-07 0.000  | 5401 0.01   
  | 17947 8.3  
   | 111E-09 1  | 1.132E-05  
   | 0.019445   | 3.788E-09 1  | L.226E-05  |
| Access Road  | RD_25_26  | 2000-2100   | 0.2100948  | 4.748E-08 (  
   | 6.623E-05   | 0.194793   | 4.402E-08  | 6.141E-05   
   | 0.1819684  | 4.112E-08   | 5.736E-05  
  | 1.507534  | 3.407E-07  | 0.0004752   | 1.4007504  | 3.165E-07  
   | 0.0004424  | 1.3122078   | 2.965E-07   | 0.0004142  
  | 1.7176287  | 3.881E-07  | 0.0005415  
   | 1.4901055  | 3.376E-07 0.000  | 471 0.02  
  | 287423 6.4   
   | .495E-09 9   | 9.061E-06  
   | 0.0311943  | 7.049E-09 9  | 9.834E-06  |
| Access Road  | RD_25_26  | 2100-2200   | 0.2107643  | 4.763E-08 (  
   | 3.644E-05 0   | 0.1950594  | 4.408E-08  | 6.149E-05   
   | 0.1819447  | 4.111E-08   | 5.736E-05  
  | 1.5126805   | 3.418E-07  | 0.0004769   | 1.4026693  | 3.17E-07   
   | 0.0004422  | 1.3120281   | 2.965E-07   | 0.0004136  
  | 1.7234447  | 3.895E-07  | 0.0005433  
   | 1.4939727  | 3.376E-07 0.000  | 471 0.02  
  | 187423 6.4   
   | 495E-09 9  | 3.061E-06  
   | J.0311943  | 7.049E-09  | 9.834E-06  |
| Access Road  | RD_25_26<br>RD_25_26  | 2200-2300   | 0.2111218  | 4.7/1E-08 E  
   | 3.656E-05 0<br>6.645E-05 (  | 0.1947261  | 4.4E-U8<br>4.384F-08   | 6.139E-05<br>6.116E-05  
   | 0.1824511  | 4.123E-08<br>4.121E-08  | 5.752E-05  
  | 1.5151467   | 3.424E-07<br>3.419E-07   | 0.0004776   | 1.4003606  | 3.164E-07<br>3.154E-07   
   | 0.0004415  | 1.3155963   | 2.973E-07<br>2.972E-07  | 0.0004147  
  | 1 7237204  | 3.901E-07<br>3.895E-07   | 0.0005442  
   | 1.4980473  | 3.385E-07 0.000<br>3.384F-07 0.000   | 4723 0.02   
  | .87423 6.4<br>287423 6.  
   | 495E-09 9  | 9.061E-06  
   | 0.0311943  | 7.049E-09 9<br>7.049E-09 9   | 9.834E-06  |
| Kam Sheung Road  | RD_27_28  | 0000-0100   | 0.0971086  | 2.374E-07 (  
   | J.0003427 C   | 0.0920492  | 2.25E-07   | 0.0003249   
   | 0.0882324  | 2.157E-07   | 0.0003114  
  | 1.441407  | 3.523E-06  | 0.0050871   | 1.3633921  | 3.333E-06  
   | 0.0048118  | 1.307968  | 3.197E-06   | 0.0046162  
  | 1.5385155  | 3.761E-06  | 0.0054299  
   | 1.3962004  | 3.413E-06 0.004  | 9276 0.02   
  | /35731 5.  
   | .762E-08 /   | 8.32E-05   
   | 0.0256711  | 5.275E-08  | 9.06E-05   |
| Kam Sheung Road  | RD_27_28  | 0100-0200   | 0.0971589  | 1.403E-07 0  
   | J.0002026 0   | 0.0911925  | 1.317E-07  | 0.0001902   
   | 0.0871827  | 1.259E-07   | 0.0001818  
  | 1.3519534   | 1.953E-06  | 0.0028195   | 1.2627065  | 1.824E-06  
   | 0.0026334  | 1.2075292   | 1.744E-06   | 0.0025183  
  | 1.4491123  | 2.093E-06  | 0.0030221  
   | 1.2947119  | 1.87E-06 0.002   | 7001 0.02   
  | :09758 3.0   
   | .03E-08 4  | 1.374E-05  
   | J.0227856  | 3.291E-08 4  | 1.752E-05  |
| Kam Sheung Road  | RD_27_28<br>RD_27_28  | 0300-0400   | 0.1047797  | 9.314E-08 C  
   | J.0001345 C   | 0.0947579  | 8.423E-08  | 0.0001235   
   | 0.0875608  | 7.783E-08   | 0.0001147  
  | 0.5220906   | 4.641E-07  | 0.00067   | 0.4429365  | 3.937E-07  
   | 0.0005685  | 0.3968191   | 3.527E-07   | 0.0005093 (  
  | ).6268703  | 5.572E-07  | 0.0008045  
   | 0.4843799  | 4.306E-07 0.000  | 6216 0.00   
  | 100412 8.  
   | .925E-09 1   | 1.289E-05  
   | 0.0109151  | 9.702E-09 1  | L.401E-05  |
| Kam Sheung Road  | RD_27_28  | 0400-0500   | 0.1033693  | 1.263E-07 C  
   | J.0001824 0   | 0.0947705  | 1.158E-07  | 0.0001672   
   | 0.0890174  | 1.088E-07   | 0.0001571  
  | 0.8473513   | 1.036E-06  | 0.0014953   | 0.7666315  | 9.37E-07   
   | 0.0013528  | 0.7194963   | 8.794E-07   | 0.0012697  
  | 0.9507206  | 1.162E-06  | 0.0016777  
   | 0.8085137  | 9.882E-07 0.001  | 4267 0.01   
  | 49715 1.   
   | .83E-08 2  | 2.642E-05  
   | J.0162657  | 1.988E-08  | 2.87E-05   |
| Kam Sheung Road<br>Kam Sheung Road   | RD_27_28<br>RD_27_28  | 0500-0600   | 0.2253105  | 4.757E-07 0<br>6.863E-07 (   
   | 0006868 0   | 0.2080476<br>0.1201324   | 4.392E-07<br>6.407E-07   | D.0005341<br>D.0009251  
   | 0.1966494  | 4.151E-07<br>6.104E-07  | 0.0005994  
  | 1.5136237<br>1.4409514  | 3.195E-06<br>7.685E-06   | 0.0046136   | 1.4006858  | 2.957E-06<br>7.159E-06   
   | 0.0042693  | 1.3329218   | 2.814E-06<br>6.845E-06  | 0.0040628 :  
  | 1.7389342<br>1.5696251   | 3.671E-06<br>8.371E-06   | 0.0053003  
   | 1.5295712  | 3.229E-06 0.004<br>7.455E-06 0.010   | 5622 0.02<br>7633 0.02  
  | 24957 1.   
   | 358E-08 8.<br>.331E-07 0   | 3.024E-05<br>0.0001922   
   | 0.0285935  | 5.036E-08 8<br>1.452E-07 0   | 3.715E-05  |
| Kam Sheung Road  | RD_27_28  | 0700-0800   | 0.1437826  | 1.63E-06 C   
   | J.0023527 C   | 0.1338977  | 1.518E-06  | 0.002191  
   | 0.1256391  | 1.424E-06   | 0.0020558  
  | 1.1322006   | 1.283E-05  | 0.0185262   | 1.0583592  | 1.199E-05  
   | 0.017318   | 1.0015864   | 1.135E-05   | 0.016389   
  | 1.2759832  | 1.446E-05  | 0.020879   
   | 1.1272255  | 1.278E-05 0.018  | 4448 0.02   
  | 21623 2./  
   | .451E-07 0   | J.0003538  
   | 0.0235889  | 2.673E-07  | 0.000386   |
| Kam Sheung Road  | RD_27_28  | 0800-0900   | 0.1305562  | 1.494E-06 0  
   | J.0021572 (   | 0.121942   | 1.396E-06  | 0.0020149   
   | 0.1138031  | 1.302E-06   | 0.0018804  
  | 1.0067008   | 1.152E-05  | 0.0166342   | 0.9413189  | 1.077E-05  
   | 0.0155538  | 0.8846147   | 1.012E-05   | 0.0146169  
  | 1.137257   | 1.302E-05  | 0.0187914  
   | 0.9984177  | 1.143E-05 0.016  | 4973 0.02   
  | 100126 2.  
   | .29E-07 0.   | 0.0003307  
   | J.0218176  | 2.497E-07 0  | 0003605  |
| Kam Sheung Road  | RD_27_28<br>RD_27_28  | 1000-1100   | 0.132455   | 1.478E-06 C  
   | J.0021345 C   | 0.1317903  | 1.391E-06  | 0.0020085   
   | 0.1229529  | 1.298E-06   | 0.0018738  
  | 1.10563   | 1.167E-05  | 0.0168499   | 1.0422786  | 1.1E-05  
   | 0.0158844  | 0.9800048   | 1.034E-05   | 0.0149354  
  | 1.2456909  | 1.315E-05  | 0.0181785  
   | 1.1029577  | 1.164E-05 0.016  | 8092 0.02   
  | 210883 2.2   
   | .226E-07 0   | 0.0003214  
   | 0.0229303  | 2.42E-07 C   | 1.0003495  |
| Kam Sheung Road  | RD_27_28  | 1100-1200   | 0.156163   | 1.614E-06 C  
   | J.0023298 0   | 0.1466967  | 1.516E-06  | 0.0021886   
   | 0.1364139  | 1.41E-06  | 0.0020352  
  | 1.107725  | 1.145E-05  | 0.0165264   | 1.0449939  | 1.08E-05   
   | 0.0155905  | 0.980866  | 1.014E-05   | 0.0146338  
  | 1.2638881  | 1.306E-05  | 0.0188563  
   | 1.1172798  | 1.155E-05 0.016  | 669 0.02  
  | 20333 2.2  
   | 277E-07 0.   | J.0003287  
   | J.0239796  | 2.478E-07 0  | .0003578   |
| Kam Sheung Road<br>Kam Sheung Road   | RD_27_28<br>RD_27_28  | 1200-1300 1300-1400   | 0.2280896  | 2./12E-06 0<br>3.475E-06 (   
   | J.0039152 0<br>J.0050168 (  | 0.2144268<br>0.2417149   | 2.549E-06<br>3.277E-06   | 0.0036807   
   | 0.1989554  | 2.365E-06<br>3.032E-06  | 0.0034151  
  | 1.3947833   | 1.658E-05<br>1.973E-05   | 0.0239417   | 1.3163497  | 1.565E-05<br>1.867E-05   
   | 0.0225953  | 1.2324169   | 1.465E-05<br>1.743E-05  | 0.0211546  
  | L.6228729<br>L.7117428   | 1.929E-05<br>2.32E-05  | 0.0278569  
   | 1.4313724  | 1.702E-05 0.024<br>2.046E-05 0.029   | 5697 0.02<br>5439 0.02  
  | 28329 3  
   | 234E-07 0.<br>3.84E-07 0   | 0.0004669<br>0.0005544   
   | 0.0308537  | 4.182E-07 0  | 1.0005086  |
| Kam Sheung Road  | RD_27_28  | 1400-1500   | 0.2514125  | 3.296E-06 (  
   | J.0047592 0   | 0.2373798  | 3.112E-06  | 0.0044936   
   | 0.2187372  | 2.868E-06   | 0.0041407  
  | 1.4063067   | 1.844E-05  | 0.0266211   | 1.3316393  | 1.746E-05  
   | 0.0252077  | 1.2384947   | 1.624E-05   | 0.0234444  
  | 1.6577192  | 2.173E-05  | 0.0313803  
   | 1.4572319  | 1.911E-05 0.027  | 5851 0.02   
  | 274123 3.5   
   | .594E-07 0   | J.0005189  
   | D.0298511  | 3.914E-07 0  | .0005651   |
| Kam Sheung Road  | RD_27_28  | 1500-1600   | 0.2338586  | 3.014E-06 0  
   | J.0043519 0   | 0.2206417  | 2.844E-06  | 0.0041059   
   | 0.2036436  | 2.625E-06   | 0.0037896  
  | 1.3336335   | 1.719E-05  | 0.0248175   | 1.2620303  | 1.627E-05  
   | 0.0234851  | 1.1752991   | 1.515E-05   | 0.0218711  
  | 1.5674922  | 2.02E-05   | 0.0291694  
   | 1.3789427  | 1.777E-05 0.025  | 6607 0.02   
  | :62625 3.3   
   | 385E-07 0.   | 1.0004887  
   | J.0286039  | 3.687E-07 C  | 00005323   |
| Kam Sheung Road  | RD_27_28<br>RD_27_28  | 1700-1800   | 0.1033409  | 1.194E-06 C  
   | J.0017241 C   | 0.0966697  | 1.117E-06  | 0.0016128   
   | 0.0899912  | 1.04E-06  | 0.0015014  
  | 0.854697  | 9.876E-06  | 0.0142597   | 0.7990364  | 9.233E-06  
   | 0.013331   | 0.7489744   | 8.655E-06   | 0.0124958  
  | 0.958038   | 1.107E-05  | 0.0159838  
   | 0.8389656  | 9.695E-06 0.013  | 9972 0.01   
  | 172709 1.9   
   | .996E-07 0   | 0.0002881  
   | 0.01887  | 2.181E-07 C  | .0003148   |
| Kam Sheung Road  | RD_27_28  | 1800-1900   | 0.1268242  | 1.564E-06 C  
   | J.0022583 0   | 0.1180684  | 1.456E-06  | 0.0021024   
   | 0.1101932  | 1.359E-06   | 0.0019622  
  | 0.9017656   | 1.112E-05  | 0.0160576   | 0.8373757  | 1.033E-05  
   | 0.014911   | 0.7848042   | 9.679E-06   | 0.0139749  
  | 1.0285898  | 1.269E-05  | 0.0183159  
   | 0.8949974  | 1.104E-05 0.015  | 9371 0.01   
  | 183075 2.2   
   | 258E-07 C  | 0.000326   
   | J.0200329  | 2.471E-07 0  | .0003567   |
| Kam Sheung Road  | RD_27_28<br>RD_27_28  | 2000-2000   | 0.1064647  | 6.362E-07 (  
   | J.0001725 0<br>J.0009186 (  | 0.0850917  | 5.956E-07  | 0.00086   
   | 0.0804664  | 5.633E-07   | 0.00015128   
  | 0.9659321   | 6.762E-06  | 0.00142725  | 0.8213148  | 6.294E-06  
   | 0.00133075   | 0.8507604   | 5.955E-06   | 0.0085983  
  | 1.0568244  | 7.398E-06  | 0.0159975  
   | 0.8694635  | 6.519E-06 0.009  | 4115 0.01   
  | .76485 1.9   
   | .225E-07 0   | 0.0002859<br>0.0001768   
   | 0.0192855 .  | 1.34E-07 C   | .0003125   |
| Kam Sheung Road  | RD_27_28  | 2100-2200   | 0.1034917  | 5.865E-07 C  
   | 1.0008467 0   | 0.0966048  | 5.474E-07  | 0.0007904   
   | 0.0911212  | 5.164E-07   | 0.0007455  
  | 1.0578532   | 5.995E-06  | 0.0086548   | 0.985578   | 5.585E-06  
   | 0.0080635  | 0.9333438   | 5.289E-06   | 0.0076362  
  | 1.161345   | 6.581E-06  | 0.0095016  
   | 1.024465   | 5.805E-06 0.008  | 3817 0.01   
  | .94153 1   
   | i.1E-07 0  | ).0001588  
   | 0.0212531  | 1.204E-07 C  | .0001739   |
| Kam Sheung Road<br>Kam Sheung Road   | RD_27_28<br>RD_27_28  | 2200-2300   | 0.1509234  | 6.372E-07<br>4.846E-07 (   
   | 0.00092 0   | 0.1411127  | 5.958E-07<br>4 506E-07   | 0.0008602   
   | 0.1340167  | 5.658E-07<br>4.28E-07   | 0.000817   
  | 1.4473156   | 6.111E-06<br>4.662E-06   | 0.0088229   | 1.3538966  | 5.716E-06<br>4 324F-06   
   | 0.0082534  | 1.2926031   | 5.458E-06<br>4.122E-06  | 0.0078798  
  | 1.5982389  | 6.748E-06<br>5 147E-06   | 0.0097429  
   | 1.4266198  | 6.024E-06 0.008<br>4.55E-06 0.006  | 5967 0.02   
  | 23969 1.0<br>199478 7'   
   | 012E-07 0.<br>979E-08 0  | 1.0001461 (<br>1.0001152   
   | ).0261409 :<br>0.0217977   | 1.104E-07 C  | 0001594  |
| Sheung Tsuen B/T and Car Park  | RD_29   | 0000-0100   | 0.1894226  | 4.047E-08 (  
   | δ.576E-05 C   | 0.1817684  | 3.884E-08  | 6.31E-05  
   | 0.1759095  | 3.759E-08   | 6.107E-05  
  | 3.6253424   | 7.746E-07  | 0.0012586   | 3.4493633  | 7.37E-07   
   | 0.0011975  | 3.3161794   | 7.086E-07   | 0.0011513  
  | 3.814765   | 8.151E-07  | 0.0013244  
   | 3.4920889  | 7.462E-07 0.001  | 2123 0.06   
  | 505089 1.2   
   | .293E-08 2   | 2.101E-05  
   | 0.0656995  | 1.404E-08  | 2.281E-05  |
| Sheung Tsuen B/T and Car Park  | RD_29   | 0100-0200   | 0.1675706  | 1.79E-08 2   
   | 2.909E-05 0   | 0.1600349  | 1.71E-08   | 2.778E-05   
   | 0.154926   | 1.655E-08   | 2.689E-05  
  | 3.2428766   | 3.465E-07  | 0.0005629   | 3.0622696  | 3.272E-07  
   | 0.0005316  | 2.9426502   | 3.144E-07   | 0.0005108  
  | 3.4104472  | 3.644E-07  | 0.000592   
   | 3.0975762  | 3.309E-07 0.000  | 5377 0.05   
  | 28971 5.6  
   | 651E-09 9  | 3.182E-06  
   | J.0574347 (  | 5.136E-09  | 9.97E-06   |
| Sheung Tsuen B/T and Car Park  | RD_29   | 0300-0400   | 0.0090001  | 9.615E-10  
   | 1.562E-06 C   | 0.0067592  | 7.221E-10  | 1.173E-06   
   | 0.0056937  | 6.083E-10   | 9.883E-07  
  | 0.312414  | 3.338E-08  | 5.423E-05   | 0.2330668  | 2.49E-08   
   | 4.046E-05  | 0.1949777   | 2.083E-08   | 3.384E-05 (  
  | ).3214141  | 3.434E-08  | 5.579E-05  
   | 0.2006714  | 2.144E-08 3.483  | E-05 0.00   
  | 03993 0.7  
   | .884E-10 4   | 4.686E-07  
   | 0.0028897  | 3.087E-10 5  | 5.016E-07  |
| Sheung Tsuen B/T and Car Park  | RD_29   | 0400-0500   | 0.003252   | 3.474E-10 5  
   | 5.645E-07 C   | 0.0025542  | 2.729E-10  | 4.434E-07   
   | 0.0022533  | 2.407E-10   | 3.911E-07  
  | 0.0602055   | 6.432E-09  | 1.045E-05   | 0.0471426  | 5.037E-09  
   | 8.183E-06  | 0.0416908   | 4.454E-09   | 7.237E-06  
  | 0.0634574  | 6.78E-09   | 1.102E-05  
   | 0.043944   | 4.695E-09 7.628  | E-06 0.00   
  | )53993 5.7   
   | 768E-10 9  | 3.372E-07  
   | J.0057794  | 5.175E-10  | L.003E-06  |
| Sheung Tsuen B/T and Car Park<br>Sheung Tsuen B/T and Car Park   | RD_29<br>RD_29  | 0600-0700   | 0.1815047  | 1.939E-08 3<br>7.705E-08 (   
   | J.151E-05 (<br>J.0001252 (  | 0.172639   | 1.844E-08<br>7.259E-08   | 2.997E-05   
   | 0.1658513  | 1.783E-08<br>6.952E-08  | 2.896E-05  
  | 4.0399699<br>3.0276244  | 4.316E-07<br>1.294E-06   | 0.0007013   | 2 8508579  | 4.068E-07<br>1.218E-06   
   | 0.000661   | 2.7310385   | 3.906E-07<br>1.167E-06  | 0.0006345 4  
  | 4.2214745<br>3.2079227   | 4.51E-07<br>1.371E-06  | 0.0007328  
   | 3.8224144<br>2.8937101   | 4.084E-07 0.000<br>1.237E-06 0.002   | 0092 0.05   
  | 51036 8.0<br>543304 2.1  
   | .322E-09 1   | 3.772E-05  
   | 0.0589882  | 2.521E-08 4  | 1.096E-05  |
| Sheung Tsuen B/T and Car Park  | RD_29   | 0700-0800   | 0.1329151  | 0.045.00 (   
   |   |  |  | 0.00011/9   
   |  |   | | |
  |   |  |   |  |  
   |  |   |   |  
  |  |  |  
   |  |  |   
  |  
   |  | 5.716E-05  
   | 0.0510639  | 3.819E-08 6  | 5.205E-05  |
| Sheung Tsuen B/T and Car Park  | RD_29   | 0800-0900   |  | 9.94E-08 (   
   | 0.0001615 6   | 0.1261103  | 9.431E-08  | 0.0001532   
   | 0.1203596  | 9.001E-08   | 0.0001462  
  | 2.4904488   | 1.863E-06  | 0.0030261   | 2.3566842  | 1.762E-06  
   | 0.0028636  | 2.2440968   | 1.678E-06   | 0.0027267  
  | 2.623364   | 1.962E-06  | 0.0031876  
   | 2.3644564  | 1.768E-06 0.002  | 873 0.04  
  | 47043 3.   
   | 518E-08 5  |  
   | 0.044.05.27  | 3 675 09   | 1.338E-05  |
| Sheung Tsuen B/T and Car Park  |   | 0900-1000   | 0.1281945  | 9.94E-08 (<br>8.218E-08 (<br>9.34E-08 (  
   | 0.0001615 (<br>0.0001335 0<br>0001518 0   | 0.1261103<br>0.1210641<br>0.1385867  | 9.431E-08<br>7.761E-08<br>8.884E-08  | 0.0001175<br>0.0001532<br>0.0001261   
   | 0.1203596<br>0.1142306<br>0.1310694  | 9.001E-08<br>7.322E-08<br>8.402E-08   | 0.0001462  
  | 2.4904488<br>1.9842855<br>2.2002208   | 1.863E-06<br>1.272E-06   | 0.0030261   | 2.3566842 1.8789238 2.0977776  | 1.762E-06<br>1.204E-06<br>1.345E-06  
   | 0.0028636<br>0.0019569   | 2.2440968<br>1.7795138  | 1.678E-06<br>1.141E-06  | 0.0027267<br>0.0018534<br>0.0020742  
  | 2.623364<br>2.11248<br>3459273   | 1.962E-06<br>1.354E-06   | 0.0031876  
   | 2.3644564<br>1.8937445<br>2.1225909  | 1.768E-06 0.002<br>1.214E-06 0.001<br>1.361E-06 0.002  | 873 0.04<br>9723 0.03   
  | 47043 3.:<br>183832 2.<br>08499 2.6  
   | .46E-08 3  | 3.998E-05  
   | 10443322   | 2.07E-08 4   | +.01/L 0.5   |
| Sheung Tsuen B/T and Car Park  | RD_29   | 0900-1000<br>1000-1100  | 0.1281945<br>0.1457065<br>0.1360751  | 9.34E-08 (<br>9.34E-08 (<br>8.723E-08 0  
   | 0.0001615 (<br>0.0001335 G<br>0.0001518 O<br>0.0001417 O  | 0.1261103<br>0.1210641<br>0.1385867<br>0.1291377   | 9.431E-08<br>7.761E-08<br>8.884E-08<br>8.278E-08   | 0.0001175<br>0.0001532<br>0.0001261<br>0.0001443<br>0.0001345   
   | 0.1203596<br>0.1142306<br>0.1310694<br>0.1216391   | 9.001E-08<br>7.322E-08<br>8.402E-08<br>7.797E-08  | 0.0001462<br>0.000119<br>0.0001365<br>0.0001267  
  | 2.4904488<br>1.9842855<br>2.2002208<br>2.1078468  | 1.863E-06<br>1.272E-06<br>1.41E-06<br>1.351E-06  | 0.0030261<br>0.0020666<br>0.0022915<br>0.0021953  | 2.3566842<br>1.8789238<br>2.0977776<br>2.0052501   | 1.762E-06<br>1.204E-06<br>1.345E-06<br>1.285E-06   
   | 0.0028636<br>0.0019569<br>0.0021848<br>0.0020885   | 2.2440968<br>1.7795138<br>1.9915216<br>1.8961342  | 1.678E-06<br>1.141E-06<br>1.277E-06<br>1.215E-06  | 0.0027267<br>0.0018534<br>0.0020742<br>0.0019748   
  | 2.623364<br>2.11248<br>2.3459273<br>2.2439219  | 1.962E-06<br>1.354E-06<br>1.504E-06<br>1.438E-06   | 0.0031876<br>0.0022001<br>0.0024433<br>0.002337  
   | 2.3644564<br>1.8937445<br>2.1225909<br>2.0177733   | 1.768E-06 0.002<br>1.214E-06 0.001<br>1.361E-06 0.002<br>1.293E-06 0.002   | 873 0.04<br>9723 0.03<br>2107 0.04<br>1015 0.04   
  | 47043 3<br>383832 2.<br>108499 2.6<br>05571 2  
   | .46E-08 3<br>619E-08 4<br>.6E-08 4   | 3.998E-05<br>1.254E-05<br>1.224E-05  
   | 0.0410527  | 2.842E-08 4  | 1.584E-05  |
| Sheung Tsuen B/T and Car Park<br>Sheung Tsuen B/T and Car Park   | RD_29<br>RD_29  | 0900-1000<br>1000-1100<br>1100-1200   | 0.1281945<br>0.1457065<br>0.1360751<br>0.1607293   | 9.94E-08 (<br>9.34E-08 (<br>9.34E-08 (<br>8.723E-08 (<br>1.03E-07 0  
   | 0.0001615 (<br>0.0001335 (<br>0.0001518 0<br>0.0001417 0<br>0.0001674 0   | 0.1261103<br>0.1210641<br>0.1385867<br>0.1291377<br>0.1521578  | 9.431E-08<br>7.761E-08<br>8.884E-08<br>8.278E-08<br>9.754E-08  | 0.0001175<br>0.0001532<br>0.0001261<br>0.0001443<br>0.0001345<br>0.0001585  
   | 0.1203596<br>0.1142306<br>0.1310694<br>0.1216391<br>0.1426721  | 9.001E-08<br>7.322E-08<br>8.402E-08<br>7.797E-08<br>9.146E-08   | 0.0001462<br>0.000119<br>0.0001365<br>0.0001267<br>0.0001486   
  | 2.4904488<br>1.9842855<br>2.2002208<br>2.1078468<br>2.2364653   | 1.863E-06<br>1.272E-06<br>1.41E-06<br>1.351E-06<br>1.434E-06   | 0.0030261<br>0.0020666<br>0.0022915<br>0.0021953<br>0.0023293   | 2.3566842<br>1.8789238<br>2.0977776<br>2.0052501<br>2.1283629  | 1.762E-06<br>1.204E-06<br>1.345E-06<br>1.285E-06<br>1.364E-06  
   | 0.0028636<br>0.0019569<br>0.0021848<br>0.0020885<br>0.0022167  | 2.2440968<br>1.7795138<br>1.9915216<br>1.8961342<br>2.0102391   | 1.678E-06<br>1.141E-06<br>1.277E-06<br>1.215E-06<br>1.289E-06   | 0.0027267<br>0.0018534<br>0.0020742<br>0.0019748<br>0.0020937  
  | 2.623364<br>2.11248<br>2.3459273<br>2.2439219<br>2.3971946   | 1.962E-06<br>1.354E-06<br>1.504E-06<br>1.438E-06<br>1.537E-06  | 0.0031876<br>0.0022001<br>0.0024433<br>0.002337<br>0.0024967   
   | 2.3644564<br>1.8937445<br>2.1225909<br>2.0177733<br>2.1529112  | 1.768E-06 0.002<br>1.214E-06 0.001<br>1.361E-06 0.002<br>1.293E-06 0.002<br>1.38E-06 0.002   | 873 0.04<br>9723 0.03<br>2107 0.04<br>1015 0.04<br>2422 0.04  
  | 47043 3.<br>383832 2.<br>408499 2.6<br>105571 2.<br>'33125 2.7   
   | .46E-08 3<br>619E-08 4<br>1.6E-08 4<br>776E-08 4   | 3.998E-05<br>4.254E-05<br>4.224E-05<br>4.511E-05   
   | 0.0416527<br>0.0443322<br>0.044017<br>0.0470156  | 2.842E-08<br>2.822E-08<br>3.014E-08  | 1.584E-05<br>1.897E-05   |
|  | RD_29<br>RD_29<br>RD_29<br>RD_29  | 0900-1000<br>1000-1100<br>1100-1200<br>1200-1300<br>1300-1400   | 0.1281945<br>0.1457065<br>0.1360751<br>0.1607293<br>0.1557266<br>0.145988  | 9.34E-08 (<br>9.34E-08 (<br>9.34E-08 (<br>8.723E-08 (<br>1.03E-07 (<br>6.655E-08 ()<br>7.799E-08 ()  
   | 0.0001615 (<br>0.0001335 (<br>0.0001518 (<br>0.0001417 0<br>0.0001674 0<br>0.0001081 0<br>0.0001267 0   | 0.1261103<br>0.1210641<br>0.1385867<br>0.1291377<br>0.1521578<br>0.1491352<br>0.1399004  | 9.431E-08<br>7.761E-08<br>8.884E-08<br>8.278E-08<br>9.754E-08<br>6.373E-08<br>7.473E-08  | 0.0001175<br>0.0001532<br>0.0001261<br>0.0001443<br>0.0001345<br>0.0001585<br>0.0001035<br>0.0001214  
   | 0.1203596<br>0.1142306<br>0.1310694<br>0.1216391<br>0.1426721<br>0.1416376<br>0.1323766  | 9.001E-08<br>7.322E-08<br>8.402E-08<br>7.797E-08<br>9.146E-08<br>6.053E-08<br>7.071E-08   | 0.0001462<br>0.000119<br>0.0001365<br>0.0001267<br>0.0001486<br>9.834E-05<br>0.0001149   
  | 2.4904488<br>1.9842855<br>2.2002208<br>2.1078468<br>2.2364653<br>3.1057198<br>3.0468346   | 1.863E-06<br>1.272E-06<br>1.41E-06<br>1.351E-06<br>1.434E-06<br>1.327E-06<br>1.628E-06   | 0.0030261<br>0.0020666<br>0.0022915<br>0.0021953<br>0.0023293<br>0.0021564<br>0.0026444   | 2.3566842<br>1.8789238<br>2.0977776<br>2.0052501<br>2.1283629<br>2.9552749<br>2.900207   | 1.762E-06<br>1.204E-06<br>1.345E-06<br>1.285E-06<br>1.364E-06<br>1.263E-06<br>1.549E-06  
   | 0.0028636<br>0.0019569<br>0.0021848<br>0.0020885<br>0.0022167<br>0.0020519<br>0.0025171  | 2.2440968<br>1.7795138<br>1.9915216<br>1.8961342<br>2.0102391<br>2.7864095<br>2.7220732   | 1.678E-06<br>1.141E-06<br>1.277E-06<br>1.215E-06<br>1.289E-06<br>1.191E-06<br>1.454E-06   | 0.0027267<br>0.0018534<br>0.0020742<br>0.0019748<br>0.0020937<br>0.0019347<br>0.0019347              
  | 2.623364<br>2.11248<br>2.3459273<br>2.2439219<br>2.3971946<br>3.2614464<br>3.1928226   | 1.962E-06<br>1.354E-06<br>1.504E-06<br>1.438E-06<br>1.537E-06<br>1.394E-06<br>1.706E-06  | 0.0031876<br>0.0022001<br>0.0024433<br>0.002337<br>0.0024967<br>0.0022645<br>0.0027711   
   | 2.3644564<br>1.8937445<br>2.1225909<br>2.0177733<br>2.1529112<br>2.928047<br>2.8544498   | 1.768E-06 0.002<br>1.214E-06 0.001<br>1.361E-06 0.002<br>1.293E-06 0.002<br>1.38E-06 0.002<br>1.38E-06 0.002<br>1.251E-06 0.002  | 873 0.04<br>9723 0.03<br>2107 0.04<br>1015 0.04<br>2422 0.04<br>033 0.05<br>4774 0.05   
  | 147043         3.1           383832         2.           108499         2.1           105571         2           133125         2.7           138011         2.2           138045         2.8  
   | -518E-08 5<br>-46E-08 3<br>-619E-08 4<br>2.6E-08 4<br>776E-08 4<br>299E-08 3<br>374E-08 4  | 3.998E-05<br>4.254E-05<br>4.224E-05<br>4.511E-05<br>3.736E-05<br>4.67E-05  
   | 0.0416527<br>0.0443322<br>0.044017<br>0.0470156<br>).0584313<br>).0584373  | 2.842E-08<br>2.822E-08<br>3.014E-08<br>2.497E-08<br>3.122E-08  | 1.584E-05<br>1.897E-05<br>1.057E-05<br>5.072E-05   |
| Sheung Tsuen B/T and Car Park  | RD_29<br>RD_29<br>RD_29<br>RD_29<br>RD_29<br>RD_29  | 0900-1000<br>1000-1100<br>1100-1200<br>1200-1300<br>1300-1400<br>1400-1500  | 0.1281945<br>0.1457065<br>0.1360751<br>0.1607293<br>0.1557266<br>0.145988<br>0.1482353   | 9.34E-08 (<br>9.34E-08 (<br>9.34E-08 (<br>8.723E-08 (<br>1.03E-07 (<br>6.655E-08 (<br>7.799E-08 (<br>6.335E-08 (   
   | 0.0001615 (<br>0.0001335 (<br>0.0001518 (<br>0.0001518 (<br>0.0001674 (<br>0.0001081 (<br>0.0001081 (<br>0.0001029 ()<br>0.0001029 ()   | 0.1261103<br>0.1210641<br>0.1385867<br>0.1291377<br>0.1521578<br>0.1491352<br>0.1399004<br>0.1423175   | 9.431E-08<br>7.761E-08<br>8.884E-08<br>8.278E-08<br>9.754E-08<br>6.373E-08<br>6.082E-08  | 0.0001175<br>0.0001532<br>0.0001261<br>0.0001443<br>0.0001345<br>0.0001345<br>0.0001035<br>0.0001035<br>0.0001214<br>9.882E-05  
   | 0.1203596<br>0.1142306<br>0.1310694<br>0.1216391<br>0.1426721<br>0.1416376<br>0.1323766<br>0.1323766   | 9.001E-08<br>7.322E-08<br>8.402E-08<br>7.797E-08<br>9.146E-08<br>6.053E-08<br>7.071E-08<br>5.749E-08  | 0.0001462<br>0.000119<br>0.0001365<br>0.0001267<br>0.0001486<br>9.834E-05<br>0.0001149<br>9.34E-05   
  | 2.4904488<br>1.9842855<br>2.2002208<br>2.1078468<br>2.2364653<br>3.1057198<br>3.0468346<br>2.9569573  | 1.863E-06<br>1.272E-06<br>1.41E-06<br>1.351E-06<br>1.327E-06<br>1.327E-06<br>1.628E-06<br>1.264E-06  | 0.0030261<br>0.0020666<br>0.0022915<br>0.0021953<br>0.0023293<br>0.0021564<br>0.0026444<br>0.0020531  | 2.3566842<br>1.8789238<br>2.0977776<br>2.0052501<br>2.1283629<br>2.9552749<br>2.900207<br>2.8217082  | 1.762E-06<br>1.204E-06<br>1.345E-06<br>1.285E-06<br>1.364E-06<br>1.263E-06<br>1.263E-06<br>1.206E-06   
   | 0.0028636<br>0.0019569<br>0.0021848<br>0.0020885<br>0.0022167<br>0.0020519<br>0.0025171<br>0.0019592   | 2.2440968<br>1.7795138<br>1.9915216<br>1.8961342<br>2.0102391<br>2.7864095<br>2.7220732<br>2.6462899  | 1.678E-06<br>1.141E-06<br>1.277E-06<br>1.215E-06<br>1.289E-06<br>1.191E-06<br>1.454E-06<br>1.131E-06  | 0.0027267<br>0.0018534<br>0.0020742 2<br>0.0019748 2<br>0.0020937 2<br>0.0020937 2<br>0.0019347
3<br>0.0023625 3<br>0.0018374 3   | 2.623364<br>2.11248<br>2.3459273<br>2.2439219<br>2.3971946<br>3.2614464<br>3.1928226<br>3.1051925  | 1.962E-06<br>1.354E-06<br>1.504E-06<br>1.438E-06<br>1.537E-06<br>1.394E-06<br>1.706E-06<br>1.327E-06   | 0.0031876<br>0.0022001<br>0.0024433<br>0.002337<br>0.0024967<br>0.0022645<br>0.0027711<br>0.002156   
   | 2.3644564<br>1.8937445<br>2.1225909<br>2.0177733<br>2.1529112<br>2.928047<br>2.8544498<br>2.7808064  | 1.768E-06 0.002<br>1.214E-06 0.001<br>1.361E-06 0.002<br>1.293E-06 0.002<br>1.38E-06 0.002<br>1.525E-06 0.002<br>1.525E-06 0.002<br>1.188E-06 0.001  | 873         0.04           9723         0.03           2107         0.04           1015         0.04           2422         0.04           033         0.05           4774         0.05           9308         0.05   
  | 147043         3.1           383832         2.           108499         2.1           105571         2           133125         2.7           138011         2.2           138045         2.8           14016         2.1  
   | 518E-08 5<br>.46E-08 3<br>.619E-08 4<br>2.6E-08 4<br>776E-08 4<br>299E-08 3<br>874E-08 4<br>197F-08 3  | 3.998E-05<br>4.254E-05<br>4.224E-05<br>4.511E-05<br>3.736E-05<br>4.67E-05<br>1.569E-05   
   | 0.0418527<br>0.0443322<br>0.044017<br>0.0470156<br>0.0584313<br>0.0584373<br>0.0558211   | 2.842E-08<br>2.842E-08<br>2.822E-08<br>3.014E-08<br>2.497E-08<br>3.122E-08<br>2.386E-08  | 1.584E-05<br>1.897E-05<br>1.057E-05<br>5.072E-05<br>3.876E-05  |
| Sheung Tsuen B/T and Car Park<br>Sheung Tsuen B/T and Car Park<br>Sheung Tsuen B/T and Car Park  | RD_29<br>RD_29<br>RD_29<br>RD_29<br>RD_29<br>RD_29<br>RD_29<br>RD_29  | 0900-1000<br>1000-1100<br>1100-1200<br>1200-1300<br>1300-1400<br>1400-1500<br>1500-1600<br>1600-1700  | 0.1281945<br>0.1457065<br>0.1360751<br>0.1607293<br>0.1557266<br>0.145988<br>0.1482353<br>0.1454652<br>0.1525888   | 9.94E-08 (<br>8.218E-08 (<br>9.34E-08 (<br>8.723E-08 (<br>1.03E-07 (<br>6.655E-08 (<br>7.799E-08 (<br>6.335E-08 (<br>7.771E-08 (<br>8.151E-08 ()   
   | 0.0001615 (<br>0.0001335 C<br>0.0001518 C<br>0.0001417 C<br>0.0001674 0<br>0.0001081 0<br>1.0001267 0<br>1.0001263 0<br>0.0001324 0   | 0.1261103<br>0.1210641<br>0.1385867<br>0.1291377<br>0.1521578<br>0.1491352<br>0.1399004<br>0.1423175<br>0.1396503<br>0.1461086   | 9.431E-08<br>7.761E-08<br>8.884E-08<br>8.278E-08<br>9.754E-08<br>6.373E-08<br>6.082E-08<br>7.46E-08<br>7.805E-08   | 0.0001173<br>0.0001532<br>0.0001261<br>0.0001443<br>0.0001345<br>0.0001345<br>0.000135<br>0.0001035<br>0.0001214<br>9.882E-05<br>0.0001212<br>0.0001212   
   | 0.1203596<br>0.1142306<br>0.1310694<br>0.1216391<br>0.1426721<br>0.1416376<br>0.1323766<br>0.1323766<br>0.1345164<br>0.1320173<br>0.1383338  | 9.001E-08<br>7.322E-08<br>8.402E-08<br>7.797E-08<br>9.146E-08<br>6.053E-08<br>7.071E-08<br>5.749E-08<br>7.052E-08<br>7.39E-08   | 0.0001462<br>0.000119<br>0.0001365<br>0.0001267<br>0.0001486<br>9.834E-05<br>0.0001149<br>9.34E-05<br>0.0001146<br>0.0001201   
  | 2.4904488<br>1.9842855<br>2.2002208<br>2.1078468<br>2.2364653<br>3.1057198<br>3.0468346<br>2.9569573<br>2.8602462<br>3.1731567  | 1.863E-06<br>1.272E-06<br>1.41E-06<br>1.351E-06<br>1.351E-06<br>1.327E-06<br>1.628E-06<br>1.264E-06<br>1.528E-06<br>1.695E-06  | 0.0030261<br>0.00220666<br>0.0022915<br>0.0021953<br>0.0023293<br>0.0021564<br>0.0026444<br>0.0020531<br>0.0024824<br>0.002754  | 2.3566842<br>1.8789238<br>2.0977776<br>2.0052501<br>2.1283629<br>2.9552749<br>2.900207<br>2.8217082<br>2.7301698<br>3.0191874  | 1.762E-06<br>1.204E-06<br>1.345E-06<br>1.285E-06<br>1.263E-06<br>1.263E-06<br>1.206E-06<br>1.458E-06<br>1.613E-06  
   | 0.0028636<br>0.0019569<br>0.0021848<br>0.0020885<br>0.0022167<br>0.0025171<br>0.0025171<br>0.0019592<br>0.0023695<br>0.0026204   | 2.2440968<br>1.7795138<br>1.9915216<br>1.8961342<br>2.0102391<br>2.7864095<br>2.7220732<br>2.6462899<br>2.5622851<br>2.8364013  | 1.678E-06<br>1.141E-06<br>1.277E-06<br>1.215E-06<br>1.289E-06<br>1.191E-06<br>1.454E-06<br>1.131E-06<br>1.369E-06<br>1.515E-06  |
0.0027267<br>0.0018534<br>0.0020742<br>0.0019748<br>0.0020937<br>0.0019347<br>0.0023625<br>0.0018374<br>0.0022238<br>0.0022238  | 2.623364<br>2.11248<br>2.3459273<br>2.2439219<br>2.3971946<br>3.2614464<br>3.1928226<br>3.1051925<br>3.0057113<br>3.3257455  | 1.962E-06<br>1.354E-06<br>1.504E-06<br>1.438E-06<br>1.337E-06<br>1.394E-06<br>1.327E-06<br>1.606E-06<br>1.777E-06  | 0.0031876<br>0.0022001<br>0.0024433<br>0.002337<br>0.0022645<br>0.0022645<br>0.00227711<br>0.002156<br>0.00226087<br>0.0028865   
   | 2.3644564<br>1.8937445<br>2.1225909<br>2.0177733<br>2.1529112<br>2.928047<br>2.8544498<br>2.7808064<br>2.6943024<br>2.9747351  | 1.768E-06 0.002<br>1.214E-06 0.001<br>1.36E-06 0.002<br>1.293E-06 0.002<br>1.251E-06 0.002<br>1.525E-06 0.002<br>1.439E-06 0.002<br>1.439E-06 0.002  | 1873         0.04           9723         0.03           2107         0.04           1015         0.04           2422         0.04           1033         0.05           4774         0.05           9308         0.05           3384         0.04           5818         0.05   
  | 147043         3.1           383832         2.           108499         2.1           105571         2           133125         2.7           138011         2.5           138045         2.8           144016         2.1           96214         2.6           65511         3.0   
   | 5.18E-08 5<br>6.619E-08 4<br>2.6E-08 4<br>776E-08 4<br>299E-08 3<br>874E-08 4<br>197E-08 3<br>551E-08 4<br>021E-08 4   | 3.998E-05<br>4.254E-05<br>4.224E-05<br>4.511E-05<br>3.736E-05<br>4.67E-05<br>3.569E-05<br>1.307E-05  
   | 0.0418327<br>0.0443322<br>0.044017<br>0.0470156<br>0.0584313<br>0.0584373<br>0.0584373<br>0.0558211<br>0.0538871<br>0.0538871  | 2.842E-08<br>2.842E-08<br>2.822E-08<br>3.014E-08<br>2.497E-08<br>3.122E-08<br>2.386E-08<br>2.879E-08<br>3.281E-08  | 1.584E-05<br>1.897E-05<br>1.057E-05<br>5.072E-05<br>3.876E-05<br>1.677E-05<br>5.331E-05  |
| Sheung Tsuen B/T and Car Park<br>Sheung Tsuen B/T and Car Park<br>Sheung Tsuen B/T and Car Park<br>Sheung Tsuen B/T and Car Park   | RD_29<br>RD_29<br>RD_29<br>RD_29<br>RD_29<br>RD_29<br>RD_29<br>RD_29<br>RD_29   | 0900-1000<br>1000-1100<br>1100-1200<br>1200-1300<br>1300-1400<br>1400-1500<br>1500-1600<br>1600-1700<br>1700-1800   | 0.1281945<br>0.1457065<br>0.1360751<br>0.1607293<br>0.1557266<br>0.145988<br>0.1482353<br>0.1454652<br>0.1525888<br>0.1332075  | 9.34E-08 (<br>8.218E-08 (<br>9.34E-08 (<br>8.723E-08 (<br>1.03E-07 (<br>6.555E-08 (<br>7.799E-08 (<br>6.335E-08 0<br>7.771E-08 0<br>8.151E-08 0<br>7.116E-08 0   
   | 0.0001615 (<br>0.0001335 (<br>0.0001518 (<br>0.0001417 (<br>0.0001674 (<br>0.0001267 (<br>0.0001267 (<br>0.0001263 (<br>0.0001263 (<br>0.0001324 (<br>0.0001156 (<br>0.000  | 0.1261103<br>0.1210641<br>0.1385867<br>0.1291377<br>0.1521578<br>0.1491352<br>0.1399004<br>0.1423175<br>0.1396503<br>0.1461086<br>0.1267454  | 9.431E-08<br>9.431E-08<br>8.884E-08<br>8.278E-08<br>9.754E-08<br>6.373E-08<br>7.473E-08<br>7.46E-08<br>7.46E-08<br>7.805E-08<br>6.771E-08  | 0.0001173<br>0.0001261<br>0.0001261<br>0.0001443<br>0.0001345<br>0.0001345<br>0.0001035<br>0.0001214<br>9.882E-05<br>0.0001212<br>0.0001268<br>0.00011  
   | 0.1203596<br>0.1142306<br>0.1310694<br>0.1216391<br>0.1426721<br>0.1416376<br>0.1323766<br>0.1345164<br>0.1320173<br>0.1383338<br>0.120155   | 9.001E-08<br>7.322E-08<br>8.402E-08<br>7.797E-08<br>9.146E-08<br>6.053E-08<br>7.071E-08<br>5.749E-08<br>7.052E-08<br>7.39E-08<br>6.419E-08  | 0.0001462<br>0.000119<br>0.0001365<br>0.0001267<br>0.0001486<br>9.834E-05<br>0.0001149<br>9.34E-05<br>0.0001146<br>0.0001201<br>0.0001201   | 2.4904488<br>1.9842855<br>2.2002208<br>2.1078468<br>2.2364653<br>3.1057198<br>3.0468346<br>2.9569573<br>2.8602462<br>3.1731567<br>2.6220547  
  | 1.863E-06<br>1.272E-06<br>1.41E-06<br>1.351E-06<br>1.351E-06<br>1.327E-06<br>1.628E-06<br>1.264E-06<br>1.528E-06<br>1.695E-06<br>1.401E-06   | 0.0030261<br>0.0022915<br>0.0021953<br>0.0021953<br>0.0022564<br>0.0026444<br>0.0020531<br>0.0024824<br>0.002754<br>0.002757  | 2.3566842<br>1.8789238<br>2.0977776<br>2.0052501<br>2.1283629<br>2.9552749<br>2.900207<br>2.8217082<br>2.7301698<br>3.0191874<br>2.4772701   | 1.762E-06<br>1.204E-06<br>1.345E-06<br>1.285E-06<br>1.263E-06<br>1.263E-06<br>1.206E-06<br>1.458E-06<br>1.458E-06<br>1.613E-06<br>1.323E-06  | 0.0028636<br>0.0019569<br>0.0021848<br>0.0020885<br>0.0022167<br>0.0025171<br>0.0025171<br>0.0019592<br>0.0023695<br>0.0023695<br>0.0026204<br>0.0021501  
  | 2.2440968<br>1.7795138<br>1.9915216<br>1.8961342<br>2.0102391<br>2.7864095<br>2.7220732<br>2.6462899<br>2.5622851<br>2.8364013<br>2.3321589   | 1.678E-06<br>1.141E-06<br>1.277E-06<br>1.215E-06<br>1.289E-06<br>1.191E-06<br>1.454E-06<br>1.31E-06<br>1.369E-06<br>1.515E-06<br>1.246E-06  | 0.0027267<br>0.0018534<br>0.0020742 2<br>0.0019748 2<br>0.002937 2<br>0.002937 2<br>0.0018374 2<br>0.0022238 2<br>0.0022238 2<br>0.00224617 2<br>0.00224617 2  
  | 2.623364<br>2.11248<br>2.3459273<br>2.2439219<br>2.3971946<br>3.2614464<br>3.1928226<br>3.1051925<br>3.0057113<br>3.3257455<br>2.7552623   | 1.962E-06<br>1.354E-06<br>1.504E-06<br>1.438E-06<br>1.537E-06<br>1.394E-06<br>1.327E-06<br>1.327E-06<br>1.606E-06<br>1.777E-06<br>1.472E-06  | 0.0031876<br>0.0022001<br>0.0024433<br>0.002337<br>0.0024967<br>0.0022645<br>0.0027711<br>0.002156<br>0.0026087<br>0.0026087<br>0.0028865<br>0.0028865   | 2.3644564<br>1.8937445<br>2.1225909<br>2.0177733<br>2.1529112<br>2.928047<br>2.8544498<br>2.7808064<br>2.6943024<br>2.9747351<br>2.4523139  
  | 1.768E-06         0.002           1.214E-06         0.001           1.361E-06         0.002           1.238E-06         0.002           1.251E-06         0.002           1.525E-06         0.002           1.188E-06         0.002           1.439E-06         0.002           1.525E-06         0.002           1.439E-06         0.002           1.439E-06         0.002           1.31E-06         0.002   | 873         0.04           9723         0.03           2107         0.04           1015         0.04           2422         0.04           033         0.05           4774         0.05           9308         0.05           3384         0.04           5818         0.05           1284         0.04   
  | 147043         3.1           383832         2.           108499         2.1           105571         2           133125         2.5           138012         2.5           138045         2.8           14016         2.1           196214         2.6           55511         3.0           57074         2.4   
   | 5.18E-08         2           2.46E-08         1           .619E-08         4           2.6E-08         4           776E-08         4           299E-08         3           874E-08         4           197E-08         3           651E-08         4           021E-08         4           422E-08         3   | 3.998E-05<br>4.254E-05<br>4.224E-05<br>4.511E-05<br>3.736E-05<br>4.67E-05<br>3.569E-05<br>1.307E-05<br>1.908E-05<br>1.908E-05  | 0.0448527<br>0.0443322<br>0.044017<br>0.0470156<br>0.0584313<br>0.0584373<br>0.0558211<br>0.0538871<br>0.0538871<br>0.0614206<br>0.0496292  
  | 2.842E-08 4<br>2.842E-08 4<br>2.822E-08 4<br>3.014E-08 4<br>2.497E-08 4<br>3.122E-08 5<br>2.386E-08 5<br>2.879E-08 4<br>3.281E-08 5<br>2.651E-08 4   | 1.584E-05<br>1.897E-05<br>1.057E-05<br>5.072E-05<br>3.876E-05<br>1.677E-05<br>5.331E-05<br>1.307E-05   |
| Sheung Tsuen B/T and Car Park<br>Sheung Tsuen B/T and Car Park  | RD_29<br>RD_29<br>RD_29<br>RD_29<br>RD_29<br>RD_29<br>RD_29<br>RD_29<br>RD_29<br>RD_29<br>RD_29<br>RD_29  | 0900-1000<br>1000-1100<br>1100-1200<br>1200-1300<br>1300-1400<br>1400-1500<br>1500-1600<br>1600-1700<br>1700-1800<br>1800-1900  | 0.1281945<br>0.1457065<br>0.1360751<br>0.1607293<br>0.1557266<br>0.145988<br>0.1482353<br>0.1454652<br>0.1525888<br>0.1332075<br>0.1309088   | 9.34E-08 (<br>8.218E-08 (<br>9.34E-08 (<br>1.03E-07 (<br>6.655E-08 (<br>7.799E-08 (<br>6.335E-08 (<br>7.771E-08 (<br>8.151E-08 (<br>7.116E-08 (<br>6.993E-08 (<br>7.069E-08 (<br>9.062) (<br>9.0   | 0.0001615 (<br>0.0001335 (<br>0.0001518 (<br>0.0001417 (<br>0.0001674 (<br>0.0001674 (<br>0.0001267 0<br>0.0001263 0<br>1.0001263 0<br>1.0001324 0<br>1.0001324 0<br>0.0001136 0<br>0.0001136 0  
  | 0.1261103<br>0.1210641<br>0.1385867<br>0.1291377<br>0.1521578<br>0.1491352<br>0.1399004<br>0.1423175<br>0.1396503<br>0.1461086<br>0.1267454<br>0.1245983   | 9.431E-08<br>7.761E-08<br>8.884E-08<br>8.278E-08<br>9.754E-08<br>6.373E-08<br>7.473E-08<br>6.082E-08<br>7.46E-08<br>7.46E-08<br>7.805E-08<br>6.771E-08<br>6.656E-08  | 0.0001173<br>0.0001532<br>0.0001261<br>0.0001443<br>0.0001345<br>0.0001385<br>0.000135<br>0.0001214<br>9.882E-05<br>0.0001212<br>0.0001268<br>0.00011081<br>0.0001081  
      | 0.1203596<br>0.1142306<br>0.1310694<br>0.1216391<br>0.1426721<br>0.1416376<br>0.1323766<br>0.1345164<br>0.1320173<br>0.1383338<br>0.120155<br>0.1189339<br>0.1208562   | 9.001E-08<br>7.322E-08<br>8.402E-08<br>7.797E-08<br>9.146E-08<br>6.053E-08<br>7.071E-08<br>5.749E-08<br>7.052E-08<br>6.419E-08<br>6.353E-08<br>6.456E-08  | 0.0001462<br>0.000119<br>0.0001365<br>0.0001267<br>0.0001486<br>9.834E-05<br>0.0001149<br>9.34E-05<br>0.0001146<br>0.0001201<br>0.0001043<br>0.0001032  | 2.4904488<br>1.9842855<br>2.2002208<br>2.1078468<br>2.2364653<br>3.1057198<br>3.0468346<br>2.9569573<br>2.8602462<br>3.1731567<br>2.620547<br>2.620547<br>2.6509615   
   | 1.863E-06<br>1.272E-06<br>1.41E-06<br>1.351E-06<br>1.434E-06<br>1.327E-06<br>1.628E-06<br>1.528E-06<br>1.695E-06<br>1.416E-06  | 0.0030261<br>0.0022915<br>0.0021953<br>0.0021953<br>0.0021564<br>0.0026444<br>0.0020531<br>0.0024824<br>0.002754<br>0.0022757<br>0.0022088  | 2.3566842<br>1.8789238<br>2.0977776<br>2.0052501<br>2.1283629<br>2.9552749<br>2.900207<br>2.8217082<br>2.7301698<br>3.0191874<br>2.4772701<br>2.5033689<br>3.502825  | 1.762E-06<br>1.204E-06<br>1.345E-06<br>1.285E-06<br>1.263E-06<br>1.263E-06<br>1.458E-06<br>1.458E-06<br>1.323E-06<br>1.337E-06   | 0.0028636<br>0.0019569<br>0.0021848<br>0.0020885<br>0.0022167<br>0.002519<br>0.002519<br>0.0025171<br>0.0019592<br>0.0023695<br>0.0026204<br>0.0021501<br>0.0021727<br>0.0027225   
   | 2.2440968<br>1.7795138<br>1.9915216<br>1.8961342<br>2.0102391<br>2.7864095<br>2.7220732<br>2.6462899<br>2.5622851<br>2.8364013<br>2.3321589<br>2.3726056<br>2.4325101   | 1.678E-06<br>1.141E-06<br>1.217E-06<br>1.215E-06<br>1.289E-06<br>1.191E-06<br>1.369E-06<br>1.369E-06<br>1.246E-06<br>1.267E-06<br>1.267E-06   | 0.0027267<br>0.0018534<br>0.0020742 2<br>0.0019748 2<br>0.002937 2<br>0.0019347 3<br>0.0023625 3<br>0.0018374 3<br>0.0022238 3<br>0.0022238 3<br>0.002241 2<br>0.0020241 2<br>0.0020241 2   
   | 2.623364<br>2.11248<br>2.3459273<br>2.2439219<br>2.3971946<br>3.2614464<br>3.1928226<br>3.1051925<br>3.0057113<br>3.3257455<br>2.7552623<br>2.7818703<br>2.8306521   | 1.962E-06<br>1.354E-06<br>1.504E-06<br>1.438E-06<br>1.537E-06<br>1.394E-06<br>1.327E-06<br>1.606E-06<br>1.777E-06<br>1.472E-06<br>1.486E-06  | 0.0031876<br>0.0022001<br>0.0024433<br>0.002337<br>0.0022645<br>0.0022645<br>0.002756<br>0.002256<br>0.0022685<br>0.0028865<br>0.0028865<br>0.0023913<br>0.0024144   | 2.3644564<br>1.8937445<br>2.1225909<br>2.0177733<br>2.1529112<br>2.928047<br>2.8544498<br>2.7808064<br>2.6943024<br>2.9747351<br>2.4523139<br>2.4915395<br>2.5583663   
   | 1.768E-06 0.002<br>1.214E-06 0.001<br>1.361E-06 0.002<br>1.386E-06 0.002<br>1.38E-06 0.002<br>1.251E-06 0.002<br>1.525E-06 0.002<br>1.188E-06 0.002<br>1.31E-06 0.002<br>1.31E-06 0.002<br>1.31E-06 0.002  | 873         0.04           9723         0.03           2107         0.04           1015         0.04           2422         0.04           033         0.05           4774         0.05           9308         0.05           3384         0.04           5818         0.05           1284         0.04           1624         0.04  
   | 147043         3.1           383832         2.           108499         2.1           105571         2           133125         2.1           133125         2.1           138045         2.8           138045         2.8           14016         2.1           196214         2.6           165511         3.0           '57074         2.4           67677         2.4   
  | 5.18E-08 2<br>2.46E-08 2<br>.619E-08 4<br>7.76E-08 4<br>299E-08 3<br>874E-08 4<br>197E-08 3<br>651E-08 4<br>021E-08 4<br>442E-08 3<br>198E-08 4  | 3.998E-05<br>4.254E-05<br>4.224E-05<br>4.511E-05<br>3.736E-05<br>4.67E-05<br>3.569E-05<br>1.307E-05<br>1.908E-05<br>1.059E-05  | 0.0443322<br>0.044017<br>0.0470156<br>0.0584313<br>0.0584313<br>0.0558211<br>0.0558211<br>0.0538871<br>0.0614206<br>0.046292<br>0.0507781<br>0.0577843   
   | 2.842E-08 4<br>2.842E-08 4<br>2.822E-08 4<br>3.014E-08 4<br>2.497E-08 4<br>3.122E-08 5<br>2.386E-08 5<br>2.879E-08 4<br>3.281E-08 5<br>2.651E-08 4<br>2.713E-08 4  | 1.584E-05<br>1.897E-05<br>1.057E-05<br>5.072E-05<br>3.876E-05<br>3.876E-05<br>5.331E-05<br>1.307E-05<br>1.407E-05<br>1.582E-05   |
| Sheung Tsuen B/T and Car Park<br>Sheung Tsuen B/T and Car Park   | RD_29<br>RD_29<br>RD_29<br>RD_29<br>RD_29<br>RD_29<br>RD_29<br>RD_29<br>RD_29<br>RD_29<br>RD_29<br>RD_29  | 0900-1000<br>1000-1100<br>1100-1200<br>1200-1300<br>1300-1400<br>1400-1500<br>1500-1600<br>1600-1700<br>1700-1800<br>1800-1900<br>1900-2000<br>2000-2100  | 0.1281945<br>0.1457065<br>0.1360751<br>0.1607293<br>0.1557266<br>0.145988<br>0.1482353<br>0.1454652<br>0.1525888<br>0.1332075<br>0.1309088<br>0.132254<br>0.132254<br>0.1352397  | 9.34E-08 (<br>8.218E-08 (<br>9.34E-08 (<br>1.03E-07 (<br>6.655E-08 (<br>7.799E-08 (<br>6.335E-08 (<br>7.771E-08 (<br>8.151E-08 (<br>7.116E-08 ()<br>6.993E-08 ()<br>4.335E-08 7  
   | 0.0001615 (<br>0.0001355 (<br>0.0001358 (<br>0.0001618 (<br>0.0001674 (<br>0.0001674 (<br>0.0001674 (<br>0.0001267 0<br>0.0001267 0<br>0.0001263 0<br>0.0001324 0<br>0.0001136 0<br>0.0001136 0<br>0.0001136 0<br>0.0001136 0   | 0.1261103<br>0.1210641<br>0.1385867<br>0.1291377<br>0.1521578<br>0.1399004<br>0.1423175<br>0.1399004<br>0.1423175<br>0.1396503<br>0.1461086<br>0.1267454<br>0.1245983<br>0.1261241<br>0.1284496  | 9.431E-08<br>7.761E-08<br>8.884E-08<br>8.278E-08<br>9.754E-08<br>6.373E-08<br>7.473E-08<br>7.473E-08<br>7.46E-08<br>7.805E-08<br>6.771E-08<br>6.556E-08<br>6.737E-08<br>4.117E-08  | 0.00011732<br>0.0001261<br>0.0001443<br>0.0001443<br>0.0001345<br>0.0001345<br>0.0001214<br>9.882E-05<br>0.0001212<br>0.0001268<br>0.00011<br>0.0001081<br>0.0001081<br>0.0001095<br>6.689E-05  
   | 0.1203596<br>0.1142306<br>0.1310694<br>0.1216391<br>0.1426721<br>0.1426721<br>0.1323766<br>0.1323766<br>0.1325766<br>0.1325173<br>0.1383338<br>0.120155<br>0.1189339<br>0.1208562<br>0.1229461   | 9.001E-08<br>7.322E-08<br>8.402E-08<br>7.797E-08<br>9.146E-08<br>6.053E-08<br>7.071E-08<br>5.749E-08<br>7.39E-08<br>6.419E-08<br>6.353E-08<br>6.353E-08<br>6.354E-08<br>3.941E-08   | 0.0001462<br>0.000119<br>0.0001365<br>0.0001267<br>0.0001486<br>9.834E-05<br>0.0001149<br>9.34E-05<br>0.0001146<br>0.0001201<br>0.0001043<br>0.0001043<br>0.0001049<br>6.402E-05   
  | 2.4904488<br>1.9842855<br>2.2002208<br>2.1078468<br>2.2364653<br>3.1057198<br>3.0468346<br>2.9569573<br>2.8602462<br>3.1731567<br>2.6220547<br>2.6520615<br>2.7073267<br>2.9117616  | 1.863E-06<br>1.272E-06<br>1.41E-06<br>1.351E-06<br>1.434E-06<br>1.327E-06<br>1.628E-06<br>1.264E-06<br>1.698E-06<br>1.401E-06<br>1.446E-06<br>9.333E-07  | 0.0030261<br>0.0022915<br>0.0021953<br>0.0021953<br>0.0021564<br>0.0026444<br>0.0020531<br>0.002754<br>0.002757<br>0.0022757<br>0.0022008<br>0.0023008<br>0.0023097<br>0.0015163  | 2.3566842<br>1.8789238<br>2.0977776<br>2.0052501<br>2.1283629<br>2.9552749<br>2.900207<br>2.8217082<br>2.7301698<br>3.0191874<br>2.4772701<br>2.5033689<br>2.5607825<br>2.7392982  | 1.762E-06<br>1.204E-06<br>1.345E-06<br>1.263E-06<br>1.263E-06<br>1.263E-06<br>1.263E-06<br>1.458E-06<br>1.613E-06<br>1.323E-06<br>1.337E-06<br>1.337E-06<br>8.78E-07   
   | 0.0028636<br>0.0019569<br>0.0021848<br>0.0020885<br>0.0022167<br>0.0025171<br>0.0025171<br>0.0019592<br>0.0023695<br>0.0023695<br>0.0022604<br>0.0021501<br>0.0021727<br>0.002225<br>0.0014265   | 2.2440968<br>1.7795138<br>1.9915216<br>1.8961342<br>2.0102391<br>2.7864095<br>2.7220732<br>2.6462899<br>2.5622851<br>2.8364013<br>2.3321589<br>2.3726056<br>2.4375101<br>2.6016999  | 1.678E-06<br>1.141E-06<br>1.277E-06<br>1.215E-06<br>1.289E-06<br>1.191E-06<br>1.369E-06<br>1.315E-06<br>1.246E-06<br>1.246E-06<br>1.302E-06<br>8.339E-07  | 0.0027267<br>0.0018534<br>0.0020742<br>0.0019748<br>0.0020937<br>0.0019347<br>0.002374<br>0.0018374<br>0.0022238<br>0.00224617<br>0.0020241<br>0.0020241<br>0.0020241<br>0.0020242<br>0.0020592<br>0.002155<br>0.0013548                 
  | 2.623364<br>2.11248<br>2.3459273<br>2.2439219<br>2.2439219<br>3.3971946<br>3.2614464<br>3.1928226<br>3.1051925<br>3.0057113<br>3.3257455<br>2.7552623<br>2.7818703<br>2.88396521<br>3.0470013  | 1.962E-06<br>1.354E-06<br>1.504E-06<br>1.537E-06<br>1.394E-06<br>1.327E-06<br>1.327E-06<br>1.606E-06<br>1.472E-06<br>1.472E-06<br>1.486E-06<br>1.517E-06<br>9.766E-07  | 0.0031876<br>0.0022001<br>0.0024433<br>0.002337<br>0.0022645<br>0.0022645<br>0.0022645<br>0.0025687<br>0.0028865<br>0.0028865<br>0.0023913<br>0.0024144<br>0.0024646<br>0.0015867  
   | 2.3644564<br>1.8937445<br>2.1225909<br>2.0177733<br>2.1529112<br>2.928047<br>2.8544498<br>2.7808064<br>2.6943024<br>2.9747351<br>2.4523139<br>2.4523139<br>2.455335<br>2.5583663<br>2.724646   | 1.768:-00         0.002           1.214:-06         0.002           1.214:-06         0.002           1.315:1-06         0.002           1.239:-06         0.002           1.38:-06         0.002           1.325:1-06         0.002           1.525:1-06         0.002           1.525:1-06         0.002           1.538:-06         0.002           1.538:-06         0.002           1.318:-06         0.002           1.318:-06         0.002           1.318:-06         0.002           1.367:-06         0.002           1.367:-00         0.002           1.338:-07         0.001   | 873         0.04           9723         0.03           2107         0.04           1015         0.04           2422         0.04           033         0.05           7774         0.05           9308         0.04           5818         0.05           1284         0.04           1624         0.04           1624         0.04           1624         0.04           1624         0.04           1624         0.04           1624         0.04           1624         0.04           1624         0.04           1624         0.04           1624         0.04           1624         0.04           1624         0.04           1624         0.04           1624         0.04           105         0.05  
  | 147043         3.1           383832         2           108499         2.1           105571         2           133125         2.1           133125         2.1           133041         2.4           138045         2.8           14016         2.1           196214         2.6           165571         3.0           157074         2.4           46276         2.5           18381         1.6   
   | 5.18E-08         2           2.46E-08         1           .619E-08         4           2.776E-08         4           299E-08         3           874E-08         4           197E-08         3           651E-08         4           021E-08         4           442E-08         3           498E-08         4           598E-08         4           561E-08         4   | 3.998E-05<br>4.254E-05<br>4.224E-05<br>4.511E-05<br>3.736E-05<br>4.67E-05<br>3.569E-05<br>4.307E-05<br>4.908E-05<br>4.059E-05<br>4.22E-05<br>2.699E-05   
   | 0.0443322<br>0.044017<br>0.0470156<br>0.0584313<br>0.0584313<br>0.0558211<br>0.0538871<br>0.0614206<br>0.0496292<br>0.0496292<br>0.0527943<br>0.0527943  | 2.842E-08 4<br>2.842E-08 4<br>3.014E-08 4<br>2.497E-08 4<br>3.122E-08 5<br>2.386E-08 5<br>2.879E-08 4<br>3.281E-08 4<br>2.651E-08 4<br>2.82E-08 4<br>1.804E-08 2   | 1.584E-05<br>1.897E-05<br>1.057E-05<br>5.072E-05<br>5.876E-05<br>1.677E-05<br>1.307E-05<br>1.307E-05<br>1.407E-05<br>1.582E-05<br>2.932E-05  |
| Sheung Tsuen B/T and Car Park<br>Sheung Tsuen B/T and Car Park   | RD_29<br>RD_29<br>RD_29<br>RD_29<br>RD_29<br>RD_29<br>RD_29<br>RD_29<br>RD_29<br>RD_29<br>RD_29<br>RD_29<br>RD_29<br>RD_29<br>RD_29<br>RD_20  | 0900-1000<br>1000-1100<br>1200-1200<br>1200-1300<br>1300-1400<br>1400-1500<br>1500-1600<br>1500-1600<br>1500-1700<br>1700-1800<br>1900-2000<br>2000-2100<br>2100-2200   | 0.1281945<br>0.1457065<br>0.1360751<br>0.1607293<br>0.1557266<br>0.145988<br>0.1482353<br>0.1454652<br>0.1325888<br>0.1332075<br>0.1309088<br>0.1323254<br>0.1352397<br>0.1792504  | 9.34E-08 (<br>8.218E-08 (<br>8.723E-08 (<br>1.03E-07 (<br>6.655E-08 (<br>7.799E-08 (<br>7.791E-08 (<br>8.151E-08 (<br>7.116E-08 (<br>6.939E-08 (<br>4.335E-08 7<br>5.745E-08 2<br>7.569E-08 (<br>7.569E-08 (<br>7.569E-   | 0.0001615 (<br>0.0001335 (<br>0.0001335 (<br>0.0001518 (<br>0.0001674 (<br>0.0001674 (<br>0.0001674 (<br>0.0001267 0<br>0.0001263 0<br>0.0001263 0<br>0.0001156 0<br>0.0001136 0<br>1.0001136 0<br>1.0001136 0<br>1.0001136 0<br>1.0001136 0   
  | 0.1261103<br>0.1210641<br>0.1385867<br>0.1291377<br>0.1521578<br>0.1399004<br>0.1423175<br>0.1396503<br>0.1461086<br>0.1267454<br>0.1267454<br>0.1267454<br>0.1261241<br>0.1284496<br>0.19957<br>0.2321300   | 9.431E-08<br>7.761E-08<br>8.884E-08<br>8.278E-08<br>9.754E-08<br>6.373E-08<br>6.032E-08<br>7.473E-08<br>6.742E-08<br>7.46E-08<br>6.757E-08<br>6.656E-08<br>6.737E-08<br>4.117E-08<br>5.447E-08<br>7.41E-08   | 0.0001532<br>0.0001532<br>0.0001532<br>0.0001443<br>0.0001345<br>0.0001345<br>0.0001258<br>0.0001214<br>9.882E-05<br>0.0001214<br>9.882E-05<br>0.000110<br>0.0001095<br>6.689E-05<br>8.885E-05<br>0.00015  
                  | 0.1203596<br>0.1142306<br>0.1310694<br>0.1216391<br>0.1426376<br>0.1323766<br>0.1323766<br>0.1320173<br>0.1320173<br>0.120155<br>0.120155<br>0.1189339<br>0.1208562<br>0.1229461<br>0.1624807  | 9.001E-08<br>7.322E-08<br>8.402E-08<br>7.797E-08<br>9.146E-08<br>6.053E-08<br>7.052E-08<br>7.052E-08<br>7.39E-08<br>6.419E-08<br>6.456E-08<br>3.941E-08<br>5.208E-08  | 0.0001462<br>0.000119<br>0.0001365<br>0.0001267<br>0.0001486<br>9.834E-05<br>0.0001149<br>9.34E-05<br>0.0001140<br>0.0001140<br>0.0001101<br>0.0001043<br>0.0001032<br>0.0001032<br>0.0001032   | 2.4904488<br>1.9842855<br>2.2002208<br>2.1078468<br>2.2364653<br>3.1057198<br>3.0468346<br>2.9569573<br>2.8602462<br>3.1731567<br>2.620547<br>2.6509615<br>2.7073267<br>2.9117616<br>4.1828878<br>4.0627072   
   | 1.863E-06<br>1.272E-06<br>1.41E-06<br>1.351E-06<br>1.351E-06<br>1.327E-06<br>1.264E-06<br>1.264E-06<br>1.264E-06<br>1.401E-06<br>1.446E-06<br>9.333E-07<br>1.341E-06   | 0.0030261<br>0.0020666<br>0.0021953<br>0.0021953<br>0.0021564<br>0.0020544<br>0.0020531<br>0.002754<br>0.002757<br>0.0022757<br>0.0022087<br>0.0023008<br>0.0023497<br>0.0015163<br>0.0021782   | 2.3566842<br>1.8789238<br>2.0977776<br>2.0052501<br>2.1283629<br>2.9552749<br>2.900207<br>2.8217082<br>2.7301698<br>3.0191874<br>2.4772701<br>2.5033689<br>2.5607825<br>2.7392982<br>3.9311649   | 1.762E-06<br>1.204E-06<br>1.345E-06<br>1.285E-06<br>1.285E-06<br>1.263E-06<br>1.263E-06<br>1.549E-06<br>1.458E-06<br>1.613E-06<br>1.323E-06<br>1.332E-06<br>1.337E-06<br>1.368E-06<br>8.78E-07<br>1.26E-06   | 0.0028636<br>0.0019569<br>0.0021848<br>0.0020885<br>0.0022167<br>0.00225171<br>0.0025171<br>0.0019592<br>0.0023695<br>0.0026204<br>0.0021501<br>0.0021501<br>0.0021277<br>0.002225<br>0.0014265<br>0.002471  
   | 2.2440968<br>1.7795138<br>1.9915216<br>1.8961342<br>2.0102391<br>2.7864095<br>2.7220732<br>2.6462899<br>2.5622851<br>2.8364013<br>2.3321589<br>2.33726056<br>2.3726056<br>2.4375101<br>2.6016999<br>3.7301495   | 1.678E-06<br>1.141E-06<br>1.277E-06<br>1.215E-06<br>1.289E-06<br>1.131E-06<br>1.369E-06<br>1.515E-06<br>1.246E-06<br>1.246E-06<br>1.246E-06<br>8.339E-07<br>1.196E-06<br>1.492E-06  | 0.0027267<br>0.0018534<br>0.0020742<br>0.0019748<br>0.0020937<br>0.0019347<br>0.0023625<br>0.00123625<br>0.0022238<br>0.0022411<br>0.0020241<br>0.0020241<br>0.0020592<br>0.002155<br>0.0013548<br>0.0013548  
   | 2.623364<br>2.11248<br>2.3459273<br>2.2439219<br>3.2614464<br>3.1928226<br>3.0051123<br>3.0257455<br>2.7552623<br>2.7818703<br>2.8396521<br>3.0470013<br>3.3621382<br>3.1070047  | 1.962E-06<br>1.354E-06<br>1.504E-06<br>1.438E-06<br>1.537E-06<br>1.334E-06<br>1.342E-06<br>1.606E-06<br>1.777E-06<br>1.472E-06<br>1.448E-06<br>1.517E-06<br>9.766E-07<br>1.338E-06<br>1.666E-06  | 0.0031876<br>0.0022001<br>0.0024433<br>0.002337<br>0.0022645<br>0.0022645<br>0.0022645<br>0.0022685<br>0.00228865<br>0.00228865<br>0.00228913<br>0.0024144<br>0.0024646<br>0.0021567<br>0.0022716  | 2.3644564<br>1.8937445<br>2.1225909<br>2.0177733<br>2.1529112<br>2.928047<br>2.75808064<br>2.6943024<br>2.9747351<br>2.4523139<br>2.4523139<br>2.4591395<br>2.5583663<br>2.724646<br>3.8926302<br>4.6016002  
   | 1.768:-00         0.002           1.214E-06         0.001           1.361E-06         0.002           1.391E-06         0.002           1.38E-06         0.002           1.525E-06         0.002           1.525E-06         0.002           1.53E-06         0.002           1.53E-06         0.002           1.53E-06         0.002           1.53E-06         0.002           1.53E-06         0.002           1.53E-06         0.002           1.33E-06         0.002           1.34E-06         0.002           1.34E-06         0.002           1.36F-06         0.002           1.36F-06         0.002           1.36F-06         0.002           1.248E-06         0.002   | 873         0.04           9723         0.03           2107         0.04           1015         0.04           0033         0.05           3038         0.05           3384         0.04           5818         0.05           1284         0.04           1624         0.04           1624         0.04           1624         0.04           1624         0.04           1624         0.04           1624         0.04           2004         0.05           2014         0.05           2024         0.04           2024         0.04           2024         0.04           2024         0.04           2024         0.04           2024         0.04           2024         0.04           4189         0.05           20271         0.07           4422         0.08  
   | 147043         3:           383832         2           408499         2:           405571         2           313125         2:           338011         2:           338045         2:           314016         2:           1514016         2:           15571         3:           165571         3:           154016         2:           157074         2:           188276         2:5           18381         1:           76164         2:4   
  | 5.18E-08         5           2.46E-08         6           6.19E-08         4           2.6E-08         4           2.776E-08         4           2.99E-08         3           874E-08         4           197E-08         3           651E-08         4           422E-08         3           98E-08         4           598E-08         4           561E-08         2           488E-08         4           561E-08         2           488E-08         4   | 3.998E-05<br>4.254E-05<br>4.224E-05<br>4.211E-05<br>3.736E-05<br>4.67E-05<br>3.569E-05<br>4.307E-05<br>3.967E-05<br>4.059E-05<br>4.022E-05<br>1.699E-05<br>1.699E-05   | 0.044322<br>0.044017<br>0.0470156<br>0.0584313<br>0.0584313<br>0.0588211<br>0.0538871<br>0.0538871<br>0.0614206<br>0.046292<br>0.0507781<br>0.0527943<br>0.052958<br>0.063224  
   | 2.842E-08<br>2.842E-08<br>3.014E-08<br>2.497E-08<br>3.014E-08<br>2.497E-08<br>3.122E-08<br>2.386E-08<br>2.3879E-08<br>2.879E-08<br>2.879E-08<br>2.879E-08<br>2.879E-08<br>2.651E-08<br>2.2651E-08<br>2.2713E-08<br>2.82E-08<br>2.82E-08<br>2.82E-08<br>2.82E-08<br>2.82E-08<br>2.82E-08<br>2.82E-08<br>2.82E-08<br>2.82E-08<br>2.82E-08<br>2.82E-08<br>2.82E-08<br>2.82E-08<br>2.82E-08<br>2.82E-08<br>2.82E-08<br>2.82E-08<br>2.82E-08<br>2.82E-08<br>2.832E-08<br>2.82E-08<br>2.82E-08<br>2.82E-08<br>2.832E-08<br>2.832E-08<br>2.832E-08<br>2.832E-08<br>2.832E-08<br>2.832E-08<br>2.832E-08<br>2.832E-08<br>2.832E-08<br>2.832E-08<br>2.832E-08<br>2.832E-08<br>2.832E-08<br>2.832E-08<br>2.832E-08<br>2.832E-08<br>2.832E-08<br>2.832E-08<br>2.832E-08<br>2.832E-08<br>2.832E-08<br>2.832E-08<br>2.832E-08<br>2.832E-08<br>2.832E-08<br>2.832E-08<br>2.832E-08<br>2.832E-08<br>2.832E-08<br>2.832E-08<br>2.832E-08<br>2.832E-08<br>2.832E-08<br>2.832E-08<br>2.832E-08<br>2.832E-08<br>2.832E-08<br>2.832E-08<br>2.832E-08<br>2.832E-08<br>2.832E-08<br>2.832E-08<br>2.832E-08<br>2.832E-08<br>2.832E-08<br>2.832E-08<br>2.832E-08<br>2.832E-08<br>2.832E-08<br>2.832E-08<br>2.832E-08<br>2.832E-08<br>2.832E-08<br>2.832E-08<br>2.832E-08<br>2.832E-08<br>2.832E-08<br>2.832E-08<br>2.832E-08<br>2.832E-08<br>2.832E-08<br>2.832E-08<br>2.832E-08<br>2.832E-08<br>2.832E-08<br>2.832E-08<br>2.832E-08<br>2.832E-08<br>2.832E-08<br>2.832E-08<br>2.832E-08<br>2.832E-08<br>2.832E-08<br>2.832E-08<br>2.832E-08<br>2.832E-08<br>2.832E-08<br>2.832E-08<br>2.832E-08<br>2.832E-08<br>2.832E-08<br>2.832E-08<br>2.832E-08<br>2.832E-08<br>2.832E-08<br>2.832E-08<br>2.832E-08<br>2.832E-08<br>2.832E-08<br>2.832E-08<br>2.832E-08<br>2.832E-08<br>2.832E-08<br>2.832E-08<br>2.832E-08<br>2.832E-08<br>2.832E-08<br>2.832E-08<br>2.832E-08<br>2.832E-08<br>2.832E-08<br>2.832E-08<br>2.832E-08<br>2.832E-08<br>2.832E-08<br>2.832E-08<br>2.832E-08<br>2.832E-08<br>2.832E-08<br>2.832E-08<br>2.832E-08<br>2.832E-08<br>2.832E-08<br>2.832E-08<br>2.832E-08<br>2.832E-08<br>2.832E-08<br>2.832E-08<br>2.832E-08<br>2.832E-08<br>2.832E-08<br>2.832E-08<br>2.832E-08<br>2.832E-08<br>2.832E-08<br>2.832E-08<br>2.832E-08<br>2.832E-08<br>2.832E-08<br>2.832E-08<br>2.832E-08<br>2.832E-08<br>2.832E-08<br>2.832E-08<br>2.832E-08<br>2.832E-08<br>2.832E-08<br>2.832E-08<br>2.832E-08<br>2.832E-08<br>2.832E-08<br>2.832E-08<br>2.832E-08<br>2.832E-08<br>2.832E-08<br>2.832E-08<br>2.832E-08<br>2.832E-08<br>2.832E-08<br>2.832E-08<br>2.832E-08<br>2.832E-08<br>2.832E-08<br>2.832E-08<br>2.832E-08<br>2.832E-08<br>2.832E-08<br>2.832E-08<br>2.832E-08<br>2.832E-08<br>2.832E-08<br>2.832E-08<br>2.832E-08<br>2.832E-08<br>2.832E-08<br>2.832E-08<br>2.832E-08<br>2.832 | 1.584E-05<br>1.897E-05<br>1.057E-05<br>3.072E-05<br>3.876E-05<br>3.876E-05<br>3.31E-05<br>1.307E-05<br>1.407E-05<br>1.582E-05<br>2.932E-05<br>1.391E-05<br>1.989E-05   |
| Sheung Tsuen B/T and Car Park<br>Sheung Tsuen B/T and Car Park   | RD_29<br>RD_29<br>RD_29<br>RD_29<br>RD_29<br>RD_29<br>RD_29<br>RD_29<br>RD_29<br>RD_29<br>RD_29<br>RD_29<br>RD_29<br>RD_29<br>RD_29<br>RD_29<br>RD_29   | 0900-1000<br>1000-1100<br>1200-1300<br>1300-1400<br>1400-1500<br>1500-1600<br>1500-1600<br>1500-1600<br>1500-1800<br>1800-1900<br>2000-2100<br>2200-2300<br>2300-0000   | 0.1281945<br>0.1457065<br>0.1360751<br>0.1607293<br>0.1557266<br>0.145988<br>0.1482353<br>0.1454552<br>0.1525888<br>0.1332075<br>0.1309088<br>0.132254<br>0.132254<br>0.1352397<br>0.1792504<br>0.234202<br>0.864332   | 9.34E-08 (<br>8.218E-08 (<br>8.218E-08 (<br>1.03E-07 (<br>6.555E-08 (<br>7.799E-08 (<br>6.335E-08 (<br>7.771E-08 (<br>8.151E-08 (<br>7.116E-08 (<br>6.993E-08 (<br>7.16E-08 (<br>4.335E-08 (<br>7.545E-08 (<br>7.545E-08 (<br>7.5975E-08 (<br>5.975E-08 (<br>8.975E-08 (<br>8.975E-   | 0.0001615 (<br>0.0001335 (<br>0.0001335 (<br>0.0001417 (<br>0.0001674 (<br>0.0001674 (<br>0.0001027 (<br>0.0001028 (<br>0.0001029 (<br>0.0001263 (<br>0.0001166 (<br>0.0001136 (<br>0.0001  | 0.1261103<br>0.1210641<br>0.1385867<br>0.1291377<br>0.1521578<br>0.1399004<br>0.1423175<br>0.1396503<br>0.1461086<br>0.1267454<br>0.1267454<br>0.1267454<br>0.1269457<br>0.2243199<br>0.2231399<br>0.2762915   
   | 9.431E-08<br>7.761E-08<br>8.884E-08<br>9.754E-08<br>9.754E-08<br>6.373E-08<br>6.082E-08<br>7.473E-08<br>6.082E-08<br>6.771E-08<br>6.656E-08<br>6.737E-08<br>9.437E-08<br>5.447E-08<br>7.152E-08<br>5.65E-08  | 0.0001532<br>0.0001532<br>0.0001261<br>0.0001443<br>0.0001345<br>0.000135<br>0.0001214<br>9.882E-05<br>0.0001214<br>9.882E-05<br>0.0001224<br>0.0001268<br>0.000110<br>0.0001081<br>0.0001081<br>0.0001162<br>9.38E-05  | 0.1203596<br>0.1142306<br>0.1310694<br>0.1216391<br>0.1426721<br>0.1426721<br>0.1320173<br>0.1320173<br>0.1320173<br>0.1320173<br>0.1320173<br>0.1320173<br>0.1320173<br>0.1320173<br>0.1320175<br>0.1209502<br>0.1229461<br>0.1224807<br>0.2148766<br>0.1594062   
   | 9.001E-08<br>7.322E-08<br>8.402E-08<br>7.797E-08<br>9.146E-08<br>6.053E-08<br>7.071E-08<br>7.39E-08<br>6.459E-08<br>6.456E-08<br>6.456E-08<br>6.456E-08<br>6.887E-08<br>5.208E-08<br>6.887E-08<br>5.43E-08  | 0.0001462<br>0.000119<br>0.0001365<br>0.0001267<br>0.0001466<br>9.834E-05<br>0.0001149<br>9.34E-05<br>0.0001140<br>0.0001140<br>0.0001103<br>0.0001032<br>0.0001032<br>0.0001032<br>0.0001043<br>0.0001032<br>0.0001032<br>0.0001043<br>0.0001032   | 2.4904488<br>1.9842855<br>2.2002208<br>2.1078468<br>2.2364653<br>3.1057198<br>3.0468346<br>2.9569573<br>2.8602462<br>3.1731567<br>2.6220547<br>2.6509615<br>2.7073267<br>2.9117616<br>4.1828878<br>4.9637927<br>4.4562831   | 1.863E-06<br>1.272E-06<br>1.371E-06<br>1.351E-06<br>1.327E-06<br>1.628E-06<br>1.628E-06<br>1.628E-06<br>1.695E-06<br>1.401E-06<br>1.446E-06<br>9.333E-07<br>1.341E-06<br>1.591E-06  
  | 0.0030261<br>0.0022915<br>0.0021953<br>0.0021953<br>0.0021564<br>0.0026444<br>0.0020544<br>0.002754<br>0.002754<br>0.0022757<br>0.0023006<br>0.002182<br>0.002182<br>0.002182   | 2.3566842<br>1.8789238<br>2.0977776<br>2.0052501<br>2.1283629<br>2.9552749<br>2.900207<br>2.8217082<br>2.7301698<br>3.0191874<br>2.4772701<br>2.5033689<br>2.5007825<br>2.7392982<br>3.9311649<br>4.68371  | 1.762E-06<br>1.204E-06<br>1.345E-06<br>1.345E-06<br>1.285E-06<br>1.263E-06<br>1.549E-06<br>1.549E-06<br>1.323E-06<br>1.337E-06<br>1.337E-06<br>1.368E-06<br>1.501E-06<br>1.501E-06   | 0.0028636<br>0.0019559<br>0.0021848<br>0.002888<br>0.002888<br>0.002819<br>0.0025171<br>0.0025519<br>0.0015522<br>0.0026204<br>0.0021501<br>0.0022125<br>0.0022225<br>0.0022471<br>0.0024239<br>0.002439   | 2.2440968<br>1.7795138<br>1.9915216<br>1.8961342<br>2.0102391<br>2.7864095<br>2.7220732<br>2.6462899<br>2.5622851<br>2.8364013<br>2.3321589<br>2.3726056<br>2.4375101<br>2.640599<br>3.7301495<br>4.4767836           
   | 1.6782-06<br>1.141E-06<br>1.277E-06<br>1.215E-06<br>1.2459E-06<br>1.454E-06<br>1.315E-06<br>1.246E-06<br>1.246E-06<br>1.302E-06<br>8.339E-07<br>1.196E-06<br>1.435E-06<br>1.427E-06   | 0.0027267<br>0.0018534<br>0.0020742<br>0.0020937<br>0.002937<br>0.0019347<br>0.0023625<br>0.0018374<br>0.002238<br>0.002241<br>0.0020241<br>0.0020241<br>0.0020592<br>0.002155<br>0.002155<br>0.0013548<br>0.002313<br>0.0023313  | 2.623364<br>2.11248<br>2.3459273<br>2.2439219<br>2.3971946<br>3.2614464<br>3.1051925<br>3.1051925<br>3.1051925<br>3.1051925<br>3.1051925<br>3.1051925<br>3.275455<br>2.7552623<br>2.7582623<br>2.7818703<br>2.8396521<br>3.0470013<br>3.3621382<br>3.1979947<br>1.6427163   
  | 1.962E-06<br>1.354E-06<br>1.504E-06<br>1.537E-06<br>1.337E-06<br>1.394E-06<br>1.706E-06<br>1.377E-06<br>1.472E-06<br>1.472E-06<br>1.486E-06<br>9.766E-07<br>1.398E-06<br>1.666E-06<br>1.488E-06  | 0.0031876<br>0.0022001<br>0.0024433<br>0.002337<br>0.002337<br>0.0022645<br>0.0027711<br>0.0022665<br>0.0025865<br>0.0028865<br>0.0028865<br>0.0024913<br>0.0024414<br>0.0026467<br>0.0022667<br>0.0022716<br>0.0027168  | 2.3644564<br>1.8937445<br>2.1225909<br>2.0177733<br>2.1529112<br>2.928047<br>2.8544498<br>2.6943024<br>2.6943024<br>2.6943024<br>2.6943024<br>2.495395<br>2.5583663<br>2.724646<br>3.8926302<br>4.6916603<br>4.1449386   | 1.768:-00         0.002           1.214E-06         0.001           1.361E-06         0.002           1.361E-06         0.002           1.38E-06         0.002           1.35E-06         0.002           1.525E-06         0.002           1.53E-06         0.002           1.53E-06         0.002           1.53E-06         0.002           1.53E-06         0.002           1.53E-06         0.002           1.33E-06         0.002           1.33E-06         0.002           1.34E-06         0.002           1.53E-66         0.002           1.54E-66         0.002           1.54E-66         0.002           1.54E-66         0.002           1.54E-66         0.002  
  | 873         0.04           9723         0.03           2107         0.04           1015         0.04           0033         0.05           3384         0.04           5818         0.05           3384         0.04           5818         0.05           1284         0.04           2200         0.44           2201         0.04           1624         0.04           1624         0.04           1624         0.04           1624         0.04           2004         0.04           2010         0.04           2024         0.04           2024         0.04           2024         0.04           2024         0.04           2024         0.04           2024         0.04           2024         0.04           4189         0.05           0.071         0.07           4432         0.08           1585         0.02  | 147043         3:           147043         3:           183832         2           408499         2:           105571         2:           133125         2:           133125         2:           133125         2:           133125         2:           1338045         2:           196214         2:           165511         3:           167677         2:4           168276         2:5           118381         1:6           76164         2:4           164276         2:5           183818         1:6           76445         2:           31928         2:6   
   
  | 5.182-08         2           2.46E-08         2           2.6E-08         4           2.6E-08         4           2.6E-08         4           2.99E-08         3           874E-08         3           651E-08         4           021E-08         3           498E-08         4           598E-08         4           661E-08         2           488E-08         4           77F-08         4           5256E-08         4   | 3.998E-05<br>4.254E-05<br>4.254E-05<br>4.511E-05<br>3.736E-05<br>3.736E-05<br>3.569E-05<br>4.307E-05<br>4.307E-05<br>4.908E-05<br>4.22E-05<br>4.22E-05<br>1.042E-05<br>1.042E-05<br>1.501E-05  | 0.044322<br>0.044017<br>0.0470156<br>0.0584313<br>0.0558211<br>0.0538871<br>0.0496292<br>0.0496292<br>0.0496292<br>0.0496292<br>0.0496292<br>0.0527943<br>0.0527943<br>0.0527943<br>0.0562958<br>0.0843242<br>0.0938934<br>0.089168  | 2.842E-08 4<br>2.842E-08 4<br>2.822E-08 4<br>2.497E-08 4<br>2.3122E-08 2<br>2.356E-08 2<br>2.879E-08 4<br>2.837E-08 4<br>2.651E-08 4<br>2.821E-08 4<br>2.821E-08 4<br>2.822E-08 4<br>2.822E-08 4<br>2.822E-08 4<br>2.832E-08 4  
  | 1.584E-05<br>1.897E-05<br>1.057E-05<br>5.072E-05<br>8.876E-05<br>1.677E-05<br>1.331E-05<br>1.307E-05<br>1.582E-05<br>1.391E-05<br>1.889E-05<br>1.889E-05<br>1.636E-05  |
| Sheung Tsuen B/T and Car Park<br>Sheung Tsuen B/T and Car Park  | RD_29<br>RD_29<br>RD_29<br>RD_29<br>RD_29<br>RD_29<br>RD_29<br>RD_29<br>RD_29<br>RD_29<br>RD_29<br>RD_29<br>RD_29<br>RD_29<br>RD_29<br>RD_29<br>RD_29<br>RD_29  | 0900-1000<br>1000-1100<br>1200-1300<br>1300-1400<br>1400-1500<br>1500-1600<br>1500-1600<br>1500-1600<br>1700-1800<br>1900-2000<br>2000-2100<br>2200-2300<br>2300-0000<br>0000-0100  | 0.1281945<br>0.1457065<br>0.1360751<br>0.1607293<br>0.1557266<br>0.1456252<br>0.1454652<br>0.1454652<br>0.132075<br>0.1302058<br>0.1322524<br>0.1322324<br>0.1322324<br>0.1322324<br>0.1322324<br>0.1324232<br>0.1264332<br>0.1684332<br>0.030941  | 9.34E-08 (<br>9.34E-08 (<br>9.34E-08 (<br>1.03E-07 (<br>6.655E-08 (<br>7.799E-08 (<br>7.799E-08 (<br>7.771E-08 (<br>8.151E-08 (<br>6.993E-08 (<br>7.116E-08 (<br>6.993E-08 (<br>7.345E-08 (<br>7.506E-08 (<br>7.506E-08 (<br>7.506E-08 (<br>7.506E-08 (<br>9.505E-08 (<br>1.505E-08 (<br>7.505E-08 (<br>1.505E-08 (<br>1.505E-08) | 0.0001615 (<br>0.0001383 (<br>0.0001318 (<br>0.0001318 (<br>0.0001614 (<br>0.0001641 (<br>0.000164 (<br>0.0001084 (<br>0.000126 (<br>0.000126 (<br>0.000126 (<br>0.000115 (<br>0.000115 (<br>0.000116 (<br>0.000116 (<br>0.000116 (<br>0.000116 (<br>0.00012 (<br>0.00016 (<br>0.000016 (<br>0.00016 (<br>0.00016 (<br>0.00016 (<br>0.00016 (<br>0.00016 (<br>0                | 0.1261103<br>0.1210641<br>0.1383867<br>0.1291377<br>0.1291377<br>0.1491352<br>0.1491352<br>0.1491352<br>0.1491352<br>0.1491352<br>0.1491352<br>0.1491353<br>0.1461086<br>0.1267454<br>0.1264496<br>0.169557<br>0.2231399<br>0.2762915<br>0.0762915   
   | 9.431E-08<br>7.761E-08<br>8.884E-08<br>8.278E-08<br>9.754E-08<br>6.373E-08<br>6.082E-08<br>6.082E-08<br>6.082E-08<br>6.73E-08<br>6.656E-08<br>6.737E-08<br>7.417E-08<br>5.447E-08<br>7.452E-08<br>5.447E-08<br>7.452E-08<br>5.447E-08<br>7.452E-08<br>5.447E-08<br>7.452E-08<br>5.455E-08  | 0.0001532<br>0.0001532<br>0.0001526<br>0.0001261<br>0.0001443<br>0.000135<br>0.000135<br>0.0001214<br>9.882E-05<br>0.0001214<br>9.882E-05<br>0.0001268<br>0.000110<br>0.0001081<br>0.0001081<br>0.0001162<br>9.3825-05<br>0.0001162<br>9.3826-05  | 0.1203596<br>0.1142306<br>0.1310694<br>0.1216391<br>0.1426721<br>0.1426721<br>0.1323766<br>0.1320173<br>0.1320173<br>0.1383338<br>0.120155<br>0.1189339<br>0.1208562<br>0.1229461<br>0.1624407<br>0.2148766<br>0.01694062<br>0.01694062  
   | 9.001E-08<br>7.322E-08<br>8.402E-08<br>7.797E-08<br>9.146E-08<br>6.053E-08<br>7.052E-08<br>7.052E-08<br>7.39E-08<br>6.419E-08<br>6.435E-08<br>5.208E-08<br>5.208E-08<br>5.208E-08<br>5.343E-08<br>5.43E-08<br>5.43E-08  | 0.0001462<br>0.000119<br>0.0001365<br>0.0001267<br>0.0001467<br>9.834E-05<br>0.0001149<br>9.34E-05<br>0.0001140<br>0.0001140<br>0.0001103<br>0.0001103<br>0.0001032<br>0.0001032<br>0.0001032<br>0.0001032<br>0.0001032<br>0.000119<br>6.402E-05<br>8.461E-05<br>0.0001119  | 2.4904488<br>1.9842855<br>2.2002208<br>2.1078468<br>2.2364653<br>3.0468346<br>2.9569573<br>2.8602462<br>3.1731567<br>2.6220547<br>2.6509615<br>2.7073267<br>2.9117616<br>4.1828878<br>4.9637927<br>4.4562831<br>0.0573728   | 1.863E-06<br>1.272E-06<br>1.41E-06<br>1.43E-06<br>1.43E-06<br>1.628E-06<br>1.528E-06<br>1.695E-06<br>1.401E-06<br>1.440E-06<br>9.333E-07<br>1.341E-06<br>9.333E-07<br>1.341E-06<br>8.711E-09  
  | 0.0030261<br>0.0020666<br>0.0022915<br>0.002295<br>0.0021953<br>0.0021954<br>0.002544<br>0.002644<br>0.002644<br>0.002744<br>0.002757<br>0.0023008<br>0.0015163<br>0.0015163<br>0.0021782<br>0.0022849<br>0.0022849<br>0.0022849<br>0.0022849<br>0.0022849<br>0.0022849   | 2.3556842<br>1.8789238<br>2.0977776<br>2.0052501<br>2.1283629<br>2.9552749<br>2.90207<br>2.8217082<br>2.7301698<br>3.0191874<br>2.4772701<br>2.5033689<br>2.5607825<br>2.5607825<br>3.9311649<br>4.688311<br>4.1688413<br>0.4168413<br>0.41684213  | 1.762E-06<br>1.204E-06<br>1.345E-06<br>1.345E-06<br>1.263E-06<br>1.263E-06<br>1.549E-06<br>1.458E-06<br>1.322E-06<br>1.332E-06<br>1.337E-06<br>1.337E-06<br>1.548E-06<br>1.501E-06<br>1.501E-06<br>1.501E-06<br>1.336E-06<br>7.03E-09  | 0.0028636<br>0.0019569<br>0.0021848<br>0.0020885<br>0.0022167<br>0.0020519<br>0.0025171<br>0.0019592<br>0.00256204<br>0.0021501<br>0.0021707<br>0.0022225<br>0.0024265<br>0.002429<br>0.002471<br>0.0024719<br>0.0024719<br>0.0024719  | 2.2440968<br>1.7795138<br>1.9915216<br>1.8961342<br>2.0102391<br>2.7864095<br>2.7220732<br>2.6462899<br>2.5622851<br>2.8364013<br>2.3321589<br>2.3726056<br>2.4375101<br>2.6016999<br>3.7301495<br>4.4767836<br>3.9755324<br>0.0410785  
   | 1.6782-06<br>1.141E-06<br>1.277E-06<br>1.235E-06<br>1.245E-06<br>1.454E-06<br>1.3454E-06<br>1.345E-06<br>1.246E-06<br>1.246E-06<br>1.246E-06<br>1.302E-06<br>8.339E-07<br>1.196E-06<br>1.435E-06<br>1.245E-06<br>1.245E-06<br>1.245E-06   | 0.0027267<br>0.0018534<br>0.0020742<br>0.0019748<br>0.0020937<br>0.0019748<br>0.0023625<br>0.0013374<br>0.002238<br>0.002238<br>0.0022328<br>0.002415<br>0.002415<br>0.002415<br>0.002415<br>0.002155<br>0.0013548<br>0.0013425<br>0.0013425<br>0.000215<br>0.000215<br>0.000213<br>0.000215<br>0.000215<br>0.000215<br>0.000215<br>0.000215<br>0.000215<br>0.000215<br>0.000215<br>0.000215<br>0.000215<br>0.000215<br>0.000215<br>0.000215<br>0.000215<br>0.000215<br>0.000215<br>0.000215<br>0.000215<br>0.000215<br>0.000215<br>0.000215<br>0.000215<br>0.000215<br>0.000215<br>0.000215<br>0.000215<br>0.000215<br>0.000215<br>0.000215<br>0.000215<br>0.000215<br>0.000215<br>0.000215<br>0.000215<br>0.000215<br>0.000215<br>0.000215<br>0.000215<br>0.000215<br>0.000215<br>0.000215<br>0.000215<br>0.000215<br>0.000215<br>0.000215<br>0.000215<br>0.000215<br>0.000215<br>0.000215<br>0.000215<br>0.000215<br>0.000215<br>0.000215<br>0.000215<br>0.000215<br>0.000215<br>0.000215<br>0.000215<br>0.000215<br>0.000215<br>0.000215<br>0.000215<br>0.000215<br>0.000215<br>0.000215<br>0.000215<br>0.000215<br>0.000215<br>0.000215<br>0.000215<br>0.000215<br>0.000215<br>0.000215<br>0.000215<br>0.000215<br>0.000215<br>0.000215<br>0.000215<br>0.000215<br>0.000215<br>0.000215<br>0.000215<br>0.000215<br>0.000215<br>0.000215<br>0.000215<br>0.000215<br>0.000215<br>0.000215<br>0.000215<br>0.000215<br>0.000205<br>0.000215<br>0.000205<br>0.000205<br>0.000205<br>0.000205<br>0.000205<br>0.000205<br>0.000205<br>0.000205<br>0.000205<br>0.000205<br>0.000205<br>0.0000000000  | 2.623364<br>2.11248<br>2.3459273<br>2.2439219<br>2.3971946<br>3.2614464<br>3.1051925<br>3.0057113<br>3.3257455<br>2.7552623<br>2.7552623<br>2.7582623<br>2.7818703<br>2.8396521<br>3.0470013<br>1.3621382<br>5.1979947<br>4.6427163<br>0.0606669  
  | 1.962E-06<br>1.354E-06<br>1.504E-06<br>1.438E-06<br>1.337E-06<br>1.394E-06<br>1.706E-06<br>1.377E-06<br>1.472E-06<br>1.472E-06<br>1.486E-06<br>9.766E-07<br>1.398E-06<br>1.666E-06<br>9.1488E-06<br>9.186-09   | 0.0031876<br>0.0022001<br>0.0024337<br>0.0024337<br>0.002437<br>0.002265<br>0.0027011<br>0.002156<br>0.0026067<br>0.0028665<br>0.0023913<br>0.0024144<br>0.0021867<br>0.0024677<br>0.0022716<br>0.0022068<br>0.0022168   | 2.3644564<br>1.8937445<br>2.1225900<br>2.0177733<br>2.1529112<br>2.928047<br>2.8544498<br>2.7808064<br>2.6943024<br>2.9747351<br>2.4523139<br>2.4915395<br>2.4915395<br>2.5838663<br>2.724646<br>3.8926302<br>4.6916603<br>4.1449386<br>0.043318   | 1.768:-00         0.002           1.214E-06         0.002           1.361E-06         0.002           1.361E-06         0.002           1.38E-06         0.002           1.38E-06         0.002           1.251E-06         0.002           1.251E-06         0.002           1.38E-06         0.002           1.38E-06         0.002           1.38E-06         0.002           1.31E-06         0.002           1.331E-06         0.002           1.504E-06         0.002           1.504E-06         0.002           5.577E-09         1.202   
  | 873         0.04           9723         0.03           2107         0.04           1015         0.04           2220         0.04           2422         0.04           033         0.05           3774         0.05           3384         0.04           5818         0.05           3384         0.04           2204         0.04           2107         0.04           2204         0.04           2184         0.05           3384         0.044           200         0.044           2180         0.04           2204         0.044           2004         0.044           2004         0.044           2004         0.044           2004         0.044           2004         0.044           2004         0.044           2005         0.002   | 147043         3.           383832         2           408499         2.1           105571         2.           133125         2.           133125         2.           133125         2.           133125         2.           133125         2.           133125         2.           133125         2.           133125         2.           143125         2.           1538045         2.           157074         2.4           165677         2.4           165677         2.4           165677         2.4           165767         2.4           16276         2.4           18381         1.6           776164         2.4           81928         2.6           539938         8.1   
   
  | 5.18E-08         2           2.46E-08         2           2.6E-08         4           2.776E-08         4           2.990-08         3           8.74E-08         -           197E-08         3           651E-08         4           442E-08         3           939E-08         4           551E-08         4           651E-08         4           651E-08         4           651E-08         4           661E-08         2           488E-08         4           526E-08         4           198E-00         1           198E-10         1  | 3.998E-05<br>4.254E-05<br>4.224E-05<br>4.224E-05<br>4.511E-05<br>3.736E-05<br>4.67E-05<br>4.67E-05<br>4.307E-05<br>4.307E-05<br>4.307E-05<br>4.098E-05<br>4.22E-05<br>1.042E-05<br>1.042E-05<br>1.501E-05<br>1.501E-05<br>1.266E-05  | 0.044322<br>0.0443322<br>0.044017<br>0.0470156<br>0.0584313<br>0.0558211<br>0.0558211<br>0.0558211<br>0.0558271<br>0.0637743<br>1.0527943<br>1.0632958<br>1.0843242<br>1.0938934<br>1.0938934  | 2.842E-08 4<br>2.822E-08 4<br>2.822E-08 4<br>2.497E-08 4<br>2.497E-08 4<br>2.3122E-08 4<br>2.381E-08 4<br>2.837E-08 4<br>2.851E-08 4<br>2.851E-08 4<br>2.82E-08 4<br>2.703E-08 4<br>2.703E-08 4<br>2.809E-08 4<br>2.809E-08 4<br>2.835E-08 4<br>2.835E-08 4<br>2.835E-08 4  
  | 1.584E-05<br>1.897E-05<br>1.057E-05<br>3.072E-05<br>3.876E-05<br>3.876E-05<br>3.31E-05<br>1.307E-05<br>4.07E-05<br>4.582E-05<br>2.932E-05<br>1.391E-05<br>1.891E-05<br>1.604E-06   |
| Sheung Towen B/T and Car Park<br>Sheung Towen B/T and Car Park<br>Village Access Road<br>Village Access Road  | RD_29<br>RD_29<br>RD_29<br>RD_29<br>RD_29<br>RD_29<br>RD_29<br>RD_29<br>RD_29<br>RD_29<br>RD_29<br>RD_29<br>RD_29<br>RD_29<br>RD_29<br>RD_29<br>RD_29<br>RD_29<br>RD_29<br>RD_29<br>RD_29<br>RD_20<br>RD_30_31<br>RD_30_31  | 0900-1000<br>1000-1100<br>1100-1200<br>1200-1300<br>1300-1400<br>1500-1600<br>1500-1600<br>1500-1700<br>1800-1900<br>1900-2000<br>2000-2100<br>2100-2200<br>2200-0000<br>0000-0100<br>0100-0200   | 0.1281945<br>0.1457065<br>0.1360751<br>0.1507253<br>0.1457266<br>0.1482353<br>0.1452588<br>0.1322075<br>0.1302075<br>0.1302075<br>0.1322075<br>0.1322079<br>0.1352297<br>0.1352397<br>0.1352397<br>0.1352397<br>0.1352397<br>0.1352397<br>0.13624322<br>0.030941<br>0<br>0   | 9.34-00 ( 8.2186-08 ( 9.34E-08 ( 9.34E-08 ( 1.03E-07 ( 6.655E-08 ( 7.79E-08 ( 6.335E-08 ( 7.79E-08 ( 6.335E-08 ( 7.77E-08 ( 6.933E-08 ( 7.71E-08 ( 7.05E-08 ( 7.16E-08 ( 1.335E-08 ( 7.50E-08 ( 1.335E-08 ( 5.975E-08 ( 0 ( 0 ( 0 ( 0 ( 0 ( 0 ( 0 ( 0 ( 0 ( 0   
  | 0.0001615 (<br>0.0001335 (<br>0.0001335 (<br>0.0001518 (<br>0.0001617 (<br>0.000167 (<br>0.000167 (<br>0.0001627 (<br>0.0001267 (<br>0.0001263 (<br>0.0001324 (<br>0.0001324 (<br>0.0001324 (<br>0.0001324 (<br>0.000132 (<br>0.000122 (<br>0.00012 (<br>0.000122 (<br>0.000122 (<br>0.000122 (<br>0.000122 (<br>0.000122 (   | 0.1261103<br>0.1210641<br>0.1383867<br>0.1291377<br>0.1291377<br>0.1491352<br>0.1491352<br>0.1491352<br>0.1491352<br>0.1491352<br>0.1491352<br>0.1491353<br>0.1461086<br>0.1267454<br>0.1264957<br>0.1264495<br>0.1262957<br>0.2231399<br>0.1762915<br>0.0025024<br>0  | 9.431E-08<br>7.761E-08<br>8.884E-08<br>8.8278E-08<br>9.754E-08<br>6.373E-08<br>7.445E-08<br>7.445E-08<br>7.465E-08<br>6.656E-08<br>6.771E-08<br>6.656E-08<br>6.737E-08<br>5.447E-08<br>7.152E-08<br>5.65E-08<br>3.799E-10<br>0<br>0  | 0.0001532<br>0.0001532<br>0.0001535<br>0.0001343<br>0.0001443<br>0.0001385<br>0.0001035<br>0.0001214<br>9.882E-05<br>0.0001212<br>0.0001212<br>0.0001212<br>0.0001162<br>9.885E-05<br>0.0001162<br>9.18E-05<br>6.944E-07<br>0<br>0  | 0.1203596<br>0.1142306<br>0.1310694<br>0.1216391<br>0.1426721<br>0.1426721<br>0.1323766<br>0.1323766<br>0.1323766<br>0.13233760<br>0.120355<br>0.1203552<br>0.1203552<br>0.1229461<br>0.1229461<br>0.1229461<br>0.1229462<br>0.01294766<br>0.01694062<br>0.01694062<br>0.01694062<br>0.01694062  
   | 9.001E-08<br>7.322E-08<br>8.402E-08<br>7.797E-08<br>9.146E-08<br>6.053E-08<br>7.052E-08<br>7.052E-08<br>7.052E-08<br>6.419E-08<br>6.435E-08<br>5.208E-08<br>5.208E-08<br>5.208E-08<br>5.208E-08<br>5.208E-08<br>5.208E-08<br>5.342E-08<br>3.44E-10<br>0   | 0.0001462<br>0.0001365<br>0.0001265<br>0.0001265<br>0.0001486<br>9.834E-05<br>0.0001446<br>9.34E-05<br>0.0001146<br>0.0001043<br>0.0001043<br>0.0001043<br>0.0001049<br>6.402E-05<br>8.461E-05<br>0.0001119<br>8.822E-05<br>6.214E-07<br>0<br>0   | 2.4904488<br>1.9842855<br>2.2002208<br>2.1078468<br>2.2364653<br>3.1057198<br>3.0468346<br>2.9569573<br>2.8602462<br>3.1731567<br>2.6220547<br>2.6509615<br>2.7073267<br>2.9117616<br>4.1828878<br>4.9637927<br>4.4562831<br>0.0573728<br>0<br>0  
   | 1.863E-06<br>1.272E-06<br>1.272E-06<br>1.41E-06<br>1.43E-06<br>1.43E-06<br>1.25E-06<br>1.628E-06<br>1.528E-06<br>1.639E-06<br>1.401E-06<br>1.440E-06<br>9.333E-07<br>1.341E-06<br>9.333E-07<br>1.341E-06<br>8.711E-09<br>0<br>0  | 0.0030261<br>0.0022056<br>0.0022953<br>0.0022953<br>0.0023293<br>0.002544<br>0.002544<br>0.002541<br>0.0022454<br>0.002757<br>0.0023008<br>0.0021782<br>0.0021782<br>0.002249<br>0.002249<br>0.002249<br>0.0022549<br>0.0022549<br>0.0022549<br>0.002255<br>0<br>0.002255<br>0<br>0   | 2.3566842<br>1.8789238<br>2.0977776<br>2.0052501<br>2.1283629<br>2.9552749<br>2.90207<br>2.8217082<br>2.7301688<br>3.0191874<br>2.4772701<br>2.5033689<br>2.5607825<br>2.5607825<br>2.39311649<br>4.688413<br>0.41688413<br>0.41688413<br>0.41688413<br>0.41688413<br>0.41688413<br>0.41688413<br>0.41688413<br>0.41688413<br>0.41688413<br>0.41688413<br>0.41688413<br>0.41688413<br>0.41688413<br>0.41688413<br>0.41688413<br>0.41688413<br>0.41688413<br>0.41688413<br>0.41688413<br>0.41688413<br>0.41688413<br>0.41688413<br>0.41688413<br>0.41688413<br>0.41688413<br>0.41688413<br>0.41688413<br>0.41688413<br>0.41688413<br>0.41688413<br>0.41688413<br>0.41688413<br>0.41688413<br>0.41688413<br>0.41688413<br>0.41688413<br>0.41688413<br>0.41688413<br>0.41688413<br>0.41688413<br>0.41688413<br>0.41688413<br>0.41688413<br>0.41688413<br>0.41688413<br>0.41688413<br>0.41688413<br>0.41688413<br>0.41688413<br>0.41688413<br>0.41688413<br>0.41688413<br>0.41688413<br>0.41688413<br>0.41688413<br>0.41688413<br>0.41688413<br>0.41688413<br>0.41688413<br>0.41688413<br>0.41688413<br>0.41688413<br>0.41688413<br>0.41688413<br>0.41688413<br>0.41688413<br>0.41688413<br>0.41688413<br>0.41688413<br>0.41688413<br>0.41688413<br>0.41688413<br>0.41688413<br>0.41688413<br>0.41688413<br>0.41688413<br>0.41688413<br>0.41688413<br>0.41688413<br>0.41688413<br>0.41688413<br>0.41688413<br>0.41688413<br>0.41688413<br>0.41688413<br>0.41688413<br>0.41688413<br>0.41688413<br>0.41688413<br>0.41688413<br>0.41688413<br>0.41688413<br>0.41688413<br>0.41688413<br>0.41688413<br>0.41688413<br>0.41688413<br>0.41688413<br>0.41688413<br>0.41688413<br>0.41688413<br>0.41688413<br>0.41688413<br>0.41688413<br>0.41688413<br>0.41688413<br>0.41688413<br>0.41688413<br>0.41688413<br>0.41688413<br>0.41688413<br>0.41688413<br>0.41688413<br>0.41688413<br>0.41688413<br>0.41688413<br>0.41688413<br>0.41688413<br>0.41688413<br>0.41688413<br>0.41688413<br>0.41688413<br>0.41688413<br>0.41688413<br>0.41688413<br>0.41688413<br>0.41688413<br>0.41688413<br>0.41688413<br>0.41688413<br>0.41688413<br>0.41688413<br>0.41688413<br>0.41688413<br>0.41688413<br>0.41688413<br>0.41688413<br>0.41688413<br>0.41688413<br>0.41688413<br>0.41688413<br>0.41688413<br>0.41688413<br>0.41688413<br>0.4168845<br>0.4168845<br>0.4168845<br>0.4168845<br>0.4168845<br>0.4168845<br>0.4168845<br>0.4168845<br>0.4168845<br>0.4168845<br>0.4168845<br>0.4168845<br>0.4168845<br>0.4168845<br>0.4168845<br>0.4168845<br>0.4168845<br>0.4168845<br>0.4168845<br>0.4168855<br>0.41688556<br>0.41688556565656565656565565656565656565656  | 1.762E-06<br>1.204F-06<br>1.345E-06<br>1.345E-06<br>1.263E-06<br>1.263E-06<br>1.263E-06<br>1.263E-06<br>1.323E-06<br>1.323E-06<br>1.323E-06<br>1.323E-06<br>1.501E-06<br>1.501E-06<br>1.501E-06<br>1.536E-06<br>7.03E-09<br>0  |
0.002636<br>0.0019569<br>0.0021848<br>0.0020885<br>0.0022167<br>0.0020519<br>0.0025171<br>0.0012592<br>0.0026204<br>0.00221501<br>0.00221501<br>0.00221501<br>0.0022471<br>0.0022471<br>0.0022471<br>0.0022471<br>0.0022471<br>0.0022471<br>0.0022471<br>0.0022471<br>0.0022471<br>0.0022471<br>0.0022471<br>0.0022471<br>0.0022471<br>0.0022471<br>0.0022471<br>0.0022471<br>0.0022471<br>0.0022471<br>0.0022471<br>0.0022471<br>0.0022471<br>0.0022471<br>0.0022471<br>0.0022471<br>0.0022471<br>0.0022471<br>0.0022471<br>0.0022471<br>0.0022471<br>0.0022471<br>0.0022471<br>0.0022471<br>0.0022471<br>0.0022471<br>0.0022471<br>0.0022471<br>0.0022471<br>0.0022471<br>0.0022471<br>0.0022471<br>0.0022471<br>0.0022471<br>0.0022471<br>0.0022471<br>0.0022471<br>0.0022471<br>0.0022471<br>0.0022471<br>0.0022471<br>0.0022471<br>0.0022471<br>0.0022471<br>0.0022471<br>0.0022471<br>0.0022471<br>0.0022471<br>0.0022471<br>0.0022471<br>0.0022471<br>0.0022471<br>0.0022471<br>0.0022471<br>0.0022471<br>0.0022471<br>0.0022471<br>0.0022471<br>0.0022471<br>0.0022471<br>0.0022471<br>0.0022471<br>0.0022471<br>0.0022471<br>0.0022471<br>0.0022471<br>0.0022471<br>0.0022471<br>0.0022471<br>0.0022471<br>0.002471<br>0.002471<br>0.002471<br>0.002471<br>0.002471<br>0.002471<br>0.002471<br>0.002471<br>0.002471<br>0.002471<br>0.002471<br>0.002471<br>0.002471<br>0.002471<br>0.002471<br>0.002471<br>0.002471<br>0.002471<br>0.002471<br>0.002471<br>0.002471<br>0.002471<br>0.002471<br>0.002471<br>0.002471<br>0.002471<br>0.002471<br>0.002471<br>0.002471<br>0.002471<br>0.002471<br>0.002471<br>0.002471<br>0.002471<br>0.002471<br>0.002471<br>0.002471<br>0.002471<br>0.002471<br>0.002471<br>0.002471<br>0.002471<br>0.002471<br>0.002471<br>0.002471<br>0.002471<br>0.002471<br>0.002471<br>0.002471<br>0.002471<br>0.002471<br>0.002471<br>0.002471<br>0.002471<br>0.002471<br>0.002471<br>0.002471<br>0.002471<br>0.002471<br>0.002471<br>0.002471<br>0.002471<br>0.002471<br>0.002471<br>0.002471<br>0.002471<br>0.002471<br>0.002471<br>0.002471<br>0.002471<br>0.002471<br>0.002471<br>0.002471<br>0.002471<br>0.002471<br>0.002471<br>0.002471<br>0.002471<br>0.002471<br>0.002471<br>0.002471<br>0.002471<br>0.002471<br>0.002471<br>0.002471<br>0.002471<br>0.002471<br>0.002471<br>0.002471<br>0.002471<br>0.002471<br>0.002471<br>0.002471<br>0.002471<br>0.002471<br>0.002471<br>0.002471<br>0.002471<br>0.002471<br>0.002471<br>0.002471<br>0.002471<br>0.002471<br>0.002471<br>0.002471<br>0.002471<br>0.002471<br>0.002471<br>0.002471<br>0.002471<br>0.002471<br>0.002471<br>0.002471<br>0.002471<br>0.002471<br>0.002471<br>0.002471<br>0.002471<br>0.002471<br>0.002471<br>0.002471<br>0.0024710000000000000000000000000000000000   | 2.2440968<br>1.7795138<br>1.9915216<br>1.8961342<br>2.7864095<br>2.726732<br>2.6462899<br>2.5622851<br>2.3321589<br>2.3726056<br>2.4375101<br>2.6016999<br>3.7301495<br>4.4767836<br>3.9755324<br>0.0410785<br>0<br>0   | 1.678E-06<br>1.141E-06<br>1.277E-06<br>1.215E-06<br>1.215E-06<br>1.454E-06<br>1.369E-06<br>1.369E-06<br>1.246E-06<br>1.246E-06<br>1.246E-06<br>1.302E-06<br>3.032E-06<br>1.302E-06<br>1.302E-06<br>6.237E-09<br>0<br>0  |
0.0027267<br>0.0018534<br>0.0020742<br>0.0019748<br>0.00209742<br>0.0019347<br>0.0020937<br>0.0019347<br>0.0023625<br>0.0022418<br>0.0022421<br>0.0022417<br>0.0020542<br>0.0022415<br>0.00202415<br>0.0020542<br>0.002155<br>0.002155<br>0.001155<br>0.001155<br>0.001155<br>0.001155<br>0.001155<br>0.001155<br>0.001155<br>0.001155<br>0.001155<br>0.001155<br>0.001155<br>0.001155<br>0.001155<br>0.001155<br>0.001155<br>0.001155<br>0.001155<br>0.001155<br>0.001155<br>0.001155<br>0.001155<br>0.001155<br>0.001155<br>0.001155<br>0.001155<br>0.001155<br>0.001155<br>0.001155<br>0.001155<br>0.001155<br>0.001155<br>0.001155<br>0.001155<br>0.001155<br>0.001155<br>0.001155<br>0.001155<br>0.001155<br>0.001155<br>0.001155<br>0.001155<br>0.001155<br>0.001155<br>0.001155<br>0.001155<br>0.001155<br>0.001155<br>0.001155<br>0.001155<br>0.001155<br>0.001155<br>0.001155<br>0.001155<br>0.001155<br>0.001155<br>0.001155<br>0.001155<br>0.001155<br>0.001155<br>0.001155<br>0.001155<br>0.001155<br>0.001155<br>0.001155<br>0.001155<br>0.001155<br>0.001155<br>0.001155<br>0.001155<br>0.001155<br>0.001155<br>0.001155<br>0.001155<br>0.001155<br>0.001155<br>0.001155<br>0.001155<br>0.001155<br>0.001155<br>0.001155<br>0.001155<br>0.001155<br>0.001155<br>0.001155<br>0.001155<br>0.001155<br>0.001155<br>0.001155<br>0.001155<br>0.001155<br>0.001155<br>0.001155<br>0.001155<br>0.001155<br>0.001155<br>0.001155<br>0.001155<br>0.001155<br>0.001155<br>0.001155<br>0.001155<br>0.001155<br>0.001155<br>0.001155<br>0.001155<br>0.001155<br>0.001155<br>0.001155<br>0.001155<br>0.001155<br>0.001155<br>0.001155<br>0.001155<br>0.001155<br>0.001155<br>0.001155<br>0.001155<br>0.001155<br>0.00155<br>0.00155<br>0.00155<br>0.00155<br>0.00155<br>0.00155<br>0.00155<br>0.00155<br>0.00155<br>0.00155<br>0.00155<br>0.00155<br>0.00155<br>0.00155<br>0.00155<br>0.00155<br>0.00155<br>0.00155<br>0.00155<br>0.00155<br>0.00155<br>0.00155<br>0.00155<br>0.00155<br>0.00155<br>0.00155<br>0.00155<br>0.00155<br>0.00155<br>0.00155<br>0.00155<br>0.00155<br>0.00155<br>0.00155<br>0.00155<br>0.00155<br>0.00155<br>0.00155<br>0.00155<br>0.00155<br>0.00155<br>0.00155<br>0.00155<br>0.00155<br>0.00155<br>0.00155<br>0.00155<br>0.00155<br>0.00155<br>0.00155<br>0.00155<br>0.00155<br>0.00155<br>0.00155<br>0.00155<br>0.00155<br>0.00155<br>0.00155<br>0.00155<br>0.00155<br>0.00155<br>0.00155<br>0.00155<br>0.00155<br>0.00155<br>0.00155<br>0.00155<br>0.00155<br>0.00155<br>0.00155<br>0.00155<br>0.00155<br>0.00155<br>0.00155<br>00  | 2.623364<br>2.11248<br>2.3459273<br>2.2439219<br>2.3971946<br>3.2561464<br>3.1051925<br>3.0057113<br>3.3257455<br>2.7552623<br>2.7818703<br>2.83965211<br>3.0470013<br>3.3621382<br>3.1979947<br>1.6427163<br>0.0604669<br>0<br>0  | 1.962E-06<br>1.354E-06<br>1.504E-06<br>1.438E-06<br>1.337E-06<br>1.337E-06<br>1.327E-06<br>1.327E-06<br>1.427E-06<br>1.427E-06<br>1.472E-06<br>1.472E-06<br>1.472E-06<br>1.517E-06<br>9.766E-07<br>1.398E-06<br>1.666E-06<br>1.488E-06<br>9.18E-09<br>0<br>0   |
0.0031876<br>0.0022001<br>0.0024337<br>0.0024337<br>0.0024337<br>0.002265<br>0.0022645<br>0.0027111<br>0.002156<br>0.0028665<br>0.0023913<br>0.0024646<br>0.0015867<br>0.0022716<br>0.0015867<br>0.0022716<br>0.0022168<br>0.0022168<br>0.0022168<br>0.002216<br>0.002216<br>0.002216<br>0.002216<br>0.002216<br>0.002216<br>0.002216<br>0.002216<br>0.002216<br>0.002216<br>0.002216<br>0.002216<br>0.002216<br>0.002216<br>0.002216<br>0.002216<br>0.002216<br>0.002216<br>0.002216<br>0.002216<br>0.002216<br>0.002216<br>0.002216<br>0.002216<br>0.002216<br>0.002216<br>0.002216<br>0.002216<br>0.002216<br>0.002216<br>0.002216<br>0.002216<br>0.002216<br>0.002216<br>0.002216<br>0.002216<br>0.002216<br>0.002216<br>0.002216<br>0.00226<br>0.002216<br>0.00226<br>0.00226<br>0.00226<br>0.00226<br>0.00226<br>0.00226<br>0.00226<br>0.00226<br>0.00226<br>0.00226<br>0.00226<br>0.00226<br>0.00226<br>0.00226<br>0.00226<br>0.00226<br>0.00226<br>0.00226<br>0.00226<br>0.00226<br>0.00226<br>0.00226<br>0.00226<br>0.00226<br>0.00226<br>0.00226<br>0.00226<br>0.00226<br>0.00226<br>0.00226<br>0.00226<br>0.00226<br>0.00226<br>0.00226<br>0.00226<br>0.00226<br>0.00226<br>0.00226<br>0.00226<br>0.00226<br>0.00226<br>0.0025<br>0.00226<br>0.00256<br>0.00256<br>0.00256<br>0.00256<br>0.00256<br>0.00256<br>0.00256<br>0.00256<br>0.00256<br>0.00256<br>0.00256<br>0.00256<br>0.00256<br>0.00256<br>0.00256<br>0.000256<br>0.00256<br>0.00256<br>0.00256<br>0.00256<br>0.00256<br>0.00256<br>0.00256<br>0.00256<br>0.00256<br>0.00256<br>0.00256<br>0.00256<br>0.00256<br>0.00256<br>0.00256<br>0.00256<br>0.00256<br>0.00256<br>0.00256<br>0.00256<br>0.00256<br>0.00256<br>0.00256<br>0.00256<br>0.00256<br>0.00256<br>0.00256<br>0.00256<br>0.00256<br>0.00256<br>0.00256<br>0.00256<br>0.00256<br>0.00256<br>0.00256<br>0.00256<br>0.00256<br>0.00256<br>0.00256<br>0.00256<br>0.00256<br>0.00256<br>0.00256<br>0.00256<br>0.00256<br>0.00256<br>0.00256<br>0.00256<br>0.00256<br>0.00256<br>0.00256<br>0.00256<br>0.00256<br>0.00256<br>0.00256<br>0.00256<br>0.00256<br>0.00256<br>0.00256<br>0.00256<br>0.00256<br>0.00256<br>0.00256<br>0.00256<br>0.00256<br>0.00256<br>0.00256<br>0.00256<br>0.00256<br>0.00256<br>0.00256<br>0.00256<br>0.00256<br>0.00256<br>0.00256<br>0.00256<br>0.00256<br>0.00256<br>0.00256<br>0.00256<br>0.00256<br>0.00256<br>0.00256<br>0.00256<br>0.00256<br>0.00256<br>0.00256<br>0.00256<br>0.00256<br>0.00256<br>0.00256<br>0.00256<br>0.00256<br>0.00256<br>0.00256<br>0.00256<br>0.00256<br>0.00256<br>0.00256<br>0.00256<br>0.00256<br>0.00256<br>0.00256<br>0.00256<br>0.00256<br>0.00000000000000000000000000000000000  | 2.3644564<br>1.8937445<br>2.1225900<br>2.0177733<br>2.1529112<br>2.928047<br>2.8544498<br>2.7808064<br>2.6943024<br>2.9747351<br>2.45231395<br>2.4915395<br>2.724646<br>3.8926302<br>4.6916603<br>4.1449386<br>0.043318<br>0<br>0  | 1.768:-06 0.002<br>1.244E-06 0.001<br>1.361E-06 0.002<br>1.238E-06 0.002<br>1.38E-06 0.002<br>1.38E-06 0.002<br>1.38E-06 0.002<br>1.38E-06 0.002<br>1.38E-06 0.002<br>1.31E-06 0.002<br>1.31E-06 0.002<br>1.31E-06 0.002<br>1.31E-06 0.002<br>1.348E-06 0.002<br>1.248E-06 0.002   | 8873         0.04           9723         0.03           2107         0.04           1015         0.04           2422         0.04           033         0.05           3774         0.05           3384         0.04           5818         0.05           3384         0.04           2204         0.04           2202         0.04           4189         0.05           2717         0.07           44189         0.05           1585         0.00           5432         0.00   
  | 147043         3:           383832         2           383832         2           483832         2           483832         2           480592         2:           313125         2:           338011         2:           338011         2:           338012         2:           338015         2:           338045         2:           314016         2:           15511         3:           165511         3:           165677         2:           186276         2:           186276         2:           186276         2:           186276         2:           186276         2:           18828         1:           6         75393           0         0  
   | 5.18E-08         2           2.46E-08         2           2.6E-08         4           2.776E-08         4           2.990-08         3           8.74E-08         -           197E-08         3           651E-08         4           442E-08         3           598E-08         4           598E-08         4           598E-08         4           661E-08         2           488E-08         4           652E-08         4           652E-08         4           661E-08         2           488E-00         4           652E-08         4           0         0  | 3.998E-05<br>4.254E-05<br>4.224E-05<br>4.224E-05<br>4.511E-05<br>3.736E-05<br>4.67E-05<br>3.569E-05<br>4.307E-05<br>4.307E-05<br>4.307E-05<br>4.307E-05<br>4.22E-05<br>1.042E-05<br>1.042E-05<br>1.501E-05<br>1.501E-05<br>1.266E-05<br>0<br>0   
   | 0.044322<br>0.044322<br>0.044017<br>0.0470156<br>0.0584373<br>0.0584373<br>0.0584373<br>0.0558211<br>0.0538871<br>0.051296<br>0.046205<br>0.046205<br>0.046202<br>0.0507781<br>0.052794<br>0.052794<br>0.0507794<br>0<br>0   | 2.872-03 4<br>2.822E-08 4<br>3.014E-08 4<br>2.822E-08 4<br>3.124E-08 4<br>3.122E-08 4<br>3.122E-08 4<br>3.122E-08 4<br>2.879E-08 4<br>2.879E-08 4<br>2.879E-08 4<br>2.828E-08 4<br>3.009E-08 4<br>3.009E-08 4<br>3.009E-08 4<br>3.853E-08 4<br>0 0   | 1.584E-05<br>1.897E-05<br>1.057E-05<br>5.072E-05<br>1.072E-05<br>1.877E-05<br>1.307E-05<br>1.307E-05<br>1.582E-05<br>1.582E-05<br>1.391E-05<br>1.899E-05<br>1.636E-05<br>1.604E-06<br>0<br>0   |
| Sheung Tsuen B/T and Car Park<br>Sheung Tsuen B/T and Car Park<br>Ullage Access Road<br>Village Access Road<br>Village Access Road   | RD_29<br>RD_29<br>RD_29<br>RD_29<br>RD_29<br>RD_29<br>RD_29<br>RD_29<br>RD_29<br>RD_29<br>RD_29<br>RD_29<br>RD_29<br>RD_29<br>RD_29<br>RD_29<br>RD_29<br>RD_30_31<br>RD_30_31<br>RD_30_31   | 0900-1000<br>1000-1100<br>1100-1200<br>1300-1400<br>1400-1500<br>1500-1600<br>1500-1600<br>1500-2000<br>2000-2100<br>2000-2100<br>2200-2300<br>2200-2300<br>2200-2300<br>0000-9100<br>0100-0200<br>0200-0300  | 0.1281945<br>0.1360751<br>0.1360751<br>0.1557266<br>0.145988<br>0.145982<br>0.1323245<br>0.1323254<br>0.1323254<br>0.1323254<br>0.1323254<br>0.1323254<br>0.1323254<br>0.1323207<br>0.1792504<br>0.1323254<br>0.1323254<br>0.1323254<br>0.1323254<br>0.1323254<br>0.1323254<br>0.1323254<br>0.1323254<br>0.1323254<br>0.1323254<br>0.1323254<br>0.1323254<br>0.1323254<br>0.1323254<br>0.1323254<br>0.1323254<br>0.1323254<br>0.1323254<br>0.1323254<br>0.1323254<br>0.1323254<br>0.1323254<br>0.1323254<br>0.1323254<br>0.1323254<br>0.1323254<br>0.1323254<br>0.1323254<br>0.1323254<br>0.1323254<br>0.1323254<br>0.1323254<br>0.1323254<br>0.1323254<br>0.1323254<br>0.1323254<br>0.1323254<br>0.1323254<br>0.1323254<br>0.1323254<br>0.1323254<br>0.1323254<br>0.1323254<br>0.1323254<br>0.1323254<br>0.1323254<br>0.1323254<br>0.1323254<br>0.1323254<br>0.1323254<br>0.1323254<br>0.1323254<br>0.1323254<br>0.1323254<br>0.1323254<br>0.1323254<br>0.1323254<br>0.1323254<br>0.1323254<br>0.1323254<br>0.1323254<br>0.1323254<br>0.1323254<br>0.1323254<br>0.1323254<br>0.1323254<br>0.1323254<br>0.1323254<br>0.1323254<br>0.1323254<br>0.1323254<br>0.1323254<br>0.1323254<br>0.1323254<br>0.1323254<br>0.1323254<br>0.1323254<br>0.1323254<br>0.1323254<br>0.1323254<br>0.1323254<br>0.1323254<br>0.1323254<br>0.1323254<br>0.1323254<br>0.1323254<br>0.1325254<br>0.1325254<br>0.1325254<br>0.1325254<br>0.1325254<br>0.1325254<br>0.1325254<br>0.1325254<br>0.1325254<br>0.1325254<br>0.1325254<br>0.1325254<br>0.1325254<br>0.1325254<br>0.1325254<br>0.132555555555555555555555555555555555555  | 3-34-00         1           9-34-00         1           9.34-08         1           9.34-08         1           9.34-08         1           1.03-07         1           6.555-08         1           7.799-08         1           6.335-08         1           7.791-08         1           8.151-08         1           7.161-08         1           4.335-08         1           7.506E-08         1           7.506E-08         1           0         0           0.306E-08         0   
   | 0.0001615 (<br>0.0001335 (<br>0.0001318 (<br>0.0001518 (<br>0.0001617 (<br>0.000167 (<br>0.000167 (<br>0.0001627 (<br>0.0001263 (<br>0.0001263 (<br>0.0001263 (<br>0.0001148 (<br>0.0001148 (<br>0.000122 (<br>0.000126 (<br>0.0000126 (  | 0.1261103<br>0.1210641<br>0.1385867<br>0.1291377<br>0.1521578<br>0.1491352<br>0.1399004<br>0.1423175<br>0.1395003<br>0.1461086<br>0.1267454<br>0.1267454<br>0.1267454<br>0.126957<br>0.1284496<br>0.169957<br>0.2231399<br>0.1762915<br>0.0025024<br>0<br>0<br>0<br>0.480112   | 9.431E-08<br>7.761E-08<br>8.884E-08<br>8.278E-08<br>9.754E-08<br>6.373E-08<br>6.632E-08<br>7.46E-08<br>7.46E-08<br>6.731E-08<br>6.751E-08<br>6.556E-08<br>7.152E-08<br>5.655E-08<br>3.799E-10<br>0<br>7.289E-08  | 0.0001532<br>0.0001532<br>0.0001535<br>0.0001343<br>0.0001443<br>0.0001385<br>0.0001385<br>0.0001235<br>0.0001212<br>0.0001212<br>0.0001212<br>0.0001162<br>9.182-05<br>6.944E-07<br>0<br>0.0001332   | 0.1203596<br>0.1142306<br>0.1310694<br>0.1216391<br>0.1426721<br>0.1426721<br>0.1323766<br>0.1323766<br>0.1323766<br>0.13233760<br>0.120155<br>0.1189339<br>0.1208562<br>0.1229461<br>0.1229461<br>0.1229461<br>0.1229462<br>0.022395<br>0<br>0<br>0.4432902              
  | 9.001E-08<br>7.322E-08<br>8.402E-08<br>9.146E-08<br>6.053E-08<br>7.071E-08<br>7.39E-08<br>7.39E-08<br>7.39E-08<br>6.439E-08<br>6.435E-08<br>3.941E-08<br>6.437E-08<br>6.837E-08<br>5.43E-08<br>3.44E-00<br>0<br>0<br>6.73E-08   | 0.0001462<br>0.0001365<br>0.0001365<br>0.0001365<br>0.0001267<br>0.0001486<br>9.344-05<br>0.0001149<br>9.342-05<br>0.0001149<br>0.0001140<br>0.0001043<br>0.0001043<br>0.0001049<br>6.402E-05<br>8.822E-05<br>6.214E-07<br>0<br>0<br>0.000123   | 2.4904488<br>1.9842855<br>2.2002208<br>2.1078468<br>2.2364653<br>3.1057198<br>3.0468346<br>2.9569573<br>2.6802462<br>3.1731567<br>2.6220547<br>2.6509615<br>2.7073267<br>2.9117616<br>4.4562831<br>0.0573728<br>0<br>0<br>0<br>1.3610254   
  | 1.863E-06<br>1.272E-06<br>1.31E-06<br>1.434E-06<br>1.628E-06<br>1.528E-06<br>1.528E-06<br>1.528E-06<br>1.401E-06<br>1.446E-06<br>9.333E-07<br>1.441E-06<br>1.541E-06<br>1.541E-06<br>1.541E-06<br>1.541E-06<br>0<br>0<br>2.066E-07   | 0.0030261<br>0.0022915<br>0.0022915<br>0.002153<br>0.0021564<br>0.0021564<br>0.0025444<br>0.0026444<br>0.002754<br>0.0024824<br>0.0022754<br>0.0023757<br>0.0015163<br>0.0023497<br>0.0021782<br>0.0023849<br>0.0023849<br>0.0023849<br>0.002377  | 2.3566842<br>1.8789238<br>2.0977756<br>2.00572501<br>2.1283629<br>2.9552749<br>2.900207<br>2.8217082<br>2.7301698<br>3.0191874<br>2.4772701<br>2.5033689<br>2.5033689<br>2.5033689<br>3.9331649<br>4.68371<br>4.168431<br>0.0463027<br>0<br>0<br>1.2345153   | 1.762E-06<br>1.204E-06<br>1.345E-06<br>1.345E-06<br>1.263E-06<br>1.263E-06<br>1.263E-06<br>1.323E-06<br>1.323E-06<br>1.323E-06<br>1.323E-06<br>1.323E-06<br>1.501E-06<br>1.501E-06<br>1.501E-06<br>1.501E-06<br>1.536E-06<br>7.03E-09<br>0<br>1.874E-07  | 0.002636<br>0.0019569<br>0.0021848<br>0.0020885<br>0.0022167<br>0.0025171<br>0.0019592<br>0.0025191<br>0.0025191<br>0.0021701<br>0.0021701<br>0.002479<br>0.0024799<br>0.0024709<br>1.285E-05<br>0<br>0<br>0.003426   
  | 2.2440968<br>1.7795138<br>1.9915216<br>1.8961342<br>2.0102391<br>2.7864095<br>2.7220732<br>2.6462899<br>2.5622851<br>2.8364013<br>2.3321589<br>2.3321589<br>2.3726056<br>2.4375101<br>2.6016999<br>3.7301495<br>4.4767836<br>3.9755324<br>0<br>0<br>1.1398552   | 1.6782-06<br>1.141E-06<br>1.2772-06<br>1.215E-06<br>1.215E-06<br>1.191E-06<br>1.3692-06<br>1.3692-06<br>1.3692-06<br>1.246E-06<br>1.246E-06<br>1.246E-06<br>1.246E-06<br>1.246E-06<br>1.302E-06<br>8.339E-07<br>1.196E-06<br>6.237E-09<br>0<br>1.731E-07  | 0.0027267<br>0.0018534<br>0.0018534<br>0.0020742<br>0.0019347<br>0.0029347<br>0.0029347<br>0.0023625<br>0.0028274<br>0.0022238<br>0.002625<br>0.0026241<br>0.0020241<br>0.0020241<br>0.0020241<br>0.0020313<br>0.002313<br>0.001345<br>0.001345<br>0.001345<br>0.001345<br>0.001345<br>0.0003163<br>0.0003163<br>0.0003163<br>0.0003163<br>0.0003163<br>0.0003163<br>0.0003163<br>0.0003163<br>0.0003163<br>0.0003163<br>0.0003163<br>0.0003163<br>0.0003163<br>0.0003163<br>0.0003163<br>0.0003163<br>0.0003163<br>0.0003163<br>0.0003163<br>0.0003163<br>0.0003163<br>0.0003163<br>0.0003163<br>0.0003163<br>0.0003163<br>0.0003163<br>0.0003163<br>0.0003163<br>0.0003163<br>0.0003163<br>0.0003163<br>0.0003163<br>0.0003163<br>0.0003163<br>0.0003163<br>0.0003163<br>0.0003163<br>0.0003163<br>0.0003163<br>0.0003163<br>0.0003163<br>0.0003163<br>0.0003163<br>0.0003163<br>0.0003163<br>0.0003163<br>0.0003163<br>0.0003163<br>0.0003163<br>0.0003163<br>0.0003163<br>0.0003163<br>0.0003163<br>0.0003163<br>0.0003163<br>0.0003163<br>0.0003163<br>0.0003163<br>0.0003165<br>0.0003165<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005  
  | 2.623364<br>2.11248<br>2.3459273<br>2.2439219<br>2.3971946<br>3.2561464<br>3.10528226<br>3.1051925<br>3.0057113<br>3.3257455<br>2.7552623<br>2.7818703<br>2.83965211<br>3.0470013<br>3.621947<br>3.6427163<br>0.0604669<br>0<br>0<br>1.8903165   | 1.962E-06<br>1.354E-06<br>1.354E-06<br>1.438E-06<br>1.337E-06<br>1.337E-06<br>1.377E-06<br>1.427E-06<br>1.427E-06<br>1.472E-06<br>1.472E-06<br>1.472E-06<br>1.472E-06<br>1.517E-06<br>9.766E-07<br>1.398E-06<br>9.166E-06<br>1.488E-06<br>9.18E-09<br>0<br>2.87E-07  | 0.0031876<br>0.0022001<br>0.0024433<br>0.0024337<br>0.0024637<br>0.0022645<br>0.0027711<br>0.002156<br>0.002571<br>0.002454<br>0.0024140<br>0.0024140<br>0.0024147<br>0.0022716<br>0.002716<br>0.002716<br>0.002716<br>0.0022716<br>0.002245   | 2.3644564<br>1.8937445<br>2.1225900<br>2.0177733<br>2.1529112<br>2.928047<br>2.8544498<br>2.7808064<br>2.6943024<br>2.7940546<br>2.5583663<br>2.724646<br>3.8826630<br>4.16496386<br>0.043318<br>0<br>0<br>1.5831454  
  | 1.768:-00         0.002           1.714E-06         0.001           1.81E-06         0.002           1.38E-06         0.002           1.238E-06         0.002           1.38E-06         0.002           1.36F-06         0.002           1.36F-06         0.002           1.373E-07         0.001           1.248E-06         0.002           5.57FE-09         1.022           0         0         0           0         0         0           0         0         0  | 8873         0.04           9723         0.03           2107         0.04           1015         0.044           2422         0.04           2422         0.04           033         0.05           4774         0.05           9308         0.05           93384         0.04           1284         0.04           1284         0.04           1438         0.05           0271         0.07           1432         0.08           E-05         0.00           4393         0.0  
   | 147043         3:           383832         2           383832         2           483832         2           483832         2           405571         2           133125         2:           338011         2:           338012         2:           1314016         2:           1514016         2:           165671         2.           165671         2.           165671         2.           167674         2.4           166276         2.4           18821         1.6           776164         2.4           164254         2:           18928         2.6           53993         8.1           0         0           14252         6.4   
  | 5.18E-08         2           6.619E-08         4           2.6E-08         2           2.6E-08         4           2.99E-08         3           8.74E-08         4           1.97E-08         3           4551E-08         4           442E-08         3           498E-08         4           558E-08         4           651E-08         4           488E-08         4           651E-08         4           198E-01         4           651E-08         4           198E-08         4           0         10           0         0           0         10           0         10  | 3.998E-05<br>4.254E-05<br>4.224E-05<br>4.211E-05<br>3.736E-05<br>4.67E-05<br>3.5569E-05<br>4.307E-05<br>4.307E-05<br>4.307E-05<br>4.307E-05<br>1.059E-05<br>4.22E-05<br>1.501E-05<br>1.501E-05<br>1.498E-06<br>0<br>0<br>1.18E-05  | 0.0443322<br>0.044017<br>0.0470156<br>0.0584313<br>0.0584313<br>0.0584373<br>0.0558211<br>0.0538271<br>0.0538271<br>0.0538271<br>0.0527943<br>0.0507781<br>0.06422<br>0.0527943<br>0.06422   
   | 2.07203 4<br>2.822E-08 4<br>2.822E-08 4<br>3.014E-08 4<br>2.822E-08 4<br>3.122E-08 4<br>3.122E-08 4<br>3.122E-08 4<br>2.8379E-08 4<br>2.8379E-08 4<br>2.832E-08 4<br>3.009E-08 4<br>8.009E-08 4<br>8.773E-10 4<br>0 0  | 1.584E-05<br>1.897E-05<br>1.057E-05<br>1.072E-05<br>3.876E-05<br>1.307E-05<br>1.307E-05<br>1.307E-05<br>1.307E-05<br>1.582E-05<br>1.391E-05<br>1.689E-05<br>1.689E-05<br>1.604E-06<br>0<br>0   |
| Sheung Tsuen B/T and Car Park<br>Sheung Tsuen B/T and Car Park  | RD_29<br>RD_29<br>RD_29<br>RD_29<br>RD_29<br>RD_29<br>RD_29<br>RD_29<br>RD_29<br>RD_29<br>RD_29<br>RD_29<br>RD_29<br>RD_29<br>RD_29<br>RD_29<br>RD_30,31<br>RD_30,31<br>RD_30,31<br>RD_30,31<br>RD_30,31<br>RD_30,31  | 9900-1000<br>1000-1100<br>1100-1200<br>1200-1300<br>1300-1400<br>1400-1500<br>1500-1600<br>1600-1700<br>2000-2100<br>2000-2100<br>2200-2200<br>2300-0000<br>0000-0100<br>0200-0200<br>0200-0300<br>0300-4400<br>400-0500<br>0400-0500   | 0.1281945<br>0.1360751<br>0.1360751<br>0.1507293<br>0.1557266<br>0.145986<br>0.14594652<br>0.132075<br>0.132075<br>0.1320254<br>0.132254<br>0.132254<br>0.132254<br>0.132254<br>0.132254<br>0.132254<br>0.132254<br>0.030941<br>0<br>0<br>0<br>0<br>0  | 9.342-006 ( 9.344-08 ( 9.344-08 ( 9.344-08 ( 8.723E-08 ( 1.032-07 ( 6.535E-08 ( 7.771E-08 ( 8.131E-08 ( 7.71E-08 ( 8.131E-08 ( 7.71E-08 ( 8.131E-08 ( 7.71E-08 ( 8.131E-08 ( 7.715-08 ( 5.774E-08 ( 7.756E-08 ( 7.5756E-08 ( 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0  
  | 0.0001615 (<br>0.0001335 (<br>0.0001518 (<br>0.0001518 (<br>0.0001614 (<br>0.000164 (<br>0.000166 (<br>0.000166 (<br>0.000165 (<br>0.000165 (<br>0.000165 (<br>0.000165 (<br>0.000166 (<br>0.0000166 (<br>0.0000166 (<br>0.0000166 (<br>0.00000000000000000000000   | 0.1261103<br>0.1210641<br>0.1385867<br>0.1291377<br>0.1521578<br>0.1491352<br>0.1399004<br>0.1421375<br>0.1396503<br>0.1461086<br>0.1267454<br>0.12612411<br>0.1284960<br>0.169957<br>0.2231399<br>0.1762915<br>0.0025024<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0   | 9.431E-08<br>7.761E-08<br>8.884E-08<br>8.278E-08<br>9.754E-08<br>6.373E-08<br>6.632E-08<br>7.46E-08<br>7.46E-08<br>6.731E-08<br>6.656E-08<br>7.46E-08<br>7.152E-08<br>7.565E-08<br>3.799E-10<br>0<br>7.289E-08<br>7.265E-08  | 0.0001532<br>0.0001532<br>0.0001261<br>0.000143<br>0.000143<br>0.0001285<br>0.0001214<br>9.882E-05<br>0.0001214<br>9.882E-05<br>0.000122<br>0.0001228<br>0.0001162<br>9.18E-05<br>6.6944E-07<br>0<br>0.0001332<br>0.0001332   | 0.1203596<br>0.1142306<br>0.1310.694<br>0.1310.694<br>0.13205721<br>0.1416376<br>0.1323766<br>0.1323766<br>0.13243760<br>0.1320173<br>0.1320373<br>0.1320373<br>0.1208562<br>0.1292461<br>0.1624807<br>0.2148766<br>0.1624807<br>0<br>0.042348202<br>0.4438849<br>0.00215***********************************   
   | 9.001E-08<br>7.322E-08<br>8.402E-08<br>9.146E-08<br>6.053E-08<br>7.071E-08<br>7.39E-08<br>7.39E-08<br>6.419E-08<br>6.435E-08<br>3.941E-08<br>5.43E-08<br>3.941E-08<br>5.43E-08<br>3.94E-08<br>0<br>0<br>6.73E-08<br>6.73E-08<br>6.73E-08<br>6.73E-08  | 0.0001462<br>0.0001365<br>0.0001365<br>0.0001365<br>0.0001267<br>0.0001486<br>9.344-05<br>0.0001149<br>9.342-05<br>0.0001149<br>9.342-05<br>0.0001043<br>0.0001043<br>0.0001049<br>0.0001049<br>8.822E-05<br>6.214E-07<br>0<br>0<br>0.000123<br>0.000123  | 2.4904488<br>1.9842855<br>2.2002208<br>2.1078468<br>2.2364653<br>3.1057198<br>3.0468346<br>2.9569573<br>2.620267<br>2.620267<br>2.6509615<br>2.7073267<br>2.9117616<br>4.4562831<br>0.0573728<br>0<br>0<br>1.3610254<br>1.3610254   
   | 1.8638-06<br>1.272E-06<br>1.31E-06<br>1.43E-06<br>1.434E-06<br>1.628E-06<br>1.628E-06<br>1.628E-06<br>1.401E-06<br>1.446E-06<br>9.333E-07<br>1.4416E-06<br>1.541E-06<br>1.541E-06<br>1.541E-06<br>1.541E-06<br>0<br>0<br>0<br>2.066E-07<br>0.0500<br>0<br>0.0500<br>0<br>0.0500<br>0<br>0.0500<br>0<br>0.0500<br>0<br>0.0500<br>0<br>0.0500<br>0<br>0.0500<br>0<br>0.0500<br>0<br>0.0500<br>0<br>0.0500<br>0<br>0.0500<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0   | 0.030261<br>0.002266<br>0.0022915<br>0.0022915<br>0.0021953<br>0.0021564<br>0.0022644<br>0.0020544<br>0.002757<br>0.0023008<br>0.0023008<br>0.0023409<br>0.0023206<br>1.592E-05<br>0<br>0<br>0<br>0<br>0.0003777<br>0.0003777   | 2.3566842<br>1.8789238<br>2.0977756<br>2.0052501<br>2.1283629<br>2.9552749<br>2.900207<br>2.8217082<br>2.7301698<br>3.0191874<br>2.4772701<br>2.5033689<br>2.5607825<br>2.739292<br>3.39311649<br>4.68371<br>4.1688413<br>0.0463027<br>0<br>1.2345153<br>1.2303809<br>0.042704   | 1.762E-06<br>1.204E-06<br>1.345E-06<br>1.345E-06<br>1.263E-06<br>1.263E-06<br>1.263E-06<br>1.263E-06<br>1.206E-06<br>1.326E-06<br>1.332E-06<br>1.332E-06<br>1.332E-06<br>1.332E-06<br>1.332E-06<br>1.332E-06<br>1.345E-07<br>1.368E-07<br>1.868E-07<br>1.462E-07<br>1.868E-07<br>1.462E-07<br>1.868E-07<br>1.462E-07<br>1.868E-07<br>1.462E-07<br>1.462E-07<br>1.462E-07<br>1.462E-07<br>1.462E-07<br>1.462E-07<br>1.462E-07<br>1.462E-07<br>1.462E-07<br>1.462E-07<br>1.462E-07<br>1.462E-07<br>1.462E-07<br>1.462E-07<br>1.462E-07<br>1.462E-07<br>1.462E-07<br>1.462E-07<br>1.462E-07<br>1.462E-07<br>1.462E-07<br>1.462E-07<br>1.462E-07<br>1.462E-07<br>1.462E-07<br>1.462E-07<br>1.462E-07<br>1.462E-07<br>1.462E-07<br>1.462E-07<br>1.462E-07<br>1.462E-07<br>1.462E-07<br>1.462E-07<br>1.462E-07<br>1.462E-07<br>1.462E-07<br>1.462E-07<br>1.462E-07<br>1.462E-07<br>1.462E-07<br>1.462E-07<br>1.462E-07<br>1.462E-07<br>1.462E-07<br>1.462E-07<br>1.462E-07<br>1.462E-07<br>1.462E-07<br>1.462E-07<br>1.462E-07<br>1.462E-07<br>1.462E-07<br>1.462E-07<br>1.462E-07<br>1.462E-07<br>1.462E-07<br>1.462E-07<br>1.462E-07<br>1.462E-07<br>1.462E-07<br>1.462E-07<br>1.462E-07<br>1.462E-07<br>1.462E-07<br>1.462E-07<br>1.462E-07<br>1.462E-07<br>1.462E-07<br>1.462E-07<br>1.462E-07<br>1.462E-07<br>1.462E-07<br>1.462E-07<br>1.462E-07<br>1.462E-07<br>1.462E-07<br>1.462E-07<br>1.462E-07<br>1.462E-07<br>1.462E-07<br>1.462E-07<br>1.462E-07<br>1.462E-07<br>1.462E-07<br>1.462E-07<br>1.462E-07<br>1.462E-07<br>1.462E-07<br>1.462E-07<br>1.462E-07<br>1.462E-07<br>1.462E-07<br>1.462E-07<br>1.462E-07<br>1.462E-07<br>1.462E-07<br>1.462E-07<br>1.462E-07<br>1.462E-07<br>1.462E-07<br>1.462E-07<br>1.462E-07<br>1.462E-07<br>1.462E-07<br>1.462E-07<br>1.462E-07<br>1.462E-07<br>1.462E-07<br>1.462E-07<br>1.462E-07<br>1.462E-07<br>1.462E-07<br>1.462E-07<br>1.462E-07<br>1.462E-07<br>1.462E-07<br>1.462E-07<br>1.462E-07<br>1.462E-07<br>1.462E-07<br>1.462E-07<br>1.462E-07<br>1.462E-07<br>1.462E-07<br>1.462E-07<br>1.462E-07<br>1.462E-07<br>1.462E-07<br>1.462E-07<br>1.462E-07<br>1.462E-07<br>1.462E-07<br>1.462E-07<br>1.462E-07<br>1.462E-07<br>1.462E-07<br>1.462E-07<br>1.462E-07<br>1.462E-07<br>1.462E-07<br>1.462E-07<br>1.462E-07<br>1.462E-07<br>1.462E-07<br>1.462E-07<br>1.462E-07<br>1.462E-07<br>1.462E-07<br>1.462E-07<br>1.462E-07<br>1.462E-07<br>1.462E-07<br>1.462E-07<br>1.462E-07<br>1.462E-07<br>1.462E-07<br>1.462E-07<br>1.462E-07<br>1.462E-07<br>1.462E-07   | 0.002635<br>0.0019569<br>0.0021848<br>0.0020885<br>0.0022167<br>0.0025171<br>0.0019592<br>0.0025191<br>0.0025191<br>0.0021701<br>0.0021701<br>0.0024790<br>0.0024790<br>0.0024790<br>0.0024709<br>1.285E-05<br>0<br>0<br>0.0003426<br>0.0003426  
   | 2.2440968<br>1.7795138<br>1.9915216<br>1.8961342<br>2.0102391<br>2.7864095<br>2.7220732<br>2.6462899<br>2.6462899<br>2.6462899<br>2.5622851<br>2.8364013<br>2.3321589<br>2.3726056<br>2.4375101<br>2.6016999<br>3.7301495<br>4.4767836<br>3.975523<br>1.0410785<br>0<br>0<br>0<br>1.398552<br>1.1413822<br>0.041078   | 1.678E-06<br>1.141E-06<br>1.277E-06<br>1.215E-06<br>1.289E-06<br>1.151E-06<br>1.454E-06<br>1.315E-06<br>1.365E-06<br>1.365E-06<br>1.302E-06<br>8.339E-07<br>1.302E-06<br>8.339E-07<br>1.196E-06<br>1.274E-06<br>0<br>0<br>1.731E-07<br>1.733E-07<br>6.232E-07<br>0  | 0.0027267<br>0.0018534<br>0.0020742<br>0.0019748<br>0.0020937<br>0.0019347<br>0.0023625<br>0.0013347<br>0.0023625<br>0.0023625<br>0.002241<br>0.002241<br>0.002241<br>0.002241<br>0.002241<br>0.002241<br>0.002241<br>0.002155<br>0.001548<br>0.002154<br>0.002313<br>0.002313<br>0.002313<br>0.002313<br>0.0003163<br>0.0003167<br>0.0003167   | 2.623364<br>2.11248<br>3.3459273<br>2.3971946<br>3.2614464<br>3.1928226<br>3.1051925<br>3.0057113<br>3.3257455<br>2.7552623<br>2.7818703<br>3.8396521<br>3.0470013<br>3.3621382<br>5.1979947<br>4.6427163<br>0.0604659<br>0<br>0.<br>8.8903165<br>1.8903165   
  | 1.962E-06<br>1.354E-06<br>1.354E-06<br>1.504E-06<br>1.438E-06<br>1.537E-06<br>1.337E-06<br>1.327E-06<br>1.327E-06<br>1.472E-06<br>1.472E-06<br>1.472E-06<br>1.472E-06<br>1.486E-06<br>1.537E-06<br>1.488E-06<br>9.188-09<br>0<br>0<br>2.87E-07<br>2.87E-07<br>0.655E-07  | 0.0031876<br>0.0022001<br>0.0024433<br>0.0024337<br>0.002463<br>0.0022645<br>0.0027711<br>0.002156<br>0.00256<br>0.0023913<br>0.0024646<br>0.0015467<br>0.002245<br>0.002245<br>0<br>0.0005245<br>0.0005245  | 2.3644564<br>1.8937445<br>2.1225900<br>2.0177733<br>2.1529112<br>2.928047<br>2.8544498<br>2.7808064<br>2.7808064<br>2.6943024<br>2.5943024<br>2.5583663<br>2.724646<br>3.8926302<br>4.6916603<br>4.1449386<br>0.043318<br>0<br>0<br>1.5831454<br>1.585267<br>0.0420825   
   | 1.768:00 0.002<br>1.231:06 0.002<br>1.233:06 0.002<br>1.233:06 0.002<br>1.231:06 0.002<br>1.231:06 0.002<br>1.231:06 0.002<br>1.251:06 0.002<br>1.589:06 0.002<br>1.589:06 0.002<br>1.311:06 0.002<br>1.313:06 0.002<br>1.313:06 0.002<br>1.313:06 0.002<br>1.313:06 0.002<br>1.329:06 0.002<br>1.329:06 0.002<br>1.329:06 0.002<br>1.329:06 0.002<br>1.329:06 0.002<br>1.329:06 0.002<br>1.329:06 0.002<br>0.00 0.002<br>0.00 0.002<br>0.00 0.002<br>0.000 0.0000<br>0.0000 0.0000<br>0.0000<br>0.0000 0.0000<br>0.00000<br>0.000000000   | 873         0.04           9723         0.03           2107         0.04           1015         0.044           2422         0.04           2422         0.04           2422         0.04           2422         0.04           2422         0.04           2422         0.04           2422         0.04           1284         0.04           1284         0.04           1284         0.04           1284         0.04           1585         0.08           1585         0.00           41390         0.0           4393         0.0           4393         0.0  | 147043         3.           383832         2           383832         2           408499         2.           313125         2.           3313125         2.           338011         2.           338012         2.           338013         2.           338014         2.           14016         2.           195717         2.4           165511         3.4           157074         2.4           164757         2.4           164254         2.           183821         1.6           177614         2.4           10         0           0         0           14252         6.4           44252         6.4  
   
  | 5.18E-08         2           6.619E-08         2           6.619E-08         4           2.6E-08         4           2.99E-08         3           8.74E-08         4           9.99E-08         3           651E-08         4           442E-08         3           442E-08         3           551E-08         4           428E-08         4           661E-08         4           598E-08         4           661E-08         4           198E-10         1           0         0           0         1           0         1           0         1           0         1           0         1           0         1           0         1           0         1           0         1           0         1           0         1           0         1           0         1           0         1           0         1           0         1           0         1  | 3.998E-05<br>4.254E-05<br>4.224E-05<br>4.211E-05<br>3.736E-05<br>4.67E-05<br>3.5569E-05<br>4.307E-05<br>4.059E-05<br>4.059E-05<br>4.059E-05<br>1.042E-05<br>1.266E-05<br>0<br>0<br>1.18E-05<br>1.18E-05  | 0.0443322<br>0.044017<br>0.0470156<br>0.0584313<br>0.0584313<br>0.0584373<br>0.0538211<br>0.0538211<br>0.0538211<br>0.0538211<br>0.0538211<br>0.0538211<br>0.053794<br>0.0557794<br>0.0843242<br>0.0843242<br>0.0843242<br>0.04622   | 2.842E-08 4<br>8.014E-08 4<br>8.014E-08 4<br>8.014E-08 4<br>2.4397E-08 4<br>2.336E-08 5<br>2.336E-08 5<br>2.336E-08 5<br>2.351E-08 4<br>2.83E-08 4<br>2.83E-08 4<br>2.83E-08 4<br>2.83E-08 4<br>2.83E-08 4<br>3.009E-08 4<br>2.8353E-08 4<br>3.009E-08 4<br>2.853E-08 4<br>0<br>0<br>0<br>0.014E-09 5<br>7.014E-09 5<br>7.0  | 1.584E-05<br>1.897E-05<br>1.057E-05<br>1.072E-05<br>3.876E-05<br>3.376E-05<br>3.331E-05<br>1.307E-05<br>1.307E-05<br>1.307E-05<br>1.391E-05<br>1.636E-05<br>1.636E-05<br>0<br>0<br>1.282E-05<br>1.282E-05  
   |
| Sheung Touen B/T and Car Park<br>Sheung Touen B/T and Car Park<br>Ullage Access Road<br>Village Access Road<br>Village Access Road<br>Village Access Road<br>Village Access Road<br>Village Access Road<br>Village Access Road   | RD_29<br>RD_29<br>RD_29<br>RD_29<br>RD_29<br>RD_29<br>RD_29<br>RD_29<br>RD_29<br>RD_29<br>RD_29<br>RD_29<br>RD_29<br>RD_29<br>RD_29<br>RD_29<br>RD_29<br>RD_29<br>RD_29<br>RD_30.31<br>RD_30.31<br>RD_30.31<br>RD_30.31<br>RD_30.31<br>RD_30.31<br>RD_30.31<br>RD_30.31   | 9900-1000<br>1000-1100<br>1100-1200<br>1200-1300<br>1300-1400<br>1400-1500<br>1500-1600<br>1600-1700<br>2000-2100<br>2100-2200<br>2200-2200<br>2200-2200<br>2200-3000<br>0000-0100<br>0000-0100<br>0000-0100<br>0000-0100<br>0000-0100<br>0000-0100<br>0000-000<br>0500-0600<br>0500-0600   | 0.1281945<br>0.1360751<br>0.13607513<br>0.1557266<br>0.145988<br>0.1454652<br>0.132205<br>0.1322055<br>0.1322075<br>0.1322075<br>0.1322075<br>0.1322375<br>0.1322375<br>0.132234<br>0.032924<br>0.032924<br>0.05292911<br>0.02230554   | 9.347-006 ( 9.347-006 ( 9.347-006 ( 9.347-006 ( 9.347-006 ( 1.037-07 ( 6.6557-08 ( 7.7997-08 ( 6.3357-08 ( 7.7717-08 ( 6.3357-08 ( 7.71167-08 ( 6.9393-08 ( 6.9393-08 ( 7.0697-08 ( 4.3357-08 ( 7.5067-08 ( 4.3357-08 ( 7.5067-08 ( 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0  
   | 0.0001615 (<br>0.0001335 (<br>0.0001518 (<br>0.0001518 (<br>0.0001614 (<br>0.000164 (<br>0.000164 (<br>0.000164 (<br>0.000165 (<br>0.000136 (<br>0.000165 (<br>0.000125 (<br>0.000146 (<br>0.000146 (<br>0.000146 (<br>0.000146 (<br>0.000128 (<br>0.00010 (<br>0.0000 (<br>0.00000 (<br>0.00000 (<br>0.0000 (<br>0.00000 ( | 0.1261103<br>0.1210641<br>0.1385867<br>0.1291377<br>0.1521578<br>0.1491352<br>0.1399004<br>0.1461086<br>0.1267454<br>0.1261241<br>0.1261241<br>0.126957<br>0.2231399<br>0.1762915<br>0.0025024<br>0<br>0<br>0<br>0<br>480112<br>0.480112<br>0.4264825<br>0.0261446   | 9.431E-08<br>7.761E-08<br>8.278E-08<br>9.754E-08<br>9.754E-08<br>6.373E-08<br>6.632E-08<br>7.473E-08<br>6.656E-08<br>6.656E-08<br>6.6771E-08<br>7.476E-08<br>7.476E-08<br>7.476E-08<br>7.476E-08<br>7.475E-08<br>3.799E-08<br>3.799E-00<br>0<br>7.289E-08<br>3.869E-10<br>6.116E-08  | 0.0001532<br>0.0001532<br>0.0001261<br>0.0001443<br>0.0001443<br>0.0001385<br>0.0001214<br>9.882E-05<br>0.000122<br>0.000122<br>0.000122<br>0.000112<br>0.0001025<br>0.6689E-05<br>8.85E-05<br>0.0001162<br>9.18E-05<br>0.944E-07<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0,0001332   | 0.1203596<br>0.1142306<br>0.1310.694<br>0.1310.694<br>0.13205721<br>0.1416376<br>0.1323766<br>0.1323766<br>0.13243760<br>0.1320135<br>0.1320135<br>0.1208562<br>0.1292461<br>0.1624807<br>0.2148766<br>0.1694062<br>0.0022395<br>0<br>0.022395<br>0.0282845 
  | 9.001E-08<br>7.322E-08<br>8.402E-08<br>9.146E-08<br>6.033E-08<br>7.071E-08<br>7.071E-08<br>7.071E-08<br>7.032E-08<br>7.35E-08<br>6.435E-08<br>5.435E-08<br>5.43E-08<br>5.43E-08<br>5.43E-08<br>5.43E-08<br>5.43E-08<br>0<br>0<br>0<br>0<br>0<br>5.73E-08<br>6.73E-08<br>5.43E-08<br>5.43E-08<br>5.43E-08<br>5.442E-08<br>5.442E-08<br>5.442E-08<br>5.442E-08<br>5.442E-08<br>5.442E-08<br>5.442E-08<br>5.442E-08<br>5.442E-08<br>5.442E-08<br>5.442E-08<br>5.442E-08<br>5.442E-08<br>5.442E-08<br>5.442E-08<br>5.442E-08<br>5.442E-08<br>5.442E-08<br>5.442E-08<br>5.444E-08<br>5.444E-08<br>5.444E-08<br>5.444E-08<br>5.444E-08<br>5.444E-08<br>5.444E-08<br>5.444E-08<br>5.444E-08<br>5.444E-08<br>5.444E-08<br>5.444E-08<br>5.444E-08<br>5.444E-08<br>5.444E-08<br>5.444E-08<br>5.444E-08<br>5.444E-08<br>5.444E-08<br>5.444E-08<br>5.444E-08<br>5.444E-08<br>5.444E-08<br>5.444E-08<br>5.444E-08<br>5.444E-08<br>5.444E-08<br>5.444E-08<br>5.444E-08<br>5.444E-08<br>5.444E-08<br>5.444E-08<br>5.444E-08<br>5.444E-08<br>5.444E-08<br>5.444E-08<br>5.444E-08<br>5.444E-08<br>5.444E-08<br>5.444E-08<br>5.444E-08<br>5.444E-08<br>5.444E-08<br>5.444E-08<br>5.444E-08<br>5.444E-08<br>5.444E-08<br>5.444E-08<br>5.444E-08<br>5.444E-08<br>5.444E-08<br>5.444E-08<br>5.444E-08<br>5.444E-08<br>5.444E-08<br>5.444E-08<br>5.444E-08<br>5.444E-08<br>5.444E-08<br>5.444E-08<br>5.444E-08<br>5.444E-08<br>5.444E-08<br>5.444E-08<br>5.444E-08<br>5.444E-08<br>5.444E-08<br>5.444E-08<br>5.444E-08<br>5.444E-08<br>5.444E-08<br>5.444E-08<br>5.444E-08<br>5.444E-08<br>5.444E-08<br>5.444E-08<br>5.444E-08<br>5.444E-08<br>5.444E-08<br>5.444E-08<br>5.444E-08<br>5.444E-08<br>5.444E-08<br>5.444E-08<br>5.444E-08<br>5.444E-08<br>5.444E-08<br>5.444E-08<br>5.444E-08<br>5.444E-08<br>5.444E-08<br>5.444E-08<br>5.444E-08<br>5.444E-08<br>5.444E-08<br>5.444E-08<br>5.444E-08<br>5.444E-08<br>5.444E-08<br>5.444E-08<br>5.444E-08<br>5.444E-08<br>5.444E-08<br>5.444E-08<br>5.444E-08<br>5.444E-08<br>5.444E-08<br>5.444E-08<br>5.444E-08<br>5.444E-08<br>5.444E-08<br>5.444E-08<br>5.444E-08<br>5.444E-08<br>5.444E-08<br>5.444E-08<br>5.444E-08<br>5.444E-08<br>5.444E-08<br>5.444E-08<br>5.444E-08<br>5.444E-08<br>5.444E-08<br>5.444E-08<br>5.444E-08<br>5.444E-08<br>5.444E-08<br>5.444E-08<br>5.444E-08<br>5.444E-08<br>5.444E-08<br>5.444E-08<br>5.444E-08<br>5.444E-08<br>5.444E-08<br>5.444E-08<br>5.444E-08<br>5.444E-08<br>5.444E-08<br>5.444E-08<br>5.444E-08<br>5.444E-08<br>5.444E-08<br>5.444E-08<br>5.444E-08<br>5.444E-08<br>5.444E-08<br>5.444E-08<br>5.444E-08<br>5.444E-08<br>5.444E-08<br>5.444E-08<br>5.444E-08<br>5.444E-08<br>5.444E-08<br>5.444E-08<br>5.444E-08<br>5.444E-08<br>5.444E-08<br>5.444E-08<br>5.444E-08<br>5.444E-08<br>5.444E-08<br>5.444E- | 0.0001462<br>0.000119<br>0.0001365<br>0.0001267<br>0.0001267<br>0.0001486<br>9.834E-05<br>0.0001149<br>9.34E-05<br>0.0001146<br>0.0001201<br>0.0001032<br>0.0001032<br>0.0001032<br>0.000113<br>8.822E-05<br>6.214E-05<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0  | 2.4904488<br>1.9842855<br>2.2002208<br>2.1078468<br>2.2364653<br>3.0468346<br>2.9569573<br>2.8602462<br>3.1731567<br>2.620547<br>2.6509615<br>2.7073267<br>2.9117616<br>4.1828878<br>4.9637927<br>4.4562831<br>0.0573728<br>0<br>0<br>1.3610254<br>1.3610254<br>0.06039622   
  | 1.863E-06<br>1.272E-06<br>1.41E-06<br>1.431E-06<br>1.434E-06<br>1.327E-06<br>1.628E-06<br>1.264E-06<br>1.401E-06<br>1.401E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.     | 0.030261<br>0.002266<br>0.0022915<br>0.002295<br>0.002295<br>0.0021564<br>0.002544<br>0.002544<br>0.002757<br>0.002757<br>0.002308<br>0.002454<br>0.0023206<br>1.59265<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0   | 2.3566842<br>1.8789238<br>2.0977776<br>2.00527501<br>2.1283629<br>2.9552749<br>2.900207<br>2.8217082<br>2.7301698<br>3.0191874<br>2.4772701<br>2.5038689<br>2.5607825<br>2.7392982<br>3.9311649<br>4.68371<br>4.16884133<br>0.0463027<br>0<br>0<br>0<br>1.2345153<br>1.2303809<br>0.04704<br>0.5415078   | 1.762E-06<br>1.204E-06<br>1.245E-06<br>1.245E-06<br>1.263E-06<br>1.263E-06<br>1.264E-06<br>1.264E-07<br>1.37E-06<br>1.32E-06<br>1.32E-06<br>1.32E-06<br>1.32E-06<br>1.32E-06<br>1.32E-06<br>1.32E-06<br>1.32E-06<br>1.32E-06<br>1.32E-06<br>1.32E-06<br>1.32E-06<br>1.32E-07<br>1.88E-07<br>7.142E-09<br>1.644E-07<br>1.644E-07  | 0.0028636<br>0.0019569<br>0.0021848<br>0.00220885<br>0.00220519<br>0.0025171<br>0.000352171<br>0.00036204<br>0.0002520<br>0.00226204<br>0.00021727<br>0.0002225<br>0.0014265<br>0.00024710<br>0.00024709<br>1.285E05<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0  
  | 2.2440968<br>1.7795138<br>1.8961342<br>2.0102391<br>2.7864095<br>2.7220732<br>2.6462899<br>2.5622851<br>2.3321589<br>2.322580<br>2.3221589<br>2.3225056<br>2.32475101<br>2.6016999<br>3.7301495<br>4.4767836<br>0.0410785<br>0<br>0<br>1.1398552<br>1.1413822<br>0.04927676   | 1.678E-06<br>1.141E-06<br>1.277E-06<br>1.215E-06<br>1.289E-06<br>1.151E-06<br>1.454E-06<br>1.316E-06<br>1.365E-06<br>1.302E-06<br>8.339E-07<br>1.302E-06<br>8.339E-07<br>1.196E-06<br>1.274E-06<br>0<br>0<br>1.731E-07<br>1.733E-07<br>6.335E-09<br>1.515E-07   | 0.0027267<br>0.0018534<br>0.0020742<br>0.0019748<br>0.0020937<br>0.0019347<br>0.0023625<br>0.0013347<br>0.0023625<br>0.0013347<br>0.002238<br>0.0020321<br>0.0020592<br>0.0020702<br>0.0023135<br>0.0023154<br>0.0023155<br>0.0023154<br>0.0023165<br>0.0023165<br>0.0003165<br>0.0003165<br>0.0002766   
  | 2.623364<br>2.11248<br>3.3459273<br>2.3971946<br>3.2614464<br>3.1928226<br>3.1051925<br>3.0057113<br>3.3257455<br>2.7552623<br>2.7818703<br>3.8396521<br>3.0470013<br>3.3621382<br>3.1979947<br>4.6427163<br>0.0604659<br>0<br>0<br>8.8903165<br>1.8903165<br>1.8903165<br>1.8270176   | 1.9622-06<br>1.354E-06<br>1.354E-06<br>1.438E-06<br>1.537E-06<br>1.357E-06<br>1.705E-06<br>1.327E-06<br>1.606E-06<br>1.777E-06<br>1.448E-06<br>9.766E-07<br>1.398E-06<br>9.18E-09<br>0<br>0<br>2.87E-07<br>2.87E-07<br>9.695E-09<br>2.531E-07  | 0.0031876<br>0.0022001<br>0.0024433<br>0.002337<br>0.002347<br>0.0022645<br>0.0022645<br>0.0022645<br>0.0022645<br>0.0022645<br>0.0025865<br>0.0025865<br>0.0024177<br>1.678E-05<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0   | 2.3644564<br>1.8937445<br>2.1225909<br>2.0177733<br>2.1529112<br>2.928047<br>2.8544498<br>2.7808064<br>2.6943024<br>2.9747351<br>2.4915395<br>2.7583663<br>2.724646<br>3.8926302<br>4.1449386<br>0.043318<br>0<br>0<br>1.5831454<br>1.583267<br>0.0438825<br>0.05846321   
  | 1.768:00 0.002<br>1.246:0 0.002<br>1.38:1-0 0.002<br>1.38:1-0 0.002<br>1.38:2-0 0.002<br>1.38:2-0 0.002<br>1.38:2-0 0.002<br>1.38:2-0 0.002<br>1.38:2-0 0.002<br>1.38:2-0 0.002<br>1.38:2-0 0.002<br>1.33:1-0 0.002<br>1.34:1-0 0.002  | 873         0.04           9723         0.03           2107         0.044           1015         0.044           1015         0.044           2422         0.03           0.033         0.05           3384         0.044           5818         0.05           1284         0.044           2204         0.044           2204         0.044           2204         0.044           2204         0.044           2204         0.044           2204         0.044           2204         0.044           2204         0.044           2204         0.044           2204         0.044           2204         0.044           2204         0.044           1585         0.026           34393         0.044           43393         0.044           0.388         0.020  | 147043         3.           383832         2           383832         2           408499         2.           313125         2.           331305         2.           338011         2.           338012         2.           338011         2.           338012         3.           14016         2.           157071         2.           165511         3.           157074         2.4           1647677         2.4           164254         2.           184254         2.           193393         8.1           0         0           14252         6.4           153939         8.1           153939         8.1           153939         8.1           153939         8.1           153939         8.1           153939         8.1           153939         8.1           153939       
 8.1           153939         8.1  
   | 5.18E-08         2           6.619E-08         4           2.6E-08         4           2.99E-08         3           8.746-08         1           8.746-08         3           8.746-08         1           9.776-08         3           8.746-08         1           0.197E-08         3           6.51E-08         4           4.42E-08         3           9.598E-08         4           6.51E-08         4           6.51E-08         4           6.52E-08         4           0         0           0         456E-09           1.98E-00         1           1.98E-00         1           1.98E-01         1           3.34E-09         1  | 3.998E-05<br>4.254E-05<br>4.224E-05<br>4.21E-05<br>3.736E-05<br>4.67E-05<br>4.307E-05<br>4.307E-05<br>4.098E-05<br>3.967E-05<br>4.098E-05<br>1.699E-05<br>1.501E-05<br>1.266E-05<br>0<br>0<br>0<br>1.18E-05<br>1.18E-05<br>1.18E-05  | 0.0443322<br>0.044017<br>0.0443323<br>0.044017<br>0.0584313<br>0.0584313<br>0.0584313<br>0.0588211<br>0.0588211<br>0.0538211<br>0.0496292<br>0.0507781<br>0.0496292<br>0.062794<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0   | 2.812F.08 4<br>8.014E.08 4<br>8.014E.08 4<br>8.014E.08 4<br>2.437F.08 4<br>2.336EF.08 5<br>2.336EF.08 5<br>2.336EF.08 5<br>2.351E.08 4<br>2.837E.08 4<br>2.832F.08 4<br>2.832F.08 4<br>2.832F.08 4<br>3.009F.08 4<br>2.8353E.08 4<br>3.009F.08 4<br>0<br>0<br>0.014E.09 5<br>7.014E.09 5<br>3.775E.10 5<br>5.775E.10 5<br>5.775E.10 5<br>5.775E.10 5<br>5.775E.10 5<br>5.775E.10 5<br>5.775E.10 5<br>5.775E.10 5<br>5.775E.10 5<br>5.775E.10 5<br>5.775E.100  |
1.584E-05<br>1.897E-05<br>1.057E-05<br>1.072E-05<br>1.077E-05<br>1.077E-05<br>1.331E-05<br>1.331E-05<br>1.307E-05<br>1.392E-05<br>1.392E-05<br>1.636E-05<br>1.636E-05<br>1.636E-05<br>1.282E-05<br>1.282E-05<br>1.282E-05<br>1.255E-05   |
| Sheung Tsuen B/T and Car Park<br>Sheung Tsuen B/T and Car Park<br>Village Access Road<br>Village Access Road  | RD_29<br>RD_29<br>RD_29<br>RD_29<br>RD_29<br>RD_29<br>RD_29<br>RD_29<br>RD_29<br>RD_29<br>RD_29<br>RD_29<br>RD_29<br>RD_29<br>RD_29<br>RD_29<br>RD_29<br>RD_29<br>RD_30_31<br>RD_30_31<br>RD_30_31<br>RD_30_31<br>RD_30_31<br>RD_30_31  | 9900-1000<br>1000-1100<br>1100-1200<br>1200-1300<br>1300-1400<br>1400-1500<br>1500-1600<br>1500-1600<br>1700-1800<br>1800-1900<br>2000-2100<br>2000-2100<br>2100-2200<br>2200-2300<br>2300-0000<br>000-0100<br>0100-0200<br>0300-0400<br>0500-0600<br>0500-0600<br>0700-0800  | 0.1281945<br>0.1360751<br>0.1360751<br>0.1557266<br>0.145988<br>0.145988<br>0.1454652<br>0.125268<br>0.1332075<br>0.1332075<br>0.1332075<br>0.1332075<br>0.1352397<br>0.1352397<br>0.1364332<br>0.030941<br>0<br>0<br>0.5292911<br>0.05292911<br>0.05292911<br>0.05292911<br>0.05292911<br>0.05292911<br>0.05292911<br>0.05292911<br>0.05292911<br>0.05292911<br>0.05292911<br>0.05292911<br>0.05292911<br>0.05292911<br>0.05292911<br>0.05292911<br>0.05292911<br>0.05292911<br>0.05292911<br>0.05292911<br>0.05292911<br>0.05292911<br>0.05292911<br>0.05292911<br>0.05292911<br>0.05292911<br>0.05292911<br>0.05292911<br>0.05292911<br>0.05292911<br>0.05292911<br>0.05292911<br>0.05292911<br>0.05292911<br>0.05292911<br>0.05292911<br>0.05292911<br>0.05292911<br>0.05292911<br>0.0529291<br>0.0529291<br>0.0529291<br>0.0529291<br>0.0529291<br>0.0529291<br>0.0529291<br>0.0529291<br>0.0529291<br>0.0529291<br>0.0529291<br>0.0529291<br>0.0529291<br>0.0529291<br>0.0529291<br>0.0529291<br>0.0529291<br>0.0529291<br>0.0529291<br>0.0529291<br>0.0529291<br>0.0529291<br>0.0529291<br>0.0529291<br>0.0529291<br>0.0529291<br>0.0529291<br>0.0529291<br>0.0529291<br>0.0529291<br>0.0529291<br>0.0529291<br>0.0529291<br>0.0529291<br>0.0529291<br>0.0529291<br>0.0529291<br>0.0529291<br>0.0529291<br>0.0529291<br>0.0529291<br>0.0529291<br>0.0529291<br>0.0529291<br>0.0529291<br>0.0529291<br>0.0529291<br>0.0529291<br>0.0529291<br>0.0529291<br>0.0529291<br>0.0529291<br>0.0529291<br>0.0529291<br>0.0529291<br>0.0529291<br>0.0529291<br>0.0529291<br>0.0529291<br>0.0529291<br>0.0529291<br>0.0529291<br>0.0529291<br>0.0529291<br>0.0529291<br>0.0529291<br>0.0529291<br>0.0529291<br>0.0529291<br>0.0529291<br>0.0529291<br>0.0529291<br>0.0529291<br>0.0529291<br>0.0529291<br>0.0529291<br>0.0529291<br>0.0529291<br>0.0529291<br>0.0529291<br>0.0529291<br>0.0529291<br>0.0529291<br>0.0529291<br>0.0529291<br>0.0529291<br>0.0529291<br>0.0529291<br>0.0529291<br>0.0529291<br>0.0529291<br>0.0529291<br>0.0529291<br>0.0529291<br>0.0529291<br>0.0529291<br>0.0529291<br>0.0529291<br>0.0529291<br>0.0529291<br>0.0529291<br>0.0529291<br>0.0529291<br>0.0529291<br>0.0529291<br>0.0529291<br>0.0529291<br>0.0529291<br>0.0529291<br>0.0529291<br>0.0529291<br>0.0529291<br>0.0529291<br>0.0529291<br>0.0529291<br>0.0529291<br>0.0529291<br>0.0529291<br>0.0529291<br>0.0529291<br>0.0529291<br>0.0529291<br>0.0529291<br>0.0529291<br>0.0529291<br>0.0529291<br>0.0529291<br>0.0529291<br>0.0529291<br>0.0529291<br>0.0529291<br>0.0529291<br>0.0529291<br>0.0529291000000000000000000000000000000000   | 9.347-006  <br>9.347-006  <br>9.347-008  <br>8.7236-08  <br>1.036-07 (<br>6.6555-08 (<br>7.7995-08 (<br>6.3357-08 (<br>7.7712-08 (<br>8.1518-08 (<br>7.7161-08 (<br>6.3357-08 (<br>7.1616-08 (<br>6.3357-08 (<br>7.0697-08 (<br>4.3356-08 (<br>0 0 (<br>0 0 (<br>8.0366-08 0)<br>8.0366-08 0)<br>8.0366-08 0)<br>8.0366-08 0)<br>5.6627-08 (<br>6.7738-08 0)<br>5.6627-08 (<br>7.738-08 0)<br>5.6627-08 (<br>7.748-08 0)<br>5.6627-08 (<br>7.748-08 0)<br>5.6627-08 (<br>7.748-08 0)<br>5.6627-08 (<br>7.748-08 0)<br>5.6627-08 (<br>7.748-08 0)<br>5.6627-08 (<br>7.748-08 0)<br>5.778-08 (<br>7.748-08 0)<br>5.778-08 (<br>7.748-08 0)<br>5.   | 0.0001615 (<br>0.0001358 (<br>0.0001358 (<br>0.0001518 (<br>0.0001518 (<br>0.0001674 (<br>0.000162 (<br>0.000   |
0.1261103<br>0.1210641<br>0.1385867<br>0.1291377<br>0.1521578<br>0.1491352<br>0.1399604<br>0.1491352<br>0.1399604<br>0.1429175<br>0.1396503<br>0.1267454<br>0.1267454<br>0.1263454<br>0.1263454<br>0.1263455<br>0.1263455<br>0.0263456<br>0.4784853<br>0.0261416<br>0.1201416<br>0.0261416<br>0.0261416<br>0.0261416<br>0.0261416<br>0.0261416<br>0.0261416<br>0.0261416<br>0.0261416<br>0.0261416<br>0.0261416<br>0.0261416<br>0.0261416<br>0.0261416<br>0.0261416<br>0.0261416<br>0.0261416<br>0.0261416<br>0.0261416<br>0.0261416<br>0.0261416<br>0.0261416<br>0.0261416<br>0.0261416<br>0.0261416<br>0.0261416<br>0.0261416<br>0.0261416<br>0.0261416<br>0.0261416<br>0.0261416<br>0.0261416<br>0.0261416<br>0.0261416<br>0.0261416<br>0.0261416<br>0.0261416<br>0.0261416<br>0.0261416<br>0.0261416<br>0.0261416<br>0.0261416<br>0.0261416<br>0.0261416<br>0.0261416<br>0.0261416<br>0.0261416<br>0.0261416<br>0.0261416<br>0.0261416<br>0.0261416<br>0.0261416<br>0.0261416<br>0.0261416<br>0.0261416<br>0.0261416<br>0.0261416<br>0.0261416<br>0.0261416<br>0.0261416<br>0.0261416<br>0.0261416<br>0.0261416<br>0.0261416<br>0.0261416<br>0.0261416<br>0.0261416<br>0.0261416<br>0.0261416<br>0.0261416<br>0.0261416<br>0.0261416<br>0.0261416<br>0.0261416<br>0.0261416<br>0.0261416<br>0.0261416<br>0.0261416<br>0.0261416<br>0.0261416<br>0.0261416<br>0.0261416<br>0.0261416<br>0.0261416<br>0.0261416<br>0.0261416<br>0.0261416<br>0.0261416<br>0.0261416<br>0.0261416<br>0.0261416<br>0.0261416<br>0.0261416<br>0.0261416<br>0.0261416<br>0.0261416<br>0.0261416<br>0.0261416<br>0.0261416<br>0.0261416<br>0.0261416<br>0.0261416<br>0.0261416<br>0.0261416<br>0.0261416<br>0.0261416<br>0.0261416<br>0.0261416<br>0.0261416<br>0.0261416<br>0.0261416<br>0.0261416<br>0.0261416<br>0.0261416<br>0.0261416<br>0.0261416<br>0.0261416<br>0.0261416<br>0.0261416<br>0.0261416<br>0.0261416<br>0.0261416<br>0.0261416<br>0.0261416<br>0.0261416<br>0.0261416<br>0.0261416<br>0.0261416<br>0.0261416<br>0.0261416<br>0.0261416<br>0.0261416<br>0.0261416<br>0.0261416<br>0.0261416<br>0.0261416<br>0.0261416<br>0.0261416<br>0.0261416<br>0.0261416<br>0.0261416<br>0.0261416<br>0.0261416<br>0.0261416<br>0.0261416<br>0.0261416<br>0.0261416<br>0.0261416<br>0.0261416<br>0.0261416<br>0.0261416<br>0.0261416<br>0.0261416<br>0.0261416<br>0.0261416<br>0.0261416<br>0.0261416<br>0.0261416<br>0.0261416<br>0.0261416<br>0.0261416<br>0.026145<br>0.026145<br>0.026145<br>0.026145<br>0.026145<br>0.026   | 9,431E-08<br>7,761E-08<br>8,284E-08<br>9,754E-08<br>6,373E-08<br>6,373E-08<br>6,632E-08<br>7,473E-08<br>6,632E-08<br>7,46E-08<br>7,46E-08<br>7,46E-08<br>7,46E-08<br>7,46E-08<br>7,46E-08<br>7,46E-08<br>7,46E-08<br>7,46E-08<br>7,46E-08<br>7,46E-08<br>7,46E-08<br>7,46E-08<br>7,46E-08<br>7,46E-08<br>7,46E-08<br>7,46E-08<br>7,46E-08<br>7,46E-08<br>7,46E-08<br>7,46E-08<br>7,46E-08<br>7,46E-08<br>7,46E-08<br>7,46E-08<br>7,46E-08<br>7,46E-08<br>7,46E-08<br>7,46E-08<br>7,46E-08<br>7,46E-08<br>7,46E-08<br>7,46E-08<br>7,46E-08<br>7,46E-08<br>7,46E-08<br>7,46E-08<br>7,46E-08<br>7,46E-08<br>7,46E-08<br>7,46E-08<br>7,46E-08<br>7,46E-08<br>7,46E-08<br>7,46E-08<br>7,46E-08<br>7,46E-08<br>7,46E-08<br>7,46E-08<br>7,46E-08<br>7,46E-08<br>7,46E-08<br>7,46E-08<br>7,46E-08<br>7,46E-08<br>7,46E-08<br>7,46E-08<br>7,46E-08<br>7,46E-08<br>7,46E-08<br>7,46E-08<br>7,46E-08<br>7,473E-08<br>6,65E-08<br>5,477E-08<br>7,473E-08<br>7,473E-08<br>7,473E-08<br>7,473E-08<br>7,473E-08<br>7,473E-08<br>7,473E-08<br>7,473E-08<br>7,473E-08<br>7,473E-08<br>7,473E-08<br>7,473E-08<br>7,473E-08<br>7,473E-08<br>7,473E-08<br>7,473E-08<br>7,473E-08<br>7,473E-08<br>7,473E-08<br>7,473E-08<br>7,473E-08<br>7,473E-08<br>7,473E-08<br>7,473E-08<br>7,473E-08<br>7,473E-08<br>7,473E-08<br>7,473E-08<br>7,473E-08<br>7,473E-08<br>7,473E-08<br>7,473E-08<br>7,473E-08<br>7,473E-08<br>7,474E-08<br>7,473E-08<br>7,473E-08<br>7,473E-08<br>7,473E-08<br>7,473E-08<br>7,473E-08<br>7,473E-08<br>7,473E-08<br>7,473E-08<br>7,473E-08<br>7,473E-08<br>7,473E-08<br>7,474E-08<br>7,474E-08<br>7,474E-08<br>7,474E-08<br>7,474E-08<br>7,474E-08<br>7,474E-08<br>7,474E-08<br>7,474E-08<br>7,474E-08<br>7,474E-08<br>7,474E-08<br>7,474E-08<br>7,474E-08<br>7,474E-08<br>7,474E-08<br>7,474E-08<br>7,474E-08<br>7,474E-08<br>7,474E-08<br>7,474E-08<br>7,474E-08<br>7,474E-08<br>7,474E-08<br>7,474E-08<br>7,474E-08<br>7,474E-08<br>7,474E-08<br>7,474E-08<br>7,474E-08<br>7,474E-08<br>7,474E-08<br>7,474E-08<br>7,474E-08<br>7,474E-08<br>7,474E-08<br>7,474E-08<br>7,474E-08<br>7,474E-08<br>7,474E-08<br>7,474E-08<br>7,474E-08<br>7,474E-08<br>7,474E-08<br>7,474E-08<br>7,474E-08<br>7,474E-08<br>7,474E-08<br>7,474E-08<br>7,474E-08<br>7,474E-08<br>7,474E-08<br>7,474E-08<br>7,474E-08<br>7,474E-08<br>7,474E-08<br>7,474E-08<br>7,474E-08<br>7,474E-08<br>7,474E-08<br>7,474E-08<br>7,474E-08<br>7,474E-08<br>7,474E-08<br>7,474E-08<br>7,474E-08<br>7,474E-08<br>7,474E-08<br>7,474E-08<br>7,474E-08<br>7,474E-08<br>7,474E-08<br>7,474E-08<br>7,474E-08<br>7,474E-08<br>7,474E-08<br>7,474E-08<br>7,474E-08<br>7,474E-08<br>7,474E-08<br>7,474E-08<br>7,474E-08<br>7,474E-08<br>7,474E-08<br>7,474E-08<br>7,474E-08<br>7,474E-08<br>7,474E-08<br>7,474E-08<br>7,474E-08 | 0.0001532<br>0.0001532<br>0.0001261<br>0.0001435<br>0.0001435<br>0.0001345<br>0.0001345<br>0.0001214<br>9.882E-05<br>0.0001122<br>0.0001122<br>0.0001128<br>0.0001105<br>6.689E-05<br>0.689E-05<br>0.0001132<br>0.0001132<br>0.0001328<br>0.0001328   | 0.1203596<br>0.1142306<br>0.1310694<br>0.1216391<br>0.14263721<br>0.1426376<br>0.13221766<br>0.1323766<br>0.1323766<br>0.13245164<br>0.1320173<br>0.1320572<br>0.1203562<br>0.1229456<br>0.1624807<br>0.02148766<br>0.1624807<br>0<br>0.2148766<br>0.1694062<br>0.022395<br>0<br>0.44328049<br>0.022575<br>0.1585645<br>0.1585933  
   | 9.0011-08<br>7.322E-08<br>8.402E-08<br>7.797E-08<br>9.146E-08<br>6.053E-08<br>7.072E-08<br>5.749E-08<br>6.439E-08<br>6.439E-08<br>6.435E-08<br>6.435E-08<br>6.435E-08<br>6.435E-08<br>6.435E-08<br>6.435E-08<br>6.435E-08<br>6.435E-08<br>6.435E-08<br>6.435E-08<br>6.738E-08<br>6.738E-08<br>6.738E-08<br>6.738E-08<br>6.738E-08   | 0.0001462<br>0.0001462<br>0.0001365<br>0.0001267<br>0.0001486<br>9.834F-05<br>0.0001148<br>9.34F-05<br>0.0001149<br>9.34F-05<br>0.0001146<br>0.0001201<br>0.0001032<br>0.0001049<br>6.402E-05<br>6.214F-07<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0  | 2.4904488<br>1.9842855<br>2.2002208<br>2.1072468<br>2.207468<br>2.2364653<br>3.1057198<br>3.0468346<br>2.9569573<br>2.8602462<br>3.1731567<br>2.620547<br>2.9117616<br>4.1828878<br>4.9637927<br>4.4562831<br>0.0573728<br>0<br>0<br>1.3610254<br>1.3610254<br>0.6039652<br>0.6039652<br>0.6039652<br>0.6039652   |
1.8638-06<br>1.2722-06<br>1.411-06<br>1.3112-06<br>1.434E-06<br>1.3272-06<br>1.628E-06<br>1.264E-06<br>1.264E-06<br>1.401E-06<br>1.440E-06<br>1.440E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.     | 0.030261<br>0.002265<br>0.0022915<br>0.0021953<br>0.0021564<br>0.002549<br>0.002544<br>0.002531<br>0.002544<br>0.002757<br>0.0021782<br>0.0021782<br>0.0021782<br>0.0023497<br>0.0021782<br>0.0023497<br>0.0021782<br>0.0023497<br>0.0003777<br>1.681E-05<br>0.0003377  | 2.3566842<br>1.8789238<br>2.0977776<br>2.00527501<br>2.1283629<br>2.9552749<br>2.900207<br>2.8217082<br>2.7301698<br>3.0191874<br>2.4772701<br>2.5033689<br>2.5607825<br>2.7392982<br>3.9311649<br>4.68371<br>4.1688413<br>0.0463027<br>0<br>0<br>1.2345153<br>0.5415078<br>0.45415074<br>0.5415078<br>0.5415078<br>0.5415078<br>0.5415078<br>0.5415078<br>0.5415078<br>0.5415078<br>0.5415078<br>0.5415078<br>0.5415078<br>0.5415078<br>0.5415078<br>0.5415078<br>0.5415078<br>0.5415078<br>0.5415078<br>0.5415078<br>0.5415078<br>0.5415078<br>0.5415078<br>0.5415078<br>0.5415078<br>0.5415078<br>0.5415078<br>0.5415078<br>0.5415078<br>0.5415078<br>0.5415078<br>0.5415078<br>0.5415078<br>0.5415078<br>0.5415078<br>0.5415078<br>0.5415078<br>0.5415078<br>0.5415078<br>0.5415078<br>0.5415078<br>0.5415078<br>0.5415078<br>0.5415078<br>0.5415078<br>0.5415078<br>0.5415078<br>0.5415078<br>0.5415078<br>0.5415078<br>0.5415078<br>0.5415078<br>0.5415078<br>0.5415078<br>0.5415078<br>0.5415078<br>0.5415078<br>0.5415078<br>0.5415078<br>0.5415078<br>0.5415078<br>0.5415078<br>0.5415078<br>0.5415078<br>0.5415078<br>0.5415078<br>0.5415078<br>0.5415078<br>0.5415078<br>0.5415078<br>0.5415078<br>0.5415078<br>0.5415078<br>0.5415078<br>0.5415078<br>0.5415078<br>0.5415078<br>0.5415078<br>0.5415078<br>0.5415078<br>0.5415078<br>0.5415078<br>0.5415078<br>0.5415078<br>0.5415078<br>0.5415078<br>0.5415078<br>0.5415078<br>0.5415078<br>0.5415078<br>0.5415078<br>0.5415078<br>0.5415078<br>0.5415078<br>0.5415078<br>0.5415078<br>0.5415078<br>0.5415078<br>0.5415078<br>0.5415078<br>0.5415078<br>0.5415078<br>0.5415078<br>0.5415078<br>0.5415078<br>0.5415078<br>0.5415078<br>0.5415078<br>0.5415078<br>0.5415078<br>0.5415078<br>0.5415078<br>0.5415078<br>0.5415078<br>0.5415078<br>0.5415078<br>0.5415078<br>0.5415078<br>0.5415078<br>0.5415078<br>0.5415078<br>0.5415078<br>0.5415078<br>0.5415078<br>0.5415078<br>0.5415078<br>0.5415078<br>0.5415078<br>0.5415078<br>0.5415078<br>0.5415078<br>0.5415078<br>0.5415078<br>0.5415078<br>0.5415078<br>0.5415078<br>0.5415078<br>0.5415078<br>0.5415078<br>0.5415078<br>0.5415078<br>0.5415078<br>0.5415078<br>0.5415078<br>0.5415078<br>0.5415078<br>0.5415078<br>0.5415078<br>0.5415078<br>0.5415078<br>0.5415078<br>0.5415078<br>0.5415078<br>0.5415078<br>0.5415078<br>0.5415078<br>0.5415078<br>0.5415078<br>0.5415078<br>0.5415078<br>0.5415078<br>0.5415078<br>0.5415078<br>0.5415078<br>0.5415078<br>0.5415078<br>0.5415078<br>0.5415   | 1.7622-06<br>1.204E-06<br>1.345E-06<br>1.345E-06<br>1.364E-06<br>1.549E-06<br>1.549E-06<br>1.549E-06<br>1.532E-06<br>1.332E-06<br>1.332E-06<br>1.33E-06<br>1.33E-06<br>1.34E-06<br>1.36E-06<br>1.36E-06<br>1.36E-06<br>1.36E-06<br>1.36E-06<br>1.36E-07<br>7.142E-09<br>1.644E-07<br>1.644E-07<br>1.644E-07<br>1.644E-07<br>1.644E-07<br>1.644E-07<br>1.644E-07<br>1.644E-07<br>1.644E-07<br>1.644E-07<br>1.644E-07<br>1.644E-07<br>1.644E-07<br>1.644E-07<br>1.644E-07<br>1.644E-07<br>1.644E-07<br>1.644E-07<br>1.644E-07<br>1.644E-07<br>1.644E-07<br>1.644E-07<br>1.644E-07<br>1.644E-07<br>1.644E-07<br>1.644E-07<br>1.644E-07<br>1.644E-07<br>1.644E-07<br>1.644E-07<br>1.644E-07<br>1.644E-07<br>1.644E-07<br>1.644E-07<br>1.644E-07<br>1.644E-07<br>1.644E-07<br>1.644E-07<br>1.644E-07<br>1.644E-07<br>1.644E-07<br>1.644E-07<br>1.644E-07<br>1.644E-07<br>1.644E-07<br>1.644E-07<br>1.644E-07<br>1.644E-07<br>1.644E-07<br>1.644E-07<br>1.644E-07<br>1.644E-07<br>1.644E-07<br>1.644E-07<br>1.644E-07<br>1.644E-07<br>1.644E-07<br>1.644E-07<br>1.644E-07<br>1.644E-07<br>1.644E-07<br>1.644E-07<br>1.644E-07<br>1.644E-07<br>1.644E-07<br>1.644E-07<br>1.644E-07<br>1.644E-07<br>1.644E-07<br>1.644E-07<br>1.644E-07<br>1.644E-07<br>1.644E-07<br>1.644E-07<br>1.644E-07<br>1.644E-07<br>1.644E-07<br>1.644E-07<br>1.644E-07<br>1.644E-07<br>1.644E-07<br>1.644E-07<br>1.644E-07<br>1.644E-07<br>1.644E-07<br>1.644E-07<br>1.644E-07<br>1.644E-07<br>1.644E-07<br>1.644E-07<br>1.644E-07<br>1.644E-07<br>1.644E-07<br>1.644E-07<br>1.644E-07<br>1.644E-07<br>1.644E-07<br>1.644E-07<br>1.644E-07<br>1.644E-07<br>1.644E-07<br>1.644E-07<br>1.644E-07<br>1.644E-07<br>1.644E-07<br>1.644E-07<br>1.644E-07<br>1.644E-07<br>1.644E-07<br>1.644E-07<br>1.644E-07<br>1.644E-07<br>1.644E-07<br>1.644E-07<br>1.644E-07<br>1.644E-07<br>1.644E-07<br>1.644E-07<br>1.644E-07<br>1.644E-07<br>1.644E-07<br>1.644E-07<br>1.644E-07<br>1.644E-07<br>1.644E-07<br>1.644E-07<br>1.644E-07<br>1.644E-07<br>1.644E-07<br>1.644E-07<br>1.644E-07<br>1.644E-07<br>1.644E-07<br>1.644E-07<br>1.644E-07<br>1.644E-07<br>1.644E-07<br>1.644E-07<br>1.644E-07<br>1.644E-07<br>1.644E-07<br>1.644E-07<br>1.644E-07<br>1.644E-07<br>1.644E-07<br>1.644E-07<br>1.644E-07<br>1.644E-07<br>1.644E-07<br>1.644E-07<br>1.644E-07<br>1.644E-07<br>1.644E-07<br>1.644E-07<br>1.644E-07<br>1.644E-07<br>1.644E-07<br>1.644E-07<br>1.644E-07<br>1.644E-07<br>1.644E-07<br>1.644E-07<br>1.644E-07<br>1.644E-07<br>1.644E-07<br>1.644E-07                                     | 0.0028636<br>0.0019569<br>0.0021848<br>0.0020885<br>0.00220519<br>0.0025171<br>0.0025171<br>0.00252171<br>0.0025204<br>0.00226204<br>0.0022204<br>0.0021200<br>0.0022205<br>0.0024205<br>0.0022471<br>0.00224709<br>1.285E-05<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0   
  | 2.2440968<br>1.7795138<br>1.9915216<br>2.8961342<br>2.0102391<br>2.7864095<br>2.7220732<br>2.6462899<br>2.5622851<br>2.8364013<br>2.3321589<br>2.3725056<br>2.3321589<br>2.3725056<br>2.4375101<br>2.6016999<br>3.7301495<br>4.4767836<br>0<br>0<br>1.1398552<br>1.1413822<br>0.0497767<br>1.1413822<br>0.0497767<br>0.42876418   | 1.678E-06<br>1.141E-06<br>1.277E-06<br>1.215E-06<br>1.215E-06<br>1.245E-06<br>1.131E-06<br>1.345E-06<br>1.345E-06<br>1.245E-06<br>1.245E-06<br>1.245E-06<br>1.245E-06<br>1.245E-06<br>1.245E-06<br>1.245E-06<br>1.245E-06<br>1.245E-06<br>1.245E-06<br>1.245E-06<br>1.245E-06<br>1.245E-06<br>1.245E-06<br>1.245E-06<br>1.245E-06<br>1.245E-06<br>1.245E-06<br>1.245E-06<br>1.245E-06<br>1.245E-06<br>1.245E-06<br>1.245E-06<br>1.245E-06<br>1.245E-06<br>1.245E-06<br>1.245E-06<br>1.245E-06<br>1.245E-06<br>1.245E-06<br>1.245E-06<br>1.245E-06<br>1.245E-06<br>1.245E-06<br>1.245E-06<br>1.245E-06<br>1.245E-06<br>1.245E-06<br>1.245E-06<br>1.245E-06<br>1.245E-06<br>1.245E-06<br>1.245E-06<br>1.245E-06<br>1.245E-06<br>1.245E-06<br>1.245E-06<br>1.245E-06<br>1.245E-06<br>1.245E-06<br>1.245E-06<br>1.245E-06<br>1.245E-06<br>1.245E-06<br>1.245E-06<br>1.245E-06<br>1.245E-06<br>1.245E-06<br>1.245E-06<br>1.245E-06<br>1.245E-06<br>1.245E-06<br>1.245E-06<br>1.245E-06<br>1.245E-06<br>1.245E-06<br>1.245E-06<br>1.245E-06<br>1.245E-06<br>1.245E-06<br>1.245E-06<br>1.245E-06<br>1.245E-06<br>1.245E-06<br>1.245E-06<br>1.245E-06<br>1.245E-06<br>1.245E-06<br>1.245E-06<br>1.245E-06<br>1.245E-06<br>1.245E-06<br>1.245E-06<br>1.245E-06<br>1.245E-06<br>1.245E-06<br>1.245E-06<br>1.245E-06<br>1.245E-06<br>1.245E-06<br>1.245E-06<br>1.245E-06<br>1.245E-06<br>1.245E-06<br>1.245E-06<br>1.245E-06<br>1.245E-06<br>1.245E-06<br>1.245E-06<br>1.245E-06<br>1.245E-06<br>1.245E-06<br>1.245E-07<br>1.245E-07<br>1.245E-07<br>1.245E-07<br>1.245E-07<br>1.245E-07<br>1.245E-07<br>1.245E-07<br>1.245E-07<br>1.245E-07<br>1.245E-07<br>1.245E-07<br>1.245E-07<br>1.245E-07<br>1.245E-07<br>1.245E-07<br>1.245E-07<br>1.245E-07<br>1.245E-07<br>1.245E-07<br>1.245E-07<br>1.245E-07<br>1.245E-07<br>1.245E-07<br>1.245E-07<br>1.245E-07<br>1.245E-07<br>1.245E-07<br>1.245E-07<br>1.245E-07<br>1.245E-07<br>1.245E-07<br>1.245E-07<br>1.245E-07<br>1.245E-07<br>1.245E-07<br>1.245E-07<br>1.245E-07<br>1.245E-07<br>1.245E-07<br>1.245E-07<br>1.245E-07<br>1.245E-07<br>1.245E-07<br>1.245E-07<br>1.245E-07<br>1.245E-07<br>1.245E-07<br>1.245E-07<br>1.245E-07<br>1.245E-07<br>1.245E-07<br>1.245E-07<br>1.245E-07<br>1.245E-07<br>1.245E-07<br>1.245E-07<br>1.245E-07<br>1.245E-07<br>1.245E-07<br>1.245E-07<br>1.245E-07<br>1.245E-07<br>1.245E-07<br>1.245E-07<br>1.245E-07<br>1.245E-07<br>1.245E-07<br>1.245E-07<br>1.245E-07<br>1.245E-07<br>1.245E-07<br>1.245E-07<br>1.245E-07<br>1.245E-07  | 0.0027267<br>0.0018534<br>0.0020742<br>0.0019748<br>0.0020974<br>0.0019748<br>0.0020974<br>0.0019748<br>0.0020874<br>0.002362<br>0.002361<br>0.002241<br>0.002241<br>0.0020241<br>0.0020241<br>0.0020241<br>0.0020241<br>0.0020341<br>0.0020341<br>0.0003163<br>1.14565<br>0.0002765<br>0.00002765<br>0.00002765<br>0.00002765<br>0.00002765  | 2.623364<br>2.11248<br>2.3459273<br>2.2439219<br>3.2614464<br>3.1051925<br>3.0057113<br>3.3051925<br>3.0057113<br>3.3257455<br>2.7818703<br>2.3836521<br>3.0470013<br>3.3621382<br>3.1979947<br>1.6427163<br>0.0604669<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0   
   | 1.962E-06<br>1.354E-06<br>1.554E-06<br>1.537E-06<br>1.537E-06<br>1.537E-06<br>1.705E-06<br>1.327E-06<br>1.428E-06<br>1.77F-06<br>1.426E-06<br>1.517E-06<br>1.488E-06<br>1.517E-06<br>1.488E-06<br>1.488E-06<br>1.488E-06<br>1.488E-06<br>1.488E-06<br>1.488E-06<br>1.488E-06<br>1.287E-07<br>9.695E-09<br>2.511E-07<br>2.251E-07   | 0.0031876<br>0.0022001<br>0.002443<br>0.002337<br>0.002347<br>0.0022645<br>0.0022645<br>0.0022645<br>0.0022645<br>0.0022645<br>0.0022687<br>0.0023865<br>0.0024865<br>0.0024865<br>0.0024177<br>1.678E-05<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0  | 2.3644564<br>1.8937445<br>2.1225909<br>2.0177733<br>2.1529112<br>2.928047<br>2.8544498<br>2.7808064<br>2.6943024<br>2.9747351<br>2.4523139<br>2.4523139<br>2.4523139<br>2.4523139<br>2.4553363<br>2.724646<br>3.8926302<br>4.16951603<br>4.1449386<br>0<br>0<br>0<br>1.5831454<br>1.583267<br>0.0433825<br>0.06846321<br>0.580351   
  | 1.7682-00         0.002           1.2446-00         0.001           1.351-00         0.002           1.351-00         0.002           1.381-00         0.002           1.381-00         0.002           1.381-00         0.002           1.381-00         0.002           1.382-00         0.002           1.525-00         0.002           1.318-00         0.002           1.318-00         0.002           1.318-00         0.002           1.318-00         0.002           1.318-00         0.002           1.318-00         0.002           1.318-00         0.002           1.318-00         0.002           1.318-00         0.002           1.318-00         0.002           1.318-00         0.002           0         0         0           0         0         0           0         0         0           0.2404-07         0.002         2.4047-07           0.02404-07         0.002         1.021-07           0.02404-07         0.002         1.021-07   | 873         0.04           9723         0.03           1010         0.044           2422         0.03           0.033         0.05           3384         0.044           5818         0.05           3384         0.044           2204         0.044           2204         0.044           2204         0.044           2204         0.044           2204         0.044           2204         0.044           2204         0.044           2204         0.044           2204         0.044           2205         0.044           1585         0.056           -0.00   
  | 147043         3:           383832         2:           383832         2:           405571         2:           313125         2:           338011         2:           338012         2:           338013         2:           338014         2:           338015         2:           313125         2:           14016         2:           150704         2:           166767         2:           186276         2:           153993         8:           0         0           0         14252           0         14252           0         14252           0         14252           0         14252           0         14252           0         3:3915           0         3:3915   
   | 5.18E-08         2           6.619E-08         4           2.6E-08         4           2.96E-08         4           2.99E-08         3           8.74E-08         3           8.74E-08         3           6.51E-08         4           0.021E-08         4           4.42E-08         3           4.42E-08         3           4.55E-08         4           651E-08         4           651E-08         4           651E-08         4           652E-08         4           652E-08         4           654E-09         1           0         0           0         4           456E-09         1           198E-10         1           384E-08         1           384E-09         1           384E-09         1           384E-09         1           384E-09         1           384E-09         1   | 3.998E-05<br>4.254E-05<br>4.224E-05<br>4.511E-05<br>3.736E-05<br>4.67E-05<br>4.307E-05<br>4.307E-05<br>4.098E-05<br>3.967E-05<br>4.098E-05<br>3.967E-05<br>4.059E-05<br>1.699E-05<br>1.266E-05<br>1.126E-05<br>1.18E-05<br>1.18E-05<br>1.18E-05<br>1.158E-05<br>1.158E-05  | 0.0443322<br>0.044017<br>0.0443323<br>0.044017<br>0.0470156<br>0.0584313<br>0.0584313<br>0.0584373<br>0.0588211<br>0.0538211<br>0.0538211<br>0.0638242<br>0.0607781<br>0.0632938<br>0.0632938<br>0.0938934<br>0.0938934<br>0.0938934<br>0.0938934<br>0.0938934<br>0.0938934<br>0.0938934<br>0.0938934<br>0.0938934<br>0.0938934<br>0.0938934<br>0.0938934<br>0.0938934<br>0.0938934<br>0.0938934<br>0.0938934<br>0.0938934<br>0.00462<br>0.04622<br>0.04622<br>0.045225<br>0.04525<br>0.04525<br>0.04525<br>0.04525<br>0.04525<br>0.04525<br>0.04525<br>0.04525<br>0.04525<br>0.04525<br>0.04525<br>0.04525<br>0.0558211<br>0.0558211<br>0.0558211<br>0.0558211<br>0.0558211<br>0.0558211<br>0.0558211<br>0.0558211<br>0.0558211<br>0.0558211<br>0.0558211<br>0.0558211<br>0.0558211<br>0.0558211<br>0.0558211<br>0.0558211<br>0.0558211<br>0.0558211<br>0.0558211<br>0.0558211<br>0.0558211<br>0.0558211<br>0.0558211<br>0.0558211<br>0.0558211<br>0.0558211<br>0.0558211<br>0.0558211<br>0.0558211<br>0.0558211<br>0.0558211<br>0.0558211<br>0.0558211<br>0.0558211<br>0.0558211<br>0.0558211<br>0.0558211<br>0.0558211<br>0.0558211<br>0.0558211<br>0.0558211<br>0.0558211<br>0.0558211<br>0.0558211<br>0.0558211<br>0.0558211<br>0.0558211<br>0.0558211<br>0.0558211<br>0.0558211<br>0.0558211<br>0.0558211<br>0.0558211<br>0.0558211<br>0.0558211<br>0.0558211<br>0.0558211<br>0.057941<br>0.0529242<br>0.0000000000000000000000000000000000   
  | 2.071-00         2.8272-00         2.8275-00         2.8275-00         2.8375-00         2.8375-00         2.8375-00         2.9375-10         2.8375-00         2.8375-00         2.8375-00         2.8375-01         2.83675-00         2.85675-00         2.6512-00         2.6512-10         <   | 1.584E-05<br>1.897E-05<br>1.057E-05<br>1.072E-05<br>1.072E-05<br>1.077E-05<br>1.331E-05<br>1.331E-05<br>1.307E-05<br>1.392E-05<br>1.392E-05<br>1.636E-05<br>1.282E-05<br>1.282E-05<br>1.282E-05<br>1.255E-05<br>1.255E-05  |
| Sheung Tsuen B/T and Car Park<br>Sheung Tsuen B/T and Car Park<br>Village Access Road<br>Village Access Road  | RD_29<br>RD_29<br>RD_29<br>RD_29<br>RD_29<br>RD_29<br>RD_29<br>RD_29<br>RD_29<br>RD_29<br>RD_29<br>RD_29<br>RD_29<br>RD_29<br>RD_29<br>RD_29<br>RD_29<br>RD_30_31<br>RD_30_31<br>RD_30_31<br>RD_30_31<br>RD_30_31<br>RD_30_31<br>RD_30_31<br>RD_30_31<br>RD_30_31<br>RD_30_31<br>RD_30_31<br>RD_30_31<br>RD_30_31<br>RD_30_31<br>RD_30_31<br>RD_30_31<br>RD_30_31<br>RD_30_31<br>RD_30_31<br>RD_30_31<br>RD_30_31<br>RD_30_31<br>RD_30_31<br>RD_30_31<br>RD_30_31<br>RD_30_31<br>RD_30_31<br>RD_30_31<br>RD_30_31<br>RD_30_31<br>RD_30_31<br>RD_30_31<br>RD_30_31<br>RD_30_31<br>RD_30_31<br>RD_30_31<br>RD_30_31<br>RD_30_31<br>RD_30_31<br>RD_30_31<br>RD_30_31<br>RD_30_31<br>RD_30_31<br>RD_30_31<br>RD_30_31<br>RD_30_31<br>RD_30_31<br>RD_30_31<br>RD_30_31<br>RD_30_31<br>RD_30_31<br>RD_30_31<br>RD_30_31<br>RD_30_31<br>RD_30_31<br>RD_30_31<br>RD_30_31<br>RD_30_31<br>RD_30_31<br>RD_30_31<br>RD_30_31<br>RD_30_31<br>RD_30_31<br>RD_30_31<br>RD_30_31<br>RD_30_31<br>RD_30_31<br>RD_30_31<br>RD_30_31<br>RD_30_31<br>RD_30_31<br>RD_30_31<br>RD_30_31<br>RD_30_31<br>RD_30_31<br>RD_30_31<br>RD_30_31<br>RD_30_31<br>RD_30_31<br>RD_30_31<br>RD_30_31<br>RD_30_31<br>RD_30_31<br>RD_30_31<br>RD_30_31<br>RD_30_31<br>RD_30_31<br>RD_30_31<br>RD_30_31<br>RD_30_31<br>RD_30_31<br>RD_30_31<br>RD_30_31<br>RD_30_31<br>RD_30_31<br>RD_30_31<br>RD_30_31<br>RD_30_31<br>RD_30_31<br>RD_30_31<br>RD_30_31<br>RD_30_31<br>RD_30_31<br>RD_30_31<br>RD_30_31<br>RD_30_31<br>RD_30_31<br>RD_30_31<br>RD_30_31<br>RD_30_31<br>RD_30_31<br>RD_30_31<br>RD_30_31<br>RD_30_31<br>RD_30_31<br>RD_30_31<br>RD_30_31<br>RD_30_31<br>RD_30_31<br>RD_30_31<br>RD_30_31<br>RD_30_31<br>RD_30_31<br>RD_30_31<br>RD_30_31<br>RD_30_31<br>RD_30_31<br>RD_30_31<br>RD_30_31<br>RD_30_31<br>RD_30_31<br>RD_30_31<br>RD_30_31<br>RD_30_31<br>RD_30_31<br>RD_30_31<br>RD_30_31<br>RD_30_31<br>RD_30_31<br>RD_30_31<br>RD_30_31<br>RD_30_31<br>RD_30_31<br>RD_30_31<br>RD_30_31<br>RD_30_31<br>RD_30_31<br>RD_30_31<br>RD_30_31<br>RD_30_31<br>RD_30_31<br>RD_30_31<br>RD_30_31<br>RD_30_31<br>RD_30_31<br>RD_30_31<br>RD_30_31<br>RD_30_31<br>RD_30_31<br>RD_30_31<br>RD_30_31<br>RD_30_31<br>RD_30_31<br>RD_30_31<br>RD_30_31<br>RD_30_31<br>RD_30_31<br>RD_30_31<br>RD_30_31<br>RD_30_31<br>RD_30_31<br>RD_30_31<br>RD_30_31<br>RD_30_31<br>RD_30_31<br>RD_30_31<br>RD_30_31<br>RD_30_31<br>RD_30_31<br>RD_30_31<br>RD_30_31<br>RD_30_31<br>RD_30_31<br>RD_30_31<br>RD_30_31<br>RD_30_31<br>RD_30_31<br>RD_30_31<br>RD_30_31<br>RD_30_31<br>RD_30_31<br>RD_30_31<br>RD_30_31<br>RD_30_31<br>RD_30_31<br>RD_30_31<br>RD_30_31<br>RD_30_31<br>RD_30_31<br>RD_30_31<br>RD_30_31<br>RD_30_31<br>RD_30_31<br>RD_30_31<br>RD_30_31<br>RD_30_31<br>RD_30_31<br>RD_30_31<br>RD_30_31<br>RD_30_31<br>RD_30_31<br>RD_30_31<br>RD_30_31<br>RD_30_31<br>RD_30_31<br>RD_30_31   | 9900-1000<br>1000-1100<br>1100-1200<br>1200-1300<br>1300-1400<br>1500-1600<br>1500-1600<br>1500-1800<br>1900-2000<br>2000-2100<br>2100-2200<br>2200-2300<br>2200-2300<br>0000-0100<br>0100-0200<br>0300-400<br>0400-0500<br>0500-6600<br>0500-6600<br>0500-6600<br>0500-6600<br>0500-6600<br>0500-6600<br>0500-6600<br>0500-6600<br>0500-6600<br>0500-6600<br>0500-6600<br>0500-6600<br>0500-6600<br>0500-6600<br>0500-6600<br>0500-6600<br>0500-6600<br>0500-6600<br>0500-6600<br>0500-6600<br>0500-6600<br>0500-6600<br>0500-6600<br>0500-6600<br>0500-6600<br>0500-6600<br>0500-6600<br>0500-6600<br>0500-6600<br>0500-6600<br>0500-6600<br>0500-6600<br>0500-6600<br>0500-6600<br>0500-6600<br>0500-6600<br>0500-6600<br>0500-6600<br>0500-6600<br>0500-6600<br>0500-6600<br>0500-6600<br>0500-6600<br>0500-6600<br>0500-6600<br>0500-6600<br>0500-6600<br>0500-6600<br>0500-6600<br>0500-6600<br>0500-6600<br>0500-6600<br>0500-6600<br>0500-6600<br>0500-6600<br>0500-6600<br>0500-6600<br>0500-6600<br>0500-6600<br>0500-6600<br>0500-6600<br>0500-6600<br>0500-6600<br>0500-6600<br>0500-6600<br>0500-6600<br>0500-6600<br>0500-6600<br>0500-6600<br>0500-6600<br>0500-6600<br>0500-6600<br>0500-6600<br>0500-6600<br>0500-6600<br>0500-6600<br>0500-6600<br>0500-6600<br>0500-6600<br>0500-6600<br>0500-6600<br>0500-6600<br>0500-6600<br>0500-6600<br>0500-6600<br>0500-6600<br>0500-6600<br>0500-6600<br>0500-6600<br>0500-6600<br>0500-6600<br>0500-6600<br>0500-6600<br>0500-6600<br>0500-6600<br>0500-6600<br>0500-6600<br>0500-6600<br>0500-6600<br>0500-6600<br>0500-6600<br>0500-6600<br>0500-6600<br>0500-6600<br>0500-6600<br>0500-6600<br>0500-6600<br>0500-6600<br>0500-6600<br>0500-6600<br>0500-6600<br>0500-6600<br>0500-6600<br>0500-6600<br>0500-6600<br>0500-6600<br>0500-6600<br>0500-6600<br>0500-6600<br>0500-6600<br>0500-6600<br>0500-6600<br>0500-6600<br>0500-6600<br>0500-6600<br>0500-6600<br>0500-6600<br>0500-6600<br>0500-6600<br>0500-6600<br>0500-6600<br>0500-6600<br>0500-6600<br>0500-6600<br>0500-6600<br>0500-6600<br>0500-6600<br>0500-6600<br>0500-6600<br>0500-6600<br>0500-6600<br>0500-6600<br>0500-6600<br>0500-6600<br>0500-6600<br>0500-6600<br>0500-6600<br>0500-6600<br>0500-6600<br>0500-6600<br>0500-6600<br>0500-6600<br>0500-6600<br>0500-6600<br>0500-6600<br>0500-6600<br>0500-6600<br>0500-6600<br>0500-6600<br>0500-6600<br>0500-6600<br>0500-6600<br>0500-6600<br>0500-6600<br>0500-6600<br>0500-6600<br>0500-660000000000 | 0.1281945<br>0.1457065<br>0.1360751<br>0.1507263<br>0.1557266<br>0.145988<br>0.1482353<br>0.1482353<br>0.132075<br>0.1332075<br>0.1332075<br>0.1332075<br>0.1323254<br>0.1322254<br>0.132237<br>0.1324254<br>0.234202<br>0.1646416<br>0.0309421<br>0.0309421<br>0.0329211<br>0.5229211<br>0.5229211<br>0.5229211<br>0.5239254<br>0.1384616<br>0.234492<br>0.234492   | 9.342-006  <br>9.342-006  <br>9.342-008  <br>9.342-008  <br>8.7232-08  <br>6.3552-08  <br>7.799-08 (<br>6.3552-08 (<br>7.799-08 (<br>8.1512-08 (<br>7.7912-08 (<br>8.1512-08 (<br>7.71162-08 (<br>6.9332-08 (<br>7.7362-08 (<br>9.57352-08 (<br>5.7452-08 (<br>8.0362-08 (<br>8.0362-08 (<br>8.0362-08 (<br>8.0362-08 (<br>8.0362-08 (<br>9.0362-08 (<br>8.0362-08 (<br>9.0362-08 (<br>9.0362 | 0.0001615 (<br>0.0001318 (<br>0.0001318 (<br>0.0001318 (<br>0.0001417 (<br>0.0001614 (<br>0.000162 (<br>0.000162 (<br>0.000162 (<br>0.000164 (<br>0.000114 (<br>0.000114 (<br>0.000165 (<br>0.0000165 (<br>0.0000165 (<br>0.0000165 (<br>0.0000165 (<br>0.0000165 (   |
0.1261103<br>0.1210641<br>0.1385867<br>0.1291377<br>0.1521578<br>0.1395807<br>0.1396503<br>0.1463157<br>0.1396503<br>0.14631686<br>0.1267454<br>0.1267454<br>0.1267454<br>0.1263454<br>0.1263454<br>0.1263454<br>0.1263454<br>0.1263454<br>0.0263464<br>0.0480112<br>0.4784853<br>0.0025442<br>0.1703611<br>0.2053644<br>0.2053644<br>0.2053644<br>0.2053644<br>0.2053644<br>0.2053644<br>0.2053644<br>0.2053644<br>0.2053644<br>0.2053644<br>0.2053644<br>0.2053644<br>0.2053644<br>0.2053644<br>0.2053644<br>0.2053644<br>0.2053644<br>0.205364<br>0.205364<br>0.205364<br>0.205364<br>0.205364<br>0.205364<br>0.205364<br>0.205364<br>0.205364<br>0.205364<br>0.205364<br>0.205364<br>0.205364<br>0.205364<br>0.205364<br>0.205364<br>0.205364<br>0.205364<br>0.205364<br>0.205364<br>0.205364<br>0.205364<br>0.205364<br>0.205364<br>0.205364<br>0.205364<br>0.205364<br>0.205364<br>0.205364<br>0.205364<br>0.205364<br>0.205364<br>0.205364<br>0.205364<br>0.205364<br>0.205364<br>0.205364<br>0.205364<br>0.205364<br>0.205364<br>0.205364<br>0.205364<br>0.205364<br>0.205364<br>0.205364<br>0.205364<br>0.205364<br>0.205364<br>0.205364<br>0.205364<br>0.205364<br>0.205364<br>0.205364<br>0.205364<br>0.205364<br>0.205364<br>0.205364<br>0.205364<br>0.205364<br>0.205364<br>0.205364<br>0.205364<br>0.205364<br>0.205364<br>0.205364<br>0.205364<br>0.205364<br>0.205364<br>0.205364<br>0.205364<br>0.205364<br>0.205364<br>0.205364<br>0.205364<br>0.205364<br>0.205364<br>0.205364<br>0.205364<br>0.205364<br>0.205364<br>0.205364<br>0.205364<br>0.205364<br>0.205364<br>0.205364<br>0.205364<br>0.205364<br>0.205364<br>0.205364<br>0.205364<br>0.205364<br>0.205364<br>0.205364<br>0.205364<br>0.205364<br>0.205364<br>0.205364<br>0.205364<br>0.205364<br>0.205364<br>0.205364<br>0.205364<br>0.205364<br>0.205364<br>0.205364<br>0.205364<br>0.205364<br>0.205364<br>0.205364<br>0.205364<br>0.205364<br>0.205364<br>0.205364<br>0.205364<br>0.205364<br>0.205364<br>0.205364<br>0.205364<br>0.205364<br>0.205364<br>0.205364<br>0.205364<br>0.205364<br>0.205364<br>0.205364<br>0.205364<br>0.205364<br>0.205364<br>0.205364<br>0.205364<br>0.205364<br>0.205364<br>0.205364<br>0.205364<br>0.205364<br>0.205364<br>0.205364<br>0.205364<br>0.205364<br>0.205364<br>0.205364<br>0.205364<br>0.205364<br>0.205364<br>0.205364<br>0.205364<br>0.205364<br>0.205364<br>0.205364<br>0.205364<br>0.205364<br>0.205364<br>0.205364<br>0.205364<br>0.205364<br>0.205364<br>0.205364<br>0.205364<br>0.205364<br>0.205364<br>0.205   | 9.431E-08<br>7.761E-08<br>8.284E-08<br>9.754E-08<br>9.754E-08<br>6.373E-08<br>6.373E-08<br>6.7475E-08<br>7.465E-08<br>6.747E-08<br>6.656E-08<br>6.737E-08<br>6.656E-08<br>6.737E-08<br>7.152E-08<br>5.65E-08<br>3.799E-10<br>0<br>7.289E-08<br>7.265E-08<br>3.869E-10<br>6.116E-08<br>5.173E-08<br>9.357E-08<br>9.357E-08<br>9.354E-08   | 0.0001532<br>0.0001532<br>0.0001261<br>0.0001345<br>0.0001345<br>0.0001345<br>0.0001385<br>0.0001214<br>9.882E-05<br>0.0001214<br>0.0001268<br>0.0001181<br>0.0001081<br>0.0001162<br>9.18E-05<br>0.0001162<br>9.18E-05<br>0.0001322<br>0.0001322<br>0.0001322<br>0.0001322<br>0.0001132  | 0.1203596<br>0.1142306<br>0.1310694<br>0.1216391<br>0.1426376<br>0.1323766<br>0.1320173<br>0.1320173<br>0.1320173<br>0.1320173<br>0.120155<br>0.1189339<br>0.1204867<br>0.1224876<br>0.1224876<br>0.1224876<br>0.04234897<br>0.0438849<br>0.0438849<br>0.0438849<br>0.1559933<br>0.1871614   
   | 9.0011-08<br>7.322E-08<br>8.402E-08<br>7.797E-08<br>9.146E-08<br>6.053E-08<br>7.071E-08<br>6.353E-08<br>7.071E-08<br>6.456E-08<br>6.456E-08<br>6.456E-08<br>6.456E-08<br>6.456E-08<br>6.456E-08<br>6.456E-08<br>6.456E-08<br>6.456E-08<br>6.456E-08<br>6.456E-08<br>6.456E-08<br>6.456E-08<br>6.456E-08<br>6.456E-08<br>6.738E-08<br>6.738E-08<br>6.738E-08<br>6.738E-08<br>8.477E-08<br>5.642E-08<br>8.525E-08<br>8.525E-08  | 0.0001462<br>0.0001462<br>0.0001365<br>0.0001267<br>0.0001486<br>9.834E-05<br>0.0001448<br>9.834E-05<br>0.0001149<br>9.34E-05<br>0.0001149<br>0.0001032<br>0.0001032<br>0.0001043<br>0.0001032<br>0.0001043<br>0.0001032<br>0.000119<br>8.822E-05<br>8.8461E-05<br>8.8461E-05<br>0<br>0<br>0.000123<br>0.0000123<br>0.0000123<br>0.0000123<br>0.0000123<br>0.0000123<br>0.0000123<br>0.0000123<br>0.0000123<br>0.0000123<br>0.0000123<br>0.0000123<br>0.0000123<br>0.0000123<br>0.0000123<br>0.0000123<br>0.0000123<br>0.0000123<br>0.0000123<br>0.0000123<br>0.0000123<br>0.0000123<br>0.0000123<br>0.0000123<br>0.0000123<br>0.0000123<br>0.0000123<br>0.0000123<br>0.0000123<br>0.0000123<br>0.0000123<br>0.0000123<br>0.0000123<br>0.0000123<br>0.0000123<br>0.0000123<br>0.0000123<br>0.0000123<br>0.0000123<br>0.0000123<br>0.0000123<br>0.0000123<br>0.0000123<br>0.0000123<br>0.0000123<br>0.0000123<br>0.0000123<br>0.0000123<br>0.0000123<br>0.0000123<br>0.0000123<br>0.0000123<br>0.0000123<br>0.0000123<br>0.0000123<br>0.0000123<br>0.0000123<br>0.0000123<br>0.0000123<br>0.0000123<br>0.0000123<br>0.0000123<br>0.0000123<br>0.0000123<br>0.0000123<br>0.0000123<br>0.0000123<br>0.0000123<br>0.0000123<br>0.0000123<br>0.0000123<br>0.0000123<br>0.0000123<br>0.0000123<br>0.0000123<br>0.0000123<br>0.0000123<br>0.0000123<br>0.0000123<br>0.0000123<br>0.0000123<br>0.0000123<br>0.0000123<br>0.0000123<br>0.0000123<br>0.0000123<br>0.0000123<br>0.0000123<br>0.0000123<br>0.0000123<br>0.0000123<br>0.0000123<br>0.0000123<br>0.0000123<br>0.0000123<br>0.0000123<br>0.0000123<br>0.0000123<br>0.0000123<br>0.0000123<br>0.0000123<br>0.0000123<br>0.0000123<br>0.0000123<br>0.0000123<br>0.0000123<br>0.0000123<br>0.0000123<br>0.0000123<br>0.0000123<br>0.0000123<br>0.0000123<br>0.0000123<br>0.0000123<br>0.0000123<br>0.0000123<br>0.0000123<br>0.0000123<br>0.0000123<br>0.0000123<br>0.0000123<br>0.0000123<br>0.0000123<br>0.0000123<br>0.0000123<br>0.0000123<br>0.0000123<br>0.0000123<br>0.0000123<br>0.0000123<br>0.0000123<br>0.0000123<br>0.0000123<br>0.0000123<br>0.0000123<br>0.0000123<br>0.0000123<br>0.0000123<br>0.0000123<br>0.0000123<br>0.0000123<br>0.0000123<br>0.0000123<br>0.0000123<br>0.0000123<br>0.0000123<br>0.0000123<br>0.0000123<br>0.0000123<br>0.0000123<br>0.0000123<br>0.00000000000000000000000000000000000  | 2.4904488<br>1.9842855<br>2.2002208<br>2.1078468<br>2.2364653<br>3.1057198<br>3.0468346<br>2.9569573<br>2.8602462<br>3.1731567<br>2.6220547<br>2.6509615<br>2.6509615<br>2.6509615<br>2.6509615<br>2.6172616<br>4.9637927<br>4.4562831<br>0.0573728<br>0<br>0<br>1.3610254<br>1.3610254<br>1.3610254<br>1.3610254<br>1.3610254<br>1.3610254<br>1.3610254<br>1.3610254<br>1.3610254<br>1.3610254<br>0.05115727<br>0.05115727   |
1.863E-06<br>1.272E-06<br>1.351E-06<br>1.351E-06<br>1.32FE-06<br>1.32FE-06<br>1.628E-06<br>1.264E-06<br>1.264E-06<br>1.264E-06<br>1.401E-06<br>1.441E-06<br>1.441E-06<br>1.441E-06<br>8.731E-09<br>0<br>0<br>2.066E-07<br>9.2066E-07<br>9.2066E-07<br>9.2066E-07<br>9.2066E-07<br>9.2066E-07<br>9.2066E-07<br>9.2066E-07<br>9.2066E-07<br>9.2066E-07<br>9.2066E-07<br>9.2066E-07<br>9.2066E-07<br>9.2066E-07<br>9.2066E-07<br>9.2066E-07<br>9.2066E-07<br>9.2066E-07<br>9.2066E-07<br>9.2066E-07<br>9.2066E-07<br>9.2066E-07<br>9.2066E-07<br>9.2066E-07<br>9.2066E-07<br>9.2066E-07<br>9.2066E-07<br>9.2066E-07<br>9.2066E-07<br>9.2066E-07<br>9.2066E-07<br>9.2066E-07<br>9.2066E-07<br>9.2066E-07<br>9.2066E-07<br>9.2066E-07<br>9.2066E-07<br>9.2066E-07<br>9.2066E-07<br>9.2066E-07<br>9.2066E-07<br>9.2066E-07<br>9.2066E-07<br>9.2066E-07<br>9.2066E-07<br>9.2066E-07<br>9.2066E-07<br>9.2066E-07<br>9.2066E-07<br>9.2066E-07<br>9.2066E-07<br>9.2066E-07<br>9.2066E-07<br>9.2066E-07<br>9.2066E-07<br>9.2066E-07<br>9.2066E-07<br>9.2066E-07<br>9.2066E-07<br>9.2066E-07<br>9.2066E-07<br>9.2066E-07<br>9.2066E-07<br>9.2066E-07<br>9.2066E-07<br>9.2066E-07<br>9.2066E-07<br>9.2066E-07<br>9.2066E-07<br>9.2066E-07<br>9.2066E-07<br>9.2066E-07<br>9.2066E-07<br>9.2066E-07<br>9.2066E-07<br>9.2765E-07<br>2.776E-07<br>2.776E-07<br>2.776E-07<br>2.776E-07<br>2.776E-07<br>2.776E-07<br>2.776E-07<br>2.776E-07<br>2.776E-07<br>2.776E-07<br>2.776E-07<br>2.776E-07<br>2.776E-07<br>2.776E-07<br>2.776E-07<br>2.776E-07<br>2.776E-07<br>2.776E-07<br>2.776E-07<br>2.776E-07<br>2.776E-07<br>2.776E-07<br>2.776E-07<br>2.776E-07<br>2.776E-07<br>2.776E-07<br>2.776E-07<br>2.776E-07<br>2.776E-07<br>2.776E-07<br>2.776E-07<br>2.776E-07<br>2.777E-07<br>2.777E-07<br>2.777E-07<br>2.777E-07<br>2.777E-07<br>2.777E-07<br>2.777E-07<br>2.777E-07<br>2.777E-07<br>2.777E-07<br>2.777E-07<br>2.777E-07<br>2.777E-07<br>2.777E-07<br>2.777E-07<br>2.777E-07<br>2.777E-07<br>2.777E-07<br>2.777E-07<br>2.777E-07<br>2.777E-07<br>2.777E-07<br>2.777E-07<br>2.777E-07<br>2.777E-07<br>2.777E-07<br>2.777E-07<br>2.777E-07<br>2.777E-07<br>2.777E-07<br>2.777E-07<br>2.777E-07<br>2.777E-07<br>2.777E-07<br>2.777E-07<br>2.777E-07<br>2.777E-07<br>2.777E-07<br>2.777E-07<br>2.777E-07<br>2.777E-07<br>2.777E-07<br>2.777E-07<br>2.777E-07<br>2.777E-07<br>2.777E-07<br>2.777E-07<br>2.777E-07<br>2.777E-07<br>2.777E-07<br>2.777E-07<br>2.777E-07<br>2.777E-07<br>2.777E-07<br>2.777E-07<br>2.777E-                 | 0.030261<br>0.002265<br>0.0022195<br>0.0021953<br>0.0021564<br>0.002531<br>0.002544<br>0.002257<br>0.002744<br>0.0022757<br>0.0021782<br>0.0023497<br>0.0015163<br>0.0023497<br>1.592E-05<br>0<br>0<br>0.0003777<br>1.681E-05<br>0.0003777<br>1.681E-05<br>0.0003728  | 2.3566842<br>1.8789238<br>2.0977776<br>2.0052501<br>2.1283629<br>2.952749<br>2.900207<br>2.8217082<br>2.7301688<br>3.0191874<br>2.4772701<br>2.5033689<br>2.5033689<br>2.5033689<br>2.5033689<br>2.5033689<br>2.7392982<br>3.39311649<br>4.688413<br>3.9311649<br>4.688413<br>0.0463027<br>0<br>0.0463027<br>0<br>0.0463057<br>0.0463057<br>0.0463057<br>0.04640617<br>0.5506791<br>0.5506791<br>0.5506791<br>0.5506791<br>0.5506791<br>0.5506791<br>0.5506791<br>0.5506791<br>0.5506791<br>0.5506791<br>0.5506791<br>0.5506791<br>0.5506791<br>0.5506791<br>0.5506791<br>0.5506791<br>0.5506791<br>0.5506791<br>0.5506791<br>0.5506791<br>0.5506791<br>0.5506791<br>0.5506791<br>0.5506791<br>0.5506791<br>0.5506791<br>0.5506791<br>0.5506791<br>0.5506791<br>0.5506791<br>0.5506791<br>0.5506791<br>0.5506791<br>0.5506791<br>0.5506791<br>0.5506791<br>0.5506791<br>0.5506791<br>0.5506791<br>0.5506791<br>0.5506791<br>0.5506791<br>0.5506791<br>0.5506791<br>0.5506791<br>0.5506791<br>0.5506791<br>0.5506791<br>0.5506791<br>0.5506791<br>0.5506791<br>0.5506791<br>0.5506791<br>0.5506791<br>0.5506791<br>0.5506791<br>0.5506791<br>0.5506791<br>0.5506791<br>0.5506791<br>0.5506791<br>0.5506791<br>0.5506791<br>0.5506791<br>0.5506791<br>0.5506791<br>0.5506791<br>0.5506791<br>0.5506791<br>0.5506791<br>0.5506791<br>0.5506791<br>0.5506791<br>0.5506791<br>0.5506791<br>0.5506791<br>0.5506791<br>0.5506791<br>0.5506791<br>0.5506791<br>0.5506791<br>0.5506791<br>0.5506791<br>0.5506791<br>0.5506791<br>0.5506791<br>0.5506791<br>0.5506791<br>0.5506791<br>0.5506791<br>0.5506791<br>0.5506791<br>0.5506791<br>0.5506791<br>0.5506791<br>0.5506791<br>0.5506791<br>0.5506791<br>0.5506791<br>0.5506791<br>0.5506791<br>0.5506791<br>0.5506791<br>0.5506791<br>0.5506791<br>0.5506791<br>0.5506791<br>0.5506791<br>0.5506791<br>0.5506791<br>0.5506791<br>0.5506791<br>0.5506791<br>0.5506791<br>0.5506791<br>0.5506791<br>0.5506791<br>0.5506791<br>0.5506791<br>0.5506791<br>0.5506791<br>0.5506791<br>0.5506791<br>0.5506791<br>0.5506791<br>0.5506791<br>0.5506791<br>0.5506791<br>0.5506791<br>0.5506791<br>0.5506791<br>0.5506791<br>0.5506791<br>0.5506791<br>0.5506791<br>0.5506791<br>0.5506791<br>0.5506791<br>0.5506791<br>0.5506791<br>0.5506791<br>0.5506791<br>0.5506791<br>0.5506791<br>0.5506791<br>0.5506791<br>0.5506791<br>0.5506791<br>0.5506791<br>0.5506791<br>0.5506791<br>0.5506791<br>0.5506791<br>0.5506791<br>0.5506791<br>0.5506791<br>0.5506791<br>0.5506791<br>0.55067   | 1.7622-06<br>1.204E-06<br>1.345E-06<br>1.345E-06<br>1.364E-06<br>1.549E-06<br>1.549E-06<br>1.549E-06<br>1.549E-06<br>1.337E-06<br>1.337E-06<br>1.337E-06<br>1.336E-06<br>1.336E-06<br>1.336E-06<br>1.336E-06<br>1.336E-06<br>1.336E-07<br>1.868E-07<br>1.648E-07<br>1.644E-07<br>1.644E-07<br>2.51E-07<br>2.508E-07<br>2.508E-07   |
0.0028636<br>0.0019569<br>0.0021848<br>0.00221848<br>0.00221848<br>0.00221848<br>0.0022184<br>0.0022151<br>0.002519<br>0.0025171<br>0.0025171<br>0.0026204<br>0.0021727<br>0.0022225<br>0.0012425<br>0.0024249<br>0.00204749<br>0.0002449<br>0.0002449<br>0.0002449<br>0.0002449<br>0.0002445<br>0.00005<br>0.00005<br>0.000055<br>0.000055<br>0.000055<br>0.000055<br>0.000055<br>0.000055<br>0.000055<br>0.000055<br>0.000055<br>0.000055<br>0.000055<br>0.000055<br>0.000055<br>0.000055<br>0.000055<br>0.000055<br>0.000055<br>0.000055<br>0.000055<br>0.000055<br>0.000055<br>0.000055<br>0.000055<br>0.000055<br>0.00005<br>0.000055<br>0.000055<br>0.000055<br>0.000055<br>0.000055<br>0.00005<br>0.00055<br>0.00005<br>0.00005<br>0.00005<br>0.00005<br>0.00005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.000 | 2.2440968<br>1.7795138<br>1.9915216<br>1.8961342<br>2.0102391<br>2.7864095<br>2.720732<br>2.6462899<br>2.5622851<br>2.38364013<br>2.3321589<br>2.332054<br>2.4375101<br>2.6016999<br>3.7301495<br>0<br>0<br>1.1398552<br>1.1398552<br>1.14138222<br>0.041725<br>0.4240418<br>0.5013024<br>0.5013024   | 1.678E-06<br>1.141E-06<br>1.277E-06<br>1.215E-06<br>1.215E-06<br>1.245E-06<br>1.1454E-06<br>1.1454E-06<br>1.315E-06<br>1.245E-06<br>1.245E-06<br>1.245E-06<br>1.245E-06<br>1.245E-06<br>1.245E-06<br>1.245E-06<br>1.245E-06<br>1.245E-06<br>1.245E-06<br>1.245E-07<br>1.285E-07<br>1.288E-07<br>1.288E-07<br>1.288E-07<br>1.288E-07<br>1.288E-07<br>1.288E-07<br>1.288E-07<br>1.288E-07<br>1.288E-07<br>1.288E-07<br>1.288E-07<br>1.288E-07<br>1.288E-07<br>1.288E-07<br>1.288E-07<br>1.288E-07<br>1.288E-07<br>1.288E-07<br>1.288E-07<br>1.288E-07<br>1.288E-07<br>1.288E-07<br>1.288E-07<br>1.288E-07<br>1.288E-07<br>1.288E-07<br>1.288E-07<br>1.288E-07<br>1.288E-07<br>1.288E-07<br>1.288E-07<br>1.288E-07<br>1.288E-07<br>1.288E-07<br>1.288E-07<br>1.288E-07<br>1.288E-07<br>1.288E-07<br>1.288E-07<br>1.288E-07<br>1.288E-07<br>1.288E-07<br>1.288E-07<br>1.288E-07<br>1.288E-07<br>1.288E-07<br>1.288E-07<br>1.288E-07<br>1.288E-07<br>1.288E-07<br>1.288E-07<br>1.288E-07<br>1.288E-07<br>1.288E-07<br>1.288E-07<br>1.288E-07<br>1.288E-07<br>1.288E-07<br>1.288E-07<br>1.288E-07<br>1.288E-07<br>1.288E-07<br>1.288E-07<br>1.288E-07<br>1.288E-07<br>1.288E-07<br>1.288E-07<br>1.288E-07<br>1.288E-07<br>1.288E-07<br>1.288E-07<br>1.288E-07<br>1.288E-07<br>1.288E-07<br>1.288E-07<br>1.288E-07<br>1.288E-07<br>1.288E-07<br>1.288E-07<br>1.288E-07<br>1.288E-07<br>1.288E-07<br>1.288E-07<br>1.288E-07<br>1.288E-07<br>1.288E-07<br>1.288E-07<br>1.288E-07<br>1.288E-07<br>1.288E-07<br>1.288E-07<br>1.288E-07<br>1.288E-07<br>1.288E-07<br>1.288E-07<br>1.288E-07<br>1.288E-07<br>1.288E-07<br>1.288E-07<br>1.288E-07<br>1.288E-07<br>1.288E-07<br>1.288E-07<br>1.288E-07<br>1.288E-07<br>1.288E-07<br>1.288E-07<br>1.288E-07<br>1.288E-07<br>1.288E-07<br>1.288E-07<br>1.288E-07<br>1.288E-07<br>1.288E-07<br>1.288E-07<br>1.288E-07<br>1.288E-07<br>1.288E-07<br>1.288E-07<br>1.288E-07<br>1.288E-07<br>1.288E-07<br>1.288E-07<br>1.288E-07<br>1.288E-07<br>1.288E-07<br>1.288E-07<br>1.288E-07<br>1.288E-07<br>1.288E-07<br>1.288E-07<br>1.288E-07<br>1.288E-07<br>1.288E-07<br>1.288E-07<br>1.288E-07<br>1.288E-07<br>1.288E-07<br>1.288E-07<br>1.288E-07<br>1.288E-07<br>1.288E-07<br>1.288E-07<br>1.288E-07<br>1.288E-07<br>1.288E-07<br>1.288E-07<br>1.288E-07<br>1.288E-07<br>1.288E-07<br>1.288E-07<br>1.288E-07<br>1.288E-07<br>1.288E-07<br>1.288E-07<br>1.288E-07<br>1.288E-07<br>1.288E-07<br>1.288E-07<br>1.288E-07<br>1.288E-07<br>1.288E-07<br>1.288E-07<br>1.288E-07<br>1.288E-07  |
0.0027267<br>0.0018534<br>0.0020742<br>0.0019748<br>0.0019748<br>0.0019748<br>0.0019748<br>0.0019347<br>0.0019347<br>0.0019347<br>0.0019347<br>0.002238<br>0.002238<br>0.002238<br>0.002238<br>0.002238<br>0.002238<br>0.002238<br>0.002238<br>0.002313<br>0.002313<br>0.002315<br>0.0003163<br>0.0003163<br>0.0003163<br>0.0003163<br>0.000238<br>0.000238<br>0.000238<br>0.000238<br>0.000238<br>0.000238<br>0.000238<br>0.000238<br>0.000238<br>0.000238<br>0.000238<br>0.000238<br>0.000238<br>0.000238<br>0.000238<br>0.000238<br>0.000238<br>0.000238<br>0.000238<br>0.000238<br>0.000238<br>0.000238<br>0.000238<br>0.000238<br>0.000238<br>0.000238<br>0.000238<br>0.000238<br>0.000238<br>0.000238<br>0.000238<br>0.000238<br>0.000238<br>0.000238<br>0.000238<br>0.000238<br>0.000238<br>0.000238<br>0.000238<br>0.000238<br>0.000238<br>0.000238<br>0.000238<br>0.000238<br>0.000238<br>0.000238<br>0.000238<br>0.000238<br>0.000238<br>0.000238<br>0.000238<br>0.000238<br>0.000238<br>0.000238<br>0.000238<br>0.000238<br>0.000238<br>0.000238<br>0.000238<br>0.000238<br>0.000238<br>0.000238<br>0.000238<br>0.000238<br>0.000238<br>0.000238<br>0.000238<br>0.000238<br>0.000238<br>0.000238<br>0.000238<br>0.000238<br>0.000238<br>0.000238<br>0.000238<br>0.000238<br>0.000238<br>0.000238<br>0.000238<br>0.000238<br>0.000238<br>0.000238<br>0.000238<br>0.000238<br>0.000238<br>0.000238<br>0.000238<br>0.000238<br>0.000238<br>0.000238<br>0.000238<br>0.000238<br>0.000238<br>0.000238<br>0.000238<br>0.000238<br>0.000238<br>0.000238<br>0.000238<br>0.000238<br>0.000238<br>0.000238<br>0.000238<br>0.000238<br>0.000238<br>0.000238<br>0.000238<br>0.000238<br>0.000238<br>0.000238<br>0.000238<br>0.000238<br>0.000238<br>0.000238<br>0.000238<br>0.000238<br>0.000238<br>0.000238<br>0.000238<br>0.000238<br>0.000238<br>0.000238<br>0.000238<br>0.000238<br>0.000238<br>0.000238<br>0.000238<br>0.000238<br>0.000238<br>0.000238<br>0.000238<br>0.000238<br>0.000238<br>0.000238<br>0.000238<br>0.000238<br>0.000238<br>0.000238<br>0.000238<br>0.000238<br>0.000238<br>0.000238<br>0.000200000000000000000000000000000000   | 2.623364<br>2.11248<br>2.3459273<br>2.2439219<br>3.2614464<br>3.1928226<br>3.1051925<br>3.0057113<br>3.3257455<br>2.7818703<br>2.8396521<br>3.0470013<br>3.3621382<br>3.1979947<br>4.6427163<br>0.0604669<br>0<br>0<br>1.8903165<br>1.8903165<br>1.68903165<br>1.68903165<br>1.68903165<br>1.68903165<br>1.68903165<br>1.68903165<br>1.68903165<br>1.68903165<br>1.68903165<br>1.68903165<br>1.68903165<br>1.68903165<br>1.68270176<br>1.6980343<br>1.82270176   | 19622.06<br>1.3542.06<br>1.3542.06<br>1.4382.06<br>1.4382.06<br>1.3572.06<br>1.3242.06<br>1.3272.06<br>1.3272.06<br>1.3272.06<br>1.4222.06<br>1.4222.06<br>1.4322.06<br>1.4322.06<br>1.4382.06<br>9.7662.07<br>1.3382.06<br>9.7662.07<br>1.3382.06<br>9.7862.07<br>2.872.07<br>9.6955.09<br>2.512.07<br>9.6955.09<br>2.512.07<br>3.7632.07<br>3.7632.07<br>3.7632.07<br>3.7632.07<br>3.7632.07<br>3.7632.07<br>3.7632.07<br>3.7632.07<br>3.7632.07<br>3.7632.07<br>3.7632.07<br>3.7632.07<br>3.7632.07<br>3.7632.07<br>3.7632.07<br>3.7632.07<br>3.7632.07<br>3.7632.07<br>3.7632.07<br>3.7632.07<br>3.7632.07<br>3.7632.07<br>3.7632.07<br>3.7632.07<br>3.7632.07<br>3.7632.07<br>3.7632.07<br>3.7632.07<br>3.7632.07<br>3.7632.07<br>3.7632.07<br>3.7632.07<br>3.7632.07<br>3.7632.07<br>3.7632.07<br>3.7632.07<br>3.7632.07<br>3.7632.07<br>3.7632.07<br>3.7632.07<br>3.7632.07<br>3.7632.07<br>3.7632.07<br>3.7632.07<br>3.7632.07<br>3.7632.07<br>3.7632.07<br>3.7632.07<br>3.7632.07<br>3.7632.07<br>3.7632.07<br>3.7632.07<br>3.7632.07<br>3.7632.07<br>3.7632.07<br>3.7632.07<br>3.7632.07<br>3.7632.07<br>3.7632.07<br>3.7632.07<br>3.7632.07<br>3.7632.07<br>3.7632.07<br>3.7632.07<br>3.7632.07<br>3.7632.07<br>3.7632.07<br>3.7632.07<br>3.7632.07<br>3.7642.07<br>3.7642.07<br>3.7642.07<br>3.7642.07<br>3.7642.07<br>3.7642.07<br>3.7642.07<br>3.7642.07<br>3.7642.07<br>3.7642.07<br>3.7642.07<br>3.7642.07<br>3.7642.07<br>3.7642.07<br>3.7642.07<br>3.7642.07<br>3.7642.07<br>3.7642.07<br>3.7642.07<br>3.7642.07<br>3.7642.07<br>3.7642.07<br>3.7642.07<br>3.7642.07<br>3.7642.07<br>3.7642.07<br>3.7642.07<br>3.7642.07<br>3.7642.07<br>3.7642.07<br>3.7642.07<br>3.7642.07<br>3.7642.07<br>3.7642.07<br>3.7642.07<br>3.7642.07<br>3.7642.07<br>3.7642.07<br>3.7642.07<br>3.7642.07<br>3.7642.07<br>3.7642.07<br>3.7642.07<br>3.7642.07<br>3.7642.07<br>3.7642.07<br>3.7642.07<br>3.7642.07<br>3.7642.07<br>3.7642.07<br>3.7642.07<br>3.7642.07<br>3.7642.07<br>3.7642.07<br>3.7642.07<br>3.7642.07<br>3.7642.07<br>3.7642.07<br>3.7642.07<br>3.7642.07<br>3.7642.07<br>3.7642.07<br>3.7642.07<br>3.7642.07<br>3.7642.07<br>3.7762.07<br>3.7762.07<br>3.7762.07<br>3.7762.07<br>3.7762.07<br>3.7762.07<br>3.7762.07<br>3.7762.07<br>3.7762.07<br>3.7762.07<br>3.7762.07<br>3.7762.07<br>3.7762.07<br>3.7762.07<br>3.7762.07<br>3.7762.07<br>3.7762.07<br>3.7762.07<br>3.7762.07<br>3.7762.07<br>3.7762.07<br>3.7762.07<br>3.7762.07<br>3.7762.07<br>3.7762.07<br>3.7762.07<br>3.7762.07<br>3.77       | 0.0031876<br>0.0022403<br>0.0022433<br>0.002337<br>0.002347<br>0.0022645<br>0.0027111<br>0.002156<br>0.0027011<br>0.0028685<br>0.0023913<br>0.0024646<br>0.0012867<br>0.002245<br>0.0022716<br>0.00024545<br>0<br>0.00024545<br>0.00025245<br>0.00005245<br>0.00005245<br>0.00005245<br>0.00005245<br>0.00005245<br>0.00005245<br>0.00005245   
   | 2.3644564<br>1.8937445<br>2.1225909<br>2.0177733<br>2.1529112<br>2.928047<br>2.8544498<br>2.7808064<br>2.6943024<br>2.9947351<br>2.4523139<br>2.4523139<br>2.4523139<br>2.4523663<br>2.72466<br>3.8826302<br>0<br>0<br>1.5831454<br>1.585267<br>0.0439825<br>0.6846321<br>0.68804531   | 1.768:-00 0.002<br>1.214:-60 0.001<br>1.351:-60 0.002<br>1.381:-60 0.002<br>1.381:-60 0.002<br>1.312:-60 0.002<br>1.312:-60 0.002<br>1.525:-60 0.002<br>1.525:-60 0.002<br>1.312:-60 0.002<br>1.313:-60 0.002<br>1.312:-00 0.002<br>1.312:-00 0.002<br>1.312:-00 0.002<br>1.312:-00 0.002<br>1.313:-00 0.002<br>1.313:   | 873         0.04           9723         0.03           2107         0.04           1015         0.03           0.33         0.53           2422         0.04           0.33         0.55           3384         0.04           1284         0.04           1284         0.04           1284         0.04           1284         0.04           1438         0.05           2204         0.04           1438         0.05           1585         0.02           64393         0.03           0.3399         0.0           14393         0.02           3384         0.021           14393         0.023           0.339         0.01           15737         0.02           5731         0.72  
  | 147043         3.           147043         3.           383822         2.           408597         2.           405571         2.           338011         2.           338012         2.           338011         2.           338014         2.1           514016         2.           155074         2.4           165511         3.1           157074         2.4           164276         2.5           164254         2.           183993         8.1           0         0           0         0           0         0           0         0           0         0           0         0           0         0           0         0           0         0           0.83015         5.5           10642         9.5           10642         9.5  
   | 5.18E-08         2           6.619E-08         2           6.619E-08         2           7.76E-08         4           2.99E-08         3           8.74E-08         -           1.97E-08         3           6.51E-08         4           0.011E-08         4           0.011E-08      | 3.998E-05<br>4.254E-05<br>4.224E-05<br>4.21E-05<br>3.569E-05<br>4.3736E-05<br>3.569E-05<br>4.307E-05<br>4.307E-05<br>4.307E-05<br>4.059E-05<br>4.22E-05<br>4.22E-05<br>4.22E-05<br>1.501E-05<br>1.501E-05<br>1.18E-05<br>1.18E-05<br>1.18E-05<br>1.18E-05<br>1.18E-05<br>1.18E-05<br>1.58E-05<br>1.52E-05<br>1.52E-05<br>1.52E-05<br>1.52E-05<br>1.52E-05<br>1.52E-05<br>1.52E-05<br>1.52E-05<br>1.52E-05<br>1.52E-05<br>1.52E-05<br>1.52E-05<br>1.52E-05<br>1.52E-05<br>1.52E-05<br>1.52E-05<br>1.52E-05<br>1.52E-05<br>1.52E-05<br>1.52E-05<br>1.52E-05<br>1.52E-05<br>1.52E-05<br>1.52E-05<br>1.52E-05<br>1.52E-05<br>1.52E-05<br>1.52E-05<br>1.52E-05<br>1.52E-05<br>1.52E-05<br>1.52E-05<br>1.52E-05<br>1.52E-05<br>1.52E-05<br>1.52E-05<br>1.52E-05<br>1.52E-05<br>1.52E-05<br>1.52E-05<br>1.52E-05<br>1.52E-05<br>1.52E-05<br>1.52E-05<br>1.52E-05<br>1.52E-05<br>1.52E-05<br>1.52E-05<br>1.52E-05<br>1.52E-05<br>1.52E-05<br>1.52E-05<br>1.52E-05<br>1.52E-05<br>1.52E-05<br>1.52E-05<br>1.52E-05<br>1.52E-05<br>1.52E-05<br>1.52E-05<br>1.52E-05<br>1.52E-05<br>1.52E-05<br>1.52E-05<br>1.52E-05<br>1.52E-05<br>1.52E-05<br>1.52E-05<br>1.52E-05<br>1.52E-05<br>1.52E-05<br>1.52E-05<br>1.52E-05<br>1.52E-05<br>1.52E-05<br>1.52E-05<br>1.52E-05<br>1.52E-05<br>1.52E-05<br>1.52E-05<br>1.52E-05<br>1.52E-05<br>1.52E-05<br>1.52E-05<br>1.52E-05<br>1.52E-05<br>1.52E-05<br>1.52E-05<br>1.52E-05<br>1.52E-05<br>1.52E-05<br>1.52E-05<br>1.52E-05<br>1.52E-05<br>1.52E-05<br>1.52E-05<br>1.52E-05<br>1.52E-05<br>1.52E-05<br>1.52E-05<br>1.52E-05<br>1.52E-05<br>1.52E-05<br>1.52E-05<br>1.52E-05<br>1.52E-05<br>1.52E-05<br>1.52E-05<br>1.52E-05<br>1.52E-05<br>1.52E-05<br>1.52E-05<br>1.52E-05<br>1.52E-05<br>1.52E-05<br>1.52E-05<br>1.52E-05<br>1.52E-05<br>1.52E-05<br>1.52E-05<br>1.52E-05<br>1.52E-05<br>1.52E-05<br>1.52E-05<br>1.52E-05<br>1.52E-05<br>1.52E-05<br>1.52E-05<br>1.52E-05<br>1.52E-05<br>1.52E-05<br>1.52E-05<br>1.52E-05<br>1.52E-05<br>1.52E-05<br>1.52E-05<br>1.52E-05<br>1.52E-05<br>1.52E-05<br>1.52E-05<br>1.52E-05<br>1.52E-05<br>1.52E-05<br>1.52E-05<br>1.52E-05<br>1.52E-05<br>1.52E-05<br>1.52E-05<br>1.52E-05<br>1.52E-05<br>1.52E-05<br>1.52E-05<br>1.52E-05<br>1.52E-05<br>1.52E-05<br>1.52E-05<br>1.52E-05<br>1.52E-05<br>1.52E-05<br>1.52E-05<br>1.52E-05<br>1.52E-05<br>1.52E-05<br>1.52E-05<br>1.52E-05<br>1.52E-05<br>1.52E-05<br>1.52E-05<br>1.52E-05<br>1.52E-05<br>1.52E-05<br>1.52E-05<br>1.52E-05<br>1.52E-05<br>1.52E-05<br>1.52E-05<br>1.52E-05<br>1.52E-05<br>1.52E-05<br>1.52E-05<br>1.52E-05<br>1.52E-05<br>1.52E-05<br>1.52E-05<br>1.52E-05<br>1.52E-05<br>1.52E-05<br>1.52E-05<br>1.52E-05<br>1.52E-05<br>1.52E-05<br>1.52E-05<br>1.52E-05<br>1.52E-05<br>1.52E-05<br>1.52E-05<br>1.52E-05<br>1.52E-05<br>1.52E-05<br>1.52E-05<br>1.52E-05<br>1.52E-05<br>1.52E-05    | 0.044322<br>0.044017<br>0.044332<br>0.044017<br>0.054813<br>0.0558211<br>0.0558213<br>0.0558213<br>0.0558213<br>0.053821<br>0.053821<br>0.056126<br>0.046222<br>0.0637764<br>0<br>0<br>0.04622<br>0.0483242<br>0<br>0.048224<br>0<br>0<br>0<br>0.04622<br>0.048214<br>0<br>0<br>0<br>0.04622<br>0.048214<br>0<br>0<br>0<br>0.04622<br>0.048324<br>0<br>0<br>0<br>0.04622<br>0.048324<br>0<br>0<br>0<br>0<br>0.04622<br>0.048324<br>0<br>0<br>0<br>0<br>0<br>0.048132<br>0<br>0<br>0<br>0<br>0<br>0<br>0.048130<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0  
  | 2.842E-08         2.822E-08           2.822E-08         2.822E-08           2.822E-08         2.437E-08           2.437E-08         2.437E-08           2.336E-08         2.827E-08           2.837E-08         2.827E-08           2.837E-08         2.827E-08           2.832E-08         2.82E-08           2.82E-08         2.82E-08           2.832E-08         2.832E-08           2.835E-08         2.835E-08           0         0           0         0           7.014E-09         2.7014E-09           5.857E-00         5.857E-00           5.857E-01         5.857E-00           5.857E-02         5.857E-00           5.857E-03         5.857E-09           5.857E-04         5.857E-09           5.857E-05         5.857E-09           5.857E-06         5.857E-09           5.857E-07         5.857E-09           5.857E-08         5.857E-09           5.857E-09         5.857E-09           5.857E-09         5.857E-09           5.857E-09         5.857E-09  | 1.584E-05<br>1.897E-05<br>1.057E-05<br>3.072E-05<br>3.876E-05<br>1.577E-05<br>3.331E-05<br>1.307E-05<br>1.307E-05<br>1.307E-05<br>1.382E-05<br>1.889E-05<br>1.889E-05<br>1.889E-05<br>1.282E-05<br>1.282E-05<br>1.282E-05<br>1.255E-05<br>1.255E-05<br>1.206E-05<br>1.901E-05  |
| Sheung Touen B/T and Car Park<br>Sheung Touen B/T and Car Park<br>Village Access Road<br>Village Access Road   | RD_29<br>RD_29<br>RD_29<br>RD_29<br>RD_29<br>RD_29<br>RD_29<br>RD_29<br>RD_29<br>RD_29<br>RD_29<br>RD_29<br>RD_29<br>RD_29<br>RD_29<br>RD_29<br>RD_29<br>RD_29<br>RD_29<br>RD_30_31<br>RD_30_31<br>RD_30_31<br>RD_30_31<br>RD_30_31<br>RD_30_31<br>RD_30_31<br>RD_30_31   | 0900-1000           1000-1000           1000-2000           1200-300           1200-300           1300-1400           1400-1500           1500-1600           1600-1700           1600-2000           2000-2100           2100-2200           2200-2000           2000-2100           2000-2100           2000-2100           0300-4000           0300-4000           0300-4000           0500-6000           0600-0700           0600-0700           0700-8000           0800-9900           0800-9900           0800-9900           0800-9900   | 0.1281945<br>0.1360751<br>0.1360751<br>0.1557266<br>0.145988<br>0.1452452<br>0.1452452<br>0.1322075<br>0.1332075<br>0.1332075<br>0.1323254<br>0.1322254<br>0.1322254<br>0.1322254<br>0.1322254<br>0.1322254<br>0.0303941<br>0.032592911<br>0.5292911<br>0.5292911<br>0.5292911<br>0.5292911<br>0.5292911<br>0.5292911<br>0.5292911<br>0.5292911<br>0.5292911<br>0.5292911<br>0.5292911<br>0.5292911<br>0.5292911<br>0.5292911<br>0.5292911<br>0.5292911<br>0.5292911<br>0.5292911<br>0.5292911<br>0.5292911<br>0.5292911<br>0.5292911<br>0.5292911<br>0.5292911<br>0.5292911<br>0.5292911<br>0.5292911<br>0.5292911<br>0.5292911<br>0.5292911<br>0.5292911<br>0.5292911<br>0.5292911<br>0.5292911<br>0.5292911<br>0.5292911<br>0.5292911<br>0.5292911<br>0.5292911<br>0.5292911<br>0.5292911<br>0.5292911<br>0.5292911<br>0.5292911<br>0.5292911<br>0.5292911<br>0.5292911<br>0.5292911<br>0.5292911<br>0.5292911<br>0.5292911<br>0.5292911<br>0.5292911<br>0.5292911<br>0.5292911<br>0.5292911<br>0.5292911<br>0.5292911<br>0.5292911<br>0.5292911<br>0.5292911<br>0.5292911<br>0.5292911<br>0.5292911<br>0.5292911<br>0.5292911<br>0.5292911<br>0.5292911<br>0.5292911<br>0.5292911<br>0.5292911<br>0.5292911<br>0.5292911<br>0.5292911<br>0.5292911<br>0.5292911<br>0.5292911<br>0.5292911<br>0.5292911<br>0.5292911<br>0.5292911<br>0.5292911<br>0.5292911<br>0.5292911<br>0.5292911<br>0.5292911<br>0.5292911<br>0.5292911<br>0.5292911<br>0.5292911<br>0.5292911<br>0.5292911<br>0.5292911<br>0.5292911<br>0.5292911<br>0.5292911<br>0.5292911<br>0.5292911<br>0.5292911<br>0.5292911<br>0.5292911<br>0.5292911<br>0.5292911<br>0.5292911<br>0.5292911<br>0.5292911<br>0.5292911<br>0.5292911<br>0.5292911<br>0.5292911<br>0.5292911<br>0.5292911<br>0.5292911<br>0.5292911<br>0.5292911<br>0.5292911<br>0.5292911<br>0.5292911<br>0.5292911<br>0.5292911<br>0.5292911<br>0.5292911<br>0.5292911<br>0.5292911<br>0.5292911<br>0.5292911<br>0.5292911<br>0.5292911<br>0.5292911<br>0.5292911<br>0.5292911<br>0.5292911<br>0.5292911<br>0.5292911<br>0.5292911<br>0.5292911<br>0.5292911<br>0.5292911<br>0.5292911<br>0.5292911<br>0.5292911<br>0.5292911<br>0.5292911<br>0.5292911<br>0.5292911<br>0.5292911<br>0.5292911<br>0.5292911<br>0.5292911<br>0.5292911<br>0.5292911<br>0.5292911<br>0.5292911<br>0.5292910<br>0.5292910<br>0.5292910<br>0.5292910<br>0.5292910<br>0.5292910<br>0.5292910<br>0.5292910<br>0.5292910<br>0.5292910<br>0.5292910<br>0.5292910<br>0.5292910<br>0.5292910<br>0.5292910<br>0.5292910<br>0.5292910<br>0.5292910<br>0.5292910<br>0.5292910<br>0.5292910<br>0.5292910<br>0.5292910<br>0.5292910<br>0.5292910<br>0.5292910<br>0.5292910000000000000000000000000000000000   | 9.342-003  <br>9.342-003  <br>9.342-003  <br>9.342-003  <br>9.342-003  <br>9.342-003  <br>9.322-003  <br>9.3325-003  <br>7.7321-003  <br>6.5335-003  <br>7.7312-003  <br>0 0<br>0 0<br>0 0<br>0 0<br>0 0<br>0 0<br>0 0<br>0  
   | 0.000135 (<br>0.0001335 (<br>0.000135 (<br>0.0001317 (<br>0.0001417 (<br>0.000164 (<br>0.000164 (<br>0.000164 (<br>0.000128 (<br>0.0001324 (<br>0.0001324 (<br>0.0001324 (<br>0.0001324 (<br>0.000126 (<br>0.000126 (<br>0.000126 (<br>0.000126 (<br>0.000128 (<br>0.000168 (<br>0.000186 (<br>0.000186 (<br>0.000186 (<br>0.000186 (<br>0.000186 (<br>0.000186 (<br>0.000186 (<br>0.000186 (<br>0.000186 (<br>0.000185 (<br>0.0000185 (<br>0.0000185 (<br>0.0000185 (<br>0   | 0.1261103<br>0.1210641<br>0.1385867<br>0.1291377<br>0.1291377<br>0.1291377<br>0.1491352<br>0.1491352<br>0.1491352<br>0.1491352<br>0.1491352<br>0.1491352<br>0.1429175<br>0.1267454<br>0.1267454<br>0.1267454<br>0.1267454<br>0.1267454<br>0.0169957<br>0.2231399<br>0.169957<br>0.025024<br>0<br>0<br>0.0480152<br>0.0025424<br>0.0025424<br>0.2073644<br>0.2073644<br>0.2073644<br>0.2073644<br>0.2063454<br>0.2063454<br>0.2063454<br>0.2063454<br>0.2063454<br>0.2063454<br>0.2063454<br>0.2063454<br>0.2063454<br>0.2063454<br>0.2063454<br>0.2063454<br>0.2063454<br>0.2063454<br>0.2063454<br>0.2063454<br>0.2063454<br>0.2063454<br>0.2063454<br>0.2063454<br>0.2063454<br>0.2063454<br>0.2063454<br>0.2063454<br>0.2063454<br>0.2063454<br>0.2063454<br>0.2063454<br>0.2063454<br>0.2063454<br>0.2063454<br>0.2063454<br>0.2063454<br>0.2063454<br>0.2063454<br>0.2063454<br>0.2063454<br>0.2063454<br>0.2063454<br>0.2063454<br>0.2063454<br>0.2063454<br>0.2063454<br>0.2063454<br>0.2063454<br>0.2063454<br>0.2063454<br>0.2063454<br>0.2063454<br>0.2063454<br>0.2063454<br>0.2063454<br>0.2063454<br>0.2063454<br>0.2063454<br>0.2063454<br>0.2063454<br>0.2063454<br>0.2063454<br>0.2063454<br>0.2063454<br>0.2063454<br>0.2063454<br>0.2063454<br>0.2063454<br>0.2063454<br>0.2063454<br>0.2063454<br>0.2063454<br>0.2063454<br>0.2063454<br>0.2063454<br>0.2063454<br>0.2063454<br>0.2063454<br>0.2063454<br>0.2063454<br>0.2063454<br>0.2063454<br>0.2063454<br>0.2063454<br>0.2063454<br>0.2063454<br>0.2063454<br>0.2063454<br>0.2063454<br>0.2063454<br>0.2063454<br>0.2063454<br>0.2063454<br>0.2063454<br>0.2063454<br>0.2063454<br>0.2063454<br>0.2063454<br>0.2063454<br>0.2063454<br>0.2063454<br>0.2063454<br>0.2063454<br>0.2063454<br>0.2063454<br>0.2063454<br>0.2063454<br>0.2063454<br>0.2063454<br>0.2063454<br>0.2063454<br>0.2063454<br>0.2063454<br>0.2063454<br>0.2063454<br>0.2063454<br>0.2063454<br>0.2063454<br>0.2063454<br>0.2063454<br>0.2063454<br>0.2063454<br>0.2063454<br>0.2063454<br>0.2063454<br>0.2063454<br>0.2063454<br>0.2063454<br>0.2063454<br>0.2063454<br>0.2063454<br>0.2063454<br>0.2063454<br>0.2063454<br>0.2063454<br>0.2063454<br>0.2063454<br>0.2063454<br>0.2063454<br>0.2063454<br>0.2063454<br>0.2063454<br>0.2063454<br>0.2063454<br>0.2063454<br>0.2063454<br>0.2063454<br>0.2063454<br>0.2063454<br>0.2063454<br>0.2063454<br>0.2063454<br>0.2063454<br>0.2063454<br>0.2063454<br>0.2063454<br>0.2063454<br>0.2063454<br>0.2063454<br>0.20635   | 9.431E-08<br>7.761E-08<br>8.278E-08<br>9.754E-08<br>9.754E-08<br>6.373E-08<br>6.373E-08<br>6.636E-08<br>6.737E-08<br>6.737E-08<br>6.737E-08<br>7.45E-08<br>7.152E-08<br>5.65E-08<br>7.12E-08<br>5.65E-08<br>7.258E-08<br>7.258E-08<br>3.369E-10<br>0<br>0<br>7.265E-08<br>3.369E-10<br>6.137E-08<br>9.357E-08<br>9.354E-08<br>9.354E-08  | 0.0001132<br>0.0001532<br>0.0001261<br>0.0001453<br>0.0001455<br>0.0001345<br>0.0001214<br>9.882E-05<br>0.0001212<br>0.00012214<br>0.00012214<br>0.00012214<br>0.0001212<br>0.0001221<br>0.000126<br>0.0001162<br>0.0001162<br>0.0001162<br>0.0001162<br>0.0001332<br>0.0001332<br>0.0001332<br>0.0001332<br>0.0001332<br>0.0001332<br>0.0001332<br>0.0001332<br>0.0001332<br>0.0001332<br>0.0001332<br>0.0001332<br>0.0001332<br>0.0001332   | 0.1203596<br>0.1142306<br>0.1310694<br>0.1216391<br>0.1426321<br>0.1426321<br>0.1426321<br>0.1345164<br>0.1323766<br>0.1323766<br>0.13233766<br>0.13233766<br>0.1208562<br>0.1208562<br>0.1208562<br>0.12084766<br>0.1624807<br>0<br>0.4438849<br>0.1559933<br>0.1559933<br>0.1559933<br>0.1873168<br>0.18710614  
  | 9.0011-08<br>7.322E-08<br>8.402E-08<br>9.146E-08<br>9.146E-08<br>9.146E-08<br>7.39F-08<br>7.39E-08<br>7.39E-08<br>7.39E-08<br>7.39E-08<br>6.439E-08<br>6.436E-08<br>3.34E-08<br>5.208E-08<br>6.438E-08<br>3.44E-10<br>0<br>0<br>0<br>5.73E-08<br>6.73B-08<br>6.73B-08<br>6.73B-08<br>6.73B-08<br>6.73B-08<br>5.342E-08<br>8.53E-08<br>8.532E-08<br>8.532E-08<br>8.532E-08   | 0.0001462<br>0.0001462<br>0.0001365<br>0.0001267<br>0.0001486<br>9.834E-05<br>0.0001486<br>9.834E-05<br>0.0001149<br>9.34E-05<br>0.0001149<br>0.0001032<br>0.0001043<br>0.0001032<br>0.0001032<br>0.0001045<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0   | 2.4904488<br>1.9842855<br>2.2002208<br>2.1078468<br>2.2364653<br>3.1057198<br>3.066346<br>2.9569573<br>2.8602462<br>3.1731567<br>2.620547<br>2.6202647<br>2.6309615<br>2.9177616<br>4.4828878<br>4.9637927<br>4.4562831<br>0<br>0.0573728<br>0<br>0<br>1.3610254<br>1.3610254<br>1.3610254<br>0.603969<br>0.6039622<br>0.5115727<br>0.6025855<br>0.5987613<br>0.6002684  
  | 1.863E-06<br>1.272E-06<br>1.351E-06<br>1.351E-06<br>1.32FE-06<br>1.628E-06<br>1.268E-06<br>1.268E-06<br>1.268E-06<br>1.268E-06<br>1.401E-06<br>1.401E-06<br>1.401E-06<br>1.401E-06<br>1.401E-06<br>1.401E-06<br>1.401E-06<br>1.401E-06<br>1.401E-06<br>1.401E-06<br>1.401E-06<br>1.401E-06<br>1.401E-06<br>1.401E-06<br>1.401E-06<br>1.401E-06<br>1.401E-06<br>1.401E-06<br>1.401E-06<br>1.401E-06<br>1.401E-06<br>1.401E-06<br>1.401E-06<br>1.401E-06<br>1.401E-06<br>1.401E-06<br>1.401E-06<br>1.401E-06<br>1.401E-06<br>1.401E-06<br>1.401E-06<br>1.401E-06<br>1.401E-06<br>1.401E-06<br>1.401E-06<br>1.401E-06<br>1.401E-06<br>1.401E-06<br>1.401E-06<br>1.401E-06<br>1.401E-06<br>1.401E-06<br>1.401E-06<br>1.401E-06<br>1.401E-06<br>1.401E-06<br>1.401E-06<br>1.401E-06<br>1.401E-06<br>1.401E-06<br>1.401E-06<br>1.401E-06<br>1.401E-06<br>1.401E-06<br>1.401E-06<br>1.401E-06<br>1.401E-06<br>1.401E-06<br>1.401E-06<br>1.401E-06<br>1.401E-06<br>1.401E-06<br>1.401E-06<br>1.401E-06<br>1.401E-06<br>1.401E-06<br>1.401E-06<br>1.401E-06<br>1.401E-06<br>1.401E-06<br>1.401E-06<br>1.401E-06<br>1.401E-06<br>1.501E-07<br>2.006E-07<br>2.006E-07<br>2.006E-07<br>2.706E-07<br>2.706E-07<br>2.74E-07<br>2.774E-07<br>2.774E-07<br>2.774E-07<br>2.774E-07<br>1.774E-07<br>1.774E-07<br>1.774E-07<br>1.774E-07<br>1.774E-07<br>1.774E-07<br>1.774E-07<br>1.774E-07<br>1.774E-07<br>1.774E-07<br>1.774E-07<br>1.774E-07<br>1.774E-07<br>1.774E-07<br>1.774E-07<br>1.774E-07<br>1.774E-07<br>1.774E-07<br>1.774E-07<br>1.774E-07<br>1.774E-07<br>1.774E-07<br>1.774E-07<br>1.774E-07<br>1.774E-07<br>1.774E-07<br>1.774E-07<br>1.774E-07<br>1.774E-07<br>1.774E-07<br>1.774E-07<br>1.774E-07<br>1.774E-07<br>1.774E-07<br>1.774E-07<br>1.774E-07<br>1.774E-07<br>1.774E-07<br>1.774E-07<br>1.774E-07<br>1.774E-07<br>1.774E-07<br>1.774E-07<br>1.774E-07<br>1.774E-07<br>1.774E-07<br>1.774E-07<br>1.774E-07<br>1.774E-07<br>1.774E-07<br>1.774E-07<br>1.774E-07<br>1.774E-07<br>1.774E-07<br>1.774E-07<br>1.774E-07<br>1.774E-07<br>1.774E-07<br>1.774E-07<br>1.774E-07<br>1.774E-07<br>1.774E-07<br>1.774E-07<br>1.774E-07<br>1.774E-07<br>1.774E-07<br>1.774E-07<br>1.774E-07<br>1.774E-07<br>1.774E-07<br>1.774E-07<br>1.774E-07<br>1.774E-07<br>1.774E-07<br>1.774E-07<br>1.774E-07<br>1.774E-07<br>1.774E-07<br>1.774E-07<br>1.774E-07<br>1.774E-07<br>1.774E-07<br>1.774E-07<br>1.774E-07<br>1.774E-07<br>1.774E-07<br>1.774E-07<br>1.774E-07<br>1.774E-07<br>1.774E-07<br>1.774E-07<br>1.774E-07<br>1     | 0.030261<br>0.002066<br>0.0022915<br>0.002193<br>0.002154<br>0.002549<br>0.002541<br>0.002544<br>0.002531<br>0.0024844<br>0.002754<br>0.00224497<br>0.0023206<br>1.592E-05<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0  | 2.3566842<br>1.8789288<br>2.0977776<br>2.0052501<br>2.1283629<br>2.9552749<br>2.9552749<br>2.90207<br>2.8217082<br>2.7301698<br>3.0191874<br>2.4772701<br>2.5033689<br>3.0311649<br>4.68371<br>4.68371<br>4.68371<br>4.6834133<br>0.0463027<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0   | 1.7622-06<br>1.204E-06<br>1.345E-06<br>1.345E-06<br>1.354E-06<br>1.268E-06<br>1.549E-06<br>1.549E-06<br>1.549E-06<br>1.549E-06<br>1.532E-06<br>1.337E-06<br>1.337E-06<br>1.337E-06<br>1.337E-06<br>1.336E-06<br>1.336E-06<br>1.336E-06<br>1.336E-06<br>1.336E-07<br>1.409E-07<br>2.516E-07<br>2.516E-07<br>2.516E-07<br>2.516E-07  |
0.0028636<br>0.0019569<br>0.0021848<br>0.00221848<br>0.00221848<br>0.0022184<br>0.0022151<br>0.002519<br>0.002519<br>0.002519<br>0.002519<br>0.0026204<br>0.0021727<br>0.002225<br>0.0012425<br>0.0024727<br>0.0022439<br>0.0020475<br>0<br>0<br>0.00024749<br>0.0002414<br>1.305E-05<br>0.0002515<br>0.000255<br>0.0002545<br>0.000255<br>0.000255<br>0.000255<br>0.000255<br>0.000255<br>0.000255<br>0.000255<br>0.000255<br>0.000255<br>0.000255<br>0.000255<br>0.000255<br>0.000255<br>0.000255<br>0.000255<br>0.000255<br>0.000255<br>0.000255<br>0.000255<br>0.000255<br>0.000255<br>0.000255<br>0.000255<br>0.000255<br>0.000255<br>0.000255<br>0.000255<br>0.000255<br>0.000255<br>0.000255<br>0.000255<br>0.000255<br>0.000255<br>0.000255<br>0.000255<br>0.000255<br>0.000255<br>0.000255<br>0.000255<br>0.000255<br>0.000255<br>0.00025<br>0.00025<br>0.00025<br>0.00025<br>0.00025<br>0.00025<br>0.00025<br>0.00025<br>0.00025<br>0.00025<br>0.00025<br>0.00025<br>0.00025<br>0.00025<br>0.00025<br>0.00025<br>0.00025<br>0.00025<br>0.00025<br>0.00025<br>0.00025<br>0.00025<br>0.00025<br>0.00025<br>0.00025<br>0.00025<br>0.00025<br>0.00025<br>0.00025<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005<br>0.0005                 | 2.2440968<br>1.7795138<br>1.9915216<br>1.8961342<br>2.0102391<br>2.7864095<br>2.720732<br>2.6462899<br>2.5622851<br>2.38364013<br>2.3221589<br>2.33726056<br>2.4375101<br>2.6016999<br>3.7301495<br>0<br>0<br>0<br>1.1398552<br>1.1398552<br>1.141438222<br>0.041725<br>0.4887676<br>0.43207676<br>0.43207676   | 1.6782-06<br>1.21412-06<br>1.21772-06<br>1.2152-06<br>1.2152-06<br>1.1312-06<br>1.1312-06<br>1.3026-06<br>1.3026-06<br>1.3026-06<br>1.2462-06<br>1.3026-06<br>1.2462-06<br>1.3026-06<br>1.3026-07<br>0<br>0<br>1.7318-07<br>1.7338-07<br>1.7338-07<br>1.5158-07<br>1.2888-07<br>2.2886-07<br>2.2828-07  | 0.0027267<br>0.0018534<br>0.0020742<br>0.0019748<br>0.0019748<br>0.0019748<br>0.0020937<br>0.0020937<br>0.0020243<br>0.0020243<br>0.0022238<br>0.002243<br>0.0022415<br>0.0020241<br>0.0020241<br>0.0020241<br>0.0020241<br>0.0002313<br>0.0020241<br>0.0002313<br>0.0002155<br>0.0002155<br>0.0002155<br>0.00023313<br>0.0002175<br>0.00002156<br>0.00002156<br>0.00002156<br>0.00002156<br>0.00002156<br>0.00002175<br>0.00002175<br>0.00002175<br>0.00002175<br>0.00002175<br>0.00002175<br>0.00002175<br>0.00002175<br>0.00002175<br>0.00002175<br>0.00002175<br>0.00002175<br>0.00002175<br>0.00002175<br>0.00002175<br>0.00002175<br>0.00002175<br>0.00002175<br>0.00002175<br>0.00002175<br>0.00002175<br>0.00002175<br>0.00002175<br>0.00002175<br>0.00002175<br>0.00002175<br>0.00002175<br>0.00002175<br>0.00002175<br>0.00002175<br>0.00002175<br>0.0000215<br>0.0000215<br>0.0000215<br>0.0000215<br>0.0000215<br>0.0000215<br>0.0000215<br>0.0000215<br>0.0000215<br>0.0000215<br>0.00002020<br>0.0000215<br>0.0000215<br>0.0000215<br>0.0000215<br>0.0000215<br>0.0000215<br>0.0000215<br>0.0000215<br>0.0000215<br>0.0000215<br>0.0000215<br>0.0000215<br>0.0000215<br>0.0000215<br>0.0000215<br>0.0000215<br>0.0000215<br>0.0000215<br>0.0000215<br>0.0000215<br>0.0000215<br>0.0000215<br>0.0000215<br>0.0000215<br>0.0000215<br>0.0000215<br>0.0000215<br>0.0000215<br>0.0000000000<br>0.0000000000000000000000   
   | 2.623364<br>2.11248<br>2.3459273<br>2.2439219<br>3.2614464<br>3.1928226<br>3.1051925<br>3.1051925<br>3.0057113<br>3.255455<br>2.7552623<br>2.7552623<br>2.7582603<br>2.3396521<br>3.04710013<br>3.3621382<br>0<br>0<br>0<br>0.<br>8903165<br>1.8903165<br>1.8903165<br>1.8903165<br>1.8903165<br>1.8903165<br>1.8903165<br>1.8903165<br>1.8903165<br>1.8903165<br>1.8903165<br>1.8903165<br>1.8903165<br>1.8903165<br>1.8903165<br>1.8903165<br>1.8903165<br>1.8903165<br>1.8903165<br>1.8903165<br>1.8903165<br>1.8903165<br>1.8903165<br>1.8903165<br>1.8903165<br>1.8903165<br>1.8903165<br>1.8903165<br>1.8903165<br>1.8903165<br>1.8903165<br>1.8903165<br>1.8903165<br>1.8903165<br>1.8903165<br>1.8903165<br>1.8903165<br>1.8903165<br>1.8903165<br>1.8903165<br>1.8903165<br>1.8903165<br>1.8903165<br>1.8903165<br>1.8903165<br>1.8903165<br>1.8903165<br>1.8903165<br>1.8903165<br>1.8903165<br>1.8903165<br>1.8903165<br>1.8903165<br>1.8903165<br>1.8903165<br>1.8903165<br>1.8903165<br>1.8903165<br>1.8903165<br>1.8903165<br>1.8903165<br>1.8903165<br>1.8903165<br>1.8903165<br>1.8903165<br>1.8903165<br>1.8903165<br>1.8903165<br>1.8903165<br>1.8903165<br>1.8903165<br>1.8903165<br>1.8903165<br>1.8903165<br>1.8903165<br>1.8903165<br>1.8903165<br>1.8903165<br>1.8903165<br>1.8903165<br>1.8903165<br>1.8903165<br>1.8903165<br>1.8903165<br>1.8903165<br>1.8903165<br>1.8903165<br>1.8903165<br>1.8903165<br>1.8903165<br>1.8903165<br>1.8903165<br>1.8903165<br>1.8903165<br>1.8903165<br>1.8903165<br>1.8903165<br>1.8903165<br>1.8903165<br>1.8903165<br>1.8903165<br>1.8903165<br>1.8903165<br>1.8903165<br>1.8903165<br>1.8903165<br>1.8903165<br>1.8903165<br>1.8903165<br>1.8903165<br>1.8903165<br>1.8903165<br>1.8903165<br>1.8903165<br>1.8903165<br>1.8903165<br>1.8903165<br>1.8903165<br>1.8903165<br>1.8903165<br>1.8903165<br>1.8903165<br>1.8903165<br>1.8903165<br>1.8903165<br>1.8903165<br>1.8903165<br>1.8903165<br>1.8903165<br>1.8903165<br>1.8903165<br>1.8903165<br>1.8903165<br>1.8903165<br>1.8903165<br>1.8903165<br>1.8903165<br>1.8903165<br>1.8903165<br>1.8903165<br>1.8903165<br>1.8903165<br>1.8903165<br>1.8903165<br>1.8903165<br>1.8903165<br>1.8903165<br>1.8903165<br>1.8903165<br>1.8903165<br>1.8903165<br>1.8903165<br>1.8903165<br>1.8903165<br>1.8903165<br>1.8903165<br>1.8903165<br>1.8903165<br>1.8903165<br>1.8903165<br>1.8903165<br>1.8903165<br>1.8903165<br>1.8903165<br>1.8903165<br>1.8903165<br>1.8903165<br>1.8903165<br>1.8903165<br>1.8903165<br>1.8903165<br>1.8903165<br>1.8903165<br>1.8903165<br>1.8903165<br>1.8903165<br>1.8903165<br>1.8903165<br>1.8903165<br>1.8903165<br>1.8903165<br>1.8903165<br>1.8903165<br>1.8903165<br>1.8903165<br>1.8903165<br>1.8903165<br>1.8903165<br>1.   | 19622.06<br>13542.06<br>1.3542.06<br>1.3542.06<br>1.3547.06<br>1.3547.06<br>1.3274.06<br>1.3274.06<br>1.3274.06<br>1.3274.06<br>1.4262.06<br>1.4262.06<br>1.4262.06<br>1.4262.06<br>1.4262.06<br>1.4262.06<br>1.4262.06<br>0<br>0<br>0<br>2.876.07<br>2.876.07<br>2.5176.07<br>2.5176.07<br>2.5176.07<br>2.5176.07<br>2.5176.07<br>2.5176.07<br>2.5176.07<br>2.5176.07<br>2.5176.07<br>2.5176.07<br>2.5176.07<br>2.5176.07<br>2.5176.07<br>2.5176.07<br>2.5176.07<br>2.5176.07<br>2.5176.07<br>2.5176.07<br>2.5176.07<br>2.5176.07<br>2.5176.07<br>2.5176.07<br>2.5176.07<br>2.5176.07<br>2.5176.07<br>2.5176.07<br>2.5176.07<br>2.5176.07<br>2.5176.07<br>2.5176.07<br>2.5176.07<br>2.5176.07<br>2.5176.07<br>2.5176.07<br>2.5176.07<br>2.5176.07<br>2.5176.07<br>2.5176.07<br>2.5176.07<br>2.5176.07<br>2.5176.07<br>2.5176.07<br>2.5176.07<br>2.5176.07<br>2.5176.07<br>2.5176.07<br>2.5176.07<br>2.5176.07<br>2.5176.07<br>2.5176.07<br>2.5176.07<br>2.5176.07<br>2.5176.07<br>2.5176.07<br>2.5176.07<br>2.5176.07<br>2.5176.07<br>2.5176.07<br>2.5176.07<br>2.5176.07<br>2.5176.07<br>2.5176.07<br>2.5176.07<br>2.5176.07<br>2.5176.07<br>2.5176.07<br>2.5176.07<br>2.5176.07<br>2.5176.07<br>2.5176.07<br>2.5176.07<br>2.5176.07<br>2.5176.07<br>2.5176.07<br>2.5176.07<br>2.5176.07<br>2.5176.07<br>2.5176.07<br>2.5176.07<br>2.5176.07<br>2.5176.07<br>2.5176.07<br>2.5176.07<br>2.5176.07<br>2.5176.07<br>2.5176.07<br>2.5176.07<br>2.5176.07<br>2.5176.07<br>2.5176.07<br>2.5176.07<br>2.5176.07<br>2.5176.07<br>2.5176.07<br>2.5176.07<br>2.5176.07<br>2.5176.07<br>2.5176.07<br>2.5176.07<br>2.5176.07<br>2.5176.07<br>2.5176.07<br>2.5176.07<br>2.5176.07<br>2.5176.07<br>2.5176.07<br>2.5176.07<br>2.5176.07<br>2.5176.07<br>2.5176.07<br>2.5176.07<br>2.5176.07<br>2.5176.07<br>2.5176.07<br>2.5176.07<br>2.5176.07<br>2.5176.07<br>2.5176.07<br>2.5176.07<br>2.5176.07<br>2.5176.07<br>2.5176.07<br>2.5176.07<br>2.5176.07<br>2.5176.07<br>2.5176.07<br>2.5176.07<br>2.5176.07<br>2.5176.07<br>2.5176.07<br>2.5176.07<br>2.5176.07<br>2.5176.07<br>2.5176.07<br>2.5176.07<br>2.5176.07<br>2.5176.07<br>2.5176.07<br>2.5176.07<br>2.5176.07<br>2.5176.07<br>2.5176.07<br>2.5176.07<br>2.5176.07<br>2.5176.07<br>2.5176.07<br>2.5176.07<br>2.5176.07<br>2.5176.07<br>2.5176.07<br>2.5176.07<br>2.5176.07<br>2.5176.07<br>2.5176.07<br>2.5176.07<br>2.5176.07<br>2.5176.07<br>2.5176.07<br>2.5176.07<br>2.5176.07<br>2.5176.07<br>2.5176.07<br>2.5176.07<br>2.5176.07<br>2.5176.07<br>2.5176.07<br>2.5176   | 0.0031876<br>0.0022403<br>0.002243<br>0.002337<br>0.002347<br>0.0022645<br>0.0027511<br>0.0025645<br>0.0027511<br>0.002687<br>0.0028687<br>0.002454<br>0.0024545<br>0.0022716<br>0.0022716<br>0.0022716<br>0.0022716<br>0.0022716<br>0.002245<br>0<br>0.0005245<br>0.00035245<br>0.00035245<br>0.00035245<br>0.00035245<br>0.00035245<br>0.00035245<br>0.00035245<br>0.00035245<br>0.00035245<br>0.00035245<br>0.00035245<br>0.00035245  |
2.3644564<br>1.8937445<br>2.1225909<br>2.0177733<br>2.1529112<br>2.928047<br>2.8544498<br>2.7808064<br>2.6943024<br>2.9747351<br>2.4523139<br>2.4915395<br>2.4523139<br>2.4523139<br>2.4523139<br>2.4523139<br>2.4523139<br>2.4523139<br>2.4523139<br>2.4523139<br>2.583663<br>2.724646<br>0.0433818<br>0<br>0.15831454<br>1.585267<br>0.043982567<br>0.043982567<br>0.043982567<br>0.043982567<br>0.043982567<br>0.043982567<br>0.043982567<br>0.043982567<br>0.043982567<br>0.043982567<br>0.043982567<br>0.043982567<br>0.043982567<br>0.043982567<br>0.043982567<br>0.043982567<br>0.043982567<br>0.043982567<br>0.043982567<br>0.043982567<br>0.043982567<br>0.043982567<br>0.043982567<br>0.043982567<br>0.043982567<br>0.043982567<br>0.043982567<br>0.043982567<br>0.043982567<br>0.043982567<br>0.043982567<br>0.043982567<br>0.043982567<br>0.043982567<br>0.043982567<br>0.043982567<br>0.043982567<br>0.043982567<br>0.043982567<br>0.043982567<br>0.043982567<br>0.043982567<br>0.043982567<br>0.043982567<br>0.043982567<br>0.043982567<br>0.043982567<br>0.043982567<br>0.043982567<br>0.043982567<br>0.043982567<br>0.043982567<br>0.043982567<br>0.043982567<br>0.043982567<br>0.043982567<br>0.043982567<br>0.043982567<br>0.043982567<br>0.043982567<br>0.043982567<br>0.043982567<br>0.043982567<br>0.043982567<br>0.043982567<br>0.043982567<br>0.043982567<br>0.043982567<br>0.043982567<br>0.043982567<br>0.043982567<br>0.043982567<br>0.043982567<br>0.043982567<br>0.043982567<br>0.043982567<br>0.043982567<br>0.043982567<br>0.043982567<br>0.043982567<br>0.043982567<br>0.043982567<br>0.043982567<br>0.043982567<br>0.043982567<br>0.043982567<br>0.043982567<br>0.043982567<br>0.043982567<br>0.043982567<br>0.043982567<br>0.043985567<br>0.043985567<br>0.043985567<br>0.043985567<br>0.043985567<br>0.04398557<br>0.04398557<br>0.04398557<br>0.04398557<br>0.04398557<br>0.04398557<br>0.04398557<br>0.04398557<br>0.04398557<br>0.04398557<br>0.04398557<br>0.06395757<br>0.06395757<br>0.06395757<br>0.06395757<br>0.06595757<br>0.05575757<br>0.05575757<br>0.05575775757575757575757575757575757575  | 1.7688-00 0.002<br>1.2446-00 0.002<br>1.3516-00 0.002<br>1.3816-00 0.002<br>1.3816-00 0.002<br>1.5258-00 0.002<br>1.5258-00 0.002<br>1.5258-00 0.002<br>1.5258-00 0.002<br>1.318-00 0.002<br>1.318-00 0.002<br>1.318-00 0.002<br>1.318-00 0.002<br>0 0 0<br>0     | 873         0.44           9723         0.24           0.2107         0.044           1015         0.03           0.33         0.55           9308         0.05           93384         0.044           1625         0.044           1624         0.05           1284         0.044           1284         0.044           11585         0.08           1285         0.08           1284         0.044           11585         0.08           14393         0.0           14393         0.0           14393         0.0           15737         0.02           25721         0.02           25731         0.02  
  | 147043         3.           147043         3.           383822         2.           408597         1           313052         2.           338011         2.           338012         2.           538011         2.           538011         2.           5511         3.           157074         2.4           167677         2.4           164254         2.           5138381         1.6           1776164         2.4           164254         2.           19282         6.4           153993         8.1           0         0           0         1.4           14252         6.4           15393         8.1           104252         6.4           15393         8.1           10452         5.1           10462         9.5           10642         9.5           10642         9.5           10642         9.5   
   | 5.18E-08         2           6.619E-08         2           6.619E-08         4           2.90E-08         3           2.99E-08         3           3.776E-08         4           6.512-08         4           6.512-08         4           6.512-08         4           6.512-08         4           6.612-08         2           9.86E-08         4           6.612-08         2           0         0           0         0           0         0           456E-09         1           1334E-09         1           334E-09         1           534E-04         1           534E-04         1           334E-05         1           534E-04         1           534E-09         1           534E-09         1           534E-09         1           534E-09         1           534E-09         1           534E-09         1   | 3.998E-05<br>4.254E-05<br>4.224E-05<br>4.224E-05<br>4.311E-05<br>3.736E-05<br>4.3776E-05<br>4.307E-05<br>4.307E-05<br>4.307E-05<br>4.307E-05<br>4.307E-05<br>4.307E-05<br>4.307E-05<br>4.307E-05<br>1.042E-05<br>1.266E-05<br>1.18E-05<br>1.18E-05<br>1.18E-05<br>1.18E-05<br>1.18E-05<br>1.18E-05<br>1.18E-05<br>1.18E-05<br>1.18E-05<br>1.18E-05<br>1.18E-05<br>1.18E-05<br>1.18E-05<br>1.18E-05<br>1.18E-05<br>1.18E-05<br>1.18E-05<br>1.18E-05<br>1.18E-05<br>1.18E-05<br>1.18E-05<br>1.18E-05<br>1.18E-05<br>1.18E-05<br>1.158E-05<br>1.158E-05<br>1.754E-05<br>1.754E-05<br>1.754E-05  |
0.041622<br>0.0443322<br>0.044017<br>0.0443323<br>0.0548313<br>0.0558211<br>0.0558211<br>0.053821<br>0.053821<br>0.0537781<br>0.0527943<br>0.0527943<br>0.0643226<br>0.0057794<br>0<br>0.04452<br>0<br>0.0457244<br>0<br>0.04577944<br>0<br>0.0457744<br>0<br>0.0457744<br>0<br>0.0457744<br>0<br>0.0457744<br>0<br>0.0457744<br>0<br>0.0457744<br>0<br>0.0457744<br>0<br>0.0457744<br>0<br>0.0457744<br>0<br>0.0457744<br>0<br>0.0457744<br>0<br>0.0457744<br>0<br>0.0457744<br>0<br>0.0457744<br>0<br>0.0457744<br>0<br>0.0457744<br>0<br>0.0457744<br>0<br>0.0457744<br>0<br>0.0457744<br>0<br>0.0457744<br>0<br>0.0457744<br>0<br>0.0457744<br>0<br>0.0457744<br>0<br>0.04223<br>0<br>0.04223<br>0<br>0.04223<br>0<br>0.04223<br>0<br>0.04223<br>0<br>0.04223<br>0<br>0.04223<br>0<br>0.04223<br>0<br>0.04223<br>0<br>0.04223<br>0<br>0.04223<br>0<br>0.04223<br>0<br>0.04223<br>0<br>0.04223<br>0<br>0.04243<br>0<br>0.055243<br>0<br>0.055221<br>0.055243<br>0<br>0.055243<br>0<br>0.0552743<br>0.055243<br>0.0552744<br>0.0552744<br>0.0552745<br>0.0552745<br>0.0552745<br>0.0552745<br>0.0552745<br>0.0552745<br>0.0552745<br>0.0552745<br>0.0552745<br>0.0552745<br>0.0552745<br>0.0552745<br>0.0552745<br>0.0552745<br>0.0552745<br>0.0552745<br>0.0552745<br>0.0552745<br>0.0552745<br>0.0552745<br>0.0552745<br>0.0552745<br>0.0557745<br>0.0557745<br>0.0557745<br>0.0557745<br>0.0557745<br>0.0557745<br>0.0557745<br>0.0557745<br>0.0557745<br>0.0557745<br>0.0557745<br>0.0557745<br>0.0557745<br>0.0557745<br>0.0557745<br>0.0557745<br>0.0557745<br>0.0557745<br>0.0557745<br>0.0557745<br>0.0557745<br>0.0557745<br>0.0557745<br>0.00577745<br>0.00577745<br>0.00577745<br>0.00577745<br>0.00577745<br>0.00577745<br>0.00577745<br>0.00577545<br>0.00577545<br>0.00577545<br>0.00577545<br>0.00577545<br>0.00577545<br>0.00577545<br>0.00577545<br>0.00577545<br>0.00577545<br>0.00577545<br>0.00577545<br>0.00577545<br>0.00577545<br>0.00577545<br>0.00577545<br>0.00577545<br>0.00577545<br>0.00577545<br>0.00577545<br>0.00577545<br>0.00577545<br>0.00577545<br>0.00577545<br>0.00577545<br>0.00577545<br>0.00577545<br>0.00577545<br>0.00577545<br>0.00577545<br>0.00577545<br>0.00577545<br>0.00577545<br>0.00577545<br>0.00577545<br>0.00577545<br>0.00577545<br>0.00577545<br>0.00577545<br>0.00577545<br>0.00577545<br>0.00577545<br>0.00577545<br>0.00577545<br>0.00577545<br>0.005775500000000000000000000000000000  | 2.01-00 - 2.201-00 - 2   | 1.584E-05<br>1.697E-05<br>1.057E-05<br>1.677E-05<br>1.307E-05<br>1.307E-05<br>1.307E-05<br>1.307E-05<br>1.307E-05<br>1.307E-05<br>1.391E-05<br>1.582E-05<br>1.604E-06<br>0<br>0<br>0<br>1.282E-05<br>1.282E-05<br>1.282E-05<br>1.282E-05<br>1.255E-05<br>1.00E-05<br>1.901E-05<br>1.901E-05<br>1.901E-05   |
| Sheung Tsuen B/T and Car Park<br>Sheung Tsuen B/T and Car Park<br>Village Access Road<br>Village Access Road   | RD_29<br>RD_29<br>RD_29<br>RD_29<br>RD_29<br>RD_29<br>RD_29<br>RD_29<br>RD_29<br>RD_29<br>RD_29<br>RD_29<br>RD_29<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20 | 9900-1000           1000-1100           1100-1200           1200-1300           1200-1300           1200-1400           1200-100           1400-1500           1500-1600           1600-1700           1700-1800           1800-2000           2000-2100           2100-2200           2200-2300           0300-4010           0100-200           0300-4000           6600-7070           0700-6800           0800-9800           0800-9000           1000-1100           1000-1000           1000-1000           1000-1000           1000-1000           1000-1000           1000-1000           1000-1000           1000-1000           1000-1000   | 0.1281945<br>0.1457065<br>0.1457065<br>0.14597263<br>0.1557266<br>0.1492353<br>0.1482353<br>0.1482452<br>0.1352848<br>0.132275<br>0.1323254<br>0.1323254<br>0.1323254<br>0.1324202<br>0.1362322<br>0.030941<br>0<br>0<br>0<br>5.292911<br>0.0352929<br>0.2230554<br>0.2345426<br>0.2234958<br>0.2234958<br>0.22249201<br>0.3075582<br>0.3167258  | 9,342-003  <br>9,342-003  <br>9,342-003  <br>9,342-003  <br>1,032-07 (<br>6,6552-08 (<br>6,6552-08 (<br>7,792-08 (<br>6,3325-08 (<br>7,7714-08 (<br>6,9332-08 (<br>7,7714-08 (<br>6,9332-08 (<br>7,7714-08 (<br>6,9332-08 (<br>7,7069-08 (<br>6,9332-08 (<br>7,7069-08 (<br>6,932-08 (<br>7,7069-08 (<br>0,0360-08 (<br>0   | 0.0001615 (<br>0.0001335 (<br>0.0001337 (<br>0.000147 (<br>0.000147 (<br>0.0001674 (<br>0.0001674 (<br>0.0001674 (<br>0.0001674 (<br>0.0001674 (<br>0.0001674 (<br>0.0001675 (<br>0.000167  |
0.1261103<br>0.1210641<br>0.1385867<br>0.1291377<br>0.1291377<br>0.1491352<br>0.1491352<br>0.1491352<br>0.1491352<br>0.1491352<br>0.1491352<br>0.1491352<br>0.1420175<br>0.1267454<br>0.1267454<br>0.1267454<br>0.1267454<br>0.1267454<br>0.1267457<br>0.2231399<br>0.762915<br>0.025024<br>0<br>0<br>0.480112<br>0.478453<br>0.0025482<br>0.0025482<br>0.0025482<br>0.201416<br>0.1703611<br>0.1703614<br>0.2053644<br>0.2063459<br>0.2063654<br>0.2063652<br>0.2063652<br>0.2063652<br>0.2063652<br>0.2063652<br>0.2063652<br>0.2063652<br>0.2063652<br>0.2063652<br>0.2063652<br>0.2063652<br>0.2063652<br>0.2063652<br>0.20652<br>0.2063652<br>0.20652<br>0.2063652<br>0.20652<br>0.20652<br>0.20652<br>0.20652<br>0.20652<br>0.20652<br>0.20652<br>0.20652<br>0.20652<br>0.20652<br>0.20652<br>0.20652<br>0.20652<br>0.20652<br>0.20652<br>0.20652<br>0.20652<br>0.20652<br>0.20652<br>0.20652<br>0.20652<br>0.20652<br>0.20652<br>0.20652<br>0.20652<br>0.20652<br>0.20652<br>0.20652<br>0.20652<br>0.20652<br>0.20652<br>0.20652<br>0.20652<br>0.20652<br>0.20652<br>0.20652<br>0.20652<br>0.20652<br>0.20652<br>0.20652<br>0.20652<br>0.20652<br>0.20652<br>0.20652<br>0.20652<br>0.20652<br>0.20652<br>0.20652<br>0.20652<br>0.20652<br>0.20652<br>0.20652<br>0.20652<br>0.20652<br>0.20652<br>0.20652<br>0.20652<br>0.20652<br>0.20652<br>0.20652<br>0.20652<br>0.20652<br>0.20652<br>0.20652<br>0.20652<br>0.20652<br>0.20652<br>0.20652<br>0.20652<br>0.20652<br>0.20652<br>0.20652<br>0.20652<br>0.20652<br>0.20652<br>0.20652<br>0.20652<br>0.20652<br>0.20652<br>0.20652<br>0.20652<br>0.20652<br>0.20652<br>0.20652<br>0.20652<br>0.20652<br>0.20652<br>0.20652<br>0.20652<br>0.20652<br>0.20652<br>0.20652<br>0.20652<br>0.20652<br>0.20652<br>0.20652<br>0.20652<br>0.20652<br>0.20652<br>0.20652<br>0.20652<br>0.20652<br>0.20652<br>0.20652<br>0.20652<br>0.20652<br>0.20652<br>0.20652<br>0.20652<br>0.20652<br>0.20652<br>0.20552<br>0.20552<br>0.20552<br>0.20552<br>0.20552<br>0.20552<br>0.20552<br>0.20552<br>0.20552<br>0.20552<br>0.20552<br>0.20552<br>0.20552<br>0.20552<br>0.20552<br>0.20552<br>0.20552<br>0.20552<br>0.20552<br>0.20552<br>0.20552<br>0.20552<br>0.20552<br>0.20552<br>0.20552<br>0.20552<br>0.20552<br>0.20552<br>0.20552<br>0.20552<br>0.20552<br>0.20552<br>0.20552<br>0.20552<br>0.20552<br>0.20552<br>0.20552<br>0.20552<br>0.20552<br>0.20552<br>0.20552<br>0.20552<br>0.20552<br>0.20552<br>0.20552<br>0.20552<br>0.205552<br>0.205552<br>0.2055552<br>0.2055555<br>0.2055555<br>0.2055555555<br>0.20555555555555555   | 9.431E.08<br>7.761E.08<br>8.284E.08<br>9.754E.08<br>6.373E.08<br>7.473E.08<br>6.032E.08<br>7.46E.08<br>6.731E.08<br>6.731E.08<br>6.731E.08<br>6.731E.08<br>6.737E.08<br>4.117E.08<br>6.655E.08<br>5.447E.08<br>7.152E.08<br>5.655E.08<br>3.799E.10<br>0<br>0<br>7.289E.08<br>5.173E.08<br>5.173E.08<br>5.173E.08<br>5.173E.08<br>9.354E.08<br>9.354E.08<br>9.354E.08<br>9.355E.08  | 0.0001532<br>0.0001532<br>0.0001261<br>0.0001261<br>0.0001345<br>0.0001345<br>0.000135<br>0.0001212<br>0.0001212<br>0.0001212<br>0.000110<br>0.00010095<br>6.944E-07<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0  | 0.1203596<br>0.1142306<br>0.1142306<br>0.11426391<br>0.1426721<br>0.1426721<br>0.1426721<br>0.1323766<br>0.1323766<br>0.1323766<br>0.1323766<br>0.1323730<br>0.128338<br>0.120155<br>0.129461<br>0.1229461<br>0.1229461<br>0.1694062<br>0.01584645<br>0.01858455<br>0.1873168<br>0.1873168<br>0.18771614<br>0.187000   
   | 9.0011-08<br>7.322E-08<br>8.402E-08<br>7.797E-08<br>9.146E-08<br>6.033E-08<br>7.071E-08<br>5.749E-08<br>7.052E-08<br>7.052E-08<br>6.439E-08<br>6.436E-08<br>3.941E-08<br>6.435E-08<br>3.941E-08<br>3.941E-08<br>3.941E-08<br>3.941E-08<br>3.941E-08<br>3.42E-10<br>0<br>0<br>0<br>6.73E-08<br>3.42E-10<br>0<br>5.73E-08<br>8.42E-08<br>5.43E-08<br>3.42E-10<br>0<br>5.644E-08<br>8.532E-08<br>8.532E-08<br>8.532E-08<br>8.532E-08<br>8.532E-08<br>8.532E-08   | 0.0001462<br>0.0001362<br>0.0001365<br>0.0001267<br>0.0001267<br>0.0001486<br>9.834F-05<br>0.0001146<br>0.0001146<br>0.0001146<br>0.0001146<br>0.0001043<br>0.0001043<br>0.0001043<br>0.0001139<br>8.451F-05<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0  | 2.4904488<br>1.9842855<br>2.2002208<br>2.1078468<br>2.2364653<br>3.1057198<br>3.0463346<br>2.9569573<br>2.6202647<br>2.6509615<br>2.7073267<br>2.9117616<br>4.1828878<br>4.9637927<br>4.4562831<br>0.0573728<br>0<br>0<br>0<br>1.3610254<br>0.06059659<br>0.60025855<br>0.5987613<br>0.60025815   |
1.8638-06<br>1.272E-06<br>1.41E-06<br>1.351E-06<br>1.434E-06<br>1.327E-06<br>1.628E-06<br>1.264E-06<br>1.264E-06<br>1.264E-06<br>1.446E-06<br>9.333E-07<br>1.341E-06<br>1.446E-06<br>9.333E-07<br>2.066E-07<br>9.2E-09<br>1.834E-07<br>1.534E-06<br>1.532E-06<br>1.532E-07<br>2.745E-07<br>2.774E-07<br>2.774E-07<br>2.774E-07<br>2.774E-07<br>2.774E-07<br>2.774E-07<br>2.774E-07<br>2.774E-07<br>2.774E-07<br>2.774E-07<br>2.774E-07<br>2.774E-07<br>2.774E-07<br>2.774E-07<br>2.774E-07<br>2.774E-07<br>2.774E-07<br>2.774E-07<br>2.774E-07<br>2.774E-07<br>2.774E-07<br>2.774E-07<br>2.774E-07<br>2.774E-07<br>2.774E-07<br>2.774E-07<br>2.774E-07<br>2.774E-07<br>2.774E-07<br>2.774E-07<br>2.774E-07<br>2.774E-07<br>2.774E-07<br>2.774E-07<br>2.774E-07<br>2.774E-07<br>2.774E-07<br>2.774E-07<br>2.774E-07<br>2.774E-07<br>2.774E-07<br>2.774E-07<br>2.774E-07<br>2.774E-07<br>2.774E-07<br>2.774E-07<br>2.774E-07<br>2.774E-07<br>2.774E-07<br>2.774E-07<br>2.774E-07<br>2.774E-07<br>2.774E-07<br>2.774E-07<br>2.774E-07<br>2.774E-07<br>2.774E-07<br>2.774E-07<br>2.774E-07<br>2.774E-07<br>2.774E-07<br>2.774E-07<br>2.774E-07<br>2.774E-07<br>2.774E-07<br>2.774E-07<br>2.774E-07<br>2.774E-07<br>2.774E-07<br>2.774E-07<br>2.774E-07<br>2.774E-07<br>2.774E-07<br>2.774E-07<br>2.774E-07<br>2.774E-07<br>2.774E-07<br>2.774E-07<br>2.774E-07<br>2.774E-07<br>2.774E-07<br>2.774E-07<br>2.774E-07<br>2.774E-07<br>2.774E-07<br>2.774E-07<br>2.774E-07<br>2.774E-07<br>2.774E-07<br>2.774E-07<br>2.774E-07<br>2.774E-07<br>2.774E-07<br>2.774E-07<br>2.774E-07<br>2.774E-07<br>2.774E-07<br>2.774E-07<br>2.774E-07<br>2.774E-07<br>2.774E-07<br>2.774E-07<br>2.774E-07<br>2.774E-07<br>2.774E-07<br>2.774E-07<br>2.774E-07<br>2.774E-07<br>2.774E-07<br>2.774E-07<br>2.774E-07<br>2.774E-07<br>2.774E-07<br>2.774E-07<br>2.774E-07<br>2.774E-07<br>2.774E-07<br>2.774E-07<br>2.774E-07<br>2.774E-07<br>2.774E-07<br>2.774E-07<br>2.774E-07<br>2.774E-07<br>2.774E-07<br>2.774E-07<br>2.774E-07<br>2.774E-07<br>2.774E-07<br>2.774E-07<br>2.774E-07<br>2.774E-07<br>2.774E-07<br>2.774E-07<br>2.774E-07<br>2.774E-07<br>2.774E-07<br>2.774E-07<br>2.774E-07<br>2.774E-07<br>2.774E-07<br>2.774E-07<br>2.774E-07<br>2.774E-07<br>2.774E-07<br>2.774E-07<br>2.774E-07<br>2.774E-07<br>2.774E-07<br>2.774E-07<br>2.774E-07<br>2.774E-07<br>2.774E-07<br>2.774E-07<br>2.774E-07<br>2.774E-07<br>2.774E-07<br>2.774E-07<br>2.774E-07<br>2.774E-07<br>2.774E-07<br>2.774E-07<br>2.774E-07<br>2.774E-07<br>2.7     | 0.003066<br>0.002265<br>0.0022915<br>0.0021953<br>0.0023393<br>0.0021953<br>0.0023493<br>0.002484<br>0.0022531<br>0.0024824<br>0.002754<br>0.0023754<br>0.0023754<br>0.002308<br>0.0021754<br>0.0023849<br>0.00025849<br>0.0003777<br>0.00033777<br>1.6816-05<br>0.0003777<br>1.6816-05<br>0.0003777<br>0.0003839<br>0.0003839<br>0.0005516<br>0.0004984<br>0.0004984<br>0.0004984<br>0.0004989   | 2.35668.2<br>1.8789238<br>2.0977776<br>2.0052501<br>2.1283629<br>2.952749<br>2.90207<br>2.8217082<br>2.7301688<br>2.7301688<br>2.5033689<br>2.5033689<br>2.5033689<br>2.503787<br>4.688413<br>0.046307<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0  | 1.7622-06<br>1.2642-06<br>1.2652-06<br>1.2852-06<br>1.2652-06<br>1.2632-06<br>1.2632-06<br>1.2632-06<br>1.2632-06<br>1.2632-06<br>1.3272-06<br>1.3372-06<br>1.3372-06<br>1.3372-06<br>1.3372-06<br>1.3372-06<br>1.3372-06<br>1.3372-06<br>1.3372-06<br>1.3372-06<br>1.3362-06<br>1.3362-06<br>1.3362-06<br>1.3362-07<br>1.8742-07<br>1.4092-07<br>2.55162-07<br>2.55162-07<br>2.55162-07<br>2.55162-07<br>2.55162-07<br>2.55162-07<br>2.55162-07<br>2.55162-07<br>2.55162-07<br>2.55162-07<br>2.55162-07<br>2.55162-07<br>2.55162-07<br>2.55162-07<br>2.55162-07<br>2.55162-07<br>2.55162-07<br>2.55162-07<br>2.55162-07<br>2.55162-07<br>2.55162-07<br>2.55162-07<br>2.55162-07<br>2.55162-07<br>2.55162-07<br>2.55162-07<br>2.55162-07<br>2.55162-07<br>2.55162-07<br>2.55162-07<br>2.55162-07<br>2.55162-07<br>2.55162-07<br>2.55162-07<br>2.55162-07<br>2.55162-07<br>2.55162-07<br>2.55162-07<br>2.55162-07<br>2.55162-07<br>2.55162-07<br>2.55162-07<br>2.55162-07<br>2.55162-07<br>2.55162-07<br>2.55162-07<br>2.55162-07<br>2.55162-07<br>2.55162-07<br>2.55162-07<br>2.55162-07<br>2.55162-07<br>2.55162-07<br>2.55162-07<br>2.55162-07<br>2.55162-07<br>2.55162-07<br>2.55162-07<br>2.55162-07<br>2.55162-07<br>2.55162-07<br>2.55162-07<br>2.55162-07<br>2.55162-07<br>2.55162-07<br>2.55162-07<br>2.55162-07<br>2.55162-07<br>2.55162-07<br>2.55162-07<br>2.55162-07<br>2.55162-07<br>2.55162-07<br>2.55162-07<br>2.55162-07<br>2.55162-07<br>2.55162-07<br>2.55162-07<br>2.55162-07<br>2.55162-07<br>2.55162-07<br>2.55162-07<br>2.55162-07<br>2.55162-07<br>2.55162-07<br>2.55162-07<br>2.55162-07<br>2.55162-07<br>2.55162-07<br>2.55162-07<br>2.55162-07<br>2.55162-07<br>2.55162-07<br>2.55162-07<br>2.55162-07<br>2.55162-07<br>2.55162-07<br>2.55162-07<br>2.55162-07<br>2.55162-07<br>2.55162-07<br>2.55162-07<br>2.55162-07<br>2.55162-07<br>2.55162-07<br>2.55162-07<br>2.55162-07<br>2.55162-07<br>2.55162-07<br>2.55162-07<br>2.55162-07<br>2.55162-07<br>2.55162-07<br>2.55162-07<br>2.55162-07<br>2.55162-07<br>2.55162-07<br>2.55162-07<br>2.55162-07<br>2.55162-07<br>2.55162-07<br>2.55162-07<br>2.55162-07<br>2.55162-07<br>2.55162-07<br>2.55162-07<br>2.55162-07<br>2.55162-07<br>2.55162-07<br>2.55162-07<br>2.55162-07<br>2.55162-07<br>2.55162-07<br>2.55162-07<br>2.55162-07<br>2.55162-07<br>2.55162-07<br>2.55162-07<br>2.55162-07<br>2.55162-07<br>2.55162-07<br>2.55162-07<br>2.55162-07<br>2.55162-07<br>2.55162-07<br>2.55162-07<br>2.55162-   | 0.0028636<br>0.0019569<br>0.0021848<br>0.0022167<br>0.0022085<br>0.0022167<br>0.002519<br>0.002519<br>0.002519<br>0.002519<br>0.002519<br>0.002519<br>0.0021709<br>0.0021709<br>1.285E05<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0  
  | 2.2440968<br>1.7795138<br>1.9915216<br>1.8961342<br>2.0102391<br>2.7864095<br>2.7220732<br>2.6462899<br>2.5622851<br>2.8364013<br>2.3321589<br>2.3321589<br>2.33726056<br>2.4375101<br>2.6016999<br>3.7301495<br>4.4767836<br>0<br>0<br>0<br>0<br>0<br>0<br>1.1398552<br>0.44767836<br>0.041725<br>0.4987676<br>0.4240418<br>0.05019047<br>0.5019047<br>0.5019047<br>0.5019047<br>0.5019047<br>0.5019047<br>0.5019047<br>0.5019047<br>0.5019047<br>0.5019047<br>0.5019047<br>0.5019047<br>0.5019047<br>0.5019047<br>0.5019047<br>0.5019047<br>0.5019047<br>0.5019047<br>0.5019047<br>0.5019047<br>0.5019047<br>0.5019047<br>0.5019047<br>0.5019047<br>0.5019047<br>0.5019047<br>0.5019047<br>0.5019047<br>0.5019047<br>0.5019047<br>0.5019047<br>0.5019047<br>0.5019047<br>0.5019047<br>0.5019047<br>0.5019047<br>0.5019047<br>0.5019047<br>0.5019047<br>0.5019047<br>0.5019047<br>0.5019047<br>0.5019047<br>0.5019047<br>0.5019047<br>0.5019047<br>0.5019047<br>0.5019047<br>0.5019047<br>0.5019047<br>0.5019047<br>0.5019047<br>0.5019047<br>0.5019047<br>0.5019047<br>0.5019047<br>0.5019047<br>0.5019047<br>0.5019047<br>0.5019047<br>0.5019047<br>0.5019047<br>0.5019047<br>0.5019047<br>0.5019047<br>0.5019047<br>0.5019047<br>0.5019047<br>0.5019047<br>0.5019047<br>0.5019047<br>0.5019047<br>0.5019047<br>0.5019047<br>0.5019047<br>0.5019047<br>0.5019047<br>0.5019047<br>0.5019047<br>0.5019047<br>0.5019047<br>0.5019047<br>0.5019047<br>0.5019047<br>0.5019047<br>0.5019047<br>0.5019047<br>0.5019047<br>0.5019047<br>0.5019047<br>0.5019047<br>0.5019047<br>0.5019047<br>0.5019047<br>0.5019047<br>0.5019047<br>0.5019047<br>0.5019047<br>0.5019047<br>0.5019047<br>0.5019047<br>0.5019047<br>0.5019047<br>0.5019047<br>0.5019047<br>0.5019047<br>0.5019047<br>0.5019047<br>0.5019047<br>0.5019047<br>0.5019047<br>0.5019047<br>0.5019047<br>0.5019047<br>0.5019047<br>0.5019047<br>0.5019047<br>0.5019047<br>0.5019047<br>0.5019047<br>0.5019047<br>0.5019047<br>0.5019047<br>0.5019047<br>0.5019047<br>0.5019047<br>0.5019047<br>0.5019047<br>0.5019047<br>0.5019047<br>0.5019047<br>0.5019047<br>0.5019047<br>0.5019047<br>0.5019047<br>0.5019047<br>0.5019047<br>0.5019047<br>0.5019047<br>0.5019047<br>0.5019047<br>0.5019047<br>0.5019047<br>0.5019047<br>0.5019047<br>0.5019047<br>0.5019047<br>0.5019047<br>0.5019047<br>0.5019047<br>0.5019047<br>0.5019047<br>0.5019047<br>0.5019047<br>0.5019047<br>0.5019047<br>0.5019047<br>0.5019047<br>0.5019047<br>0.5019047<br>0.5019047<br>0.5019047<br>0.5019047<br>0.5019047<br>0.5019047<br>0.5019047<br>0.5019047<br>0.5019047<br>0.5019047<br>0.5019047<br>0.5019047<br>0.5019047<br>0.5019047<br>0.5019047<br>0.5019047<br>0.5019047<br>0.5019047<br>0.5019047<br>0.5019047<br>0.5019047<br>0.5019047<br>0.   | 1.6782-06<br>1.1412-06<br>1.2797-06<br>1.2152-06<br>1.2152-06<br>1.1912-06<br>1.1912-06<br>1.1312-06<br>1.1312-06<br>1.1312-06<br>1.1312-06<br>1.132462-06<br>1.3022-06<br>8.3392-07<br>1.13662-06<br>1.2742-06<br>0<br>0<br>0<br>1.7312-07<br>1.7332-07<br>6.3352-09<br>1.5155-07<br>1.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2822-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07   | 0.0027267<br>0.0018534<br>0.0020742<br>0.0019748<br>0.0020937<br>0.0023937<br>0.0023625<br>0.0023238<br>0.0022238<br>0.0022238<br>0.0022421<br>0.0022421<br>0.0020241<br>0.0020242<br>0.0020242<br>1.0020242<br>1.14E-05<br>0<br>0<br>0.0003163<br>1.158E-05<br>0.0002758<br>0.0002758<br>0.0002758<br>0.00024175<br>0.0002758<br>0.0002758<br>0.0002758<br>0.00004175<br>0.00004175<br>0.0002758   | 2.623364<br>2.12248<br>2.3459273<br>2.2439219<br>2.3971946<br>3.2614464<br>3.1928226<br>3.1051925<br>3.0057113<br>3.3257455<br>2.7552623<br>2.7552623<br>2.7552623<br>2.7552623<br>2.7552623<br>2.7552623<br>2.7552623<br>2.752623<br>2.752623<br>2.752623<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0  
   | 19622-06<br>1354E-06<br>1354E-06<br>1438E-06<br>1438E-06<br>1337E-06<br>1377E-06<br>1606E-06<br>1377E-06<br>1472E-06<br>1472E-06<br>1472E-06<br>1472E-06<br>1488E-06<br>9.18E-09<br>0<br>0<br>0<br>2.57E-07<br>9.665E-09<br>2.511E-07<br>2.77E-07<br>3.763E-07<br>3.763E-07<br>3.749E-07<br>3.749E-07<br>3.749E-07<br>3.749E-07<br>3.749E-07<br>3.749E-07<br>3.749E-07<br>3.749E-07<br>3.749E-07<br>3.749E-07<br>3.749E-07<br>3.749E-07<br>3.749E-07<br>3.749E-07<br>3.749E-07<br>3.749E-07<br>3.749E-07<br>3.749E-07<br>3.749E-07<br>3.749E-07<br>3.749E-07<br>3.749E-07<br>3.749E-07<br>3.749E-07<br>3.749E-07<br>3.749E-07<br>3.749E-07<br>3.749E-07<br>3.749E-07<br>3.749E-07<br>3.749E-07<br>3.749E-07<br>3.749E-07<br>3.749E-07<br>3.749E-07<br>3.749E-07<br>3.749E-07<br>3.749E-07<br>3.749E-07<br>3.749E-07<br>3.749E-07<br>3.749E-07<br>3.749E-07<br>3.749E-07<br>3.749E-07<br>3.749E-07<br>3.749E-07<br>3.749E-07<br>3.749E-07<br>3.749E-07<br>3.749E-07<br>3.749E-07<br>3.749E-07<br>3.749E-07<br>3.749E-07<br>3.749E-07<br>3.749E-07<br>3.749E-07<br>3.749E-07<br>3.749E-07<br>3.749E-07<br>3.749E-07<br>3.749E-07<br>3.749E-07<br>3.749E-07<br>3.749E-07<br>3.749E-07<br>3.749E-07<br>3.749E-07<br>3.749E-07<br>3.749E-07<br>3.749E-07<br>3.749E-07<br>3.749E-07<br>3.749E-07<br>3.749E-07<br>3.749E-07<br>3.749E-07<br>3.749E-07<br>3.749E-07<br>3.749E-07<br>3.749E-07<br>3.749E-07<br>3.749E-07<br>3.749E-07<br>3.749E-07<br>3.749E-07<br>3.749E-07<br>3.749E-07<br>3.749E-07<br>3.749E-07<br>3.749E-07<br>3.749E-07<br>3.749E-07<br>3.749E-07<br>3.749E-07<br>3.749E-07<br>3.749E-07<br>3.749E-07<br>3.749E-07<br>3.749E-07<br>3.749E-07<br>3.749E-07<br>3.749E-07<br>3.749E-07<br>3.749E-07<br>3.749E-07<br>3.749E-07<br>3.749E-07<br>3.749E-07<br>3.749E-07<br>3.749E-07<br>3.749E-07<br>3.749E-07<br>3.749E-07<br>3.749E-07<br>3.749E-07<br>3.749E-07<br>3.749E-07<br>3.749E-07<br>3.749E-07<br>3.749E-07<br>3.749E-07<br>3.749E-07<br>3.749E-07<br>3.749E-07<br>3.749E-07<br>3.749E-07<br>3.749E-07<br>3.749E-07<br>3.749E-07<br>3.749E-07<br>3.749E-07<br>3.749E-07<br>3.749E-07<br>3.749E-07<br>3.749E-07<br>3.749E-07<br>3.749E-07<br>3.749E-07<br>3.749E-07<br>3.749E-07<br>3.749E-07<br>3.749E-07<br>3.749E-07<br>3.749E-07<br>3.749E-07<br>3.749E-07<br>3.749E-07<br>3.749E-07<br>3.749E-07<br>3.749E-07<br>3.749E-07<br>3.749E-07<br>3.749E-07<br>3.749E-07<br>3.749E-07<br>3.749E-07<br>3.749E-07<br>3.749E-07<br>3.749E-07<br>3.749E-07<br>3.749E-07<br>3.749E-07<br>3.749E-07<br>3.749E-0 | 0.0031876<br>0.0022403<br>0.002243<br>0.002337<br>0.002347<br>0.0022645<br>0.0027711<br>0.002156<br>0.0025865<br>0.0023913<br>0.002645<br>0.0023913<br>0.002646<br>0.0015867<br>0.0015867<br>0.0015867<br>0.001587<br>0.001587<br>0.002245<br>0.0005245<br>0.0005245<br>0.0005245<br>0.0005245<br>0.0005245<br>0.0005245<br>0.0005245<br>0.0005245<br>0.0005245<br>0.0005245<br>0.0005245<br>0.0005245<br>0.0005245<br>0.0005874<br>0.0006887<br>0.0006887<br>0.0006883<br>0.0005883<br>0.0005883<br>0.0005883<br>0.0005883<br>0.0005883<br>0.000583<br>0.000583<br>0.000583<br>0.000583<br>0.000583<br>0.000583<br>0.000583<br>0.000583<br>0.000583<br>0.000583<br>0.000583<br>0.000583<br>0.000583<br>0.000583<br>0.000583<br>0.000583<br>0.000583<br>0.000583<br>0.000583<br>0.000583<br>0.000583<br>0.000583<br>0.000583<br>0.000583<br>0.000583<br>0.000583<br>0.000583<br>0.000583<br>0.000583<br>0.000583<br>0.000583<br>0.000583<br>0.000583<br>0.000583<br>0.000585<br>0.000585<br>0.000585<br>0.00058<br>0.00058<br>0.00058<br>0.00058<br>0.00058<br>0.00058<br>0.00058<br>0.00058<br>0.00058<br>0.00058<br>0.00058<br>0.00058<br>0.00058<br>0.00058<br>0.00058<br>0.00058<br>0.00058<br>0.00058<br>0.00058<br>0.00058<br>0.00058<br>0.00058<br>0.00058<br>0.00058<br>0.00058<br>0.00058<br>0.00058<br>0.00058<br>0.00058<br>0.00058<br>0.00058<br>0.00058<br>0.00058<br>0.00058<br>0.00058<br>0.00058<br>0.00058<br>0.00058<br>0.00058<br>0.00058<br>0.00058<br>0.00058<br>0.00058<br>0.00058<br>0.00058<br>0.00058<br>0.00058<br>0.00058<br>0.00058<br>0.00058<br>0.00058<br>0.00058<br>0.00058<br>0.00058<br>0.00058<br>0.00058<br>0.00058<br>0.00058<br>0.00058<br>0.00058<br>0.00058<br>0.00058<br>0.00058<br>0.00058<br>0.00058<br>0.00058<br>0.00058<br>0.00058<br>0.00058<br>0.00058<br>0.00058<br>0.00058<br>0.00058<br>0.00058<br>0.00058<br>0.00058<br>0.00058<br>0.00058<br>0.00058<br>0.00058<br>0.00058<br>0.00058<br>0.00058<br>0.00058<br>0.00058<br>0.00058<br>0.00058<br>0.00058<br>0.00058<br>0.00058<br>0.00058<br>0.00058<br>0.00058<br>0.00058<br>0.00058<br>0.00058<br>0.00058<br>0.00058<br>0.00058<br>0.00058<br>0.00058<br>0.00058<br>0.00058<br>0.00058<br>0.00058<br>0.00058<br>0.00058<br>0.00058<br>0.00058<br>0.00058<br>0.00058<br>0.00058<br>0.00058<br>0.00058<br>0.00058<br>0.00058<br>0.00058<br>0.00058<br>0.00058<br>0.00058<br>0.00058<br>0.00058<br>0.00058<br>0.00058<br>0.00058<br>0.00058<br>0.00058<br>0.00058<br>0.00058<br>0.00058<br>0.00058<br>0.00058<br>0.00058<br>0.00058<br>0.00058<br>0.00058<br>0.00058<br>0.00058<br>0.00058<br>0.00058<br>0.00058<br>0.00058<br>0.00058<br>0.00058<br>0.00058<br>0.00058<br>0.00058<br>0.00058<br>0.00058<br>0.00058<br>0.00058<br>0.00058<br>0.00058<br>0.00058<br>0.00058<br>0.00058<br>0.00058<br>0.00058<br>0.00058<br>0.00058<br>0.00058<br>0.00058<br>0.00058<br>0.00058 |
2.3644564<br>1.8937445<br>2.1225909<br>2.0177733<br>2.1529112<br>2.928047<br>2.928047<br>2.928047<br>2.928047<br>2.928047<br>2.928047<br>2.928147<br>2.928147<br>2.928147<br>2.928147<br>2.928147<br>2.928147<br>2.928147<br>2.928147<br>2.928147<br>2.928147<br>2.928147<br>2.928147<br>2.928147<br>2.928147<br>2.928147<br>2.928147<br>2.928147<br>2.928147<br>2.928147<br>2.928147<br>2.928147<br>2.928147<br>2.928147<br>2.928147<br>2.928147<br>2.928147<br>2.928147<br>2.928147<br>2.928147<br>2.928147<br>2.928147<br>2.928147<br>2.928147<br>2.928147<br>2.928147<br>2.928147<br>2.928147<br>2.928147<br>2.928147<br>2.928147<br>2.928147<br>2.928147<br>2.928147<br>2.928147<br>2.928147<br>2.928147<br>2.928147<br>2.928147<br>2.928147<br>2.928147<br>2.928147<br>2.928147<br>2.928147<br>2.928147<br>2.928147<br>2.928147<br>2.928147<br>2.928147<br>2.928147<br>2.928147<br>2.928147<br>2.928147<br>2.928147<br>2.928147<br>2.928147<br>2.928147<br>2.928147<br>2.928147<br>2.928147<br>2.928147<br>2.928147<br>2.928147<br>2.928147<br>2.928147<br>2.928147<br>2.928147<br>2.928147<br>2.928147<br>2.928147<br>2.928147<br>2.928147<br>2.928147<br>2.928147<br>2.928147<br>2.928147<br>2.928147<br>2.928147<br>2.928147<br>2.928147<br>2.928147<br>2.928147<br>2.928147<br>2.928147<br>2.928147<br>2.928147<br>2.928147<br>2.928147<br>2.928147<br>2.928147<br>2.928147<br>2.928147<br>2.928147<br>2.928147<br>2.928147<br>2.928147<br>2.928147<br>2.928147<br>2.928147<br>2.928147<br>2.928147<br>2.928147<br>2.928147<br>2.928147<br>2.928147<br>2.928147<br>2.928147<br>2.928147<br>2.928147<br>2.928147<br>2.928147<br>2.928147<br>2.928147<br>2.928147<br>2.928147<br>2.928147<br>2.928147<br>2.928147<br>2.928147<br>2.928147<br>2.928147<br>2.928147<br>2.928147<br>2.928147<br>2.928147<br>2.928147<br>2.928147<br>2.928147<br>2.928147<br>2.928147<br>2.928147<br>2.928147<br>2.928147<br>2.928147<br>2.928147<br>2.928147<br>2.928147<br>2.928147<br>2.928147<br>2.928147<br>2.928147<br>2.928147<br>2.928147<br>2.928147<br>2.928147<br>2.928147<br>2.928147<br>2.928147<br>2.928147<br>2.928147<br>2.928147<br>2.928147<br>2.928147<br>2.928147<br>2.928147<br>2.928147<br>2.928147<br>2.928147<br>2.928147<br>2.928147<br>2.928147<br>2.928147<br>2.928147<br>2.928147<br>2.928147<br>2.928147<br>2.928147<br>2.928147<br>2.928147<br>2.928147<br>2.928147<br>2.928147<br>2.928147<br>2.928147<br>2.928147<br>2.928147<br>2.928147<br>2.928147<br>2.928147<br>2.928147<br>2.928147<br>2.928147<br>2.928147<br>2.928147<br>2.928147<br>2.928147<br>2.928147<br>2.928147<br>2.928147<br>2.928147<br>2.928147<br>2.928147<br>2.928147<br>2.928147<br>2.928147<br>2.928147<br>2.928147<br>2.929147<br>2.929147<br>2.929147<br>2.929147<br>2.929147<br>2.929147<br>2.929147<br>2.929147<br>2.929147<br>2.92914 | 1.7688-00 0.002<br>1.3681-00 0.002<br>1.381-00 0.002<br>1.381-00 0.002<br>1.382-00 0.002<br>1.525-00 0.002<br>1.525-00 0.002<br>1.535-00 0.002<br>1.535-00 0.002<br>1.536-00 0.002<br>1.536-00 0.002<br>1.536-00 0.002<br>1.537-00 0.002<br>1.548-00 0.002<br>1.548-00000<br>1.548-0000<br>1.548-00000<br>1.548-00000<br>1.548-000   | 873         0.04           8723         0.03           9723         0.03           1010         0.04           1015         0.04           1033         0.05           10422         0.04           1033         0.05           1015         0.04           1033         0.05           1016         0.04           1033         0.05           1110         0.04           1111         0.05           1111         0.04           1111         0.04           1111         0.04           1111         0.04           1111         0.04           1111         0.04           1111         0.04           1111         0.04           1111         0.04           1111         0.04           1111         0.04           1111         0.04           1111         0.02           1111         0.02           1111         0.02           1111         0.02           1111         0.02           11111         0.02           11111<  
   | 147043         3.           147043         3.           828322         2.           405571         1           313125         2.           338011         2.           338012         3.           338013         2.           338014         2.           195214         2.4           167677         2.4           18381         1.4           1776164         2.4           153993         8.1           0         0           4/4252         6.4           153993         8.1           108061         6.3           8/3915         5.5           10642         9.5           10642         9.5           10642         9.5           10642         9.5           10642         9.5           10642         9.5           10642         9.5           70592         1.6           70672         1.6   
  | 5.18E-08         2           6.619E-08         2           6.619E-08         4           2.99E-08         3           8.74E-08         3           8.74E-08         3           8.74E-08         3           8.74E-08         3           4.99E-08         3           4.98E-08         4           6.51E-08         1           0         0           0         0           0         1           3.34E-09         1           3.34E-09         1           3.54E-09         1           3.54E-09         1           5.34E-08         3           6.34E-08         3   | 3.998E-05<br>4.254E-05<br>4.224E-05<br>4.224E-05<br>4.511E-05<br>3.736E-05<br>3.569E-05<br>4.672-05<br>4.672-05<br>4.672-05<br>4.052E-05<br>4.052E-05<br>4.052E-05<br>4.042E-05<br>1.266E-05<br>1.18E-05<br>0<br>0<br>1.18E-05<br>1.18E-05<br>1.18E-05<br>1.18E-05<br>1.18E-05<br>1.18E-05<br>1.18E-05<br>1.1254E-05<br>1.754E-05<br>1.754E-05<br>1.754E-05<br>1.754E-05<br>1.754E-05<br>1.754E-05<br>1.754E-05<br>1.754E-05<br>1.754E-05<br>1.754E-05<br>1.754E-05<br>1.754E-05<br>1.754E-05<br>1.754E-05<br>1.754E-05<br>1.754E-05<br>1.754E-05<br>1.754E-05<br>1.754E-05<br>1.754E-05<br>1.754E-05<br>1.754E-05<br>1.754E-05<br>1.754E-05<br>1.754E-05<br>1.754E-05<br>1.754E-05<br>1.754E-05<br>1.754E-05<br>1.754E-05<br>1.754E-05<br>1.754E-05<br>1.754E-05<br>1.754E-05<br>1.754E-05<br>1.754E-05<br>1.754E-05<br>1.754E-05<br>1.754E-05<br>1.754E-05<br>1.754E-05<br>1.754E-05<br>1.754E-05<br>1.754E-05<br>1.754E-05<br>1.754E-05<br>1.754E-05<br>1.754E-05<br>1.754E-05<br>1.754E-05<br>1.754E-05<br>1.754E-05<br>1.754E-05<br>1.754E-05<br>1.754E-05<br>1.754E-05<br>1.754E-05<br>1.754E-05<br>1.754E-05<br>1.754E-05<br>1.754E-05<br>1.754E-05<br>1.754E-05<br>1.754E-05<br>1.754E-05<br>1.754E-05<br>1.754E-05<br>1.754E-05<br>1.754E-05<br>1.754E-05<br>1.754E-05<br>1.754E-05<br>1.754E-05<br>1.754E-05<br>1.754E-05<br>1.754E-05<br>1.754E-05<br>1.754E-05<br>1.754E-05<br>1.754E-05<br>1.754E-05<br>1.754E-05<br>1.754E-05<br>1.754E-05<br>1.754E-05<br>1.754E-05<br>1.754E-05<br>1.754E-05<br>1.754E-05<br>1.754E-05<br>1.754E-05<br>1.754E-05<br>1.754E-05<br>1.754E-05<br>1.754E-05<br>1.754E-05<br>1.754E-05<br>1.754E-05<br>1.754E-05<br>1.754E-05<br>1.754E-05<br>1.754E-05<br>1.754E-05<br>1.754E-05<br>1.754E-05<br>1.754E-05<br>1.754E-05<br>1.754E-05<br>1.754E-05<br>1.754E-05<br>1.754E-05<br>1.754E-05<br>1.754E-05<br>1.754E-05<br>1.754E-05<br>1.754E-05<br>1.754E-05<br>1.754E-05<br>1.754E-05<br>1.754E-05<br>1.754E-05<br>1.754E-05<br>1.754E-05<br>1.754E-05<br>1.754E-05<br>1.754E-05<br>1.754E-05<br>1.754E-05<br>1.754E-05<br>1.754E-05<br>1.754E-05<br>1.754E-05<br>1.754E-05<br>1.754E-05<br>1.754E-05<br>1.754E-05<br>1.754E-05<br>1.754E-05<br>1.754E-05<br>1.754E-05<br>1.754E-05<br>1.754E-05<br>1.754E-05<br>1.754E-05<br>1.754E-05<br>1.754E-05<br>1.754E-05<br>1.754E-05<br>1.754E-05<br>1.754E-05<br>1.754E-05<br>1.754E-05<br>1.754E-05<br>1.754E-05<br>1.754E-05<br>1.754E-05<br>1.754E-05<br>1.754E-05<br>1.754E-05<br>1.754E-05<br>1.754E-05<br>1.754E-05<br>1.754E-05<br>1.754E-05<br>1.754E-05<br>1.754E-05<br>1.754E-05<br>1.754E-05<br>1.754E-05<br>1.754E-05<br>1.754E-05<br>1.754E-05<br>1.754E-05<br>1.754E-05<br>1.754E-05<br>1.754E-05<br>1.754E-05<br>1.754E-05<br>1.754E-05<br>1.754E-05<br>1.754E-05<br>1.  |
0.0443322<br>0.044017<br>0.0443322<br>0.044017<br>0.0584373<br>0.05584373<br>0.0558211<br>0.0538871<br>0.054262<br>0.0461220<br>0.046222<br>0.06507581<br>0.06507581<br>0.06507581<br>0.06507581<br>0.06507584<br>0.06507584<br>0.06507584<br>0.06507584<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.045774<br>1.0225166<br>0.04226146<br>0.04226146<br>0.04226146<br>0.04226146<br>0.04226146<br>0.04226146<br>0.04226146<br>0.04226146<br>0.04226146<br>0.04226146<br>0.04226146<br>0.04226146<br>0.04226146<br>0.04226146<br>0.04226146<br>0.04226146<br>0.04226146<br>0.04226146<br>0.04226146<br>0.04226146<br>0.04226146<br>0.04226146<br>0.04226146<br>0.04226146<br>0.04226146<br>0.04226146<br>0.04226146<br>0.04226146<br>0.04226146<br>0.04226146<br>0.04226146<br>0.04226146<br>0.04226146<br>0.04226146<br>0.04226146<br>0.04226146<br>0.04226146<br>0.04226146<br>0.04226146<br>0.04226146<br>0.04226146<br>0.04226146<br>0.04226146<br>0.04226146<br>0.04226146<br>0.04226146<br>0.04226146<br>0.04226146<br>0.04226146<br>0.04226146<br>0.04226146<br>0.04226146<br>0.04226146<br>0.04226146<br>0.04226146<br>0.04226146<br>0.04226146<br>0.04226146<br>0.04226146<br>0.04226146<br>0.04226146<br>0.04226146<br>0.04226146<br>0.04226146<br>0.04226146<br>0.04226146<br>0.04226146<br>0.04226146<br>0.04226146<br>0.04226146<br>0.04226146<br>0.04226146<br>0.04226146<br>0.04226146<br>0.04226146<br>0.04226146<br>0.04226146<br>0.04226146<br>0.04226146<br>0.04226146<br>0.04226146<br>0.04226146<br>0.04226146<br>0.04226146<br>0.04226146<br>0.04226146<br>0.04226146<br>0.04226146<br>0.04226146<br>0.04226146<br>0.04226146<br>0.04226146<br>0.04226146<br>0.04226146<br>0.04226146<br>0.04226146<br>0.04226146<br>0.04226146<br>0.04226146<br>0.04226146<br>0.04226146<br>0.04226146<br>0.04226146<br>0.04226146<br>0.04226146<br>0.04226146<br>0.04226146<br>0.04226146<br>0.04226146<br>0.04226146<br>0.04226146<br>0.04226146<br>0.04226146<br>0.04226146<br>0.04226146<br>0.04226146<br>0.04226146<br>0.04226146<br>0.04226146<br>0.04226146<br>0.04226146<br>0.04226146<br>0.04226146<br>0.0426146<br>0.0426146<br>0.0426146<br>0.0426146<br>0.0426146<br>0.0426146<br>0.0426146<br>0.0426146<br>0.0426146<br>0.0426146<br>0.0426146<br>0.0426146<br>0.0426146<br>0.0426146<br>0.0426146<br>0.0426146<br>0.0426146<br>0.0426146<br>0.0426146<br>0.0426146<br>0.0426146<br>0.0426146<br>0.0426146<br>0.0426146<br>0.0426146<br>0.0426146<br>0.0426146<br>0.0426146<br>0.0426146<br>0.0426146<br>0.0426146<br>0.0426146<br>0.0426146<br>0.0426146<br>0.0426146<br>0.0426146<br>0.04260   | 2.842-0.8<br>8.822-08<br>8.822-08<br>8.822-08<br>4.877-08<br>2.879E-08<br>2.879E-08<br>2.879E-08<br>2.879E-08<br>2.83281-08<br>2.821-08<br>2.821-08<br>2.821-08<br>2.822-08<br>4.2713E-08<br>2.822-08<br>4.2732-08<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0   | 1.584-05<br>1.897E-05<br>1.057E-05<br>3.072E-05<br>3.876E-05<br>3.331E-05<br>1.307E-05<br>1.307E-05<br>1.307E-05<br>1.393E-05<br>1.393E-05<br>1.393E-05<br>1.636E-05<br>0<br>0<br>0<br>1.282E-05<br>1.282E-05<br>1.282E-05<br>1.282E-05<br>1.282E-05<br>1.201E-05<br>1.901E-05<br>3.259E-05<br>1.267E-05   |
| Sheung Tsuen B/T and Car Park<br>Sheung Tsuen B/T and Car Park<br>Village Access Road<br>Village Access Road  | RD_29<br>RD_29<br>RD_29<br>RD_29<br>RD_29<br>RD_29<br>RD_29<br>RD_29<br>RD_29<br>RD_29<br>RD_29<br>RD_29<br>RD_29<br>RD_29<br>RD_29<br>RD_29<br>RD_29<br>RD_29<br>RD_29<br>RD_29<br>RD_29<br>RD_29<br>RD_29<br>RD_29<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20 | 9900.1000           1000-1100           1100-1200           1200-1300           1200-1300           1200-1300           1200-1300           1200-100           1400-1500           1500-1600           1600-1700           1700-1800           1800-2000           2000-2100           2100-2200           2000-2100           2000-2100           2000-2100           2000-2000           0000-0100           0000-0100           0500-6600           0500-6600           0800-9000           0800-9000           0800-9000           1000-1100           1100-1200           1200-1300  | 0.1281945<br>0.1457065<br>0.1360751<br>0.1607293<br>0.1557266<br>0.1482353<br>0.14524652<br>0.1525488<br>0.1323275<br>0.1325488<br>0.1323275<br>0.1325288<br>0.1323275<br>0.1325292<br>0.13652928<br>0.0329421<br>0.032942<br>0.25292911<br>0.0326927<br>0.1264616<br>0.2234554<br>0.2224179<br>0.2224179<br>0.2224179<br>0.2224179<br>0.2224179<br>0.2224179<br>0.2224179<br>0.2224179<br>0.2224179<br>0.2224179<br>0.2224179<br>0.2224179<br>0.2224179<br>0.2224179<br>0.2224179<br>0.2224179<br>0.2224179<br>0.2224179<br>0.2224179<br>0.2224179<br>0.2224179<br>0.2224179<br>0.2224179<br>0.2224179<br>0.2224179<br>0.2224179<br>0.2224179<br>0.2224179<br>0.2224179<br>0.2224179<br>0.2224179<br>0.2224179<br>0.2224179<br>0.2224179<br>0.2224179<br>0.2224179<br>0.2224179<br>0.2224179<br>0.2224179<br>0.2224179<br>0.2224179<br>0.2224179<br>0.2224179<br>0.2224179<br>0.2224179<br>0.2224179<br>0.2224179<br>0.2224179<br>0.2224179<br>0.2224179<br>0.2224179<br>0.2224179<br>0.2224179<br>0.2224179<br>0.2224179<br>0.2224179<br>0.2224179<br>0.2224179<br>0.2224179<br>0.2224179<br>0.2224179<br>0.2224179<br>0.2224179<br>0.2224179<br>0.2224179<br>0.2224179<br>0.2224179<br>0.2224179<br>0.2224179<br>0.2224179<br>0.2224179<br>0.2224179<br>0.2224179<br>0.2224179<br>0.2224179<br>0.2224179<br>0.2224179<br>0.2224179<br>0.2224179<br>0.2224179<br>0.2224179<br>0.2224179<br>0.2224179<br>0.2224179<br>0.2224179<br>0.2224179<br>0.2224179<br>0.2224179<br>0.2224179<br>0.2224179<br>0.2224179<br>0.2224179<br>0.2224179<br>0.2224179<br>0.2224179<br>0.2224179<br>0.2224179<br>0.2224179<br>0.2224179<br>0.2224179<br>0.2224179<br>0.2224179<br>0.2224179<br>0.2224179<br>0.2224179<br>0.2224179<br>0.2224179<br>0.2224179<br>0.2224179<br>0.2224179<br>0.2224179<br>0.2224179<br>0.2224179<br>0.2224179<br>0.2224179<br>0.2224179<br>0.2224179<br>0.2224179<br>0.2224179<br>0.2224179<br>0.2224179<br>0.2224179<br>0.2224179<br>0.2224179<br>0.2224179<br>0.224179<br>0.224179<br>0.224179<br>0.224179<br>0.224179<br>0.224179<br>0.224179<br>0.224179<br>0.224179<br>0.224179<br>0.224179<br>0.224179<br>0.224179<br>0.224179<br>0.224179<br>0.224179<br>0.224179<br>0.224179<br>0.224179<br>0.224179<br>0.224179<br>0.224179<br>0.224179<br>0.224179<br>0.224179<br>0.224179<br>0.224179<br>0.224179<br>0.224179<br>0.224179<br>0.224179<br>0.224179<br>0.224179<br>0.224179<br>0.224179<br>0.224179<br>0.224179<br>0.224179<br>0.224179<br>0.224179<br>0.224179<br>0.224179<br>0.224179<br>0.224179<br>0.224179<br>0.224179<br>0.224179<br>0.224179<br>0.224179<br>0.224179<br>0.224179<br>0.224179<br>0.224179<br>0.224179<br>0.224179<br>0.224179<br>0.224179<br>0.224179<br>0.224179<br>0.224179<br>0.224179<br>0.224179<br>0.224179<br>0.224179<br>0.224179             | 9.342.003         1           9.342.003         1           9.342.003         1           9.342.003         1           9.342.003         1           0.352.00  
  | 0.0001615 (<br>0.0001335 (<br>0.000135 (<br>0.0001674 (<br>0.0001674 (<br>0.0001674 (<br>0.000167 (<br>0.000167 (<br>0.000167 (<br>0.0001675 (<br>0.0001162 (<br>0.0001162 (<br>0.0001162 (<br>0.000122 (<br>0.000122 (<br>0.000122 (<br>0.000122 (<br>0.000162 (<br>0.000162 (<br>0.000162 (<br>0.000165 (<br>0.0000165 (<br>0.0000165 (   | 0.1261103<br>0.1210641<br>0.1385867<br>0.1291377<br>0.1291377<br>0.1399004<br>0.1423175<br>0.1399004<br>0.1423175<br>0.1399004<br>0.1423175<br>0.1399004<br>0.1420185<br>0.1267454<br>0.1263453<br>0.1263454<br>0.0169957<br>0.2231399<br>0.1762915<br>0.0254420<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0  | 9.43114.08<br>7.761E-08<br>8.284E-08<br>9.754E-08<br>9.754E-08<br>6.373E-08<br>7.465-08<br>6.632E-08<br>7.8055-08<br>6.771E-08<br>6.656E-08<br>6.656E-08<br>6.771E-08<br>7.45E-08<br>7.152E-08<br>7.152E-08<br>7.255E-08<br>3.869E-10<br>6.116E-08<br>5.173E-08<br>9.357E-08<br>9.354E-08<br>9.354E-08<br>9.354E-08<br>9.354E-08<br>9.354E-08<br>9.354E-08<br>9.354E-08<br>9.354E-08<br>9.354E-08<br>9.354E-08<br>9.354E-08<br>9.354E-08<br>9.354E-08<br>9.354E-08<br>9.354E-08<br>9.354E-08<br>9.354E-08<br>9.354E-08   | 0.0001532<br>0.0001532<br>0.0001261<br>0.0001453<br>0.0001345<br>0.0001345<br>0.000135<br>0.0001214<br>9.8822-05<br>0.0001214<br>9.8822-05<br>0.0001121<br>0.000128<br>0.00011<br>0.0001095<br>6.689E-05<br>8.855-05<br>0.0001162<br>9.18E-05<br>0.0001328<br>0.0001328<br>0.0001328<br>0.0001328<br>0.0001328<br>0.000171<br>0.0000711<br>0.0001715<br>0.0001715<br>0.0001715  |
0.1203596<br>0.1142306<br>0.1142306<br>0.1310694<br>0.1216391<br>0.1426721<br>0.1426721<br>0.1426721<br>0.1323766<br>0.1323766<br>0.1323766<br>0.1323766<br>0.1323376<br>0.123838<br>0.120155<br>0.1229461<br>0.1229461<br>0.01228452<br>0.01624807<br>0.01624807<br>0.01624807<br>0.01624807<br>0.01624807<br>0.01624807<br>0.01624807<br>0.01624807<br>0.01624807<br>0.01624807<br>0.01624807<br>0.01624807<br>0.01624807<br>0.01624807<br>0.01624807<br>0.01624807<br>0.01624807<br>0.01624807<br>0.01624807<br>0.01624807<br>0.01624807<br>0.01624807<br>0.01624807<br>0.01624807<br>0.01624807<br>0.01624807<br>0.01624807<br>0.01624807<br>0.01624807<br>0.01624807<br>0.01624807<br>0.01624807<br>0.01624807<br>0.01624807<br>0.01624807<br>0.01624807<br>0.01624807<br>0.01624807<br>0.01624807<br>0.01624807<br>0.01624807<br>0.01624807<br>0.01624807<br>0.01624807<br>0.01624807<br>0.01624807<br>0.01624807<br>0.01624807<br>0.01624807<br>0.01624807<br>0.01624807<br>0.01624807<br>0.01624807<br>0.01624807<br>0.01624807<br>0.01624807<br>0.01624807<br>0.01624807<br>0.01624807<br>0.01624807<br>0.01624807<br>0.01624807<br>0.01624807<br>0.01624807<br>0.01624807<br>0.01624807<br>0.01624807<br>0.01624807<br>0.01624807<br>0.01624807<br>0.01624807<br>0.01624807<br>0.01624807<br>0.01624807<br>0.01624807<br>0.01624807<br>0.01624807<br>0.01624807<br>0.01624807<br>0.01624807<br>0.01624807<br>0.01624807<br>0.01624807<br>0.01624807<br>0.01624807<br>0.01624807<br>0.01624807<br>0.01624807<br>0.01624807<br>0.01624807<br>0.01624807<br>0.01624807<br>0.01624807<br>0.01624807<br>0.01624807<br>0.01624807<br>0.01624807<br>0.01624807<br>0.01624807<br>0.01624807<br>0.01624807<br>0.01624807<br>0.01624807<br>0.01624807<br>0.01624807<br>0.01624807<br>0.01624807<br>0.01624807<br>0.01624807<br>0.01624807<br>0.01624807<br>0.01624807<br>0.01624807<br>0.01624807<br>0.01624807<br>0.01624807<br>0.01624807<br>0.01624807<br>0.01624807<br>0.01624807<br>0.01624807<br>0.01624807<br>0.01624807<br>0.01624807<br>0.01624807<br>0.01624807<br>0.01624807<br>0.01624807<br>0.01624807<br>0.01624807<br>0.01624807<br>0.01624807<br>0.01624807<br>0.01624807<br>0.01624807<br>0.01624807<br>0.01624807<br>0.01624807<br>0.01624807<br>0.01624807<br>0.01624807<br>0.01624807<br>0.01624807<br>0.01624807<br>0.01624807<br>0.01624807<br>0.01624807<br>0.01624807<br>0.01624807<br>0.01624807<br>0.01624807<br>0.01624807<br>0.01624807<br>0.01624807<br>0.01624807<br>0.01624807<br>0.01624807<br>0.01624807<br>0.01624807<br>0.01624807<br>0.01624807<br>0.01624807<br>0.01624807<br>0.01644807<br>0.01644807<br>0.01644807<br>0.01644807<br>0.01644807<br>0.01644807<br>0.016480  | 9.001-08<br>7.322E-08<br>8.402E-08<br>8.402E-08<br>9.146E-08<br>9.146E-08<br>5.749E-08<br>7.39E-08<br>6.353E-08<br>6.353E-08<br>6.456E-08<br>5.248E-08<br>5.438E-08<br>5.438E-08<br>5.438E-08<br>6.739E-08<br>6.739E-08<br>6.739E-08<br>6.739E-08<br>6.739E-08<br>8.542E-08<br>8.542E-08<br>8.542E-08<br>8.542E-08<br>8.542E-08<br>8.542E-08<br>8.542E-08<br>8.542E-08<br>8.542E-08<br>8.542E-08<br>8.542E-08<br>8.542E-08<br>8.542E-08<br>8.542E-08<br>8.542E-08<br>8.542E-08<br>8.542E-08<br>8.542E-08<br>8.542E-08<br>8.542E-08<br>8.542E-08<br>8.542E-08<br>8.542E-08<br>8.542E-08<br>8.542E-08<br>8.542E-08<br>8.542E-08<br>8.542E-08<br>8.542E-08<br>8.542E-08<br>8.542E-08<br>8.542E-08<br>8.542E-08<br>8.542E-08<br>8.542E-08<br>8.542E-08<br>8.542E-08<br>8.542E-08<br>8.542E-08<br>8.542E-08<br>8.542E-08<br>8.542E-08<br>8.542E-08<br>8.542E-08<br>8.542E-08<br>8.542E-08<br>8.542E-08<br>8.542E-08<br>8.542E-08<br>8.542E-08<br>8.542E-08<br>8.542E-08<br>8.542E-08<br>8.542E-08<br>8.542E-08<br>8.542E-08<br>8.542E-08<br>8.542E-08<br>8.542E-08<br>8.542E-08<br>8.542E-08<br>8.542E-08<br>8.542E-08<br>8.542E-08<br>8.542E-08<br>8.542E-08<br>8.542E-08<br>8.542E-08<br>8.542E-08<br>8.542E-08<br>8.542E-08<br>8.542E-08<br>8.542E-08<br>8.542E-08<br>8.542E-08<br>8.542E-08<br>8.542E-08<br>8.542E-08<br>8.542E-08<br>8.542E-08<br>8.542E-08<br>8.542E-08<br>8.542E-08<br>8.542E-08<br>8.542E-08<br>8.542E-08<br>8.542E-08<br>8.542E-08<br>8.542E-08<br>8.542E-08<br>8.542E-08<br>8.542E-08<br>8.542E-08<br>8.542E-08<br>8.542E-08<br>8.542E-08<br>8.542E-08<br>8.542E-08<br>8.542E-08<br>8.542E-08<br>8.542E-08<br>8.542E-08<br>8.542E-08<br>8.542E-08<br>8.542E-08<br>8.542E-08<br>8.542E-08<br>8.542E-08<br>8.542E-08<br>8.542E-08<br>8.542E-08<br>8.542E-08<br>8.542E-08<br>8.542E-08<br>8.542E-08<br>8.542E-08<br>8.542E-08<br>8.542E-08<br>8.542E-08<br>8.542E-08<br>8.542E-08<br>8.542E-08<br>8.542E-08<br>8.542E-08<br>8.542E-08<br>8.542E-08<br>8.542E-08<br>8.542E-08<br>8.542E-08<br>8.542E-08<br>8.542E-08<br>8.542E-08<br>8.542E-08<br>8.542E-08<br>8.542E-08<br>8.542E-08<br>8.542E-08<br>8.542E-08<br>8.542E-08<br>8.542E-08<br>8.542E-08<br>8.542E-08<br>8.542E-08<br>8.542E-08<br>8.542E-08<br>8.542E-08<br>8.542E-08<br>8.542E-08<br>8.542E-08<br>8.542E-08<br>8.542E-08<br>8.542E-08<br>8.542E-08<br>8.542E-08<br>8.542E-08<br>8.542E-08<br>8.542E-08<br>8.542E-08<br>8.542E-08<br>8.542E-08<br>8.542E-08<br>8.542E-08<br>8.542E-08<br>8.542E-08<br>8.542E-08<br>8.542E-08<br>8.542E-08<br>8.542E-08<br>8.542E-08<br>8.542E-08<br>8.542E-08<br>8.542E-08<br>8.542E-08<br>8.542E-08<br>8.542E-08<br>8.542E-08<br>8.542E-08<br>8.542E-08<br>8.542E-08<br>8.542E-08<br>8.542E-08<br>8.542E-08<br>8.542E-08<br>8.542E-08<br>8.542E-0                | 0.0001462<br>0.0001365<br>0.0001267<br>0.0001267<br>0.0001267<br>0.0001246<br>9.834E-05<br>0.0001248<br>9.34E-05<br>0.0001149<br>9.34E-05<br>0.0001210<br>0.0001149<br>6.402E-05<br>8.461E-05<br>0.000123<br>0.0001139<br>0.000123<br>0.000123<br>0.000123<br>0.000123<br>0.000123<br>0.000123<br>0.000123<br>0.000123<br>0.0001557<br>0.0001557<br>0.0001557<br>0.0001557  | 2.4904488<br>1.9842855<br>2.2002208<br>2.1078468<br>2.33646334<br>2.3569573<br>2.8602462<br>3.1731567<br>2.620547<br>2.6309615<br>2.7073267<br>2.9117616<br>4.4562831<br>0.0573728<br>0<br>0<br>1.3610254<br>1.3610254<br>1.3610254<br>1.3610254<br>0.06039632<br>0.6039622<br>0.5137727<br>0.6039622<br>0.5137727<br>0.6039622<br>0.5137727<br>0.6039622<br>0.5137727<br>0.6039622<br>0.5137727<br>0.6039622<br>0.5137727<br>0.6025855<br>0.50987613<br>0.6002628<br>0.8039808<br>0.8324163  
   | 1.8634-06<br>1.272E-06<br>1.41E-06<br>1.351E-06<br>1.4344-06<br>1.327E-06<br>1.628E-06<br>1.628E-06<br>1.628E-06<br>1.639E-06<br>1.401E-06<br>1.401E-06<br>1.401E-06<br>1.440E-06<br>1.440E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>1.446E-06<br>2.066E-07<br>2.066E-07<br>2.745E-07<br>2.745E-07<br>2.745E-07<br>2.745E-07<br>2.745E-07<br>2.745E-07<br>2.745E-07<br>2.745E-07<br>2.745E-07<br>2.745E-07<br>2.745E-07<br>2.745E-07<br>2.745E-07<br>2.745E-07<br>2.745E-07<br>2.745E-07<br>2.745E-07<br>2.745E-07<br>2.745E-07<br>2.745E-07<br>2.745E-07<br>2.745E-07<br>2.745E-07<br>2.745E-07<br>2.745E-07<br>2.745E-07<br>2.745E-07<br>2.745E-07<br>2.745E-07<br>2.745E-07<br>2.745E-07<br>2.745E-07<br>2.745E-07<br>2.745E-07<br>2.745E-07<br>2.745E-07<br>2.745E-07<br>2.745E-07<br>2.745E-07<br>2.745E-07<br>2.745E-07<br>2.745E-07<br>2.745E-07<br>2.745E-07<br>2.758E-07<br>2.758E-07<br>2.758E-07<br>2.758E-07<br>2.758E-07<br>2.758E-07<br>2.758E-07<br>2.758E-07<br>2.758E-07<br>2.758E-07<br>2.758E-07<br>2.758E-07<br>2.758E-07<br>2.758E-07<br>2.758E-07<br>2.758E-07<br>2.758E-07<br>2.758E-07<br>2.758E-07<br>2.758E-07<br>2.758E-07<br>2.758E-07<br>2.758E-07<br>2.758E-07<br>2.758E-07<br>2.758E-07<br>2.758E-07<br>2.758E-07<br>2.758E-07<br>2.758E-07<br>2.758E-07<br>2.758E-07<br>2.758E-07<br>2.758E-07<br>2.758E-07<br>2.758E-07<br>2.758E-07<br>2.758E-07<br>2.758E-07<br>2.758E-07<br>2.758E-07<br>2.758E-07<br>2.758E-07<br>2.758E-07<br>2.758E-07<br>2.758E-07<br>2.758E-07<br>2     | 0.003066<br>0.002265<br>0.0022915<br>0.0021953<br>0.0021553<br>0.0021554<br>0.0024544<br>0.002531<br>0.0024844<br>0.002754<br>0.0022754<br>0.0023200<br>0.0021782<br>0.0023200<br>0.0021782<br>0.0023200<br>0.0003777<br>1.6816-05<br>0.0003777<br>1.6816-05<br>0.0003777<br>0.0003777<br>0.0003777<br>0.0003777<br>0.0003777<br>0.0003777<br>0.0003777<br>0.0003777<br>0.0003777<br>0.0003777<br>0.0003777<br>0.0003777<br>0.0003777<br>0.0003777<br>0.0003777<br>0.0003777<br>0.0003777<br>0.0003777<br>0.0003777<br>0.0003777<br>0.0003777<br>0.0003777<br>0.0003777<br>0.0003777<br>0.0003777<br>0.0003777<br>0.0003777<br>0.0003777<br>0.0003777<br>0.0003777<br>0.0003777<br>0.0003777<br>0.0003777<br>0.0003777<br>0.0003777<br>0.0003777<br>0.0003777<br>0.0003777<br>0.0003777<br>0.000377<br>0.000377<br>0.000377<br>0.000377<br>0.000377<br>0.000464<br>0.000464<br>0.000464<br>0.000464 | - 3566842<br>- 3566842<br>1.8789238<br>2.0977776<br>2.0977776<br>2.28217082<br>2.790207<br>2.8217082<br>2.7301698<br>3.0191874<br>2.4772701<br>2.5037825<br>2.7392982<br>3.3311649<br>4.68371<br>4.68371<br>4.68371<br>0.0463027<br>0<br>0.0463027<br>0<br>0.0463027<br>0<br>0.0450174<br>0.551144<br>0.5506791<br>0.5506791<br>0.5506791<br>0.5506791<br>0.5506794<br>0.7736461<br>0.77736461<br>0.77736461<br>0.77736461<br>0.77736461<br>0.77736461<br>0.77736461<br>0.77736461<br>0.77736461<br>0.77736461<br>0.77736461<br>0.77736461<br>0.77736461<br>0.77736461<br>0.77736461<br>0.77736461<br>0.77736461<br>0.77736461<br>0.77736461<br>0.77736461<br>0.77736461<br>0.77736461<br>0.77736461<br>0.77736461<br>0.77736461<br>0.77736461<br>0.77736461<br>0.77736461<br>0.77736461<br>0.77736461<br>0.77736461<br>0.77736461<br>0.77736461<br>0.77736461<br>0.77736461<br>0.77736461<br>0.77736461<br>0.77736461<br>0.77736461<br>0.77736461<br>0.77736461<br>0.77736461<br>0.77736461<br>0.77736461<br>0.77736461<br>0.77736461<br>0.77736461<br>0.77736461<br>0.77736461<br>0.77736461<br>0.77736461<br>0.77736461<br>0.77736461<br>0.77736461<br>0.77736461<br>0.77736461<br>0.77736461<br>0.77736461<br>0.77736461<br>0.77736461<br>0.77736461<br>0.77736461<br>0.77736461<br>0.77736461<br>0.77736461<br>0.77736461<br>0.77736461<br>0.77736461<br>0.77736461<br>0.77736461<br>0.77736461<br>0.77736461<br>0.77736461<br>0.77736461<br>0.77736461<br>0.77736461<br>0.77736461<br>0.77736461<br>0.77736461<br>0.57736451<br>0.77736461<br>0.77736461<br>0.77736461<br>0.77736461<br>0.77736461<br>0.77736461<br>0.77736461<br>0.57736451<br>0.57736451<br>0.57736451<br>0.57736451<br>0.77736451<br>0.77736451<br>0.77736451<br>0.77736451<br>0.77736451<br>0.77736451<br>0.77736451<br>0.77736451<br>0.77736451<br>0.77736451<br>0.77736451<br>0.77736451<br>0.77736451<br>0.77736451<br>0.77736451<br>0.77736451<br>0.77736451<br>0.77736451<br>0.77736451<br>0.77736451<br>0.77736451<br>0.77736451<br>0.77736451<br>0.77736451<br>0.77736451<br>0.77736451<br>0.77736451<br>0.77736451<br>0.77736451<br>0.77736451<br>0.77736451<br>0.77736451<br>0.77736451<br>0.77736451<br>0.77736451<br>0.77736451<br>0.77736451<br>0.77736451<br>0.77745451<br>0.77745451<br>0.77745451<br>0.77745451<br>0.77745451<br>0.77745451<br>0.77745451<br>0.77745451<br>0.77745451<br>0.7774555<br>0.7774555<br>0.7775555<br>0.7775555<br>0.7775555<br>0.77755555<br>0.77755555555555555555555   | 1.7622-06<br>1.2642-06<br>1.2852-06<br>1.3452-06<br>1.2852-06<br>1.2632-06<br>1.2632-06<br>1.2632-06<br>1.2632-06<br>1.3272-06<br>1.3372-06<br>1.3372-06<br>1.3372-06<br>1.3372-06<br>1.3372-06<br>1.3372-06<br>1.3372-06<br>1.3372-06<br>1.3372-06<br>1.3372-06<br>1.3372-06<br>1.3372-06<br>1.3372-06<br>1.3372-06<br>1.3372-06<br>1.3372-06<br>1.3372-06<br>1.3372-06<br>1.3372-06<br>1.3372-06<br>1.3372-06<br>1.3372-07<br>2.542-07<br>2.3492-07<br>2.3492-07<br>2.3492-07<br>2.3492-07<br>2.3492-07  | 0.0028636<br>0.0021569<br>0.0021848<br>0.0020855<br>0.0022167<br>0.0020815<br>0.0025171<br>0.0019592<br>0.0025204<br>0.002150<br>0.002150<br>0.002150<br>0.0022425<br>0.002429<br>0.0020429<br>0.0002419<br>0.0000411<br>1.305E-05<br>0.00004584<br>0.0004584<br>0.0004584<br>0.0004584<br>0.0004584<br>0.0008316  
   | 2.2440968<br>1.7795138<br>1.9915216<br>1.8861342<br>2.0102391<br>2.7864095<br>2.7720732<br>2.6462899<br>2.5622851<br>2.8364013<br>2.3212589<br>2.37226056<br>2.3321589<br>2.3726056<br>2.4375101<br>2.6016999<br>3.7701495<br>4.4767836<br>3.3755324<br>0.040785<br>0<br>1.1398552<br>1.14138252<br>0.041775<br>0.04987676<br>0.42840418<br>0.5019047<br>0.5019047<br>0.5019047<br>0.5019047<br>0.5019047<br>0.5019047<br>0.5019047<br>0.5019047  | 1.6782-06<br>1.1412-06<br>1.2772-06<br>1.2152-06<br>1.12892-06<br>1.1312-06<br>1.1312-06<br>1.1312-06<br>1.3452-06<br>1.3452-06<br>1.3452-06<br>1.3222-06<br>8.3392-07<br>1.1362-06<br>0<br>0<br>0<br>0<br>1.7312-07<br>1.7332-07<br>6.3352-09<br>1.2382-07<br>1.2382-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2892-07<br>2.2892-07<br>2.2892-07  | 0.0027267<br>0.0018534<br>0.0020742<br>0.0019748<br>0.0019748<br>0.0019748<br>0.0019347<br>0.0019347<br>0.0019347<br>0.0019347<br>0.002238<br>0.002248<br>0.002248<br>0.002248<br>0.002248<br>0.002248<br>0.002248<br>0.002248<br>0.002248<br>0.002248<br>0.002415<br>0.001947<br>0<br>0.0003163<br>0.0002754<br>0.0002754<br>0.0002754<br>0.0002417<br>0.0002417<br>0.0002417<br>0.0002417<br>0.0002417<br>0.0002417<br>0.0002417<br>0.0002417<br>0.0002417<br>0.0002417<br>0.0000417<br>0.0000417<br>0.0000417<br>0.0000417<br>0.000045<br>0.0000554  
   | 2.623364<br>2.623364<br>2.11248<br>2.419219<br>2.2439219<br>3.2614464<br>3.1928226<br>3.1051925<br>3.0057113<br>3.3257455<br>2.7552623<br>2.7582623<br>2.7582623<br>2.7818703<br>3.3627176<br>3.0470013<br>3.3621382<br>3.1979947<br>4.6427163<br>0.064669<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0   | 19622.06<br>1.3542.06<br>1.3542.06<br>1.4382.06<br>1.4382.06<br>1.3372.06<br>1.3772.06<br>1.3772.06<br>1.3772.06<br>1.4722.06<br>1.4722.06<br>1.4722.06<br>1.4722.06<br>1.4722.06<br>1.4822.06<br>1.4882.06<br>1.6662.06<br>9.182.09<br>0<br>0<br>2.8772.07<br>2.8772.07<br>2.8772.07<br>2.8772.07<br>3.7632.07<br>3.7632.07<br>3.7632.07<br>3.7452.07<br>3.7452.07<br>3.7452.07<br>3.7452.07<br>3.7452.07<br>3.7452.07<br>3.7452.07<br>3.7452.07<br>3.7452.07<br>3.7452.07<br>3.7452.07<br>3.7452.07<br>3.7452.07<br>3.7452.07<br>3.7452.07<br>3.7452.07<br>3.7452.07<br>3.7452.07<br>3.7452.07<br>3.7452.07<br>3.7452.07<br>3.7452.07<br>3.7452.07<br>3.7452.07<br>3.7452.07<br>3.7452.07<br>3.7452.07<br>3.7452.07<br>3.7452.07<br>3.7452.07<br>3.7452.07<br>3.7452.07<br>3.7452.07<br>3.7452.07<br>3.7452.07<br>3.7452.07<br>3.7452.07<br>3.7452.07<br>3.7452.07<br>3.7452.07<br>3.7452.07<br>3.7452.07<br>3.7452.07<br>3.7452.07<br>3.7452.07<br>3.7452.07<br>3.7452.07<br>3.7452.07<br>3.7452.07<br>3.7452.07<br>3.7452.07<br>3.7452.07<br>3.7452.07<br>3.7452.07<br>3.7452.07<br>3.7452.07<br>3.7452.07<br>3.7452.07<br>3.7452.07<br>3.7452.07<br>3.7452.07<br>3.7452.07<br>3.7452.07<br>3.7452.07<br>3.7452.07<br>3.7452.07<br>3.7452.07<br>3.7452.07<br>3.7452.07<br>3.7452.07<br>3.7452.07<br>3.7452.07<br>3.7452.07<br>3.7452.07<br>3.7452.07<br>3.7452.07<br>3.7452.07<br>3.7452.07<br>3.7452.07<br>3.7452.07<br>3.7452.07<br>3.7452.07<br>3.7452.07<br>3.7452.07<br>3.7452.07<br>3.7452.07<br>3.7452.07<br>3.7452.07<br>3.7452.07<br>3.7452.07<br>3.7452.07<br>3.7452.07<br>3.7452.07<br>3.7452.07<br>3.7452.07<br>3.7452.07<br>3.7452.07<br>3.7452.07<br>3.7452.07<br>3.7452.07<br>3.7452.07<br>3.7452.07<br>3.7452.07<br>3.7452.07<br>3.7452.07<br>3.7452.07<br>3.7452.07<br>3.7452.07<br>3.7452.07<br>3.7452.07<br>3.7452.07<br>3.7452.07<br>3.7452.07<br>3.7452.07<br>3.7452.07<br>3.7452.07<br>3.7452.07<br>3.7452.07<br>3.7452.07<br>3.7452.07<br>3.7452.07<br>3.7452.07<br>3.7452.07<br>3.7452.07<br>3.7452.07<br>3.7452.07<br>3.7452.07<br>3.7452.07<br>3.7452.07<br>3.7452.07<br>3.7452.07<br>3.7452.07<br>3.7452.07<br>3.7452.07<br>3.7452.07<br>3.7452.07<br>3.7452.07<br>3.7452.07<br>3.7452.07<br>3.7452.07<br>3.7452.07<br>3.7452.07<br>3.7452.07<br>3.7452.07<br>3.7452.07<br>3.7452.07<br>3.7452.07<br>3.7452.07<br>3.7452.07<br>3.7452.07<br>3.7452.07<br>3.7452.07<br>3.7452.07<br>3.7452.07<br>3.7452.07<br>3.7452.07<br>3.7452.07<br>3.7452.07<br>3.7452.07<br>3.7452.07<br>3.7452.     | 0.0031876<br>0.0022001<br>0.0024433<br>0.002347<br>0.0023497<br>0.00224967<br>0.0022645<br>0.002711<br>0.002156<br>0.002545<br>0.0026885<br>0.0024144<br>0.0012867<br>0.00227068<br>0.00227068<br>0.0022716<br>0.0022706<br>0.00025245<br>0.0005245<br>0.0005245<br>0.0005245<br>0.0005245<br>0.0005245<br>0.0005245<br>0.0005245<br>0.0005245<br>0.0005245<br>0.0005245<br>0.0005245<br>0.0005245<br>0.0005245<br>0.0005245<br>0.0005245<br>0.00055245<br>0.00055245<br>0.00055245<br>0.00055245<br>0.00055245<br>0.00055245<br>0.00055245<br>0.00055245<br>0.00055245<br>0.00055245<br>0.00055245<br>0.0005537   | 2.3644564<br>1.8937445<br>2.1225909<br>2.0177733<br>2.1529112<br>2.928047<br>2.8544498<br>2.7808064<br>2.6943024<br>2.9747351<br>2.452139<br>2.4915395<br>2.724646<br>3.8926302<br>4.6916603<br>4.1449386<br>0<br>0.043318<br>0<br>0.043318<br>0.058454<br>0.0584545<br>0.6884638<br>0.6879172<br>0.982715<br>0.982715   
   | 1.768:0-0 0.002<br>1.361:0-0 0.002<br>1.381:0-0 0.002<br>1.381:0-0 0.002<br>1.381:0-0 0.002<br>1.525:0-0 0.002<br>1.525:0-0 0.002<br>1.525:0-0 0.002<br>1.535:0-0 0.002<br>1.535:0-0 0.002<br>1.531:0-0 0.002<br>1.551:0-0 0.002<br>1.551:   | 873         0.04           9723         0.04           9723         0.03           2107         0.04           1015         0.04           1016         0.04           2422         0.04           033         0.05           9308         0.05           9308         0.05           9318         0.05           1284         0.04           1284         0.04           1284         0.04           1284         0.04           1284         0.04           1284         0.05           0.271         0.07           4139         0.05           41393         0.04           4393         0.01           7371         0.02           7341         0.02           7341         0.02           1244         0.02  | 447043         3.           447043         3.           838322         2.           408499         2.           405571         1           343125         2.           338011         2.           338014         2.           338015         3.1           314016         2.           4655511         3.4           157074         2.           167677         2.           188276         2.           183811         1.           167677         2.           167677         2.           163831         1.     
     163831         1.           165767         2.           164767         2.           165767         2.           165393         8.1           0         0           0         1.           104252         6.4           105429         9.5           10642         9.5           10642         9.5           10642         9.5           107571         8.4           76717         8.4  
  | 5.18E-00         2           6.619E-03         2           6.619E-03         2           6.619E-03         2           7.76E-08         4           2.99E-08         3           6.51E-08         4           0.21E-08         4           0.21E-08         4           0.21E-08         4           4.42E-08         3           5.558E-08         4           6.558E-08         4           6.558E-08         4           6.61E-08         2           4.48E-08         4           5.58E-08         4           6.51E-08         4           6.51E-08         4           6.61E-08         2           4.48E-08         4           5.58E-08         4           6.61E-08         2           4.56E-09         1           5.33E-00         1           5.33E-00         1           5.34E-00         1           5.34E-00         1           5.34E-00         1           5.34E-00         1           5.34E-00         1           5.34E-00         1                | 3.998E-05<br>4.254E-05<br>4.224E-05<br>4.224E-05<br>4.511E-05<br>3.736E-05<br>4.67E-05<br>4.67E-05<br>4.307E-05<br>4.307E-05<br>4.307E-05<br>4.098E-05<br>4.098E-05<br>4.098E-05<br>4.098E-05<br>4.098E-05<br>4.098E-05<br>4.098E-05<br>4.098E-05<br>4.501E-05<br>1.188E-05<br>0<br>0<br>0<br>1.188E-05<br>1.188E-05<br>1.158E-05<br>1.1558E-05<br>1.536E-05<br>1.536E-05<br>1.536E-05   | 0.041322<br>0.0443322<br>0.044017<br>0.0470156<br>0.0584373<br>0.05584373<br>0.0558211<br>0.0538871<br>0.061206<br>0.0496292<br>0.0507781<br>0.0496292<br>0.0527943<br>0.0527943<br>0.0452794<br>0<br>0.04662<br>0.04662<br>0.0452794<br>0<br>0.04662<br>0.0452794<br>0<br>0.04662<br>0.0452794<br>0<br>0.04662<br>0.0452794<br>0.0452794<br>0.0228169<br>0.0228169<br>0.0228169<br>0.0228169<br>0.0228169<br>0.0228169<br>0.0228169<br>0.0228169<br>0.0228169<br>0.0228169<br>0.0228169<br>0.0228169<br>0.0228169<br>0.0228169<br>0.0228169<br>0.0228169<br>0.0228169<br>0.0228169<br>0.0228169<br>0.0228169<br>0.0228169<br>0.0228169<br>0.0228169<br>0.0228169<br>0.0228169<br>0.0228169<br>0.0228169<br>0.0228169<br>0.0228169<br>0.0228169<br>0.0228169<br>0.0228169<br>0.0228169<br>0.0228169<br>0.0228169<br>0.0228169<br>0.0228169<br>0.0228169<br>0.0228169<br>0.0228169<br>0.0228169<br>0.0228169<br>0.0228169<br>0.0228169<br>0.0228169<br>0.0228169<br>0.0228169<br>0.0228169<br>0.0228169<br>0.0228169<br>0.0228169<br>0.0228169<br>0.0228169<br>0.0228169<br>0.0228169<br>0.0228169<br>0.0228169<br>0.0228169<br>0.0228169<br>0.0228169<br>0.0228169<br>0.0228169<br>0.0228169<br>0.0228169<br>0.0228169<br>0.0228169<br>0.0228169<br>0.0228169<br>0.0228169<br>0.0228169<br>0.0228169<br>0.0228169<br>0.0228169<br>0.0228169<br>0.0228169<br>0.0228169<br>0.0228169<br>0.0228169<br>0.0228169<br>0.0228169<br>0.0228169<br>0.0228169<br>0.0228169<br>0.0228169<br>0.0228169<br>0.0228169<br>0.0228169<br>0.0228169<br>0.0228169<br>0.0228169<br>0.0228169<br>0.0228169<br>0.0228169<br>0.0228169<br>0.0228169<br>0.0228169<br>0.0228169<br>0.0228169<br>0.0228169<br>0.0228169<br>0.0228169<br>0.0228169<br>0.0228169<br>0.0228169<br>0.0228169<br>0.0228169<br>0.0228169<br>0.0228169<br>0.0228169<br>0.0228169<br>0.0228169<br>0.0228169<br>0.0228169<br>0.0228169<br>0.0228169<br>0.0228169<br>0.0228169<br>0.0228169<br>0.0228169<br>0.0228169<br>0.0228169<br>0.0228169<br>0.0228169<br>0.0228169<br>0.0228169<br>0.0228169<br>0.0228169<br>0.0228169<br>0.0228169<br>0.0228169<br>0.0228169<br>0.0228169<br>0.0228169<br>0.0228169<br>0.0228169<br>0.0228169<br>0.0228169<br>0.0228169<br>0.0228169<br>0.0228169<br>0.0228169<br>0.0228169<br>0.0228169<br>0.0228169<br>0.0228169<br>0.0228169<br>0.0228169<br>0.0228169<br>0.0228169<br>0.0228169<br>0.0228169<br>0.0228169<br>0.0228169<br>0.0228169<br>0.0228169<br>0.0228169<br>0.0228169<br>0.0228169<br>0.0228169<br>0.0228169<br>0.0228169<br>0.0228169<br>0.0228169<br>0.0228169<br>0.0228169<br>0.0228169<br>0.0228169<br>0.0228169<br>0.0228169<br>0.0228169<br>0.0228169<br>0.0228169<br>0.0228169<br>0.0228169<br>0.0228169<br>0.0228169<br>0.0228169<br>0.0228169<br>0.0228169<br>0.02281690   | 2.071-003         2           2.8212-03         2           2.8212-03         2           3.814-03         2           2.497-08         2           2.385-08         2           2.385-08         2           2.385-08         2           2.381-08         2           2.379-08         2           2.321-08         2           2.713-08         2        
  2.703-08         2           3.005-08         2           2.703-08         3           3.005-08         4           2.825-08         4           3.005-08         4           3.775-10         0           0         0           0         0           0         0           0         0           0         0           0         0           0         0           0         0           0         0           0         0           0         0           0         0           0         0           0         0           0  | 1.584-05<br>1.897E-05<br>1.057E-05<br>3.072E-05<br>3.376E-05<br>3.331E-05<br>1.307E-05<br>3.331E-05<br>1.307E-05<br>1.307E-05<br>1.393E-05<br>1.393E-05<br>1.393E-05<br>1.636E-05<br>0<br>0<br>0<br>1.282E-05<br>1.282E-05<br>1.282E-05<br>1.282E-05<br>1.282E-05<br>1.282E-05<br>1.2901E-05<br>1.901E-05<br>1.901E-05<br>1.901E-05<br>1.901E-05<br>1.901E-05<br>1.901E-05<br>1.901E-05<br>1.901E-05<br>1.901E-05<br>1.901E-05<br>1.901E-05<br>1.901E-05<br>1.901E-05<br>1.901E-05<br>1.901E-05<br>1.901E-05<br>1.901E-05<br>1.901E-05<br>1.901E-05<br>1.901E-05<br>1.901E-05<br>1.901E-05<br>1.901E-05<br>1.667E-05   |
| Sheung Touen B/T and Car Park<br>Sheung Touen B/T and Car Park<br>Ullage Access Road<br>Village Access Road                         | RD_29<br>RD_29<br>RD_29<br>RD_29<br>RD_29<br>RD_29<br>RD_29<br>RD_29<br>RD_29<br>RD_29<br>RD_29<br>RD_29<br>RD_29<br>RD_29<br>RD_29<br>RD_29<br>RD_29<br>RD_29<br>RD_29<br>RD_29<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20 | 9900.1000           1100-1100           1100-1200           1100-1200           1100-1200           1200-1300           1200-1300           1401-100           1500-1600           1600-1700           1700-1800           1800-2000           200-2100           2100-2200           2200-3000           000-0100           000-0100           000-0100           000-0100           000-0100           000-0100           000-0000           000-0100           000-0000           000-0100           000-0100           000-0100           000-0100           000-0100           000-0000           0000-0100           1000-1200           1000-1200           1000-1200           100-1200           100-1200           1300-1400           1400-1500  | 0.1281945<br>0.1457065<br>0.150723<br>0.1607293<br>0.1452756<br>0.1452756<br>0.1482253<br>0.1482452<br>0.1452588<br>0.132275<br>0.1792504<br>0.232202<br>0.1352588<br>0.0323275<br>0.1792504<br>0.232402<br>0.0305911<br>0.5929911<br>0.5929911<br>0.5929911<br>0.5929911<br>0.1864616<br>0.233492<br>0.236540<br>0.234492<br>0.2224179<br>0.2224179<br>0.2224179<br>0.2224179<br>0.2224179<br>0.2224179<br>0.2224179<br>0.2224179<br>0.2224179<br>0.2224179<br>0.2224179<br>0.2224179<br>0.2224179<br>0.2224179<br>0.2224179<br>0.2224179<br>0.2224179<br>0.2224179<br>0.2224179<br>0.2224179<br>0.2224179<br>0.2224179<br>0.2224179<br>0.2224179<br>0.2224179<br>0.2224179<br>0.2224179<br>0.2224179<br>0.2224179<br>0.2224179<br>0.2224179<br>0.2224179<br>0.2224179<br>0.2224179<br>0.2224179<br>0.2224179<br>0.2224179<br>0.2224179<br>0.2224179<br>0.2224179<br>0.2224179<br>0.2224179<br>0.2224179<br>0.2224179<br>0.2224179<br>0.2224179<br>0.2224179<br>0.2224179<br>0.2224179<br>0.2224179<br>0.2224179<br>0.2224179<br>0.2224179<br>0.2224179<br>0.2224179<br>0.2224179<br>0.2224179<br>0.2224179<br>0.2224179<br>0.2224179<br>0.2224179<br>0.2224179<br>0.2224179<br>0.2224179<br>0.2224179<br>0.2224179<br>0.2224179<br>0.2224179<br>0.2224179<br>0.2224179<br>0.2224179<br>0.2224179<br>0.2224179<br>0.2224179<br>0.2224179<br>0.2224179<br>0.2224179<br>0.2224179<br>0.2224179<br>0.2224179<br>0.2224179<br>0.2224179<br>0.2224179<br>0.2224179<br>0.2224179<br>0.2224179<br>0.2224179<br>0.2224179<br>0.2224179<br>0.2224179<br>0.2224179<br>0.2224179<br>0.2224179<br>0.2224179<br>0.2224179<br>0.2224179<br>0.2224179<br>0.2224179<br>0.224179<br>0.224179<br>0.224179<br>0.224179<br>0.224179<br>0.224179<br>0.224179<br>0.224179<br>0.224179<br>0.224179<br>0.224179<br>0.224179<br>0.224179<br>0.224179<br>0.224179<br>0.224179<br>0.224179<br>0.224179<br>0.224179<br>0.224179<br>0.224179<br>0.224179<br>0.224179<br>0.224179<br>0.224179<br>0.224179<br>0.224179<br>0.224179<br>0.224179<br>0.224179<br>0.224179<br>0.224179<br>0.224179<br>0.224179<br>0.224179<br>0.224179<br>0.224179<br>0.224179<br>0.224179<br>0.224179<br>0.224179<br>0.224179<br>0.224179<br>0.224179<br>0.224179<br>0.224179<br>0.224179<br>0.224179<br>0.224179<br>0.224179<br>0.224179<br>0.224179<br>0.224179<br>0.224179<br>0.224179<br>0.224179<br>0.224179<br>0.224179<br>0.224179<br>0.224179<br>0.224179<br>0.224179<br>0.224179<br>0.224179<br>0.224179<br>0.224179<br>0.224179<br>0.224179<br>0.224179<br>0.224179<br>0.224179<br>0.224179<br>0.224179<br>0.224179<br>0.224179<br>0.224179<br>0.224179<br>0.224179<br>0.224179<br>0.224179<br>0.224179<br>0.224179<br>0.224179<br>0.224179<br>0.224179<br>0.224179<br>0.224179<br>0.224179<br>0.224179<br>0.224179<br>0.224179<br>0.24 | 9.342-08   8.2188-08   9.342-08   9.342-08   9.342-08   8.2188-08   8.2188-08   8.6352-08   6.6552-08   6.6552-08   6.6552-08   6.6352-08   6.3352-08   7.3352-08  
7.3352-08   7.352-08   7.352-08   7.352-08   7.352-08   7.352-08   7.3   | 0.0001315 (<br>0.0001335 (<br>0.0001345 (<br>0.0001417 (<br>0.0001417 (<br>0.000147 (<br>0.000147 (<br>0.000147 (<br>0.000126 (<br>0.000126 (<br>0.0001324 (<br>0.000136 (<br>0.000136 (<br>0.000146 (<br>0.000146 (<br>0.000127 (<br>0.000146 (<br>0.000147 (<br>0.000   |  | 9.43114.08<br>9.43114.08<br>7.76114.08<br>8.28444.08<br>9.75444.08<br>9.75444.08<br>6.37344.08<br>7.4734-08<br>6.08214.08<br>7.465-08<br>6.77114-08<br>7.465-08<br>6.77144.08<br>7.4174-08<br>7.41524.08<br>7.452-08<br>3.4694-10<br>0<br>7.2894-08<br>7.2654-08<br>3.3694-10<br>6.1164-08<br>5.1734-08<br>9.3574-08<br>9.3574-08<br>9.3574-08<br>9.3574-08<br>9.3574-08<br>9.3574-08<br>9.3574-08<br>9.3574-08<br>9.3574-08<br>9.3544-08<br>9.3544-08<br>9.3544-08<br>9.3544-08<br>9.3544-08<br>9.3544-08<br>9.3544-08<br>9.3544-08<br>9.3544-08<br>9.3544-08<br>9.3544-08<br>9.3544-08<br>9.3544-08<br>9.3544-08<br>9.3544-08<br>9.3544-08<br>9.3544-08<br>9.3544-08<br>9.3544-08<br>9.3544-08<br>9.3544-08<br>9.3544-08<br>9.3544-08<br>9.3544-08<br>9.3544-08<br>9.3544-08<br>9.3544-08<br>9.3544-08<br>9.3544-08<br>9.3544-08<br>9.3544-08<br>9.3544-08<br>9.3544-08<br>9.3544-08<br>9.3544-08<br>9.3544-08<br>9.3544-08<br>9.3544-08<br>9.3544-08<br>9.3544-08<br>9.3544-08<br>9.3544-08<br>9.3544-08<br>9.3544-08<br>9.3544-08<br>9.3544-08<br>9.3544-08<br>9.3544-08<br>9.3544-08<br>9.3544-08<br>9.3544-08<br>9.3544-08<br>9.3544-08<br>9.3544-08<br>9.3544-08<br>9.3544-08<br>9.3544-08<br>9.3544-08<br>9.3544-08<br>9.3544-08<br>9.3544-08<br>9.3544-08<br>9.3544-08<br>9.3544-08<br>9.3544-08<br>9.3544-08<br>9.3544-08<br>9.3544-08<br>9.3544-08<br>9.3544-08<br>9.3544-08<br>9.3544-08<br>9.3544-08<br>9.3544-08<br>9.3544-08<br>9.3544-08<br>9.3544-08<br>9.3544-08<br>9.3544-08<br>9.3544-08<br>9.3544-08<br>9.3544-08<br>9.3544-08<br>9.3544-08<br>9.3544-08<br>9.3544-08<br>9.3544-08<br>9.3544-08<br>9.3544-08<br>9.3544-08<br>9.3544-08<br>9.3544-08<br>9.3544-08<br>9.3544-08<br>9.3544-08<br>9.3544-08<br>9.3544-08<br>9.3544-08<br>9.3544-08<br>9.3544-08<br>9.3544-08<br>9.3544-08<br>9.3544-08<br>9.3544-08<br>9.3544-08<br>9.3544-08<br>9.3544-08<br>9.3544-08<br>9.3544-08<br>9.3544-08<br>9.3544-08<br>9.3544-08<br>9.3544-08<br>9.3544-08<br>9.3544-08<br>9.3544-08<br>9.3544-08<br>9.3544-08<br>9.3544-08<br>9.3544-08<br>9.3544-08<br>9.3544-08<br>9.3544-08<br>9.3544-08<br>9.3544-08<br>9.3544-08<br>9.3544-08<br>9.3544-08<br>9.3544-08<br>9.3544-08<br>9.3544-08<br>9.3544-08<br>9.3544-08<br>9.3544-08<br>9.3544-08<br>9.3544-08<br>9.3544-08<br>9.3544-08<br>9.3544-08<br>9.3544-08<br>9.3544-08<br>9.3544-08<br>9.3544-08<br>9.3544-08<br>9.3544-08<br>9.3544-08<br>9.3544-08<br>9.3544-08<br>9.3544-08<br>9.3544-08<br>9.3544-08<br>9.3544-08<br>9.3544-08<br>9.3544-08   | 0.0001532<br>0.0001532<br>0.0001261<br>0.0001435<br>0.0001345<br>0.0001345<br>0.000135<br>0.0001285<br>0.0001268<br>0.000112<br>0.000122<br>0.000122<br>0.00011095<br>6.689F-05<br>0.000113<br>0.0001095<br>6.944E-07<br>0<br>0<br>0.0001332<br>0.0001332<br>0.000132<br>0.000132<br>0.000132<br>0.000132<br>0.000132<br>0.000132<br>0.000132<br>0.0001118<br>9.455E-05<br>0.000171<br>0.000171<br>0.0001715<br>0.000171<br>0.0001715<br>0.000171   |
0.1203596<br>0.1142030<br>0.1140591<br>0.1140591<br>0.1426721<br>0.1426721<br>0.1426721<br>0.1323766<br>0.13435164<br>0.1320155<br>0.120155<br>0.120155<br>0.120155<br>0.120155<br>0.120155<br>0.1224876<br>0.1248766<br>0.1549462<br>0.0022355<br>0.1559933<br>0.1559933<br>0.1559933<br>0.1559933<br>0.1559933<br>0.1559933<br>0.1559933<br>0.1559933<br>0.1559933<br>0.1559933<br>0.1559933<br>0.1559933<br>0.1559933<br>0.1559933<br>0.1559933<br>0.1559933<br>0.1559933<br>0.1559933<br>0.1559933<br>0.1559933<br>0.1559933<br>0.1559933<br>0.1559933<br>0.255644<br>0.255644<br>0.256645<br>0.2648624<br>0.2648624<br>0.2648624<br>0.2648624<br>0.2648624<br>0.2648624<br>0.2648624<br>0.2648624<br>0.2648624<br>0.2648624<br>0.2648624<br>0.2648624<br>0.2648624<br>0.2648624<br>0.2648624<br>0.2648624<br>0.2648624<br>0.2648624<br>0.2648624<br>0.2648624<br>0.2648624<br>0.2648624<br>0.2648624<br>0.2648624<br>0.2648624<br>0.2648624<br>0.2648624<br>0.2648624<br>0.2648624<br>0.2648624<br>0.2648624<br>0.2648624<br>0.2648624<br>0.2648624<br>0.2648624<br>0.2648624<br>0.2648624<br>0.2648624<br>0.2648624<br>0.2648624<br>0.2648624<br>0.2648624<br>0.2648624<br>0.2648624<br>0.2648624<br>0.2648624<br>0.2648624<br>0.2648624<br>0.2648624<br>0.2648624<br>0.2648624<br>0.2648624<br>0.2648624<br>0.2648624<br>0.2648624<br>0.2648624<br>0.2648624<br>0.2648624<br>0.2648624<br>0.2648624<br>0.2648624<br>0.2648624<br>0.2648624<br>0.2648624<br>0.2648624<br>0.2648624<br>0.2648624<br>0.2648624<br>0.2648624<br>0.2648624<br>0.2648624<br>0.2648624<br>0.2648624<br>0.2648624<br>0.2648624<br>0.2648624<br>0.2648624<br>0.2648624<br>0.2648624<br>0.2648624<br>0.2648624<br>0.2648624<br>0.2648624<br>0.2648624<br>0.2648624<br>0.2648624<br>0.2648624<br>0.2648624<br>0.2648624<br>0.2648624<br>0.2648624<br>0.2648624<br>0.2648624<br>0.2648624<br>0.2648624<br>0.2648624<br>0.2648624<br>0.2648624<br>0.2648624<br>0.2648624<br>0.2648624<br>0.2648624<br>0.2648624<br>0.2648624<br>0.2648624<br>0.2648624<br>0.2648624<br>0.2648624<br>0.2648624<br>0.2648624<br>0.2648624<br>0.2648624<br>0.2648624<br>0.2648624<br>0.2648624<br>0.2648624<br>0.2648624<br>0.2648624<br>0.2648624<br>0.2648624<br>0.2648624<br>0.2648624<br>0.2648624<br>0.2648624<br>0.2648624<br>0.2648624<br>0.2648624<br>0.2648624<br>0.2648624<br>0.2648624<br>0.2648624<br>0.2648624<br>0.2648624<br>0.2648624<br>0.2648624<br>0.2648624<br>0.2648624<br>0.2648624<br>0.2648624<br>0.2648624<br>0.2648624<br>0.2648624<br>0.2648624<br>0.2648624<br>0.2648624<br>0.2648624<br>0.2648624<br>0.2648624<br>0.2648624<br>0.2648624<br>0.2648624<br>0.2648624<br>0.2648624<br>0.2648624<br>0.2648624<br>0.2648624<br>0.2648624<br>0.2648624<br>0.2648624<br>0.2648624<br>0.2  | 9.001-08<br>7.322E.08<br>7.707E.08<br>9.146E.08<br>6.053E.08<br>7.707E.08<br>7.39E.08<br>6.439E.08<br>6.439E.08<br>6.439E.08<br>6.439E.08<br>6.439E.08<br>6.439E.08<br>6.439E.08<br>6.439E.08<br>6.439E.08<br>6.439E.08<br>6.439E.08<br>6.837E.08<br>6.73E.08<br>6.73E.08<br>6.73E.08<br>6.73E.08<br>6.73E.08<br>6.73E.08<br>6.73E.08<br>8.542E.08<br>8.542E.08<br>8.542E.08<br>8.542E.08<br>8.525E.08<br>8.525E.08<br>8.525E.08<br>8.527E.08<br>8.527E.08<br>8.527E.08<br>8.527E.08<br>8.527E.08<br>8.527E.08<br>8.527E.08<br>8.527E.08<br>8.527E.08<br>8.527E.08<br>8.527E.08<br>8.527E.08<br>8.527E.08<br>8.527E.08<br>8.527E.08<br>8.527E.08<br>8.527E.08<br>8.527E.08<br>8.527E.08<br>8.527E.08<br>8.527E.08<br>8.527E.08<br>8.527E.08<br>8.527E.08<br>8.527E.08<br>8.527E.08<br>8.527E.08<br>8.527E.08<br>8.527E.08<br>8.527E.08<br>8.527E.08<br>8.527E.08<br>8.527E.08<br>8.527E.08<br>8.527E.08<br>8.527E.08<br>8.527E.08<br>8.527E.08<br>8.527E.08<br>8.527E.08<br>8.527E.08<br>8.527E.08<br>8.527E.08<br>8.527E.08<br>8.527E.08<br>8.527E.08<br>8.527E.08<br>8.527E.08<br>8.527E.08<br>8.527E.08<br>8.527E.08<br>8.527E.08<br>8.527E.08<br>8.527E.08<br>8.527E.08<br>8.527E.08<br>8.527E.08<br>8.527E.08<br>8.527E.08<br>8.527E.08<br>8.527E.08<br>8.527E.08<br>8.527E.08<br>8.527E.08<br>8.527E.08<br>8.527E.08<br>8.527E.08<br>8.527E.08<br>8.527E.08<br>8.527E.08<br>8.527E.08<br>8.527E.08<br>8.527E.08<br>8.527E.08<br>8.527E.08<br>8.527E.08<br>8.527E.08<br>8.527E.08<br>8.527E.08<br>8.527E.08<br>8.527E.08<br>8.527E.08<br>8.527E.08<br>8.527E.08<br>8.527E.08<br>8.527E.08<br>8.527E.08<br>8.527E.08<br>8.527E.08<br>8.527E.08<br>8.527E.08<br>8.527E.08<br>8.527E.08<br>8.527E.08<br>8.527E.08<br>8.527E.08<br>8.527E.08<br>8.527E.08<br>8.527E.08<br>8.527E.08<br>8.527E.08<br>8.527E.08<br>8.527E.08<br>8.527E.08<br>8.527E.08<br>8.527E.08<br>8.527E.08<br>8.527E.08<br>8.527E.08<br>8.527E.08<br>8.527E.08<br>8.527E.08<br>8.527E.08<br>8.527E.08<br>8.527E.08<br>8.527E.08<br>8.527E.08<br>8.527E.08<br>8.527E.08<br>8.527E.08<br>8.527E.08<br>8.527E.08<br>8.527E.08<br>8.527E.08<br>8.527E.08<br>8.527E.08<br>8.527E.08<br>8.527E.08<br>8.527E.08<br>8.527E.08<br>8.527E.08<br>8.527E.08<br>8.527E.08<br>8.527E.08<br>8.527E.08<br>8.527E.08<br>8.527E.08<br>8.527E.08<br>8.527E.08<br>8.527E.08<br>8.527E.08<br>8.527E.08<br>8.527E.08<br>8.527E.08<br>8.527E.08<br>8.527E.08<br>8.527E.08<br>8.527E.08<br>8.527E.08<br>8.527E.08<br>8.527E.08<br>8.527E.08<br>8.527E.08<br>8.527E.08<br>8.527E.08<br>8.527E.08<br>8.527E.08<br>8.527E.08<br>8.527E.08<br>8.527E.08<br>8.527E.08<br>8.527E.08<br>8.527E.08<br>8.527E.08<br>8.527E.08<br>8.527E.08<br>8.527E.08<br>8.527E.08<br>8.527E.08<br>8.527E.08<br>8.527E.08<br>8.527E.08<br>8.527             | 0.0001462<br>0.0001462<br>0.0001365<br>0.0001267<br>0.0001267<br>0.0001246<br>9.334-05<br>0.0001247<br>9.34-05<br>0.0001032<br>0.0001032<br>0.0001032<br>0.0001032<br>0.0001032<br>0.0001032<br>0.000103<br>8.822E-05<br>6.214E-07<br>0<br>0.0000123<br>0.000123<br>0.000123<br>0.000123<br>0.000123<br>0.000123<br>0.000123<br>0.000123<br>0.000123<br>0.000123<br>0.000123<br>0.000123<br>0.000123<br>0.000123<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.0000125<br>0.0000125<br>0.0000125<br>0    |
2.4904488<br>1.9842855<br>2.2002208<br>2.1073468<br>2.2364653<br>3.1057198<br>3.0468346<br>2.9569573<br>2.8602462<br>2.2620547<br>2.6509615<br>2.27073267<br>2.9117616<br>4.1828878<br>4.9637927<br>4.4562881<br>0.0573728<br>0<br>0<br>1.3610254<br>1.3610254<br>1.3610254<br>0.6059690<br>0.6039622<br>0.5115727<br>0.5025855<br>0.5987613<br>0.60398062<br>0.8324163<br>0.60398262<br>0.59877613<br>0.6025855<br>0.5987613<br>0.6025825<br>0.8324163<br>0.8324163<br>0.8324163<br>0.8324163<br>0.8324163<br>0.8324163<br>0.8324163<br>0.8324163<br>0.8324163<br>0.8324163<br>0.8324163<br>0.8324163<br>0.8324163<br>0.8324163<br>0.8324163<br>0.8324163<br>0.8324163<br>0.8324163<br>0.8324163<br>0.8324163<br>0.8324163<br>0.8324163<br>0.8324163<br>0.8324163<br>0.8324163<br>0.8324163<br>0.8324163<br>0.8324163<br>0.8324163<br>0.8324163<br>0.8324163<br>0.8324163<br>0.8324163<br>0.8324163<br>0.8324163<br>0.8324163<br>0.8324163<br>0.8324163<br>0.8324163<br>0.8324163<br>0.8324163<br>0.8324163<br>0.8324163<br>0.8324163<br>0.8324163<br>0.8324163<br>0.8324163<br>0.8324163<br>0.8324163<br>0.8324163<br>0.8324163<br>0.8324163<br>0.8324163<br>0.8324163<br>0.8324163<br>0.8324163<br>0.8324163<br>0.8324163<br>0.8324163<br>0.8324163<br>0.8324163<br>0.8324163<br>0.8324163<br>0.8324163<br>0.8324163<br>0.8324163<br>0.8324163<br>0.8324163<br>0.8324163<br>0.8324163<br>0.8324163<br>0.8324163<br>0.8324163<br>0.8324163<br>0.8324163<br>0.8324163<br>0.8324163<br>0.8324163<br>0.8324163<br>0.8324163<br>0.8324163<br>0.8324163<br>0.8324163<br>0.8324163<br>0.8324163<br>0.8324163<br>0.8324163<br>0.8324163<br>0.8324163<br>0.8324163<br>0.8324163<br>0.8324163<br>0.8324163<br>0.8324163<br>0.8324163<br>0.8324163<br>0.8324163<br>0.8324163<br>0.8324163<br>0.8324163<br>0.8324163<br>0.8324163<br>0.8324163<br>0.8324163<br>0.8324163<br>0.8324163<br>0.8324163<br>0.8324163<br>0.8324163<br>0.8324163<br>0.8324163<br>0.8324163<br>0.8324163<br>0.8324163<br>0.8324163<br>0.8324163<br>0.8324163<br>0.8324163<br>0.8324163<br>0.8324163<br>0.8324163<br>0.8324163<br>0.8324163<br>0.8324163<br>0.8324163<br>0.8324163<br>0.8324163<br>0.8324163<br>0.8324163<br>0.8324163<br>0.8324163<br>0.8324163<br>0.8324163<br>0.8324163<br>0.8324163<br>0.8324163<br>0.8324163<br>0.8324163<br>0.8324163<br>0.8324163<br>0.8324163<br>0.8324163<br>0.8324163<br>0.8324163<br>0.8324163<br>0.8324163<br>0.8324163<br>0.8324163<br>0.8324163<br>0.8324163<br>0.8344163<br>0.8344163<br>0.8344163<br>0.8344163<br>0.8344163<br>0.8344163<br>0.8344163<br>0.8344163<br>0.8344163<br>0.8344163<br>0.8344163<br>0.8344163<br>0.8344163<br>0.8344163<br>0.8344163<br>0.8344163<br>0.8344163<br>0.8344163<br>0.8344163<br>0.8344163<br>0.8344163<br>0.834416 | 1.863E-06<br>1.272E-06<br>1.41E-06<br>1.351E-06<br>1.351E-06<br>1.32FE-06<br>1.264E-06<br>1.264E-06<br>1.264E-06<br>1.264E-06<br>1.401E-06<br>1.401E-06<br>1.440E-06<br>9.338E-07<br>1.341E-06<br>9.338E-07<br>0<br>0<br>0<br>2.066E-07<br>9.2065E-07<br>1.533E-07<br>2.72FE-07<br>2.72FE-07<br>2.72FE-07<br>2.72FE-07<br>2.528E-07<br>2.528E-07<br>2.528E-07<br>2.528E-07<br>2.528E-07<br>2.528E-07<br>2.528E-07<br>2.528E-07<br>2.528E-07<br>2.528E-07<br>2.528E-07<br>2.528E-07<br>2.528E-07<br>2.528E-07<br>2.528E-07<br>2.528E-07<br>2.528E-07<br>2.528E-07<br>2.528E-07<br>2.528E-07<br>2.528E-07<br>2.528E-07<br>2.528E-07<br>2.528E-07<br>2.528E-07<br>2.528E-07<br>2.528E-07<br>2.528E-07<br>2.528E-07<br>2.528E-07<br>2.528E-07<br>2.528E-07<br>2.528E-07<br>2.528E-07<br>2.528E-07<br>2.528E-07<br>2.528E-07<br>2.528E-07<br>2.528E-07<br>2.528E-07<br>2.528E-07<br>2.528E-07<br>2.528E-07<br>2.528E-07<br>2.528E-07<br>2.528E-07<br>2.528E-07<br>2.528E-07<br>2.528E-07<br>2.528E-07<br>2.528E-07<br>2.528E-07<br>2.528E-07<br>2.528E-07<br>2.528E-07<br>2.528E-07<br>2.528E-07<br>2.528E-07<br>2.528E-07<br>2.528E-07<br>2.528E-07<br>2.528E-07<br>2.528E-07<br>2.528E-07<br>2.528E-07<br>2.528E-07<br>2.528E-07<br>2.528E-07<br>2.528E-07<br>2.528E-07<br>2.528E-07<br>2.528E-07<br>2.528E-07<br>2.528E-07<br>2.528E-07<br>2.528E-07<br>2.528E-07<br>2.528E-07<br>2.528E-07<br>2.528E-07<br>2.528E-07<br>2.528E-07<br>2.528E-07<br>2.528E-07<br>2.528E-07<br>2.528E-07<br>2.528E-07<br>2.528E-07<br>2.528E-07<br>2.528E-07<br>2.528E-07<br>2.528E-07<br>2.528E-07<br>2.528E-07<br>2.528E-07<br>2.528E-07<br>2.528E-07<br>2.528E-07<br>2.528E-07<br>2.528E-07<br>2.528E-07<br>2.528E-07<br>2.528E-07<br>2.528E-07<br>2.528E-07<br>2.528E-07<br>2.528E-07<br>2.528E-07<br>2.528E-07<br>2.528E-07<br>2.528E-07<br>2.528E-07<br>2.528E-07<br>2.528E-07<br>2.528E-07<br>2.528E-07<br>2.528E-07<br>2.528E-07<br>2.528E-07<br>2.528E-07<br>2.528E-07<br>2.528E-07<br>2.528E-07<br>2.528E-07<br>2.528E-07<br>2.528E-07<br>2.528E-07<br>2.528E-07<br>2.528E-07<br>2.528E-07<br>2.528E-07<br>2.528E-07<br>2.528E-07<br>2.528E-07<br>2.528E-07<br>2.528E-07<br>2.528E-07<br>2.528E-07<br>2.528E-07<br>2.528E-07<br>2.528E-07<br>2.528E-07<br>2.528E-07<br>2.528E-07<br>2.528E-07<br>2.528E-07<br>2.528E-07<br>2.528E-07<br>2.528E-07<br>2.528E-07<br>2.528E-07<br>2.528E-07<br>2.528E-07<br>2.528E-07<br>2.528E-07<br>2.528E-07<br>2.528E-07<br>2.528E-07<br>2.528E-07<br>2.528E-07<br>2.528E-07<br>2.528E-07<br>2. | 0.0030261<br>0.0020660<br>0.0022915<br>0.0021953<br>0.0023293<br>0.0023293<br>0.0023293<br>0.0022542<br>0.0022542<br>0.00224824<br>0.00224824<br>0.00224824<br>0.00224824<br>0.00224824<br>0.00224824<br>0.00224824<br>0.00224824<br>0.00224824<br>0.00224824<br>0.00024824<br>0.0003777<br>0.0003777<br>0.0003777<br>0.0003777<br>0.0003777<br>0.000352<br>0.0002616<br>0.0002635<br>0.0002616<br>0.0002635<br>0.0002616<br>0.0004644<br>0.0004614<br>0.0004614<br>0.0004614   | 2.556842<br>1.8769248<br>2.0977776<br>2.0052501<br>2.1283629<br>2.9552749<br>2.9552749<br>2.900207<br>2.8217082<br>2.7301698<br>3.0191874<br>2.5033689<br>2.5607825<br>2.33931649<br>4.688371<br>4.16884130<br>0.0450027<br>0<br>1.2345153<br>1.2303809<br>0.04704<br>0.5112078<br>0.04704<br>0.5506791<br>0.55506791<br>0.55506791<br>0.55506791<br>0.55506791<br>0.55506791<br>0.55506791<br>0.55506791<br>0.55506791<br>0.55506791<br>0.55506791<br>0.55506791<br>0.55506791<br>0.55506791<br>0.55506791<br>0.55506791<br>0.55506791<br>0.55506791<br>0.55506791<br>0.55506791<br>0.55506791<br>0.55506791<br>0.55506791<br>0.55506791<br>0.55506791<br>0.55506791<br>0.55506791<br>0.55506791<br>0.55506791<br>0.55506791<br>0.55506791<br>0.55506791<br>0.55506791<br>0.55506791<br>0.55506791<br>0.55506791<br>0.55506791<br>0.55506791<br>0.55506791<br>0.55506791<br>0.55506791<br>0.5550791<br>0.5550791<br>0.5550791<br>0.5550791<br>0.5550791<br>0.5550791<br>0.5550791<br>0.5550791<br>0.5550791<br>0.5550791<br>0.5550791<br>0.5550791<br>0.5550791<br>0.5550791<br>0.5550791<br>0.5550791<br>0.5550791<br>0.5550791<br>0.5550791<br>0.5550791<br>0.5550791<br>0.5550791<br>0.5550791<br>0.5550791<br>0.5550791<br>0.5550791<br>0.5550791<br>0.5550791<br>0.5550791<br>0.5550791<br>0.5550791<br>0.5550791<br>0.5550791<br>0.5550791<br>0.5550791<br>0.5550791<br>0.5550791<br>0.5550791<br>0.5550791<br>0.5550791<br>0.5550791<br>0.5550791<br>0.5550791<br>0.5550791<br>0.5550791<br>0.5550791<br>0.5550791<br>0.5550791<br>0.5550791<br>0.5550791<br>0.5550791<br>0.5550791<br>0.5550791<br>0.5550791<br>0.5550791<br>0.5550791<br>0.5550791<br>0.5550791<br>0.5550791<br>0.5550791<br>0.5550791<br>0.5550791<br>0.5550791<br>0.5550791<br>0.5550791<br>0.5550791<br>0.5550791<br>0.5550791<br>0.5550791<br>0.5550791<br>0.5550791<br>0.5550791<br>0.5550791<br>0.5550791<br>0.5550791<br>0.5550791<br>0.5550791<br>0.5550791<br>0.5550791<br>0.5550791<br>0.5550791<br>0.5550791<br>0.5550791<br>0.5550791<br>0.5550791<br>0.5550791<br>0.5550791<br>0.5550791<br>0.5550791<br>0.5550791<br>0.5550791<br>0.5550791<br>0.5550791<br>0.5550791<br>0.5550791<br>0.5550791<br>0.5550791<br>0.5550791<br>0.5550791<br>0.5550791<br>0.5550791<br>0.5550791<br>0.5550791<br>0.5550791<br>0.5550791<br>0.5550791<br>0.5550791<br>0.5550791<br>0.5550791<br>0.5550791<br>0.5550791<br>0.5550791<br>0.5550791<br>0.5550791<br>0.5550791<br>0.5550791<br>0.5550791<br>0.5550791<br>0.5550791<br>0.5550791<br>0.5550791<br>0.5550791<br>0.5550791<br>0.5550791<br>0.5550791<br>0.5550791<br>0.5550791<br>0.5550791<br>0.5550791<br>0.5550791<br>0.5550791<br>0.5550791<br>0.5550791<br>0.5550791<br>0.5550791<br>0.5550791<br>0.5550791<br>0.5550790  |
1.7622-06<br>1.2642-06<br>1.3452-06<br>1.2852-06<br>1.2852-06<br>1.2852-06<br>1.2632-06<br>1.2632-06<br>1.2632-06<br>1.2632-06<br>1.3322-06<br>1.3322-06<br>1.3322-06<br>1.3322-06<br>1.3682-07<br>1.262-07<br>2.5162-07<br>2.5452-07<br>2.3492-07<br>2.3492-07<br>2.3492-07<br>2.3492-07<br>2.3492-07<br>2.3492-07<br>2.3492-07<br>2.3492-07<br>2.3492-07<br>2.3492-07<br>2.3492-07<br>2.3492-07<br>2.3492-07<br>2.3492-07<br>2.3492-07<br>2.3492-07<br>2.3492-07<br>2.3492-07<br>2.3492-07<br>2.3492-07<br>2.3492-07<br>2.3492-07<br>2.3492-07<br>2.3492-07<br>2.3492-07<br>2.3492-07<br>2.3492-07<br>2.3492-07<br>2.3492-07<br>2.3492-07<br>2.3492-07<br>2.3492-07<br>2.3492-07<br>2.3492-07<br>2.3492-07<br>2.3492-07<br>2.3492-07<br>2.3492-07<br>2.3492-07<br>2.3492-07<br>2.3492-07<br>2.3492-07<br>2.3492-07<br>2.3492-07<br>2.3492-07<br>2.3492-07<br>2.3492-07<br>2.3492-07<br>2.3492-07<br>2.3492-07<br>2.3492-07<br>2.3492-07<br>2.3492-07<br>2.3492-07<br>2.3492-07<br>2.3492-07<br>2.3492-07<br>2.3492-07<br>2.3492-07<br>2.3492-07<br>2.3492-07<br>2.3492-07<br>2.3492-07<br>2.3492-07<br>2.3492-07<br>2.3492-07<br>2.3492-07<br>2.3492-07<br>2.3492-07<br>2.3492-07<br>2.3492-07<br>2.3492-07<br>2.3492-07<br>2.3492-07<br>2.3492-07<br>2.3492-07<br>2.3492-07<br>2.3492-07<br>2.3492-07<br>2.3492-07<br>2.3492-07<br>2.3492-07<br>2.3492-07<br>2.3492-07<br>2.3492-07<br>2.3492-07<br>2.3492-07<br>2.3492-07<br>2.3492-07<br>2.3492-07<br>2.3492-07<br>2.3492-07<br>2.3492-07<br>2.3492-07<br>2.3492-07<br>2.3492-07<br>2.3492-07<br>2.3492-07<br>2.3492-07<br>2.3492-07<br>2.3492-07<br>2.3492-07<br>2.3492-07<br>2.3492-07<br>2.3492-07<br>2.3492-07<br>2.3492-07<br>2.3492-07<br>2.3492-07<br>2.3492-07<br>2.3492-07<br>2.3492-07<br>2.3492-07<br>2.3492-07<br>2.3492-07<br>2.3492-07<br>2.3492-07<br>2.3492-07<br>2.3492-07<br>2.3492-07<br>2.3492-07<br>2.3492-07<br>2.3492-07<br>2.3492-07<br>2.3492-07<br>2.3492-07<br>2.3492-07<br>2.3492-07<br>2.3492-07<br>2.3492-07<br>2.3492-07<br>2.3492-07<br>2.3492-07<br>2.3492-07<br>2.3492-07<br>2.3492-07<br>2.3492-07<br>2.3492-07<br>2.3492-07<br>2.3492-07<br>2.3492-07<br>2.3492-07<br>2.3492-07<br>2.3492-07<br>2.3492-07<br>2.3492-07<br>2.3492-07<br>2.3492-07<br>2.3492-07<br>2.3492-07<br>2.3492-07<br>2.3492-07<br>2.3492-07<br>2.3492-07<br>2.3492-07<br>2.3492-07<br>2.3492-07<br>2.3492-07<br>2.3492-07<br>2.3492-07<br>2.3492-07<br>2.3492-07<br>2.3492-07<br>2.3492-07<br>2.3492-07<br>2.3492-07<br>2.3492-07<br>2.3492-07<br>0                                     | 0.0028636<br>0.0021569<br>0.0021848<br>0.0022167<br>0.0020815<br>0.0025171<br>0.0025171<br>0.001552<br>0.0025471<br>0.0021501<br>0.0021501<br>0.0021501<br>0.0021501<br>0.0021501<br>0.002150<br>0.002450<br>0.002450<br>0.000411<br>1.2556-05<br>0.00004514<br>1.305E-05<br>0.00004588<br>0.00004588<br>0.00004588<br>0.00004588<br>0.00004588<br>0.00004588<br>0.00004588<br>0.00004588<br>0.00004588<br>0.00004588<br>0.00004588<br>0.00004588<br>0.00004588<br>0.00004588<br>0.00004588<br>0.00004588<br>0.00004588<br>0.00004588<br>0.00004588<br>0.00004588<br>0.00004588<br>0.00004588<br>0.00004588<br>0.00004588<br>0.00004588<br>0.00004588<br>0.00004588<br>0.00004588<br>0.00004588<br>0.00004588<br>0.00004588<br>0.00004588<br>0.00004588<br>0.00004588<br>0.00004588<br>0.00004588<br>0.00004588<br>0.00004588<br>0.00004588<br>0.00004588<br>0.00004588<br>0.00004588<br>0.00004588<br>0.00004588<br>0.00004588<br>0.00004588<br>0.00004588<br>0.00004588<br>0.00004588<br>0.00004588<br>0.00004588<br>0.00004588<br>0.00004588<br>0.00004588<br>0.00004588<br>0.00004588<br>0.00004588<br>0.00004588<br>0.00004588<br>0.00004588<br>0.00004588<br>0.00004588<br>0.00004588<br>0.00004588<br>0.00004588<br>0.00004588<br>0.00004588<br>0.00004588<br>0.00004588<br>0.00004588<br>0.00004588<br>0.00004588<br>0.00004588<br>0.00004588<br>0.00004588<br>0.00004588<br>0.00004588<br>0.00004588<br>0.00004588<br>0.00004588<br>0.00004588<br>0.00004588<br>0.00004588<br>0.00004588<br>0.00004588<br>0.00004588<br>0.00004588<br>0.00004588<br>0.00004588<br>0.00004588<br>0.00004588<br>0.00004588<br>0.00004588<br>0.00004588<br>0.00004588<br>0.00004588<br>0.00004588<br>0.00004588<br>0.00004588<br>0.00004588<br>0.00004588<br>0.00004588<br>0.00004588<br>0.00004588<br>0.00004588<br>0.00004588<br>0.00004588<br>0.00004588<br>0.00004588<br>0.00004588<br>0.00004588<br>0.00004588<br>0.00004588<br>0.00004588<br>0.00004588<br>0.00004588<br>0.00004588<br>0.00004588<br>0.00004588<br>0.00004588<br>0.00004588<br>0.00004588<br>0.00004588<br>0.00004588<br>0.00004588<br>0.00004588<br>0.00004588<br>0.00004588<br>0.00004588<br>0.00004588<br>0.00004588<br>0.00004588<br>0.00004588<br>0.00004588<br>0.00004588<br>0.00004588<br>0.00004588<br>0.00004588<br>0.00004588<br>0.00004588<br>0.00004588<br>0.00004588<br>0.00004588<br>0.00004588<br>0.00004588<br>0.00004588<br>0.00004588<br>0.00004588<br>0.00004588<br>0.00004588<br>0.00004588<br>0.00004588<br>0.00004588<br>0.00004588<br>0.00004588<br>0.00004588<br>0.00004588<br>0.00004588<br>0.00004588<br>0.00004588<br>0.00004588<br>0.00004588<br>0.00004588<br>0.00004588<br>0.00004588<br>0.00004588<br>0.00004588<br>0.0   | 2.2440968<br>1.7795138<br>1.9915216<br>1.8961342<br>2.720732<br>2.720732<br>2.6462899<br>2.5522851<br>2.8364013<br>2.325289<br>2.3726056<br>2.3321589<br>2.3726056<br>2.4375101<br>2.6416999<br>3.7301495<br>4.4757836<br>0<br>0<br>1.1398552<br>1.14138222<br>0.041725<br>0.4807676<br>0.4240418<br>0.5013024<br>0.50513024<br>0.50513024<br>0.50513024<br>0.5051371<br>0.66950701   |
1.6782-06<br>1.21412-06<br>1.2175-06<br>1.2152-06<br>1.21912-06<br>1.3112-06<br>1.3112-06<br>1.3112-06<br>1.3125-06<br>1.3125-06<br>1.3022-06<br>1.3022-06<br>1.3022-06<br>1.3022-06<br>1.3022-06<br>0<br>0<br>1.7312-07<br>1.3582-07<br>1.5152-07<br>1.5152-07<br>1.5282-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2882-07<br>2.2892-07<br>2.2892-07<br>2.2892-07<br>2.2892-07<br>2.2892-07<br>2.2  | 0.0027267<br>0.0018534<br>0.0020742<br>0.0029742<br>0.0029742<br>0.002937<br>0.002937<br>0.002937<br>0.002937<br>0.002937<br>0.002937<br>0.002937<br>0.002937<br>0.002937<br>0.002937<br>0.002937<br>0.002937<br>0.002941<br>0.002941<br>0.002941<br>0.002941<br>0.002941<br>0.002941<br>0.002941<br>0.002941<br>0.002941<br>0.002941<br>0.002941<br>0.002941<br>0.002941<br>0.002941<br>0.002941<br>0.002941<br>0.002941<br>0.0002163<br>0.0002163<br>0.0002163<br>0.0002163<br>0.0000175<br>0.0000417<br>0.0000417<br>0.0000417<br>0.000055<br>0.000055<br>0.000055<br>0.000055<br>0.000055<br>0.000055<br>0.000055<br>0.000055<br>0.000055<br>0.000055<br>0.000055<br>0.000055<br>0.000055<br>0.000055<br>0.000055<br>0.000055<br>0.000055<br>0.000055<br>0.000055<br>0.000055<br>0.000055<br>0.000055<br>0.000055<br>0.000055<br>0.000055<br>0.000055<br>0.000055<br>0.000055<br>0.000055<br>0.000055<br>0.000055<br>0.000055<br>0.000055<br>0.000055<br>0.000055<br>0.000055<br>0.000055<br>0.000055<br>0.000055<br>0.000055<br>0.000055<br>0.00005<br>0.000055<br>0.000055<br>0.000055<br>0.000055<br>0.000055<br>0.000055<br>0.00005<br>0.00005<br>0.00055<br>0.00055<br>0.00055<br>0.00055<br>0.00055<br>0.00055<br>0.00055<br>0.00055<br>0.00055<br>0.00055<br>0.00055<br>0.00055<br>0.00055<br>0.00055<br>0.00055<br>0.00055<br>0.00055<br>0.00055<br>0.00055<br>0.00055<br>0.00055<br>0.00055<br>0.00055<br>0.00055<br>0.00055<br>0.00055<br>0.00055<br>0.00055<br>0.00055<br>0.00055<br>0.00055<br>0.00055<br>0.00055<br>0.00055<br>0.00055<br>0.00055<br>0.00055<br>0.00055<br>0.00055<br>0.00055<br>0.00055<br>0.00055<br>0.00055<br>0.00055<br>0.00055<br>0.00055<br>0.00055<br>0.00055<br>0.00055<br>0.00055<br>0.00055<br>0.00055<br>0.00055<br>0.00055<br>0.00055<br>0.00055<br>0.00055<br>0.00055<br>0.00055<br>0.00055<br>0.00055<br>0.00055<br>0.00055<br>0.00055<br>0.00055<br>0.00055<br>0.00055<br>0.00055<br>0.00055<br>0.00055<br>0.00055<br>0.00055<br>0.00055<br>0.00055<br>0.00055<br>0.00055<br>0.00055<br>0.00055<br>0.00055<br>0.00055<br>0.00055<br>0.00055<br>0.00055<br>0.00055<br>0.00055<br>0.00055<br>0.00055<br>0.00055<br>0.00055<br>0.00055<br>0.00055<br>0.00055<br>0.00055<br>0.00055<br>0.00055<br>0.00055<br>0.00055<br>0.00055<br>0.00055<br>0.00055<br>0.00055<br>0.00055<br>0.00055<br>0.00055<br>0.00055<br>0.00055<br>0.00055<br>0.00055<br>0.00055<br>0.00055<br>0.00055<br>0.00055<br>0.00055<br>0.00055<br>0.00055<br>0.00055<br>0.00055<br>0.00055<br>0.00055<br>0.00055<br>0.00055<br>0.00055<br>0.00055<br>0.00055<br>0.00055<br>0.00055<br>0.00055<br>0.00055<br>0.00055<br>0.00055<br>0.00055<br>0.00055<br>0.00055<br>0.00055<br>0.00055<br>0.00055<br>0.00055<br>0.00055<br>0.00055<br>0.00055<br>0.00055<br>0.00055<br>0.00055<br>0.00055<br>0.00055<br>0.00055<br>0.00055<br>0.00055<br>0.00055<br>0.00055<br>0.00055 | 2.623364<br>2.623364<br>2.3450273<br>2.439219<br>2.3971946<br>3.2614464<br>3.105125<br>3.1051925<br>3.0057113<br>3.3257455<br>2.7816703<br>3.3257455<br>2.7816703<br>3.3257455<br>2.7816703<br>3.3257455<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0   |
19622-06<br>1.354E-06<br>1.438E-06<br>1.438E-06<br>1.337E-06<br>1.337E-06<br>1.327E-06<br>1.327E-06<br>1.327E-06<br>1.436E-06<br>1.517E-06<br>1.436E-06<br>1.438E-06<br>9.766E-07<br>1.398E-06<br>0<br>0<br>2.87E-07<br>2.87F-07<br>2.87F-07<br>2.517E-07<br>2.517E-07<br>2.517E-07<br>2.517E-07<br>2.517E-07<br>2.517E-07<br>2.517E-07<br>2.517E-07<br>2.517E-07<br>2.517E-07<br>2.517E-07<br>2.517E-07<br>2.517E-07<br>2.517E-07<br>2.517E-07<br>2.517E-07<br>2.517E-07<br>2.517E-07<br>2.517E-07<br>2.517E-07<br>2.517E-07<br>2.517E-07<br>2.517E-07<br>2.517E-07<br>2.517E-07<br>2.517E-07<br>2.517E-07<br>2.517E-07<br>2.517E-07<br>2.517E-07<br>2.517E-07<br>2.517E-07<br>2.517E-07<br>2.517E-07<br>2.517E-07<br>2.517E-07<br>2.517E-07<br>2.517E-07<br>2.517E-07<br>2.517E-07<br>2.517E-07<br>2.517E-07<br>2.517E-07<br>2.517E-07<br>2.517E-07<br>2.517E-07<br>2.517E-07<br>2.517E-07<br>2.517E-07<br>2.517E-07<br>2.517E-07<br>2.517E-07<br>2.517E-07<br>2.517E-07<br>2.517E-07<br>2.517E-07<br>2.517E-07<br>2.517E-07<br>2.517E-07<br>2.517E-07<br>2.517E-07<br>2.517E-07<br>2.517E-07<br>2.517E-07<br>2.517E-07<br>2.517E-07<br>2.517E-07<br>2.517E-07<br>2.517E-07<br>2.517E-07<br>2.517E-07<br>2.517E-07<br>2.517E-07<br>2.517E-07<br>2.517E-07<br>2.517E-07<br>2.517E-07<br>2.517E-07<br>2.517E-07<br>2.517E-07<br>2.517E-07<br>2.517E-07<br>2.517E-07<br>2.517E-07<br>2.517E-07<br>2.517E-07<br>2.517E-07<br>2.517E-07<br>2.517E-07<br>2.517E-07<br>2.517E-07<br>2.517E-07<br>2.517E-07<br>2.517E-07<br>2.517E-07<br>2.517E-07<br>2.517E-07<br>2.517E-07<br>2.517E-07<br>2.517E-07<br>2.517E-07<br>2.517E-07<br>2.517E-07<br>2.517E-07<br>2.517E-07<br>2.517E-07<br>2.517E-07<br>2.517E-07<br>2.517E-07<br>2.517E-07<br>2.517E-07<br>2.517E-07<br>2.517E-07<br>2.517E-07<br>2.517E-07<br>2.517E-07<br>2.517E-07<br>2.517E-07<br>2.517E-07<br>2.517E-07<br>2.517E-07<br>2.517E-07<br>2.517E-07<br>2.517E-07<br>2.517E-07<br>2.517E-07<br>2.517E-07<br>2.517E-07<br>2.517E-07<br>2.517E-07<br>2.517E-07<br>2.517E-07<br>2.517E-07<br>2.517E-07<br>2.517E-07<br>2.517E-07<br>2.517E-07<br>2.517E-07<br>2.517E-07<br>2.517E-07<br>2.517E-07<br>2.517E-07<br>2.517E-07<br>2.517E-07<br>2.517E-07<br>2.517E-07<br>2.517E-07<br>2.517E-07<br>2.517E-07<br>2.517E-07<br>2.517E-07<br>2.517E-07<br>2.517E-07<br>2.517E-07<br>2.517E-07<br>2.517E-07<br>2.517E-07<br>2.517E-07<br>2.517E-07<br>2.517E-07<br>2.517E-07<br>2.517E-07<br>2.517E-07<br>2.517E-07<br>2.517E-07<br>2.517E-07<br>2.517E-07<br>2.517E-07     | 0.0031876<br>0.0022001<br>0.0024433<br>0.002347<br>0.002347<br>0.0022645<br>0.0027711<br>0.002156<br>0.0025015<br>0.002885<br>0.0023913<br>0.002464<br>0.0015867<br>0.002885<br>0.002454<br>0.002716<br>0.002705<br>0<br>0.0005245<br>0.00005245<br>0.00005245<br>0.00005245<br>0.00005245<br>0.00005245<br>0.00005245<br>0.00005245<br>0.00005245<br>0.00005245<br>0.00005245<br>0.00005245<br>0.00005245<br>0.00005245<br>0.0000557<br>0.0000557<br>0.0000557<br>0.0000557<br>0.0000557<br>0.0000557<br>0.0000557<br>0.0000557<br>0.0000557<br>0.0000555<br>0.0000555<br>0.0000555<br>0.0000555<br>0.0000555<br>0.0000555<br>0.0000555<br>0.0000555<br>0.0000555<br>0.0000555<br>0.0000555<br>0.0000555<br>0.0000555<br>0.0000555<br>0.0000555<br>0.0000555<br>0.0000555<br>0.0000555<br>0.0000555<br>0.0000555<br>0.0000555<br>0.0000555<br>0.0000555<br>0.0000555<br>0.0000555<br>0.0000555<br>0.0000555<br>0.0000555<br>0.0000555<br>0.0000555<br>0.0000555<br>0.0000555<br>0.0000555<br>0.0000555<br>0.0000555<br>0.0000555<br>0.0000555<br>0.0000555<br>0.0000555<br>0.0000555<br>0.0000555<br>0.0000555<br>0.0000555<br>0.0000555<br>0.0000555<br>0.0000555<br>0.0000555<br>0.0000555<br>0.000555<br>0.0000555<br>0.0000555<br>0.0000555<br>0.0000555<br>0.0000555<br>0.000555<br>0.000555<br>0.000555<br>0.000555<br>0.000555<br>0.000555<br>0.000555<br>0.000555<br>0.000555<br>0.000555<br>0.000555<br>0.000555<br>0.000555<br>0.000555<br>0.000555<br>0.000555<br>0.000555<br>0.000555<br>0.0005555<br>0.0005555<br>0.0005555<br>0.0005555<br>0.0005555<br>0.0005555<br>0.0005555<br>0.0005555<br>0.0005555<br>0.0005555<br>0.0005555<br>0.0005555<br>0.0005555<br>0.0005555<br>0.0005555<br>0.0005555<br>0.0005555<br>0.0005555<br>0.0005555<br>0.0005555<br>0.0005555<br>0.0005555<br>0.0005555<br>0.0005555<br>0.0005555<br>0.0005555<br>0.0005555<br>0.0005555<br>0.0005555<br>0.0005555<br>0.0005555<br>0.0005555<br>0.0005555<br>0.0005555<br>0.0005555<br>0.0005555<br>0.0005555<br>0.0005555<br>0.0005555<br>0.0005555<br>0.0005555<br>0.0005555<br>0.0005555<br>0.0005555<br>0.0005555<br>0.00055555<br>0.00055555<br>0.00055555<br>0.00055555<br>0.00055555<br>0.00055555<br>0.00055555<br>0.00055555<br>0.00055555<br>0.00055555<br>0.00055555<br>0.000555555<br>0.000555555<br>0.000555555<br>0.000555555<br>0.00055555555   | 2.3644564<br>1.8937445<br>2.1225909<br>2.0177733<br>2.1529112<br>2.928047<br>2.8544498<br>2.6943024<br>2.6943024<br>2.6943024<br>2.594363<br>2.724646<br>3.8826302<br>4.6916603<br>4.1449386<br>0.043318<br>0<br>0<br>0<br>1.5831454<br>1.583267<br>0.043982567<br>0.043982567<br>0.043982567<br>0.043982567<br>0.043982567<br>0.043982567<br>0.043982567<br>0.043982567<br>0.043982567<br>0.043982567<br>0.043982567<br>0.043982567<br>0.043982567<br>0.043982567<br>0.043982567<br>0.043982567<br>0.043982567<br>0.043982567<br>0.043982567<br>0.043982567<br>0.043982567<br>0.043982567<br>0.043982567<br>0.043982567<br>0.043982567<br>0.043982567<br>0.043982567<br>0.043982567<br>0.043982567<br>0.043982567<br>0.043982567<br>0.043982567<br>0.043982567<br>0.043982567<br>0.043982567<br>0.043982567<br>0.043982567<br>0.043982567<br>0.043982567<br>0.043982567<br>0.043982567<br>0.043982567<br>0.043982567<br>0.043982567<br>0.043982567<br>0.043982567<br>0.043982567<br>0.043982567<br>0.043982567<br>0.043982567<br>0.043982567<br>0.043982567<br>0.043982567<br>0.043982567<br>0.043982567<br>0.043982567<br>0.043982567<br>0.043982567<br>0.043982567<br>0.043982567<br>0.043982567<br>0.043982567<br>0.043982567<br>0.043982567<br>0.043982567<br>0.043982567<br>0.043982567<br>0.043982567<br>0.043982567<br>0.043982567<br>0.043982567<br>0.043982567<br>0.043982567<br>0.043982567<br>0.043982567<br>0.043982567<br>0.043982567<br>0.043982567<br>0.043982567<br>0.043982567<br>0.043982567<br>0.043982567<br>0.043982567<br>0.043982567<br>0.043982567<br>0.043982567<br>0.043982567<br>0.043982567<br>0.043982567<br>0.043982567<br>0.043982567<br>0.043982567<br>0.043982567<br>0.043982567<br>0.043982567<br>0.043982567<br>0.043982567<br>0.043982567<br>0.043982567<br>0.043982567<br>0.043982567<br>0.043982567<br>0.043982567<br>0.043982567<br>0.043982567<br>0.043982567<br>0.043982567<br>0.043982567<br>0.043982567<br>0.043982567<br>0.043982567<br>0.043982567<br>0.043982567<br>0.043982567<br>0.043982567<br>0.043982567<br>0.043982567<br>0.043982567<br>0.043982567<br>0.043982567<br>0.043982567<br>0.043982567<br>0.043982567<br>0.043982567<br>0.043982567<br>0.043982567<br>0.043982567<br>0.043982567<br>0.043982567<br>0.043982567<br>0.043982567<br>0.043982567<br>0.043982567<br>0.043982567<br>0.043982567<br>0.043982567<br>0.043982567<br>0.043982567<br>0.043982567<br>0.043982567<br>0.043982567<br>0.043982567<br>0.043982567<br>0.043982567<br>0.043982567<br>0.043982567<br>0.043982567<br>0.043982567<br>0.043982567<br>0.043982567<br>0.0439857567<br>0.043985567<br>0.043985567<br>0.04398575785757757575757575   | 1.7681-00 0.002<br>1.2446-00 0.002<br>1.3516-00 0.002<br>1.3516-00 0.002<br>1.3516-00 0.002<br>1.5256-00 0.002<br>1.5256-00 0.002<br>1.5256-00 0.002<br>1.5256-00 0.002<br>1.5256-00 0.002<br>1.316-00 0.002<br>1.316-00 0.002<br>1.316-00 0.002<br>1.316-00 0.002<br>0 0 0 0<br>0 0<br>0 0 0<br>0 0<br>0 0 0<br>0 0 0<br>0 0 0<br>0 0<br>0 0<br>0 0<br>0 0 0<br>0 0 | 773         0.04           773         0.03           0.010         0.04           0.015         0.04           0.015         0.04           0.015         0.04           0.015         0.04           0.015         0.04           0.015         0.04           0.015         0.04           0.016         0.04           0.017         0.05           0.018         0.01           0.019         0.01           0.010         0.01           0.010         0.01           0.010         0.01           0.010         0.01           0.010         0.01           0.010         0.01           0.010         0.01           0.010         0.01           0.010         0.01           0.010         0.01           0.010         0.01           0.010         0.01           0.010         0.01           0.010         0.01           0.010         0.01           0.010         0.01           0.010         0.01           0.010         0.01  
   | 147043         3.           147043         3.           1483832         2.           408499         2.           408499         2.           403125         2.           338017         2.           338017         2.           338017         2.           338016         2.           140162         2.           155511         3.           155707         2.           167677         2.           186276         2.           195381         1.           167677         2.           81928         2.           153993         8.           0         0           104252         6.4           19395         5.           10640         9.           110642         9.           110642         9.           110642         9.           110642         9.           76717         8.           76717         8.           76717         8.   
  | 5.18E-08         2           6.619E-08         2           6.619E-08         4           6.776E-08         4           9.996-08         3           8.74E-08         3           8.74E-08         3           6.512-08         4           0.21E-08         4          
0.21E-08         4           4.42E-08         3           5.936-08         4           5.936-08         4           7.77E-08         4           6.26E-09         1           4.56E-09         1           3.34E-09         1           5.54E-09         1           5.54E-09         1           5.54E-09         1           5.54E-09         1           6.34E-08         1           6.34E-08         1           6.34E-08         1           6.34E-08         1           6.34E-09         1                   | 3.998E-05<br>4.254E-05<br>4.224E-05<br>4.224E-05<br>4.224E-05<br>4.511E-05<br>3.736E-05<br>4.377E-05<br>4.307E-05<br>4.307E-05<br>4.307E-05<br>4.307E-05<br>4.307E-05<br>4.307E-05<br>4.307E-05<br>4.307E-05<br>4.501E-05<br>4.22E-05<br>0<br>0<br>1.18E-05<br>1.18E-05<br>1.18E-05<br>1.18E-05<br>1.158E-05<br>1.254E-05<br>1.254E-05<br>1.254E-05<br>1.254E-05<br>1.254E-05<br>1.254E-05<br>1.254E-05<br>1.254E-05<br>1.254E-05<br>1.254E-05<br>1.254E-05<br>1.254E-05<br>1.254E-05<br>1.254E-05<br>1.254E-05<br>1.254E-05<br>1.254E-05<br>1.254E-05<br>1.254E-05<br>1.254E-05<br>1.254E-05<br>1.254E-05<br>1.254E-05<br>1.254E-05<br>1.254E-05<br>1.254E-05<br>1.254E-05<br>1.254E-05<br>1.254E-05<br>1.254E-05<br>1.254E-05<br>1.254E-05<br>1.254E-05<br>1.254E-05<br>1.254E-05<br>1.254E-05<br>1.254E-05<br>1.254E-05<br>1.254E-05<br>1.254E-05<br>1.254E-05<br>1.254E-05<br>1.254E-05<br>1.254E-05<br>1.254E-05<br>1.254E-05<br>1.254E-05<br>1.254E-05<br>1.254E-05<br>1.254E-05<br>1.254E-05<br>1.254E-05<br>1.254E-05<br>1.254E-05<br>1.254E-05<br>1.254E-05<br>1.254E-05<br>1.254E-05<br>1.254E-05<br>1.254E-05<br>1.254E-05<br>1.254E-05<br>1.254E-05<br>1.254E-05<br>1.254E-05<br>1.254E-05<br>1.254E-05<br>1.254E-05<br>1.254E-05<br>1.254E-05<br>1.254E-05<br>1.254E-05<br>1.254E-05<br>1.254E-05<br>1.254E-05<br>1.254E-05<br>1.254E-05<br>1.254E-05<br>1.254E-05<br>1.2554E-05<br>1.254E-05<br>1.254E-05<br>1.254E-05<br>1.254E-05<br>1.254E-05<br>1.254E-05<br>1.254E-05<br>1.254E-05<br>1.254E-05<br>1.254E-05<br>1.254E-05<br>1.254E-05<br>1.254E-05<br>1.254E-05<br>1.254E-05<br>1.254E-05<br>1.254E-05<br>1.254E-05<br>1.254E-05<br>1.254E-05<br>1.254E-05<br>1.254E-05<br>1.254E-05<br>1.254E-05<br>1.254E-05<br>1.254E-05<br>1.254E-05<br>1.254E-05<br>1.254E-05<br>1.254E-05<br>1.254E-05<br>1.254E-05<br>1.254E-05<br>1.254E-05<br>1.254E-05<br>1.254E-05<br>1.254E-05<br>1.254E-05<br>1.254E-05<br>1.254E-05<br>1.254E-05<br>1.254E-05<br>1.254E-05<br>1.254E-05<br>1.254E-05<br>1.254E-05<br>1.254E-05<br>1.254E-05<br>1.254E-05<br>1.254E-05<br>1.254E-05<br>1.254E-05<br>1.254E-05<br>1.254E-05<br>1.254E-05<br>1.254E-05<br>1.254E-05<br>1.254E-05<br>1.254E-05<br>1.254E-05<br>1.254E-05<br>1.254E-05<br>1.254E-05<br>1.254E-05<br>1.254E-05<br>1.254E-05<br>1.254E-05<br>1.254E-05<br>1.254E-05<br>1.254E-05<br>1.254E-05<br>1.254E-05<br>1.254E-05<br>1.254E-05<br>1.254E-05<br>1.254E-05<br>1.254E-05<br>1.254E-05<br>1.254E-05<br>1.254E-05<br>1.254E-05<br>1.254E-05<br>1.254E-05<br>1.254E-05<br>1.254E-05<br>1.254E-05<br>1.254E-05<br>1.254E-05<br>1.254E-05<br>1.254E-05<br>1.254E-05<br>1.254E-05<br>1.254E-05<br>1.254E-05<br>1.254E-05<br>1.254E-054E-054E-054E-054E-054E-054E-054E-0  | 0.0443322<br>0.044017<br>0.0443322<br>0.044017<br>0.054333<br>0.05584333<br>0.0558211<br>0.0538373<br>0.0558211<br>0.0538731<br>0.0538741<br>0.0538741<br>0.05527943<br>0.05527943<br>0.05527943<br>0.05527943<br>0.05527943<br>0.05527943<br>0.05527943<br>0.05527943<br>0.05527944<br>0.05527944<br>0.0938934<br>0.0938934<br>0.0938934<br>0.04622<br>0.04622<br>0.04622<br>0.04622<br>0.04622<br>0.04622<br>0.04622<br>0.04622<br>0.04622<br>0.04622<br>0.04622<br>0.04622<br>0.04622<br>0.04622<br>0.04622<br>0.04622<br>0.04622<br>0.04622<br>0.04622<br>0.04622<br>0.04622<br>0.04622<br>0.04622<br>0.04622<br>0.04622<br>0.04622<br>0.04622<br>0.04622<br>0.04622<br>0.04622<br>0.04622<br>0.04622<br>0.04622<br>0.04622<br>0.04622<br>0.04622<br>0.04622<br>0.04622<br>0.04622<br>0.04622<br>0.04622<br>0.04622<br>0.04622<br>0.04622<br>0.04622<br>0.04622<br>0.04622<br>0.04622<br>0.04622<br>0.04622<br>0.04622<br>0.04622<br>0.04622<br>0.04622<br>0.04622<br>0.04622<br>0.04622<br>0.04622<br>0.04622<br>0.04622<br>0.04622<br>0.04622<br>0.04622<br>0.04622<br>0.04622<br>0.04622<br>0.04622<br>0.04622<br>0.04622<br>0.04622<br>0.04622<br>0.04622<br>0.04622<br>0.04622<br>0.04622<br>0.04622<br>0.04622<br>0.04622<br>0.04622<br>0.04622<br>0.04622<br>0.04622<br>0.04622<br>0.04622<br>0.04622<br>0.04622<br>0.04622<br>0.04622<br>0.04622<br>0.04622<br>0.04622<br>0.04622<br>0.04622<br>0.04622<br>0.04622<br>0.04622<br>0.04622<br>0.04622<br>0.04622<br>0.04622<br>0.04622<br>0.04622<br>0.04622<br>0.04622<br>0.04622<br>0.04622<br>0.04622<br>0.04622<br>0.04622<br>0.04622<br>0.04622<br>0.04622<br>0.04622<br>0.04622<br>0.04622<br>0.04622<br>0.04622<br>0.04622<br>0.04622<br>0.04622<br>0.04622<br>0.04622<br>0.04622<br>0.04622<br>0.04622<br>0.04622<br>0.04622<br>0.04622<br>0.04622<br>0.04622<br>0.04622<br>0.02636<br>0.04622<br>0.02636<br>0.02636<br>0.02636<br>0.02636<br>0.04622<br>0.02636<br>0.02636<br>0.02636<br>0.02636<br>0.04622<br>0.02636<br>0.04622<br>0.02636<br>0.04622<br>0.02636<br>0.04622<br>0.02636<br>0.02636<br>0.04622<br>0.02636<br>0.02636<br>0.02636<br>0.04622<br>0.02786<br>0.02786<br>0.02786<br>0.04622<br>0.02786<br>0.02786<br>0.02786<br>0.02786<br>0.02786<br>0.02786<br>0.02786<br>0.02786<br>0.02786<br>0.02786<br>0.02786<br>0.02786<br>0.02786<br>0.02786<br>0.02786<br>0.02786<br>0.02786<br>0.02786<br>0.02786<br>0.02786<br>0.02786<br>0.02786<br>0.02786<br>0.02786<br>0.02786<br>0.02786<br>0.02786<br>0.02786<br>0.02786<br>0.02786<br>0.02786<br>0.02786<br>0.02786<br>0.02786<br>0.02786<br>0.02786<br>0.02786<br>0.02786<br>0.02786<br>0.02786<br>0.02786<br>0.02786<br>0.02786<br>0.02786<br>0.02786<br>0.02786<br>0.02786<br>0.02786<br>0.02786<br>0.02786<br>0.02786<br>0.02786<br>0.02786<br>0.02786<br>0.02786<br>0.02786<br>0.02786<br>0.02786<br>0.02786<br>0.02786<br>0.02786<br>0.02786<br>0.02786<br>0.00   | 2.8422-08<br>2.8222-08<br>2.8222-08<br>2.8222-08<br>2.8222-08<br>2.4972-08<br>2.4972-08<br>2.3865-08<br>2.3365-08<br>2.3385-08<br>2.3385-08<br>2.713E-08<br>2.222-08<br>2.222-08<br>2.713E-08<br>2.222-08<br>2.713E-08<br>2.222-08<br>2.713E-08<br>2.222-08<br>2.713E-08<br>2.713E-08<br>2.713E-08<br>2.775E-10<br>0<br>7.014E-09<br>2.704E-09<br>2.704E-09<br>2.704E-09<br>2.704E-09<br>2.704E-09<br>2.704E-09<br>2.704E-09<br>2.704E-09<br>2.704E-09<br>2.704E-09<br>2.704E-09<br>2.704E-09<br>2.704E-09<br>2.704E-09<br>2.704E-09<br>2.704E-09<br>2.704E-09<br>2.704E-09<br>2.704E-09<br>2.704E-09<br>2.704E-09<br>2.704E-09<br>2.704E-09<br>2.704E-09<br>2.704E-09<br>2.704E-09<br>2.704E-09<br>2.104E-08<br>2.104E-08<br>2.104E-08<br>2.104E-08<br>2.104E-08<br>2.104E-08<br>2.104E-08<br>2.104E-08<br>2.104E-08<br>2.104E-08<br>2.104E-08<br>2.104E-08<br>2.104E-08<br>2.104E-08<br>2.104E-08<br>2.104E-08<br>2.104E-08<br>2.104E-08<br>2.104E-08<br>2.104E-08<br>2.104E-08<br>2.104E-08<br>2.104E-08<br>2.104E-08<br>2.104E-08<br>2.104E-08<br>2.104E-08<br>2.104E-08<br>2.104E-08<br>2.104E-08<br>2.104E-08<br>2.104E-08<br>2.104E-08<br>2.104E-08<br>2.104E-08<br>2.104E-08<br>2.104E-08<br>2.104E-08<br>2.104E-08<br>2.104E-08<br>2.104E-08<br>2.104E-08<br>2.104E-08<br>2.104E-08<br>2.104E-08<br>2.104E-08<br>2.104E-08<br>2.104E-08<br>2.104E-08<br>2.104E-08<br>2.104E-08<br>2.104E-08<br>2.104E-08<br>2.104E-08<br>2.104E-08<br>2.104E-08<br>2.104E-08<br>2.104E-08<br>2.104E-08<br>2.104E-08<br>2.104E-08<br>2.104E-08<br>2.104E-08<br>2.104E-08<br>2.104E-08<br>2.104E-08<br>2.104E-08<br>2.104E-08<br>2.104E-08<br>2.104E-08<br>2.104E-08<br>2.104E-08<br>2.104E-08<br>2.104E-08<br>2.104E-08<br>2.104E-08<br>2.104E-08<br>2.104E-08<br>2.104E-08<br>2.104E-08<br>2.104E-08<br>2.104E-08<br>2.104E-08<br>2.104E-08<br>2.104E-08<br>2.104E-08<br>2.104E-08<br>2.104E-08<br>2.104E-08<br>2.104E-08<br>2.104E-08<br>2.104E-08<br>2.104E-08<br>2.104E-08<br>2.104E-08<br>2.104E-08<br>2.104E-08<br>2.104E-08<br>2.104E-08<br>2.104E-08<br>2.104E-08<br>2.104E-08<br>2.104E-08<br>2.104E-08<br>2.104E-08<br>2.104E-08<br>2.104E-08<br>2.104E-08<br>2.104E-08<br>2.104E-08<br>2.104E-08<br>2.104E-08<br>2.104E-08<br>2.104E-08<br>2.104E-08<br>2.104E-08<br>2.104E-08<br>2.104E-08<br>2.104E-08<br>2.104E-08<br>2.104E-08<br>2.104E-08<br>2.104E-08<br>2.104E-08<br>2.104E-08<br>2.104E-08<br>2.104E-08<br>2.104E-08<br>2.104E-08<br>2.104E-08<br>2.104E-08<br>2.104E-08<br>2.104E-08<br>2.104E-08<br>2.   |
1.584E-05<br>1.897E-05<br>1.057E-05<br>1.072E-05<br>1.876E-05<br>1.877E-05<br>1.331E-05<br>1.331E-05<br>1.331E-05<br>1.331E-05<br>1.331E-05<br>1.332E-05<br>1.638E-05<br>1.638E-05<br>1.282E-05<br>1.282E-05<br>1.282E-05<br>1.282E-05<br>1.255E-05<br>1.201E-05<br>1.901E-05<br>1.901E-05<br>1.901E-05<br>1.901E-05<br>1.901E-05<br>1.901E-05<br>1.901E-05<br>1.901E-05<br>1.901E-05<br>1.901E-05<br>1.901E-05<br>1.901E-05<br>1.901E-05<br>1.901E-05<br>1.901E-05<br>1.901E-05<br>1.901E-05<br>1.901E-05<br>1.901E-05<br>1.901E-05<br>1.901E-05<br>1.901E-05<br>1.901E-05<br>1.901E-05<br>1.901E-05<br>1.901E-05<br>1.901E-05<br>1.901E-05<br>1.901E-05<br>1.901E-05<br>1.901E-05<br>1.901E-05<br>1.901E-05<br>1.901E-05<br>1.901E-05<br>1.901E-05<br>1.901E-05<br>1.901E-05<br>1.901E-05<br>1.901E-05<br>1.901E-05<br>1.901E-05<br>1.901E-05<br>1.901E-05<br>1.901E-05<br>1.901E-05<br>1.901E-05<br>1.901E-05<br>1.901E-05<br>1.901E-05<br>1.901E-05<br>1.901E-05<br>1.901E-05<br>1.901E-05<br>1.901E-05<br>1.901E-05<br>1.901E-05<br>1.901E-05<br>1.901E-05<br>1.901E-05<br>1.901E-05<br>1.901E-05<br>1.901E-05<br>1.901E-05<br>1.901E-05<br>1.901E-05<br>1.901E-05<br>1.901E-05<br>1.901E-05<br>1.901E-05<br>1.901E-05<br>1.901E-05<br>1.901E-05<br>1.901E-05<br>1.901E-05<br>1.901E-05<br>1.901E-05<br>1.901E-05<br>1.901E-05<br>1.901E-05<br>1.901E-05<br>1.901E-05<br>1.901E-05<br>1.901E-05<br>1.901E-05<br>1.901E-05<br>1.901E-05<br>1.901E-05<br>1.901E-05<br>1.901E-05<br>1.901E-05<br>1.901E-05<br>1.901E-05<br>1.901E-05<br>1.901E-05<br>1.901E-05<br>1.901E-05<br>1.901E-05<br>1.901E-05<br>1.901E-05<br>1.901E-05<br>1.901E-05<br>1.901E-05<br>1.901E-05<br>1.901E-05<br>1.901E-05<br>1.901E-05<br>1.901E-05<br>1.901E-05<br>1.901E-05<br>1.901E-05<br>1.901E-05<br>1.901E-05<br>1.901E-05<br>1.901E-05<br>1.901E-05<br>1.901E-05<br>1.901E-05<br>1.901E-05<br>1.901E-05<br>1.901E-05<br>1.901E-05<br>1.901E-05<br>1.901E-05<br>1.901E-05<br>1.901E-05<br>1.901E-05<br>1.901E-05<br>1.901E-05<br>1.901E-05<br>1.901E-05<br>1.901E-05<br>1.901E-05<br>1.901E-05<br>1.901E-05<br>1.901E-05<br>1.901E-05<br>1.901E-05<br>1.901E-05<br>1.901E-05<br>1.901E-05<br>1.901E-05<br>1.901E-05<br>1.901E-05<br>1.901E-05<br>1.901E-05<br>1.901E-05<br>1.901E-05<br>1.901E-05<br>1.901E-05<br>1.901E-05<br>1.901E-05<br>1.901E-05<br>1.901E-05<br>1.901E-05<br>1.901E-05<br>1.901E-05<br>1.901E-05<br>1.901E-05<br>1.901E-05<br>1.901E-05<br>1.901E-05<br>1.901E-05<br>1.901E-05<br>1.901E-05<br>1.901E-05   |
| Sheung Touen B/T and Car Park<br>Sheung Touen B/T and Car Park<br>Village Access Road<br>Village Access Road              | R0_29<br>R0_29<br>R0_29<br>R0_29<br>R0_29<br>R0_29<br>R0_29<br>R0_29<br>R0_29<br>R0_29<br>R0_29<br>R0_29<br>R0_29<br>R0_29<br>R0_29<br>R0_29<br>R0_29<br>R0_29<br>R0_29<br>R0_29<br>R0_29<br>R0_29<br>R0_29<br>R0_29<br>R0_29<br>R0_29<br>R0_29<br>R0_29<br>R0_29<br>R0_29<br>R0_29<br>R0_29<br>R0_29<br>R0_29<br>R0_29<br>R0_29<br>R0_29<br>R0_29<br>R0_29<br>R0_29<br>R0_29<br>R0_29<br>R0_29<br>R0_29<br>R0_29<br>R0_29<br>R0_29<br>R0_29<br>R0_29<br>R0_29<br>R0_29<br>R0_29<br>R0_29<br>R0_29<br>R0_29<br>R0_29<br>R0_29<br>R0_29<br>R0_29<br>R0_29<br>R0_29<br>R0_29<br>R0_29<br>R0_29<br>R0_29<br>R0_29<br>R0_29<br>R0_29<br>R0_29<br>R0_29<br>R0_29<br>R0_29<br>R0_29<br>R0_29<br>R0_29<br>R0_29<br>R0_29<br>R0_29<br>R0_29<br>R0_29<br>R0_29<br>R0_29<br>R0_29<br>R0_29<br>R0_29<br>R0_29<br>R0_29<br>R0_29<br>R0_29<br>R0_29<br>R0_29<br>R0_29<br>R0_29<br>R0_29<br>R0_29<br>R0_29<br>R0_29<br>R0_29<br>R0_29<br>R0_29<br>R0_29<br>R0_29<br>R0_29<br>R0_29<br>R0_29<br>R0_29<br>R0_29<br>R0_29<br>R0_29<br>R0_29<br>R0_29<br>R0_29<br>R0_29<br>R0_29<br>R0_29<br>R0_29<br>R0_29<br>R0_29<br>R0_29<br>R0_29<br>R0_29<br>R0_29<br>R0_29<br>R0_29<br>R0_29<br>R0_29<br>R0_29<br>R0_29<br>R0_29<br>R0_29<br>R0_29<br>R0_29<br>R0_29<br>R0_29<br>R0_29<br>R0_29<br>R0_29<br>R0_29<br>R0_29<br>R0_29<br>R0_29<br>R0_29<br>R0_29<br>R0_20<br>R0_20<br>R0_20<br>R0_20<br>R0_20<br>R0_20<br>R0_20<br>R0_20<br>R0_20<br>R0_20<br>R0_20<br>R0_20<br>R0_20<br>R0_20<br>R0_20<br>R0_20<br>R0_20<br>R0_20<br>R0_20<br>R0_20<br>R0_20<br>R0_20<br>R0_20<br>R0_20<br>R0_20<br>R0_20<br>R0_20<br>R0_20<br>R0_20<br>R0_20<br>R0_20<br>R0_20<br>R0_20<br>R0_20<br>R0_20<br>R0_20<br>R0_20<br>R0_20<br>R0_20<br>R0_20<br>R0_20<br>R0_20<br>R0_20<br>R0_20<br>R0_20<br>R0_20<br>R0_20<br>R0_20<br>R0_20<br>R0_20<br>R0_20<br>R0_20<br>R0_20<br>R0_20<br>R0_20<br>R0_20<br>R0_20<br>R0_20<br>R0_20<br>R0_20<br>R0_20<br>R0_20<br>R0_20<br>R0_20<br>R0_20<br>R0_20<br>R0_20<br>R0_20<br>R0_20<br>R0_20<br>R0_20<br>R0_20<br>R0_20<br>R0_20<br>R0_20<br>R0_20<br>R0_20<br>R0_20<br>R0_20<br>R0_20<br>R0_20<br>R0_20<br>R0_20<br>R0_20<br>R0_20<br>R0_20<br>R0_20<br>R0_20<br>R0_20<br>R0_20<br>R0_20<br>R0_20<br>R0_20<br>R0_20<br>R0_20<br>R0_20<br>R0_20<br>R0_20<br>R0_20<br>R0_20<br>R0_20<br>R0_20<br>R0_20<br>R0_20<br>R0_20<br>R0_20<br>R0_20<br>R0_20<br>R0_20<br>R0_20<br>R0_20<br>R0_20<br>R0_20<br>R0_20<br>R0_20<br>R0_20<br>R0_20<br>R0_20<br>R0_20<br>R0_20<br>R0_20<br>R0_20<br>R0_20<br>R0_20<br>R0_20<br>R0_20<br>R0_20<br>R0_20<br>R0_20<br>R0_20<br>R0_20<br>R0_20<br>R0_20<br>R0_20<br>R0_20<br>R0_20<br>R0_20<br>R0_20<br>R0_20<br>R0_20<br>R0_20<br>R0_20<br>R0_20<br>R0_20<br>R0_20<br>R0_20<br>R0_20<br>R0_20<br>R0_20<br>R0_20<br>R0_20<br>R0_20<br>R0_20<br>R0_20<br>R0_20<br>R0_20<br>R0_20<br>R0_20<br>R0_20<br>R0_20<br>R0_20<br>R0_20<br>R0_20<br>R0_20<br>R0_20<br>R0_20<br>R0_20<br>R0_20<br>R0_20<br>R0_20<br>R0_20<br>R0_20<br>R0_20<br>R0_20<br>R0_20<br>R0_20<br>R0_20<br>R0_20<br>R0_20<br>R0_20<br>R0_20<br>R0_20<br>R0_20<br>R0_20<br>R0_20<br>R0_20<br>R0_20<br>R0_20<br>R0_20<br>R0_20<br>R0_20<br>R0_20<br>R0_20<br>R0_20<br>R0_20<br>R0_20<br>R0_20<br>R0_20 | 9900-1000           1900-1100           1100-1200           1200-1300           1300-1400           1300-1600           1500-1600           1500-1700           1700-1800           1900-2000           2000-2100           2100-2200           2200-2200           2200-2000           0000-0100           0100-200           0200-2100           0300-400           0400-9500           0500-6900           0500-6900           0500-6900           0500-1001           1000-1100           1100-1200           1200-1300           1300-1400           1400-1500           1500-1600           1600-1700   | 0.1281945<br>0.1457065<br>0.1360751<br>0.1607293<br>0.1557266<br>0.145988<br>0.1424823<br>0.1454852<br>0.1454852<br>0.1332075<br>0.1332075<br>0.1392054<br>0.1322354<br>0.1322354<br>0.030941<br>0<br>0.5292911<br>0.0329291<br>0.0329291<br>0.0329292<br>0.2230554<br>0.234202<br>0.2234954<br>0.2346456<br>0.2324928<br>0.2234954<br>0.2234954<br>0.2234954<br>0.2234954<br>0.2234954<br>0.2234954<br>0.2234954<br>0.2234954<br>0.2234954<br>0.2234954<br>0.2234954<br>0.2234954<br>0.2234954<br>0.2234954<br>0.2234954<br>0.3156128<br>0.3156128<br>0.3156128<br>0.3156128<br>0.3156128<br>0.3156128<br>0.3156128<br>0.3156128<br>0.3156128<br>0.3156128<br>0.3156128<br>0.3156128<br>0.3156128<br>0.3156128<br>0.3156128<br>0.3156128<br>0.3156128<br>0.3156128<br>0.3156128<br>0.3156128<br>0.3156128<br>0.3156128<br>0.3156128<br>0.3156128<br>0.3156128<br>0.3156128<br>0.3156128<br>0.3156128<br>0.3156128<br>0.3156128<br>0.3156128<br>0.3156128<br>0.3156128<br>0.3156128<br>0.3156128<br>0.3156128<br>0.3156128<br>0.3156128<br>0.3156128<br>0.3156128<br>0.3156128<br>0.3156128<br>0.3156128<br>0.3156128<br>0.3156128<br>0.3156128<br>0.3156128<br>0.3156128<br>0.3156128<br>0.3156128<br>0.3156128<br>0.3156128<br>0.3156128<br>0.3156128<br>0.3156128<br>0.3156128<br>0.3156128<br>0.3156128<br>0.3156128<br>0.3156128<br>0.3156128<br>0.3156128<br>0.3156128<br>0.3156128<br>0.3156128<br>0.3156128<br>0.3156128<br>0.3156128<br>0.3156128<br>0.3156128<br>0.3156128<br>0.3156128<br>0.3156128<br>0.3156128<br>0.3156128<br>0.3156128<br>0.3156128<br>0.3156128<br>0.3156128<br>0.3156128<br>0.3156128<br>0.3156128<br>0.3156128<br>0.3156128<br>0.3156128<br>0.3156128<br>0.3156128<br>0.3156128<br>0.3156128<br>0.3156128<br>0.3156128<br>0.3156128<br>0.3156128<br>0.3156128<br>0.3156128<br>0.3156128<br>0.3156128<br>0.3156128<br>0.3156128<br>0.3156128<br>0.3156128<br>0.3156128<br>0.3156128<br>0.3156128<br>0.3156128<br>0.3156128<br>0.3156128<br>0.3156128<br>0.3156128<br>0.3156128<br>0.3156128<br>0.3156128<br>0.3156128<br>0.3156128<br>0.3156128<br>0.3156128<br>0.3156128<br>0.3156128<br>0.3156128<br>0.3156128<br>0.3156128<br>0.3156128<br>0.3156128<br>0.3156128<br>0.3156128<br>0.3156128<br>0.3156128<br>0.3156128<br>0.3156128<br>0.3156128<br>0.3156128<br>0.3156128<br>0.3156128<br>0.3156128<br>0.3156128<br>0.3156128<br>0.3156128<br>0.3156128<br>0.3156128<br>0.3156128<br>0.3156128<br>0.3156128<br>0.3156128<br>0.3156128<br>0.3156128<br>0.3156128<br>0.3156128   | 9.342.003  <br>9.342.003  <br>9.342.003  <br>9.342.003  <br>9.342.003  <br>9.342.003  <br>9.342.003  <br>1.032.007  <br>6.5552-08  <br>6.5552-08  <br>7.7712-08  <br>6.5552-08  <br>7.7712-08  <br>6.3352-08  <br>7.7121-08  <br>6.3352-08  <br>7.7121-08  <br>6.3352-08  <br>7.75062-08  <br>0  <br>8.0362-08  <br>0  <br>8.0362-08  <br>0  <br>8.0362-08  <br>0  <br>8.0362-08  <br>0  <br>8.0362-08  <br>9.6322-08  <br>1.0328-07  <br>9.6322-08  <br>9.5562-08  <br>9.556   | 0.000135 (<br>0.000135 (<br>0.000135 (<br>0.000147 (<br>0.000147 (<br>0.000147 (<br>0.000164 (<br>0.000164 (<br>0.000164 (<br>0.000164 (<br>0.000164 (<br>0.000164 (<br>0.000164 (<br>0.000165 (<br>0.000164 (<br>0.000165 (  |
b.1261(10)<br>1.121(14)<br>1.121(14)<br>1.1338(7)<br>1.121(14)<br>1.1338(7)<br>1.121(14)<br>1.1338(7)<br>1.121(14)<br>1.121(14)<br>1.121(14)<br>1.121(14)<br>1.121(14)<br>1.121(14)<br>1.121(14)<br>1.121(14)<br>1.121(14)<br>1.121(14)<br>1.121(14)<br>1.121(14)<br>1.121(14)<br>1.121(14)<br>1.121(14)<br>1.121(14)<br>1.121(14)<br>1.121(14)<br>1.121(14)<br>1.121(14)<br>1.121(14)<br>1.121(14)<br>1.121(14)<br>1.121(14)<br>1.121(14)<br>1.121(14)<br>1.121(14)<br>1.121(14)<br>1.121(14)<br>1.121(14)<br>1.121(14)<br>1.121(14)<br>1.121(14)<br>1.121(14)<br>1.121(14)<br>1.121(14)<br>1.121(14)<br>1.121(14)<br>1.121(14)<br>1.121(14)<br>1.121(14)<br>1.121(14)<br>1.121(14)<br>1.121(14)<br>1.121(14)<br>1.121(14)<br>1.121(14)<br>1.121(14)<br>1.121(14)<br>1.121(14)<br>1.121(14)<br>1.121(14)<br>1.121(14)<br>1.121(14)<br>1.121(14)<br>1.121(14)<br>1.121(14)<br>1.121(14)<br>1.121(14)<br>1.121(14)<br>1.121(14)<br>1.121(14)<br>1.121(14)<br>1.121(14)<br>1.121(14)<br>1.121(14)<br>1.121(14)<br>1.121(14)<br>1.121(14)<br>1.121(14)<br>1.121(14)<br>1.121(14)<br>1.121(14)<br>1.121(14)<br>1.121(14)<br>1.121(14)<br>1.121(14)<br>1.121(14)<br>1.121(14)<br>1.121(14)<br>1.121(14)<br>1.121(14)<br>1.121(14)<br>1.121(14)<br>1.121(14)<br>1.121(14)<br>1.121(14)<br>1.121(14)<br>1.121(14)<br>1.121(14)<br>1.121(14)<br>1.121(14)<br>1.121(14)(14)(14)(14)(14)(14)(14)(14)(14)(14  | 9 4 3114 -08<br>9 4 3114 -08<br>7 7 7615 -08<br>8 8 845 -08<br>8 2 786 -08<br>9 7345 -08<br>6 3 736 -08<br>7 4 737 -08<br>6 0 325 -08<br>7 7465 -08<br>6 -0526 -08<br>6 -0526 -08<br>6 -0526 -08<br>6 -0526 -08<br>6 -0526 -08<br>6 -0526 -08<br>7 -1526 -08<br>7 -1526 -08<br>7 -2556 -08<br>3 -8967 -10<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0   | 0.0001719<br>0.0001752<br>0.0001752<br>0.0001752<br>0.0001752<br>0.0001752<br>0.0001752<br>0.0001752<br>0.0001752<br>0.0001752<br>0.0001752<br>0.0001752<br>0.0001752<br>0.0001752<br>0.0001752<br>0.0001752<br>0.0001752<br>0.0001752<br>0.0001752<br>0.0001752<br>0.0001752<br>0.0001752<br>0.0001752<br>0.0001752<br>0.0001752<br>0.0001752<br>0.0001752<br>0.0001752<br>0.0001752<br>0.0001752<br>0.0001752<br>0.0001752<br>0.0001752<br>0.0001752<br>0.0001752<br>0.0001752<br>0.0001752<br>0.0001752<br>0.0001752<br>0.0001752<br>0.0001752<br>0.0001752<br>0.0001752<br>0.0001752<br>0.0001752<br>0.0001752<br>0.0001752<br>0.0001752<br>0.0001752<br>0.0001752<br>0.0001752<br>0.0001752<br>0.0001752<br>0.0001752<br>0.0001752<br>0.0001752<br>0.0001752<br>0.0001752<br>0.0001752<br>0.0001752<br>0.0001752<br>0.0001752<br>0.0001752<br>0.0001752<br>0.0001752<br>0.0001752<br>0.0001752<br>0.0001752<br>0.0001752<br>0.0001752<br>0.0001752<br>0.0001752<br>0.0001752<br>0.0001752<br>0.0001752<br>0.0001752<br>0.0001752<br>0.0001752<br>0.0001752<br>0.0001752<br>0.0001752<br>0.0001752<br>0.0001752<br>0.0001752<br>0.0001752<br>0.0001752<br>0.0001752<br>0.0001752<br>0.0001752<br>0.0001752<br>0.0001752<br>0.0001752<br>0.0001752<br>0.0001752<br>0.0001752<br>0.0001752<br>0.0001752<br>0.0001752<br>0.0001752<br>0.0001752<br>0.0001752<br>0.0001752<br>0.0001752<br>0.0001752<br>0.0001752<br>0.0001752<br>0.0001752<br>0.0001752<br>0.0001752<br>0.0001752<br>0.0001752<br>0.0001752<br>0.0001752<br>0.0001752<br>0.0001752<br>0.0001752<br>0.0001752<br>0.0001752<br>0.0001752<br>0.0001752<br>0.0001752<br>0.0001752<br>0.0001752<br>0.0001752<br>0.0001752<br>0.0001752<br>0.0001752<br>0.0001752<br>0.0001752<br>0.0001752<br>0.0001752<br>0.0001752<br>0.0001752<br>0.0001752<br>0.0001752<br>0.0001752<br>0.0001752<br>0.0001752<br>0.0001752<br>0.0001752<br>0.0001752<br>0.0001752<br>0.0001752<br>0.0001752<br>0.0001752<br>0.0001752<br>0.0001752<br>0.0001752<br>0.0001752<br>0.0001752<br>0.0001752<br>0.0001752<br>0.0001752<br>0.0001752<br>0.0001752<br>0.0001752<br>0.0001752<br>0.0001752<br>0.0001752<br>0.0001752<br>0.0001752<br>0.0001752<br>0.0001752<br>0.0001752<br>0.0001752<br>0.0001752<br>0.0001752<br>0.0001752<br>0.0001752<br>0.0001752<br>0.0001752<br>0.0001752<br>0.0001752<br>0.0001752<br>0.0001752<br>0.0001752<br>0.0001752<br>0.0001752<br>0.0001752<br>0.0001752<br>0.0001752<br>0.0001752<br>0.0001752<br>0.0001752<br>0.0001752<br>0.0001752  | 1203596<br>0.1142206<br>0.1310694<br>0.1216391<br>0.1426721<br>0.1426721<br>0.1426721<br>0.1323766<br>0.1323766<br>0.1323766<br>0.1323765<br>0.1320375<br>0.1320375<br>0.1201552<br>0.1201552<br>0.1201552<br>0.1201562<br>0.1202452<br>0.1202452<br>0.1202452<br>0.1202452<br>0.1202452<br>0.01624067<br>0.01624067<br>0.1858453<br>0.1873168<br>0.18771614<br>0.2555644<br>0.2655655<br>0.26552665   
   | 9.001-08<br>7.322E.08<br>8.402E.08<br>7.797E.08<br>9.146E.08<br>6.033E.08<br>7.391-60<br>8.5749E.08<br>7.39E.08<br>6.333E.08<br>6.439E.08<br>6.439E.08<br>6.439E.08<br>6.439E.08<br>6.439E.08<br>6.439E.08<br>6.439E.08<br>6.439E.08<br>6.439E.08<br>6.439E.08<br>6.439E.08<br>6.439E.08<br>6.439E.08<br>6.439E.08<br>6.439E.08<br>6.439E.08<br>6.439E.08<br>6.439E.08<br>6.439E.08<br>6.439E.08<br>6.439E.08<br>8.427E.00<br>8.427E.08<br>8.542E.08<br>8.542E.08<br>8.542E.08<br>8.542E.08<br>8.542E.08<br>8.542E.08<br>8.542E.08<br>8.542E.08<br>8.542E.08<br>8.542E.08<br>8.542E.08<br>8.542E.08<br>8.542E.08<br>8.542E.08<br>8.542E.08<br>8.542E.08<br>8.542E.08<br>8.542E.08<br>8.542E.08<br>8.542E.08<br>8.542E.08<br>8.542E.08<br>8.542E.08<br>8.542E.08<br>8.542E.08<br>8.542E.08<br>8.542E.08<br>8.542E.08<br>8.542E.08<br>8.542E.08<br>8.542E.08<br>8.542E.08<br>8.542E.08<br>8.542E.08<br>8.542E.08<br>8.542E.08<br>8.542E.08<br>8.542E.08<br>8.542E.08<br>8.542E.08<br>8.542E.08<br>8.542E.08<br>8.542E.08<br>8.542E.08<br>8.542E.08<br>8.542E.08<br>8.542E.08<br>8.542E.08<br>8.542E.08<br>8.542E.08<br>8.542E.08<br>8.542E.08<br>8.542E.08<br>8.542E.08<br>8.542E.08<br>8.542E.08<br>8.542E.08<br>8.542E.08<br>8.542E.08<br>8.542E.08<br>8.542E.08<br>8.542E.08<br>8.542E.08<br>8.542E.08<br>8.542E.08<br>8.542E.08<br>8.542E.08<br>8.542E.08<br>8.542E.08<br>8.542E.08<br>8.542E.08<br>8.542E.08<br>8.542E.08<br>8.542E.08<br>8.542E.08<br>8.542E.08<br>8.542E.08<br>8.542E.08<br>8.542E.08<br>8.542E.08<br>8.542E.08<br>8.542E.08<br>8.542E.08<br>8.542E.08<br>8.542E.08<br>8.542E.08<br>8.542E.08<br>8.542E.08<br>8.542E.08<br>8.542E.08<br>8.542E.08<br>8.542E.08<br>8.542E.08<br>8.542E.08<br>8.542E.08<br>8.542E.08<br>8.542E.08<br>8.542E.08<br>8.542E.08<br>8.542E.08<br>8.542E.08<br>8.542E.08<br>8.542E.08<br>8.542E.08<br>8.542E.08<br>8.542E.08<br>8.542E.08<br>8.542E.08<br>8.542E.08<br>8.542E.08<br>8.542E.08<br>8.542E.08<br>8.542E.08<br>8.542E.08<br>8.542E.08<br>8.542E.08<br>8.542E.08<br>8.542E.08<br>8.542E.08<br>8.542E.08<br>8.542E.08<br>8.542E.08<br>8.542E.08<br>8.542E.08<br>8.542E.08<br>8.542E.08<br>8.542E.08<br>8.542E.08<br>8.542E.08<br>8.542E.08<br>8.542E.08<br>8.542E.08<br>8.542E.08<br>8.542E.08<br>8.542E.08<br>8.542E.08<br>8.542E.08<br>8.542E.08<br>8.542E.08<br>8.542E.08<br>8.542E.08<br>8.542E.08<br>8.542E.08<br>8.542E.08<br>8.542E.08<br>8.542E.08<br>8.542E.08<br>8.542E.08<br>8.542E.08<br>8.542E.08<br>8.542E.08<br>8.542E.08<br>8.542E.08<br>8.542E.08<br>8.542E.08<br>8.542E.08<br>8.542E.08<br>8.542E.08<br>8.542E.08<br>8.542E.08<br>8.542E.08<br>8.542E.08<br>8.542E.08<br>8.542E.08<br>8.542E.08<br>8.542E.08<br>8.542E.08<br>8.542E.08<br>8.552E.08<br>8.552E.08<br>8.552E.08<br>8.552E.0                | 0.0001462<br>0.0001365<br>0.0001287<br>0.0001486<br>9.834-05<br>0.0001486<br>9.834-05<br>0.0001480<br>9.345-05<br>0.0001143<br>0.0001021<br>0.0001021<br>0.0001021<br>0.0001021<br>0.0001021<br>0.000123<br>0.000123<br>0.000123<br>0.000123<br>0.000123<br>0.000123<br>0.000123<br>0.000123<br>0.000123<br>0.000123<br>0.000123<br>0.000123<br>0.000123<br>0.000123<br>0.000123<br>0.000123<br>0.000123<br>0.000123<br>0.000123<br>0.000123<br>0.000123<br>0.000123<br>0.000123<br>0.000123<br>0.000123<br>0.000123<br>0.000123<br>0.000123<br>0.000123<br>0.000123<br>0.000123<br>0.000123<br>0.000123<br>0.000123<br>0.000123<br>0.000123<br>0.000123<br>0.000123<br>0.000123<br>0.000123<br>0.000123<br>0.000123<br>0.000123<br>0.000123<br>0.000123<br>0.000123<br>0.000123<br>0.000123<br>0.000123<br>0.000123<br>0.000123<br>0.000123<br>0.000123<br>0.000123<br>0.000123<br>0.000123<br>0.000123<br>0.000123<br>0.000123<br>0.000123<br>0.000123<br>0.000123<br>0.000123<br>0.000123<br>0.000123<br>0.000123<br>0.000123<br>0.000123<br>0.000123<br>0.000123<br>0.000123<br>0.000123<br>0.000123<br>0.000123<br>0.000123<br>0.000123<br>0.000123<br>0.000123<br>0.000123<br>0.000123<br>0.000123<br>0.000123<br>0.000123<br>0.000123<br>0.000123<br>0.000123<br>0.000123<br>0.000123<br>0.000123<br>0.000123<br>0.000123<br>0.000123<br>0.000123<br>0.000123<br>0.000123<br>0.000123<br>0.000123<br>0.000123<br>0.000123<br>0.000123<br>0.000123<br>0.000123<br>0.000123<br>0.000123<br>0.000123<br>0.000123<br>0.000123<br>0.000123<br>0.000123<br>0.000123<br>0.000123<br>0.000123<br>0.000123<br>0.000123<br>0.000123<br>0.000123<br>0.000123<br>0.000123<br>0.000123<br>0.000123<br>0.000123<br>0.000123<br>0.000123<br>0.000123<br>0.000123<br>0.000123<br>0.000123<br>0.000123<br>0.000123<br>0.000123<br>0.000123<br>0.000123<br>0.000123<br>0.000123<br>0.000123<br>0.000123<br>0.000123<br>0.000123<br>0.000123<br>0.000123<br>0.000123<br>0.000123<br>0.000123<br>0.000123<br>0.000123<br>0.000123<br>0.000123<br>0.000123<br>0.000123<br>0.000123<br>0.000123<br>0.000123<br>0.000123<br>0.000123<br>0.000123<br>0.000123<br>0.000123<br>0.000123<br>0.000123<br>0.000123<br>0.000123<br>0.000123<br>0.000123<br>0.000123<br>0.000123<br>0.000123<br>0.000123<br>0.000123<br>0.000123<br>0.000123<br>0.000123<br>0.000123<br>0.000123<br>0.000123<br>0.000123<br>0.000123<br>0.000123<br>0.000123<br>0.000123<br>0.000123<br>0.000123<br>0.000123<br>0.000123<br>0.000123<br>0.000123<br>0.000123<br>0.000123<br>0.000123<br>0.000123<br>0.000123<br>0.000123<br>0.000123<br>0.000123<br>0.000123<br>0.000123<br>0.000123<br>0.000123<br>0.000123<br>0.000123<br>0.000123<br>0.000123<br>0.000123<br>0.000123<br>0.000123<br>0.0000123<br>0.0000123<br>0.0000123<br>0.0000123<br>0.0000123<br>0.00000000000          | 2.4904488<br>1.9842855<br>2.2002208<br>2.1078468<br>2.2364653<br>3.1057198<br>3.0668346<br>2.9569573<br>2.8602462<br>2.9569573<br>2.9569573<br>2.9117616<br>4.9637927<br>4.4562831<br>0.0573728<br>0<br>0<br>0<br>1.3610254<br>0.06059669<br>0.6039622<br>0.511577<br>0.06025855<br>0.9897613<br>0.6002628<br>0.8313783<br>0.8331573<br>0.8332163<br>0.8323763  | 1.8634-06<br>1.272E-06<br>1.41E-06<br>1.351E-06<br>1.327E-06<br>1.232E-06<br>1.264E-06<br>1.264E-06<br>1.264E-06<br>1.264E-06<br>1.401E-06<br>1.440E-06<br>1.440E-06<br>1.440E-06<br>1.440E-06<br>1.341E-06<br>1.341E-06<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0   
  | 0.0030261<br>0.0020660<br>0.0022915<br>0.002393<br>0.002393<br>0.002393<br>0.002393<br>0.0021564<br>0.0022424<br>0.00274<br>0.002374<br>0.002374<br>0.002374<br>0.002340<br>0.002374<br>0.002340<br>0.002374<br>0.0023849<br>0.0023849<br>0.0003777<br>0.0003777<br>0.0003777<br>0.0003777<br>0.0003777<br>0.0003777<br>0.0003777<br>0.0003777<br>0.0003777<br>0.0003777<br>0.0003777<br>0.0003352<br>0.0004589<br>0.000459<br>0.000462<br>0.000462<br>0.000462<br>0.0004655  | 2.3566842<br>2.997776<br>2.005201<br>2.005201<br>2.952749<br>2.952749<br>2.952749<br>2.952749<br>2.952749<br>2.952749<br>2.950787<br>2.8217082<br>2.7301698<br>3.0191874<br>2.5038689<br>2.7392982<br>3.3311649<br>4.688413<br>0.0463027<br>0<br>0<br>0.04704<br>0.5415078<br>1.2303809<br>0.047046<br>0.5415078<br>0.5415078<br>0.5415078<br>0.5415078<br>0.55234311<br>0.7528431<br>0.7704794<br>0.5523431<br>0.7704794<br>0.7736863<br>0.777488<br>0.7776883<br>0.7776883   | 17622-06<br>13452-06<br>13452-06<br>13452-06<br>13452-06<br>13452-06<br>13452-06<br>13452-06<br>13452-06<br>13452-06<br>13452-06<br>13372-06<br>13372-06<br>13372-06<br>13372-06<br>13372-06<br>13372-06<br>13372-06<br>13372-06<br>13372-06<br>13372-06<br>13372-06<br>13372-06<br>13372-06<br>13372-06<br>13372-06<br>13372-06<br>13372-06<br>13372-06<br>13372-06<br>13372-06<br>13372-06<br>13372-06<br>13372-06<br>13372-06<br>13372-06<br>13372-06<br>13372-06<br>13372-06<br>13372-06<br>13372-06<br>13372-06<br>13372-06<br>13372-06<br>13372-06<br>13372-06<br>13372-06<br>13372-06<br>13372-06<br>13372-06<br>13372-06<br>13372-06<br>13372-06<br>13372-06<br>13372-06<br>13372-06<br>13372-06<br>13372-06<br>13372-06<br>13372-06<br>13372-06<br>13372-06<br>13372-06<br>13372-06<br>13372-06<br>13372-06<br>13372-06<br>13372-06<br>13372-06<br>13372-06<br>13372-06<br>13372-06<br>13372-06<br>13372-06<br>13372-06<br>13372-06<br>13372-06<br>13372-06<br>13372-06<br>13372-06<br>13372-06<br>13372-06<br>13372-06<br>13372-06<br>13372-06<br>13372-06<br>13372-06<br>13372-06<br>13372-06<br>13372-06<br>13372-06<br>13372-06<br>13372-06<br>13372-06<br>13372-06<br>13372-06<br>13372-06<br>13372-06<br>13372-06<br>13372-06<br>13372-06<br>13372-06<br>13372-06<br>13372-06<br>13372-06<br>13372-06<br>13372-06<br>13372-06<br>13372-06<br>13372-06<br>13372-06<br>13372-06<br>13372-06<br>13372-06<br>13372-06<br>13372-06<br>13372-06<br>13372-06<br>13372-06<br>13372-06<br>13372-06<br>13372-06<br>13372-06<br>13372-06<br>13372-06<br>13372-06<br>13372-06<br>13372-06<br>13372-06<br>13372-06<br>13372-06<br>13372-06<br>13372-06<br>13372-06<br>13372-06<br>13372-06<br>13372-06<br>13372-06<br>13372-06<br>13372-06<br>13372-06<br>13372-06<br>13372-06<br>13372-06<br>13372-06<br>13372-06<br>13372-06<br>13372-06<br>13372-06<br>13372-06<br>13372-06<br>13372-06<br>13372-06<br>13372-06<br>13372-06<br>13372-06<br>13372-06<br>13372-06<br>13372-06<br>13372-06<br>13372-06<br>13372-06<br>13372-06<br>13372-06<br>13372-06<br>13372-06<br>13372-06<br>13372-06<br>13372-06<br>13372-06<br>13372-06<br>13372-06<br>13372-06<br>13372-06<br>13372-06<br>13372-06<br>13372-06<br>13372-06<br>13372-06<br>13372-06<br>13372-06<br>13372-06<br>13372-06<br>13372-06<br>13372-06<br>13372-06<br>13372-06<br>13372-06<br>13372-06<br>13372-06<br>13372-06<br>13372-06<br>13372-06<br>13372-06<br>13372-06<br>13372-06<br>13372-06<br>13372-06<br>13372-06<br>13372-06<br>13372-06<br>13372-06<br>13372-06<br>13372-06<br>13372- | 0.0028636<br>0.0021560<br>0.0021848<br>0.00221847<br>0.0020885<br>0.0022167<br>0.0023695<br>0.0025171<br>0.0025171<br>0.0021501<br>0.0021501<br>0.0021501<br>0.0021201<br>0.0021201<br>0.0021201<br>0.0022225<br>0.0012425<br>0.0024249<br>0.0003426<br>0.0003426<br>0.0003426<br>0.0003414<br>1.305E-05<br>0.0004555<br>0.0004555<br>0.0004555<br>0.0004555<br>0.0004555<br>0.0004555<br>0.0004555<br>0.0004555<br>0.0004555<br>0.0004255<br>0.0004255<br>0.0004255<br>0.0004255<br>0.0004255<br>0.0004255<br>0.0004255<br>0.0004255<br>0.0004255<br>0.0004255  
   | 2.2440968<br>1.795138<br>1.9915216<br>1.8951342<br>2.0102391<br>2.7864095<br>2.7624095<br>2.7624095<br>2.7624095<br>2.3221589<br>2.3221589<br>2.3221589<br>2.3725056<br>2.4375101<br>2.6016999<br>3.7301495<br>0<br>0<br>0<br>0<br>1398552<br>0.4375312<br>0.43753324<br>0.0410785<br>0.4397553<br>0.4397553<br>0.4397553<br>0.439755701<br>0.50509171<br>0.50509171<br>0.50509171<br>0.50509171<br>0.50509171<br>0.50509171<br>0.50509171<br>0.50509171<br>0.50509171<br>0.50509171<br>0.50509171<br>0.50509171<br>0.50509171<br>0.50509171<br>0.50509171<br>0.50509171<br>0.50509171<br>0.50509171<br>0.50509171<br>0.50509171<br>0.50509171<br>0.50509171<br>0.50509171<br>0.50509171<br>0.50509171<br>0.50509171<br>0.50509171<br>0.50509171<br>0.50509171<br>0.50509171<br>0.50509171<br>0.50509171<br>0.50509171<br>0.50509171<br>0.50509171<br>0.50509171<br>0.50509171<br>0.50509171<br>0.50509171<br>0.50509171<br>0.50509171<br>0.50509171<br>0.50509171<br>0.50509171<br>0.50509171<br>0.50509171<br>0.50509171<br>0.50509171<br>0.50509171<br>0.50509171<br>0.50509171<br>0.50509171<br>0.50509171<br>0.50509171<br>0.50509171<br>0.50509171<br>0.50509171<br>0.50509171<br>0.50509171<br>0.50509171<br>0.50509171<br>0.50509171<br>0.50509171<br>0.50509171<br>0.50509171<br>0.50509171<br>0.50509171<br>0.50509171<br>0.50509171<br>0.50509171<br>0.50509171<br>0.50509171<br>0.50509171<br>0.50509171<br>0.50509171<br>0.50509171<br>0.50509171<br>0.50509171<br>0.50509171<br>0.50509171<br>0.50509171<br>0.50509171<br>0.50509171<br>0.50509171<br>0.50509171<br>0.50509171<br>0.50509171<br>0.50509171<br>0.50509171<br>0.50509171<br>0.50509171<br>0.50509171<br>0.50509171<br>0.50509171<br>0.50509171<br>0.50509171<br>0.50509171<br>0.50509171<br>0.50509171<br>0.50509171<br>0.50509171<br>0.50509171<br>0.50509171<br>0.50509171<br>0.50509171<br>0.50509171<br>0.50509171<br>0.50509171<br>0.50509171<br>0.50509171<br>0.50509171<br>0.50509171<br>0.50509171<br>0.50509171<br>0.50509171<br>0.50509171<br>0.50509171<br>0.50509171<br>0.50509171<br>0.50509171<br>0.50509171<br>0.50509171<br>0.50509171<br>0.50509171<br>0.50509171<br>0.50509171<br>0.50509171<br>0.50509171<br>0.50509171<br>0.50509171<br>0.50509171<br>0.50509171<br>0.50509171<br>0.50509171<br>0.50509171<br>0.50509171<br>0.50509171<br>0.50509171<br>0.50509171<br>0.50509171<br>0.50509171<br>0.50509171<br>0.50509171<br>0.50509171<br>0.50509171<br>0.50509171<br>0.50509171<br>0.50509171<br>0.50509171<br>0.50509171<br>0.50509171<br>0.50509171<br>0.50509171<br>0.50509171<br>0.50509171<br>0.50509171<br>0.50509171<br>0.50509171<br>0.50509171<br>0.50509171<br>0.50509171<br>0.50509171<br>0.50509171<br>0.505   | 1.6782.06<br>1.2412.06<br>1.2412.06<br>1.2412.06<br>1.2412.06<br>1.2412.06<br>1.2412.06<br>1.2412.06<br>1.2412.06<br>1.2442.06<br>1.2442.06<br>1.2442.06<br>1.2442.06<br>1.2442.06<br>1.2442.06<br>0<br>0<br>1.7312.07<br>0<br>0<br>1.7312.07<br>1.7312.07<br>1.7312.07<br>1.2312.07<br>1.2312.07<br>1.2312.07<br>1.2312.07<br>1.2312.07<br>1.2312.07<br>1.2312.07<br>1.2312.07<br>1.2312.07<br>1.2312.07<br>1.2312.07<br>1.2312.07<br>1.2312.07<br>1.2312.07<br>1.2312.07<br>1.2312.07<br>1.2312.07<br>1.2312.07<br>1.2312.07<br>1.2312.07<br>1.2312.07<br>1.2312.07<br>1.2312.07<br>1.2312.07<br>1.2312.07<br>1.2312.07<br>1.2312.07<br>1.2312.07<br>1.2312.07<br>1.2312.07<br>1.2312.07<br>1.2312.07<br>1.2312.07<br>1.2312.07<br>1.2312.07<br>1.2312.07<br>1.2312.07<br>1.2312.07<br>1.2312.07<br>1.2312.07<br>1.2312.07<br>1.2312.07<br>1.2312.07<br>1.2312.07<br>1.2312.07<br>1.2312.07<br>1.2312.07<br>1.2312.07<br>1.2312.07<br>1.2312.07<br>1.2312.07<br>1.2312.07<br>1.2312.07<br>1.2312.07<br>1.2312.07<br>1.2312.07<br>1.2312.07<br>1.2312.07<br>1.2312.07<br>1.2312.07<br>1.2312.07<br>1.2312.07<br>1.2312.07<br>1.2312.07<br>1.2312.07<br>1.2312.07<br>1.2312.07<br>1.2312.07<br>1.2312.07<br>1.2312.07<br>1.2312.07<br>1.2312.07<br>1.2312.07<br>1.2312.07<br>1.2312.07<br>1.2312.07<br>1.2312.07<br>1.2312.07<br>1.2312.07<br>1.2312.07<br>1.2312.07<br>1.2312.07<br>1.2312.07<br>1.2312.07<br>1.2312.07<br>1.2312.07<br>1.2312.07<br>1.2312.07<br>1.2312.07<br>1.2312.07<br>1.2312.07<br>1.2312.07<br>1.2312.07<br>1.2312.07<br>1.2312.07<br>1.2312.07<br>1.2312.07<br>1.2312.07<br>1.2312.07<br>1.2312.07<br>1.2312.07<br>1.2312.07<br>1.2312.07<br>1.2312.07<br>1.2312.07<br>1.2312.07<br>1.2312.07<br>1.2312.07<br>1.2312.07<br>1.2312.07<br>1.2312.07<br>1.2312.07<br>1.2312.07<br>1.2312.07<br>1.2312.07<br>1.2312.07<br>1.2312.07<br>1.2312.07<br>1.2312.07<br>1.2312.07<br>1.2312.07<br>1.2312.07<br>1.2312.07<br>1.2312.07<br>1.2312.07<br>1.2312.07<br>1.2312.07<br>1.2312.07<br>1.2312.07<br>1.2312.07<br>1.2312.07<br>1.2312.07<br>1.2312.07<br>1.2312.07<br>1.2312.07<br>1.2312.07<br>1.2312.07<br>1.2312.07<br>1.2312.07<br>1.2312.07<br>1.2312.07<br>1.2312.07<br>1.2312.07<br>1.2312.07<br>1.2312.07<br>1.2312.07<br>1.2312.07<br>1.2312.07<br>1.2312.07<br>1.2312.07<br>1.2312.07<br>1.2312.07<br>1.2312.07<br>1.2312.07<br>1.2312.07<br>1.2312.07<br>1.2312.07<br>1.2312.07<br>1.2312.07<br>1.2312.07<br>1.2312.07<br>1.2312.07<br>1.2312.07<br>1.2312.07<br>1.2312.07<br>1.2312.07<br>1.2312.07<br>1.2312.07<br>1.2312.07<br>1.2312.07<br>1.2312.07<br>1.2312.07<br>1.2312.07<br>1.2312.07<br>1.2312.07<br>1.2312.07<br>1.2312.07<br>1.2312.07<br>1.2312.07<br>1.2312.07<br>1.2312.07<br>1.2312.07<br>1.2312.07<br>1.2312.07<br>1.2312.0 | 0.0027267<br>0.0018534<br>0.0020742<br>0.0020742<br>0.0020937<br>0.0023625<br>0.0020937<br>0.0023625<br>0.0023625<br>0.002241<br>0.002262<br>0.0021155<br>0.0020542<br>0.0021155<br>0.0020542<br>0.0021155<br>0.0020542<br>0.0021155<br>0.0020542<br>0.0021313<br>0.0020702<br>1.1586-05<br>0.00003163<br>0.00003163<br>0.00003163<br>0.00002780<br>0.00003163<br>0.00002780<br>0.00003163<br>0.00002780<br>0.00003163<br>0.00002780<br>0.00003163<br>0.00002780<br>0.0000385<br>0.0000385<br>0.0000385<br>0.00003863   |
2.623364<br>2.623324<br>2.3452273<br>2.2439219<br>2.261464<br>2.261464<br>2.1051925<br>2.1051925<br>2.1051925<br>2.1052925<br>2.1052927<br>2.105292<br>2.105292<br>2.105292<br>2.105292<br>2.105292<br>2.105292<br>2.105292<br>2.105292<br>2.105292<br>2.105292<br>2.105292<br>2.105292<br>2.105292<br>2.105292<br>2.105292<br>2.105292<br>2.105292<br>2.105292<br>2.105292<br>2.105292<br>2.105292<br>2.105292<br>2.105292<br>2.105292<br>2.105292<br>2.105292<br>2.105292<br>2.105292<br>2.105292<br>2.105292<br>2.105292<br>2.105292<br>2.105292<br>2.105292<br>2.105292<br>2.105292<br>2.105292<br>2.105292<br>2.105292<br>2.105292<br>2.105292<br>2.105292<br>2.105292<br>2.105292<br>2.105292<br>2.105292<br>2.105292<br>2.105292<br>2.105292<br>2.105292<br>2.105292<br>2.105292<br>2.105292<br>2.105292<br>2.105292<br>2.105292<br>2.105292<br>2.105292<br>2.105292<br>2.105292<br>2.105292<br>2.105292<br>2.105292<br>2.105292<br>2.105292<br>2.105292<br>2.105292<br>2.105292<br>2.105292<br>2.105292<br>2.105292<br>2.105292<br>2.105292<br>2.105292<br>2.105292<br>2.105292<br>2.105292<br>2.105292<br>2.105292<br>2.105292<br>2.105292<br>2.105292<br>2.105292<br>2.105292<br>2.105292<br>2.105292<br>2.105292<br>2.105292<br>2.105292<br>2.105292<br>2.105292<br>2.105292<br>2.105292<br>2.105292<br>2.105292<br>2.105292<br>2.105292<br>2.105292<br>2.105292<br>2.105292<br>2.105292<br>2.105292<br>2.105292<br>2.105292<br>2.105292<br>2.105292<br>2.105292<br>2.105292<br>2.105292<br>2.105292<br>2.105292<br>2.105292<br>2.105292<br>2.105292<br>2.105292<br>2.105292<br>2.105292<br>2.105292<br>2.105292<br>2.105292<br>2.105292<br>2.105292<br>2.105292<br>2.105292<br>2.105292<br>2.105292<br>2.105292<br>2.105292<br>2.105292<br>2.105292<br>2.105292<br>2.105292<br>2.105292<br>2.105292<br>2.105292<br>2.105292<br>2.105292<br>2.105292<br>2.105292<br>2.105292<br>2.105292<br>2.105292<br>2.105292<br>2.105292<br>2.105292<br>2.105292<br>2.105292<br>2.105292<br>2.105292<br>2.105292<br>2.105292<br>2.105292<br>2.105292<br>2.105292<br>2.105292<br>2.105292<br>2.105292<br>2.105292<br>2.105292<br>2.105292<br>2.105292<br>2.105292<br>2.105292<br>2.105292<br>2.105292<br>2.105292<br>2.105292<br>2.105292<br>2.105292<br>2.105292<br>2.105292<br>2.105292<br>2.105292<br>2.105292<br>2.105292<br>2.105292<br>2.105292<br>2.105292<br>2.105292<br>2.105292<br>2.105292<br>2.105292<br>2.105292<br>2.105292<br>2.105292<br>2.105292<br>2.105292<br>2.105292<br>2.105292<br>2.105292<br>2.105292<br>2.105292<br>2.105292<br>2.105292<br>2.105292<br>2.105292<br>2.105292<br>2.105292<br>2.105292<br>2.105292<br>2.105292<br>2.105292<br>2.105292<br>2.105292<br>2.105292<br>2.105292<br>2.105292<br>2.105292<br>2.105292<br>2.105292<br>2.105292<br>2.105292<br>2.105292<br>2.105292<br>2.105292<br>2.105 | 19622-06<br>13542-06<br>13542-06<br>14382-06<br>13772-06<br>13772-06<br>13772-06<br>13772-06<br>13772-06<br>14772-06<br>14772-06<br>14772-06<br>14772-06<br>14772-06<br>14772-06<br>14772-06<br>14772-06<br>14772-06<br>14772-06<br>14772-06<br>14772-06<br>14772-06<br>14772-06<br>14772-06<br>14772-07<br>15772-07<br>16952-09<br>2.511E-07<br>2.7632-07<br>3.7482-07<br>3.7482-07<br>3.7482-07<br>3.7482-07<br>3.7482-07<br>3.7482-07<br>3.7482-07<br>3.7482-07<br>3.7482-07<br>3.7482-07<br>3.7482-07<br>3.7482-07<br>3.7482-07<br>3.7482-07<br>3.7482-07<br>3.7482-07<br>3.7482-07<br>3.7482-07<br>3.7482-07<br>3.7482-07<br>3.7482-07<br>3.7482-07<br>3.7482-07<br>3.7482-07<br>3.7482-07<br>3.7482-07<br>3.7482-07<br>3.7482-07<br>3.7482-07<br>3.7482-07<br>3.7482-07<br>3.7482-07<br>3.7482-07<br>3.7482-07<br>3.7482-07<br>3.7482-07<br>3.7482-07<br>3.7482-07<br>3.7482-07<br>3.7482-07<br>3.7482-07<br>3.7482-07<br>3.7482-07<br>3.7482-07<br>3.7482-07<br>3.7482-07<br>3.7482-07<br>3.7482-07<br>3.7482-07<br>3.7482-07<br>3.7482-07<br>3.7482-07<br>3.7482-07<br>3.7482-07<br>3.7482-07<br>3.7482-07<br>3.7482-07<br>3.7482-07<br>3.7482-07<br>3.7482-07<br>3.7482-07<br>3.7482-07<br>3.7482-07<br>3.7482-07<br>3.7482-07<br>3.7482-07<br>3.7482-07<br>3.7482-07<br>3.7482-07<br>3.7482-07<br>3.7482-07<br>3.7482-07<br>3.7482-07<br>3.7482-07<br>3.7482-07<br>3.7482-07<br>3.7482-07<br>3.7482-07<br>3.7482-07<br>3.7482-07<br>3.7482-07<br>3.7482-07<br>3.7482-07<br>3.7482-07<br>3.7482-07<br>3.7482-07<br>3.7482-07<br>3.7482-07<br>3.7482-07<br>3.7482-07<br>3.7482-07<br>3.7482-07<br>3.7482-07<br>3.7482-07<br>3.7482-07<br>3.7482-07<br>3.7482-07<br>3.7482-07<br>3.7482-07<br>3.7482-07<br>3.7482-07<br>3.7482-07<br>3.7482-07<br>3.7482-07<br>3.7482-07<br>3.7482-07<br>3.7482-07<br>3.7482-07<br>3.7482-07<br>3.7482-07<br>3.7482-07<br>3.7482-07<br>3.7482-07<br>3.7482-07<br>3.7482-07<br>3.7482-07<br>3.7482-07<br>3.7482-07<br>3.7482-07<br>3.7482-07<br>3.7482-07<br>3.7482-07<br>3.7482-07<br>3.7482-07<br>3.7482-07<br>3.7482-07<br>3.7482-07<br>3.7482-07<br>3.7482-07<br>3.7482-07<br>3.7482-07<br>3.7482-07<br>3.7482-07<br>3.7482-07<br>3.7482-07<br>3.7482-07<br>3.7482-07<br>3.7482-07<br>3.7482-07<br>3.7482-07<br>3.7482-07<br>3.7482-07<br>3.7482-07<br>3.7482-07<br>3.7482-07<br>3.7482-07<br>3.7482-07<br>3.7482-07<br>3.7482-07<br>3.7482-07<br>3.7482-07<br>3.7482-07<br>3.7482-07<br>3.7482-07<br>3.7482-07<br>3.7482-07<br>3.7482-07<br>3.7482-07<br>3.7482-07<br>3.748   | 0.0031876<br>0.0022001<br>0.0024433<br>0.00234433<br>0.0023457<br>0.002367<br>0.002367<br>0.002865<br>0.0027711<br>0.002865<br>0.00254144<br>0.0025867<br>0.002845<br>0.002464<br>0.0015867<br>0.002464<br>0.0015867<br>0.0022716<br>0.0022716<br>0.002245<br>1.777E-05<br>0<br>0.0005245<br>1.777E-05<br>0.0005837<br>0.0006835<br>0.0006835<br>0.0006835<br>0.0006835<br>0.0006835<br>0.0006837<br>0.0006837<br>0.0006837  |
2.3644564<br>1.8937445<br>2.1225909<br>2.0177733<br>2.1529112<br>2.928047<br>2.8544098<br>2.6943024<br>2.6943024<br>2.6943024<br>2.6943024<br>2.6943024<br>2.7453139<br>2.4915395<br>2.724646<br>3.8925400<br>0<br>0<br>0<br>1.5831454<br>1.583567<br>0.0439825<br>0.6840221<br>0.5800351<br>0.68892151<br>0.68892151<br>0.68892151<br>0.68879172<br>0.98623154<br>0.9623154<br>0.9623154<br>0.9623154<br>0.9623154<br>0.9623154<br>0.96203751<br>0.96203751<br>0.96203751<br>0.96203751<br>0.96203751<br>0.96203751<br>0.96203751<br>0.96203751<br>0.96203751<br>0.96203751<br>0.96203751<br>0.96203751<br>0.96203751<br>0.96203751<br>0.96203751<br>0.96203751<br>0.96203751<br>0.9620374<br>0.96203751<br>0.96203751<br>0.96203751<br>0.96203751<br>0.96203751<br>0.96203751<br>0.96203751<br>0.96203751<br>0.96203751<br>0.96203751<br>0.96203751<br>0.96203751<br>0.96203751<br>0.96203751<br>0.96203751<br>0.96203751<br>0.96203751<br>0.96203751<br>0.96203751<br>0.96203751<br>0.96203751<br>0.96203751<br>0.96203751<br>0.96203751<br>0.96203751<br>0.96203751<br>0.96203751<br>0.96203751<br>0.96203751<br>0.96203751<br>0.96203751<br>0.96203751<br>0.96203751<br>0.96203751<br>0.96203751<br>0.96203751<br>0.96203751<br>0.96203751<br>0.96203751<br>0.96203751<br>0.96203751<br>0.96203751<br>0.96203751<br>0.96203751<br>0.96203751<br>0.96203751<br>0.96203751<br>0.96203751<br>0.96203751<br>0.96203751<br>0.96203751<br>0.96203751<br>0.96203751<br>0.96203751<br>0.96203751<br>0.96203751<br>0.96203751<br>0.96203751<br>0.96203751<br>0.96203751<br>0.96203751<br>0.96203751<br>0.96203751<br>0.96203751<br>0.96203751<br>0.96203751<br>0.96203751<br>0.96203751<br>0.96203751<br>0.96203751<br>0.96203751<br>0.96203751<br>0.96203751<br>0.96203751<br>0.96203751<br>0.96203751<br>0.96203751<br>0.96203751<br>0.96203751<br>0.96203751<br>0.96203751<br>0.96203751<br>0.96203751<br>0.96203751<br>0.96203751<br>0.96203751<br>0.96203751<br>0.96203751<br>0.96203751<br>0.96203751<br>0.96203751<br>0.96203751<br>0.96203751<br>0.96203751<br>0.96203751<br>0.96203751<br>0.96203751<br>0.96203751<br>0.96203751<br>0.96203751<br>0.96203751<br>0.96503751<br>0.96503751<br>0.96503751<br>0.96503751<br>0.96503751<br>0.96503751<br>0.96503751<br>0.96503751<br>0.96503751<br>0.96503751<br>0.96503751<br>0.96503751<br>0.96503751<br>0.96503751<br>0.96503751<br>0.96503751<br>0.96503751<br>0.96503751<br>0.96503751<br>0.96503751<br>0.96503751<br>0.96503751<br>0.96503751<br>0.96503751<br>0.96503751<br>0.96503751<br>0.96503751<br>0.96503751<br>0.96503751<br>0.96503751<br>0.96503751<br>0.96503751<br>0.96503751<br>0.96503751<br>0.96503751<br>0.96505751<br>0.96505751<br>0.96505751<br>0.  | 1.7688-00 0.002<br>1.3681-00 0.002<br>1.381-00 0.002<br>1.381-00 0.002<br>1.382-00 0.002<br>1.525-00 0.002<br>1.525-00 0.002<br>1.525-00 0.002<br>1.535-00 0.002<br>1.5387-00 0.002<br>1.5387-00 0.002<br>1.318-00 0.002<br>1.328-00 0   | 873         0.04           8773         0.03           9773         0.03           1015         0.04           1016         0.05           1017         0.02           1018         0.02           1019         0.05           1010         0.03           1010         0.03           1011         0.04           1012         0.04           1013 <td>447043         3.           447043         3.           828322         2.           408499         2.           408499         2.           408499         2.           433125         2.           338041         2.           338041         2.           338041         2.           338041         2.           196214         2.           1967677         2.           186767         2.           181831         1.           1677764         2.           153993         8.1           0         0           0         0           104252         6.4           153993         8.1           108061         6.3           110642         9.5           110642         9.5           110642         9.5           110642         9.5           110642         9.5           110642         9.5           110642         9.5           110642         9.5           110642         9.5           110642         9.5           110</td> <td>5.18E-008         5           6.619E-008         6           6.619E-008         6           7.76E-088         4           2.990E-08         3           8.74E-068         4           9.071E-08         4           9.072E-08         4           9.072E-08         4           9.08E-00         1           9.08E-00         1           9.08E-00         1           9.08E-009         1           9.04E-009         1           9.04E-009         1           9.04E-009         1           9.04E-009         1           9.04E-009</td> <td>3.998E-05<br/>4.254E-05<br/>4.254E-05<br/>4.214E-05<br/>4.3174E-05<br/>3.736E-05<br/>4.377E-05<br/>4.307E-05<br/>4.307E-05<br/>4.307E-05<br/>4.307E-05<br/>4.307E-05<br/>4.059E-05<br/>4.22E-05<br/>4.22E-05<br/>4.242E-05<br/>4.242E-05<br/>4.242E-05<br/>4.242E-05<br/>4.242E-05<br/>4.242E-05<br/>1.18E-05<br/>1.18E-05<br/>1.18E-05<br/>1.18E-05<br/>1.18E-05<br/>1.18E-05<br/>1.18E-05<br/>1.18E-05<br/>1.18E-05<br/>1.18E-05<br/>1.18E-05<br/>1.158E-05<br/>1.536E-05<br/>1.536E-05<br/>1.536E-05<br/>1.536E-05</td> <td>0.0443322<br/>0.0443322<br/>0.044332<br/>0.0470156<br/>0.0584373<br/>0.05584373<br/>0.0558211<br/>0.0538871<br/>0.0538871<br/>0.0538871<br/>0.0538871<br/>0.0527943<br/>0.0527943<br/>0.0527943<br/>0.052794<br/>0.048222<br/>0.0607794<br/>0.04822<br/>0.04822<br/>0.0462<br/>0.0462<br/>0.0462<br/>0.0462<br/>0.0462<br/>0.0462<br/>0.0462<br/>0.0462<br/>0.0226166<br/>0.0462<br/>0.0226166<br/>0.0462<br/>0.0226166<br/>0.0462<br/>0.0226166<br/>0.0462<br/>0.0226166<br/>0.0462<br/>0.0462<br/>0.0462<br/>0.0462<br/>0.0462<br/>0.0462<br/>0.0462<br/>0.0462<br/>0.0462<br/>0.0462<br/>0.0462<br/>0.0462<br/>0.0462<br/>0.0462<br/>0.0462<br/>0.0462<br/>0.0462<br/>0.0462<br/>0.0462<br/>0.0462<br/>0.0462<br/>0.0462<br/>0.0462<br/>0.0462<br/>0.0462<br/>0.0462<br/>0.0462<br/>0.0462<br/>0.0462<br/>0.0462<br/>0.0462<br/>0.0462<br/>0.0462<br/>0.0462<br/>0.0462<br/>0.0462<br/>0.0462<br/>0.0462<br/>0.0462<br/>0.0462<br/>0.0462<br/>0.0462<br/>0.0462<br/>0.0462<br/>0.0462<br/>0.0462<br/>0.0462<br/>0.0462<br/>0.0462<br/>0.0462<br/>0.0462<br/>0.0462<br/>0.0462<br/>0.0462<br/>0.0462<br/>0.0462<br/>0.0462<br/>0.0462<br/>0.0462<br/>0.0462<br/>0.0462<br/>0.0462<br/>0.0462<br/>0.0462<br/>0.0462<br/>0.0462<br/>0.0462<br/>0.0462<br/>0.0462<br/>0.0462<br/>0.0462<br/>0.0462<br/>0.0462<br/>0.0462<br/>0.0462<br/>0.0462<br/>0.0462<br/>0.0462<br/>0.0462<br/>0.0462<br/>0.0462<br/>0.0462<br/>0.0462<br/>0.0462<br/>0.0462<br/>0.0462<br/>0.0462<br/>0.0462<br/>0.0462<br/>0.0462<br/>0.0462<br/>0.0462<br/>0.0462<br/>0.0462<br/>0.0462<br/>0.0462<br/>0.0462<br/>0.0462<br/>0.0462<br/>0.0462<br/>0.0462<br/>0.0462<br/>0.0462<br/>0.0462<br/>0.0462<br/>0.0462<br/>0.0462<br/>0.0462<br/>0.0462<br/>0.0462<br/>0.0462<br/>0.0462<br/>0.0462<br/>0.0462<br/>0.0462<br/>0.0462<br/>0.0462<br/>0.0462<br/>0.0462<br/>0.0462<br/>0.0462<br/>0.0462<br/>0.0462<br/>0.0462<br/>0.0462<br/>0.0462<br/>0.0462<br/>0.0462<br/>0.0462<br/>0.0462<br/>0.0462<br/>0.0462<br/>0.0462<br/>0.0462<br/>0.0462<br/>0.0462<br/>0.0462<br/>0.0462<br/>0.0462<br/>0.0462<br/>0.0462<br/>0.0462<br/>0.0462<br/>0.0462<br/>0.0462<br/>0.0462<br/>0.0462<br/>0.0462<br/>0.0462<br/>0.0462<br/>0.0462<br/>0.0462<br/>0.0462<br/>0.0462<br/>0.0462<br/>0.0462<br/>0.0462<br/>0.0462<br/>0.0462<br/>0.0462<br/>0.0462<br/>0.0462<br/>0.0462<br/>0.0462<br/>0.0462<br/>0.0462<br/>0.0462<br/>0.0462<br/>0.0462<br/>0.0462<br/>0.0462<br/>0.0462<br/>0.0462<br/>0.0462<br/>0.0462<br/>0.0462<br/>0.0462<br/>0.0462<br/>0.0462<br/>0.0462<br/>0.0462<br/>0.0462<br/>0.0462<br/>0.0462<br/>0.0462<br/>0.0462<br/>0.0462<br/>0.0462<br/>0.0462<br/>0.0462<br/>0.0462<br/>0.0462<br/>0.0462<br/>0.0462<br/>0.0462<br/>0.0462<br/>0.0462<br/>0.0462<br/>0.0462<br/>0.0462<br/>0.0462<br/>0.0462<br/>0.0462<br/>0.0462<br/>0.0462<br/>0.0462<br/>0.0462<br/>0.0462<br/>0.0462<br/>0.0462<br/>0.0462<br/>0.0462<br/>0.0462<br/>0.0462<br/>0.0462<br/>0.0462<br/>0.0462<br/>0.0462<br/>0.0462<br/>0.0462<br/>0.0462<br/>0.0462<br/>0.0462<br/>0.0462<br/>0.0462<br/>0.0462<br/>0.0462<br/>0.0462<br/>0.0462<br/>0.0462<br/>0.0462<br/>0.0462<br/>0.0462<br/>0.0462<br/>0.0462<br/>0.0462<br/>0.0462<br/>0.0462<br/>0.0462<br/>0.0462<br/>0.0462<br/>0.0462<br/>0.0462<br/>0.0462<br/>0.0462<br/>0.0462<br/>0.0462</td> <td>2.8422-08         2           8.822-08         2           8.822-08         2           8.822-08         2           8.912-08         2           8.312-08         2           8.312-08         2           8.312-08         2           8.312-08         2           8.251-08         2           8.271-08         2           8.312-08         2           8.004-08         2           7.013-08         2           8.004-08         2           7.014-09         2           9.014E-09         2           7.014E-09         2           8.8775E-10         0           0         0    
      7.014E-09         1           8.8775E-09         1           5.867E-09         1           1.04E-08         1           1.04E-08         1           1.04E-08         1           1.104E-08         1           3.119E-09         1           3.119E-09         1           3.119E-09         1           3.119E-09         1           3.119E-09         1<td>1.584E-05<br/>1.897E-05<br/>1.077E-05<br/>1.077E-05<br/>1.077E-05<br/>1.377E-05<br/>1.377E-05<br/>1.307E-05<br/>1.307E-05<br/>1.307E-05<br/>1.307E-05<br/>1.307E-05<br/>1.889E-05<br/>1.636E-05<br/>1.636E-05<br/>1.604E-06<br/>1.252E-05<br/>1.255E-05<br/>1.901E-05<br/>1.901E-05<br/>1.901E-05<br/>1.901E-05<br/>1.901E-05<br/>1.901E-05<br/>1.901E-05<br/>1.901E-05<br/>1.901E-05<br/>1.901E-05<br/>1.901E-05<br/>1.667E-05<br/>1.667E-05<br/>1.667E-05<br/>1.667E-05</td></td> | 447043         3.           447043         3.           828322         2.           408499         2.           408499         2.           408499         2.           433125         2.           338041         2.           338041         2.           338041         2.           338041         2.           196214         2.           1967677         2.           186767         2.           181831         1.           1677764         2.           153993         8.1           0         0           0         0           104252         6.4           153993         8.1           108061         6.3           110642         9.5           110642         9.5           110642         9.5           110642         9.5           110642         9.5           110642         9.5           110642         9.5           110642         9.5           110642         9.5           110642         9.5           110   
  | 5.18E-008         5           6.619E-008         6           6.619E-008         6           7.76E-088         4           2.990E-08         3           8.74E-068         4           9.071E-08         4           9.072E-08         4           9.072E-08         4           9.08E-00         1           9.08E-00         1           9.08E-00         1           9.08E-009         1           9.04E-009         1           9.04E-009         1           9.04E-009         1           9.04E-009         1           9.04E-009 | 3.998E-05<br>4.254E-05<br>4.254E-05<br>4.214E-05<br>4.3174E-05<br>3.736E-05<br>4.377E-05<br>4.307E-05<br>4.307E-05<br>4.307E-05<br>4.307E-05<br>4.307E-05<br>4.059E-05<br>4.22E-05<br>4.22E-05<br>4.242E-05<br>4.242E-05<br>4.242E-05<br>4.242E-05<br>4.242E-05<br>4.242E-05<br>1.18E-05<br>1.18E-05<br>1.18E-05<br>1.18E-05<br>1.18E-05<br>1.18E-05<br>1.18E-05<br>1.18E-05<br>1.18E-05<br>1.18E-05<br>1.18E-05<br>1.158E-05<br>1.536E-05<br>1.536E-05<br>1.536E-05<br>1.536E-05  |
0.0443322<br>0.0443322<br>0.044332<br>0.0470156<br>0.0584373<br>0.05584373<br>0.0558211<br>0.0538871<br>0.0538871<br>0.0538871<br>0.0538871<br>0.0527943<br>0.0527943<br>0.0527943<br>0.052794<br>0.048222<br>0.0607794<br>0.04822<br>0.04822<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0226166<br>0.0462<br>0.0226166<br>0.0462<br>0.0226166<br>0.0462<br>0.0226166<br>0.0462<br>0.0226166<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462<br>0.0462 | 2.8422-08         2           8.822-08         2           8.822-08         2           8.822-08         2           8.912-08         2           8.312-08         2           8.312-08         2           8.312-08         2           8.312-08         2           8.251-08         2           8.271-08         2           8.312-08         2           8.004-08         2           7.013-08         2           8.004-08         2           7.014-09         2           9.014E-09         2           7.014E-09         2           8.8775E-10         0           0         0           7.014E-09         1           8.8775E-09         1           5.867E-09         1           1.04E-08         1           1.04E-08         1           1.04E-08         1           1.104E-08         1           3.119E-09         1           3.119E-09         1           3.119E-09         1           3.119E-09         1           3.119E-09         1 <td>1.584E-05<br/>1.897E-05<br/>1.077E-05<br/>1.077E-05<br/>1.077E-05<br/>1.377E-05<br/>1.377E-05<br/>1.307E-05<br/>1.307E-05<br/>1.307E-05<br/>1.307E-05<br/>1.307E-05<br/>1.889E-05<br/>1.636E-05<br/>1.636E-05<br/>1.604E-06<br/>1.252E-05<br/>1.255E-05<br/>1.901E-05<br/>1.901E-05<br/>1.901E-05<br/>1.901E-05<br/>1.901E-05<br/>1.901E-05<br/>1.901E-05<br/>1.901E-05<br/>1.901E-05<br/>1.901E-05<br/>1.901E-05<br/>1.667E-05<br/>1.667E-05<br/>1.667E-05<br/>1.667E-05</td>   | 1.584E-05<br>1.897E-05<br>1.077E-05<br>1.077E-05<br>1.077E-05<br>1.377E-05<br>1.377E-05<br>1.307E-05<br>1.307E-05<br>1.307E-05<br>1.307E-05<br>1.307E-05<br>1.889E-05<br>1.636E-05<br>1.636E-05<br>1.604E-06<br>1.252E-05<br>1.255E-05<br>1.901E-05<br>1.901E-05<br>1.901E-05<br>1.901E-05<br>1.901E-05<br>1.901E-05<br>1.901E-05<br>1.901E-05<br>1.901E-05<br>1.901E-05<br>1.901E-05<br>1.667E-05<br>1.667E-05<br>1.667E-05<br>1.667E-05  |
| Sheung Tsuen B/T and Car Park<br>Sheung Tsuen B/T and Car Park<br>Village Access Road<br>Village Access Road | RD_29<br>RD_29<br>RD_29<br>RD_29<br>RD_29<br>RD_29<br>RD_29<br>RD_29<br>RD_29<br>RD_29<br>RD_29<br>RD_29<br>RD_29<br>RD_29<br>RD_29<br>RD_29<br>RD_29<br>RD_29<br>RD_29<br>RD_29<br>RD_29<br>RD_29<br>RD_29<br>RD_29<br>RD_29<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20<br>RD_20 | 9900-1000 1100-1200 1200-1200 1200-1200 1300-400 1400-1500 1500-1600 1500-1600 1500-1800 1800-2000 2000-2100 2000-2100 2000-2100 2000-2100 2000-2100 2000-2100 2000-200 2000-20   | 0.1281945<br>0.1457065<br>0.1607293<br>0.1557266<br>0.1459828<br>0.1452452<br>0.1452452<br>0.1452452<br>0.1332075<br>0.1302075<br>0.1302075<br>0.1302075<br>0.1302082<br>0.0325292<br>0.030540<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0   | 9.342-003 9.342-003 9.342-003 9.342-003 14.552-03
14.552-03 14.552   | 0.0001335 (<br>0.0001335 (<br>0.0001345 (<br>0.0001417 (<br>1.000147 (<br>1.0001647 (<br>1.000164 (<br>1.000164 (<br>1.000165 (<br>1.000164 (<br>1.000165 (<br>1.000165 (<br>1.000165 (<br>1.000165 (<br>1.000165 (<br>1.000165 (<br>1.000165 (<br>1.000165 (<br>1.000165 (<br>1.000175 (<br>1.0001  | 1.1261(10)<br>3.1210641<br>3.1210641<br>3.1210641<br>3.1291377<br>3.1383667<br>3.1383667<br>3.1383667<br>3.139307<br>3.1461086<br>3.1461086<br>3.1461086<br>0.1462157<br>3.1267454<br>3.1267454<br>3.1267454<br>0.166957<br>0.1267454<br>0.166957<br>0.1267454<br>0.0052024<br>0.0052024<br>0.0052024<br>0.0052024<br>0.0052024<br>0.0052024<br>0.0052024<br>0.0052024<br>0.0052024<br>0.0052024<br>0.0052024<br>0.0052024<br>0.0052024<br>0.0052024<br>0.0052024<br>0.0052024<br>0.0052024<br>0.0052024<br>0.0052024<br>0.0052024<br>0.0052024<br>0.0052024<br>0.0052024<br>0.0052024<br>0.0052024<br>0.0052024<br>0.0052024<br>0.0052024<br>0.0052024<br>0.0052024<br>0.0052024<br>0.0052024<br>0.0052024<br>0.0052024<br>0.0052024<br>0.0052024<br>0.0052024<br>0.0052024<br>0.0052024<br>0.0052024<br>0.0052024<br>0.0052024<br>0.0052024<br>0.0052024<br>0.0052024<br>0.0052024<br>0.0052024<br>0.0052024<br>0.0052024<br>0.0052024<br>0.0052024<br>0.0052024<br>0.0052024<br>0.0052024<br>0.0052024<br>0.0052024<br>0.0052024<br>0.0052024<br>0.0052024<br>0.0052024<br>0.0052024<br>0.0052024<br>0.0052024<br>0.0052024<br>0.0052024<br>0.0052024<br>0.0052024<br>0.0052024<br>0.0052024<br>0.0052024<br>0.0052024<br>0.0052024<br>0.0052024<br>0.0052024<br>0.0052024<br>0.0052024<br>0.0052024<br>0.0052024<br>0.0052024<br>0.0052024<br>0.0052024<br>0.0052024<br>0.0052024<br>0.0052024<br>0.0052024<br>0.0052024<br>0.0052024<br>0.0052024<br>0.0052024<br>0.0052024<br>0.0052024<br>0.0052024<br>0.0052024<br>0.0052024<br>0.0052024<br>0.0052024<br>0.0052024<br>0.0052024<br>0.0052024<br>0.0052024<br>0.0052024<br>0.0052024<br>0.0052024<br>0.0052024<br>0.0052024<br>0.0052024<br>0.0052024<br>0.0052024<br>0.0052024<br>0.0052024<br>0.0052024<br>0.0052024<br>0.0052024<br>0.0052024<br>0.0052024<br>0.0052024<br>0.0052024<br>0.0052024<br>0.0052024<br>0.0052024<br>0.0052024<br>0.0052024<br>0.0052024<br>0.0052024<br>0.0052024<br>0.0052024<br>0.0052024<br>0.0052024<br>0.0052024<br>0.0052024<br>0.0052024<br>0.0052024<br>0.0052024<br>0.0052024<br>0.0052024<br>0.0052024<br>0.0052024<br>0.0052024<br>0.0052024<br>0.0052024<br>0.0052024<br>0.0052024<br>0.0052024<br>0.0052024<br>0.0052024<br>0.0052024<br>0.0052024<br>0.0052024<br>0.0052024<br>0.0052024<br>0.0052024<br>0.0052024<br>0.0052024<br>0.0052024<br>0.0052024<br>0.0052024<br>0.0052024<br>0.0052024<br>0.0052024<br>0.0052024<br>0.0052024<br>0.0052024<br>0.0052024<br>0.0052024<br>0.0052024<br>0.0052024<br>0.0052024<br>0.0052024<br>0.0052024<br>0.0052024<br>0.0052024<br>0.0052024<br>0.0052024<br>0.0052024<br>0.0052024<br>0.0052024<br>0.0052024<br>0.0052024<br>0.0052024<br>0.0052024<br>0.00520240000000000000000000000000000000 | 9.4311-08<br>8.8947-08<br>8.8947-08<br>9.7547-08<br>9.7547-08<br>9.7547-08<br>9.7747-08<br>9.7747-08<br>9.7747-08<br>9.7747-08<br>9.7747-08<br>9.7747-08<br>9.7747-08<br>9.7745-08<br>9.7745-08<br>9.7745-08<br>9.7745-08<br>9.7745-08<br>9.7555-08<br>9.7555-08<br>9.7555-08<br>9.7555-08<br>9.7555-08<br>9.7555-08<br>9.7555-08<br>9.7555-08<br>9.7555-08<br>9.7555-08<br>9.7555-08<br>9.7555-08<br>9.7555-08<br>9.7555-08<br>9.7555-08<br>9.7555-08<br>9.7555-08<br>9.7555-08<br>9.7555-08<br>9.7555-08<br>9.7555-08<br>9.7555-08<br>9.7555-08<br>9.7555-08<br>9.7555-08<br>9.7555-08<br>9.7555-08<br>9.7555-08<br>9.7555-08<br>9.7555-08<br>9.7555-08<br>9.7555-08<br>9.7555-08<br>9.7555-08<br>9.7555-08<br>9.7555-08<br>9.7555-08<br>9.7555-08<br>9.7555-08<br>9.7555-08<br>9.7555-08<br>9.7555-08<br>9.7555-08<br>9.7555-08<br>9.7555-08<br>9.7555-08<br>9.7555-08<br>9.7555-08<br>9.7555-08<br>9.7555-08<br>9.7555-08<br>9.7555-08<br>9.7555-08<br>9.7555-08<br>9.7555-08<br>9.7555-08<br>9.7555-08<br>9.7555-08<br>9.7555-08<br>9.7555-08<br>9.7555-08<br>9.7555-08<br>9.7555-08<br>9.7555-08<br>9.7555-08<br>9.7555-08<br>9.7555-08<br>9.7555-08<br>9.7555-08<br>9.7555-08<br>9.7555-08<br>9.7555-08<br>9.7555-08<br>9.7555-08<br>9.7555-08<br>9.7555-08<br>9.7555-08<br>9.7555-08<br>9.7555-08<br>9.7555-08<br>9.7555-08<br>9.7555-08<br>9.7555-08<br>9.7555-08<br>9.7555-08<br>9.7555-08<br>9.7555-08<br>9.7555-08<br>9.7555-08<br>9.7555-08<br>9.7555-08<br>9.7555-08<br>9.7555-08<br>9.7555-08<br>9.7555-08<br>9.7555-08<br>9.7555-08<br>9.7555-08<br>9.7555-08<br>9.7555-08<br>9.7555-08<br>9.7555-08<br>9.7555-08<br>9.7555-08<br>9.7555-08<br>9.7555-08<br>9.7555-08<br>9.7555-08<br>9.7555-08<br>9.7555-08<br>9.7555-08<br>9.7555-08<br>9.7555-08<br>9.7555-08<br>9.7555-08<br>9.7555-08<br>9.7555-08<br>9.7555-08<br>9.7555-08<br>9.7555-08<br>9.7555-08<br>9.7555-08<br>9.7555-08<br>9.7555-08<br>9.7555-08<br>9.7555-08<br>9.7555-08<br>9.7555-08<br>9.7555-08<br>9.7555-08<br>9.7555-08<br>9.7555-08<br>9.7555-08<br>9.7555-08<br>9.7555-08<br>9.7555-08<br>9.7555-08<br>9.7555-08<br>9.7555-08<br>9.7555-08<br>9.7555-08<br>9.7555-08<br>9.7555-08<br>9.7555-08<br>9.7555-08<br>9.7555-08<br>9.7555-08<br>9.7555-08<br>9.7555-08<br>9.7555-08<br>9.7555-08<br>9.7555-08<br>9.7555-08<br>9.7555-08<br>9.7555-08<br>9.7555-08<br>9.7555-08<br>9.7555-08<br>9.7555-08<br>9.7555-08<br>9.7555-08<br>9.7555-08<br>9.7555-08<br>9.7555-08<br>9.7555-08<br>9.7555-08<br>9.7555-08<br>9.7555-08<br>9.7555-08<br>9.7555-08<br>9.7555-08<br>9.7555-08<br>9.7555-08<br>9.7555-08<br>9.7555-08<br>9.7555-08<br>9.7555-08<br>9.7555-08<br>9.7555-08<br>9.7555-08<br>9.7555-08<br>9.7555-08<br>9.7555-08<br>9.7555-08<br>9.7555-08<br>9.7555-08<br>9.7555                   | 0.0001332<br>0.0001252<br>0.0001253<br>0.0001253<br>0.0001253<br>0.0001355<br>0.0001055<br>0.0001055<br>0.0001152<br>0.000125<br>0.0001152<br>0.0001152<br>0.0001152<br>0.0001152<br>0.0001152<br>0.0001152<br>0.0001152<br>0.0001152<br>0.0001152<br>0.0001152<br>0.0001152<br>0.0001152<br>0.0001152<br>0.0001152<br>0.0001152<br>0.0001152<br>0.0001152<br>0.0001152<br>0.0001152<br>0.0001152<br>0.0001152<br>0.0001152<br>0.0001152<br>0.0001152<br>0.0001152<br>0.0001152<br>0.0001152<br>0.0001152<br>0.0001152<br>0.0001152<br>0.0001152<br>0.0001152<br>0.0001152<br>0.0001152<br>0.0001152<br>0.0001152<br>0.0001152<br>0.0001152<br>0.0001152<br>0.0001152<br>0.0001152<br>0.0001152<br>0.0001152<br>0.0001152<br>0.0001152<br>0.0001152<br>0.0001152<br>0.0001152<br>0.0001152<br>0.0001152<br>0.0001152<br>0.0001152<br>0.0001152<br>0.0001152<br>0.0001152<br>0.0001152<br>0.0001152<br>0.0001152<br>0.0001152<br>0.0001152<br>0.0001152<br>0.0001152<br>0.0001152<br>0.0001152<br>0.0001152<br>0.0001152<br>0.0001152<br>0.0001152<br>0.0001152<br>0.0001152<br>0.0001152<br>0.0001152<br>0.0001152<br>0.0001152<br>0.0001152<br>0.0001152<br>0.0001152<br>0.0001152<br>0.0001152<br>0.0001152<br>0.0001152<br>0.0001152<br>0.0001152<br>0.0001152<br>0.0001152<br>0.0001152<br>0.0001152<br>0.0001152<br>0.0001152<br>0.0001152<br>0.0001152<br>0.0001152<br>0.0001152<br>0.0001152<br>0.0001152<br>0.0001152<br>0.0001152<br>0.0001152<br>0.0001152<br>0.0001152<br>0.0001152<br>0.0001152<br>0.0001152<br>0.0001152<br>0.0001152<br>0.0001152<br>0.0001152<br>0.0001152<br>0.0001152<br>0.0001152<br>0.0001152<br>0.0001152<br>0.0001152<br>0.0001152<br>0.0001152<br>0.0001152<br>0.0001152<br>0.0001152<br>0.0001152<br>0.0001152<br>0.0001152<br>0.0001152<br>0.0001152<br>0.0001152<br>0.0001152<br>0.0001152<br>0.0001152<br>0.0001152<br>0.0001152<br>0.0001152<br>0.0001152<br>0.0001152<br>0.0001152<br>0.0001152<br>0.0001152<br>0.0001152<br>0.0001152<br>0.0001152<br>0.0001152<br>0.0001152<br>0.0001152<br>0.0001152<br>0.0001152<br>0.0001152<br>0.0001152<br>0.0001152<br>0.0001152<br>0.0001152<br>0.0001152<br>0.0001152<br>0.0001152<br>0.0001152<br>0.0001152<br>0.0001152<br>0.0001152<br>0.0001152<br>0.000152<br>0.000152<br>0.000152<br>0.000152<br>0.000152<br>0.000152<br>0.000152<br>0.000152<br>0.000152<br>0.000152<br>0.000152<br>0.000152<br>0.000152<br>0.000152<br>0.000152<br>0.000152<br>0.000152<br>0.000152<br>0.000152<br>0.000152<br>0.000152<br>0.000152<br>0.000152<br>0.000152<br>0.000152<br>0.000152<br>0.000152<br>0.000152<br>0.000152<br>0.000152<br>0.000152<br>0.000152<br>0.000152<br>0.000152<br>0.000152<br>0.000152<br>0.000152<br>0.000152<br>0.000152<br>0.000152<br>0.000152<br>0.000152<br>0.000152 |
0.121356<br>0.1142306<br>0.1142306<br>0.1142306<br>0.1142307<br>0.114237<br>0.114237<br>0.114237<br>0.114237<br>0.112376<br>0.112376<br>0.112376<br>0.112376<br>0.112376<br>0.112376<br>0.112376<br>0.112376<br>0.112376<br>0.112376<br>0.112376<br>0.112376<br>0.112376<br>0.112376<br>0.112376<br>0.112376<br>0.112376<br>0.112376<br>0.112376<br>0.112376<br>0.112376<br>0.112376<br>0.112376<br>0.112376<br>0.112376<br>0.112376<br>0.112376<br>0.112376<br>0.112376<br>0.112376<br>0.112376<br>0.112376<br>0.112376<br>0.112376<br>0.112376<br>0.112376<br>0.112376<br>0.112376<br>0.112376<br>0.112376<br>0.112376<br>0.112376<br>0.112376<br>0.112376<br>0.112376<br>0.112376<br>0.112376<br>0.112376<br>0.112376<br>0.112376<br>0.112376<br>0.112376<br>0.112376<br>0.112376<br>0.112376<br>0.112376<br>0.112376<br>0.112376<br>0.112376<br>0.112376<br>0.112376<br>0.112376<br>0.112376<br>0.112376<br>0.112376<br>0.112376<br>0.112376<br>0.112376<br>0.112376<br>0.112376<br>0.112376<br>0.112376<br>0.112376<br>0.112376<br>0.112376<br>0.112376<br>0.112376<br>0.112376<br>0.112376<br>0.112376<br>0.112376<br>0.112376<br>0.112376<br>0.112376<br>0.112376<br>0.112376<br>0.112376<br>0.112376<br>0.112376<br>0.112376<br>0.112376<br>0.112376<br>0.112376<br>0.112376<br>0.112376<br>0.112376<br>0.112376<br>0.112376<br>0.112376<br>0.112376<br>0.112376<br>0.112376<br>0.112376<br>0.112376<br>0.112376<br>0.112376<br>0.112376<br>0.112376<br>0.112376<br>0.112376<br>0.112376<br>0.112376<br>0.112376<br>0.112376<br>0.112376<br>0.112376<br>0.112376<br>0.112376<br>0.112376<br>0.112376<br>0.112376<br>0.112376<br>0.112376<br>0.112376<br>0.112376<br>0.112376<br>0.112376<br>0.112376<br>0.112376<br>0.112376<br>0.112376<br>0.112376<br>0.112376<br>0.112376<br>0.112376<br>0.112376<br>0.112376<br>0.112376<br>0.112376<br>0.112376<br>0.112376<br>0.112376<br>0.112376<br>0.112376<br>0.112376<br>0.112376<br>0.112376<br>0.112376<br>0.112376<br>0.112376<br>0.112376<br>0.112376<br>0.112376<br>0.112376<br>0.112376<br>0.112376<br>0.112376<br>0.112376<br>0.112376<br>0.112376<br>0.112376<br>0.112376<br>0.112376<br>0.112376<br>0.112376<br>0.112376<br>0.112376<br>0.112376<br>0.112376<br>0.112376<br>0.112376<br>0.112376<br>0.112376<br>0.112376<br>0.112376<br>0.112376<br>0.112376<br>0.112376<br>0.112376<br>0.112376<br>0.112376<br>0.112376<br>0.112376<br>0.112376<br>0.112376<br>0.112376<br>0.112376<br>0.112376<br>0.112376<br>0.112376<br>0.112376<br>0.112376<br>0.112376<br>0.112376<br>0.112376<br>0.112376<br>0.112376<br>0.112376<br>0.112376<br>0.112376<br>0.112376<br>0.112376<br>0.112376<br>0.112376<br>0.112376<br>0.112376<br>0.112376<br>0.112376<br>0.112376<br>0.112376<br>0.112376<br>0.112376<br>0.112376<br>0.112376<br>0.112376<br>0.112376<br>0.112376<br>0.112376 | 9.001-08<br>9.001-08<br>9.146-08<br>9.146-08<br>10.531-08<br>9.146-08<br>10.531-08<br>9.146-08<br>10.521-08<br>9.146-08<br>10.521-08<br>10.521-08<br>10.521-08<br>10.521-08<br>10.521-08<br>10.521-08<br>10.521-08<br>10.521-08<br>10.521-08<br>10.521-08<br>10.521-08<br>10.521-08<br>10.521-08<br>10.521-08<br>10.521-08<br>10.521-08<br>10.521-08<br>10.521-08<br>10.521-08<br>10.521-08<br>10.521-08<br>10.521-08<br>10.521-08<br>10.521-08<br>10.521-08<br>10.521-08<br>10.521-08<br>10.521-08<br>10.521-08<br>10.521-08<br>10.521-08<br>10.521-08<br>10.521-08<br>10.521-08<br>10.521-08<br>10.521-08<br>10.521-08<br>10.521-08<br>10.521-08<br>10.521-08<br>10.521-08<br>10.521-08<br>10.521-08<br>10.521-08<br>10.521-08<br>10.521-08<br>10.521-08<br>10.521-08<br>10.521-08<br>10.521-08<br>10.521-08<br>10.521-08<br>10.521-08<br>10.521-08<br>10.521-08<br>10.521-08<br>10.521-08<br>10.521-08<br>10.521-08<br>10.521-08<br>10.521-08<br>10.521-08<br>10.521-08<br>10.521-08<br>10.521-08<br>10.521-08<br>10.521-08<br>10.521-08<br>10.521-08<br>10.521-08<br>10.521-08<br>10.521-08<br>10.521-08<br>10.521-08<br>10.521-08<br>10.521-08<br>10.521-08<br>10.521-08<br>10.521-08<br>10.521-08<br>10.521-08<br>10.521-08<br>10.521-08<br>10.521-08<br>10.521-08<br>10.521-08<br>10.521-08<br>10.521-08<br>10.521-08<br>10.521-08<br>10.521-08<br>10.521-08<br>10.521-08<br>10.521-08<br>10.521-08<br>10.521-08<br>10.521-08<br>10.521-08<br>10.521-08<br>10.521-08<br>10.521-08<br>10.521-08<br>10.521-08<br>10.521-08<br>10.521-08<br>10.521-08<br>10.521-08<br>10.521-08<br>10.521-08<br>10.521-08<br>10.521-08<br>10.521-08<br>10.521-08<br>10.521-08<br>10.521-08<br>10.521-08<br>10.521-08<br>10.521-08<br>10.521-08<br>10.521-08<br>10.521-08<br>10.521-08<br>10.521-08<br>10.521-08<br>10.521-08<br>10.521-08<br>10.521-08<br>10.521-08<br>10.521-08<br>10.521-08<br>10.521-08<br>10.521-08<br>10.521-08<br>10.521-08<br>10.521-08<br>10.521-08<br>10.521-08<br>10.521-08<br>10.521-08<br>10.521-08<br>10.521-08<br>10.521-08<br>10.521-08<br>10.521-08<br>10.521-08<br>10.521-08<br>10.521-08<br>10.521-08<br>10.521-08<br>10.521-08<br>10.521-08<br>10.521-08<br>10.521-08<br>10.521-08<br>10.521-08<br>10.521-08<br>10.521-08<br>10.521-08<br>10.521-08<br>10.521-08<br>10.521-08<br>10.521-08<br>10.521-08<br>10.521-08<br>10.521-08<br>10.521-08<br>10.521-08<br>10.521-08<br>10.521-08<br>10.521-08<br>10.521-08<br>10.521-08<br>10.521-08<br>10.521-08<br>10.521-08<br>10.521-08<br>10.521-  | 0.000142<br>0.000135<br>0.000135<br>0.000135<br>0.000125<br>0.000126<br>0.000125<br>0.000126<br>0.000125<br>0.000125<br>0.000125<br>0.000121<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.000125<br>0.0 |
2.490.485<br>1984285<br>2.700268<br>2.730653<br>31.057188<br>4.800262<br>2.836653<br>31.057188<br>4.800262<br>2.850553<br>31.057188<br>4.800262<br>2.820547<br>2.810563<br>4.650361<br>2.810254<br>4.650281<br>3.13150254<br>4.650281<br>3.13150254<br>4.650281<br>3.13150254<br>4.650281<br>3.13150254<br>4.650281<br>3.13150254<br>4.650281<br>3.13150254<br>4.650281<br>3.13150254<br>4.650281<br>3.13150254<br>4.650281<br>3.13150254<br>4.650281<br>3.13150254<br>4.650281<br>3.13150254<br>4.650281<br>3.13150254<br>4.650281<br>3.13150254<br>4.650281<br>3.13150254<br>4.650281<br>3.13150254<br>4.650281<br>3.13150254<br>4.650281<br>3.13150254<br>4.650281<br>3.13150254<br>4.650281<br>3.13150254<br>4.650281<br>3.13150254<br>4.650281<br>3.13150254<br>4.650281<br>3.13150254<br>4.650281<br>3.13150254<br>4.650281<br>3.13150254<br>4.650281<br>3.13150254<br>4.650281<br>3.13150254<br>4.650281<br>3.13150254<br>4.650281<br>3.13150254<br>4.650281<br>3.13150254<br>4.650281<br>3.13150254<br>4.650281<br>3.13150254<br>4.650281<br>3.13150254<br>4.650281<br>3.13150254<br>4.650281<br>3.13150254<br>4.650281<br>3.13150254<br>4.650281<br>3.13150254<br>4.650281<br>3.13150254<br>4.650281<br>3.13150254<br>4.650281<br>3.13150254<br>4.650281<br>3.13150254<br>4.650281<br>3.13150254<br>4.650281<br>3.13150254<br>4.650281<br>3.13150254<br>4.650281<br>3.13150254<br>4.650281<br>3.13150254<br>4.650281<br>3.13150254<br>4.650281<br>3.13150254<br>4.650281<br>3.13150254<br>4.650281<br>3.13150254<br>3.13150254<br>3.13150254<br>3.13150254<br>3.13150254<br>3.13150254<br>3.13150254<br>3.13150254<br>3.13150254<br>3.13150254<br>3.13150254<br>3.13150254<br>3.13150254<br>3.13150254<br>3.13150254<br>3.13150254<br>3.13150254<br>3.13150254<br>3.13150254<br>3.13150254<br>3.13150254<br>3.13150254<br>3.13150254<br>3.13150254<br>3.13150254<br>3.13150254<br>3.13150254<br>3.13150254<br>3.13150254<br>3.13150254<br>3.13150254<br>3.13150254<br>3.13150254<br>3.13150254<br>3.13150254<br>3.13150254<br>3.13150254<br>3.13150254<br>3.13150254<br>3.13150254<br>3.1315025555555555555555555555555555555555   | 1.2681-06<br>1.2722-06<br>1.412-06<br>1.412-06<br>1.4342-06<br>1.4342-06<br>1.4342-06<br>1.5282-06<br>1.5282-06<br>1.5282-06<br>1.5282-06<br>1.5282-06<br>1.5282-06<br>1.5282-06<br>1.5282-06<br>1.5282-06<br>1.5282-06<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0   | 0.003261<br>0.002266<br>0.0022915<br>0.0021953<br>0.0023793<br>0.0023293<br>0.002354<br>0.002554<br>0.002554<br>0.002554<br>0.002757<br>0.002757<br>0.0015163<br>0.002757<br>0.0015163<br>0.002757<br>0.0015163<br>0.002757<br>0.00152549<br>0.0023206<br>0.0023206<br>0.0003777<br>1.6816.05<br>0.0003777<br>1.6816.05<br>0.0003777<br>0.0003777<br>0.0003777<br>0.0003777<br>0.0003777<br>0.0003777<br>0.0003777<br>0.0003839<br>0.0004614<br>0.0004997<br>0.0004614<br>0.0004626<br>0.0004626<br>0.0004646   | 2.356(842)<br>2.356(842)<br>1.8789238<br>2.00977776<br>2.0055501<br>2.1283629<br>2.9552749<br>2.9002077<br>2.952749<br>2.900207<br>2.952749<br>2.900207<br>2.952749<br>2.900207<br>2.952749<br>2.950749<br>2.900207<br>2.950749<br>2.950749<br>2.950749<br>2.950749<br>2.950749<br>2.950749<br>2.950749<br>2.950749<br>2.950749<br>2.950749<br>2.950749<br>2.950749<br>2.950749<br>2.950749<br>2.950749<br>2.950749<br>2.950749<br>2.950749<br>2.950749<br>2.950749<br>2.950749<br>2.950749<br>2.950749<br>2.950749<br>2.950749<br>2.950749<br>2.950749<br>2.950749<br>2.950749<br>2.950749<br>2.950749<br>2.950749<br>2.950749<br>2.950749<br>2.950749<br>2.950749<br>2.950749<br>2.950749<br>2.950749<br>2.950749<br>2.950749<br>2.950749<br>2.950749<br>2.950749<br>2.950749<br>2.950749<br>2.950749<br>2.950749<br>2.950749<br>2.950749<br>2.950749<br>2.950749<br>2.950749<br>2.950749<br>2.950749<br>2.950749<br>2.950749<br>2.950749<br>2.950749<br>2.950749<br>2.950749<br>2.950749<br>2.950749<br>2.950749<br>2.950749<br>2.950749<br>2.950749<br>2.950749<br>2.950749<br>2.950749<br>2.950749<br>2.950749<br>2.950749<br>2.950749<br>2.950749<br>2.950749<br>2.950749<br>2.950749<br>2.950749<br>2.950749<br>2.950749<br>2.950749<br>2.950749<br>2.950749<br>2.950749<br>2.950749<br>2.950749<br>2.950749<br>2.950749<br>2.950749<br>2.950749<br>2.950749<br>2.950749<br>2.950749<br>2.950749<br>2.950749<br>2.950749<br>2.950749<br>2.950749<br>2.950749<br>2.950749<br>2.950749<br>2.950749<br>2.950749<br>2.950749<br>2.950749<br>2.950749<br>2.950749<br>2.950749<br>2.950749<br>2.950749<br>2.950749<br>2.950749<br>2.950749<br>2.950749<br>2.950749<br>2.950749<br>2.950749<br>2.950749<br>2.950749<br>2.950749<br>2.950749<br>2.950749<br>2.950749<br>2.950749<br>2.950749<br>2.950749<br>2.950749<br>2.950749<br>2.950749<br>2.950749<br>2.950749<br>2.950749<br>2.950749<br>2.950749<br>2.950749<br>2.950749<br>2.950749<br>2.950749<br>2.950749<br>2.950749<br>2.950749<br>2.950749<br>2.950749<br>2.950749<br>2.950749<br>2.950749<br>2.950749<br>2.950749<br>2.950749<br>2.950749<br>2.950749<br>2.950749<br>2.950749<br>2.950749<br>2.950749<br>2.950749<br>2.950749<br>2.950749<br>2.950749<br>2.950749<br>2.950749<br>2.950749<br>2.950749<br>2.950749<br>2.950749<br>2.950749<br>2.950749<br>2.950749<br>2.950749<br>2.950749<br>2.950749<br>2.950749<br>2.950749<br>2.950749<br>2.950749<br>2.950749<br>2.950749<br>2.950749<br>2.950749<br>2.950749<br>2.950749<br>2.950749<br>2.950749<br>2.950749<br>2.950749<br>2.950749<br>2.950749<br>2.950749<br>2.950749<br>2.950749<br>2.950749<br>2.950749<br>2.950749<br>2.950749<br>2.950749<br>2.950749<br>2.950749<br>2.950749<br>2.950749<br>2.950749<br>2.950749<br>2.950749<br>2.950749<br>2.950749<br>2.950749<br>2.950749<br>2.950749<br>2.950749<br>2 | 17022-06<br>13452-06<br>13452-06<br>13452-06<br>13642-06<br>13642-06<br>13642-06<br>13642-06<br>13642-06<br>13642-06<br>13642-06<br>13172-06<br>13172-06<br>13172-06<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0   
   | 0.026860<br>0.001369<br>0.002148<br>0.002088<br>0.002088<br>0.002085<br>0.002085<br>0.002085<br>0.002085<br>0.002085<br>0.002085<br>0.002085<br>0.002285<br>0.002285<br>0.002285<br>0.002285<br>0.002285<br>0.002285<br>0.002285<br>0.002285<br>0.002285<br>0.002285<br>0.002285<br>0.002285<br>0.002285<br>0.002285<br>0.002285<br>0.002285<br>0.002285<br>0.002285<br>0.002285<br>0.002285<br>0.002285<br>0.002285<br>0.002285<br>0.002285<br>0.002285<br>0.002285<br>0.002285<br>0.002285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.0000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.0000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.0000285<br>0.0000285<br>0.0000285<br>0.000085<br>0.000085<br>0.000085<br>0.00008   | 2.2440968<br>1.9915216<br>1.9915216<br>2.7220732<br>2.6462890<br>2.7220732<br>2.6462890<br>2.5622851<br>2.7220732<br>2.6462890<br>2.5622851<br>2.7220732<br>2.6462890<br>2.5622851<br>2.522851<br>2.522851<br>2.522851<br>2.522851<br>2.522851<br>2.522851<br>2.522851<br>2.522851<br>2.522851<br>2.522851<br>2.522851<br>2.522851<br>2.522851<br>2.522851<br>2.522851<br>2.522851<br>2.522851<br>2.522851<br>2.522851<br>2.522851<br>2.522851<br>2.522851<br>2.522851<br>2.522851<br>2.522851<br>2.522851<br>2.522851<br>2.522851<br>2.522851<br>2.522851<br>2.522851<br>2.522851<br>2.522851<br>2.522851<br>2.522851<br>2.522851<br>2.522851<br>2.522851<br>2.522851<br>2.522851<br>2.522851<br>2.522851<br>2.522851<br>2.522851<br>2.522851<br>2.522851<br>2.522851<br>2.522851<br>2.522851<br>2.522851<br>2.522851<br>2.522851<br>2.522851<br>2.522851<br>2.522851<br>2.522851<br>2.522851<br>2.522851<br>2.522851<br>2.522851<br>2.522851<br>2.522851<br>2.522851<br>2.522851<br>2.522851<br>2.522851<br>2.522851<br>2.522851<br>2.522851<br>2.522851<br>2.522851<br>2.522851<br>2.522851<br>2.522851<br>2.522851<br>2.522851<br>2.522851<br>2.522851<br>2.522851<br>2.522851<br>2.522851<br>2.522851<br>2.522851<br>2.522851<br>2.522851<br>2.522851<br>2.522851<br>2.522851<br>2.522851<br>2.522851<br>2.522851<br>2.522851<br>2.522851<br>2.522851<br>2.522851<br>2.522851<br>2.522851<br>2.522851<br>2.522851<br>2.522851<br>2.522851<br>2.522851<br>2.522851<br>2.522851<br>2.522851<br>2.522851<br>2.522851<br>2.522851<br>2.522851<br>2.522851<br>2.522851<br>2.522851<br>2.522851<br>2.522851<br>2.522851<br>2.522851<br>2.522851<br>2.522851<br>2.522851<br>2.522851<br>2.522851<br>2.522851<br>2.522851<br>2.522851<br>2.522851<br>2.522851<br>2.522851<br>2.522851<br>2.522851<br>2.522851<br>2.522851<br>2.522851<br>2.522851<br>2.522851<br>2.522851<br>2.522851<br>2.522851<br>2.522851<br>2.522851<br>2.522851<br>2.522851<br>2.522851<br>2.522851<br>2.522851<br>2.522851<br>2.522851<br>2.522851<br>2.522851<br>2.522851<br>2.522851<br>2.522851<br>2.522851<br>2.522851<br>2.522851<br>2.522851<br>2.522851<br>2.522851<br>2.522851<br>2.522851<br>2.522851<br>2.522851<br>2.522851<br>2.522851<br>2.522851<br>2.522851<br>2.522851<br>2.522851<br>2.522851<br>2.522851<br>2.522851<br>2.522851<br>2.522851<br>2.522851<br>2.522851<br>2.522851<br>2.522851<br>2.522851<br>2.522851<br>2.522851<br>2.522851<br>2.522851<br>2.522851<br>2.522851<br>2.522851<br>2.522851<br>2.522851<br>2.522851<br>2.522851<br>2.522851<br>2.522851<br>2.522851<br>2.522851<br>2.522851<br>2.522851<br>2.522851<br>2.522851<br>2.522851<br>2.522851<br>2.522851<br>2.522851<br>2.522851<br>2.522851<br>2.522851<br>2.522851<br>2.522851<br>2.522851<br>2.522851<br>2.522851<br>2.522851<br>2.522851<br>2.522851<br>2.52285 | 1.6782.06<br>1.2412.06<br>1.2412.06<br>1.2412.06<br>1.2412.06<br>1.2412.06<br>1.2412.06<br>1.2412.06<br>1.2412.06<br>1.2412.07<br>1.2442.06<br>1.2442.06<br>1.2442.06<br>1.2412.07<br>0<br>0<br>0<br>1.2412.07<br>0<br>0<br>0<br>1.2412.07<br>0<br>0<br>0<br>0<br>1.2412.07<br>0<br>0<br>0<br>0<br>1.2412.07<br>0<br>0<br>0<br>0<br>0<br>1.2412.07<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0  |
0.0027267<br>0.0018534<br>0.0020742<br>0.0020742<br>0.0020742<br>0.0020742<br>0.0020742<br>0.0020742<br>0.0020742<br>0.0020742<br>0.002037<br>0.002062<br>0.0020542<br>0.0020542<br>0.0020542<br>0.0020542<br>0.0002163<br>0.0002072<br>0.0002163<br>0.0002163<br>0.0002163<br>0.0002163<br>0.0002163<br>0.0002163<br>0.0002163<br>0.0002163<br>0.0002163<br>0.0002163<br>0.0002163<br>0.0002175<br>0.0002175<br>0.0002175<br>0.0002175<br>0.0002175<br>0.0002175<br>0.0002175<br>0.0002175<br>0.0002175<br>0.0002175<br>0.0002175<br>0.0002175<br>0.0002175<br>0.0002175<br>0.0002175<br>0.0002175<br>0.0002175<br>0.0002175<br>0.0002175<br>0.0002175<br>0.0002175<br>0.0002175<br>0.0002175<br>0.0002175<br>0.0002175<br>0.0002175<br>0.0002175<br>0.0002175<br>0.0002175<br>0.0002175<br>0.0002175<br>0.0002185<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.000285<br>0.00028  | 2 6.23364<br>2.1248<br>2.3459273<br>2.397196<br>2.154464<br>1.1028226<br>2.161051925<br>2.161051925<br>2.161051925<br>2.161051925<br>2.161051925<br>2.161057113<br>2.161057113<br>2.161057113<br>2.161057113<br>2.161057113<br>2.161057113<br>2.161057113<br>2.161057113<br>2.161057113<br>2.161057113<br>2.161057113<br>2.161057113<br>2.161057113<br>2.161057113<br>2.161057113<br>2.161057113<br>2.161057113<br>2.161057113<br>2.161057113<br>2.161057113<br>2.161057113<br>2.161057113<br>2.161057113<br>2.161057113<br>2.161057113<br>2.161057113<br>2.161057113<br>2.161057113<br>2.161057113<br>2.161057113<br>2.161057113<br>2.161057113<br>2.161057113<br>2.161057113<br>2.161057113<br>2.161057113<br>2.161057113<br>2.161057113<br>2.161057113<br>2.161057113<br>2.161057113<br>2.161057113<br>2.161057113<br>2.161057113<br>2.161057113<br>2.161057113<br>2.161057113<br>2.161057113<br>2.161057113<br>2.161057113<br>2.161057113<br>2.161057113<br>2.161057113<br>2.161057113<br>2.161057113<br>2.161057113<br>2.161057113<br>2.161057113<br>2.161057113<br>2.161057113<br>2.161057113<br>2.161057113<br>2.161057113<br>2.161057113<br>2.161057113<br>2.161057113<br>2.161057113<br>2.161057113<br>2.161057113<br>2.161057113<br>2.161057113<br>2.161057113<br>2.161057113<br>2.161057113<br>2.161057113<br>2.161057113<br>2.161057113<br>2.161057113<br>2.161057113<br>2.161057113<br>2.161057113<br>2.161057113<br>2.161057113<br>2.161057113<br>2.161057113<br>2.161057113<br>2.161057113<br>2.161057113<br>2.161057113<br>2.161057113<br>2.161057113<br>2.161057113<br>2.161057113<br>2.161057113<br>2.161057113<br>2.161057113<br>2.161057113<br>2.161057113<br>2.161057113<br>2.161057113<br>2.161057113<br>2.161057113<br>2.161057113<br>2.161057113<br>2.161057113<br>2.161057113<br>2.161057113<br>2.161057113<br>2.161057113<br>2.161057113<br>2.161057113<br>2.161057113<br>2.161057113<br>2.161057113<br>2.161057113<br>2.161057113<br>2.161057113<br>2.161057113<br>2.161057113<br>2.161057113<br>2.161057113<br>2.161057113<br>2.161057113<br>2.161057113<br>2.161057113<br>2.161057113<br>2.161057113<br>2.161057113<br>2.161057113<br>2.161057113<br>2.161057113<br>2.161057113<br>2.161057113<br>2.161057113<br>2.161057113<br>2.161057113<br>2.161057113<br>2.161057113<br>2.161057113<br>2.161057113<br>2.161057113<br>2.161057113<br>2.161057113<br>2.161057113<br>2.161057113<br>2.161057113<br>2.161057113<br>2.161057113<br>2.161057113<br>2.161057113<br>2.161057113<br>2.161057113<br>2.161057113<br>2.161057113<br>2.161057113<br>2.161057113<br>2.161057113<br>2.161057113<br>2.161057113<br>2.161057113<br>2.161057110  | 19622-06<br>19642-06<br>19542-06<br>19542-06<br>19542-06<br>19542-06<br>19542-06<br>19542-06<br>19542-06<br>19542-06<br>19542-06<br>19542-06<br>19542-07<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0  | 0.0031876<br>0.0022001<br>0.0024433<br>0.0024433<br>0.0023967<br>0.0023967<br>0.0022695<br>0.0022711<br>0.0022605<br>0.0022605<br>0.0023913<br>0.0024544<br>0.0024540<br>0.0015867<br>0.002716<br>0.0005245<br>0.0005245<br>0.0005245<br>0.0005387<br>0.0005853<br>0.0005853<br>0.0012404<br>0.0006853<br>0.0006835<br>0.0006835<br>0.0006835<br>0.0006836<br>0.0006835<br>0.0006835<br>0.0006836<br>0.0006835<br>0.0006835<br>0.0006836   
   | 2.3644564<br>1.8937445<br>2.1225909<br>2.0177733<br>2.1529112<br>2.928047<br>2.8544498<br>2.6943024<br>2.6943024<br>2.6943024<br>2.6943024<br>2.4951395<br>2.5583663<br>2.724646<br>0.493165<br>0.633218<br>0.043318<br>0.6349325<br>0.6846321<br>0.5803511<br>0.684632<br>0.684632<br>0.684632<br>0.684632<br>0.6846321<br>0.5803515<br>0.6846438<br>0.6879215<br>0.6846323<br>0.68479172<br>0.5932154<br>0.5932154<br>0.5932154<br>0.5932154<br>0.5932154<br>0.5932154<br>0.5932154<br>0.5932154<br>0.5932154<br>0.5932154<br>0.5932154<br>0.5932154<br>0.5932154<br>0.5932154<br>0.5932154<br>0.5932154<br>0.5932154<br>0.5932154<br>0.5932154<br>0.5932154<br>0.5932154<br>0.5932154<br>0.5932154<br>0.5932154<br>0.5932154<br>0.5932154<br>0.5932154<br>0.5932154<br>0.5932154<br>0.5932154<br>0.5932154<br>0.5932154<br>0.5932154<br>0.5932154<br>0.5932154<br>0.5932154<br>0.5932154<br>0.5932154<br>0.5932154<br>0.5932154<br>0.5932154<br>0.5932154<br>0.5932154<br>0.5932154<br>0.5932154<br>0.5932154<br>0.5932154<br>0.5932154<br>0.5932154<br>0.5932154<br>0.5932154<br>0.5932154<br>0.5932154<br>0.5932154<br>0.5932154<br>0.5932154<br>0.5932154<br>0.5932154<br>0.5932154<br>0.5932154<br>0.5932154<br>0.5932154<br>0.5932154<br>0.5932154<br>0.5932154<br>0.5932154<br>0.5932154<br>0.5932154<br>0.5932154<br>0.5932154<br>0.5932154<br>0.5932154<br>0.5932154<br>0.5932154<br>0.5932154<br>0.5932154<br>0.5932154<br>0.5932154<br>0.5932154<br>0.5932154<br>0.5932154<br>0.5932154<br>0.5932154<br>0.5932154<br>0.5932154<br>0.5932154<br>0.5932154<br>0.5932154<br>0.5932154<br>0.5932154<br>0.5932154<br>0.5932154<br>0.5932154<br>0.5932154<br>0.5932154<br>0.5932154<br>0.5932154<br>0.5932154<br>0.5932154<br>0.5932154<br>0.5932154<br>0.5932154<br>0.5932154<br>0.5932154<br>0.5932154<br>0.5932154<br>0.5932154<br>0.5932154<br>0.5932154<br>0.5932154<br>0.5932154<br>0.5932154<br>0.5932154<br>0.5932154<br>0.5932154<br>0.5932154<br>0.5932154<br>0.5932154<br>0.5932154<br>0.5932154<br>0.5932154<br>0.5932154<br>0.5932154<br>0.5932154<br>0.5932154<br>0.5932154<br>0.5932154<br>0.5932154<br>0.5932154<br>0.5932154<br>0.5932154<br>0.5932154<br>0.5932154<br>0.5932154<br>0.5932154<br>0.5932154<br>0.5932154<br>0.5932154<br>0.5932154<br>0.5932154<br>0.5932154<br>0.5932154<br>0.5932154<br>0.5932154<br>0.5932154<br>0.5932154<br>0.5932154<br>0.5932154<br>0.5932154<br>0.5932154<br>0.5932154<br>0.5932154<br>0.5932154<br>0.5932154<br>0.5932154<br>0.5932154<br>0.5932154<br>0.5932154<br>0.5932154<br>0.5932154<br>0.5932154<br>0.5932154<br>0.5932154<br>0.5932154<br>0.5932154<br>0.5932154<br>0.5932154<br>0.5932154<br>0.5932154<br>0.5932154<br>0.5932154<br>0.5932154<br>0.5932154<br>0.5932154<br>0.5932154<br>0.59  | 1.768:0-0 0002<br>1.361:0-0 0002<br>1.381:0-0 002<br>1.381:0-0 002<br>1.382:0-0 002<br>1.382:0-0 002<br>1.525:0-0 002<br>1.525:0-0 002<br>1.525:0-0 002<br>1.535:0-0 000<br>1.535:0-0 000<br>1.535:0-0 000<br>1.335:0-0 0000<br>1.335:0-0 0000<br>1.335:0-0 0000<br>1.335:0-0 000<br>1   | 3         0.44           7773         0.54           0.1015         0.04           1015         0.04           1015         0.04           1015         0.04           1015         0.04           1015         0.04           1015         0.04           1015         0.04           1015         0.04           1015         0.04           1015         0.04           1015         0.04           1015         0.04           1015         0.04           1015         0.04           1015         0.04           1116         0.05           1116         0.05           1116         0.05           1116         0.05           1116         0.05           1117         0.02           1118         0.05           1118         0.05           1119         0.05           1119         0.05           1119         0.05           1119         0.05           1119         0.05           119         0.05           119   
  | 47043         3.           8382         2         2           8382         2         3           8382         2         3           83812         2         3           83812         2         3           83812         2         3           83812         2         3           84212         21         3           84215         6         6           84252         2         1           84252         6         1           84252         6         1           84252         6         1           84252         6         1           84252         6         1           84252         6         1           84252         6         1           84252         6         1           84252         6         1           84252         6         1           84352         6         1           84352         6         1           84364         7         1           843         7         1           843         7         1 <td>5.18±.00         5           6.619±.00         5           6.619±.00         4           2.62-08         4           2.94±.03         3           3.874±.08         4           3.874±.08         4           3.874±.08         3           6.51±.08         4           0.01±.08         4           0.21±.08         4           0.442±.08         3           4.42±.08         3           4.42±.08         4           4.62±.08         4           6.51±.08         2           0         0           0         0           0         0           0         0           0         1           0.534±.00         1           0.534±.00         1           534±.00         1           534±.00         1           534±.00         1           534±.00         1           534±.00         1           643±.00         1           643±.00         1           643±.00         1           643±.00         1           643±.00</td> <td>3.998E-05<br/>4.254E-05<br/>4.254E-05<br/>4.254E-05<br/>3.736E-05<br/>4.511E-05<br/>3.669E-05<br/>4.307E-05<br/>4.307E-05<br/>4.008E-05<br/>4.22E-05<br/>4.22E-05<br/>4.22E-05<br/>4.22E-05<br/>4.22E-05<br/>4.22E-05<br/>4.22E-05<br/>4.22E-05<br/>4.22E-05<br/>4.22E-05<br/>4.22E-05<br/>4.22E-05<br/>4.22E-05<br/>4.22E-05<br/>4.22E-05<br/>4.22E-05<br/>4.22E-05<br/>4.22E-05<br/>4.22E-05<br/>4.22E-05<br/>4.22E-05<br/>4.22E-05<br/>4.22E-05<br/>4.22E-05<br/>4.22E-05<br/>4.22E-05<br/>4.22E-05<br/>4.22E-05<br/>4.22E-05<br/>4.22E-05<br/>4.22E-05<br/>4.22E-05<br/>4.22E-05<br/>4.22E-05<br/>4.22E-05<br/>4.22E-05<br/>4.22E-05<br/>4.22E-05<br/>4.22E-05<br/>4.22E-05<br/>4.22E-05<br/>4.22E-05<br/>4.22E-05<br/>4.22E-05<br/>4.22E-05<br/>4.22E-05<br/>4.22E-05<br/>4.22E-05<br/>4.22E-05<br/>4.22E-05<br/>4.22E-05<br/>4.22E-05<br/>4.22E-05<br/>4.22E-05<br/>4.22E-05<br/>4.22E-05<br/>4.22E-05<br/>4.22E-05<br/>4.22E-05<br/>4.22E-05<br/>4.22E-05<br/>4.22E-05<br/>4.22E-05<br/>4.22E-05<br/>4.22E-05<br/>4.22E-05<br/>4.22E-05<br/>4.22E-05<br/>4.22E-05<br/>4.22E-05<br/>4.22E-05<br/>4.22E-05<br/>4.22E-05<br/>4.22E-05<br/>4.22E-05<br/>4.22E-05<br/>4.22E-05<br/>4.22E-05<br/>4.22E-05<br/>4.22E-05<br/>4.22E-05<br/>4.22E-05<br/>4.22E-05<br/>4.22E-05<br/>4.22E-05<br/>4.22E-05<br/>4.22E-05<br/>4.22E-05<br/>4.22E-05<br/>4.22E-05<br/>4.22E-05<br/>4.22E-05<br/>4.22E-05<br/>4.22E-05<br/>4.22E-05<br/>4.22E-05<br/>4.22E-05<br/>4.22E-05<br/>4.22E-05<br/>4.22E-05<br/>4.22E-05<br/>4.22E-05<br/>4.22E-05<br/>4.22E-05<br/>4.22E-05<br/>4.22E-05<br/>4.22E-05<br/>4.22E-05<br/>4.22E-05<br/>4.22E-05<br/>4.22E-05<br/>4.22E-05<br/>4.22E-05<br/>4.22E-05<br/>4.22E-05<br/>4.22E-05<br/>4.22E-05<br/>4.22E-05<br/>4.22E-05<br/>4.22E-05<br/>4.22E-05<br/>4.22E-05<br/>4.22E-05<br/>4.22E-05<br/>4.22E-05<br/>4.22E-05<br/>4.22E-05<br/>4.22E-05<br/>4.22E-05<br/>4.22E-05<br/>4.22E-05<br/>4.22E-05<br/>4.22E-05<br/>4.22E-05<br/>4.22E-05<br/>4.22E-05<br/>4.22E-05<br/>4.22E-05<br/>4.22E-05<br/>4.22E-05<br/>4.22E-05<br/>4.22E-05<br/>4.22E-05<br/>4.22E-05<br/>4.22E-05<br/>4.22E-05<br/>4.22E-05<br/>4.22E-05<br/>4.22E-05<br/>4.22E-05<br/>4.22E-05<br/>4.22E-05<br/>4.22E-05<br/>4.22E-05<br/>4.22E-05<br/>4.22E-05<br/>4.22E-05<br/>4.22E-05<br/>4.22E-05<br/>4.22E-05<br/>4.22E-05<br/>4.22E-05<br/>4.22E-05<br/>4.22E-05<br/>4.22E-05<br/>4.22E-05<br/>4.22E-05<br/>4.22E-05<br/>4.22E-05<br/>4.22E-05<br/>4.22E-05<br/>4.22E-05<br/>4.22E-05<br/>4.22E-05<br/>4.22E-05<br/>4.22E-05<br/>4.22E-05<br/>4.22E-05<br/>4.22E-05<br/>4.22E-05<br/>4.22E-05<br/>4.22E-05<br/>4.22E-05<br/>4.22E-05<br/>4.22E-05<br/>4.22E-05<br/>4.22E-05<br/>4.22E-05<br/>4.22E-05<br/>4.22E-05<br/>4.22E-05<br/>4.22E-05<br/>4.22E-05<br/>4.22E-05<br/>4.22E-05<br/>4.22E-05<br/>4.22E-05<br/>4.22E-05<br/>4.22E-05<br/>4.22E-05<br/>4.22E-05<br/>4.22E-05<br/>4.22E-05<br/>4.22E-05<br/>4.22E-05<br/>4.22E-05<br/>4.22E-05<br/>4.22E-05<br/>4.22E-05<br/>4.22E-05<br/>4.22E-05<br/>4.22E-05<br/>4.22E-05<br/>4.22E-05<br/>4.22E-05<br/>4.22E-05<br/>4.</td> <td>0.0443322<br/>0.0443322<br/>0.044332<br/>0.0470156<br/>0.0584373<br/>0.05584373<br/>0.0558211<br/>0.0538871<br/>0.0538871<br/>0.061206<br/>0.0496292<br/>0.0527943<br/>0.062794<br/>0.04822<br/>0.0657794<br/>0<br/>0.04622<br/>0.0457794<br/>0<br/>0.04622<br/>0.0457794<br/>0<br/>0.04622<br/>0.0457794<br/>0<br/>0.04622<br/>0.0457794<br/>0<br/>0.04622<br/>0.0457794<br/>0<br/>0.04622<br/>0.0457794<br/>0<br/>0.04622<br/>0.0457794<br/>0<br/>0.04622<br/>0.0457794<br/>0<br/>0.04525889<br/>0.0225869<br/>0.0228369<br/>0.0228369<br/>0.0228369<br/>0.0228369<br/>0.0228369<br/>0.0228369<br/>0.0228369<br/>0.0228369<br/>0.0228369<br/>0.0228369<br/>0.0228369<br/>0.0228369<br/>0.0228369<br/>0.0228369<br/>0.0228369<br/>0.0228369<br/>0.0228369<br/>0.0228369<br/>0.0228369<br/>0.0228369<br/>0.0228369<br/>0.0228369<br/>0.0228369<br/>0.0228369<br/>0.0228369<br/>0.0228369<br/>0.0228369<br/>0.022837<br/>0.0228387<br/>0.0228387<br/>0.0228387<br/>0.0228387<br/>0.0228387<br/>0.0228387<br/>0.0228387<br/>0.0228387<br/>0.0228387<br/>0.0228387<br/>0.0228387<br/>0.0228387<br/>0.0228387<br/>0.0228387<br/>0.0228387<br/>0.0228387<br/>0.0228387<br/>0.0228387<br/>0.0228387<br/>0.022837<br/>0.022837<br/>0.022837<br/>0.022837<br/>0.022837<br/>0.022837<br/>0.022837<br/>0.022837<br/>0.022837<br/>0.022837<br/>0.022837<br/>0.022837<br/>0.022837<br/>0.022837<br/>0.022837<br/>0.022837<br/>0.022837<br/>0.022837<br/>0.022837<br/>0.022837<br/>0.022837<br/>0.022837<br/>0.022837<br/>0.022837<br/>0.022837<br/>0.022837<br/>0.022837<br/>0.022837<br/>0.022837<br/>0.022837<br/>0.022837<br/>0.022837<br/>0.022837<br/>0.022837<br/>0.022837<br/>0.022837<br/>0.022837<br/>0.022837<br/>0.022837<br/>0.022837<br/>0.022837<br/>0.022837<br/>0.022837<br/>0.022837<br/>0.022837<br/>0.022837<br/>0.022837<br/>0.022837<br/>0.022837<br/>0.022837<br/>0.022837<br/>0.022837<br/>0.022837<br/>0.022837<br/>0.022837<br/>0.022837<br/>0.022837<br/>0.022837<br/>0.022837<br/>0.022837<br/>0.022837<br/>0.022837<br/>0.022837<br/>0.022837<br/>0.022837<br/>0.022837<br/>0.022837<br/>0.022837<br/>0.022837<br/>0.022837<br/>0.022837<br/>0.022837<br/>0.022837<br/>0.022837<br/>0.022837<br/>0.022837<br/>0.022837<br/>0.022837<br/>0.022837<br/>0.022837<br/>0.022837<br/>0.022837<br/>0.022837<br/>0.022837<br/>0.022837<br/>0.022837<br/>0.022837<br/>0.022837<br/>0.022837<br/>0.022837<br/>0.022837<br/>0.022837<br/>0.022837<br/>0.022837<br/>0.022837<br/>0.022837<br/>0.022837<br/>0.022837<br/>0.022837<br/>0.022837<br/>0.022837<br/>0.022837<br/>0.022837<br/>0.022837<br/>0.022837<br/>0.022837<br/>0.022837<br/>0.022837<br/>0.022837<br/>0.022837<br/>0.022837<br/>0.022837<br/>0.022837<br/>0.022837<br/>0.022837<br/>0.022837<br/>0.022837<br/>0.022837<br/>0.022837<br/>0.022837<br/>0.022837<br/>0.022837<br/>0.022837<br/>0.022837<br/>0.022837<br/>0.022837<br/>0.022837<br/>0.022837<br/>0.022837<br/>0.022837<br/>0.022837<br/>0.022837<br/>0.022837<br/>0.022837<br/>0.022837<br/>0.022837<br/>0.022837<br/>0.022837<br/>0.0228</td>
<td>2.842E-0.8<br/>2.822E-0.8<br/>2.822E-0.8<br/>2.822E-0.8<br/>2.837E-0.8<br/>2.338E-0.8<br/>2.338E-0.8<br/>2.338E-0.8<br/>2.338E-0.8<br/>2.338E-0.8<br/>2.328E-0.8<br/>2.328E-0.8<br/>2.328E-0.8<br/>2.328E-0.8<br/>2.328E-0.8<br/>2.328E-0.8<br/>2.3275E-0.0<br/>2.0<br/>0<br/>0<br/>0<br/>0.014E-09<br/>2.014E-09<br/>2.014E-09<br/>2.014E-09<br/>2.014E-09<br/>2.014E-09<br/>2.014E-09<br/>2.014E-09<br/>2.014E-09<br/>2.014E-09<br/>2.014E-09<br/>2.014E-09<br/>2.014E-09<br/>2.014E-09<br/>2.014E-09<br/>2.014E-09<br/>2.014E-09<br/>2.014E-09<br/>2.014E-09<br/>2.014E-09<br/>2.014E-09<br/>2.014E-09<br/>2.014E-09<br/>2.014E-09<br/>2.014E-09<br/>2.014E-09<br/>2.014E-09<br/>2.014E-09<br/>2.014E-09<br/>2.014E-09<br/>2.014E-09<br/>2.014E-09<br/>2.014E-09<br/>2.014E-09<br/>2.014E-09<br/>2.014E-09<br/>2.014E-09<br/>2.014E-09<br/>2.014E-09<br/>2.014E-09<br/>2.014E-09<br/>2.014E-09<br/>2.014E-09<br/>2.014E-09<br/>2.014E-09<br/>2.014E-09<br/>2.014E-09<br/>2.014E-09<br/>2.014E-09<br/>2.014E-09<br/>2.014E-09<br/>2.014E-09<br/>2.014E-09<br/>2.014E-09<br/>2.014E-09<br/>2.014E-09<br/>2.014E-09<br/>2.014E-09<br/>2.014E-09<br/>2.014E-09<br/>2.014E-09<br/>2.014E-09<br/>2.014E-09<br/>2.014E-09<br/>2.014E-09<br/>2.014E-09<br/>2.014E-09<br/>2.014E-09<br/>2.014E-09<br/>2.014E-09<br/>2.014E-09<br/>2.014E-09<br/>2.014E-09<br/>2.014E-09<br/>2.014E-09<br/>2.014E-09<br/>2.014E-09<br/>2.014E-09<br/>2.014E-09<br/>2.014E-09<br/>2.014E-09<br/>2.014E-09<br/>2.014E-09<br/>2.014E-09<br/>2.014E-09<br/>2.014E-09<br/>2.014E-09<br/>2.014E-09<br/>2.014E-09<br/>2.014E-09<br/>2.014E-09<br/>2.014E-09<br/>2.014E-09<br/>2.014E-09<br/>2.014E-09<br/>2.014E-09<br/>2.014E-09<br/>2.014E-09<br/>2.014E-09<br/>2.014E-09<br/>2.014E-09<br/>2.014E-09<br/>2.014E-09<br/>2.014E-09<br/>2.014E-09<br/>2.014E-09<br/>2.014E-09<br/>2.014E-09<br/>2.014E-09<br/>2.014E-09<br/>2.014E-09<br/>2.014E-09<br/>2.014E-09<br/>2.014E-09<br/>2.014E-09<br/>2.014E-09<br/>2.014E-09<br/>2.014E-09<br/>2.014E-09<br/>2.014E-09<br/>2.014E-09<br/>2.014E-09<br/>2.014E-09<br/>2.014E-09<br/>2.014E-09<br/>2.014E-09<br/>2.014E-09<br/>2.014E-09<br/>2.014E-09<br/>2.014E-09<br/>2.014E-09<br/>2.014E-09<br/>2.014E-09<br/>2.014E-09<br/>2.014E-09<br/>2.014E-09<br/>2.014E-09<br/>2.014E-09<br/>2.014E-09<br/>2.014E-09<br/>2.014E-09<br/>2.014E-09<br/>2.014E-09<br/>2.014E-09<br/>2.014E-09<br/>2.014E-09<br/>2.014E-09<br/>2.014E-09<br/>2.014E-09<br/>2.014E-09<br/>2.014E-09<br/>2.014E-09<br/>2.014E-09<br/>2.014E-09<br/>2.014E-09<br/>2.014E-09<br/>2.014E-09<br/>2.014E-09<br/>2.014E-09<br/>2.014E-09<br/>2.014E-09<br/>2.014E-09<br/>2.014E-09<br/>2.014E-09<br/>2.014E-09<br/>2.014E-09<br/>2.014E-09<br/>2.014E-09<br/>2.014E-09<br/>2.014E-09<br/>2.014E-09<br/>2.014E-09<br/>2.014E-09<br/>2.014E-09<br/>2.014E-09<br/>2.014E-09<br/>2.014E-09<br/>2.014E-09<br/>2.014E-09<br/>2.014E-09<br/>2.014E-09<br/>2.014E-09<br/>2.014E-09<br/>2.014E-09<br/>2.014E-0</td> <td>15346-05<br/>1897-05<br/>1897-05<br/>1876-05<br/>1876-05<br/>1876-05<br/>1876-05<br/>1876-05<br/>18307-05<br/>18307-05<br/>18307-05<br/>18307-05<br/>18307-05<br/>18307-05<br/>18307-05<br/>18307-05<br/>18307-05<br/>18307-05<br/>18307-05<br/>18307-05<br/>18307-05<br/>18307-05<br/>18307-05<br/>18307-05<br/>18307-05<br/>18307-05<br/>18307-05<br/>18307-05<br/>18307-05<br/>18307-05<br/>18307-05<br/>18307-05<br/>18307-05<br/>18307-05<br/>18307-05<br/>18307-05<br/>18307-05<br/>18307-05<br/>18307-05<br/>18307-05<br/>18307-05<br/>18307-05<br/>18307-05<br/>18307-05<br/>18307-05<br/>18307-05<br/>18307-05<br/>18307-05<br/>18307-05<br/>18307-05<br/>18307-05<br/>18307-05<br/>18307-05<br/>18307-05<br/>18307-05<br/>18307-05<br/>18307-05<br/>18307-05<br/>18307-05<br/>18307-05<br/>18307-05<br/>18307-05<br/>18307-05<br/>18307-05<br/>18307-05<br/>18307-05<br/>18307-05<br/>18307-05<br/>18307-05<br/>18307-05<br/>18307-05<br/>18307-05<br/>18307-05<br/>18307-05<br/>18307-05<br/>18307-05<br/>18307-05<br/>18307-05<br/>18307-05<br/>18307-05<br/>18307-05<br/>18307-05<br/>18307-05<br/>18307-05<br/>18307-05<br/>18307-05<br/>18307-05<br/>18307-05<br/>18307-05<br/>18307-05<br/>18307-05<br/>18307-05<br/>18307-05<br/>18307-05<br/>18307-05<br/>18307-05<br/>18307-05<br/>18307-05<br/>18307-05<br/>18307-05<br/>18307-05<br/>18307-05<br/>18307-05<br/>18307-05<br/>18307-05<br/>18307-05<br/>18307-05<br/>18307-05<br/>18307-05<br/>18307-05<br/>18307-05<br/>18307-05<br/>18307-05<br/>18307-05<br/>18307-05<br/>18307-05<br/>18307-05<br/>18307-05<br/>18307-05<br/>18307-05<br/>18307-05<br/>18307-05<br/>18307-05<br/>18307-05<br/>18307-05<br/>18307-05<br/>18307-05<br/>18307-05<br/>18307-05<br/>18307-05<br/>18307-05<br/>18307-05<br/>18307-05<br/>18307-05<br/>18307-05<br/>18307-05<br/>18307-05<br/>18307-05<br/>18307-05<br/>18307-05<br/>18307-05<br/>18307-05<br/>18307-05<br/>18307-05<br/>18307-05<br/>18307-05<br/>18307-05<br/>18307-05<br/>18307-05<br/>18307-05<br/>18307-05<br/>18307-05<br/>18307-05<br/>18307-05<br/>18307-05<br/>18307-05<br/>18307-05<br/>18307-05<br/>18307-05<br/>18307-05<br/>18307-05<br/>18307-05<br/>18307-05<br/>18307-05<br/>18307-05<br/>18307-05<br/>18307-05<br/>18307-05<br/>18307-05<br/>18307-05<br/>18307-05<br/>19307-05<br/>19307-05<br/>19307-05<br/>19307-05<br/>19307-05<br/>19307-05<br/>19307-05<br/>19307-05<br/>19307-05<br/>19307-05<br/>19307-05<br/>19307-05<br/>19307-05<br/>19307-05<br/>19307-05<br/>19307-05<br/>19307-05<br/>19307-05<br/>19307-05<br/>19307-05<br/>19307-05<br/>19307-05<br/>19307-05<br/>19307-05<br/>19307-05<br/>19307-05<br/>19307-05<br/>19307-05<br/>19307-05<br/>19307-05<br/>19307-05<br/>19307-05<br/>19307-05<br/>19307-05<br/>19307-05<br/>19307-05<br/>19307-05<br/>19307-05<br/>19307-05<br/>19307-05<br/>19307-05<br/>19307-05<br/>19307-05<br/>19307-05<br/>19307-05<br/>19307-05<br/>19307-05<br/>19307-05<br/>19307-05<br/>19307-05<br/>19307-05<br/>19307-05<br/>19307-05<br/>19307-05<br/>19307-05<br/>19307-05<br/>19307-05<br/>1</td> | 5.18±.00         5           6.619±.00         5           6.619±.00         4           2.62-08         4           2.94±.03         3           3.874±.08         4           3.874±.08         4           3.874±.08         3           6.51±.08         4           0.01±.08         4           0.21±.08         4           0.442±.08         3           4.42±.08         3           4.42±.08         4           4.62±.08         4           6.51±.08         2           0         0           0         0           0         0           0         0           0         1           0.534±.00         1           0.534±.00         1           534±.00         1           534±.00         1           534±.00         1           534±.00         1           534±.00         1           643±.00         1           643±.00         1           643±.00         1           643±.00         1           643±.00   | 3.998E-05<br>4.254E-05<br>4.254E-05<br>4.254E-05<br>3.736E-05<br>4.511E-05<br>3.669E-05<br>4.307E-05<br>4.307E-05<br>4.008E-05<br>4.22E-05<br>4.22E-05<br>4.22E-05<br>4.22E-05<br>4.22E-05<br>4.22E-05<br>4.22E-05<br>4.22E-05<br>4.22E-05<br>4.22E-05<br>4.22E-05<br>4.22E-05<br>4.22E-05<br>4.22E-05<br>4.22E-05<br>4.22E-05<br>4.22E-05<br>4.22E-05<br>4.22E-05<br>4.22E-05<br>4.22E-05<br>4.22E-05<br>4.22E-05<br>4.22E-05<br>4.22E-05<br>4.22E-05<br>4.22E-05<br>4.22E-05<br>4.22E-05<br>4.22E-05<br>4.22E-05<br>4.22E-05<br>4.22E-05<br>4.22E-05<br>4.22E-05<br>4.22E-05<br>4.22E-05<br>4.22E-05<br>4.22E-05<br>4.22E-05<br>4.22E-05<br>4.22E-05<br>4.22E-05<br>4.22E-05<br>4.22E-05<br>4.22E-05<br>4.22E-05<br>4.22E-05<br>4.22E-05<br>4.22E-05<br>4.22E-05<br>4.22E-05<br>4.22E-05<br>4.22E-05<br>4.22E-05<br>4.22E-05<br>4.22E-05<br>4.22E-05<br>4.22E-05<br>4.22E-05<br>4.22E-05<br>4.22E-05<br>4.22E-05<br>4.22E-05<br>4.22E-05<br>4.22E-05<br>4.22E-05<br>4.22E-05<br>4.22E-05<br>4.22E-05<br>4.22E-05<br>4.22E-05<br>4.22E-05<br>4.22E-05<br>4.22E-05<br>4.22E-05<br>4.22E-05<br>4.22E-05<br>4.22E-05<br>4.22E-05<br>4.22E-05<br>4.22E-05<br>4.22E-05<br>4.22E-05<br>4.22E-05<br>4.22E-05<br>4.22E-05<br>4.22E-05<br>4.22E-05<br>4.22E-05<br>4.22E-05<br>4.22E-05<br>4.22E-05<br>4.22E-05<br>4.22E-05<br>4.22E-05<br>4.22E-05<br>4.22E-05<br>4.22E-05<br>4.22E-05<br>4.22E-05<br>4.22E-05<br>4.22E-05<br>4.22E-05<br>4.22E-05<br>4.22E-05<br>4.22E-05<br>4.22E-05<br>4.22E-05<br>4.22E-05<br>4.22E-05<br>4.22E-05<br>4.22E-05<br>4.22E-05<br>4.22E-05<br>4.22E-05<br>4.22E-05<br>4.22E-05<br>4.22E-05<br>4.22E-05<br>4.22E-05<br>4.22E-05<br>4.22E-05<br>4.22E-05<br>4.22E-05<br>4.22E-05<br>4.22E-05<br>4.22E-05<br>4.22E-05<br>4.22E-05<br>4.22E-05<br>4.22E-05<br>4.22E-05<br>4.22E-05<br>4.22E-05<br>4.22E-05<br>4.22E-05<br>4.22E-05<br>4.22E-05<br>4.22E-05<br>4.22E-05<br>4.22E-05<br>4.22E-05<br>4.22E-05<br>4.22E-05<br>4.22E-05<br>4.22E-05<br>4.22E-05<br>4.22E-05<br>4.22E-05<br>4.22E-05<br>4.22E-05<br>4.22E-05<br>4.22E-05<br>4.22E-05<br>4.22E-05<br>4.22E-05<br>4.22E-05<br>4.22E-05<br>4.22E-05<br>4.22E-05<br>4.22E-05<br>4.22E-05<br>4.22E-05<br>4.22E-05<br>4.22E-05<br>4.22E-05<br>4.22E-05<br>4.22E-05<br>4.22E-05<br>4.22E-05<br>4.22E-05<br>4.22E-05<br>4.22E-05<br>4.22E-05<br>4.22E-05<br>4.22E-05<br>4.22E-05<br>4.22E-05<br>4.22E-05<br>4.22E-05<br>4.22E-05<br>4.22E-05<br>4.22E-05<br>4.22E-05<br>4.22E-05<br>4.22E-05<br>4.22E-05<br>4.22E-05<br>4.22E-05<br>4.22E-05<br>4.22E-05<br>4.22E-05<br>4.22E-05<br>4.22E-05<br>4.22E-05<br>4.22E-05<br>4.22E-05<br>4.22E-05<br>4.22E-05<br>4.22E-05<br>4.22E-05<br>4.22E-05<br>4.22E-05<br>4.22E-05<br>4.22E-05<br>4.22E-05<br>4.22E-05<br>4.22E-05<br>4.22E-05<br>4.22E-05<br>4.22E-05<br>4.22E-05<br>4.22E-05<br>4.22E-05<br>4.22E-05<br>4. |
0.0443322<br>0.0443322<br>0.044332<br>0.0470156<br>0.0584373<br>0.05584373<br>0.0558211<br>0.0538871<br>0.0538871<br>0.061206<br>0.0496292<br>0.0527943<br>0.062794<br>0.04822<br>0.0657794<br>0<br>0.04622<br>0.0457794<br>0<br>0.04622<br>0.0457794<br>0<br>0.04622<br>0.0457794<br>0<br>0.04622<br>0.0457794<br>0<br>0.04622<br>0.0457794<br>0<br>0.04622<br>0.0457794<br>0<br>0.04622<br>0.0457794<br>0<br>0.04622<br>0.0457794<br>0<br>0.04525889<br>0.0225869<br>0.0228369<br>0.0228369<br>0.0228369<br>0.0228369<br>0.0228369<br>0.0228369<br>0.0228369<br>0.0228369<br>0.0228369<br>0.0228369<br>0.0228369<br>0.0228369<br>0.0228369<br>0.0228369<br>0.0228369<br>0.0228369<br>0.0228369<br>0.0228369<br>0.0228369<br>0.0228369<br>0.0228369<br>0.0228369<br>0.0228369<br>0.0228369<br>0.0228369<br>0.0228369<br>0.0228369<br>0.022837<br>0.0228387<br>0.0228387<br>0.0228387<br>0.0228387<br>0.0228387<br>0.0228387<br>0.0228387<br>0.0228387<br>0.0228387<br>0.0228387<br>0.0228387<br>0.0228387<br>0.0228387<br>0.0228387<br>0.0228387<br>0.0228387<br>0.0228387<br>0.0228387<br>0.0228387<br>0.022837<br>0.022837<br>0.022837<br>0.022837<br>0.022837<br>0.022837<br>0.022837<br>0.022837<br>0.022837<br>0.022837<br>0.022837<br>0.022837<br>0.022837<br>0.022837<br>0.022837<br>0.022837<br>0.022837<br>0.022837<br>0.022837<br>0.022837<br>0.022837<br>0.022837<br>0.022837<br>0.022837<br>0.022837<br>0.022837<br>0.022837<br>0.022837<br>0.022837<br>0.022837<br>0.022837<br>0.022837<br>0.022837<br>0.022837<br>0.022837<br>0.022837<br>0.022837<br>0.022837<br>0.022837<br>0.022837<br>0.022837<br>0.022837<br>0.022837<br>0.022837<br>0.022837<br>0.022837<br>0.022837<br>0.022837<br>0.022837<br>0.022837<br>0.022837<br>0.022837<br>0.022837<br>0.022837<br>0.022837<br>0.022837<br>0.022837<br>0.022837<br>0.022837<br>0.022837<br>0.022837<br>0.022837<br>0.022837<br>0.022837<br>0.022837<br>0.022837<br>0.022837<br>0.022837<br>0.022837<br>0.022837<br>0.022837<br>0.022837<br>0.022837<br>0.022837<br>0.022837<br>0.022837<br>0.022837<br>0.022837<br>0.022837<br>0.022837<br>0.022837<br>0.022837<br>0.022837<br>0.022837<br>0.022837<br>0.022837<br>0.022837<br>0.022837<br>0.022837<br>0.022837<br>0.022837<br>0.022837<br>0.022837<br>0.022837<br>0.022837<br>0.022837<br>0.022837<br>0.022837<br>0.022837<br>0.022837<br>0.022837<br>0.022837<br>0.022837<br>0.022837<br>0.022837<br>0.022837<br>0.022837<br>0.022837<br>0.022837<br>0.022837<br>0.022837<br>0.022837<br>0.022837<br>0.022837<br>0.022837<br>0.022837<br>0.022837<br>0.022837<br>0.022837<br>0.022837<br>0.022837<br>0.022837<br>0.022837<br>0.022837<br>0.022837<br>0.022837<br>0.022837<br>0.022837<br>0.022837<br>0.022837<br>0.022837<br>0.022837<br>0.022837<br>0.022837<br>0.022837<br>0.022837<br>0.022837<br>0.022837<br>0.0228  | 2.842E-0.8<br>2.822E-0.8<br>2.822E-0.8<br>2.822E-0.8<br>2.837E-0.8<br>2.338E-0.8<br>2.338E-0.8<br>2.338E-0.8<br>2.338E-0.8<br>2.338E-0.8<br>2.328E-0.8<br>2.328E-0.8<br>2.328E-0.8<br>2.328E-0.8<br>2.328E-0.8<br>2.328E-0.8<br>2.3275E-0.0<br>2.0<br>0<br>0<br>0<br>0.014E-09<br>2.014E-09<br>2.014E-09<br>2.014E-09<br>2.014E-09<br>2.014E-09<br>2.014E-09<br>2.014E-09<br>2.014E-09<br>2.014E-09<br>2.014E-09<br>2.014E-09<br>2.014E-09<br>2.014E-09<br>2.014E-09<br>2.014E-09<br>2.014E-09<br>2.014E-09<br>2.014E-09<br>2.014E-09<br>2.014E-09<br>2.014E-09<br>2.014E-09<br>2.014E-09<br>2.014E-09<br>2.014E-09<br>2.014E-09<br>2.014E-09<br>2.014E-09<br>2.014E-09<br>2.014E-09<br>2.014E-09<br>2.014E-09<br>2.014E-09<br>2.014E-09<br>2.014E-09<br>2.014E-09<br>2.014E-09<br>2.014E-09<br>2.014E-09<br>2.014E-09<br>2.014E-09<br>2.014E-09<br>2.014E-09<br>2.014E-09<br>2.014E-09<br>2.014E-09<br>2.014E-09<br>2.014E-09<br>2.014E-09<br>2.014E-09<br>2.014E-09<br>2.014E-09<br>2.014E-09<br>2.014E-09<br>2.014E-09<br>2.014E-09<br>2.014E-09<br>2.014E-09<br>2.014E-09<br>2.014E-09<br>2.014E-09<br>2.014E-09<br>2.014E-09<br>2.014E-09<br>2.014E-09<br>2.014E-09<br>2.014E-09<br>2.014E-09<br>2.014E-09<br>2.014E-09<br>2.014E-09<br>2.014E-09<br>2.014E-09<br>2.014E-09<br>2.014E-09<br>2.014E-09<br>2.014E-09<br>2.014E-09<br>2.014E-09<br>2.014E-09<br>2.014E-09<br>2.014E-09<br>2.014E-09<br>2.014E-09<br>2.014E-09<br>2.014E-09<br>2.014E-09<br>2.014E-09<br>2.014E-09<br>2.014E-09<br>2.014E-09<br>2.014E-09<br>2.014E-09<br>2.014E-09<br>2.014E-09<br>2.014E-09<br>2.014E-09<br>2.014E-09<br>2.014E-09<br>2.014E-09<br>2.014E-09<br>2.014E-09<br>2.014E-09<br>2.014E-09<br>2.014E-09<br>2.014E-09<br>2.014E-09<br>2.014E-09<br>2.014E-09<br>2.014E-09<br>2.014E-09<br>2.014E-09<br>2.014E-09<br>2.014E-09<br>2.014E-09<br>2.014E-09<br>2.014E-09<br>2.014E-09<br>2.014E-09<br>2.014E-09<br>2.014E-09<br>2.014E-09<br>2.014E-09<br>2.014E-09<br>2.014E-09<br>2.014E-09<br>2.014E-09<br>2.014E-09<br>2.014E-09<br>2.014E-09<br>2.014E-09<br>2.014E-09<br>2.014E-09<br>2.014E-09<br>2.014E-09<br>2.014E-09<br>2.014E-09<br>2.014E-09<br>2.014E-09<br>2.014E-09<br>2.014E-09<br>2.014E-09<br>2.014E-09<br>2.014E-09<br>2.014E-09<br>2.014E-09<br>2.014E-09<br>2.014E-09<br>2.014E-09<br>2.014E-09<br>2.014E-09<br>2.014E-09<br>2.014E-09<br>2.014E-09<br>2.014E-09<br>2.014E-09<br>2.014E-09<br>2.014E-09<br>2.014E-09<br>2.014E-09<br>2.014E-09<br>2.014E-09<br>2.014E-09<br>2.014E-09<br>2.014E-09<br>2.014E-09<br>2.014E-09<br>2.014E-09<br>2.014E-09<br>2.014E-09<br>2.014E-09<br>2.014E-09<br>2.014E-09<br>2.014E-09<br>2.014E-09<br>2.014E-09<br>2.014E-09<br>2.014E-09<br>2.014E-09<br>2.014E-09<br>2.014E-09<br>2.014E-09<br>2.014E-09<br>2.014E-0    | 15346-05<br>1897-05<br>1897-05<br>1876-05<br>1876-05<br>1876-05<br>1876-05<br>1876-05<br>18307-05<br>18307-05<br>18307-05<br>18307-05<br>18307-05<br>18307-05<br>18307-05<br>18307-05<br>18307-05<br>18307-05<br>18307-05<br>18307-05<br>18307-05<br>18307-05<br>18307-05<br>18307-05<br>18307-05<br>18307-05<br>18307-05<br>18307-05<br>18307-05<br>18307-05<br>18307-05<br>18307-05<br>18307-05<br>18307-05<br>18307-05<br>18307-05<br>18307-05<br>18307-05<br>18307-05<br>18307-05<br>18307-05<br>18307-05<br>18307-05<br>18307-05<br>18307-05<br>18307-05<br>18307-05<br>18307-05<br>18307-05<br>18307-05<br>18307-05<br>18307-05<br>18307-05<br>18307-05<br>18307-05<br>18307-05<br>18307-05<br>18307-05<br>18307-05<br>18307-05<br>18307-05<br>18307-05<br>18307-05<br>18307-05<br>18307-05<br>18307-05<br>18307-05<br>18307-05<br>18307-05<br>18307-05<br>18307-05<br>18307-05<br>18307-05<br>18307-05<br>18307-05<br>18307-05<br>18307-05<br>18307-05<br>18307-05<br>18307-05<br>18307-05<br>18307-05<br>18307-05<br>18307-05<br>18307-05<br>18307-05<br>18307-05<br>18307-05<br>18307-05<br>18307-05<br>18307-05<br>18307-05<br>18307-05<br>18307-05<br>18307-05<br>18307-05<br>18307-05<br>18307-05<br>18307-05<br>18307-05<br>18307-05<br>18307-05<br>18307-05<br>18307-05<br>18307-05<br>18307-05<br>18307-05<br>18307-05<br>18307-05<br>18307-05<br>18307-05<br>18307-05<br>18307-05<br>18307-05<br>18307-05<br>18307-05<br>18307-05<br>18307-05<br>18307-05<br>18307-05<br>18307-05<br>18307-05<br>18307-05<br>18307-05<br>18307-05<br>18307-05<br>18307-05<br>18307-05<br>18307-05<br>18307-05<br>18307-05<br>18307-05<br>18307-05<br>18307-05<br>18307-05<br>18307-05<br>18307-05<br>18307-05<br>18307-05<br>18307-05<br>18307-05<br>18307-05<br>18307-05<br>18307-05<br>18307-05<br>18307-05<br>18307-05<br>18307-05<br>18307-05<br>18307-05<br>18307-05<br>18307-05<br>18307-05<br>18307-05<br>18307-05<br>18307-05<br>18307-05<br>18307-05<br>18307-05<br>18307-05<br>18307-05<br>18307-05<br>18307-05<br>18307-05<br>18307-05<br>18307-05<br>18307-05<br>18307-05<br>18307-05<br>18307-05<br>18307-05<br>19307-05<br>19307-05<br>19307-05<br>19307-05<br>19307-05<br>19307-05<br>19307-05<br>19307-05<br>19307-05<br>19307-05<br>19307-05<br>19307-05<br>19307-05<br>19307-05<br>19307-05<br>19307-05<br>19307-05<br>19307-05<br>19307-05<br>19307-05<br>19307-05<br>19307-05<br>19307-05<br>19307-05<br>19307-05<br>19307-05<br>19307-05<br>19307-05<br>19307-05<br>19307-05<br>19307-05<br>19307-05<br>19307-05<br>19307-05<br>19307-05<br>19307-05<br>19307-05<br>19307-05<br>19307-05<br>19307-05<br>19307-05<br>19307-05<br>19307-05<br>19307-05<br>19307-05<br>19307-05<br>19307-05<br>19307-05<br>19307-05<br>19307-05<br>19307-05<br>19307-05<br>19307-05<br>19307-05<br>19307-05<br>19307-05<br>19307-05<br>1 |

						N	O2 Emission								N	O Emission							NOX En	nission			P	M25 Emissio	n	P	V10 Emission	ı
Road Name	Road ID	Hour	Ani a/km/uoh	nualHourMi	in toppo/woor	Mo	nthlyHourM	in tonno/woor	Monti a/km/uoh	nlyHourAver	rage	An a/km/uch	nualHourM	lin toppo/uppr	Mor	nthlyHourM	lin tonno/woor	Monti	nlyHourAver	rage	Anr a/km/uoh	ualHourMi	n tonno/woor	Month	nlyHourAver	rage toppo/woor	Ar a /km (uob	nnualHourMi	n tonno/woor	Ar a /km /uoh	inualHourMir	n toppo/woor
Village Access Road	RD_30_31	2000-2100	0.0029628	8.997E-10	1.644E-06	0.0024888	7.557E-10	1.381E-06	0.0022174	6.733E-10	1.231E-06	0.055384	1.682E-08	3.074E-05	0.0456636	1.387E-08	2.534E-05	0.0407685	1.238E-08	2.263E-05	0.0583468	L.772E-08	3.238E-05	0.0429859	L.305E-08	2.386E-05	0.0053993	1.64E-09	2.996E-06	0.0057794	1.755E-09	3.207E-06
Village Access Road	RD_30_31	2100-2200	0.2653357	8.057E-08	0.0001473	0.2409592	7.317E-08	0.0001337	0.221684	5.731E-08	0.000123	0.7065235	2.145E-07	0.0003921	0.6392872	1.941E-07	0.0003548	0.5875665	1.784E-07	0.0003261	0.9718592	2.951E-07	0.0005394	0.8092504	2.457E-07	0.0004491	0.0239596	7.275E-09	1.33E-05	0.0259897	7.892E-09	1.442E-05
Village Access Road	RD_30_31 RD_30_31	2200-2300	0.0030941	9.395E-10 9.395E-10	1.717E-06	0.0024999	7.591E-10 7.566E-10	1.387E-06	0.0022216	5.746E-10	1.233E-06	0.0575628	1.748E-08	3.195E-05 3.191E-05	0.0461523	1.401E-08	2.561E-05	0.0408885	1.242E-08	2.269E-05	0.0606569 1	1.842E-08	3.366E-05	0.0431101	L.309E-08	2.392E-05	0.0053993	1.64E-09	2.996E-06	0.0057794	1.755E-09	3.207E-06
Emission Point 19 Paved Road	EP19_1_7	0000-0100	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0
Emission Point 19 Paved Road	EP19_1_7	0100-0200	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0
Emission Point 19 Paved Road Emission Point 19 Paved Road	EP19_1_7 EP19_1_7	0200-0300	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0
Emission Point 19 Paved Road	EP19_1_7	0400-0500	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0
Emission Point 19 Paved Road	EP19_1_7	0500-0600	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0
Emission Point 19 Paved Road Emission Point 19 Paved Road	EP19_1_7 EP19_1_7	0700-0800	1.2417505	0 2.606E-06	0.0038378	1.1431613	0 2.399E-06	0.0035331	1.0542111	U 2.213E-06	0.0032582	7.0412124	0 1.478E-05	0.0217621	6.5608874	U 1.377E-05	0.0202776	6.1275841	0 1.286E-05	0.0189384	8.2829629 1	0 L.738E-05	0.0255999	7.1817951	U 1.507E-05	0.0221966	0.14789	0 3.104E-07	0.0004571	0.16078	0 3.374E-07	0.0004969
Emission Point 19 Paved Road	EP19_1_7	0800-0900	1.2382505	2.599E-06	0.003827	1.146112	2.405E-06	0.0035423	1.0527439	2.209E-06	0.0032537	7.0242124	1.474E-05	0.0217096	6.5753194	1.38E-05	0.0203222	6.1202932	1.285E-05	0.0189158	8.2624629	L.734E-05	0.0255366	7.1730371	1.505E-05	0.0221695	0.14789	3.104E-07	0.0004571	0.16078	3.374E-07	0.0004969
Emission Point 19 Paved Road	EP19_1_7	0900-1000	1.2330605	2.588E-06	0.003811	1.1458003	2.405E-06	0.0035413	1.0518817	2.208E-06	0.003251	6.9988024	1.469E-05	0.021631	6.5737021	1.38E-05	0.0203172	6.116295	1.284E-05	0.0189035	8.2318629 1	L.728E-05	0.025442	7.1681768	1.504E-05	0.0221545	0.14789	3.104E-07	0.0004571	0.16078	3.374E-07	0.0004969
Emission Point 19 Paved Road	EP19_1_7 EP19_1_7	1100-1100	1.2358505	2.594E-06 2.597E-06	0.0038196	1.1535482	2.412E-06 2.421E-06	0.0035522	1.0511094	2.208E-06 2.213E-06	0.0032593	7.0122124	1.472E-05 1.474E-05	0.0216725	6.6114412	1.383E-05	0.0203707	6.1293638	1.285E-05	0.0188917	8.2583629	L.733E-05	0.0255239	7.1839324	1.503E-05	0.0222032	0.14789	3.104E-07 3.104E-07	0.0004571	0.16078	3.374E-07 3.374E-07	0.0004969
Emission Point 19 Paved Road	EP19_1_7	1200-1300	1.2375505	2.597E-06	0.0038249	1.1555057	2.425E-06	0.0035713	1.0533248	2.211E-06	0.0032555	7.0208124	1.474E-05	0.021699	6.6210505	1.39E-05	0.0204635	6.1232761	1.285E-05	0.0189251	8.2583629	L.733E-05	0.0255239	7.1766009	1.506E-05	0.0221805	0.14789	3.104E-07	0.0004571	0.16078	3.374E-07	0.0004969
Emission Point 19 Paved Road	EP19_1_7	1300-1400	1.2390505	2.6E-06	0.0038295	1.1599369	2.434E-06	0.003585	1.050765	2.205E-06	0.0032476	7.0280124	1.475E-05	0.0217213	6.6426406	1.394E-05	0.0205302	6.1107885	1.283E-05	0.0188865	8.2670629 1	L.735E-05	0.0255508	7.1615536	L.503E-05	0.022134	0.14789	3.104E-07	0.0004571	0.16078	3.374E-07	0.0004969
Emission Point 19 Paved Road	EP19_1_7 EP19_1_7	1500-1500	1.2334605	2.589E-06	0.0038341	1.1628084	2.44E-06 2.437E-06	0.0035883	1.0515387	2.207E-06 2.209E-06	0.0032525	7.0006024	1.477E-05 1.469E-05	0.0217432	6.6477452	1.397E-05	0.0205733	6.1185408	1.283E-05	0.018898	8.2340629 1	L.728E-05	0.0255774	7.1000535	1.504E-05	0.0221479	0.14789	3.104E-07 3.104E-07	0.0004571	0.16078	3.374E-07 3.374E-07	0.0004969
Emission Point 19 Paved Road	EP19_1_7	1600-1700	1.2360505	2.594E-06	0.0038202	1.1549561	2.424E-06	0.0035696	1.0473401	2.198E-06	0.003237	7.0136124	1.472E-05	0.0216768	6.618445	1.389E-05	0.0204555	6.094155	1.279E-05	0.018835	8.2496629	L.731E-05	0.025497	7.1414951	1.499E-05	0.022072	0.14789	3.104E-07	0.0004571	0.16078	3.374E-07	0.0004969
Emission Point 19 Paved Road	EP19_1_7	1700-1800	1.2413505	2.605E-06	0.0038366	1.1489235	2.411E-06	0.0035509	1.0474911	2.198E-06	0.0032375	7.0392124	1.477E-05	0.0217559	6.5889874	1.383E-05	0.0203644	6.0948965	1.279E-05	0.0188373	8.2805629 1	L.738E-05	0.0255925	7.1423876	L.499E-05	0.0220748	0.14789	3.104E-07	0.0004571	0.16078	3.374E-07	0.0004969
Emission Point 19 Paved Road	EP19_1_7 EP19_1_7	1900-2000	1.2396505	2.602E-06	0.0038300	1.1433085	2.4E-06 2.395E-06	0.0035336	1.0508467	2.205E-06 2.208E-06	0.0032519	7.0392124	1.477E-05	0.0217559	6.5507383	1.377E-05	0.0202797	6.1176633	1.283E-05	0.0188878	8.2704629 1	L.736E-05	0.0255925	7.1620686	1.505E-05	0.0221556	0.14789	3.104E-07 3.104E-07	0.0004571	0.16078	3.374E-07 3.374E-07	0.0004969
Emission Point 19 Paved Road	EP19_1_7	2000-2100	1.2396505	2.602E-06	0.0038314	1.1393857	2.391E-06	0.0035215	1.0534549	2.211E-06	0.0032559	7.0308124	1.476E-05	0.02173	6.5424317	1.373E-05	0.0202205	6.1238575	1.285E-05	0.0189269	8.2704629	.736E-05	0.0255613	7.1773124	1.506E-05	0.0221827	0.14789	3.104E-07	0.0004571	0.16078	3.374E-07	0.0004969
Emission Point 19 Paved Road	EP19_1_7	2100-2200	1.2431505	2.609E-06	0.0038422	1.1373811	2.387E-06	0.0035153	1.0533069	2.211E-06	0.0032554	7.0477124	1.479E-05	0.0217822	6.5326681	1.371E-05	0.0201904	6.1232096	1.285E-05	0.0189248	8.2908629	1.74E-05	0.0256244	7.1765166	1.506E-05	0.0221803	0.14789	3.104E-07	0.0004571	0.16078	3.374E-07	0.0004969
Emission Point 19 Paved Road Emission Point 19 Paved Road	EP19_1_7 EP19_1_7	2200-2300	1.2451505	2.613E-U6	0.0038484	1.1385454	2.39E-06	0.0035189	1.0526379 .	2.209E-06 0	0.0032534	7.0577124	1.481E-05 0	0.0218131	0.5384429	1.372E-05	0.0202082	6.1199055	1.284E-U5 0	0.0189146	8.3028629 1	0.743E-05	0.0256614	7.1725434 : 0	0	0.022168	0.14789	3.104E-07	0.0004571	0.16078	3.3/4E-0/	0.0004969
Emission Point 19 Paved Road	EP19_8	0000-0100	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0
Emission Point 19 Paved Road	EP19_8	0100-0200	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0
Emission Point 19 Paved Road Emission Point 19 Paved Road	EP19_8 EP19_8	0200-0300	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0
Emission Point 19 Paved Road	EP19_8	0400-0500	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0
Emission Point 19 Paved Road	EP19_8	0500-0600	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0
Emission Point 19 Paved Road Emission Point 19 Paved Road	EP19_8 EP19_8	0500-0700	0	0 7.665E-07	0	0	0 7 0575-07	0 8 372F-05	0	0 5 507E-07	0 7 72E-05	0	0 4 3465-06	0 0005157	0	0	0 0004805	6 1275841	0 3 7825-06	0	8 2829629	0	0 0006066	7 1817951	0	0	0	0 9.129F-08	0	0	0 0255.08	0 1 1775-05
Emission Point 19 Paved Road	EP19_8	0800-0900	1.2382505	7.644E-07	9.068E-05	1.146112	7.075E-07	8.393E-05	1.0527439	5.498E-07	7.71E-05	7.0242124	4.336E-06	0.0005144	6.5753194	4.059E-06	0.0004815	6.1202932	3.778E-06	0.0004482	8.2624629	5.1E-06	0.0006051	7.1730371	1.428E-06	0.0005253	0.14789	9.129E-08	1.083E-05	0.16078	9.925E-08	1.177E-05
Emission Point 19 Paved Road	EP19_8	0900-1000	1.2330605	7.611E-07	9.03E-05	1.1458003	7.073E-07	8.391E-05	1.0518817	6.493E-07	7.703E-05	6.9988024	4.32E-06	0.0005126	6.5737021	4.058E-06	0.0004814	6.116295	3.775E-06	0.0004479	8.2318629	5.081E-06	0.0006029	7.1681768	1.425E-06	0.000525	0.14789	9.129E-08	1.083E-05	0.16078	9.925E-08	1.177E-05
Emission Point 19 Paved Road Emission Point 19 Paved Road	EP19_8 EP19_8	1000-1100	1.2358505	7.629E-07 7.639E-07	9.051E-05 9.063E-05	1.149343	7.095E-07 7.121E-07	8.417E-05 8.448E-05	1.0511094	6.488E-07	7.698E-05 7.723E-05	7.0122124	4.329E-06 4.334E-06	0.0005135	6.591007	4.069E-06 4.081E-06	0.0004827	6.1124795	3.773E-06 3.784F-06	0.0004476	8.2480629 5	5.091E-06	0.000604	7.1635889 4	1.422E-06 1.435E-06	0.0005246	0.14789	9.129E-08 9.129E-08	1.083E-05	0.16078	9.925E-08 9.925E-08	1.177E-05
Emission Point 19 Paved Road	EP19_8	1200-1300	1.2375505	7.639E-07	9.063E-05	1.1555057	7.133E-07	8.462E-05	1.0533248	6.502E-07	7.714E-05	7.0208124	4.334E-06	0.0005142	6.6210505	4.087E-06	0.0004849	6.1232761	3.78E-06	0.0004484	8.2583629 5	5.098E-06	0.0006048	7.1766009	4.43E-06	0.0005256	0.14789	9.129E-08	1.083E-05	0.16078	9.925E-08	1.177E-05
Emission Point 19 Paved Road	EP19_8	1300-1400	1.2390505	7.648E-07	9.074E-05	1.1599369	7.16E-07	8.495E-05	1.050765	5.486E-07	7.695E-05	7.0280124	4.338E-06	0.0005147	6.6426406	4.1E-06	0.0004865	6.1107885	3.772E-06	0.0004475	8.2670629 5	5.103E-06	0.0006054	7.1615536	1.421E-06	0.0005245	0.14789	9.129E-08	1.083E-05	0.16078	9.925E-08	1.177E-05
Emission Point 19 Paved Road Emission Point 19 Paved Road	EP19_8 EP19_8	1400-1500	1.2405505	7.658E-07	9.085E-05 9.033E-05	1.1628084	7.178E-07	8.516E-05 8.502E-05	1.0515387	5.491E-07	7.701E-05 7.707E-05	7.0351124	4.343E-06	0.0005152	6.656583	4.109E-06	0.0004875	6.1145148	3.774E-06	0.0004478	8.2756629 5	5.108E-06	0.0006061	7.1660535 4	1.423E-06	0.0005248	0.14789	9.129E-08	1.083E-05	0.16078	9.925E-08	1.177E-05
Emission Point 19 Paved Road	EP19_8	1600-1700	1.2360505	7.63E-07	9.052E-05	1.1549561	7.129E-07	8.458E-05	1.0473401	6.465E-07	7.67E-05	7.0136124	4.329E-06	0.0005136	6.618445	4.085E-06	0.0004808	6.094155	3.762E-06	0.0004463	8.2496629 5	5.092E-06	0.0006042	7.1414951	1.408E-06	0.0005232	0.14789	9.129E-08	1.083E-05	0.16078	9.925E-08	1.177E-05
Emission Point 19 Paved Road	EP19_8	1700-1800	1.2413505	7.663E-07	9.091E-05	1.1489235	7.092E-07	8.414E-05	1.0474911	6.466E-07	7.671E-05	7.0392124	4.345E-06	0.0005155	6.5889874	4.067E-06	0.0004825	6.0948965	3.762E-06	0.0004464	8.2805629 5	5.111E-06	0.0006064	7.1423876	1.409E-06	0.0005231	0.14789	9.129E-08	1.083E-05	0.16078	9.925E-08	1.177E-05
Emission Point 19 Paved Road	EP19_8	1800-1900	1.2413505	7.663E-07	9.091E-05	1.1433085	7.057E-07	8.373E-05	1.0508467	5.487E-07	7.696E-05	7.0392124	4.345E-06	0.0005155	6.5615731	4.05E-06	0.0004805	6.111222	3.772E-06	0.0004476	8.2805629 5	5.111E-06	0.0006064	7.1620686	1.421E-06	0.0005245	0.14789	9.129E-08	1.083E-05	0.16078	9.925E-08	1.177E-05
Emission Point 19 Paved Road	EP19_8 EP19_8	2000-2000	1.2396505	7.652E-07	9.078E-05	1.1393857	7.044E-07 7.033E-07	8.344E-05	1.0521795	6.503E-07	7.715E-05	7.0308124	4.34E-06	0.0005149	6.5424317	4.044E-06 4.039E-06	0.0004797	6.1238575	3.78E-06	0.0004485	8.2704629 5	5.105E-06	0.0006057	7.1098428 4	4.426E-06	0.0005251	0.14789	9.129E-08 9.129E-08	1.083E-05	0.16078	9.925E-08	1.177E-05
Emission Point 19 Paved Road	EP19_8	2100-2200	1.2431505	7.674E-07	9.104E-05	1.1373811	7.021E-07	8.33E-05	1.0533069	6.502E-07	7.714E-05	7.0477124	4.35E-06	0.0005161	6.5326681	4.033E-06	0.0004784	6.1232096	3.78E-06	0.0004484	8.2908629 5	5.118E-06	0.0006072	7.1765166	4.43E-06	0.0005256	0.14789	9.129E-08	1.083E-05	0.16078	9.925E-08	1.177E-05
Emission Point 19 Paved Road	EP19_8	2200-2300	1.2451505	7.686E-07	9.119E-05	1.1385454	7.028E-07	8.338E-05	1.0526379	5.498E-07	7.709E-05	7.0577124	4.357E-06	0.0005169	6.5384429	4.036E-06	0.0004788	6.1199055	3.778E-06	0.0004482	8.3028629 5	5.125E-06	0.0006081	7.1725434	1.427E-06	0.0005253	0.14789	9.129E-08	1.083E-05	0.16078	9.925E-08	1.177E-05
Emission Point 19 Paved Road	EP19_8 EP19 9 13	2300-0000	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0
Emission Point 19 Paved Road	EP19_9_13	0100-0200	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0
Emission Point 19 Paved Road	EP19_9_13	0200-0300	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0
Emission Point 19 Paved Road	EP19_9_13 EP19 9 13	0400-0400	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0
Emission Point 19 Paved Road	EP19_9_13	0500-0600	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0
Emission Point 19 Paved Road	EP19_9_13	0600-0700	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0
Emission Point 19 Paved Road	EP19_9_13 EP19 9 13	0800-0900	1.2382505	3.822E-07	0.0003655	1.1451615	3.528E-07 3.537E-07	0.0003383	1.0542111	3.254E-07 3.249E-07	0.0003108	7.0242124	2.1/3E-06 2.168E-06	0.0020785	6.5753194	2.025E-06	0.0019367	6.1202932	1.891E-06	0.0018088	8.2624629	2.55E-06	0.0024451	7.1730371	2.217E-06 2.214E-06	0.00212	0.14789	4.565E-08	4.366E-05	0.16078	4.962E-08	4.746E-05
Emission Point 19 Paved Road	EP19_9_13	0900-1000	1.2330605	3.806E-07	0.000364	1.1458003	3.536E-07	0.0003382	1.0518817	3.247E-07	0.0003105	6.9988024	2.16E-06	0.002066	6.5737021	2.029E-06	0.0019405	6.116295	1.888E-06	0.0018055	8.2318629	2.541E-06	0.00243	7.1681768	2.212E-06	0.002116	0.14789	4.565E-08	4.366E-05	0.16078	4.962E-08	4.746E-05
Emission Point 19 Paved Road	EP19_9_13	1000-1100	1.2358505	3.814E-07	0.0003648	1.149343	3.547E-07	0.0003393	1.0511094	3.244E-07	0.0003103	7.0122124	2.164E-06	0.00207	6.591007	2.034E-06	0.0019456	6.1124795	1.887E-06	0.0018044	8.2480629 2	2.546E-06	0.0024348	7.1635889	2.211E-06	0.0021146	0.14789	4.565E-08	4.366E-05	0.16078	4.962E-08	4.746E-05
Emission Point 19 Paved Road	EP19_9_13	1200-1200	1.2375505	3.82E-07	0.0003653	1.1555057	3.566E-07	0.0003411	1.0533248	3.251E-07	0.0003109	7.0208124	2.167E-06	0.0020725	6.6210505	2.044E-06	0.0019545	6.1232761	1.89E-06	0.0018076	8.2583629	2.549E-06	0.0024378	7.1766009	2.215E-06	0.0021185	0.14789	4.565E-08	4.366E-05	0.16078	4.962E-08	4.746E-05
Emission Point 19 Paved Road	EP19_9_13	1300-1400	1.2390505	3.824E-07	0.0003658	1.1599369	3.58E-07	0.0003424	1.050765	3.243E-07	0.0003102	7.0280124	2.169E-06	0.0020746	6.6426406	2.05E-06	0.0019609	6.1107885	1.886E-06	0.0018039	8.2670629	2.552E-06	0.0024404	7.1615536	2.21E-06	0.002114	0.14789	4.565E-08	4.366E-05	0.16078	4.962E-08	4.746E-05
Emission Point 19 Paved Road	EP19_9_13 EP19_9_12	1400-1500	1.2405505	3.829E-07	0.0003662	1.1628084	3.589E-07	0.0003433	1.0515387	3.245E-07	0.0003104	7.0351124	2.171E-06	0.0020767	6.656583	2.055E-06	0.001965	6.1145148	1.887E-06	0.001805	8.2756629	2.554E-06	0.0024429	7 1708852	2.212E-06	0.0021154	0.14789	4.565E-08	4.366E-05	0.16078	4.962E-08	4.746E-05
Emission Point 19 Paved Road	EP19_9_13	1600-1000	1.2360505	3.815E-07	0.0003649	1.1549561	3.565E-07	0.0003409	1.0473401	3.233E-07	0.0003092	7.0136124	2.165E-06	0.0020005	6.618445	2.043E-06	0.0019537	6.094155	1.881E-06	0.001799	8.2496629	2.546E-06	0.0024352	7.1414951	2.204E-06	0.0021081	0.14789	4.565E-08	4.366E-05	0.16078	4.962E-08	4.746E-05
Emission Point 19 Paved Road	EP19_9_13	1700-1800	1.2413505	3.831E-07	0.0003664	1.1489235	3.546E-07	0.0003392	1.0474911	3.233E-07	0.0003092	7.0392124	2.173E-06	0.0020779	6.5889874	2.034E-06	0.001945	6.0948965	1.881E-06	0.0017992	8.2805629	2.556E-06	0.0024444	7.1423876	2.204E-06	0.0021084	0.14789	4.565E-08	4.366E-05	0.16078	4.962E-08	4.746E-05
Emission Point 19 Paved Road Emission Point 19 Paved Road	EP19_9_13 EP19_9_13	1800-1900	1.2413505	3.831E-07 3.826E-07	0.0003664	1.1433085	3.529E-07 3.522E-07	0.0003375	1.0508467	3.243E-07 3.247E-07	0.0003102	7.0392124	2.173E-06	0.0020779	6.5615731	2.025E-06	0.0019369	6.111222	1.886E-06	0.001804	8.2805629 2	2.556E-06	0.0024444	7 1698428	2.211E-06	0.0021142	0.14789 0.14789	4.565E-08	4.366E-05	0.16078	4.962E-08	4.746E-05
Emission Point 19 Paved Road	EP19_9_13	2000-2100	1.2396505	3.826E-07	0.0003659	1.1393857	3.517E-07	0.0003363	1.0534549	3.251E-07	0.000311	7.0308124	2.17E-06	0.0020755	6.5424317	2.019E-06	0.0019313	6.1238575	1.89E-06	0.0018077	8.2704629	2.553E-06	0.0024414	7.1773124	2.215E-06	0.0021187	0.14789	4.565E-08	4.366E-05	0.16078	4.962E-08	4.746E-05
Emission Point 19 Paved Road	EP19_9_13	2100-2200	1.2431505	3.837E-07	0.000367	1.1373811	3.51E-07	0.0003357	1.0533069	3.251E-07	0.0003109	7.0477124	2.175E-06	0.0020804	6.5326681	2.016E-06	0.0019284	6.1232096	1.89E-06	0.0018075	8.2908629	2.559E-06	0.0024474	7.1765166	2.215E-06	0.0021185	0.14789	4.565E-08	4.366E-05	0.16078	4.962E-08	4.746E-05
Emission Point 19 Paved Road Emission Point 19 Paved Road	EP19_9_13 EP19_9_13	2200-2300	1.2451505	3.843E-07	0.0003676	1.1385454	3.514E-07	0.0003361	1.0526379	3.249E-07	0.0003107	7.0577124	2.178E-06	0.0020834	6.5384429	2.018E-06	0.0019301	6.1199055	1.889E-06	0.0018066	8.3028629 2	2.563E-06	0.002451	7.1725434	2.214E-06	0.0021173	0.14789	4.565E-08	4.366E-05	0.16078	4.962E-08	4.746E-05
Emission Point 19 Paved Road	EP19_14	0000-0100	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0
Emission Point 19 Paved Road	EP19_14	0100-0200	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0
Emission Point 19 Paved Road	EP19_14 FP19_14	0200-0300	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0
Emission Point 19 Paved Road	EP19_14	0400-0500	ō	ō	0	0	ō	0	ō	ō	0	0	õ	0	0	ō	0	ō	ō	0	ō	ō	ő	ō	õ	ő	ō	0	ō	ō	ō	ō
Emission Point 19 Paved Road	EP19_14	0500-0600	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0
Emission Point 19 Paved Road	EP19_14	0600-0700	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0
Emission Point 19 Paved Road Emission Point 19 Paved Road	EP19_14 EP19_14	0800-0900	1.2382505	1.834E-06	0.0006295	1.1451613	1.698E-06	0.0005811	1.0542111	1.56E-06	0.0005352	7.0412124	1.043E-US	0.0035794	6.5753194	9.741E-06	0.0033425	6.1202932	9.067E-06	0.0031149	8.2624629 1	L227E-05	0.0042106	7.1730371	1.063E-05	0.0036464	0.14789	2.191E-07 2.191E-07	7.518E-05	0.16078	2.382E-07	8.173E-05
Emission Point 19 Paved Road	EP19_14	0900-1000	1.2330605	1.827E-06	0.0006268	1.1458003	1.697E-06	0.0005825	1.0518817	1.558E-06	0.0005347	6.9988024	1.037E-05	0.0035578	6.5737021	9.739E-06	0.0033417	6.116295	9.061E-06	0.0031092	8.2318629	1.22E-05	0.0041846	7.1681768	1.062E-05	0.0036439	0.14789	2.191E-07	7.518E-05	0.16078	2.382E-07	8.173E-05
Emission Point 19 Paved Road	EP19_14	1000-1100	1.2358505	1.831E-06	0.0006282	1.149343	1.703E-06	0.0005843	1.0511094	1.557E-06	0.0005343	7.0122124	1.039E-05	0.0035646	6.591007	9.764E-06	0.0033505	6.1124795	9.056E-06	0.0031072	8.2480629	1.222E-05	0.0041929	7.1635889	1.061E-05	0.0036416	0.14789	2.191E-07	7.518E-05	0.16078	2.382E-07	8.173E-05
Emission Point 19 Paved Road	EP19_14 EP19_14	1200-1200	1.2375505	1.833E-06	0.0006291	1.1555057	1.712E-06	0.0005874	1.0533248	1.56E-06	0.0005355	7.0208124	1.04E-05	0.003569	6.6210505	9.809E-06	0.0033658	6.1232761	9.072E-06	0.0031158	8.2583629 1		0.0041981	7.1766009	1.063E-05	0.0036482	0.14789	2.191E-07	7.518E-05	0.16078	2.382E-07	8.173E-05
Emission Point 19 Paved Road	EP19_14	1300-1400	1.2390505	1.836E-06	0.0006299	1.1599369	1.718E-06	0.0005896	1.050765	1.557E-06	0.0005342	7.0280124	1.041E-05	0.0035726	6.6426406	9.841E-06	0.0033767	6.1107885	9.053E-06	0.0031064	8.2670629	L.225E-05	0.0042025	7.1615536	1.061E-05	0.0036405	0.14789	2.191E-07	7.518E-05	0.16078	2.382E-07	8.173E-05
Emission Point 19 Paved Road	EP19_14	1400-1500	1.2405505	1.838E-06	0.0006306	1.1628084	1.723E-06	0.0005911	1.0515387	1.558E-06	0.0005345	7.0351124	1.042E-05	0.0035763	6.656583	9.862E-06	0.0033838	6.1145148	9.059E-06	0.0031083	8.2756629	.226E-05	0.0042069	7.1660535	1.062E-05	0.0036428	0.14789	2.191E-07	7.518E-05	0.16078	2.382E-07	8.173E-05
Emission Point 19 Paved Road Emission Point 19 Paved Road	EP19_14 EP19_14	1600-1600	1.2360505	1.831E-06	0.0006283	1.1549561	1.711E-06	0.0005902	1.0523445	1.552E-06	0.0005324	7.0136124	1.037E-05	0.0035587	6.618445	9.805E-06	0.0033644	6.094155	9.005E-06	0.0031103	6.2340629 8.2496629 1	1.222E-05	0.0041857	7.1414951	1.058E-05	0.0036303	0.14789	2.191E-07 2.191E-07	7.518E-05	0.16078	2.382E-07	8.173E-05
Emission Point 19 Paved Road	EP19_14	1700-1800	1.2413505	1.839E-06	0.000631	1.1489235	1.702E-06	0.000584	1.0474911	1.552E-06	0.0005325	7.0392124	1.043E-05	0.0035783	6.5889874	9.761E-06	0.0033495	6.0948965	9.029E-06	0.0030983	8.2805629	L.227E-05	0.0042094	7.1423876	1.058E-05	0.0036308	0.14789	2.191E-07	7.518E-05	0.16078	2.382E-07	8.173E-05
Emission Point 19 Paved Road	EP19_14	1800-1900	1.2413505	1.839E-06	0.000631	1.1433085	1.694E-06	0.0005812	1.0508467	1.557E-06	0.0005342	7.0392124	1.043E-05	0.0035783	6.5615731	9.721E-06	0.0033355	6.111222	9.054E-06	0.0031066	8.2805629	L.227E-05	0.0042094	7.1620686	1.061E-05	0.0036408	0.14789	2.191E-07	7.518E-05	0.16078	2.382E-07	8.173E-05

Road Name	Road ID	Hour	А	nnualHourN	Ain	M	NO2 Emissic onthlyHourf	in Min	Mon	thlyHourAve	erage	Ar	nnualHourN	1in	I Mo	NO Emission onthlyHourN	tin	Mont	thlyHourAv	erage	A	nnualHourN	NOX E	mission Mont	thlyHourAv	erage	F A	M25 Emissi	on Ain	1	'M10 Emissio InnualHourM	n
			g/km/veh	g/s/m2	tonne/year	g/km/veh	g/s/m2	tonne/year	g/km/veh	g/s/m2	tonne/year	g/km/veh	g/s/m2	tonne/year	g/km/veh	g/s/m2	tonne/year	g/km/veh	g/s/m2	tonne/year	g/km/veh	g/s/m2	tonne/year	g/km/veh	g/s/m2	tonne/year	g/km/veh	g/s/m2	tonne/year	g/km/veh	g/s/m2	tonne/year
Emission Point 19 Paved Road	EP19_14	1900-2000	1.2396505	1.837E-06	0.0006302	1.1410852	1.69E-06	0.0005801	1.0521795	1.559E-06	0.0005349	7.0308124	1.042E-05	0.0035741	6.5507383	9.705E-06	0.00333	6.1176633	9.063E-06	0.0031099	8.2704629	1.225E-05	0.0042042	7.1698428	1.062E-05	0.0036447	0.14789	2.191E-07	7.518E-05	0.16078	2.382E-07	8.173E-05
Emission Point 19 Paved Road	EP19_14	2000-2100	1.2396505	1.837E-06	0.0006302	1.1393857	1.688E-06	0.0005792	1.0534549	1.561E-06	0.0005355	7.0308124	1.042E-05	0.0035741	6.5424317	9.692E-06	0.0033258	6.1238575	9.072E-06	0.003113	8.2704629	1.225E-05	0.0042042	7.1773124	1.063E-05	0.0036485	0.14789	2.191E-07	7.518E-05	0.16078	2.382E-07	8.173E-05
Emission Point 19 Paved Road	EP19_14	2100-2200	1.2431505	1.842E-06	0.0006319	1.1373811	1.685E-06	0.0005782	1.0533069	1.56E-06	0.0005354	7.0477124	1.044E-05	0.0035827	6.5326681	9.678E-06	0.0033208	6.1232096	9.071E-06	0.0031127	8.2908629	1.228E-05	0.0042146	7.1765166	1.063E-05	0.0036481	0.14789	2.191E-07	7.518E-05	0.16078	2.382E-07	8.173E-05
Emission Point 19 Paved Road	EP19_14	2200-2300	1.2451505	1.845E-06	0.000633	1.1385454	1.687E-06	0.0005788	1.0526379	1.559E-06	0.0005351	7.0577124	1.046E-05	0.0035877	6.5384429	9.687E-06	0.0033238	6.1199055	9.067E-06	0.003111	8.3028629	1.23E-05	0.0042207	7.1725434	1.063E-05	0.0036461	0.14789	2.191E-07	7.518E-05	0.16078	2.382E-07	8.173E-05
Emission Point 19 Paved Road	EP19_14	2300-0000	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0



# Appendix C-3Derivation of Cumulative Annual<br/>Average Nox to NO2 Conversion<br/>Equation using Jenkin Method by<br/>SAMP



Annual Average NOx, NO<sub>2</sub> and OX concentration in Recent Five Years (Year 201 – 2023) at Selected EPD AQMS

		AQMS Data of the	e Past 5 Years	
Year	Station	NO2 (ug/m3)	NOx (ug/m3)	Conversion
2021	NORTH	36	51	39.1
2022	NORTH	31	42	33.0
2023	NORTH	30	41	32.3
2019	TAP MUN	10	12	10.0
2020	TAP MUN	9	11	9.2
2021	TAP MUN	10	12	10.0
2022	TAP MUN	8	13	10.8
2023	TAP MUN	8	14	11.6
2019	MONG KOK	78	154	78.0
2020	MONG KOK	74	162	79.2
2021	MONG KOK	70	139	75.3
2022	MONG KOK	64	131	73.6
2023	MONG KOK	68	131	73.6



OX = 95.57 J/K = 17.114



# **Appendix D**

**Cumulative Results comparing to Updated Pollutants of Proposed AQOs** 



Table D-21	Predicted FSP and RSP	Concentrations at Representative	ASRs (2025 AQOs)
------------	-----------------------	----------------------------------	------------------

					RSP, I	ug/m³	FSP, µ	ւg/m³
ASR ID	DESCRIPTION	PATH GRID	HEIGHT (mPD)	HEIGHT (mAG)	10 <sup>th</sup> HIGHEST DAILY AVERAGE	ANNUAL AVERAGE	19 <sup>th</sup> HIGHEST DAILY AVERAGE	ANNUAL AVERAGE
ASR1	Village House 1	(33,46)	53.1	1.5	56.5	21.0	33.7	13.2
ASR1	Village House 1	(33,46)	56.1	4.5	56.5	21.0	33.7	13.1
ASR2	Auto Repair Shop 1	(33,46)	51.8	1.5	56.6	21.3	33.9	13.3
ASR2	Auto Repair Shop 1	(33,46)	54.8	4.5	56.5	21.3	33.8	13.2
ASR3	Auto Repair Shop 2	(33,46)	54.3	1.5	56.5	21.0	33.7	13.1
ASR3	Auto Repair Shop 2	(33,46)	57.3	4.5	56.4	20.9	33.7	13.1
ASR4	Warehouse 1	(33,46)	53.1	1.5	56.4	20.9	33.6	13.1
ASR4	Warehouse 1	(33,46)	56.1	4.5	56.4	20.9	33.6	13.1
ASR4	Warehouse 1	(33,46)	59.1	7.5	56.4	20.9	33.6	13.1
ASR5	Warehouse 2	(33,46)	56.2	1.5	56.4	20.9	33.6	13.1
ASR5	Warehouse 2	(33,46)	59.2	4.5	56.4	20.9	33.6	13.1
ASR5	Warehouse 2	(33,46)	62.2	7.5	56.4	20.9	33.6	13.1
ASR6	Corporate Office	(33,46)	62.7	1.5	56.4	20.9	33.6	13.1
ASR6	Corporate Office	(33,46)	65.7	4.5	56.4	20.9	33.6	13.1
ASR7	Village House 2	(33,46)	68.5	1.5	56.4	20.9	33.6	13.1
ASR7	Village House 2	(33,46)	71.5	4.5	56.4	20.9	33.6	13.1
ASR8	Village House 3	(33,46)	83.4	1.5	56.4	20.9	33.6	13.1
ASR9	Auto Repair Shop 3	(33,46)	58.6	1.5	57.9	23.2	34.1	13.8
ASR9	Auto Repair Shop 3	(33,46)	61.6	4.5	57.1	22.2	34.0	13.7
ASR10	Village House 4	(33,46)	47	1.5	57.3	21.5	34.5	13.6
ASR10	Village House 4	(33,46)	50	4.5	57.2	21.4	34.4	13.6
ASR11	Construction equipment rental	(33,46)	44.6	1.5	56.6	21.2	33.9	13.3



					RSP, J	ug/m³	FSP, µ	ւg/m³
ASR ID	DESCRIPTION	PATH GRID	HEIGHT (mPD)	HEIGHT (mAG)	10 <sup>th</sup> HIGHEST DAILY AVERAGE	ANNUAL AVERAGE	19 <sup>th</sup> HIGHEST DAILY AVERAGE	ANNUAL AVERAGE
ASR 11	Construction equipment rental	(33,46)	47.6	4.5	56.6	21.1	33.8	13.3
ASR 12	Tire Shop	(33,46)	46.2	1.5	56.4	20.9	33.7	13.1
ASR 12	Tire Shop	(33,46)	49.2	4.5	56.4	20.9	33.7	13.1
ASR 13	Village House 5	(33,46)	43.8	1.5	56.6	21.0	33.9	13.2
ASR 13	Village House 5	(33,46)	46.8	4.5	56.6	21.0	33.9	13.2
ASR 13	Village House 5	(33,46)	49.8	7.5	56.6	21.0	33.9	13.2
ASR 14	Food Manufacturer Office	(33,46)	59.5	1.5	56.6	21.0	33.8	13.2
ASR 14	Food Manufacturer Office	(33,46)	62.5	4.5	56.5	21.0	33.8	13.2
	Propose	d 2025 AC	QOs		75	30	37.5	15

Table D-32	Predicted NO <sub>2</sub> Concentrations	at Representative	ASRs (2025 AQOs)
		achepicschiative	15115 [2025 /1005]

	DESCRIPTION	PATH GRID	HEIGHT, mPD	HEIGHT,	NO₂, μg/m³			
ASR ID				mAG	19 <sup>th</sup> HIGHEST HOURLY AVERAGE	10 <sup>th</sup> HIGHEST DAILY AVERAGE	ANNUAL AVERAGE	
ASR1	Village House 1	(33,46)	53.1	1.5	55.3	22.0	11.2	
ASR1	Village House 1	(33,46)	56.1	4.5	54.2	21.6	10.9	
ASR2	Auto Repair Shop 1	(33,46)	51.8	1.5	58.1	22.5	10.8	
ASR2	Auto Repair Shop 1	(33,46)	54.8	4.5	56.0	22.2	10.8	
ASR3	Auto Repair Shop 2	(33,46)	54.3	1.5	53.3	22.7	10.9	
ASR3	Auto Repair Shop 2	(33,46)	57.3	4.5	52.7	21.6	10.6	
ASR4	Warehouse 1	(33,46)	53.1	1.5	53.1	20.7	10.4	
ASR4	Warehouse 1	(33,46)	56.1	4.5	53.1	20.6	10.4	
ASR4	Warehouse 1	(33,46)	59.1	7.5	52.4	20.6	10.4	
ASR5	Warehouse 2	(33,46)	56.2	1.5	51.9	20.5	10.3	



				HEIGHT,	NO₂, μg/m³		
ASR ID	DESCRIPTION	PATH GRID	HEIGHT, mPD	mAG	19 <sup>th</sup> HIGHEST HOURLY AVERAGE	10 <sup>th</sup> HIGHEST DAILY AVERAGE	ANNUAL AVERAGE
ASR5	Warehouse 2	(33,46)	59.2	4.5	51.9	20.4	10.3
ASR5	Warehouse 2	(33,46)	62.2	7.5	50.4	20.2	10.3
ASR6	Corporate Office	(33,46)	62.7	1.5	49.4	19.7	10.2
ASR6	Corporate Office	(33,46)	65.7	4.5	49.4	19.7	10.2
ASR7	Village House 2	(33,46)	68.5	1.5	49.0	19.7	10.2
ASR7	Village House 2	(33,46)	71.5	4.5	49.0	19.6	10.1
ASR8	Village House 3	(33,46)	83.4	1.5	49.0	19.5	10.1
ASR9	Auto Repair Shop 3	(33,46)	58.6	1.5	64.5	24.5	11.9
ASR9	Auto Repair Shop 3	(33,46)	61.6	4.5	59.9	23.2	11.4
ASR10	Village House 4	(33,46)	47	1.5	168.3	57.4	27.1
ASR10	Village House 4	(33,46)	50	4.5	156.4	50.3	24.4
ASR11	Construction equipment rental	(33,46)	44.6	1.5	98.6	39.2	15.0
ASR 11	Construction equipment rental	(33,46)	47.6	4.5	93.4	35.8	14.3
ASR 12	Tire Shop	(33,46)	46.2	1.5	59.0	22.5	10.9
ASR 12	Tire Shop	(33,46)	49.2	4.5	59.1	22.5	10.9
ASR 13	Village House 5	(33,46)	43.8	1.5	102.5	28.3	15.3
ASR 13	Village House 5	(33,46)	46.8	4.5	100.9	27.3	14.9
ASR 13	Village House 5	(33,46)	49.8	7.5	85.6	25.2	13.9
ASR 14	Food Manufacturer Office	(33,46)	59.5	1.5	65.1	24.3	11.5
ASR 14	Food Manufacturer Office	(33,46)	62.5	4.5	62.6	23.7	11.4
Proposed 2025 AQOs Standard					200	120	40



# Appendix E Background Noise Monitoring Data

BN1



Sampling Date: 10 December 2024			Noise Measurement			
Sampling Period:	0700 to 1900	Noise Meter Model:		Rion, NL-52		
Weather Conditions:	Sunny	<b>Calibration</b>				
Wind Speed:	<5 m/s	Calib	rator Model:	Rion,	NC-74	
Noise Meter Model:	Rion, NL-52	Befor	re:	94.0 dB(A)		
Façade/ Free-field:	Free-field	After	:	94.0 dB(A)		
Start Time	Lea (30min)	Lmax	Lmin	L10	L90 (1hr)	
10/12/2024 7:00	48.2	67.8	42.1	51.3		
10/12/2024 7:30	49.4	78.9	42.1	43.7		
10/12/2024 8:00	50.3	79	42.6	52.5		
10/12/2024 8:30	48.5	68.8	42.5	51.6	44.4	
10/12/2024 9:00	51.5	74.8	42.5	53.8		
10/12/2024 9:30	56	83.6	42.5	56.1	44.3	
10/12/2024 10:00	51.5	76.8	42.2	54.4	44.0	
10/12/2024 10:30	52.4	73	42.3	54.8	.8 44.9	
10/12/2024 11:00	63.5	91.9	44	66.7	AC 1	
10/12/2024 11:30	56.4	85.3	42.9	55.6	40.1	
10/12/2024 12:00	48.3	62.6	41.9	51.3	AA A	
10/12/2024 12:30	54.9	79.6	42.7	53.9	44.4	
10/12/2024 13:00	56.3	80.3	41.4	53.5	11.2	
10/12/2024 13:30	57.6	81.6	41	55.8	44.2	
10/12/2024 14:00	48.3	77.8	41.4	49.3	12 0	
10/12/2024 14:30	51.8	76.9	43.5	54.2	43.9	
10/12/2024 15:00	49.7	72.2	43.8	49.9	46	
10/12/2024 15:30	50.3	71.7	44.6	51.1	40	
10/12/2024 16:00	47.6	66.6	41.5	49.6		
10/12/2024 16:30	54.2	78.9	41.7	53.5	43.4	
10/12/2024 17:00	50.6	74.6	40.5	48.7	/12 1	
10/12/2024 17:30	46.7	78.2	41.6	47	45.1	
10/12/2024 18:00	51.6	81.7	42.2	50.7	15	
10/12/2024 18:30	48.2	73.8	42.8	49.2	45	
Day (0700 40)	00)					
		62 6	<i>1</i> 0 F	17	/12 1	
	40.7 51 0	02.0 76 E	40.5	4/ 527	45.1 // E	
Avera	ike DT'Q	70.5	42.3	52.7	44.5	



Sampling Date:	10 December 2024		Noise Measurement			
Sampling Period:	070	0 to 1900	Noise M	eter Model:	Rion, NL-52	
Weather Conditions: Sunny		<b>Calibration</b>				
Wind Speed:	<5 r	n/s	Calibrato	or Model:	Rion, NC-74	
Noise Meter Model:	Rio	n, NL-52	Before:		93.8 dB(A)	
Façade/ Free-field:	Free	e-field	After:		93.8 dB(A)	
Start Time		Lea (30min)	Lmax	Lmin	L10	L90 (1hr)
10/12/2024 7	2:00	48.5	66.8	36.1	52.2	
10/12/2024 7	/:30	61.4	91.8	34.7	51.5	39.1
10/12/2024 8	8:00	47.5	71.1	36.2	50.2	
10/12/2024 8	8:30	51.5	76	37.2	51.4	40.2
10/12/2024 9	00:	52	76.7	36.8	50.1	
10/12/2024 9	:30	50.8	77.1	38	51.6	41.2
10/12/2024 10	00:00	50.5	76.8	40.1	51.8	
10/12/2024 10	):30	51.6	75.7	40.3	54.2	43.6
10/12/2024 11	:00	51.7	71.4	41.3	54.9	
10/12/2024 11	:30	52.8	80.7	38.9	52.9	43.6
10/12/2024 12	2:00	49.5	75.5	38.3	51.1	
10/12/2024 12	2:30	49.5	72.7	37.4	51.5	42
10/12/2024 13	8:00	49.8	83.3	35.7	48.7	
10/12/2024 13	3:30	47.9	68	37.2	50.4	40
10/12/2024 14	:00	45.8	63.2	38.6	48.3	
10/12/2024 14	:30	60.5	72.4	47.1	62	42.5
10/12/2024 15	5:00	62.6	73.4	58.7	65.1	
10/12/2024 15	5:30	64.6	83.6	58.8	66.3	60.5
10/12/2024 16	5:00	55.1	68	40.1	62.5	
10/12/2024 16	5:30	59.9	83	38.2	66.8	42.3
10/12/2024 17	2:00	55.3	67.2	38	58.2	
10/12/2024 17	':30	53.9	72.1	37.9	57.6	40.4
10/12/2024 18	8:00	44.8	64.9	35.1	46.4	
10/12/2024 18	8:30	43.6	68.9	34.9	45.4	38.1
Day (0700-19	00)					
Minim	um	43.6	63.2	34.7	45.4	38.1
Avera	age	52.5	74.2	39.8	54.2	42.8

BN3



Sampling Date: Sampling Period: Weather Conditions: Wind Speed: Noise Meter Model: Façade/ Free-field:	10 070 Sur <5 1 Rio Fre	December 2024 00 to 1900 nny m/s n, NL-52 e-field	Noise Me Noise Me <u>Calibratic</u> Calibrator Before: After:	easurement eter Model: on r Model:	Rion, NL-5 Rion, NC-7 94.0 dB(A) 94.0 dB(A)	2 '4
Start Time		Leq (30min)	Lmax	Lmin	L10	L90 (1hr)
10/12/2024	7:00	55.8	81.2	48.7	62.7	<b>E1</b>
10/12/2024	7:30	56.1	79.2	49.2	62.6	51
10/12/2024	8:00	55.2	83.4	47.9	62.9	40.0
10/12/2024	8:30	54.6	78.7	47.6	60.3	49.9
10/12/2024	9:00	56.8	77.3	49.5	60.4	E1 0
10/12/2024	9:30	56.3	82.3	49.1	62.4	51.0
10/12/2024 1	0:00	53.7	86.4	47.4	61.2	40
10/12/2024 1	0:30	53.3	79.7	47.9	60	49
10/12/2024 1	1:00	54.1	81.6	47.4	61.9	40 E
10/12/2024 1	1:30	54.5	75.5	47.1	60	49.5
10/12/2024 1	2:00	53.7	81.4	46.6	61.6	19 6
10/12/2024 1	2:30	53.9	76.5	47.2	61.5	40.0
10/12/2024 1	3:00	52.7	77.3	46.6	59.3	10 7
10/12/2024 1	3:30	53.2	76	46.1	60.3	40.2
10/12/2024 1	4:00	52.5	77.5	45.9	59.5	17 6
10/12/2024 1	4:30	52.1	79.4	46.3	58.8	47.0
10/12/2024 1	5:00	51.4	78.7	46.3	57.7	17.2
10/12/2024 1	5:30	51.2	74.8	46	57.6	47.5
10/12/2024 1	6:00	51.4	78.7	46.2	57.9	17.2
10/12/2024 1	6:30	52.3	76.9	46.1	60.6	47.5
10/12/2024 1	7:00	51.5	77.2	45.9	60.6	16.9
10/12/2024 1	7:30	50.1	75.8	46.1	57.6	40.0
10/12/2024 1	8:00	50.7	76.9	46.2	58.9	16.0
10/12/2024 1	8:30	49.1	79.7	46.1	54	40.9
Day (0700-1	900)					
Minim	num	49.1	74.8	45.9	54	46.8
Aver	age	53.2	78.8	47.1	60.0	48.7



# Photographic Record of Prevailing Background Noise Measurement




# Appendix F

# **Noise Meter Calibration Certificates**



# **Calibration Certificate**

25 727		
Customer		
Address	Date of	receipt : 14-Dec-23
Order No.: Q34412	Date of	
tem Tested		
Description : Sound Level Meter	10	
Manufacturer : RION	I.D. Sorial N	
Model : NL-52	Senaria	
Test Conditions		
Date of Test: 9-Jan-24	Supply	Voltage :
Ambient Temperature : (23 ± 3)°C	Relative	e Humidity : (50 ± 25) %
Test Specifications		
Collibration abook		
Calibration check.	s to IEC 61672-1:2002 Class 1	
The UUT has an indication that it conform	s to IEC 61672-1:2002 Class 1 2-1:2013	
The UUT has an indication that it conform Ref. Document/Procedure: Z01, IEC 6167	s to IEC 61672-1:2002 Class 1 2-1:2013.	
The UUT has an indication that it conform Ref. Document/Procedure: Z01, IEC 6167	s to IEC 61672-1:2002 Class 1 2-1:2013.	
The UUT has an indication that it conform Ref. Document/Procedure: Z01, IEC 6167 Test Results All results were within the IEC 61672 Clas	s to IEC 61672-1:2002 Class 1 2-1:2013. s 1, manufacturer's specification or	Tolerance.
The UUT has an indication that it conform Ref. Document/Procedure: Z01, IEC 6167 Test Results All results were within the IEC 61672 Clas The results are shown in the attached page	s to IEC 61672-1:2002 Class 1 2-1:2013. s 1, manufacturer's specification or re(s).	Tolerance.
The UUT has an indication that it conform Ref. Document/Procedure: Z01, IEC 6167 <b>Test Results</b> All results were within the IEC 61672 Clas The results are shown in the attached pag	s to IEC 61672-1:2002 Class 1 2-1:2013. s 1, manufacturer's specification or le(s).	Tolerance.
The UUT has an indication that it conform Ref. Document/Procedure: Z01, IEC 6167 Test Results All results were within the IEC 61672 Class The results are shown in the attached page Main Test equipment used: Equipment No. Description	s to IEC 61672-1:2002 Class 1 2-1:2013. s 1, manufacturer's specification or re(s).	Tolerance. Traceable to
The UUT has an indication that it conform Ref. Document/Procedure: Z01, IEC 6167 <b>Test Results</b> All results were within the IEC 61672 Class The results are shown in the attached page Main Test equipment used: <u>Equipment No.</u> <u>Description</u> Sound Level Calibrator	s to IEC 61672-1:2002 Class 1 2-1:2013. s 1, manufacturer's specification or le(s). <u>Cert. No.</u> 303941	Tolerance. <u>Traceable to</u> NIM-PRC & SCL-HKSAR
Calibration check.   The UUT has an indication that it conform   Ref. Document/Procedure: Z01, IEC 6167 <b>Test Results</b> All results were within the IEC 61672 Class   The results are shown in the attached page   Main Test equipment used:   Equipment No. Description   S240 Sound Level Calibrator   S017 Multi-Eunction Generator	s to IEC 61672-1:2002 Class 1 2-1:2013. s 1, manufacturer's specification or re(s). <u>Cert. No.</u> 303941 C211339	Tolerance. <u>Traceable to</u> NIM-PRC & SCL-HKSAR SCL-HKSAR
Calibration check.   The UUT has an indication that it conform   Ref. Document/Procedure: Z01, IEC 6167 <b>Test Results</b> All results were within the IEC 61672 Class   The results are shown in the attached page   Main Test equipment used: <u>Equipment No.</u> S240 Sound Level Calibrator   S017 Multi-Function Generator	s to IEC 61672-1:2002 Class 1 2-1:2013. s 1, manufacturer's specification or re(s). <u>Cert. No.</u> 303941 C211339	Tolerance. Traceable to NIM-PRC & SCL-HKSAR SCL-HKSAR
The UUT has an indication that it conform   Ref. Document/Procedure: Z01, IEC 6167   Test Results   All results were within the IEC 61672 Class   The results are shown in the attached page   Main Test equipment used:   Equipment No.   S240 Sound Level Calibrator   S017 Multi-Function Generator	s to IEC 61672-1:2002 Class 1 2-1:2013. s 1, manufacturer's specification or re(s). <u>Cert. No.</u> 303941 C211339	Tolerance. <u>Traceable to</u> NIM-PRC & SCL-HKSAR SCL-HKSAR
The UUT has an indication that it conform   Ref. Document/Procedure: Z01, IEC 6167   Test Results   All results were within the IEC 61672 Class   The results are shown in the attached page   Main Test equipment used:   Equipment No. Description   S240 Sound Level Calibrator   S017 Multi-Function Generator	s to IEC 61672-1:2002 Class 1 2-1:2013. s 1, manufacturer's specification or re(s). <u>Cert. No.</u> 303941 C211339	Tolerance. <u>Traceable to</u> NIM-PRC & SCL-HKSAR SCL-HKSAR
The UUT has an indication that it conform   Ref. Document/Procedure: Z01, IEC 6167   Test Results   All results were within the IEC 61672 Class   The results are shown in the attached page   Main Test equipment used:   Equipment No. Description   S240 Sound Level Calibrator   S017 Multi-Function Generator	s to IEC 61672-1:2002 Class 1 2-1:2013. s 1, manufacturer's specification or ie(s). <u>Cert. No.</u> 303941 C211339	Tolerance. <u>Traceable to</u> NIM-PRC & SCL-HKSAR SCL-HKSAR
The UUT has an indication that it conform Ref. Document/Procedure: Z01, IEC 6167 <b>Test Results</b> All results were within the IEC 61672 Class The results are shown in the attached pag Main Test equipment used: Equipment No. Description S240 Sound Level Calibrator S017 Multi-Function Generator	s to IEC 61672-1:2002 Class 1 2-1:2013. s 1, manufacturer's specification or re(s). <u>Cert. No.</u> 303941 C211339	Tolerance. <u>Traceable to</u> NIM-PRC & SCL-HKSAR SCL-HKSAR
The values given in this Calibration Certificate only will not include allowance for the equipment long to the comparent long to the	s to IEC 61672-1:2002 Class 1 2-1:2013. s 1, manufacturer's specification or re(s). <u>Cert. No.</u> 303941 C211339	Tolerance. <u>Traceable to</u> NIM-PRC & SCL-HKSAR SCL-HKSAR f the test and any uncertainties quoted ges, vibration and shock during transportation.
The values given in this Calibration Certificate only will not include allowance for the equipment long te overloading, mis-handling, or the capability of any of the capab	s to IEC 61672-1:2002 Class 1 2-1:2013. s 1, manufacturer's specification or re(s). <u>Cert. No.</u> 303941 C211339 relate to the values measured at the time of rm drift, variations with environmental chang ther laboratory to repeat the measurement.	Tolerance. <u>Traceable to</u> NIM-PRC & SCL-HKSAR SCL-HKSAR fthe test and any uncertainties quoted ges, vibration and shock during transportation, Hong Kong Calibration Ltd. shall not be liable
The UUT has an indication that it conform   Ref. Document/Procedure: Z01, IEC 6167   Test Results   All results were within the IEC 61672 Class   The results are shown in the attached page   Main Test equipment used:   Equipment No.   Description   S240 Sound Level Calibrator   S017 Multi-Function Generator   The values given in this Calibration Certificate only will not include allowance for the equipment long to overloading, mis-handling, or the capability of any of for any loss or damage resulting from the use of the second secon	s to IEC 61672-1:2002 Class 1 2-1:2013. s 1, manufacturer's specification or re(s). <u>Cert. No.</u> 303941 C211339 relate to the values measured at the time of rm dritt, variations with environmental chang ther laboratory to repeat the measurement. e equipment.	Tolerance. <u>Traceable to</u> NIM-PRC & SCL-HKSAR SCL-HKSAR fthe test and any uncertainties quoted ges, vibration and shock during transportation, Hong Kong Calibration Ltd. shall not be liable
The UUT has an indication that it conform   Ref. Document/Procedure: Z01, IEC 6167   Test Results   All results were within the IEC 61672 Class   The results are shown in the attached page   Main Test equipment used:   Equipment No.   Description   S240 Sound Level Calibrator   S017 Multi-Function Generator   The values given in this Calibration Certificate only   will not include allowance for the equipment long te   overloading, mis-handling, or the capability of any of for any loss or damage resulting from the use of the   The test equipment used for calibration are traceated	s to IEC 61672-1:2002 Class 1 2-1:2013. s 1, manufacturer's specification or le(s). <u>Cert. No.</u> 303941 C211339 relate to the values measured at the time of rm drift, variations with environmental chang ther laboratory to repeat the measurement. e equipment.	Tolerance. <u>Traceable to</u> NIM-PRC & SCL-HKSAR SCL-HKSAR f the test and any uncertainties quoted ges, vibration and shock during transportation, Hong Kong Calibration Ltd. shall not be liable by reference to a natural constant.

This Certificate is issued by. D Hong Kong Calibration Ltd Unit 88, 24/F Well Fung Industrial Centre, No. 58-76, 'Ta Chuen Ping Street,Kwai Chung, NT.Hong Kong Tet 2425 8801 Fax 2425 8646

The copyright of this certificate is owned by Hong Kong Calibration Ltd.. It may not be reproduced except in full.

Е

Date:

9-Jan-24





# **Calibration Certificate**

Certificate No. 311869

Page 2 of 3 Pages

Results :

Acoustical signal test

1. Indication at the Calibration Check Frequency ( 1kHz )

UUT Setting Appl.		Applied Value (dB)	UUT Reading (dB)	
Weight Response	Afte	After Adjust.*		
A	F	, 94.0	94.0	
	S		94.0	
C	F		94.0	
7	-		94.0	

\*Adjustment using the customer's sound calibrator was performed immediately before test.

Tolerance :  $\pm 1.0 \text{ dB}$ Uncertainty :  $\pm 0.1 \text{ dB}$ 

2. Self-generated noise (Microphone Installed, most sensitive range): 14.5 dBA (Mfr's Spec.  $\leq$  17 dBA)

### Electrical signal tests

3. Frequency weightings (A,F)

Freq	uency	Attenuation (dB)	IEC 61672-1 Class 1 Spec.
31.5	Hz	-39.6	- 39.4 dB, ± 1.5 dB
63	Hz	-26.2	- 26.2 dB, ± 1.0 dB
125	Hz	-16.2	- 16.1 dB, ± 1.0 dB
250	Hz	-8.7	- 8.6 dB, ± 1.0 dB
500	Hz	-3.2	- 3.2 dB, ± 1.0 dB
1	kHz	0.0 (Ref)	$0 \text{ dB}, \pm 0.7 \text{ dB}$
2	kHz	+1.0	$+$ 1.2 dB, $\pm$ 1.0 dB
4	kHz	+0.6	$+$ 1.0 dB, $\pm$ 1.0 dB
8	kHz	-1.2	- 1.1 dB, + 1.5 dB ~ -2.5 dB
16	kHz	-8.6	- 6.6 dB, + 2.5 dB ~ - 16.0 dB

Uncertainty :  $\pm 0.1 \text{ dB}$ 





# **Calibration Certificate**

### Certificate No. 311869

Page 3 of 3 Pages

### 4. Frequency & Time weightings

4.1 Frequency Weighting (1kHz)

UUT Setting				8
Time Weight.	Freq. Weight.	Anticipated Value (dB)	UUT Reading (dB)	IEC 61672-1 Class 1 Spec.
F	А	94.0	94.0 (Ref.)	
	С		94.0	± 0.2 dB
	Z		94.0	

Uncertainty : ± 0.1 dB

### 4.2 Time Weighting (1kHz)

UUT Setting				Call International ACCEPTION OF
Time Weight.	Freq. Weight.	Anticipated Value (dB)	UUT Reading (dB)	IEC 61672-1 Class 1 Spec.
. F	A	94.0	94.0 (Ref.)	
S		1. Sec. 1	94.0	± 0.1 dB
eq			94.0	

Uncertainty : ± 0.1 dB

### 5. Level Linearity on the Reference Level Range (8 kHz, A, F)

Anticipated Value (dB)	UUT Reading (dB)	IEC 61672-1 Class 1 Spec.
124.0	124.0	± 0.8 dB
114.0	114.0	
104.0	104.0	
94.0	94.0 (Ref.)	
84.0	84.0	
74.0	74.0	
64.0	64.0	-
54.0	54.0	
44.0	44.1	

Uncertainty :  $\pm 0.1 \text{ dB}$ 

# 6. Level Linearity including the level range control (1 kHz, A, F) N.A. (UUT is single range )

Remarks: 1. UUT: Unit-Under-Test

- 2. The uncertainty claimed is for a confidence probability of not less than 95%.
- 3. Atmospheric Pressure: 1 008 hPa.
- 4. Microphone model: UC-59, S/N: 11558.
- 5. Preamplifier model: NH-25, S/N: 43502.

### ----- END -----



# **Calibration Certificate**

Certificate No. 411654	v	Page 1	of 3 Pages
Customer :			
Address :			
Order No.: Q44338	C	Date of receipt	: 8-Nov-24
Item Tested			
Description : Sound Level Meter			
Manufacturer : RION	L	.D.	: N15-RION-008
Model : NL-52	S	Serial No.	: 01143485
Test Conditions			
Date of Test: 18-Nov-24	5	Supply Voltage	:
Ambient Temperature : (23 ± 3)°C	F	Relative Humidity	r : (50 ± 25) %
Test Specifications			
Calibration check.			
The UUT has an indication that it conforms to II	EC 61672-1:2013 Class	s 1	
Ref. Document/Procedure: Z01, IEC 61672-1:2	.013.		
Test Results			
The results are shown in the attached page(s).			
Main Test equinment used			
Equipment No Description	Cert No	T	aceable to
S240 Sound Level Calibrator	405380	N	IM-PRC & SCL-HKSAR
S017 Multi-Function Generator	C211339	S	CL-HKSAR
The values given in this Calibration Certificate only relate t will not include allowance for the equipment long term drift	o the values measured at the	e time of the test and a tal changes, vibration a	any uncertainties quoted and shock during transportation,
overloading, mis-handling, or the capability of any other lal	boratory to repeat the measument	urement. Hong Kong (	Calibration Ltd. shall not be liable
	n Del have no seen se l		
The test equipment used for calibration are traceable to In The test results apply to the above Unit-Under-Test only	ternational System of Units (	(SI), or by reference to	a natural constant.
Calibrated by			
This Certificate is issued by: Hann Konn Calibration Ltd	Date:	18-Nov-24	
Unit 8B, 24/F., Well Fung Industrial Centre, No. 58-76, Ta Chuen Ping Street Tel: 2425 8801 Fax: 2425 8646	Kwai Chung, NT, Hong Kong,		
The copyright of this certificate is owned by Hong Kong Calibration Ltd It may	ay not be reproduced except in full.		E





# **Calibration Certificate**

Certificate No. 411654

Page 2 of 3 Pages

Results :

Acoustical signal test

1. Indication at the Calibration Check Frequency (1kHz)

UUT Setting		Applied Value (dB)	UUT Reading (dB)	
Weight.	Response		After Adjust.*	
А	F	94.0	93.8	
	S		93.8	
С	F		93.8	
Z			93.8	

\*Adjustment using the customer's sound calibrator was performed immediately before test.

Tolerance :  $\pm$  1.0 dB Uncertainty :  $\pm$  0.1 dB

2. Self-generated noise (Microphone Installed, most sensitive range): 16.6 dBA (Mfr's Spec. ≤ 17 dBA)

### **Electrical signal tests**

3. Frequency weightings (A,F)

Freq	uency	Attenuation (dB)	IEC 61672-1 Class 1 Spec.
31.5	Hz	-39.5	- 39.4 dB, ± 1.5 dB
63	Hz	-26.1	- 26.2 dB, ± 1.0 dB
125	Hz	-16.1	- 16.1 dB, ± 1.0 dB
250	Hz	-8.6	- 8.6 dB, ± 1.0 dB
500	Hz	-3.2	- 3.2 dB, ± 1.0 dB
1	kHz	0.0 (Ref)	$0  dB, \pm 0.7  dB$
2	kHz	+1.2	$+$ 1.2 dB, $\pm$ 1.0 dB
4	kHz	+1.3	$+$ 1.0 dB, $\pm$ 1.0 dB
8	kHz	-1.0	- 1.1 dB, + 1.5 dB ~ -2.5 dB
16	kHz	-2.5	- 6.6 dB, + 2.5 dB ~ - 16.0 dF

Uncertainty : ± 0.1 dB



# **Calibration Certificate**

### Certificate No. 411654

Page 3 of 3 Pages

### 4. Frequency & Time weightings

4.1 Frequency Weighting (1kHz)

UUT Setting				
Time Weight.	Freq. Weight.	Anticipated Value (dB)	UUT Reading (dB)	IEC 61672-1 Class 1 Spec.
F	Α	94.0	94.0 (Ref.)	1 <u>000</u>
	С		94.0	± 0.2 dB
	Z		94.0	

Uncertainty : ± 0.1 dB

### 4.2 Time Weighting (1kHz)

UUT Setting				
Time Weight.	Freq. Weight.	Anticipated Value (dB)	UUT Reading (dB)	IEC 61672-1 Class 1 Spec.
F	А	94.0	94.0 (Ref.)	
S	1		94.0	± 0.1 dB
eq			94.0	

Uncertainty :  $\pm 0.1 \text{ dB}$ 

### 5. Level Linearity on the Reference Level Range (8 kHz, A, F)

Anticipated Value (dB)	UUT Reading (dB)	IEC 61672-1 Class 1 Spec.
124.0	123.9	± 0.8 dB
114.0	113.9	
104.0	104.0	
94.0	94.0 (Ref.)	
84.0	84.0	
74.0	74.0	
64.0		
54.0	54.0	
44.0	44.1	

 $Uncertainty:\pm 0.1 \ dB$ 

# 6. Level Linearity including the level range control (1 kHz, A, F) N.A. (UUT is single range )

Remarks : 1. UUT : Unit-Under-Test

- 2. The uncertainty claimed is for a confidence probability of not less than 95%.
- 3. Atmospheric Pressure: 1 007 hPa.
- 4. Microphone model: UC-59, S/N: 04030.
- 5. Preamplifier model: NH-25, S/N: 21113.

#### ----- END -----



# **Calibration Certificate**

Certificate No. 411000	÷	Page	1 of 4	Pages
Customer :				
Address :				
Order No.: Q44338	_	Date of receipt	:	8-Nov-24
Item Tested				
Description : Sound Level Meter				
Manufacturer : RION		I.D.	:	
Model : NL-52		Serial No.	: 00175	560
Test Conditions				
Date of Test: 18-Nov-24		Supply Voltage	:	
Ambient Temperature : (23 ± 3)°C		Relative Humid	ity: (50 ± 2	25) %
Test Specifications				
Calibration check				
The LILLT has an indirection that it conform				
Pof Document/Presedure: 701 /50 0107	IS TO TEC 61672-1:2013/200	JZ GIASS T		
Test Results				
The results are shown in the attached page	ge(s).			
Main Test equipment used:				
Equipment No. Description	Cert. No.		Traceable	to
S240 Sound Level Calibrator	405380		NIM-PRC	SCL-HKSAR
S017 Multi-Function Generator	C211339		SCL-HKSA	R
The values given in this Calibration Certificate only i will not include allowance for the equipment long ter overloading, mis-handling, or the capability of any o for any loss or damage resulting from the use of the	relate to the values measured at rm drift, variations with environme ther laboratory to repeat the mea e equipment.	the time of the test and ental changes, vibratio surement. Hong Kong	d any uncerta n and shock o g Calibration I	inties quoted luring transportation, .td. shall not be liable
The values given in this Calibration Certificate only in will not include allowance for the equipment long ter overloading, mis-handling, or the capability of any o for any loss or damage resulting from the use of the The test equipment used for calibration are traceable the test equipment used for calibration are traceable	relate to the values measured at rm drift, variations with environme ther laboratory to repeat the mea equipment. le to International System of Unit	the time of the test an ental changes, vibratio surement. Hong Kong s (SI), or by reference	d any uncerta n and shock o g Calibration I to a natural c	inties quoted luring transportation, .td. shall not be liable onstant.
The values given in this Calibration Certificate only in will not include allowance for the equipment long ter overloading, mis-handling, or the capability of any o for any loss or damage resulting from the use of the The test equipment used for calibration are traceabl The test results apply to the above Unit-Under-Test	relate to the values measured at rm drift, variations with environme ther laboratory to repeat the mea e equipment. le to International System of Unit only	the time of the test an intal changes, vibratio surement. Hong Kong s (SI), or by reference	d any uncerta n and shock o g Calibration I to a natural c	inties quoted luring transportation, td. shall not be liable onstant.
The values given in this Calibration Certificate only in will not include allowance for the equipment long ter overloading, mis-handling, or the capability of any o for any loss or damage resulting from the use of the The test equipment used for calibration are traceabl The test results apply to the above Unit-Under-Test	relate to the values measured at rm drift, variations with environme ther laboratory to repeat the mea e equipment. le to International System of Unit only	the time of the test an intal changes, vibratio surement. Hong Kong s (SI), or by reference	d any uncerta n and shock o g Calibration I to a natural c	inties quoted luring transportation, t.d. shall not be liable onstant.
The values given in this Calibration Certificate only in will not include allowance for the equipment long ter overloading, mis-handling, or the capability of any o for any loss or damage resulting from the use of the The test equipment used for calibration are traceabl The test results apply to the above Unit-Under-Test Calibrated by :	relate to the values measured at rm drift, variations with environme ther laboratory to repeat the mea e equipment. le to International System of Unit only	the time of the test and intal changes, vibratio surement. Hong Kong s (SI), or by reference	d any uncerta n and shock o g Calibration I to a natural c	inties quoted luring transportation, t.d. shall not be liable onstant.
The values given in this Calibration Certificate only in will not include allowance for the equipment long ter overloading, mis-handling, or the capability of any o for any loss or damage resulting from the use of the The test equipment used for calibration are traceable. The test results apply to the above Unit-Under-Test	relate to the values measured at rm drift, variations with environme ther laboratory to repeat the mea e equipment. le to International System of Unit only Date	the time of the test and intal changes, vibratio surement. Hong Kong s (SI), or by reference 18-Nov-24	d any uncerta n and shock o g Calibration I to a natural c	inties quoted luring transportation, t.d. shall not be liable onstant.
The values given in this Calibration Certificate only in will not include allowance for the equipment long ter overloading, mis-handling, or the capability of any o for any loss or damage resulting from the use of the The test equipment used for calibration are traceabl The test results apply to the above Unit-Under-Test Calibrated by : Init Be Certificate is issued by: Hong Kong Calibration Ltd Jnit 8B, 24/F., Weil Fung Industrial Centre, No. 58-76, Ta Chuen Pin Fer 2425 Ref.	relate to the values measured at rm drift, variations with environme ther laboratory to repeat the mea e equipment. le to International System of Unit only Date	the time of the test and intal changes, vibratio surement. Hong Kong s (SI), or by reference 18-Nov-24	d any uncerta n and shock o g Calibration I to a natural c	inties quoted luring transportation, td. shall not be liable onstant.





# **Calibration Certificate**

Certificate No. 411655

Page 2 of 4 Pages

Results :

Acoustical signal test

1. Indication at the Calibration Check Frequency (1kHz)

UUT	Setting	Applied Value (dB)	UUT Reading (dB)	
Weight.	Response		After Adjust.*	
A	F	94.0	94.0	
	S		94.0	
С	F		94.0	
Z			94.0	

\*Adjustment using the customer's sound calibrator was performed immediately before test.

Tolerance :  $\pm$  1.0 dB Uncertainty :  $\pm$  0.1 dB

2. Self-generated noise (Microphone Installed, most sensitive range) : 23.1 dBA (Mfr's Spec. ≤ 17 dBA)

### Electrical signal tests

3. Frequency weightings (A,F)

Frequ	lency	Attenuation (dB)	IEC 61672-1 Class 1 Spec.
31.5	Hz	-39.5	- 39.4 dB, ± 1.5 dB
63	Hz	-26.2	- 26.2 dB, ± 1.0 dB
125	Hz	-16.2	- 16.1 dB, ± 1.0 dB
250	Hz	-8.7	- $8.6  dB, \pm 1.0  dB$
500	Hz	-3.2	$-3.2 \text{ dB}, \pm 1.0 \text{ dB}$
1	kHz	0.0 (Ref)	$0  dB \pm 0.7  dB$
2	kHz	+1.2	$+$ 1.2 dB, $\pm$ 1.0 dB
4	kHz	+1.3	+ 1.0  dB + 1.0  dB
8	kHz	-1.0	- 1.1 dB, +1.5 dB ~ -2.5 dB
16	kHz	-2.5	- 6.6 dB, + 2.5 dB ~ - 16.0 dI

Uncertainty : ± 0.1 dB



# **Calibration Certificate**

# Certificate No. 411655

Page 3 of 4 Pages

### 4. Frequency & Time weightings

4.1 Frequency Weighting (1kHz)

UUTS	Setting			
Time Weight.	Freq. Weight.	Anticipated Value (dB)	UUT Reading (dB)	IEC 61672-1 Class 1 Spec.
F	A	94.0	94.0 (Ref.)	
	С		94.0	± 0.2 dB
	Z		94.0	

Uncertainty :  $\pm 0.1 \text{ dB}$ 

4.2 Time Weighting (1kHz)

UUT Setting				
Time Weight.	Freq. Weight.	Anticipated Value (dB)	UUT Reading (dB)	IEC 61672-1 Class 1 Spec.
F	А	94.0	94.0 (Ref.)	
S			94.0	± 0.1 dB
eq			94.0	

Uncertainty : ± 0.1 dB

### 5. Level Linearity on the Reference Level Range (8 kHz, A, F)

Anticipated Value (dB)	UUT Reading (dB)	IEC 61672-1 Class 1 Spec.
124.0	123.9	± 0.8 dB
114.0	114.0	
104.0	104.0	
94.0	94.0 (Ref.)	
84.0	84.0	
74.0	74.0	
64.0	64.0	
54.0	54.0	
44.0	44.1	

Uncertainty : ± 0.1 dB

### 6. Level Linearity including the level range control (1 kHz, A, F) N.A. (UUT is single range )



# **Calibration Certificate**

# Certificate No. 411655

Page 4 of 4 Pages

## 7. Filter Characteristics

Frequency	Attenuation (dB)	Tolerance (dB) (Ref : IEC 61260-1 Class 1 Spec.)
125 Hz	-76.7	<- 60
250 Hz	-71.4	< - 40.5
500 Hz	-39.9	< - 16.6
707 Hz	-3.3	$+0.4 \sim -5.3$
1 kHz (Ref)		1.44
1.414 kHz	-3.3	$+0.4 \sim -5.3$
2 kHz	-40.9	< - 16.6
4 kHz	-85.7	< - 40.5
8 kHz	-86.3	< - 60

### Uncertainty : $\pm 0.25 \text{ dB}$

7.2 1/3 – Octave Filter

Frequency	Attenuation (dB)	Tolerance (dB) (Ref: IEC 61260-1 Class 1 Spec.)
326 Hz	-65.3	< - 60
530 Hz	-47.3	<- 40.5
772 Hz	-22.5	< - 16.6
891 Hz	-3.6	+ 0.4 ~ - 5.3
1 kHz (Ref)		
1.122 kHz	-3.8	$+0.4 \sim -5.3$
1.296 kHz	-22.8	< - 16.6
1.887 kHz	-47.7	< - 40.5
3.070 kHz	-92.6	<- 60

Uncertainty :  $\pm 0.25 \text{ dB}$ 

Remarks: 1. UUT: Unit-Under-Test

2. The uncertainty claimed is for a confidence probability of not less than 95%.

3. Atmospheric Pressure: 1 007 hPa.

4. Microphone model: UC-59, S/N: 10989.

5. Preamplifier model: NH-25, S/N: 65662.

----- END ------



# **Calibration Certificate**

Address :					
Order No. :	Q44338		Date of recei	pt :	8-Nov-24
tem Tested					
Description	: Sound Calibrator				
Manufacturer	: RION		I.D.	:	
Viodel	: NC-74		Serial No.	: 3485	7296
Test Condit	ions				
Date of Test :	18-Nov-24		Supply Volta	ige :	
Ambient Tem	perature : (23 ± 3)°C		Relative Hun	nidity : (50 ±	25) %
lest Specif	Ications				
Calibration che	ck.				
The UUT has a	an indication that it conforms to	o IEC 60942:2003 C	lass 1.		
The UUT has a Ref. Document	an indication that it conforms to /Procedure : F21, Z02, IEC 60	o IEC 60942:2003 C 0942:2003.	lass 1.		
The UUT has a Ref. Document <b>Test Result</b>	an indication that it conforms to t/Procedure : F21, Z02, IEC 60 s	o IEC 60942:2003 C 0942:2003.	lass 1.		
The UUT has a Ref. Document <b>Test Result</b> All results were	an indication that it conforms to t/Procedure : F21, Z02, IEC 60 <b>s</b> within the IEC 60942 Class 1	o IEC 60942:2003 C 0942:2003.	lass 1.		
The UUT has a Ref. Document Test Result All results were The results are	an indication that it conforms to VProcedure : F21, Z02, IEC 60 s within the IEC 60942 Class 1 shown in the attached page(s	o IEC 60942:2003 C 0942:2003. specification.	lass 1.		
The UUT has a Ref. Document Test Result All results were The results are	an indication that it conforms to t/Procedure : F21, Z02, IEC 60 s within the IEC 60942 Class 1 shown in the attached page(s	o IEC 60942:2003 C 0942:2003. specification. s).	lass 1.		
The UUT has a Ref. Document Test Result All results were The results are Main Test equi	an indication that it conforms to //Procedure : F21, Z02, IEC 6( s within the IEC 60942 Class 1 shown in the attached page(s pment used: Description	o IEC 60942:2003 C 0942:2003. specification. s).	lass 1.	Tanaashi	
The UUT has a Ref. Document Test Result All results were The results are Main Test equi Equipment No.	an indication that it conforms to VProcedure : F21, Z02, IEC 60 s within the IEC 60942 Class 1 shown in the attached page(s pment used: <u>Description</u> Spectrum Analyzer	o IEC 60942:2003 C 0942:2003. specification. s). <u>Cert. No.</u> 405219	lass 1.		1 SOL HKEAD
The UUT has a <u>Ref. Document</u> <b>Test Result</b> All results were The results are Main Test equi <u>Equipment No.</u> 5014 5240	an indication that it conforms to //Procedure : F21, Z02, IEC 6/ s e within the IEC 60942 Class 1 e shown in the attached page(s pment used: <u>Description</u> Spectrum Analyzer Sound Level Calibrator	o IEC 60942:2003 C 0942:2003. specification. s). <u>Cert. No.</u> 405219 405380	lass 1.	Traceable NIM-PRC	e to & SCL-HKSAR
The UUT has a <u>Ref. Document</u> <b>Test Result</b> All results were The results are Main Test equi <u>Equipment No.</u> 5014 5240 5041	an indication that it conforms to //Procedure : F21, Z02, IEC 6/ s e within the IEC 60942 Class 1 e shown in the attached page(s pment used: . <u>Description</u> Spectrum Analyzer Sound Level Calibrator Universal Counter	o IEC 60942:2003 C 0942:2003. specification. s). <u>Cert. No.</u> 405219 405380 402289	lass 1.	Traceable NIM-PRC NIM-PRC SCI -HKS	<u>≥ to</u> & SCL-HKSAR & SCL-HKSAR
The UUT has a Ref. Document Test Result All results were The results are Vain Test equi Equipment No. 5014 5240 5041 5206	an indication that it conforms to //Procedure : F21, Z02, IEC 6/ s within the IEC 60942 Class 1 e shown in the attached page(s pment used: <u>Description</u> Spectrum Analyzer Sound Level Calibrator Universal Counter Sound Level Meter	c IEC 60942:2003 C 0942:2003. specification. s). <u>Cert. No.</u> 405219 405380 402289 405379	lass 1.	Traceable NIM-PRC NIM-PRC SCL-HKS SCL-HKS	e to & SCL-HKSAR & SCL-HKSAR & SCL-HKSAR
The UUT has a <u>Ref. Document</u> <b>Test Result</b> All results were The results are Vain Test equi <u>Equipment No.</u> \$014 \$240 \$041 \$206	an indication that it conforms to //Procedure : F21, Z02, IEC 6/ s e within the IEC 60942 Class 1 e shown in the attached page(s pment used: <u>Description</u> Spectrum Analyzer Sound Level Calibrator Universal Counter Sound Level Meter	o IEC 60942:2003 C 0942:2003. specification. s). <u>Cert. No.</u> 405219 405380 402289 405379	lass 1.	Traceable NIM-PRC NIM-PRC SCL-HKS SCL-HKS	e to & SCL-HKSAR & SCL-HKSAR AR AR
The UUT has a Ref. Document Test Result All results were The results are Main Test equi <u>Equipment No.</u> 3014 5240 5041 5206	an indication that it conforms to VProcedure : F21, Z02, IEC 60 <b>s</b> within the IEC 60942 Class 1 a shown in the attached page(s pment used: <u>Description</u> Spectrum Analyzer Sound Level Calibrator Universal Counter Sound Level Meter	o IEC 60942:2003 C 0942:2003. specification. s). <u>Cert. No.</u> 405219 405380 402289 405379	lass 1.	Traceable NIM-PRC NIM-PRC SCL-HKS SCL-HKS	2 to & SCL-HKSAR & SCL-HKSAR & AR

The test equipment used for calibration are traceable to International System of Units (SI), or by reference to a natural constant. The test results apply to the above Unit-Under-Test only

Calibrated by				
Cambrated by				
This Certificate is issued by:	Date:	18-Nov-24	20 A	
Hong Kong Calibration Ltd.				
Unit 8B, 24/F., Well Fung Industrial Centre, No. 58-76, Ta Chuen Ping Street, Kwai Chung, NT	Hong Kong.			
Tel: 2425 8801 Fax: 2425 8646				8
The copyright of this certificate is owned by Hong Kong Calibration Ltd It may not be reprodu	ced except in full.			E



# **Calibration Certificate**

Certificate No. 411656

Page 2 of 2 Pages

Results :

1. Generated Sound Pressure Level

UUT Nominal Value (dB)	Measured Value (dB)	IEC 60942 Class 1 Spec.
94.0	94.0	± 0.4 dB

Uncertainty : ± 0.2 dB

- Short-term Level Fluctuation : 0.0 dB IEC 60942 Class 1 Spec. : ± 0.1 dB Uncertainty : ± 0.05 dB
- 3. Frequency

UUT Nominal Value (kHz)	Measured Value (kHz)	IEC 60942 Class 1 Spec.
1	1.002	±1%

Uncertainty :  $\pm$  3.6 x 10 <sup>-6</sup>

4. Total Distortion + Noise : < 1.4 % IEC 60942 Class 1 Spec. : < 3.0 % Uncertainty : ± 2.3 % of reading

Remark : 1. UUT : Unit-Under-Test

- 2. The uncertainty claimed is for a confidence probability of not less than 95%.
- 3. Atmospheric Pressure : 1 007 hPa.

----- END -----



# **Appendix G**

# **Result of Industrial Noise Impact** Assessment



### Daytime – Scenario 1

#### Predicted Noise Level at IN1

						At-source Noise	Noise Reduction									Corr	ection				Total	
	Fixed-noise Sources	Source Type	Ref. Code	SWL/unit, dB(A)	Quantity	Mitigation Measures	from Mitigation Measures, dB(A)	Corrected SWL, dB(A)	Distance, m	Speed, km/h	Angle of view, degree	Operation (30mins)	% on time	Time	Distance	Angle of view	Speed	Screening Effect, dB(A) <sup>[1]</sup>	Facade	CNL, dB(A)	CNL, dB(A)	Criterion , dB(A)
	Concrete mixer 1(electric)	Stationary	CNP 045	96	1	Enclosure	-20	76.0	121.7	NA	NA	30	100.0%	0.0	-49.7	NA	NA	0	3	29.3		
	Concrete mixer 2(electric)	Stationary	CNP 045	96	1	Enclosure	-20	76.0	111.8	NA	NA	30	100.0%	0.0	-48.9	NA	NA	0	3	30.1		1
Filter F	an (mixing unit) - blower (mixer 1)	Stationary	Other PME	95	1	Enclosure	-20	75.0	121.7	NA	NA	30	100.0%	0.0	-49.7	NA	NA	0	3	28.3		
Filter	Fan (mixing unit) - blower (mixer 2)	Stationary	Other PME	95	1	Enclosure	-20	75.0	111.8	NA	NA	30	100.0%	0.0	-48.9	NA	NA	0	3	29.1		
	Truck (Concrete Collection)	Stationary	CNP044	109	5	Enclosure	-20	96.0	121.7	NA	NA	3	10.0%	-10.0	-49.7	NA	NA	0	3	39.3		
	Truck (Concrete Collection)	Stationary	CNP044	109	5	Enclosure	-20	96.0	111.8	NA	NA	3	10.0%	-10.0	-48.9	NA	NA	0	3	40.1		1
	Conveyor / Screw Conveyors	Stationary	CNP 041	90	2	Enclosure	-20	73.0	103.4	NA	NA	30	100.0%	0.0	-48.3	NA	NA	0	3	27.7		1
	Conveyor / Screw Conveyors	Stationary	CNP 041	90	1	Semi-enclosure	-15	75.0	92.7	NA	NA	30	100.0%	0.0	-47.3	NA	NA	0	3	30.7		1
	Water / Admixture Pump	Stationary	CNP 281	88	1	Semi-enclosure	-15	73.0	98.1	NA	NA	30	100.0%	0.0	-47.8	NA	NA	0	3	28.2		1
	Pump (silo)	Stationary	CNP 281	88	4	Enclosure	-20	74.0	120.8	NA	NA	30	100.0%	0.0	-49.6	NA	NA	0	3	27.4		1
Rd1	Concrete Truck	Haul Road	CNP044	109	10	NA	NA	NA	128.9	15	3.6	NA	N/A	NA	-21.1	-17	-11.8	-10	3	29.1		1
Rd2	Concrete truck	Haul Road	CNP044	109	10	NA	NA	NA	119.3	15	9.4	NA	N/A	NA	-20.8	-12.8	-11.8	-10	3	33.6	46	46
Rd3	Concrete truck	Haul Road	CNP044	109	10	NA	NA	NA	115.8	15	9.8	NA	N/A	NA	-20.6	-12.6	-11.8	-10	3	34.0	-10	
Rd4	Concrete truck	Haul Road	CNP044	109	10	NA	NA	NA	112.1	15	3.7	NA	N/A	NA	-20.5	-16.8	-11.8	-10	3	29.9		1
RdS	Concrete truck	Haul Road	CNP044	109	10	NA	NA	NA	103.4	15	0.7	NA	N/A	NA	-20.1	-24.2	-11.8	-10	3	22.9		1
Rdb	Concrete truck	Haul Road	CNP044	109	10	NA	NA	NA	92.4	15	4.6	NA	N/A	NA	-19.7	-15.9	-11.8	-10	3	31.6		1
Rd7	Concrete truck	Haul Road	CNP044	109	10	NA	NA	NA	88.7	15	21.7	NA	N/A	NA	-19.5	-9.2	-11.8	-10	3	38.5		1
Ras	Concrete truck	Haul Koad	CNP044	109	10	NA	NA	NA	96.1	15	5.1	NA	N/A	NA	- 19.8	-15.4	-11.8	-10	3	32.0		1
Rd9	Concrete truck	Haul Road	CNP044	109	5	NA	NA	NA	103.1	15	3.1	NA	N/A	NA	-20.1	-17.7	-11.8	-10	3	26.4		
Rd 10	Concrete truck	Haul Road	CNP044	109	5	NA	NA	NA	114.5	15	4.0	NA	N/A	NA	-20.6	-16.6	-11.8	-10	3	27.0		
Rd11	Concrete truck	Haul Road	CNP044	109	5	NA	NA	NA	116.7	15	2.6	NA	N/A	NA	-20.7	-18.4	-11.8	-10	3	25.1		
Rd12	Concrete truck	Haul Road	CNP044	109	5	NA	NA	NA	128.6	15	3.3	NA	N/A	NA	-21.1	-17.3	-11.8	-10	3	25.8		
Rd13	Concrete truck	Haul Road	CNP044	109	5	NA	NA	NA	128.6	15	3.4	NA	N/A	NA	-21.1	-17.3	-11.8	-10	3	25.8		

Notes:

1 A 10dB(A) noise reduction was applied, as IN1 is 1 storey building, and was blocked by the existing soild wall of the Open Storage located between the Site and the IN1, thus, no line of sight to the entire proposed development as well as the haul road.

#### Predicted Noise Level at IN2

						At course Naise	Noice Deduction									Corr	ection				Total	
	Fixed-noise Sources	Source Type	Ref. Code	SWL/unit, dB(A)	Quantity	Mitigation Measures	from Mitigation Measures, dB(A)	Corrected SWL, dB(A)	Distance, m	Speed, km/h	Angle of view, degree	Operation (30mins)	% on time	Time	Distance	Angle of view	Speed	Screening Effect, dB(A)	Facade	CNL, dB(A)	CNL, dB(A)	Criterion , dB(A)
	Concrete mixer 1(electric)	Stationary	CNP 045	96	1	Enclosure	-20	76.0	221.4	NA	NA	30	100.0%	0.0	-54.9	NA	NA	0	3	24.1		
	Concrete mixer 2(electric)	Stationary	CNP 045	96	1	Enclosure	-20	76.0	213.4	NA	NA	30	100.0%	0.0	-54.6	NA	NA	0	3	24.4		
Filter F	an (mixing unit) - blower (mixer 1)	Stationary	Other PME	95	1	Enclosure	-20	75.0	121.7	NA	NA	30	100.0%	0.0	-49.7	NA	NA	0	3	28.3		
Filter F	an (mixing unit) - blower (mixer 2)	Stationary	Other PME	95	1	Enclosure	-20	75.0	111.8	NA	NA	30	100.0%	0.0	-48.9	NA	NA	0	3	29.1		
	Truck (Concrete Collection)	Stationary	CNP044	109	5	Enclosure	-20	96.0	221.4	NA	NA	3	10.0%	-10.0	-54.9	NA	NA	0	3	34.1		
	Truck (Concrete Collection)	Stationary	CNP044	109	5	Enclosure	-20	96.0	213.4	NA	NA	3	10.0%	-10.0	-54.6	NA	NA	0	3	34.4		
	Conveyor / Screw Conveyors	Stationary	Other PME	90	2	Enclosure	-20	73.0	173.7	NA	NA	30	100.0%	0.0	-52.8	NA	NA	0	3	23.2		
	Conveyor / Screw Conveyors	Stationary	Other PME	90	1	Semi-enclosure	-15	75.0	145.3	NA	NA	30	100.0%	0.0	-51.2	NA	NA	0	3	26.8		
	Water / Admixture Pump	Stationary	CNP044	88	1	Semi-enclosure	-15	73.0	178.1	NA	NA	30	100.0%	0.0	-53.0	NA	NA	0	3	23.0		
	Pump (silo)	Stationary	CNP 281	88	4	Enclosure	-20	74.0	187.2	NA	NA	30	100.0%	0.0	-53.4	NA	NA	0	3	23.6		
Rd1	Concrete truck	Haul Road	CNP044	109	10	NA	NA	NA	213.4	15	2.8	NA	N/A	NA	-23.3	-18.1	-11.8	-10	3	25.8		
Rd2	Concrete truck	Haul Road	CNP044	109	10	NA	NA	NA	197.4	15	0.5	NA	N/A	NA	-23	-25.5	-11.8	0	3	28.7	45	48
Rd3	Concrete truck	Haul Road	CNP044	109	10	NA	NA	NA	177.4	15	0.6	NA	N/A	NA	-22.5	-25	-11.8	0	3	29.7		
Rd4	Concrete truck	Haul Road	CNP044	109	10	NA	NA	NA	164.3	15	2.6	NA	N/A	NA	-22.2	-18.5	-11.8	0	3	36.5		
Rd5	Concrete truck	Haul Road	CNP044	109	10	NA	NA	NA	161.9	15	4.0	NA	N/A	NA	-22.1	-16.6	-11.8	0	3	38.5		
Rd6	Concrete truck	Haul Road	CNP044	109	10	NA	NA	NA	167.7	15	2.9	NA	N/A	NA	-22.2	-18	-11.8	-10	3	27.0		
Rd7	Concrete truck	Haul Road	CNP044	109	10	NA	NA	NA	189.5	15	2.4	NA	N/A	NA	-22.8	-18.7	-11.8	-10	3	25.7		
Rd8	Concrete truck	Haul Road	CNP044	109	10	NA	NA	NA	211.2	15	0.8	NA	N/A	NA	-23.2	-23.3	-11.8	-10	3	20.7		
Rd9	Concrete truck	Haul Road	CNP044	109	5	NA	NA	NA	219.9	15	1.7	NA	N/A	NA	-23.4	-20.2	-11.8	-10	3	20.6		
Rd10	Concrete truck	Haul Road	CNP044	109	5	NA	NA	NA	222.5	15	4.1	NA	N/A	NA	-23.5	-16.4	-11.8	-10	3	24.3	]	
Rd11	Concrete truck	Haul Road	CNP044	109	5	NA	NA	NA	217.4	15	2.0	NA	N/A	NA	-23.4	-19.4	-11.8	-10	4	22.4		
Rd12	Concrete truck	Haul Road	CNP044	109	5	NA	NA	NA	220.2	15	4.1	NA	N/A	NA	-23.4	-16.4	-11.8	0	3	34.4		
Rd13	Concrete truck	Haul Road	CNP044	109	5	NA	NA	NA	220.3	15	4.1	NA	N/A	NA	-23.4	-16.4	-11.8	0	3	34.4		

### J24.00171.HK.01 S.16 Planning Application for Proposed Temporary Concrete Batching Plant with Ancillary Facilities for a Period of 5 Years at Lots 573 RP and 1710 in D.D. 114, Shek Kong, Yuen Long Environmental Assessment



#### Predicted Noise Level at IN3

		1											1	-						-	1	1
	Fixed-noise Sources	Source Type	Ref. Code	SWL/unit, dB(A)	Quantity	At-source Noise Mitigation Measures	Noise Reduction from Mitigation Measures, dB(A)	Corrected SWL, dB(A)	Distance, n	n Speed, km/h	Angle of view, degree	Operation (30mins)	% on time	Time	Distance	Cor Angle of view	Speed	Screening Effect, dB(A)	Facade	CNL, dB(A)	Total CNL, dB(A)	Criterion , dB(A)
	Concrete mixer 1(electric)	Stationary	CNP 045	96	1	Enclosure	-20	76.0	248.1	NA	NA	30	100.0%	0.0	-55.9	NA	NA	0	3	23.1		
	Concrete mixer 2(electric)	Stationary	CNP 045	96	1	Enclosure	-20	76.0	241.4	NA	NA	30	100.0%	0.0	-55.6	NA	NA	0	3	23.4	1	
Filter	Fan (mixing unit) - blower (mixer 1)	Stationary	Other PME	95	1	Enclosure	-20	75.0	121.7	NA	NA	30	100.0%	0.0	-49.7	NA	NA	0	3	28.3	1	
Filter	Fan (mixing unit) - blower (mixer 2)	Stationary	Other PME	95	1	Enclosure	-20	75.0	111.8	NA	NA	30	100.0%	0.0	-48.9	NA	NA	0	3	29.1	1	
	Truck (Concrete Collection)	Stationary	CNP044	109	5	Enclosure	-20	96.0	248.1	NA	NA	3	10.0%	-10.0	-55.9	NA	NA	0	3	33.1		
	Truck (Concrete Collection)	Stationary	CNP044	109	5	Enclosure	-20	96.0	241.4	NA	NA	3	10.0%	-10.0	-55.6	NA	NA	0	3	33.4		
	Conveyor / Screw Conveyors	Stationary	Other PME	90	2	Enclosure	-20	73.0	202.2	NA	NA	30	100.0%	0.0	-54.1	NA	NA	0	3	21.9		
	Conveyor / Screw Conveyors	Stationary	Other PME	90	1	Semi-enclosure	-15	75.0	182.5	NA	NA	30	100.0%	0.0	-53.2	NA	NA	0	3	24.8		
	Water / Admixture Pump	Stationary	CNP044	88	1	Semi-enclosure	-15	73.0	208.1	NA	NA	30	100.0%	0.0	-54.3	NA	NA	0	3	21.7		
	Pump (silo)	Stationary	CNP 281	88	4	Enclosure	-20	74.0	213.3	NA	NA	30	100.0%	0.0	-54.6	NA	NA	0	3	22.4		
Rd1	Concrete truck	Haul Road	CNP044	109	10	NA	NA	NA	238.7	15	2.9	NA	N/A	NA	-23.8	-18	-11.8	-10	3	25.4		
Rd2	Concrete truck	Haul Road	CNP044	109	10	NA	NA	NA	223.9	15	1.3	NA	N/A	NA	-23.5	-21.5	-11.8	-10	3	22.2	45	48
Rd3	Concrete truck	Haul Road	CNP044	109	10	NA	NA	NA	204.4	15	1.5	NA	N/A	NA	-23.1	-20.9	-11.8	0	3	33.2		
Rd4	Concrete truck	Haul Road	CNP044	109	10	NA	NA	NA	192.4	15	2.5	NA	N/A	NA	-22.8	-18.6	-11.8	0	3	35.8		
Rd5	Concrete truck	Haul Road	CNP044	109	10	NA	NA	NA	191.6	15	3.2	NA	N/A	NA	-22.8	-17.5	-11.8	0	3	36.9		
Rd6	Concrete truck	Haul Road	CNP044	109	10	NA	NA	NA	199.0	15	1.9	NA	N/A	NA	-23.0	-19.7	-11.8	0	3	34.5		
Rd7	Concrete truck	Haul Road	CNP044	109	10	NA	NA	NA	220.7	15	3.3	NA	N/A	NA	-23.4	-17.3	-11.8	0	3	36.5		
Rd8	Concrete truck	Haul Road	CNP044	109	10	NA	NA	NA	241.3	15	1.0	NA	N/A	NA	-23.8	-22.4	-11.8	-10	3	21.0		
Rd9	Concrete truck	Haul Road	CNP044	109	5	NA	NA	NA	249.3	15	1.7	NA	N/A	NA	-24.0	-20.1	-11.8	-10	3	20.1		
Rd10	Concrete truck	Haul Road	CNP044	109	5	NA	NA	NA	250.4	15	3.6	NA	N/A	NA	-24.0	-17	-11.8	-10	3	23.2		
Rd11	Concrete truck	Haul Road	CNP044	109	5	NA	NA	NA	244.8	15	2.1	NA	N/A	NA	-23.9	-19.4	-11.8	-10	3	20.9		
Rd12	Concrete truck	Haul Road	CNP044	109	5	NA	NA	NA	245.9	15	3.6	NA	N/A	NA	-23.9	-17	-11.8	0	3	33.3	]	
Rd13	Concrete truck	Haul Road	CNP044	109	5	NA	NA	NA	245.9	15	3.6	NA	N/A	NA	-23.9	-17	-11.8	0	3	33.3	1	

#### Predicted Noise Level at IN4

																Corr	ection					
	Fixed-noise Sources	Source Type	Ref. Code	SWL/unit, dB(A)	Quantity	At-source Noise Mitigation Measures	Noise Reduction from Mitigation Measures, dB(A)	Corrected SWL, dB(A)	Distance, m	Speed, km/h	Angle of view, degree	Operation (30mins)	% on time	Time	Distance	Angle of view	Speed	Screening Effect, dB(A) <sup>[1]</sup>	Facade	CNL, dB(A)	Total CNL, dB(A)	Criterion , dB(A)
	Concrete mixer 1(electric)	Stationary	CNP 045	96	1	Enclosure	-20	76.0	253.5	NA	NA	30	100.0%	0.0	-56.1	NA	NA	0	3	22.9		
	Concrete mixer 2(electric)	Stationary	CNP 045	96	1	Enclosure	-20	76.0	259.9	NA	NA	30	100.0%	0.0	-56.3	NA	NA	0	3	22.7		
Filter I	an (mixing unit) - blower (mixer 1)	Stationary	Other PME	95	1	Enclosure	-20	75.0	121.7	NA	NA	30	100.0%	0.0	-49.7	NA	NA	0	3	28.3		
Filter	Fan (mixing unit) - blower (mixer 2)	Stationary	Other PME	95	1	Enclosure	-20	75.0	111.8	NA	NA	30	100.0%	0.0	-48.9	NA	NA	0	3	29.1		
	Truck (Concrete Collection)	Stationary	CNP044	109	5	Enclosure	-20	96.0	253.5	NA	NA	3	10.0%	-10.0	-56.1	NA	NA	0	3	32.9		
	Truck (Concrete Collection)	Stationary	CNP044	109	5	Enclosure	-20	96.0	259.9	NA	NA	3	10.0%	-10.0	-56.3	NA	NA	0	3	32.7		
	Conveyor / Screw Conveyors	Stationary	Other PME	90	2	Enclosure	-20	73.0	245.1	NA	NA	30	100.0%	0.0	-55.8	NA	NA	0	3	20.2		
	Conveyor / Screw Conveyors	Stationary	Other PME	90	1	Semi-enclosure	-15	75.0	272.1	NA	NA	30	100.0%	0.0	-56.7	NA	NA	0	4	22.3		
	Water / Admixture Pump	Stationary	Other PME	88	1	Semi-enclosure	-15	73.0	294.0	NA	NA	30	100.0%	0.0	-57.4	NA	NA	0	3	18.6		
	Pump (silo)	Stationary	CNP 281	88	4	Enclosure	-20	74.0	288.4	NA	NA	30	100.0%	0.0	-57.2	NA	NA	0	3	19.8		
Rd1	Concrete truck	Haul Road	CNP044	109	10	NA	NA	NA	263.9	15	2.9	NA	N/A	NA	-24.2	-17.9	-11.8	-10	3	25.1		
Rd2	Concrete truck	Haul Road	CNP044	109	10	NA	NA	NA	277.8	15	1.3	NA	N/A	NA	-24.4	-21.4	-11.8	-10	3	21.4	39	52
Rd3	Concrete truck	Haul Road	CNP044	109	10	NA	NA	NA	296.9	15	1.1	NA	N/A	NA	-24.7	-22.1	-11.8	-10	3	20.4		
Rd4	Concrete truck	Haul Road	CNP044	109	10	NA	NA	NA	309.0	15	1.5	NA	N/A	NA	-24.9	-20.7	-11.8	-10	3	21.6		
Rd5	Concrete truck	Haul Road	CNP044	109	10	NA	NA	NA	310.4	15	2.0	NA	N/A	NA	-24.9	-19.5	-11.8	-10	3	22.8		
Rd6	Concrete truck	Haul Road	CNP044	109	10	NA	NA	NA	304.4	15	1.6	NA	N/A	NA	-24.8	-20.4	-11.8	-10	3	22.0		
Rd7	Concrete truck	Haul Road	CNP044	109	10	NA	NA	NA	282.6	15	1.4	NA	N/A	NA	-24.5	-21.1	-11.8	-10	3	21.6		
Rd8	Concrete truck	Haul Road	CNP044	109	10	NA	NA	NA	260.9	15	0.7	NA	N/A	NA	-24.2	-24	-11.8	-10	3	19.0		
Rd9	Concrete truck	Haul Road	CNP044	109	5	NA	NA	NA	252.4	15	1.6	NA	N/A	NA	-24.0	-20.5	-11.8	-10	3	19.7		
Rd10	Concrete truck	Haul Road	CNP044	109	5	NA	NA	NA	250.9	15	3.5	NA	N/A	NA	-24.0	-17.1	-11.8	-10	3	23.1		
Rd11	Concrete truck	Haul Road	CNP044	109	5	NA	NA	NA	256.6	15	2.0	NA	N/A	NA	-24.1	-19.5	-11.8	-10	3	20.6		1
Rd12	Concrete truck	Haul Road	CNP044	109	5	NA	NA	NA	256.6	15	3.3	NA	N/A	NA	-24.1	-17.4	-11.8	-10	3	22.7		
Rd13	Concrete truck	Haul Road	CNP044	109	5	NA	NA	NA	256.5	15	3.3	NA	N/A	NA	-24.1	-17.4	-11.8	-10	3	22.7		1

Notes:

1 A 10dB(A) noise reduction was applied, as 1N4 is 1 storey building, and was blocked by the existing buildings located between the Site and the IN4, thus, no line of sight to the entire proposed development as well as the haul road.



### Daytime – Scenario 2

#### Predicted Noise Level at IN1

						At course Noise	Noice Reduction									Corr	rection			_	Total	
	Fixed-noise Sources	Source Type	Ref. Code	SWL/unit, dB(A)	Quantity	Mitigation Measures	from Mitigation Measures, dB(A)	Corrected SWL, dB(A)	Distance, m	Speed, km/h	Angle of view, degree	Operation (30mins)	% on time	Time	Distance	Angle of view	Speed	Screening Effect, dB(A) <sup>[1]</sup>	Facade	CNL, dB(A)	CNL, dB(A)	Criterion , dB(A)
	Conveyor / Screw Conveyors	Stationary	CNP 041	90	2	Enclosure	-20	73.0	103.4	NA	NA	30	100.0%	0.0	-48.3	NA	NA	0	3	27.7		
	Conveyor / Screw Conveyors	Stationary	CNP 041	90	1	Semi-enclosure	-15	75.0	92.7	NA	NA	30	100.0%	0.0	-47.3	NA	NA	0	3	30.7	'	
	Water / Admixture Pump	Stationary	CNP 281	88	1	Semi-enclosure	-15	73.0	98.1	NA	NA	30	100.0%	0.0	-47.8	NA	NA	0	3	28.2	] '	
P	ump (mounted on unloading tank)	Stationary	CNP 281	88	4	Enclosure	-20	74.0	120.8	NA	NA	30	100.0%	0.0	-49.6	NA	NA	0	3	27.4	_ '	
	Filter Fan (silo) - blower	Stationary	Other PME	95	4	Semi-enclosure	-15	86.0	120.8	NA	NA	30	100.0%	0.0	-49.6	NA	NA	0	3	39.4	] '	
	Aggregates Truck	Haul Road	Other PME	105	7	NA	NA	NA	128.9	15	3.6	NA	N/A	NA	-21.1	-17.0	-11.8	-10	3	23.6	1 '	
Rd1	Cement / PFA / GGBS/ Admxture tanker	Haul Road	Other PME	105	2	NA	NA	NA	128.9	15	3.6	NA	N/A	NA	-21.1	-17	-11.8	-10	3	18.1	_ '	
	Lorry	Haul Road	Other PME	105	7	NA	NA	NA	128.9	15	3.6	NA	N/A	NA	-21.1	-17	-11.8	-10	3	23.6	'	
Pd3	Aggregates Truck	Haul Road	Other PME	105	9	NA	NA	NA	119.3	15	9.4	NA	N/A	NA	-20.8	-12.8	-11.8	-10	3	29.1	] '	
Ruz	Cement / PFA / GGBS/ Admxture tanker	Haul Road	Other PME	105	2	NA	NA	NA	119.3	15	9.4	NA	N/A	NA	-20.8	-12.8	-11.8	-10	3	22.6	1	
	Aggregates Truck	Haul Road	Other PME	105	7	NA	NA	NA	115.8	15	9.8	NA	N/A	NA	-20.6	-12.6	-11.8	-10	3	28.5	43	46
Rd3	Cement / PFA / GGBS/ Admxture tanker	Haul Road	Other PME	105	1	NA	NA	NA	115.8	15	9.8	NA	N/A	NA	-20.6	-12.6	-11.8	-10	3	20.0		10
	Lorry	Haul Road	Other PME	105	1	NA	NA	NA	115.8	15	9.8	NA	N/A	NA	-20.6	-12.6	-11.8	-10	3	20.0	] '	
	Aggregates Truck	Haul Road	Other PME	105	4	NA	NA	NA	112.1	15	3.7	NA	N/A	NA	-20.5	-16.8	-11.8	-10	3	21.9	_ '	
Rd4	Cement / PFA / GGBS/ Admxture tanker	Haul Road	Other PME	105	1	NA	NA	NA	112.1	15	3.7	NA	N/A	NA	-20.5	-16.8	-11.8	-10	3	15.9	1 '	
	Lorry	Haul Road	Other PME	105	5	NA	NA	NA	112.1	15	3.7	NA	N/A	NA	-20.5	-16.8	-11.8	-10	3	22.9	] '	
	Aggregates Truck	Haul Road	Other PME	105	4	NA	NA	NA	103.4	15	0.7	NA	N/A	NA	-20.1	-24.2	-11.8	-10	3	14.9	_ '	
Rd5	Cement / PFA / GGBS/ Admxture tanker	Haul Road	Other PME	105	1	NA	NA	NA	103.4	15	0.7	NA	N/A	NA	-20.1	-24.2	-11.8	-10	3	8.9	1	
	Lorry	Haul Road	Other PME	105	5	NA	NA	NA	103.4	15	0.7	NA	N/A	NA	-20.1	-24.2	-11.8	-10	3	15.9	_ '	
Rd6	Lorry	Haul Road	Other PME	105	9	NA	NA	NA	92.4	15	4.6	NA	N/A	NA	-19.7	-15.9	-11.8	-10	3	27.1	1	
Rd7	Lorry	Haul Road	Other PME	105	9	NA	NA	NA	88.7	15	21.7	NA	N/A	NA	-19.5	-9.2	-11.8	- 10	3	34.0	1 '	
Rd14	Lorry	Haul Road	Other PME	105	9	NA	NA	NA	107.9	15	2.7	NA	N/A	NA	-20.3	-18.3	-11.8	-10	3	24.1		

Notes:

1 A 10dB(A) noise reduction was applied, as IN1 is 1 storey building, and was blocked by the existing soild wall of the Open Storage located between the Site and the IN1, thus, no line of sight to the entire proposed development as well as the haul road.

#### Predicted Noise Level at IN2

																Corr	ection					
	Fixed-noise Sources	Source Type	Ref. Code	SWL/unit, dB(A)	Quantity	At-source Noise Mitigation Measures	from Mitigation Measures, dB(A)	Corrected SWL, dB(A)	Distance, m	Speed, km/h	Angle of view, degree	Operation (30mins)	% on time	Time	Distance	Angle of view	Speed	Screening Effect, dB(A)	Facade	CNL, dB(A)	Dotal CNL, dB(A)	Criterion , dB(A)
	Conveyor / Screw Conveyors	Stationary	CNP 041	90	2	Enclosure	-20	73.0	173.7	NA	NA	30	100.0%	0.0	-52.8	NA	NA	0	3	23.2		
	Conveyor / Screw Conveyors	Stationary	CNP 041	90	1	Semi-enclosure	-15	75.0	145.3	NA	NA	30	100.0%	0.0	-51.2	NA	NA	0	3	26.8		
	Water / Admixture Pump	Stationary	CNP 281	88	1	Semi-enclosure	-15	73.0	178.1	NA	NA	30	100.0%	0.0	-53.0	NA	NA	0	3	23.0		
P	ump (mounted on unloading tank)	Stationary	CNP 281	88	4	Enclosure	-20	74.0	187.2	NA	NA	30	100.0%	0.0	-53.4	NA	NA	0	3	23.6		
	Filter Fan (silo) - blower	Stationary	Other PME	95	4	Semi-enclosure	-15	86.0	187.2	NA	NA	30	100.0%	0.0	-53.4	NA	NA	0	3	35.6		
	Aggregates Truck	Haul Road	Other PME	105	7	NA	NA	NA	213.4	15	2.8	NA	N/A	NA	-23.3	-18.1	-11.8	-10	3	20.3		
Rd1	Cement / PFA / GGBS/ Admxture tanker	Haul Road	Other PME	105	2	NA	NA	NA	213.4	15	2.8	NA	N/A	NA	-23.3	-18.1	-11.8	- 10	3	14.8		1
	Lorry	Haul Road	Other PME	105	7	NA	NA	NA	213.4	15	2.8	NA	N/A	NA	-23.3	-18.1	-11.8	-10	3	20.3		
Pd2	Aggregates Truck	Haul Road	Other PME	105	9	NA	NA	NA	197.4	15	0.5	NA	N/A	NA	-23	-25.5	-11.8	0	3	24.2		
Ruz	Cement / PFA / GGBS/ Admxture tanker	Haul Road	Other PME	105	2	NA	NA	NA	197.4	15	0.5	NA	N/A	NA	-23	-25.5	-11.8	0	3	17.7		1
	Aggregates Truck	Haul Road	Other PME	105	7	NA	NA	NA	177.4	15	0.6	NA	N/A	NA	-22.5	-25	-11.8	0	3	24.2	40	48
Rd3	Cement / PFA / GGBS/ Admxture tanker	Haul Road	Other PME	105	1	NA	NA	NA	177.4	15	0.6	NA	N/A	NA	-22.5	-25	-11.8	0	3	15.7	-10	
	Lorry	Haul Road	Other PME	105	1	NA	NA	NA	177.4	15	0.6	NA	N/A	NA	-22.5	-25	-11.8	0	3	15.7		1
	Aggregates Truck	Haul Road	Other PME	105	4	NA	NA	NA	164.3	15	2.6	NA	N/A	NA	-22.2	-18.5	-11.8	0	3	28.5		1
Rd4	Cement / PFA / GGBS/ Admxture tanker	Haul Road	Other PME	105	1	NA	NA	NA	164.3	15	2.6	NA	N/A	NA	-22.2	-18.5	-11.8	0	3	22.5		
	Lorry	Haul Road	Other PME	105	5	NA	NA	NA	164.3	15	2.6	NA	N/A	NA	-22.2	-18.5	-11.8	0	3	29.5		
	Aggregates Truck	Haul Road	Other PME	105	4	NA	NA	NA	161.9	15	4.0	NA	N/A	NA	-22.1	-16.6	-11.8	0	3	30.5		
Rd5	Cement / PFA / GGBS/ Admxture tanker	Haul Road	Other PME	105	1	NA	NA	NA	161.9	15	4.0	NA	N/A	NA	-22.1	-16.6	-11.8	0	3	24.5		
	Lorry	Haul Road	Other PME	105	5	NA	NA	NA	161.9	15	4.0	NA	N/A	NA	-22.1	-16.6	-11.8	0	3	31.5		1
Rd6	Lorry	Haul Road	Other PME	105	9	NA	NA	NA	167.7	15	2.9	NA	N/A	NA	-22.2	-18	-11.8	-10	3	22.5		
Rd7	Lorry	Haul Road	Other PME	105	9	NA	NA	NA	189.5	15	2.4	NA	N/A	NA	-22.8	-18.7	-11.8	-10	3	21.2		
Rd14	Lorry	Haul Road	Other PME	105	9	NA	NA	NA	206.4	15	8.0	NA	N/A	NA	-23.1	-13.5	-11.8	-10	3	26.1		i i

### J24.00171.HK.01 S.16 Planning Application for Proposed Temporary Concrete Batching Plant with Ancillary Facilities for a Period of 5 Years at Lots 573 RP and 1710 in D.D. 114, Shek Kong, Yuen Long Environmental Assessment



#### Predicted Noise Level at IN3

						At cource Noice	Noico Poduction									Corr	ection				Total	
	Fixed-noise Sources	Source Type	Ref. Code	SWL/unit, dB(A)	Quantity	Mitigation Measures	from Mitigation Measures, dB(A)	Corrected SWL, dB(A)	Distance, m	Speed, km/h	Angle of view, degree	Operation (30mins)	% on time	Time	Distance	Angle of view	Speed	Screening Effect, dB(A)	Facade	CNL, dB(A)	CNL, dB(A)	Criterion , dB(A)
	Conveyor / Screw Conveyors	Stationary	CNP 041	90	2	Enclosure	-20	73.0	202.2	NA	NA	30	100.0%	0.0	-54.1	NA	NA	0	3	21.9		
	Conveyor / Screw Conveyors	Stationary	CNP 041	90	1	Semi-enclosure	-15	75.0	182.5	NA	NA	30	100.0%	0.0	-53.2	NA	NA	0	3	24.8		
	Water / Admixture Pump	Stationary	CNP 281	88	1	Semi-enclosure	-15	73.0	208.1	NA	NA	30	100.0%	0.0	-54.3	NA	NA	0	3	21.7		
P	ump (mounted on unloading tank)	Stationary	CNP 281	88	4	Enclosure	-20	74.0	213.3	NA	NA	30	100.0%	0.0	-54.6	NA	NA	0	3	22.4		
	Filter Fan (silo) - blower	Stationary	Other PME	95	4	Semi-enclosure	-15	86.0	213.3	NA	NA	30	100.0%	0.0	-54.6	NA	NA	0	3	34.4		
	Aggregates Truck	Haul Road	Other PME	105	7	NA	NA	NA	238.7	15	2.9	NA	N/A	NA	-23.8	-18.0	-11.8	-10	3	19.9		
Rd1	Cement / PFA / GGBS/ Admxture tanker	Haul Road	Other PME	105	2	NA	NA	NA	238.7	15	2.9	NA	N/A	NA	-23.8	-18	-11.8	-10	3	14.4		
	Lorry	Haul Road	Other PME	105	7	NA	NA	NA	238.7	15	2.9	NA	N/A	NA	-23.8	-18	-11.8	-10	3	19.9		
Pd2	Aggregates Truck	Haul Road	Other PME	105	9	NA	NA	NA	223.9	15	1.3	NA	N/A	NA	-23.5	-21.5	-11.8	-10	3	17.7		
Ru2	Cement / PFA / GGBS/ Admxture tanker	Haul Road	Other PME	105	2	NA	NA	NA	223.9	15	1.3	NA	N/A	NA	-23.5	-21.5	-11.8	-10	3	11.2		
	Aggregates Truck	Haul Road	Other PME	105	7	NA	NA	NA	204.4	15	1.5	NA	N/A	NA	-23.1	-20.9	-11.8	0	3	27.7	40	48
Rd3	Cement / PFA / GGBS/ Admxture tanker	Haul Road	Other PME	105	1	NA	NA	NA	204.4	15	1.5	NA	N/A	NA	-23.1	-20.9	-11.8	0	3	19.2	-10	10
	Lorry	Haul Road	Other PME	105	1	NA	NA	NA	204.4	15	1.5	NA	N/A	NA	-23.1	-20.9	-11.8	0	3	19.2		
	Aggregates Truck	Haul Road	Other PME	105	4	NA	NA	NA	192.4	15	2.5	NA	N/A	NA	-22.8	-18.6	-11.8	0	3	27.8		
Rd4	Cement / PFA / GGBS/ Admxture tanker	Haul Road	Other PME	105	1	NA	NA	NA	192.4	15	2.5	NA	N/A	NA	-22.8	-18.6	-11.8	0	3	21.8		
	Lorry	Haul Road	Other PME	105	5	NA	NA	NA	192.4	15	2.5	NA	N/A	NA	-22.8	-18.6	-11.8	0	3	28.8		
	Aggregates Truck	Haul Road	Other PME	105	4	NA	NA	NA	191.6	15	3.2	NA	N/A	NA	-22.8	-17.5	-11.8	0	3	28.9		
Rd5	Cement / PFA / GGBS/ Admxture tanker	Haul Road	Other PME	105	1	NA	NA	NA	191.6	15	3.2	NA	N/A	NA	-22.8	-17.5	-11.8	0	3	22.9		
	Lorry	Haul Road	Other PME	105	5	NA	NA	NA	191.6	15	3.2	NA	N/A	NA	-22.8	-17.5	-11.8	0	3	29.9		
Rd6	Lorry	Haul Road	Other PME	105	9	NA	NA	NA	199.0	15	1.9	NA	N/A	NA	-23	-19.7	-11.8	0	3	30.0		
Rd7	Lorry	Haul Road	Other PME	105	9	NA	NA	NA	220.7	15	3.3	NA	N/A	NA	-23.4	-17.3	-11.8	0	3	32.0		
Rd14	Lorry	Haul Road	Other PME	105	9	NA	NA	NA	234.8	15	7.0	NA	N/A	NA	-23.7	-14.1	-11.8	-10	3	24.9	1	

#### Predicted Noise Level at IN4

														% on Correction								
	Fixed-noise Sources	Source Type	Ref. Code	SWL/unit, dB(A)	Quantity	At-source Noise Mitigation Measures	Noise Reduction from Mitigation Measures, dB(A)	Corrected SWL, dB(A)	Distance, m	Speed, km/h	Angle of view, degree	Operation (30mins)	% on time	Time	Distance	Angle of view	Speed	Screening Effect, dB(A) <sup>[1]</sup>	Facade	CNL, dB(A)	Total CNL, dB(A)	Criterion , dB(A)
	Conveyor / Screw Conveyors	Stationary	CNP 041	90	2	Enclosure	-20	73.0	245.1	NA	NA	30	100.0%	0.0	-55.8	NA	NA	0	3	20.2		
	Conveyor / Screw Conveyors	Stationary	CNP 041	90	1	Semi-enclosure	-15	75.0	272.1	NA	NA	30	100.0%	0.0	-56.7	NA	NA	0	3	21.3		
	Water / Admixture Pump	Stationary	CNP 281	88	1	Semi-enclosure	-15	73.0	294.0	NA	NA	30	100.0%	0.0	-57.4	NA	NA	0	3	18.6		
	Pump (mounted on unloading tank)	Stationary	CNP 281	88	4	Enclosure	-20	74.0	288.4	NA	NA	30	100.0%	0.0	-57.2	NA	NA	0	3	19.8		
	Filter Fan (silo) - blower	Stationary	CNP044	109	4	Semi-enclosure	-15	100.0	288.4	NA	NA	5	16.7%	-7.8	-57.2	NA	NA	0	3	38.0		
	Aggregates Truck	Haul Road	Other PME	105	7	NA	NA	NA	263.9	15	2.9	NA	N/A	NA	-24.2	-17.9	-11.8	-10	3	19.6		
Rd1	Cement / PFA / GGBS/ Admxture tanker	Haul Road	Other PME	105	2	NA	NA	NA	263.9	15	2.9	NA	N/A	NA	-24.2	-17.9	-11.8	-10	3	14.1		
	Lorry	Haul Road	Other PME	105	7	NA	NA	NA	263.9	15	2.9	NA	N/A	NA	-24.2	-17.9	-11.8	-10	3	19.6		
Ch C	Aggregates Truck	Haul Road	Other PME	105	9	NA	NA	NA	277.8	15	1.3	NA	N/A	NA	-24.4	-21.4	-11.8	-10	3	16.9		
Ruz	Cement / PFA / GGBS/ Admxture tanker	Haul Road	Other PME	105	2	NA	NA	NA	277.8	15	1.3	NA	N/A	NA	-24.4	-21.4	-11.8	-10	3	10.4		
	Aggregates Truck	Haul Road	Other PME	105	7	NA	NA	NA	296.9	15	1.1	NA	N/A	NA	-24.7	-22.1	-11.8	-10	3	14.9	39	52
Rd3	Cement / PFA / GGBS/ Admxture tanker	Haul Road	Other PME	105	1	NA	NA	NA	296.9	15	1.1	NA	N/A	NA	-24.7	-22.1	-11.8	-10	3	6.4	35	52
	Lorry	Haul Road	Other PME	105	1	NA	NA	NA	296.9	15	1.1	NA	N/A	NA	-24.7	-22.1	-11.8	-10	3	6.4		
	Aggregates Truck	Haul Road	Other PME	105	4	NA	NA	NA	309.0	15	1.5	NA	N/A	NA	-24.9	-20.7	-11.8	-10	3	13.6		
Rd4	Cement / PFA / GGBS/ Admxture tanker	Haul Road	Other PME	105	1	NA	NA	NA	309.0	15	1.5	NA	N/A	NA	-24.9	-20.7	-11.8	-10	3	7.6		
	Lorry	Haul Road	Other PME	105	5	NA	NA	NA	309.0	15	1.5	NA	N/A	NA	-24.9	-20.7	-11.8	-10	3	14.6		
	Aggregates Truck	Haul Road	Other PME	105	4	NA	NA	NA	310.4	15	2.0	NA	N/A	NA	-24.9	-19.5	-11.8	-10	3	14.8		
Rd5	Cement / PFA / GGBS/ Admxture tanker	Haul Road	Other PME	105	1	NA	NA	NA	310.4	15	2.0	NA	N/A	NA	-24.9	-19.5	-11.8	-10	3	8.8		
	Lorry	Haul Road	Other PME	105	5	NA	NA	NA	310.4	15	2.0	NA	N/A	NA	-24.9	-19.5	-11.8	-10	3	15.8		
Rd6	Lorry	Haul Road	Other PME	105	9	NA	NA	NA	304.4	15	1.6	NA	N/A	NA	-24.8	-20.4	-11.8	-10	3	17.5		
Rd7	Lorry	Haul Road	Other PME	105	9	NA	NA	NA	282.6	15	1.4	NA	N/A	NA	-24.5	-21.1	-11.8	-10	3	17.1		
Rd14	Lorry	Haul Road	Other PME	105	9	NA	NA	NA	266.5	15	6.2	NA	N/A	NA	-24.3	-14.6	-11.8	-10	3	23.8		

Notes:

1 A 10dB(A) noise reduction was applied, as IN4 is 1 storey building, and was blocked by the existing buildings located between the Site and the IN4, thus, no line of sight to the entire proposed development as well as the haul road.



### Evening – Scenario 1

#### Predicted Noise Level at IN1

						At-source Noise	Noise Reduction									Corr	rection				Total	
	Fixed-noise Sources	Source Type	Ref. Code	SWL/unit, dB(A)	Quantity	Mitigation Measures	from Mitigation Measures, dB(A)	Corrected SWL, dB(A)	Distance, m	Speed, km/h	Angle of view, degree	Operation (30mins)	% on time	Time	Distance	Angle of view	Speed	Screening Effect, dB(A) <sup>[1]</sup>	Facade	CNL, dB(A)	CNL, dB(A)	Criterion , dB(A)
	Concrete mixer 2(electric)	Stationary	CNP 045	96	1	Enclosure	-20	76.0	111.8	NA	NA	30	100.0%	0.0	-48.9	NA	NA	0	3	30.1	· · · · ·	
Filter	Fan (mixing unit) - blower (mixer 2)	Stationary	Other PME	95	1	Enclosure	-20	75.0	111.8	NA	NA	30	100.0%	0.0	-48.9	NA	NA	0	3	29.1	, · ·	
	Truck (Concrete Collection)	Stationary	CNP044	109	1	Enclosure	-20	89.0	111.8	NA	NA	3	10.0%	-10.0	-48.9	NA	NA	0	3	33.1	, I	1
	Conveyor / Screw Conveyors	Stationary	CNP 041	90	2	Enclosure	-20	73.0	103.4	NA	NA	30	100.0%	0.0	-48.3	NA	NA	0	3	27.7	۱ <sup>۱</sup>	1
	Conveyor / Screw Conveyors	Stationary	CNP 041	90	1	Semi-enclosure	-15	75.0	92.7	NA	NA	30	100.0%	0.0	-47.3	NA	NA	0	3	30.7	۱ <sup>۱</sup>	
	Water / Admixture Pump	Stationary	CNP 281	88	1	Semi-enclosure	-15	73.0	98.1	NA	NA	30	100.0%	0.0	-47.8	NA	NA	0	3	28.2	۱ <sup>۱</sup>	1
	Pump (silo)	Stationary	CNP 281	88	2	Enclosure	-20	71.0	120.8	NA	NA	30	100.0%	0.0	-49.6	NA	NA	0	3	24.4	4 <sup>,</sup>	
Rd1	Concrete Truck	Haul Road	CNP044	109	2	NA	NA	NA	128.9	15	3.6	NA	N/A	NA	-21.1	-17	-11.8	-10	3	22.1	4 '	1
Rd2	Concrete truck	Haul Road	CNP044	109	2	NA	NA	NA	119.3	15	9.4	NA	N/A	NA	-20.8	-12.8	-11.8	-10	3	26.6	, '	
Rd3	Concrete truck	Haul Road	CNP044	109	2	NA	NA	NA	115.8	15	9.8	NA	N/A	NA	-20.6	-12.6	-11.8	-10	3	27.0		
Rd4	Concrete truck	Haul Road	CNP044	109	2	NA	NA	NA	112.1	15	3.7	NA	N/A	NA	-20.5	-16.8	-11.8	-10	3	22.9	40	40
Rd5	Concrete truck	Haul Road	CNP044	109	2	NA	NA	NA	103.4	15	0.7	NA	N/A	NA	-20.1	-24.2	-11.8	-10	3	15.9	, I	1
Rd6	Concrete truck	Haul Road	CNP044	109	2	NA	NA	NA	92.4	15	4.6	NA	N/A	NA	-19.7	-15.9	-11.8	-10	3	24.6	, · · ·	
Rd7	Concrete truck	Haul Road	CNP044	109	2	NA	NA	NA	88.7	15	21.7	NA	N/A	NA	-19.5	-9.2	-11.8	-10	3	31.5		1
Rd8	Concrete truck	Haul Road	CNP044	109	2	NA	NA	NA	96.1	15	5.1	NA	N/A	NA	-19.8	-15.4	-11.8	-10	3	25.0	۱	
Rd9	Concrete truck	Haul Road	CNP044	109	2	NA	NA	NA	103.1	15	3.1	NA	N/A	NA	-20.1	-17.7	-11.8	-10	3	22.4	1 '	
Rd10	Concrete truck	Haul Road	CNP044	109	2	NA	NA	NA	114.5	15	4.0	NA	N/A	NA	-20.6	-16.6	-11.8	-10	3	23.0	1 '	
Rd11	Concrete truck	Haul Road	CNP044	109	2	NA	NA	NA	116.7	15	2.6	NA	N/A	NA	-20.7	-18.4	-11.8	-10	3	21.1	, <sup>,</sup>	1
Rd12	Concrete truck	Haul Road	CNP044	109	2	NA	NA	NA	128.6	15	3.3	NA	N/A	NA	-21.1	-17.3	-11.8	-10	3	21.8	<u>ا</u> ا	1
Rd13	Concrete truck	Haul Road	CNP044	109	2	NA	NA	NA	128.6	15	3.4	NA	N/A	NA	-21.1	-17.3	-11.8	-10	3	21.8	۱	1

Notes: 1 A 10dB(A) noise reduction was applied, as IN1 is 1 storey building, and was blocked by the existing solid wall of the Open Storage located between the Site and the IN1, thus, no line of sight to the entire proposed development as well as the haul road.

#### Predicted Noise Level at IN2

																Corr	rection					
	Fixed-noise Sources	Source Type	Ref. Code	SWL/unit, dB(A)	Quantity	At-source Noise Mitigation Measures	from Mitigation Measures, dB(A)	Corrected SWL, dB(A)	Distance, m	Speed, km/h	Angle of view, degree	Operation (30mins)	% on time	Time	Distance	Angle of view	Speed	Screening Effect, dB(A)	Facade	CNL, dB(A)	CNL, dB(A)	Criterion , dB(A)
	Concrete mixer 1(electric)	Stationary	CNP 045	96	0	Enclosure	-20	0.0	221.4	NA	NA	30	100.0%	0.0	-54.9	NA	NA	0	3	-51.9		
	Concrete mixer 2(electric)	Stationary	CNP 045	96	1	Enclosure	-20	76.0	213.4	NA	NA	30	100.0%	0.0	-54.6	NA	NA	0	3	24.4		
Filter F	an (mixing unit) - blower (mixer 1)	Stationary	Other PME	95	0	Enclosure	-20	0.0	121.7	NA	NA	30	100.0%	0.0	-49.7	NA	NA	0	3	-46.7		
Filter	an (mixing unit) - blower (mixer 2)	Stationary	Other PME	95	1	Enclosure	-20	75.0	111.8	NA	NA	30	100.0%	0.0	-48.9	NA	NA	0	3	29.1		
	Truck (Concrete Collection)	Stationary	CNP044	109	0	sed Semi-enclosu	-20	0.0	221.4	NA	NA	2	6.7%	-11.8	-54.9	NA	NA	0	3	-63.7		
	Truck (Concrete Collection)	Stationary	CNP044	109	1	Enclosure	-20	89.0	213.4	NA	NA	3	10.0%	-10.0	-54.6	NA	NA	0	3	27.4		
	Conveyor / Screw Conveyors	Stationary	Other PME	90	2	Enclosure	-20	73.0	173.7	NA	NA	30	100.0%	0.0	-52.8	NA	NA	0	3	23.2		
	Conveyor / Screw Conveyors	Stationary	Other PME	90	1	Semi-enclosure	-15	75.0	145.3	NA	NA	30	100.0%	0.0	-51.2	NA	NA	0	3	26.8		
	Water / Admixture Pump	Stationary	CNP044	88	1	Semi-enclosure	-15	73.0	178.1	NA	NA	30	100.0%	0.0	-53.0	NA	NA	0	3	23.0		
	Pump (silo)	Stationary	CNP 281	88	2	Enclosure	-20	71.0	187.2	NA	NA	30	100.0%	0.0	-53.4	NA	NA	0	3	20.6		
Rd1	Concrete truck	Haul Road	CNP044	109	2	NA	NA	NA	213.4	15	2.8	NA	N/A	NA	-23.3	-18.1	-11.8	-10	3	18.8		
Rd2	Concrete truck	Haul Road	CNP044	109	2	NA	NA	NA	197.4	15	0.5	NA	N/A	NA	-23	-25.5	-11.8	0	3	21.7	39	47
Rd3	Concrete truck	Haul Road	CNP044	109	2	NA	NA	NA	177.4	15	0.6	NA	N/A	NA	-22.5	-25	-11.8	0	3	22.7		
Rd4	Concrete truck	Haul Road	CNP044	109	2	NA	NA	NA	164.3	15	2.6	NA	N/A	NA	-22.2	-18.5	-11.8	0	3	29.5		
Rd5	Concrete truck	Haul Road	CNP044	109	2	NA	NA	NA	161.9	15	4.0	NA	N/A	NA	-22.1	-16.6	-11.8	0	3	31.5		
Rd6	Concrete truck	Haul Road	CNP044	109	2	NA	NA	NA	167.7	15	2.9	NA	N/A	NA	-22.2	-18	-11.8	-10	3	20.0		
Rd7	Concrete truck	Haul Road	CNP044	109	2	NA	NA	NA	189.5	15	2.4	NA	N/A	NA	-22.8	-18.7	-11.8	-10	3	18.7		
Rd8	Concrete truck	Haul Road	CNP044	109	2	NA	NA	NA	211.2	15	0.8	NA	N/A	NA	-23.2	-23.3	-11.8	-10	3	13.7		
Rd9	Concrete truck	Haul Road	CNP044	109	2	NA	NA	NA	219.9	15	1.7	NA	N/A	NA	-23.4	-20.2	-11.8	-10	3	16.6		
Rd10	Concrete truck	Haul Road	CNP044	109	2	NA	NA	NA	222.5	15	4.1	NA	N/A	NA	-23.5	-16.4	-11.8	-10	3	20.3		
Rd11	Concrete truck	Haul Road	CNP044	109	2	NA	NA	NA	217.4	15	2.0	NA	N/A	NA	-23.4	-19.4	-11.8	-10	4	18.4		
Rd12	Concrete truck	Haul Road	CNP044	109	2	NA	NA	NA	220.2	15	4.1	NA	N/A	NA	-23.4	-16.4	-11.8	0	3	30.4		
Rd13	Concrete truck	Haul Road	CNP044	109	2	NA	NA	NA	220.3	15	4.1	NA	N/A	NA	-23.4	-16.4	-11.8	0	3	30.4		

### J24.00171.HK.01 S.16 Planning Application for Proposed Temporary Concrete Batching Plant with Ancillary Facilities for a Period of 5 Years at Lots 573 RP and 1710 in D.D. 114, Shek Kong, Yuen Long Environmental Assessment



#### Predicted Noise Level at IN3

		1	1	1	1	1		1	1											1	1	1
	Fixed-noise Sources	Source Type	Ref. Code	SWL/unit, dB(A)	Quantity	At-source Noise Mitigation Measures	Noise Reduction from Mitigation Measures, dB(A)	Corrected SWL, dB(A)	Distance, n	n Speed, km/h	Angle of view, degree	Operation (30mins)	% on time	Time	Distance	Angle of view	Speed	Screening Effect, dB(A)	Facade	CNL, dB(A)	Total CNL, dB(A)	Criterion , dB(A)
	Concrete mixer 1(electric)	Stationary	CNP 045	96	0	Enclosure	-20	0.0	248.1	NA	NA	30	100.0%	0.0	-55.9	NA	NA	0	3	-52.9		
	Concrete mixer 2(electric)	Stationary	CNP 045	96	1	Enclosure	-20	76.0	241.4	NA	NA	30	100.0%	0.0	-55.6	NA	NA	0	3	23.4		
Filter	Fan (mixing unit) - blower (mixer 1)	Stationary	Other PME	95	0	Enclosure	-20	0.0	121.7	NA	NA	30	100.0%	0.0	-49.7	NA	NA	0	3	-46.7		
Filter	Fan (mixing unit) - blower (mixer 2)	Stationary	Other PME	95	1	Enclosure	-20	75.0	111.8	NA	NA	30	100.0%	0.0	-48.9	NA	NA	0	3	29.1		
	Truck (Concrete Collection)	Stationary	CNP044	109	0	sed Semi-enclosu	-20	0.0	248.1	NA	NA	2	6.7%	-11.8	-55.9	NA	NA	0	3	-64.7		
	Truck (Concrete Collection)	Stationary	CNP044	109	1	Enclosure	-20	89.0	241.4	NA	NA	3	10.0%	-10.0	-55.6	NA	NA	0	3	26.4		
	Conveyor / Screw Conveyors	Stationary	Other PME	90	2	Enclosure	-20	73.0	202.2	NA	NA	30	100.0%	0.0	-54.1	NA	NA	0	3	21.9		
	Conveyor / Screw Conveyors	Stationary	Other PME	90	1	Semi-enclosure	-15	75.0	182.5	NA	NA	30	100.0%	0.0	-53.2	NA	NA	0	3	24.8		
	Water / Admixture Pump	Stationary	CNP044	88	1	Semi-enclosure	-15	73.0	208.1	NA	NA	30	100.0%	0.0	-54.3	NA	NA	0	3	21.7		
	Pump (silo)	Stationary	CNP 281	88	2	Enclosure	-20	71.0	213.3	NA	NA	30	100.0%	0.0	-54.6	NA	NA	0	3	19.4		
Rd1	Concrete truck	Haul Road	CNP044	109	2	NA	NA	NA	238.7	15	2.9	NA	N/A	NA	-23.8	-18	-11.8	-10	3	18.4		
Rd2	Concrete truck	Haul Road	CNP044	109	2	NA	NA	NA	223.9	15	1.3	NA	N/A	NA	-23.5	-21.5	-11.8	-10	3	15.2	39	47
Rd3	Concrete truck	Haul Road	CNP044	109	2	NA	NA	NA	204.4	15	1.5	NA	N/A	NA	-23.1	-20.9	-11.8	0	3	26.2		
Rd4	Concrete truck	Haul Road	CNP044	109	2	NA	NA	NA	192.4	15	2.5	NA	N/A	NA	-22.8	-18.6	-11.8	0	3	28.8		
Rd5	Concrete truck	Haul Road	CNP044	109	2	NA	NA	NA	191.6	15	3.2	NA	N/A	NA	-22.8	-17.5	-11.8	0	3	29.9		
Rd6	Concrete truck	Haul Road	CNP044	109	2	NA	NA	NA	199.0	15	1.9	NA	N/A	NA	-23.0	-19.7	-11.8	0	3	27.5		
Rd7	Concrete truck	Haul Road	CNP044	109	2	NA	NA	NA	220.7	15	3.3	NA	N/A	NA	-23.4	-17.3	-11.8	0	3	29.5		
Rd8	Concrete truck	Haul Road	CNP044	109	2	NA	NA	NA	241.3	15	1.0	NA	N/A	NA	-23.8	-22.4	-11.8	-10	3	14.0		
Rd9	Concrete truck	Haul Road	CNP044	109	2	NA	NA	NA	249.3	15	1.7	NA	N/A	NA	-24.0	-20.1	-11.8	-10	3	16.1		
Rd10	Concrete truck	Haul Road	CNP044	109	2	NA	NA	NA	250.4	15	3.6	NA	N/A	NA	-24.0	-17	-11.8	-10	3	19.2		
Rd11	Concrete truck	Haul Road	CNP044	109	2	NA	NA	NA	244.8	15	2.1	NA	N/A	NA	-23.9	-19.4	-11.8	-10	3	16.9		1
Rd12	Concrete truck	Haul Road	CNP044	109	2	NA	NA	NA	245.9	15	3.6	NA	N/A	NA	-23.9	-17	-11.8	0	3	29.3	]	1
Rd13	Concrete truck	Haul Road	CNP044	109	2	NA	NA	NA	245.9	15	3.6	NA	N/A	NA	-23.9	-17	-11.8	0	3	29.3	1	

#### Predicted Noise Level at IN4

																Corr	ection					
	Fixed-noise Sources	Source Type	Ref. Code	SWL/unit, dB(A)	Quantity	At-source Noise Mitigation Measures	Noise Reduction from Mitigation Measures, dB(A)	Corrected SWL, dB(A)	Distance, m	Speed, km/h	Angle of view, degree	Operation (30mins)	% on time	Time	Distance	Angle of view	Speed	Screening Effect, dB(A) <sup>[1]</sup>	Facade	CNL, dB(A)	Total CNL, dB(A)	Criterion , dB(A)
	Concrete mixer 1(electric)	Stationary	CNP 045	96	0	Enclosure	-20	0.0	253.5	NA	NA	30	100.0%	0.0	-56.1	NA	NA	0	3	-53.1		
	Concrete mixer 2(electric)	Stationary	CNP 045	96	1	Enclosure	-20	76.0	259.9	NA	NA	30	100.0%	0.0	-56.3	NA	NA	0	3	22.7		
Filter	Fan (mixing unit) - blower (mixer 1)	Stationary	Other PME	95	0	Enclosure	-20	0.0	121.7	NA	NA	30	100.0%	0.0	-49.7	NA	NA	0	3	-46.7		
Filter	Fan (mixing unit) - blower (mixer 2)	Stationary	Other PME	95	1	Enclosure	-20	75.0	111.8	NA	NA	30	100.0%	0.0	-48.9	NA	NA	0	3	29.1		
	Truck (Concrete Collection)	Stationary	CNP044	109	0	sed Semi-enclosu	-20	0.0	253.5	NA	NA	2	6.7%	-11.8	-56.1	NA	NA	0	3	-64.9		
	Truck (Concrete Collection)	Stationary	CNP044	109	1	Enclosure	-20	89.0	259.9	NA	NA	3	10.0%	-10.0	-56.3	NA	NA	0	3	25.7		
	Conveyor / Screw Conveyors	Stationary	Other PME	90	2	Enclosure	-20	73.0	245.1	NA	NA	30	100.0%	0.0	-55.8	NA	NA	0	3	20.2		
	Conveyor / Screw Conveyors	Stationary	Other PME	90	1	Semi-enclosure	-15	75.0	272.1	NA	NA	30	100.0%	0.0	-56.7	NA	NA	0	4	22.3		
	Water / Admixture Pump	Stationary	Other PME	88	1	Semi-enclosure	-15	73.0	294.0	NA	NA	30	100.0%	0.0	-57.4	NA	NA	0	3	18.6		
	Pump (silo)	Stationary	CNP 281	88	2	Enclosure	-20	71.0	288.4	NA	NA	30	100.0%	0.0	-57.2	NA	NA	0	3	16.8		
Rd1	Concrete truck	Haul Road	CNP044	109	2	NA	NA	NA	263.9	15	2.9	NA	N/A	NA	-24.2	-17.9	-11.8	-10	3	18.1		
Rd2	Concrete truck	Haul Road	CNP044	109	2	NA	NA	NA	277.8	15	1.3	NA	N/A	NA	-24.4	-21.4	-11.8	-10	3	14.4	34	49
Rd3	Concrete truck	Haul Road	CNP044	109	2	NA	NA	NA	296.9	15	1.1	NA	N/A	NA	-24.7	-22.1	-11.8	-10	3	13.4	-	
Rd4	Concrete truck	Haul Road	CNP044	109	2	NA	NA	NA	309.0	15	1.5	NA	N/A	NA	-24.9	-20.7	-11.8	-10	3	14.6		
Rd5	Concrete truck	Haul Road	CNP044	109	2	NA	NA	NA	310.4	15	2.0	NA	N/A	NA	-24.9	-19.5	-11.8	-10	3	15.8		
Rd6	Concrete truck	Haul Road	CNP044	109	2	NA	NA	NA	304.4	15	1.6	NA	N/A	NA	-24.8	-20.4	-11.8	-10	3	15.0		
Rd7	Concrete truck	Haul Road	CNP044	109	2	NA	NA	NA	282.6	15	1.4	NA	N/A	NA	-24.5	-21.1	-11.8	-10	3	14.6		
Rd8	Concrete truck	Haul Road	CNP044	109	2	NA	NA	NA	260.9	15	0.7	NA	N/A	NA	-24.2	-24	-11.8	-10	3	12.0		
Rd9	Concrete truck	Haul Road	CNP044	109	2	NA	NA	NA	252.4	15	1.6	NA	N/A	NA	-24.0	-20.5	-11.8	-10	3	15.7		
Rd10	Concrete truck	Haul Road	CNP044	109	2	NA	NA	NA	250.9	15	3.5	NA	N/A	NA	-24.0	-17.1	-11.8	-10	3	19.1		
Rd11	Concrete truck	Haul Road	CNP044	109	2	NA	NA	NA	256.6	15	2.0	NA	N/A	NA	-24.1	-19.5	-11.8	-10	3	16.6		
Rd12	Concrete truck	Haul Road	CNP044	109	2	NA	NA	NA	256.6	15	3.3	NA	N/A	NA	-24.1	-17.4	-11.8	-10	3	18.7		1
Rd13	Concrete truck	Haul Road	CNP044	109	2	NA	NA	NA	256.5	15	3.3	NA	N/A	NA	-24.1	-17.4	-11.8	-10	3	18.7		1

Notes:

1 A 10dB(A) noise reduction was applied, as 1N4 is 1 storey building, and was blocked by the existing buildings located between the Site and the IN4, thus, no line of sight to the entire proposed development as well as the haul road.



### Evening – Scenario 2

#### Predicted Noise Level at IN1

					At course Noise	Noice Reduction									Corr	ection				Total		
Fixed-noise Sources		Source Type	Ref. Code	SWL/unit, dB(A)	Quantity	Mitigation Measures	from Mitigation Measures, dB(A)	Corrected SWL, dB(A)	Distance, m	Speed, km/h	Angle of view, degree	Operation (30mins)	% on time	Time	Distance	Angle of view	Speed	Screening Effect, dB(A) <sup>[1]</sup>	Facade	CNL, dB(A)	CNL, dB(A)	Criterion , dB(A)
	Conveyor / Screw Conveyors	Stationary	CNP 041	90	2	Enclosure	-20	73.0	103.4	NA	NA	30	100.0%	0.0	-48.3	NA	NA	0	3	27.7		
Conveyor / Screw Conveyors		Stationary	CNP 041	90	1	Semi-enclosure	-15	75.0	92.7	NA	NA	30	100.0%	0.0	-47.3	NA	NA	0	3	30.7		
	Water / Admixture Pump	Stationary	CNP 281	88	1	Semi-enclosure	-15	73.0	98.1	NA	NA	30	100.0%	0.0	-47.8	NA	NA	0	3	28.2		
	Pump (mounted on unloading tank)	Stationary	CNP 281	88	2	Enclosure	-20	71.0	120.8	NA	NA	30	100.0%	0.0	-49.6	NA	NA	0	3	24.4		
	Filter Fan (silo) - blower	Stationary	Other PME	95	2	Semi-enclosure	-15	83.0	120.8	NA	NA	30	100.0%	0.0	-49.6	NA	NA	0	3	36.4		
	Aggregates Truck	Haul Road	Other PME	105	5	NA	NA	NA	128.9	15	3.6	NA	N/A	NA	-21.1	-17.0	-11.8	-10	3	22.1		
Rd1	Cement / PFA / GGBS/ Admxture tanker	Haul Road	Other PME	105	1	NA	NA	NA	128.9	15	3.6	NA	N/A	NA	-21.1	-17	-11.8	-10	3	15.1		
	Lorry	Haul Road	Other PME	105	6	NA	NA	NA	128.9	15	3.6	NA	N/A	NA	-21.1	-17	-11.8	-10	3	22.9		
Pd2	Aggregates Truck	Haul Road	Other PME	105	5	NA	NA	NA	119.3	15	9.4	NA	N/A	NA	-20.8	-12.8	-11.8	-10	3	26.6		
Ruz	Cement / PFA / GGBS/ Admxture tanker	Haul Road	Other PME	105	1	NA	NA	NA	119.3	15	9.4	NA	N/A	NA	-20.8	-12.8	-11.8	-10	3	19.6	40	40
Rd3	Aggregates Truck	Haul Road	Other PME	105	5	NA	NA	NA	115.8	15	9.8	NA	N/A	NA	-20.6	-12.6	-11.8	-10	3	27.0		
nus	Lorry	Haul Road	Other PME	105	1	NA	NA	NA	115.8	15	9.8	NA	N/A	NA	-20.6	-12.6	-11.8	-10	3	20.0		
Rd4	Aggregates Truck	Haul Road	Other PME	105	2	NA	NA	NA	112.1	15	3.7	NA	N/A	NA	-20.5	-16.8	-11.8	-10	3	18.9		
	Lorry	Haul Road	Other PME	105	4	NA	NA	NA	112.1	15	3.7	NA	N/A	NA	-20.5	-16.8	-11.8	-10	3	21.9		
Rd5	Aggregates Truck	Haul Road	Other PME	105	2	NA	NA	NA	103.4	15	0.7	NA	N/A	NA	-20.1	-24.2	-11.8	-10	3	11.9		
nus	Lorry	Haul Road	Other PME	105	4	NA	NA	NA	103.4	15	0.7	NA	N/A	NA	-20.1	-24.2	-11.8	-10	3	14.9		
Rd6	Lorry	Haul Road	Other PME	105	6	NA	NA	NA	92.4	15	4.6	NA	N/A	NA	-19.7	-15.9	-11.8	-10	3	25.4		
Rd7	Lorry	Haul Road	Other PME	105	6	NA	NA	NA	88.7	15	21.7	NA	N/A	NA	-19.5	-9.2	-11.8	-10	3	32.3		
Rd14	Lorry	Haul Road	Other PME	105	6	NA	NA	NA	107.9	15	2.7	NA	N/A	NA	-20.3	-18.3	-11.8	-10	3	22.4		

Notes:

1 A 10dB(A) noise reduction was applied, as IN1 is 1 storey building, and was blocked by the existing solid wall of the Open Storage located between the Site and the IN1, thus, no line of sight to the entire proposed development as well as the haul road.

#### Predicted Noise Level at IN2

							Notes Badantian									Corr	ection				Tetal	
Fixed-noise Sou	Fixed-noise Sources	Source Type	Ref. Code	SWL/unit, dB(A)	Quantity	At-source Noise Mitigation Measures	from Mitigation Measures, dB(A)	Corrected SWL, dB(A)	Distance, m	Speed, km/h	Angle of view, degree	Operation (30mins)	% on time	Time	Distance	Angle of view	Speed	Screening Effect, dB(A)	Facade	CNL, dB(A)	CNL, dB(A)	Criterion , dB(A)
	Conveyor / Screw Conveyors	Stationary	CNP 041	90	2	Enclosure	-20	73.0	173.7	NA	NA	30	100.0%	0.0	-52.8	NA	NA	0	3	23.2		
	Conveyor / Screw Conveyors	Stationary	CNP 041	90	1	Semi-enclosure	-15	75.0	145.3	NA	NA	30	100.0%	0.0	-51.2	NA	NA	0	3	26.8		
	Water / Admixture Pump	Stationary	CNP 281	88	1	Semi-enclosure	-15	73.0	178.1	NA	NA	30	100.0%	0.0	-53.0	NA	NA	0	3	23.0		
	Pump (mounted on unloading tank)	Stationary	CNP 281	88	2	NA	-20	71.0	187.2	NA	NA	30	100.0%	0.0	-53.4	NA	NA	0	3	20.6		
	Filter Fan (silo) - blower	Stationary	Other PME	95	2	Semi-enclosure	-15	83.0	187.2	NA	NA	30	100.0%	0.0	-53.4	NA	NA	0	3	32.6		
	Aggregates Truck	Haul Road	Other PME	105	5	NA	NA	NA	213.4	15	2.8	NA	N/A	NA	-23.3	-18.1	-11.8	-10	3	18.8		
Rd1	Cement / PFA / GGBS/ Admxture tanker	Haul Road	Other PME	105	1	NA	NA	NA	213.4	15	2.8	NA	N/A	NA	-23.3	-18.1	-11.8	-10	3	11.8		
	Lorry	Haul Road	Other PME	105	6	NA	NA	NA	213.4	15	2.8	NA	N/A	NA	-23.3	-18.1	-11.8	-10	3	19.6		
Rd2	Aggregates Truck	Haul Road	Other PME	105	5	NA	NA	NA	197.4	15	0.5	NA	N/A	NA	-23	-25.5	-11.8	0	3	21.7		
Nuz	Cement / PFA / GGBS/ Admxture tanker	Haul Road	Other PME	105	1	NA	NA	NA	197.4	15	0.5	NA	N/A	NA	-23	-25.5	-11.8	0	3	14.7	40	47
Rd3	Aggregates Truck	Haul Road	Other PME	105	5	NA	NA	NA	177.4	15	0.6	NA	N/A	NA	-22.5	-25	-11.8	0	3	22.7		
Rus	Lorry	Haul Road	Other PME	105	1	NA	NA	NA	177.4	15	0.6	NA	N/A	NA	-22.5	-25	-11.8	0	3	15.7		
Rd4	Aggregates Truck	Haul Road	Other PME	105	2	NA	NA	NA	164.3	15	2.6	NA	N/A	NA	-22.2	-18.5	-11.8	0	3	25.5		
Nu4	Lorry	Haul Road	Other PME	105	4	NA	NA	NA	164.3	15	2.6	NA	N/A	NA	-22.2	-18.5	-11.8	0	3	28.5		
Rd5	Aggregates Truck	Haul Road	Other PME	105	2	NA	NA	NA	161.9	15	4.0	NA	N/A	NA	-22.1	-16.6	-11.8	0	3	27.5		
Rus	Lorry	Haul Road	Other PME	105	4	NA	NA	NA	161.9	15	4.0	NA	N/A	NA	-22.1	-16.6	-11.8	0	3	30.5		
Rd6	Lorry	Haul Road	Other PME	105	6	NA	NA	NA	167.7	15	2.9	NA	N/A	NA	-22.2	-18	-11.8	-10	3	20.8		
Rd7	Lorry	Haul Road	Other PME	105	6	NA	NA	NA	189.5	15	2.4	NA	N/A	NA	-22.8	-18.7	-11.8	-10	3	19.5		
Rd14	Lorry	Haul Road	Other PME	105	6	NA	NA	NA	206.4	15	8.0	NA	N/A	NA	-23.1	-13.5	-11.8	0	3	34.4		

### J24.00171.HK.01 S.16 Planning Application for Proposed Temporary Concrete Batching Plant with Ancillary Facilities for a Period of 5 Years at Lots 573 RP and 1710 in D.D. 114, Shek Kong, Yuen Long Environmental Assessment



#### Predicted Noise Level at IN3

					At-source Noise Reduction			j	Total													
Fixed-noise Sources		Source Type	Ref. Code	SWL/unit, dB(A)	Quantity	Mitigation Measures	from Mitigation Measures, dB(A)	SWL, dB(A)	Distance, m	Speed, km/h	Angle of view, degree	Operation (30mins)	% on time	Time	Distance	Angle of view	Speed	Screening Effect, dB(A)	Facade	CNL, dB(A)	CNL, dB(A)	Criterion , dB(A)
	Conveyor / Screw Conveyors	Stationary	CNP 041	90	2	Enclosure	-20	73.0	202.2	NA	NA	30	100.0%	0.0	-54.1	NA	NA	0	3	21.9		
	Conveyor / Screw Conveyors	Stationary	CNP 041	90	1	Semi-enclosure	-15	75.0	182.5	NA	NA	30	100.0%	0.0	-53.2	NA	NA	0	3	24.8		
	Water / Admixture Pump	Stationary	CNP 281	88	1	Semi-enclosure	-15	73.0	208.1	NA	NA	30	100.0%	0.0	-54.3	NA	NA	0	3	21.7		
F	Pump (mounted on unloading tank)	Stationary	CNP 281	88	2	NA	-20	71.0	213.3	NA	NA	30	100.0%	0.0	-54.6	NA	NA	0	3	19.4	l	
	Filter Fan (silo) - blower	Stationary	Other PME	95	2	Semi-enclosure	-15	83.0	213.3	NA	NA	30	100.0%	0.0	-54.6	NA	NA	0	3	31.4		
	Aggregates Truck	Haul Road	Other PME	105	5	NA	NA	NA	238.7	15	2.9	NA	N/A	NA	-23.8	-18.0	-11.8	-10	3	18.4		
Rd1	Cement / PFA / GGBS/ Admxture tanker	Haul Road	Other PME	105	1	NA	NA	NA	238.7	15	2.9	NA	N/A	NA	-23.8	-18	-11.8	-10	3	11.4		
	Lorry	Haul Road	Other PME	105	6	NA	NA	NA	238.7	15	2.9	NA	N/A	NA	-23.8	-18	-11.8	- 10	3	19.2		
Ch C	Aggregates Truck	Haul Road	Other PME	105	5	NA	NA	NA	223.9	15	1.3	NA	N/A	NA	-23.5	-21.5	-11.8	-10	3	15.2	1	
Ruz	Cement / PFA / GGBS/ Admxture tanker	Haul Road	Other PME	105	1	NA	NA	NA	223.9	15	1.3	NA	N/A	NA	-23.5	-21.5	-11.8	-10	3	8.2	39	47
chia	Aggregates Truck	Haul Road	Other PME	105	5	NA	NA	NA	204.4	15	1.5	NA	N/A	NA	-23.1	-20.9	-11.8	0	3	26.2	1	
Rus	Lorry	Haul Road	Other PME	105	1	NA	NA	NA	204.4	15	1.5	NA	N/A	NA	-23.1	-20.9	-11.8	0	3	19.2		
Rd4	Aggregates Truck	Haul Road	Other PME	105	2	NA	NA	NA	192.4	15	2.5	NA	N/A	NA	-22.8	-18.6	-11.8	0	3	24.8		
Ku4	Lorry	Haul Road	Other PME	105	4	NA	NA	NA	192.4	15	2.5	NA	N/A	NA	-22.8	-18.6	-11.8	0	3	27.8		
DdE	Aggregates Truck	Haul Road	Other PME	105	2	NA	NA	NA	191.6	15	3.2	NA	N/A	NA	-22.8	-17.5	-11.8	0	3	25.9		
Rus	Lorry	Haul Road	Other PME	105	4	NA	NA	NA	191.6	15	3.2	NA	N/A	NA	-22.8	-17.5	-11.8	0	3	28.9		
Rd6	Lorry	Haul Road	Other PME	105	6	NA	NA	NA	199.0	15	1.9	NA	N/A	NA	-23	-19.7	-11.8	0	3	28.3		
Rd7	Lorry	Haul Road	Other PME	105	6	NA	NA	NA	220.7	15	3.3	NA	N/A	NA	-23.4	-17.3	-11.8	0	3	30.3		
Rd14	Lorry	Haul Road	Other PME	105	6	NA	NA	NA	234.8	15	7.0	NA	N/A	NA	-23.7	-14.1	-11.8	0	3	33.2	1	

#### Predicted Noise Level at IN4

				At-source Noise Reduction								Correction										
	Fixed-noise Sources		Ref. Code	SWL/unit, dB(A)	Quantity	At-source Noise Mitigation Measures	Noise Reduction from Mitigation Measures, dB(A)	Corrected SWL, dB(A)	Distance, m	Speed, km/h	Angle of view, degree	Operation (30mins)	% on time	Time	Distance	Angle of view	Speed	Screening Effect, dB(A) <sup>[1]</sup>	Facade	CNL, dB(A)	Total CNL, dB(A)	Criterion , dB(A)
	Conveyor / Screw Conveyors	Stationary	CNP 041	90	2	Enclosure	-20	73.0	245.1	NA	NA	30	100.0%	0.0	-55.8	NA	NA	0	3	20.2		
	Conveyor / Screw Conveyors	Stationary	CNP 041	90	1	Semi-enclosure	-15	75.0	272.1	NA	NA	30	100.0%	0.0	-56.7	NA	NA	0	4	22.3		
	Water / Admixture Pump	Stationary	CNP 281	88	1	Semi-enclosure	-15	73.0	294.0	NA	NA	30	100.0%	0.0	-57.4	NA	NA	0	3	18.6		
P	ump (mounted on unloading tank)	Stationary	CNP 281	88	2	NA	-20	71.0	288.4	NA	NA	30	100.0%	0.0	-57.2	NA	NA	0	3	16.8		
	Filter Fan (silo) - blower	Stationary	CNP044	109	2	Semi-enclosure	-15	97.0	288.4	NA	NA	5	16.7%	-7.8	-57.2	NA	NA	0	3	35.0		
	Aggregates Truck	Haul Road	Other PME	105	5	NA	NA	NA	263.9	15	2.9	NA	N/A	NA	-24.2	-17.9	-11.8	-10	3	18.1		
Rd1	Cement / PFA / GGBS/ Admxture tanker	Haul Road	Other PME	105	1	NA	NA	NA	263.9	15	2.9	NA	N/A	NA	-24.2	-17.9	-11.8	-10	3	11.1		
	Lorry	Haul Road	Other PME	105	6	NA	NA	NA	263.9	15	2.9	NA	N/A	NA	-24.2	-17.9	-11.8	-10	3	18.9		
Rd2	Aggregates Truck	Haul Road	Other PME	105	5	NA	NA	NA	277.8	15	1.3	NA	N/A	NA	-24.4	-21.4	-11.8	-10	3	14.4		
nuz	Cement / PFA / GGBS/ Admxture tanker	Haul Road	Other PME	105	1	NA	NA	NA	277.8	15	1.3	NA	N/A	NA	-24.4	-21.4	-11.8	-10	3	7.4	36	49
Rd3	Aggregates Truck	Haul Road	Other PME	105	5	NA	NA	NA	296.9	15	1.1	NA	N/A	NA	-24.7	-22.1	-11.8	-10	3	13.4		
nus	Lorry	Haul Road	Other PME	105	1	NA	NA	NA	296.9	15	1.1	NA	N/A	NA	-24.7	-22.1	-11.8	-10	3	6.4		
Rd4	Aggregates Truck	Haul Road	Other PME	105	2	NA	NA	NA	309.0	15	1.5	NA	N/A	NA	-24.9	-20.7	-11.8	-10	3	10.6		
	Lorry	Haul Road	Other PME	105	4	NA	NA	NA	309.0	15	1.5	NA	N/A	NA	-24.9	-20.7	-11.8	-10	3	13.6		
Rd5	Aggregates Truck	Haul Road	Other PME	105	2	NA	NA	NA	310.4	15	2.0	NA	N/A	NA	-24.9	-19.5	-11.8	-10	3	11.8		
	Lorry	Haul Road	Other PME	105	4	NA	NA	NA	310.4	15	2.0	NA	N/A	NA	-24.9	-19.5	-11.8	-10	3	14.8		
Rd6	Lorry	Haul Road	Other PME	105	6	NA	NA	NA	304.4	15	1.6	NA	N/A	NA	-24.8	-20.4	-11.8	-10	3	15.8		
Rd7	Lorry	Haul Road	Other PME	105	6	NA	NA	NA	282.6	15	1.4	NA	N/A	NA	-24.5	-21.1	-11.8	-10	3	15.4		
Rd14	Lorry	Haul Road	Other PME	105	6	NA	NA	NA	266.5	15	6.2	NA	N/A	NA	-24.3	-14.6	-11.8	-10	3	22.1		

Notes:

1 A 10dB(A) noise reduction was applied, as IN4 is 1 storey building, and was blocked by the existing buildings located between the Site and the IN4, thus, no line of sight to the entire proposed development as well as the haul road.



# Appendix H Traffic Forecasts for Year 2031

### J24.00171.HK.01 S.16 Planning Application for Proposed Temporary Concrete Batching Plant with Ancillary Facilities for a Period of 5 Years at Lots 573 RP and 1710 in D.D. 114, Shek Kong, Yuen Long Environmental Assessment



Link ID	Pood Soction			With th	ne Site			Without the Site								
	Rodu Section	AM Peak I	Hour		PM Peak H	lour		AM Peak I	Hour		PM Peak Hour					
		Traffic Flows (veh/hr)	%LV	%HV	Traffic Flows (veh/hr)	%LV	%HV	Traffic Flows (veh/hr)	%LV	%HV	Traffic Flows (veh/hr)	%LV	%HV			
RD_001	Chun Yiu Road	10	50.0%	50.0%	30	40.0%	60.0%	10	50.0%	50.0%	30	40.0%	60.0%			
RD_002	Chun Yiu Road	15	69.2%	30.8%	25	44.0%	56.0%	15	69.2%	30.8%	25	44.0%	56.0%			
RD_003	Chun Yiu Road	25	68.2%	31.8%	25	54.5%	45.5%	25	68.2%	31.8%	25	54.5%	45.5%			
RD_004	Chun Yiu Road	10	100.0%	0.0%	35	37.4%	62.6%	10	100.0%	0.0%	35	37.4%	62.6%			
RD_005	Chun Yiu Road	45	34.1%	65.9%	60	20.0%	80.0%	25	68.2%	31.8%	25	54.5%	45.5%			
RD_006	Chun Yiu Road	30	21.4%	78.6%	70	17.1%	82.9%	10	100.0%	0.0%	35	37.4%	62.6%			
RD_007	Chun Yiu Road	60	41.4%	58.6%	85	26.2%	73.8%	40	66.7%	33.3%	50	47.9%	52.1%			
RD_008	Chun Yiu Road	40	30.8%	69.2%	100	22.2%	77.8%	20	70.5%	29.5%	65	36.0%	64.0%			
RD_009	Kam Tin Road	485	67.4%	32.6%	690	51.0%	49.0%	460	70.4%	29.6%	655	53.9%	46.1%			
RD_010	Kam Tin Road	500	69.1%	30.9%	635	46.4%	53.6%	480	72.4%	27.6%	595	49.3%	50.7%			
RD_011	Kam Tin Road	470	66.9%	33.1%	510	66.9%	33.1%	435	72.8%	27.2%	460	75.0%	25.0%			
RD_012	Kam Tin Road	505	68.6%	31.4%	445	66.3%	33.7%	470	74.2%	25.8%	390	75.5%	24.5%			

#### J24.00171.HK.01

S.16 Planning Application for Proposed Temporary Concrete Batching Plant with Ancillary Facilities for a Period of 5 Years at Lots 573 RP and 1710 in D.D. 114, Shek Kong, Yuen Long Environmental Assessment







# **Appendix I**

# **Result of Traffic Noise Impact** Assessment



# Scenario: Road Traffic Noise - "Before" Scenario

		Existin	g NSRs	
Floor	IN1	IN2	IN3	IN4
G/F	67.8	54.1	52.6	68.7
1/F	-	54.2	-	-
Criterion	70	70	70	70
Max	68	54	53	69
Compliance	Y	Y	Y	Y

# Scenario: Road Traffic Noise - "After" Scenario

		Existin	g NSRs	
Floor	IN1	IN2	IN3	IN4
G/F	67.9	54.4	53.2	69.5
1/F	-	54.5	-	-
Criterion	70	70	70	70
Max	68	55	53	70
Compliance	Ŷ	Y	Y	Y

# Appendix J Calculation of Sewage Generation

### J24.00171.HK.01

S.16 Planning Application for Proposed Temporary Concrete Batching Plant with Ancillary Facilities for a Period of 5 Years at Lots 573 RP and 1710 in D.D. 114, Shek Kong, Yuen Long Environmental Assessment



Ca	culation of Sewage Generation from the Proposed Developm	nent				Remarks/ Justification
1)	Sewage generated by Staff					
	No of onsite staff	=	12	staff		As advised by the Applicant
	Unit Flow Factor (UFF) per staff	=	0.23	m <sup>3</sup> /day-staff		Refer to "Commercial Employee" and "J9 Construction" of Table T-2 of reference 1 [Note 1].
	Total Sewage Generation	=	2,760.0	L/day	(a)	
2)	Sewage generated by Truck Drivers					
	Total No of toilet visits per day	=	100.0	toilet visit/day		As advised by the Applicant, no more than 100 truck drivers (i.e. non-site staff) use the toilet on- site each day
	Flow rate per flushing	=	7.5	L/flush	(b)	Refer to BEAM Plus New Buildings Version 1.2), the estimated toilet flush of 7.5 L/flush. [Note 3]
	Flow rate per hand washing	=	1.4	L/per hand washing	(c)	Refer to BEAM Plus New Buildings Version 1.2, Wastewater used for handwashing (8.3L/min x
	Flow rate from micturition per visit	=	0.2	L/per visit	(d)	Refer to Human's micturition is assumed to be 200mL in accordance with p. 3081 of "Magill's
	Unit flow rate (for truck driver)	=	9.1	L/visit	(e)	Medical Guide". 6th ed various medical editors. Salem Press. USA. 2011. [ <sup>NOIE 4]</sup> (b) + (c) + (d)
	Total Sewage Generation	=	908.3	L/day	(f)	
	Total Average Daily Dry Weather Flow	=	3,668	L/day		(a) + (f)
		=	3.67	m³/day		
	Containter Toilets					
	The Minimum Total Storage Capacity of Sewage Storage Tank	=	3	m <sup>3</sup> / tank		As advised by the Applicant
	No. of containers required (3,000L each)	=	2	tanks		

Notes:

1 Guidelines for Estimating Sewage Flows for Sewage Infrastructure Planning Version 1.0, Environmental Protection Department of HK Government, March 2005

2 The volume of flushing system is advised by the supplier of chemical/container toilet.

3 With reference to BEAM Plus New Buildings Version 1.2 in July 2012, it is assumed that water required for hand washing = 8.3L/mins x 10s. And the estimated toilet flush volume of toilet is about 7.5 L/fush.

4 Human's micturition is assumed to be 200mL in accordance with p. 3081 of "Magill's Medical Guide", 6th ed., various medical editors, Salem Press, USA, 2011.





# Appendix K Historical Aerial Photographs



### Figure J-1: Historical Aerial Photos in Year 1963\_Photo No.: 8394





Figure J-2: Historical Aerial Photos in Year 1973\_Photo ID: 07328





Figure J-3: Historical Aerial Photos in Year 1982\_Photo ID: 42976





Figure J-4: Historical Aerial Photos in Year 1993\_Photo ID: CN05931





*Figure J-5: Historical Aerial Photos in Year 2003\_Photo ID: CW52218* 





Figure J-6: Historical Aerial Photos in Year 2013\_Photo ID: S43741




Figure J-7: Historical Aerial Photos in Year 2023\_Photo ID: E188907C





### Appendix L

# Site Walkover Checklist and Site Photos





#### **GENERAL SITE DETAILS**

SITE OWNER	/CLIENT Join Bright Warehousing Limited
PROPERTY A	DDRESS
PERSON COM	IDUCTING THE QUESTIONNAIRE
NAME	
POSITION	
AUTHORIZED	OWNER/CLIENT REPRESENTATIVE (IF APPLICABLE)
NAME	N/A
POSITION	N/A
TELEPHONE	N/A
SITE ACTIV	<b>TITIES</b>
<del>-Briefly descri</del> Obtain a flo	be activities carried out on site, including types of products/chemicals/materials handled, w schematic if possible.
Number of e	mployees: Full-time:
	Part-time:
	Temporary/Seasonal:
Maximum no	. of people on site at any time: N/Δ
Typical hours	s of operation:
Number of sl	nifts:
Days per wee	ek:
Weeks per ye	ear:



#### Detail the main sources of energy at the site:

Gas	<del>-Yes</del> /No
Electricity	Yes/ <del>No</del>
Coal	<del>Yes</del> /No
Oil	<del>Yes</del> /No
Other	- <del>Yes</del> /No

#### SITE DESCRIPTION

This section is intended to gather information on site setting and environmental receptors on, adjacent or close to the site.

What is the total site area:	<u>4,411 m<sup>2</sup></u>	
What area of the site is covered by buildings (%):	0%	
Please list all current and previous owners/occupiers if possible.	N/A	

Is a site plan available? If yes, please attach. <del>Yes</del>/No

Are there any other parties on site as tenants or sub-tenants? Yes/No

If yes, identify those parties:

Describe surrounding land use (residential, industrial, rural, etc.) and identify neighbouring facilities and types of industry.

North: Temporary structures and access road

South: Temporary structures and stream

East: Temporary structures, vegetation and access road

West: Temporary structures and access road

Practice Guide for Investigation and Remediation of Contaminated Land



### Annex C1 Site Walkover Checklist

Describe the topography of the area (flat terrain, rolling hills, mountains, by a large body of water, vegetation, etc.).

The site is a flat area paved with concrete.

State the size and location of the nearest residential communities.

The nearest residential use is a village house located 140m away from the Site.

Are there any sensitive habitats nearby, such as nature reserves, parks, wetlands or sites of special scientific interest?

The Site is surrounded by temporary structures with non-sensitive use. No sensitive use is identified in

the vicinity.

#### Questionnaire with Existing/Previous Site Owner or Occupier

		Yes/No	Notes
1.	What are the main activities/operations at the above address?		Open storage for construction materials and plants
2.	How long have you been occupying the site?		More than 20 years
3.	Were you the first occupant on site? (If yes, what was the usage of the site prior to occupancy.)		More than 20 years
4.	Prior to your occupancy, who occupied the site?		N/A
5.	What were the main activities/operations during their occupancy?		N/A
6.	Have there been any major changes in operations carried out at the site in the last 10 years?	Yes	The site was used as open storage for abandoned vehicles and vehicle parts
7.	Have any polluting activities been carried out in the vicinity of the site in the past?	No	
8.	To the best of your knowledge, has the site ever been used as a petrol filling station/car service garage?	No	
9.	Are there any boreholes/wells or natural springs either on the site or in the surrounding area?	No	
10.	Do you have any registered hazardous installations as defined under relevant ordinances? (If yes, please provide details.)	No	
11.	Are any chemicals used in your daily operations? (If yes, please provide details.)	Yes	
	Where do you store these chemicals?		Storage in the oil drums
12.	Material inventory lists, including quantities and locations available? (If yes, how often are these inventories updated?)	N/A	
13.	Has the facility produced a separate hazardous substance inventory?	N/A	
14.	Have there ever been any incidents or accidents (e.g. spills, fires, injuries, etc.) involving any of these materials? (If yes, please provide details.)	No	





		Yes/No	Notes
15.	How are materials received (e.g. rail, truck, etc.) and stored on site (e.g. drums, tanks, carboys, bags, silos, cisterns, vaults and cylinders)?	N/A	
16.	Do you have any underground storage tanks? (If yes, please provide details.)	No	
	<ul> <li>How many underground storage tanks do you have on site?</li> </ul>	N/A	
	<ul> <li>What are the tanks constructed of?</li> </ul>	N/A	
	What are the contents of these tanks?	N/A	
	<ul> <li>Are the pipelines above or below ground?</li> </ul>	N/A	
	• If the pipelines are below ground, has any leak and integrity testing been performed?	N/A	
	<ul> <li>Have there been any spills associated with these tanks?</li> </ul>	N/A	
17.	Are there any disused underground storage tanks?	No	
18.	Do you have regular check for any spillage and monitoring of chemicals handled? (If yes, please provide details.)	N/A	
19.	How are the wastes disposed of?		General refuse will be collected an disposed of regularly
20.	Have you ever received any notices of violation of environmental regulations or received public complaints? (If yes, please provide details.)	No	
21.	Have any spills occurred on site? (If yes, please provide details.)	No	
	When did the spill occur?	N/A	
	What were the substances spilled?	N/A	
	<ul> <li>What was the quantity of material spilled?</li> </ul>	N/A	
	<ul> <li>Did you notify the relevant departments of the spill?</li> </ul>	N/A	
	<ul> <li>What were the actions taken to clean up the spill?</li> </ul>	N/A	
	What were the areas affected?	N/A	
22.	Do you have any records of major renovation of your site or re- arrangement of underground utilities, pipe work/underground tanks (If yes, please provide details.)	No	
23.	Have disused underground tanks been removed or otherwise secured (e.g. concrete, sand, etc.)?	No	
24.	Are there any known contaminations on site? (If yes, please provide details.)	No	
25.	Has the site ever been remediated? (If yes, please provide details.)	No	

Practice Guide for Investigation and Remediation of Contaminated Land



### Annex C1 Site Walkover Checklist

#### Observations

		Yes/No	Notes
1.	Are chemical storage areas provided with secondary containment (i.e. bund walls and floors)?	Yes	
2.	What are the conditions of the bund walls and floors?		The site is paved with concrete and in good conc
3.	Are any surface water drains located near to drum storage and unloading areas?	No	
1.	Are any solid or liquid waste (other than wastewater) generated at the site? (If yes, please provide details.)		General refuse will be generated by on-site staff.
5.	Is there a storage site for the wastes?	No	
5.	Is there an on-site landfill?	No	
7.	Were any stressed vegetation noted on site during the site reconnaissance? (If yes, please indicate location and approximate size.)	No	
3.	Were any stained surfaces noted on-site during the site reconnaissance? (If yes, please provide details.)	No	
).	Are there any potential off-site sources of contamination?	No	
10.	Does the site have any equipment which might contain polychlorinated biphenyls (PCBs)?	No	
11.	Are there any sumps, effluent pits, interceptors or lagoons on site?	No	
12.	Any noticeable odours during site walkover?	No	
13.	Are any of the following chemicals used on site: fuels, lubricating oils, hydraulic fluids, cleaning solvents, used chemical solutions, acids, anti-corrosive paints, thinners, coal, ash, oily tanks and bilge sludge, metal wastes, wood preservatives and polyurethane foam?	No	

45

#### J24.00171.HK.01 S.16 Planning Application for Proposed Temporary Concrete Batching Plant with Ancillary Facilities for a Period of 5 Years at Lots 573 RP and 1710 in D.D. 114, Shek Kong, Yuen Long Environmental Assessment



### Figure K-1 Site Photos





## Appendix M Correspondence with EPD and FSD



### Enquiry email to EPD dated 04/07/2024

ESC

Our Ref. J24.00171.HK.01/ L00113/AW/JC

4 July 2024

Yuen Long - Regional Office (North) Environmental Compliance Division Environmental Protection Department 10th floor, Shatin Government Offices No.1 Sheung Wo Che Road, Sha Tin N.T., Hong Kong

#### **EnviroSolutions & Consulting Ltd**

Solutions for Environment | Safety | Sustainability

- 16/F & 17/F, 700 Nathan Road, Mong Kok, Kowloon, Hong Kong
- +852 3960 7000
- +852 3960 7111
   www.envirosc.co
- www.envirosc.com | www.simplyehs.com
   enquiries@envirosc.com

By Email (wsfung@epd.gov.hk)

Attention: Mr. FUNG Wai Shun

Dear Sir

Section 16 Planning Application for Proposed Temporary Concrete Batching Plant with Ancillary Facilities for a Period of 5 Years at Lots 573 RP and 1710 in D.D. 114, Shek Kong, New Territories Environmental Assessment - Information Request to EPD

Join Bright Warehousing Limited proposes to relocate the existing "Concrete Batching Plant" which operated by Glorious Concrete (H.K.) Ltd. From Lot 1824 RP in D.D. 130 in Hung Shui Kiu to Lot 573 RP and 1710 in D.D. 114 in Shek Kong (the Site) due to the land resumption to facilitate the development of Hung Shui Kiu/ Ha Tsuen New Development Area.

The Site currently falls within an area zoned as "Industrial (Group D)" ("I(D)") on the Approved Shek Kong Outline Zoning Plan (OZP) No. S/YL-SK/9. According to the Notes of the OZP, the proposed use is a Column 2 use within "i(D)" zone, which requires planning permission from the Town Planning Board under Section 16 of the Town Planning Ordinance. We have been appointed by Join Bright Warehousing Limited as the Environmental Consultant to prepare an Environmental Assessment (EA) report for the captioned project. The location of the Site is shown in **Figure 1**. The subcontract for the consultancy service is also enclosed for your information.

To address the potential land contamination issue, we would appreciate if you could provide us with a list of records of Chemical Waste Producers Registration or incidents of chemical spillage/leakage, etc, related to the site if any, including the company name, type of chemical, location etc.



#### J24.00171.HK.01

S.16 Planning Application for Proposed Temporary Concrete Batching Plant with Ancillary Facilities for a Period of 5 Years at Lots 573 RP and 1710 in D.D. 114, Shek Kong, Yuen Long Environmental Assessment





ESC Project No. J24.00171.HK.01 | D01 | Rev 0

Page 2

Issued on 28/6/2024

ESC Project No. J24.00171.HK.01 | D01 | Rev 1



Section 16 Planning Application for Proposed Temporary Concrete Batching Plant with Ancillary Facilities for a Period of 5 Years at Lot 573 RP in D.D. 114, Kam Tin Environmental Assessment Report





ESC Project No. J24.00171.HK.01 | D01 | Rev 0

Page 3

Issued on 28/6/2024



EPD's reply is pending



### Enquiry email to FSD dated 04/07/2024

FSC

Our Ref. J24.00171.HK.01/ L00114/AW/JC

4 July 2024

**Corporate Services Division Corporate Strategy Command** Fire Services Department 9/F, Fire Services Headquarters Building 1 Hong Chong Road, Tsim Sha Tsui East Kowloon, Hong Kong

#### **EnviroSolutions & Consulting Ltd**

Solutions for Environment | Safety | Sustainability 0 16/F & 17/F, 700 Nathan Road, Mong Kok, Kowloon,

- Hong Kong
- ٩ +852 3960 7000
- +852 3960 7111
- 0 www.envirosc.com | www.simplyehs.com ۲ enquiries@envirosc.com

By Email (ado\_lad@hkfsd.gov.hk)

Attention: Mr. LAI Kin Man

Dear Sir

Section 16 Planning Application for Proposed Temporary Concrete Batching Plant with Ancillary Facilities for a Period of 5 Years at Lot 573 RP and 1710 in D.D. 114, Shek Kong, New Territories **Environmental Assessment - Information Request to FSD** 

Join Bright Warehousing Limited proposes to relocate the existing "Concrete Batching Plant" which operated by Glorious Concrete (H.K.) Ltd. From Lot 1824 RP in D.D. 130 in Hung Shui Kiu to Lot 573 RP and 1710 in D.D. 114 in Shek Kong (the Site) due to the land resumption to facilitate the development of Hung Shui Kiu/ Ha Tsuen New Development Area.

The Site currently falls within an area zoned as "Industrial (Group D)" ("I(D)") on the Approved Shek Kong Outline Zoning Plan (OZP) No. S/YL-SK/9. According to the Notes of the OZP, the proposed use is a Column 2 use within "i(D)" zone, which requires planning permission from the Town Planning Board under Section 16 of the Town Planning Ordinance. We have been appointed by Join Bright Warehousing Limited as the Environmental Consultant to prepare an Environmental Assessment (EA) report for the captioned project. The location of the Site is shown in Figure 1. The subcontract for the consultancy service is also enclosed for your information.

To address the potential land contamination issue, we would appreciate if you could provide us with a list of records of dangerous goods license, fire incidents or incidents of dangerous goods spillage/leakage, etc, related to the site if any, including the company name, type of chemical, location etc.

Yours faithfully for and on behalf of ESC



#### J24.00171.HK.01

S.16 Planning Application for Proposed Temporary Concrete Batching Plant with Ancillary Facilities for a Period of 5 Years at Lots 573 RP and 1710 in D.D. 114, Shek Kong, Yuen Long Environmental Assessment





ESC Project No. J24.00171.HK.01 | D01 | Rev 0

Page 2

Issued on 28/6/2024

ESC Project No. J24.00171.HK.01 | D01 | Rev 1



Section 16 Planning Application for Proposed Temporary Concrete Batching Plant with Ancillary Facilities for a Period of 5 Years at Lot 573 RP in D.D. 114, Kam Tin Environmental Assessment Report



- ----

-		ESC	EnviroSolutions & Consulting Ltd Solutions for Environment   Safety   Sustainability © 16/# & 17/#, 700 Nathan Road, Mong Kok, Kowkoon, Hong Kong * +852 3960 7000 * +852 3960 7111 © www.envirosc.com   www.simplyehs.com © enquiries@envirosc.com
an ann an		Our Ref. P24.00171.HK.01/L00095/AB/AW Your Ref. 2 May 2024 Join Bright Warebousing Limited	
- 1000 - 1000		Joll Dright wateriousing chingen	By Email (paul@manfungItd.com)
		Dear Sir Section 16 Planning Application for Proposed Tempora with Ancillary Facilities for a Period of 5 Years at Lot 57 Environmental Assessment and Drainage Proposal Proposal for Services	ry Concrete Batching Plant 3 RP in D.D. 114, Kam Tin
		Thank you for your invitation to submit a proposal for th EnviroSolutions & Consulting Ltd ("ESC") is pleased to su Proposal to prepare Environmental Assessment ("EA") a project.	e captioned services. Ibmit herewith this Technical and Fee nd Drainage Proposal for the captioned
		We trust you find this proposal acceptable and are then proposal back to the undersigned at your earliest conve to receive your Works Order/Purchase Order for the Ser	efore able to send a signed copy of this nience. Alternatively, we would be pleased vices, referencing this Proposal.
	-	Yours Faithfully for and on behalf of ESC	Accepted for and on behalf of loin Bright Warehousing Limited
		Enci.	Date: - 3 MAY 2024
ı			

ESC Project No. J24.00171.HK.01 | D01 | Rev 0

Page 3

Issued on 28/6/2024



### FSD's reply dated 30/07/2024

消防處 香港九龍尖沙咀東部康	ŧ莊	ii 1 th	FIRE SERVICES DEPARTMENT FIRE SERVICES HEADQUARTERS BUILDING, No.1 Hong Chong Road,
<b>冲的</b> 處認即入	厦	PICK ONC	Tsim Sha Tsui East, Kowloon, Hong Kong.
本處檔號 OUR REF.	:	(79) in FSD GR 6-5/4 R Pt. 54	
來函檔號 YOUR REF.	:	J24.00171.HK.01/L00114/AW/JC	
電子郵件 E-mail	÷	hkfsdenq@hkfsd.gov.hk	
圖文傳真 FAX NO.	:	2988 1196	
電話 TEL NO.	:	2733 7570	
			30 July 2024

EnviroSolutions & Consulting Ltd

Dear Mr. WONG,

#### Section 16 Planning Application for Proposed Temporary Concrete Batching Plant with Ancillary Facilities for a Period of 5 years at Lot 573 RP and 1710 in D.D. 114, Shek Kong, New Territories Environmental Assessment – Information Request to FSD Request for Information of Dangerous Goods & Incident Records Records

I refer to your letter of 4.7.2024 regarding the captioned request and reply below in response to your questions:-

Please be advised that neither records of dangerous goods license, fire incidents nor incidents of spillage / leakage of dangerous goods were found in connection with the given conditions of your request at the subject location.

If you have further questions, please feel free to contact the undersigned.

Yours sincerely,

(LAI Kin-man) for Director of Fire Service

Ref. number and date should be quoted in reference to this letter 凡提及本信時請引述編號及日期



#### **EnviroSolutions & Consulting Ltd**

Solutions for Environment | Safety | Sustainability

### **EnviroSolutions & Consulting Ltd**

### 000

### Accountability

We understand the importance of being accountable to each other and our clients.



#### Passion

We are completely passionate about providing practical solutions and outcomes that deliver for our clients.



#### Insight

We work in an environment that encourages and values insight as a critical quality which informs our decisions and our clients and supports practical solutions and project delivery.



#### Integrity

We behave with respect and honesty toward each other, our clients and our stakeholders. Appendix V

Drainage Proposal







### J24.00171.HK.01

S.16 Planning Application for Proposed Temporary Concrete Batching Plant with Ancillary Facilities for a Period of 5 Years at Lots 573 RP and 1710 in D.D. 114, Shek Kong, Yuen Long, New Territories Drainage Proposal

Prepared for: Join Bright Warehousing Limited

**30 December 2024** 

### S.16 Planning Application for Proposed Temporary Concrete Batching Plant with Ancillary Facilities for a Period of 5 Years at Lots 573 RP and 1710 in D.D. 114, Shek Kong, Yuen Long, New Territories Drainage Proposal

Prepared for Join Bright Warehousing Limited

For	For and on behalf of						
Env	iroSolutio	ns & Consulting					
Alex	ci BHANJA	L .					
Gro	up COO						
<b>FCC</b>	Ducient N	- 124 00171	UK 01				
ESC	Project No	<b>b</b> . J24.00171	.пк.01				
Deli	verable N	<b>o</b> . D02					
Revi	sion No.	1					
File I	ocation						
https	//envirosc sl	narenoint com/tean	ns/newsharenoint/sha	red documer	nts/new share	enoint/05	
jobs/j	24.00171.hk	.01 kam tin cbp s16,	/06. deliverables/d01	drainage prop	posal rev0.2_	aw.docx	
Rev.	Description		Prenared	Reviewed	Approved	Date	
nen	Description	•	Teparca	heneneu	Approved	Dute	
0	Drainage P	roposal	MJW	CL	AW	01/08/2024	
1	Drainage P	roposal	MJW	CL	AW	30/12/2024	
Distri	bution	Internal	🛛 Confidentia	l	Public		
This r	eport has be	en prepared by Envi	roSolutions & Consulti	ng Limited w	ith all reasona	able skill, care,	
and d	liligence witl	hin the terms of the	e Contract with Client	, incorporati	ng our Gene	ral Terms and	
We di	'e disclaim any responsibility to Client and others in respect or any matters outside the scope of the						
above	ove. This report is confidential to Client and we accept no responsibility of whatsoever nature to						

any third parties to whom this report, or any part thereof, is made known.



### CONTENTS

1	PROJECT BACKGROUND		1-1	
	1.1	Introduction	1-1	
	1.2	Objectives of this Penert	1 <sup>-</sup> 1	
	1.5 1.4	Reference Materials	1-1 1-1	
2	DESCRI	PTION OF EXISTING ENVIRONMENT AND DRAINAGE CONDITIONS	2-1	
	2.1 Site Location and Topography		2-1	
	2.2	Existing Baseline Conditions	2-1	
	2.3	Proposed Discharge Point	2-1	
3	DRAINA	GE ANALYSIS	3-1	
	3.1	Assumptions and Methodology	3-1	
	3.2	Assessment Assumptions	3-2	
	3.3	Estimated Runoff	3-4	
	3.4	Capacity of Proposed Perimeter Surface Drains and Underground Pipe	3-4	
4	CONCLU	JSION	4-1	

### **APPENDICES**

Appendix A	Runoff Calculations
Appendix B	Calculation of Drainage Capacity
Appendix C	Cross Section of the Site and the Surrounding Area After the Proposed Development

### **FIGURES**

Figure 1-1	Site Location and its Environs	. 1-2
Figure 2-1	Proposed Discharge Point	.2-2
Figure 3-1	Identification of Catchments	.3-6
Figure 3-2	Indicative Location of Proposed Terminal Manhole and Stormwater Pipe	.3-7
Figure 3-3	Typical Details of Catchpit and Sand Trap	.3-8
Figure 3-4	Typical Details of Sand Trap	.3-9

### **TABLES**

Table 3-1	Runoff Coefficients with Different Surface Characteristics and	3-2
Table 3-2	Surface Characteristics and Runoff Coefficients of the Site	3-3
Table 3-3	Surface Characteristics and Runoff Coefficients of Surrounding Catchments	3-3
Table 3-3	Estimated Peak Runoff of the Site	3-4
Table 3-5	Estimated Cumulative Runoff of the Site and Catchment B	3-4
Table 3-6	Summary of Indicative Perimeter Drainage System	3-5
Table 3-6	Summary of Indicative Perimeter Drainage System	3-5



### **1 PROJECT BACKGROUND**

### 1.1 Introduction

- 1.1.1 Land resumption will be carried out to facilitate the development of Hung Shui Kiu/ Ha Tsuen New Development Area. An existing Concrete Batching Plant ("CBP") in Hung Shui Kiu will be affected by the land resumption. Thus, a site located at Lots 573 RP and 1710 in D.D. 114, Shek Kong ("the Site") has been identified for relocating the existing CBP to the Site ("the Proposed Development").
- 1.1.2 The Site is currently zoned "Industrial (Group D)" ("I(D)") on the Approved Shek Kong Outline Zoning Plan ("OZP") No. S/YL-SK/9 which "Concrete Batching Plant" is under Column 2 in accordance with the Schedule of Use for I(D) zone. As such, a planning application is required to submit and obtain permission from the Town Planning Board under Section 16 of the *Town Planning Ordinance* for the Proposed Development.
- 1.1.3 EnviroSolutions & Consulting Limited ("ESC") has been engaged to carry out a Drainage Proposal to support the abovementioned application.

### **1.2** Site Description

- 1.2.1 The Site locations and its environs are shown in **Figure 1-1** which the uses surrounding the Site include:
  - To the North: temporary structures with industrial use and vegetated slope
  - To the East: temporary structures with industrial use and vegetation
  - To the South: temporary structures with industrial use
  - To the West: temporary structures with industrial use

### **1.3** Objectives of this Report

- 1.3.1 The objectives of this Drainage Proposal are to:
  - Assess the potential drainage impacts arising from the Proposed Development.
  - Recommend the necessary mitigation measures to alleviate any impacts, if any.

### **1.4 Reference Materials**

- 1.4.1 In evaluating the drainage impact arising from the Proposed Development, the following materials have been referred to:
  - Stormwater Drainage Manual (with Eurocodes incorporated) Planning, Design and Management (Fifth Edition, January 2018)
  - Stormwater Drainage Manual Corrigenda Nos. 1/2022, 1/2024 and 2/2024

Page 1-1

- DSD Advice Note No. 1 Application of the Drainage Impact Assessment Process to Private Sector Projects
- Technical Note to prepare a Drainage Submission
- Drainage Data of GeoInfo Map reviewed on 11 July 2024.



### *Figure 1-1 Site Location and its Environs*





### 2 DESCRIPTION OF EXISTING ENVIRONMENT AND DRAINAGE CONDITIONS

### 2.1 Site Location and Topography

- 2.1.1 As illustrated in **Figure 1-1**, the Site is surrounded by various temporary structures with industrial use and vegetated slopes to the north and east.
- 2.1.2 The Site is situated at Lots 573 RP and 1710 in D.D. 114. The Site area is about 4,411m<sup>2</sup> with elevation ranging from +52.9mPD to +48.1mPD.

### 2.2 Existing Baseline Conditions

2.2.1 According to the drainage record provided in Geoinfo Map, no public drainage system was observed in the vicinity of the Site. A site inspection was conducted on 10 January 2023 to review the existing site condition and identify the drainage connection of the surrounding catchments. The entire Site is currently paved, with 100% of the Site being paved with concrete. The detailed discussion about surrounding catchments is provided in the following sections.

### 2.3 Proposed Discharge Point

2.3.1 It is proposed to discharge the stormwater runoff from the Site to the existing stream to the south of the Site. The indicative discharge point is shown in **Figure 2-1**.



### Figure 2-1 Proposed Discharge Point





### **3 DRAINAGE ANALYSIS**

### 3.1 Assumptions and Methodology

- 3.1.1 Peak instantaneous run off before and after the Proposed Development was calculated based on the Rational Method. The recommended physical parameters, including runoff coefficient (C) and storm constants for different return periods, are as per the *Stormwater Drainage Manual*.
- 3.1.2 The Rational Method has been adopted for hydraulic analysis and the peak runoff is given by the following expression:

 $Q_p = 0.278 C i A$  --- Equation 1

where  $Q_p = peak runoff in m^3/s$  C = runoff coefficient i = rainfall intensity in mm/hr $A = catchment area in km^2$ 

3.1.3 Rainfall intensity is calculated using the following expression:

$$i = \frac{a}{(t_d + b)^c} \qquad \qquad \text{--- Equation 2}$$

where

i = rainfall intensity in mm/hr td = duration in minutes (td≤240) a,b,c = storm constants given in table 3 of SDM

3.1.4 For a single catchment, duration (t<sub>d</sub>) can be assumed equal to the time of concentration (t<sub>c</sub>) which is calculated as follows:

$$t_c = t_0 + t_f$$
 --- Equation 3

where  $t_c$  = time of correction  $t_0$  = inlet time (time taken for flow from the remotest point to reach the most upstream point of the urban drainage system)  $t_f$  = flow time

3.1.5 Generally,  $t_0$  is much larger than  $t_f$ . As shown in Equation 2,  $t_d$  is the divisor. Therefore, larger  $t_d$  will result in smaller rainfall intensity (i) as well as smaller  $Q_p$ . For the worst-case scenario,  $t_f$  is assumed to be negligible and so:

$$t_c = t_0 = t_f$$
  
 $t_0 = \frac{0.14465 L}{H^{0.2} A^{0.1}}$  ---- Equation 4

where A = catchment area (m<sup>2</sup>)
 H = average slope (m per 100m), measured along the line of natural flow, from the summit of the catchment to the point under consideration
 L = distance (on plan) measured on the line of natural flow between the summit and the point under consideration (m)



3.1.6 The Colebrook-White Equation was adopted for calculation of drainage capacity of pipes. Full bore flow with no surcharge is assumed, and 10% sedimentation was incorporated in the calculation of drainage capacity in accordance with the *Stormwater Drainage Manual*.

$$V = -\sqrt{8gDsg} * \log(\frac{ks}{3.7D} + \frac{2.51v}{D\sqrt{2gDs}})$$

where

V = mean velocity (m/s) g = gravitational acceleration (m/s<sup>2</sup>)

D = internal pipe diameter (m)

- ks = hydraulic pipeline roughness (m)
- v = kinematic viscosity of fluid (m<sup>2</sup>/s)
- s = hydraulic gradient (energy loss per unit length due to friction)
- 3.1.7 On the other hand, the capacity of open channel has been calculated using the Manning's Equation:

$$V = \frac{R^{1/6}}{n} \times \sqrt{Rs}$$

--- Equation 6

--- Equation 5

where

V = mean velocity (m/s)

R = hydraulic radius (m)

n = Meaning coefficient ( $s/m^{1/3}$ )

s = hydraulic gradient (energy loss per unit length due to friction)

### 3.2 Assessment Assumptions

### Identification of Catchments

3.2.1 Based on the site visit and the topographic maps obtained from the Lands Department, seven (7) catchments namely Catchments A to G were identified as shown in **Figure 3-1**.

### Internal Catchment (the Site)

- 3.2.2 The Site includes Catchment A. The entire Site is currently 100% paved with some concrete-paved roads and temporary structures.
- 3.2.3 After the Proposed Development, the Site will be occupied by the CBP and other structures and will still be entirely paved. A peripheral drainage is proposed to intercept the stormwater runoff from the Site. It is assumed that the runoff will be finally discharged to the stream to the south of the Site via an underground pipe. The details will be further discussed in the following sections.
- 3.2.4 With reference to the *Stormwater Drainage Manual*, the runoff coefficients vary from different surface characteristics, as summarized in **Table 3-1** below.

### Table 3-1 Runoff Coefficients with Different Surface Characteristics and

SURFACE CHARACTERISTICS	RUNOFF COEFFICIENT		
Concrete	0.80 – 0.95		
Flat Grassland (heavy soil)	0.13 - 0.25		
Steep Grassland (heavy soil)	0.25 – 0.35		



3.2.5 As the Site is 100% paved before and after development, runoff coefficient of 0.95 before and after development should be adopted. The surface characteristics and runoff coefficients of Catchment A are summarised in **Table 3-2** below.

Table 3-2	Surface Characteristics	and Runoff	Coefficients of the Site
-----------	-------------------------	------------	--------------------------

SCENARIO	AREA	SURFACE CHARACTERISTICS	RUNOFF COEFFICIENT
Before and After Development	A: 4,411m <sup>2</sup>	100% paved + 0% unpaved	0.95

### **Surrounding Catchments**

- 3.2.6 **Catchment B** According to the site visit, Catchment B is occupied by some temporary structures with industrial use, which is assumed to be 100% paved with concrete. As there is no drainage system observed in the vicinity, it is assumed that stormwater runoff from Catchment B would overflow to Catchment A as conservative approach. Runoff from Catchment B will be taken into account in the proposed perimeter drainage system of Catchment A.
- 3.2.7 **Catchment C** Similarly, Catchment C is occupied by some temporary structures with industrial use, which is assumed to be 100% paved with concrete. According to level differences, it is assumed that stormwater runoff from Catchment C would overflow to Catchment A as conservative approach. Runoff from Catchment C will be taken into account in the proposed perimeter drainage system of Catchment A.
- 3.2.8 **Catchment D** Catchment D is a piece of vegetated slope located to the north of the Site. The stormwater runoff from Catchment D should be collected and drained by its toe drain and will not flow into the Site. The runoff from this catchment will not have any drainage impact on the Site.
- 3.2.9 **Catchment E** Catchment E is located to the east of the Site. The runoff from Catchment E would be discharged to the existing stream. There is no drainage connect between Catchment E and the Site, the runoff from this catchment will not have any drainage impact on the Site.
- 3.2.10 Catchment F Catchment F is located to the south of the Site. According to the topographic levels, the runoff from Catchment F would be discharged to existing stream to the south. There is no drainage connect between Catchment F and the Site, the runoff from this catchment will not have any drainage impact on the Site.
- 3.2.11 **Catchment G** Catchment G is located to the west of the Site. According to the topographic levels, the runoff from Catchment G would go further to the west. There is no drainage connect between Catchment F and the Site, the runoff from this catchment will not have any drainage impact on the Site.
- 3.2.12 The runoff of Catchments B and C are estimated using the Rational Method. The surface characteristics and runoff coefficients of the surrounding catchments are summarised in **Table 3-3** below.

### Table 3-3 Surface Characteristics and Runoff Coefficients of Surrounding Catchments

CATCHMENTS	AREA	SURFACE CHARACTERISTICS	RUNOFF COEFFICIENT



В	1,654m²	100% paved (flat) + 0% unpaved	0.95
С	1,577m <sup>2</sup>	100% paved (flat) + 0% unpaved	0.95

### 3.3 Estimated Runoff

### Peak Runoff from the Site

3.3.1 Based on the assumptions as described in **Section 3.2**, the runoff from the Site before and after development has been estimated based on the return periods of 2, 10 and 50 years. The increase in rainfall due to climate change effect of 11.1% has been considered in the runoff estimation. As summarised in **Table 3-4**, there will be no change on the stormwater runoff before and after the Proposed Development under all assessed return periods. The detailed calculation is provided in **Appendix A**.

Table 3-4 Estimated Peak Runoff of the Site	able 3-4	Estimated Peak Runoff of the S	ite
---	----------	--------------------------------	-----

RETURN	ESTIMATED PEAK RUNOFF (m³/s)				
PERIOD	BEFORE DEVELOPMENT	AFTER DEVELOPMENT	INCREMENT		
2 Years	0.214	0.214	0%		
10 Years	0.271	0.271	0%		
50 Years	0.307	0.307	0%		

### **Cumulative Peak Runoff**

3.3.2 As mentioned in **Paragraph 3.2.6**, it is assumed that the runoff from Catchments B and C may overflow to Catchment A in the worst-case scenario. Therefore, runoff from Catchments B and C will be regarded as the cumulative runoff. The estimated cumulative runoff is summarised in **Table 3-5.** below and detailed in **Appendix A**.

### Table 3-5Estimated Cumulative Runoff of the Site and Catchment B

	ESTIMATED PEAK RUNOFF (m³/s)					
RETORN PERIOD	CATCHMENT A	CATCHMENT B	CATCHMENT C	CUMULATIVE		
2 Years	0.214	0.094	0.099	0.408		
10 Years	0.271	0.118	0.123	0.512		
50 Years	0.307	0.205	0.135	0.572		

### 3.4 Capacity of Proposed Perimeter Surface Drains and Underground Pipe

- 3.4.1 A series of perimeter surface drains with sand trap/catch pit was proposed to collect the cumulative runoff, which will finally connect to proposed discharge point at the stream to the south of the Site via a stormwater drainage pipe. The indicative location of surface drains, catchpits, terminal manhole and proposed Ø650mm stormwater drainage pipe is shown in **Figure 3-2**.
- 3.4.2 The calculation on the capacity of the indicative perimeter surface drains and proposed Ø650 stormwater pipe are summarised in **Table 3-6** below and detailed in **Appendix B**.



### Table 3-6Summary of Indicative Perimeter Drainage System

DESCRIPTION	SIZE (mm)	RELATED CATCHMENT	RUNOFF (m³/s)	CAPACITY (m³/s)	% OF CAPACITY	SUFFICIENT CAPACITY?
U-Channel with gradient of 1:200	Ø600mm	Catchment A and B	0.284	0.449	63%	Yes
U-Channel with gradient of 1:200	Ø600mm	Catchment A and C	0.288	0.449	64%	Yes
Underground Pipe with gradient of 1:200	Ø650mm	Catchment A, B and C	0.572	0.663	86%	Yes

3.4.3 The calculation shows that both the proposed perimeter surface drains and Ø650mm stormwater drainage pipe have sufficient capacity for the cumulative runoff. Therefore, no adverse drainage impact due to the Proposed Development is anticipated.

### *Figure 3-1 Identification of Catchments*









Note: Some buffer is provided between the southern border of the Site and the perimeter surface drain as there are some concrete blocks located on the south of the Site.











### Figure 3-4 Typical Details of Sand Trap





### 4 CONCLUSION

- 4.1.1 Potential drainage impacts that may arise from the Site after the Proposed Development have been assessed.
- 4.1.2 The peak runoff before and after development of the Site has been estimated using the Rational Method and based on the catchment surface characteristics for the existing environment and the Proposed Development. There will be no change to runoff before and after the Proposed Development under all assessed return periods. The estimated peak runoff generated from the Site is 0.307m<sup>3</sup>/s under a 50-year return period.
- 4.1.3 The indicative location of proposed perimeter surface drains, terminal manhole and Ø650 stormwater pipe shown on **Figure 3-2** will properly divert the runoff arising from the Site including cumulative runoff from Catchments B and C, which may overflow into the Site. The runoff would finally be discharged to the stream to the south of the Site via the proposed Ø650 stormwater drainage pipe.
- 4.1.4 The capacity of proposed perimeter surface drains and proposed Ø650 stormwater drainage pipe has been checked. The calculation shows that it can handle the cumulative runoff from the Site and surrounding catchments. As such, no adverse drainage impact is anticipated.
- 4.1.5 This Drainage Proposal Report indicates the initial findings regarding drainage impact and indicative drainage layout. A qualified engineer should be engaged by the Architect/Contractor of the Proposed Development to review and provide detailed designs for the internal Site drainage layout.
- 4.1.6 Overall, adverse drainage impact from the Proposed Development is not anticipated.


# Appendix A Runoff Calculations



### Calculation of Runoff for Return Period of 10 Years

Cost of warmen to ID	Catchment Area (A),	Average slope (H),	Flow path length	Inlet time (t <sub>0</sub> ), min Du	Duration (t.) min	Storm Constants [Note 2]		Runoff intensity (i)	Dura off an affiniant (C)		Peak runoff (Q <sub>p</sub> ),	Peak runoff with Climate	
Catchment ID	km <sup>2</sup>	m/100m	(L), m		Duration $(t_d)$ , min	а	b	с	mm/hr	Runon coemcient (c)	CAA	m³/s	Change (Q' <sub>p</sub> ), m <sup>3</sup> /s <sup>[Note 3]</sup>
Before the Proposed Develop	fore the Proposed Development												
Site Area (Catchment A)	0.0044	4.64	112.0	5.15	5.15	485	3.11	0.397	209.76	0.95	0.0042	0.244	0.271
Catchment B	0.0017	11.45	62.0	2.62	2.62	485	3.11	0.397	242.44	0.95	0.0016	0.106	0.118
Catchment C	0.0016	14.44	36.0	1.46	1.46	485	3.11	0.397	265.27	0.95	0.0015	0.110	0.123
											Total	0.461	0.512
After the Proposed Developm	ent												
Site Area (Catchment A)	0.0044	4.64	112.0	5.15	5.15	485	3.11	0.397	209.76	0.95	0.0042	0.244	0.271
Catchment B	0.0017	11.45	62.0	2.62	2.62	485	3.11	0.397	242.44	0.95	0.0016	0.106	0.118
Catchment C	0.0016	14.44	36.0	1.46	1.46	485	3.11	0.397	265.27	0.95	0.0015	0.110	0.123
											Total	0.461	0.512

### Calculation of Runoff for Return Period of 50 Years

Catchmont ID	Catchment Area (A),	Average slope (H),	Flow path length	Inlot time (t.) min	Duration (t.) min	Storm	Constants	[Note 2]	Runoff intensity (i)	Bunoff coofficient (C)	C * A	Peak runoff (Q <sub>p</sub> ),	Peak runoff with Climate
Catchment ID	km <sup>2</sup>	m/100m	(L), m	iniet time (t <sub>0</sub> ), min	Duration (t <sub>d</sub> ), min	а	Ь	с	mm/hr	Kunon coendent (C)	CAR	m³/s	Change (Q' <sub>p</sub> ), m <sup>3</sup> /s [Note 3]
Before the Proposed Develop	fore the Proposed Development												
Site Area (Catchment A)	0.0044	4.64	112.0	5.15	5.15	505.5	3.29	0.355	237.08	0.95	0.0042	0.276	0.307
Catchment B	0.0017	11.45	62.0	2.62	2.62	505.5	3.29	0.355	268.96	0.95	0.0016	0.117	0.131
Catchment C	0.0016	14.44	36.0	1.46	1.46	505.5	3.29	0.355	290.69	0.95	0.0015	0.121	0.135
											Total	0.515	0.572
After the Proposed Developm	ent												
Site Area (Catchment A)	0.0044	4.64	112.0	5.15	5.15	505.5	3.29	0.355	237.08	0.95	0.0042	0.276	0.307
Catchment B	0.0017	11.45	62.0	2.62	2.62	505.5	3.29	0.355	268.96	0.95	0.0016	0.117	0.131
Catchment C	0.0016	14.44	36.0	1.46	1.46	505.5	3.29	0.355	290.69	0.95	0.0015	0.121	0.135
											Total	0.515	0.572

Note:

1. Runoff is calculated in accordance with DSD's "Stormwater Drainage Manual (with Eurocodes incorporated) - Planning, Design and Management" (SDM), fifth edition, January 2018.

2. Storm Constants were adopted from Table 3a Storm Constants for Different Return Periods of HKO Headquarters of DSD's Corrigendum No. 1/2024.

3. Table 28 Rainfall Increase due to Climate Change of DSD's *Corrigendum No. 1/2022* of 11.1% for mid-21st Century is adopted.



# **Appendix B** Calculation of Drainage Capacity



#### Calculation of Drainage Capacity for Return Period of 50 Years

#### Indicative Drainage capacity of the Internal Drainage System

Channel	Shape	Catchment Description	d	Depth, m	Slope (s)	Aw	Pw	R	n	V	Qc	Qp'	Is Qc > Qp' ?	% of capacity
U-Channel 1	U-Shape	1/2 Catchment A, Catchment B	0.600	0.300	0.005	0.321	1.542	0.208	0.016	1.553	0.449	0.284	OK	63%
U-Channel 2	U-Shape	1/2 Catchment A, Catchment C	0.600	0.300	0.005	0.321	1.542	0.208	0.016	1.553	0.449	0.288	OK	64%

### Legend

d = diameter, m	n = Manning's roughness coefficient
A <sub>w</sub> = Cross Section Area of Flow, m <sup>2</sup>	V = Mean Velocity, m/s
P <sub>w</sub> = Wetted Perimeter, m	Q <sub>c</sub> = Flow Capacity (10% sedimentation inclusive), m <sup>3</sup> /s
R = Hydraulic Radius = A <sub>w</sub> /P <sub>w</sub> , m	Q <sub>p</sub> = Estimated Peak Flow, m <sup>3</sup> /s
s = Hydraulic Gradient	

#### Note

1. Flow capacity of pipe segment is calculated based on Manning's Equation.

2. The diameter and gradient of the proposed stormwater pipe is indicative only. Its details will be subject to change during the detailed design stage.

#### Drainage Capacity of Proposed Stormwater Drainage Pipe

Description	Shape	Catchment Description	d	r	Aw	Pw	R	S	ks	V	Qc	Qp'	Is Qc > Qp' ?	% of capacity
Proposed Stormwater Pipe Connecting to Proposed Discharge Point	Circular Pipe	Catchment A, Catchment B	0.650	0.325	0.332	2.042	0.163	0.005	0.06	2.220	0.663	0.572	Y	86%

#### Where

d = pipe diameter, m	k <sub>s</sub> = hydraulic pipeline roughness, mm
r = pipe radius (m) = 0.5d	V = Velocity of flow calculated based on Colebrook-White Equation, m/s
$A_w = wetted area (m^2) = (r^2/2) (b + sinq)$	Q <sub>c</sub> = Flow Capacity including 10% for siltation, m <sup>3</sup> /s
P <sub>w</sub> = wetted perimeter (m) = br	Q <sub>p</sub> = Estimated total peak flow from the Site during peak season, m <sup>3</sup> /s
R = Hydraulic radius (m) = A <sub>w</sub> /P <sub>w</sub>	

#### Note

1. Flow capacity of pipe segment is calculated based on Colebrook-White Equation.

2. The diameter and gradient of the proposed stormwater pipe is indicative only. Its details will be subject to change during the detailed design stage.

3. The ks value of 0.06 in good condition for precast concrete pipes with 'O' ring joints recommended in Table 14 of the SDM for design purpose is adopted.



# Appendix C

Cross Section of the Site and the Surrounding Area After the Proposed Development



## Figure C-1Cross Section of the Site





## **EnviroSolutions & Consulting Ltd**

Solutions for Environment | Safety | Sustainability

# EnviroSolutions & Consulting Ltd

# 000

# Accountability

We understand the importance of being accountable to each other and our clients.



### Passion

We are completely passionate about providing practical solutions and outcomes that deliver for our clients.



### Insight

We work in an environment that encourages and values insight as a critical quality which informs our decisions and our clients and supports practical solutions and project delivery.



### Integrity

We behave with respect and honesty toward each other, our clients and our stakeholders.



 Our Ref.:
 DD 114 Lots 573 RP & 1710

 Your Ref.:
 TPB/A/YL-SK/410

The Secretary, Town Planning Board, 15/F, North Point Government Offices, 333 Java Road, North Point, Hong Kong

Dear Sir,

# Further Information

# Proposed Temporary Concrete Batching Plant with Ancillary Facilities for a Period of 5 Years in "Industrial (Group D)" Zone, Lots 573 RP and 1710 in D.D. 114, Shek Kong, Yuen Long, New Territories

# (S.16 Planning Application No. A/YL-SK/410)

We write to submit further information to provide clarifications on the subject application.

Should you require more information regarding the application, please contact our or the undersigned at your convenience.

Thank you for your kind attention.

Yours faithfully,

For and on behalf of R-riches Property Consultants Limited

Christian CHIM Town Planner

cc DPO/FSYLE, PlanD



<u>By Email</u>

08 April 2025

# **Further Information**

# Proposed Temporary Concrete Batching Plant with Ancillary Facilities for a Period of 5 Years in "Industrial (Group D)" Zone, Lots 573 RP and 1710 in D.D. 114, Shek Kong, Yuen Long, New Territories

# (Application No. A/YL-SK/410)

# (i) The applicant provides responses to the following <u>departmental comments</u>:

	Departmental Comments	Applicant's Responses
1. C	Comments of the Director of Agriculture, Fishe	eries and Conservation (DAFC)
(a)	According to the site plan, it is noted that the subject site would encroach into existing stream along its southern boundary. In this regard, the proposed development will pose direct impact on the stream while stream and watercourse should be preserved as far as possible from nature conservation perspective.	Although there is a minor encroachment of Lot 573 RP in D.D. 114 into the watercourse along its southern boundary, all structures of the proposed development maintain a distance of <u>at</u> <u>least 3.8 m from the top bank of the watercourse</u> (please refer to the proposed layout at Plan 10 of the original submission). Besides, <u>no filling of</u> <u>land will be carried out in any part of the</u> <u>watercourse</u> .
		In order to protect the watercourse as far as possible, the following measures are proposed in the reports of the Environmental Assessment (EA) and Drainage Proposal:
		<ul> <li>sewage arising from the proposed development would be temporarily stored in sewage holding tanks and then tankered away for off-site disposal in a sewage treatment facility; and</li> </ul>
		<ul> <li>perimeter channels with sand traps/catch pits are proposed to collect the surface runoff from the application site (the Site), so that no surface runoff would be washed directly into the watercourse.</li> </ul>
		In view of the above, it is expected that the proposed development would not impose adverse environmental and drainage impacts on the watercourse.



	Departmental Comments	Applicant's Responses					
2. 0	comments of the District Lands Officer/Yuen I	Long, Lands Department (DLO/YL, LandsD)					
(a)	By desktop checking, the boundary of the Application Site deviates from the boundary of the Lots and encroached on Lot No. 574 S.A in D.D. 114 and the adjoining Government land. The actual lot boundary of the private lots under the application have to be verified at the land application stage if any STW and/or STT is/are applied for by the Applicant to LandsD.	Noted. Matters on land administration will be liaised with LandsD upon the submission of applications for Short Term Waiver and/or Short Term Tenancy after obtaining the planning permission from the Town Planning Board (the Board). Detailed survey was conducted to delineate the lot boundaries, i.e. Lot Nos. 573 RP and 1710 in D.D. 114. The Site consists of 2 private lots with private land area of 4,332 m <sup>2</sup> (about) of Old Schedule Lot held under Block Government Lease (Lot No. 573 RP in D.D. 114) and 79 m <sup>2</sup> (about) of New Grant Lot (Lot No. 1710 in D.D. 114). On 4.10.2024, the submission of Land Boundary Plan and Survey Record Plan of Lot Nos. 573 RP and 1710 in D.D. 114 was acknowledged by the District Survey Office/Yuen Long, LandsD (Annex 1).					

- (ii) The applicant provides the "Memorandum of Understanding" signed by the applicant and the affected business operator, who has <u>authorised</u> the applicant to submit the current application on its behalf (**Annex 2**).
- (iii) The applicant provides the following clarifications on the <u>justifications</u> of the current application:
  - with reference to the implementation programme, the affected premises falls within the land resumption limit for the second phase development of the Hung Shui Kiu/Ha Tsuen New Development Area (HSK/HT NDA). As land where the premises will be developed for industrial and government, Institution or community uses, the concerned parcel of land is expected to be resumed and reverted to the Government in 2025. Therefore, the applicant desperately needs to identify a suitable site for relocation in order to continue its business operation;
  - ready-mixed concrete is one of the most commonly-used raw materials for construction projects in Hong Kong. With the continued development across various NDAs in the New Territories (such as HSK/HT NDA, Kwu Tung North NDA etc.), the amount of construction works and the demand for concrete arising from planned development projects will remain high. The continued operation of the concrete batching plant can ensure a steady and



sustainable concrete supply and reduce cross-region traffic for concrete delivery; and

- in view of the continued development across various NDAs in the New Territories, the increase in silo capacity and production capacity is to meet the growing demand for readymixed concrete production. The applicant will adopt mitigation measures to alleviate the potential environmental impacts whilst maintaining a more efficient production of readymixed concrete.
- (iv) The applicant provides the following clarifications on the <u>proposed road work</u> on the local access i.e. Chun Yiu Road:
  - the local access is an existing paved road with a minimum width of about 6 m, which allows 2-lane 2-way traffic between the Site and Kam Tin Road. However, portion of the local access has fallen into disrepair and is currently covered with debris/leftovers. Some vehicles, with or without valid vehicle licences, have been illegally parked/stored on the side of the local access;
  - as mentioned in Section 5.4 of the Planning Statement, the applicant intends to carry out upgrading work on the local access, including the removal of debris/leftovers and illegallyparked vehicles. Therefore, the proposed upgrading work solely comprises minor maintenance works to the existing local access. Figure 3.1 from the Traffic Impact Assessment showing the proposed upgrading work of the local access has been revised (Annex 3);
  - since the proposed road upgrading work is currently at the preliminary planning stage, the final alignment of the local access shall subject to detailed design, which will be provided at a later stage upon obtaining planning permission from the Board; and
  - the applicant undertakes to submit detailed proposal for the proposed road upgrading work after obtaining the planning permission from the Board. The subsequent implementation works for the road upgrading work would be carried out in accordance with relevant government guidelines and regulations; and such works would be completed by the applicant before commencement of the proposed development. The applicant also commits to manage and carry out maintenance work for the local access.
- (v) The applicant provides the following clarifications on <u>development parameters</u> of the proposed development:
  - the proposed concrete batching plant comprises 6 silos with total capacity of about 2,100 tonnes for storage of raw materials for concrete production (i.e. cement, pulverised fuel ash (PFA), ground blastfurnace slag (GGBS), and silica fume (CSF)); and
  - the operation hours are from 07:00 to 23:00 daily (i.e. from Mondays to Sundays (including public holidays)), which consist of a 12-hour production period between 07:00 to 19:00, and a raw material delivery period from 09:00 to 19:00. An evening delivery period of raw materials beyond peak hours between 19:00 and 23:00 will also take place. Except for delivery of ice which cannot be produced/stored on-site, no other raw materials will be



delivered between 07:00 and 09:00. It is estimated that the Site would be able to accommodate about 12 workers. The main source of visitors will be drivers who deliver raw materials/concrete products. Outside visitor is not anticipated.

- (vi) The applicant provides the following clarifications on the <u>drainage matters</u> of the proposed development:
  - all drainage works will be placed at least 3 m away from the top of the bank of the watercourse. All structures will maintain a distance of at least 3.8 m from the top bank of the watercourse. Appropriate flooding mitigation measures will be provided at the resources of the applicant to the satisfaction of the Drainage Services Department, if necessary;
  - perimeter channels with sand traps/catch pits are proposed to collect the surface runoff from the Site, so that no surface runoff would be washed directly into the watercourse. The applicant also ensures that adequate openings will be provided along the site hoardings to intercept the existing overland flow passing through the Site;
  - the surface runoff from water sprinklers and wheel washing facilities will be diverted to sedimentation tank for silt removal. After that, the silt/sediment (i.e. cement cake) will be disposed of at fill bank reception facility managed by the Civil Engineering and Development Department. The treated water will then be reused for daily operation of the concrete batching plant, including concrete mixing, vehicle washing etc;
  - the drainage system for surface runoff from the Site and overland flow passing through the Site (i.e. perimeter channels and sand trap/catch pit) will be separated from the drainage system for surface runoff from water sprinklers and wheel washing facilities (i.e. sedimentation tank); and
  - the applicant commits to submit a revised drainage proposal with the provision of on-site drainage facilities and sedimentation tanks before commencement of the proposed development. The applicant will comply with relevant ordinances and regulations for the provision of on-site drainage facilities. The applicant also commits to manage and carry out maintenance work for the on-site drainage facilities.
- (vii) The applicant provides the following clarifications on the <u>environmental matters</u> of the proposed development:
  - some temporary structures are found beyond the northern periphery of the Site. In order to
    mitigate the potential adverse noise, air quality and visual impacts on these structures, the
    applicant proposes Aggregate Stockpile No. 2 (which will serve to stockpile finer aggregates)
    to be enclosed on the top and the 3 sides;
  - coarser aggregates will be stockpiled in Aggregate Stockpile No. 1. Albeit not to be enclosed on the top, metal walls will be erected along the 3-sides of Aggregate Stockpile No. 1;
  - the following mitigation measures on both aggregate stockpiles are proposed in the EA to



alleviate the potential adverse impacts to the surroundings (refer to Sections 2.3 and 3.4 of the EA for details):

- provision of hard paving on open area and regular watering to reduce dust emissions;
- stockpile of dusty materials shall be either covered entirely by impervious sheeting, placed in an area sheltered on the top and 3 sides, or sprayed with water so as to maintain the entire surface wet;
- tarpaulin covering of all dusty vehicles loads transported to, from and between site locations; and
- major noise sources will be enclosed/semi enclosed; and
- the applicant will follow relevant ordinances and regulations for the operation of the proposed development. The applicant will obtain relevant licences such as Specified Process (SP) Licence and relevant waste disposal permits from the appropriate authorities; including Chemical Waste Permits/licenses under the *Waste Disposal Ordinance (Cap 354), Water Pollution Control Ordinance (Cap 358),* Public Dumping Licence under the *Land (Miscellaneous Provisions) Ordinance (Cap 28),* Effluent Discharge Licence under the *Water Pollution Control Ordinance.*
- (viii) The applicant provides the following clarifications on the <u>concrete production workflow</u>:
  - raw materials for concrete production delivered to the Site will be stored in respective silos and aggregate stockpiles;
  - aggregates, cement, ice and other raw materials (including PFA, GGBS and CFS) will be transferred to the mixing towers via the receiving hoppers, bucket elevators and conveyor belts for mixing;
  - the mixed concrete will be loaded onto concrete trucks for delivery; and
  - please refer to the schematic diagram enclosed at **Annex 4** for details of the production workflow.
- (ix) The applicant provides a <u>revised layout plan</u> to include indications of the site boundary, and the width and location of the site ingress/egress (**Annex 5**).



Annex 1

Land Boundary Plan and Survey Record Plan



窇	話	Tel:	3547 1341					
圖文傳真 Fax;		2473 6996						
電郵地址 Email:		lsy14@landsd.gov.hk						
本署初	當號	Our Ref:	(	) in LD DSO/YL/W/426/2021				
本:ふ	14 HA	Your Bef	(	) in LD DSO/YL/W/227/2017				

來집請註明本署檔號 Please quote our reference in your reply



地政總署 元朗測量處 DISTRICT SURVEY OFFICE YUEN LONG LANDS DEPARTMENT

我們矢志努力不讓,提供盡善盡美的土地行政服務。 We strive to achieve excellence in land administration.

新界元朝元政約二號 NO. 2 YUEN CHING ROAD, YUEN LONG, N.T.

網站 Website : www.landsd.gov.hk

By Fax: 2666 9921

4 - OCT 2024

Dear Sir,

# Survey Plan of Lot 573 RP & Lot 1710 in D.D. 114

I acknowledge receipt of your duly completed standard form SMF-0043, survey record plan no. <u>SRP/YL/057/06741/D1</u>, land boundary plan no. <u>LBP/YL/057/06741/D1</u> and the survey report voluntarily submitted in respect of the land boundary survey of the captioned lot(s) with the consent of the \*registered land owner / authorized representative of the registered land owner.

Your plan(s) and survey report will be handled in accordance with SMO Practice Note No. 1/2024.

Yours faithfull

(CHOY Hoi/shing, Ivan) for Senior Land Surveyor/Yuen Long Lands Department



測繪處 Survey and Mapping Office

本信息及任何附件只供收件人使用,而其中可能载有機盛及/或屬法律特権的資料。敬請注意,未獲許可,不得擅自披露或使用本信息。倘本信息誤 傳給你,請立即通知本署,並刪除或納毀本信息。本署絕不承擔因使用本信息而引致的任何法律責任。

This message including any attachment is intended for the use of the addressee only. It may contain information which is confidential and/or legally privileged. You are hereby notified that no unauthorized disclosure or use of this message is permitted. If you have received this message by mistake, please notify us immediately and delete or destroy this message, as appropriate. Any liability arising from the use of this information is excluded. 自願提交土地界線圖及測量記錄圖予分區測量處供分享資料之用

Voluntary Submission of Land Boundary Plan and Survey Record Plan to District Survey Office for Sharing Purpose

本表格由以下人士提交(只可在其中一項填上「√」號): This form is submitted by (tick only one of the boxes below):

# \_\_\_\_\_ 認可土地測量師/註冊專業測量師(土地測量)根據測繪處作業備考編號1/2024自願向分區測 量處提交文件。

Authorized Land Surveyor [ALS]/ Registered Professional Surveyor (Land Surveying) [RPS(LS)] for making Voluntary Submission to District Survey Office in accordance with SMO Practice Note No. 1/2024.

# 1

認可人士根據屋宇署作業備考編號ADV-33及ADM-21向屋宇署提交,以供分區測量處備存土地測量圖則。

Authorized Person to Buildings Department for keeping Land Survey Plan in District Survey Office in accordance with BD's PNAP ADV-33 and PNAP ADM-21.

{甲部由註冊土地擁有人/認可人士/獲註冊土地擁有人授權的代表填寫;乙部則由認可土地 測量師或註冊專業測量師(土地測量)填寫}

{Part A to be completed by the registered land owner / Authorized Person /authorized representative of the registered land owner; and Part B to be completed by ALS or RPS(LS)}

甲部 - *註冊土地擁有人/認可人士/獲註冊土地擁有人授權代表的資料							
PART A - PARTICULARS OF * REGISTERED LANE	OWNER / AUTHORIZED PERSON /						
AUTHORIZED REPRESENTATIVE OF TH 1. 姓名 (*先生/太太/女士) Name (* Mr / Mrs / Ms) 認可人士註冊編號,如適用: Authorized Person Registration number, if applicable:	2. 地段編號 Lot Nos. Lot 573 RP & Lot 1710 in D.D.114						
3. (i)* 我/我們已聘用以下專業土地測量師	測量土地界線。						
* I / We have engaged the following professional	* I / We have engaged the following professional land surveyor to carry out the land boundary survey.						
<ul> <li>(ii) * 我/我們謹此准許测量處備存這次提交的文件,而且不論收費與否, 均可酌情決定把提交的圖則、測量報告及任何其他文件的副本分發給第三方。</li> <li>* I / We hereby give permission to District Survey Office/<u>Yuen Long</u> to keep this submission and to distribute copies of the plans, survey report and any other documents in this submission, with or without charges, to third parties at its discretion.</li> </ul>							
<ul> <li>(iii)* 我/我們明白提交文件屬自願性質,僅供備存於測量處作分享測量資料之用。一般而言,測量處只會審視夾附於相關土地或屋宇發展申請內的土地界線圖及測量記錄圖。故此,自願提交的文件不應被視為已經被政府審查、核准、批註或接納。</li> <li>* I/We understand that the submission is on a voluntary basis and is keeping in District Survey Office for sharing of survey information only. Generally, District Survey Office will only scrutinise the land boundary plan and survey record plan attached to the application for land or housing development. Therefore, the voluntary submission shall not be regarded as scrutinised, approved, endorsed or accepted by the Government.</li> </ul>							
簽署 Signature	日期 Date						

乙部· PART	- 測量師及提交 B - PARTICUI	交件的資料 ARS OF SURV	VEYOR AND S	UBMISSION			
1. 測 註 N	量師姓名 * 認 E冊專業測量師( <u>-</u> ame of Surveyor * A) WONG YIU	可土地測量的 土地測量) LS/RPS(LS) CHOJOS	₩ SEPH	2. 註冊編號 Registration No 057	Э.		
3. 關	於地段編號						的土地界
線 Is lan	測量結果,現向 ubmit a set of the fol nd boundary survey o	lowing documer f Lot no. 573 RP	測量處 its to District Su & 1710 in D.D.114	提交以下文件:- rvey Office/ <u>Yuen Long</u> :-	5	in res	pect of the
(i)	測量記錄圖,	編號		055 1005 11 /D	•	_ ( 頁);	及
	the survey record	plan, Number	SRP/YL/	057/06741/D	L	$(\frac{3}{2}$ shee	ets); and
(ii)	<ul> <li>) 土地界線或土 測量圖,編號</li> </ul>	_地				(頁);	及
	the land boundary	y or land survey	plan, Number	LSP/YL/057/	0674	$\frac{1}{D1}$ ( $\frac{2}{}$ sh	eets); and
(iii	) *由註冊土地 非由註冊土地	擁有人簽署」 擁有人或認	以委任獲授權 可人士填寫)	l代表的有效證明 。	文件(	頁) (倘	若甲部並
	*a valid documer representative (	ntary proof ( if Part A is not c	_ sheets) signed completed by the	by the registered land eregistered land	d owner r or Aut	for appointment o horized Person).	fauthorized
中,漫 測 備 Is bou	量工作已經完成 動測量報告( ubmit a survey repor undary survey which h the Code of Practic	(第475章 , 現就這次 <u></u> 頁)和證明 rt ( <u>4</u> sheet: have been prepa- se under the Lan	測量所界定 文件:- s) and supportin ared and perform d Survey Ordina	的土地界線提交的 ng documents on the ned by myself or undo nnce (Cap. 473) :-	我直我親 由我親 land bo er my d	自或在我直接 undary definition f	エ地介 み 監督下擬 for this land compliance
		頁 數 No. of sheet		頁 數 No. of sheet			頁數 No. of sheet
	土地查冊資料 Land Search data	<u>    10                                </u>	轉讓圖 Assignment	t Plan		地段索引圖 Lot Index Plan	
	土地界線圖 Land Boundary Plan		土地分割 Division Pla	圖 an		屋地圖則 House Lot Plan	: 
	測量記錄圖 Survey Record Plan	4	✓ 丈量約份 DD Sheet	· · · · · · · · · · · · · · · · · · ·	$\checkmark$	航空照片 Aerial photograph	3
	導線測量控制圖 Control Traverse Plan	:	✓ 大量約份 DD Control	控制圖 <u>1</u> Sheet		照 片 Photograph	
	永久土地記錄圖 Permanent Land Record Plan		"A"圖 "A"Sheet			地籍測量圖 Cadastral Survey Plan	
$\checkmark$	其他圖則 Other plans	2	其他 ✔ Others	8			

ş : 5

5.(i)	我特此准許测量處備存這次提交的文件,而且不論收費與否,均 可酌情決定把提交的圖則、測量報告及任何其他文件的副本分發給第三方。我亦明 白如本人曾就上述地段向測繪處提交圖則及/或測量報告,所有先前版本的圖則及/或 測量報告均會被本次提交的版本取代,而測繪處並不會分發任何被取代的版本給第 三方。
	I hereby give permission to District Survey Office/ to keep this submission and to distribute copies of the plans, survey report and any other documents in this submission, with or without charges, to third parties at its discretion. I also understand that in case there exists any previous version of plan(s) and/or survey report(s) submitted by me for the same captioned lot(s), all these previous
(ii)	versions will be superseded and will not be supplied to others for sharing purpose. 我明白提交文件屬自願性質,僅供備存於測量處作分享測量資料之用。一般而言, 測量處只會審視夾附於相關土地或屋宇發展申請內的土地界線圖及測量記錄圖。故 此,自願提交的文件不應被視為已經被政府審查、核准、批註或接納。
	I understand that the submission is on a voluntary basis and is keeping in District Survey Office for sharing of surve rally, District Survey Office will only scrutinise the land boundary plan and he application for land or housing development. Therefore, the voluntary subm
	日期 Date
*删去	不 適 戶 te

收集個人資料聲明:本表格所收集的個人資料僅供地政總署用於處理與本表格相關的事宜。提供個人資料純屬自願性質。若你未能提供所須個人資料,地 政總署可能無法處理本次申請。在未取得你的同意前,地政總署不會把你的個人資料向第三者披露或轉移。你有權要求查閱及更改你在本表格中提供的個 人資料。在香港法例第486章《個人資料(私隱)條例》第20條指定的情況下,地政總署或會拒絕查閱資料的要求。請透過以下方式向部門個人資料管制主 任(經辦人:高級土地測量師/技術資訊)提出查閱或更改個人資料的要求。電郵:smohq@landsd.gov.hk;地址:香港北角渣華道333號北角政府合署23樓地 政總署測繪處。

**Personal Information Collection Statement:** The personal data collected in this form will only be used by Lands Department for the purposes directly related to the handling of your submission made in this form. The provision of your personal data is voluntary. If you do not provide personal information, Lands Department may not be able to process your application. The personal data collected in this form will not be disclosed or transferred to any other third parties without your prior consent. You have the right to request access to and correction of your personal data provided in this form. Please note that we may refuse to comply with a data access request in the circumstances specified in section 20 of the Personal Data (Privacy) Ordinance (Cap. 486). Your requests should be made to the Departmental Personal Data Controlling Officer (Attn: Senior Land Surveyor/Technical Information). Email: smohq@landsd.gov.hk; Address: Survey and Mapping Office, Lands Department, 23/F North Point Government Offices, 333 Java Road, North Point, Hong Kong.

## Survey Report

## Boundary Survey of Lot 573 RP & Lot 1710 in D.D. 114, Kam Tin

### 1. Purpose

LAND MARKER (1980) H.K. CO. LTD. was appointed by the Land Owner to prepare a boundary plan for the subject lot – Lot 573 RP & Lot 1710 in D.D. 114, Kam Tin.

## 2. Root of Title

The history of the subject lot and registered area were extracted in the following table:

D.D.114	Lot	Lot 1710					
Registered Area	1.09 ac. ~4411	. sq.m. (about)	850 sq. ft. ~79.0 sq. m.(about)				
Held Under	Governn	Government lease					
Subdivided lot	Lot 573 S.A (Surrendered)	Lot 573 RP					
	850 sq. ft.	46630 sq. ft					
Registered Area	~79.0 sq. m. (about)	~4332 sq.m. (about)					
Hold Updat	YL1180						
Tiela Olider	On 22 <sup>nd</sup> July 1966						

### 3. List of Available Boundary Evidences

Documentary evidence of the subject lot and adjacent lots were searched and reviewed as follows:

- 1. 1:1000 Lot Index Plan
- 2. D.D. Field Sheet no. 114
- 3. D.D. Control Sheet no. 114
- 4. D.D. Retrace no. 114
- 5. Block Government Lease of D.D. 114
- 6. Field area statement of D.D. 114
- 7. Aerial photo no. H31/0027 taken in year 1924, 681/4\_4222 taken in year 1945 and orthophoto taken in 1963
- ALS's survey record plan no. SRP/YL/047/5762/D1, SRP/YL/053/4018/D1 & SRP/YL/002/3465/D2
- 9. New Grant no. YL1180
- 10. Searches from Land Registry
- 11. Station summary of the geodetic survey stations

## 4. Survey Origin

The geodetic survey stations OUSM8159.004, OUSM8159.005 and OUSM8159.006 were checked and adopted as the survey stations. New traverses were run and closed within the allowable misclosures.

## 5. Intention of Grant/Subdivision

Lot 573 was held under government lease. Later in July 1966, Lot 573 S.A was surrendered to the government and was regrant as Lot 1710 in the form of condition of exchange. Lot 1710 was granted under New Grant No.1180 with an area of 850 sq. ft. (79 sq. m.); Area of Lot 573 RP was calculated by simple deduction.

### 6. Evaluation of Boundary Evidence

### 6.1 Ground occupation

A site survey was conducted in January 2024. At the dates of the survey, the lots were occupied by temporary structures and surrounded by walls and fences. All surrounding features were surveyed for reference.

## 6.2 Boundary record searched from DSO

D.D. Field Sheet, Control Sheet, and Retrace Sheet were available in the District Survey Office. Furthermore, it was found that the subject lots were previously defined by other ALSs, i.e. Lot 573RP was defined and recorded in SRP/YL/047/5762/D1 and Lot 1710 was defined and recorded in SRP/YL/053/4018/D1. In addition, the adjoining lot was also defined by other ALS and record in SRP/YL/002/3465/D2.

### 6.3 Correlation exercise of parent lot

The subject lot boundaries were delineated from the D.D. control sheet and the area was consistent with the area as stated in the New Grant and was compatible to the deducted area. As a result, the correlated boundaries were adopted in the following boundary determination.

Then, the correlated boundaries were overlaid onto the aerial photos to define the location, size, and shape of the lot boundary. Field bunds were observed from the aerial photos taken in years 1924 and 1945, also, in orthophoto taken in 1963. It was found that the shape of the field bunds tallied with the correlated boundaries, and similar findings were recognized over the years.

Consistent field pattern was observed over years of aerial photos. Without noticeable alteration to the size and shape of the subject field parcels throughout the year, it is reasonable to believe the observed field bunds were the original occupation of the subject lots.

## 6.4 Checking and adoption of existing survey record of subject lots

The defined boundaries and area of the subject lots were found compatible to the correlation result and the search record. Therefore, they were adopted. 6.5 Checking of adjoining lot boundaries

Adjoining lot, which was previously defined by the ALS was checked. It was found consistent with the correlations result. Therefore, the common boundaries were adopted.

# 7. Conclusion

The previously defined boundaries of subject lots were checked and adopted. The surveyed areas of the subject lots are summarized as follows:

D.D.114	Lot 573 RP	Lot 1710		
Registered area (R)	46630 sq. ft.	850 sq. ft.		
	~4332 sq. m. (about)	~79.0 sq. m.(about)		
	(By deduction)			
Recorded in/	SRP/YL/047/5762/D1	New Grant no.1180 & record		
Held under		in SRP/YL/002/3465/D2		
Surveyed area (S)	4332 sq. m. (about)	790 sq. m. (about)		
Allowable tolerance#	451.1 sq. m. (about)	27.9 sq. m. (about)		
Discrepancy (S-R)	0 sq. m. (about)	0 sq. m. (about)		
Within tolerance	Yes	Yes		

#Remarks: Allowable discrepancy = +/- (20 sq. m. + 10% of register area) for 1:3960 D.D. Sheet

# 8. Enclosures

- 1. 1:1000 Lot Index Plan
- 2. D.D. Field Sheet no. 114
- 3. D.D. Control Sheet no. 114
- 4. D.D. Retrace no. 114
- 5. Block Government Lease of D.D. 114
- 6. Field area statement of D.D. 114
- 7. Aerial photo no. H31/0027 taken in year 1924, 681/4\_4222 taken in year 1945 and orthophoto taken in 1963
- ALS's survey record plan no. SRP/YL/047/5762/D1, SRP/YL/053/4018/D1 & SRP/YL/002/3465/D2
- 9. New Grant no. YL1180
- 10. Searches from Land Registry
- 11. Station summary of the geodetic survey stations

# 9. Authorized Land Surveyor's Certification

I, <u>Wong Yiu Cho Joseph</u>, an Authorized Land Surveyor registered under the Land Survey Ordinance (Cap. 473), hereby certify that this survey for the definition of <u>Lot 573 RP & Lot 1710</u> <u>in D.D. 114, Kam</u> Tin has been carried out by me, or under my direct supervision in conformity with the Code of Practice approved by the Land Survey Authority under the above Ordinance, and that this report correctly represents my work completed on the twenty-fifth of January, 2024.

Dated this fifteenth day of July, 2024.







SCALE 1:5000

	and the second se
Ref	Colour
В	BROWN
G	GREEN
1	INDIGO
0	ORANGE
Р	PINK
R	RED
V	VIOLET
Y	YELLOW
GY	GREY

I, and boundary plan has been prepared from land boundary surveys that were carried out by me or under my direct supervision in conformity with the Code of Practice approved by the Land Survey Authority under the above Ordinance, and that this plan correctly represents that survey completed on the <u>twenty-fifth</u> of <u>January</u> 20 <u>24</u>.

0

Dated this <u>fifteenth</u> day of <u>July</u> 20 <u>24</u>

FOR OFFICIAL USE

Land Survey Plan No. : LSP/YL/057/06741/D1

	LOT COORDI	NATES & D	IMENSIONS:				LC	DT COORDII	NATES & D	IMENSIONS:		
Boundar	y Point Bearing	Distance in metres	Northing	Easting	Bou	ndary Po	oint	Bearing	Distance in metres	Northing	Easting	
Boundar LOT 57: A B C D E F G H J K L M N P Q R S T U V W X Y Z A A B C D E F G H J K L M N P Q R S T U V W X Y Z A A B C D E F G H J K L M N P Q R S T U V W X Y Z A A B C D E F G H J K L M N P Q R S T U V W X Y Z A A B C D E F G H J K L M N P Q R S T U V W X Y Z A A B C D E F G H J K L M N P Q R S T U V W X Y Z A A B C D E F G H J K L M N P Q R S T U V W X Y Z A B C D E F G H J K L M N P Q R S T U V W X Y Z A B C D E F G H J K L M N P Q R S T U V W X Y Z A B C D E F G A H A Z A B C A A B C A A B C A A B C A A B C A A B C A A B C A A B C A A B C A A B C A A B C A A B C A A B C A A B C A B C A A B C A A B C A A B C A B C A A B C A A B C A A C A C	y Point         Bearing           3 RP IN D.D.114         52'34'42'           65'32'43'         77'11'56'           70'39'50'         20'00'33'           93'52'40'         93'52'40'           78'14'41'         82'23'19'           92'10'54'         91'09'53'           92'03'35'         92'03'35'           94'50'42'         148'35'02'           148'35'02'         148'35'02'           199'08'01'         300'26'23'           177'32'14'         125'06'43'           129'26'27'         150'07'42'           161'43'21'         172'06'26'           180'02'32'         191'57'50'           171'59'38'         274'58'29'           260'26'13'         350'26'25'           260'26'25'         260'26'25'           350'26'25'         260'26'25'           260'26'25'         260'26'25'           350'26'25'         260'26'25'           260'26'25'         260'26'25'           260'26'25'         260'26'25'           260'26'25'         260'26'25'           260'26'25'         260'26'25'           270'58'29'         294'57'41'           279'32'58'         282'08'12'           285'58'53' <td>in metres         9.388         9.290         9.722         12.748         4.317         8.680         13.699         10.154         10.613         14.464         5.982         11.816         7.213         5.187         11.251         9.309         15.647         6.269         5.694         2.647         5.221         2.720         5.397         7.952         19.847         6.266         4.161         2.151         7.273         5.341         7.273         5.341         7.273         5.341         7.273         5.341         7.273         5.341         7.273         5.341         7.273         5.341         7.273         5.341         7.273         5.924         4.623         13.705         8.298         0</td> <td>Northing 832935.736 832941.441 832945.287 832947.441 832955.718 832955.718 832955.718 832955.202 832958.863 832958.863 832958.354 832957.356 832958.354 832957.356 832957.356 832952.000 832942.700 832932.000 832933.700 832933.700 832934.009 832931.496 832933.409 832931.496 832912.170 832911.129 832912.170 832912.170 832911.129 832912.170 832912.170 832912.170 832915.232 832914.874 83292.046 83291.160 832913.988 832913.630 832905.957 832905.957 832905.957 832905.957 832905.957 832907.761 832908.604 832912.344 832912.344 832912.344 832912.344 832912.344 832912.344 832912.344 832912.344 832912.344 832912.344 832912.344 832912.344 83293.209 832935.736</td> <td>Easting 828686.286 828693.742 828702.199 828711.679 828723.708 828725.185 828733.845 828747.257 828757.322 828767.927 828767.927 828782.388 828788.366 828800.140 828803.900 828802.200 828792.500 828792.500 828792.900 828811.798 828812.628 828813.345 828813.345 828812.628 828813.345 828813.345 828813.345 828813.345 828813.345 828813.345 828813.345 828813.345 828813.345 828813.345 828813.345 828777.871 828786.690 828784.569 828784.569 828777.181 828777.871 828777.871 828777.871 828755.021 828749.878 828741.877 828755.021 828749.878 828741.877 828739.445 828741.877 828739.445 828741.877 828739.445 828741.877 828739.445 828741.877 828739.445 828741.877 828739.445 828741.877</td> <td>Bour AH AG AF AL AL AK AJ AH</td> <td>ndary Pc 1710 II</td> <td>bint N D.C</td> <td>Bearing 0.114 80°26'25' 170°26'25' 260°26'25' 350°26'25' 350°26'25' 350°26'25'</td> <td>in metres 5.341 7.273 2.151 4.161 9.644 4.161 2.151 7.273</td> <td><u>Northing</u> 832921.160 832922.046 832915.232 832913.630 832913.630 832913.988 832921.160</td> <td><u>Easting</u> 828778.094 828783.361 828784.569 828787.381 828777.181 828779.302 828778.094</td> <td></td>	in metres         9.388         9.290         9.722         12.748         4.317         8.680         13.699         10.154         10.613         14.464         5.982         11.816         7.213         5.187         11.251         9.309         15.647         6.269         5.694         2.647         5.221         2.720         5.397         7.952         19.847         6.266         4.161         2.151         7.273         5.341         7.273         5.341         7.273         5.341         7.273         5.341         7.273         5.341         7.273         5.341         7.273         5.341         7.273         5.341         7.273         5.924         4.623         13.705         8.298         0	Northing 832935.736 832941.441 832945.287 832947.441 832955.718 832955.718 832955.718 832955.202 832958.863 832958.863 832958.354 832957.356 832958.354 832957.356 832957.356 832952.000 832942.700 832932.000 832933.700 832933.700 832934.009 832931.496 832933.409 832931.496 832912.170 832911.129 832912.170 832912.170 832911.129 832912.170 832912.170 832912.170 832915.232 832914.874 83292.046 83291.160 832913.988 832913.630 832905.957 832905.957 832905.957 832905.957 832905.957 832907.761 832908.604 832912.344 832912.344 832912.344 832912.344 832912.344 832912.344 832912.344 832912.344 832912.344 832912.344 832912.344 832912.344 83293.209 832935.736	Easting 828686.286 828693.742 828702.199 828711.679 828723.708 828725.185 828733.845 828747.257 828757.322 828767.927 828767.927 828782.388 828788.366 828800.140 828803.900 828802.200 828792.500 828792.500 828792.900 828811.798 828812.628 828813.345 828813.345 828812.628 828813.345 828813.345 828813.345 828813.345 828813.345 828813.345 828813.345 828813.345 828813.345 828813.345 828813.345 828777.871 828786.690 828784.569 828784.569 828777.181 828777.871 828777.871 828777.871 828755.021 828749.878 828741.877 828755.021 828749.878 828741.877 828739.445 828741.877 828739.445 828741.877 828739.445 828741.877 828739.445 828741.877 828739.445 828741.877 828739.445 828741.877	Bour AH AG AF AL AL AK AJ AH	ndary Pc 1710 II	bint N D.C	Bearing 0.114 80°26'25' 170°26'25' 260°26'25' 350°26'25' 350°26'25' 350°26'25'	in metres 5.341 7.273 2.151 4.161 9.644 4.161 2.151 7.273	<u>Northing</u> 832921.160 832922.046 832915.232 832913.630 832913.630 832913.988 832921.160	<u>Easting</u> 828778.094 828783.361 828784.569 828787.381 828777.181 828779.302 828778.094	
Survey	y District	: YUE	N LONG									=
Date	of survey	: JAN	2024					PLAN	OFL	_01 57	3 RP	
Survey	/ Sheet No.	: 6-1	IE-15C								KAM	IIN

Survey Record Plan : SRP/YL/057/06741/D1

710 IN D.D.114, KAM TIN

(SHEET 2 OF 2)

I, an Authorized Land Surveyor registered dinance (Cap.473), hereby certify that the second surveys that were carried out by me or under my direct supervision in conformity with the Code of Practice approved by the Land Survey Authority under the above Ordinance, and that this plan correctly represents that survey completed on the twenty-fifth of January 20 24. Dated this <u>fifteenth</u> day of <u>Jub</u> 20

FOR OFFICIAL USE

Land Survey Plan No. : LSP/YL/057/06741/D1



```
Survey District: YUEN LONG
Date of survey: JAN 2024
Scale 1:400 Field Bk: 6741
Survey Sheet: 6-NE-15C
Reference SRP's: SRP/YL/002/3465/D2
```

# PLAN OF LOT 57

LOT 573 RP



under the Land Survey Ordinance (Cap.473), hereby certify that





832860 N 832860 N 832860 N	828720 E
Survey District: YUEN LONG Date of survey: JAN 2024 Scale 1:400 Field Bk: 6741 Survey Sheet: 6-NE-15C Reference SRP's: SRP/YL/002/3465/D2	PLAN OF LOT 57



SUBJECT Le Boundary F	OT COORDINATES & Point Bearing	c DIMENS Distance	IONS: Northing	Easting	TRAVER Station	SE	Bearing	Distance	Northing	Easting	RADIAT Survey	IONS ed point	Be
SUBJECT L	OT NO LOT 573	3 RP IN	D.D.114	-	OUSM8	159.005	5		833023.572	828413.817	FROM	NAIL13	
А			832935.736	828686.286	NAIL1		149°10'55"	98.134	832939.294	828464.092	TO	S1	245
В	52°34'42"	9.388	832941.441	828693.742	NAIL2		154 <b>°</b> 01'59"	85.089	832862.794	828501.347		S2	243
С	65°32′43″	9.290	832945.287	828702.199	NAIL3		42'03'19"	53.060	832902.190	828536.889		S5	244*2
D	// 11 5b 70°30'50"	9.722	832947.441	828/11.6/9	NAIL4		105*50'02"	21.079	832896.439	828557.168		S6	241
E F	20°00'33"	12.740	832951.002	828725 185	NAIL5	150.000	215'02'26	/3.901	832835.932	828514.736		59 S10	244 230°
G	93°52'40"	8.680	832955.131	828733.845	00200	159.006	320 42 30	102.920	032923.000	020401.290		S10	198"
Ĥ	78°14'41"	13.699	832957.922	828747.257	NAIL4				832896.439	828557.168		512	1002
J	82"23'19"	10.154	832959.267	828757.322	NAIL6		61°53'23"	45.717	832917.979	828597.493	FROM	NAIL12	
K	92°10'54"	10.613	832958.863	828767.927	NAIL7		67 <b>°</b> 09'32"	44.890	832935.405	828638.864	TO	S14	2"4
L	91°09'53"	14.464	832958.569	828782.388	NAIL8		249*40'24"	57.369	832915.477	828585.068		S15	16"(
М	92°03'35"	5.982	832958.354	828788.366	NAIL3		254 <b>°</b> 34'55"	49.979	832902.190	828536.889		S16	19"(
N	94°50′42″	11.816	832957.356	828800.140	NAIL 6				832917 979	828597 493		517 C18	213
P 0	140 30 02	7.21J	832931.200	828802 200	NAIL9		71°55'37"	41.583	832930.879	828637.025		S10	22 2
R	300°26'23"	11 251	832952 000	828792 500	NAIL10		108'32'37"	23.770	832923.320	828659.561		S20	25.4
S	177°32'14"	9.309	832942.700	828792.900	NAIL11		74 <b>°</b> 39'00"	44.754	832935.166	828702.719		S21	24°(
Т	125°06'43"	15.647	832933.700	828805.700	NAIL12		95°46'25"	41.207	832931.021	828743.718		S22	25°(
U	3°18'04"	6.269	832939.959	828806.061	NAIL13		324°32'47"	35.089	832959.604	828723.365		S23	26
V	129°26'27"	5.694	832936.342	828810.458	NAIL14		264 54 43	38.456	832956.193	828685.061		S24	29"3
W	150°07'42"	2.690	832934.009	828811.798	NAIL /		243 46 29	000.00	832933.403	828638.864		S25	38
X	161 43 21"	2.647	832931.496	828812.628	NAIL14				832956.193	828685.061		320 977	42 ( र/*/
ו 7	1/2 UO 20 120°02'32"	5.221 2.221	032920.324	02001J.J4J 828813 313	NAIL15	(PSM1)	62°36'32"	42.477	832975.734	828722.776		527 S28	.37*
∠ AA	100 02 32 191°57'50"	2.720	832018 321	828812 2240	NAIL16	(PSM2)	90°52'26"	47.400	832975.010	828770.170		S29	4.3*.
AB	171°59'38"	7.952	832910.449	828813.332	NAIL17		92'50'59"	40.062	832973.018	828810.182		S30	45"2
AC	274°58'29"	19.847	832912.170	828793.560	NAIL18		108'35'46"	27.952	832964.103	828836.674		S31	46"
AD	260°26'13"	6.266	832911.129	828787.381	NAIL19		1841/50″	/5.189	832889.124	828831.039		S32	53°.
AE	350°26'25"	4.161	832915.232	828786.690	NAILZU		JU/ UO 11 21°37'/5"	09.010 17 575	832941 260	020/03./33 828790 231		S33	53"(
AF	260°26'25"	2.151	832914.874	828784.569			21 37 40 257 <b>°</b> 34'30"	47 628	832931 021	828743 718		S54	141
AG	350°26'25"	7.273	832922.046	828783.361			20,0100	.,.020	55200 1.021	520, 10,710		500 656	1453
AH A.I	260 26 25	5.341 7 777	002921.160	020//0.094 828770 302	NAIL13				832959.604	828723.365		550 557	140 C 15 7°(
AK	1702023 260°26'25"	1.2/J 2 151	002910.900 832013 630	828777 181	NAIL22		129'35'52"	78.596	832909.510	828783.926		S58	152"
AL	170°26'25"	4.161	832909.527	828777.871	NAIL20		359"21"29"	15.423	832924.932	828783.753		S59	157
AM	256'01'53"	14.789	832905.957	828763.519	RADIATI	ONS						S60	177 (
AN	285*58'53"	6.552	832907.761	828757.220	Surveye	ed point	Bearing	Distance	Northing	Easting		S61	186*2
AP	290°58'29"	2.355	832908.604	828755.021	FROM	NAIL11	-		832935.166	828702.719		S62	192'4
AQ	294°57'41"	5.673	832910.998	828749.878	TO	S3	1'35'50"	7.802	832942.965	828702.937		S63	211
AR	279"32'58"	8.113	832912.344	828741.877		S4	343°57'21"	7.848	832942.709	828700.550	FROM	NAII 21	
AD AT	202 UN 12 286"57'56"	∠.4ŏŏ 2.047	002912.00/	020/JY.445 828736 626		S7	21°19'17"	12.180	832946.513	828707.148	TO	S34	352"
ALI	200 07 00 295'19'31"	∠.∀4/ १ ७००	002910./2/ 832014 886	020700.020 828734 177		S8	6'49'40"	10.598	832945.689	828703.979		S35	349"
AV	300°28'17"	19.976	832925.016	828716.960		511	35"12"21"	15.188	832947.422	828/11.691		S36	6"4
AW	304°23'35"	5.924	832928.362	828712.072		513	514/59 19391'22"	20./40 20.577	002901./05 832019.007	020/23./32 828727 200		S37	11° <i>°</i>
AX	292'39'36"	4.623	832930.143	828707.806		504 565	1202100 194°11'46"	∠9.000 25.615	002910.92/ 832020 770	020121.JOD 828723 QAR		S38	40'3
AY	282°55'37"	13.705	832933.209	828694.448		S66	124°24'31"	21.867	832922 810	828720 760		S39	47°4
AZ	282'00'16"	8.298	832934.935	828686.331		S67	125'39'01"	16.678	832925.446	828716.272		54U 541	6/2
А	356°47'04"	0.802	832935.736	828686.286		S68	127'44'50"	16.605	832925.001	828715.849		341 S12	92.7
						S69	123'35'12"	16.605	832925.981	828716.552		543	30 ( 100°
SUBJECT I	OT NO. – LOT 171		).114			S70	123°07'14"	15.282	832926.816	828715.519		S44	109"!
AH			832921.160	828778.094		S71	127'38'31"	15.276	832925.837	828714.816		S45	113'4
AG	80°26'25"	5.341	832922.046	828783.361		S72	125'42'54"	14.434	832926.741	828714.439		S46	116°(
AF	170°26'25"	7.273	832914.874	828784.569		5/3	150 05 24"	9.132	832929.285	828704 647		S89	353'4
AE	80°26'25"	2.151	832915.232	828786.690		3/4 875	104 12 Ub 227*20'53"	4.429 1 615	002901.1/9 839039 M99	020/U4.04/ 828600 205		S90	354*5
AD	1/0.26.25"	4.161	832911.129	828/8/.381		576	227 29 00 248°06'06"	6.518	832932.020	828696 671		S91	29°C
AL AK	200 20 20 350°26'25"	9.044 1 1 6 1	002909.52/ 832013 630	020///.0/1 828777 191		S77	256°36'53"	8.860	832933.115	828694.100		592	32"(
	JJU ZO ZO 80°26'25"	4.101 2 151	002910.000 832013 088	020///.101 828779 302		S78	257°22'25"	8.948	832933.210	828693.988	FROM	NAII 20	
AH	350°26'25"	7 273	832921 160	828778 094		S79	246'02'50"	9.966	832931.120	828693.612	TO	S47	<b>87°</b> 7
	000 20 20	1.210	002021.100	020770.00T		S80	254°49'34"	12.751	832931.829	828690.413		S48	92*0
						S81	263°31'06"	11.894	832933.824	828690.901		S49	97°5
						S82	263'29'44"	11.449	832933.869	828691.344		S50	103°1
						S83	266"16'48"	13.758	832934.274	828688.990		S51	112"(
						504 585	200 32 5/"	14.000	832934.324	020000./45 828688 356		S52	120°2
						300 S86	200 37 04 268°59'43"	14.009	0J2934.318 832934 RRN	020000.JOD 828686 372		503	209"2
						S87	268 <b>°</b> 58'23"	16.655	832934.868	828686.067			
						S88	270'08'17"	19.035	832935.212	828683.685			
					T	·····		-					
Survey	y District	:: YU	EN LO	NG									
)ate d	of survey	: JAI	N 202	4								T E	.7
	1.400	Fic	Id RIV	67/1					ГLA	IN UF	LU		)/、
					-								
survey	/ Sheet:	6-	NE-1										
	$\sim \sim CDD'$	S. CDF	ινι ινν	) / < / 65 / 177	1								

Distance	Northing	Easting
26.200 23.931 22.428 19.214 18.954 15.305	832959.604 832948.823 832949.079 832949.903 832950.481 832951.348 832951.836	828723.365 828699.486 828701.872 828703.144 828706.455 828706.304 828710.178
3.702	832956.090	828722.200
27.999 28.914 29.883 31.536 30.801 30.558 31 373	832931.021 832958.987 832958.802 832959.260 832960.264 832959.512 832959.196 832959.283	828743.718 828745.078 828751.732 828753.492 828755.523 828755.423 828755.546 828757.337
30.355 30.933 30.869 32.236 35.749	832958.746 832959.054 832958.635 832958.942 832959.083	828756.077 828756.794 828757.515 828759.828 828765 866
38.041 39.706 41.894 38.627 38.718	832959.083 832959.264 832963.892 832964.221 832958.977 832958.170	828769.202 828765.990 828769.269 828770.374 828771 323
41.784 45.639 46.493 32.091	832958.170 832959.577 832958.126 832958.906 832905.941	828774.222 828780.436 828780.920 828763.739
30.319 27.338 25.059 24.848 22.693	832906.496 832907.598 832908.681 832908.985 832909.997 832912.024	828751.544 828757.816 828755.070 828755.202 828752.260 828744 713
17.832 17.938 18.718	832913.302 832913.522 832915.066	828741.714 828739.769 828733.931
21.040 16.892 16.428 11.378 21.481 18.925 13.008 14.398 16.285 18.643 23.006 23.861	832941.269 832957.899 832957.582 832957.576 832953.989 832946.267 832940.419 832939.559 832937.968 832933.455 832931.672	828790.231 828787.538 828787.263 828792.172 828792.449 828804.214 828804.244 828802.241 828804.603 828806.426 828808.579 828811.869 828812.076
17.243 19.600 21.458 19.325	832958.410 832960.795 832960.024 832957.640 832924.932	828788.355 828788.521 828800.657 828800.500 828783.753



-\$



# 地段索引圖 LOT INDEX PLAN

摘要說明:本地段索引圖在其背景的地形圖上標示了各種永久和短期持有的土地 的圖像界線。這些土地包括私人地段、政府撥地、短期租約批地,以及其他作核 准用途的土地。請注意:(1)本索引圖上的資料會被不時更新而不作事先通知; (2)索引圖的更新或會延後於有關資料的實際變更;以及(3)本索引圖中顯示的界 線僅供識別之用,資料是否準確可靠,應徵詢專業土地測量師的意見。 免責說明:如因使用本地段索引圖,或因所依據的本索引圖資料出錯、遺漏、過 時或有誤差而引致任何損失或損害,政府概不承擔任何法律責任。

Explanatory notes : This plan shows the graphical boundaries of different kinds of permanent and temporary land holdings with the topographic map in the backdrop. The land holdings as shown may include private lots, government land allocations, short term tenancies and other permitted uses of land. It must be noted that: (1) the information shown on this plan is subject to update without prior notification; (2) there may be time lag between an update and the related changes taken place; and (3) the graphical boundaries as shown are for identification purpose only and interpretation of their accuracy and reliability requires the advice from professional land surveyor. Disclaimer: The Government shall not be responsible for any loss or damage howsoever arising from the use of this plan or in reliance upon its correctness, completeness, timeliness or accuracy.

# U

# 地政總署測繪處 Survey and Mapping Office Lands Department

香港特別行政區政府 一 版權所有 © Copyright reserved — Hong Kong SAR Government

> 比例尺 SCALE 1:1000 10 20 50 metres 30 40 metres 10

Locality : Lot Index Plan No. : ags\_S00000123290\_0001 District Survey Office : Land Information Centre Date :02-Feb-2024 Reference No. : 6-NE-15A,6-NE-15C

# D.D. FIELD SHEET



# D.D. CONTROL SHEET







1 11 11

# SCHEDULE OF CROWN LESSEES.

DISTRICT No. 114

Area of pat claw 5.07 recases

.

					Name of Owner.		Cro	ma Rent.							
I.	nt Nn.	Term of Iona in Şmrit,	Area	Duringtion of 1.05	Chinene ' Trustaliteration	Address	Clum	*	# Rpattr.						
Y	よよる	45	.10	andi	2 to wide in the	上村	15	Э	30						
1	55H		1.05	1	1/2 A the there Is		4	3. 1	15						
	155	4	.01	-	In the Sander Chi Tai		4	0	03						
	\$ 56	-	.05	-	the set that the se	n	-	1	15						
	ILY	1	ale	•	the to the this Tai	7	1	14	12-						
	5.58	-1	.12	1	A la to de the		4	3	36						
	2 Kg	4	05	7	the the staid in the	7	4	1	15						
	007	4	-v g,		in the second		-								
	560		1.41		A la to dai the Is	4	м	4.2	23 .						
	561	-	.56		\$ & R dai Chung Hay.	41	4	1.6	68						
	562	1	.14		御殿庭祖在下了了下	台山村	4	4	42						
		1		,	司理确像tay tak traitintes	12									
;	563	4	.14	4	the & A Lai Chury Hay.	上云村	1	4	42 <u>.</u>						
	564	ý.	.42		A le todai din tai		N.	1.2	26						
	565	47	-10	1	御鼎真祖 Gy Ly Y To	台山村.	4	9	30 .						
-					引建翰维大在, Cat Tinta	- "									
~	566	y	.01	5	御國英祖lay kat 7.5 to	永净里	. 7	0	03						
					are the tag you to truty			-							
	564	7	.30	-5	御臣郡祖在了了一人。下	台内村	4	9	<i>q b</i>						
					HE Star to King the King Structory	57 5-3		0	01						
	668	7	·LY	мĵ	帮加 法 ~ ~ ~	上初	1	a	a (						
	569	4	. 10	46	A to A The day Mar the the		Ħ	0	30						
	810	` ₽	./0	и	how wald . n. T. I.	2	4	C.	30						
4					(1) 332 the All Allow Many Car (turker)	<b>41</b>		2	20 WESER Lithe al all all all all all all all all all						
	571	4	10	1	BB TOT The Change Known Car	· /* 18	१ जन्मरी	3 3	22 Margue adai sing ( 10) " Mara) for mo - surgade - provident						
*	572	7	1-61	*1	BP B 3 TABlay Kind Kan The	水子坐	a, 4.	0.0	P.C.						
	le lu a				7) 32 The They for she ( ton ( )	~1		24	18						
3	573	*	1.04	"1	v de	P	ч	a /	10						
	EVL		.10		(et in a sivi name	r " 23	1.	6	60						
	217	ų	.00	м	13 13 13 13 14 5 1 - 2 · el et	44		-							
	245		. 11		25 St : in Chang 2: Huns	.,		2	22						
	576	14	.16	4	省并持冠石石上石	台山村	4	3	32.						
1	0,0	7	10	-1	司理御子前一 王王二	1	-		,						
	544		.1%	M 3	38 An is Chan 2: they	上村	1	. 2	52						
	578			· · · ·	御垂桂祖るいではたいて	台山村	y	10	04						
					月理時大福になるのない	51		1							
							••••	· • •							
. 5	7	VIouz	ah	Sm FIF	tbey 1 L D	0 	f Indi REA	a so	Dep PAT man	EI	tment. n E n T. Distric	t'	ho 114	<b>F</b>	age 10th
-------------------	-------------------	----------------------------	--	------------------------------	------------------------------	----------------------------------	--	-----------------------------	---------------------------	-------------------	--	---------------------	------------------------------------	--------------------------	--
No. on Plan.	Area Acres.	in Dec.	Record if otherwise than Cultivated.	No. on Plan	Area Acres,	in Dec.	Record if otherwise than Cultivated.	No. on Plan.	Area Acres.	in Dec.	Record if otherwise than Cultivated.	No. on Plan.	.Area i Acres.	n Dec.	Record if otherwise than Cultivated,
a1 2		16	zed	(652 (652	20'K 17' 10'X 34'	340 5	1. L.O. 9 m 1/22 1. 8 m 1/22	1701 2 3	8,000	·17 . 17.	LNT 11 6/4/27/59 LNT-298/4/27/60 LNT 109/4/27/59	751 2 3	700 700 200	.02 .02 .01	LNT. 3. 7 14PA/61 2NT 353/ 1PA/61 WT. 110/107/67
3 4 5 6	2	09 04 10	tallow ;	1654 1655 1656	30 x 32	992 . 22a	= 205 - 1/25 = 5/1.0- 2 g in 1/25 = p 5. LO 44 for 1/23	4 5 6	5,9,00 64,440 5,000	-14	LNT 141/427/59 105 66/427/61	4 5 6	700 116,400 700	.02 2.68 .02	1007 3019407/67 1007 344/4L7/65 1007 1344/4P7/60
7 8 9		91 · 05 21	basti	1657 1658 1659		9054 100" 100"	" " I quinifat	7 8 9 <i>F</i> 7/0	20,000 9,100 510	·46 ·21 ·02	LAT 200/4-1/61 2NT 300/4-1/59 LAT 21/4-1/63 LAT 21/4-1/63	9 9 1760	700 3870 600 495	.02 .09 .02 .02	107 75/107/60 107. 7/807 /60 1007 201/407/60
70 1 2 3	. {.	1 01 1 01 5 28 12	2 mi Flerr Anllogo 2 mi - Jallow	1661 . 1662 1663		3415; 310 * 432 ** 182-	" " 265 in 1/26 " " 265 in 1/26 " " 15 in 1/19. " " 18 in 1/19.	1 2 3	1,000 700 700	·03 ·02 ·02	4077887/60 2×7 48888980	1 2 3	495 700 9440	.02 .02 :22	107308/19/100 107331/4/17/60 101331/4/17/60 1017 4826
4 5 6	500 ( 60 686 1	12		1664 1665 1666 1667	126×10' 20×10' 150×17'	338 400 850	" " 96 - 1/28. " " 15 - 1/20. 	4 5 6 7	510 640 700 700	·02 ·02 ·02	WT 491/YPA/0 WT 493/YPA/0 WT 3/YPA/0 WT 3/YPA/0		2.18/ 100 700 st 320	.06 .02 .02 .01	1NT 8/1177/6 1NT 480/1177/6 1NT 480/1177/6 LWT 307/10/2
7 8 9			Lo. 63/09 Lo. 22/10	1669 1659	30×25	750.	415 - 1/31. 	8 1719 1720	600 700 700	· 02	2 LAT 96/ YLTC int. 85/ YPA/6 LNT. 120/ YPA/6	54 S 1 9 1770	700 37.92M 408 SF 65.03H2	-02 -01	- 2NT 303/177/61 LNT 355/ YPT/61 F LNT 302/ YPT/61

教派特別行政証政所 活躍時近 Corenabi Reserved Hary Kong SAR Government

No,		Mo	nzab		FII	GLD P	AREA	ST	TA	IG M	IG N T. Distric	t	14		
Plan.	Area	la	Record if	Plan.	Are	a in	Record if otherwise film	0. Plan.	Are	i în	Record if otherwise than	te Plan.	Are	n lu	Becord otherwise
Na. ca	Acres.	Deat	Cultavated.	No. of	Aarca,	Dag.	Cultivaled.	Mo. 0	Aures.	Dec.	Caltivated.	No. c	Acres.	Dea.	Culliva
401 3 8 8 8 9 410 12 8 8 9 410 12 8 8 9 410 12 8 8 9 410 12 8 8 9 9 12 8 8 9 9 12 8 8 9 9 12 8 8 9 9 12 8 8 9 12 8 8 9 12 8 12 12 12 12 12 12 12 12 12 12		110200220126101000001078044400084300111202381111		4.51 3 4 5 6 7 8 9 4 7 8 9 4 7 8 9 4 7 8 9 4 7 8 9 4 7 8 9 4 7 8 9 4 7 8 9 4 4 7 8 9 4 4 6 7 8 9 4 4 6 7 8 9 4 4 6 7 8 9 4 4 6 7 8 9 4 4 6 7 8 9 4 4 6 7 8 9 4 4 6 7 8 9 4 4 6 7 8 9 4 4 6 7 8 9 1 1 1 1 1 1 1 1 1 1 1 1 1		0332322120021212100300102001075030316005412101000		521 3 3 4 5 5 1 9 3 4 5 7 1 9 5 7 1 9 5 7 1 9 5 7 1 9 5 7 1 9 5 5 7 1 9 5 7 1 9 5 7 1 9 5 7 1 9 5 7 1 9 5 7 1 9 5 7 1 9 5 7 1 9 5 7 1 9 5 7 1 9 5 7 1 9 5 7 1 9 5 7 1 9 5 7 5 7 1 9 5 7 5 7 5 7 5 7 5 7 5 7 5 7 5 7 5 7 5 7 5 7 5 7 5 7 5 7 5 7 5 7 5 7 7 7 7 7 7 7 7 7 7 7 7 7		2363537315885232572075982762885884378115866756	2" " " " " " " " " " " " " " " " " " "	551 284 0 07 8 0 01 2 3 4 0 07 8 0 0 1 2 3 4 0 07 8 0 0 1 2 3 4 0 07 8 0 0 1 2 3 4 0 07 8 0 0 1 2 3 4 0 07 8 0 0 1 2 3 4 0 07 8 0 0 1 2 3 4 0 0 7 8 0 0 7 8 0 0 1 2 3 4 0 0 7 8 0 0 1 2 3 4 0 0 7 8 0 0 7 8 0 0 1 2 3 4 0 0 7 8 0 0 7 8 0 0 1 2 3 4 0 0 7 8 0 0 0 7 8 0 0 7 8 0 0	1	23051542516442010700019016628870744003744352686584	2. Jalle Jall
		1.50		-	1		1-204-02		1.2	ila.	2	1	120	201	1

## Aerial photo no. H31/0027 taken in year 1924



## Aerial photo no. 681/4\_4222 taken in year 1945



# Orthophoto taken in 1963





		and the second se
Notes:		_
SUBJECT LOT COORDINATES & DIMENSIO Boundary Point Bearing Distance	Northing	Easting
SUBJECT LOT NO LOT 573 RP IN D.	D.114 832935.736	828686.286
B 52°34′42″ 9.388 C 65°32′43″ 9.290	832941.441 832945.287	828693.742 828702.199
E 70°39′50″ 12.748	832951.662	828723.708
G 93°52′40″ 8.680	832955.131	828733.845
J 82°23′19″ 10.154	832959.267	828757.322
L 91°09′53″ 14.464 M 92°03′35″ 5.982	832958.569	828782.388
N 94°50′42″ 11.816 P 148°35′02″ 7.213	832957.356	828800,140 828803,900
Q 199°08'01" 5.187 R 300°26'23" 11.251	832946.300 832952.000	828802.200 828792.500
S 177°32′14″ 9.309 T 125°06′43″ 15.647	832942.700 832933.700	828792.900 828805.700
U 3°18'04" 6.269 V 129°26'27" 5.694	832939.959 832936.342	828806.061 828810.458
W 150°07'42" 2.690 X 161°43'21" 2.647	832934.009 832931.496	828811.798 828812.628
Y 172°06'26" 5.221 Z 180°02'32" 2.720	832926.324 832923.604	828813.345 828813.343
AA 191°57′50″ 5.397 AB 171°59′38″ 7.952	832918.324 832910.449	828812.224 828813.332
AC 274°58′29″ 19.847 AD 260°26′13″ 6.266	832912.170 832911.129	828793.560 828787.381
AE 350°26'50″ 4.161 AF 260°26'50″ 2.151	832915.232 832914.874	828786.690
AG 550°26'50° 7.275 AH 260°26'50″ 5.341	832922.046	828778.094
AK 260°26′50″ 2.151 AK 170°26′50″ 4.161	832913.630	828777.181
AM 256°01'53" 14.789 AN 285°58'53" 6.552	832905.957	828763.519
AP 290°58'29" 2.355 AQ 294°57'41" 5.673	832908.604 832910.998	828755.021 828749.878
AR 279°32′58″ 8.113 AS 282°08′12″ 2.488	832912.344 832912.867	828741.877 828739.445
AT 286°57′56″ 2.947 AU 295°19′34″ 2.709	832913.727 832914.886	828736.626 828734.177
AV 300°28'17" 19.976 AW 304°23'35" 5.924	832925.016 832928.362	828716.960 828712.072
AX 292°39′36″ 4.623 AY 282°55′37″ 13.705	832930.143 832933.209	828707.806 828694.448
AZ 282-00 16 8.298 A 356°47′04″ 0.802	832935.736	828686.286
		1.461
I, an Authorized L	and Surveyor	registered
that this survey record plan has be boundary surveys that were carried	en prepared	from land
direct supervision in conformity wi	th the Code	of Practice
Ordinance, and that this plan corresurvey completed on the18thday of	ofMay	nts that
Dated this26thday ofMay		
ERIC TANG & A	SSOCIATE	S LTD.
◎ 氏測量師行	有限公司	1
FOR OFFICIAL USE		
Survey Record Plan No.: SRP/YL	/047/5	762/D1





S:	
Star -	
	· · · · · · · · · · · · · · · · · · ·
TIIC	<u> </u>
lable of	Subdivisions
SECTION	AREA
_OT 574 S. A	701.7 sg.m (About)
_OT 574 S.B	305.9 sq.m (About)
_01 5/4 RP	225.1 sq.m (About)
otal Area	1232.7 sq.m (About)
	. an Authorized Land Surveyor
under the Land Sur	vey Ordinance ( Cap. 473 ), hereby
at this survey record	plan has been prepared from land
surveys that were ca	rried out by me, or under my direct
n, in conformity w	ith the Code of Practice approved
Land Survey Autho	prity under the above Ordinance,
t this plan corrections and a corrections of August 201	y represents that survey completed
a day of August 201	/.
s 16th day of Augus	t 2018.
	_
220	
工測量有限	公司
工 <b>測量有限</b> N MA SURVEYIN	公司 IG LTD. 12 凱 2018
工 <b>測量有限</b> N MA SURVEYIN	公司 IG LTD. 12.肌瘤
工測量有限 N MA SURVEYIN FICIAL IUSE FORMAN	公司 IG LTD. 追題證 Survey Authority due to

Survey Record Plan No. : SRP/YL/002/3465/D2(Sheet 1 of 2)

						Notes:
		Boundary Point LOT 574 RP IN D.	Bearing D. 114	Distance N	Ε	SUBJECT LOT COORDINATES & DIMENSIONS: Boundary Point Bearing Distance N E
		L M AR AS AT AU AV Y Z AA AB AQ L	125° 40' 05' 156° 14' 41' 106° 20' 43' 174° 06' 07' 149° 12' 18' 208° 23' 33' 305° 40' 31' 326° 08' 50' 314° 24' 09' 128° 53' 37' 55° 30' 30'	832899.7           9.000         832894.4           10.538         832884.8           2.669         832884.9           4.288         832878.6           0.407         832878.3           5.922         832873.1           12.069         832880.15           6.818         832890.15           6.818         832890.6           7.128         832896.20           6.175         832899.70	00         828827, 300           52         828834, 612           7         828838, 857           56         828841, 418           828841, 418         828847, 612           70         828848, 053           21         828848, 261           828845, 445         828845, 445           2828831, 843         828831, 843           9         828816, 663           03         828822, 211           00         828827, 300	LUT 574 S. A IN D. D. 114 A B 120° 26′ 23′ 11. 251 832946. 300 828802. 200 C D 126° 59′ 44′ 13. 908 832942. 831 828815. 008 E 131° 52′ 32′ 1. 885 832941. 572 828816. 412 F 5 139° 30′ 34′ 5. 026 832931. 249 828819. 676 G 151° 10′ 14′ 7. 420 832931. 249 828823. 254 H 164° 31′ 55′ 10. 811 832920. 829 828825. 329 J 1 83° 13′ 10′ 9. 787 832911. 057 828825. 380 K 178° 16° 46′ 10′ 4. 554 832899. 700 828827. 300 L 1 60° 46′ 10′ 4. 554 832899. 700 828827. 300 AQ 2 35° 30′ 30′ 6. 175 832896. 203 828822. 211 AB 3 00° 53′ 37′ 7. 128 832900. 679 828816. 663
		TRAVERSE Station UNail6/* Nail1 Nail2 UNail5/* Nail6/* Nail3	Bearing 290° 45' 27' 124° 42' 40' 141° 08' 32' 345° 22' 48'	Distance Northing 832892.95 13.421 832897.7 17.287 832887.86 19.400 832872.76 832892.95 26.120 832918.22	Easting 5828829.737 2828817.187 888831.398 2828843.568 5828829.737 29828829.737	AC       28/*41'22'       2.008       832901.289       82814.751         AD       338*47'37"       4.557       832905.537       828813.102         AE       02*40'30"       4.918       832910.449       828813.332         AF       351*59'38"       7.952       832916.324       828813.343         AG       11*57'50"       5.397       832926.324       828813.343         AH       00°02'32"       2.720       832931.496       828812.628         AJ       352*06'26"       5.221       832934.009       828812.628         AK       341*43'21"       2.647       832934.009       828811.798         AL       330*07'42"       2.690       832939.959       828810.458         AM       309*26'27"       5.694       832939.959       828806.061
		Na1(4 DNa1(5/*	164° 26' 02" 147° 49' 13"	24, 241 832894, 8 26, 130 832872, 76	8 828829, 651 62 828843, 568	AP         305*06' 43"         15, 647         832942, 700         828792, 900           A         357*32' 14"         9, 309         832952, 000         828792, 500
		□Nail4/₩ □Nail10/₩ □Nail11/₩	122° 26′ 51″ 103° 46′ 07″	832887, 69 11, 169 832881, 69 14, 148 832878, 33	91 828836,722 99 828846,147 82 828859,889	LOT 574 S. B IN D. D. 114 M 832894, 452 828834, 612 N 125* 40( 05* 2, 250, 822893, 140, 828834, 612
		PSM RADIATIONS: Surveyed Point PSM1(ONail5/*, N PSM2(Nail1, Nail	all on ground) on ground)	North In 832872, 7 832897, 7	e Easting 2 828843.568 2 828817.187	N         123         40         2         230         323         14         620         633         64           P         105*58'08"         20,137         832887.600         828855.800         828855.800         828855.800         828857.300         828867.300         828867.300         828867.300         828867.300         828867.300         828863.300         828866.142         832875.884         828866.142         7         832875.51'36"         828865.591         926         832875.288         828863.590         828863.590         828863.590         828865.591         828865.
		RADIATIONS: Surveyed Point Fr. ONail6/* To 8 13 17	Bearing 287° 48' 15' 347° 05' 20' 163° 34' 34'	Distance Northing 832892, 95 12, 904 832896, 90 18, 572 832911, 05 7, 446 832885, 81	Easting 5 828829,737 11 828817,451 57 828825,588 3 828831,843	U         270* 07' 49"         3. 340         832875. 296         828860. 251           V         258* 04' 40"         4. 859         832874. 292         828855. 497           W         266* 18' 05"         6. 228         832876. 040         828849. 520           X         229* 06' 01"         4. 889         832872. 839         828845. 824           Y         305* 40' 31"         0. 467         832873. 112         828845. 445           AV         28* 23' 33"         5. 922         832878. 321         828848. 261           AU         329* 12' 18"         0. 407         832878. 670         828848. 053
		Fr. Nail1 To 26 27 30 33 35 35 36	320° 56′ 46″ 332° 26′ 00″ 325° 44′ 07″ 349° 59′ 02″ 106° 42′ 50″ 119° 56′ 31″	832897, 7 11, 443 832906, 5 8, 827 832905, 5 4, 328 832901, 28 3, 013 832900, 6 5, 245 832896, 20 15, 410 832890, 02	2         828817.187           8         828809.978           87         828813.102           89         828814.751           98         828814.663           828822.211         828822.211           828830.541         828830.541	AT         354*06'07"         4.288         832882.936         828847.612           AS         280*14'52"         6.295         832884.056         828841.418           AR         286*20'43"         2.669         832884.807         828838.857           M         336*14'41"         10.538         832894.452         828834.612
		Fr. Nail3 To 62 64 65 66 67 82 83 83 84 85 86 87 88 88 89	49° 00' 42' 00° 28' 40' 349° 55' 17' 343° 54' 35' 231° 35' 35' 270° 29' 49' 298° 44' 07' 309° 33' 16' 321° 35' 35' 324° 16' 45' 324° 59' 22' 321° 49' 28'	832918, 22 3, 964 832920, 82 13, 020 832931, 24 19, 826 832937, 74 24, 294 832941, 57 25, 912 832942, 83 12, 524 832910, 44 10, 922 832918, 32 11, 179 832923, 60 12, 711 832918, 32 16, 930 832931, 49 19, 436 832934, 00 22, 114 832936, 34 27, 642 832939, 95	9         828823.146           9         828826.137           9         828823.254           9         828819.676           2         828815.008           9         828813.332           4         828813.343           4         828813.343           4         828813.345           6         828812.628           9         828811.798           828810.458         828810.458           9         828810.458	I, an Authorized Land Surveyor registered under the Land Survey Ordinance ( Cap. 473 ), hereby certify that this survey record plan has been prepared from land boundary surveys that were carried out by me, or under my direct
		Fr. ONail5/* To 42 43 45 47	37° 11′ 53″ 40° 10′ 06″ 101° 09′ 05″ 312° 59′ 01″	832872.76 7.417 832878.67 7.275 832878.32 1.555 832872.46 10.836 832880.15	2 828843.568 0 828848.053 1 828848.261 1 828845.094 0 828835.641	supervision, in conformity with the Code of Practice approved by the Land Survey Authority under the above Ordinance, and that this plan correctly represents that survey completed on the 1st day of August 2017.
		Fr. DNa110/* To 94 96 98 106 107	114° 25' 04" 110° 10' 44" 106° 12' 54" 61° 10' 27" 58° 15' 10"	832881, 69 15, 489 832875, 29 18, 583 832875, 29 20, 823 832875, 88 14, 878 832888, 87 11, 983 832888, 00	9         828846.147           6         828860.251           8         828863.590           4         828866.142           2         828859.181           4         828856.337	Dated this 16th day of August 2018.
		/* represent to	SRP/YL/002/114	4/581-D		馬容江測量有限公司 HELICON MA SURVEYING LTD. 12 JUL 2018
Survey District: Yuen Long Date of Survey: August 2017 Scale 1: 400 Field Bk: YL114_574	PLAN OF LOT N	NOS. 574 S.A TO S.	B AND	RP IN D	.D.114	FOR OFFICIAL USE Deposited with the Land Survey Authonity due to section 30(4) of the Land Survey Ordinance (Cap.473) Legislation Section, SMO, Lands Department
Reference SRP's: SRP/YL/002/114/574-D SRP/YL/002/114/581-D	BEING SU	JEDIVISION OF LOT	5/4 IN	D.D.114		Survey Record Plan No. : SRP/YL/002/3465/D2(Sheet 2 of 2)



#### PARTICULARS AND CONDITIONS OF EXCHANGE

L.N.T. No. 101/YLT/63

PARTICULARS AND CONDITIONS for the GRANT of the Lot described in the First Schedule hereto and shown coloured red on the attached planIfor the residue of a term of 99 years less three days commencing from the lst day of July, 1898, at the rent specified in such Schedule, in exchange for the surrender of the OLD LOT<del>OR LOTE</del> described in the Second Schedule and shown edged coloured blue on plan II.

#### First Schedule

7	Lot No.	Location	Site	Aréa in sq.ft.	Annual Rental	Prenium
114	1710	Nga Yiu Hau, Fat Hsung	As delineated and coloured Red on Plan <u></u> annexed hereto	850 (approx)	\$ 10.c0	Gross Premium \$2,975.00 Less value of lot stated in 2nd Schedule \$425.00 Not Fremium \$2,550.00

#### PARTICULARS OF THE LOT

Second Schedule OLD LOT OP LOTS TO HE SURRENDERED

D.D. No.	Lot No.	Area in sq. ft.	Annua l Renta l	Remarks	
114	573 S.A.	850 sq.ft.	-	850 sq.ft. of agricultura land valued \$425.00	<u>1</u> 

1/1/YLLF.





印製於 PRINTED AT: INTERNET SEARCH (DOWNLOAD) 查冊日期及時間 SEARCH DATE AND TIME: 12/07/2024 17:27 查冊者姓名/名稱 NAME OF SEARCHER: 查冊種類 SEARCH TYPE: HISTORICAL AND CURRENT

本登記冊列明有關物業截至12/07/2024 07:30 之資料 THE INFORMATION SET OUT BELOW CONTAINS PARTICULARS OF THE PROPERTY UP TO 07:30 ON 12/07/2024.

備存土地紀錄以供市民查閱旨在防止秘密及有欺詐成分的物業轉易,以及提供容易追溯和確定土地財產及不動產業權的方法。土地紀錄內載的資料不得用於與土地紀錄的宗旨 無關之目的,使用所提供的資料須符合《個人資料(私隱)條例》的規定。

The land records are kept and made available to members of the public to prevent secret and fraudulent conveyances, and to provide means whereby the title to real and immovable property may be easily traced and ascertained. The information contained in the land records shall not be used for purposes that are not related to the purposes of the land records. The use of information provided is subject to the provisions in the Personal Data (Privacy) Ordinance.

《政府租契續期條例》已經生效、土地登記冊的適用政府租契年期資料不會相應更新。適用政府租契的年期屆滿日期,以地政總署根據該條例刊憲的任何相關「續期公告」為準。

The Extension of Government Leases Ordinance is in force. Information on lease term for an applicable lease is not updated in the land register accordingly. For the expiry date of the lease term of an applicable Government lease, please refer to the relevant "Extension Notice" published by the Lands Department which should prevail.

進行任何交易前,應先向土地註冊處查閱最新的土地紀錄。

BEFORE ANY DEALINGS, UP-TO-DATE LAND SEARCH SHOULD BE CONDUCTED WITH THE LAND REGISTRY.

#### 物業資料 PROPERTY PARTICULARS

物業参考編號 PROPERTY REFERENCE NUMBER (PRN): C2640801

地段編號

LOT NO.: LOT NO. 573 IN D.D. 114

批約 HELD UNDER: GOVERNMENT LEASE 年期 LEASE TERM: 75 YEARS RENEWABLE FOR 24 YEARS 開始日期 COMMENCEMENT OF LEASE TERM: 01/07/1898 每年地税 RENT PER ANNUM: \$2.18

物業参考編號 PRN: C2640801 (12/07/2024) 第 1 頁,共 4 頁 PAGE 1 OF 4

土地註冊處THE LAND REGISTRY 土地登記冊LAND REGISTER

印製編號 PRINT CONTROL: ESN240712016311

物業參考編號 PRN: C2640801 (12/07/2024) 第 2 頁,共 4 頁 PAGE 2 OF 4

土地註冊處THE LAND REGISTRY 土地登記冊LAND REGISTER

印製編號 PRINT CONTROL: ESN240712016311

#### 業主資料

OWNER PARTICULARS

業主姓名 NAME OF OWNER	身分 (如非唯一擁有人) CAPACITY (IF NOT SOLE OWNER)	註冊摘要編號 MEMORIAL NO.	文書日期 DATE OF INSTRUMENT	註冊日期 DATE OF REGISTRATION	代價 CONSIDERATION	-

土地註冊處THE LAND REGISTRY 土地登記冊LAND REGISTER 印製編號 PRINT CONTROL: ESN240712016311

物業参考編號 PRN: C2640801 (12/07/2024) 第 4 頁,共 4 頁 PAGE 4 OF 4

#### 土地註冊處THE LAND REGISTRY 土地登記冊LAND REGISTER

印製於 PRINTED AT: INTERNET SEARCH (DOWNLOAD) 查冊日期及時間 SEARCH DATE AND TIME: 12/07/2024 17:27 查冊者姓名/名稱 NAME OF SEARCHER 查冊種類 SEARCH TYPE: HISTORICAL AND CURRENT

本登記冊列明有關物業截至12/07/2024 07:30 之資料 THE INFORMATION SET OUT BELOW CONTAINS PARTICULARS OF THE PROPERTY UP TO 07:30 ON 12/07/2024.

備存土地紀錄以供市民查閱旨在防止秘密及有欺詐成分的物業轉易,以及提供容易追溯和確定土地財產及不動產業權的方法。土地紀錄內載的資料不得用於與土地紀錄的宗旨 無關之目的,使用所提供的資料須符合《個人資料(私隱)條例》的規定。

The land records are kept and made available to members of the public to prevent secret and fraudulent conveyances, and to provide means whereby the title to real and immovable property may be easily traced and ascertained. The information contained in the land records shall not be used for purposes that are not related to the purposes of the land records. The use of information provided is subject to the provisions in the Personal Data (Privacy) Ordinance.

《政府租契續期條例》已經生效,土地登記冊的適用政府租契年期資料不會相應更新。適用政府租契的年期屆滿日期,以地政總署根據該條例刊憲的任何相關「續期公告」為 準。

The Extension of Government Leases Ordinance is in force. Information on lease term for an applicable lease is not updated in the land register accordingly. For the expiry date of the lease term of an applicable Government lease, please refer to the relevant "Extension Notice" published by the Lands Department which should prevail.

進行任何交易前,應先向土地註冊處查閱最新的土地紀錄。

BEFORE ANY DEALINGS, UP-TO-DATE LAND SEARCH SHOULD BE CONDUCTED WITH THE LAND REGISTRY.

物業資料 PROPERTY PARTICULARS

物業参考編號 PROPERTY REFERENCE NUMBER (PRN): C6258306

地段編號

LOT NO.: SECTION A OF LOT NO. 573 IN D.D. 114

批約 HELD UNDER: GOVERNMENT LEASE 年期 LEASE TERM: 75 YEARS RENEWABLE FOR 24 YEARS 開始日期 COMMENCEMENT OF LEASE TERM: 01/07/1898 每年地稅 RENT PER ANNUM: -

印製編號 PRINT CONTROL: ESN240712016313

物業參考編號 PRN: C6258306 (12/07/2024) 第 1 頁,共 3 頁 PAGE 1 OF 3

土地註冊處THE LAND REGISTRY 土地登記冊LAND REGISTER



土地註冊處THE LAND REGISTRY 土地登記冊LAND REGISTER 印製編號 PRINT CONTROL: ESN240712016313 等待註冊的契約 DEEDS PENDING REGISTRATION

1 物業參考編號 PRN: C6258306 (12/07/2024) 第 3 頁,共 3 頁 PAGE 3 OF 3

4

印製於 PRINTED AT: INTERNET SEARCH (DOWNLOAD) 查冊日期及時間 SEARCH DATE AND TIME: 12/07/2024 17:27 查冊者姓名/名稱 NAME OF SEARCHEI 查冊種類 SEARCH TYPE: HISTORICAL AND CURRENT

本登記冊列明有關物業截至 12/07/2024 07:30 之資料 THE INFORMATION SET OUT BELOW CONTAINS PARTICULARS OF THE PROPERTY UP TO 07:30 ON 12/07/2024.

備存土地紀錄以供市民查閱旨在防止秘密及有欺詐成分的物業轉易,以及提供容易追溯和確定土地財產及不動產業權的方法。土地紀錄內載的資料不得用於與土地紀錄的宗旨 無關之目的,使用所提供的資料須符合《個人資料(私隱)條例》的規定。

The land records are kept and made available to members of the public to prevent secret and fraudulent conveyances, and to provide means whereby the title to real and immovable property may be easily traced and ascertained. The information contained in the land records shall not be used for purposes that are not related to the purposes of the land records. The use of information provided is subject to the provisions in the Personal Data (Privacy) Ordinance.

《政府租契續期條例》已經生效,土地登記冊的適用政府租契年期資料不會相應更新。適用政府租契的年期屆滿日期,以地政總署根據該條例刊憲的任何相關「續期公告」為 準。

The Extension of Government Leases Ordinance is in force. Information on lease term for an applicable lease is not updated in the land register accordingly. For the expiry date of the lease term of an applicable Government lease, please refer to the relevant "Extension Notice" published by the Lands Department which should prevail.

進行任何交易前,應先向土地註冊處查閱最新的土地紀錄。

BEFORE ANY DEALINGS, UP-TO-DATE LAND SEARCH SHOULD BE CONDUCTED WITH THE LAND REGISTRY.

物業資料 PROPERTY PARTICULARS

物業参考編號 PROPERTY REFERENCE NUMBER (PRN): C2640810

地段編號

LOT NO.: THE REMAINING PORTION OF LOT NO. 573 IN D.D. 114

批約 HELD UNDER: GOVERNMENT LEASE 年期 LEASE TERM: 75 YEARS RENEWABLE FOR 24 YEARS 開始日期 COMMENCEMENT OF LEASE TERM: 01/07/1898 每年地税 RENT PER ANNUM: -

物業參考編號 PRN: C2640810 (12/07/2024) 第 1 頁,共 6 頁 PAGE 1 OF 6

土地註冊處THE LAND REGISTRY



NAME OF OWNER (IF NOT SOLE OWNER) INSTRUMENT REGISTRATION MEMORIAL NO. CONSIDERATION

身分 (如非唯一擁有人) 文書日期 註冊日期 業主姓名 CAPACITY 註冊摘要編號 DATE OF DATE OF

料資主業 OWNER PARTICULARS

土地註冊處THE LAND REGISTRY 土地登記冊LAND REGISTER

印製編號 PRINT CONTROL: ESN240712016312

代價

土地註冊處THE LAND REGISTRY 土地登記冊LAND REGISTER 物業涉及的轇轕

印製編號 PRINT CONTROL: ESN240712016312

### INCUMBRANCES

文書日期 註冊日期 註冊摘要編號 文書性質 DATE OF DATE OF 受惠各方 代價 INSTRUMENT REGISTRATION MEMORIAL NO. NATURE IN FAVOUR OF CONSIDERATION

印製編號 PRINT CONTROL: ESN240712016314

印製於 PRINTED AT: INTERNET SEARCH (DOWNLOAD) 查冊日期及時間 SEARCH DATE AND TIME: 12/07/2024 17:27 查冊者姓名/名稱 NAME OF SEARCHER: 查冊種類 SEARCH TYPE: HISTORICAL AND CURRENT

本登記冊列明有關物衆截至12/07/2024 07:30 之資料 THE INFORMATION SET OUT BELOW CONTAINS PARTICULARS OF THE PROPERTY UP TO 07:30 ON 12/07/2024.

備存土地紀錄以供市民查閱旨在防止秘密及有欺詐成分的物業轉易,以及提供容易追溯和確定土地財產及不動產業權的方法。土地紀錄內載的資料不得用於與土地紀錄的宗旨 無關之目的,使用所提供的資料須符合《個人資料(私隱)條例》的規定。

The land records are kept and made available to members of the public to prevent secret and fraudulent conveyances, and to provide means whereby the title to real and immovable property may be easily traced and ascertained. The information contained in the land records shall not be used for purposes that are not related to the purposes of the land records. The use of information provided is subject to the provisions in the Personal Data (Privacy) Ordinance.

《政府租契續期條例》已經生效,土地登記冊的適用政府租契年期資料不會相應更新。適用政府租契的年期屆滿日期,以地政總署根據該條例刊憲的任何相關「續期公告」為 準。

The Extension of Government Leases Ordinance is in force. Information on lease term for an applicable lease is not updated in the land register accordingly. For the expiry date of the lease term of an applicable Government lease, please refer to the relevant "Extension Notice" published by the Lands Department which should prevail.

進行任何交易前,應先向土地註冊處查閱最新的土地紀錄。

BEFORE ANY DEALINGS, UP-TO-DATE LAND SEARCH SHOULD BE CONDUCTED WITH THE LAND REGISTRY.

物業資料 PROPERTY PARTICULARS

物業參考編號

-3

PROPERTY REFERENCE NUMBER (PRN): C5226899

地段編號 LOT NO.: LOT NO. 1710 IN D.D. 114

批約 HELD UNDER: NEW GRANT NO.1180 年期 LEASE TERM: 99 YEARS 開始日期 COMMENCEMENT OF LEASE TERM: 01/07/1898 每年地稅 RENT PER ANNUM: \$10.00

物業參考編號 PRN:C5226899 (12/07/2024) 第 1 頁·共 6 頁 PAGE 1 OF 6

土地註冊處THE LAND REGISTRY 土地登記冊LAND REGISTRP



業主姓名 NAME OF OWNER	身分 (如非唯一擁有人) CAPACITY (IF NOT SOLE OWNER)	註冊摘要編號 MEMORIAL NO.	文書日期 DATE OF INSTRUMENT	註冊日期 DATE OF REGISTRATION	代價 CONSIDERATION

物業參考編號 PRN: C5226899 (12/07/2024) 第 4 頁,共 6 頁 PAGE 4 OF 6

## 土地註冊處THE LAND REGISTRY 土地登記冊LAND REGISTER 印製編號 PRINT CONTROL: ESN240712016314

業主資料 OWNER PARTICULARS

文書日期 註冊日期 文書性質 註冊摘要編號 DATE OF DATE OF INSTRUMENT REGISTRATION NATURE MEMORIAL NO.

INCUMBRANCES

物業涉及的轇轕

IN FAVOUR OF

受惠各方

代價 CONSIDERATION

印製編號 PRINT CONTROL: ESN240712016314

土地註冊處THE LAND REGISTRY 土地登記冊LAND REGISTER

## TRAVERSE STATION SUMMARY



Prepared by : C Y LAU

Checked by : K K LAW

Approved by : CHENG WC, LS/G(NT) Date : 15/09/2010 Geodetic Survey Section SMO, Lands Department © Copyright reserved

### TRAVERSE STATION SUMMARY



Prepared by : C Y LAU

Checked by : K K LAW

Approved by : CHENG WC, LS/G(NT) Date : 15/09/2010 Geodetic Survey Section SMO, Lands Department © Copyright reserved

## TRAVERSE STATION SUMMARY



Prepared by : LAU C Y

Checked by : LAU Y K

Approved by : **KWOK K H, LS/G(NT)** Date : **30/05/2012**  Geodetic Survey Section SMO, Lands Department © Copyright Reserved

#### Annex 2

Memorandum of Understanding



#### 規劃申請意向書

#### 受新發展區發展影響的在地經營業務搬遷 - 規劃許可申請

業務經營者 (甲方)	-	永輝混凝土(香港)有限公司 Glorious Concrete (H.K.) Limited	
公司註冊證明書號碼	:		_
申請人 (乙方)	:	俊怡嘉貨倉管理有限公司 Join Bright Warehousing Limited	
公司註冊證明書號碼	:		

<u>永輝混凝土(香港)有限公司 (甲方)</u>現時為洪水橋 / 厦村新發展區範圍內的業務經營者,由於 受到政府新發展區收地影響,因此,需要覓地搬遷以繼續經營。<u>永輝混凝土(香港)有限公司 (甲方)</u> 初步與<u>俊怡嘉貨倉管理有限公司 (乙方)</u>達成共識,同意<u>俊怡嘉貨倉管理有限公司 (乙方)</u>作 為規劃申請的申請人,並根據《城市規劃條例》第 16 條,向城市規劃委員會(城規會)提交規劃申請, 於*丈量約份第 114 約地段第 573 號餘段及第 1710 號*作「擬議臨時混凝土配料廠連附屬設施(為期 5 年)」。

甲乙雙方亦承諾於規劃申請獲得城規會批准後,申請地點將會由<u>永輝混凝土(香港)有限公司</u> (甲方) 使用營運。

備注: 上述標題地段將會因應規劃許可的需要而有所修訂。

For and on behalf of GLORIOUS CONCRETE (H.K.) LIMITED	For and on behalf of Join Bright Warehousing Limited 俊怡嘉貨倉管理有限公司
Authorised Stgnature(s)	Authorized Signature(s)
永輝混凝土(香港)有限公司(甲方) 業務經營者簽署	後怡嘉貨倉管理有限公司(乙方) 規劃申請的申請人簽署

2025年1月20日

Annex 3

Revised Figure 3.1 from the Traffic Impact Assessment





#### Annex 4

Schematic Diagram of Concrete Production Workflow





Annex 5 Revised Layout Plan





Appendix Ib of RNTPC Paper No. A/YL-SK/410



 Our Ref.:
 DD 114 Lots 573 RP & 1710

 Your Ref.:
 TPB/A/YL-SK/410

The Secretary, Town Planning Board, 15/F, North Point Government Offices, 333 Java Road, North Point, Hong Kong

Dear Sir,

#### 2<sup>nd</sup> Further Information

### Proposed Temporary Concrete Batching Plant with Ancillary Facilities for a Period of 5 Years in "Industrial (Group D)" Zone, Lots 573 RP and 1710 in D.D. 114, Shek Kong, Yuen Long, New Territories

#### (S.16 Planning Application No. A/YL-SK/410)

We write to submit further information to provide clarification, as per the request of the Planning Department, on the operation hours of the proposed development under the subject application.

Should you require more information regarding the application, please contact or the undersigned at your convenience.

Thank you for your kind attention.

Yours faithfully,

For and on behalf of R-riches Property Consultants Limited



Christian CHIM Town Planner

cc DPO/FSYLE, PlanD



<u>By Email</u>

09 April 2025

#### 2<sup>nd</sup> Further Information

#### Proposed Temporary Concrete Batching Plant with Ancillary Facilities for a Period of 5 Years in "Industrial (Group D)" Zone, Lots 573 RP and 1710 in D.D. 114, Shek Kong, Yuen Long, New Territories

#### (Application No. A/YL-SK/410)

- (i) The applicant provides the following clarification on the operation hours of the proposed development:
  - the delivery of ice will take place between 07:00 to 19:00 (i.e. the 12-hour production period).



#### **Government Departments' General Comments**

#### 1. Traffic

Comments of the Chief Highway Engineer/New Territories West, Highways Department (HyD):

- no in-principle objection to the application;
- the proposed access arrangement of the application site (the Site) should be commented and approved by Transport Department (TD);
- HyD is not/ shall not be responsible for the maintenance of any access connecting the Site and Kam Tin Road; and
- advisory comments are detailed in Appendix III.

### 2. Landscape

Comments of the Chief Town Planner/Urban Design and Landscape, Planning Department:

• the Site falls within "Industrial (Group D)" ("I(D)") zone, which is a non-landscape sensitive zoning. No significant landscape impact arising from the proposed use is anticipated.

### 3. <u>Fire Safety</u>

Comments of the Director of Fire Services (D of FS):

- no specific comment on the application; and
- advisory comments are detailed in Appendix III.

#### 4. <u>Geotechnical</u>

Comments of the Head of Geotechnical Engineering Office, Civil Engineering and Development Department (CEDD):

- no adverse geotechnical comment on the application;
- it is noted that the applicant has included a 'no-build' zone in the western part of the Site, within which no critical facilities are proposed to be located; and
- advisory comments are detailed in Appendix III.

#### 5. <u>Other Departments</u>

The following government departments have no objection to/no adverse comment on the application and their advisory comments, if any, are in **Appendix III**:

• Chief Building Surveyor/New Territories West, Buildings Department;
- Chief Engineer/Railway Development 1-1, HyD;
- Project Manager (West), CEDD;
- Chief Engineer/Construction, Water Supplies Department;
- Commissioner of Police; and
- Director of Electrical and Mechanical Services.

#### **Recommended Advisory Clauses**

- (a) the applicant is reminded to resolve any land issues relating to the proposed use with the Lands Department (LandsD);
- (b) the applicant is reminded to obtain relevant licences in relation to the operation of concrete batching plant, including the Specified Process Licence prior to the operation of the proposed use, and to follow and comply with relevant environmental-related ordinances, guidelines and regulations during the construction and operation of the concrete batching plant;
- (c) the applicant is reminded to submit details of the proposed road improvement works to relevant authorities and/or government departments for agreement before its implementation. Relevant ordinances (including the Town Planning Ordinance and the Environmental Impact Assessment Ordinance (EIAO)) and regulations for the proposed road improvement works shall be duly observed and complied with. Approval of the planning application by the Town Planning Board does not imply approval of the proposed road improvement works;
- (d) to note the comments of the District Lands Officer/Yuen Long, LandsD that:
  - by desktop checking, the boundary of the application site (the Site) deviates from the boundary of Lot Nos. 573 RP and 1710 both in D.D. 114 and encroaches on Lot No. 574 S.A in D.D. 114 and the adjoining Government land. The actual lot boundary of the private lots under the application have to be verified at the land application stage if any Short Term Waiver (STW) and/or Short Term Tenancy (STT) is/are applied for by the applicant to LandsD;
- (e) to note the comments of the Commissioner for Transport that:
  - no vehicle is allowed to queue back to or reverse onto / from public road at any time during the planning approval period; and
  - the Site is connected to the public road network via a section of a local access road which is not managed by Transport Department. The land status of the local access road should be checked with the LandsD. Moreover, the management and maintenance responsibilities of the local access road should be clarified with the relevant lands and maintenance authorities accordingly;
- (f) to note the comments of the Director of Agriculture, Fisheries and Conservation:
  - the applicant is advised to adopt appropriate measures to avoid causing pollution or disturbance the adjacent watercourse;
- (g) to note the comments of the Director of Fire Services that:
  - detailed fire services requirements will be formulated upon receipt of formal submission of STT / STW, general building plans or referral of application via relevant licensing authority as appropriate. Furthermore, the EVA provision in the captioned work shall comply with the standard as stipulated in Section 6, Part D of the Code of Practice for Fire Safety in Buildings 2011, which is administered by the Buildings Department (BD);

- (h) to note the comments of the Chief Highway Engineer/New Territories West, Highways Department that:
  - adequate drainage measures shall be provided to prevent surface water running from the Site to the nearby public roads and drains;
- (i) to note the comments of the Director of Environmental Protection that:
  - the applicant is advised that the proposed road upgrading works may potentially constitute a Designated Project under EIAO by virtue of Item Q.1 (i.e., All projects involving earthworks, dredging works and other building works partly or wholly in an existing or gazetted proposed country park or special area, a conservation area, an existing or gazetted proposed marine park or marine reserve, a site of cultural heritage, and a site of special scientific interest.). In this regard, the applicant is reminded to liaise with his Department on the potential EIAO implication of the proposed road upgrading works;
- (j) to note the Comments of the Head of Geotechnical Engineering Office, Civil Engineering and Development Department that:
  - an existing registered man-made geotechnical feature No. 6NE-D/R27 is found in the vicinity of the proposed use, which may affect or be affected by the proposed works. The applicant is reminded that any modification to the existing geotechnical feature No. 6NE-D/R27 should be submitted to the Building Authority (BA) for approval; and
- (k) to note the comments of the Chief Building Surveyor/New Territories West, BD that:
  - it is noted that temporary concrete batching plant with ancillary facilities are proposed in the application. Before any new building works (including containers/ open sheds as temporary buildings, demolition and land filling, etc.) are to be carried out on the Site, prior approval and consent of the BA should be obtained, otherwise they are unauthorised building works (UBWs) under the BO. An Authorised Person should be appointed as the co-ordinator for the proposed building works in accordance with the BO;
  - the Site shall be provided with means of obtaining access thereto from a street and emergency vehicular access in accordance with Regulations 5 and 41D of the Building (Planning) Regulation (B(P)R) respectively;
  - the Site does not abut on a specified street of not less than 4.5m wide and its permitted development intensity shall be determined under Regulation 19(3) of the B(P)R at building plan submission stage;
  - if the existing structures are erected on leased land without the approval of the BA, they are UBW under the BO and should not be designated for any proposed use under the application;
  - for UBW erected on leased land, enforcement action may be taken by BD to effect their removal in accordance with the prevailing enforcement policy against UBW as and when necessary. The granting of any planning approval should not be construed as an acceptance of any existing building woks or UBW on the Site under the BO;

- any temporary shelters or converted containers for office, storage, washroom or other uses are considered as temporary buildings are subject to the control of Part VII of the B(P)R; and
- detailed checking under the BO will be carried out at building plan submission stage.

<sup>3</sup>25-03-12 13:54 FROM- PAT HEUNG RC 水小水回安只冒從音 首治北内這等追 >>> 航北内政府管省 <u>No. A/YL-SK/410</u> (傳真: 2877 0245 及 電郵:www.tpbpd@pland.gov.hk)

> <u>強烈反對元朗石崗丈量約份第114 约地段第573 號餘段及第1710 號</u> 擬議臨時混凝土配料廠連附屬設施,申請編號: A/YL-SK/410 - 申請人提交進一步資料

執事先生 / 女士:

本人是元朗八鄉上村居民,就上標計劃改建為混凝土配料廠的提案作出強烈和堅決的反對。

1. 對環境造成影響環境影響

空氣污染:混凝土配料廠可能產生粉塵和廢氣,影響空氣品質,對居民健康造成威脅,尤其是 老人和兒童。

噪音污染:工廠運作可能帶來持續噪音,影響居民生活品質,特別是夜間作業時。 水污染:混凝土生產可能產生廢水,若處理不當,可能污染附近水源,影響生態和農業。

2. 交通壓力

重型車輛增加:混凝土配料廠需要大量原材料運輸,可能導致重型車輛頻繁出入,加劇交通擁 堵,並增加道路損壞風險。

交通安全:重型車輛增多可能提升交通事故風險,尤其對行人、騎單車者和學生構成威脅。

3. 社區影響

生活品質下降:工廠帶來的污染和噪音可能降低居民生活品質。健康風險:長期暴露於粉塵和 廢氣可能導致呼吸系統疾病,增加醫療負擔。

4. 生態破壞

自然環境受損:混凝土配料廠可能破壞周邊自然環境,影響動植物棲息地,降低生物多樣性。 農業影響:若附近有農地,工廠的污染可能影響農作物生長,損害農民生計。

5. 基礎設施壓力

工廠運作可能對水電等基礎設施造成額外壓力,影響居民正常使用。

作為當區的持份者,我們的反對是基於對環境、交通、社區健康和長遠發展的擔憂。規劃處在 審批時需全面評估這些影響,並考慮居民意見,確保項目符合可持續發展原則

八鄉上村居民簽署 姓名(正楷).

2025年	月_12 日	
通訊地址	:	
聯絡電話	•	

 '25-03-12 13:54 FROM PAT HEUNG RC
 T-678 P0002/0042 F-544

 城市規劃安貝質祕書
 資港北甪遺筆通 333 號北月政府含者 13 楼

 (傳真: 2877 0245 及 電郵: www. tpbpd@pland.gov.hk)
 2

<u>強烈反對元朗石崗丈量約份第114 约地段第573 號餘段及第1710 號</u> 擬藏臨時混凝土配料廠連附屬設施,申請編號: A/YL-SK/410 - 申請人提交進一步資料

執事先生 / 女士:

本人是元朗八鄉上村居民,就上標計劃改建為混凝土配料廠的提案作出強烈和堅決的反對。 1. 對環境造成影響環境影響

空氣污染:混凝土配料廠可能產生粉塵和廢氣,影響空氣品質,對居民健康造成威脅,尤其是 老人和兒童。

噪音污染:工廠運作可能帶來持續噪音,影響居民生活品質,特別是夜間作業時。

水污染:混凝土生產可能產生廢水,若處理不當,可能污染附近水源,影響生態和農業。

2. 交通壓力

重型車輛增加:混凝土配料廠需要大量原材料運輸,可能導致重型車輛頻繁出入,加劇交通擁 堵,並增加道路損壞風險。

交通安全:重型車輛增多可能提升交通事故風險,尤其對行人、騎單車者和學生構成威脅。

3. 社區影響

生活品質下降;工廠帶來的污染和噪音可能降低居民生活品質。健康風險;長期暴露於粉塵和 廢氣可能導致呼吸系統疾病,增加醫療負擔。

4. 生態破壞

自然環境受損:混凝土配料廠可能破壞周邊自然環境,影響動植物棲息地,降低生物多樣性。 農業影響:若附近有農地,工廠的污染可能影響農作物生長,損害農民生計。

5. 基礎設施壓力

工廠運作可能對水電等基礎設施造成額外壓力,影響居民正常使用。

作為當區的持份者,我們的反對是基於對環境、交通、社區健康和長遠發展的擔憂。規劃處在 審批時需全面評估這些影響,並考慮居民意見,確保項目符合可持續發展原則

八鄉上村居民簽署: 76 师 姓名(正楷): 揭明董

2025年	3	月_	12	_B			
通訊地址	:						ſ
聯絡電話	:						

<sup>25-03-12</sup> 13:55 FROM- PAT HEUNG RC T-678 P0003/0042 F-544 城市規劃委員曾祕畜 沓港北角澄季道 *335* 號北角政府合者 13 楼 (傳真: 2877 0245 及 電郵:www.tobpd@pland.gov.hk)

##UL 3

<u>強烈反對元明石崗丈量約份第114 约地段第573 號餘段及第1710 號</u> 擬議臨時混凝土配料廢達附屬設施,申請編號: A/YL-SK/410 - 申請人提交進一步資料

執事先生 / 女士:

本人是元朗八鄉上村居民,就上標計劃改建為混凝土配料廠的提案作出強烈和堅決的反對。 1. 對環境造成影響環境影響

空氣污染:混凝土配料廠可能產生粉塵和廢氣,影響空氣品質,對居民健康造成威脅,尤其是 老人和兒童。

噪音污染:工廠運作可能帶來持續噪音,影響居民生活品質,特別是夜間作業時。

水污染:混凝土生產可能產生廢水,若處理不當,可能污染附近水源,影響生態和農業。

2. 交通壓力

重型車輛增加;混凝土配料廠需要大量原材料運輸,可能導致重型車輛頻繁出入,加劇交通擁 堵,並增加道路損壞風險。

交通安全:重型車輛增多可能提升交通事故風險,尤其對行人、騎單車者和學生構成威脅。

3. 社區影響

生活品質下降;工廠帶來的污染和噪音可能降低居民生活品質。健康風險;長期暴露於粉塵和 廢氣可能導致呼吸系統疾病,增加醫療負擔。

4. 生態破壞

自然環境受損:混凝土配料廠可能破壞周邊自然環境,影響動植物棲息地,降低生物多樣性。 農業影響:若附近有農地,工廠的污染可能影響農作物生長,損害農民生計。

5. 基礎設施壓力

工廠運作可能對水電等基礎設施造成額外壓力,影響居民正常使用。

作為當區的持份者,我們的反對是基於對環境、交通、社區健康和長邃發展的擔憂。規劃處在 審批時需全面評估這些影響,並考慮居民意見,確保項目符合可持續發展原則

本人懇請城市規劃委員會及政府相關部門重視八鄉居民的意見,聆聽居民的聲音,否決上標申請。

八鄉上村居民簽署: <u>乘 拘弃</u> 姓名(正楷): <u>蒙 拘弃</u>

2025 年	3月12日	8	
通訊地址			
聯絡電話			

'25-03-12 13:55 FROM- PAT HEUNG RC
 T-678 P0004/0042 F-544
 城市規劃委員會秘書
 香港北角渣華道 333 號北角政府含著 15 樓
 (傳真: 2877 0245 及 電郵: www. tpbpd@pland.gov.hk)

-3 R.G. 4

強烈反對元朗石崗丈量约份第114约地段第573號餘段及第1710號

擬議臨時混凝土配料廠連附屬設施,申請編號: A/YL-SK/410

- 申請人提交進一步資料

執事先生 / 女士:

本人是元朗八鄉上村居民,就上標計劃改建為混凝土配料廠的提案作出強烈和堅決的反對。 1. 對環境造成影響環境影響

空氣污染:混凝土配料廠可能產生粉塵和廢氣,影響空氣品質,對居民健康造成威脅,尤其是 老人和兒童。

噪音污染:工廠運作可能帶來持續噪音,影響居民生活品質,特別是夜間作業時。

水污染:混凝土生產可能產生廢水,若處理不當,可能污染附近水源,影響生態和農業。

2. 交通壓力

重型車輛增加:混凝土配料廠需要大量原材料運輸,可能導致重型車輛頻繁出入,加劇交通擁 端,站,並增加道路損壞風險。

交通安全:重型車輛增多可能提升交通事故風險,尤其對行人、騎單車者和學生構成威脅。

3. 社區影響

生活品質下降:工廠帶來的污染和噪音可能降低居民生活品質。健康風險:長期暴露於粉塵和 廢氣可能導致呼吸系統疾病,增加醫療負擔。

4. 生態破壞

自然環境受損:混凝土配料廠可能破壞周邊自然環境,影響動植物棲息地,降低生物多樣性。 農業影響:若附近有農地,工廠的污染可能影響農作物生長,損害農民生計。

5. 基礎設施壓力

工廠運作可能對水電等基礎設施造成額外壓力,影響居民正常使用。

作為當區的持份者,我們的反對是基於對環境、交通、社區健康和長遠發展的擔憂。規劃處在 審批時需全面評估這些影響,並考慮居民意見,確保項目符合可持續發展原則

八鄉上村居民簽署: 姓名(正楷):

2025_年	7月12	Ð		
通訊地址				
聯絡電話				
		10.15		

25-03-12 13:55 FROM- PAT HEUNG RC T-678 P0005/0042 F-544 香港北角渣華道 333 號北角政府合署 15 楼 城市規劃委員會秘書 (傳真: 2877 0245 及 電郵: www.tpbpd@pland.gov.hk) -; d8 .

> 強烈反對元朗石崗丈量约份第114约地段第573號餘段及第1710號 擬議臨時混凝土配料廠連附屬設施,申請編號: A/YL-SK/410 - 申請人提交進一步資料

執事先生 / 女士:

本人是元朗八鄉上村居民,就上標計劃改建為混凝土配料廠的提案作出強烈和堅決的反對。

1. 對環境造成影響環境影響

空氣污染:混凝土配料廠可能產生粉塵和廢氣,影響空氣品質,對居民健康造成威脅,尤其是 老人和兒童。

噪音污染:工廠運作可能帶來持續噪音,影響居民生活品質,特別是夜間作業時。 水污染:混凝土生產可能產生廢水,若處理不當,可能污染附近水源,影響生態和農業。

2. 交通壓力

重型車輛增加:混凝土配料廠需要大量原材料運輸,可能導致重型車輛頻繁出入,加劇交通擁 堵,並增加道路損壞風險。

交通安全:重型車輛增多可能提升交通事故風險,尤其對行人、騎單車者和學生構成威脅。

3. 社區影響

生活品質下降:工廠帶來的污染和噪音可能降低居民生活品質。健康風險:長期暴露於粉塵和 廢氣可能導致呼吸系統疾病,增加醫療負擔。

4. 生態破壞

自然環境受損:混凝上配料廠可能破壞問邊自然環境,影響動植物棲息地,降低生物多樣性。 農業影響:若附近有農地,工廠的污染可能影響農作物生長,損害農民生計。

5. 基礎設施壓力

工廠運作可能對水電等基礎設施造成額外壓力,影響居民正常使用。

作為當區的持份者,我們的反對是基於對環境、交通、社區健康和長遠發展的擔憂。規劃處在 審批時需全面評估這些影響,並考慮居民意見,確保項目符合可持續發展原則

本人懇請城市規劃委員會及政府相關部門重視八鄉居民的意見,聆聽居民的聲音,否決 上標申請。

八鄉上村居民簽署 姓名(正楷): LA

2025 年	2	n	-			
通訊地址	:					
聯絡電話	:					
	5	5				

5

'25-03-12 13:55 FROM- PAT HEUNG RC T-678
 城市規劃委員會秘書 香港北角渣築道 333 號北角政府含者 13 後
 (傳真: 2877 0245 及 電郵: www. tobpd@pland. gov. hk)

T-678 P0006/0042 F-544

S 18 1 4

6

### <u>強烈反對元朗石崗丈量約份第114 约地段第573 號餘段及第1710 號</u> 擬議臨時混凝土配料廠連附屬設施,申請編號: A/YL-SK/410 - 申請人提交進一步資料

執事先生 / 女士:

本人是元朗八鄉上村居民,就上標計劃改建為混凝土配料廠的提案作出強烈和堅決的反對。

1. 對環境造成影響環境影響

空氣污染:混凝土配料廠可能產生粉塵和廢氣,影響空氣品質,對居民健康造成威脅,尤其是 老人和兒童。

噪音污染:工廠運作可能帶來持續噪音,影響居民生活品質,特別是夜間作業時。

水污染:混凝土生產可能產生廢水,若處理不當,可能污染附近水源,影響生態和農業。

2. 交通壓力

重型車輛增加:混凝土配料廠需要大量原材料運輸,可能導致重型車輛頻繁出入,加劇交通擁 堵,並增加道路損壞風險。

交通安全:重型車輛增多可能提升交通事故風險,尤其對行人、騎單車者和學生構成威脅。

3. 社區影響

生活品質下降:工廠帶來的污染和噪音可能降低居民生活品質。健康風險:長期暴露於粉塵和 廢氣可能導致呼吸系統疾病,增加醫療負擔。

4. 生態破壞

自然環境受損:混凝土配料廠可能破壞周邊自然環境,影響動植物棲息地,降低生物多樣性。 農業影響:若附近有農地,工廠的污染可能影響農作物生長,損害農民生計。

5. 基礎設施壓力

工廠運作可能對水電等基礎設施造成額外壓力,影響居民正常使用。

作為當區的持份者,我們的反對是基於對環境、交通、社區健康和長遠發展的擔憂。規劃處在 審批時需全面評估這些影響,並考慮居民意見,確保項目符合可持續發展原則

八郷上村居民簽署: ()のハロ, 姓名(正楷): 黎守直

2025年	3月	12	H		
通訊地址	:				
聯絡電話	:				

'25-03-12 13:55 FROM- PAT HEUNG RC T-678 PO 城市規劃委員會秘書 香港北角渣華道 333 號北角政府合署 15 樓 (傳真: 2877 0245 及 電郵: www. tpbpd@pland.gov.hk)

T-678 P0007/0042 F-544

7

<u>強烈反對元朗石崗丈量约份第114 约地段第573 號餘段及第1710 號</u> 擬議臨時混凝土配料廠連附屬設施,申請編號: A/YL-SK/410 - 申請人提交進一步資料

執事先生 / 女士:

本人是元朗八鄉上村居民,就上標計劃改建為混凝土配料廠的提案作出強烈和堅決的反對。

1. 對環境造成影響環境影響

空氣污染:混凝土配料廠可能產生粉塵和廢氣,影響空氣品質,對居民健康造成威脅,尤其是 老人和兒童。

噪音污染:工廠運作可能帶來持續噪音,影響居民生活品質,特別是夜間作業時。

水污染:混凝土生產可能產生廢水,若處理不當,可能污染附近水源,影響生態和農業。

2. 交通壓力

重型車輛增加:混凝土配料廠需要大量原材料運輸,可能導致重型車輛頻繁出入,加劇交通擁 堵,並增加道路損壞風險。

交通安全:重型車輛增多可能提升交通事故風險,尤其對行人、騎單車者和學生構成威脅。

3. 社區影響

生活品質下降:工廠帶來的污染和噪音可能降低居民生活品質。健康風險:長期暴露於粉塵和 廢氣可能導致呼吸系統疾病,增加醫療負擔。

4. 生態破壞

自然環境受損:混凝土配料廠可能破壞周邊自然環境,影響動植物棲息地,降低生物多樣性。 農業影響:若附近有農地,工廠的污染可能影響農作物生長,損害農民生計。

5. 基礎設施壓力

工廠運作可能對水電等基礎設施造成額外壓力,影響居民正常使用。

作為當區的持份者,我們的反對是基於對環境、交通、社區健康和長遠發展的擔憂。規劃處在 審批時需全面評估這些影響,並考慮居民意見,確保項目符合可持續發展原則

八鄉上村居民簽署: 姓名(正楷): 我不不算

2025 年	2	B	12	Я	а а (я	21	
通訊地址	:						
聯絡電話	:						

·25-03-12 13:55 FROM- PAT HEUNG RC T-678 P0008/0042 F-544 城市規劃委員會秘書 香港北角渣華道 333 號北角政府合著 15 樓 (傳真: 2877 0245 及 電郵: www. tpbpd@pland, gov. hk)

17 M 43 . 8

強烈反對元朗石崗丈量約份第114约地段第573號餘段及第1710號 擬議臨時混凝土配料廠連附屬設施,申請編號: A/YL-SK/410 - 申請人提交進一步資料

執事先生 / 女士:

本人是元朗八鄉上村居民,就上標計劃改建為混凝土配料廠的提案作出強烈和堅決的反對。

1. 對環境造成影響環境影響

空氣污染:混凝土配料廠可能產生粉塵和廢氣,影響空氣品質,對居民健康造成威脅,尤其是 老人和兒童。

噪音污染:工廠運作可能帶來持續噪音,影響居民生活品質,特別是夜間作業時。

水污染:混凝土生產可能產生廢水,若處理不當,可能污染附近水源,影響生態和農業。

2. 交通壓力

重型車輛增加:混凝土配料廠需要大量原材料運輸,可能導致重型車輛頻繁出入,加劇交通擁 堵,並增加道路損壞風險。

交通安全:重型車輛增多可能提升交通事故風險,尤其對行人、騎單車者和學生構成威脅。

3. 社區影響

生活品質下降:工廠帶來的污染和噪音可能降低居民生活品質。健康風險;長期暴露於粉塵和 廢氣可能導致呼吸系統疾病,增加醫療負擔。

4. 生態破壞

自然環境受損:混凝土配料廠可能破壞周邊自然環境,影響動植物棲息地,降低生物多樣性。 農業影響:若附近有農地,工廠的污染可能影響農作物生長,損害農民生計。

5. 基礎設施壓力

工廠運作可能對水電等基礎設施造成額外壓力,影響居民正常使用。

作為當區的持份者,我們的反對是基於對環境、交通、社區健康和長遠發展的擔憂。規劃處在 審批時需全面評估這些影響,並考慮居民意見,確保項目符合可持續發展原則

八鄉上村居民簽署: 姓名(正楷): 王文

2025	年_	3	A	12	B		
通訊地	址	:					
聯絡電	話	:					

T-678 P0009/0042 F-544

-1 -0

'25-03-12 13:56 FROM- PAT HEUNG RC T-678 F
 城市規劃委員會秘書 香港北角渣華道 333 號北角政府合署 15 樓
 (傳真: 2877 0245 及 電郵: www.tpbpd@pland.gov.hk)

9

<u>強烈反對元朗石崗丈量約份第114的地段第573號餘段及第1710號</u> 擬議臨時混凝土配料廠連附屬設施,申請編號: A/YL-SK/410 - 申請人提交進一步資料

執事先生 / 女士:

本人是元朗八鄉上村居民,就上標計劃改建為混凝土配料廠的提案作出強烈和堅決的反對。

1. 對環境造成影響環境影響

空氣污染: 混凝土配料廠可能產生粉塵和廢氣,影響空氣品質,對居民健康造成威脅,尤其是 老人和兒童。

嗓音污染:工廠運作可能帶來持續嗓音,影響居民生活品質,特別是夜間作業時。

水污染:混凝土生產可能產生廢水,若處理不當,可能污染附近水源,影響生態和農業。

2. 交通壓力

重型車輛增加:混凝土配料廠需要大量原材料運輸,可能導致重型車輛頻繁出入,加劇交通擁 堵,並增加道路損壞風險。

交通安全:重型車輛增多可能提升交通事故風險,尤其對行人、騎單車者和學生構成威脅。

3. 社區影響

生活品質下降:工廠帶來的污染和噪音可能降低居民生活品質。健康風險:長期暴露於粉塵和 廢氣可能導致呼吸系統疾病,增加醫療負擔。

4. 生態破壞

自然環境受損:混凝土配料廠可能破壞周邊自然環境,影響動植物棲息地,降低生物多樣性。 農業影響:若附近有農地,工廠的污染可能影響農作物生長,損害農民生計。

5. 基礎設施壓力

工廠運作可能對水電等基礎設施造成額外壓力,影響居民正常使用。

作為當區的持份者,我們的反對是基於對環境、交通、社區健康和長遠發展的擔憂。規劃處在 審批時需全面評估這些影響,並考慮居民意見,確保項目符合可持續發展原則

八鄉上村居民簽署: 姓名(正楷):

2025 年	7	p	12	D			
通訊地址	:						
聯絡電話	:						

'25-03-12 13:56 FROM- PAT HEUNG RC
 丁-678 P0010/0042 F-544
 城市規劃委員會秘書
 香港北角渣華道 333 號北角政府含著 15 樓
 (傳真: 2877 0245 及 電郵: www. tpbpd@pland. gov. hk)

10

1 1

<u>強烈反對元朗石崗丈量约份第114 约地段第573 號餘段及第1710 號</u> 擬議臨時混凝土配料廠連附屬設施,申請編號: A/YL-SK/410 - 申請人提交進一步資料

執事先生 / 女士:

本人是元朗八鄉上村居民,就上標計劃改建為混凝土配料廠的提案作出強烈和堅決的反對。
 1. 對環境造成影響環境影響

空氣污染:混凝土配料廠可能產生粉壓和廢氣,影響空氣品質,對居民健康造成威脅,尤其是 \* 老人和兒童。

噪音污染;工廠運作可能帶來持續噪音,影響居民生活品質,特別是夜間作業時。

水污染:混凝土生產可能產生廢水,若處理不當,可能污染附近水源,影響生態和農業。

2. 交通壓力

重型車輛增加:混凝土配料廠需要大量原材料運輸,可能導致重型車輛頻繁出入,加劇交通擁 堵,並增加道路損壞風險。

交通安全:重型車輛增多可能提升交通事故風險,尤其對行人、騎單車者和學生構成威脅。

3. 社區影響

生活品質下降;工廠帶來的污染和噪音可能降低居民生活品質。健康風險:長期暴露於粉塵和 廢氣可能導致呼吸系統疾病,增加醫療負擔。

4. 生態破壞

自然環境受損:混凝土配料廠可能破壞周邊自然環境,影響動植物棲息地,降低生物多樣性。 農業影響:若附近有農地,工廠的污染可能影響農作物生長,損害農民生計。

5. 基礎設施壓力

工廠運作可能對水電等基礎設施造成額外壓力,影響居民正常使用。

作為當區的持份者,我們的反對是基於對環境、交通、社區健康和長遠發展的擔憂。規劃處在 審批時需全面評估這些影響,並考慮居民意見,確保項目符合可持續發展原則

本人懇請城市規劃委員會及政府相關部門重視八鄉居民的意見,聆聽居民的聲音,否決上標申請。

八鄉上村居民簽署:

2025 <u>年 3 月 12 日</u> 通訊地址: 聯絡電話:

25-03-12 13:56 FROM- PAT HELING RC T-678 P0011/0042 F-544 城市規劃委員會秘書 香港北角渣華道 333 號北角政府合署 15 樓 (傳真: 2877 0245 及 電郵: www.tpbpd@pland.gov.hk) -1 --

11

### 強烈反對元朗石崗丈量約份第114约地段第573號餘段及第1710號 擬議臨時混凝土配料廠連附屬設施,申請編號: A/YL-SK/410 - 申請人提交進一步資料

執事先生 / 女士:

本人是元朗八鄉上村居民,就上標計劃改建為混凝土配料廠的提案作出強烈和堅決的反對。

1. 對環境造成影響環境影響

空氣污染:混凝土配料廠可能產生粉塵和廢氣,影響空氣品質,對居民健康造成威脅,尤其是 老人和兒童。

嗓音污染:工廠運作可能帶來持續嗓音,影響居民生活品質,特別是夜間作業時。

水污染:混凝土生產可能產生廢水,若處理不當,可能污染附近水源,影響生態和農業。

2. 交通壓力

重型車輛增加:混凝土配料廠需要大量原材料運輸,可能導致重型車輛頻繁出入,加劇交通擁 堵,並增加道路損壞風險。

交通安全:重型車輛增多可能提升交通事故風險,尤其對行人、騎單車者和學生構成威脅。

3. 社區影響

生活品質下降;工廠帶來的污染和噪音可能降低居民生活品質。健康風險:長期暴露於粉塵和 廢氣可能導致呼吸系統疾病,增加醫療負擔。

4. 生態破壞

2025年3日

12-MAR-2025 14:00

自然環境受損:混凝土配料廠可能破壞周邊自然環境,影響動植物棲息地,降低生物多樣性。 農業影響:若附近有農地,工廠的污染可能影響農作物生長,損害農民生計。

5. 基礎設施壓力

工廠運作可能對水電等基礎設施造成額外壓力,影響居民正常使用。

作為當區的持份者,我們的反對是基於對環境、交通、社區健康和長遠發展的捲臺。規劃處在 審批時需全面評估這些影響,並考慮居民意見,確保項目符合可持續發展原則

八鄉上村居民簽署: 白 維素 姓名(正楷): 近錐毫

T-678 P0012/0042 F-544

·25-03-12 13:56 FROM- PAT HEUNG RC T-678 城市規劃委員會秘書 香港北角渣華道 333 號北角政府合署 15 樓 (傳真: 2877 0245 及 電郵: www. tobpd@pland.gov.hk)

12

### <u>強烈反對元朗石崗丈量约份第114 约地段第573 號餘段及第1710 號</u> 擬議臨時混凝土配料廠連附屬設施,申請編號: A/YL-SK/410 - 申請人提交進一步資料

執事先生 / 女士:

本人是元朝八鄉鄉事委員會委員,就上標計劃改建為混凝土配料廠的提案作出強烈和堅決的反對。

1. 對環境造成影響環境影響

空氣污染:混凝土配料廠可能產生粉塵和廢氣,影響空氣品質,對居民健康造成威脅,尤其是 老人和兒童。

嗓音污染:工廠運作可能帶來持續噪音,影響居民生活品質,特別是夜間作業時。 水污染:混凝土生產可能產生廢水,若處理不當,可能污染附近水源,影響生態和農業。

2. 交通壓力

重型車輛增加:混凝土配料廠需要大量原材料運輸,可能導致重型車輛頻繁出入,加劇交通擁 站,並增加道路損壞風險。

交通安全:重型車輛增多可能提升交通事故風險,尤其對行人、騎單車者和學生構成威脅。

3. 社區影響

生活品質下降:工廠帶來的污染和噪音可能降低居民生活品質。健康風險:長期暴露於粉塵和 廢氣可能導致呼吸系統疾病,增加醫療負擔。

4. 生態破壞

自然環境受損:混凝土配料廠可能破壞周邊自然環境,影響動植物棲息地,降低生物多樣性。 農業影響:若附近有農地,工廠的污染可能影響農作物生長,損害農民生計。

5. 基礎設施壓力

工廠運作可能對水電等基礎設施造成額外壓力,影響居民正常使用。

作為當區的持份者,我們的反對是基於對環境、交通、社區健康和長遠發展的擔憂。規劃處在 審批時需全面評估這些影響,並考慮居民意見,確保項目符合可持續發展原則

本人懇請城市規劃委員會及政府相關部門重視八鄉居民的意見,聆聽居民的聲音,否決上標申請。

簽署:<u>/252</u> 姓名(正楷): 鄧 志 光

## 2025年3月12日

通訊地址:

聯絡電話:

12-MAR-2025 14:00

T-678 P0013/0042 F-544

 '25-03-12 13:56 FROM PAT HEUNG RC
 T-678

 城市規劃委員會秘書
 香港北角渣華道 333 號北角政府合署 15 樓

 (傳真: 2877 0245 及 電郵: www.tpbpd@pland.gov.hk)

13

### <u>強烈反對元朗石崗丈量约份第114 约地段第573 號餘段及第1710 號</u> 擬議臨時混凝土配料廠連附屬設施,申請編號: A/YL-SK/410 - 申請人提交進一步資料

執事先生 / 女士:

本人是元朗八鄉鄉事委員會委員,就上標計劃改建為混凝土配料廠的提案作出強烈和堅決的反對。

1. 對環境造成影響環境影響

空氣污染:混凝土配料廠可能產生粉塵和廢氣,影響空氣品質,對居民健康造成威脅,尤其是 老人和兒童。

嗓音污染:工廠運作可能帶來持續嗓音,影響居民生活品質,特別是夜間作業時。 水污染:混凝土生產可能產生廢水,若處理不當,可能污染附近水源,影響生態和農業。

2. 交通壓力

重型車輛增加:混凝土配料廠需要大量原材料運輸,可能導致重型車輛頻繁出入,加劇交通擁 堵,並增加道路損壞風險。

交通安全:重型車輛增多可能提升交通事故風險,尤其對行人、騎單車者和學生構成威脅。

3. 社區影響

生活品質下降:工廠帶來的污染和噪音可能降低居民生活品質。健康風險:長期暴露於粉塵和 廢氣可能導致呼吸系統疾病,增加醫療負擔。

4. 生態破壞

自然環境受損:混凝土配料廠可能破壞周邊自然環境,影響動植物棲息地,降低生物多樣性。 農業影響:若附近有農地,工廠的污染可能影響農作物生長,損害農民生計。

5. 基礎設施壓力

工廠運作可能對水電等基礎設施造成額外壓力,影響居民正常使用。

作為當區的持份者,我們的反對是基於對環境、交通、社區健康和長遠發展的擔憂。規劃處在 審批時需全面評估這些影響,並考慮居民意見,確保項目符合可持續發展原則

本人懇請城市規劃委員會及政府相關部門重視八鄉居民的意見,聆聽居民的聲音,否決上標申請。

發書: <u>楊礼策</u> 姓名(正楷): 楊禧举

# 2025年3月12日

通訊地址:

聯絡電話:

'25-03-12 13:56 FROM- PAT HEUNG RC
 T-678 POC
 城市規劃委員會秘書
 香港北角渣華道 333 號北角政府合署 15 樓
 (傳真: 2877 0245 及 電郵: www.tpbpd@pland.gov.hk)

T-678 P0014/0042 F-544

14

<u>強烈反對元朗石崗丈量約份第114 约地段第573 號餘段及第1710 號</u> 擬議臨時混凝土配料廠連附屬設施,申請編號: A/YL-SK/410

#### - 申請人提交進一步資料

執事先生 / 女士:

本人是元朗八鄉鄉事委員會委員,就上標計劃改建為混凝土配料廠的提案作出強烈和堅決的反對。

1. 對環境造成影響環境影響

空氣污染: 混凝土配料廠可能產生粉塵和廢氣,影響空氣品質,對居民健康造成威脅,尤其是 老人和兒童。

嗓音污染:工廠運作可能帶來持續嗓音,影響居民生活品質,特別是夜間作業時。 水污染:混凝土生產可能產生廢水,若處理不當,可能污染附近水源,影響生態和農業。

2. 交通壓力

重型車輛增加:混凝土配料廠需要大量原材料運輸,可能導致重型車輛頻繁出入,加劇交通擁 堵,並增加道路損壞風險。

交通安全:重型車輛增多可能提升交通事故風險,尤其對行人、騎單車者和學生構成威脅。

3. 社區影響

生活品質下降:工廠帶來的污染和噪音可能降低居民生活品質。健康風險:長期暴露於粉塵和 廢氣可能導致呼吸系統疾病,增加醫療負擔。

4. 生態破壞

自然環境受損;混凝土配料廠可能破壞周邊自然環境,影響動植物棲息地,降低生物多樣性。 農業影響:若附近有農地,工廠的污染可能影響農作物生長,損害農民生計。

5. 基礎設施壓力

工廠運作可能對水電等基礎設施造成額外壓力,影響居民正常使用。

作為當區的持份者,我們的反對是基於對環境、交通、社區健康和長遠發展的擔憂。規劃處在 審批時需全面評估這些影響,並考慮居民意見,確保項目符合可持續發展原則

本人懇請城市規劃委員會及政府相關部門重視八鄉居民的意見,聆聽居民的聲音,否決上標申請。

姓名(正楷):

2<u>025 年3月12</u>日 通訊地址:

聯絡電話:

·25-03-12 13:57 FROM- PAT HEUNG RC T-678 P0015/0042 F-544 城市規劃委員會秘書 香港北角渣華道 333 號北角政府合署 15 樓 (傳真: 2877 0245 及 電郵: www.tpbpd@pland.gov.hk)

> <u>強烈反對元朗石崗丈量約份第114 约地段第573 號餘段及第1710 號</u> 擬議臨時混凝土配料廠連附屬設施,申請編號: A/YL-SK/410 - 申請人提交進一步資料

執事先生 / 女士:

本人是元朗八鄉鄉事委員會委員,就上標計劃改建為混凝土配料廠的提案作出強烈和堅決的反對。

1. 對環境造成影響環境影響

空氣污染:混凝土配料廠可能產生粉塵和廢氣,影響空氣品質,對居民健康造成威脅,尤其是 老人和兒童。

噪音污染:工廠運作可能帶來持續噪音,影響居民生活品質,特別是夜間作業時。

水污染:混凝土生產可能產生廢水,若處理不當,可能污染附近水源,影響生態和農業。

2. 交通壓力

重型車輛增加:混凝土配料廠需要大量原材料運輸,可能導致重型車輛頻繁出入,加劇交通擁 堵,並增加道路損壞風險。

交通安全:重型車輛增多可能提升交通事故風險,尤其對行人、騎單車者和學生構成威脅。

3. 社區影響

生活品質下降:工廠帶來的污染和噪音可能降低居民生活品質。健康風險:長期暴露於粉塵和 廢氣可能導致呼吸系統疾病,增加醫療負擔。

4. 生態破壞

自然環境受損:混凝土配料廠可能破壞周邊自然環境,影響動植物棲息地,降低生物多樣性。 農業影響:若附近有農地,工廠的污染可能影響農作物生長,損害農民生計。

5. 基礎設施壓力

工廠運作可能對水電等基礎設施造成額外壓力,影響居民正常使用。

作為當區的持份者,我們的反對是基於對環境、交通、社區健康和長遠發展的擔憂。規劃處在 審批時需全面評估這些影響,並考慮居民意見,確保項目符合可持續發展原則

本人懇請城市規劃委員會及政府相關部門重視八鄉居民的意見,聆聽居民的聲音,否決上標申請。

2025年3	月_12_日	
通訊地址:		
聯絡電話:		
12-MAR-2025	14:08	

97%

15

16

2 .\*

### <u>強烈反對元朗石崗丈量約份第114 约地段第573 號餘段及第1710 號</u> 擬鐵臨時混凝土配料廠連附屬設施,申請編號: A/YL-SK/410 - 申請人提交進一步資料

執事先生 / 女士:

本人是元朗八鄉鄉事委員會委員,就上標計劃改建為混凝土配料廠的提案作出強烈和堅決的反對。

1. 對環境造成影響環境影響

空氣污染:混凝土配料廠可能產生粉塵和廢氣,影響空氣品質,對居民健康造成威脅,尤其是 老人和兒童。

噪音污染:工廠運作可能帶來持續噪音,影響居民生活品質,特別是夜間作業時。

水污染:混凝土生產可能產生廢水,若處理不當,可能污染附近水源,影響生態和農業。

2. 交通壓力

重型車輛增加:混凝土配料廠需要大量原材料運輸,可能導致重型車輛頻繁出入,加劇交通擁 堵,並增加道路損壞風險。

交通安全:重型車輛增多可能提升交通事故風險,尤其對行人、騎單車者和學生構成威脅。

3. 社區影響

生活品質下降:工廠帶來的污染和噪音可能降低居民生活品質。健康風險:長期暴露於粉塵和 廢氣可能導致呼吸系統疾病,增加醫療負擔。

4. 生態破壞

自然環境受損: 混凝土配料廠可能破壞周邊自然環境,影響動植物棲息地,降低生物多樣性。 農業影響:若附近有農地,工廠的污染可能影響農作物生長,損害農民生計。

5. 基礎設施壓力

工廠運作可能對水電等基礎設施造成額外壓力,影響居民正常使用。

作為當區的持份者,我們的反對是基於對環境、交通、社區健康和長遠發展的擔憂。規劃處在審批時需全面評估這些影響,並考慮居民意見,確保項目符合可持續發展原則

本人懇請城市規劃委員會及政府相關部門重視八鄉居民的意見,聆聽居民的聲音,否決上標申請。

20x ( # 3	月_12百	
通訊地址:		
聯絡電話:		
12-MAR-2025	14:08	

25-03-12 13:57 FROM- PAT HEUNG RC T-678 P0017/0042 F-544 珈児規劃安貝官秘書 首応北月燈準退 333 硫北月政府含者 13 後 (傳真: 2877 0245 及 電郵:www.tpbpd@pland.gov.hk)

> <u>強烈反對元朗石崗丈量約份第114 约地段第573 號餘段及第1710 號</u> 擬議臨時混凝土配料廠連附屬設施,申請編號: A/YL-SK/410 - 申請人提交進一步資料

17

#### 一个词八处.

執事先生 / 女士:

本人是元朗八鄉鄉事委員會委員,就上標計劃改建為混凝土配料廠的提案作出強烈和堅決的反對。

1. 對環境造成影響環境影響

空氣污染:混凝土配料廠可能產生粉塵和廢氣,影響空氣品質,對居民健康造成威脅,尤其是 老人和兒童。

噪音污染:工廠運作可能帶來持鎖噪音,影響居民生活品質,特別是夜間作業時。

水污染:混凝土生產可能產生廢水,若處理不當,可能污染附近水源,影響生態和農業。

2. 交通壓力

重型車輛增加:混凝土配料廠需要大量原材料運輸,可能導致重型車輛頻繁出入,加劇交通擁 堵,並增加道路損壞風險。

交通安全:重型車輛增多可能提升交通事故風險,尤其對行人、騎單車者和學生構成威脅。

3. 社區影響

生活品質下降:工廠帶來的污染和噪音可能降低居民生活品質。健康風險:長期暴露於粉塵和 廢氣可能導致呼吸系統疾病,增加醫療負擔。

4. 生態破壞

自然環境受損:混凝土配料廠可能破壞周邊自然環境,影響動植物棲息地,降低生物多樣性。 農業影響:若附近有農地,工廠的污染可能影響農作物生長,損害農民生計。

5. 基礎設施壓力

工廠運作可能對水電等基礎設施造成額外壓力,影響居民正常使用。

作為當區的持份者,我們的反對是基於對環境、交通、社區健康和長遠發展的擔憂。規劃處在 審批時需全面評估這些影響,並考慮居民意見,確保項目符合可持續發展原則

本人懇請城市規劃委員會及政府相關部門重視八鄉居民的意見,聆聽居民的聲音,否決 上標申請。

姓名(正楷): 簡成中寧

<u>2025年</u>月<u>12</u>日 通訊地址: 聯絡電話:

12-MAR-2025 14:09

'25-03-12 13:57 FROM- PAT HEUNG RC

 T-678 P0018/0042 F-544

 城 中規劃 姿員 曾秘書
 香港北角 澄華 道 333 號北角政府 含著 15 楼

 (傳真: 2877 0245 及 電郵: www.tpbpd@pland.gov.hk)

- 4 18

### 強烈反對元朗石崗丈量約份第114约地段第573號餘段及第1710號 擬議臨時混凝土配料廠連附屬設施,申請編號: A/YL-SK/410 - 申請人提交進一步資料

執事先生 / 女士:

本人是元朗八鄉鄉事委員會委員,就上標計劃改建為混凝土配料廠的提案作出強烈和堅決的反對。

1. 對環境造成影響環境影響

空氣污染:混凝土配料廠可能產生粉塵和廢氣,影響空氣品質,對居民健康造成威脅,尤其是 老人和兒童。

噪音污染:工廠運作可能帶來持續噪音,影響居民生活品質,特別是夜間作業時。 水污染:混凝土生產可能產生廢水,若處理不當,可能污染附近水源,影響生態和農業。

2. 交通壓力

重型車輛增加:混凝土配料廠需要大量原材料運輸,可能導致重型車輛頻繁出入,加劇交通擁 堵,並增加道路損壞風險。

交通安全:重型車輛增多可能提升交通事故風險,尤其對行人、騎單車者和學生構成威脅。

3. 社區影響

生活品質下降:工廠帶來的污染和噪音可能降低居民生活品質。健康風險:長期暴露於粉塵和 廢氣可能導致呼吸系統疾病,增加醫療負擔。

4. 生態破壞

自然環境受損:混凝土配料廠可能破壞周邊自然環境,影響動植物棲息地,降低生物多樣性。 農業影響:若附近有農地,工廠的污染可能影響農作物生長,損害農民生計。

5. 基礎設施壓力

工廠運作可能對水電等基礎設施造成額外壓力,影響居民正常使用。

作為當區的持份者,我們的反對是基於對環境、交通、社區健康和長遠發展的擔憂。規劃處在審批時需全面評估這些影響,並考慮居民意見,確保項目符合可持續發展原則

本人懇請城市規劃委員會及政府相關部門重視八鄉居民的意見, 聆聽居民的聲音, 否決上標申請。

姓名(正楷):

2025年了	月_12	_日	
通訊地址:			
聯絡電話:			
12-MAR-2025	14:09	¢	

<sup>25-03-12 13:57 FROM-</sup> PAT HEUNG RC T-678 P0019/0042 F-544 取化规则安良智秘音 合応北月進学道 233 疏北月政府合者 13 後 (傳真: 2877 0245 及電郵:www.tpbpd@pland.gov.hk) 19

### <u>強烈反對元朗石崗丈量约份第114 约地段第573 號餘段及第1710 號</u> 擬議臨時混凝土配料廠連附屬設施,申請編號: A/YL-SK/410 - 申請人提交進一步資料

執事先生 / 女士:

本人是元朗八鄉鄉事委員會委員,就上標計劃改建為混凝土配料廠的提案作出強烈和堅決的反對。

1. 對環境造成影響環境影響

空氣污染:混凝土配料廠可能產生粉塵和廢氣,影響空氣品質,對居民健康造成威脅,尤其是 老人和兒童。

噪音污染:工廠運作可能帶來持續噪音,影響居民生活品質,特別是夜間作業時。

水污染:混凝土生產可能產生廢水,若處理不當,可能污染附近水源,影響生態和農業。

2. 交通壓力

重型車輛增加:混凝土配料廠需要大量原材料運輸,可能導致重型車輛頻繁出入,加劇交通擁 堵,並增加道路損壞風險。

交通安全:重型車輛增多可能提升交通事故風險,尤其對行人、騎單車者和學生構成威脅。

3. 社區影響

生活品質下降;工廠帶來的污染和噪音可能降低居民生活品質。健康風險:長期暴露於粉塵和 廢氣可能導致呼吸系統疾病,增加醫療負擔。

4. 生態破壞

自然環境受損:混凝土配料廠可能破壞周邊自然環境,影響動植物棲息地,降低生物多樣性。 農業影響:若附近有農地,工廠的污染可能影響農作物生長,損害農民生計。

5. 基礎設施壓力

工廠運作可能對水電等基礎設施造成額外壓力,影響居民正常使用。

作為當區的持份者,我們的反對是基於對環境、交通、社區健康和長遠發展的擔憂。規劃處在 審批時需全面評估這些影響,並考慮居民意見,確保項目符合可持續發展原則

本人懇請城市規劃委員會及政府相關部門重視八鄉居民的意見,聆聽居民的聲音,否決上標申請。

簽署: 姓名(正楷): 蒸發食

2025 <u>F</u> 3	月12日	
通訊地址:		
聯絡電話:		
12-MAR-2025	14:09	

T-678 P0020/0042 F-544

· 1 12

20

### <u>強烈反對元朗石崗丈量約份第114约地段第573號餘段及第1710號</u> 擬議臨時混凝土配料廠連附屬設施,申請編號: A/YL-SK/410 - 申請人提交進一步資料

執事先生 / 女士:

本人是元朗八鄉鄉事委員會委員,就上標計劃改建為混凝土配料廠的提案作出強烈和堅決的反對。

1. 對環境造成影響環境影響

空氣污染:混凝土配料廠可能產生粉塵和廢氣,影響空氣品質,對居民健康造成威脅,尤其是 老人和兒童。

嗓音污染:工廠運作可能帶來持續嗓音,影響居民生活品質,特別是夜間作業時。

水污染:混凝土生產可能產生廢水,若處理不當,可能污染附近水源,影響生態和農業。

2. 交通壓力

重型車輛增加:混凝土配料廠需要大量原材料運輸,可能導致重型車輛頻繁出入,加劇交通擁 堵,並增加道路損壞風險。

交通安全:重型車輛增多可能提升交通事故風險,尤其對行人、騎單車者和學生構成威脅。

3. 社區影響

生活品質下降:工廠帶來的污染和嗓音可能降低居民生活品質。健康風險:長期暴露於粉塵和 廢氣可能導致呼吸系統疾病,增加醫療負擔。

#### 4. 生態破壞

自然環境受損:混凝土配料廠可能破壞周遊自然環境,影響動植物棲息地,降低生物多樣性。 農業影響:若附近有農地,工廠的污染可能影響農作物生長,損害農民生計。

5. 基礎設施壓力

工廠運作可能對水電等基礎設施造成額外壓力,影響居民正常使用。

作為當區的持份者,我們的反對是基於對環境、交通、社區健康和長遠發展的擔憂。規劃處在 審批時需全面評估這些影響,並考慮居民意見,確保項目符合可持續發展原則

本人懇請城市規劃委員會及政府相關部門重視八鄉居民的意見,聆聽居民的聲音,否決 上標申請。

(a) (b) (b) (b) (b) (b) (b) (b) (b) (b) (b				
2025年3	月_12	_8		
通訊地址:				
聯絡電話:				
		- 1 - E	×	
12-MAR-2025	14:09			

25-03-12 13:58 FROM- PAT HEUNG RC
 1-6/8
 3.5 就北角政府合者 13 接
 (傳真: 2877 0245 及電郵: www.tpbpd@pland.gov.hk)

1-678 P0021/0042 F-544

21

### <u>強烈反對元朗石崗丈量約份第114约地段第573號餘段及第1710號</u> 擬議臨時混凝土配料廠連附屬設施,申請編號: A/YL-SK/410 - 申請人提交進一步資料

執事先生 / 女士:

本人是元朗八鄉鄉事委員會委員,就上標計劃改建為混凝土配料廠的提案作出強烈和堅決的反對。

1. 對環境造成影響環境影響

空氣污染:混凝土配料廠可能產生粉塵和廢氣,影響空氣品質,對居民健康造成威脅,尤其是 老人和兒童。

嗓音污染:工廠運作可能帶來持續噪音,影響居民生活品質,特別是夜間作業時。

水污染:混凝土生產可能產生廢水,若處理不當,可能污染附近水源,影響生態和農業。

2. 交通壓力

重型車輛增加:混凝土配料廠需要大量原材料運輸,可能導致重型車輛頻繁出入,加劇交通擁 堵,並增加道路損壞風險。

交通安全:重型車輛增多可能提升交通事故風險,尤其對行人、騎單車者和學生構成威脅。

3. 社區影響

生活品質下降:工廠帶來的污染和嗓音可能降低居民生活品質。健康風險;長期暴露於粉塵和 廢氣可能導致呼吸系統疾病,增加醫療負擔。

4. 生態破壞

自然環境受損:混凝土配料廠可能破壞周邊自然環境,影響動植物棲息地,降低生物多樣性。 農業影響:若附近有農地,工廠的污染可能影響農作物生長,損害農民生計。

5. 基礎設施壓力

工廠運作可能對水電等基礎設施造成額外壓力,影響居民正常使用。

作為當區的持份者,我們的反對是基於對環境、交通、社區健康和長遠發展的擔憂。規劃處在 審批時需全面評估這些影響,並考慮居民意見,確保項目符合可持續發展原則

本人懇請城市規劃委員會及政府相關部門重視八鄉居民的意見,聆聽居民的聲音,否決上標申請。

簽署: \_ \_ \_ 姓名(正楷): 宋瑞



<sup>12-</sup>MAR-2025 14:09

25-03-12 13:58 FROM- PAT HEUNG RC T-678 P0022/0042 F-544 城 F 观 型 安 貝 冒 秘 音 首応北 用 進 辛 退 333 疏 北 用 収 村 谷 者 13 夜 (傳真: 2877 0245 及 電郵: www. tobpd@pland.gov.hk)

22

### <u>強烈反對元朗石崗丈量約份第114 约地段第573 號餘段及第1710 號</u> 擬議臨時混凝土配料廠連附屬設施,申請編號: A/YL-SK/410 - 申請人提交進一步資料

執事先生 / 女士:

本人是元朗八鄉鄉事委員會委員,就上標計劃改建為混凝土配料廠的提案作出強烈和堅決的反對。

1. 對環境造成影響環境影響

空氣污染:混凝土配料廠可能產生粉塵和廢氣,影響空氣品質,對居民健康造成威脅,尤其是 老人和兒童。

噪音污染:工廠運作可能帶來持續噪音,影響居民生活品質,特別是夜間作業時。

水污染:混凝土生產可能產生廢水,若處理不當,可能污染附近水源,影響生態和農業。

2. 交通壓力

重型車輛增加:混凝土配料廠需要大量原材料運輸,可能導致重型車輛頻繁出入,加劇交通擁 堵,並增加道路損壞風險。

交通安全:重型車輛增多可能提升交通事故風險,尤其對行人、騎單車者和學生構成威脅。

3. 社區影響

生活品質下降:工廠帶來的污染和噪音可能降低居民生活品質。健康風險:長期暴露於粉塵和 廢氣可能導致呼吸系統疾病,增加醫療負擔。

4. 生態破壞

自然環境受損:混凝土配料廠可能破壞問過自然環境,影響動植物棲息地,降低生物多樣性。 農業影響:若附近有農地,工廠的污染可能影響農作物生長,損害農民生計。

5. 基礎設施壓力

工廠運作可能對水電等基礎設施造成額外壓力,影響居民正常使用。

作為當區的持份者,我們的反對是基於對環境、交通、社區健康和長遠發展的擔憂。規劃處在 審批時需全面評估這些影響,並考慮居民意見,確保項目符合可持續發展原則

资晋: 大家里了, 姓名(正楷): 美人) 第一年1

2025年了	月12	_日	
通訊地址:			
聯絡電話:			
		5 2 4 <sup>3</sup>	
12-MAR-2025	14:09		

97%

25-03-12 13:58 FROM- PAT HEUNG RC 1-6/8 P0023/0042 F-544 省港北用准举理 >>> 號北用以府合省 13 接 城平规则安良官秘音 (傳真: 2877 0245 及 電郵: www.tpbpd@pland.gov.hk)

23

### 強烈反對元朗石崗丈量約份第114约地段第573號餘段及第1710號 擬議臨時混凝土配料廠連附屬設施,申請編號: A/YL-SK/410 - 申請人提交進一步資料

#### 執事先生 / 女士:

本人是元朗八鄉鄉事委員會委員,就上標計劃改建為混凝土配料廠的提案作出強烈和堅決 的反對。

1. 對環境造成影響環境影響

空氣污染:混凝土配料廠可能產生粉糜和廢氣,影響空氣品質,對居民健康造成威脅,尤其是 老人和兒童。

嗓音污染:工廠運作可能帶來持續嗓音,影響居民生活品質,特別是夜間作業時。

水污染:混凝土生產可能產生廢水,若處理不當,可能污染附近水源,影響生態和農業。

2. 交通壓力

重型車輛增加:混凝土配料廠需要大量原材料運輸,可能導致重型車輛頻繁出入,加劇交通擁 堵,並增加道路損壞風險。

交通安全:重型車輛增多可能提升交通事故風險,尤其對行人、騎單車者和學生構成威脅。

3. 社區影響

生活品質下降:工廠帶來的污染和噪音可能降低居民生活品質。健康風險:長期暴露於粉塵和 廢氣可能導致呼吸系統疾病,增加醫療負擔。

4. 生態破壞

自然環境受損:混凝土配料廠可能破壞周邊自然環境,影響動植物棲息地,降低生物多樣性。 農業影響:若附近有農地,工廠的污染可能影響農作物生長,損害農民生計。

5. 基礎設施壓力

工廠運作可能對水電等基礎設施造成額外壓力,影響居民正常使用。

作為當區的持份者,我們的反對是基於對環境、交通、社區健康和長遠發展的擴疊。規劃處在 審批時需全面評估這些影響,並考慮居民意見,確保項目符合可持續發展原則

本人懇請城市規劃委員會及政府相關部門重視八鄉居民的意見,聆聽居民的聲音,否決 上標申請。

姓名(正楷): LEE CHI MING

2025 年 3 月 元

通訊地址: 聯絡電話: 25-03-12 13:58 FROM- PAT HEUNG RC T-678 P0024/0042 F-544 城中成副安良盲松盲 首心に用温書を見つつ 派し月以内 古者 10 後 (傳真: 2877 0245 及 電郵:www.tpbpd@pland.gov.hk)

24

3 ...

### <u>強烈反對元朗石崗丈量约份第114 约地段第573 號餘段及第1710 號</u> 擬議臨時混凝土配料廠連附屬設施,申請編號: A/YL-SK/410 - 申請人提交進一步資料

執事先生 / 女士:

本人是元朗八鄉鄉事委員會委員,就上標計劃改建為混凝土配料廠的提案作出強烈和堅決的反對。

1. 對環境造成影響環境影響

空氣污染:混凝土配料廠可能產生粉塵和廢氣,影響空氣品質,對居民健康造成威脅,尤其是 老人和兒童。

嗓音污染:工廠運作可能帶來持續嗓音,影響居民生活品質,特別是夜間作業時。

水污染:混凝土生產可能產生廢水,若處理不當,可能污染附近水源,影響生態和農業。

2. 交通壓力

重型車輛增加:混凝土配料廠需要大量原材料運輸,可能導致重型車輛頻繁出入,加劇交通擁 堵,並增加道路損壞風險。

交通安全:重型車輛增多可能提升交通事故風險,尤其對行人、騎單車者和學生構成威脅。

3. 社區影響

生活品質下降:工廠帶來的污染和噪音可能降低居民生活品質。健康風險:長期暴露於粉塵和 廢氣可能導致呼吸系統疾病,增加醫療負擔。

4. 生態破壞

自然環境受損:混凝土配料廠可能破壞周邊自然環境,影響動植物棲息地,降低生物多樣性。 農業影響:若附近有農地,工廠的污染可能影響農作物生長,損害農民生計。

5. 基礎設施壓力

工廠運作可能對水電等基礎設施造成額外壓力,影響居民正常使用。

作為當區的持份者,我們的反對是基於對環境、交通、社區健康和長遠發展的擔憂。規劃處在 審批時需全面評估這些影響,並考慮居民意見,確保項目符合可持續發展原則

本人懇請城市規劃委員會及政府相關部門重視八鄉居民的意見,聆聽居民的聲音,否決 上標申請。

姓名(正楷): 登 蒋乳 休。

2015年3月12

通訊地址:

聯絡電話:

25-03-12 13:58 FROM- PAT HEUNG RC T-678 P0025/0042 F-544 かないからくますがす すべいしのほすといいかいいしょうないしょうない (得真: 2877 0245 及 電郵:www.tpbpd@pland.gov.hk)

強烈反對元朗石崗丈量約份第114约地段第573號餘段及第1710號

擬議臨時混凝土配料廠連附屬設施,申請編號: A/YL-SK/410

25

#### - 申請人提交進一步資料

執事先生 / 女士:

本人是元朗八鄉鄉事委員會委員,就上標計劃改建為混凝土配料廠的提案作出強烈和堅決的反對。

1. 對環境造成影響環境影響

空氣污染:混凝土配料廠可能產生粉塵和廢氣,影響空氣品質,對居民健康造成威脅,尤其是 老人和兒童。

嗓音污染:工廠運作可能帶來持續嗓音,影響居民生活品質,特別是夜間作業時。

水污染:混凝土生產可能產生廢水,若處理不當,可能污染附近水源,影響生態和農業。

2. 交通壓力

重型車輛增加:混凝土配料廠需要大量原材料運輸,可能導致重型車輛頻繁出入,加劇交通擁 堵,並增加道路損壞風險。

交通安全:重型車輛增多可能提升交通事故風險,尤其對行人、騎軍車者和學生構成威脅。

3、社區影響

生活品質下降:工廠帶來的污染和噪音可能降低居民生活品質。健康風險:長期暴露於粉塵和 廢氣可能導致呼吸系統疾病,增加醫療負擔。

#### 4. 生態破壞

自然環境受損:混凝土配料廠可能破壞周邊自然環境,影響動植物棲息地,降低生物多樣性。 農業影響:若附近有農地,工廠的污染可能影響農作物生長,損害農民生計。

5. 基礎設施壓力

工廠運作可能對水電等基礎設施造成額外壓力,影響居民正常使用。

作為當區的持份者,我們的反對是基於對環境、交通、社區健康和長遠發展的擔憂。規劃處在 審批時需全面評估這些影響,並考慮居民意見,確保項目符合可持續發展原則

本人懇請城市規劃委員會及政府相關部門重視八鄉居民的意見,聆聽居民的聲音,否決上標申請。

簽署: 4 姓名(正楷): \_

2025 年 3月 12 日

通訊地址: 聯絡電話:

<sup>12-</sup>MAR-2025 14:10

(傳真: 2877 0245 及 電郵: www. tpbpd@pland. gov. hk)

強烈反對元朗石崗丈量約份第114约地段第573號餘段及第1710號 擬議臨時混凝土配料廠連附屬設施,申請編號: A/YL-SK/410

- 申請人提交進一步資料

執事先生 / 女士:

本人是元朗八鄉鄉事委員會委員,就上標計劃改建為混凝土配料廠的提案作出強烈和堅決的反對。

1. 對環境造成影響環境影響

空氣污染:混凝土配料廠可能產生粉塵和廢氣,影響空氣品質,對居民健康造成威脅,尤其是 老人和兒童。

噪音污染:工廠運作可能帶來持續噪音,影響居民生活品質,特別是夜間作業時。 水污染:混凝土生產可能產生廢水,若處理不當,可能污染附近水源,影響生態和農業。

2. 交通壓力

重型車輛增加:混凝土配料廠需要大量原材料運輸,可能導致重型車輛頻繁出入,加劇交通擁 堵,並增加道路損壞風險。

交通安全:重型車輛增多可能提升交通事故風險,尤其對行人、騎單車者和學生構成威脅。

3. 社區影響

生活品質下降;工廠帶來的污染和噪音可能降低居民生活品質。健康風險;長期暴露於粉塵和 廢氣可能導致呼吸系統疾病,增加醫療負擔。

4. 生態破壞

自然環境受損: 混凝土配料廠可能破壞周邊自然環境,影響動植物棲息地,降低生物多樣性。 農業影響:若附近有農地,工廠的污染可能影響農作物生長,損害農民生計。

5. 基礎設施壓力

工廠運作可能對水電等基礎設施造成額外壓力,影響居民正常使用。

作為當區的持份者,我們的反對是基於對環境、交通、社區健康和長遠發展的擔憂。規劃處在 審批時需全面評估這些影響,並考慮居民意見,確保項目符合可持續發展原則

本人懇請城市規劃委員會及政府相關部門重視八鄉居民的意見,聆聽居民的聲音,否決上標申請。

簽署:\_\_\_\_\_\_ 姓名(正楷): 第一部 考生

2025年3月12日



 25-03-12 13:59 FROM PAI HEUNG RC
 I-6/8 P002//0042 F-544

 城中規劃安貝賞秘書
 香港北角澄華道 333 號北角政府合著 15 樓

 (傳真: 2877 0245 及 電郵: www.tpbpd@pland.gov.hk)
 27

### <u>強烈反對元朗石崗丈量約份第114 约地段第573 號餘段及第1710 號</u> 擬議臨時混凝土配料廠連附屬設施,申請編號: A/YL-SK/410 - 申請人提交進一步資料

執事先生 / 女士:

本人是元朗八鄉橫台山羅屋村居民,就上標計劃改建為混凝土配料廠的提案作出強烈和堅決的反對。

1. 對環境造成影響環境影響

空氣污染:混凝土配料廠可能產生粉塵和廢氣,影響空氣品質,對居民健康造成威脅,尤其是 老人和兒童。

噪音污染:工廠運作可能帶來持續噪音,影響居民生活品質,特別是夜間作業時。

水污染:混凝土生產可能產生廢水,若處理不當,可能污染附近水源,影響生態和農業。

2. 交通壓力

重型車輛增加:混凝土配料廠需要大量原材料運輸,可能導致重型車輛頻繁出入,加劇交通擁 堵,並增加道路損壞風險。

交通安全:重型車輛增多可能提升交通事故風險,尤其對行人、騎單車者和學生構成威脅。

3. 社區影響

生活品質下降:工廠帶來的污染和噪音可能降低居民生活品質。健康風險:長期暴露於粉塵和 廢氣可能導致呼吸系統疾病,增加醫療負擔。

4. 生態破壞

自然環境受損:混凝土配料廠可能破壞周邊自然環境,影響動植物棲息地,降低生物多樣性。 農業影響:若附近有農地,工廠的污染可能影響農作物生長,損害農民生計。

5. 基礎設施壓力

工廠運作可能對水電等基礎設施造成額外壓力,影響居民正常使用。

作為當區的持份者,我們的反對是基於對環境、交通、社區健康和長遠發展的擔憂。規劃處在 審批時需全面評估這些影響,並考慮居民意見,確保項目符合可持續發展原則

本人懇請城市規劃委員會及政府相關部門重視八鄉居民的意見, 聆聽居民的聲音, 否決上標申請。

姓名(正楷):

2025年_	3	月	12	8			
通訊地址	:						
聯絡電話							
			1.1		-	λ.	

1.00

25-03-12 13:59 FROM- PAT HEUNG RC
 工-678 P0028/0042 F-544
 城中規劃委員曾秘書
 香港北角渣華道 333 號北角政府合署 15 樓
 (傳真: 2877 0245 及 電郵: www.tpbpd@pland.gov.hk)

. 28

### <u>強烈反對元朗石崗丈量約份第114 约地段第573 號餘段及第1710 號</u> 擬議臨時混凝土配料廠連附屬設施,申請編號: A/YL-SK/410 - 申請人提交進一步資料

執事先生 / 女士:

本人是元朗八鄉橫台山羅屋村居民,就上標計劃改建為混凝土配料廠的提案作出強烈和堅決的反對。

1. 對環境造成影響環境影響

空氣污染:混凝土配料廠可能產生粉塵和廢氣,影響空氣品質,對居民健康造成威脅,尤其是 老人和兒童。

嗓音污染:工廠運作可能帶來持續噪音,影響居民生活品質,特別是夜間作業時。

水污染:混凝土生產可能產生廢水,若處理不當,可能污染附近水源,影響生態和農業。

2. 交通壓力

重型車輛增加:混凝土配料廠需要大量原材料運輸,可能導致重型車輛頻繁出入,加劇交通擁 堵,並增加道路損壞風險。

交通安全:重型車輛增多可能提升交通事故風險,尤其對行人、騎單車者和學生構成威脅。

3. 社區影響

生活品質下降:工廠帶來的污染和噪音可能降低居民生活品質。健康風險:長期暴露於粉塵和 廢氣可能導致呼吸系統疾病,增加醫療負擔。

4. 生態破壞

自然環境受損:混凝土配料廠可能破壞周邊自然環境,影響動植物棲息地,降低生物多樣性。 農業影響:若附近有農地,工廠的污染可能影響農作物生長,損害農民生計。

5. 基礎設施壓力

工廠運作可能對水電等基礎設施造成額外壓力,影響居民正當使用。

作為當區的持份者,我們的反對是基於對環境、交通、社區健康和長遠發展的擔憂。規劃處在 審批時需全面評估這些影響,並考慮居民意見,確保項目符合可持續發展原則

姓名(正楷);

2015年_	3月	12 5	}	
通訊地址	:			
聯絡電話	;			

 '25-03-12 13:59 FROM- PAT HEUNG RC
 「一678 P0029/0042 F-544 城市規劃要員會秘書
 「傳真: 2877 0245 及 富郵:www.tobod@pland.gov.hk)

23

### <u>強烈反對元朗石崗丈量約份第114</u>约地段第573號餘段及第1710號 擬議臨時混凝土配料廠連附屬設施,申請編號: A/YL-SK/410 - 申請人提交進一步資料

執事先生 / 女士:

本人是元朗八鄉橫台山羅屋村居民,就上標計劃改建為混凝土配料廠的提案作出強烈和堅決的反對。

1. 對環境造成影響環境影響

空氣污染:混凝土配料廠可能產生粉塵和廢氣,影響空氣品質,對居民健康造成威脅,尤其是 老人和兒童。

噪音污染:工廠運作可能帶來持續噪音,影響居民生活品質,特別是夜間作業時。

水污染:混凝土生產可能產生廢水,若處理不當,可能污染附近水源,影響生態和農業。

2. 交通壓力

重型車輛增加:混凝土配料廠需要大量原材料運輸,可能導致重型車輛頻繁出入,加劇交通擁 堵,並增加道路損壞風險。

交通安全:重型車輛增多可能提升交通事故風險,尤其對行人、騎單車者和學生構成威脅。

3. 社區影響

生活品質下降:工廠帶來的污染和噪音可能降低居民生活品質。健康風險:長期暴露於粉塵和 廢氣可能導致呼吸系統疾病,增加醫療負擔。

4. 生態破壞

自然環境受損:混凝土配料廠可能破壞周邊自然環境,影響動植物棲息地,降低生物多樣性。 農業影響:若附近有農地,工廠的污染可能影響農作物生長,損害農民生計。

5. 基礎設施壓力

工廠運作可能對水電等基礎設施造成額外壓力,影響居民正常使用。

作為當區的持份者,我們的反對是基於對環境、交通、社區健康和長遠發展的搪臺。規劃處在 審批時需全面評估這些影響,並考慮居民意見,確保項目符合可持續發展原則

本人懇請城市規劃委員會及政府相關部門重視八鄉居民的意見,聆聽居民的聲音,否決上標申請。



2025年了月12日

通訊地址:

 '25-03-12 13:59 FROM PAT HEUNG RC
 T-678 P0030/0042 F-544

 城市規劃委員會秘書
 香港北角渣華道 333 號北角政府合署 15 樓

 (傳真: 2877 0245 及 電郵: www.tpbpd@pland.gov.hk)
 30

<u>強烈反對元朗石崗丈量約份第114 约地段第573 號餘段及第1710 號</u> 擬議臨時混凝土配料廠連附屬設施,申請編號: A/YL-SK/410

#### - 申請人提交進一步資料

執事先生 / 女士:

本人是元朗八鄉橫台山羅屋村居民,就上標計劃改建為混凝土配料廠的提案作出強烈和堅決的反對。

1. 對環境造成影響環境影響

空氣污染:混凝土配料廠可能產生粉塵和廢氣,影響空氣品質,對居民健康造成威脅,尤其是 老人和兒童。

嗓音污染;工廠運作可能帶來持續嗓音,影響居民生活品質,特別是夜間作業時。

水污染:混凝土生產可能產生廢水,若處理不當,可能污染附近水源,影響生態和農業。

2. 交通壓力

重型車輛增加:混凝土配料廠需要大量原材料運輸,可能導致重型車輛頻繁出入,加劇交通擁 堵,並增加道路損壞風險。

交通安全:重型車輛增多可能提升交通事故風險,尤其對行人、騎單車者和學生構成威脅。

3. 社區影響

生活品質下降:工廠帶來的污染和噪音可能降低居民生活品質。健康風險:長期暴露於粉塵和 廢氣可能導致呼吸系統疾病,增加醫療員擔。

4. 生態破壞

自然環境受損:混凝土配料廠可能破壞周邊自然環境,影響動植物棲息地,降低生物多樣性。 農業影響:若附近有農地,工廠的污染可能影響農作物生長,損害農民生計。

5. 基礎設施壓力

工廠運作可能對水電等基礎設施造成額外壓力,影響居民正常使用。

作為當區的持份者,我們的反對是基於對環境、交通、社區健康和長遠發展的擔憂。規劃處在 審批時需全面評估這些影響,並考慮居民意見,確保項目符合可持續發展原則

2025年了)	9 <u>12</u> E	3	
通訊地址:			
聯絡電話:			
50 - 10 - 10 - 10 - 10 - 10 - 10 - 10 -			
12-MAR-2025	14:11		

 '25-03-12 13:59 FROM- PAT HEUNG RC
 T-678 P0031/0042 F-544

 城市規劃委員會秘書
 香港北角渣華道 333 號北角政府合署 15 樓

(傳真 : 2877 0245 及 電郵:www.tpbpd@pland.gov.hk)

• • 31

強烈反對元朗石崗丈量約份第114约地段第573號餘段及第1710號 擬議臨時混凝土配料廠連附屬設施,申請編號: A/YL-SK/410

#### - 申請人提交進一步資料

執事先生 / 女士:

本人是元朗八鄉橫台山羅屋村居民,就上標計劃改建為混凝土配料廠的提案作出強烈和堅決的反對。

1. 對環境造成影響環境影響

空氣污染:混凝土配料廠可能產生粉塵和廢氣,影響空氣品質,對居民健康造成威脅,尤其是 老人和兒童。

噪音污染︰工廠運作可能帶來持續噪音,影響居民生活品質,特別是夜間作業時。

水污染:混凝土生產可能產生廢水,若處理不當,可能污染附近水源,影響生態和農業。

2. 交通壓力

重型車輛增加:混凝土配料廠需要大量原材料運輸,可能導致重型車輛頻繁出入,加劇交通擁 堵,並增加道路損壞風險。

交通安全:重型車輛增多可能提升交通事故風險,尤其對行人、騎單車者和學生構成威脅。

3. 社區影響

生活品質下降:工廠帶來的污染和噪音可能降低居民生活品質。健康風險:長期暴露於粉塵和廢氣可能導致呼吸系統疾病,增加醫療負擔。

4. 生態破壞

自然環境受損:混凝土配料廠可能破壞周邊自然環境,影響動植物棲息地,降低生物多樣性。 農業影響:若附近有農地,工廠的污染可能影響農作物生長,損害農民生計。

5. 基礎設施壓力

工廠運作可能對水電等基礎設施造成額外壓力,影響居民正常使用。

作為當區的持份者,我們的反對是基於對環境、交通、社區健康和長遠發展的擔憂。規劃處在審批時需全面評估這些影響,並考慮居民意見,確保項目符合可持續發展原則

本人懇請城市規劃委員會及政府相關部門重視八鄉居民的意見,聆聽居民的聲音,否決上標申請。

Anf--Rma 姓名(正楷):

2025年3月	12日	
通訊地址:		
聯絡電話:		
12-MAR-2025	14:11	

 25-03-12 13:59 FROM PAT HEUNG RC
 T-678 P0032/0042 F-544

 城市規劃委員會秘書
 香港北角渣華道 333 號北角政府合署 15 樓

 (傳真: 2877 0245 及 電郵:www.tpbpd@pland.gov.hk)
 32

<u>強烈反對元朗石崗丈量约份第114 约地段第573 號餘段及第1710 號</u> 擬議臨時混凝土配料廠連附屬設施,申請編號: A/YL-SK/410 - 申請人提交進一步資料

執事先生 / 女士:

本人是元朗八鄉橫台山羅屋村居民,就上標計劃改建為混凝土配料廠的提案作出強烈和堅決的反對。

1. 對環境造成影響環境影響

空氣污染:混凝土配料廠可能產生粉塵和廢氣,影響空氣品質,對居民健康造成威脅,尤其是 老人和兒童。

嗓音污染:工廠運作可能帶來持續嗓音,影響居民生活品質,特別是夜間作業時。

水污染:混凝土生產可能產生廢水,若處理不當,可能污染附近水源,影響生態和農業。

2. 交通壓力

重型車輛增加:混凝土配料廠需要大量原材料運輸,可能導致重型車輛頻繁出入,加劇交通擁 堵,並增加道路損壞風險。

交通安全:重型車輛增多可能提升交通事故風險,尤其對行人、騎單車者和學生構成威脅。

3. 社區影響

生活品質下降:工廠帶來的污染和噪音可能降低居民生活品質。健康風險:長期暴露於粉塵和 廢氣可能導致呼吸系統疾病,增加醫療負擔。

4. 生態破壞

自然環境受損:混凝土配料廠可能破壞周邊自然環境,影響動植物棲息地,降低生物多樣性。 農業影響:若附近有農地,工廠的污染可能影響農作物生長,損害農民生計。

5. 基礎設施壓力

工廠運作可能對水電等基礎設施造成額外壓力,影響居民正常使用。

作為當區的持份者,我們的反對是基於對環境、交通、社區健康和長遠發展的擔憂。規劃處在 審批時需全面評估這些影響,並考慮居民意見,確保項目符合可持續發展原則

本人懇請城市規劃委員會及政府相關部門重視八鄉居民的意見,聆聽居民的聲音,否決上標申請。

簽署: <u>华</u> 姓名(正楷): <u>那</u> 围华

2025年_3	月_12日	Ę		
通訊地址:				
聯絡電話:				
12-MAR-2025	14:11			
25-03-12 14:00 FROM PAT HEUNG RC
 T-678 P0033/0042 F-544

 城市規劃委員會秘書
 香港北角渣莓道 333 號北角政府合署 15 樓

 (傳真: 2877 0245 及 電郵:www.tpbpd@pland.gov.hk)
 33

<u>強烈反對元朗石崗丈量约份第114 约地段第573 號餘段及第1710 號</u> 擬議臨時混凝土配料廠連附屬設施,申請編號: A/YL-SK/410 - 申請人提交進一步資料

#### 執事先生 / 女士:

本人是元朗八鄉橫台山羅屋村居民,就上標計劃改建為混凝土配料廠的提案作出強烈和堅決的反對。

1. 對環境造成影響環境影響

空氣污染:混凝土配料廠可能產生粉塵和廢氣,影響空氣品質,對居民健康造成威脅,尤其是 老人和兒童。

噪音污染:工廠運作可能帶來持續噪音,影響居民生活品質,特別是夜間作業時。

水污染:混凝土生產可能產生廢水,若處理不當,可能污染附近水源,影響生態和農業。

2. 交通壓力

重型車輛增加:混凝土配料廠需要大量原材料運輸,可能導致重型車輛頻繁出入,加劇交通擁 堵,並增加道路損壞風險。

交通安全:重型車輛增多可能提升交通事故風險,尤其對行人、騎單車者和學生構成威脅。

3. 社區影響

生活品質下降:工廠帶來的污染和噪音可能降低居民生活品質。健康風險:長期暴露於粉塵和 廢氣可能導致呼吸系統疾病,增加醫療負擔。

4. 生態破壞

自然環境受損:混凝土配料廠可能破壞周邊自然環境,影響動植物棲息地,降低生物多樣性。 農業影響:若附近有農地,工廠的污染可能影響農作物生長,損害農民生計。

5. 基礎設施壓力

工廠運作可能對水電等基礎設施造成額外壓力,影響居民正常使用。

作為當區的持份者,我們的反對是基於對環境、交通、社區健康和長遠發展的擔憂。規劃處在審批時需全面評估這些影響,並考慮居民意見,確保項目符合可持續發展原則

本人懇請城市規劃委員會及政府相關部門重視八鄉居民的意見,聆聽居民的聲音,否決 上標申請。

簽署:\_\_\_\_\_ 姓名(正楷):

12-MAR-2025 14:11

2025年3月12日

通訊地址:

聯絡電話:

·25-03-12 14:00 FROM- PAT HEUNG RC I-6/8 PUU34/UU42 F-544 城市規劃委員會秘書 香港北角渣華道 333 號北角政府合署 15 樓 (傳真: 2877 0245 及 電郵: www.tpbpd@pland.gov.hk)

34

1. 10

#### <u>強烈反對元朗石崗丈量約份第114 约地段第573 號餘段及第1710 號</u> 擬議臨時混凝土配料廠速附屬設施,申請編號: A/YL-SK/410 - 申請人提交進一步資料

執事先生 / 女士:

本人是元朗八鄉橫台山羅屋村居民,就上標計劃改建為混凝土配料廠的提案作出強烈和堅決 的反對。

1. 對環境造成影響環境影響

空氣污染:混凝土配料廠可能產生粉塵和廢氣,影響空氣品質,對居民健康造成威脅,尤其是 老人和兒童。

嗓音污染:工廠運作可能帶來持續噪音,影響居民生活品質,特別是夜間作業時。 水污染:混凝上生產可能產生廢水,若處理不當,可能污染附近水源,影響生態和農業。

2. 交通壓力

重型車輛增加:混凝土配料廠需要大量原材料運輸,可能導致重型車輛頻繁出入,加劇交通擁 堵,並增加道路損壞風險。

交通安全:重型車輛增多可能提升交通事故風險,尤其對行人、騎單車者和學生構成威脅。

3. 社區影響

生活品質下降:工廠帶來的污染和嗓音可能降低居民生活品質。健康風險;長期暴露於粉塵和 廢氣可能導致呼吸系統疾病,增加醫療負擔。

#### 4. 生態破壞

自然環境受損:混凝土配料廠可能破壞周邊自然環境,影響動植物棲息地,降低生物多樣性。 農業影響:若附近有農地,工廠的污染可能影響農作物生長,損害農民生計。

5. 基礎設施壓力

工廠運作可能對水電等基礎設施造成額外壓力,影響居民正常使用。

作為當區的持份者,我們的反對是基於對環境、交通、社區健康和長遠發展的擔憂。規劃處在 審批時需全面評估這些影響,並考慮居民意見,確保項目符合可持續發展原則

本人懸請城市規劃委員會及政府相關部門重視八鄉居民的意見,聆聽居民的聲音,否決上標申請。

姓名(正楷):



聯絡電話:

通訊地址:

 25-03-12 14:00 FROM- PAT HEUNG RC
 1-6/8 P0035/0042 F-544

 城市規劃委員會秘書
 香港北角渣華道 333 號北角政府含署 15 樓

 (傳真: 2877 0245 及 電郵:www.tpbpd@pland.gov.hk)
 35

#### <u>強烈反對元朗石崗丈量約份第114 约地段第573 號餘段及第1710 號</u> 擬議臨時混凝土配料廠連附屬設施,申請編號: A/YL-SK/410 - 申請人提交進一步資料

執事先生 / 女士:

本人是元朗八鄉橫台山羅屋村居民,就上標計劃改建為混凝土配料廠的提案作出強烈和堅決 的反對。

1. 對環境造成影響環境影響

空氣污染:混凝土配料廠可能產生粉壓和廢氣,影響空氣品質,對居民健康造成威脅,尤其是 老人和兒童。

噪音污染:工廠運作可能帶來持續噪音,影響居民生活品質,特別是夜間作業時。

水污染:混凝土生產可能產生廢水,若處理不當,可能污染附近水源,影響生態和農業。

2. 交通壓力

重型車輛增加:混凝土配料廠需要大量原材料運輸,可能導致重型車輛頻繁出入,加劇交通擁 堵,並增加道路損壞風險。

交通安全:重型車輛增多可能提升交通事故風險,尤其對行人、騎單車者和學生構成威脅。

3. 社區影響

生活品質下降:工廠帶來的污染和噪音可能降低居民生活品質。健康風險:長期暴露於紛塵和 廢氣可能導致呼吸系統疾病,增加醫療負擔。

4. 生態破壞

自然環境受損:混凝土配料廠可能破壞周邊自然環境,影響動植物棲息地,降低生物多樣性。 農業影響:若附近有農地,工廠的污染可能影響農作物生長,損害農民生計。

5. 基礎設施壓力

工廠運作可能對水電等基礎設施造成額外壓力,影響居民正常使用。

作為當區的持份者,我們的反對是基於對環境、交通、社區健康和長遠發展的擔憂。規劃處在 審批時需全面評估這些影響,並考慮居民意見,確保項目符合可持續發展原則

本人懇請城市規劃委員會及政府相關部門重視八鄉居民的意見,聆聽居民的聲音,否決上標申請。

簽署:



2025年3月12

通訊地址

聯絡電話

A

 '25-03-12 14:00 FROM- PAT HEUNG RC
 I-6/8 P0036/0042 F-544

 城市規劃委員會秘書
 香港北角渣華道 333 號北角政府合署 15 樓

 (傳真: 2877 0245 及:電郵:www.tpbpd@pland.gov.hk)
 36

#### <u>強烈反對元朗石崗丈量約份第114 约地段第573 號餘段及第1710 號</u> 擬議臨時混凝土配料廠連附屬設施,申請編號: A/YL-SK/410 - 申請人提交進一步資料

執事先生 / 女士:

本人是元朗八鄉橫台山羅屋村居民,就上標計劃改建為混凝土配料廠的提案作出強烈和堅決 的反對。

1. 對環境造成影響環境影響

空氣污染:混凝土配料廠可能產生粉塵和廢氣,影響空氣品質,對居民健康造成威脅,尤其是 老人和兒童。

噪音污染:工廠運作可能帶來持續噪音,影響居民生活品質,特別是夜間作業時。 水污染:混凝土生產可能產生廢水,若處理不當,可能污染附近水源,影響生態和農業。

2. 交通壓力

重型車輛增加:混凝土配料廠需要大量原材料運輸,可能導致重型車輛頻繁出入,加劇交通擁 堵,並增加道路損壞風險。

交通安全:重型車輛增多可能提升交通事故風險,尤其對行人、騎單車者和學生構成威脅。

3. 社區影響

生活品質下降:工廠帶來的污染和噪音可能降低居民生活品質。健康風險:長期暴露於粉塵和 廢氣可能導致呼吸系統疾病,增加醫療負擔。

4. 生態破壞

自然環境受損: 混凝土配料廠可能破壞周邊自然環境,影響動植物棲息地,降低生物多樣性。 農業影響:若附近有農地,工廠的污染可能影響農作物生長,損害農民生計。

5. 基礎設施壓力

工廠運作可能對水電等基礎設施造成額外壓力,影響居民正常使用。

作為當區的持份者,我們的反對是基於對環境、交通、社區健康和長遠發展的擔憂。規劃處在 審批時需全面評估這些影響,並考慮居民意見,確保項目符合可持續發展原則

簽署: 一座 花



 25-03-12 14:00 FROM PAT HEUNG RC
 I-6/8 P003//0042 F-544

 城市規劃委員會秘書
 香港北角渣華道 333 號北角政府合署 15 樓

 (傳真: 2877 0245 及 電郵: www.tpbpd@pland.gov.hk)
 37

#### <u>強烈反對元朗石崗丈量約份第114约地段第573號餘段及第1710號</u> 擬議臨時混凝土配料廠連附屬設施,申請編號: A/YL~SK/410 - 申請人提交進一步資料

#### 執事先生 / 女士:

本人是元朗八鄉橫台山羅屋村居民,就上標計劃改建為混凝土配料廠的提案作出強烈和堅決的反對。

1. 對環境造成影響環境影響

空氣污染:混凝土配料廠可能產生粉塵和廢氣,影響空氣品質,對居民健康造成威脅,尤其是 老人和兒童。

噪音污染:工廠運作可能帶來持續噪音,影響居民生活品質,特別是夜間作業時。

水污染:混凝土生產可能產生廢水,若處理不當,可能污染附近水源,影響生態和農業。

2. 交通壓力

重型車輛增加:混凝土配料廠需要大量原材料運輸,可能導致重型車輛頻繁出入,加劇交通擁 堵,並增加道路損壞風險。

交通安全:重型車輛增多可能提升交通事故風險,尤其對行人、騎單車者和學生構成威脅。

3. 社區影響

生活品質下降:工廠帶來的污染和噪音可能降低居民生活品質。健康風險:長期暴露於粉塵和 廢氣可能導致呼吸系統疾病,增加醫療負擔。

4. 生態破壞

自然環境受損: 混凝土配料廠可能破壞周邊自然環境,影響動植物棲息地,降低生物多樣性。 農業影響:若附近有農地,工廠的污染可能影響農作物生長,損害農民生計。

5. 基礎設施壓力

工廠運作可能對水電等基礎設施造成額外壓力,影響居民正常使用。

作為當區的持份者,我們的反對是基於對環境、交通、社區健康和長遠發展的擔憂。規劃處在 審批時需全面評估這些影響,並考慮居民意見,確保項目符合可持續發展原則

本人懇請城市規劃委員會及政府相關部門重視八鄉居民的意見,聆聽居民的聲音,否決上標申請。

簽署: <u>文正</u> 姓名(正楷): <u><u></u>, <u>文正</u></u>



 25-03-12 14:00 FROM- PAT HEUNG RC
 1-6/8 P0038/0042 F-544

 城市規劃委員會秘書
 香港北角渣華道 333 號北角政府合署 15 樓

 (傳真: 2877 0245 及 電郵:www.tpbpd@pland.gov.hk)
 33

#### <u>強烈反對元朗石崗丈量約份第114 约地段第573 號餘段及第1710 號</u> 擬議臨時混凝土配料廠連附屬設施,申請編號: A/YL-SK/410 - 申請人提交進一步資料

執事先生 / 女士:

本人是元朗八鄉橫台山羅屋村居民,就上標計劃改建為混凝土配料廠的提案作出強烈和堅決 的反對。

1. 封環境造成影響環境影響

空氣污染:混凝土配料廠可能產生粉塵和廢氣,影響空氣品質,對居民健康造成威脅,尤其是 老人和兒童。

嗓音污染:工廠運作可能帶來持續嗓音,影響居民生活品質,特別是夜間作業時。

水污染:混凝土生產可能產生廢水,若處理不當,可能污染附近水源,影響生態和農業。

2、交通壓力

重型車輛增加:混凝土配料廠需要大量原材料運輸,可能導致重型車輛頻繁出入,加劇交通擁 堵,並增加道路損壞風險。

交通安全:重型車輛增多可能提升交通事故風險,尤其對行人、騎單車者和學生構成威脅。

3. 社區影響

生活品質下降:工廠帶來的污染和噪音可能降低居民生活品質。健康風險:長期暴露於粉塵和 廢氣可能導致呼吸系統疾病,增加醫療負擔。

4. 生態破壞

自然環境受損: 混凝土配料廠可能破壞周邊自然環境,影響動植物棲息地,降低生物多樣性。 農業影響:若附近有農地,工廠的污染可能影響農作物生長,損害農民生計。

5. 基礎設施壓力

工廠運作可能對水電等基礎設施造成額外壓力,影響居民正常使用。

作為當區的持份者,我們的反對是基於對環境、交通、社區健康和長遠發展的擔憂。規劃處在 審批時需全面評估這些影響,並考慮居民意見,確保項目符合可持續發展原則

本人懇請城市規劃委員會及政府相關部門重視八鄉居民的意見,聆聽居民的聲音,否決 上標申請。

簽署:\_\_\_\_\_ 姓名(正楷): 霸國昌



 25-03-12 14:01 FROM- PAT HEUNG RC
 T-678 P0039/0042 F-544

 城市規劃委員會秘書
 香港北角渣華道 333 號北角政府合署 15 樓

 (傳真: 2877 0245 及 電郵: www.tpbpd@pland.gov.hk)
 20

···· 33

## <u>強烈反對元朗石崗丈量约份第114</u>约地段第573號餘段及第1710號 擬議臨時混凝土配料廠連附屬設施,申請編號: A/YL-SK/410 - 申請人提交進一步資料

執事先生 / 女士:

本人是元朗八鄉橫台山羅屋村居民,就上標計劃改建為混凝土配料廠的提案作出強烈和堅決的反對。

1. 對環境造成影響環境影響

空氣污染:混凝土配料廠可能產生粉塵和廢氣,影響空氣品質,對居民健康造成威脅,尤其是 老人和兒童。

嗓音污染:工廠運作可能帶來持續嗓音,影響居民生活品質,特別是夜間作業時。

水污染:混凝土生產可能產生廢水,若處理不當,可能污染附近水源,影響生態和農業。

2. 交通壓力

重型車輛增加:混凝土配料廠 需要大量原材料運輸,可能導致重型車輛頻繁出入,加劇交通擁 堵,並增加道路損壞風險。

交通安全:重型車輛增多可能提升交通事故風險,尤其對行人、騎單車者和學生構成威脅。

3. 社區影響

生活品質下降:工廠帶來的污染和噪音可能降低居民生活品質。健康風險:長期暴露於粉塵和 廢氣可能導致呼吸系統疾病,增加醫療負擔。

4. 生態破壞

自然環境受損:混凝土配料廠可能破壞周邊自然環境,影響動植物棲息地,降低生物多樣性。 農業影響:若附近有農地,工廠的污染可能影響農作物生長,損害農民生計。

5. 基礎設施壓力

工廠運作可能對水電等基礎設施造成額外壓力,影響居民正常使用。

作為當區的持份者,我們的反對是基於對環境、交通、社區健康和長遠發展的擔憂。規劃處在審批時需全面評估這些影響,並考慮居民意見,確保項目符合可持續發展原則

本人懇請城市規劃委員會及政府相關部門重視八鄉居民的意見,聆聽居民的聲音,否決上標申請。

簽署: つ 姓名(正楷): 死世昌

2025年_5	月12日	8	1.5
通訊地址:			
聯絡電話:			
12-MAR-202	5 14:12		

1-010 100-10/00 -----25-03-12 14:01 FROM- PAT HEUNG RC 杳港北角澄樂道 555 號北角政府合看 15 樓 城市規劃委員曾松菁 (傳真 : 2877 0245 及 電郵 : www.tpbpd@pland.gov.hk) 1. 10

40

## 強烈反對元朗石崗丈量约份第114约地段第573號餘段及第1710號 擬議臨時混凝土配料廠連附屬設施,申請編號: A/YL-SK/410 - 申請人提交進一步資料

執事先生 / 女士:

本人是元朗八鄉橫台山羅屋村居民,就上標計劃改建為混凝土配料廠的提案作出強烈和堅決 的反對。

1. 對環境造成影響環境影響

空氣污染:混凝土配料廠可能產生粉歷和廢氣,影響空氣品質,對居民健康造成威脅,尤其是 老人和兒童。

嗓音污染:工廠運作可能帶來持續嗓音,影響居民生活品質,特別是夜間作業時。

水污染:混凝土生產可能產生廢水,若處理不當,可能污染附近水源,影響生態和農業。

2. 交通壓力

重型車輛增加:混凝土配料廠需要大量原材料運輸,可能導致重型車輛頻繁出入,加劇交通擁 堵,並增加道路損壞風險。 交通安全:重型車輛增多可能提升交通事故風險,尤其對行人、騎單車者和學生構成威脅。

3. 社區影響

生活品質下降:工廠帶來的污染和噪音可能降低居民生活品質。健康風險:長期暴露於粉塵和 廢氣可能導致呼吸系統疾病,增加醫療負擔。

4. 生態破壞

自然環境受損:混凝土配料廠可能破壞周邊自然環境,影響動植物棲息地,降低生物多樣性。 農業影響:若附近有農地,工廠的污染可能影響農作物生長,損害農民生計。

5 基礎設施壓力

工廠運作可能對水電等基礎設施造成額外壓力,影響居民正常使用。

作為當區的持份者,我們的反對是基於對環境、交通、社區健康和長遠發展的擔憂。規劃處在 審批時黨全面評估這些影響,並考慮居民意見,確保項目符合可持續發展原則

本人懇請城市規劃委員會及政府相關部門重視八鄉居民的意見,聆聽居民的聲音,否決 上標申請。

簽署: <u>Temen</u> 姓名(正楷): <u>第 浩</u> 号

2025年3月12日

诵訊地址:

聯絡電話:

25-03-12 14:01 FRUM- PAI HEUNG RC 城甲規劃安貝官秘書 百啓北用渲华返 333 號北用政府含者 13 禄 (傳真: 2877 0245 及 電郵:www.tpbpd@pland.gov.hk)

41

3.9

## <u>強烈反對元朗石崗丈量約份第114 约地段第573 號餘段及第1710 號</u> 擬議臨時混凝土配料廠連附屬設施,申請編號: A/YL-SK/410 - 申請人提交進一步資料

執事先生 / 女士:

本人是元朗八鄉橫台山羅屋村居民,就上標計劃改建為混凝土配料廠的提案作出強烈和堅決的反對。

1. 對環境造成影響環境影響

空氣污染:混凝土配料廠可能產生粉塵和廢氣,影響空氣品質,對居民健康造成威脅,尤其是 老人和兒童。

噪音污染:工廠運作可能帶來持續噪音,影響居民生活品質,特別是夜間作業時。

水污染:混凝土生產可能產生廢水,若處理不當,可能污染附近水源,影響生態和農業。

2. 交通壓力

重型車輛增加:混凝土配料廠需要大量原材料運輸,可能導致重型車輛頻繁出入,加劇交通擁 堵,並增加道路損壞風險。

交通安全:重型車輛增多可能提升交通事故風險,尤其對行人、騎單車者和學生構成威脅。

3. 社區影響

生活品質下降:工廠帶來的污染和噪音可能降低居民生活品質。健康風險:長期暴露於粉塵和 廢氣可能導致呼吸系統疾病,增加醫療負擔。

4. 生態破壞

自然環境受損:混凝土配料廠可能破壞周邊自然環境,影響動植物棲息地,降低生物多樣性。 農業影響:若附近有農地,工廠的污染可能影響農作物生長,損害農民生計。

5. 基礎設施壓力

工廠運作可能對水電等基礎設施造成額外壓力,影響居民正常使用。

作為當區的持份者,我們的反對是基於對環境、交通、社區健康和長遠發展的擔憂。規劃處在 審批時需全面評估這些影響,並考慮居民意見,確保項目符合可持續發展原則

本人懇請城市規劃委員會及政府相關部門重視八鄉居民的意見,聆聽居民的聲音,否決上標申請。

簽署: <u>Ann Lann</u> 姓名(正楷): <u>第2 修 号</u>

通訊地址:

2025年3月12日

聯絡電話:

25-03-12 14:01 FRUM- PAI HEUNG RC 水中水回安只留加雪 省心工用進軍建 333 號工用政府合省 13 優 (傳真: 2877 0245 及 電郵:www.tpbpd@pland.gov.hk)

42

### <u>強烈反對元朗石崗丈量约份第114 约地段第573 號餘段及第1710 號</u> 擬議臨時混凝土配料廠連附屬設施,申請編號: A/YL-SK/410 - 申請人提交進一步資料

執事先生 / 女士:

本人是元朗八鄉橫台山羅屋村居民,就上標計劃改建為混凝土配料廠的提案作出強烈和堅決的反對。

1. 對環境造成影響環境影響

空氣污染:混凝土配料廠可能產生粉塵和廢氣,影響空氣品質,對居民健康造成威脅,尤其是 老人和兒童。

噪音污染:工廠運作可能帶來持續噪音,影響居民生活品質,特別是夜間作業時。

水污染:混凝土生產可能產生廢水,若處理不當,可能污染附近水源,影響生態和農業。

2. 交通壓力

重型車輛增加:混凝土配料廠需要大量原材料運輸,可能导致重型車輛頻繁出入,加劇交通擁 堵,並增加道路損壞風險。

交通安全:重型車輛增多可能提升交通事故風險,尤其對行人、騎單車者和學生構成威脅。

3. 社區影響

生活品質下降:工廠帶來的污染和噪音可能降低居民生活品質。健康風險:長期暴露於粉塵和 廢氣可能導致呼吸系統疾病,增加醫療負擔。

4. 生態破壞

自然環境受損:混凝土配料廠可能破壞周邊自然環境,影響動植物棲息地,降低生物多樣性。 農業影響:若附近有農地,工廠的污染可能影響農作物生長,損害農民生計。

5. 基礎設施壓力

工廠運作可能對水電等基礎設施造成額外壓力,影響居民正常使用。

作為當區的持份者,我們的反對是基於對環境、交通、社區健康和長遠發展的擔憂。規劃處在審批時需全面評估這些影響,並考慮居民意見,確保項目符合可持續發展原則

本人懇請城市規劃委員會及政府相關部門重視八鄉居民的意見,聆聽居民的聲音,否決上標申請。

簽署: <u>大</u>

2025年3月12日

通訊地址

25-03-15 09:37 FROM- PAT HEUNG RC T-680 P0001/0010 F-547 城中祝副安貝官秘書 首応北月進挙追ここ 疏北月政府合省 に 使 (傳真: 2877 0245 及 電郵:www.tpbpd@pland.gov.hk)

51

強烈反對元朗石崗丈量约份第114约地段第573號餘段及第1710號

#### 擬議臨時混凝土配料廠連附屬設施,申請編號: A/YL-SK/410 - 申請人提交進一步資料

執事先生 / 女士:

本人是元朗八鄉上村居民,就上標計劃改建為混凝土配料廠的提案作出強烈和堅決的反對。

1. 對環境造成影響環境影響

空氣污染:混凝土配料廠可能產生粉塵和廢氣,影響空氣品質,對居民健康造成威脅,尤其是 老人和兒童。

噪音污染:工廠運作可能帶來持續噪音,影響居民生活品質,特別是夜間作業時。

水污染:混凝土生產可能產生廢水,若處理不當,可能污染附近水源,影響生態和農業。

2. 交通壓力

重型車輛增加:混凝土配料廠需要大量原材料運輸,可能導致重型車輛頻繁出入,加劇交通擁 堵,並增加道路損壞風險。

交通安全:重型車輛增多可能提升交通事故風險,尤其對行人、騎單車者和學生構成威脅。

3. 社區影響

生活品質下降:工廠帶來的污染和噪音可能降低居民生活品質。健康風險:長期暴露於粉塵和 廢氣可能導致呼吸系統疾病,增加醫療負擔。

4. 生態破壞

自然環境受損:混凝土配料廠可能破壞周邊自然環境,影響動植物棲息地,降低生物多樣性。 農業影響:若附近有農地,工廠的污染可能影響農作物生長,損害農民生計。

5. 基礎設施壓力

工廠運作可能對水電等基礎設施造成額外壓力,影響居民正常使用。

作為當區的持份者,我們的反對是基於對環境、交通、社區健康和長遠發展的擔憂。規劃處在 審批時需全面評估這些影響,並考慮居民意見,確保項目符合可持續發展原則

八鄉上村居民簽署: 姓名(正楷): \_



25-03-15 09:37 FROM- PAT HEUNG RC T-680 P0002/0010 F-547 加いアルションタス 目が言 (存真: 2877 0245 及 電郵:www.tpbpd@pland.gov.hk)

> <u>強烈反對元朗石崗丈量約份第114 约地段第573 號餘段及第1710 號</u> 擬議臨時混凝土配料廠速附屬設施,申請編號: <u>A/YL-SK/410</u>

> > - 申請人提交進一步資料

執事先生 / 女士:

本人是元朗八鄉上村居民,就上標計劃改建為混凝土配料廠的提案作出強烈和堅決的反對。

1. 對環境造成影響環境影響

空氣污染:混凝土配料廠可能產生粉塵和廢氣,影響空氣品質,對居民健康造成威脅,尤其是 老人和兒童。

嗓音污染:工廠運作可能帶來持續嗓音,影響居民生活品質,特別是夜間作業時。

水污染:混凝土生產可能產生廢水,若處理不當,可能污染附近水源,影響生態和農業。

2. 交通壓力

重型車輛增加:混凝土配料廠需要大量原材料運輸,可能導致重型車輛頻繁出入,加劇交通擁 堵,並增加道路損壞風險。

交通安全:重型車輛增多可能提升交通事故風險,尤其對行人、騎單車者和學生構成威脅。

3. 社區影響

生活品質下降:工廠帶來的污染和噪音可能降低居民生活品質。健康風險:長期暴露於粉塵和 廢氣可能導致呼吸系統疾病,增加醫療負擔。

4. 生態破壞

自然環境受損:混凝土配料廠可能破壞周邊自然環境,影響動植物棲息地,降低生物多樣性。 農業影響:若附近有農地,工廠的污染可能影響農作物生長,損害農民生計。

5. 基礎設施壓力

工廠運作可能對水電等基礎設施造成額外壓力,影響居民正常使用。

作為當區的持份者,我們的反對是基於對環境、交通、社區健康和長遠發展的擔憂。規劃處在 審批時需全面評估這些影響,並考慮居民意見,確保項目符合可持續發展原則

本人懇請城市規劃委員會及政府相關部門重視八鄉居民的意見,聆聽居民的聲音,否決 上標申請。

八鄉上村居民簽署: 拉久保 姓名(正楷): 龙久保

	年	月	Ħ		8
	·				
通訊地:	让:				
	15				
聯絡電	話:				

\*25-03-15 09:38 FROM- PAT HEUNG RC T-680 P0003/0010 F-547 地小の回安見冒が盲 首体ル内但学程ココー統ル内理、17-680 P0003/0010 F-547 (傳真: 2877 0245 及 電郵:www.tpbpd@pland.gov.hk)

> <u>強烈反對元朗石崗丈量約份第114 约地段第573 號餘段及第1710 號</u> 擬議臨時混凝土配料廠連附屬設施,申請編號: A/YL-SK/410

53

#### - 申請人提交進一步資料

執事先生 / 女士:

本人是元朗八鄉上村居民,就上標計劃改建為混凝土配料廠的提案作出強烈和堅決的反對。

1. 對環境造成影響環境影響

空氣污染:混凝土配料廠可能產生粉塵和廢氣,影響空氣品質,對居民健康造成威脅,尤其是 老人和兒童。

嗓音污染:工廠運作可能帶來持續噪音,影響居民生活品質,特別是夜間作業時。

水污染:混凝土生產可能產生廢水,若處理不當,可能污染附近水源,影響生態和農業。

2. 交通壓力

重型車輛增加:混凝土配料廠需要大量原材料運輸,可能導致重型車輛頻繁出入,加劇交通擁 堵,並增加道路損壞風險。

交通安全:重型車輛增多可能提升交通事故風險,尤其對行人、騎單車者和學生構成威脅。

3. 社區影響

生活品質下降:工廠帶來的污染和噪音可能降低居民生活品質。健康風險:長期暴露於粉塵和 廢氣可能導致呼吸系統疾病,增加醫療負擔。

4. 生態破壞

自然環境受損:混凝土配料廠可能破壞周邊自然環境,影響動植物棲息地,降低生物多樣性。 農業影響:若附近有農地,工廠的污染可能影響農作物生長,損害農民生計。

5. 基礎設施壓力

工廠運作可能對水電等基礎設施造成額外壓力,影響居民正常使用。

作為當區的持份者,我們的反對是基於對環境、交通、社區健康和長遠發展的擔憂。規劃處在 審批時需全面評估這些影響,並考慮居民意見,確保項目符合可持續發展原則

八鄉上村居民簽署: 新客北



25-03-15 09:38 FROM- PAT HEUNG RC T-680 P0004/0010 F-547 城市規劃安貝質祕音 首応に用進業現 333 航北用政府で省 13 接 (傳真: 2877 0245 及 電郵:www.tpbpd@pland.gov.hk)

> <u>強烈反對元朗石崗丈量約份第114 约地段第573 號餘段及第1710 號</u> 擬議臨時混凝土配料廠連附屬設施,申請編號: A/YL-SK/410 - 申請人提交進一步資料

執事先生 / 女士:

本人是元朗八鄉上村居民,就上標計劃改建為混凝土配料廠的提案作出強烈和堅決的反對。

1. 對環境造成影響環境影響

空氣污染:混凝土配料廠可能產生粉塵和廢氣,影響空氣品質,對居民健康造成威脅,尤其是 老人和兒童。

嗓音污染:工廠運作可能帶來持續嗓音,影響居民生活品質,特別是夜間作業時。

水污染:混凝土生產可能產生廢水,若處理不當,可能污染附近水源,影響生態和農業。

2. 交通壓力

重型車輛增加:混凝土配料廠需要大量原材料運輸,可能導致重型車輛頻繁出入,加劇交通擁 堵,並增加道路損壞風險。

交通安全:重型車輛增多可能提升交通事故風險,尤其對行人、騎單車者和學生構成威脅。

3. 社區影響

生活品質下降:工廠帶來的污染和噪音可能降低居民生活品質。健康風險:長期暴露於粉塵和 廢氣可能導致呼吸系統疾病,增加醫療負擔。

4. 生態破壞

自然環境受損:混凝土配料廠可能破壞周邊自然環境,影響動植物棲息地,降低生物多樣性。 農業影響:若附近有農地,工廠的污染可能影響農作物生長,損害農民生計。

5. 基礎設施壓力

工廠運作可能對水電等基礎設施造成額外壓力,影響居民正常使用。

作為當區的持份者,我們的反對是基於對環境、交通、社區健康和長遠發展的擔憂。規劃處在 審批時需全面評估這些影響,並考慮居民意見,確保項目符合可持續發展原則

八鄉上村居民簽署 民 姓名(正楷):



25-03-15 09:38 FROM- PAT HEUNG RC T-680 P0005/0010 F-547 城中祝到安县省地省 省心北月進季返しつ 北北月以内百有 10 後 (傳真: 2877 0245 及電郵:www.tpbpd@pland.gov.hk)

> <u>強烈反對元朗石崗丈量約份第114 约地段第573 號餘段及第1710 號</u> 擬議臨時混凝土配料廠速附屬設施,申請編號: A/YL-SK/410 - 申請人提交進一步資料

執事先生 / 女士:

本人是元朗八郷上村居民,就上標計劃改建為混凝土配料廠的提案作出強烈和堅決的反對。

1. 對環境造成影響環境影響

空氣污染:混凝土配料廠可能產生粉塵和廢氣,影響空氣品質,對居民健康造成威脅,尤其是 老人和兒童。

噪音污染:工廠運作可能帶來持續噪音,影響居民生活品質,特別是夜間作業時。

水污染:混凝土生產可能產生廢水,若處理不當,可能污染附近水源,影響生態和農業。

2. 交通壓力

重型車輛增加:混凝土配料廠需要大量原材料運輸,可能導致重型車輛頻繁出入,加劇交通擁 堵,並增加道路損壞風險。

交通安全:重型車輛增多可能提升交通事故風險,尤其對行人、騎單車者和學生構成威脅。

3. 社區影響

生活品質下降:工廠帶來的污染和噪音可能降低居民生活品質。健康風險:長期暴露於粉塵和 廢氣可能導致呼吸系統疾病,增加醫療負擔。

4. 生態破壞

自然環境受損:混凝土配料廠可能破壞周邊自然環境,影響動植物棲息地,降低生物多樣性。 農業影響:若附近有農地,工廠的污染可能影響農作物生長,損害農民生計。

5. 基礎設施壓力

工廠運作可能對水電等基礎設施造成額外壓力,影響居民正常使用。

作為當區的持份者,我們的反對是基於對環境、交通、社區健康和長遠發展的擔憂。規劃處在 審批時需全面評估這些影響,並考慮居民意見,確保項目符合可持續發展原則

八鄉上村居民簽署

> <u>強烈反對元朗石崗丈量約份第114 约地段第573 號餘段及第1710 號</u> 擬議臨時混凝土配料廠連附屬設施,申請編號: A/YL-SK/410 - 申請人提交進一步資料

執事先生 / 女士:

本人是元朗八鄉上村居民,就上標計劃改建為混凝土配料廠的提案作出強烈和堅決的反對。

1. 對環境造成影響環境影響

空氣污染:混凝土配料廠可能產生粉塵和廢氣,影響空氣品質,對居民健康造成威脅,尤其是 老人和兒童。

噪音污染:工廠運作可能帶來持續嗓音,影響居民生活品質,特別是夜間作業時。 水污染:混凝土生產可能產生廢水,若處理不當,可能污染附近水源,影響生態和農業。

2. 交通壓力

重型車輛增加:混凝土配料廠需要大量原材料運輸,可能導致重型車輛頻繁出入,加劇交通擁 堵,並增加道路損壞風險。

交通安全:重型車輛增多可能提升交通事故風險,尤其對行人、騎單車者和學生構成威脅。

3. 社區影響

生活品質下降:工廠帶來的污染和噪音可能降低居民生活品質。健康風險:長期暴露於粉塵和 廢氣可能導致呼吸系統疾病,增加醫療負擔。

4. 生態破壞

自然環境受損:混凝土配料廠可能破壞周邊自然環境,影響動植物棲息地,降低生物多樣性。 農業影響:若附近有農地,工廠的污染可能影響農作物生長,損害農民生計。

5. 基礎設施壓力

工廠運作可能對水電等基礎設施造成額外壓力,影響居民正常使用。

作為當區的持份者,我們的反對是基於對環境、交通、社區健康和長遠發展的擔憂。規劃處在 審批時需全面評估這些影響,並考慮居民意見,確保項目符合可持續發展原則

本人懇請城市規劃委員會及政府相關部門重視八鄉居民的意見,聆聽居民的聲音,否決 上標申請。

八鄉上村居民簽署

姓名(正楷).



25-03-15 09:38 FROM- PAT HEUNG RC 日本ロロロローモーション アレンレアレス 1-680 PUUU//UUIU F-54/ (得真: 2877 0245 及電郵:www.tpbpd@pland.gov.hk)

> 強烈反對元朗石崗丈量約份第114 約地段第573 號餘段及第1710 號 擬議臨時混凝土配料廠連附屬設施,申請編號: A/YL-SK/410 - 申請人提交進一步資料

執事先生 / 女士:

本人是元朗八鄉上村居民,就上標計劃改建為混凝土配料廠的提案作出強烈和堅決的反對。

1. 對環境造成影響環境影響

空氣污染:混凝土配料廠可能產生粉塵和廢氣,影響空氣品質,對居民健康造成威脅,尤其是 老人和兒童。

噪音污染:工廠運作可能帶來持續噪音,影響居民生活品質,特別是夜間作業時。

水污染:混凝土生產可能產生廢水,若處理不當,可能污染附近水源,影響生態和農業。

2. 交通壓力

重型車輛增加:混凝土配料廠需要大量原材料運輸,可能導致重型車輛頻繁出入,加劇交通擁 堵,並增加道路損壞風險。

交通安全:重型車輛增多可能提升交通事故風險,尤其對行人、騎單車者和學生構成威脅。

3. 社區影響

生活品質下降:工廠帶來的污染和噪音可能降低居民生活品質。健康風險:長期暴露於粉塵和 廢氣可能導致呼吸系統疾病,增加醫療負擔。

4. 生態破壞

自然環境受損:混凝土配料廠可能破壞周邊自然環境,影響動植物棲息地,降低生物多樣性。 農業影響:若附近有農地,工廠的污染可能影響農作物生長,損害農民生計。

5. 基礎設施壓力

工廠運作可能對水電等基礎設施造成額外壓力,影響居民正常使用。

作為當區的持份者,我們的反對是基於對環境、交通、社區健康和長遠發展的擔憂。規劃處在 審批時需全面評估這些影響,並考慮居民意見,確保項目符合可持續發展原則

本人懇請城市規劃委員會及政府相關部門重視八鄉居民的意見, 聆聽居民的聲音, 否決 上標申請。

八鄉上村居民簽署 姓名(正楷):

年_	В	Я		
通訊地址				
聯絡電話				

25-03-15 09:38 FROM- PAT HEUNG RC T-680 P0008/0010 F-547 1970日 19700日 1970日

### <u>強烈反對元朗石崗丈量约份第114 约地段第573 號餘段及第1710 號</u> 擬議臨時混凝土配料廠連附屬設施,申請編號: A/YL-SK/410 - 申請人提交進一步資料

執事先生 / 女士:

本人是元朗八鄉上村居民,就上標計劃改建為混凝土配料廠的提案作出強烈和堅決的反對。

1. 對環境造成影響環境影響

空氣污染:混凝土配料廠可能產生粉塵和廢氣,影響空氣品質,對居民健康造成威脅,尤其是 老人和兒童。

噪音污染:工廠運作可能帶來持續噪音,影響居民生活品質,特別是夜間作業時。

水污染:混凝土生產可能產生廢水,若處理不當,可能污染附近水源,影響生態和農業。

2. 交通壓力

重型車輛增加:混凝土配料廠需要大量原材料運輸,可能導致重型車輛頻繁出入,加劇交通擁 堵,並增加道路損壞風險。

交通安全:重型車輛增多可能提升交通事故風險,尤其對行人、騎單車者和學生構成威脅。

3. 社區影響

生活品質下降:工廠帶來的污染和噪音可能降低居民生活品質。健康風險:長期暴露於粉塵和 廢氣可能導致呼吸系統疾病,增加醫療負擔。

4. 生態破壞

自然環境受損:混凝土配料廠可能破壞周邊自然環境,影響動植物棲息地,降低生物多樣性。 農業影響:若附近有農地,工廠的污染可能影響農作物生長,損害農民生計。

5. 基礎設施壓力

工廠運作可能對水電等基礎設施造成額外壓力,影響居民正常使用。

作為當區的持份者,我們的反對是基於對環境、交通、社區健康和長遠發展的擔憂。規劃處在 審批時需全面評估這些影響,並考慮居民意見,確保項目符合可持續發展原則

本人懇請城市規劃委員會及政府相關部門重視八鄉居民的意見,聆聽居民的聲音,否決上標申請。

八鄉上村居民簽署 姓名(正楷);

年_	_		
通訊地址	4		
聯絡電話	a 1		

(傳真: 2877 0245 及 電郵: www.tpbpd@pland.gov.hk)

25-03-15 09:39 FROM- PAT HEUNG RC

#### <u>強烈反對元朗石崗丈量约份第114 约地段第573 號餘段及第1710 號</u> 擬議臨時混凝土配料廠連附屬設施,申請編號: A/YL-SK/410 - 申請人提交進一步資料

日临儿内但中地了了那儿儿内以小口泊儿子

#### 執事先生 / 女士:

本人是元朗八鄉上村居民,就上標計劃改建為混凝土配料廠的提案作出強烈和堅決的反對。

1. 對環境造成影響環境影響

空氣污染:混凝土配料廠可能產生粉塵和廢氣,影響空氣品質,對居民健康造成威脅,尤其是 老人和兒童。

噪音污染:工廠運作可能帶來持續噪音,影響居民生活品質,特別是夜間作業時。

水污染:混凝土生產可能產生廢水,若處理不當,可能污染附近水源,影響生態和農業。

#### 2. 交通壓力

重型車輛增加:混凝土配料廠需要大量原材料運輸,可能導致重型車輛頻繁出入,加劇交通擁 堵,並增加道路損壞風險。

交通安全:重型車輛增多可能提升交通事故風險,尤其對行人、騎單車者和學生構成威脅。

3、社區影響

生活品質下降:工廠帶來的污染和噪音可能降低居民生活品質。健康風險:長期暴露於粉塵和 廢氣可能導致呼吸系統疾病,增加醫療負擔。

#### 4. 生態破壞

自然環境受損:混凝土配料廠可能破壞周邊自然環境,影響動植物棲息地,降低生物多樣性。 農業影響:若附近有農地,工廠的污染可能影響農作物生長,損害農民生計。

5. 基礎設施壓力

工廠運作可能對水電等基礎設施造成額外壓力,影響居民正常使用。

作為當區的持份者,我們的反對是基於對環境、交通、社區健康和長遠發展的擔憂。規劃處在 審批時需全面評估這些影響,並考慮居民意見,確保項目符合可持續發展原則

八鄉上村居民簽署: 姓名(正楼).

年_	-	A	8			
通訊地址	2					
聯絡電話	:					

25-03-15 09:39 FROM- PAT HEUNG RC T-680 P0010/0010 F-547 (傳真: 2877 0245 及 電郵:www.tpbpd@pland.gov.hk)

## <u>強烈反對元朗石崗丈量約份第114</u>约地段第573號餘段及第1710號 擬議臨時混凝土配料廠連附屬設施,申請編號: A/YL-SK/410 一申請人提交進一步資料

執事先生 / 女士:

本人是元朗八鄉上村居民,就上標計劃改建為混凝土配料廠的提案作出強烈和堅決的反對。

1. 對環境造成影響環境影響

空氣污染:混凝土配料廠可能產生粉塵和廢氣,影響空氣品質,對居民健康造成威脅,尤其是 老人和兒童。

噪音污染:工廠運作可能帶來持續噪音,影響居民生活品質,特別是夜間作業時。

水污染:混凝土生產可能產生廢水,若處理不當,可能污染附近水源,影響生態和農業。

#### 2. 交通壓力

重型車輛增加:混凝土配料廠需要大量原材料運輸,可能導致重型車輛頻繁出入,加劇交通擁 堵,並增加道路損壞風險。

交通安全:重型車輛增多可能提升交通事故風險,尤其對行人、騎單車者和學生構成威脅。

3. 社區影響

生活品質下降:工廠帶來的污染和噪音可能降低居民生活品質。健康風險:長期暴露於粉塵和 廢氣可能導致呼吸系統疾病,增加醫療負擔。

4. 生態破壞

自然環境受損:混凝土配料廠可能破壞周邊自然環境,影響動植物棲息地,降低生物多樣性。 農業影響:若附近有農地,工廠的污染可能影響農作物生長,損害農民生計。

5. 基礎設施壓力

工廠運作可能對水電等基礎設施造成額外壓力,影響居民正常使用。

作為當區的持份者,我們的反對是基於對環境、交通、社區健康和長遠發展的擔憂。規劃處在 審批時需全面評估這些影響,並考慮居民意見,確保項目符合可持續發展原則

本人懇請城市規劃委員會及政府相關部門重視八鄉居民的意見,聆聽居民的聲音,否決上標申請。

八鄉上村居民簽署: 雪陽 為 姓名(正楷): 雪勒(例

60.

Appendix IVb of RNTPC Paper No. A/YL-SK/410

F3 to 50

# 申請編號 A/YL-SK/410

- 1

## 擬議臨時混凝土配料廠連附屬設施

# (為期5年)

# 8 份公眾意見

RECEIVED 1 8 MAR 2025 Board Plannin

## 城市規劃委員會:

我就 A/YL-SK/410 混凝土廠的申請表達支持。

這些年政府收了很多棕地來做住宅發展,對於在棕地上經 營的企業和工人來說,影響真的很大。他們很多人辛辛苦 苦營運了好多年,結果土地被收回,只好另尋出路。我覺 得在八鄉安置受影響的混凝土廠,就是一個很好的替代方 案,既可以讓這些工業活動繼續下去,也能支持香港的經 濟發展。

我認為這個混凝土廠的位置選擇合理,遠離民居,而且就 在棕地帶裡,和周邊的物流、貨倉業務能形成合作,這樣 的規劃真的非常實際。香港的工業用地資源越來越少,每 一塊工業地的使用都應該充分發揮它的功能。希望城規會 能批准這個計劃,讓棕地經營者有更多發展的空間和信 心。

居民

鄭太 n Planning

2025/03/07

城市規劃委員會鈞鑒:

本人是八鄉一帶的貨倉營運者,我支持 A/YL-SK/410 規劃申請。

混凝土廠所位處的土地閑置已久,並未有加以善用,可算是浪費土地資源。此塊土地十分適合營 運工業相關用途,因為他為處於工業地帶,混凝 土廠的用途絕對符合該土地的規劃意向。而且, 土地鄰近於附近的貨倉及物流中心,能夠與附近 的企業產生協同效應。另外,項目鄰近錦田公 路,並且遠離民居,所以有利大型車輛出入。再 者,土地已經進行平整,有利工業發展。香港的 工業用地已經一直減少,因此城規會應該批准該 項目申請,保護香港的工業發展。

棕地作業者

鄧太謹啟

13/03/2025



我想就 A/YL-SK/410 混凝土厂的申请发表一些个人意见,并 支持这个计划。

香港的混凝土供应市场中,其实很多小型建筑公司和中小型企业经常面临价格谈判的压力,因为供应商少、选择少,香港只有十多间混凝土公司。香港建筑业物料联会早前都提过,混凝土 2024 年年底开始变得紧张,而且「做一个面包,要有足够的面粉,但香港的面粉量不多。」这个混凝土厂如果能顺利建成,意味着市场上会多一个供应来源,对于中小企业来说,这是一个可以减轻成本压力的好消息。更稳定的供应和竞争,对我们这些小公司来说真的很重要。

再者,该项目选址合理,周边没有民居,对环境影响有限,还能促进本地经济和中小企建造业发展,所以希望城 规会能批准这个申请。

市民

陈先生

二零二五年三月十一日



## 城市規劃委員會:

本人特此來函就 A/YL-SK/410 項目表達意見。

本人支持擬建的混凝土廠,未來香港即將建設北部都會 區,預計混凝土的需求將會大幅上升,因此十分迫切需要 建造更多混凝土廠,應付未來工程高峰期,對於促進香港 基建發展尤其重要。

本人並不擔心混凝土廠會帶來負面的環境影響,因為混凝 土廠遠離附近的村落及民居,並且位於工業地帶上,附近 很多土地都是棕地作業用途,因此不會與附近的土地用途 產生衝突。相比不少混凝土廠亦設置在市區及住宅附近, 設置於工業地帶更貼合混凝土廠的營運需求。

因此,我懇請城規會處理混凝土廠申請時,能夠考慮長遠 香港基建需要,不要忽略混凝土廠對香港經濟發展的重要 性。

市民

羅先生

1 8 MAR 2025 own Planning Board

二零二五年三月十一日

#### 城市規劃委員會:

我是一名普通市民,對 A/YL-SK/410 混凝土廠的規劃申請表示支持。

香港作為一個國際城市,很多東西都依賴進口,但我認為有些基建 原材料,比如混凝土,應該盡量提升本地生產的能力。這不僅可以 減少對外依賴,還能確保供應穩定。最近幾年,全球物流經常出現 問題,當運輸受阻時,我們的基建項目可能會因此拖延,尤其是北 都不少公屋發展,但如果本地有更多混凝土廠,這些問題就可以大 幅減少。

這個廠的位置交通便利,可以就近供應北部都會區和其他新發展 區,尤其混凝土混合後會在一段短時間內凝固,需趕快運送。這樣 一來,效率高了,整體成本也會降低。這樣的安排,對香港未來很 有幫助。

希望相關部門能批准這次申請,支持香港的自給能力和長遠發展。

黄小姐 ICCEIVED 18 MAR 2025 Town Planning Board

市民

二零二五年三月十六日

#### 城市規劃委員會鈞鑒:

本人支持 A/YL-SK/410 混凝土廠的申請項目。

近年來,政府先後開展了不少新發展區項目,導致區內不少的棕地 被徵用做住宅發展項目,嚴重影響棕地作業者的生計,亦打破不少 工人的飯碗。政府必須理解棕地作業(例如貨倉、工場)對於香港 經濟貢獻良多,意義重大,若然政府不作出合理補償,則大大削弱 棕地作業者的信心以及損害香港經濟發展。項目前身位處於洪水橋 新發展區,但因為政府收地而被迫放棄經營,因此政府應該批准這 項規劃申請,讓原本相安無事的混凝土廠能夠經營下去。

因此,城規會應該批准混凝土廠發展,以確保他們獲得合理補償, 並讓他們繼續經營及支撐香港經濟,謝謝!

棕地作業者

陳先生謹啟

1 8 MAR 2025 fown Planning Board

二零二五年三月十日

城市規劃委員會秘書:

本人謹就 A/YL-SK/410 規劃申請表達支持意見。

香港正積極推動北部都會區及多個新發展區項目,包 括新田科技城和牛潭尾地區,未來十年將迎來基建工 程高峰期,混凝土需求勢必大幅增加。根據申請文件 所述,擬議的混凝土配料廠不僅能滿足本地建設需 求,更可減少跨區運輸造成的環境污染及交通壓力, 這亦代表減少貨車長途運輸所帶來的懸浮物及廢氣排 放。石崗鄰近附近擬建發展區,將能直接為項目提供 穩定材料供應。此外,文件提到廠房將採用現代化設 備,大幅提升生產效率至每小時200 立方米,遠高於 過往廠房,有助縮短工程週期。本人認為,此舉能直 接支持香港基建發展,鞏固城市的經濟競爭力。

因此,我懇請城規會考量長遠社會效益,批准此申 請。

市民

鄧先生



二零二五年三月五日

城規會:

本人想就 A/YL-SK/410 規劃申請提供意見,希望本人意見能夠供為 規劃署參考。

本人居住於八鄉一帶的物流業經營者,我支持混凝土廠的設立。現 時八鄉和錦田一帶有多個物流中心、倉庫及露天存放,混凝土裝的 設立能夠與這些研究發展產生協同效應,降低業界的運輸成本,因 為混凝土可以直接供應附近棕地發展項目,減少跨區運輸需求。尤 其近年不少新發展區的棕地被徵收,不少棕地作業者遷移到八鄉錦 田一帶繼續經營他們的業務,預期一帶的棕地發展項目將會不斷上 升,混凝土的需求只會不跌反升,因此混凝土廠的設立將有利於業 界發展。

本人亦留意到混凝土廠有與物流業的作業時間錯開,廠方將會集中 於晚間配送貨物,相信避免了錦田公路一帶日間貨車擠塞的情況, 不會影響物流業間的日間運送服務。

我認為項目是一個對棕地業界是一個雙贏方案,因此,希望規劃署 能批准該項目,謝謝。

棕地作業者

鄧先生



二零二五年三月十二日